JOURNAL

OF THE

SENATE OF VIRGINIA

2021 REGULAR SESSION

Beginning Wednesday, January 13, 2021 and Ending Monday, February 8, 2021

2021 RECONVENED SESSION OF REGULAR SESSION

Wednesday, March 17, 2021

2021 SPECIAL SESSION I

Beginning Wednesday, February 10, 2021 and Ending Monday, March 1, 2021

2021 RECONVENED SESSION OF SPECIAL SESSION I

Wednesday, April 7, 2021

2021 SPECIAL SESSION II

Beginning Monday, August 2, 2021 and Ending Tuesday, August 10, 2021

Held at the Science Museum of Virginia in the City of Richmond

Volume 1

Compiled in the office of

Susan Clarke Schaar, Clerk of the Senate

Pursuant to Article IV, Section 10 of the Constitution of Virginia

Tara H. Perkinson Chief Deputy Clerk

Joanna W. Bolstad Journal Clerk Corinne F. Sloan Assistant Journal Clerk Liz K. Greenwood Journal Assistant

COMMONWEALTH OF VIRGINIA RICHMOND 2021

TABLE OF CONTENTS

VOLUME 1

DAILY SESSIONS OF THE SENATE, 2021 REGULAR SESSION	1-506
TABLE OF CODE TITLES	507-510
NUMERICAL INDEX, 2021 REGULAR SESSION	511-712
SUBJECT INDEX, 2021 REGULAR SESSION	713-956
PATRONAGE OF SENATE LEGISLATION	957-1000
DAILY SESSIONS OF THE SENATE, 2021 SPECIAL SESSION I	1-582
VOLUME 2	
DAILY SESSIONS OF THE SENATE, 2021 SPECIAL SESSION I	583-1224
TABLE OF CODE TITLES	
NUMERICAL INDEX, 2021 SPECIAL SESSION I	
SUBJECT INDEX, 2021 SPECIAL SESSION I	
PATRONAGE OF SENATE LEGISLATION	
DAILY SESSIONS OF THE SENATE, 2021 SPECIAL SESSION II	
TABLE OF CODE TITLES	
NUMERICAL INDEX, 2021 SPECIAL SESSION II	
SUBJECT INDEX, 2021 SPECIAL SESSION II	
PATRONAGE OF SENATE LEGISLATION	
TAIRONAGE OF SENATE LEGISLATION	133-100
APPENDIX	
STATE OFFICIALS	1
MEMBERS OF THE SENATE	
OFFICERS AND EMPLOYEES OF THE SENATE	6-7
LIST OF SENATORS IN ORDER OF SENIORITY	8
SEATING ARRANGEMENT OF THE SENATE	9
SENATE OFFICES, POCAHONTAS BUILDING	
COUNTIES AND CITIES SHOWING SENATORS AND DELEGATES	
HOUSE AND SENATE BILLS APPROVED, 2021 REGULAR SESSION	
HOUSE BILLS APPROVED, 2021 SPECIAL SESSION I	
SENATE BILLS APPROVED, 2021 SPECIAL SESSION I	
HOUSE BILL APPROVED, 2021 SPECIAL SESSION II	
SUMMARY OF 2021 REGULAR SESSION LEGISLATION	
SUMMARY OF 2021 SPECIAL SESSION I LEGISLATION	
SUMMARY OF 2021 SPECIAL SESSION II LEGISLATION	
SUPREME COURT JUSTICES AND COURT OF APPEALS OF VIRGINIA JUDGES .	
SESSIONS OF THE GENERAL ASSEMBLY, 2017-2021	28-32

JOURNAL

OF

THE

SENATE

2021 REGULAR SESSION

WEDNESDAY, JANUARY 13, 2021

The Senate met at 12 m. at the Science Museum of Virginia in Richmond, Virginia, and was called to order by Lieutenant Governor Justin E. Fairfax.

Nathan Hatfield, Assistant Clerk-LINCS, Senate Clerk's Office, offered the following prayer:

On this opening day of session, a blessing upon each head assembled here to do and support the business of the people: our Senators, our Lieutenant Governor, our Clerk and her staff. To the Capitol Police who protect us, the journalists who report our actions, the Science Museum of Virginia staff who provide space for us to be distanced yet together, and to Phyllis and her staff who prepared the meal to nourish us. Let us pray.

Almighty God, we pause in this moment to ask for Your hand of comfort as we continue to reflect on the life and remember the loss of one of our own in the Senate family, Senator Chafin.

I offer as today's invocation these words from American author, theologian, educator, and civil rights leader, Howard Thurman:

Creating God, open to us Light for our darkness, Courage for our fear, Hope for our despair. Loving God, open to us Wisdom for our confusion, Forgiveness for our failings, Love for our hate. Spirit of Peace, open to us Peace for our turmoil, Joy for our sorrow, Strength for our weakness. Generous God, open our hearts to receive the gifts of Your Spirit. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Susan Clarke Schaar, Clerk of the Senate.

STATEMENT BY SENATOR

Senator Norment offered the following statement:

The Governor has declared a "state of emergency" in Virginia as a result of the COVID-19 pandemic that has resulted in some temporary adjustments to the Senate of Virginia's normal procedures. Because some senators are more susceptible to serious implications of COVID-19, the Senate Democratic Caucus and the Senate Republican Caucus have mutually agreed as follows only for the 2021 Session of General Assembly:

- 1. The senators previously agreed upon by the respective leadership of each caucus shall be permitted to participate in the 2021 General Assembly session remotely and electronically, subject to the following requirements.
- 2. Each day the Senate is actually in session, pursuant to Rule 5 of the Rules of the Senate, a "quorum" must be physically present in the Chamber.
- 3. Each day the Senate is actually in session during the "morning hour," a motion will be made to temporarily suspend the Rules of the Senate and, approved by 27 senators actually physically present, to allow previously agreed upon senators to participate remotely and electronically in that daily session.
- 4. No proxy votes will be allowed by a senator participating electronically and remotely.
- 5. On any vote taken in committee or on the floor of the Senate, a senator participating electronically and remotely must personally cast her or his vote; and
- A. No surrogate may ever cast any votes.
- B. A senator participating electronically or remotely must personally cast her or his vote on video; and
- C. A senator participating electronically or remotely will have her or his vote orally taken by the Clerk on a roll call.
- 6. The number of senators participating electronically and remotely may be expanded by the mutual agreement of the Senate Democratic and Republican leadership upon receipt of written confirmation from a licensed healthcare provider.
- 7. This temporary change in the procedures of the Senate is a direct response to the Declared State of Emergency and shall not be considered a precedent for any future consideration of temporarily or permanently adjusting the Senate's procedures or Rules.
- 8. It is mutually agreed there will be no future effort on the part of either Caucus to amend the Rules of the Senate to provide for electronic and remote participation. This provision shall remain in effect only until the earlier of the end of the Declared State of Emergency or the end of the current terms of the Senators including the years 2021-2024, regardless of any special election, without the concurrence of a majority, recorded vote of both caucuses.

9. No Senator may collect per diem on days they participate electronically or remotely.

On motion of Senator Norment, the Rules were suspended to allow Senators Lewis and Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--37.

NAYS--0. RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

RULES OF THE SENATE

The President announced that the Senate was operating under the Rules of the Senate as adopted at the 2020 Session, which state, "The Rules of the Senate shall be adopted at the commencement of the first regular session of the General Assembly after the election of the Senate, and shall be in force for the succeeding four years unless amended or suspended as provided by these Rules."

COMMUNICATION

The following communication was received and read:

COMMONWEALTH OF VIRGINIA Executive Department

Proclamation

WHEREAS, the Commonwealth of Virginia remains under a state of emergency due to the novel coronavirus 19 (COVID-19) pandemic. Medical experts advise that the virus that causes COVID-19 is highly contagious. In the wake of the holidays, case positivity rates in Virginia are above 16 percent. Since the beginning of the pandemic, more than 5,300 Virginias have lost their lives to this virus. The seven-day moving average for new, confirmed cases in Virginia is almost 4,000 per day. COVID-19 is a dangerous contagious disease; and

WHEREAS, on January 13, 2021, members of the Virginia Senate are scheduled to meet to convene the 2021 regular session of General Assembly; and

WHEREAS, § 30-1 of the *Code of Virginia* mandates that the General Assembly meet for regular session in the Capitol building in the City of Richmond, Virginia; and

WHEREAS, § 30-2 of the *Code of Virginia* provides that if, however, the General Assembly cannot meet at the Capitol building as prescribed by law because of a "dangerous contagious disease," the Governor, by proclamation, may appoint another place for the General Assembly to meet to conduct regular session; and

WHEREAS, by letter dated November 8, 2020, the Majority Leader and the Caucus Chairman of the Senate advised that they believed "holding session in the Senate Chamber of the State Capitol poses an unnecessary risk" in light of the COVID-19 pandemic; and

WHEREAS, the Majority Leader and the Caucus Chairman of the Senate requested, by the aforementioned letter, I issue a proclamation appointing the Science Museum of Virginia in the City of Richmond, Virginia as the place for the Senate to meet to conduct the 2021 regular session.

NOW, THEREFORE, it being apparent that the Senate cannot safely meet to conduct the 2021 regular session at the place prescribed by law because of a dangerous contagious disease, I, Ralph S. Northam, by virtue of the authority vested in me pursuant to § 30-2 of the *Code of Virginia*, give notice that beginning January 13, 2021, the Senate may meet to conduct the 2021 regular session at the Science Museum of Virginia in the City of Richmond, Virginia.

Given under my hand and under the Lesser Seal of the Commonwealth at Richmond, this 12th day of January in the year Two Thousand and Twenty-One and in the 245th year of the Commonwealth.

/s/ Ralph S. Northam, Governor

[SEAL]

Attest:

/s/ Kelly Thomasson, Secretary of the Commonwealth

COMMITTEE REPORTS

Senator Locke, from the Committee on Rules, presented the following reports:

March 23, 2020

Report of the Rules Subcommittee on Financial Disclosure Review

The above-named subcommittee has reviewed all Senate members' Statements of Economic Interest filed with the Virginia Conflict of Interest and Ethics Advisory Council. The subcommittee found that all members of the Senate have filed disclosure forms.

/s/ Scott A. Surovell, Chairman

December 20, 2020

Report of the Rules Subcommittee on Financial Disclosure Review

The above-named subcommittee has reviewed all Senate gift report forms filed with the Virginia Conflict of Interest and Ethics Advisory Council. The subcommittee found that all members of the Senate have filed disclosure forms.

/s/ Scott A. Surovell, Chairman

IMMEDIATE CONSIDERATION

Senator Locke moved that the Rules be suspended, the Committee on Rules be discharged from further consideration of **S.R. 90** (ninety), the readings of the title and engrossment be waived, and the resolution be taken up for immediate consideration.

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 90

2021 Senate operating resolution.

RESOLVED by the Senate of Virginia, That the Comptroller is directed to issue his warrants on the Treasurer, payable from the contingent fund of the Senate, to accomplish the work of the Senate of Virginia as reported by the Clerk of the Senate to the Senate Committee on Rules during the 2021 Session. Necessary payments to cover salaries of temporary employees, per diem for legislative assistants who establish a temporary residence, and certain employees designated by the Clerk and reported to the Chair of the Senate Committee on Rules, as well as other contingent and incidental expenses, will be certified by the Clerk of the Senate or her designee.

S.R. 90, on motion of Senator Locke, was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Saslaw was ordered to inform the House of Delegates that the Senate was duly organized and ready to proceed to business.

LEGISLATION CONTINUED

The following report was received from the Clerk:

TO THE MEMBERS OF THE SENATE:

Pursuant to the provisions of Senate Rules 20 (f) and 20 (g) and House Joint Resolution No. 99 of the 2020 Session, certain bills and joint resolutions were continued from the 2020 Session of the General Assembly in the several Senate committees, as follows:

COMMITTEE ON AGRICULTURE, CONSERVATION AND NATURAL RESOURCES

S.B. 409, S.B. 626, S.B. 669.

H.B. 672, H.B. 1329, H.B. 1674.

COMMITTEE ON COMMERCE AND LABOR

S.B. 27, S.B. 274, S.B. 295, S.B. 411, S.B. 427, S.B. 474, S.B. 719, S.B. 765, S.B. 785, S.B. 842, S.B. 867.

H.B. 333, H.B. 416, H.B. 800.

COMMITTEE ON EDUCATION AND HEALTH

S.B. 80, S.B. 249, S.B. 390, S.B. 464, S.B. 898, S.B. 917, S.B. 1050, S.B. 1068, S.B. 1078, S.B. 1079.

S.J.R. 25.

H.B. 332, H.B. 634.

COMMITTEE ON FINANCE AND APPROPRIATIONS

S.B. 128, S.B. 135, S.B. 143, S.B. 190, S.B. 191, S.B. 218, S.B. 234, S.B. 327, S.B. 363, S.B. 367, S.B. 387, S.B. 446, S.B. 456, S.B. 457, S.B. 460, S.B. 461, S.B. 510, S.B. 536, S.B. 606, S.B. 637, S.B. 756, S.B. 779, S.B. 854, S.B. 863, S.B. 946, S.B. 965, S.B. 1033, S.B. 1052.

S.J.R. 58.

H.B. 281, H.B. 547, H.B. 729, H.B. 734, H.B. 1194, H.B. 1619, H.B. 1621, H.B. 1633.

COMMITTEE ON GENERAL LAWS AND TECHNOLOGY

S.B. 533, S.B. 627, S.B. 777, S.B. 906, S.B. 1061.

H.B. 7, H.B. 1134, H.B. 1650.

COMMITTEE ON THE JUDICIARY

S.B. 13, S.B. 15, S.B. 32, S.B. 91, S.B. 223, S.B. 285, S.B. 353, S.B. 359, S.B. 449, S.B. 490, S.B. 537, S.B. 550, S.B. 618, S.B. 641, S.B. 660, S.B. 699, S.B. 721, S.B. 780, S.B. 802, S.B. 803, S.B. 808, S.B. 813, S.B. 819, S.B. 914, S.B. 947, S.B. 973, S.B. 1060.

H.B. 251, H.B. 304, H.B. 610, H.B. 673, H.B. 869, H.B. 961, H.B. 1063, H.B. 1288.

COMMITTEE ON LOCAL GOVERNMENT

S.B. 419, S.B. 1092.

H.B. 707.

COMMITTEE ON PRIVILEGES AND ELECTIONS

S.B. 166, S.B. 887.

S.J.R. 2, S.J.R. 3, S.J.R. 6, S.J.R. 7, S.J.R. 8, S.J.R. 11, S.J.R. 14, S.J.R. 16, S.J.R. 19, S.J.R. 59, S.J.R. 63, S.J.R. 75.

H.B. 177.

COMMITTEE ON REHABILITATION AND SOCIAL SERVICES

S.B. 502, S.B. 1095.

H.B. 1187.

COMMITTEE ON TRANSPORTATION

S.B. 1021.

H.B. 157, H.B. 845.

COMMITTEE ON RULES

S.B. 150, S.B. 1062.

H.B. 553, H.B. 1010.

Pursuant to the provisions of House Joint Resolution No. 99 of the 2020 Session, certain Senate bills and resolution were continued from the 2020 Session of the General Assembly in the several House committees, as follows:

COMMITTEE ON AGRICULTURE, CHESAPEAKE AND NATURAL RESOURCES

S.B. 304, S.B. 318.

COMMITTEE ON APPROPRIATIONS

S.B. 366, S.B. 723, S.B. 1080.

COMMITTEE FOR COURTS OF JUSTICE

S.B. 148, S.B. 253, S.B. 306, S.B. 326, S.B. 624, S.B. 681, S.B. 805, S.B. 810, S.B. 811, S.B. 1042.

COMMITTEE ON EDUCATION

S.B. 420.

COMMITTEE ON GENERAL LAWS

S.B. 97, S.B. 346, S.B. 362, S.B. 824, S.B. 826.

COMMITTEE ON HEALTH, WELFARE AND INSTITUTIONS

S.B. 317, S.B. 352, S.B. 569, S.B. 983, S.B. 993, S.B. 1049.

COMMITTEE ON LABOR AND COMMERCE

S.B. 291, S.B. 382.

COMMITTEE ON TRANSPORTATION

S.B. 1011.

COMMITTEE ON RULES

S.B. 486.

S.J.R. 39.

CALENDAR

MEMORIAL RESOLUTION

S.J.R. 284 (two hundred eighty-four) was taken up, as follows:

SENATE JOINT RESOLUTION NO. 284

Celebrating the life of the Honorable Augustus Benton Chafin, Jr.

WHEREAS, the Honorable Augustus Benton Chafin, Jr., a passionate advocate for Southwest Virginia and an accomplished state legislator who earned bipartisan admiration for his commitment to the betterment of the Commonwealth as a whole, died on January 1, 2021; and

WHEREAS, a proud native of Russell County, Benton "Ben" Chafin grew up on his family's farm and learned the value of hard work and responsibility at a young age; he worked at a local meat packing plant as a teenager, and he returned home most weekends while attending college to help on the farm; and

WHEREAS, a first-generation college student in his family, Ben Chafin received a bachelor's degree from East Tennessee State University and a legal degree from the University of Richmond; and

WHEREAS, Ben Chafin established what is now Chafin Law Firm, P.C., in the 1980s and practiced law throughout Southwest Virginia for more than 30 years; in addition to his work as an attorney, he served on the board of First Bank & Trust Company and held close to his roots in agriculture as the owner and operator of a beef cattle farm; and

WHEREAS, desirous to be of further service to his community and to the Commonwealth, Ben Chafin ran for and was elected to the Virginia House of Delegates in 2013, becoming the first Republican to represent the Fourth District in more than 20 years; and

WHEREAS, Ben Chafin was subsequently elected to the Senate of Virginia during a special election in 2014 and ably served the residents of Southwest Virginia in the 38th District until the time of his passing; and

WHEREAS, Ben Chafin adapted to the unique challenges of representing a district that is geographically larger than some federal congressional districts, encompassing all of the Counties of Bland, Buchanan, Dickenson, Pulaski, Russell, and Tazewell; parts of the Counties of Montgomery, Smyth, and Wise; and the Cities of Norton and Radford; and

WHEREAS, during his tenure as a state lawmaker, Ben Chafin introduced and supported many pieces of important legislation to benefit all Virginians and provided his expertise to the Agriculture, Conservation and Natural Resources, Commerce and Labor, Education and Health, Judiciary, Privileges and Elections, and Rehabilitation and Social Services committees; and

WHEREAS, Ben Chafin was a staunch proponent of public education, responsible economic growth, Second Amendment rights, and access to health care for all Virginians; and

WHEREAS, Ben Chafin consistently supported job creation in his district, helping to establish the InvestSWVA partnership to promote the coalfields region as a future site of renewable energy infrastructure and high-technology business development; and

WHEREAS, Ben Chafin served the residents of Southwest Virginia and the entire Commonwealth with the utmost dedication, integrity, and distinction; and

WHEREAS, outside of his careers, Ben Chafin was an avid outdoorsman, who relished every opportunity to appreciate Southwest Virginia's natural splendor; he volunteered his time and leadership as former president of the Russell County Rotary Club, and he enjoyed fellowship and worship with the congregation of Gracewood Community Church in Lebanon; and

WHEREAS, Ben Chafin will be fondly remembered and greatly missed by his wife of 38 years, Lora; his children, Sophia, Audra, Augustus III, and their families; and numerous other family members, friends, and colleagues; now, therefore, be it

RESOLVED by the Senate, the House of Delegates concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Augustus Benton Chafin, Jr., a respected statesman and a champion for Southwest Virginia; and, be it

RESOLVED FURTHER, That the Clerk of the Senate prepare a copy of this resolution for presentation to the family of Augustus Benton Chafin, Jr., as an expression of the General Assembly's respect for his memory.

S.J.R. 284, on motion of Senator Pillion, was ordered to be engrossed and was agreed to by a unanimous standing vote.

HOUSE COMMUNICATION

The following communication was received and read:

In the House of Delegates January 13, 2021

THE HOUSE OF DELEGATES IS DULY ORGANIZED AND READY TO PROCEED TO BUSINESS.

/s/ Suzette Denslow Clerk of the House of Delegates

RECESS

At 1:00 p.m., Senator Norment moved that the Senate recess until 2:50 p.m.

The motion was agreed to.

The hour of 2:50 p.m. having arrived, the Chair was resumed.

HOUSE COMMUNICATIONS

The following communications were received:

In the House of Delegates January 13, 2021

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTION:

H.J.R. 575. Providing for a Joint Assembly, establishing a schedule for the conduct of business coming before the 2021 Regular Session of the General Assembly of Virginia, and providing for legislative continuity between the 2021 Regular Session of the General Assembly and a special session.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

In the House of Delegates January 13, 2021

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTION:

H.J.R. 576. Establishing a schedule for the conduct of business for the prefiling period of the 2022 Regular Session of the General Assembly of Virginia.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communications from the House of Delegates was waived.

The recorded vote is as follows: YEAS--37. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Deeds--1.

RULE 36--0.

The House joint resolutions, communicated as agreed to by the House of Delegates, the first reading of their titles having been waived, were referred as follows:

H.J.R. 575 and H.J.R. 576 were referred to the Committee on Rules.

IMMEDIATE CONSIDERATION

Senator Locke moved that the Rules be suspended, the Committee on Rules be discharged from further consideration of **H.J.R. 575** (five hundred seventy-five), the second reading of the title be waived, and the resolution be taken up for immediate consideration.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.J.R. 575 was read by title the third time.

HOUSE JOINT RESOLUTION NO. 575

Providing for a Joint Assembly, establishing a schedule for the conduct of business coming before the 2021 Regular Session of the General Assembly of Virginia, and providing for legislative continuity between the 2021 Regular Session of the General Assembly and a special session.

RESOLVED by the House of Delegates, the Senate concurring, That the Speaker of the House of Delegates, President pro tempore of the Senate, and President of the Senate shall assemble in the Hall of the House of Delegates on Wednesday, January 13, 2021, at such time as specified by the Speaker of the House of Delegates, to receive the Governor of Virginia, and such address as he may desire to make, and that the rules for the government of the House of Delegates and the Senate, when convened in joint session for such purpose, shall be as follows:

Rule I. At the hour fixed for the meeting of the Joint Assembly, the President pro tempore of the Senate and the President of the Senate, accompanied by the Clerk of the Senate, shall proceed to the Hall of the House of Delegates and shall be received by the Speaker of the House of Delegates and the Clerk of the House of Delegates standing. An appropriate seat shall be assigned to the President pro tempore of the Senate by the Sergeant at Arms of the House. The Speaker of the House of Delegates shall assign an appropriate seat for the President of the Senate.

Rule II. The Speaker of the House of Delegates shall be President of the Joint Assembly. In case it shall be necessary for the Speaker to vacate the Chair, the President of the Senate shall serve as the presiding officer.

Rule III. The Clerk of the House of Delegates shall be Clerk of the Joint Assembly and shall be assisted by the Clerk of the Senate. The Clerk of the Joint Assembly shall enter the proceedings of the Joint Assembly in the Journal of the House and shall certify a copy of the same to the Clerk of the Senate, who shall enter the same in the Journal of the Senate.

Rule IV. The Sergeant at Arms and Doorkeepers of the House shall act as such for the Joint Assembly.

Rule V. The Rules of the House of Delegates, as far as applicable, shall be the rules of the Joint Assembly.

Rule VI. In calling the roll of the Joint Assembly, the names of the Senators shall be called in alphabetical order, then the names of the Delegates in like order, except that the name of the Speaker of the House of Delegates shall be called last.

Rule VII. If, when the Joint Assembly meets, it shall be ascertained that a majority of each house is not present, the Joint Assembly may take measures to secure the attendance of absentees, or adjourn to a succeeding day, as a majority of those present may determine.

Rule VIII. When the Joint Assembly adjourns, the President pro tempore of the Senate and the President of the Senate, accompanied by the Clerk of the Senate, shall return to their chamber, and the business of the House shall be continued in the same order as at the time of the entrance of the President pro tempore of the Senate, the President of the Senate, and the Clerk of the Senate; and, be it

RESOLVED FURTHER, That notwithstanding any other provision of this resolution and in accordance with the practices of each house, with the exception of commending and memorial joint resolutions, a request to be added as a co-patron shall be received prior to the first vote on the passage of

a bill or agreement to a joint resolution or, if the bill or joint resolution is not reported from committee, then prior to the last action on such legislation. A request to be removed as a co-patron shall be received no later than 5:00 p.m., Monday, February 1, 2021; and, be it

RESOLVED FURTHER, That any member offering for introduction a bill or joint resolution not submitted to the Division of Legislative Services for drafting is encouraged to submit an electronic version no later than 5:00 p.m. on the day the legislation is introduced; and, be it

RESOLVED FURTHER, That for purposes of the procedural deadlines established herein for the 2021 Regular Session of the General Assembly:

"Budget Bill" means the general appropriation bill introduced in each house that authorizes the biennial expenditure of public revenues for the period from July 1, 2020, through June 30, 2022.

"Legislative day" means the period of time that begins with the call to order by the presiding officer and ends when declared adjourned by the presiding officer. Unless another time is specified, any deadline established in this resolution shall expire at the end of the legislative day.

"Prefiled legislation" means any bill or joint resolution requested from the Division of Legislative Services no later than 5:00 p.m., Monday, November 30, 2020, and prefiled no later than 10:00 a.m., Wednesday, January 13, 2021, or any bill or joint resolution not requested from the Division of Legislative Services and prefiled no later than 10:00 a.m., Wednesday, January 13, 2021.

"Unanimous consent" means the affirmation of all the members present in the house of origin. Any legislation intended to be offered for introduction with unanimous consent or with the written request of the Governor shall not require the consent of the house in order for the member to request the Division of Legislative Services to draft such legislation. The Division of Legislative Services shall return such legislation after the original introduction deadline.

"Virginia Retirement System bill" means any bill that amends, adds, repeals, or modifies any provision of any retirement system established in Title 51.1 of the Code of Virginia; and, be it

RESOLVED FINALLY, That the 2021 Regular Session of the General Assembly shall be governed by the following procedural rules, which establish time limitations for elections and for all legislation prefiled and introduced for or continued to the 2021 Regular Session except:

- (i) House and Senate resolutions;
- (ii) Bills and joint resolutions affecting the rules of procedure or the schedule of business of the General Assembly, either of its houses, or any of its committees;
- (iii) Bills and joint resolutions introduced with unanimous consent to exceed the time limitations established in Rules 2, 3, 6, and 9;
 - (iv) Joint resolutions confirming appointments subject to the confirmation of the General Assembly;
- (v) Joint commending and memorial resolutions, except for the time limitations established in Rules 8 and 9:
- (vi) Bills and joint resolutions regarding elections held by the General Assembly during the 2021 Regular Session; or

- (vii) Bills and joint resolutions requested in writing by the Governor.
- Rule 1. Neither house of the General Assembly shall receive from any committee any bill or joint resolution that was continued on the agenda of such committee and acted upon later than midnight, Thursday, December 3, 2020. For purposes of this rule, a motion to refer a measure to another committee shall be treated as an action by a committee.
- Rule 2. No bill or joint resolution creating or continuing a study shall be offered in either house after the adjournment of that house on Wednesday, January 13, 2021.
- Rule 3. No Virginia Retirement System bill shall be offered in either house after adjournment of that house on Wednesday, January 13, 2021.
- Rule 4. Except for bills and joint resolutions required to be requested earlier, requests for the drafting, redrafting, or correction of any bill or joint resolution shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Friday, January 15, 2021.
- Rule 5. No later than Monday, January 18, 2021, each house shall begin its consideration of any election to fill any judicial seat in the courts of the Commonwealth, or to fill a seat on any commission or office elected by the General Assembly. In the event that the houses cannot agree on such election before Tuesday, January 19, 2021, such election shall become the subject of a special and continuing joint order in each house, and such special and continuing joint order shall have precedence over all other business of either house, until such time as both houses reach agreement on such election or agree to hold it at another specific time. The Rules of each house, as far as applicable, shall be the rules governing such election.
- Rule 6. Except for bills required to be filed earlier, no bill or joint resolution shall be offered in either house after 3:00 p.m., Friday, January 22, 2021.
- Rule 7. No later than Friday, January 22, 2021, the Board of Trustees of the Virginia Retirement System shall submit, in accordance with § 30-19.1:7, impact statements for all Virginia Retirement System bills filed by the first day of session. For any Virginia Retirement System bill filed later than the first day of session, the Board of Trustees shall use due diligence in preparing the impact statement in time for review by the standing committees.
- Rule 8. Requests for the drafting, redrafting, or correction of any joint commending or memorial resolution shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Saturday, January 30, 2021.
- Rule 9. No joint commending or memorial resolution shall be offered in either house after 5:00 p.m., Tuesday, February 2, 2021.
- Rule 10. Except for the Budget Bill, beginning Saturday, February 6, 2021, the House of Delegates shall consider only Senate bills, Senate joint resolutions, House bills with Senate amendments, and House joint resolutions with Senate amendments; the Senate shall consider only House bills, House joint resolutions, Senate bills with House amendments, and Senate joint resolutions with House amendments; and each house may consider conference reports and other privileged matters relating thereto to the end that the work of each house may be disposed of by the other.
- Rule 11. The committees responsible for the consideration of the Budget Bill in the houses of introduction shall complete their work on such bill no later than midnight, Sunday, February 7, 2021.

Rule 12. This session of the General Assembly shall adjourn sine die no later than midnight on Thursday, February 11, 2021.

Rule 13. Pursuant to Section 6 of Article IV of the Constitution of Virginia, the General Assembly shall reconvene Wednesday, March 17, 2021, for the purpose of considering bills that may have been returned by the Governor with recommendations for their amendment and bills and items of appropriation bills, including the general appropriation act, that may have been returned by the Governor with his objections.

Rule 14. That, notwithstanding any applicable rules of the House of Delegates or Senate, the General Assembly does hereby provide legislative continuity pursuant to Section 7 of Article IV of the Constitution of Virginia. Any bill, joint resolution, or resolution introduced during the 2021 Regular Session may, by motion, be continued to the first 2021 Special Session of the General Assembly convened subsequent to the 2021 Regular Session, for action by any committee or subcommittee of the General Assembly, either house of the General Assembly, or any conference committee.

Rule 15. That members of the General Assembly and credentialed legislative staff attending floor sessions, committees, subcommittees, and any other meetings of the General Assembly in person or meeting in a legislative office in person shall be required to wear a facemask or face shield, and (i) at the discretion of the Speaker of the House of Delegates for Delegates and House credentialed staff and the discretion of the chair of the Senate Committee on Rules for Senators and Senate credentialed staff, have their temperature taken daily when initially entering legislative space; and (ii) at the discretion of the Speaker of the House of Delegates for Delegates and House credentialed staff and the discretion of the chair of the Senate Committee on Rules for Senators and Senate credentialed staff, be tested weekly for Covid-19 by the Virginia Department of Health.

Rule 16. The conduct of the business of any subcommittee of any House committee, any joint subcommittee of House and Senate committees, and any interim study commission created pursuant to a House measure shall be governed by the Rules of the House of Delegates; the conduct of the business of any subcommittee of any Senate committee, any joint subcommittee of Senate and House committees, and any interim study commission created pursuant to a Senate measure shall be governed by the Rules of the Senate. If a House measure and a Senate measure create the same study, the conduct of the business of the study shall be governed by the rules of the house of the chairman of the study, or in the case of co-chairmen, the rules of the house as agreed upon by the co-chairmen.

Rule 17. Interim meetings of any standing committee, joint committee, joint subcommittee, legislative commission, or any other interim study subcommittee or study commission shall be held on such day as called by the chair if electronically, or if in person on Monday, Tuesday, or Wednesday during the first and third full weeks of the month, unless otherwise authorized by the Speaker of the House of Delegates or the Chairman of the Senate Committee on Rules, as may be appropriate for the house in which the chairman serves.

Rule 18. Any staff member assigned to work for, and support the efforts of, any committee of the House or Senate, any subcommittee of any such committee, any joint subcommittee of House and Senate committees, or any interim study commission shall work under the direction of the chairman of such committee, subcommittee, joint subcommittee, or interim study commission.

Senator Surovell offered an amendment in the nature of a substitute, as follows:

HOUSE JOINT RESOLUTION NO. 575 AMENDMENT IN THE NATURE OF A SUBSTITUTE

Providing for a Joint Assembly and establishing a schedule for the conduct of business coming before the 2021 Regular Session of the General Assembly of Virginia.

RESOLVED by the House of Delegates, the Senate concurring, That the Speaker of the House of Delegates, President pro tempore of the Senate, and President of the Senate shall assemble in the Hall of the House of Delegates on Wednesday, January 13, 2021, at such time as specified by the Speaker of the House of Delegates, to receive the Governor of Virginia, and such address as he may desire to make, and that the rules for the government of the House of Delegates and the Senate, when convened in joint session for such purpose, shall be as follows:

Rule I. At the hour fixed for the meeting of the Joint Assembly, the President pro tempore of the Senate and the President of the Senate, accompanied by the Clerk of the Senate, shall proceed to the Hall of the House of Delegates and shall be received by the Speaker of the House of Delegates and the Clerk of the House of Delegates standing. An appropriate seat shall be assigned to the President pro tempore of the Senate by the Sergeant at Arms of the House. The Speaker of the House of Delegates shall assign an appropriate seat for the President of the Senate.

Rule II. The Speaker of the House of Delegates shall be President of the Joint Assembly. In case it shall be necessary for the Speaker to vacate the Chair, the President of the Senate shall serve as the presiding officer.

Rule III. The Clerk of the House of Delegates shall be Clerk of the Joint Assembly and shall be assisted by the Clerk of the Senate. The Clerk of the Joint Assembly shall enter the proceedings of the Joint Assembly in the Journal of the House and shall certify a copy of the same to the Clerk of the Senate, who shall enter the same in the Journal of the Senate.

Rule IV. The Sergeant at Arms and Doorkeepers of the House shall act as such for the Joint Assembly.

Rule V. The Rules of the House of Delegates, as far as applicable, shall be the rules of the Joint Assembly.

Rule VI. In calling the roll of the Joint Assembly, the names of the Senators shall be called in alphabetical order, then the names of the Delegates in like order, except that the name of the Speaker of the House of Delegates shall be called last.

Rule VII. If, when the Joint Assembly meets, it shall be ascertained that a majority of each house is not present, the Joint Assembly may take measures to secure the attendance of absentees, or adjourn to a succeeding day, as a majority of those present may determine.

Rule VIII. When the Joint Assembly adjourns, the President pro tempore of the Senate and the President of the Senate, accompanied by the Clerk of the Senate, shall return to their chamber, and the business of the House shall be continued in the same order as at the time of the entrance of the President pro tempore of the Senate, the President of the Senate, and the Clerk of the Senate; and, be it

RESOLVED FURTHER, That notwithstanding any other provision of this resolution and in accordance with the practices of each house, with the exception of commending and memorial joint resolutions, a request to be added as a co-patron shall be received prior to the first vote on the passage of

a bill or agreement to a joint resolution or, if the bill or joint resolution is not reported from committee, then prior to the last action on such legislation. A request to be removed as a co-patron shall be received no later than 3:00 p.m., Friday, February 12, 2021; and, be it

RESOLVED FURTHER, That any member offering for introduction a bill or joint resolution not submitted to the Division of Legislative Services for drafting is encouraged to submit an electronic version no later than 5:00 p.m. on the day the legislation is introduced; and, be it

RESOLVED FURTHER, That for purposes of the procedural deadlines established herein for the 2021 Regular Session of the General Assembly:

"Budget Bill" means the general appropriation bill introduced in each house that authorizes the biennial expenditure of public revenues for the period from July 1, 2020, through June 30, 2022.

"Debt bill" means any bill that authorizes the issuance of debt.

"Legislative day" means the period of time that begins with the call to order by the presiding officer and ends when declared adjourned by the presiding officer. Unless another time is specified, any deadline established in this resolution shall expire at the end of the legislative day.

"Prefiled legislation" means any bill or joint resolution requested from the Division of Legislative Services no later than 5:00 p.m., Monday, November 30, 2020, and prefiled no later than 10:00 a.m., Wednesday, January 13, 2021, or any bill or joint resolution not requested from the Division of Legislative Services and prefiled no later than 10:00 a.m., Wednesday, January 13, 2021.

"Revenue bill" means any bill, except the Budget Bill and debt bills, that increases or decreases the total revenues available for appropriation.

"Unanimous consent" means the affirmation of all the members present in the house of origin. Any legislation intended to be offered for introduction with unanimous consent or with the written request of the Governor shall not require the consent of the house in order for the member to request the Division of Legislative Services to draft such legislation. The Division of Legislative Services shall return such legislation after the original introduction deadline.

"Virginia Retirement System bill" means any bill that amends, adds, repeals, or modifies any provision of any retirement system established in Title 51.1 of the Code of Virginia; and, be it

RESOLVED FINALLY, That the 2021 Regular Session of the General Assembly shall be governed by the following procedural rules, which establish time limitations for elections and for all legislation prefiled and introduced for or continued to the 2021 Regular Session except:

- (i) House and Senate resolutions, except for the time limitations established in Rules 18 and 20;
- (ii) Bills and joint resolutions affecting the rules of procedure or the schedule of business of the General Assembly, either of its houses, or any of its committees;
- (iii) Bills and joint resolutions introduced with unanimous consent to exceed the time limitations established in Rules 2, 3, 6, and 16;
 - (iv) Joint resolutions confirming appointments subject to the confirmation of the General Assembly;

- (v) Joint commending and memorial resolutions, except for the time limitations established in Rules 14 and 16;
- (vi) Bills and joint resolutions regarding elections held by the General Assembly during the 2021 Regular Session; or
 - (vii) Bills and joint resolutions requested in writing by the Governor.
- Rule 1. Neither house of the General Assembly shall receive from any committee any bill or joint resolution that was continued on the agenda of such committee and acted upon later than midnight, Thursday, December 3, 2020. For purposes of this rule, a motion to refer a measure to another committee shall be treated as an action by a committee.
- Rule 2. No bill or joint resolution creating or continuing a study shall be offered in either house after the adjournment of that house on Wednesday, January 13, 2021.
- Rule 3. No Virginia Retirement System bill shall be offered in either house after adjournment of that house on Wednesday, January 13, 2021.
- Rule 4. Except for bills and joint resolutions required to be requested earlier, requests for the drafting, redrafting, or correction of any bill or joint resolution shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Friday, January 15, 2021.
- Rule 5. No later than Monday, January 18, 2021, each house shall begin its consideration of any election to fill any judicial seat in the courts of the Commonwealth, or to fill a seat on any commission or office elected by the General Assembly. In the event that the houses cannot agree on such election before Tuesday, January 19, 2021, such election shall become the subject of a special and continuing joint order in each house, and such special and continuing joint order shall have precedence over all other business of either house, until such time as both houses reach agreement on such election or agree to hold it at another specific time. The Rules of each house, as far as applicable, shall be the rules governing such election.
- Rule 6. Except for bills required to be filed earlier, no bill or joint resolution shall be offered in either house after 3:00 p.m., Friday, January 22, 2021.
- Rule 7. No later than Friday, January 22, 2021, the Board of Trustees of the Virginia Retirement System shall submit, in accordance with § 30-19.1:7, impact statements for all Virginia Retirement System bills filed by the first day of session. For any Virginia Retirement System bill filed later than the first day of session, the Board of Trustees shall use due diligence in preparing the impact statement in time for review by the standing committees.
- Rule 8. The committees responsible for the consideration of the Budget Bill in the houses of introduction shall complete their work on such bill no later than midnight, Sunday, February 7, 2021, and any amendments proposed by such committees shall be made available to their respective houses no later than noon, Tuesday, February 9, 2021.
- Rule 9. Except for the Budget Bill, beginning Wednesday, February 10, 2021, the House of Delegates shall consider only Senate bills, Senate joint resolutions, House bills with Senate amendments, and House joint resolutions with Senate amendments; the Senate shall consider only House bills, House joint resolutions, Senate bills with House amendments, and Senate joint resolutions with House amendments; and each house may consider conference reports and other privileged matters relating thereto to the end that the work of each house may be disposed of by the other.

- Rule 10. The houses of introduction shall complete their consideration of the Budget Bill, except for conference reports and other privileged matters relating thereto, no later than Thursday, February 11, 2021.
- Rule 11. The committees responsible for the consideration of revenue bills of the other house shall complete their consideration of such bills no later than midnight, Tuesday, February 16, 2021.
- Rule 12. No later than Wednesday, February 17, 2021, each house shall complete its consideration of the Budget Bill and all revenue bills of the other house, except for conference reports and other privileged matters relating thereto, and the appointing authority shall appoint the conferees to such bills.
- Rule 13. No later than Wednesday, February 17, 2021, each house shall begin its consideration of any election to fill any judicial seat in the courts of the Commonwealth, or to fill a seat on any commission or office elected by the General Assembly. In the event that the houses cannot agree on such election before Thursday, February 18, 2021, such election shall become the subject of a special and continuing joint order in each house, and such special and continuing joint order shall have precedence over all other business of either house, until such time as both houses reach agreement on such election, or either house votes to suspend or discharge the order. The Rules of each house, as far as applicable, shall be the rules governing such election.
- Rule 14. Requests for the drafting, redrafting, or correction of any joint commending or memorial resolution shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Thursday, February 18, 2021.
- Rule 15. Any conference committee on any revenue bills shall complete its deliberations and make the report of such conference available to the General Assembly as soon as practicable.
- Rule 16. No joint commending or memorial resolution shall be offered in either house after 5:00 p.m., Monday, February 22, 2021.
- Rule 17. Beginning Tuesday, February 23, 2021, neither house shall receive from any committee any bill or joint resolution acted on by any committee later than midnight, Monday, February 22, 2021.
- Rule 18. Requests for the drafting, redrafting, or correction of any single-house commending or memorial resolution shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Tuesday, February 23, 2021.
- Rule 19. Any conference committee on the Budget Bill shall complete its deliberations and make the report of such conference available to the General Assembly as soon as practicable. Neither house shall receive, consider, or vote on any Budget Bill that is in conference unless it has been agreed to in writing or signed electronically by a majority of conferees from each house. Neither house shall consider such conference report earlier than 48 hours after receipt, unless both houses respectively determine to proceed earlier by a vote of two-thirds of the members voting in each house. No engrossment of the Budget Bill shall be required in either house, and any conference on the Budget Bill shall consider, as the basis of its deliberations, the Budget Bill as recommended by the Governor and introduced in the House and the amendments thereto proposed by each house. A report shall be issued concurrently with the report of the conference committee that identifies the following by item number, narrative description, and dollar amount: (i) any nonstate agency appropriation, (ii) any item in the conference report that was not included in a general appropriation bill as passed by either the House or the Senate, and (iii) any item that represents legislation that failed in either house during the regular or a special session.
- Rule 20. No single-house commending or memorial resolution shall be offered in either house after 5:00 p.m., Thursday, February 25, 2021.

Rule 21. Except for joint resolutions affecting the rules of procedure or the schedule of business of the General Assembly, beginning Friday, February 26, 2021, the House of Delegates shall consider only Senate joint resolutions and House joint resolutions with Senate amendments; the Senate shall consider only House joint resolutions and Senate joint resolutions with House amendments; and each house may consider conference reports or joint resolutions and other privileged matters relating thereto, to the end that the work of each house may be disposed of by the other.

Rule 22. This session of the General Assembly shall be extended beyond the 30-day period provided in Section 6 of Article IV of the Constitution of Virginia and shall adjourn sine die no later than Saturday, February 27, 2021.

Rule 23. Pursuant to Section 6 of Article IV of the Constitution of Virginia, the General Assembly shall reconvene Wednesday, April 7, 2021, for the purpose of considering bills and items of appropriation bills that may have been returned by the Governor with recommendations for their amendment, and bills and items of appropriation bills, including the general appropriation act, that may have been returned by the Governor with his objections.

Rule 24. That members of the General Assembly and credentialed legislative staff attending floor sessions, committees, subcommittees, and any other meetings of the General Assembly in person or meeting in a legislative office in person shall be required to wear a facemask or face shield, and (i) at the discretion of the Speaker of the House of Delegates for Delegates and House credentialed staff and the discretion of the chair of the Senate Committee on Rules for Senators and Senate credentialed staff, have their temperature taken daily when initially entering legislative space; and (ii) at the discretion of the Speaker of the House of Delegates for Delegates and House credentialed staff and the discretion of the chair of the Senate Committee on Rules for Senators and Senate credentialed staff, be tested weekly for Covid-19 by the Virginia Department of Health.

Rule 25. The conduct of the business of any subcommittee of any House committee, any joint subcommittee of House and Senate committees, and any interim study commission created pursuant to a House measure shall be governed by the Rules of the House of Delegates; the conduct of the business of any subcommittee of any Senate committee, any joint subcommittee of Senate and House committees, and any interim study commission created pursuant to a Senate measure shall be governed by the Rules of the Senate. If a House measure and a Senate measure create the same study, the conduct of the business of the study shall be governed by the rules of the house of the chairman of the study, or in the case of co-chairmen, the rules of the house as agreed upon by the co-chairmen.

Rule 26. Interim meetings of any standing committee, joint committee, joint subcommittee, legislative commission, or any other interim study subcommittee or study commission shall be held on such day as called by the chair if electronically, or if in person on Monday, Tuesday, or Wednesday during the first and third full weeks of the month, unless otherwise authorized by the Speaker of the House of Delegates or the Chairman of the Senate Committee on Rules, as may be appropriate for the house in which the chairman serves.

Rule 27. Any staff member assigned to work for, and support the efforts of, any committee of the House or Senate, any subcommittee of any such committee, any joint subcommittee of House and Senate committees, or any interim study commission shall work under the direction of the chairman of such committee, subcommittee, joint subcommittee, or interim study commission.

On motion of Senator Surovell, the reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

Senator Locke offered the following amendment to the substitute:

1. Line 249, substitute, after space strike

; and (ii) through line 252

insert

. (period)

On motion of Senator Locke, the reading of the amendment was waived.

On motion of Senator Locke, the amendment was agreed to.

The substitute with amendment was ordered to be engrossed.

Senator Surovell moved that H.J.R. 575 be agreed to.

The question was put on agreeing to H.J.R. 575.

H.J.R. 575 was rejected, having failed to receive the necessary affirmative votes required by Article IV, Section 6, of the Constitution.

The recorded vote is as follows:

YEAS--22, NAYS--17, RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Stuart, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--0.

RECONSIDERATION

Senator Suetterlein moved to reconsider the vote by which **H.J.R.** 575 (five hundred seventy-five) was rejected.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

RECONSIDERATION

Senator Locke moved to reconsider the vote by which the amendment offered by Senator Locke to the substitute to **H.J.R.** 575 (five hundred seventy-five) was agreed to.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Locke withdrew the amendment.

RECONSIDERATION

Senator Surovell moved to reconsider the vote by which the substitute offered by Senator Surovell to **H.J.R.** 575 (five hundred seventy-five) was agreed to.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Surovell withdrew the substitute.

Senator Locke offered the following amendment:

```
1. Line 140, engrossed, after space
```

strike

; and (ii) through line 143

insert

. (period)

On motion of Senator Locke, the reading of the amendment was waived.

On motion of Senator Locke, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.J.R. 575, on motion of Senator Locke, was agreed to.

The recorded vote is as follows: YEAS--30. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Reeves, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--30.

NAYS--Chase, DeSteph, Dunnavant, McDougle, Obenshain, Peake, Pillion, Ruff, Stuart--9. RULE 36--0.

Senator Locke was ordered to inform the House of Delegates thereof.

IMMEDIATE CONSIDERATION

Senator Locke moved that the Rules be suspended, the Committee on Rules be discharged from further consideration of **H.J.R. 576** (five hundred seventy-six), the readings of the title be waived, and the resolution be taken up for immediate consideration.

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

HOUSE JOINT RESOLUTION NO. 576

Establishing a schedule for the conduct of business for the prefiling period of the 2022 Regular Session of the General Assembly of Virginia.

RESOLVED by the House of Delegates, the Senate concurring, That the prefiling period of the 2022 Regular Session of the General Assembly shall be governed by the following rules:

- Rule 1. Requests for drafts of any bill or joint resolution to be prefiled shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Monday, November 29, 2021. The Division shall make such drafts available for review no later than midnight, Friday, December 31, 2021.
- Rule 2. Requests for the drafting, redrafting, or correction of any bill or joint resolution creating or continuing a study shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Friday, January 7, 2022, in order to be filed on the first day of the 2022 Regular Session.
- Rule 3. Requests for redrafts and corrections of any draft prepared for prefiling shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Friday, January 7, 2022. The Division shall make such drafts available no later than noon, Tuesday, January 11, 2022.
- Rule 4. Bills and joint resolutions offered for prefiling shall be prefiled in either house no later than 10:00 a.m., Wednesday, January 12, 2022. Any member offering for prefiling a bill or joint resolution not submitted to the Division of Legislative Services for drafting is encouraged to submit an electronic version no later than 5:00 p.m. on the day the legislation is prefiled.

H.J.R. 576, on motion of Senator Locke, was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Locke was ordered to inform the House of Delegates thereof.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 13, 2021

THE HOUSE OF DELEGATES HAS AGREED TO THE AMENDMENT PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE JOINT RESOLUTION:

H.J.R. 575. Providing for a Joint Assembly, establishing a schedule for the conduct of business coming before the 2021 Regular Session of the General Assembly of Virginia, and providing for legislative continuity between the 2021 Regular Session of the General Assembly and a special session.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--37. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Deeds--1.

RULE 36--0.

INTRODUCTION OF LEGISLATION

The following were prefiled on the dates indicated, ordered to be printed, and referred pursuant to \S 30-19.3 of the Code of Virginia:

S.B. 1097. A BILL to amend and reenact §§ 24.2-702.1, 24.2-704, 24.2-706, and 24.2-707 of the Code of Virginia, relating to absentee voting; witness signature not required.

(Prefiled November 9, 2020)

Patron--Favola

Referred to Committee on Privileges and Elections

S.B. 1098. A BILL to amend and reenact § 5.1-5 of the Code of Virginia, relating to aircraft registration; unmanned aircraft.

(Prefiled November 18, 2020)

Patron--Favola

Referred to Committee on Transportation

S.B. 1099. A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; special education programs.

(Prefiled November 19, 2020)

Patron--Stuart

Referred to Committee on Education and Health

S.B. 1100. A BILL to amend and reenact Chapter 1289 of the 2020 Acts of Assembly, as amended by Chapter 56 of the 2020 Acts of Assembly, Special Session I, which appropriated funds for the 2020-22 Biennium and provided a portion of revenues for the two years ending, respectively, on the thirtieth day of June, 2021, and the thirtieth day of June, 2022.

(Prefiled December 16, 2020)

Patron--Howell

Referred to Committee on Finance and Appropriations

S.B. 1101. A BILL to amend the Code of Virginia by adding in Chapter 2 of Title 24.2 an article numbered 2.1, consisting of sections numbered 24.2-209.1 and 24.2-209.2, relating to the presidential electors and the Agreement Among the States to Elect the President by National Popular Vote Compact.

(Prefiled November 27, 2020)

Patrons--Ebbin and Lucas

Referred to Committee on Privileges and Elections

S.B. 1102. A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 10 of Title 32.1 a section numbered 32.1-331.04, relating to Department of Medical Assistance Services; personal care aides; orientation program.

(Prefiled December 1, 2020)

Patron--Locke

Referred to Committee on Education and Health

S.B. 1103. A BILL to amend and reenact § 2.2-3703 of the Code of Virginia, relating to the Virginia Freedom of Information Act; Virginia Parole Board member votes.

(Prefiled December 2, 2020)

Patron--Suetterlein

Referred to Committee on General Laws and Technology

S.B. 1104. A BILL to amend and reenact §§ 53.1-136 and 53.1-155 of the Code of Virginia, relating to parole; notice and certification; monthly reports.

(Prefiled December 2, 2020)

Patron--Obenshain

Referred to Committee on Rehabilitation and Social Services

S.B. 1105. A BILL to amend the Code of Virginia by adding in Title 19.2 a chapter numbered 19.4, consisting of a section numbered 19.2-327.15, relating to post-conviction relief; previously admitted forensic scientific evidence.

(Prefiled December 2, 2020)

Patron--Stanley

Referred to Committee on the Judiciary

S.B. 1106. A BILL to amend the Code of Virginia by adding in Article 3 of Chapter 9 of Title 22.1 a section numbered 22.1-141.3, relating to the establishment of the Public School Assistance Fund and Program.

(Prefiled December 2, 2020)

Patron--Stanley

Referred to Committee on Education and Health

S.B. 1107. A BILL to amend and reenact § 8.01-581.15 of the Code of Virginia, relating to limitation on recovery in certain medical malpractice actions.

(Prefiled December 2, 2020)

Patron--Stanley

Referred to Committee on the Judiciary

S.B. 1108. A BILL to amend and reenact §§ 8.01-195.4, 16.1-77, 43-34, and 46.2-644.03 of the Code of Virginia, relating to general district courts; jurisdictional limits.

(Prefiled December 2, 2020)

Patron--Stanley

Referred to Committee on the Judiciary

S.B. 1109. A BILL to provide for a statewide advisory referendum relating to the issuance of state general obligation bonds for school facility modernization.

(Prefiled December 7, 2020)

Patron--Stanley

Referred to Committee on Privileges and Elections

S.B. 1110. A BILL to amend and reenact § 55.1-1004 of the Code of Virginia, relating to property; duties of real estate settlement agents.

(Prefiled December 10, 2020)

Patron--Spruill

Referred to Committee on General Laws and Technology

S.B. 1111. A BILL to amend and reenact § 24.2-606 of the Code of Virginia, relating to elections; preservation of order at the polls; powers of officers of election.

(Prefiled December 15, 2020)

Patron--Spruill

Referred to Committee on Privileges and Elections

S.B. 1112. A BILL to amend and reenact §§ 58.1-439.12:08 and 58.1-439.12:11 of the Code of Virginia, relating to research and development tax credits.

(Prefiled December 15, 2020)

Patron--Locke

Referred to Committee on Finance and Appropriations

S.B. 1113. A BILL to amend the Code of Virginia by adding a section numbered 18.2-83.1, relating to communicating threats of death or bodily injury to a person at any place of assembly, any building or other structure, or any means of transportation; penalty.

(Prefiled December 18, 2020)

Patron--Spruill

Referred to Committee on the Judiciary

S.B. 1114. A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; eligibility for state pool of funds.

(Prefiled December 21, 2020)

Patron--Peake

Referred to Committee on Education and Health

S.B. 1115. A BILL to amend and reenact §§ 3.2-4112 and 54.1-3401 of the Code of Virginia, relating to industrial hemp; promotion of commerce.

(Prefiled December 21, 2020)

Patron--Peake

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1116. A BILL to amend and reenact §§ 32.1-46, as it is currently effective and as it shall become effective, and 32.1-48 of the Code of Virginia, relating to powers of State Health Commissioner in epidemic; vaccine; religious tenets or practices.

(Prefiled December 21, 2020)

Patron--Peake

Referred to Committee on Education and Health

S.B. 1117. A BILL to amend and reenact § 32.1-46, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to immunizations; religious tenets or practices.

(Prefiled December 21, 2020)

Patron--Peake

Referred to Committee on Education and Health

S.B. 1118. A BILL to amend and reenact § 24.2-653 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 24.2-418.01 and 24.2-653.4, relating to voter registration; verification of social security numbers; provisional registration status.

(Prefiled December 21, 2020)

Patron--Peake

Referred to Committee on Privileges and Elections

S.B. 1119. A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 1 of Title 9.1 a section numbered 9.1-116.7, relating to law-enforcement agencies; body-worn camera systems.

(Prefiled December 22, 2020)

Patron--Reeves

Referred to Committee on the Judiciary

S.B. 1120. A BILL to amend and reenact §§ 15.2-520 and 15.2-2506 of the Code of Virginia, relating to county executive form of government; local budgets.

(Prefiled December 22, 2020)

Patron--Reeves

Referred to Committee on Local Government

S.B. 1121. A BILL to amend and reenact § 32.1-269 of the Code of Virginia and to amend the Code of Virginia by adding in Article 6 of Chapter 7 of Title 32.1 a section numbered 32.1-269.2, relating to birth certificates; amendments.

(Prefiled December 22, 2020)

Patron--Locke

Referred to Committee on Education and Health

S.B. 1122. A BILL to amend and reenact §§ 8.01-9, 8.01-407, 16.1-77, 16.1-305, 17.1-213, 19.2-389, as it is currently effective and as it shall become effective, 46.2-301, 46.2-301.1, 46.2-411, and 53.1-21 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 46.2-301.2; and to repeal Article 9 (§§ 46.2-355.1 through 46.2-363) of Chapter 3 of Title 46.2 of the Code of Virginia, relating to habitual offenders; repeal.

(Prefiled December 22, 2020)

Patron--Stanley

Referred to Committee on the Judiciary

S.B. 1123. A BILL to amend the Code of Virginia by adding in Article 5 of Chapter 4 of Title 64.2 a section numbered 64.2-454.1, relating to will contest; presumption of undue influence.

(Prefiled December 22, 2020)

Patron--Obenshain

Referred to Committee on the Judiciary

S.B. 1124. A BILL to amend and reenact § 64.2-403 of the Code of Virginia, relating to execution of wills; witnesses.

(Prefiled December 22, 2020)

Patron--Obenshain

Referred to Committee on the Judiciary

S.B. 1125. A BILL to amend and reenact §§ 53.1-136 and 53.1-155 of the Code of Virginia, relating to Parole Board; notice to victim.

(Prefiled December 22, 2020)

Patron--Obenshain

Referred to Committee on Rehabilitation and Social Services

S.B. 1126. A BILL to amend and reenact § 33.2-1907 of the Code of Virginia, relating to the Transportation District Commission of Hampton Roads; membership.

(Prefiled December 23, 2020)

Patron--Spruill

Referred to Committee on Rules

S.B. 1127. A BILL to amend and reenact §§ 18.2-340.28 and 18.2-340.28:1 of the Code of Virginia, relating to charitable gaming; conduct of instant bingo, network bingo, pull tabs, and seal cards. (Prefiled December 23, 2020)

Patron--Reeves

Referred to Committee on General Laws and Technology

S.B. 1128. A BILL to amend and reenact § 44 and § 133, as amended, of Chapter 34 of the Acts of Assembly of 1918 and to repeal § 61 of Chapter 34 of the Acts of Assembly of 1918, which provided a charter for the City of Norfolk, relating to general updates.

(Prefiled December 30, 2020)

Patrons--Spruill; Delegate: Guy

Referred to Committee on Local Government

S.B. 1129. A BILL to amend and reenact §§ 18.2-282 and 18.2-433.2 of the Code of Virginia, relating to veterans service organizations; paramilitary activities.

(Prefiled December 30, 2020)

Patron--Reeves

Referred to Committee on the Judiciary

S.B. 1130. A BILL to amend the Code of Virginia by adding in Article 5 of Chapter 36 of Title 58.1 a section numbered 58.1-3668, relating to personal property tax exemption; motor vehicle of a disabled veteran.

(Prefiled December 30, 2020)

Patron--Reeves

Referred to Committee on Finance and Appropriations

S.B. 1131. A BILL to amend and reenact § 44-146.17, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to Emergency Services and Disaster Law; limitation on duration of executive orders.

(Prefiled December 31, 2020)

Patron--Suetterlein

Referred to Committee on General Laws and Technology

S.B. 1132. A BILL to amend and reenact § 22.1-98 of the Code of Virginia, relating to public schools; severe weather conditions and other emergency situations; unscheduled remote learning days.

(Prefiled December 31, 2020)

Patron--Suetterlein

Referred to Committee on Education and Health

S.B. 1133. A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; eligibility for state pool of funds; pilot program related to educational placement transition for certain students with disabilities.

(Prefiled December 31, 2020)

Patron--Suetterlein

Referred to Committee on Education and Health

S.B. 1134. A BILL to amend and reenact §§ 2 and 4 of the first enactment of Chapter 265 and §§ 2 and 4 of the first enactment of Chapter 408 of the Acts of Assembly of 1992, relating to the issuance of Commonwealth of Virginia Article X, Section 9 (c) Refunding Bonds, subject to the provisions of Article X, Sections 9 (a) and 9 (c) of the Constitution of Virginia; emergency.

EMERGENCY

(Prefiled December 31, 2020)

Patron--Howell

Referred to Committee on Finance and Appropriations

S.B. 1135. A BILL to amend and reenact §§ 3.2-6540 and 3.2-6542 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 3.2-6540.01 through 3.2-6540.04, 3.2-6541.1, 3.2-6542.1, 3.2-6542.2, 3.2-6543.1, 3.2-6562.2, and 18.2-52.2 relating to dangerous dogs; penalty.

(Prefiled January 1, 2021)

Patron--Marsden

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1136. A BILL to amend and reenact § 46.2-1063 of the Code of Virginia and to repeal §§ 46.2-746.6, 46.2-746.9, 46.2-746.12, 46.2-747, 46.2-748, 46.2-749.10, and 46.2-749.69:1 of the Code of Virginia and to repeal § 1 of Chapter 776 of the Acts of Assembly of 2010, relating to special license plates; removal.

(Prefiled January 2, 2021)

Patron--Marsden

Referred to Committee on Transportation

S.B. 1137. A BILL to amend and reenact § 51.1-155 of the Code of Virginia, relating to Virginia Retirement System; retired law-enforcement officers employed as school security officers.

(Prefiled January 3, 2021)

Patron--Cosgrove

Referred to Committee on Finance and Appropriations

S.B. 1138. A BILL to amend and reenact §§ 16.1-69.55, 17.1-805, 18.2-52.1, 18.2-346.1, 19.2-8, 19.2-299, 19.2-310.2, 19.2-392.02, as it is currently effective and as it shall become effective, 32.1-291.16, 54.1-2982, 54.1-2983, and 57-48 of the Code of Virginia and to repeal §§ 18.2-62, 18.2-67.4:1, and 32.1-289.2 of the Code of Virginia, relating to sexually transmitted infections; infected sexual battery; repeal.

(Prefiled January 3, 2021)

Patron--Locke

Referred to Committee on the Judiciary

S.B. 1139. A BILL to amend the Code of Virginia by adding in Title 46.2 a chapter numbered 10.1, consisting of a section numbered 46.2-1193, relating to animal-drawn vehicle equipment.

(Prefiled January 4, 2021)

Patron--Peake

Referred to Committee on Transportation

S.B. 1140. A BILL to amend the Code of Virginia by adding a section numbered 55.1-604.1, relating to gifts of real estate; requirements.

(Prefiled January 4, 2021)

Patron--Cosgrove

Referred to Committee on the Judiciary

S.B. 1141. A BILL to amend and reenact § 15.2-5102.1 of the Code of Virginia, relating to the Hampton Roads area refuse collection authority; financial planning.

(Prefiled January 4, 2021)

Patron--Cosgrove

Referred to Committee on Local Government

S.B. 1142. A BILL to amend and reenact § 20-25 of the Code of Virginia, relating to persons who may celebrate rites of marriage; members of the General Assembly.

(Prefiled January 5, 2021)

Patron--Cosgrove

Referred to Committee on the Judiciary

S.B. 1143. A BILL to extend certain wetlands permits through 2021.

(Prefiled January 5, 2021)

Patron--Cosgrove

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1144. A BILL to amend and reenact §§ 5.1-5, 5.1-51, and 58.1-1509 of the Code of Virginia, relating to aircraft; registration and licensing.

(Prefiled January 5, 2021)

Patron--Stuart

Referred to Committee on Transportation

S.B. 1145. A BILL to authorize the issuance of bonds, in an amount up to \$34,136,000 plus financing costs, pursuant to Article X, Section 9 (c) of the Constitution of Virginia, for paying costs of acquiring, constructing and equipping revenue-producing capital projects at institutions of higher learning of the Commonwealth; to authorize the Treasury Board, by and with the consent of the Governor, to fix the details of such bonds, to provide for the sale of such bonds, and to issue notes to borrow money in anticipation of the issuance of the bonds; to provide for the pledge of the net revenues of such capital projects and the full faith, credit and taxing power of the Commonwealth for the payment of such bonds; to provide that the interest income on such bonds and notes shall be exempt from all taxation by the Commonwealth and any political subdivision thereof; emergency.

EMERGENCY

(Prefiled January 5, 2021)

Patron--Howell

Referred to Committee on Finance and Appropriations

S.B. 1146. A BILL to amend and reenact § 58.1-301 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code; emergency.

EMERGENCY

(Prefiled January 5, 2021)

Patron--Howell

Referred to Committee on Finance and Appropriations

S.B. 1147. A BILL to amend and reenact § 32.1-122.6:04 of the Code of Virginia, relating to Nurse Loan Repayment Program; certified nurse aide.

(Prefiled January 6, 2021)

Patron--Kiggans

Referred to Committee on Education and Health

S.B. 1148. A BILL to amend and reenact §§ 24.2-311, 24.2-503, 24.2-507, 24.2-510, 24.2-515, and 24.2-515.1 of the Code of Virginia, relating to elections; date of June primary election.

(Prefiled January 6, 2021)

Patron--Kiggans

Referred to Committee on Privileges and Elections

S.B. 1149. A BILL to amend and reenact § 32.1-127 of the Code of Virginia, relating to nursing home standards of care and staff requirements; regulations.

(Prefiled January 6, 2021)

Patron--Kiggans

Referred to Committee on Education and Health

S.B. 1150. A BILL to amend the Code of Virginia by adding a section numbered 2.2-2002.2, relating to Department of Veterans Services; Military Spouse Liaison; position created.

(Prefiled January 6, 2021)

Patron--Kiggans

Referred to Committee on General Laws and Technology

S.B. 1151. A BILL to amend and reenact § 58.1-322.02 of the Code of Virginia, relating to income tax subtraction; veteran retirement income.

(Prefiled January 6, 2021)

Patron--Kiggans

Referred to Committee on Finance and Appropriations

S.B. 1152. A BILL to amend and reenact §§ 1 and 4 of the charter of the Town of Appomattox, which was granted by order of the Circuit Court of the County of Appomattox on June 2, 1925, and as amended by Chapter 43 of the Acts of Assembly of 1980, relating to election and appointment of officers; time of election.

(Prefiled January 7, 2021)

Patron--Peake

Referred to Committee on Local Government

S.B. 1153. A BILL to amend and reenact § 24.2-712 of the Code of Virginia, relating to elections; absentee voting; ballots to be sorted and counted by precinct.

(Prefiled January 7, 2021)

Patron--Suetterlein

Referred to Committee on Privileges and Elections

S.B. 1154. A BILL to amend and reenact § 37.2-304 of the Code of Virginia, relating to Commissioner of Behavioral Health and Developmental Services; reports to designated protection and advocacy system.

(Prefiled January 7, 2021)

Patron--Favola

Referred to Committee on Education and Health

S.B. 1155. A BILL to create a six-year capital outlay plan for projects to be funded entirely or partially from general fund-supported resources and to repeal Chapter 1134 of the Acts of Assembly of 2020. (Prefiled January 7, 2021)

Patron--Howell

Referred to Committee on Finance and Appropriations

S.B. 1156. A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 22.17, consisting of a section numbered 59.1-284.38, relating to creation of the Technology Development Grant Fund.

(Prefiled January 7, 2021)

Patron--Howell

Referred to Committee on Finance and Appropriations

S.B. 1157. A BILL to amend and reenact § 15.2-1400 of the Code of Virginia, relating to time of certain local elections.

(Prefiled January 7, 2021)

Patron--Spruill

Referred to Committee on Local Government

S.B. 1158. A BILL to amend and reenact §§ 58.1-439.12:06, 58.1-439.12:09, and 58.1-439.12:10 of the Code of Virginia, relating to Port of Virginia tax credits; sunset.

(Prefiled January 7, 2021)

Patron--Spruill

Referred to Committee on Finance and Appropriations

S.B. 1159. A BILL to amend the Code of Virginia by adding a section numbered 40.1-28.6:1, relating to the ability of employees to use sick leave for the care of immediate family members.

(Prefiled January 7, 2021)

Patron--Favola

Referred to Committee on Commerce and Labor

S.B. 1160. A BILL to amend and reenact § 46.2-1212.1 of the Code of Virginia, relating to removal of vehicles and cargoes involved in accidents; liability.

(Prefiled January 7, 2021)

Patrons--Suetterlein and Edwards

Referred to Committee on Transportation

S.B. 1161. A BILL to amend and reenact § 10.1-502 of the Code of Virginia, relating to Soil and Water Conservation Board; membership.

(Prefiled January 8, 2021)

Patron--Hanger

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1162. A BILL to amend and reenact §§ 58.1-339.3 and 58.1-439.5 of the Code of Virginia, relating to tax credit; agricultural best management practices.

(Prefiled January 8, 2021)

Patron--Hanger

Referred to Committee on Finance and Appropriations

S.B. 1163. A BILL to amend and reenact §§ 58.1-334, 58.1-337, 58.1-432, and 58.1-436 of the Code of Virginia, relating to tax credits of agricultural equipment.

(Prefiled January 8, 2021)

Patron--Hanger

Referred to Committee on Finance and Appropriations

S.B. 1164. A BILL to amend and reenact § 10.1-1400 of the Code of Virginia, relating to advanced recycling; definition; not considered solid waste management.

(Prefiled January 8, 2021)

Patron--Hanger

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1165. A BILL to amend and reenact §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, as it is currently effective and as it shall become effective, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307 of the Code of Virginia and to repeal §§ 8.01-654.1, 8.01-654.2, 17.1-313, and 18.2-17, Article 4.1 (§§ 19.2-163.7 and 19.2-163.8) of Chapter 10 of Title 19.2, Article 4.1 (§§ 19.2-264.2 through 19.2-264.5) of Chapter 15 of Title 19.2, § 53.1-230, and Chapter 13 (§§ 53.1-232 through 53.1-236) of Title 53.1 of the Code of Virginia, relating to abolition of the death penalty.

(Prefiled January 8, 2021)

Patron--Surovell

Referred to Committee on the Judiciary

S.B. 1166. A BILL to amend the Code of Virginia by adding in Article 3 of Chapter 3 of Title 58.1 a section numbered 58.1-339.13, relating to nursing professional tax credit.

(Prefiled January 8, 2021)

Patron--Kiggans

Referred to Committee on Finance and Appropriations

S.B. 1167. A BILL to amend and reenact §§ 54.1-3018, 54.1-3021, and 54.1-3024 of the Code of Virginia, relating to Board of Nursing; licensure or certification by endorsement for members of the United States military.

(Prefiled January 8, 2021)

Patron--Kiggans

Referred to Committee on Education and Health

S.B. 1168. A BILL to amend and reenact § 16.1-228 of the Code of Virginia, relating to definition of abused or neglected child.

(Prefiled January 8, 2021)

Patron--Lucas

Referred to Committee on the Judiciary

S.B. 1169. A BILL to amend the Code of Virginia by adding a section numbered 22.1-205.1, relating to high school student parking passes; valid driver's license or driver privilege card required.

(Prefiled January 8, 2021)

Patron--Norment

Referred to Committee on Education and Health

S.B. 1170. A BILL to amend and reenact §§ 58.1-602 and 58.1-605 of the Code of Virginia, relating to additional local sales and use tax to support schools.

(Prefiled January 8, 2021)

Patron--Norment

Referred to Committee on Finance and Appropriations

S.B. 1171. A BILL to amend and reenact § 2.2-3115 of the Code of Virginia, relating to the State and Local Government Conflict of Interests Act; disclosure requirements; industrial development authorities and economic development authorities.

(Prefiled January 8, 2021)

Patron--Norment

Referred to Committee on General Laws and Technology

S.B. 1172. A BILL to amend and reenact § 30-194 of the Code of Virginia, relating to the Capitol Square Preservation Council; powers and duties; review and approval of plans for changes to artifacts contained within the Capitol Building.

(Prefiled January 8, 2021)

Patron--Norment

Referred to Committee on Rules

S.B. 1173. A BILL to amend and reenact §§ 3.2-3100 and 3.2-3102 of the Code of Virginia, relating to Tobacco Region Revitalization Commission; membership.

(Prefiled January 8, 2021)

Patron--Norment

Referred to Committee on Rules

S.B. 1174. A BILL to amend and reenact § 2.2-435.12 of the Code of Virginia, relating to the Director of Diversity, Equity, and Inclusion; reporting requirements.

(Prefiled January 8, 2021)

Patron--Norment

Referred to Committee on General Laws and Technology

S.B. 1175. A BILL to amend and reenact § 22.1-32 of the Code of Virginia, relating to Brunswick County school board; appointed school board salaries.

(Prefiled January 8, 2021)

Patron--Ruff

Referred to Committee on Education and Health

S.B. 1176. A BILL to amend and reenact §§ 37.2-314, 37.2-416, and 37.2-506 of the Code of Virginia, relating to the Department of Behavioral Health and Developmental Services; barrier crimes.

(Prefiled January 8, 2021)

Patron--Ruff

Referred to Committee on Education and Health

S.B. 1177. A BILL to amend and reenact § 3.2-5802 of the Code of Virginia, relating to public weighmasters; license.

(Prefiled January 8, 2021)

Patron--Ruff

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1178. A BILL to repeal § 54.1-2957.21 of the Code of Virginia, relating to genetic counseling; conscience clause.

(Prefiled January 8, 2021)

Patron--Ebbin

Referred to Committee on Education and Health

S.B. 1179. A BILL to amend and reenact §§ 2.2-1837, 2.2-3703, 8.01-195.10, 8.01-690, 53.1-1, as it is currently effective and as it shall become effective, 53.1-31.1, 53.1-261, 53.1-262, and 53.1-265 of the Code of Virginia, and to repeal §§ 53.1-263, 53.1-264, and 53.1-266 of the Code of Virginia, relating to Corrections Private Management Act; name change; private management prohibited.

(Prefiled January 8, 2021)

Patron--Ebbin

Referred to Committee on Rehabilitation and Social Services

S.B. 1180. A BILL to amend and reenact § 8.01-267.1 of the Code of Virginia, relating to civil actions filed on behalf of multiple persons.

(Prefiled January 8, 2021)

Patron--Surovell

Referred to Committee on the Judiciary

S.B. 1181. A BILL to amend and reenact § 16.1-241 of the Code of Virginia, relating to special immigrant juvenile status; jurisdiction.

(Prefiled January 8, 2021)

Patron--Surovell

S.B. 1182. A BILL to amend and reenact §§ 46.2-419, 46.2-472, and 46.2-2057 of the Code of Virginia, relating to motor vehicle liability insurance coverage limits.

(Prefiled January 8, 2021)

Patron--Surovell

Referred to Committee on Commerce and Labor

S.B. 1183. A BILL to amend and reenact §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953 of the Code of Virginia, relating to the Property Owners' Association Act; the Condominium Act; use of electronic means for meetings and voting. (Prefiled January 8, 2021)

Patron--Dunnavant

Referred to Committee on General Laws and Technology

S.B. 1184. A BILL to amend and reenact §§ 16.1-351, 16.1-352, and 16.1-353 of the Code of Virginia, relating to standby guardianship; triggering event.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on the Judiciary

S.B. 1185. A BILL to amend and reenact § 51.5-160 of the Code of Virginia, relating to auxiliary grants; assisted living facilities.

(Prefiled January 8, 2021)

Patron--Dunnavant

Referred to Committee on Rehabilitation and Social Services

S.B. 1186. A BILL to amend and reenact § 10.1-1408.4 of the Code of Virginia, relating to landfill siting; historic preservation.

(Prefiled January 10, 2021)

Patron--Hashmi

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1187. A BILL to amend and reenact § 54.1-3482 of the Code of Virginia, relating to the Department of Health Professions; practice of physical therapy.

(Prefiled January 10, 2021)

Patron--Hashmi

Referred to Committee on Education and Health

S.B. 1188. A BILL to amend the Code of Virginia by adding in Title 3.2 a chapter numbered 47.1, consisting of sections numbered 3.2-4780 through 3.2-4783, relating to the Virginia Agriculture Food Assistance Program and Fund; established.

(Prefiled January 10, 2021)

Patron--Hashmi

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1189. A BILL to amend the Code of Virginia by adding a section numbered 54.1-2956.7:1, relating to Occupational Therapy Interjurisdictional Licensure Compact.

(Prefiled January 10, 2021)

Patron--Hashmi

Referred to Committee on Education and Health

S.B. 1190. A BILL to direct the Board of Education to include advanced directive education in its curriculum framework for the Health Standards of Learning for high school students.

(Prefiled January 11, 2021)

Patron--Kiggans

Referred to Committee on Education and Health

S.B. 1191. A BILL to amend and reenact §§ 22.1-253.13:2, 22.1-274, and 22.1-274.01:1 of the Code of Virginia, relating to school personnel; school nurses.

(Prefiled January 11, 2021)

Patron--Kiggans

Referred to Committee on Education and Health

S.B. 1192. A BILL to direct the Department of Health Professions to amend its regulations related to naturopathic doctors.

(Prefiled January 11, 2021)

Patron--Kiggans

Referred to Committee on Education and Health

S.B. 1193. A BILL to amend the Code of Virginia by adding in Title 3.2 a chapter numbered 33.1, consisting of sections numbered 3.2-3304, 3.2-3305, and 3.2-3306, relating to Dairy Producer Margin Coverage Premium Assistance Program.

(Prefiled January 11, 2021)

Patron--Obenshain

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1194. A BILL to amend and reenact the second enactment of Chapter 574 of the Acts of Assembly of 2017, relating to produce safety; sunset.

(Prefiled January 11, 2021)

Patron--Obenshain

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1195. A BILL to amend and reenact § 38.2-2206 of the Code of Virginia, relating to motor vehicle insurance; underinsured motor vehicle.

(Prefiled January 11, 2021)

Patron--Obenshain

Referred to Committee on Commerce and Labor

S.B. 1196. A BILL to amend and reenact §§ 22.1-253.13:5 and 22.1-298.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-298.7, relating to teachers and other licensed school board employees; cultural competency.

(Prefiled January 11, 2021)

Patron--Locke

Referred to Committee on Education and Health

S.B. 1197. A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 58.1 an article numbered 13.4, consisting of sections numbered 58.1-439.29 and 58.1-439.30, relating to Virginia housing opportunity tax credit.

(Prefiled January 11, 2021)

Patron--Locke

Referred to Committee on Finance and Appropriations

S.B. 1198. A BILL to amend and reenact §§ 2.2-3800, 2.2-3801, 2.2-3802, and 52-48 of the Code of Virginia, relating to Government Data Collection and Dissemination Practices Act; license plate readers.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on General Laws and Technology

S.B. 1199. A BILL to amend the Code of Virginia by adding in Chapter 10.1 of Title 10.1 a section numbered 10.1-1016.1 and by adding in Chapter 17 of Title 10.1 a section numbered 10.1-1705.1, relating to conservation easements; construction.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1200. A BILL to amend and reenact §§ 10.1-1408.1 and 10.1-1426 of the Code of Virginia, relating to waste disposal; local approval.

(Prefiled January 11, 2021)

Patron--Hashmi

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1201. A BILL to amend and reenact §§ 58.1-2600 and 58.1-2628 of the Code of Virginia, relating to taxation of public service corporations; electric suppliers; storage.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Finance and Appropriations

S.B. 1202. A BILL to amend and reenact §§ 8.01-66.1 and 38.2-2206 of the Code of Virginia, relating to uninsured and underinsured motorist insurance policies; bad faith.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Commerce and Labor

S.B. 1203. A BILL to amend and reenact §§ 8.01-42.1, 8.01-49.1, 18.2-51, 18.2-57, 18.2-121, and 52-8.5 of the Code of Virginia, relating to hate crimes; associational relationships; penalty.

(Prefiled January 11, 2021)

Patron--Hashmi

Referred to Committee on the Judiciary

S.B. 1204. A BILL providing a management agreement between the Commonwealth and George Mason University pursuant to the Restructured Higher Education Financial and Administrative Operations Act (§ 23.1-1000 et seq.).

(Prefiled January 11, 2021)

Patron--Barker

Referred to Committee on Finance and Appropriations

S.B. 1205. A BILL to amend and reenact §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909 of the Code of Virginia, relating to programs to address career fatigue and wellness in certain health care providers; civil immunity.

EMERGENCY

(Prefiled January 11, 2021)

Patron--Barker

Referred to Committee on Education and Health

S.B. 1206. A BILL to amend and reenact § 16.1-305 of the Code of Virginia, relating to confidentiality of juvenile court records; exceptions.

(Prefiled January 11, 2021)

Patron--Barker

Referred to Committee on the Judiciary

S.B. 1207. A BILL to amend and reenact §§ 15.2-2316.6 through 15.2-2316.9 of the Code of Virginia, relating to solar projects and energy storage projects; siting agreements throughout the Commonwealth.

(Prefiled January 11, 2021)

Patron--Barker

Referred to Committee on Local Government

S.B. 1208. A BILL to amend and reenact § 15.2-1413 of the Code of Virginia, relating to continuity of government.

(Prefiled January 11, 2021)

Patron--Barker

Referred to Committee on Local Government

S.B. 1209. A BILL to amend and reenact § 11-4.6 of the Code of Virginia, relating to liability of contractor for wages of subcontractor's employees.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on the Judiciary

S.B. 1210. A BILL to amend and reenact §§ 10.1-1402 and 62.1-44.15:6 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 10.1-1402.1:2, relating to environmental permit fees.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1211. A BILL to amend and reenact §§ 46.2-686 and 46.2-694, as it is currently effective and as it may become effective, of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 46.2-694.2, relating to establishment of the Public Safety Trust Fund.

(Prefiled January 11, 2021)

Patron--Edwards

Referred to Committee on Finance and Appropriations

S.B. 1212. A BILL to amend the Code of Virginia by adding in Title 33.2 a chapter numbered 38, consisting of sections numbered 33.2-3800 through 33.2-3816, relating to creation of the New River Valley Passenger Rail Station Authority.

(Prefiled January 11, 2021)

Patron--Edwards

Referred to Committee on Transportation

S.B. 1213. A BILL to amend and reenact § 18.2-271.1 of the Code of Virginia, relating to driver's license suspensions; restricted licenses; drug offenses.

(Prefiled January 11, 2021)

Patron--Edwards

S.B. 1214. A BILL to repeal § 5.1-178 of the Code of Virginia, relating to the Metropolitan Washington Airports Authority; effective date.

(Prefiled January 11, 2021)

Patron--Edwards

Referred to Committee on Transportation

S.B. 1215. A BILL to amend the Code of Virginia by adding a section numbered 55.1-1243.1 and to repeal § 55.1-1243 of the Code of Virginia, relating to Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion from dwelling unit, interruption of services, or actions taken to make premises unsafe.

(Prefiled January 11, 2021)

Patron--Ebbin

Referred to Committee on General Laws and Technology

S.B. 1216. A BILL to amend and reenact §§ 3.1, as amended, 3.2, 3.7, as amended, and 5.1 of Chapter 669 of the Acts of Assembly of 1972, which provided a charter for the Town of Crewe in Nottoway County, and to repeal Chapter 6 (§§ 6.1 through 6.4) of Chapter 669 of the Acts of Assembly of 1972, relating to town council; elections and powers.

(Prefiled January 11, 2021)

Patron--Ruff

Referred to Committee on Local Government

S.B. 1217. A BILL to amend and reenact § 33.2-335 of the Code of Virginia, relating to taking certain private roads into the secondary state highway system.

(Prefiled January 11, 2021)

Patron--Ruff (By Request)

Referred to Committee on Transportation

S.B. 1218. A BILL to amend and reenact §§ 54.1-2900, 54.1-2901, 54.1-2914, 54.1-2973.1, and 54.1-3401 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 54.1-2956.15 through 54.1-2956.18, relating to licensure of naturopathic doctors.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Education and Health

S.B. 1219. A BILL to direct the Bureau of Insurance to review and make recommendations regarding paid family leave.

(Prefiled January 11, 2021)

Patron--Favola

Referred to Committee on Commerce and Labor

S.B. 1220. A BILL to repeal § 37.2-827 of the Code of Virginia, relating to state hospitals; admission of certain aliens.

(Prefiled January 11, 2021)

Patrons--Favola and Barker

Referred to Committee on Education and Health

S.B. 1221. A BILL to provide for the operation of the Loudoun County local health department.

(Prefiled January 11, 2021)

Patron--Favola

Referred to Committee on Education and Health

S.B. 1222. A BILL to amend the Code of Virginia by adding a section numbered 2.2-4328.1, relating to the Virginia Public Procurement Act; purchase of personal protective equipment.

(Prefiled January 11, 2021)

Patron--DeSteph

Referred to Committee on General Laws and Technology

S.B. 1223. A BILL to amend and reenact §§ 67-102, 67-201, and 67-202 of the Code of Virginia, relating to transportation electrification; Virginia Energy Plan.

(Prefiled January 11, 2021)

Patrons--Boysko and Howell

Referred to Committee on Commerce and Labor

S.B. 1224. A BILL to amend and reenact § 36-99 of the Code of Virginia, relating to the Uniform Statewide Building Code; amendments; energy efficiency and conservation.

(Prefiled January 11, 2021)

Patron--Boysko

Referred to Committee on General Laws and Technology

S.B. 1225. A BILL to amend the Code of Virginia by adding in Article 5 of Chapter 9 of Title 15.2 a section numbered 15.2-986 and by adding a section numbered 22.1-79.9, relating to promotion of broadband service for educational purposes.

(Prefiled January 11, 2021)

Patron--Boysko

Referred to Committee on Commerce and Labor

S.B. 1226. A BILL to amend and reenact §§ 15.2-1626 and 15.2-1636.8 of the Code of Virginia, relating to the Compensation Board determining staffing and salaries for an attorney for the Commonwealth. (Prefiled January 11, 2021)

Patron--Boysko

Referred to Committee on Local Government

S.B. 1227. A BILL to amend and reenact § 32.1-325 of the Code of Virginia, relating to state plan for medical assistance; payment of medical assistance; 12-month supply of hormonal contraceptives.

(Prefiled January 11, 2021)

Patron--Boysko

Referred to Committee on Education and Health

S.B. 1228. A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 40.1 an article numbered 1.2, consisting of sections numbered 40.1-28.13 through 40.1-28.17, and to repeal § 40.1-28.6 of the Code of Virginia, relating to the Virginia Equal Pay Act; civil penalties.

(Prefiled January 11, 2021)

Patron--Boysko

Referred to Committee on Commerce and Labor

S.B. 1229. A BILL to authorize the issuance of special license plates for supporters of Ducks Unlimited bearing the legend DUCKS UNLIMITED; fees.

(Prefiled January 11, 2021)

Patron--Stuart

Referred to Committee on Transportation

S.B. 1230. A BILL to amend and reenact § 16.1-241 of the Code of Virginia, relating to criminal cases; transfer to general district court.

(Prefiled January 11, 2021)

Patron--Chase

Referred to Committee on the Judiciary

S.B. 1231. A BILL to amend the Code of Virginia by adding a section numbered 16.1-305.3, relating to filing an order of disposition from a criminal case in general district courts.

(Prefiled January 11, 2021)

Patron--Chase

Referred to Committee on the Judiciary

S.B. 1232. A BILL to amend and reenact § 53.1-40.01 of the Code of Virginia, relating to release of geriatric prisoners; exceptions.

(Prefiled January 11, 2021)

Patron--Chase

Referred to Committee on Rehabilitation and Social Services

S.B. 1233. A BILL to amend the Code of Virginia by adding in Chapter 9.3 of Title 24.2 an article numbered 3.1, consisting of sections numbered 24.2-948.5 through 24.2-948.8, and by adding in Article 8 of Chapter 9.3 of Title 24.2 a section numbered 24.2-953.6, relating to campaign finance; campaign contribution limits; civil penalty.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Privileges and Elections

S.B. 1234. A BILL to amend and reenact § 54.1-3931 of the Code of Virginia, relating to attorneys; granting certificates without examination.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on the Judiciary

S.B. 1235. A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 1 of Title 32.1 a section numbered 32.1-4.1, relating to Department of Health; certain communication prohibited. (Prefiled January 11, 2021)

Patron--Peake

Referred to Committee on Education and Health

S.B. 1236. A BILL to amend the Code of Virginia by adding a section numbered 24.2-947.4:2, relating to campaign finance; prohibited contributions to candidates.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Privileges and Elections

S.B. 1237. A BILL to amend and reenact § 32.-48.010 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 44-146.17:1.1, relating to certain emergency and quarantine orders; additional procedural requirements.

(Prefiled January 11, 2021)

Patron--Petersen

Referred to Committee on Education and Health

S.B. 1238. A BILL to amend and reenact § 22.1-280.2:1 of the Code of Virginia, relating to employment of school security officers; United States Armed Forces veterans; employees of local school boards. (Prefiled January 11, 2021)

Patron--Chase

Referred to Committee on Education and Health

S.B. 1239. A BILL to amend and reenact § 24.2-706 of the Code of Virginia, relating to absentee voting; third-party absentee ballot assembly and distribution.

(Prefiled January 11, 2021)

Patron--Bell

Referred to Committee on Privileges and Elections

S.B. 1240. A BILL to amend and reenact §§ 18.2-67.1, 18.2-67.2, and 18.2-366 of the Code of Virginia, relating to criminal sexual assault committed by parents, stepparents, grandparents, and step-grandparents; penalty.

(Prefiled January 11, 2021)

Patron--Stuart

Referred to Committee on the Judiciary

S.B. 1241. A BILL to amend and reenact § 8.01-417 of the Code of Virginia, relating to personal injury claim; disclosure of insurance policy limits.

(Prefiled January 11, 2021)

Patron--Stuart

Referred to Committee on the Judiciary

S.B. 1242. A BILL to amend and reenact § 19.2-3.1 of the Code of Virginia, relating to personal appearance by two-way electronic video and audio communication; entry of plea or nolle prosequi; adjudication of probation violations.

(Prefiled January 11, 2021)

Patron--Edwards

Referred to Committee on the Judiciary

S.B. 1243. A BILL to amend and reenact §§ 2.2-3705.3, 2.2-3711, 3.2-4113, 4.1-225, 15.2-1627, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-69.48:3, 16.1-228, as it is currently effective and as it shall become effective, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 16.1-309.1, 17.1-275, 17.1-275.8, 17.1-805, 18.2-46.1, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258through 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.09, 18.2-308.1:5, 18.2-308.4, 18.2-460, 18.2-474.1, 18.2-513, 19.2-11.2, 19.2-66, 19.2-81.1, 19.2-83.1, 19.2-120, 19.2-120.1, 19.2-188.1, 19.2-215.1, 19.2-291.1, 19.2-299, 19.2-299.2, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-386.28, 19.2-389, as it is currently effective and as it shall become effective, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.2, 22.1-277.08, 22.1-315, 24.2-233, 37.2-314, 37.2-416, 37.2-506, 48-17, 52-8.1:1, 52-35, 53.1-220.1, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, and 54.1-3442.8 of the Code of Virginia; to amend the Code of Virginia by adding in Title 3.2 a chapter numbered 41.2, containing articles numbered 1 through 7, consisting of sections numbered 3.2-4122 through 3.2-4199.6, and by adding a section numbered 19.2-392.2:1; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1 through 18.2-251.1:3, and 19.2-389.3 of the Code of Virginia, relating to the cultivation, manufacture, sale, possession, and testing of marijuana; penalties.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on Rehabilitation and Social Services

S.B. 1244. A BILL to amend and reenact §§ 2.2-3703, 2.2-3705.2, 2.2-3711, 9.1-101, 9.1-177.1, 9.1-907, 9.1-908, 10.1-104.7, 16.1-69.55, 16.1-300, 16.1-305, 17.1-213, 17.1-805, 19.2-11.01, 19.2-169.3, 19.2-174.1, 19.2-299, 19.2-301, 19.2-388, 19.2-389, as it is currently effective and as it shall become effective, 19.2-389.1, 19.2-392.02, as it is currently effective and as it shall become effective, 37.2-844, 37.2-845, 37.2-846, 37.2-1102, 44-146.18:4, 44-146.22, 53.1-136, 53.1-145, and 63.2-105 of the Code of Virginia and to repeal Chapter 9 (§§ 37.2-900 through 37.2-921) of Title 37.2 of the Code of Virginia, relating to civil commitment of sexually violent predators.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on the Judiciary

S.B. 1245. A BILL to amend and reenact §§ 24.2-706, 24.2-707, 24.2-709, 24.2-709.1, and 24.2-710 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 24.2-707.1, relating to absentee voting; establishment of drop-off locations; ballot defects; cure process.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on Privileges and Elections

S.B. 1246. A BILL to amend and reenact §§ 24.2-101, 24.2-709.1, and 24.2-712 of the Code of Virginia, relating to absentee voting; mandatory processing of returned absentee ballots before election day; central absentee voter precinct in the office of the general registrar.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on Privileges and Elections

S.B. 1247. A BILL to amend and reenact § 56-580 of the Code of Virginia, relating to electric utilities; closure of carbon-emitting generating units.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on Commerce and Labor

S.B. 1248. A BILL to amend and reenact § 16.1-356 of the Code of Virginia, relating to juveniles; competency evaluation; receipt of court order.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on the Judiciary

S.B. 1249. A BILL to amend and reenact § 15.2-2259 of the Code of Virginia, relating to local planning commissions; review deadlines.

(Prefiled January 11, 2021)

Patron--Stuart (By Request)

Referred to Committee on Local Government

S.B. 1250. A BILL to amend and reenact § 18.2-308.2:2, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to criminal history record information check required for firearm rentals; penalty.

(Prefiled January 11, 2021)

Patron--Deeds

S.B. 1251. A BILL to amend and reenact §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308 of the Code of Virginia, relating to Virginia Retirement System; technical amendments.

(Prefiled January 12, 2021)

Patron--Newman

Referred to Committee on Finance and Appropriations

S.B. 1252. A BILL to amend and reenact §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1 of the Code of Virginia, relating to sunset of coal tax credits.

(Prefiled January 12, 2021)

Patrons--McPike; Delegate: Plum

Referred to Committee on Finance and Appropriations

S.B. 1253. A BILL to amend and reenact § 33.2-1509 of the Code of Virginia, relating to funds for access roads to economic development sites; criteria for use of funds.

(Prefiled January 12, 2021)

Patrons--McPike; Delegate: Plum

Referred to Committee on Transportation

S.B. 1254. A BILL to amend and reenact §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100 of the Code of Virginia, relating to sports betting; technical amendments.

(Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on Finance and Appropriations

S.B. 1255. A BILL to amend and reenact § 38.2-200 of the Code of Virginia, relating to State Corporation Commission; issuance or renewal of insurance licenses or registrations during an emergency.

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Commerce and Labor

S.B. 1256. A BILL to amend and reenact §§ 9.1-102, 9.1-108, and 9.1-112, as they shall become effective, of the Code of Virginia, relating to membership on Criminal Justice Services Board and Committee on Training; law-enforcement training.

(Prefiled January 12, 2021)

Patron--Marsden

Referred to Committee on the Judiciary

S.B. 1257. A BILL to amend and reenact §§ 22.1-129, 22.1-199.1, as it is currently effective and as it shall become effective, 22.1-253.13:1, 22.1-253.13:2, 22.1-253.13:3, 22.1-253.13:5, 22.1-274, 22.1-274.01:1, 22.1-294, 22.1-299.4, as it is currently effective, 22.1-303, and 51.1-617, as it is currently effective and as it shall become effective, of the Code of Virginia and to repeal § 22.1-305.1 of the Code of Virginia, relating to the Standards of Quality; work-based learning; teacher leaders and mentors; principal mentors; certain personnel positions and initiatives.

(Prefiled January 12, 2021)

Patron--McClellan

Referred to Committee on Education and Health

S.B. 1258. A BILL to amend the Code of Virginia by adding a section numbered 62.1-44.15:55.1, relating to solar projects; erosion and sediment control.

(Prefiled January 11, 2021)

Patron--Marsden

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1259. A BILL to amend and reenact §§ 56-539 and 56-542 of the Code of Virginia, relating to Virginia Highway Corporation Act; alteration of certificate of authority; powers and duties of the State Corporation Commission.

(Prefiled January 11, 2021)

Patron--Bell

Referred to Committee on Transportation

S.B. 1260. A BILL to amend and reenact § 33.2-1011 of the Code of Virginia, relating to entry onto land for inspection for transportation purposes.

(Prefiled January 11, 2021)

Patron--Bell

Referred to Committee on Transportation

S.B. 1261. A BILL to amend and reenact §§ 2.2-511, 8.01-36, 8.01-267.8, 8.01-383.1, 8.01-555, 8.01-626, 8.01-670, 8.01-671, 8.01-675.3, 8.01-676.1, 9.1-909, 15.2-1627, 15.2-1643, 15.2-2139, 15.2-2140, 15.2-2656, 15.2-3104, 15.2-3217, 15.2-3221, 15.2-3222, 15.2-3227, 15.2-3244, 15.2-3308, 15.2-3528, 15.2-3605, 15.2-3809, 15.2-3909, 15.2-4108, 15.2-4120, 15.2-5218, 15.2-5367, 15.2-6606, 15.2-6632, 15.2-7406, 16.1-279.1, 17.1-309, 17.1-400 through 17.1-403, 17.1-405 through 17.1-408, 17.1-410, 17.1-413, 17.1-503, 17.1-513, 18.2-308.08, 18.2-384, 19.2-152.10, 19.2-321, 19.2-321.1, 19.2-321.2, 19.2-322.1, 19.2-386.13, 19.2-402, 19.2-403, 19.2-404, 22.1-97, 22.1-289.024, as it shall become effective, 24.2-237, 24.2-422, 24.2-433, 25.1-239, 32.1-48.010, 32.1-48.013, 33.2-928, 33.2-2917, 37.2-920, 45.1-161.322, 55.1-1833, 55.1-1966, 55.1-2211, 57-2.02, 58.1-527, 58.1-1828, 58.1-2282, 58.1-3147, 58.1-3992, and 63.2-1710 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 26.1 of Title 8.01 sections numbered 8.01-675.5 and 8.01-675.6; and to repeal §§ 8.01-670.1 and 8.01-672 of the Code of Virginia, relating to the Court of Appeals; jurisdiction; number of judges.

(Prefiled January 12, 2021)

Patron--Edwards

Referred to Committee on the Judiciary

S.B. 1262. A BILL to amend and reenact § 18.2-271.1 of the Code of Virginia, relating to restricted licenses; payment of fines and costs.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on the Judiciary

S.B. 1263. A BILL to amend and reenact §§ 46.2-839 and 46.2-905 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 46.2-903.1, relating to traffic regulation; bicycles.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on Transportation

S.B. 1264. A BILL to amend and reenact § 19.2-392.4 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 19.2-392.2:1, relating to expungement of emergency and preliminary protective orders.

(Prefiled January 11, 2021)

Patron--Morrissey

S.B. 1265. A BILL to amend and reenact §§ 62.1-44.15:37.1 and 62.1-44.15:58.1 of the Code of Virginia, relating to natural gas pipelines; stop work orders.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1266. A BILL to amend and reenact §§ 19.2-120 and 19.2-124 of the Code of Virginia and to repeal § 19.2-120.1 of the Code of Virginia, relating to admission to bail; rebuttable presumptions against bail.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on the Judiciary

S.B. 1267. A BILL to amend and reenact §§ 2.02, 5.021, and 6.02, § 12.01, as amended, and § 15.03 of Chapter 227, as amended, of the Acts of Assembly of 1954, which provided a charter for the City of Covington, and to amend and reenact § 22.1-32 of the Code of Virginia, relating to consolidated school board of Alleghany County and the City of Covington; school board salaries.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on Local Government

S.B. 1268. A BILL to amend and reenact §§ 32.1-309.1, 54.1-2800, 54.1-2807, and 54.1-2825 of the Code of Virginia; to amend the Code of Virginia by adding in Article 5 of Chapter 28 of Title 54.1 a section numbered 54.1-2825.1; and to repeal §§ 54.1-2807.01 and 54.1-2807.02 of the Code of Virginia, relating to disposition of the remains of a decedent; persons to make arrangements for funeral and disposition of remains.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on General Laws and Technology

S.B. 1269. A BILL to amend and reenact § 38.2-3407.15:2 of the Code of Virginia, relating to health insurance; authorization of drug prescribed for the treatment of a mental disorder.

(Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on Commerce and Labor

S.B. 1270. A BILL to amend and reenact § 25.1-306 of the Code of Virginia, relating to eminent domain; notice of intent to file certificate.

(Prefiled January 12, 2021)

Patron--Cosgrove

Referred to Committee on the Judiciary

S.B. 1271. A BILL to amend and reenact § 2.2-3708.2 of the Code of Virginia, relating to the Virginia Freedom of Information Act; meetings held by electronic communication means during a state of emergency.

(Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on General Laws and Technology

S.B. 1272. A BILL to amend the Code of Virginia by adding a section numbered 19.2-169.3:1, relating to disposition of the unrestorably incompetent defendant; capital murder charge; inpatient custody of the Commissioner of the Department of Behavioral Health and Developmental Services.

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on the Judiciary

S.B. 1273. A BILL to amend the Code of Virginia by adding in Title 30 a chapter numbered 63, consisting of sections numbered 30-401 through 30-408, relating to the Behavioral Health Commission created.

(Prefiled January 12, 2021)

Patrons--Deeds and Hanger; Delegates: Bell, Price and Watts

Referred to Committee on Rules

S.B. 1274. A BILL to amend and reenact §§ 10.1-207, 10.1-1105, 29.1-579, and 33.2-353 of the Code of Virginia, relating to government planning; wildlife corridors.

(Prefiled January 12, 2021)

Patron--Marsden

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1275. A BILL to amend and reenact § 65.2-402 of the Code of Virginia, relating to workers' compensation; presumption of compensation for certain diseases; applicable to salaried and volunteer emergency medical services personnel.

(Prefiled January 12, 2021)

Patron--Marsden

Referred to Committee on Commerce and Labor

S.B. 1276. A BILL to amend and reenact § 38.2-3451 of the Code of Virginia, relating to health insurance; essential health benefits; abortion coverage.

(Prefiled January 12, 2021)

Patron--McClellan

Referred to Committee on Education and Health

S.B. 1277. A BILL to repeal the second enactment of Chapter 228 of the Acts of Assembly of 2015, relating to repeal of reporting requirement; Department of Motor Vehicles and Supreme Court of Virginia.

(Prefiled January 12, 2021)

Patron--Newman

Referred to Committee on Transportation

S.B. 1278. A BILL to amend and reenact §§ 18.2-334.2, 18.2-340.16, 18.2-340.18, 18.2-340.19, 18.2-340.22, 18.2-340.28:2, and 18.2-340.34:2 of the Code of Virginia, relating to charitable gaming; regulations; Texas Hold'em poker games.

(Prefiled January 11, 2021)

Patron--Bell

Referred to Committee on General Laws and Technology

S.B. 1279. A BILL to amend and reenact § 2.2-2001.2 of the Code of Virginia, relating to Department of Veterans Services; initiatives to reduce unemployment among veterans; comprehensive transition program.

(Prefiled January 11, 2021)

Patron--Bell

Referred to Committee on General Laws and Technology

S.B. 1280. A BILL to amend the Code of Virginia by adding in Article 2 of Chapter 6 of Title 10.1 a section numbered 10.1-613.6, relating to dams; negotiated settlement agreements.

(Prefiled January 11, 2021)

Patron--Bell

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1281. A BILL to amend and reenact § 24.2-110 of the Code of Virginia, relating to elections; qualifications of the general registrar, residency.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on Privileges and Elections

S.B. 1282. A BILL to amend the Code of Virginia by adding a section numbered 10.1-1307.04, relating to greenhouse gas emissions inventory.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1283. A BILL to amend and reenact §§ 9.1-101, as it is currently effective and as it shall become effective, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-389.3, and 19.2-390 of the Code of Virginia and to amend the Code of Virginia by adding in Title 19.2 a chapter numbered 23.2, consisting of sections numbered 19.2-392.5 through 19.2-392.14, relating to automatic expungement of criminal records.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on the Judiciary

S.B. 1284. A BILL to amend and reenact §§ 56-46.1, 56-585.1, 56-598, 56-601, 62.1-199, 67-103, 67-104, and 67-201 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 67-101.1; and to repeal §§ 67-101 and 67-102 of the Code of Virginia, relating to the Commonwealth Energy Policy.

(Prefiled January 11, 2021)

Patron--Favola

Referred to Committee on Commerce and Labor

S.B. 1285. A BILL to amend and reenact § 15.2-1127 of the Code of Virginia, relating to vacant buildings; registration.

(Prefiled January 11, 2021)

Patron--Locke

Referred to Committee on Local Government

S.B. 1286. A BILL to amend and reenact § 58.1-320 of the Code of Virginia and to amend the Code of Virginia by adding in Article 2 of Chapter 3 of Title 58.1 a section numbered 58.1-327, relating to income tax; rate increase; funding for schools and law-enforcement officer salaries.

(Prefiled January 11, 2021)

Patron--Deeds

Referred to Committee on Finance and Appropriations

S.B. 1287. A BILL to amend and reenact § 18.2-340.19 of the Code of Virginia, relating to the Charitable Gaming Board; regulations; electronic pull tabs.

(Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on General Laws and Technology

S.B. 1288. A BILL to amend and reenact §§ 22.1-214, 22.1-215, 22.1-253.13:4, and 22.1-298.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-214.4, relating to Department of Education; special education.

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on Education and Health

S.B. 1289. A BILL to amend and reenact § 38.2-3407.15 of the Code of Virginia, relating to health insurance; carrier business practices; provider contracts.

(Prefiled January 12, 2021)

Patron--Surovell

Referred to Committee on Commerce and Labor

S.B. 1290. A BILL to amend and reenact §§ 10.1-1018.1 and 10.1-1021 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 1 of Title 10.1 a section numbered 10.1-104.6:1, relating to ConserveVirginia program; established.

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1291. A BILL to amend and reenact §§ 62.1-44.15:22 and 62.1-262 of the Code of Virginia, relating to Virginia Water Protection Permit; efficient water use.

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1292. A BILL to amend and reenact § 56-585.1 of the Code of Virginia, relating to electric utilities; overearnings; customer bill credits.

(Prefiled January 12, 2021)

Patron--McClellan

Referred to Committee on Commerce and Labor

S.B. 1293. A BILL to amend and reenact § 53.1-136 of the Code of Virginia, relating to Parole Board; monthly reports.

(Prefiled January 12, 2021)

Patron--DeSteph

Referred to Committee on Rehabilitation and Social Services

S.B. 1294. A BILL to amend and reenact §§ 17.1-275 and 19.2-392.2 of the Code of Virginia, relating to expungement of police and court records; fees.

(Prefiled January 12, 2021)

Patron--DeSteph

Referred to Committee on the Judiciary

S.B. 1295. A BILL to amend and reenact §§ 56-585.1:11 and 56-585.5 of the Code of Virginia, relating to electric utilities; procurement.

(Prefiled January 12, 2021)

Patron--DeSteph

Referred to Committee on Commerce and Labor

S.B. 1296. A BILL to amend and reenact § 44-146.18 of the Code of Virginia, relating to the State Coordinator of Emergency Management; establishment of Emergency Management Equity Working Group.

(Prefiled January 11, 2021)

Patron--Spruill

Referred to Committee on General Laws and Technology

S.B. 1297. A BILL to amend and reenact §§ 63.2-1603 and 63.2-1609 of the Code of Virginia, relating to emergency order for adult protective services; acts of violence, force, or threat or financial exploitation; penalty.

(Prefiled January 11, 2021)

Patron--Obenshain

Referred to Committee on Rehabilitation and Social Services

S.B. 1298. A BILL to amend the Code of Virginia by adding in Chapter 24 of Title 15.2 an article numbered 3, consisting of sections numbered 15.2-2413.1 through 15.2-2413.11, relating to tourism improvement districts.

(Prefiled January 11, 2021)

Patrons--Bell and Lucas

Referred to Committee on Local Government

S.B. 1299. A BILL to amend and reenact §§ 4.1-119, as it is currently effective and as it shall become effective, 4.1-204, as it is currently effective and as it shall become effective, 4.1-206.1, as it shall become effective, 4.1-207, 4.1-210, 4.1-212.1, as it is currently effective and as it shall become effective, and 4.1-221 of the Code of Virginia, relating to alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption.

(Prefiled January 11, 2021)

Patron--Bell

Referred to Committee on Rehabilitation and Social Services

S.B. 1300. A BILL to direct the Board of Local and Regional Jails to review services provided to inmates during pregnancy, pregnancy termination, labor and delivery, and postpartum recovery; report.

(Prefiled January 11, 2021)

Patron--Favola

Referred to Committee on Rehabilitation and Social Services

S.B. 1301. A BILL to amend the Code of Virginia by adding sections numbered 53.1-39.2 and 66-20.1, relating to correctional facilities; use of isolated confinement.

(Prefiled January 11, 2021)

Patron--Morrissey

Referred to Committee on Rehabilitation and Social Services

S.B. 1302. A BILL to amend and reenact §§ 37.2-311.1, as it shall become effective, 56-484.12, 56-484.17, and 56-484.17:1 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 37.2-311.2, 37.2-311.3, and 37.2-311.4, relating to crisis call centers; Crisis Call Center Fund established.

(Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on Education and Health

S.B. 1303. A BILL to direct each local school division in the Commonwealth to make in-person learning available to all students.

EMERGENCY

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on Education and Health

S.B. 1304. A BILL to amend and reenact § 37.2-505 of the Code of Virginia, relating to community services boards; discharge planning.

(Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on Education and Health

S.B. 1305. A BILL to amend the Code of Virginia by adding a section numbered 2.2-4303.02, relating to the Virginia Public Procurement Act; public works contracts; subcontractor workforce requirements. (Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on General Laws and Technology

S.B. 1306. A BILL to amend and reenact § 18.2-57 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 19.2-9.2, relating to assault and battery; penalty.

(Prefiled January 12, 2021)

Patron--Morrissey

Referred to Committee on the Judiciary

S.B. 1307. A BILL to amend and reenact § 32.1-325 of the Code of Virginia, relating to Department of Medical Assistance Services; school-based health services; telemedicine.

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on Education and Health

S.B. 1308. A BILL to amend and reenact § 46.2-923, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to pedestrians; interference with traffic; penalty.

(Prefiled January 12, 2021)

Patron--DeSteph

Referred to Committee on Transportation

S.B. 1309. A BILL to amend and reenact § 15.2-2114.01 of the Code of Virginia, relating to local stormwater assistance; flood mitigation and protection.

(Prefiled January 12, 2021)

Patron--Ebbin

Referred to Committee on Local Government

S.B. 1310. A BILL to amend and reenact §§ 2.2-3905, 40.1-2, 40.1-29, 40.1-49.3, 40.1-49.8, 65.2-101, and 65.2-305 of the Code of Virginia, relating to the employees providing domestic service; the Virginia Human Rights Act; application of laws applicable to employee safety and workers' compensation.

(Prefiled January 12, 2021)

Patrons--McClellan, Boysko and Lucas

Referred to Committee on Commerce and Labor

S.B. 1311. A BILL to amend and reenact § 62.1-44.15:81 of the Code of Virginia and to amend the Code of Virginia by adding in Article 2.6 of Chapter 3.1 of Title 62.1 a section numbered 62.1-44.15:85, relating to water quality standards; modification of permits and certifications.

(Prefiled January 12, 2021)

Patron--McClellan

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1312. A BILL to amend and reenact § 3.2-3011 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 3.2-3012.1 and 3.2-3012.2, relating to Virginia Spirits Promotion Fund; assessment on distillers; emergency.

EMERGENCY

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Rehabilitation and Social Services

S.B. 1313. A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; special education programs.

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Education and Health

S.B. 1314. A BILL to amend and reenact § 2.2-2238 of the Code of Virginia, relating to Virginia Economic Development Partnership Authority; Office of Education and Labor Market Alignment established; workforce and higher education alignment.

(Prefiled January 12, 2021)

Patron--Hashmi

Referred to Committee on General Laws and Technology

S.B. 1315. A BILL to amend and reenact §§ 19.2-120, 19.2-163.03, and 19.2-299 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 16 of Title 19.2 a section numbered 19.2-271.6, relating to criminal proceedings; consideration of mental condition and intellectual and developmental disabilities.

(Prefiled January 12, 2021)

Patrons--McClellan and Lucas

S.B. 1316. A BILL to amend and reenact §§ 19.2-389, as it is currently effective and as it shall become effective, 22.1-289.035, as it shall become effective, 22.1-289.039, as it shall become effective, 63.2-1720.1, and 63.2-1724 of the Code of Virginia, relating to child care providers; background check portability; subsidy pilot program; report.

(Prefiled January 12, 2021)

Patrons--McClellan and Boysko

Referred to Committee on the Judiciary

S.B. 1317. A BILL to amend and reenact § 22.1-7.1 of the Code of Virginia, relating to local school boards; open enrollment policy required.

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on Education and Health

S.B. 1318. A BILL to amend and reenact § 2.2-234 of the Code of Virginia and to amend the Code of Virginia by adding in Article 12 of Chapter 2 of Title 2.2 sections numbered 2.2-236 and 2.2-237, relating to environmental justice; interagency working group.

(Prefiled January 12, 2021)

Patron--Hashmi

Referred to Committee on General Laws and Technology

S.B. 1319. A BILL to study waste control and recycling; permits.

(Prefiled January 12, 2021)

Patron--Hashmi

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1320. A BILL to amend and reenact § 54.1-2900 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 54.1-2957.04, relating to licensed certified midwives; licensure; practice.

(Prefiled January 12, 2021)

Patron--Lucas

Referred to Committee on Education and Health

S.B. 1321. A BILL to amend and reenact § 63.2-1241 of the Code of Virginia, relating to confirmatory adoption.

(Prefiled January 12, 2021)

Patrons--Boysko and Barker

Referred to Committee on Rehabilitation and Social Services

S.B. 1322. A BILL to amend and reenact § 8.01-225 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-274.6, relating to public schools; seizure management and action plan; biennial training.

(Prefiled January 12, 2021)

Patron--DeSteph

Referred to Committee on Education and Health

S.B. 1323. A BILL to amend and reenact § 40.1-28.7:7 of the Code of Virginia, relating to worker classification; independent contractors.

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on Commerce and Labor

S.B. 1324. A BILL to amend the Code of Virginia by adding in Chapter 20.1 of Title 54.1 a section numbered 54.1-2019.1 and by adding in Chapter 20.2 of Title 54.1 a section numbered 54.1-2024, relating to actions against real estate appraisers or appraisal management companies; statute of limitations.

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on General Laws and Technology

S.B. 1325. A BILL to amend and reenact § 20-124.2 of the Code of Virginia, relating to visitation; petition of grandparent.

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on the Judiciary

S.B. 1326. A BILL to amend and reenact § 58.1-3830 of the Code of Virginia and to amend the Code of Virginia by adding in Article 7 of Chapter 38 of Title 58.1 a section numbered 58.1-3832.1, relating to local cigarette taxes; regional cigarette tax boards.

(Prefiled January 12, 2021)

Patron--Hanger

Referred to Committee on Finance and Appropriations

S.B. 1327. A BILL to amend and reenact §§ 8.01-462, 36-139, 55.1-320, 55.1-321, and 55.1-1303 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-2223.5, relating to housing protections; foreclosures; manufactured housing.

(Prefiled January 12, 2021)

Patron--McClellan

Referred to Committee on General Laws and Technology

S.B. 1328. A BILL to amend and reenact §§ 16.1-282.1, 63.2-100, as it is currently effective and as it shall become effective, 63.2-905, 63.2-906, and 63.2-1305 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 13 of Title 63.2 a section numbered 63.2-1306, relating to State-Funded Kinship Guardianship Assistance program.

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Rehabilitation and Social Services

S.B. 1329. A BILL to amend and reenact §§ 46.2-936 and 46.2-940 of the Code of Virginia, relating to promises to appear after the issuance of a summons.

(Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Transportation

S.B. 1330. A BILL to amend the Code of Virginia by adding in Title 60.2 a chapter numbered 8, consisting of sections numbered 60.2-800 through 60.2-820, relating to the establishment of family and medical leave insurance program; financing through payroll taxes.

(Prefiled January 12, 2021)

Patrons--Boysko and McClellan

Referred to Committee on Commerce and Labor

S.B. 1331. A BILL to amend and reenact § 24.2-704 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 24.2-103.2, relating to absentee voting; accessibility for voters with a visual impairment or print disability.

(Prefiled January 12, 2021)

Patron--Reeves

Referred to Committee on Privileges and Elections

S.B. 1332. A BILL to amend and reenact § 19.2-83.5, as it shall become effective, of the Code of Virginia, relating to use of deadly force by a law-enforcement officer during an arrest or detention. (Prefiled January 12, 2021)

Patron--Reeves

Referred to Committee on the Judiciary

S.B. 1333. A BILL to amend and reenact §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8 of the Code of Virginia, relating to pharmaceutical processors; cannabis products.

(Prefiled January 12, 2021)

Patron--Lucas

Referred to Committee on Education and Health

S.B. 1334. A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to pilot program for broadband capacity to unserved areas of the Commonwealth; municipal broadband authorities.

(Prefiled January 12, 2021)

Patrons--Edwards and Lewis

Referred to Committee on Commerce and Labor

S.B. 1335. A BILL to amend and reenact § 46.2-335, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to learner's permits; use of personal communication devices.

(Prefiled January 12, 2021)

Patron--Stuart

Referred to Committee on Transportation

S.B. 1336. A BILL to amend the Code of Virginia by adding a section numbered 18.2-271.5, relating to restricted permits to operate a motor vehicle; ignition interlock systems.

(Prefiled January 12, 2021)

Patron--Stuart

Referred to Committee on the Judiciary

S.B. 1337. A BILL to amend and reenact § 46.2-916.3 of the Code of Virginia, relating to golf carts and utility vehicles; County of Westmoreland.

(Prefiled January 12, 2021)

Patron--Stuart

Referred to Committee on Transportation

S.B. 1338. A BILL to amend and reenact §§ 32.1-325 and 38.2-3418.16 of the Code of Virginia, relating to telemedicine services; remote patient monitoring services.

(Prefiled January 12, 2021)

Patron--Barker

Referred to Committee on Education and Health

S.B. 1339. A BILL to amend and reenact §§ 9.1-128, 9.1-134, 17.1-502, and 19.2-392.1 through 19.2-392.4 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 1-229.1 and 8.01-40.5, by adding in Article 1 of Chapter 2 of Title 17.1 a section numbered 17.1-205.1, by adding sections numbered 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, and 19.2-392.4:1, and by adding in Chapter 23.1 of Title 19.2 sections numbered 19.2-392.5 and 19.2-392.6, relating to expungement and sealing of police and court records; Expungement Fee Fund created; protection of public record information; penalties.

(Prefiled January 12, 2021)

Patron--Surovell

Referred to Committee on the Judiciary

S.B. 1340. A BILL to amend and reenact § 8.01-246 of the Code of Virginia, relating to statute of limitations to collect medical debt.

(Prefiled January 12, 2021)

Patron--Hashmi

Referred to Committee on the Judiciary

S.B. 1341. A BILL to amend and reenact § 38.2-3521.1 of the Code of Virginia, relating to health insurance; association health plan for real estate salespersons.

(Prefiled January 12, 2021)

Patron--Barker

Referred to Committee on Commerce and Labor

S.B. 1342. A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability from COVID-19.

(Prefiled January 12, 2021)

Patron--Vogel

Referred to Committee on Commerce and Labor

S.B. 1343. A BILL to amend and reenact § 2.2-3705.6 of the Code of Virginia, relating to the Virginia Freedom of Information Act; record exclusion for proprietary records and trade secrets; carbon sequestration agreements.

(Prefiled January 12, 2021)

Patron--Vogel

Referred to Committee on General Laws and Technology

S.B. 1344. A BILL to amend and reenact § 18.2-251.03 of the Code of Virginia, relating to arrest and prosecution when experiencing or reporting overdoses.

(Prefiled January 12, 2021)

Patron--Vogel

Referred to Committee on the Judiciary

S.B. 1345. A BILL to amend and reenact § 4.1-230, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to alcoholic beverage control; license applications; notice requirements.

(Prefiled January 12, 2021)

Patron--Vogel

Referred to Committee on Rehabilitation and Social Services

S.B. 1346. A BILL to amend and reenact §§ 4.1-206.3, 4.1-209, as it is currently effective, 4.1-233, and 4.1-233.1 of the Code of Virginia, relating to alcoholic beverage control; mobile retailer license.

(Prefiled January 13, 2021)

Patron--Stuart

Referred to Committee on Rehabilitation and Social Services

S.B. 1347. A BILL to amend and reenact § 46.2-1012 of the Code of Virginia, relating to motorcycles; auxiliary lighting.

(Prefiled January 13, 2021)

Patron--Reeves

Referred to Committee on Transportation

S.B. 1348. A BILL to amend and reenact § 24.2-684 of the Code of Virginia, relating to elections; referenda; local advisory referenda.

(Prefiled January 13, 2021)

Patron--Newman

Referred to Committee on Privileges and Elections

S.B. 1349. A BILL to amend and reenact § 2.2-3802 of the Code of Virginia, relating to the Government Data Collection and Dissemination Practices Act; exemptions; email addresses of licensed professionals.

(Prefiled January 13, 2021)

Patron--Newman (By Request)

Referred to Committee on General Laws and Technology

S.B. 1350. A BILL to amend and reenact §§ 33.2-214.1, 33.2-370, 33.2-371, and 33.2-1503 of the Code of Virginia, relating to statewide prioritization process; projection selection.

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on Transportation

S.B. 1351. A BILL to amend the Code of Virginia by adding a section numbered 65.2-706.2, relating to workers' compensation; claims not barred.

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on Commerce and Labor

S.B. 1352. A BILL to amend and reenact §§ 2.2-215, 10.1-603.25, 10.1-659, and 10.1-1330 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 10.1-658.1 through 10.1-658.7, relating to flood control; Department of Flood Control and Commonwealth Flood Control Board established; report.

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1353. A BILL to amend and reenact §§ 58.1-400.1, 58.1-406, 58.1-442, and 58.1-443 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 3 of Title 58.1 an article numbered 11, consisting of sections numbered 58.1-424 through 58.1-427, relating to corporate income tax; combined reporting requirements.

(Prefiled January 13, 2021)

Patron--Marsden

Referred to Committee on Finance and Appropriations

S.B. 1354. A BILL to amend and reenact §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14 of the Code of Virginia, relating to Chesapeake Bay Phase III Watershed Improvement Plan; nutrient removal; regulations.

(Prefiled January 12, 2021)

Patron--Hanger

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1355. A BILL to allow the withdrawal of a member of the Rapidan Service Authority.

(Prefiled January 12, 2021)

Patron--Hanger

Referred to Committee on Local Government

S.B. 1356. A BILL to amend and reenact §§ 32.1-127, 32.1-162.5, and 63.2-1732 of the Code of Virginia, relating to hospitals, nursing homes, certified nursing facilities, hospices, and assisted living facilities; visits by clergy; public health emergency.

(Prefiled January 12, 2021)

Patron--Kiggans

Referred to Committee on Education and Health

S.B. 1357. A BILL to direct local school divisions to develop and administer diagnostic and growth assessments to each student to assess core subject competency.

(Prefiled January 12, 2021)

Patron--Dunnavant

Referred to Committee on Education and Health

S.B. 1358. A BILL to amend the Code of Virginia by adding a section numbered 40.1-44.2, relating to employment health and safety standards; heat illness prevention.

(Prefiled January 12, 2021)

Patron--Hashmi

Referred to Committee on Commerce and Labor

S.B. 1359. A BILL to amend and reenact § 23.1-1301 of the Code of Virginia, relating to public institutions of higher education; governing boards; fees; collegiate athletics.

(Prefiled January 12, 2021)

Patron--DeSteph

Referred to Committee on Education and Health

S.B. 1360. A BILL to amend and reenact §§ 2.2-522, 2.2-3902, and 2.2-3905 through 2.2-3908 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-3905.1, relating to the Virginia Human Rights Act; nondiscrimination in employment; sexual harassment and workplace harassment.

(Prefiled January 12, 2021)

Patrons--McClellan; Delegate: Watts

Referred to Committee on General Laws and Technology

S.B. 1361. A BILL to amend and reenact § 9.1-601, as it shall become effective, of the Code of Virginia, relating to law-enforcement civilian oversight bodies.

(Prefiled January 13, 2021)

Patron--Reeves

S.B. 1362. A BILL to require employers to report outbreaks of COVID-19; emergency.

EMERGENCY

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on Education and Health

S.B. 1363. A BILL to amend and reenact § 53.1-5 of the Code of Virginia and to amend the Code of Virginia by adding in Article 2 of Chapter 1 of Title 53.1 a section numbered 53.1-7.1, relating to Board of Local and Regional Jails; powers and duties.

(Prefiled January 13, 2021)

Patron--Marsden

Referred to Committee on Rehabilitation and Social Services

S.B. 1364. A BILL to amend and reenact § 33.2-1526.3 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 33.2-1526.3:1, relating to the creation of the Commonwealth Transit Ridership Fund.

(Prefiled January 13, 2021)

Patron--Cosgrove

Referred to Committee on Transportation

S.B. 1365. A BILL to amend and reenact § 2.2-203.2:4 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 25 of Title 2.2 an article numbered 13, consisting of sections numbered 2.2-2558 through 2.2-2564, relating to data governance; Office of Data Governance and Analytics; Chief Data Officer; Virginia Data Commission; report.

(Prefiled January 13, 2021)

Patron--Barker

Referred to Committee on General Laws and Technology

S.B. 1366. A BILL to amend and reenact §§ 51.5-134 and 51.5-135 of the Code of Virginia, relating to aging services; social need.

(Prefiled January 13, 2021)

Patron--Barker

Referred to Committee on Rehabilitation and Social Services

S.B. 1367. A BILL to amend and reenact §§ 9.1-404 and 9.1-405 of the Code of Virginia, relating to Line of Duty Act; requiring Virginia licensed health practitioners to conduct medical reviews.

(Prefiled January 13, 2021)

Patron--DeSteph

Referred to Committee on the Judiciary

S.B. 1368. A BILL to amend and reenact § 18.2-371.1 of the Code of Virginia, relating to abuse and neglect of children; contact with other parent; penalty.

(Prefiled January 13, 2021)

Patron--DeSteph

Referred to Committee on the Judiciary

S.B. 1369. A BILL to amend and reenact §§ 2.2-1604 and 2.2-4310 of the Code of Virginia, relating to the Department of Small Business and Supplier Diversity; definitions; small business.

(Prefiled January 13, 2021)

Patrons--Obenshain, Deeds and Hanger

Referred to Committee on General Laws and Technology

S.B. 1370. A BILL to amend and reenact § 53.1-165.1 of the Code of Virginia, relating to the application of parole statutes for juveniles and persons committed upon felony offenses committed on or after January 1, 1995.

(Prefiled January 13, 2021)

Patrons--Edwards and Morrissey

Referred to Committee on Rehabilitation and Social Services

S.B. 1371. A BILL to amend and reenact §§ 8.01-226.5:2, 16.1-228, 18.2-371, 18.2-371.1, 40.1-103, and 63.2-100, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to safe haven protections; newborn safety device.

(Prefiled January 13, 2021)

Patron--Ruff (By Request)

Referred to Committee on the Judiciary

S.B. 1372. A BILL to amend and reenact §§ 9.1-101, as it is currently effective and as it shall become effective, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-389.3, and 19.2-390 of the Code of Virginia and to amend the Code of Virginia by adding in Title 19.2 a chapter numbered 23.2, consisting of sections numbered 19.2-392.5 through 19.2-392.14, relating to automatic expungement of criminal records.

(Prefiled January 13, 2021)

Patron--Lucas

Referred to Committee on the Judiciary

S.B. 1373. A BILL to amend and reenact § 10.1-1408.1 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 10.1-1321.2 and 10.1-1437.1, by adding in Article 1 of Chapter 3.1 of Title 62.1 a section numbered 62.1-44.6:1, and by adding a section numbered 62.1-266.1, relating to environmental permits; community and environmental justice outreach.

(Prefiled January 13, 2021)

Patron--McClellan

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1374. A BILL to establish the Carbon Sequestration Task Force; report.

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1375. A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability from COVID-19.

(Prefiled January 13, 2021)

Patron--Saslaw

Referred to Committee on Commerce and Labor

S.B. 1376. A BILL to amend and reenact §§ 24.2-101, 24.2-603, 24.2-709.1, 24.2-710, 24.2-711, and 24.2-712 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 24.2-667.1, relating to absentee voting; processing of absentee ballots before election day; reporting of absentee ballot vote totals; establishment of central absentee voter precincts.

(Prefiled January 13, 2021)

Patron--Saslaw

Referred to Committee on Privileges and Elections

S.B. 1377. A BILL to amend and reenact § 24.2-947.11 of the Code of Virginia, relating to campaign finance; special report for large pre-legislative session contributions; contributions in aggregate.

(Prefiled January 13, 2021)

Patron--Suetterlein

Referred to Committee on Privileges and Elections

S.B. 1378. A BILL to amend and reenact § 44-146.17, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to Emergency Services and Disaster Law; limitation on duration of executive orders.

(Prefiled January 13, 2021)

Patron--Newman

Referred to Committee on General Laws and Technology

S.B. 1379. A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-575, relating to Humane Cosmetics Act; civil penalties.

(Prefiled January 12, 2021)

Patron--Boysko

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1380. A BILL to amend and reenact § 58.1-3660 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 56-585.1:13, relating to electric utilities; electric school bus projects; report.

(Prefiled January 12, 2021)

Patron--Lucas

Referred to Committee on Commerce and Labor

S.B. 1381. A BILL to amend the Code of Virginia by adding a section numbered 18.2-283.2, relating to possessing or transporting a weapon within Capitol Square or into building owned or leased by the Commonwealth; penalty.

(Prefiled January 12, 2021)

Patron--Ebbin

Referred to Committee on the Judiciary

S.B. 1382. A BILL to amend and reenact §§ 18.2-308.09, 18.2-308.2:1, as it is currently effective and as it shall become effective, 18.2-308.2:2, as it is currently effective and as it shall become effective, 18.2-308.2:3, as it is currently effective and as it shall become effective, and 19.2-386.28 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-308.1:8, relating to purchase, possession, or transportation of firearms following conviction for assault and battery of a family or household member; penalties.

(Prefiled January 12, 2021)

Patron--Favola

Referred to Committee on the Judiciary

S.B. 1383. A BILL to amend and reenact §§ 19.2-120 and 19.2-299 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 16 of Title 19.2 a section numbered 19.2-271.6, relating to criminal proceedings; consideration of mental condition and intellectual and developmental disabilities.

(Prefiled January 12, 2021)

Patron--Favola

S.B. 1384. A BILL to amend the Code of Virginia by adding in Chapter 43 of Title 2.2 an article numbered 7, consisting of sections numbered 2.2-4377.1 through 2.2-4377.5, relating to the Virginia Public Procurement Act; local arbitration agreements.

(Prefiled January 12, 2021)

Patron--Surovell

Referred to Committee on General Laws and Technology

S.B. 1385. A BILL to amend and reenact § 15.2-816.1 of the Code of Virginia, relating to underground utility lines; Fairfax County.

(Prefiled January 12, 2021)

Patron--Surovell

Referred to Committee on Local Government

S.B. 1386. A BILL to amend the Code of Virginia by adding in Article 1.1 of Chapter 8 of Title 46.2 a section numbered 46.2-819.11, relating to enforcement of private tolls; vehicle registration.

(Prefiled January 12, 2021)

Patron--Spruill

Referred to Committee on Transportation

S.B. 1387. A BILL to amend and reenact § 23.1-506 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 23.1-505.1, relating to the State Council of Higher Education for Virginia; eligibility for in-state tuition.

(Prefiled January 12, 2021)

Patrons--Boysko and Hashmi

Referred to Committee on Education and Health

S.B. 1388. A BILL to amend and reenact §§ 4.1-119, as it is currently effective and as it shall become effective, 4.1-204, as it is currently effective and as it shall become effective, 4.1-206.1, as it shall become effective, 4.1-207, 4.1-210, 4.1-212.1, as it is currently effective and as it shall become effective, and 4.1-221 of the Code of Virginia, relating to alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption.

(Prefiled January 13, 2021)

Patron--Reeves

Referred to Committee on Rehabilitation and Social Services

S.B. 1389. A BILL to amend and reenact § 55.1-703 of the Code of Virginia and to amend the Code of Virginia by adding in Article 6 of Chapter 6 of Title 10.1 a section numbered 10.1-660 and by adding a section numbered 55.1-708.2, relating to property; required disclosures for buyer to exercise due diligence; flood risk report.

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on General Laws and Technology

S.B. 1390. A BILL to amend and reenact §§ 3.2-6500, 3.2-6524, 3.2-6546, and 3.2-6551 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 3.2-6579.1, relating to cats; trap, neuter, and return programs; civil penalty.

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1391. A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 9 of Title 19.2 a section numbered 19.2-134.1, relating to pretrial data collection.

(Prefiled January 13, 2021)

Patron--Lucas

Referred to Committee on the Judiciary

S.B. 1392. A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-581, relating to Consumer Data Protection Act.

(Prefiled January 13, 2021)

Patron--Marsden

Referred to Committee on General Laws and Technology

S.B. 1393. A BILL to amend and reenact §§ 15.2-961 and 15.2-961.1 of the Code of Virginia, relating to replacement and conservation of trees during development.

(Prefiled January 13, 2021)

Patron--Marsden

Referred to Committee on Local Government

S.B. 1394. A BILL to amend and reenact § 58.1-321 of the Code of Virginia, relating to income tax exclusion for Paycheck Protection Plan loan forgiveness; deductibility.

(Prefiled January 13, 2021)

Patrons--Petersen and Pillion

Referred to Committee on Finance and Appropriations

S.B. 1395. A BILL to amend and reenact §§ 24.2-105, as it shall become effective, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 24.2-1005.2, by adding in Article 5 of Chapter 2 of Title 24.2 a section numbered 24.2-224.1, and by adding in Title 24.2 a chapter numbered 1.1, consisting of sections numbered 24.2-125 through 24.2-129; and to repeal § 24.2-124, as it shall become effective, of the Code of Virginia, relating to elections; prohibited discrimination in voting and elections administration; required process for enacting certain covered practices; civil causes of action.

(Prefiled January 13, 2021)

Patrons--McClellan; Delegate: Price

Referred to Committee on Privileges and Elections

S.B. 1396. A BILL to amend and reenact §§ 32.1-164 and 32.1-164.1:01 of the Code of Virginia and to amend the Code of Virginia by adding in Title 62.1 a chapter numbered 21.1, consisting of sections numbered 62.1-223.1, 62.1-223.2, and 62.1-223.3, relating to sewage; Onsite Sewage Indemnification Fund; Wastewater Infrastructure Policy Working Group; regulations; report.

(Prefiled January 13, 2021)

Patron--Hashmi

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1397. A BILL to amend and reenact § 53.1-136 of the Code of Virginia, relating to conditional release of geriatric or terminally ill inmates; notice to attorney for the Commonwealth.

(Prefiled January 13, 2021)

Patron--Norment

Referred to Committee on Rehabilitation and Social Services

S.B. 1398. A BILL to amend and reenact §§ 58.1-602, 58.1-603, as it is currently effective and as it may become effective, 58.1-3819, as it shall become effective, 58.1-3823, as it shall become effective, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, as it shall become effective, 58.1-3826, 58.1-3842, and 58.1-3843 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 2.2-2320.2 and 58.1-612.2 and by adding in Article 6 of Chapter 38 of Title 58.1 a section numbered 58.1-3818.8, relating to retail sales and transient occupancy taxes on room rentals.

(Prefiled January 13, 2021)

Patron--Norment

Referred to Committee on Finance and Appropriations

S.B. 1399. A BILL to amend and reenact §§ 15.2-5500, 15.2-5505, 15.2-5506, and 45.1-246 of the Code of Virginia, relating to the Tourism Development Authority; name change.

(Prefiled January 13, 2021)

Patron--Pillion

Referred to Committee on Local Government

S.B. 1400. A BILL to authorize a quitclaim and release of interest and the conveyance of an easement by the Board of Wildlife Resources in Tazewell County.

(Prefiled January 13, 2021)

Patron--Pillion

Referred to Committee on the Judiciary

S.B. 1401. A BILL to amend and reenact § 22.1-253.13:3 of the Code of Virginia, relating to Standards of Learning assessments; reduction; report.

(Prefiled January 13, 2021)

Patron--Pillion

Referred to Committee on Education and Health

S.B. 1402. A BILL to amend and reenact § 29.1-311 of the Code of Virginia, relating to trout fishing in stocked waters.

(Prefiled January 13, 2021)

Patron--Pillion

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1403. A BILL to amend the Code of Virginia by adding a section numbered 58.1-609.14, relating to sales tax; exemption for personal protective equipment; emergency.

EMERGENCY

(Prefiled January 13, 2021)

Patron--Pillion

Referred to Committee on Finance and Appropriations

S.B. 1404. A BILL to amend and reenact § 62.1-44.15:29.1 of the Code of Virginia, relating to Stormwater Local Assistance Fund; grant requirements.

(Prefiled January 13, 2021)

Patron--Lewis

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1405. A BILL to amend the Code of Virginia by adding a section numbered 23.1-2911.2, relating to the establishment of the Get Skilled, Get a Job, Give Back (G3) Fund and Program. (Prefiled January 13, 2021)

Patron--Saslaw

Referred to Committee on Education and Health

S.B. 1406. A BILL to amend and reenact §§ 1-404, 2.2-221, 2.2-507, 2.2-509.1, 2.2-511, 2.2-1119, 2.2-2696, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 2.2-4345, 3.2-1010, 3.2-3008, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101, 4.1-116, 4.1-121, 4.1-124, as it is currently effective and as it shall become effective, 4.1-225, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-102, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2288.3, as it is currently effective and as it shall become effective, 15.2-2288.3:1, as it is currently effective and as it shall become effective, 15.2-2288.3:2, as it is currently effective and as it shall become effective, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-246.6, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, $18.2 - 258.1, \quad 18.2 - 265.1, \quad 18.2 - 265.2, \quad 18.2 - 265.3, \quad 18.2 - 287.2, \quad 18.2 - 308.03, \quad 18.2 - 308.09, \quad 18.2 - 30$ 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.21 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 32.1-357, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 58.1-623.2, 58.1-1007, 58.1-1017.4, 58.1-3651, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29 consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6 consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a chapter numbered 6, containing articles numbered 1 through 10, consisting of sections numbered 4.1-600 through 4.1-696, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding a section numbered 19.2-392.2:1, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

(Prefiled January 13, 2021)

Patrons--Ebbin and Lucas

Referred to Committee on Rehabilitation and Social Services

S.B. 1407. A BILL to amend and reenact §§ 2.2-419, 2.2-2905, 2.2-3114, 2.2-3202, 2.2-3705.3, 2.2-3705.7, 2.2-3711, 2.2-4002, 8.01-195.11, 9.1-101, 9.1-801, 18.2-308.016, 18.2-325, 18.2-334.3, 18.2-340.22, 19.2-389, as it is currently effective and as it shall become effective, 37.2-314.2, 58.1-3, 58.1-302, 58.1-32.02, 58.1-460, 58.1-4000, 58.1-4002, 58.1-4003, 58.1-4006, 58.1-4007, 58.1-4008, 58.1-4009, 58.1-4011, 58.1-4012, 58.1-4020.1, 58.1-4025, 58.1-4027, 58.1-4100, and 59.1-148.3 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 3 of Title 11 a section numbered 11-16.3 and by adding in Chapter 40 of Title 58.1 an article numbered 3, consisting of sections numbered 58.1-4048 through 58.1-4075, relating to regulating electronic gaming devices; penalties.

(Prefiled January 13, 2021)

Patron--Bell

Referred to Committee on General Laws and Technology

S.B. 1408. A BILL to repeal § 30-170 of the Code of Virginia, relating to the Joint Commission on Health Care; sunset.

(Prefiled January 13, 2021)

Patrons--Barker; Delegates: Hodges, Hope and Sickles

Referred to Committee on Rules

S.B. 1409. A BILL to amend and reenact § 51.1-155 of the Code of Virginia, relating to Virginia Retirement System; retired law-enforcement officers employed as fingerprint examiners.

(Prefiled January 13, 2021)

Patron--Cosgrove

Referred to Committee on Finance and Appropriations

S.B. 1410. A BILL to amend and reenact §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310 of the Code of Virginia, relating to public accommodations, employment, and housing; prohibited discrimination on the basis of status as active military or a military spouse.

(Prefiled January 13, 2021)

Patrons--Bell; Delegates: Ayala and Reid

Referred to Committee on General Laws and Technology

S.B. 1411. A BILL to amend and reenact § 3.2-1905 of the Code of Virginia, relating to the excise tax on peanuts.

(Prefiled January 13, 2021)

Patron--Lucas

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1412. A BILL to amend and reenact §§ 3.2-6511.1 and 3.2-6511.2 of the Code of Virginia, relating to pet shops, dealers, and dog breeders; employees convicted of animal abuse; penalty.

(Prefiled January 13, 2021)

Patron--Stanley

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1413. A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to provision of broadband services by investor-owned electric utilities.

(Prefiled January 13, 2021)

Patron--Boysko

Referred to Committee on Commerce and Labor

S.B. 1414. A BILL to amend and reenact § 2.2-2543 of the Code of Virginia, relating to Henrietta Lacks Commission; sunset.

(Prefiled January 13, 2021)

Patron--Stanley

Referred to Committee on Rules

S.B. 1415. A BILL to amend and reenact §§ 16.1-253 and 16.1-253.2 of the Code of Virginia, relating to violations of protective orders; preliminary child protective order.

(Prefiled January 13, 2021)

Patron--Stanley

S.B. 1416. A BILL to amend and reenact § 32.1-325 of the Code of Virginia, relating to Department of Medical Assistance Services; remote patient monitoring; rural and underserved populations.

(Prefiled January 13, 2021)

Patron--Stanley

Referred to Committee on Education and Health

S.B. 1417. A BILL to amend and reenact § 3.2-6591 of the Code of Virginia and to amend the Code of Virginia by adding in Article 13 of Chapter 65 of Title 3.2 a section numbered 3.2-6593.1, relating to animal testing facilities; adoption of dogs and cats; civil penalty.

(Prefiled January 13, 2021)

Patron--Stanley

Referred to Committee on Agriculture, Conservation and Natural Resources

S.J.R. 270. Proposing the repeal of Section 15-A of Article I of the Constitution of Virginia, relating to marriage.

(Prefiled July 20, 2020)

Patron--Ebbin

Referred to Committee on Privileges and Elections

S.J.R. 271. Proposing an amendment to Section 1 of Article V of the Constitution of Virginia, relating to executive power; Governor's term of office.

(Prefiled July 20, 2020)

Patron--Ebbin

Referred to Committee on Privileges and Elections

S.J.R. 272. Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters and the right to vote.

(Prefiled December 1, 2020)

Patron--Locke

Referred to Committee on Privileges and Elections

S.J.R. 274. Proposing an amendment to Section 9 of Article IV of the Constitution of Virginia, relating to immunity of legislators.

(Prefiled December 8, 2020)

Patron--Reeves

Referred to Committee on Privileges and Elections

S.J.R. 275. Proposing amendments to Section 1 and Section 2 of Article VIII of the Constitution of Virginia, relating to public schools of the Commonwealth; equal educational opportunities.

(Prefiled December 9, 2020)

Patron--Stanley

Referred to Committee on Privileges and Elections

S.J.R. 276. Designating September, in 2021 and in each succeeding year, as Brain Aneurysm Awareness Month in Virginia.

(Prefiled December 29, 2020)

Patron--Favola

Referred to Committee on Rules

S.J.R. 285. Continuing the Joint Subcommittee to Study Barrier Crimes and Criminal History Records Checks. Report.

(Prefiled January 11, 2021)

Patron--Edwards

Referred to Committee on Rules

S.J.R. 286. Designating the third week in March, in 2021 and in each succeeding year, as Emergency Management Professionals Week in Virginia.

(Prefiled January 12, 2021)

Patron--McPike

Referred to Committee on Rules

S.J.R. 288. Memorializing the Joint Committee of Congress on the Library to approve the request of the Commonwealth to fill its vacant spot in the National Statuary Hall Collection at the United States Capitol with a statue to commemorate Barbara Rose Johns.

(Prefiled January 12, 2021)

Patron--Lucas

Referred to Committee on Rules

S.J.R. 289. Proposing an amendment to Section 9 of Article VII of the Constitution of Virginia, relating to local government; sale of property; authority to grant perpetual easements to units of government. (Prefiled January 12, 2021)

Patron--Mason

Referred to Committee on Privileges and Elections

S.J.R. 290. Proposing an amendment to Section 5 of Article IV of the Constitution of Virginia and proposing an amendment to the Constitution of Virginia by adding in Article IV a section numbered 5-A, relating to legislative compensation; Legislative Compensation Commission.

(Prefiled January 12, 2021)

Patrons--McPike and Hashmi; Delegate: Plum

Referred to Committee on Privileges and Elections

S.J.R. 291. Supporting the establishment of a commemorative commission to honor Booker T. Washington with a statue in the State Capitol.

(Prefiled January 12, 2021)

Patron--Suetterlein

Referred to Committee on Rules

S.J.R. 292. Designating August, in 2021 and in each succeeding year, as Women's Suffrage Month in Virginia.

(Prefiled January 12, 2021)

Patrons--Vogel; Delegate: Cole, M.L.

Referred to Committee on Rules

S.J.R. 293. Directing the Joint Commission on Health Care to study available data regarding assisted living and auxiliary grants and develop a blueprint for implementing recommendations that will allow the Commonwealth to provide a more realistic system of addressing housing and care needs. Report.

(Prefiled January 13, 2021)

Patron--Spruill

Referred to Committee on Rules

S.R. 90. 2021 Senate operating resolution.

(Prefiled January 12, 2021)

Patron--Locke

Referred to Committee on Rules

The following, by leave, were prefiled, presented, and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 273. Celebrating the life of Bruce Winston Edwards.

(Prefiled December 7, 2020)

Patron--DeSteph

S.J.R. 277. Commending Marymount University.

(Prefiled December 30, 2020)

Patrons--Favola, Ebbin and Howell; Delegates: Hope, Levine, Lopez and Sullivan

S.J.R. 278. Commending the Honorable Bruce D. White.

(Prefiled January 8, 2021)

Patron--Surovell

S.J.R. 279. Commending the Honorable Steven Selwyn Smith.

(Prefiled January 8, 2021)

Patron--Surovell

S.J.R. 280. Commending the Honorable Kimberly J. Daniel.

(Prefiled January 8, 2021)

Patron--Surovell

S.J.R. 281. Commending the Honorable Janine M. Saxe.

(Prefiled January 8, 2021)

Patron--Surovell

S.J.R. 282. Commending the Crater Planning District Commission.

(Prefiled January 8, 2021)

Patron--Ruff

S.J.R. 283. Celebrating the life of Marisa L. Fleck.

(Prefiled January 11, 2021)

Patron--Surovell

$\textbf{S.J.R. 284.} \ \ \textbf{Celebrating the life of the Honorable Augustus Benton Chafin, Jr.}$

(Prefiled January 11, 2021)

Patrons--Pillion, Norment and Stuart; Delegates: Kilgore and Morefield

S.J.R. 287. Celebrating the life of the Honorable Bernard S. Cohen.

(Prefiled January 12, 2021)

Patron--Saslaw

S.R. 84. Celebrating the life of William C. Washington.

(Prefiled December 22, 2020)

Patron--Lucas

S.R. 85. Celebrating the life of Jeffrey Peter Powell, M.D.

(Prefiled January 3, 2021)

Patron--Cosgrove

S.R. 86. Celebrating the life of Dora McGlone Harris.

(Prefiled January 4, 2021)

Patron--Lucas

S.R. 87. Celebrating the life of Ruth Ann Whitby Greene.

(Prefiled January 10, 2021)

Patron--Hashmi

S.R. 88. Celebrating the life of Robert Gene LeRosen.

(Prefiled January 10, 2021)

Patron--Hashmi

S.R. 89. Commending Tressie McMillan Cottom.

(Prefiled January 11, 2021)

Patron--Hashmi

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1418. A BILL to amend and reenact § 2.2-621 of the Code of Virginia, relating to implementation of federal mandates; grants by the Commonwealth; certification of employment; waiver or reduction of capital investment and local match requirements.

Patron--McPike

Referred to Committee on General Laws and Technology

S.B. 1419. A BILL to amend and reenact § 2.2-4321.2, as it shall become effective, of the Code of Virginia, relating to project labor agreements; public interest.

Patron--Marsden

Referred to Committee on General Laws and Technology

S.B. 1420. A BILL to amend and reenact § 1 of the first enactment of Chapters 358 and 382 of the Acts of Assembly of 2013, as amended by Chapter 803 of the Acts of Assembly of 2017, and as amended by Chapters 1187, 1188, 1189, 1193, 1194, and 1239 of the Acts of Assembly of 2020, relating to electric utilities; nonjurisdictional customers; third party power purchase agreements.

Patron--Edwards

Referred to Committee on Commerce and Labor

S.B. 1421. A BILL to amend and reenact § 37.2-403 of the Code of Virginia, relating to brain injury; definition.

Patron--Edwards

Referred to Committee on Education and Health

S.B. 1422. A BILL to amend and reenact §§ 24.2-408 and 24.2-427 of the Code of Virginia, relating to voter registration; list maintenance; lists of decedents transmitted by State Registrar of Vital Records.

Patron--Kiggans

Referred to Committee on Privileges and Elections

S.B. 1423. A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

Patron--McPike

Referred to Committee on Finance and Appropriations

S.B. 1424. A BILL to amend and reenact §§ 54.1-2800 and 54.1-2810 of the Code of Virginia, relating to funeral service establishments; manager of record.

Patron--Cosgrove

Referred to Committee on General Laws and Technology

S.J.R. 294. Directing the Joint Legislative Audit and Review Commission to study the true cost of education in the Commonwealth and provide an accurate assessment of the costs to implement the Standards of Quality. Report.

Patron--Lewis

Referred to Committee on Rules

S.R. 91. Expressing the censure of the Senate.

Patron--Bell

Referred to Committee on Privileges and Elections

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 295. Commending Spotsylvania County.

Patron--Reeves

S.J.R. 296. Commending Barry M. Barnard.

Patron--Reeves

RECESS

At 4:40 p.m., Senator Saslaw moved that the Senate recess until 6:45 p.m.

The motion was agreed to.

The hour of 6:45 p.m. having arrived, the Chair was resumed, the President pro tempore, Senator Lucas, presiding.

JOINT ASSEMBLY

The President pro tempore requested that the President of the Senate, the President pro tempore, and the Clerk of the Senate proceed to the Hall of the House of Delegates for a Joint Assembly to receive the Governor.

THE JOINT ASSEMBLY

The hour of 6:45 p.m. having arrived, being the time designated by the Speaker of the House of Delegates to receive the Governor of Virginia, Ralph S. Northam, the President of the Senate, Justin E. Fairfax, the President pro tempore of the Senate, L. Louise Lucas, and the Clerk of the Senate, Susan Clarke Schaar, entered the Hall of the House of Delegates and were received by the Speaker of the House of Delegates and the Clerk of the House of Delegates standing.

The Clerk of the House of Delegates announced that members of the Senate of Virginia were present in the virtual session.

The roll of the Senate was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel.

There were 36 Senators present.

The roll of the House of Delegates was called and the following Delegates answered to their names:

Adams, D.M., Adams, L.R., Aird, Askew, Ayala, Bagby, Bourne, Bulova, Campbell, R.R., Carr, Carter, Cole, J.G., Convirs-Fowler, Cox, Coyner, Davis, Delaney, Fariss, Gilbert, Gooditis, Guy, Guzman, Hayes, Helmer, Heretick, Herring, Hope, Hudson, Hurst, Jenkins, Jones, Keam, Kory, Krizek, Levine, Lopez, McNamara, McQuinn, Mugler, Mullin, Mundon King, Murphy, Plum, Poindexter, Price, Reid, Roem, Samirah, Scott, Sickles, Simon, Simonds, Sullivan, Torian, Tran, Tyler, Ward, Watts, Wiley, Willett, Williams Graves, Madam Speaker.

There were 62 Delegates present.

A majority of each house being present, the President of the Joint Assembly, Eileen Filler-Corn, declared the Joint Assembly duly organized and ready to proceed to business.

The President of the Joint Assembly presented the First Lady of the Commonwealth, Pam Northam, who was received by the members of the Joint Assembly.

The President of the Joint Assembly presented the following distinguished guests, in attendance virtually, who were received by the members of the Joint Assembly:

The Supreme Court of Virginia

The State Corporation Commission

The Governor's Cabinet and Executive Branch Officials

The President of the Joint Assembly presented the Governor, Ralph S. Northam, who was formally received by the Joint Assembly.

The Governor addressed the Joint Assembly.

On motion of the Senator from Portsmouth, Senator Lucas, one thousand copies of the Governor's address were ordered to be printed as Senate Document No. 1.

On motion of the Senator from Portsmouth, Senator Lucas, the Joint Assembly adjourned sine die, whereupon the President pro tempore of the Senate, the President of the Senate, and the Clerk of the Senate returned to the Senate Chamber.

I hereby certify that the foregoing is a true copy of the Journal of the House of Delegates in relation to the address of the Governor to the Joint Assembly on January 13, 2021.

/s/ Suzette Denslow Clerk of the House of Delegates and Clerk of the Joint Assembly

Upon the Senators' return to the Senate Chamber, the President resumed the Chair.

Jacke Ochson

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

THURSDAY, JANUARY 14, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Mamie E. Locke, Second Senatorial District, offered the following prayer:

I begin with a brief meditation. May God be above you for blessings, below you for support, before you for guidance, behind you for protection, beside you for comfort, and inside you for strength and determination.

Now let us pray together. Almighty God, thank You for this day. We ask now that You guide our hearts and minds as we govern our people. We ask that You grant us the wisdom to govern wisely, the empathy to govern compassionately, and the integrity to govern justly. Help us to be levelheaded and straightforward in purpose. Help us conduct the business of Your people with clarity and honor. Let us be slow to speech, deliberate in thought, and always swift to reconciliation. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Susan Clarke Schaar, Clerk of the Senate of Virginia.

On motion of Senator Norment, the Rules were suspended to allow Senator Lewis, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--37.

NAYS--0. RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell.

A quorum was present.

On motion of Senator Edwards, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--34. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell--34.

NAYS--Deeds, Petersen, Stanley--3. RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Lucas from the Committee on Education and Health:

- **S.B.** 1102 (one thousand one hundred two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B.** 1106 (one thousand one hundred six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B.** 1132 (one thousand one hundred thirty-two).
- **S.B. 1147** (one thousand one hundred forty-seven) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1175 (one thousand one hundred seventy-five).
 - S.B. 1190 (one thousand one hundred ninety).
- **S.B. 1191** (one thousand one hundred ninety-one) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1196** (one thousand one hundred ninety-six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bill, having been considered by the committee in session, was recommended for rereferral by the Committee on Education and Health:

S.B. 1276 (one thousand two hundred seventy-six) with the recommendation that it be rereferred to the Committee on Commerce and Labor.

The following bills, having been considered by the committee in session, were reported by Senator Barker from the Committee on General Laws and Technology:

- S.B. 1110 (one thousand one hundred ten) with amendment.
- **S.B.** 1150 (one thousand one hundred fifty) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1171 (one thousand one hundred seventy-one) with amendment.
- S.B. 1102, S.B. 1106, S.B. 1147, S.B. 1150, S.B. 1191, and S.B. 1196 were rereferred to the Committee on Finance and Appropriations.
 - S.B. 1276 was rereferred to the Committee on Commerce and Labor.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1425. A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

Patron--Ruff

Referred to Committee on Finance and Appropriations

S.B. 1426. A BILL to amend and reenact §§ 19.2-305.1, 19.2-305.2, 19.2-349, and 19.2-354 of the Code of Virginia, relating to orders of restitution; enforcement.

Patron--Stanley

Referred to Committee on the Judiciary

S.B. 1427. A BILL to amend the Code of Virginia by adding a section numbered 37.2-313.2, relating to Department of Behavioral Health and Developmental Services; Coordinated Specialty Care Program Advisory Board established.

Patron--Morrissey

Referred to Committee on Education and Health

S.B. 1428. A BILL to amend and reenact § 4.1-119, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to alcoholic beverage control; operation of government stores; sale of low alcohol beverage coolers.

Patron--Locke

Referred to Committee on Rehabilitation and Social Services

S.J.R. 297. Proposing an amendment to Section 6-A of Article X of the Constitution of Virginia, relating to authority of General Assembly to exempt property owned by certain veterans' and service members' services organizations from taxation.

Patron--Bell

Referred to Committee on Finance and Appropriations

S.J.R. 298. Confirming appointments by the Governor of certain persons communicated to the General Assembly June 1, 2020.

Patron--Deeds

Referred to Committee on Privileges and Elections

S.J.R. 299. Confirming appointments by the Governor of certain persons communicated to the General Assembly August 1, 2020.

Patron--Deeds

Referred to Committee on Privileges and Elections

S.J.R. 300. Confirming appointments by the Governor of certain persons communicated to the General Assembly October 1, 2020.

Patron--Deeds

Referred to Committee on Privileges and Elections

S.J.R. 301. Confirming appointments by the Governor of certain persons communicated to the General Assembly April 1, 2020.

Patron--Deeds

Referred to Committee on Privileges and Elections

S.J.R. 302. Confirming appointments by the Governor of certain persons communicated to the General Assembly December 1, 2020.

Patron--Deeds

Referred to Committee on Privileges and Elections

S.J.R. 303. Confirming appointments by the Governor of certain persons communicated to the General Assembly January 1, 2021.

Patron--Deeds

Referred to Committee on Privileges and Elections

CALENDAR

MEMORIAL RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions and resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions and resolutions were taken up, ordered to be engrossed, and agreed to en bloc by a unanimous standing vote:

S.J.R. 273 (two hundred seventy-three).

S.J.R. 283 (two hundred eighty-three).

S.R. 84 (eighty-four).

S.R. 86 (eighty-six).

S.R. 87 (eighty-seven).

S.R. 88 (eighty-eight).

S.J.R. 287 (two hundred eighty-seven) was taken up, as follows:

SENATE JOINT RESOLUTION NO. 287

Celebrating the life of the Honorable Bernard S. Cohen.

WHEREAS, the Honorable Bernard S. Cohen, a former member of the House of Delegates who had previously served as co-counsel for the appellants in the historic *Loving v. Virginia* case, in which the Supreme Court of the United States struck down laws banning interracial marriage, died on October 12, 2020; and

WHEREAS, the child of Jewish immigrants from Eastern Europe, Bernard "Bernie" Cohen grew up in Brooklyn and graduated from the City College of New York; he subsequently worked for the U.S. Department of Labor while studying to earn a law degree from Georgetown University; and

WHEREAS, Bernie Cohen was working as a volunteer attorney for the American Civil Liberties Union when he met Richard and Mildred Loving, who had been arrested under Virginia's anti-miscegenation laws and been given a suspended jail sentence on the condition that they leave the Commonwealth and not return for a period of 25 years; and

WHEREAS, during the *Loving v. Virginia* case, Bernie Cohen and co-counsel Philip Hirschkop argued that laws prohibiting interracial marriage violated the due process and equal protection clauses of the Fourteenth Amendment to the Constitution of the United States, and in 1967, the Supreme Court of the United States ruled unanimously in favor of the Lovings; and

WHEREAS, the landmark case was a significant victory for the civil rights movement in the United States, and Bernie Cohen's contributions have been immortalized in numerous books, documentaries, and dramatizations, including the 2016 major motion picture *Loving*; and

WHEREAS, desirous to be of further service to the Commonwealth, Bernie Cohen ran for and was elected to the House of Delegates in 1979; he represented residents of Alexandria and Chesapeake in the 21st District until 1983, when he took office as the delegate for the 46th District, representing parts of Alexandria; and

WHEREAS, during his tenure as a state lawmaker, Bernie Cohen offered his expertise to several standing committees and introduced many important pieces of legislation, including a bill to impose restrictions on indoor smoking and measures to ensure that terminally ill patients would have the right to decline life-prolonging treatments; and

WHEREAS, Bernie Cohen also introduced a bill to decriminalize same-sex marriage in the Commonwealth, which was unsuccessful, but the decision in *Loving v. Virginia* was ultimately cited as a precedent in the Supreme Court's decision in another landmark case, *Obergefell v. Hodges* (2015), which guaranteed same-sex couples the right to marry throughout the nation; and

WHEREAS, Bernie Cohen continued to practice law in Alexandria for many years, specializing in environmental and employment cases until his well-earned retirement in 2006; and

WHEREAS, a man of great integrity, Bernie Cohen served the Commonwealth with the utmost dedication and distinction as both an attorney and a state legislator; and

WHEREAS, Bernie Cohen will be fondly remembered and greatly missed by his wife of 61 years, Rae Rose; his children, Bennett and Karen, and their families; and numerous other family members and friends; now, therefore, be it

RESOLVED by the Senate, the House of Delegates concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Bernard S. Cohen, a highly admired attorney and a dedicated public servant who touched the lives of generations of Americans through his contributions to civil rights; and, be it

RESOLVED FURTHER, That the Clerk of the Senate prepare a copy of this resolution for presentation to the family of the Honorable Bernard S. Cohen as an expression of the General Assembly's respect for his memory.

S.J.R. 287, on motion of Senator Saslaw, was ordered to be engrossed and was agreed to by a unanimous standing vote.

S.R. 85 (eighty-five), on motion of Senator Cosgrove, was passed by for the day.

COMMENDING RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions and resolution that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions and resolution were taken up, ordered to be engrossed, and agreed to en bloc:

S.J.R. 277 (two hundred seventy-seven).

S.J.R. 278 (two hundred seventy-eight).

S.J.R. 279 (two hundred seventy-nine).

S.J.R. 280 (two hundred eighty).

S.J.R. 281 (two hundred eighty-one).

S.J.R. 282 (two hundred eighty-two).

S.R. 89 (eighty-nine).

On motion of Senator McDougle, a leave of absence for the day was granted Senator Vogel on account of pressing personal business.

On motion of Senator Lucas, the Senate adjourned until tomorrow at $12\ m.$

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

FRIDAY, JANUARY 15, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Siobhan S. Dunnavant, Twelfth Senatorial District, offered the following prayer:

Heavenly Father, we gather this morning to work for a better Commonwealth and happier lives for her citizens. Though we are inadequate to the task, through You we are strengthened. We are unified. We pray to have Your hand on our shoulders, to lead us with Your love, Your wisdom, and Your understanding. It may have best been said by Your servant, St. Francis, when he asked of You, "Lord, make me an instrument of Your peace. Where there is hatred, let me sow love. Where there is injury, pardon. Where there is doubt, faith. Where there is despair, hope. Where there is darkness, light. And where there is sadness, joy. O Divine Master, grant that I may not so much seek to be consoled as to console, to be understood as to understand, to be loved as to love. For it is in giving that we receive. It is in pardoning that we are pardoned. And it is in dying that we are born to eternal life." Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Susan Clarke Schaar, Clerk of the Senate of Virginia.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows: YEAS--34. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Stanley, Suetterlein, Surovell--34.

NAYS--0. RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Morrissey, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--34. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, Stanley--2. RULE 36--0.

COMMITTEE REPORTS

Senator Deeds, from the Committee on Privileges and Elections, presented the following report:

SENATE OF VIRGINIA

January 14, 2021

TO THE SENATE OF VIRGINIA:

The Committee on Privileges and Elections hereby certifies that the following person is qualified to be elected as the Auditor of Public Accounts as follows:

Staci A. Henshaw, of Dinwiddie, as the Auditor of Public Accounts for a term of four years commencing February 1, 2021.

Respectfully submitted,

/s/ R. Creigh Deeds, Chairman Committee on Privileges and Elections

The following bills, having been considered by the committee in session, were reported by Senator Favola from the Committee on Rehabilitation and Social Services:

- **S.B. 1104** (one thousand one hundred four) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1125** (one thousand one hundred twenty-five) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1299 (one thousand two hundred ninety-nine) with substitute.
- **S.B. 1312** (one thousand three hundred twelve) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1328** (one thousand three hundred twenty-eight) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- S.B. 1104, S.B. 1125, S.B. 1312, and S.B. 1328 were rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1429. A BILL to authorize the Commonwealth to lease a portion of property previously used by the Department of Behavioral Health and Developmental Services as the Southwestern Virginia Mental Health Institute and to amend and reenact §§ 1 and 2 of Chapter 678 of the Acts of Assembly of 2019.

Patron--Pillion

Referred to Committee on General Laws and Technology

S.B. 1430. A BILL to amend and reenact § 18.2-308.2:5 of the Code of Virginia, relating to criminal history record information check required to sell firearm; National Firearms Act.

Patron--DeSteph

Referred to Committee on the Judiciary

S.B. 1431. A BILL to amend and reenact § 19.2-169.1 of the Code of Virginia, relating to unrestorably incompetent defendant; competency report.

Patron--Mason

Referred to Committee on the Judiciary

S.B. 1432. A BILL to amend and reenact §§ 24.2-202, 24.2-203, 24.2-542, 24.2-542.1, and 24.2-673 of the Code of Virginia, relating to electors for President and Vice President; allocation of electoral votes.

Patron--Chase

Referred to Committee on Privileges and Elections

S.B. 1433. A BILL to amend the Code of Virginia by adding in Chapter 8 of Title 22.1 an article numbered 7, consisting of sections numbered 22.1-124.1 through 22.1-124.6, relating to public education; voucher program; emergency.

EMERGENCY

Patron--Chase

Referred to Committee on Education and Health

S.B. 1434. A BILL to amend and reenact § 22.1-212.1 of the Code of Virginia, relating to public schools; mandatory virtual learning; provision of required technology and Internet service; emergency.

EMERGENCY

Patron--Chase

Referred to Committee on Education and Health

S.B. 1435. A BILL to amend and reenact §§ 32.1-291.7, 54.1-2982, 54.1-2983, 54.1-2984, 59.1-481, 64.2-100, 64.2-403, 64.2-404, 64.2-407, 64.2-450, and 64.2-701 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 64.2-453.1 and by adding in Chapter 4 of Title 64.2 an article numbered 7, consisting of sections numbered 64.2-459 through 64.2-468, relating to electronic execution of estate planning documents; Uniform Electronic Wills Act.

Patron--Hanger

Referred to Committee on the Judiciary

S.B. 1436. A BILL to amend the Code of Virginia by adding in Article 2 of Chapter 1 of Title 23.1 a section numbered 23.1-110, by adding in Article 3 of Chapter 1 of Title 32.1 a section numbered 32.1-23.3, and by adding a section numbered 54.1-2400.03, relating to Department of Health; Eligible Health Care Provider Reserve Directory.

Patron--Hanger

Referred to Committee on Education and Health

S.B. 1437. A BILL to amend and reenact §§ 2.2-311, 19.2-74, 28.2-901, 29.1-210, 46.2-936, 46.2-940, 54.1-306, 54.1-2506, and 54.1-4407 of the Code of Virginia, relating to promises to appear after the issuance of a summons.

Patron--Hanger

Referred to Committee on the Judiciary

S.B. 1438. A BILL to amend and reenact § 58.1-3842 of the Code of Virginia, relating to combined transient occupancy and food and beverage tax; technical amendments.

Patron--Hanger

Referred to Committee on Finance and Appropriations

S.B. 1439. A BILL to amend and reenact § 22.1-254 of the Code of Virginia, relating to the Department of Education; guidelines on excused student absences; civic or political engagement.

Patron--McClellan

Referred to Committee on Education and Health

S.J.R. 308. Directing the Joint Legislative Audit and Review Commission to study the impact of COVID-19 on Virginia's public schools, students, and school employees. Report.

Patron--Lucas

Referred to Committee on Rules

S.J.R. 309. Proposing an amendment to Section 4 of Article IV of the Constitution of Virginia, relating to qualifications for members of the General Assembly; term limits.

Patron--Chase

Referred to Committee on Rules

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 304. Celebrating the life of the Honorable Mamye E. BaCote.

Patron--Locke

S.J.R. 305. Celebrating the life of Billy Joe Roberts.

Patron--Locke

S.J.R. 306. Commending Moneta Elementary School.

Patron--Newman

S.J.R. 307. Commending Pete's Pizza.

Patron--Reeves

CALENDAR

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

S.B. 1110 (one thousand one hundred ten).

S.B. 1175 (one thousand one hundred seventy-five).

S.B. 1132 (one thousand one hundred thirty-two).

S.B. 1171 (one thousand one hundred seventy-one).

S.B. 1190 (one thousand one hundred ninety).

The motion was agreed to.

The recorded vote is as follows:

YEAS--35. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Stanley, Suetterlein, Surovell, Vogel--35.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

S.B. 1110 (one thousand one hundred ten).

S.B. 1175 (one thousand one hundred seventy-five).

S.B. 1132 (one thousand one hundred thirty-two).

S.B. 1171 (one thousand one hundred seventy-one).

S.B. 1190 (one thousand one hundred ninety).

On motion of Senator Surovell, leaves of absence for the day were granted Senator Lewis and Senator Spruill on account of pressing personal business.

On motion of Senator McDougle, a leave of absence for the day was granted Senator Stuart on account of pressing personal business.

HONORARY ADJOURNMENT

Senator McClellan addressed the Senate in memory of the Reverend Dr. Martin Luther King, Jr.

Senator McClellan requested that when the Senate adjourns today, it adjourn in memory of the Reverend Dr. Martin Luther King, Jr.

On motion of Senator Lucas, the Senate, in memory of the Reverend Dr. Martin Luther King, Jr., adjourned until Monday, January 18,2021, at $12\,\mathrm{m}$.

Jacke Solman

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

MONDAY, JANUARY 18, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Barbara A. Favola, Thirty-first Senatorial District, offered the following prayer:

I ask you to join me in a moment of prayer and reflection.

O God, grant us the wisdom to find that moral universe and the courage to bend the arc of that universe towards justice as Dr. Martin Luther King, Jr. implored us to do. On our journey towards justice, teach us to embrace our diversity. For that will strengthen our common humanity. In our hearts we know that every one of us desires and deserves to be treated with dignity and respect. Few are granted the privileges we enjoy as members of this deliberative body and with those privileges comes the responsibility to govern with compassion and humility and fairness. Always remind us to never let pride get the best of us. Fill our hearts with love so we may mend what divides us and give us the will to find the bonds that unite us. As we embark on the work of making laws and caring for our fellow Virginians, remind us to not be so quick to judge. Many of us can recall the words of the Pulitzer Prize winning author, Harper Lee. Quote, "You never really understand a person until you consider things from his or her point of view. Until you climb into that person's skin and walk around in it." O God, give us kindness so we can make every day bright and bring joy to those we encounter. We ask for these gifts with sincerity and a singleness of purpose and we ask You to bless the Commonwealth of Virginia and the United States of America. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Tara H. Perkinson, Chief Deputy Clerk, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--37.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Hanger, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--34. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--34. NAYS--Cosgrove, Deeds, DeSteph, Petersen--4. RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- **S.B.** 1108 (one thousand one hundred eight) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1119** (one thousand one hundred nineteen) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1122 (one thousand one hundred twenty-two) with substitute.
 - **S.B.** 1123 (one thousand one hundred twenty-three).
 - S.B. 1142 (one thousand one hundred forty-two).
- **S.B. 1165** (one thousand one hundred sixty-five) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1168 (one thousand one hundred sixty-eight).
 - **S.B.** 1181 (one thousand one hundred eighty-one).
 - **S.B. 1241** (one thousand two hundred forty-one).
 - **S.B. 1270** (one thousand two hundred seventy).

The following bills, having been considered by the committee in session, were recommended for rereferral by the Committee on the Judiciary:

- **S.B. 1316** (one thousand three hundred sixteen) with the recommendation that it be rereferred to the Committee on Education and Health.
- **S.B. 1367** (one thousand three hundred sixty-seven) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

Senator Edwards, from the Committee on the Judiciary, presented the following reports:

SENATE OF VIRGINIA

January 18, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following persons are qualified to be elected to the respective circuit court judgeships as follows:

Brenda C. Spry, Esquire, of Portsmouth, as a judge of the Third Judicial Circuit for a term of eight years commencing February 16, 2021.

The Honorable Junius P. Fulton, III, of Norfolk, as a judge of the Fourth Judicial Circuit for a term of eight years commencing February 1, 2021.

The Honorable Kimberley S. White, of Halifax, as a judge of the Tenth Judicial Circuit for a term of eight years commencing February 1, 2021.

The Honorable Patricia Kelly, of Caroline, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Michael E. Levy, of Stafford, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Lisa Bondareff Kemler, of Alexandria, as a judge of the Eighteenth Judicial Circuit for a term of eight years commencing March 1, 2021.

The Honorable John M. Tran, of Fairfax County, as a judge of the Nineteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable G. Carter Greer, of Martinsville, as a judge of the Twenty-first Judicial Circuit for a term of eight years commencing March 1, 2021.

The Honorable James J. Reynolds, of Danville, as a judge of the Twenty-second Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable David B. Carson, of Roanoke City, as a judge of the Twenty-third Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Bruce D. Albertson, of Rockingham, as a judge of the Twenty-sixth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Deanis L. Simmons, of Bristol, as a judge of the Twenty-eighth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Jack S. Hurley, Jr., of Tazewell, as a judge of the Twenty-ninth Judicial Circuit for a term of eight years commencing February 1, 2021.

The Honorable Tammy S. McElyea, of Lee, as a judge of the Thirtieth Judicial Circuit for a term of eight years commencing April 1, 2021.

The Honorable Carroll A. Weimer, Jr., of Prince William, as a judge of the Thirty-first Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

SENATE OF VIRGINIA

January 18, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following persons are qualified to be elected to the respective general district court judgeships as follows:

The Honorable Vivian F. Henderson, of Virginia Beach, as a judge of the Second Judicial District for a term of six years commencing February 1, 2021.

The Honorable Tasha D. Scott, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Tyneka L. D. Flythe, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing July 1, 2021.

The Honorable M. Scott Stein, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing July 1, 2021.

The Honorable Tonya Henderson-Stith, of Hampton, as a judge of the Eighth Judicial District for a term of six years commencing May 1, 2021.

The Honorable David M. Hicks, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Jacqueline S. McClenney, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing December 1, 2021.

The Honorable Mansi J. Shah, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing February 1, 2021.

The Honorable B. Craig Dunkum, of Henrico, as a judge of the Fourteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable John K. Honey, Jr., of Henrico, as a judge of the Fourteenth Judicial District for a term of six years commencing November 1, 2021.

The Honorable Manuel A. Capsalis, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Michael J. Lindner, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable William J. Minor, Jr., of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing April 1, 2021.

The Honorable Tina L. Snee, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Jacqueline F. Ward Talevi, of Roanoke, as a judge of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.

The Honorable Stephanie S. Maddox, of Amherst, as a judge of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Amy B. Tisinger, of Shenandoah, as a judge of the Twenty-sixth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Erin J. DeHart, of Bland, as a judge of the Twenty-seventh Judicial District for a term of six years commencing July 1, 2021.

The Honorable Gino W. Williams, of Floyd, as a judge of the Twenty-seventh Judicial District for a term of six years commencing April 1, 2021.

The Honorable Robert P. Coleman, of Prince William, as a judge of the Thirty-first Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

SENATE OF VIRGINIA

January 18, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following persons are qualified to be elected to the respective juvenile and domestic relations district court judgeships as follows:

The Honorable Lyn M. Simmons, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing September 17, 2021.

The Honorable Jacqueline R. Waymack, of Prince George, as a judge of the Sixth Judicial District for a term of six years commencing June 1, 2021.

The Honorable John E. Franklin, of Fredericksburg, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Frank G. Uvanni, of Hanover, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Thomas P. Sotelo, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing February 1, 2021.

The Honorable Timothy W. Allen, of Franklin County, as a judge of the Twenty-second Judicial District for a term of six years commencing July 1, 2021.

The Honorable Hilary D. Griffith, of Salem, as a judge of the Twenty-third Judicial District for a term of six years commencing July 1, 2021.

The Honorable Melissa W. Friedman, of Roanoke City, as a judge of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.

The Honorable Jeffrey P. Bennett, of Amherst, as a judge of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Hugh David O'Donnell, of Rockingham, as a judge of the Twenty-sixth Judicial District for a term of six years commencing April 1, 2021.

The Honorable Robert C. Viar, Jr., of Montgomery, as a judge of the Twenty-seventh Judicial District for a term of six years commencing May 1, 2021.

The Honorable Joseph B. Lyle, of Washington, as a judge of the Twenty-eighth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Michael J. Bush, of Russell, as a judge of the Twenty-ninth Judicial District for a term of six years commencing April 1, 2021.

The Honorable Martha P. Ketron, of Russell, as a judge of the Twenty-ninth Judicial District for a term of six years commencing February 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

SENATE OF VIRGINIA

January 18, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following persons are qualified as members of the Judicial Inquiry and Review Commission as follows:

The Honorable James E. Plowman, of Loudoun, as a member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.

Terrie N. Thompson, of Chesapeake, as a member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

The following bills, having been considered by the committee in session, were reported by Senator Lewis from the Committee on Local Government:

- S.B. 1120 (one thousand one hundred twenty).
- S.B. 1128 (one thousand one hundred twenty-eight) with substitute.
- **S.B. 1141** (one thousand one hundred forty-one).
- S.B. 1152 (one thousand one hundred fifty-two).
- S.B. 1157 (one thousand one hundred fifty-seven).
- **S.B. 1207** (one thousand two hundred seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1208 (one thousand two hundred eight).
 - S.B. 1216 (one thousand two hundred sixteen).
 - **S.B. 1285** (one thousand two hundred eighty-five) with amendment.
- $\mathbf{S.B.}\ 1298$ (one thousand two hundred ninety-eight) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

- **S.B.** 1309 (one thousand three hundred nine) with substitute.
- **S.B.** 1355 (one thousand three hundred fifty-five) with substitute.

The following bill, having been considered by the committee in session, was recommended for rereferral by the Committee on Local Government:

- **S.B. 1226** (one thousand two hundred twenty-six) with the recommendation that it be rereferred to the Committee on the Judiciary.
- S.B. 1108, S.B. 1119, S.B, 1165, S.B. 1207, S.B. 1298, and S.B. 1367 were rereferred to the Committee on Finance and Appropriations.
 - **S.B.** 1226 was rereferred to the Committee on the Judiciary.
 - S.B. 1316 was rereferred to the Committee on Education and Health.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1440. A BILL to amend the Code of Virginia by adding a section numbered 8.01-42.6, relating to civil action for unlawful acts of force or failure to intervene by a law-enforcement officer or correctional officer.

Patron--Surovell

Referred to Committee on the Judiciary

S.B. 1441. A BILL to amend and reenact §§ 30-105, 64.2-1200, and 64.2-1202 of the Code of Virginia, relating to appointment of commissioners of accounts; members of the General Assembly.

Patron--Lucas

Referred to Committee on Rules

S.B. 1442. A BILL to amend and reenact § 19.2-163.04 of the Code of Virginia, relating to public defender offices; County of Chesterfield.

Patron--Morrissey

Referred to Committee on the Judiciary

S.B. 1443. A BILL to amend and reenact §§ 3.2-4212, 4.1-302, 16.1-253.2, 18.2-12.1, 18.2-31, 18.2-36.1, 18.2-36.2, 18.2-46.3:3, 18.2-51.1, 18.2-53.1,18.2-57, 18.2-60.4, 18.2-61, 18.2-67.1, 18.2-67.2, 18.2-67.5:2, 18.2-67.5:3, 18.2-154, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5, 18.2-255, 18.2-255.2, 18.2-270, 18.2-308.1, 18.2-308.2, 18.2-308.2:2, as it is currently effective and as it shall become effective, 18.2-308.4, 18.2-374.1, 18.2-374.1:1, 18.2-374.3, 19.2-297.1, 46.2-341.28, 46.2-357, 46.2-391, 46.2-865.1, and 53.1-203 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 19.2-303.03, relating to elimination of mandatory minimum sentences; modification of sentence to mandatory minimum term of confinement for felony offenses.

Patrons--Edwards and Morrissey

Referred to Committee on the Judiciary

S.B. 1444. A BILL to amend and reenact § 24.2-947.10 of the Code of Virginia, relating to filing of campaign finance reports.

Patron--Saslaw

Referred to Committee on Privileges and Elections

S.J.R. 310. Proposing an amendment to Section 6 of Article IV of the Constitution of Virginia, relating to legislative sessions; length of regular sessions convened in odd-numbered years.

Patron--Saslaw

Referred to Committee on Privileges and Elections

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 311. Celebrating the life of Robert Edward Mann.

Patron--Locke

S.J.R. 312. Commending O. R. Singleton, Jr.

Patron--Hashmi

S.J.R. 313. Commending Virginia's Crossroads.

Patron--Lucas

CALENDAR

SENATE BILLS ON SECOND READING

S.B. 1110 (one thousand one hundred ten) was read by title the second time.

The following amendment proposed by the Committee on General Laws and Technology was offered:

1. Line 29, introduced, after any

strike

person

insert

party to the real estate transaction

The reading of the amendment was waived.

On motion of Senator Spruill, the amendment was agreed to.

On motion of Senator Spruill, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1175** (one thousand one hundred seventy-five) was read by title the second time and, on motion of Senator Ruff, was ordered to be engrossed and read by title the third time.
 - S.B. 1132 (one thousand one hundred thirty-two) was read by title the second time.

Senator Suetterlein offered the following amendment:

1. Line 48, introduced, after *provides* strike

the remainder of line 48, all of lines 49, 50, and 51, and through day on line 52

insert

instruction and student services that are consistent with guidelines established by the Department of Education to ensure the equitable provision of such services. School divisions may not claim more than 10 unscheduled remote learning days in a school year unless the Superintendent of Public Instruction grants an extension

On motion of Senator Suetterlein, the reading of the amendment was waived.

On motion of Senator Suetterlein, the amendment was agreed to.

On motion of Senator Suetterlein, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1171** (one thousand one hundred seventy-one), on motion of Senator Norment, was passed by for the day.
- S.B. 1190 (one thousand one hundred ninety), on motion of Senator Kiggans, was passed by for the day.

SENATE BILL ON FIRST READING

S.B. 1299 (one thousand two hundred ninety-nine) was read by title the first time.

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

wan Carle Golman

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

TUESDAY, JANUARY 19, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Richard H. Stuart, Twenty-eighth Senatorial District, offered the following prayer:

Heavenly Father, we thank You for this day that You have given us and the ability to gather in this place. We ask for Your blessings for all, but especially for those less fortunate or with afflictions, and we ask You for wisdom and strength to do Your will so that we are truly able to see our endeavor of being one nation under God, indivisible, with liberty and justice for all. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Tara H. Perkinson, Chief Deputy Clerk, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Reeves, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--35. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, Petersen, Stanley--3.

RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 18, 2021

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTIONS:

- H.J.R. 512. Commending William Mann, M.D.
- H.J.R. 517. Celebrating the life of Lue Ethel Lapsley Carson.
- H.J.R. 518. Celebrating the life of Nathaniel Cridlin.
- H.J.R. 519. Celebrating the life of Frederick Barger, Jr.
- H.J.R. 520. Celebrating the life of Phillip G. Osborne.
- H.J.R. 521. Commending the Crater Planning District Commission.
- H.J.R. 532. Commending the McLean Volunteer Fire Department.
- H.J.R. 533. Celebrating the life of George Markey Keating.
- **H.J.R. 534.** Celebrating the life of Lilla Dunovant McCutchen Richards.
- H.J.R. 535. Commending Mark Stripe.
- H.J.R. 540. Commending Frank Tamberrino.
- H.J.R. 544. Commending Peter Clark.
- H.J.R. 547. Commending the Norfolk and Western Railway Section Foreman's House.
- H.J.R. 553. Commending Colonel William Joseph Flanagan, Jr., USA, Ret.
- H.J.R. 554. Commending Kent Stigall.
- H.J.R. 565. Celebrating the life of the Honorable Augustus Benton Chafin, Jr.
- H.J.R. 570. Commending Jill Harrell.
- **H.J.R. 571.** Celebrating the life of Alan Zimm.
- H.J.R. 580. Commending the Lake Ridge Occoquan Coles Civic Association.
- H.J.R. 581. Celebrating the life of Karl David Stoltzfus, Sr.
- H.J.R. 584. Commending Edwin Eagle.
- H.J.R. 585. Celebrating the life of Harold Reid.
- H.J.R. 586. Celebrating the life of the Reverend Alphonso Hamilton.

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING SENATE JOINT RESOLUTIONS:

- S.J.R. 273. Celebrating the life of Bruce Winston Edwards.
- S.J.R. 277. Commending Marymount University.
- S.J.R. 278. Commending the Honorable Bruce D. White.
- S.J.R. 279. Commending the Honorable Steven Selwyn Smith.
- S.J.R. 280. Commending the Honorable Kimberly J. Daniel.
- S.J.R. 281. Commending the Honorable Janine M. Saxe.
- S.J.R. 282. Commending the Crater Planning District Commission.
- S.J.R. 283. Celebrating the life of Marisa L. Fleck.
- S.J.R. 284. Celebrating the life of the Honorable Augustus Benton Chafin, Jr.
- **S.J.R. 287.** Celebrating the life of the Honorable Bernard S. Cohen.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, DeSteph, McDougle, Petersen, Reeves--5.

RULE 36--0.

The House joint resolutions, communicated as agreed to by the House of Delegates, were laid on the Clerk's Desk under Senate Rule 26 (g) as follows:

H.J.R. 512, H.J.R. 517, H.J.R. 518, H.J.R. 519, H.J.R. 520, H.J.R. 521, H.J.R. 532, H.J.R. 533, H.J.R. 534, H.J.R. 535, H.J.R. 540, H.J.R. 544, H.J.R. 547, H.J.R. 553, H.J.R. 554, H.J.R. 565, H.J.R. 570, H.J.R. 571, H.J.R. 580, H.J.R. 581, H.J.R. 584, H.J.R. 585, and H.J.R. 586.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Saslaw from the Committee on Commerce and Labor:

- S.B. 1182 (one thousand one hundred eighty-two).
- S.B. 1202 (one thousand two hundred two).
- S.B. 1219 (one thousand two hundred nineteen) with amendment.
- S.B. 1255 (one thousand two hundred fifty-five) with amendment.
- **S.B. 1276** (one thousand two hundred seventy-six).

The following bill, having been considered by the committee in session, was recommended for rereferral by the Committee on Commerce and Labor:

S.B. 1225 (one thousand two hundred twenty-five) with the recommendation that it be rereferred to the Committee on Education and Health.

Senator Saslaw, from the Committee on Commerce and Labor, presented the following reports:

SENATE OF VIRGINIA

January 18, 2021

To The Senate of Virginia:

The Committee on Commerce and Labor hereby certifies that the following person is qualified as a member of the State Corporation Commission, as follows:

The Honorable Jehmal T. Hudson, of the City of Richmond, as a member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2026.

Respectfully submitted,

/s/ Richard L. Saslaw, Chair

SENATE OF VIRGINIA

January 18, 2021

To The Senate of Virginia:

The Committee on Commerce and Labor hereby certifies that the following person is qualified as a member of the State Corporation Commission, as follows:

The Honorable Angela L. Navarro, of the City of Charlottesville, to succeed Mark C. Christie as a member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2022.

Respectfully submitted,

/s/ Richard L. Saslaw, Chair

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

S.B. 1130 (one thousand one hundred thirty) with amendment.

S.B. 1134 (one thousand one hundred thirty-four).

- S.B. 1145 (one thousand one hundred forty-five) with amendment.
- **S.B.** 1155 (one thousand one hundred fifty-five) with amendment.
- S.B. 1156 (one thousand one hundred fifty-six).
- **S.B. 1211** (one thousand two hundred eleven).
- **S.B.** 1225 was rereferred to the Committee on Education and Health.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1445. A BILL to facilitate the administration of the COVID-19 vaccine; emergency.

EMERGENCY

Patron--Dunnavant

Referred to Committee on Education and Health

S.B. 1446. A BILL to amend the Code of Virginia by adding a section numbered 8.01-401.4, relating to the practice of medicine and other healing arts; provision of litigation assistance.

Patron--Surovell

Referred to Committee on Education and Health

S.B. 1447. A BILL to amend and reenact § 15.2-2159 of the Code of Virginia, relating to fees for disposal of solid waste; Buckingham County.

Patron--Peake

Referred to Committee on Local Government

S.B. 1448. A BILL to amend and reenact § 58.1-322.02 of the Code of Virginia, relating to income tax; subtraction for military retirement income and survivor benefits.

Patron--Chase

Referred to Committee on Finance and Appropriations

S.B. 1449. A BILL to amend and reenact §§ 2.2-2901.1, 2.2-3004, 15.2-1500.1, 15.2-1507, as it is currently effective and as it shall become effective, 15.2-1604, 22.1-295.2, 22.1-306, 32.1-127, and 65.2-402.1 of the Code of Virginia and to amend the Code of Virginia by adding in Article 2 of Chapter 1 of Title 32.1 a section numbered 32.1-15.2, by adding in Chapter 2 of Title 37.2 a section numbered 37.2-205, by adding a section numbered 40.1-27.4, and by adding in Article 2 of Chapter 2 of Title 63.2 a section numbered 63.2-221.1, relating to COVID-19 immunization; prohibition on requirement; employment discrimination prohibited.

Patron--Chase

Referred to Committee on General Laws and Technology

S.B. 1450. A BILL to amend and reenact §§ 2.2-2203.3, 2.2-2901.1, 2.2-3004, 15.2-1500.1, 15.2-1507, as it is currently effective and as it shall become effective, 15.2-1604, 22.1-295.2, 22.1-306, 23.1-2415, and 33.2-290 of the Code of Virginia, relating to COVID-19 vaccination; discrimination in employment prohibited.

Patron--Chase

Referred to Committee on General Laws and Technology

S.B. 1451. A BILL to amend the Code of Virginia by adding a section numbered 2.2-1147.4, relating to Department of General Services; equal access to state and local public property by persons who choose not to receive COVID-19 vaccine.

Patron--Chase

Referred to Committee on General Laws and Technology

S.B. 1452. A BILL to authorize the issuance of special license plates for supporters of the Southwest Regional Recreation Authority bearing the legend SPEARHEAD TRAILS; fees.

Patron--Pillion (By Request)

Referred to Committee on Transportation

S.B. 1453. A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 10.1 an article numbered 5, consisting of sections numbered 10.1-1332 and 10.1-1333, by adding in Chapter 1 of Title 33.2 a section numbered 33.2-120, by adding in Article 2 of Chapter 2 of Title 33.2 a section numbered 33.2-221.1, by adding a title numbered 45.2, containing a subtitle numbered I, consisting of chapters numbered 1 through 4, containing sections numbered 45.2-100 through 45.2-402, a subtitle numbered II, consisting of chapters numbered 5 through 10, containing sections numbered 45.2-500 through 45.2-1051, a subtitle numbered III, consisting of chapters numbered 11 through 15, containing sections numbered 45.2-1100 through 45.2-1505, a subtitle numbered IV, consisting of a chapter numbered 16, containing sections numbered 45.2-1600 through 45.2-1649, and a subtitle numbered V, consisting of chapters numbered 17 through 21, containing sections numbered 45.2-1700 through 45.2-2119, by adding sections numbered 55.1-1820.1, 55.1-1951.1, and 55.1-2133.1, and by adding in Title 56 a chapter numbered 29, consisting of sections numbered 56-614 through 56-624, and to repeal Chapter 6.1 (§§ 11-34.1 through 11-34.4) of Title 11, Title 45.1 (§§ 45.1-161.1 through 45.1-399), §§ 62.1-195.1 and 62.1-195.3, and Title 67 (§§ 67-100 through 67-1700) of the Code of Virginia, relating to administration of the Department of Mines, Minerals and Energy, coal mining, mineral mines, gas and oil, and other sources of energy and energy policy.

Patron--Edwards

Referred to Committee on Agriculture, Conservation and Natural Resources

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 314. Commending the Virginia Chapter of American Promise and Take Back Our Republic. Patron--Deeds

S.J.R. 315. Celebrating the life of Theodore Carter DeLaney, Jr.

Patron--Deeds

S.J.R. 316. Celebrating the life of Edith Luke Martin.

Patron--Deeds

S.J.R. 317. Celebrating the life of Walter Lowrie Martin III.

Patron--Deeds

S.J.R. 318. Celebrating the life of Stephen Teel Goodwin.

Patrons--Reeves and Boysko; Delegates: Adams, D.M., Cole, J.G., Cole, M.L., Coyner, Davis, Fowler, Freitas, Heretick, Hope, Mundon King, Rasoul, Reid, Simon, Simonds, Ware, Webert and Wiley

S.R. 92. Celebrating the life of Luther Hilton Foster.

Patron--Ruff

JOINT ORDER FOR ELECTIONS

Senator Edwards moved that, pursuant to Rule 5 of **H.J.R.** 575 (five hundred seventy-five), the Senate agree to hold the special and continuing joint order relating to judicial elections on Tuesday, January 26, 2021 at 3:00 p.m.

The motion was agreed to.

The recorded vote is as follows:

YEAS--33. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--33.

NAYS--Chase, Morrissey, Newman, Obenshain, Suetterlein--5.

RULE 36--0.

Senator Edwards was ordered to inform the House of Delegates thereof.

COMMITTEE NOMINATIONS REPORT

Senator Locke presented the following report:

2021 SENATE STANDING COMMITTEES

Education and Health

Lucas, Chair; Saslaw; Howell; Newman; Locke; Barker; Petersen; Cosgrove; Lewis; Dunnavant; Suetterlein; Peake; Edwards; Hashmi and *Pillion*

Judiciary

Edwards, Chair; Saslaw; Norment; Lucas; Obenshain; McDougle; Stuart; Stanley; Deeds; Petersen; Surovell; McClellan; Boysko; Morrissey and *Peake*

Local Government

Lewis, Chair; Hanger; Stanley; Favola; DeSteph; Dunnavant; McPike; Howell; Spruill; Morrissey; Pillion; Bell; Hashmi; Kiggans and *Reeves*

Privileges and Elections

Deeds, Chair; Howell; Vogel; Reeves; Ebbin; Ruff; Spruill; Peake; McDougle; Surovell; Mason; McClellan; Boysko; Bell and *Dunnavant*

Rehabilitation and Social Services

Favola, Chair; Hanger; Locke; McDougle; Reeves; Cosgrove; Surovell; DeSteph; McPike; Spruill; Mason; Boysko; Morrissey; Kiggans and *Stuart*

Transportation

Marsden, Chair; Newman; Favola; Cosgrove; DeSteph; Suetterlein; McClellan; Peake; Boysko; Obenshain; Ebbin; Surovell; McPike; Spruill and *Stanley*

/s/ Mamie E. Locke

Senator Locke moved that the Committee Nominations Report be adopted.

RECESS

-101-

At 12:40 p.m., Senator Norment moved that the Senate recess until 12:55 p.m.

The motion was agreed to.

The hour of 12:55 p.m. having arrived, the Chair was resumed.

COMMITTEE NOMINATIONS REPORT

The Committee Nominations Report was taken up.

The question was put on adopting the Committee Nominations Report.

The Committee Nominations Report was adopted.

The recorded vote is as follows:

YEAS--37. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Chase--1.

RULE 36--0.

CALENDAR

SENATE BILLS ON THIRD READING

S.B. 1110 (one thousand one hundred ten) was read by title the third time and, on motion of Senator Spruill, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1175 (one thousand one hundred seventy-five) was read by title the third time and, on motion of Senator Ruff, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1132 (one thousand one hundred thirty-two) was read by title the third time and, on motion of Senator Suetterlein, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE BILLS ON SECOND READING

S.B. 1299 (one thousand two hundred ninety-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 4.1-119, as it is currently effective and as it shall become effective, 4.1-204, as it is currently effective and as it shall become effective, 4.1-206.1, as it shall become effective, 4.1-206.3, as it shall become effective, 4.1-207, 4.1-210, 4.1-212.1, as it is currently effective and as it shall become effective, and 4.1-221 of the Code of Virginia, relating to alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption.

The reading of the substitute was waived.

On motion of Senator Bell, the substitute was agreed to.

On motion of Senator Bell, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1171** (one thousand one hundred seventy-one), on motion of Senator Norment, was passed by for the day.
- **S.B. 1190** (one thousand one hundred ninety) was read by title the second time and, on motion of Senator Kiggans, was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- **S.B. 1120** (one thousand one hundred twenty).
- **S.B.** 1152 (one thousand one hundred fifty-two).
- S.B. 1168 (one thousand one hundred sixty-eight).
- **S.B. 1216** (one thousand two hundred sixteen).
- S.B. 1241 (one thousand two hundred forty-one).
- S.B. 1270 (one thousand two hundred seventy).
- **S.B.** 1122 (one thousand one hundred twenty-two).
- S.B. 1123 (one thousand one hundred twenty-three).
- S.B. 1128 (one thousand one hundred twenty-eight).
- S.B. 1141 (one thousand one hundred forty-one).
- S.B. 1142 (one thousand one hundred forty-two).
- **S.B.** 1157 (one thousand one hundred fifty-seven).
- S.B. 1181 (one thousand one hundred eighty-one).
- S.B. 1208 (one thousand two hundred eight).
- S.B. 1285 (one thousand two hundred eighty-five).
- **S.B. 1309** (one thousand three hundred nine).
- S.B. 1355 (one thousand three hundred fifty-five).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- S.B. 1120 (one thousand one hundred twenty).
- S.B. 1152 (one thousand one hundred fifty-two).
- **S.B.** 1168 (one thousand one hundred sixty-eight).
- S.B. 1216 (one thousand two hundred sixteen).
- S.B. 1241 (one thousand two hundred forty-one).
- **S.B. 1270** (one thousand two hundred seventy).
- S.B. 1122 (one thousand one hundred twenty-two).
- S.B. 1123 (one thousand one hundred twenty-three).
- S.B. 1128 (one thousand one hundred twenty-eight).
- S.B. 1141 (one thousand one hundred forty-one).
- S.B. 1142 (one thousand one hundred forty-two).
- S.B. 1157 (one thousand one hundred fifty-seven).
- **S.B.** 1181 (one thousand one hundred eighty-one).
- S.B. 1208 (one thousand two hundred eight).
- S.B. 1285 (one thousand two hundred eighty-five).

S.B. 1309 (one thousand three hundred nine).

S.B. 1355 (one thousand three hundred fifty-five).

HOUSE COMMUNICATION

The following communication was received and read:

In the House of Delegates January 19, 2021

THE HOUSE OF DELEGATES HAS POSTPONED THE SPECIAL AND CONTINUING JOINT ORDER UNTIL TUESDAY, JANUARY 26, 2021, AT 3:00 P.M., PURSUANT TO RULE 5 OF HJR 575.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

ysan Carle Delsan

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

WEDNESDAY, JANUARY 20, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Jennifer L. McClellan, Ninth Senatorial District, offered the following prayer:

Heavenly Father, on this historic day may You bless President Joe Biden and Vice President Kamala Harris and extend to them Your amazing grace as they begin the difficult task of leading this land and binding up divisions. May You also send Your blessings and grace on this Chamber, make us ambassadors of comfort and reconciliation, open our ears that we may listen with respect and understanding, touch our lips that we may speak words of wisdom, warm our hearts that we may bring wholeness to the broken hearted, comfort to the suffering, and dissolve the barriers of division. Guide our work and renew us with the spirit of Your love. Help us shape a Commonwealth where all will have a place, where the flames of hatred are quenched, and where all can grow together as one. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Tara H. Perkinson, Chief Deputy Clerk, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Pillion, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--29. NAYS--10. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--29.

NAYS--Cosgrove, Deeds, DeSteph, Dunnavant, Kiggans, McDougle, Petersen, Reeves, Stanley, Suetterlein--10.

RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 19, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1747.** A BILL to amend and reenact §§ 54.1-2900, 54.1-2901, 54.1-2957, 54.1-2957.01, and 54.1-3000 of the Code of Virginia and to repeal § 54.1-3018.1 of the Code of Virginia, relating to clinical nurse specialist; licensure by the Boards of Medicine and Nursing.
- **H.B. 1751.** A BILL to amend and reenact § 3.2-1905 of the Code of Virginia, relating to the excise tax on peanuts.
- **H.B. 1805.** A BILL to amend and reenact §§ 51.5-134 and 51.5-135 of the Code of Virginia, relating to aging services; economic and social need.
- **H.B. 1807.** A BILL to amend and reenact § 38.2-4310 of the Code of Virginia, relating to health maintenance organizations; insolvency.
- H.B. 1808. A BILL to amend and reenact § 37.2-304 of the Code of Virginia, relating to Commissioner of Behavioral Health and Developmental Services; reports to designated protection and advocacy system.
- **H.B. 1810.** A BILL to amend and reenact § 24.2-416 of the Code of Virginia, relating to voter registration; failure of online voter registration system; deadline extension.
- **H.B. 1813.** A BILL to amend and reenact § 33.2-234 of the Code of Virginia, relating to highway construction by state or local employees; limit.
- **H.B. 1816.** A BILL to amend and reenact §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953 of the Code of Virginia, relating to the Property Owners' Association Act; the Condominium Act; use of electronic means for meetings and voting.
- **H.B. 1829.** A BILL to amend and reenact §§ 38.2-4319 and 38.2-4509 of the Code of Virginia, relating to health insurance; credentialing; health care providers.
- **H.B. 1836.** A BILL to amend and reenact §§ 2.2-200, 2.2-205, 2.2-206, 2.2-215, 2.2-220.1, 2.2-220.2, 2.2-220.3, 2.2-1156, 2.2-1176, 2.2-2316, 2.2-2338, 2.2-2481, 2.2-2699.10, 2.2-2699.11, 2.2-2699.13, 10.1-603.25, 10.1-704, 10.1-1018, 10.1-1181.15, 10.1-1188, 10.1-1329, 10.1-1402.03, 10.1-1402.04, 10.1-1405, 10.1-2129, 10.1-2202.3, 15.2-2295.1, 28.2-207, 29.1-102, 29.1-573, 29.1-579, 30-377, 56-596.2, 58.1-344.3, 62.1-44.15:68, 62.1-44.34:25, 62.1-44.34:28, 62.1-44.117, 62.1-44.118, 62.1-44.119:1, 62.1-69.31, 62.1-69.32, 62.1-69.33, 62.1-69.41, 62.1-69.52, and 62.1-195.1 of the Code of Virginia, relating to the Secretary of Natural Resources.
- H.B. 1842. A BILL to amend and reenact §§ 55.1-1819 and 55.1-1959 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 55.1-1819.1 and 55.1-1960.1, relating to the Property Owners' Association Act; the Condominium Act; rulemaking authority of property owners' associations and unit owners' associations; smoking.

- H.B. 1845. A BILL to amend and reenact §§ 4.1-230, as it shall become effective, and 4.1-233.1 of the Code of Virginia and to amend and reenact the third, fifth, and eighth enactments of Chapter 1113 of the Acts of Assembly of 2020 and the third, fifth, and eighth enactments of Chapter 1114 of the Acts of Assembly of 2020, relating to alcoholic beverage control; license fee reform; delay; emergency. EMERGENCY
- **H.B. 1850.** A BILL to amend and reenact § 46.2-1129.2 of the Code of Virginia, relating to motor vehicle weight limits; vehicles powered primarily by electric battery power or fueled primarily by natural gas.
- **H.B. 1855.** A BILL to amend and reenact §§ 2.2-204, 2.2-604.2, 2.2-1157, 2.2-1176.1, 2.2-3705.6, 2.2-4006, 10.1-606.3, 10.1-659, 10.1-1194, 10.1-1329, 10.1-1330, 10.1-1406.2, 11-34.3, 15.2-958.3, 15.2-980, 15.2-2224, 23.1-2626, 23.1-2627, 28.2-1208, 30-275, 33.2-236, 45.1-161.1, 45.1-161.2, 45.1-161.5, 45.1-161.15, 45.1-161.179, 45.1-161.292:2, 45.1-180, 45.1-229, 45.1-230, 45.1-270.4:1, 45.1-361.28, 45.1-361.41, 45.1-383, 45.1-390, 56-265.15:1, 56-576, 56-585.5, 56-594.3, 56-596.2, 58.1-439.2, 58.1-439.12:02, 58.1-3660, 58.1-3706, 58.1-3745, 62.1-44.15:21, 62.1-44.15:66, 62.1-195.1, 62.1-243, 62.1-256, 62.1-259, 63.2-805, 67-200, 67-202.1, 67-602, 67-900, 67-1000, 67-1206, 67-1208, 67-1209, 67-1403, and 67-1506 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 45.1-161.4:1, relating to Department of Mines, Minerals and Energy.
- **H.B. 1873.** A BILL to amend and reenact § 37.2-403 of the Code of Virginia, relating to brain injury; definition.
- **H.B. 1881.** A BILL to amend and reenact § 59.1-547 of the Code of Virginia, relating to enterprise zone job creation grants.
- H.B. 1888. A BILL to amend and reenact §§ 24.2-101, 24.2-416.1, 24.2-603, 24.2-653.1, 24.2-704, 24.2-705, 24.2-706 through 24.2-711, and 24.2-712 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 24.2-103.2, 24.2-667.1, and 24.2-707.1, relating to absentee voting; procedural and process reforms; availability and accessibility reforms; penalty.
- **H.B. 1892.** A BILL to amend and reenact § 38.2-317 of the Code of Virginia, relating to approval of property and casualty insurance policy forms and endorsements.
- **H.B. 1902.** A BILL to amend and reenact §§ 10.1-1414 and 10.1-1422.01 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 10.1-1424.3, relating to expanded polystyrene food service containers; prohibition; civil penalty.
- **H.B. 1903.** A BILL to amend and reenact § 46.2-1300, as it is effective and as it shall become effective, of the Code of Virginia, relating to local government authority; reduction of speed limits.
- H.B. 1908. A BILL to amend and reenact § 55.1-1245, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; noncompliance with rental agreement; prohibition on using negative credit information that arose during a closure of the United States Government against certain applicants for tenancy; penalty.
- IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--36. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Vogel--36.

NAYS--Deeds, Petersen, Surovell--3.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1747, H.B. 1808, and H.B. 1873 were referred to the Committee on Education and Health.
- H.B. 1751, H.B. 1836, H.B. 1855, and H.B. 1902 were referred to the Committee on Agriculture, Conservation and Natural Resources.
 - H.B. 1805 and H.B. 1845 were referred to the Committee on Rehabilitation and Social Services.
- H.B. 1807, H.B. 1829, H.B. 1881, and H.B. 1892 were referred to the Committee on Commerce and Labor.
 - H.B. 1810 and H.B. 1888 were referred to the Committee on Privileges and Elections.
 - H.B. 1813, H.B. 1850, and H.B. 1903 were referred to the Committee on Transportation.
- H.B. 1816, H.B. 1842, and H.B. 1908 were referred to the Committee on General Laws and Technology.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Petersen from the Committee on Agriculture, Conservation and Natural Resources:

- S.B. 1161 (one thousand one hundred sixty-one) with substitute.
- **S.B. 1188** (one thousand one hundred eighty-eight) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1193** (one thousand one hundred ninety-three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1194 (one thousand one hundred ninety-four).
 - **S.B.** 1199 (one thousand one hundred ninety-nine) with amendments.
- **S.B. 1258** (one thousand two hundred fifty-eight) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1274 (one thousand two hundred seventy-four) with amendments.
 - S.B. 1280 (one thousand two hundred eighty).
 - S.B. 1290 (one thousand two hundred ninety) with substitute.

- S.B. 1291 (one thousand two hundred ninety-one) with amendments.
- **S.B. 1402** (one thousand four hundred two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- **S.B.** 1112 (one thousand one hundred twelve) with amendments.
- S.B. 1158 (one thousand one hundred fifty-eight).

The following bills, joint resolutions, and resolution, having been considered by the committee in session, were reported by Senator Deeds from the Committee on Privileges and Elections:

- S.B. 1097 (one thousand ninety-seven).
- **S.B. 1111** (one thousand one hundred eleven).
- **S.B. 1239** (one thousand two hundred thirty-nine).
- S.B. 1281 (one thousand two hundred eighty-one) with amendments.
- S.J.R. 298 (two hundred ninety-eight).
- S.J.R. 299 (two hundred ninety-nine) with amendment.
- S.J.R. 300 (three hundred).
- S.J.R. 301 (three hundred one).
- S.J.R. 302 (three hundred two).
- S.J.R. 303 (three hundred three).
- S.R. 91 (ninety-one).
- S.B. 1188, S.B. 1193, S.B. 1258, and S.B. 1402 were rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1454. A BILL to amend the Code of Virginia by adding in Article 7 of Chapter 14 of Title 8.01 a section numbered 8.01-413.03, relating to evidence of mental competence; medical scans of brain. Patron--Ruff (By Request)

Referred to Committee on the Judiciary

S.B. 1455. A BILL to amend and reenact §§ 24.2-702.1, 24.2-706, and 24.2-707 of the Code of Virginia, relating to absentee ballots; witness requirement; printed name and residence address.

Patron--Ruff

Referred to Committee on Privileges and Elections

S.B. 1456. A BILL to amend and reenact §§ 16.1-248.1, 16.1-249, 16.1-278.7, and 16.1-278.8 of the Code of Virginia, relating to juveniles; eligibility for commitment to the Department of Juvenile Justice; eligibility for predispositional confinement in a secure facility.

Patron--Marsden

Referred to Committee on the Judiciary

S.B. 1457. A BILL to amend and reenact § 15.2-2306 of the Code of Virginia, relating to preservation of historic sites.

Patron--Surovell

Referred to Committee on Local Government

S.B. 1458. A BILL to amend and reenact §§ 2.2-436 and 2.2-437 of the Code of Virginia, relating to Secretary of Commerce and Trade; Identity Management Standards Advisory Council.

Patron--Boysko

Referred to Committee on General Laws and Technology

S.R. 93. Nominating a person to be elected as the Auditor of Public Accounts.

Patron--Deeds

Referred to Committee on Privileges and Elections

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 319. Celebrating the life of Michael Stephen Horwatt.

Patrons--Howell, Barker, Bell, Boysko and Saslaw; Delegates: Delaney, Hope, Plum, Reid, Simon, Sullivan and Watts

S.J.R. 320. Celebrating the life of Dr. John E. Thomasson.

Patrons--Reeves, Boysko and Howell; Delegates: Cole, J.G., Cole, M.L., Davis, Heretick, Reid, Simonds and Watts

CALENDAR

SENATE BILLS ON THIRD READING

S.B. 1299 (one thousand two hundred ninety-nine) was read by title the third time and, on motion of Senator Bell, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1.

RULE 36--0.

S.B. 1190 (one thousand one hundred ninety) was read by title the third time and, on motion of Senator Kiggans, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--31.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Obenshain, Peake, Stuart, Suetterlein--8. RULE 36--0.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1120 (one thousand one hundred twenty).
- S.B. 1152 (one thousand one hundred fifty-two).
- S.B. 1168 (one thousand one hundred sixty-eight).
- S.B. 1216 (one thousand two hundred sixteen).
- S.B. 1241 (one thousand two hundred forty-one).
- S.B. 1270 (one thousand two hundred seventy).

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1120 (one thousand one hundred twenty).
- S.B. 1152 (one thousand one hundred fifty-two).
- S.B. 1168 (one thousand one hundred sixty-eight).
- S.B. 1216 (one thousand two hundred sixteen).
- S.B. 1241 (one thousand two hundred forty-one).
- S.B. 1270 (one thousand two hundred seventy).
- S.B. 1171 (one thousand one hundred seventy-one) was read by title the second time.

The following amendment proposed by the Committee on General Laws and Technology was offered:

1. Line 22, introduced, after 1.

insert

The executive director and members of such industrial development authorities and economic development authorities shall also comply with the provisions of § 2.2-3112.

The reading of the amendment was waived.

Senator Barker moved that the amendment be rejected.

The question was put on agreeing to the amendment.

The amendment was rejected.

Senator Barker offered the following amendments:

1. Line 27, introduced, after et seq.),

strike

the remainder of line 27, all of line 28, and through 25,000, on line 29

2. Line 33, introduced, after 2.2-3117.

insert

The executive director and members of each industrial development authority and economic development authority, as created by the Industrial Development and Revenue Bond Act (§ 15.2-4900 et seq.), in a single locality with a population of 25,000 or fewer or in a region serving more than one locality that, in the aggregate, has a population of 25,000 or fewer, shall not be required to file, as a condition to assuming office or annually thereafter, a disclosure statement of their personal interests pursuant to § 2.2-3117 or 2.2-3118, but shall at all times comply with the provisions of § 2.2-3112.

On motion of Senator Barker, the reading of the amendments was waived.

On motion of Senator Barker, the amendments were agreed to.

Senator Norment offered the following amendments:

1. Line 25, introduced, after year,

strike

other than

insert

and

2. Line 28, introduced, after *population* strike

in excess

3. Line 28, introduced, after 25,000

insert

or fewer

4. Line 29, introduced, after population

strike

in excess

5. Line 29, introduced, after 25,000

insert

 $or \, fewer$

Senator Norment withdrew the amendments.

On motion of Senator Norment, the bill was ordered to be engrossed and read by title the third time.

S.B. 1122 (one thousand one hundred twenty-two) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 8.01-9, 8.01-407, 16.1-77, 16.1-305, 17.1-213, 19.2-389, as it is currently effective and as it shall become effective, 46.2-301, 46.2-301.1, 46.2-411, and 53.1-21 of the Code of Virginia and to repeal Article 9 (§§ 46.2-355.1 through 46.2-363) of Chapter 3 of Title 46.2 of the Code of Virginia, relating to habitual offenders; repeal.

The reading of the substitute was waived.

On motion of Senator Stanley, the substitute was agreed to.

On motion of Senator Stanley, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1123** (one thousand one hundred twenty-three) was read by title the second time and, on motion of Senator Obenshain, was ordered to be engrossed and read by title the third time.
 - **S.B.** 1128 (one thousand one hundred twenty-eight) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Local Government was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 44 and § 133, as amended, of Chapter 34 of the Acts of Assembly of 1918 and to repeal § 61 of Chapter 34 of the Acts of Assembly of 1918, which provided a charter for the City of Norfolk, relating to general updates.

The reading of the substitute was waived.

On motion of Senator Spruill, the substitute was agreed to.

On motion of Senator Spruill, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1141** (one thousand one hundred forty-one) was read by title the second time and, on motion of Senator Cosgrove, was ordered to be engrossed and read by title the third time.
- **S.B. 1142** (one thousand one hundred forty-two) was read by title the second time and, on motion of Senator Cosgrove, was ordered to be engrossed and read by title the third time.
 - S.B. 1157 (one thousand one hundred fifty-seven) was read by title the second time.

Senator Petersen offered the following amendment:

- 1. After line 31, introduced insert
 - 3. That the provisions of this act shall not apply to any locality located in Planning District 8.

Senator Petersen withdrew the amendment.

On motion of Senator Spruill, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1181** (one thousand one hundred eighty-one) was read by title the second time and, on motion of Senator Surovell, was ordered to be engrossed and read by title the third time.
- **S.B. 1208** (one thousand two hundred eight) was read by title the second time and, on motion of Senator Barker, was ordered to be engrossed and read by title the third time.
 - S.B. 1285 (one thousand two hundred eighty-five) was read by title the second time.

The following amendment proposed by the Committee on Local Government was offered:

1. Line 15, introduced, after more,

strike

and which

insert

and which may endanger the public health, safety, or welfare

The reading of the amendment was waived.

On motion of Senator Locke, the amendment was agreed to.

On motion of Senator Locke, the bill was ordered to be engrossed and read by title the third time.

S.B. 1309 (one thousand three hundred nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Local Government was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 15.2-2114.01 of the Code of Virginia, relating to local stormwater assistance; flood mitigation and protection.

The reading of the substitute was waived.

On motion of Senator Ebbin, the substitute was agreed to.

On motion of Senator Ebbin, the bill was ordered to be engrossed and read by title the third time.

S.B. 1355 (one thousand three hundred fifty-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Local Government was offered, having been printed separately, with its title reading as follows:

A BILL to allow the withdrawal of a member of the Rapidan Service Authority.

The reading of the substitute was waived.

On motion of Senator Hanger, the substitute was agreed to.

S.B. 1355, on motion of Senator Hanger, was passed by for the day.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1130 (one thousand one hundred thirty).
- S.B. 1134 (one thousand one hundred thirty-four).
- S.B. 1145 (one thousand one hundred forty-five).
- S.B. 1155 (one thousand one hundred fifty-five).
- **S.B.** 1156 (one thousand one hundred fifty-six).
- S.B. 1255 (one thousand two hundred fifty-five).

S.B. 1182 (one thousand one hundred eighty-two).

S.B. 1202 (one thousand two hundred two).

S.B. 1211 (one thousand two hundred eleven).

S.B. 1219 (one thousand two hundred nineteen).

S.B. 1276 (one thousand two hundred seventy-six).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

S.B. 1130 (one thousand one hundred thirty).

S.B. 1134 (one thousand one hundred thirty-four).

S.B. 1145 (one thousand one hundred forty-five).

S.B. 1155 (one thousand one hundred fifty-five).

S.B. 1156 (one thousand one hundred fifty-six).

S.B. 1255 (one thousand two hundred fifty-five).

S.B. 1182 (one thousand one hundred eighty-two).

S.B. 1202 (one thousand two hundred two).

S.B. 1211 (one thousand two hundred eleven).

S.B. 1219 (one thousand two hundred nineteen).

S.B. 1276 (one thousand two hundred seventy-six).

SENATE BILL ON SECOND READING RECONSIDERATION

Senator Cosgrove moved to reconsider the vote by which **S.B. 1142** (one thousand one hundred forty-two) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1142, on motion of Senator Cosgrove, was passed by for the day.

On motion of Senator Lucas, the Senate adjourned until tomorrow at $12\ m.$

Susan Carle Gelman

-116-

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

THURSDAY, JANUARY 21, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Ryan T. McDougle, Fourth Senatorial District, offered the following prayer:

A reading from the Psalms 143:1 and 8, "Lord, hear my prayer, listen to my cry for mercy; in Your faithfulness and righteousness, come to my relief. Let the morning bring me word of Your unfailing love for I have put my trust in You. Show me the way I should go, for to You I lift up my soul." Lord, as we go through the trials and tribulations of life and this pursuit of good government, as we go through also the joys and exultations of life, and certainly the challenges that we have been through over the last year, we ask that You give us wisdom and discernment. We ask that You give us calmness and discretion. Lord, we are all sinners and we are so thankful for the blessings that You have bestowed upon us. So as we go through these, as we have the heated discussions, late days, lack of sleep, we ask that You remind us of those blessings. Even in difficult times the simple blessings of being able to be together to make decisions, the opportunities to serve You and our fellow Virginians, and Lord, as we continue to go forward, remind us constantly of what You have given us. Not for us to focus on what we do not have or cannot do. Thank You for Your love. Thank You for always being with us. It is in Your holy name that we pray. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Tara H. Perkinson, Chief Deputy Clerk, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Mason, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Deeds, Petersen--2. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 20, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1743.** A BILL to amend and reenact § 15.2-4904 of the Code of Virginia, relating to industrial development authorities; size of board in certain towns; quorum.
- **H.B. 1768.** A BILL to amend and reenact § 5, as amended, of Chapter 76 of the Acts of Assembly of 1974, which created the Lynchburg Parking Authority, relating to terms of members.
- **H.B. 1825.** A BILL to amend and reenact § 15.2-5931 of the Code of Virginia, relating to Virginia Beach Sports or Entertainment Project; bond issuance.
- **H.B. 1859.** A BILL to amend and reenact § 15.2-958.3 of the Code of Virginia, relating to financing clean energy and other programs; when owner costs incurred.
- **H.B. 1894.** A BILL to amend and reenact § 54.1-3408 of the Code of Virginia, relating to certain employees of the Department of Juvenile Justice; naloxone or other opioid antagonist.
- **H.B. 1927.** A BILL to amend and reenact §§ 3, 4, and 5 of Chapter 643 of the Acts of Assembly of 1964, as amended by Chapter 882 of the Acts of Assembly of 2003, relating to economic development authorities.
- **H.B. 1948.** A BILL to amend and reenact § 52-30.1 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 15.2-1704.1 and 15.2-1704.2, relating to law-enforcement officer; duty to render aid; duty to report wrongdoing by another law-enforcement officer.
- **H.B. 1949.** A BILL to amend and reenact §§ 15.2-520 and 15.2-2506 of the Code of Virginia, relating to county executive form of government; local budgets.
- **H.B. 2085.** A BILL to amend and reenact § 44-146.19 of the Code of Virginia, relating to Emergency Services and Disaster Law; local and interjurisdictional emergency operations plans.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--36. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Deeds, Petersen, Stanley--3.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1743, H.B. 1768, H.B. 1825, H.B. 1859, H.B. 1927, and H.B. 1949 were referred to the Committee on Local Government.
 - H.B. 1894 was referred to the Committee on Education and Health.
 - H.B. 1948 was referred to the Committee on the Judiciary.
 - H.B. 2085 was referred to the Committee on General Laws and Technology.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Lucas from the Committee on Education and Health:

- **S.B. 1121** (one thousand one hundred twenty-one) with substitute.
- **S.B. 1133** (one thousand one hundred thirty-three) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1154 (one thousand one hundred fifty-four) with substitute.
 - S.B. 1169 (one thousand one hundred sixty-nine).
 - S.B. 1187 (one thousand one hundred eighty-seven).
 - S.B. 1189 (one thousand one hundred eighty-nine) with amendment.
 - S.B. 1205 (one thousand two hundred five).
 - S.B. 1220 (one thousand two hundred twenty).
- **S.B. 1221** (one thousand two hundred twenty-one) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1227** (one thousand two hundred twenty-seven) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1302** (one thousand three hundred two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1313** (one thousand three hundred thirteen) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1320 (one thousand three hundred twenty) with amendment.
 - S.B. 1322 (one thousand three hundred twenty-two) with substitute.
 - **S.B.** 1338 (one thousand three hundred thirty-eight) with substitute.
- **S.B. 1387** (one thousand three hundred eighty-seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

- **S.B. 1405** (one thousand four hundred five) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1421** (one thousand four hundred twenty-one).
 - **S.B. 1427** (one thousand four hundred twenty-seven) with substitute.
 - S.B. 1445 (one thousand four hundred forty-five) with substitute.

The following bill, having been considered by the committee in session, was recommended for rereferral by the Committee on Education and Health:

S.B. 1446 (one thousand four hundred forty-six) with the recommendation that it be rereferred to the Committee on the Judiciary.

The following bills, having been considered by the committee in session, were reported by Senator Barker from the Committee on General Laws and Technology:

- **S.B. 1103** (one thousand one hundred three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1271 (one thousand two hundred seventy-one).
- **S.B. 1279** (one thousand two hundred seventy-nine) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1296** (one thousand two hundred ninety-six).
- **S.B. 1305** (one thousand three hundred five) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1314** (one thousand three hundred fourteen) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1318** (one thousand three hundred eighteen) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B.** 1324 (one thousand three hundred twenty-four) with amendments.
 - **S.B.** 1343 (one thousand three hundred forty-three).
 - **S.B.** 1369 (one thousand three hundred sixty-nine).

The following bill, having been considered by the committee in session, was recommended for rereferral by the Committee on General Laws and Technology:

S.B. 1419 (one thousand four hundred nineteen) with the recommendation that it be rereferred to the Committee on Commerce and Labor.

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- **S.B. 1213** (one thousand two hundred thirteen).
- S.B. 1242 (one thousand two hundred forty-two).
- **S.B. 1256** (one thousand two hundred fifty-six).
- **S.B. 1336** (one thousand three hundred thirty-six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- S.B. 1103, S.B. 1133, S.B. 1221, S.B. 1227, S.B. 1279, S.B. 1302, S.B. 1305, S.B. 1313, S.B. 1314, S.B. 1318, S.B. 1336, S.B. 1387, and S.B. 1405 were rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1419** was rereferred to the Committee on Commerce and Labor.

S.B. 1446 was rereferred to the Committee on the Judiciary.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1459. A BILL to amend and reenact §§ 24.2-101, 24.2-404, 24.2-411.3, 24.2-643, 24.2-653, 24.2-653.01, 24.2-701, 24.2-701.1, 24.2-702.1, 24.2-706, 24.2-707, 24.2-709.1, and 24.2-712 of the Code of Virginia, relating to conduct of elections, voter identification and absentee voting; process for applying for, casting, and counting absentee ballots.

Patron--Norment

Referred to Committee on Privileges and Elections

S.B. 1460. A BILL to amend and reenact § 29.1-519 of the Code of Virginia, relating to deer hunting; rifle caliber.

Patron--Lewis

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1461. A BILL to amend the Code of Virginia by adding a section numbered 18.2-474.2, relating to bribery in correctional facilities; penalty.

Patron--Lewis

Referred to Committee on the Judiciary

S.B. 1462. A BILL to amend the Code of Virginia by adding in Chapter 8 of Title 63.2 a section numbered 63.2-806, relating to Virginia Digital Equity Pilot Program and Fund.

Patron--Mason

Referred to Committee on Rehabilitation and Social Services

S.B. 1463. A BILL to amend and reenact § 67-701 of the Code of Virginia, relating to covenants regarding solar power; planned unit developments.

Patron--Cosgrove

Referred to Committee on General Laws and Technology

S.B. 1464. A BILL to amend and reenact § 54.1-3446 of the Code of Virginia, relating to Drug Control Act; Schedule I.

Patron--Newman

Referred to Committee on Education and Health

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 321. Commending Wolf Trap Foundation for the Performing Arts.

Patrons--Howell, Barker, Bell, Boysko, Marsden, Petersen, Saslaw and Surovell; Delegates: Bulova, Guzman, Helmer, Hope, Keam, Levine, Lopez, Plum, Reid, Roem, Samirah, Sickles, Simon, Sullivan, Tran and Watts

S.R. 94. Celebrating the life of Madison Boyd, Jr.

Patron--McDougle

CALENDAR

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1120 (one thousand one hundred twenty).
- S.B. 1152 (one thousand one hundred fifty-two).
- **S.B.** 1168 (one thousand one hundred sixty-eight).
- **S.B. 1216** (one thousand two hundred sixteen).
- **S.B. 1241** (one thousand two hundred forty-one).
- S.B. 1270 (one thousand two hundred seventy).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- S.B. 1120 (one thousand one hundred twenty).
- S.B. 1152 (one thousand one hundred fifty-two).
- **S.B.** 1168 (one thousand one hundred sixty-eight).
- **S.B. 1216** (one thousand two hundred sixteen).
- S.B. 1241 (one thousand two hundred forty-one).
- S.B. 1270 (one thousand two hundred seventy).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1171 (one thousand one hundred seventy-one) was read by title the third time and, on motion of Senator Norment, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Dunnavant, Edwards, Favola, Hanger, Lewis, Locke, Lucas, Marsden, Mason, McDougle, Morrissey, Newman, Norment, Peake, Pillion, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--24.

NAYS--Chase, Cosgrove, Deeds, DeSteph, Ebbin, Hashmi, Howell, Kiggans, McClellan, McPike, Obenshain, Petersen, Reeves, Stanley, Suetterlein--15.

RULE 36--0.

- **S.B. 1122** (one thousand one hundred twenty-two), on motion of Senator Stanley, was passed by for the day.
- **S.B. 1123** (one thousand one hundred twenty-three) was read by title the third time and, on motion of Senator Obenshain, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Edwards--1.

RULE 36--0.

S.B. 1128 (one thousand one hundred twenty-eight) was read by title the third time and, on motion of Senator Spruill, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1141 (one thousand one hundred forty-one) was read by title the third time and, on motion of Senator Cosgrove, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1157 (one thousand one hundred fifty-seven) was read by title the third time.

Senator Spruill moved that S.B. 1157 be passed with its title.

The question was put on passing **S.B. 1157** with its title.

The recorded vote is as follows:

YEAS--19. NAYS--19. RULE 36--0.

YEAS--Barker, Bell, Chase, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, McClellan, McPike, Morrissey, Saslaw, Spruill, Surovell--19.

NAYS--Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Mason, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--19.

RULE 36--0.

There being an equal division, the President cast his vote in accordance with Article V, Section 14, of the Constitution.

The President voted yea.

The final vote is as follows:

YEAS--20. NAYS--19. RULE 36--0.

S.B. 1157 was passed with its title.

S.B. 1181 (one thousand one hundred eighty-one) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

S.B. 1208 (one thousand two hundred eight) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1285 (one thousand two hundred eighty-five) was read by title the third time and, on motion of Senator Locke, was passed with its title.

The recorded vote is as follows: YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Chase, Obenshain--2.

RULE 36--0.

S.B. 1309 (one thousand three hundred nine) was read by title the third time and, on motion of Senator Ebbin, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

S.B. 1130 (one thousand one hundred thirty).

S.B. 1134 (one thousand one hundred thirty-four).

S.B. 1145 (one thousand one hundred forty-five).

S.B. 1155 (one thousand one hundred fifty-five).

S.B. 1156 (one thousand one hundred fifty-six).

S.B. 1255 (one thousand two hundred fifty-five).

The motion was agreed to.

S.B. 1130 (one thousand one hundred thirty) was taken up.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. Line 21, introduced, after *100* insert

percent

The reading of the amendment was waived.

On motion of Senator Reeves, the amendment was agreed to.

S.B. 1145 (one thousand one hundred forty-five) was taken up.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. Line 46, introduced, after Academic

strike

18142

insert

18412

The reading of the amendment was waived.

On motion of Senator Howell, the amendment was agreed to.

S.B. 1155 (one thousand one hundred fifty-five) was taken up.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. Line 85, introduced

strike

all of lines 85, 86, and 87

The reading of the amendment was waived.

On motion of Senator Howell, the amendment was agreed to.

S.B. 1255 (one thousand two hundred fifty-five) was taken up.

The following amendment proposed by the Committee on Commerce and Labor was offered:

1. Line 35, introduced, after order

insert

, not to exceed 120 days unless renewed by the Commission

The reading of the amendment was waived.

On motion of Senator Mason, the amendment was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

S.B. 1130 (one thousand one hundred thirty) as amended.

S.B. 1134 (one thousand one hundred thirty-four).

S.B. 1145 (one thousand one hundred forty-five) as amended.

S.B. 1155 (one thousand one hundred fifty-five) as amended.

S.B. 1156 (one thousand one hundred fifty-six).

S.B. 1255 (one thousand two hundred fifty-five) as amended.

S.B. 1142 (one thousand one hundred forty-two) was taken up.

Senator Ebbin offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 20-25 of the Code of Virginia, relating to persons who may celebrate rites of marriage; members of the General Assembly.

On motion of Senator Ebbin, the reading of the substitute was waived.

On motion of Senator Ebbin, the substitute was agreed to.

On motion of Senator Cosgrove, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1355** (one thousand three hundred fifty-five), on motion of Senator Hanger, was passed by for the day.
- S.B. 1182 (one thousand one hundred eighty-two), on motion of Senator Surovell, was passed by for the day.
- **S.B. 1202** (one thousand two hundred two) was read by title the second time and, on motion of Senator Petersen, was ordered to be engrossed and read by title the third time.
- **S.B. 1211** (one thousand two hundred eleven) was read by title the second time and, on motion of Senator Edwards, was ordered to be engrossed and read by title the third time.
 - S.B. 1219 (one thousand two hundred nineteen) was read by title the second time.

The following amendment proposed by the Committee on Commerce and Labor was offered:

1. Line 15, introduced, after *offer* insert *up to*

The reading of the amendment was waived.

On motion of Senator Favola, the amendment was agreed to.

On motion of Senator Favola, the bill was ordered to be engrossed and read by title the third time.

S.B. 1276 (one thousand two hundred seventy-six) was read by title the second time and, on motion of Senator McClellan, was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

S.B. 1111 (one thousand one hundred eleven).

S.B. 1112 (one thousand one hundred twelve).

S.B. 1158 (one thousand one hundred fifty-eight).

S.B. 1161 (one thousand one hundred sixty-one).

S.B. 1280 (one thousand two hundred eighty).

S.B. 1097 (one thousand ninety-seven).

- **S.B.** 1194 (one thousand one hundred ninety-four).
- S.B. 1199 (one thousand one hundred ninety-nine).
- S.B. 1239 (one thousand two hundred thirty-nine).
- S.B. 1274 (one thousand two hundred seventy-four).
- S.B. 1281 (one thousand two hundred eighty-one).
- S.B. 1290 (one thousand two hundred ninety).
- S.B. 1291 (one thousand two hundred ninety-one).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- S.B. 1111 (one thousand one hundred eleven).
- S.B. 1112 (one thousand one hundred twelve).
- S.B. 1158 (one thousand one hundred fifty-eight).
- **S.B. 1161** (one thousand one hundred sixty-one).
- S.B. 1280 (one thousand two hundred eighty).
- S.B. 1097 (one thousand ninety-seven).
- **S.B.** 1194 (one thousand one hundred ninety-four).
- S.B. 1199 (one thousand one hundred ninety-nine).
- **S.B. 1239** (one thousand two hundred thirty-nine).
- S.B. 1274 (one thousand two hundred seventy-four).
- S.B. 1281 (one thousand two hundred eighty-one).
- S.B. 1290 (one thousand two hundred ninety).
- S.B. 1291 (one thousand two hundred ninety-one).

SENATE JOINT RESOLUTIONS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate joint resolutions be waived:

- S.J.R. 298 (two hundred ninety-eight).
- S.J.R. 299 (two hundred ninety-nine).
- S.J.R. 300 (three hundred).
- S.J.R. 301 (three hundred one).
- S.J.R. 302 (three hundred two).
- S.J.R. 303 (three hundred three).

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0. RULE 36--0.

On motion of Senator Saslaw, the following Senate joint resolutions were passed by for the day:

```
S.J.R. 298 (two hundred ninety-eight). S.J.R. 299 (two hundred ninety-nine). S.J.R. 300 (three hundred). S.J.R. 301 (three hundred one). S.J.R. 302 (three hundred two). S.J.R. 303 (three hundred three).
```

SENATE RESOLUTION ON FIRST READING

S.R. 91 (ninety-one), on motion of Senator Bell, was passed by for the day.

MEMORIAL RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following House joint resolutions were taken up and agreed to en bloc by a unanimous standing vote:

```
H.J.R. 517 (five hundred seventeen).
H.J.R. 518 (five hundred eighteen).
H.J.R. 519 (five hundred nineteen).
H.J.R. 520 (five hundred twenty).
H.J.R. 533 (five hundred thirty-three).
H.J.R. 534 (five hundred thirty-four).
H.J.R. 571 (five hundred seventy-one).
H.J.R. 581 (five hundred eighty-one).
H.J.R. 585 (five hundred eighty-five).
H.J.R. 586 (five hundred eighty-six).
```

H.J.R. 565 (five hundred sixty-five) was taken up, as follows:

HOUSE JOINT RESOLUTION NO. 565

Celebrating the life of the Honorable Augustus Benton Chafin, Jr.

WHEREAS, the Honorable Augustus Benton Chafin, Jr., a passionate advocate for Southwest Virginia and an accomplished state legislator who earned bipartisan admiration for his commitment to the betterment of the Commonwealth as a whole, died on January 1, 2021; and

WHEREAS, a proud native of Russell County, Augustus Benton "Ben" Chafin, Jr., grew up on his family's farm and learned the value of hard work and responsibility at a young age; he worked at a local meat packing plant as a teenager, and he returned home most weekends while attending college to help on the farm; and

WHEREAS, a first-generation college student in his family, Ben Chafin received a bachelor's degree from East Tennessee State University and a law degree from the University of Richmond; and

WHEREAS, Ben Chafin established what is now Chafin Law Firm, P.C., in the 1980s and practiced law throughout Southwest Virginia for more than 30 years; in addition to his work as an attorney, he served on the board of First Bank & Trust Company and held close to his roots in agriculture as the owner and operator of a beef cattle farm; and

WHEREAS, desirous to be of further service to his community and to the Commonwealth, Ben Chafin ran for and was elected to the Virginia House of Delegates in 2013, becoming the first Republican to represent the Fourth District in more than 20 years; and

WHEREAS, Ben Chafin was subsequently elected to the Senate of Virginia during a special election in 2014 and ably served the residents of Southwest Virginia in the 38th District until the time of his passing; and

WHEREAS, Ben Chafin adapted to the unique challenges of representing a district that is geographically larger than some federal congressional districts, encompassing all of the Counties of Bland, Buchanan, Dickenson, Pulaski, Russell, and Tazewell; parts of the Counties of Montgomery, Smyth, and Wise; and the Cities of Norton and Radford; and

WHEREAS, during his tenure as a state lawmaker, Ben Chafin introduced and supported many pieces of important legislation to benefit all Virginians and provided his expertise to the Agriculture, Conservation and Natural Resources; Commerce and Labor; Education and Health; Judiciary; Privileges and Elections; and Rehabilitation and Social Services committees; and

WHEREAS, Ben Chafin was a staunch proponent of public education, responsible economic growth, Second Amendment rights, and access to health care for all Virginians; and

WHEREAS, Ben Chafin consistently supported job creation in his district, helping to establish the InvestSWVA partnership to promote the coalfields region as a future site of renewable energy infrastructure and high-technology business development; and

WHEREAS, Ben Chafin served the residents of Southwest Virginia and the entire Commonwealth with the utmost dedication, integrity, and distinction; and

WHEREAS, outside of his careers, Ben Chafin was an avid outdoorsman, who relished every opportunity to appreciate Southwest Virginia's natural splendor; he volunteered his time and leadership as former president of the Russell County Rotary Club; and he enjoyed fellowship and worship with the congregation of Gracewood Community Church in Lebanon; and

WHEREAS, Ben Chafin will be fondly remembered and greatly missed by his wife of 38 years, Lora; his children, Sophia, Audra, Augustus III, and their families; and numerous other family members, friends, and colleagues; now, therefore, be it

RESOLVED by the House of Delegates, the Senate concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Augustus Benton Chafin, Jr., a respected statesman and a champion for Southwest Virginia; and, be it

RESOLVED FURTHER, That the Clerk of the House of Delegates prepare a copy of this resolution for presentation to the family of Augustus Benton Chafin, Jr., as an expression of the General Assembly's respect for his memory.

H.J.R. 565, on motion of Senator Pillion, was agreed to by a unanimous standing vote.

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions and resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions and resolutions were taken up, ordered to be engrossed, and agreed to en bloc by a unanimous standing vote:

S.J.R. 311 (three hundred eleven).

S.J.R. 315 (three hundred fifteen).

S.J.R. 318 (three hundred eighteen).

S.R. 85 (eighty-five).

S.R. 92 (ninety-two).

S.J.R. 304 (three hundred four) was taken up, as follows:

SENATE JOINT RESOLUTION NO. 304

Celebrating the life of the Honorable Mamye E. BaCote.

WHEREAS, the Honorable Mamye E. BaCote, an inspirational educator and a highly admired public servant who represented the residents of Newport News and Hampton in the House of Delegates for more than a decade, died on December 14, 2020; and

WHEREAS, a native of Halifax County, Mamye BaCote received a bachelor's degree from Virginia Union University and a master's degree from what is now Hampton University; and

WHEREAS, while a student at Virginia Union University in 1960, Mamye BaCote played a significant role in the civil rights movement as a member of the Richmond 34, a group of students and activists who staged a sit-in at the Thalhimers lunch counter in downtown Richmond; and

WHEREAS, Mamye BaCote pursued a career in education with Newport News Public Schools, teaching social studies at Collis P. Huntington High School and Menchville High School, where she also became a department head; and

WHEREAS, Mamye BaCote fostered trust among her students and worked tirelessly to help them achieve success in and out of the classroom; she also inspired young adults as an adjunct professor of political science at Hampton University; and

WHEREAS, after her retirement as a teacher in 1994, Mamye BaCote was elected to the Newport News City Council and served the city in that capacity from 1996 to 2003; and

WHEREAS, desirous to be of further service to the Commonwealth, Mamye BaCote ran for and was elected to the House of Delegates in 2003 and represented the residents of the 95th District until January 2016; and

WHEREAS, Mamye BaCote introduced and supported numerous important pieces of legislation to benefit all Virginians and offered her leadership and expertise to several standing committees, including Appropriations; Health, Welfare and Institutions; and Transportation; and

WHEREAS, during her tenure as a state lawmaker, Mamye BaCote focused on education and support for public schools, gun safety measures, and helping to rehabilitate nonviolent drug offenders by funding the Newport News Drug Court; and

WHEREAS, Mamye BaCote worked diligently to cultivate mutual respect across party lines, providing both Democrats and Republicans alike with her wisdom and insights, which helped to build bipartisan consensus on critical issues; and

WHEREAS, Mamye BaCote was well-known among constituents and colleagues for her grace, dignity, and genuine kindness, but her soft-spoken nature belied her tenacity and commitment to the public good; and

WHEREAS, Mamye BaCote served the Commonwealth with the utmost integrity, dedication, and distinction and inspired other members of her community to pursue a life of public service; and

WHEREAS, Mamye BaCote enjoyed fellowship and worship with the congregation of Saint Vincent de Paul Catholic Church and shared her faith with the community through song as a member of the choir; and

WHEREAS, predeceased by her husband, Theodore, and one son, Kyle, Mamye BaCote will be fondly remembered and greatly missed by her sons, Theodore III, Derek, and Marlon, and their families; numerous other family members and friends, and colleagues on both sides of the aisle; now, therefore, be it

RESOLVED by the Senate, the House of Delegates concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Mamye E. BaCote, a respected public servant and former member of the House of Delegates; and, be it

RESOLVED FURTHER, That the Clerk of the Senate prepare a copy of this resolution for presentation to the family of the Honorable Mamye E. BaCote as an expression of the General Assembly's respect for her memory.

- **S.J.R. 304**, on motion of Senator Locke, was ordered to be engrossed and was agreed to by a unanimous standing vote.
 - S.J.R. 305 (three hundred five), on motion of Senator Locke, was passed by for the day.
 - S.J.R. 316 (three hundred sixteen), on motion of Senator Deeds, was passed by for the day.
 - S.J.R. 317 (three hundred seventeen), on motion of Senator Deeds, was passed by for the day.

COMMENDING RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following House joint resolutions were taken up and agreed to en bloc:

```
H.J.R. 512 (five hundred twelve).
```

H.J.R. 521 (five hundred twenty-one).

H.J.R. 532 (five hundred thirty-two).

H.J.R. 535 (five hundred thirty-five).

H.J.R. 540 (five hundred forty).

H.J.R. 544 (five hundred forty-four).

H.J.R. 547 (five hundred forty-seven).

H.J.R. 553 (five hundred fifty-three).

H.J.R. 554 (five hundred fifty-four).

H.J.R. 570 (five hundred seventy).

H.J.R. 580 (five hundred eighty).

H.J.R. 584 (five hundred eighty-four).

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions were taken up, ordered to be engrossed, and agreed to en bloc:

```
S.J.R. 295 (two hundred ninety-five).
```

S.J.R. 296 (two hundred ninety-six).

S.J.R. 306 (three hundred six).

S.J.R. 307 (three hundred seven).

S.J.R. 312 (three hundred twelve).

S.J.R. 313 (three hundred thirteen).

S.J.R. 314 (three hundred fourteen).

HONORARY ADJOURNMENT

Senator Cosgrove addressed the Senate in memory of former Delegate Mamye E. BaCote.

Senator Cosgrove requested that when the Senate adjourns today, it adjourn in memory of former Delegate Mamye E. BaCote.

On motion of Senator Lucas, the Senate, in memory of former Delegate Mamye E. BaCote, adjourned until tomorrow at 12 m. The Clerk was ordered to receive the committee report.

COMMITTEE REPORT

The following bill, having been considered by the committee in session, was reported by Senator Lewis from the Committee on Local Government on January 18, 2021:

S.B. 1399 (one thousand three hundred ninety-nine).

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

FRIDAY, JANUARY 22, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable R. Creigh Deeds, Twenty-fifth Senatorial District, offered the following prayer:

Our Heavenly Father, we are in a period of great consequence. Please bring our country together and bless its leaders. Give us a sense of hope and revival and a chance to do Your will and bring greater understanding and reconciliation to Your people. We ask that You watch over us and help us to understand that whatever power each of us has is derived from the people we represent. Help us remember that even though we come to this place from forty different points of view and forty different places we are elected to work together. That we are much stronger as one than as forty. Help us look for solutions instead of spotlights. Help us to work together to achieve hope and a better life and a path forward for the people we represent. Please bless richly those who enable us to do this great work—our aides, the staffs of our Clerk's Office, the staff of the Finance Committee, and Legislative Services. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Tara H. Perkinson, Chief Deputy Clerk, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell.

A quorum was present.

On motion of Senator Peake, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--33. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell--33.

NAYS--Deeds, DeSteph, McDougle, Petersen, Stanley--5. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 21, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

H.B. 1774. A BILL to amend and reenact § 58.1-3506 of the Code of Virginia, relating to tangible personal property taxes; classification of certain motor vehicles, trailers, and semitrailers.

- **H.B. 1798.** A BILL to amend and reenact § 22.1-32 of the Code of Virginia, relating to Brunswick County school board; appointed school board salaries.
- H.B. 1823. A BILL to amend the Code of Virginia by adding a section numbered 22.1-138.2, by adding in Article 8 of Chapter 14.1 of Title 22.1 a section numbered 22.1-289.058, and by adding a section numbered 63.2-1705.2, relating to public schools, child day programs, and certain other programs; carbon monoxide detectors required.
- **H.B. 1904.** A BILL to amend and reenact §§ 22.1-253.13:5 and 22.1-298.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-298.7, relating to teachers and other licensed school board employees; cultural competency.
- **H.B. 2031.** A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 17 of Title 15.2 a section numbered 15.2-1723.2 and by adding a section numbered 23.1-815.1, relating to facial recognition technology; authorization of use by local law-enforcement agencies and public institutions of higher education.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--34. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--34.

NAYS--Deeds, DeSteph, McDougle, Petersen--4. RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- **H.B.** 1774 was referred to the Committee on Finance and Appropriations.
- H.B. 1798, H.B. 1823, and H.B. 1904 were referred to the Committee on Education and Health.
- H.B. 2031 was referred to the Committee on General Laws and Technology.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Favola from the Committee on Rehabilitation and Social Services:

- **S.B. 1185** (one thousand one hundred eighty-five) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1297** (one thousand two hundred ninety-seven).
- **S.B. 1300** (one thousand three hundred) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1321** (one thousand three hundred twenty-one) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B.** 1366 (one thousand three hundred sixty-six) with substitute.
- **S.B. 1406** (one thousand four hundred six) with substitute with the recommendation that it be rereferred to the Committee on the Judiciary.

The following bills, having been considered by the committee in session, were reported by Senator Marsden from the Committee on Transportation:

- S.B. 1098 (one thousand ninety-eight).
- **S.B.** 1136 (one thousand one hundred thirty-six) with amendment.
- **S.B. 1212** (one thousand two hundred twelve) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1214** (one thousand two hundred fourteen).
 - S.B. 1229 (one thousand two hundred twenty-nine) with substitute.
 - S.B. 1253 (one thousand two hundred fifty-three).
 - **S.B. 1263** (one thousand two hundred sixty-three).
 - S.B. 1277 (one thousand two hundred seventy-seven).
 - **S.B.** 1329 (one thousand three hundred twenty-nine).
 - **S.B.** 1335 (one thousand three hundred thirty-five).
- **S.B.** 1350 (one thousand three hundred fifty) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills and joint resolutions, having been considered by the committee in session, were reported by Senator Locke from the Committee on Rules:

- **S.B.** 1126 (one thousand one hundred twenty-six).
- **S.B.** 1172 (one thousand one hundred seventy-two).
- S.B. 1408 (one thousand four hundred eight).
- S.B. 1414 (one thousand four hundred fourteen).
- **S.J.R. 276** (two hundred seventy-six).
- S.J.R. 285 (two hundred eighty-five).
- S.J.R. 286 (two hundred eighty-six).
- S.J.R. 292 (two hundred ninety-two).
- S.J.R. 294 (two hundred ninety-four) with amendments.
- S.J.R. 308 (three hundred eight) with amendments.
- S.B. 1185, S.B. 1212, S.B. 1300, S.B. 1321, and S.B. 1350 were rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1406** was rereferred to the Committee on the Judiciary.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented, ordered to be printed, and referred under Senate Rule 11 (b):

S.B. 1465. A BILL to amend and reenact § 18.2-325, as it is currently effective and as it shall become effective, of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-331.1, relating to illegal gambling; skills games; civil penalty; enforcement by localities and Attorney General.

Patrons--Reeves and Morrissey

Referred to Committee on the Judiciary

S.B. 1466. A BILL to amend and reenact § 15.2-915 of the Code of Virginia, relating to local control of firearms; concealed handgun permit.

Patron--Obenshain

Referred to Committee on the Judiciary

S.B. 1467. A BILL to amend and reenact § 28.2-409 of the Code of Virginia, relating to menhaden fishing in territorial waters prohibited.

Patron--DeSteph

Referred to Committee on Agriculture, Conservation and Natural Resources

S.B. 1468. A BILL to amend the Code of Virginia by adding in Title 9.1 a chapter numbered 15, consisting of sections numbered 9.1-1500, 9.1-1501, and 9.1-1502, relating to certifications for victims of qualifying criminal activity.

Patron--Surovell

Referred to Committee on the Judiciary

S.B. 1469. A BILL to amend and reenact § 2.2-212 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-507.3, by adding in Chapter 22 of Title 2.2 an article numbered 12, consisting of sections numbered 2.2-2365 through 2.2-2376, and by adding in Article 3.1 of Chapter 1 of Title 51.1 a section numbered 51.1-124.40, relating to establishing an Opioid Abatement Authority.

Patron--Barker

Referred to Committee on General Laws and Technology

S.B. 1470. A BILL to amend and reenact § 46.2-600.1 of the Code of Virginia, relating to vehicle registration; special communication needs indicator.

Patron--Barker

Referred to Committee on Transportation

S.B. 1471. A BILL to amend and reenact §§ 4.1-128 and 4.1-308 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 4.1-128.1 and by adding in Article 1 of Chapter 9 of Title 15.2 a section numbered 15.2-926.5, relating to alcoholic beverage control; local outdoor refreshment areas.

Patron--Dunnavant

Referred to Committee on Rehabilitation and Social Services

S.B. 1472. A BILL to require the Department of Medical Assistance Services to establish a work group to study options for increasing access to virtual supports and services for individuals with intellectual and developmental disabilities.

Patron--Suetterlein

Referred to Committee on Rehabilitation and Social Services

S.B. 1473. A BILL to amend and reenact § 30-343 of the Code of Virginia, relating to the Health Insurance Reform Commission; mandated health insurance benefit or provider.

Patron--Saslaw

Referred to Committee on Rules

S.B. 1474. A BILL to amend and reenact the second enactment of Chapter 342 of the Acts of Assembly of 2017 and the third enactment of Chapter 362 of the Acts of Assembly of 2017, relating to nonrepairable and rebuilt vehicles.

Patron--Newman

Referred to Committee on Transportation

S.B. 1475. A BILL to amend and reenact § 19.2-56 of the Code of Virginia, relating to execution of search warrants.

Patron--Stuart

Referred to Committee on the Judiciary

S.J.R. 322. Requesting the Secretary of Administration to oversee and develop a charter and directives for the State Board of Elections to form a working group to study implementation of electronic return of voted military-overseas ballots. Report.

Patron--DeSteph

Referred to Committee on General Laws and Technology

S.J.R. 323. Designating June 19 through the third Monday in July, in 2021 and in each succeeding year, as Liberty Amendments Month in Virginia.

Patron--Locke

Referred to Committee on Rules

S.R. 95. Supporting the establishment of a commemorative commission to honor Booker T. Washington with a statue in the Old Senate Chamber.

Patron--Suetterlein

Referred to Committee on Rules

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 324. Celebrating the life of Edward S. Garcia.

Patron--Cosgrove

S.J.R. 325. Commending Harold J. Roesch II.

Patron--McPike

S.J.R. 326. Celebrating the life of David Anthony Rice.

Patrons--Reeves, Peake and Ruff; Delegates: Cole, M.L., Coyner, Fowler and Orrock

S.R. 96. Commending Elizabeth Stamoulis Via-Gossman.

Patron--McPike

S.R. 97. Celebrating the life of Gloria Mae Benton Walker.

Patron--Lucas

S.R. 98. Celebrating the life of Mary Holley.

Patron--Lucas

CALENDAR

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1130 (one thousand one hundred thirty).
- S.B. 1134 (one thousand one hundred thirty-four).
- S.B. 1145 (one thousand one hundred forty-five).
- S.B. 1155 (one thousand one hundred fifty-five).
- S.B. 1156 (one thousand one hundred fifty-six).
- S.B. 1255 (one thousand two hundred fifty-five).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- **S.B.** 1130 (one thousand one hundred thirty).
- **S.B.** 1134 (one thousand one hundred thirty-four).
- S.B. 1145 (one thousand one hundred forty-five).
- S.B. 1155 (one thousand one hundred fifty-five).
- **S.B.** 1156 (one thousand one hundred fifty-six).

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell--37.

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator Suetterlein stated that he was recorded as not voting on the question of the passage en bloc of S.B. 1130, S.B. 1134, S.B. 1145, S.B. 1155, and S.B. 1156. He intended to vote yea on the question of the passage en bloc of S.B. 1130, S.B. 1134, S.B. 1145, and S.B. 1155. He intended to vote nay on the question of the passage en bloc of S.B. 1156.

S.B. 1255 (one thousand two hundred fifty-five), on motion of Senator Mason, was passed with its title.

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

- **S.B. 1122** (one thousand one hundred twenty-two), on motion of Senator Stanley, was passed by for the day.
- **S.B. 1142** (one thousand one hundred forty-two) was read by title the third time and, on motion of Senator Cosgrove, was passed with its title.

The recorded vote is as follows:

YEAS--33. NAYS--4. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell--33.

NAYS--Chase, Deeds, Petersen, Suetterlein--4.

RULE 36--Stuart--1.

S.B. 1202 (one thousand two hundred two) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--29. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Petersen, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--29.

NAYS--Chase, Cosgrove, DeSteph, Kiggans, Newman, Peake, Pillion, Reeves, Ruff--9. RULE 36--0.

S.B. 1211 (one thousand two hundred eleven) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--7. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell--31.

NAYS--Chase, Hanger, McDougle, Newman, Obenshain, Stuart, Suetterlein--7.

RULE 36--0.

S.B. 1219 (one thousand two hundred nineteen) was read by title the third time and, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--24.

NAYS--Chase, Cosgrove, DeSteph, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--13.

RULE 36--0.

STATEMENT ON VOTE

Senator McDougle stated that he was recorded as not voting on the question of the passage of **S.B.** 1219, whereas he intended to vote nay.

S.B. 1276 (one thousand two hundred seventy-six) was read by title the third time and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Morrissey, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--17.

RULE 36--0.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

S.B. 1111 (one thousand one hundred eleven).

S.B. 1112 (one thousand one hundred twelve).

S.B. 1158 (one thousand one hundred fifty-eight).

S.B. 1161 (one thousand one hundred sixty-one).

S.B. 1280 (one thousand two hundred eighty).

The motion was agreed to.

S.B. 1112 (one thousand one hundred twelve) was taken up.

The following amendments proposed by the Committee on Finance and Appropriations were offered:

```
1. Line 155, introduced, after 58.1-320 strike

or
insert
, [a comma]

2. Line 155, introduced, after 58.1-400 insert
```

, or 58.1-1202

The reading of the amendments was waived.

On motion of Senator Locke, the amendments were agreed to.

S.B. 1161 (one thousand one hundred sixty-one) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 10.1-502 of the Code of Virginia, relating to Soil and Water Conservation Board; membership.

The reading of the substitute was waived.

On motion of Senator Hanger, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1111 (one thousand one hundred eleven).
- S.B. 1112 (one thousand one hundred twelve) as amended.
- S.B. 1158 (one thousand one hundred fifty-eight).
- S.B. 1161 (one thousand one hundred sixty-one) as amended.
- S.B. 1280 (one thousand two hundred eighty).
- **S.B. 1355** (one thousand three hundred fifty-five), on motion of Senator Hanger, was passed by for the day.
- **S.B. 1182** (one thousand one hundred eighty-two) was read by title the second time and, on motion of Senator Surovell, was ordered to be engrossed and read by title the third time.
- **S.B. 1097** (one thousand ninety-seven) was read by title the second time and, on motion of Senator Favola, was ordered to be engrossed and read by title the third time.
- **S.B. 1194** (one thousand one hundred ninety-four) was read by title the second time and, on motion of Senator Obenshain, was ordered to be engrossed and read by title the third time.
 - S.B. 1199 (one thousand one hundred ninety-nine) was taken up.

PARLIAMENTARY INQUIRY

Senator Petersen propounded a parliamentary inquiry as to whether the committee amendments to **S.B. 1199** could be agreed to before taking up the substitute offered by Senator Stuart to **S.B. 1199**.

The Chair stated that the committee amendments to **S.B. 1199** could not be agreed to before taking up the substitute offered by Senator Stuart to **S.B. 1199**.

- S.B. 1199, on motion of Senator Petersen, was passed by for the day.
- **S.B. 1239** (one thousand two hundred thirty-nine) was read by title the second time and, on motion of Senator Bell, was ordered to be engrossed and read by title the third time.
 - S.B. 1274 (one thousand two hundred seventy-four) was read by title the second time.

The following amendments proposed by the Committee on Agriculture, Conservation and Natural Resources were offered:

1. Line 75, introduced, after action.

insert

The Department shall publish the plan and any subsequent updates on its website

2. After line 104, introduced

insert

2. That in the development of the Virginia Wildlife Action Plan, or any successor publication, the Department of Wildlife Resources shall consider and incorporate, where applicable, wildlife corridors and any recommendation of the Wildlife Corridor Action Plan developed pursuant to § 29.1-579 of the Code of Virginia, as amended by this act.

The reading of the amendments was waived.

On motion of Senator Marsden, the amendments were agreed to.

On motion of Senator Marsden, the bill was ordered to be engrossed and read by title the third time.

S.B. 1281 (one thousand two hundred eighty-one) was read by title the second time.

The following amendments proposed by the Committee on Privileges and Elections were offered:

1. Line 15, introduced, after the

strike

the remainder of line 15 through population of on line 16

insert

county or city for which he is appointed unless such county or city has a population of

2. Line 16, introduced, after 25,000

insert

50,000

3. Line 16, introduced, after 25,000

strike

the remainder of line 16, all of line 17, and through registrar on line 18

insert

or less. In the case of a city that is wholly contained within one county, the city electoral board may appoint a qualified voter of that county to serve as city general registrar

4. Line 18, introduced, after registrar

strike

Commonwealth

The reading of the amendments was waived.

On motion of Senator Morrissey, the amendments were agreed to.

On motion of Senator Morrissey, the bill was ordered to be engrossed and read by title the third time.

S.B. 1290 (one thousand two hundred ninety) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 10.1-1018.1 and 10.1-1021 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 1 of Title 10.1 a section numbered 10.1-104.6:1, relating to ConserveVirginia program; established.

The reading of the substitute was waived.

On motion of Senator Mason, the substitute was agreed to.

On motion of Senator Mason, the bill was ordered to be engrossed and read by title the third time.

S.B. 1291 (one thousand two hundred ninety-one) was read by title the second time.

The following amendments proposed by the Committee on Agriculture, Conservation and Natural Resources were offered:

1. Line 22, introduced, after *plan and* strike

a

2. Line 67, introduced, after act.

strike

the remainder of line 67 and all of lines 68 through 81

The reading of the amendments was waived.

On motion of Senator Mason, the amendments were agreed to.

On motion of Senator Mason, the bill was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1154 (one thousand one hundred fifty-four).
- S.B. 1169 (one thousand one hundred sixty-nine).
- S.B. 1187 (one thousand one hundred eighty-seven).
- S.B. 1189 (one thousand one hundred eighty-nine).
- S.B. 1205 (one thousand two hundred five).
- S.B. 1213 (one thousand two hundred thirteen).
- S.B. 1242 (one thousand two hundred forty-two).
- S.B. 1271 (one thousand two hundred seventy-one).
- S.B. 1296 (one thousand two hundred ninety-six).
- S.B. 1320 (one thousand three hundred twenty).
- S.B. 1324 (one thousand three hundred twenty-four).
- S.B. 1338 (one thousand three hundred thirty-eight).
- **S.B. 1369** (one thousand three hundred sixty-nine).
- S.B. 1399 (one thousand three hundred ninety-nine).
- **S.B. 1421** (one thousand four hundred twenty-one).
- S.B. 1427 (one thousand four hundred twenty-seven).
- **S.B. 1445** (one thousand four hundred forty-five).
- S.B. 1121 (one thousand one hundred twenty-one).
- S.B. 1220 (one thousand two hundred twenty).
- **S.B. 1256** (one thousand two hundred fifty-six).
- S.B. 1322 (one thousand three hundred twenty-two).
- S.B. 1343 (one thousand three hundred forty-three).

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--37.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- S.B. 1154 (one thousand one hundred fifty-four).
- S.B. 1169 (one thousand one hundred sixty-nine).
- S.B. 1187 (one thousand one hundred eighty-seven).
- S.B. 1189 (one thousand one hundred eighty-nine).
- S.B. 1205 (one thousand two hundred five).
- S.B. 1213 (one thousand two hundred thirteen).
- S.B. 1242 (one thousand two hundred forty-two).
- **S.B. 1271** (one thousand two hundred seventy-one).
- **S.B.** 1296 (one thousand two hundred ninety-six).
- **S.B. 1320** (one thousand three hundred twenty).

- S.B. 1324 (one thousand three hundred twenty-four).
- **S.B. 1338** (one thousand three hundred thirty-eight).
- S.B. 1369 (one thousand three hundred sixty-nine).
- S.B. 1399 (one thousand three hundred ninety-nine).
- S.B. 1421 (one thousand four hundred twenty-one).
- S.B. 1427 (one thousand four hundred twenty-seven).
- **S.B. 1121** (one thousand one hundred twenty-one).
- S.B. 1220 (one thousand two hundred twenty).
- S.B. 1256 (one thousand two hundred fifty-six).
- **S.B. 1322** (one thousand three hundred twenty-two).
- S.B. 1343 (one thousand three hundred forty-three).
- S.B. 1445 (one thousand four hundred forty-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to facilitate the administration of the COVID-19 vaccine; emergency.

The reading of the substitute was waived.

On motion of Senator Dunnavant, the substitute was agreed to.

Senator Mason offered the following amendments to the substitute:

1. After line 65, substitute

insert

4. § 1. That localities with fire departments or emergency medical services departments employing full-time, non-volunteer emergency medical technicians (EMT) or paramedics may establish and staff vaccine administration clinics for the purpose of administering COVID-19 vaccines. Vaccines shall be administered at such clinics only by EMTs or paramedics trained in the administration of vaccines and may be provided under the existing operating medical director (OMD) license for such local fire department or emergency medical services department. The Department of Health or hospitals serving the locality are authorized to provide vaccines to locality-created vaccine administration clinics upon the request of the locality, provided that such clinics meet the requirements under this act. In the absence of gross negligence or willful misconduct, any locality and OMD overseeing or EMT or paramedic who administers the COVID-19 vaccine pursuant to this act shall not be liable for any actual or alleged injury or wrongful death or any civil cause of action arising from any act or omission arising out of, related to, or alleged to have resulted from the administration of the COVID-19 vaccine.

2. At the beginning of line 66, substitute

strike

4.

insert

5.

On motion of Senator Mason, the reading of the amendments was waived.

On motion of Senator Mason, the amendments were agreed to.

On motion of Senator Dunnavant, the bill was ordered to be engrossed and read by title the third time.

Senator Dunnavant moved that the Rules be suspended and the third reading of the title of **S.B. 1445** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--1.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--36.

NAYS--0.

RULE 36--Howell--1.

S.B. 1445, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

SENATE JOINT RESOLUTIONS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate joint resolutions that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate joint resolutions, the titles of the joint resolutions having been printed in the Calendar for their second reading, be placed before the Senate by number only:

S.J.R. 298 (two hundred ninety-eight).

S.J.R. 299 (two hundred ninety-nine).

S.J.R. 300 (three hundred).

S.J.R. 301 (three hundred one).

S.J.R. 302 (three hundred two).

S.J.R. 303 (three hundred three).

The motion was agreed to.

S.J.R. 299 (two hundred ninety-nine) was taken up.

The following amendment proposed by the Committee on Privileges and Elections was offered:

1. Line 198, introduced, after $\mathbf{Marquez}$ of

strike

McLean

insert

Penhook

The reading of the amendment was waived.

On motion of Senator Deeds, the amendment was agreed to.

On motion of Senator Saslaw, the following Senate joint resolutions were ordered en bloc to be engrossed and read by title the third time:

S.J.R. 298 (two hundred ninety-eight).

S.J.R. 299 (two hundred ninety-nine) as amended.

S.J.R. 300 (three hundred).

S.J.R. 301 (three hundred one).

S.J.R. 302 (three hundred two).

S.J.R. 303 (three hundred three).

SENATE RESOLUTION ON FIRST READING

S.R. 91 (ninety-one), on motion of Senator Bell, was passed by for the day.

RECESS

At 1:35 p.m., Senator Saslaw moved that the Senate recess until 2:00 p.m.

The motion was agreed to.

The hour of 2:00 p.m. having arrived, the Chair was resumed.

On motion of Senator McDougle, a leave of absence for the day was granted Senator Vogel on account of pressing personal business.

Senator Lucas moved that the Senate adjourn until Monday, January 25, 2021, at 12 m. and that the Rules be suspended and, pursuant to the provisions of **H.J.R. 575** (five hundred seventy-five), the Clerk be directed to accept legislation to be introduced until 3:00 p.m. on Friday, January 22, 2021.

The motion was agreed to.

The recorded vote is as follows: YEAS--27. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Obenshain, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--27.

NAYS--0. RULE 36--0.

> Justin E. Fairfax President of the Senate

Jacke Solmon

Susan Clarke Schaar Clerk of the Senate

MONDAY, JANUARY 25, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Bryce E. Reeves, Seventeenth Senatorial District, offered the following prayer:

Before we go to prayer let me do a quick reading from our Bible and we are going to be in Psalms. Psalms 86:15 says "but You, O Lord, are a God merciful and gracious, slow to anger, and abounding in steadfast love and faithfulness." Let us go.

Lord, I just thank You for this day that You have brought before us. Your word tells us that You are the Alpha, the Omega, the beginning, and the end. Lord, I humbly ask for the grace to find the space between my impulse and my action as Your word just told us in the Psalms. Lord, to let the flow of the cooling breeze when I would respond with heat. Lord, to interrupt the fierceness sometimes that I have, with gentleness and to accept the moment that allows judgment to become discernment. To defer to silence when my tongue would rush to attack or defend. Father, we take on a lot of issues here, Lord, but let us all remember that we are one body of one accord. Father, Your word also tells us that a strand of three cords seldom breaks. Father, I thank You for what You have done in my life, my family's life, and I ask You to put a hedge of protection around all those who gather here today and their families as we are away. Father, give those who live the farthest away traveling mercy as they go home today. I thank You for what You do and what You continue to do for us each and every day. In Your Son's most precious and holy name I pray, Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Michael Adams, Director of Human Resources, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Petersen, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--36, NAYS--2, RULE 36--1.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Deeds, DeSteph--2. RULE 36--Petersen--1.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 22, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1824.** A BILL to amend and reenact § 55.1-703 of the Code of Virginia, relating to the Virginia Residential Property Disclosure Act; required disclosures for buyer to beware; mold.
- **H.B. 1830.** A BILL to amend and reenact § 2.2-2282 of the Code of Virginia, relating to the Virginia Small Business Financing Authority; Board of Directors; membership; small business lending experience.
- **H.B. 1848.** A BILL to amend and reenact §§ 2.2-3902, 2.2-3905, and 51.5-41 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-3905.1, relating to the Virginia Human Rights Act; discrimination on the basis of disability.
- **H.B. 1851.** A BILL to amend and reenact § 5.1-5 of the Code of Virginia, relating to aircraft registration; unmanned aircraft.
- **H.B. 1900.** A BILL to amend the Code of Virginia by adding a section numbered 55.1-1243.1 and to repeal § 55.1-1243 of the Code of Virginia, relating to Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion from dwelling unit, interruption of services, or actions taken to make premises unsafe.
- **H.B. 1923.** A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to electric utilities; broadband capacity pilot program.
- **H.B. 1940.** A BILL to amend and reenact § 22.1-254 of the Code of Virginia, relating to the Department of Education; guidelines on excused student absences; civic or political engagement.
- **H.B. 1942.** A BILL to amend and reenact §§ 38.2-1866, 38.2-1867, 38.2-1868.1, 38.2-1869, 38.2-1871, and 38.2-1873 of the Code of Virginia and to repeal § 38.2-1845.9 of the Code of Virginia, relating to public adjusters; continuing education.
- **H.B. 1964.** A BILL to amend and reenact § 6.2-1317 of the Code of Virginia and to repeal § 6.2-1318 of the Code of Virginia, relating to State Corporation Commission; supervisory merger or transfer of assets of financially unstable credit union.

- **H.B. 1967.** A BILL to amend and reenact § 2.2-2240.3 of the Code of Virginia, relating to the Virginia Jobs Investment Program and Fund; minimum wage requirements.
- **H.B. 1994.** A BILL to amend and reenact § 56-594.2 of the Code of Virginia, relating to small agricultural generators; definition.
- **H.B. 2034.** A BILL to amend and reenact § 1 of the first enactment of Chapters 358 and 382 of the Acts of Assembly of 2013, as amended by Chapter 803 of the Acts of Assembly of 2017, and as amended by Chapters 1187, 1188, 1189, 1193, 1194, and 1239 of the Acts of Assembly of 2020, relating to electric utilities; nonjurisdictional customers; third party power purchase agreements.
- **H.B. 2130.** A BILL to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29, consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, relating to establishment of the Virginia LGBTQ+ Advisory Board.
- **H.B. 2170.** A BILL to amend the Code of Virginia by adding a section numbered 2.2-2312.1, relating to the Virginia Small Business Financing Authority; risk-based review of outstanding loans; report.
- **H.B. 2171.** A BILL to amend and reenact § 2.2-2312 of the Code of Virginia, relating to the Virginia Small Business Financing Authority; annual report; utilization or award of loan and grant program funds.
- **H.B. 2172.** A BILL to amend and reenact § 2.2-1606 of the Code of Virginia, relating to the Department of Small Business and Supplier Diversity; certification of small, women-owned, and minority-owned businesses; right to appeal denial of initial certification.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, DeSteph, McDougle, Petersen--4.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1824, H.B. 1830, H.B. 1848, H.B. 1900, H.B. 1967, H.B. 2130, H.B. 2170, H.B. 2171, and H.B. 2172 were referred to the Committee on General Laws and Technology.
 - H.B. 1851 was referred to the Committee on Transportation.

- H.B. 1923, H.B. 1942, H.B. 1964, H.B. 1994, and H.B. 2034 were referred to the Committee on Commerce and Labor.
 - H.B. 1940 was referred to the Committee on Education and Health.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- S.B. 1206 (one thousand two hundred six) with substitute.
- **S.B. 1226** (one thousand two hundred twenty-six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1248 (one thousand two hundred forty-eight) with amendment.
- **S.B. 1250** (one thousand two hundred fifty) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1261** (one thousand two hundred sixty-one) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1262** (one thousand two hundred sixty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1391** (one thousand three hundred ninety-one) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1400 (one thousand four hundred).
- **S.B. 1442** (one thousand four hundred forty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1443** (one thousand four hundred forty-three) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills, having been considered by the committee in session, were reported by Senator Lewis from the Committee on Local Government:

- S.B. 1393 (one thousand three hundred ninety-three) with substitute.
- S.B. 1447 (one thousand four hundred forty-seven).
- S.B. 1457 (one thousand four hundred fifty-seven) with amendments.
- **S.B. 1226, S.B. 1250, S.B. 1261, S.B. 1262, S.B. 1391, S.B. 1442,** and **S.B. 1443** were rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

Pursuant to the provisions of House Joint Resolution No. 575 and Senate Rule 11 (b), Senator Deeds introduced a joint resolution; subsequently, the following was presented, ordered to be printed, and referred:

S.J.R. 327. Confirming appointments by the Governor of certain persons communicated to the General Assembly January 24, 2021.

Patron--Deeds

Referred to Committee on Privileges and Elections

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 328. Celebrating the life of Roger Hamilton Brown.

Patrons--Howell, Barker and Boysko; Delegates: Helmer, Plum, Simon and Watts

S.J.R. 329. Celebrating the life of Joseph Richard Stowers.

Patrons--Howell, Barker, Boysko and Saslaw; Delegates: Helmer, Plum, Simon, Sullivan and Watts

CALENDAR

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1111 (one thousand one hundred eleven).
- S.B. 1112 (one thousand one hundred twelve).
- S.B. 1158 (one thousand one hundred fifty-eight).
- **S.B.** 1161 (one thousand one hundred sixty-one).
- S.B. 1280 (one thousand two hundred eighty).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- **S.B. 1111** (one thousand one hundred eleven).
- S.B. 1112 (one thousand one hundred twelve).
- S.B. 1158 (one thousand one hundred fifty-eight).
- **S.B. 1161** (one thousand one hundred sixty-one).
- S.B. 1280 (one thousand two hundred eighty).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1122 (one thousand one hundred twenty-two) was taken up.

RECONSIDERATION

Senator Stanley moved to reconsider the vote by which **S.B. 1122** (one thousand one hundred twenty-two) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Stanley offered the following amendment to the substitute:

1. After line 1042, substitute

insert

4. That the Virginia Alcohol and Safety Action Program (VASAP) shall be authorized to administer intervention interviews pursuant to former § 46.2-355.1 of the Code of Virginia for individuals who were ordered to attend an intervention interview on or before June 30, 2021. The Department of Motor Vehicles shall suspend the driving privileges of any person who fails to attend such intervention interview within 60 days of the date of such notice for an intervention interview, in accordance with former § 46.2-355.1 of the Code of Virginia.

On motion of Senator Stanley, the reading of the amendment was waived.

On motion of Senator Stanley, the amendment was agreed to.

On motion of Senator Stanley, the bill was ordered to be engrossed and read by title the third time.

Senator Stanley moved that the Rules be suspended and the third reading of the title of **S.B. 1122** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1122, on motion of Senator Stanley, was passed with its title.

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Petersen, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--26.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stuart--13.

RULE 36--0.

S.B. 1182 (one thousand one hundred eighty-two) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--11. RULE 36--1.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Petersen, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--27.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, Marsden, Newman, Peake, Pillion, Ruff, Vogel--11.

RULE 36--Reeves--1.

S.B. 1097 (one thousand ninety-seven) was read by title the third time and, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1194 (one thousand one hundred ninety-four) was read by title the third time and, on motion of Senator Obenshain, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

NAYS--McPike, Suetterlein--2.

RULE 36--0.

S.B. 1239 (one thousand two hundred thirty-nine) was read by title the third time and, on motion of Senator Bell, was passed with its title.

YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Reeves, Saslaw, Spruill, Surovell, Vogel--27.

NAYS--Chase, Cosgrove, Deeds, McDougle, Newman, Obenshain, Peake, Pillion, Ruff, Stanley, Stuart, Suetterlein--12.

RULE 36--0.

S.B. 1274 (one thousand two hundred seventy-four) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--29. NAYS--10. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--29.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Obenshain, Peake, Reeves, Ruff, Suetterlein--10.

RULE 36--0.

S.B. 1281 (one thousand two hundred eighty-one) was read by title the third time and, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Saslaw, Spruill, Surovell, Vogel--31.

NAYS--Chase, Cosgrove, McDougle, Obenshain, Ruff, Stanley, Stuart, Suetterlein--8. RULE 36--0.

S.B. 1290 (one thousand two hundred ninety) was read by title the third time and, on motion of Senator Mason, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1291 (one thousand two hundred ninety-one) was read by title the third time.

Senator Mason moved that S.B. 1291 be passed with its title.

S.B. 1291, on motion of Senator Newman, was passed by for the day.

SENATE BILLS ON SECOND READING

S.B. 1338 (one thousand three hundred thirty-eight), on motion of Senator Barker, was rereferred to the Committee on Finance and Appropriations.

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1154 (one thousand one hundred fifty-four).
- **S.B.** 1169 (one thousand one hundred sixty-nine).
- **S.B.** 1187 (one thousand one hundred eighty-seven).
- S.B. 1189 (one thousand one hundred eighty-nine).
- S.B. 1205 (one thousand two hundred five).
- S.B. 1213 (one thousand two hundred thirteen).
- S.B. 1242 (one thousand two hundred forty-two).
- S.B. 1271 (one thousand two hundred seventy-one).
- S.B. 1296 (one thousand two hundred ninety-six).
- **S.B. 1320** (one thousand three hundred twenty).
- **S.B.** 1324 (one thousand three hundred twenty-four).
- **S.B.** 1369 (one thousand three hundred sixty-nine).
- S.B. 1399 (one thousand three hundred ninety-nine).
- S.B. 1421 (one thousand four hundred twenty-one).
- S.B. 1427 (one thousand four hundred twenty-seven).

The motion was agreed to.

S.B. 1154 (one thousand one hundred fifty-four) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 37.2-304 of the Code of Virginia, relating to Commissioner of Behavioral Health and Developmental Services; reports to designated protection and advocacy system.

The reading of the substitute was waived.

On motion of Senator Favola, the substitute was agreed to.

S.B. 1189 (one thousand one hundred eighty-nine) was taken up.

The following amendment proposed by the Committee on Education and Health was offered:

1. Line 416, introduced, after *the* insert

grossly negligent,

The reading of the amendment was waived.

On motion of Senator Hashmi, the amendment was agreed to.

S.B. 1320 (one thousand three hundred twenty) was taken up.

The following amendment proposed by the Committee on Education and Health was offered:

1. Line 220, introduced, after *in* insert

consultation with a licensed physician in accordance with a practice agreement between the licensed certified midwife and the licensed physician. Such practice agreement shall address the availability of the physician for routine and urgent consultation on patient care. Evidence of a practice agreement shall be maintained by the licensed certified midwife and provided to the Board upon request. The Board shall adopt regulations for the practice of licensed certified midwives, which shall be in

The reading of the amendment was waived.

On motion of Senator Lucas, the amendment was agreed to.

S.B. 1324 (one thousand three hundred twenty-four) was taken up.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 20, introduced, after *within* strike

the remainder of line 20, all of line 21, and through than on line 22

2. Line 32, introduced, after *within* strike

the remainder of line 32, all of line 33, and through than on line 34

The reading of the amendments was waived.

On motion of Senator Dunnavant, the amendments were agreed to.

S.B. 1427 (one thousand four hundred twenty-seven) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding a section numbered 37.2-313.2, relating to Department of Behavioral Health and Developmental Services; Early Psychosis Intervention and Coordinated Specialty Care Advisory Board established.

The reading of the substitute was waived.

On motion of Senator Morrissey, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- **S.B.** 1154 (one thousand one hundred fifty-four) as amended.
- S.B. 1169 (one thousand one hundred sixty-nine).
- **S.B.** 1187 (one thousand one hundred eighty-seven).
- S.B. 1189 (one thousand one hundred eighty-nine) as amended.
- S.B. 1205 (one thousand two hundred five).
- **S.B. 1213** (one thousand two hundred thirteen).
- S.B. 1242 (one thousand two hundred forty-two).
- S.B. 1271 (one thousand two hundred seventy-one).
- **S.B. 1296** (one thousand two hundred ninety-six).
- **S.B. 1320** (one thousand three hundred twenty) as amended.
- S.B. 1324 (one thousand three hundred twenty-four) as amended.
- S.B. 1369 (one thousand three hundred sixty-nine).
- **S.B. 1399** (one thousand three hundred ninety-nine).
- S.B. 1421 (one thousand four hundred twenty-one).
- S.B. 1427 (one thousand four hundred twenty-seven) as amended.
- **S.B. 1355** (one thousand three hundred fifty-five), on motion of Senator Hanger, was passed by for the day.
 - S.B. 1199 (one thousand one hundred ninety-nine) was read by title the second time.

The following amendments proposed by the Committee on Agriculture, Conservation and Natural Resources were offered:

1. Line 17, introduced, after *be* strike

liberally

2. Line 20, introduced, after *be* strike

liberally

The reading of the amendments was waived.

Senator Petersen moved that the amendments be agreed to.

- S.B. 1199, on motion of Senator Peake, was passed by for the day.
- S.B. 1121 (one thousand one hundred twenty-one) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 32.1-269 of the Code of Virginia, relating to birth certificates; amendments.

The reading of the substitute was waived.

On motion of Senator Locke, the substitute was agreed to.

On motion of Senator Locke, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1220** (one thousand two hundred twenty) was read by title the second time and, on motion of Senator Favola, was ordered to be engrossed and read by title the third time.
- **S.B. 1256** (one thousand two hundred fifty-six) was read by title the second time and, on motion of Senator Marsden, was ordered to be engrossed and read by title the third time.
 - **S.B. 1322** (one thousand three hundred twenty-two) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 8.01-225 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-274.6, relating to public schools; seizure management and action plan; biennial training.

The reading of the substitute was waived.

On motion of Senator DeSteph, the substitute was agreed to.

On motion of Senator DeSteph, the bill was ordered to be engrossed and read by title the third time.

S.B. 1343 (one thousand three hundred forty-three) was read by title the second time and, on motion of Senator Vogel, was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1098 (one thousand ninety-eight).
- S.B. 1126 (one thousand one hundred twenty-six).
- **S.B.** 1172 (one thousand one hundred seventy-two).
- **S.B. 1214** (one thousand two hundred fourteen).
- S.B. 1253 (one thousand two hundred fifty-three).
- S.B. 1277 (one thousand two hundred seventy-seven).
- **S.B. 1297** (one thousand two hundred ninety-seven).
- S.B. 1335 (one thousand three hundred thirty-five).
- **S.B. 1408** (one thousand four hundred eight).
- **S.B. 1414** (one thousand four hundred fourteen).
- **S.B.** 1136 (one thousand one hundred thirty-six).
- S.B. 1229 (one thousand two hundred twenty-nine).
- S.B. 1263 (one thousand two hundred sixty-three).
- **S.B.** 1329 (one thousand three hundred twenty-nine).
- **S.B.** 1366 (one thousand three hundred sixty-six).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- S.B. 1098 (one thousand ninety-eight).
- **S.B.** 1126 (one thousand one hundred twenty-six).
- S.B. 1172 (one thousand one hundred seventy-two).
- **S.B. 1214** (one thousand two hundred fourteen).
- S.B. 1253 (one thousand two hundred fifty-three).
- **S.B. 1277** (one thousand two hundred seventy-seven).
- S.B. 1297 (one thousand two hundred ninety-seven).
- **S.B.** 1335 (one thousand three hundred thirty-five).
- **S.B. 1408** (one thousand four hundred eight).
- **S.B. 1414** (one thousand four hundred fourteen).
- **S.B.** 1136 (one thousand one hundred thirty-six).
- S.B. 1229 (one thousand two hundred twenty-nine).
- S.B. 1263 (one thousand two hundred sixty-three).
- S.B. 1329 (one thousand three hundred twenty-nine).
- S.B. 1366 (one thousand three hundred sixty-six).

SENATE JOINT RESOLUTIONS ON THIRD READING

Senator Saslaw moved that the following Senate joint resolutions, the titles of the joint resolutions having been printed in the Calendar for their third reading, be placed before the Senate by number only:

```
S.J.R. 298 (two hundred ninety-eight).
```

- S.J.R. 299 (two hundred ninety-nine).
- S.J.R. 300 (three hundred).
- S.J.R. 301 (three hundred one).
- S.J.R. 302 (three hundred two).
- S.J.R. 303 (three hundred three).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate joint resolutions that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate joint resolutions were agreed to en bloc:

S.J.R. 298 (two hundred ninety-eight).

S.J.R. 302 (three hundred two).

S.J.R. 303 (three hundred three).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.J.R. 299 (two hundred ninety-nine), on motion of Senator Ebbin, was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator Deeds stated that he was abstaining pursuant to Rule 36 on lines 21-22, but voting yea on **S.J.R. 299** as a whole.

S.J.R. 300 (three hundred), on motion of Senator Deeds, was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator McClellan stated that she was abstaining pursuant to Rule 36 on lines 259-260, but voting yea on **S.J.R. 300** as a whole.

S.J.R. 301 (three hundred one), on motion of Senator Deeds, was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator Hashmi stated that she was abstaining pursuant to Rule 36 on lines 171-172, but voting yea on **S.J.R. 301** as a whole.

SENATE JOINT RESOLUTIONS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate joint resolutions be waived:

S.J.R. 276 (two hundred seventy-six).

S.J.R. 286 (two hundred eighty-six).

S.J.R. 292 (two hundred ninety-two).

S.J.R. 294 (two hundred ninety-four).

S.J.R. 308 (three hundred eight).

S.J.R. 285 (two hundred eighty-five).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate joint resolutions were passed by for the day:

S.J.R. 276 (two hundred seventy-six).

S.J.R. 286 (two hundred eighty-six).

S.J.R. 292 (two hundred ninety-two).

S.J.R. 294 (two hundred ninety-four).

S.J.R. 308 (three hundred eight).

S.J.R. 285 (two hundred eighty-five).

SENATE RESOLUTION ON FIRST READING

S.R. 91 (ninety-one) was read by title the first time.

On motion of Senator Lucas, the Senate adjourned until tomorrow at $12\ m.$

Susan Carle Gelman

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

TUESDAY, JANUARY 26, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

RECESS

At 12:20 p.m., Senator Saslaw moved that the Senate recess until 3:10 p.m.

The motion was agreed to.

The hour of 3:10 p.m. having arrived, the Chair was resumed.

The Honorable John S. Edwards, Twenty-first Senatorial District, offered the following prayer:

Mr. President, in our country we've experienced divisiveness, depression, and disenheartment pouring into anger, hatred, and even violence that we have seen in our Nation's capital. Now we need reconciliation and unity. So it's appropriate to hear the serenity prayer. The serenity prayer was written by Reinhold Niebuhr, my favorite theologian, whom I studied and who taught at Union Theological Seminary in New York City when I studied there before I went to law school. So please bow your heads to hear the serenity prayer.

God, grant us the serenity to accept the things we cannot change, the courage to change the things we can, and the wisdom to know the difference. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Michael Adams, Director of Human Resources, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Norment, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Petersen--1. RULE 36--0.

HOUSE COMMUNICATIONS

The following communications were received:

In the House of Delegates January 25, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1736.** A BILL to amend and reenact § 22.1-274 of the Code of Virginia, relating to school nurses; requirements.
- **H.B. 1760.** A BILL to amend the Code of Virginia by adding in Chapter 10.1 of Title 10.1 a section numbered 10.1-1016.1 and by adding in Chapter 17 of Title 10.1 a section numbered 10.1-1705.1, relating to conservation easements; construction.
- **H.B. 1775.** A BILL to amend and reenact § 17.1-276 of the Code of Virginia, relating to the State Corporation Commission; exemption from fees for remote access to local land records.
- **H.B. 1790.** A BILL to amend and reenact § 22.1-98 of the Code of Virginia, relating to public schools; severe weather conditions and other emergency situations; unscheduled remote learning days.
- **H.B. 1804.** A BILL to direct the Department of Conservation and Recreation to recommend a dedicated funding source for state parks.
- **H.B. 1806.** A BILL to amend and reenact § 19.2-303 of the Code of Virginia, relating to suspension or modification of sentence; transfer to the Department of Corrections.
- **H.B. 1814.** A BILL to amend and reenact § 34-29 of the Code of Virginia, relating to garnishment of wages; protected portion of disposable earnings.
- **H.B. 1821.** A BILL to amend and reenact § 18.2-251.03 of the Code of Virginia, relating to arrest and prosecution when experiencing or reporting overdoses.
- H.B. 1833. A BILL to amend and reenact § 10.1-109 of the Code of Virginia, relating to Department of Conservation and Recreation; leasing of land.
- **H.B. 1837.** A BILL to amend and reenact § 10.1-502 of the Code of Virginia, relating to Soil and Water Conservation Board; membership.
- **H.B. 1838.** A BILL to amend and reenact § 22.1-57.3:1.1 of the Code of Virginia, relating to Loudoun County school board; staggered terms.

- **H.B. 1852.** A BILL to amend the Code of Virginia by adding in Title 20 a chapter numbered 11, consisting of sections numbered 20-168 through 20-187, relating to the Uniform Collaborative Law Act.
- **H.B. 1853.** A BILL to amend and reenact § 54.1-3916 of the Code of Virginia and to repeal § 54.1-3915.1 of the Code of Virginia, relating to lawyers; client accounts.
- **H.B. 1865.** A BILL to amend and reenact § 22.1-253.13:1 of the Code of Virginia, relating to certain students in kindergarten through grade 3; reading intervention services.
- **H.B. 1878.** A BILL to amend and reenact §§ 16.1-256 and 16.1-260 of the Code of Virginia, relating to juvenile intake and petition; appeal to a magistrate on a finding of no probable cause.
- **H.B. 1897.** A BILL to amend and reenact § 8.01-126 of the Code of Virginia, relating to summons for unlawful detainer; notice; adverse employment actions prohibited.
- **H.B. 1905.** A BILL to amend and reenact § 22.1-200.03 of the Code of Virginia, relating to economic education and financial literacy required in middle and high school grades; employment arrangements.
- **H.B. 1911.** A BILL to amend and reenact §§ 20-99 and 20-106 of the Code of Virginia, relating to no-fault divorces; corroboration requirement.
- **H.B. 1916.** A BILL to amend and reenact §§ 58.1-439.12:08 and 58.1-439.12:11 of the Code of Virginia, relating to research and development tax credits.
- **H.B. 1921.** A BILL to amend and reenact §§ 24.2-638, 24.2-646.1, and 24.2-649 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 24.2-649.1, relating to assistance for certain voters; curbside voting.
- **H.B. 1951.** A BILL to amend the Code of Virginia by adding a section numbered 18.2-16.1, relating to abolishing the common-law crime of suicide.
- **H.B. 1958.** A BILL to amend the Code of Virginia by adding a section numbered 10.1-417.1, relating to designation of a segment of the South River as a state scenic river.
- **H.B. 1968.** A BILL to amend and reenact § 24.2-701.1 of the Code of Virginia, relating to absentee voting; early in person; availability on Sundays.
- **H.B. 1982.** A BILL to amend and reenact § 62.1-44.19:21 of the Code of Virginia, relating to nutrient credits; use by facility with permit for stormwater discharges.
- **H.B. 1983.** A BILL to amend and reenact §§ 62.1-44.15:23 and 62.1-44.15:23.1 of the Code of Virginia, relating to wetland and stream mitigation banks; proximity of impacted site.
- **H.B. 1999.** A BILL to amend and reenact § 58.1-112 of the Code of Virginia, relating to authorizing Tax Commissioner to waive accrual of interest in the event that the Governor declares a state of emergency.
- **H.B. 2012.** A BILL to amend and reenact §§ 16.1-253 and 16.1-253.2 of the Code of Virginia, relating to violations of protective orders; preliminary child protective order.

- **H.B. 2058.** A BILL to amend the Code of Virginia by adding in Title 22.1 a chapter numbered 26, consisting of sections numbered 22.1-364 through 22.1-368, relating to Virginia STEM Education Advisory Board; established; report.
- **H.B. 2060.** A BILL to direct the Department of Taxation to analyze the prospect of establishing an online portal for tax practitioners.
- **H.B. 2081.** A BILL to amend and reenact §§ 24.2-604, 24.2-671, and 24.2-802.1 of the Code of Virginia, relating to polling places; prohibited activities; possession of a firearm; penalty.
- **H.B. 2159.** A BILL to amend and reenact § 29.1-556.1 of the Code of Virginia, relating to release of balloon; prohibition; civil penalty.
- H.B. 2178. A BILL to authorize the issuance of bonds, in an amount up to \$34,136,000 plus financing costs, pursuant to Article X, Section 9 (c) of the Constitution of Virginia, for paying costs of acquiring, constructing and equipping revenue-producing capital projects at institutions of higher learning of the Commonwealth; to authorize the Treasury Board, by and with the consent of the Governor, to fix the details of such bonds, to provide for the sale of such bonds, and to issue notes to borrow money in anticipation of the issuance of the bonds; to provide for the pledge of the net revenues of such capital projects and the full faith, credit and taxing power of the Commonwealth for the payment of such bonds; to provide that the interest income on such bonds and notes shall be exempt from all taxation by the Commonwealth and any political subdivision thereof; emergency. EMERGENCY
- H.B. 2179. A BILL to amend and reenact §§ 2 and 4 of the first enactment of Chapter 265 and §§ 2 and 4 of the first enactment of Chapter 408 of the Acts of Assembly of 1992, relating to the issuance of Commonwealth of Virginia Article X, Section 9 (c) Refunding Bonds, subject to the provisions of Article X, Sections 9 (a) and 9 (c) of the Constitution of Virginia; emergency.
 EMERGENCY
- **H.B. 2181.** A BILL to amend and reenact §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308 of the Code of Virginia, relating to Virginia Retirement System; technical amendments.
- **H.B. 2223.** A BILL to amend and reenact §§ 2.2-2809, 5.1-1.3, 10.1-2006, 21-163, 30-131, 33.2-205, 36-111, 42.1-16, 44-21, 46.2-202, 52-3, 53.1-11, 54.1-305, 58.1-201, and 60.2-109 of the Code of Virginia, relating to Department of the Treasury and State Treasurer; bonds.
- THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTIONS:
- H.J.R. 587. Commending Sim Ewing.
- H.J.R. 588. Celebrating the life of Dave Andrew Weeks, Jr.
- **H.J.R. 589.** Celebrating the life of the Honorable Mamye E. BaCote.
- H.J.R. 590. Celebrating the life of Stephen Teel Goodwin.
- H.J.R. 591. Celebrating the life of Billy Joe Roberts.
- H.J.R. 592. Celebrating the life of Captain Charles F. Noll, Sr., USN, Ret.

H.J.R. 593. Celebrating the life of William Burford Warrick.

H.J.R. 594. Commending Pierce's Pitt Bar-B-Que.

H.J.R. 595. Celebrating the life of Charlotte Melton.

H.J.R. 597. Commending Virginia's Crossroads.

H.J.R. 598. Commending Richmond Raceway.

H.J.R. 599. Commending the Virginia Chapter of American Promise and Take Back Our Republic.

H.J.R. 600. Commending Lisa Rode.

H.J.R. 601. Celebrating the life of Marvin Howard Wagner.

H.J.R. 602. Commending Tiffany J. Alfred.

H.J.R. 603. Commending Christchurch School.

H.J.R. 608. Commending Robert Sykes.

H.J.R. 609. Commending Dennis K. Morris.

H.J.R. 610. Commending Richard D. Pillow.

H.J.R. 611. Commending Andrew's Grill.

H.J.R. 612. Commending Beulah E. Thomas.

H.J.R. 613. Celebrating the life of the Reverend Leroy A. Cherry.

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING SENATE JOINT RESOLUTIONS:

S.J.R. 295. Commending Spotsylvania County.

S.J.R. 296. Commending Barry M. Barnard.

S.J.R. 304. Celebrating the life of the Honorable Mamye E. BaCote.

S.J.R. 306. Commending Moneta Elementary School.

S.J.R. 307. Commending Pete's Pizza.

S.J.R. 311. Celebrating the life of Robert Edward Mann.

S.J.R. 312. Commending O. R. Singleton, Jr.

S.J.R. 313. Commending Virginia's Crossroads.

S.J.R. 314. Commending the Virginia Chapter of American Promise and Take Back Our Republic.

S.J.R. 315. Celebrating the life of Theodore Carter DeLaney, Jr.

S.J.R. 318. Celebrating the life of Stephen Teel Goodwin.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

> In the House of Delegates January 26, 2021

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTION:

H.J.R. 628. Election of Circuit Court Judges, General District Court Judges, Juvenile and Domestic Relations District Court Judges, members of the Judicial Inquiry and Review Commission, members of the State Corporation Commission, and the Auditor of Public Accounts.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communications from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--36. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Deeds, McDougle, Petersen--3.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1736, H.B. 1790, H.B. 1838, H.B. 1865, H.B. 1905, and H.B. 2058 were referred to the Committee on Education and Health.
- H.B. 1760, H.B. 1804, H.B. 1833, H.B. 1837, H.B. 1958, H.B. 1982, H.B. 1983, and H.B. 2159 were referred to the Committee on Agriculture, Conservation and Natural Resources.
 - H.B. 1775 was referred to the Committee on Commerce and Labor.
- H.B. 1806, H.B. 1814, H.B. 1821, H.B. 1852, H.B. 1853, H.B. 1878, H.B. 1897, H.B. 1911, H.B. 1951, H.B. 2012, and H.B. 2081 were referred to the Committee on the Judiciary.

- H.B. 1916, H.B. 1999, H.B. 2060, H.B. 2178, H.B. 2179, H.B. 2181, and H.B. 2223 were referred to the Committee on Finance and Appropriations.
 - H.B. 1921 and H.B. 1968 were referred to the Committee on Privileges and Elections.

The House joint resolutions, communicated as agreed to by the House of Delegates, were laid on the Clerk's Desk under Senate Rule 26 (g) as follows:

H.J.R. 587, H.J.R. 588, H.J.R. 589, H.J.R. 590, H.J.R. 591, H.J.R. 592, H.J.R. 593, H.J.R. 594, H.J.R. 595, H.J.R. 597, H.J.R. 598, H.J.R. 599, H.J.R. 600, H.J.R. 601, H.J.R. 602, H.J.R. 603, H.J.R. 608, H.J.R. 609, H.J.R. 610, H.J.R. 611, H.J.R. 612, and H.J.R. 613.

The House joint resolution, communicated as agreed to by the House of Delegates, the first reading of its title having been waived, was referred as follows:

H.J.R. 628 was referred to the Committee on the Judiciary.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Saslaw from the Committee on Commerce and Labor:

- **S.B. 1159** (one thousand one hundred fifty-nine) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1195 (one thousand one hundred ninety-five) with substitute.
 - S.B. 1223 (one thousand two hundred twenty-three).
 - S.B. 1269 (one thousand two hundred sixty-nine) with substitute.
- **S.B. 1275** (one thousand two hundred seventy-five) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1310** (one thousand three hundred ten) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1334 (one thousand three hundred thirty-four).
- **S.B. 1342** (one thousand three hundred forty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1351** (one thousand three hundred fifty-one).
- **S.B. 1375** (one thousand three hundred seventy-five) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1380** (one thousand three hundred eighty) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1413** (one thousand four hundred thirteen) with substitute.

The following bill, having been considered by the committee in session, was recommended for rereferral by the Committee on Commerce and Labor:

S.B. 1323 (one thousand three hundred twenty-three) with the recommendation that it be rereferred to the Committee on the Judiciary.

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- S.B. 1102 (one thousand one hundred two) with substitute.
- **S.B.** 1147 (one thousand one hundred forty-seven).
- **S.B.** 1150 (one thousand one hundred fifty) with amendment.
- **S.B.** 1165 (one thousand one hundred sixty-five).
- **S.B.** 1170 (one thousand one hundred seventy).
- **S.B. 1204** (one thousand two hundred four) with amendments.
- S.B. 1207 (one thousand two hundred seven).
- S.B. 1221 (one thousand two hundred twenty-one).
- **S.B. 1227** (one thousand two hundred twenty-seven).
- **S.B. 1251** (one thousand two hundred fifty-one).
- S.B. 1298 (one thousand two hundred ninety-eight).
- S.B. 1302 (one thousand three hundred two).
- **S.B. 1321** (one thousand three hundred twenty-one).
- S.B. 1328 (one thousand three hundred twenty-eight) with amendments.
- **S.B.** 1367 (one thousand three hundred sixty-seven).
- S.B. 1403 (one thousand four hundred three).
- S.B. 1438 (one thousand four hundred thirty-eight).
- S.B. 1159, S.B. 1275, S.B. 1310, S.B. 1342, S.B. 1375, and S.B. 1380 were rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1323** was rereferred to the Committee on the Judiciary.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 330. Celebrating the life of Robert Emmett Mannion, Sr.

Patron--Petersen

S.J.R. 331. Commending Richard D. Pillow.

Patron--Newman

JUDICIAL NOMINATION FORMS RECEIVED

Pursuant to Rule 18 (f), the following judicial nomination forms were filed with the Clerk:

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Third Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Brenda C. Spry, of Portsmouth, as a judge of the Third Judicial Circuit for a term of eight years commencing February 16, 2021.

Respectfully submitted,

/s/ L. Louise Lucas /s/ Mamie E. Locke John A. Cosgrove, Jr.

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fourth Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Junius P. Fulton, III, of Norfolk, as a judge of the Fourth Judicial Circuit for a term of eight years commencing February 1, 2021.

Respectfully submitted,

/s/ Lynwood W. Lewis, Jr. /s/ Lionell Spruill, Sr. /s/ Jen A. Kiggans

$\begin{array}{c} \text{COMMONWEALTH OF VIRGINIA} \\ \text{SENATE} \end{array}$

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Tenth Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Kimberley S. White, of Halifax, as a judge of the Tenth Judicial Circuit for a term of eight years commencing February 1, 2021.

Respectfully submitted,

/s/ Frank M. Ruff, Jr. /s/ William M. Stanley, Jr. /s/ Mark J. Peake

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fifteenth Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Patricia Kelly, of Caroline, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ Ryan T. McDougle

/s/ Jill H. Vogel

/s/ Richard H. Stuart

/s/ Bryce E. Reeves

/s/ Scott A. Surovell

/s/ Siobhan S. Dunnavant

/s/ Jennifer L. McClellan

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fifteenth Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Michael E. Levy, of Stafford, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ Ryan T. McDougle

/s/ Jill H. Vogel

/s/ Richard H. Stuart

/s/ Bryce E. Reeves

/s/ Scott A. Surovell

/s/ Siobhan S. Dunnavant

/s/ Jennifer L. McClellan

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Eighteenth Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Lisa Bondareff Kemler, of Alexandria, as a judge of the Eighteenth Judicial Circuit for a term of eight years commencing March 1, 2021.

Respectfully submitted,

/s/ Richard L. Saslaw /s/ George L. Barker

/s/ Adam P. Ebbin

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Nineteenth Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

John M. Tran, of Fairfax County, as a judge of the Nineteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ Richard L. Saslaw

/s/ Janet D. Howell

/s/ J. Chapman Petersen

/s/ George L. Barker

/s/ David W. Marsden

/s/ Adam P. Ebbin

/s/ Barbara A. Favola

/s/ Scott A. Surovell

/s/ Jennifer B. Boysko

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senator representing the Twenty-first Judicial Circuit hereby nominates, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

G. Carter Greer, of Martinsville, as a judge of the Twenty-first Judicial Circuit for a term of eight years commencing March 1, 2021.

Respectfully submitted,

/s/ William M. Stanley, Jr.

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-second Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

James J. Reynolds, of Danville, as a judge of the Twenty-second Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ Frank M. Ruff, Jr. /s/ William M. Stanley, Jr. /s/ David R. Suetterlein

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-third Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

David B. Carson, of Roanoke City, as a judge of the Twenty-third Judicial Circuit for a term of eight years commencing July 1, 2021.

/s/ Stephen D. Newman /s/ John S. Edwards /s/ David R. Suetterlein

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-sixth Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Bruce D. Albertson, of Rockingham, as a judge of the Twenty-sixth Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ Emmett W. Hanger, Jr. /s/ Mark D. Obenshain /s/ Jill H. Vogel

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senator representing the Twenty-eighth Judicial Circuit hereby nominates, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Deanis L. Simmons, of Bristol, as a judge of the Twenty-eighth Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ Todd E. Pillion Vacancy

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senator representing the Thirtieth Judicial Circuit hereby nominates, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Tammy S. McElyea, of Lee, as a judge of the Thirtieth Judicial Circuit for a term of eight years commencing April 1, 2021.

Respectfully submitted,

/s/ Todd E. Pillion Vacancy

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM CIRCUIT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Thirty-first Judicial Circuit hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the circuit court judgeship listed below:

Carroll A. Weimer, Jr., of Prince William, as a judge of the Thirty-first Judicial Circuit for a term of eight years commencing July 1, 2021.

Respectfully submitted,

/s/ George L. Barker /s/ Richard H. Stuart /s/ Scott A. Surovell

/s/ Jeremy S. McPike

/s/ John J. Bell

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Second Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Vivian F. Henderson, of Virginia Beach, as a judge of the Second Judicial District for a term of six years commencing February 1, 2021.

/s/ John A. Cosgrove, Jr. /s/ Lynwood W. Lewis, Jr. /s/ William R. DeSteph, Jr. /s/ Jen A. Kiggans

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fourth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Tasha D. Scott, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Lynwood W. Lewis, Jr. /s/ Lionell Spruill, Sr. /s/ Jen A. Kiggans

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Seventh Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Tyneka L. D. Flythe, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Mamie E. Locke /s/ T. Montgomery Mason

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Seventh Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

M. Scott Stein, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Mamie E. Locke /s/ T. Montgomery Mason

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Eighth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Tonya Henderson-Stith, of Hampton, as a judge of the Eighth Judicial District for a term of six years commencing May 1, 2021.

Respectfully submitted,

/s/ Thomas K. Norment, Jr. /s/ Mamie E. Locke /s/ T. Montgomery Mason

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Thirteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

David M. Hicks, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing July 1, 2021.

/s/ Jennifer L. McClellan /s/ Joseph D. Morrissey /s/ Ghazala F. Hashmi

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Thirteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Jacqueline S. McClenney, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing December 1, 2021.

Respectfully submitted,

/s/ Jennifer L. McClellan /s/ Joseph D. Morrissey /s/ Ghazala F. Hashmi

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Thirteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Mansi J. Shah, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing February 1, 2021.

Respectfully submitted,

/s/ Jennifer L. McClellan /s/ Joseph D. Morrissey /s/ Ghazala F. Hashmi

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fourteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

B. Craig Dunkum, of Henrico, as a judge of the Fourteenth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Siobhan S. Dunnavant /s/ Jennifer L. McClellan

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fourteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

John K. Honey, Jr., of Henrico, as a judge of the Fourteenth Judicial District for a term of six years commencing November 1, 2021.

Respectfully submitted,

/s/ Siobhan S. Dunnavant /s/ Jennifer L. McClellan

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Nineteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Manuel A. Capsalis, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

- /s/ Richard L. Saslaw
- /s/ Janet D. Howell
- /s/ J. Chapman Petersen
- /s/ George L. Barker
- /s/ David W. Marsden
- /s/ Adam P. Ebbin
- /s/ Barbara A. Favola
- /s/ Scott A. Surovell
- /s/ Jennifer B. Boysko

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Nineteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Michael J. Lindner, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

- /s/ Richard L. Saslaw
- /s/ Janet D. Howell
- /s/ J. Chapman Petersen
- /s/ George L. Barker
- /s/ David W. Marsden
- /s/ Adam P. Ebbin
- /s/ Barbara A. Favola /s/ Scott A. Surovell
- /s/ Jennifer B. Boysko

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Nineteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

William J. Minor, Jr., of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing April 1, 2021.

- /s/ Richard L. Saslaw
- /s/ Janet D. Howell
- /s/ J. Chapman Petersen
- /s/ George L. Barker
- /s/ David W. Marsden
- /s/ Adam P. Ebbin
- /s/ Barbara A. Favola
- /s/ Scott A. Surovell
- /s/ Jennifer B. Boysko

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Nineteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Tina L. Snee, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

- /s/ Richard L. Saslaw
- /s/ Janet D. Howell
- /s/ J. Chapman Petersen
- /s/ George L. Barker
- /s/ David W. Marsden
- /s/ Adam P. Ebbin
- /s/ Barbara A. Favola
- /s/ Scott A. Surovell
- /s/ Jennifer B. Boysko

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-third Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Jacqueline F. Ward Talevi, of Roanoke, as a judge of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.

/s/ Stephen D. Newman /s/ John S. Edwards /s/ David R. Suetterlein

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-fourth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Stephanie S. Maddox, of Amherst, as a judge of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Stephen D. Newman /s/ Frank M. Ruff, Jr. R. Creigh Deeds /s/ David R. Suetterlein /s/ Mark J. Peake

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-sixth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Amy B. Tisinger, of Shenandoah, as a judge of the Twenty-sixth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Emmett W. Hanger, Jr. /s/ Mark D. Obenshain /s/ Jill H. Vogel

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-seventh Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Erin J. DeHart, of Bland, as a judge of the Twenty-seventh Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ John S. Edwards /s/ William M. Stanley, Jr. /s/ David R. Suetterlein /s/ Todd E. Pillion

Vacancy

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-seventh Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Gino W. Williams, of Floyd, as a judge of the Twenty-seventh Judicial District for a term of six years commencing April 1, 2021.

Respectfully submitted,

/s/ John S. Edwards /s/ William M. Stanley, Jr. /s/ David R. Suetterlein

/s/ Todd E. Pillion

Vacancy

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM GENERAL DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Thirty-first Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the general district court judgeship listed below:

Robert P. Coleman, of Prince William, as a judge of the Thirty-first Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ George L. Barker /s/ Richard H. Stuart /s/ Scott A. Surovell /s/ Jeremy S. McPike /s/ John J. Bell

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fourth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Lyn M. Simmons, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing September 17, 2021.

Respectfully submitted,

/s/ Lynwood W. Lewis, Jr. /s/ Lionell Spruill, Sr. /s/ Jen A. Kiggans

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM
JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Sixth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Jacqueline R. Waymack, of Prince George, as a judge of the Sixth Judicial District for a term of six years commencing June 1, 2021.

Respectfully submitted,

/s/ L. Louise Lucas /s/ Thomas K. Norment, Jr. /s/ Frank M. Ruff, Jr. /s/ Joseph D. Morrissey

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fifteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

John E. Franklin, of Fredericksburg, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Ryan T. McDougle /s/ Jill H. Vogel /s/ Richard H. Stuart /s/ Bryce E. Reeves Scott A. Surovell /s/ Siobhan S. Dunnavant /s/ Jennifer L. McClellan

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Fifteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Frank G. Uvanni, of Hanover, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Ryan T. McDougle /s/ Jill H. Vogel

/s/ Richard H. Stuart

/s/ Bryce E. Reeves

/s/ Scott A. Surovell

/s/ Siobhan S. Dunnavant

/s/ Jennifer L. McClellan

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Nineteenth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Thomas P. Sotelo, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing February 1, 2021.

Respectfully submitted,

/s/ Richard L. Saslaw

/s/ Janet D. Howell

/s/ J. Chapman Petersen

/s/ George L. Barker

/s/ David W. Marsden

/s/ Adam P. Ebbin

/s/ Barbara A. Favola

/s/ Scott A. Surovell

/s/ Jennifer B. Boysko

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-second Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Timothy W. Allen, of Franklin County, as a judge of the Twenty-second Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Frank M. Ruff, Jr. /s/ William M. Stanley, Jr. /s/ David R. Suetterlein

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-third Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Hilary D. Griffith, of Salem, as a judge of the Twenty-third Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted, /s/ Stephen D. Newman /s/ John S. Edwards /s/ David R. Suetterlein

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-third Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Melissa W. Friedman, of Roanoke City, as a judge of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.

Respectfully submitted,

/s/ Stephen D. Newman /s/ John S. Edwards /s/ David R. Suetterlein

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-fourth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Jeffrey P. Bennett, of Amherst, as a judge of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Stephen D. Newman /s/ Frank M. Ruff, Jr. R. Creigh Deeds /s/ David R. Suetterlein /s/ Mark J. Peake

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-sixth Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Hugh David O'Donnell, of Rockingham, as a judge of the Twenty-sixth Judicial District for a term of six years commencing April 1, 2021.

Respectfully submitted,

/s/ Emmett W. Hanger, Jr. /s/ Mark D. Obenshain /s/ Jill H. Vogel

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM
JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senators representing the Twenty-seventh Judicial District hereby nominate, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Robert C. Viar, Jr., of Montgomery, as a judge of the Twenty-seventh Judicial District for a term of six years commencing May 1, 2021.

Respectfully submitted,

/s/ John S. Edwards /s/ William M. Stanley, Jr. /s/ David R. Suetterlein /s/ Todd E. Pillion Vacancy

COMMONWEALTH OF VIRGINIA SENATE

JUDICIAL NOMINATION FORM JUVENILE AND DOMESTIC RELATIONS DISTRICT COURT

TO THE SENATE OF VIRGINIA:

The undersigned Senator representing the Twenty-eighth Judicial District hereby nominates, pursuant to Senate Rule 18 (f), the following person to be elected to the juvenile and domestic relations district court judgeship listed below:

Joseph B. Lyle, of Washington, as a judge of the Twenty-eighth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ Todd E. Pillion Vacancy

INTRODUCTION OF LEGISLATION

Senator Edwards, by leave, under Senate Rule 11 (b) presented the following resolutions which were ordered to be printed and referred:

S.R. 99. Nominating persons to be elected to circuit court judgeships.

Patron--Edwards

Referred to Committee on the Judiciary

S.R. 100. Nominating persons to be elected to general district court judgeships.

Patron--Edwards

Referred to Committee on the Judiciary

S.R. 101. Nominating persons to be elected to juvenile and domestic relations district court judgeships. Patron--Edwards

Referred to Committee on the Judiciary

S.R. 102. Nominating a person to be elected as a member of the Judicial Inquiry and Review Commission.

Patron--Edwards

Referred to Committee on the Judiciary

Senator Saslaw, by leave, under Senate Rule 11 (b) presented the following resolution which was ordered to be printed and referred:

S.R. 103. Nominating persons to be elected as members of the State Corporation Commission.

Patron--Saslaw

Referred to Committee on Commerce and Labor

IMMEDIATE CONSIDERATION

Senator Edwards moved that the Rules be suspended, the Committee on the Judiciary be discharged from further consideration of **H.J.R. 628** (six hundred twenty-eight), the readings of the title be waived, and the joint resolution be taken up for immediate consideration.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

HOUSE JOINT RESOLUTION NO. 628

Election of Circuit Court Judges, General District Court Judges, Juvenile and Domestic Relations District Court Judges, members of the Judicial Inquiry and Review Commission, members of the State Corporation Commission, and the Auditor of Public Accounts.

RESOLVED by the House of Delegates, the Senate concurring, That the General Assembly shall proceed at $3:00~\mathrm{p.m.}$

To the election of Circuit Court judges for terms of eight years commencing as follows:

One judge for the Third Judicial Circuit, term commencing February 16, 2021.

One judge for the Fourth Judicial Circuit, term commencing February 1, 2021.

One judge for the Tenth Judicial Circuit, term commencing February $1,\,2021.$

One judge for the Fifteenth Judicial Circuit, term commencing July 1, 2021. One judge for the Fifteenth Judicial Circuit, term commencing July 1, 2021.

One judge for the Eighteenth Judicial Circuit, term commencing March 1, 2021.

One judge for the Nineteenth Judicial Circuit, term commencing July 1, 2021.

One judge for the Twenty-first Judicial Circuit, term commencing March 1, 2021.

```
One judge for the Twenty-second Judicial Circuit, term commencing July 1, 2021. One judge for the Twenty-third Judicial Circuit, term commencing July 1, 2021. One judge for the Twenty-sixth Judicial Circuit, term commencing July 1, 2021. One judge for the Twenty-eighth Judicial Circuit, term commencing July 1, 2021. One judge for the Twenty-ninth Judicial Circuit, term commencing February 1, 2021. One judge for the Thirtieth Judicial Circuit, term commencing April 1, 2021. One judge for the Thirty-first Judicial Circuit, term commencing July 1, 2021.
```

To the election of General District Court judges for terms of six years commencing as follows: One judge for the Second Judicial District, term commencing February 1, 2021. One judge for the Fourth Judicial District, term commencing July 1, 2021. One judge for the Seventh Judicial District, term commencing July 1, 2021. One judge for the Seventh Judicial District, term commencing July 1, 2021. One judge for the Eighth Judicial District, term commencing May 1, 2021. One judge for the Twelfth Judicial District, term commencing April 1, 2021. One judge for the Thirteenth Judicial District, term commencing July 1, 2021. One judge for the Thirteenth Judicial District, term commencing December 1, 2021. One judge for the Thirteenth Judicial District, term commencing February 1, 2021. One judge for the Fourteenth Judicial District, term commencing July 1, 2021. One judge for the Fourteenth Judicial District, term commencing November 1, 2021. One judge for the Nineteenth Judicial District, term commencing July 1, 2021. One judge for the Nineteenth Judicial District, term commencing July 1, 2021. One judge for the Nineteenth Judicial District, term commencing April 1, 2021. One judge for the Nineteenth Judicial District, term commencing July 1, 2021. One judge for the Twenty-third Judicial District, term commencing February 1, 2021. One judge for the Twenty-fourth Judicial District, term commencing July 1, 2021. One judge for the Twenty-sixth Judicial District, term commencing July 1, 2021. One judge for the Twenty-seventh Judicial District, term commencing July 1, 2021. One judge for the Twenty-seventh Judicial District, term commencing April 1, 2021.

To the election of Juvenile and Domestic Relations District Court judges for terms of six years commencing as follows:

```
One judge for the Fourth Judicial District, term commencing September 17, 2021.

One judge for the Sixth Judicial District, term commencing June 1, 2021.

One judge for the Fifteenth Judicial District, term commencing July 1, 2021.

One judge for the Fifteenth Judicial District, term commencing July 1, 2021.

One judge for the Nineteenth Judicial District, term commencing February 1, 2021.

One judge for the Twenty-second Judicial District, term commencing July 1, 2021.

One judge for the Twenty-third Judicial District, term commencing July 1, 2021.

One judge for the Twenty-third Judicial District, term commencing February 1, 2021.

One judge for the Twenty-fourth Judicial District, term commencing July 1, 2021.

One judge for the Twenty-sixth Judicial District, term commencing April 1, 2021.

One judge for the Twenty-seventh Judicial District, term commencing May 1, 2021.

One judge for the Twenty-eighth Judicial District, term commencing July 1, 2021.

One judge for the Twenty-ninth Judicial District, term commencing April 1, 2021.

One judge for the Twenty-ninth Judicial District, term commencing April 1, 2021.
```

One judge for the Thirty-first Judicial District, term commencing July 1, 2021.

To the election of members of the Judicial Inquiry and Review Commission for terms of four years commencing as follows:

```
One member, term commencing July 1, 2021.
```

One member, term commencing July 1, 2021.

To the election of members of the State Corporation Commission for terms as follows:

One member, term commencing February 1, 2021, and ending January 31, 2022.

One member, term commencing February 1, 2021, and ending January 31, 2026.

To the election of the Auditor of Public Accounts for a term of four years commencing February 1, 2021.

And that in the execution of the joint order nominations shall be made in the order herein named, and that each house shall be notified of said nominations, and when the rolls shall be called for the whole number, the presiding officers of each house shall appoint a committee of three, which together shall constitute the joint committee to count the vote of each house in each case and report the results to their respective houses. The joint order may be suspended by the presiding officer of either house at any time but for no longer than twenty-four hours to receive the report of the joint committee.

H.J.R. 628, on motion of Senator Edwards, was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Edwards was ordered to inform the House of Delegates thereof.

JOINT ORDER FOR ELECTIONS

The President stated that the Senate on its part was ready to proceed, pursuant to House Joint Resolution No. 628, with the execution of the Joint Order to the election of certain judges and other officers of the Commonwealth.

The President stated that nominations were in order for judges of the respective circuit courts.

On motion of Senator Edwards, the Rules were suspended and S.R. 99 (ninety-nine) was taken up for immediate consideration, discharging the Committee on the Judiciary from further consideration of the resolution, and waiving the readings of the title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 99

Nominating persons to be elected to circuit court judgeships.

RESOLVED by the Senate of Virginia, That the following persons are hereby nominated to be elected to the respective circuit court judgeships as follows:

Brenda C. Spry, Esquire, of Portsmouth, as a judge of the Third Judicial Circuit for a term of eight years commencing February 16, 2021.

The Honorable Junius P. Fulton, III, of Norfolk, as a judge of the Fourth Judicial Circuit for a term of eight years commencing February 1, 2021.

The Honorable Kimberley S. White, of Halifax, as a judge of the Tenth Judicial Circuit for a term of eight years commencing February 1, 2021.

The Honorable Patricia Kelly, of Caroline, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Michael E. Levy, of Stafford, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Lisa Bondareff Kemler, of Alexandria, as a judge of the Eighteenth Judicial Circuit for a term of eight years commencing March 1, 2021.

The Honorable John M. Tran, of Fairfax County, as a judge of the Nineteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable G. Carter Greer, of Martinsville, as a judge of the Twenty-first Judicial Circuit for a term of eight years commencing March 1, 2021.

The Honorable James J. Reynolds, of Danville, as a judge of the Twenty-second Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable David B. Carson, of Roanoke City, as a judge of the Twenty-third Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Bruce D. Albertson, of Rockingham, as a judge of the Twenty-sixth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Deanis L. Simmons, of Bristol, as a judge of the Twenty-eighth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Tammy S. McElyea, of Lee, as a judge of the Thirtieth Judicial Circuit for a term of eight years commencing April 1, 2021.

The Honorable Carroll A. Weimer, Jr., of Prince William, as a judge of the Thirty-first Judicial Circuit for a term of eight years commencing July 1, 2021.

Senator Edwards offered the following amendment:

1. After line 33, introduced

insert

The Honorable Jack S. Hurley, Jr., of Tazewell, as a judge of the Twenty-ninth Judicial Circuit for a term of eight years commencing February 1, 2021.

On motion of Senator Edwards, the reading of the amendment was waived.

On motion of Senator Edwards, the amendment was agreed to.

The resolution was ordered to be engrossed.

SENATE RESOLUTION NO. 99

Nominating persons to be elected to circuit court judgeships.

RESOLVED by the Senate of Virginia, That the following persons are hereby nominated to be elected to the respective circuit court judgeships as follows:

Brenda C. Spry, Esquire, of Portsmouth, as a judge of the Third Judicial Circuit for a term of eight years commencing February 16, 2021.

The Honorable Junius P. Fulton, III, of Norfolk, as a judge of the Fourth Judicial Circuit for a term of eight years commencing February 1, 2021.

The Honorable Kimberley S. White, of Halifax, as a judge of the Tenth Judicial Circuit for a term of eight years commencing February 1, 2021.

The Honorable Patricia Kelly, of Caroline, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Michael E. Levy, of Stafford, as a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Lisa Bondareff Kemler, of Alexandria, as a judge of the Eighteenth Judicial Circuit for a term of eight years commencing March 1, 2021.

The Honorable John M. Tran, of Fairfax County, as a judge of the Nineteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable G. Carter Greer, of Martinsville, as a judge of the Twenty-first Judicial Circuit for a term of eight years commencing March 1, 2021.

The Honorable James J. Reynolds, of Danville, as a judge of the Twenty-second Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable David B. Carson, of Roanoke City, as a judge of the Twenty-third Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Bruce D. Albertson, of Rockingham, as a judge of the Twenty-sixth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Deanis L. Simmons, of Bristol, as a judge of the Twenty-eighth Judicial Circuit for a term of eight years commencing July 1, 2021.

[The Honorable Jack S. Hurley, Jr., of Tazewell, as a judge of the Twenty-ninth Judicial Circuit for a term of eight years commencing February 1, 2021.]

The Honorable Tammy S. McElyea, of Lee, as a judge of the Thirtieth Judicial Circuit for a term of eight years commencing April 1, 2021.

The Honorable Carroll A. Weimer, Jr., of Prince William, as a judge of the Thirty-first Judicial Circuit for a term of eight years commencing July 1, 2021.

S.R. 99, on motion of Senator Edwards, was agreed to.

The President stated that nominations were in order for judges of the respective general district courts.

On motion of Senator Edwards, the Rules were suspended and **S.R. 100** (one hundred) was taken up for immediate consideration, discharging the Committee on the Judiciary from further consideration of the resolution, and waiving the readings of the title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 100

Nominating persons to be elected to general district court judgeships.

RESOLVED by the Senate of Virginia, That the following persons are hereby nominated to be elected to the respective general district court judgeships as follows:

The Honorable Vivian F. Henderson, of Virginia Beach, as a judge of the Second Judicial District for a term of six years commencing February 1, 2021.

The Honorable Tasha D. Scott, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Tyneka L. D. Flythe, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing July 1, 2021.

The Honorable M. Scott Stein, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing July 1, 2021.

The Honorable Tonya Henderson-Stith, of Hampton, as a judge of the Eighth Judicial District for a term of six years commencing May 1, 2021.

The Honorable David M. Hicks, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Jacqueline S. McClenney, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing December 1, 2021.

The Honorable Mansi J. Shah, of Richmond, as a judge of the Thirteenth Judicial District for a term of six years commencing February 1, 2021.

The Honorable B. Craig Dunkum, of Henrico, as a judge of the Fourteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable John K. Honey, Jr., of Henrico, as a judge of the Fourteenth Judicial District for a term of six years commencing November 1, 2021.

The Honorable Manuel A. Capsalis, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Michael J. Lindner, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable William J. Minor, Jr., of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing April 1, 2021.

The Honorable Tina L. Snee, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Jacqueline F. Ward Talevi, of Roanoke, as a judge of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.

The Honorable Stephanie S. Maddox, of Amherst, as a judge of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Amy B. Tisinger, of Shenandoah, as a judge of the Twenty-sixth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Erin J. DeHart, of Bland, as a judge of the Twenty-seventh Judicial District for a term of six years commencing July 1, 2021.

The Honorable Gino W. Williams, of Floyd, as a judge of the Twenty-seventh Judicial District for a term of six years commencing April 1, 2021.

The Honorable Robert P. Coleman, of Prince William, as a judge of the Thirty-first Judicial District for a term of six years commencing July 1, 2021.

S.R. 100, on motion of Senator Edwards, was ordered to be engrossed and was agreed to.

The President stated that nominations were in order for judges of the respective juvenile and domestic relations district courts.

On motion of Senator Edwards, the Rules were suspended and **S.R. 101** (one hundred one) was taken up for immediate consideration, discharging the Committee on the Judiciary from further consideration of the resolution, and waiving the readings of the title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 101

Nominating persons to be elected to juvenile and domestic relations district court judgeships.

RESOLVED by the Senate of Virginia, That the following persons are hereby nominated to be elected to the respective juvenile and domestic relations district court judgeships as follows:

The Honorable Lyn M. Simmons, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing September 17, 2021.

The Honorable Jacqueline R. Waymack, of Prince George, as a judge of the Sixth Judicial District for a term of six years commencing June 1, 2021.

The Honorable John E. Franklin, of Fredericksburg, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Frank G. Uvanni, of Hanover, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Thomas P. Sotelo, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing February 1, 2021.

The Honorable Timothy W. Allen, of Franklin County, as a judge of the Twenty-second Judicial District for a term of six years commencing July 1, 2021.

The Honorable Hilary D. Griffith, of Salem, as a judge of the Twenty-third Judicial District for a term of six years commencing July 1, 2021.

The Honorable Melissa W. Friedman, of Roanoke City, as a judge of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.

The Honorable Jeffrey P. Bennett, of Amherst, as a judge of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Hugh David O'Donnell, of Rockingham, as a judge of the Twenty-sixth Judicial District for a term of six years commencing April 1, 2021.

The Honorable Robert C. Viar, Jr., of Montgomery, as a judge of the Twenty-seventh Judicial District for a term of six years commencing May 1, 2021.

The Honorable Joseph B. Lyle, of Washington, as a judge of the Twenty-eighth Judicial District for a term of six years commencing July 1, 2021.

Senator Edwards offered the following amendments:

1. After line 33, introduced

insert

The Honorable Michael J. Bush, of Russell, as a judge of the Twenty-ninth Judicial District for a term of six years commencing April 1, 2021.

2. After line 33, introduced

insert

The Honorable Martha P. Ketron, of Russell, as a judge of the Twenty-ninth Judicial District for a term of six years commencing February 1, 2021.

On motion of Senator Edwards, the reading of the amendments was waived.

On motion of Senator Edwards, the amendments were agreed to.

The resolution was ordered to be engrossed.

SENATE RESOLUTION NO. 101

Nominating persons to be elected to juvenile and domestic relations district court judgeships.

RESOLVED by the Senate of Virginia, That the following persons are hereby nominated to be elected to the respective juvenile and domestic relations district court judgeships as follows:

The Honorable Lyn M. Simmons, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing September 17, 2021.

The Honorable Jacqueline R. Waymack, of Prince George, as a judge of the Sixth Judicial District for a term of six years commencing June 1, 2021.

The Honorable John E. Franklin, of Fredericksburg, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Frank G. Uvanni, of Hanover, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Thomas P. Sotelo, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing February 1, 2021.

The Honorable Timothy W. Allen, of Franklin County, as a judge of the Twenty-second Judicial District for a term of six years commencing July 1, 2021.

The Honorable Hilary D. Griffith, of Salem, as a judge of the Twenty-third Judicial District for a term of six years commencing July 1, 2021.

The Honorable Melissa W. Friedman, of Roanoke City, as a judge of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.

The Honorable Jeffrey P. Bennett, of Amherst, as a judge of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.

The Honorable Hugh David O'Donnell, of Rockingham, as a judge of the Twenty-sixth Judicial District for a term of six years commencing April 1, 2021.

The Honorable Robert C. Viar, Jr., of Montgomery, as a judge of the Twenty-seventh Judicial District for a term of six years commencing May 1, 2021.

The Honorable Joseph B. Lyle, of Washington, as a judge of the Twenty-eighth Judicial District for a term of six years commencing July 1, 2021.

[The Honorable Michael J. Bush, of Russell, as a judge of the Twenty-ninth Judicial District for a term of six years commencing April 1, 2021.]

[The Honorable Martha P. Ketron, of Russell, as a judge of the Twenty-ninth Judicial District for a term of six years commencing February 1, 2021.]

S.R. 101, on motion of Senator Edwards, was agreed to.

The President stated that nominations were in order for members of the Judicial Inquiry and Review Commission.

On motion of Senator Edwards, the Rules were suspended and **S.R. 102** (one hundred two) was taken up for immediate consideration, discharging the Committee on the Judiciary from further consideration of the resolution, and waiving the readings of the title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 102

Nominating persons to be elected as members of the Judicial Inquiry and Review Commission.

RESOLVED by the Senate of Virginia, That the following persons are hereby nominated to be elected as members of the Judicial Inquiry and Review Commission as follows:

The Honorable James E. Plowman, of Loudoun, as a member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.

Terrie N. Thompson, of Chesapeake, as a member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.

Senator Surovell offered the following amendment:

1. After line 9, introduced strike

lines 10-11

On motion of Senator Surovell, the reading of the amendment was waived.

Senator Surovell moved that the amendment be agreed to.

The question was put on agreeing to the amendment.

The yeas and nays were called for by the Chair.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

The amendment was agreed to.

The resolution was ordered to be engrossed.

SENATE RESOLUTION NO. 102

Nominating a person to be elected as a member of the Judicial Inquiry and Review Commission.

RESOLVED by the Senate of Virginia, That the following person is hereby nominated to be elected as a member of the Judicial Inquiry and Review Commission as follows:

[The Honorable James E. Plowman, of Loudoun, as a member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.]

Terrie N. Thompson, of Chesapeake, as a member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.

S.R. 102, on motion of Senator Edwards, was agreed to.

The President stated that nominations were in order for members of the State Corporation Commission.

On motion of Senator Saslaw, the Rules were suspended and **S.R. 103** (one hundred three) was taken up for immediate consideration, discharging the Committee on Commerce and Labor from further consideration of the resolution, and waiving the readings of the title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 103

Nominating persons to be elected as members of the State Corporation Commission.

RESOLVED by the Senate of Virginia, That the following persons are hereby nominated to be elected as members of the State Corporation Commission as follows:

The Honorable Angela L. Navarro, of the City of Charlottesville, to succeed Mark C. Christie as a member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2022.

The Honorable Jehmal T. Hudson, of the City of Richmond, as a member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2026.

S.R. 103, on motion of Senator Saslaw, was ordered to be engrossed and was agreed to.

The President stated that nominations were in order for the Auditor of Public Accounts.

On motion of Senator Deeds, the Rules were suspended and S.R. 93 (ninety-three) was taken up for immediate consideration, discharging the Committee on Privileges and Elections from further consideration of the resolution, and waiving the readings of the title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 93

Nominating a person to be elected as the Auditor of Public Accounts.

RESOLVED by the Senate of Virginia, That the following person is hereby nominated to be elected as the Auditor of Public Accounts as follows:

Staci A. Henshaw, of Dinwiddie, as the Auditor of Public Accounts for a term of four years commencing February 1, 2021.

S.R. 93, on motion of Senator Deeds, was ordered to be engrossed and was agreed to.

Senator Edwards was ordered to inform the House of Delegates of the nominations made by the Senate.

Under the provisions of the Joint Order, the Senate proceeded with the Calendar.

CALENDAR

-207-

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- **S.B.** 1154 (one thousand one hundred fifty-four).
- **S.B.** 1169 (one thousand one hundred sixty-nine).
- S.B. 1187 (one thousand one hundred eighty-seven).
- S.B. 1189 (one thousand one hundred eighty-nine).
- **S.B. 1205** (one thousand two hundred five).
- S.B. 1213 (one thousand two hundred thirteen).
- S.B. 1242 (one thousand two hundred forty-two).
- **S.B. 1271** (one thousand two hundred seventy-one).
- **S.B. 1296** (one thousand two hundred ninety-six).
- **S.B. 1320** (one thousand three hundred twenty).
- **S.B. 1324** (one thousand three hundred twenty-four).
- S.B. 1369 (one thousand three hundred sixty-nine).
- **S.B.** 1399 (one thousand three hundred ninety-nine).
- S.B. 1421 (one thousand four hundred twenty-one).
- S.B. 1427 (one thousand four hundred twenty-seven).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- S.B. 1154 (one thousand one hundred fifty-four).
- S.B. 1169 (one thousand one hundred sixty-nine).
- S.B. 1187 (one thousand one hundred eighty-seven).
- S.B. 1189 (one thousand one hundred eighty-nine).
- S.B. 1205 (one thousand two hundred five).
- S.B. 1213 (one thousand two hundred thirteen).
- S.B. 1242 (one thousand two hundred forty-two).
- S.B. 1271 (one thousand two hundred seventy-one).
- **S.B. 1296** (one thousand two hundred ninety-six).
- **S.B. 1320** (one thousand three hundred twenty).
- **S.B. 1324** (one thousand three hundred twenty-four).
- S.B. 1369 (one thousand three hundred sixty-nine).
- **S.B. 1399** (one thousand three hundred ninety-nine).
- $\textbf{S.B. 1421} \ (\textbf{one thousand four hundred twenty-one}).$
- **S.B. 1427** (one thousand four hundred twenty-seven).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1291 (one thousand two hundred ninety-one) was taken up and, on motion of Senator Mason, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

S.B. 1121 (one thousand one hundred twenty-one) was read by title the third time and, on motion of Senator Locke, was passed with its title.

The recorded vote is as follows:

YEAS--29. NAYS--10. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Ruff, Saslaw, Spruill, Surovell, Vogel--29.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Obenshain, Peake, Reeves, Stanley, Stuart, Suetterlein--10.

RULE 36--0.

S.B. 1220 (one thousand two hundred twenty) was read by title the third time and, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--14.

RULE 36--0.

S.B. 1256 (one thousand two hundred fifty-six) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1322 (one thousand three hundred twenty-two) was read by title the third time and, on motion of Senator DeSteph, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Kiggans, Lewis, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Barker, Howell, Locke, Lucas--4.

RULE 36--0.

S.B. 1343 (one thousand three hundred forty-three) was read by title the third time and, on motion of Senator Vogel, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

S.B. 1098 (one thousand ninety-eight).

S.B. 1126 (one thousand one hundred twenty-six).

S.B. 1172 (one thousand one hundred seventy-two).

S.B. 1214 (one thousand two hundred fourteen).

S.B. 1253 (one thousand two hundred fifty-three).

- **S.B.** 1277 (one thousand two hundred seventy-seven).
- **S.B. 1297** (one thousand two hundred ninety-seven).
- **S.B.** 1335 (one thousand three hundred thirty-five).
- S.B. 1408 (one thousand four hundred eight).
- **S.B. 1414** (one thousand four hundred fourteen).

The motion was agreed to.

S.B. 1335 (one thousand three hundred thirty-five) was taken up.

Senator Stuart offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 46.2-335, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to learner's permits; use of personal communication devices.

On motion of Senator Stuart, the reading of the substitute was waived.

On motion of Senator Stuart, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1098 (one thousand ninety-eight).
- **S.B.** 1126 (one thousand one hundred twenty-six).
- **S.B.** 1172 (one thousand one hundred seventy-two).
- S.B. 1214 (one thousand two hundred fourteen).
- S.B. 1253 (one thousand two hundred fifty-three).
- **S.B. 1277** (one thousand two hundred seventy-seven).
- S.B. 1297 (one thousand two hundred ninety-seven).
- S.B. 1335 (one thousand three hundred thirty-five) as amended.
- S.B. 1408 (one thousand four hundred eight).
- S.B. 1414 (one thousand four hundred fourteen).
- **S.B. 1355** (one thousand three hundred fifty-five) was taken up, the committee substitute having been agreed to on January 20, 2021.
 - S.B. 1355, on motion of Senator Hanger, was recommitted to the Committee on Local Government.
- **S.B. 1199** (one thousand one hundred ninety-nine) was taken up, the committee amendments having been offered on January 25, 2021.

Senator Petersen moved that the amendments be agreed to.

The question was put on agreeing to the amendments.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--23. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Peake, Petersen, Saslaw, Spruill, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Surovell--16.

RULE 36--0.

The amendments were agreed to.

Senator Stuart offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Chapter 10.1 of Title 10.1 a section numbered 10.1-1016.1 and by adding in Chapter 17 of Title 10.1 a section numbered 10.1-1705.1, relating to conservation easements; construction.

RULING OF THE CHAIR

The Chair ruled that the substitute offered by Senator Stuart to S.B. 1199 was out of order.

On motion of Senator Petersen, the bill was ordered to be engrossed and read by title the third time.

- **S.B.** 1136 (one thousand one hundred thirty-six), on motion of Senator Marsden, was passed by for the day.
 - S.B. 1229 (one thousand two hundred twenty-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Transportation was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 46.2-749.7 of the Code of Virginia, relating to special license plates for supporters of Ducks Unlimited; fees.

The reading of the substitute was waived.

On motion of Senator Stuart, the substitute was agreed to.

On motion of Senator Stuart, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1263** (one thousand two hundred sixty-three) was read by title the second time and, on motion of Senator Morrissey, was ordered to be engrossed and read by title the third time.
- **S.B. 1329** (one thousand three hundred twenty-nine) was read by title the second time and, on motion of Senator Mason, was ordered to be engrossed and read by title the third time.
- **S.B. 1366** (one thousand three hundred sixty-six), on motion of Senator Barker, was passed by for the day.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1248 (one thousand two hundred forty-eight).
- S.B. 1400 (one thousand four hundred).
- S.B. 1447 (one thousand four hundred forty-seven).
- S.B. 1457 (one thousand four hundred fifty-seven).
- S.B. 1206 (one thousand two hundred six).
- S.B. 1393 (one thousand three hundred ninety-three).

The motion was agreed to.

The recorded vote is as follows:

YEAS--36, NAYS--0, RULE 36--0.

YEAS--Barker, Bell, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--36.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- S.B. 1248 (one thousand two hundred forty-eight).
- S.B. 1400 (one thousand four hundred).
- S.B. 1447 (one thousand four hundred forty-seven).
- S.B. 1457 (one thousand four hundred fifty-seven).
- S.B. 1206 (one thousand two hundred six).
- S.B. 1393 (one thousand three hundred ninety-three).

SENATE JOINT RESOLUTIONS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate joint resolutions that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate joint resolutions, the titles of the joint resolutions having been printed in the Calendar for their second reading, be placed before the Senate by number only:

- S.J.R. 276 (two hundred seventy-six).
- S.J.R. 286 (two hundred eighty-six).
- S.J.R. 292 (two hundred ninety-two).
- S.J.R. 294 (two hundred ninety-four).
- S.J.R. 308 (three hundred eight).

The motion was agreed to.

S.J.R. 294 (two hundred ninety-four) was taken up.

The following amendments proposed by the Committee on Rules were offered:

```
1. Line 39, introduced, after November 30, [first instance] strike

2021
insert

2022
```

2. Line 39, introduced, after by November 30,

strike

2022

insert

2023

The reading of the amendments was waived.

On motion of Senator Lewis, the amendments were agreed to.

S.J.R. 308 (three hundred eight) was taken up.

The following amendments proposed by the Committee on Rules were offered:

1. At the beginning of line 73, introduced

strike

2021

insert

2022

2. Line 74, introduced, after of the

strike

2022

insert

2023

The reading of the amendments was waived.

On motion of Senator Lucas, the amendments were agreed to.

On motion of Senator Saslaw, the following Senate joint resolutions were ordered en bloc to be engrossed and read by title the third time:

```
S.J.R. 276 (two hundred seventy-six).
```

S.J.R. 286 (two hundred eighty-six).

S.J.R. 292 (two hundred ninety-two).

S.J.R. 294 (two hundred ninety-four) as amended.

S.J.R. 308 (three hundred eight) as amended.

S.J.R. 285 (two hundred eighty-five) was read by title the second time and, on motion of Senator Edwards, was ordered to be engrossed and read by title the third time.

SENATE RESOLUTION ON SECOND READING

S.R. 91 (ninety-one) was read by title the second time.

SENATE RESOLUTION NO. 91

Expressing the censure of the Senate.

WHEREAS, Section 3 of the Fourteenth Amendment to the Constitution of the United States provides that no person shall, "hold any office, civil or military, under the United States, or under any State, who, having previously taken an oath . . . as a member of any State legislature . . . to support the Constitution of the United States, shall have engaged in insurrection or rebellion against the same, or given aid or comfort to the enemies thereof"; and

WHEREAS, Senator Amanda F. Chase, a member of the Senate of Virginia since 2016, took such an oath; and

WHEREAS, on the morning of Wednesday, January 6, 2021, Senator Amanda F. Chase addressed a crowd gathered in Washington, D.C., to urge that action be taken to overturn the lawfully conducted 2020 presidential election; and

WHEREAS, the morning's events were a catalyst to the deadly insurrection that followed, when members of the crowd forcibly entered the United States Capitol where the Vice President of the United States and members of Congress were convening in a Joint Session of Congress to count the certified electoral votes for the President and Vice President of the United States, as prescribed by the Constitution of the United States for the orderly transition of presidential power; and

WHEREAS, in the aftermath of the insurrection that led to multiple deaths, including that of a Capitol Police officer; injured numerous others; desecrated the United States Capitol; threatened the lives and safety of those who are entrusted with carrying out the will of the American people; and required an emergency response from 200 Virginia State Troopers and 1,300 members of the Virginia National Guard, who were put in harm's way to quell the violence and restore order, Senator Amanda F. Chase continued to voice support for those who participated, calling them "[p]atriots who love their country" and propagating unfounded claims regarding the nature of the events, the identities of those who took part, and the validity of the presidential election; and

WHEREAS, Article IV, Section 7 of the Constitution of Virginia provides that each house of the General Assembly may "punish [its members] for disorderly behavior"; and

WHEREAS, Rules 18 (h) and 53 (b) of the Senate provide that the Senate may, by a majority vote, censure one of its members; and

WHEREAS, the inflammatory statements and actions of Senator Amanda F. Chase before, during, and after the events that led to the insurrection at the United States Capitol on January 6, 2021, constitute a failure to uphold her oath of office and conduct unbecoming of a Senator; now, therefore, be it

RESOLVED, by the Senate, That the Senate of Virginia does censure Senator Amanda F. Chase for fomenting insurrection against the United States.

Senator Bell offered an amendment in the nature of a substitute, having been printed separately, as follows:

SENATE RESOLUTION NO. 91 FLOOR AMENDMENT IN THE NATURE OF A SUBSTITUTE

Expressing the censure of the Senate.

WHEREAS, Senator Amanda F. Chase, a member of the Senate of Virginia since 2016, has exhibited conduct unbecoming of a Senator during her terms in office by displaying a disregard for civility in discourse with colleagues, making false and misleading statements both in committee and on the Senate floor, and displaying a disregard for the significance of her duty to the citizens of the Commonwealth as an elected representative in the Senate of Virginia; and

WHEREAS, Senator Amanda F. Chase's pattern of unacceptable conduct has been publicly displayed in a series of incendiary incidents during her tenure, including the following:

- 1. On March 22, 2019, Senator Chase berated a Capitol Police officer on duty when the Senator was not given access to a restricted parking area in front of the Capitol, reportedly stating, "Do you know who I am? . . . I'm Senator Chase . . . Don't you see it on my license plate? . . . But I guess you don't care." The report of the incident indicates that Senator Chase addressed the officer profanely and made offensive remarks regarding the Clerk of the Senate. Following the incident, Senator Chase suggested that the officer's actions, which were in the normal course of duty, had been racially motivated in reaction to the Senator's "white privilege"; and
- 2. Senator Chase used her social media page to recklessly identify the names and office contact information of colleagues, Democratic and Republican, whose legislation and votes she disagreed with. Expressing outrage, and in an attempt to intimidate fellow legislators over policy differences, Senator Chase urged her social media followers to voice their dissent directly, resulting in several legislators and aides receiving threats; and
- 3. In discussing on social media her personal choice to carry a weapon, Senator Chase blamed victims of sexual assault for not fending off their attackers, making the damaging and indefensible claim that "It's those who are naive and unprepared that end [up] raped"; and
- 4. After another Senator announced her candidacy for Governor of Virginia, Senator Chase decried "identity politics" in the gubernatorial race and suggested that, because her fellow Senator is vice-chair of the Virginia Legislative Black Caucus, she is "not for all Virginians"; and
- 5. During the COVID-19 pandemic, Senator Chase undermined the seriousness of the pandemic by stating, "I don't do COVID"; and
- 6. Following the 2020 presidential election, Senator Chase implicated both major political parties in baseless claims of a "stolen" election, asserting, "Make no mistake. We are at war. The Democratic Party hijacked our 2020 Presidential Election and [has] committed treason. Where the hell are the Republicans?"; and
- 7. Senator Chase claimed that "The Virginia Democratic Party is racist to its core," after Democratic officials asked a white local registrar to step down over issues relating to management of the 2020 election; and
- 8. In the aftermath of the unfortunate events and riots at the United States Capitol on January 6, 2021, Senator Chase voiced support for those who participated in storming the United States Capitol, calling them "[p]atriots who love their country" and propagating unfounded claims regarding the nature of the events, the identities of those who took part, and the validity of the presidential election. When given the opportunity to specifically denounce and repudiate those individuals at the January 6 riot who (i) were proponents of white supremacy; (ii) wore T-shirts and other attire emblazoned with anti-Semitic messaging; and (iii) breached the United States Capitol grounds barricades, broke into the Capitol when the United States Congress was in session, and disturbed and defaced the Capitol, Senator Chase declined to do so. The destruction and violence at the United States Capitol was the first occasion that the defenses of the United States Capitol have been breached since the War of 1812; led to multiple deaths, including that of a United States Capitol Police officer; injured numerous others; desecrated the United States Capitol; resulted in property destruction; threatened the lives and safety of those who are entrusted with carrying out the will of the American people; and required an emergency response from 200 Virginia State Troopers and 1,300 members of the Virginia National Guard, who were put in harm's way to quell the violence and restore order; and

WHEREAS, Article IV, Section 7 of the Constitution of Virginia provides that each house of the General Assembly may "punish [its members] for disorderly behavior"; and

WHEREAS, Rules 18 (h) and 53 (b) of the Senate provide that the Senate may, by a majority vote, censure one of its members and place the member last in seniority; and

WHEREAS, the inflammatory statements and actions of Senator Amanda F. Chase during her tenure in the Senate of Virginia have created and aggravated tensions, misled constituents and citizens, and obstructed the Senate's business in service of the Commonwealth, and such behavior constitutes a failure to uphold her oath of office, misuse of office, and conduct unbecoming of a Senator and, collectively, has caused a material effect upon the conduct of her office; now, therefore, be it

RESOLVED by the Senate, That the Senate of Virginia does censure Senator Amanda F. Chase and place Senator Amanda F. Chase last in seniority for failure to uphold her oath of office, misuse of office, and conduct unbecoming of a Senator.

On motion of Senator Bell, the reading of the substitute was waived.

Senator Bell moved that the substitute be agreed to.

RULING OF THE CHAIR

Senator Chase propounded a parliamentary inquiry as to whether the substitute offered by Senator Bell to S.R. 91 was germane.

RECESS

At 4:45 p.m., Senator Chase moved that the Senate recess until 5:25 p.m.

The motion was agreed to.

The hour of 5:25 p.m. having arrived, the Chair was resumed.

RULING OF THE CHAIR

The Chair ruled that the substitute offered by Senator Bell to S.R. 91 was germane.

On motion of Senator Bell, the substitute was agreed to.

Senator Bell moved that the resolution be ordered to be engrossed and read by title the third time.

S.R. 91, on motion of Senator Chase, was passed by for the day.

JOINT ORDER FOR ELECTIONS RESUMED

The Joint Order under House Joint Resolution No. 628 was resumed.

HOUSE COMMUNICATION

The following communication was received and read:

In the House of Delegates January 26, 2021

THE HOUSE OF DELEGATES HAS MADE CERTAIN NOMINATIONS (ATTACHED) PURSUANT TO H.J.R. 628.

/s/ Suzette Denslow Clerk of the House of Delegates

For judges of the respective circuit courts:
Brenda C. Spry, Third Judicial Circuit.
Junius P. Fulton, III, Fourth Judicial Circuit.
Kimberley S. White, Tenth Judicial Circuit.
Patricia Kelly, Fifteenth Judicial Circuit.
Michael E. Levy, Fifteenth Judicial Circuit.
Lisa Bondareff Kemler, Eighteenth Judicial Circuit.
John M. Tran, Nineteenth Judicial Circuit.

G. Carter Greer, Twenty-first Judicial Circuit.

James J. Reynolds, Twenty-second Judicial Circuit.

David B. Carson, Twenty-third Judicial Circuit.

Bruce D. Albertson, Twenty-sixth Judicial Circuit.

Deanis L. Simmons, Twenty-eighth Judicial Circuit.

Jack S. Hurley, Jr., Twenty-ninth Judicial Circuit.

Tammy S. McElyea, Thirtieth Judicial Circuit.

Carroll A. Weimer, Jr., Thirty-first Judicial Circuit.

For judges of the respective general district courts:

Vivian F. Henderson, Second Judicial District.

Tasha D. Scott, Fourth Judicial District.

Tyneka L. D. Flythe, Seventh Judicial District.

M. Scott Stein, Seventh Judicial District.

Tonya Henderson-Stith, Eighth Judicial District.

Pamela Y. O'Berry, Twelfth Judicial District.

David M. Hicks, Thirteenth Judicial District.

Jacqueline S. McClenney, Thirteenth Judicial District.

Mansi J. Shah, Thirteenth Judicial District.

B. Craig Dunkum, Fourteenth Judicial District.

John K. Honey, Jr., Fourteenth Judicial District.

Manuel A. Capsalis, Nineteenth Judicial District.

Michael J. Lindner, Nineteenth Judicial District.

William J. Minor, Jr., Nineteenth Judicial District.

Tina L. Snee, Nineteenth Judicial District.

Jacqueline F. Ward Talevi, Twenty-third Judicial District.

Stephanie S. Maddox, Twenty-fourth Judicial District.

Amy B. Tisinger, Twenty-sixth Judicial District.

Erin J. DeHart, Twenty-seventh Judicial District.

Gino W. Williams, Twenty-seventh Judicial District.

Robert P. Coleman, Thirty-first Judicial District.

For judges of the respective juvenile and domestic relations district courts:

Lyn M. Simmons, Fourth Judicial District.

Jacqueline R. Waymack, Sixth Judicial District.

John E. Franklin, Fifteenth Judicial District.

Frank G. Uvanni, Fifteenth Judicial District.

Thomas P. Sotelo, Nineteenth Judicial District.

Timothy W. Allen, Twenty-second Judicial District.

Hilary D. Griffith, Twenty-third Judicial District.

Melissa W. Friedman, Twenty-third Judicial District.

Jeffrey P. Bennett, Twenty-fourth Judicial District.

Hugh David O'Donnell, Twenty-sixth Judicial District.

Robert C. Viar, Jr., Twenty-seventh Judicial District.

Joseph B. Lyle, Twenty-eighth Judicial District.

Michael J. Bush, Twenty-ninth Judicial District.

Martha P. Ketron, Twenty-ninth Judicial District.

For a member of the Judicial Inquiry and Review Commission:

Terrie N. Thompson.

For members of the State Corporation Commission:

Angela L. Navarro.

Jehmal T. Hudson.

For the Auditor of Public Accounts:

Staci A. Henshaw.

The roll was called with the following results:

For judges of the respective circuit courts for the terms set forth:

The nominees by Senate Resolution No. 99, excluding lines 10 and 11, received an affirmative vote of 37.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--Reeves--1.

The nominee by Senate Resolution No. 99, lines 10 and 11, as follows:

Brenda C. Spry, Esquire, of Portsmouth, as a judge of the Third Judicial Circuit for a term of eight years commencing February 16, 2021

received an affirmative vote of 33.

The recorded vote is as follows:

YEAS--33. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--33.

NAYS--0.

RULE 36--0.

For judges of the respective general district courts for the terms set forth:

The nominees by Senate Resolution No. 100 received an affirmative vote of 38.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

For judges of the respective juvenile and domestic relations district courts for the terms set forth:

The nominees by Senate Resolution No. 101 received an affirmative vote of 38.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

For a member of the Judicial Inquiry and Review Commission for the term set forth:

The nominee by Senate Resolution No. 102 received an affirmative vote of 38.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

For members of the State Corporation Commission for the terms set forth:

The nominees by Senate Resolution No. 103 received an affirmative vote of 32.

The recorded vote is as follows:

YEAS--32. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--32.

NAYS--0.

RULE 36--0.

For the Auditor of Public Accounts for the term set forth:

The nominee by Senate Resolution No. 93 received an affirmative vote of 38.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

Subsequently, the committee reported as follows:

The President appointed Senators Boysko, Morrissey, and Peake, the committee on the part of the Senate to count and report the vote of each house in each case.

Whole number of votes necessary to elect: In the House of Delegates.....51 In the Senate 20 For a judge of the Third Judicial Circuit for a term of eight years commencing February 16, 2021: Brenda C. Spry received: In the House of Delegates.....51 For a judge of the Fourth Judicial Circuit for a term of eight years commencing February 1, 2021: Junius P. Fulton, III received: In the House of Delegates.....96 For a judge of the Tenth Judicial Circuit for a term of eight years commencing February 1, 2021: Kimberley S. White received: In the House of Delegates....96 For a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021: Patricia Kelly received: In the House of Delegates.....96 For a judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021: Michael E. Levy received: In the House of Delegates.....96 For a judge of the Eighteenth Judicial Circuit for a term of eight years commencing March 1, 2021: Lisa Bondareff Kemler received: In the House of Delegates....96

For a judge of the Nineteenth Judicial Circuit for a term of eight years commencing July 1, 2021: John M. Tran received: In the House of Delegates.....96 For a judge of the Twenty-first Judicial Circuit for a term of eight years commencing March 1, 2021: G. Carter Greer received: In the House of Delegates....96 For a judge of the Twenty-second Judicial Circuit for a term of eight years commencing July 1, 2021: James J. Reynolds received: In the House of Delegates.....96 For a judge of the Twenty-third Judicial Circuit for a term of eight years commencing July 1, 2021: David B. Carson received: In the House of Delegates.....96 For a judge of the Twenty-sixth Judicial Circuit for a term of eight years commencing July 1, 2021: Bruce D. Albertson received: In the House of Delegates....96 For a judge of the Twenty-eighth Judicial Circuit for a term of eight years commencing July 1, 2021: Deanis L. Simmons received: In the House of Delegates.....96 For a judge of the Twenty-ninth Judicial Circuit for a term of eight years commencing February 1, 2021: Jack S. Hurley, Jr. received: In the House of Delegates....96 For a judge of the Thirtieth Judicial Circuit for a term of eight years commencing April 1, 2021:

Tammy S. McElyea received:
In the House of Delegates96 In the Senate
For a judge of the Thirty-first Judicial Circuit for a term of eight years commencing July 1, 2021:
Carroll A. Weimer, Jr. received:
In the House of Delegates51 In the Senate37
For a judge of the General District Court of the Second Judicial District for a term of six year commencing February 1, 2021:
Vivian F. Henderson received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Fourth Judicial District for a term of six year commencing July 1, 2021:
Tasha D. Scott received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Seventh Judicial District for a term of six year commencing July 1, 2021:
Tyneka L. D. Flythe received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Seventh Judicial District for a term of six year commencing July 1, 2021:
M. Scott Stein received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Eighth Judicial District for a term of six year commencing May 1, 2021:
Tonya Henderson-Stith received:
In the House of Delegates 98 In the Senate

For a judge of the General District Court of the Twelfth Judicial District for a term of six years commencing April 1, 2021:

Pamela Y. O'Berry received:

```
In the House of Delegates.....98
In the Senate......No vote taken
```

For a judge of the General District Court of the Thirteenth Judicial District for a term of six years commencing July 1, 2021:

David M. Hicks received:

```
In the House of Delegates.....98
In the Senate..............38
```

For a judge of the General District Court of the Thirteenth Judicial District for a term of six years commencing December 1, 2021:

Jacqueline S. McClenney received:

For a judge of the General District Court of the Thirteenth Judicial District for a term of six years commencing February 1, 2021:

Mansi J. Shah received:

```
In the House of Delegates.....98
In the Senate.............38
```

For a judge of the General District Court of the Fourteenth Judicial District for a term of six years commencing July 1, 2021:

B. Craig Dunkum received:

```
In the House of Delegates.... 98
In the Senate..... 38
```

For a judge of the General District Court of the Fourteenth Judicial District for a term of six years commencing November 1, 2021:

John K. Honey, Jr. received:

```
In the House of Delegates.....98
In the Senate.............38
```

For a judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing July 1, 2021:

Manuel A. Capsalis received:

In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing July 1, 2021:
Michael J. Lindner received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing April 1, 2021:
William J. Minor, Jr. received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing July 1, 2021:
Tina L. Snee received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Twenty-third Judicial District for a term of six years commencing February 1, 2021:
Jacqueline F. Ward Talevi received:
In the House of Delegates 98 In the Senate 38
For a judge of the General District Court of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021:
Stephanie S. Maddox received:
In the House of Delegates 98 In the Senate
For a judge of the General District Court of the Twenty-sixth Judicial District for a term of six years commencing July 1, 2021:
Amy B. Tisinger received:

For a judge of the General District Court of the Twenty-seventh Judicial District for a term of six years commencing July 1, 2021:

Erin J. DeHart received:

In the House of Delegates.....98
In the Senate.......38

For a judge of the General District Court of the Twenty-seventh Judicial District for a term of six years commencing April 1, 2021:

Gino W. Williams received:

In the House of Delegates.....98 In the Senate.............38

For a judge of the General District Court of the Thirty-first Judicial District for a term of six years commencing July 1, 2021:

Robert P. Coleman received:

In the House of Delegates.....98
In the Senate.............38

For a judge of the Juvenile and Domestic Relations District Court of the Fourth Judicial District for a term of six years commencing September 17, 2021:

Lyn M. Simmons received:

In the House of Delegates.... 98 In the Senate..... 38

For a judge of the Juvenile and Domestic Relations District Court of the Sixth Judicial District for a term of six years commencing June 1, 2021:

Jacqueline R. Waymack received:

In the House of Delegates....96
In the Senate.....38

For a judge of the Juvenile and Domestic Relations District Court of the Fifteenth Judicial District for a term of six years commencing July 1, 2021:

John E. Franklin received:

In the House of Delegates.....96 In the Senate............38

For a judge of the Juvenile and Domestic Relations District Court of the Fifteenth Judicial District for a term of six years commencing July 1, 2021:

Frank G. Uvanni received:

In the House of Delegates.....96 For a judge of the Juvenile and Domestic Relations District Court of the Nineteenth Judicial District for a term of six years commencing February 1, 2021: Thomas P. Sotelo received: In the House of Delegates....96 For a judge of the Juvenile and Domestic Relations District Court of the Twenty-second Judicial District for a term of six years commencing July 1, 2021: Timothy W. Allen received: In the House of Delegates.....96 For a judge of the Juvenile and Domestic Relations District Court of the Twenty-third Judicial District for a term of six years commencing July 1, 2021: Hilary D. Griffith received: In the House of Delegates.....96 For a judge of the Juvenile and Domestic Relations District Court of the Twenty-third Judicial District for a term of six years commencing February 1, 2021: Melissa W. Friedman received: In the House of Delegates.....96 For a judge of the Juvenile and Domestic Relations District Court of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021: Jeffrey P. Bennett received: In the House of Delegates....96

For a judge of the Juvenile and Domestic Relations District Court of the Twenty-sixth Judicial District for a term of six years commencing April 1, 2021:

Hugh David O'Donnell received:

In the House of Delegates.....96 In the Senate............38 For a judge of the Juvenile and Domestic Relations District Court of the Twenty-seventh Judicial District for a term of six years commencing May 1, 2021:

For a judge of the Juvenile and Domestic Relations District Court of the Twenty-eighth Judicial District for a term of six years commencing July 1, 2021:

Joseph B. Lyle received:

In the House of Delegates.....96 In the Senate............38

For a judge of the Juvenile and Domestic Relations District Court of the Twenty-ninth Judicial District for a term of six years commencing April 1, 2021:

Michael J. Bush received:

In the House of Delegates....96
In the Senate.....38

For a judge of the Juvenile and Domestic Relations District Court of the Twenty-ninth Judicial District for a term of six years commencing February 1, 2021:

Martha P. Ketron received:

In the House of Delegates.....96 In the Senate............38

For a member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021:

Terrie N. Thompson received:

In the House of Delegates.....99
In the Senate.................38

For a member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2022:

Angela L. Navarro received:

For a member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2026:

Jehmal T. Hudson received:

In the House of Delegates.....96 In the Senate.......................32

For the Auditor of Public Accounts for a term of four years commencing February 1, 2021:

Staci A. Henshaw received:

In the House of Delegates.....98
In the Senate38

On motion of Senator Edwards, the reading of the report was waived.

The recorded vote is as follows: YEAS--33. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Petersen, Pillion, Ruff, Saslaw, Spruill, Suetterlein, Surovell, Vogel--33.

NAYS--Deeds--1. RULE 36--0.

The nominees, having received the vote of a majority of the members elected to each house of the General Assembly, were declared by the President duly elected judges of the respective circuit courts, general district courts, and juvenile and domestic relations district courts; a member of the Judicial Inquiry and Review Commission; members of the State Corporation Commission; and the Auditor of Public Accounts, as follows:

Brenda C. Spry, judge of the Third Judicial Circuit for a term of eight years commencing February 16, 2021.

Junius P. Fulton, III, judge of the Fourth Judicial Circuit for a term of eight years commencing February 1, 2021.

Kimberley S. White, judge of the Tenth Judicial Circuit for a term of eight years commencing February 1, 2021.

Patricia Kelly, judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

Michael E. Levy, judge of the Fifteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

Lisa Bondareff Kemler, judge of the Eighteenth Judicial Circuit for a term of eight years commencing March 1, 2021.

John M. Tran, judge of the Nineteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

G. Carter Greer, judge of the Twenty-first Judicial Circuit for a term of eight years commencing March 1, 2021.

James J. Reynolds, judge of the Twenty-second Judicial Circuit for a term of eight years commencing July 1, 2021.

David B. Carson, judge of the Twenty-third Judicial Circuit for a term of eight years commencing July 1, 2021.

- Bruce D. Albertson, judge of the Twenty-sixth Judicial Circuit for a term of eight years commencing July 1, 2021.
- Deanis L. Simmons, judge of the Twenty-eighth Judicial Circuit for a term of eight years commencing July 1, 2021.
- Jack S. Hurley, Jr., judge of the Twenty-ninth Judicial Circuit for a term of eight years commencing February 1, 2021.
- Tammy S. McElyea, judge of the Thirtieth Judicial Circuit for a term of eight years commencing April 1, 2021.
- Carroll A. Weimer, Jr., judge of the Thirty-first Judicial Circuit for a term of eight years commencing July 1, 2021.
- Vivian F. Henderson, judge of the General District Court of the Second Judicial District for a term of six years commencing February 1, 2021.
- Tasha D. Scott, judge of the General District Court of the Fourth Judicial District for a term of six years commencing July 1, 2021.
- Tyneka L. D. Flythe, judge of the General District Court of the Seventh Judicial District for a term of six years commencing July 1, 2021.
- M. Scott Stein, judge of the General District Court of the Seventh Judicial District for a term of six years commencing July 1, 2021.
- Tonya Henderson-Stith, judge of the General District Court of the Eighth Judicial District for a term of six years commencing May 1, 2021.
- David M. Hicks, judge of the General District Court of the Thirteenth Judicial District for a term of six years commencing July 1, 2021.
- Jacqueline S. McClenney, judge of the General District Court of the Thirteenth Judicial District for a term of six years commencing December 1, 2021.
- Mansi J. Shah, judge of the General District Court of the Thirteenth Judicial District for a term of six years commencing February 1, 2021.
- B. Craig Dunkum, judge of the General District Court of the Fourteenth Judicial District for a term of six years commencing July 1, 2021.
- John K. Honey, Jr., judge of the General District Court of the Fourteenth Judicial District for a term of six years commencing November 1, 2021.
- Manuel A. Capsalis, judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.
- Michael J. Lindner, judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.
- William J. Minor, Jr., judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing April 1, 2021.
- Tina L. Snee, judge of the General District Court of the Nineteenth Judicial District for a term of six years commencing July 1, 2021.

- Jacqueline F. Ward Talevi, judge of the General District Court of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.
- Stephanie S. Maddox, judge of the General District Court of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.
- Amy B. Tisinger, judge of the General District Court of the Twenty-sixth Judicial District for a term of six years commencing July 1, 2021.
- Erin J. DeHart, judge of the General District Court of the Twenty-seventh Judicial District for a term of six years commencing July 1, 2021.
- Gino W. Williams, judge of the General District Court of the Twenty-seventh Judicial District for a term of six years commencing April 1, 2021.
- Robert P. Coleman, judge of the General District Court of the Thirty-first Judicial District for a term of six years commencing July 1, 2021.
- Lyn M. Simmons, judge of the Juvenile and Domestic Relations District Court of the Fourth Judicial District for a term of six years commencing September 17, 2021.
- Jacqueline R. Waymack, judge of the Juvenile and Domestic Relations District Court of the Sixth Judicial District for a term of six years commencing June 1, 2021.
- John E. Franklin, judge of the Juvenile and Domestic Relations District Court of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.
- Frank G. Uvanni, judge of the Juvenile and Domestic Relations District Court of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.
- Thomas P. Sotelo, judge of the Juvenile and Domestic Relations District Court of the Nineteenth Judicial District for a term of six years commencing February 1, 2021.
- Timothy W. Allen, judge of the Juvenile and Domestic Relations District Court of the Twenty-second Judicial District for a term of six years commencing July 1, 2021.
- Hilary D. Griffith, judge of the Juvenile and Domestic Relations District Court of the Twenty-third Judicial District for a term of six years commencing July 1, 2021.
- Melissa W. Friedman, judge of the Juvenile and Domestic Relations District Court of the Twenty-third Judicial District for a term of six years commencing February 1, 2021.
- Jeffrey P. Bennett, judge of the Juvenile and Domestic Relations District Court of the Twenty-fourth Judicial District for a term of six years commencing July 1, 2021.
- Hugh David O'Donnell, judge of the Juvenile and Domestic Relations District Court of the Twenty-sixth Judicial District for a term of six years commencing April 1, 2021.
- Robert C. Viar, Jr., judge of the Juvenile and Domestic Relations District Court of the Twenty-seventh Judicial District for a term of six years commencing May 1, 2021.
- Joseph B. Lyle, judge of the Juvenile and Domestic Relations District Court of the Twenty-eighth Judicial District for a term of six years commencing July 1, 2021.
- Michael J. Bush, judge of the Juvenile and Domestic Relations District Court of the Twenty-ninth Judicial District for a term of six years commencing April 1, 2021.

Martha P. Ketron, judge of the Juvenile and Domestic Relations District Court of the Twenty-ninth Judicial District for a term of six years commencing February 1, 2021.

Terrie N. Thompson, member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.

Angela L. Navarro, to succeed Mark C. Christie as a member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2022.

Jehmal T. Hudson, member of the State Corporation Commission for an unexpired term commencing February 1, 2021, and ending January 31, 2026.

Staci A. Henshaw, Auditor of Public Accounts for a term of four years commencing February 1, 2021.

No nominees having received the vote of a majority of the members elected to each house of the General Assembly, it was declared by the President that no election resulted for the offices listed as follows:

Judge of the General District Court of the Twelfth Judicial District for a term of six years commencing April 1, 2021.

A member of the Judicial Inquiry and Review Commission for a term of four years commencing July 1, 2021.

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

san Carle Delman

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

WEDNESDAY, JANUARY 27, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Frank M. Ruff, Jr., Fifteenth Senatorial District, offered the following prayer:

As we were instructed in Deuteronomy 6:18, "you shall do what is right and good in the sight of the Lord so that it may go well for you."

Heavenly Father, we come before You asking that You lead us through heart, mind, and soul to do that which is right and pleasing to You. We ask that You give us the determination to follow Your guidance in our treatment of one another and that we deal with honesty and integrity. As well, we ask for Your guidance as we do our work for the people of Virginia. You know that at times we will fail, but with Your help, we will rise up to continue our service to You and for those we have been entrusted to represent. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Michael Adams, Director of Human Resources, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows: YEAS--36. NAYS--0. RULE 36--0.

YEAS--Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--36.

NAYS--0. RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Suetterlein, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--35.

NAYS--Deeds, McDougle, Petersen, Suetterlein--4. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 26, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1754.** A BILL to amend and reenact §§ 40.1-27.3 and 65.2-308 of the Code of Virginia, relating to employment; retaliatory discharge by employer; Workers' Compensation.
- **H.B. 1811.** A BILL to amend the Code of Virginia by adding a section numbered 2.2-4328.1, relating to the Virginia Public Procurement Act; preference for energy-efficient and water-efficient goods.
- H.B. 1819. A BILL to amend and reenact § 10.1-415 of the Code of Virginia, relating to Rappahannock State Scenic River.
- **H.B. 1828.** A BILL to amend and reenact § 46.2-223 of the Code of Virginia, relating to the Commissioner of the Department of Motor Vehicles; powers and duties.
- **H.B. 1831.** A BILL to amend and reenact § 32.1-162.12 of the Code of Virginia, relating to home care organizations; personal care services; supervision; regulations.
- H.B. 1841. A BILL to direct the Department of Transportation to convene a working group to determine model policies for crosswalk design; report.
- **H.B. 1868.** A BILL to amend and reenact §§ 46.2-341.18, 46.2-382, and 46.2-1702 of the Code of Virginia, relating to commercial driver's licenses.
- **H.B. 1874.** A BILL to amend and reenact § 53.1-68 of the Code of Virginia, relating to behavioral health assessments in local correctional facilities.
- **H.B. 1887.** A BILL to amend and reenact § 46.2-602 of the Code of Virginia, relating to titling and registration of foreign market vehicles.
- **H.B. 1896.** A BILL to amend and reenact § 38.2-3451 of the Code of Virginia, relating to health insurance; essential health benefits; abortion coverage.
- **H.B. 1901.** A BILL to amend and reenact § 46.2-325 of the Code of Virginia, relating to online Virginia Driver's Manual course; training school.
- H.B. 1913. A BILL to amend and reenact §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909 of the Code of Virginia, relating to programs to address career fatigue and wellness in certain health care providers; civil immunity.
 EMERGENCY
- **H.B. 1931.** A BILL to amend and reenact § 2.2-3708.2 of the Code of Virginia, relating to the Virginia Freedom of Information Act; electronic meetings.
- **H.B. 1943.** A BILL to amend and reenact §§ 18.2-340.19 and 18.2-340.28 of the Code of Virginia, relating to the Charitable Gaming Board; regulations; electronic pull tabs.

- **H.B. 1944.** A BILL to amend and reenact § 58.1-4110 of the Code of Virginia, relating to casino gaming; requirements for issuance of operator's license; human trafficking training.
- **H.B. 1960.** A BILL to amend and reenact § 46.2-600.1 of the Code of Virginia, relating to vehicle registration; special communication needs indicator.
- **H.B. 1961.** A BILL to amend and reenact § 46.2-345 of the Code of Virginia, relating to special identification cards; application by guardian.
- **H.B. 1976.** A BILL to amend and reenact §§ 32.1-122.7 and 32.1-122.7:1 of the Code of Virginia, relating to Virginia Health Workforce Development Authority; mission; membership.
- **H.B. 1996.** A BILL to amend and reenact § 2.2-4302.1 of the Code of Virginia, relating to the Virginia Public Procurement Act; determination of responsibility; local option to include criteria in Invitation to Bid.
- **H.B. 2025.** A BILL to amend and reenact § 2.2-3705.1 of the Code of Virginia, relating to the Virginia Freedom of Information Act; record exclusion for personal contact information provided to a public body.
- **H.B. 2059.** A BILL to amend the Code of Virginia by adding a section numbered 58.1-1802.2, relating to delinquent returns; enforcement; when approval required.
- **H.B. 2061.** A BILL to amend and reenact § 32.1-46.01 of the Code of Virginia, relating to Virginia Immunization Information System; health care entities; required participation.
- **H.B. 2070.** A BILL to amend and reenact § 37.2-500 of the Code of Virginia, relating to community services boards; contracts with private providers.
- H.B. 2116. A BILL to amend the Code of Virginia by adding sections numbered 32.1-42.2 and 44-146.17:1.1, relating to declaration of emergency; essential workers; access to personal protective equipment and immunization; emergency.
 EMERGENCY

H.B. 2121. A BILL to amend and reenact §§ 13.1-609 and 13.1-610, as they shall become effective, § 13.1-615, § 13.1-615.1, as it shall become effective, §§ 13.1-625 and 13.1-628, §§ 13.1-630 and 13.1-636, as they shall become effective, §§ 13.1-639, 13.1-658, 13.1-661, 13.1-710, 13.1-711, and 13.1-716, §§ 13.1-718, 13.1-721.1, 13.1-722.6, and 13.1-722.12:1, as they shall become effective, §§ 13.1-759, 13.1-765, 13.1-775.1, 13.1-803, 13.1-806, 13.1-807, 13.1-809, 13.1-815, 13.1-815.1, 13.1-816, 13.1-829, 13.1-830, 13.1-831, 13.1-835, 13.1-894, 13.1-897.1, 13.1-898.7, 13.1-921, 13.1-927, 13.1-936.1, and 13.1-944.7, §§ 13.1-1002, 13.1-1004, 13.1-1005, and 13.1-1012, as they shall become effective, §§ 13.1-1017, 13.1-1052, and 13.1-1054, §§ 13.1-1062 and 13.1-1065, as they shall become effective, §§ 13.1-1073.1, 13.1-1074, 13.1-1075, 13.1-1080, and 13.1-1087, §§ 13.1-1096, 13.1-1099.14, and 13.1-1099.26, as they shall become effective, and §§ 13.1-1201, 13.1-1203, 13.1-1212, 13.1-1214, 13.1-1222, 13.1-1242, 13.1-1252, 13.1-1255, 13.1-1264, $13.1-1265,\ 13.1-1271,\ 13.1-1277,\ 15.2-5112,\ 15.2-5431.9,\ 50-73.1,\ 50-73.2,\ 50-73.6,\ 50-73.17,$ 50-73.54, 50-73.67, 50-73.70, 50-73.83, and 50-73.135 of the Code of Virginia; to amend the Code of Virginia by adding in Article 11.1 of Chapter 10 of Title 13.1 a section numbered 13.1-898.1:1, by adding in Article 11 of Chapter 14 of Title 13.1 a section numbered 13.1-1263.1, by adding in Chapter 54.1 of Title 15.2 sections numbered 15.2-5431.8:1, 15.2-5431.9:1, and 15.2-5431.35:1, and by adding in Article 7.1 of Chapter 2.1 of Title 50 a section numbered 50-73.48:5; and to repeal Article 17 (§§ 13.1-941.01 through 13.1-944) of Chapter 10 of Title 13.1 of the Code of Virginia, relating to business entities; filings with the State Corporation Commission; Virginia Stock Corporation Act.

- **H.B. 2138.** A BILL to amend and reenact §§ 24.2-411.3, 24.2-643, 46.2-203.2, 46.2-216.1, 46.2-323.01, 46.2-323.1, 46.2-346, 46.2-600.1, and 58.1-3 of the Code of Virginia and to amend the Code of Virginia by adding in Article 7 of Chapter 3 of Title 46.2 a section numbered 46.2-345.3, relating to identification privilege cards; fee; confidentiality; penalties.
- **H.B. 2174.** A BILL to amend and reenact § 23.1-701 of the Code of Virginia and to amend the Code of Virginia by adding in Title 2.2 a chapter numbered 27.1, consisting of sections numbered 2.2-2744 through 2.2-2756, relating to VirginiaSaves Program; establishment.
- H.B. 2197. A BILL to require the Department of Medical Assistance Services to establish a work group to study options for the permanent use of virtual supports and increasing access to virtual supports and services for individuals with intellectual and developmental disabilities.
- **H.B. 2333.** A BILL to facilitate the administration of the COVID-19 vaccine; emergency. EMERGENCY
- THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTIONS:
- **H.J.R. 525.** Authorizing, and directing the submission to the Joint Committee of Congress on the Library, that the vacant spot of the Commonwealth in the National Statuary Hall Collection at the United States Capitol be filled with a statue to commemorate Barbara Rose Johns.
- **H.J.R. 527.** Requesting the Department of Conservation and Recreation, jointly with the Virginia Department of Agriculture and Consumer Services, to study the sale and use of invasive plant species. Report.
- H.J.R. 537. Recognizing that racism is a public health crisis in Virginia.
- H.J.R. 538. Recognizing that access to clean, potable, and affordable water is a necessary human right.
- **H.J.R. 549.** Directing the Joint Legislative Audit and Review Commission to study the impact of COVID-19 on Virginia's public schools, students, and school employees. Report.
- **H.J.R. 562.** Designating August 31, in 2021 and in each succeeding year, as International Overdose Awareness Day in Virginia and directing that flags be lowered to half-mast on August 31.
- **H.J.R. 567.** Directing the Joint Legislative Audit and Review Commission to study increasing the progressivity of Virginia's individual income tax system. Report.
- H.J.R. 572. Extending state recognition to the Wolf Creek Cherokee Tribe of Virginia.
- **H.J.R. 579.** Confirming an appointment by the Speaker of the House of Delegates to the Virginia Commonwealth University Health System Authority Board of Directors.
- **H.J.R. 583.** Designating June 19 through the third Monday in July, in 2021 and in each succeeding year, as Liberty Amendments Month in Virginia.

- **H.J.R. 596.** Designating September, in 2021 and in each succeeding year, as Brain Aneurysm Awareness Month in Virginia.
- **H.J.R. 604.** Confirming the appointment of Hal E. Greer as Director of the Joint Legislative Audit and Review Commission.
- **H.J.R. 605.** Designating March 14, in 2021 and in each succeeding year, as Victims of COVID-19 Remembrance Day in Virginia.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--36. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Deeds, McDougle, Petersen--3.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1754, H.B. 1896, and H.B. 2121 were referred to the Committee on Commerce and Labor.
- H.B. 1811, H.B. 1931, H.B. 1943, H.B. 1944, H.B. 1996, and H.B. 2025 were referred to the Committee on General Laws and Technology.
 - H.B. 1819 was referred to the Committee on Agriculture, Conservation and Natural Resources.
- H.B. 1828, H.B. 1841, H.B. 1868, H.B. 1887, H.B. 1901, H.B. 1960, H.B. 1961, and H.B. 2138 were referred to the Committee on Transportation.
- **H.B. 1831, H.B. 1874, H.B. 2070,** and **H.B. 2197** were referred to the Committee on Rehabilitation and Social Services.
- H.B. 1913, H.B. 1976, H.B. 2061, H.B. 2116, and H.B. 2333 were referred to the Committee on Education and Health.
 - **H.B. 2059** and **H.B. 2174** were referred to the Committee on Finance and Appropriations.

The House joint resolutions, communicated as agreed to by the House of Delegates, the first reading of their titles having been waived, were referred as follows:

H.J.R. 525, H.J.R. 527, H.J.R. 537, H.J.R. 538, H.J.R. 549, H.J.R. 562, H.J.R. 567, H.J.R. 572, H.J.R. 579, H.J.R. 583, H.J.R. 596, H.J.R. 604, and H.J.R. 605 were referred to the Committee on Rules.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Petersen from the Committee on Agriculture, Conservation and Natural Resources:

- **S.B.** 1115 (one thousand one hundred fifteen) with substitute.
- **S.B.** 1135 (one thousand one hundred thirty-five) with substitute.
- **S.B.** 1143 (one thousand one hundred forty-three).
- **S.B. 1210** (one thousand two hundred ten) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1282** (one thousand two hundred eighty-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1379 (one thousand three hundred seventy-nine) with substitute.
 - S.B. 1390 (one thousand three hundred ninety) with amendments.
- **S.B. 1396** (one thousand three hundred ninety-six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1412 (one thousand four hundred twelve) with substitute.
 - S.B. 1417 (one thousand four hundred seventeen) with substitute.
 - S.B. 1453 (one thousand four hundred fifty-three).

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- S.B. 1103 (one thousand one hundred three).
- S.B. 1104 (one thousand one hundred four).
- S.B. 1106 (one thousand one hundred six) with substitute.
- S.B. 1108 (one thousand one hundred eight) with substitute.
- **S.B.** 1119 (one thousand one hundred nineteen).
- S.B. 1125 (one thousand one hundred twenty-five).
- S.B. 1188 (one thousand one hundred eighty-eight).
- S.B. 1193 (one thousand one hundred ninety-three) with substitute.
- **S.B. 1254** (one thousand two hundred fifty-four) with substitute. **S.B. 1279** (one thousand two hundred seventy-nine).
- S.B. 1300 (one thousand three hundred).
- **S.B.** 1305 (one thousand three hundred five) with amendment.
- **S.B. 1314** (one thousand three hundred fourteen).
- S.B. 1387 (one thousand three hundred eighty-seven).
- S.B. 1402 (one thousand four hundred two).
- S.B. 1423 (one thousand four hundred twenty-three) with substitute.

The following bills and joint resolution, having been considered by the committee in session, were reported by Senator Deeds from the Committee on Privileges and Elections:

- S.B. 1148 (one thousand one hundred forty-eight).
- S.B. 1245 (one thousand two hundred forty-five) with substitute.
- S.B. 1246 (one thousand two hundred forty-six) with substitute.
- **S.B. 1331** (one thousand three hundred thirty-one).
- **S.B. 1395** (one thousand three hundred ninety-five) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

S.B. 1422 (one thousand four hundred twenty-two).

S.J.R. 327 (three hundred twenty-seven).

S.B. 1210, S.B. 1282, S.B. 1395, and S.B. 1396 were rereferred to the Committee on Finance and Appropriations.

CALENDAR

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1098 (one thousand ninety-eight).
- **S.B.** 1126 (one thousand one hundred twenty-six).
- **S.B.** 1172 (one thousand one hundred seventy-two).
- S.B. 1214 (one thousand two hundred fourteen).
- S.B. 1253 (one thousand two hundred fifty-three).
- S.B. 1277 (one thousand two hundred seventy-seven).
- **S.B. 1297** (one thousand two hundred ninety-seven).
- S.B. 1335 (one thousand three hundred thirty-five).
- S.B. 1408 (one thousand four hundred eight).
- **S.B. 1414** (one thousand four hundred fourteen).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- S.B. 1098 (one thousand ninety-eight).
- **S.B.** 1126 (one thousand one hundred twenty-six).
- **S.B.** 1172 (one thousand one hundred seventy-two).
- S.B. 1214 (one thousand two hundred fourteen).
- S.B. 1253 (one thousand two hundred fifty-three).
- S.B. 1277 (one thousand two hundred seventy-seven).
- **S.B.** 1297 (one thousand two hundred ninety-seven).
- S.B. 1335 (one thousand three hundred thirty-five).
- S.B. 1408 (one thousand four hundred eight).
- S.B. 1414 (one thousand four hundred fourteen).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1199 (one thousand one hundred ninety-nine) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Peake, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Reeves, Ruff, Stanley, Stuart, Suetterlein--14.

RULE 36--0.

S.B. 1229 (one thousand two hundred twenty-nine) was read by title the third time and, on motion of Senator Stuart, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1.

RULE 36--0.

S.B. 1263 (one thousand two hundred sixty-three) was read by title the third time.

Senator Morrissey moved that **S.B. 1263** be passed with its title.

The question was put on passing S.B. 1263 with its title.

S.B. 1263 was defeated with its title.

The recorded vote is as follows:

YEAS--16. NAYS--22. RULE 36--0.

YEAS--Barker, Bell, Boysko, DeSteph, Ebbin, Favola, Hashmi, Lewis, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Spruill, Surovell--16.

NAYS--Chase, Cosgrove, Deeds, Dunnavant, Edwards, Hanger, Howell, Kiggans, Locke, Lucas, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Stuart, Suetterlein, Vogel--22.

RULE 36--0.

S.B. 1329 (one thousand three hundred twenty-nine) was read by title the third time and, on motion of Senator Mason, was passed with its title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

RECESS

At 12:45 p.m., Senator Norment moved that the Senate recess until 1:20 p.m.

The motion was agreed to.

The hour of 1:20 p.m. having arrived, the Chair was resumed.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1248 (one thousand two hundred forty-eight).
- S.B. 1400 (one thousand four hundred).
- S.B. 1447 (one thousand four hundred forty-seven).
- **S.B.** 1457 (one thousand four hundred fifty-seven).

The motion was agreed to.

S.B. 1248 (one thousand two hundred forty-eight) was taken up.

The following amendment proposed by the Committee on the Judiciary was offered:

1. Line 50, introduced, after *order*. insert

If the appointed evaluator or the director of the community services board, behavioral health authority, hospital, or private evaluator is unable to conduct the evaluation, he shall inform the court on the acknowledgment form.

The reading of the amendment was waived.

On motion of Senator Deeds, the amendment was agreed to.

S.B. 1457 (one thousand four hundred fifty-seven) was taken up.

The following amendments proposed by the Committee on Local Government were offered:

```
1. Line 114, introduced, after section
strike
requiring
insert
that would allow

2. Line 119, introduced, after park
strike
, (ii) owned by a common owner exempt from taxation, and (iii)
insert
and (ii)
```

The reading of the amendments was waived.

On motion of Senator Surovell, the amendments were agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1248 (one thousand two hundred forty-eight) as amended.
- S.B. 1400 (one thousand four hundred).
- **S.B. 1447** (one thousand four hundred forty-seven).
- S.B. 1457 (one thousand four hundred fifty-seven) as amended.
- S.B. 1136 (one thousand one hundred thirty-six) was read by title the second time.

The following amendment proposed by the Committee on Transportation was offered:

1. Line 68, introduced, after Nutrition.

insert

The remainder of each annual fee for renewal shall be paid to, and received by, the Commissioner of the Department of Motor Vehicles and paid by the Commissioner into the state treasury and set aside as a special fund to be used to meet the necessary expenses incurred by the Department of Motor Vehicles.

The reading of the amendment was waived.

On motion of Senator Marsden, the amendment was agreed to.

Senator Marsden offered the following amendment:

1. Line 59, introduced, after 3.

insert

That notwithstanding any other provision of law, the range of bumper heights for any street rod bearing a street rod license plate issued pursuant to § 46.2-747 of the Code of Virginia, as it was in effect on June 30, 2021, shall be nine to 22 inches. For purposes of this enactment, "street rod" means a modernized passenger motor vehicle either manufactured prior to 1949 or designed or manufactured to resemble a vehicle manufactured prior to 1949.

4.

On motion of Senator Marsden, the reading of the amendment was waived.

On motion of Senator Marsden, the amendment was agreed to.

On motion of Senator Marsden, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1366** (one thousand three hundred sixty-six), on motion of Senator Barker, was passed by for the day.
 - S.B. 1206 (one thousand two hundred six) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 16.1-300 of the Code of Virginia, relating to confidentiality of juvenile records; exceptions.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

Senator Barker offered the following amendments to the substitute:

1. Line 34, substitute, after records.

insert

Prior to making any report or record open for inspection, the court service unit or Department of Juvenile Justice shall determine which reports or records are relevant to the treatment, services, or care of such juvenile and shall limit such inspection to such relevant reports or records.

2. Line 115, substitute, after shall

insert

review current data and record sharing provisions with regard to youth served by the juvenile justice and child welfare systems and

3. Line 116, substitute, after recommendations

strike

the remainder of line 116, all of line 117, and through systems on line 118

insert

on best practices for the sharing, collection, and use of such data and records while respecting the privacy interest of youth and families

On motion of Senator Barker, the reading of the amendments was waived.

On motion of Senator Barker, the amendments were agreed to.

On motion of Senator Barker, the bill was ordered to be engrossed and read by title the third time.

S.B. 1393 (one thousand three hundred ninety-three), on motion of Senator Marsden, was passed by for the day.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- **S.B.** 1102 (one thousand one hundred two).
- **S.B.** 1150 (one thousand one hundred fifty).
- S.B. 1204 (one thousand two hundred four).
- S.B. 1221 (one thousand two hundred twenty-one).
- S.B. 1227 (one thousand two hundred twenty-seven).
- S.B. 1251 (one thousand two hundred fifty-one).
- S.B. 1269 (one thousand two hundred sixty-nine).
- S.B. 1328 (one thousand three hundred twenty-eight).
- **S.B.** 1351 (one thousand three hundred fifty-one).
- S.B. 1438 (one thousand four hundred thirty-eight).
- S.B. 1147 (one thousand one hundred forty-seven).
- **S.B.** 1165 (one thousand one hundred sixty-five).
- **S.B.** 1170 (one thousand one hundred seventy).
- S.B. 1195 (one thousand one hundred ninety-five).
- **S.B. 1207** (one thousand two hundred seven).
- S.B. 1223 (one thousand two hundred twenty-three).
- S.B. 1298 (one thousand two hundred ninety-eight).
- S.B. 1302 (one thousand three hundred two).
- **S.B. 1321** (one thousand three hundred twenty-one).
- **S.B.** 1334 (one thousand three hundred thirty-four).
- **S.B. 1367** (one thousand three hundred sixty-seven).
- S.B. 1403 (one thousand four hundred three).
- S.B. 1413 (one thousand four hundred thirteen).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- S.B. 1102 (one thousand one hundred two).
- S.B. 1150 (one thousand one hundred fifty).
- S.B. 1204 (one thousand two hundred four).
- S.B. 1221 (one thousand two hundred twenty-one).
- S.B. 1227 (one thousand two hundred twenty-seven).
- S.B. 1251 (one thousand two hundred fifty-one).
- **S.B. 1269** (one thousand two hundred sixty-nine).
- **S.B. 1328** (one thousand three hundred twenty-eight).
- S.B. 1351 (one thousand three hundred fifty-one).

- S.B. 1438 (one thousand four hundred thirty-eight).
- S.B. 1147 (one thousand one hundred forty-seven).
- **S.B.** 1165 (one thousand one hundred sixty-five).
- **S.B. 1170** (one thousand one hundred seventy).
- **S.B.** 1195 (one thousand one hundred ninety-five).
- **S.B. 1207** (one thousand two hundred seven).
- **S.B. 1223** (one thousand two hundred twenty-three).
- S.B. 1298 (one thousand two hundred ninety-eight).
- S.B. 1302 (one thousand three hundred two).
- S.B. 1321 (one thousand three hundred twenty-one).
- **S.B.** 1334 (one thousand three hundred thirty-four).
- **S.B.** 1367 (one thousand three hundred sixty-seven).
- S.B. 1403 (one thousand four hundred three).
- S.B. 1413 (one thousand four hundred thirteen).

SENATE JOINT RESOLUTIONS ON THIRD READING

-244-

Senator Saslaw moved that the following Senate joint resolutions, the titles of the joint resolutions having been printed in the Calendar for their third reading, be placed before the Senate by number only:

- S.J.R. 276 (two hundred seventy-six).
- S.J.R. 286 (two hundred eighty-six).
- S.J.R. 292 (two hundred ninety-two).
- S.J.R. 294 (two hundred ninety-four).
- S.J.R. 308 (three hundred eight).

The motion was agreed to.

Senator Saslaw moved that the questions on agreeing to the Senate joint resolutions that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate joint resolutions were agreed to en bloc:

- S.J.R. 276 (two hundred seventy-six).
- S.J.R. 286 (two hundred eighty-six).
- S.J.R. 292 (two hundred ninety-two).
- S.J.R. 294 (two hundred ninety-four).
- S.J.R. 308 (three hundred eight).

S.J.R. 285 (two hundred eighty-five) was read by title the third time and, on motion of Senator Edwards, was agreed to.

SENATE RESOLUTION ON SECOND READING

S.R. 91 (ninety-one) was taken up, the substitute offered by Senator Bell having been agreed to on January 26, 2021.

On motion of Senator Norment, the resolution was ordered to be engrossed and read by title the third time.

Senator Norment moved that the Rules be suspended and the third reading of the title of S.R. 91 be waived.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1.

RULE 36--0.

Senator Bell moved that S.R. 91 be agreed to.

PARLIAMENTARY INQUIRIES

Senator Morrissey propounded a parliamentary inquiry as to whether the Senate would consider only S.R. 91, relating to the censure of a Senator, or other remarks made outside of S.R. 91.

The Chair stated that the Senate would consider only S.R. 91.

Senator Morrissey propounded a further parliamentary inquiry as to whether the Senate would vote on each separate point in S.R. 91, relating to the censure of a Senator, or would cast one vote on censure.

The Chair stated that the Senate would cast one vote on S.R. 91, relating to the censure of a Senator.

RULING OF THE CHAIR

Senator Morrissey propounded a parliamentary inquiry as to whether S.R. 91 was properly before the Senate.

The Chair ruled that **S.R. 91** was not properly before the Senate, pursuant to Senate Rules 18 (h) and 53 (b), which state, "The Committee shall consider the matter, conduct such hearings as it shall deem necessary, and, in all cases report its determination of the matter, together with its recommendations and reasons for its resolves, to the Senate." The Chair stated that "each of the matters contained in the censure resolution must be investigated, the facts must be established, and the conclusions and the resolutions of the Privileges and Elections Committee then be reported out to the Senate as a whole."

Senator Saslaw appealed from the Ruling of the Chair.

The question was put on sustaining the Ruling of the Chair.

The Ruling of the Chair was not sustained.

The recorded vote is as follows: YEAS--8. NAYS--29. RULE 36--0.

YEAS--Chase, DeSteph, Edwards, McDougle, Morrissey, Newman, Stuart, Vogel--8.

NAYS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Suetterlein, Surovell--29.

RULE 36--0.

S.R. 91, on motion of Senator Bell, was agreed to.

The recorded vote is as follows: YEAS--24. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Reeves, Saslaw, Spruill, Surovell, Vogel--24.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Newman, Peake, Ruff, Stuart--9. RULE 36--0.

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

san Carle Delsa

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

THURSDAY, JANUARY 28, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Lynwood W. Lewis, Jr., Sixth Senatorial District, offered the following prayer:

Heavenly Father, as we gather here today we ask for Your guidance and grace in doing the work of our Commonwealth. We realize the awesome responsibility given to us by the people who empowered each of us to serve in this body and we ask that You give us the will to lead with humility and the strength to govern justly. Father, in these times of uncertainties we are reminded of our common purpose through You as we work together to seek the welfare of the people of Virginia. We ask that You grant us the wisdom to carry out this common purpose and that Your grace will guide each of us in our willingness to work together for the betterment of all Virginians. Today and all days we thank You for the opportunity to serve and we ask for Your blessings and continued protection of our Commonwealth and her people. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Michael Adams, Director of Human Resources, Senate Clerk's Office.

On motion of Senator McDougle, the Rules were suspended to allow Senators Norment and Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows: YEAS--35. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein--35.

NAYS--Surovell--1. RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Mason, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--35. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--McDougle, Petersen--2. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 27, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1749.** A BILL to amend and reenact § 3.1 of Chapters 398 and 520 of the Acts of Assembly of 2009, which provided a charter for the Town of Nassawadox, relating to town elections.
- **H.B. 1764.** A BILL to amend and reenact §§ 3.1, as amended, 3.2, 3.7, as amended, and 5.1 of Chapter 669 of the Acts of Assembly of 1972, which provided a charter for the Town of Crewe in Nottoway County, and to repeal Chapter 6 (§§ 6.1 through 6.4) of Chapter 669 of the Acts of Assembly of 1972, relating to town council; elections and powers.
- **H.B. 1783.** A BILL to provide a new charter for the Town of Glasgow in Rockbridge County and to repeal Chapter 486, as amended, of the Acts of Assembly of 1892, which provided a charter for the Town of Glasgow.
- **H.B. 1812.** A BILL to amend and reenact §§ 2.2-3711, 58.1-4100, 58.1-4109, 58.1-4110, 58.1-4114, 58.1-4122, 58.1-4124, and 58.1-4125 of the Code of Virginia, relating to casino gaming; technical amendments.
- **H.B. 1847.** A BILL to amend and reenact §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100 of the Code of Virginia, relating to sports betting; technical amendments.
- **H.B. 1858.** A BILL to amend and reenact §§ 1 and 4 of the charter of the Town of Appomattox, which was granted by order of the Circuit Court of the County of Appomattox on June 2, 1925, and as amended by Chapter 43 of the Acts of Assembly of 1980, relating to election and appointment of officers; time of election.
- **H.B. 1866.** A BILL to amend and reenact §§ 9.1-153 and 9.1-156 of the Code of Virginia, relating to court-appointed special advocates; information sharing.
- H.B. 1882. A BILL to amend and reenact § 55.1-319 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 55.1-318.1, relating to deeds of trust; amendment to loan document; statement of interest rate of a refinanced mortgage.
- **H.B. 1898.** A BILL to amend and reenact § 15.2-2308 of the Code of Virginia, relating to board of zoning appeals; appointments.
- **H.B. 1909.** A BILL to amend the Code of Virginia by adding a section numbered 22.1-131.1, relating to certain school board property; establishment of gun-free zone permitted.
- **H.B. 1991.** A BILL to amend and reenact §§ 16.1-285.1 and 16.1-285.2 of the Code of Virginia, relating to juveniles; release and review hearing for serious offender; plea agreement.
- **H.B. 2018.** A BILL to amend and reenact §§ 63.2-1603, 63.2-1606, and 63.2-1609 of the Code of Virginia, relating to emergency order for adult protective services; acts of violence, force, or threat or financial exploitation; penalty.

- **H.B. 2042.** A BILL to amend and reenact §§ 15.2-961 and 15.2-961.1 of the Code of Virginia, relating to replacement and conservation of trees during development.
- **H.B. 2054.** A BILL to amend and reenact § 15.2-2223.4 of the Code of Virginia, relating to comprehensive plan; transit-oriented development.
- H.B. 2055. A BILL to amend and reenact §§ 20-108.1 and 63.2-1918 of the Code of Virginia, relating to child support obligations; party's incarceration not deemed voluntary unemployment or underemployment.
- **H.B. 2095.** A BILL to amend and reenact §§ 2.06, 5.04, as amended, 6.03, 8.04, as amended, 10.06, as amended, 10.07, and 15.03, as amended, of Chapter 542 of the Acts of Assembly of 1990, which provided a charter for the City of Bristol, relating to powers and organization.
- **H.B. 2099.** A BILL to amend and reenact §§ 8.01-251, 8.01-458, and 55.1-339 of the Code of Virginia, relating to limitations on enforcement of judgments; judgment liens; settlement agents.
- **H.B. 2128.** A BILL to amend and reenact § 18.2-308.2:2, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to sale and transfer of firearms; criminal history record information checks.
- **H.B. 2133.** A BILL to amend the Code of Virginia by adding in Title 19.2 a chapter numbered 19.4, consisting of sections numbered 19.2-327.15 through 19.2-327.20, relating to issuance of writ of vacatur for victims of commercial sex trafficking.
- **H.B. 2150.** A BILL to amend and reenact §§ 16.1-123.1, 16.1-241, and 17.1-513 of the Code of Virginia, relating to jurisdiction over criminal cases; certification or appeal of charges.
- **H.B. 2169.** A BILL to amend and reenact §§ 8.01-42.4, 9.1-116.5, 9.1-902, 16.1-69.48:6, 16.1-69.55, 17.1-275.13, 17.1-805, 18.2-46.1, 18.2-346, 18.2-346.1, 18.2-350, 18.2-357.1, 18.2-513, 19.2-10.2, 19.2-215.1, 19.2-268.3, 19.2-386.16, 19.2-386.35, 19.2-392.02, as it is currently effective and as it shall become effective, 32.1-58, 37.2-314, 37.2-416, and 37.2-506 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-346.01, relating to prostitution; solicitation.
- **H.B. 2180.** A BILL to amend and reenact § 7, as amended, of Chapter 343 of the Acts of Assembly of 1928, which provided a charter for the City of Lynchburg, relating to salaries.
- H.B. 2186. A BILL to amend and reenact § 15.2-4904 of the Code of Virginia, relating to Industrial Development and Revenue Bond Act; directors; Mathews County.
- **H.B. 2192.** A BILL to amend and reenact §§ 20-60.3 and 63.2-1916 of the Code of Virginia, relating to domestic relations; contents of support orders; unemployment benefits.
- **H.B. 2201.** A BILL to amend and reenact §§ 15.2-2288.8 and 15.2-2316.6 through 15.2-2316.9 of the Code of Virginia, relating to solar projects and energy storage projects; siting agreements throughout the Commonwealth.
- H.B. 2208. A BILL to direct the Department of General Services to remove the statue of Harry F. Byrd, Sr., from Capitol Square.

- **H.B. 2216.** A BILL to amend and reenact §§ 52-34.13, 52-34.14, and 52-34.15 of the Code of Virginia, relating the Virginia Missing Person with Autism Alert Program.
- **H.B. 2233.** A BILL to amend and reenact §§ 19.2-305.1, 19.2-305.2, 19.2-349, and 19.2-354 of the Code of Virginia, relating to orders of restitution; enforcement.
- H.B. 2276. A BILL to amend and reenact § 18.2-308.5 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-308.5:2, relating to manufacture, import, sale, transfer, or possession of plastic firearms and unfinished frames or receivers and unserialized firearms; penalties.
- **H.B. 2287.** A BILL to amend and reenact § 15.2-4904 of the Code of Virginia, relating to economic development authorities; size of board in Powhatan County; quorum.

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTIONS:

- **H.J.R. 578.** Requesting the Department of Behavioral Health and Developmental Services to study the feasibility of developing a secure, de-identified, renewable, and relational database of criminal justice, behavioral health, and other human services records to facilitate the development of more effective interventions. Report.
- **H.J.R. 606.** Designating November, in 2021 and in each succeeding year, as Equal Citizens Month in Virginia.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--33. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--33.

NAYS--DeSteph, McDougle, Petersen, Stanley--4.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

H.B. 1749, H.B. 1764, H.B. 1783, H.B. 1858, H.B. 1898, H.B. 2042, H.B. 2054, H.B. 2095, H.B. 2180, H.B. 2186, H.B. 2201, and H.B. 2287 were referred to the Committee on Local Government.

H.B. 1812, H.B. 1847, and **H.B. 1882** were referred to the Committee on General Laws and Technology.

H.B. 1866, H.B. 1991, H.B. 2018, H.B. 2055, H.B. 2099, H.B. 2128, H.B. 2133, H.B. 2150, H.B. 2169, H.B. 2192, H.B. 2233, and H.B. 2276 were referred to the Committee on the Judiciary.

- H.B. 1909 was referred to the Committee on Education and Health.
- H.B. 2208 was referred to the Committee on Rules.
- **H.B. 2216** was referred to the Committee on Transportation.

The House joint resolutions, communicated as agreed to by the House of Delegates, the first reading of their titles having been waived, were referred as follows:

H.J.R. 578 and H.J.R. 606 were referred to the Committee on Rules.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Lucas from the Committee on Education and Health:

- **S.B.** 1176 (one thousand one hundred seventy-six) with substitute.
- **S.B.** 1178 (one thousand one hundred seventy-eight).
- **S.B. 1225** (one thousand two hundred twenty-five).
- S.B. 1235 (one thousand two hundred thirty-five) with amendments.
- S.B. 1237 (one thousand two hundred thirty-seven).
- **S.B. 1257** (one thousand two hundred fifty-seven) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1288 (one thousand two hundred eighty-eight) with substitute.
 - **S.B. 1303** (one thousand three hundred three).
 - **S.B. 1304** (one thousand three hundred four) with amendments.
- **S.B. 1307** (one thousand three hundred seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1316** (one thousand three hundred sixteen) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1317** (one thousand three hundred seventeen).
- **S.B. 1333** (one thousand three hundred thirty-three) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1356** (one thousand three hundred fifty-six) with substitute.
- **S.B. 1357** (one thousand three hundred fifty-seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1362** (one thousand three hundred sixty-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1401** (one thousand four hundred one) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1436** (one thousand four hundred thirty-six) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1439 (one thousand four hundred thirty-nine) with amendment.
 - S.B. 1464 (one thousand four hundred sixty-four).
 - H.B. 2333 (two thousand three hundred thirty-three) with substitute.

The following bills, having been considered by the committee in session, were reported by Senator Barker from the Committee on General Laws and Technology:

- **S.B. 1127** (one thousand one hundred twenty-seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1183 (one thousand one hundred eighty-three).
 - **S.B.** 1198 (one thousand one hundred ninety-eight) with amendments.
 - S.B. 1215 (one thousand two hundred fifteen) with amendment.
- **S.B. 1222** (one thousand two hundred twenty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1278** (one thousand two hundred seventy-eight) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1287** (one thousand two hundred eighty-seven) with substitute.
 - S.B. 1327 (one thousand three hundred twenty-seven) with substitute.
 - S.B. 1349 (one thousand three hundred forty-nine).
- **S.B. 1360** (one thousand three hundred sixty) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1365** (one thousand three hundred sixty-five) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1384 (one thousand three hundred eighty-four).
- **S.B. 1389** (one thousand three hundred eighty-nine) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1392** (one thousand three hundred ninety-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1410 (one thousand four hundred ten) with substitute.
- **S.B. 1418** (one thousand four hundred eighteen) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1429** (one thousand four hundred twenty-nine) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1458 (one thousand four hundred fifty-eight).
- **S.B. 1469** (one thousand four hundred sixty-nine) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following joint resolution, having been considered by the committee in session, was recommended for rereferral by the Committee on General Laws and Technology:

S.J.R. 322 (three hundred twenty-two) with the recommendation that it be rereferred to the Committee on Privileges and Elections.

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- S.B. 1138 (one thousand one hundred thirty-eight).
- **S.B.** 1180 (one thousand one hundred eighty) with substitute.
- **S.B.** 1184 (one thousand one hundred eighty-four) with substitute.
- S.B. 1266 (one thousand two hundred sixty-six) with amendment.
- S.B. 1272 (one thousand two hundred seventy-two).
- **S.B. 1315** (one thousand three hundred fifteen) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1325 (one thousand three hundred twenty-five) with substitute.
 - **S.B. 1381** (one thousand three hundred eighty-one) with substitute.

- **S.B. 1382** (one thousand three hundred eighty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1431 (one thousand four hundred thirty-one).
 - **S.B. 1437** (one thousand four hundred thirty-seven).
 - **S.B. 1456** (one thousand four hundred fifty-six).
- **S.B. 1461** (one thousand four hundred sixty-one) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1468 (one thousand four hundred sixty-eight) with substitute.
- S.B. 1127, S.B. 1222, S.B. 1257, S.B. 1278, S.B. 1307, S.B. 1315, S.B. 1316, S.B. 1333, S.B. 1357, S.B. 1360, S.B. 1362, S.B. 1365, S.B. 1382, S.B. 1389, S.B. 1392, S.B. 1401, S.B. 1418, S.B. 1429, S.B. 1436, S.B. 1461, and S.B. 1469 were rereferred to the Committee on Finance and Appropriations.
 - S.J.R. 322 was rereferred to the Committee on Privileges and Elections.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 332. Commending the Honorable Mark C. Christie.

Patron--McDougle

S.J.R. 333. Celebrating the life of Frederick Cameron.

Patron--DeSteph

CALENDAR

SENATE BILLS ON THIRD READING

S.B. 1447 (one thousand four hundred forty-seven), on motion of Senator Peake, was passed by for the day.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1248 (one thousand two hundred forty-eight).
- S.B. 1400 (one thousand four hundred).
- S.B. 1457 (one thousand four hundred fifty-seven).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- S.B. 1248 (one thousand two hundred forty-eight).
- S.B. 1400 (one thousand four hundred).
- S.B. 1457 (one thousand four hundred fifty-seven).

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1136 (one thousand one hundred thirty-six) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Chase, DeSteph, Stanley--3.

RULE 36--0.

S.B. 1206 (one thousand two hundred six) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--27, NAYS--11, RULE 36--0.

YEAS--Barker, Bell, Boysko, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Obenshain, Peake, Petersen, Saslaw, Spruill, Suetterlein, Surovell--27.

NAYS--Chase, Cosgrove, DeSteph, Newman, Norment, Pillion, Reeves, Ruff, Stanley, Stuart, Vogel--11.

RULE 36--0.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

S.B. 1102 (one thousand one hundred two).

S.B. 1150 (one thousand one hundred fifty).

S.B. 1204 (one thousand two hundred four).

S.B. 1221 (one thousand two hundred twenty-one).

S.B. 1227 (one thousand two hundred twenty-seven).

S.B. 1251 (one thousand two hundred fifty-one).

S.B. 1269 (one thousand two hundred sixty-nine).

S.B. 1328 (one thousand three hundred twenty-eight).

S.B. 1351 (one thousand three hundred fifty-one).

S.B. 1438 (one thousand four hundred thirty-eight).

The motion was agreed to.

S.B. 1102 (one thousand one hundred two) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

-255-

A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 10 of Title 32.1 a section numbered 32.1-331.04, relating to Department of Medical Assistance Services; personal care aides; orientation program.

The reading of the substitute was waived.

On motion of Senator Locke, the substitute was agreed to.

S.B. 1150 (one thousand one hundred fifty) was taken up.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. After line 30, introduced

insert

2. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendment was waived.

On motion of Senator Kiggans, the amendment was agreed to.

Senator Kiggans offered the following amendment:

1. Line 22, introduced, after of

strike

Social Services

insert

Education

On motion of Senator Kiggans, the reading of the amendment was waived.

On motion of Senator Kiggans, the amendment was agreed to.

S.B. 1204 (one thousand two hundred four) was taken up.

The following amendments proposed by the Committee on Finance and Appropriations were offered:

1. Line 285, introduced, after *but* strike

the remainder of line 285 and through and on line 286

2. Line 287, introduced, after subsection

strike

В

insert

D

3. Line 287, introduced, after §

strike

2.2-3708

insert

2.2-3708.2

4. Line 2205, introduced, after §

strike

33.1-387

insert

33.2-1103

The reading of the amendments was waived.

On motion of Senator Barker, the amendments were agreed to.

S.B. 1227 (one thousand two hundred twenty-seven) was taken up.

The following amendment proposed by the Committee on Education and Health was offered:

1. Line 172, introduced, after time

insert

for Medicaid and FAMIS enrollees

The reading of the amendment was waived.

On motion of Senator Boysko, the amendment was agreed to.

S.B. 1269 (one thousand two hundred sixty-nine) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 38.2-3407.15:2 of the Code of Virginia, relating to health insurance; authorization of drug prescribed for the treatment of a mental disorder.

The reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

S.B. 1328 (one thousand three hundred twenty-eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 16.1-228, 16.1-282.1, 63.2-100, as it is currently effective and as it shall become effective, 63.2-905, 63.2-906, and 63.2-1305 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 13 of Title 63.2 a section numbered 63.2-1306, relating to State-Funded Kinship Guardianship Assistance program.

The reading of the substitute was waived.

On motion of Senator Mason, the substitute was agreed to.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

1. Line 141, substitute, after 673

insert

or the State-Funded Kinship Guardianship Assistance program set forth in § 63.2-1306

2. Line 1059, substitute, after 673

insert

or the State-Funded Kinship Guardianship Assistance program set forth in \$63.2-1306

3. Line 1111, substitute, after state-funded maintenance payments,

strike

the remainder of line 1111

4. Line 1173, substitute, after §

strike

63.2-1306

insert

63.2-1305

5. Line 1182, substitute, after check

strike

in accordance with § 63.2-1720

insert

and has not been convicted of any barrier crime as outlined in 42 U.S.C. $\S 671(a)(20)$

The reading of the amendments was waived.

On motion of Senator Mason, the amendments were agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1102 (one thousand one hundred two) as amended.
- S.B. 1150 (one thousand one hundred fifty) as amended.
- S.B. 1204 (one thousand two hundred four) as amended.
- S.B. 1221 (one thousand two hundred twenty-one).
- S.B. 1227 (one thousand two hundred twenty-seven) as amended.
- S.B. 1251 (one thousand two hundred fifty-one).
- S.B. 1269 (one thousand two hundred sixty-nine) as amended.
- S.B. 1328 (one thousand three hundred twenty-eight) as amended.
- **S.B. 1351** (one thousand three hundred fifty-one).
- S.B. 1438 (one thousand four hundred thirty-eight).
- **S.B. 1366** (one thousand three hundred sixty-six), on motion of Senator Barker, was passed by for the day.
- **S.B. 1393** (one thousand three hundred ninety-three), on motion of Senator Marsden, was passed by for the day.
- **S.B. 1147** (one thousand one hundred forty-seven) was read by title the second time and, on motion of Senator Kiggans, was ordered to be engrossed and read by title the third time.
- $\mathbf{S.B.}\ 1165$ (one thousand one hundred sixty-five), on motion of Senator Surovell, was passed by for the day.
- **S.B. 1170** (one thousand one hundred seventy), on motion of Senator Norment, was passed by for the day.
 - S.B. 1195 (one thousand one hundred ninety-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 38.2-2206 of the Code of Virginia, relating to motor vehicle insurance; underinsured motor vehicle.

The reading of the substitute was waived.

On motion of Senator Obenshain, the substitute was agreed to.

On motion of Senator Obenshain, the bill was ordered to be engrossed and read by title the third time.

S.B. 1207 (one thousand two hundred seven) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Local Government was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 15.2-2288.8 and 15.2-2316.6 through 15.2-2316.9 of the Code of Virginia, relating to solar projects and energy storage projects; siting agreements throughout the Commonwealth.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

On motion of Senator Barker, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1223** (one thousand two hundred twenty-three) was read by title the second time and, on motion of Senator Boysko, was ordered to be engrossed and read by title the third time.
- **S.B. 1298** (one thousand two hundred ninety-eight) was read by title the second time and, on motion of Senator Bell, was ordered to be engrossed and read by title the third time.
 - S.B. 1302 (one thousand three hundred two) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 37.2-311.1, as it shall become effective, 56-484.12, 56-484.17, and 56-484.17:1 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 37.2-311.2, 37.2-311.3, and 37.2-311.4 and by adding in Article 7 of Chapter 15 of Title 56 a section numbered 56-484.18:1, relating to crisis call centers; Crisis Call Center Fund established.

The reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

On motion of Senator McPike, the bill was ordered to be engrossed and read by title the third time.

S.B. 1321 (one thousand three hundred twenty-one) was read by title the second time.

The following amendments proposed by the Committee on Rehabilitation and Social Services were offered:

1. Line 16, introduced, after *interest* strike

, as defined in § 20-124.1,

2. After line 54, introduced

insert

E. For the purposes of this section, "person with a legitimate interest" means the same as that term is defined in § 20-124.1.

The reading of the amendments was waived.

On motion of Senator Boysko, the amendments were agreed to.

On motion of Senator Boysko, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1334** (one thousand three hundred thirty-four) was read by title the second time and, on motion of Senator Edwards, was ordered to be engrossed and read by title the third time.
- **S.B. 1367** (one thousand three hundred sixty-seven) was read by title the second time and, on motion of Senator DeSteph, was ordered to be engrossed and read by title the third time.
- **S.B. 1403** (one thousand four hundred three) was read by title the second time and, on motion of Senator Pillion, was ordered to be engrossed and read by title the third time.
 - S.B. 1413 (one thousand four hundred thirteen) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to provision of broadband services by investor-owned electric utilities.

The reading of the substitute was waived.

On motion of Senator Boysko, the substitute was agreed to.

On motion of Senator Boysko, the bill was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1104 (one thousand one hundred four).
- **S.B.** 1119 (one thousand one hundred nineteen).
- **S.B.** 1125 (one thousand one hundred twenty-five).
- S.B. 1135 (one thousand one hundred thirty-five).
- S.B. 1143 (one thousand one hundred forty-three).
- **S.B.** 1188 (one thousand one hundred eighty-eight).
- **S.B. 1193** (one thousand one hundred ninety-three).
- S.B. 1246 (one thousand two hundred forty-six).
- S.B. 1279 (one thousand two hundred seventy-nine).
- **S.B.** 1331 (one thousand three hundred thirty-one).
- S.B. 1417 (one thousand four hundred seventeen).
- S.B. 1423 (one thousand four hundred twenty-three).
- **S.B.** 1453 (one thousand four hundred fifty-three).
- **S.B.** 1103 (one thousand one hundred three).
- **S.B.** 1106 (one thousand one hundred six).
- S.B. 1108 (one thousand one hundred eight).
- S.B. 1115 (one thousand one hundred fifteen).
- S.B. 1148 (one thousand one hundred forty-eight).
- S.B. 1245 (one thousand two hundred forty-five).
- **S.B. 1254** (one thousand two hundred fifty-four).
- **S.B. 1300** (one thousand three hundred).
- S.B. 1305 (one thousand three hundred five).

```
S.B. 1314 (one thousand three hundred fourteen).
```

S.B. 1379 (one thousand three hundred seventy-nine).

S.B. 1387 (one thousand three hundred eighty-seven).

S.B. 1390 (one thousand three hundred ninety).

S.B. 1402 (one thousand four hundred two).

S.B. 1412 (one thousand four hundred twelve).

S.B. 1422 (one thousand four hundred twenty-two).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

```
S.B. 1104 (one thousand one hundred four).
```

S.B. 1119 (one thousand one hundred nineteen).

S.B. 1125 (one thousand one hundred twenty-five).

S.B. 1135 (one thousand one hundred thirty-five).

S.B. 1143 (one thousand one hundred forty-three).

S.B. 1188 (one thousand one hundred eighty-eight).

S.B. 1193 (one thousand one hundred ninety-three).

S.B. 1246 (one thousand two hundred forty-six).

S.B. 1279 (one thousand two hundred seventy-nine).

S.B. 1331 (one thousand three hundred thirty-one).

S.B. 1417 (one thousand four hundred seventeen).

S.B. 1423 (one thousand four hundred twenty-three).

S.B. 1453 (one thousand four hundred fifty-three).

S.B. 1103 (one thousand one hundred three).

S.B. 1106 (one thousand one hundred six).

S.B. 1108 (one thousand one hundred eight).

S.B. 1115 (one thousand one hundred fifteen).

S.B. 1148 (one thousand one hundred forty-eight).

S.B. 1245 (one thousand two hundred forty-five).

S.B. 1254 (one thousand two hundred fifty-four).

S.B. 1300 (one thousand three hundred).

S.B. 1305 (one thousand three hundred five).

S.B. 1314 (one thousand three hundred fourteen).

S.B. 1379 (one thousand three hundred seventy-nine).

S.B. 1387 (one thousand three hundred eighty-seven).

S.B. 1390 (one thousand three hundred ninety).

S.B. 1402 (one thousand four hundred two).

S.B. 1412 (one thousand four hundred twelve).

S.B. 1422 (one thousand four hundred twenty-two).

SENATE JOINT RESOLUTION ON FIRST READING

S.J.R. 327 (three hundred twenty-seven) was read by title the first time.

MEMORIAL RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following House joint resolutions were taken up and agreed to en bloc by a unanimous standing vote:

H.J.R. 588 (five hundred eighty-eight).

H.J.R. 590 (five hundred ninety).

H.J.R. 591 (five hundred ninety-one).

H.J.R. 592 (five hundred ninety-two).

H.J.R. 593 (five hundred ninety-three).

H.J.R. 595 (five hundred ninety-five).

H.J.R. 601 (six hundred one).

H.J.R. 613 (six hundred thirteen).

H.J.R. 589 (five hundred eighty-nine) was taken up, as follows:

HOUSE JOINT RESOLUTION NO. 589

Celebrating the life of the Honorable Mamye E. BaCote.

WHEREAS, the Honorable Mamye E. BaCote, an inspirational educator and a highly admired public servant who represented the residents of Newport News and Hampton in the Virginia House of Delegates for more than a decade, died on December 14, 2020; and

WHEREAS, a native of Halifax County, Mamye BaCote received a bachelor's degree from Virginia Union University and a master's degree from what is now Hampton University; and

WHEREAS, while a student at Virginia Union University in 1960, Mamye BaCote played a significant role in the civil rights movement as a member of the Richmond 34, a group of students and activists who staged a sit-in at the Thalhimers lunch counter in downtown Richmond; and

WHEREAS, Mamye BaCote pursued a career in education with Newport News Public Schools, teaching social studies at Collis P. Huntington High School and Menchville High School, where she became a department head; and

WHEREAS, Mamye BaCote fostered trust among her students and worked tirelessly to help them achieve success in and out of the classroom; she inspired young adults as an adjunct professor of political science at Hampton University; and

WHEREAS, after her retirement as a teacher in 1994, Mamye BaCote was elected to the Newport News City Council and served the city in that capacity from 1996 to 2003; and

WHEREAS, desirous to be of further service to the Commonwealth, Mamye BaCote ran for and was elected to the Virginia House of Delegates in 2003 and represented the residents of the 95th District until January 2016; and

WHEREAS, Mamye BaCote introduced and supported numerous important pieces of legislation to benefit all Virginians and offered her leadership and expertise to several standing committees, including Appropriations; Health, Welfare and Institutions; and Transportation; and

WHEREAS, during her tenure as a state lawmaker, Mamye BaCote focused on education and support for public schools, gun safety measures, and helping to rehabilitate nonviolent drug offenders by funding the Newport News Drug Court; and

WHEREAS, Mamye BaCote worked diligently to cultivate mutual respect across party lines, providing both Democrats and Republicans alike with her wisdom and insights, which helped to build bipartisan consensus on critical issues; and

WHEREAS, Mamye BaCote was well-known among constituents and colleagues for her grace, dignity, and genuine kindness, but her soft-spoken nature belied her tenacity and commitment to the public good; and

WHEREAS, Mamye BaCote served the Commonwealth with the utmost integrity, dedication, and distinction and inspired other members of her community to pursue a life of public service; and

WHEREAS, Mamye BaCote enjoyed fellowship and worship with the congregation of Saint Vincent de Paul Catholic Church and shared her faith with the community through song as a member of the choir; and

WHEREAS, predeceased by her husband, Theodore, and one son, Kyle, Mamye BaCote will be fondly remembered and greatly missed by her sons, Theodore III, Derek, and Marlon, and their families; numerous other family members and friends; and colleagues on both sides of the aisle; now, therefore, be it

RESOLVED by the House of Delegates, the Senate concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Mamye E. BaCote, a respected public servant and former member of the Virginia House of Delegates; and, be it

RESOLVED FURTHER, That the Clerk of the House of Delegates prepare a copy of this resolution for presentation to the family of the Honorable Mamye E. BaCote as an expression of the General Assembly's respect for her memory.

H.J.R. 589, on motion of Senator Ebbin, was agreed to by a unanimous standing vote.

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions and resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions and resolutions were taken up, ordered to be engrossed, and agreed to en bloc by a unanimous standing vote:

S.J.R. 305 (three hundred five).

S.J.R. 319 (three hundred nineteen).

```
S.J.R. 320 (three hundred twenty).
S.J.R. 324 (three hundred twenty-four).
S.J.R. 326 (three hundred twenty-six).
S.J.R. 328 (three hundred twenty-eight).
S.J.R. 329 (three hundred twenty-nine).
S.J.R. 330 (three hundred thirty).
S.R. 94 (ninety-four).
S.R. 97 (ninety-seven).
S.R. 98 (ninety-eight).
```

S.J.R. 316 (three hundred sixteen), on motion of Senator Ebbin, was passed by for the day.

COMMENDING RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following House joint resolutions were taken up and agreed to en bloc:

```
H.J.R. 587 (five hundred eighty-seven).
H.J.R. 594 (five hundred ninety-four).
H.J.R. 597 (five hundred ninety-seven).
H.J.R. 598 (five hundred ninety-eight).
H.J.R. 599 (five hundred ninety-nine).
H.J.R. 600 (six hundred).
H.J.R. 602 (six hundred two).
H.J.R. 603 (six hundred three).
H.J.R. 608 (six hundred eight).
H.J.R. 609 (six hundred nine).
H.J.R. 610 (six hundred ten).
H.J.R. 611 (six hundred eleven).
H.J.R. 612 (six hundred twelve).
```

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions and resolution that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions and resolution were taken up, ordered to be engrossed, and agreed to en bloc:

```
S.J.R. 321 (three hundred twenty-one).S.J.R. 325 (three hundred twenty-five).S.R. 96 (ninety-six).
```

S.J.R. 331 (three hundred thirty-one), on motion of Senator Newman, was passed by for the day.

On motion of Senator Ebbin, a leave of absence for the day was granted Senator Deeds on account of pressing personal business.

On motion of Senator Lucas, the Senate adjourned until tomorrow at $12\ m.$

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

FRIDAY, JANUARY 29, 2021

The Senate met at 12 m. and was called to order by the President pro tempore, Senator L. Louise Lucas.

The Honorable Bill DeSteph, Eighth Senatorial District, offered the following prayer:

Let us go to God's word to seek the truth. In Matthew 5:9 it said, "blessed are the peacekeepers for they will be called children of God." In Galatians 3:28, "there is neither Jew nor Greek, slave nor free, male nor female. For you are all one in Christ Jesus." In Romans 10:12, "for there is no difference between Jew and gentile. The same Lord is Lord of all and richly blessed. All who call on him will be richly blessed."

Lord God, we pray today for our peacekeepers. As Your word says, as I have just read, blessed are the peacemakers. What I pray for is our first responders, military, and those who come to the aid of our neighbors, families, friends, cities, counties, towns, and those who protect us and protect our country. Lord, I want to pray for understanding of all, equality for all, so we treat all of God's children as one. Let us not judge or we too will be judged. But, give us wisdom, discernment, and guidance from You, Lord, and You alone. Father, we know Your word tells us that we are to treat others as we want to be treated. Lord, we ask that we are all treated equally with respect, kindness, and above all love. In His most precious, holy name, Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by John McE. Garrett, former Chief Deputy Clerk, Senate Clerk's Office.

RECESS

At 12:35 p.m., Senator Saslaw moved that the Senate recess until 12:40 p.m.

The motion was agreed to.

The hour of 12:40 p.m. having arrived, the Chair was resumed.

On motion of Senator Norment, the Rules were suspended to allow Senator Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--34. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--34.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell.

A quorum was present.

After the roll call, Senators Reeves and Vogel notified the Clerk of their presence.

On motion of Senator Locke, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--31. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Saslaw, Spruill, Suetterlein, Surovell--31.

NAYS--Cosgrove, Deeds, DeSteph, Petersen, Stanley--5.

RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 28, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1846.** A BILL to amend and reenact § 46.2-334.01, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to license restrictions for minors; use of handheld personal communications devices.
- **H.B. 1893.** A BILL to amend the Code of Virginia by adding in Title 33.2 a chapter numbered 38, consisting of sections numbered 33.2-3800 through 33.2-3816, relating to creation of the New River Valley Passenger Rail Station Authority.
- **H.B. 1918.** A BILL to amend and reenact § 22.1-205 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-205.1, relating to student driver safety.
- **H.B. 1925.** A BILL to amend the Code of Virginia by adding in Title 67 a chapter numbered 18, consisting of a section numbered 67-1800, relating to Virginia Brownfield and Coal Mine Renewable Energy Grant Fund and Program; handbook.
- **H.B. 1930.** A BILL to amend the Code of Virginia by adding a section numbered 23.1-407.1, relating to public institutions of higher education; admissions applications; criminal history.
- **H.B. 1950.** A BILL to require the Office of the Chief Medical Examiner to convene a work group to develop a plan for the establishment of a Fetal and Infant Mortality Review Team; report.
- **H.B. 1957.** A BILL to amend and reenact § 63.2-1244 of the Code of Virginia, relating to adult adoption; investigation and report.
- **H.B. 1986.** A BILL providing a management agreement between the Commonwealth and George Mason University pursuant to the Restructured Higher Education Financial and Administrative Operations Act (§ 23.1-1000 et seq.).

- **H.B. 1995.** A BILL to amend the Code of Virginia by adding in Chapter 2 of Title 32.1 an article numbered 19, consisting of sections numbered 32.1-73.14 through 32.1-73.17, relating to Rare Disease Council; Rare Disease Council Fund; report.
- **H.B. 1998.** A BILL to amend and reenact § 22.1-137.2 of the Code of Virginia, relating to public schools; lock-down drills; annual requirement.
- **H.B. 2001.** A BILL to amend and reenact §§ 2.2-1182 and 2.2-1183 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-1804.1, relating to building standards for certain state and local buildings.
- **H.B. 2013.** A BILL to amend and reenact § 22.1-79.7 of the Code of Virginia, relating to school board policies; school meal debt; enforcement.
- H.B. 2019. A BILL to amend and reenact §§ 8.01-225, 22.1-274.2, and 54.1-3408 of the Code of Virginia, relating to public elementary and secondary schools; possession and administration of undesignated stock albuterol inhalers and valved holding chambers.
- **H.B. 2068.** A BILL to amend and reenact §§ 3.2-303, 3.2-304, and 3.2-310 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 3.1 of Title 3.2 a section numbered 3.2-311, relating to Local Food and Farming Infrastructure Grant Program.
- **H.B. 2092.** A BILL to amend and reenact §§ 19.2-389, as it is currently effective and as it shall become effective, 37.2-416, and 37.2-506 of the Code of Virginia, relating to Department of Behavioral Health and Developmental Services; background checks; persons providing contractual services.
- H.B. 2105. A BILL to amend and reenact § 2.2-208.1 of the Code of Virginia and to amend and reenact the second and eleventh enactments of Chapter 860 and the second and eleventh enactments of Chapter 861 of the Acts of Assembly of 2020, relating to early childhood education; quality rating and improvement system participation; School Readiness Committee.
- H.B. 2111. A BILL to establish the Task Force on Maternal Health Data and Quality Measures; report.
- **H.B. 2119.** A BILL to amend and reenact § 22.1-205 of the Code of Virginia, relating to student driver education program; parent/student component exemption.
- **H.B. 2135.** A BILL to amend the Code of Virginia by adding a section numbered 22.1-207.4:2, relating to certain school boards; student meals; participation in the Afterschool Meal Program.
- H.B. 2154. A BILL to amend and reenact § 32.1-127 of the Code of Virginia, relating to hospitals, nursing homes, and certified nursing facilities; regulations; patient access to intelligent personal assistant.
- H.B. 2177. A BILL to create a six-year capital outlay plan for projects to be funded entirely or partially from general fund—supported resources and to repeal Chapter 1134 of the Acts of Assembly of 2020.
- **H.B. 2182.** A BILL to require the Board of Education to amend the regulatory definition of traumatic brain injury.
- **H.B. 2187.** A BILL to direct study topics for the Commonwealth Center for Recurrent Flooding Resiliency.

- **H.B. 2203.** A BILL to amend the Code of Virginia by adding in Title 3.2 a chapter numbered 47.1, consisting of sections numbered 3.2-4780 through 3.2-4783, relating to the Virginia Agriculture Food Assistance Program and Fund; established.
- **H.B. 2204.** A BILL to amend the Code of Virginia by adding a section numbered 23.1-2911.2, relating to the establishment of the Get Skilled, Get a Job, Give Back (G3) Fund and Program.
- **H.B. 2271.** A BILL to require the Joint Commission on Health Care to enter into a contract for a study of options for financing universal health care in the Commonwealth.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--33. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--33.

NAYS--Deeds, DeSteph, McDougle, Petersen--4.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1846 and H.B. 1893 were referred to the Committee on Transportation.
- H.B. 1918, H.B. 1930, H.B. 1950, H.B. 1986, H.B. 1995, H.B. 1998, H.B. 2013, H.B. 2019, H.B. 2105, H.B. 2111, H.B. 2119, H.B. 2135, H.B. 2182, and H.B. 2204 were referred to the Committee on Education and Health.
 - H.B. 1925 was referred to the Committee on Commerce and Labor.
- **H.B. 1957**, **H.B. 2092**, and **H.B. 2154** were referred to the Committee on Rehabilitation and Social Services.
 - H.B. 2001 was referred to the Committee on General Laws and Technology.
- H.B. 2068, H.B. 2187, and H.B. 2203 were referred to the Committee on Agriculture, Conservation and Natural Resources.
 - H.B. 2177 was referred to the Committee on Finance and Appropriations.

H.B. 2271 was referred to the Committee on Rules.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Favola from the Committee on Rehabilitation and Social Services:

- **S.B. 1301** (one thousand three hundred one) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1363** (one thousand three hundred sixty-three) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B.** 1397 (one thousand three hundred ninety-seven) with amendments.
- **S.B. 1428** (one thousand four hundred twenty-eight) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1462** (one thousand four hundred sixty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1471** (one thousand four hundred seventy-one) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **S.B. 1472** (one thousand four hundred seventy-two) with substitute.

The following bills, having been considered by the committee in session, were reported by Senator Marsden from the Committee on Transportation:

- S.B. 1144 (one thousand one hundred forty-four) with substitute.
- **S.B.** 1160 (one thousand one hundred sixty) with substitute.
- **S.B. 1259** (one thousand two hundred fifty-nine) with substitute.
- S.B. 1260 (one thousand two hundred sixty) with substitute.
- S.B. 1470 (one thousand four hundred seventy) with amendment.
- S.B. 1474 (one thousand four hundred seventy-four) with substitute.

The following bills, joint resolutions, and resolution, having been considered by the committee in session, were reported by Senator Locke from the Committee on Rules:

- S.B. 1273 (one thousand two hundred seventy-three) with amendments.
- S.B. 1473 (one thousand four hundred seventy-three).
- S.J.R. 288 (two hundred eighty-eight).
- S.J.R. 293 (two hundred ninety-three) with amendments.
- S.J.R. 323 (three hundred twenty-three).
- S.R. 95 (ninety-five) with amendment.
- S.B. 1301, S.B. 1363, S.B. 1428, S.B. 1462, and S.B. 1471 were rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 334. Commending Major Wayne R. Lee, USA, Ret.

Patrons--Reeves, Bell, Boysko, Hashmi, Howell, Ruff and Stanley; Delegates: Adams, D.M., Avoli, Byron, Cole, M.L., Fowler, Heretick, Hope, Mundon King, Rasoul, Reid, Simonds and Ware

- **S.R. 104.** Celebrating the life of the Honorable Leslie D. Campbell, Jr. Patron--McDougle
- S.R. 105. Commending the Virginia Piedmont Heritage Area Association. Patron--Vogel

CALENDAR

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1447 (one thousand four hundred forty-seven).
- S.B. 1102 (one thousand one hundred two).
- S.B. 1150 (one thousand one hundred fifty).
- S.B. 1204 (one thousand two hundred four).
- S.B. 1221 (one thousand two hundred twenty-one).
- S.B. 1227 (one thousand two hundred twenty-seven).
- S.B. 1251 (one thousand two hundred fifty-one).
- **S.B. 1269** (one thousand two hundred sixty-nine).
- **S.B.** 1328 (one thousand three hundred twenty-eight).
- S.B. 1351 (one thousand three hundred fifty-one).
- S.B. 1438 (one thousand four hundred thirty-eight).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- S.B. 1102 (one thousand one hundred two).
- **S.B.** 1150 (one thousand one hundred fifty).
- S.B. 1204 (one thousand two hundred four).
- S.B. 1221 (one thousand two hundred twenty-one).
- S.B. 1251 (one thousand two hundred fifty-one).
- S.B. 1269 (one thousand two hundred sixty-nine).
- **S.B.** 1328 (one thousand three hundred twenty-eight).
- S.B. 1351 (one thousand three hundred fifty-one).

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--37.

NAYS--0.

RULE 36--0.

S.B. 1447 (one thousand four hundred forty-seven) was taken up,

RECONSIDERATION

Senator Peake moved to reconsider the vote by which S.B. 1447 (one thousand four hundred forty-seven) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--36.

NAYS--0.

RULE 36--0.

Senator Peake offered the following amendment:

1. Line 12, introduced, after Augusta,

insert

Buckingham,

On motion of Senator Peake, the reading of the amendment was waived.

On motion of Senator Peake, the amendment was agreed to.

On motion of Senator Peake, the bill was ordered to be engrossed and read by title the third time.

Senator Peake moved that the Rules be suspended and the third reading of the title of **S.B. 1447** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1447, on motion of Senator Peake, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0. RULE 36--0.

S.B. 1227 (one thousand two hundred twenty-seven), on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0. RULE 36--0.

S.B. 1438 (one thousand four hundred thirty-eight), on motion of Senator Hanger, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0. RULE 36--0.

THE PRESIDENT PRESIDING

The President assumed the Chair.

S.B. 1147 (one thousand one hundred forty-seven) was read by title the third time and, on motion of Senator Kiggans, was passed with its title.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--37.

NAYS--0.

RULE 36--0.

S.B. 1195 (one thousand one hundred ninety-five) was read by title the third time and, on motion of Senator Obenshain, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--5. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Petersen, Saslaw, Spruill, Stanley, Suetterlein, Surovell--31.

NAYS--Chase, Newman, Peake, Pillion, Ruff--5.

RULE 36--Reeves--1.

S.B. 1207 (one thousand two hundred seven) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--37.

NAYS--0.

RULE 36--0.

S.B. 1223 (one thousand two hundred twenty-three) was read by title the third time and, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Surovell--22. NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Suetterlein--15.

RULE 36--0.

S.B. 1298 (one thousand two hundred ninety-eight) was read by title the third time and, on motion of Senator Bell, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--37.

NAYS--0.

RULE 36--0.

S.B. 1302 (one thousand three hundred two) was read by title the third time and, on motion of Senator McPike, was passed with its title.

The recorded vote is as follows:

YEAS--30. NAYS--7. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Peake, Petersen, Pillion, Reeves, Saslaw, Spruill, Suetterlein, Surovell--30.

NAYS--Chase, Cosgrove, McDougle, Newman, Obenshain, Ruff, Stanley--7. RULE 36--0.

S.B. 1321 (one thousand three hundred twenty-one) was read by title the third time and, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--24.

NAYS--Chase, Cosgrove, Hanger, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--13.

RULE 36--0.

STATEMENT ON VOTE

Senator Kiggans stated that she voted yea on the question of the passage of S.B. 1321, whereas she intended to vote nay.

S.B. 1334 (one thousand three hundred thirty-four) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--36.

NAYS--Newman--1.

RULE 36--0.

S.B. 1367 (one thousand three hundred sixty-seven) was read by title the third time and, on motion of Senator DeSteph, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--35.

NAYS--Deeds, Locke--2.

RULE 36--0.

- **S.B. 1403** (one thousand four hundred three), on motion of Senator Petersen, was passed by for the day.
- **S.B. 1413** (one thousand four hundred thirteen) was read by title the third time and, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--28. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Surovell--28.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, Newman, Norment, Reeves, Suetterlein--9. RULE 36--0.

STATEMENT ON VOTE

Senator Chase stated that she voted nay on the question of the passage of S.B. 1413, whereas she intended to vote yea.

HOUSE BILL ON SECOND READING

H.B. 2333 (two thousand three hundred thirty-three) was read by title the second time.

SENATE BILLS ON SECOND READING

- **S.B. 1104** (one thousand one hundred four), on motion of Senator Morrissey, was passed by for the day.
- **S.B. 1246** (one thousand two hundred forty-six), on motion of Senator Deeds, was passed by for the day.

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1119 (one thousand one hundred nineteen).
- **S.B.** 1125 (one thousand one hundred twenty-five).
- **S.B.** 1135 (one thousand one hundred thirty-five).
- S.B. 1143 (one thousand one hundred forty-three).
- S.B. 1188 (one thousand one hundred eighty-eight).
- S.B. 1193 (one thousand one hundred ninety-three).
- **S.B. 1279** (one thousand two hundred seventy-nine).
- S.B. 1331 (one thousand three hundred thirty-one).
- S.B. 1417 (one thousand four hundred seventeen).
- S.B. 1423 (one thousand four hundred twenty-three).
- **S.B.** 1453 (one thousand four hundred fifty-three).

The motion was agreed to.

S.B. 1125 (one thousand one hundred twenty-five) was taken up.

The following amendments proposed by the Committee on Rehabilitation and Social Services were offered:

1. Line 33, introduced, after notice

insert

. Nothing in this subdivision shall be construed to delay the date upon which a prisoner is scheduled to be released on parole

2. After line 55, introduced

insert

For the purposes of this section, "victim" means the same as that term is defined in § 19.2-11.01.

The reading of the amendments was waived.

On motion of Senator Obenshain, the amendments were agreed to.

S.B. 1135 (one thousand one hundred thirty-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 3.2-6540 and 3.2-6542 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 3.2-6540.01 through 3.2-6540.04, 3.2-6541.1, 3.2-6542.1, 3.2-6542.2, 3.2-6543.1, 3.2-6562.2, and 18.2-52.2 relating to dangerous dogs; penalty.

The reading of the substitute was waived.

Senator Marsden moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

Senator Marsden offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 3.2-6540 and 3.2-6542 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 3.2-6540.01 through 3.2-6540.04, 3.2-6541.1, 3.2-6542.1, 3.2-6542.2, 3.2-6543.1, 3.2-6562.2, and 18.2-52.2 relating to dangerous dogs; penalty.

On motion of Senator Marsden, the reading of the substitute was waived.

On motion of Senator Marsden, the substitute was agreed to.

S.B. 1188 (one thousand one hundred eighty-eight) was taken up.

Senator Norment offered the following amendment:

1. Line 45, introduced, after *chapter*: insert

Such regulations shall prohibit any charitable food assistance organization that contracts with or receives a grant from the Commissioner from attributing more than five percent of the total grant or contract amount to administrative costs.

On motion of Senator Norment, the reading of the amendment was waived.

On motion of Senator Norment, the amendment was agreed to.

S.B. 1193 (one thousand one hundred ninety-three) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Title 3.2 a chapter numbered 33.1, consisting of sections numbered 3.2-3304 through 3.2-3307, relating to Dairy Producer Margin Coverage Premium Assistance Program.

The reading of the substitute was waived.

On motion of Senator Obenshain, the substitute was agreed to.

S.B. 1417 (one thousand four hundred seventeen) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 3.2-6591 of the Code of Virginia and to amend the Code of Virginia by adding in Article 13 of Chapter 65 of Title 3.2 a section numbered 3.2-6593.1, relating to animal testing facilities; adoption of dogs and cats; civil penalty.

The reading of the substitute was waived.

On motion of Senator Reeves, the substitute was agreed to.

S.B. 1423 (one thousand four hundred twenty-three) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

The reading of the substitute was waived.

Senator McPike moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

Senator McPike offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

On motion of Senator McPike, the reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- **S.B.** 1119 (one thousand one hundred nineteen).
- **S.B.** 1125 (one thousand one hundred twenty-five) as amended.
- S.B. 1135 (one thousand one hundred thirty-five) as amended.
- **S.B.** 1143 (one thousand one hundred forty-three).
- **S.B.** 1188 (one thousand one hundred eighty-eight) as amended.
- **S.B.** 1193 (one thousand one hundred ninety-three) as amended.
- **S.B. 1279** (one thousand two hundred seventy-nine).
- **S.B. 1331** (one thousand three hundred thirty-one).
- **S.B. 1417** (one thousand four hundred seventeen) as amended.
- S.B. 1423 (one thousand four hundred twenty-three) as amended.
- S.B. 1453 (one thousand four hundred fifty-three).
- **S.B.** 1366 (one thousand three hundred sixty-six) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 51.5-134 and 51.5-135 of the Code of Virginia, relating to aging services; economic and social need.

The reading of the substitute was waived.

Senator Barker moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

Senator Barker offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 51.5-134 and 51.5-135 of the Code of Virginia, relating to aging services; economic and social need.

On motion of Senator Barker, the reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

On motion of Senator Barker, the bill was ordered to be engrossed and read by title the third time.

S.B. 1393 (one thousand three hundred ninety-three) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Local Government was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 15.2-961 and 15.2-961.1 of the Code of Virginia, relating to replacement and conservation of trees during development.

The reading of the substitute was waived.

On motion of Senator Marsden, the substitute was agreed to.

Senator Marsden offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 15.2-961 and 15.2-961.1 of the Code of Virginia, relating to replacement and conservation of trees during development.

Senator Marsden withdrew his substitute.

Senator Marsden offered the following amendments to the committee substitute:

1. Line 66, substitute, after Area

insert

to address recurrent flooding

2. Line 282, substitute, after Preservation Area

insert

to address recurrent flooding

3. Line 295, substitute, after including

strike

the remainder of line 295 and through Virginia and on line 296

4. Line 301, substitute, after development

strike

and construction

5. Line 307, substitute, after trees, [first instance]

strike

or

insert

and

On motion of Senator Marsden, the reading of the amendments was waived.

On motion of Senator Marsden, the amendments were agreed to.

On motion of Senator Marsden, the bill was ordered to be engrossed and read by title the third time.

S.B. 1165 (one thousand one hundred sixty-five), on motion of Senator Surovell, was passed by for the day.

- **S.B. 1170** (one thousand one hundred seventy), on motion of Senator Norment, was passed by for the day.
- **S.B.** 1103 (one thousand one hundred three) was read by title the second time and, on motion of Senator Suetterlein, was ordered to be engrossed and read by title the third time.
 - S.B. 1106 (one thousand one hundred six), on motion of Senator Stanley, was passed by for the day.
 - S.B. 1108 (one thousand one hundred eight) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 8.01-195.4, 16.1-77, and 16.1-107 of the Code of Virginia, relating to general district courts; jurisdictional limits; appeal bond; emergency.

The reading of the substitute was waived.

Senator Stanley moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 8.01-195.4, 16.1-77, and 16.1-107 of the Code of Virginia, relating to general district courts; jurisdictional limits; appeal bond; emergency.

The reading of the substitute was waived.

On motion of Senator Stanley, the substitute was agreed to.

On motion of Senator Stanley, the bill was ordered to be engrossed and read by title the third time.

S.B. 1115 (one thousand one hundred fifteen) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446 of the Code of Virginia, relating to industrial hemp; emergency.

The reading of the substitute was waived.

On motion of Senator Peake, the substitute was agreed to.

On motion of Senator Peake, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1148** (one thousand one hundred forty-eight) was read by title the second time and, on motion of Senator Kiggans, was ordered to be engrossed and read by title the third time.
 - S.B. 1245 (one thousand two hundred forty-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Privileges and Elections was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 24.2-706, 24.2-707, 24.2-709, 24.2-709.1, and 24.2-710 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 24.2-707.1, relating to absentee voting; establishment of drop-off locations; ballot defects; cure process.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

On motion of Senator Deeds, the bill was ordered to be engrossed and read by title the third time.

S.B. 1254 (one thousand two hundred fifty-four) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100 of the Code of Virginia, relating to sports betting; technical amendments.

The reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

On motion of Senator McPike, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1300** (one thousand three hundred) was read by title the second time and, on motion of Senator Favola, was ordered to be engrossed and read by title the third time.
 - S.B. 1305 (one thousand three hundred five) was read by title the second time.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 16, introduced, after than

strike

\$250,000

insert

\$500,000

2. Line 35, introduced

strike

all of lines 35, 36, and 37

The reading of the amendments was waived.

Senator McPike moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. Line 28, introduced, after subcontractor's

strike

workforce

insert

work

The reading of the amendment was waived.

Senator McPike moved that the amendment be rejected.

The question was put on agreeing to the amendment.

The amendment was rejected.

Senator McPike offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding a section numbered 2.2-4303.02, relating to the Virginia Public Procurement Act; construction contracts; subcontractor workforce requirements.

On motion of Senator McPike, the reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

On motion of Senator McPike, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1314** (one thousand three hundred fourteen) was read by title the second time and, on motion of Senator Hashmi, was ordered to be engrossed and read by title the third time.
 - S.B. 1379 (one thousand three hundred seventy-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-574, relating to Humane Cosmetics Act; civil penalties.

The reading of the substitute was waived.

On motion of Senator Boysko, the substitute was agreed to.

On motion of Senator Boysko, the bill was ordered to be engrossed and read by title the third time.

S.B. 1387 (one thousand three hundred eighty-seven) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 23.1-506 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 23.1-505.1, relating to the State Council of Higher Education for Virginia; eligibility for in-state tuition.

The reading of the substitute was waived.

On motion of Senator Boysko, the substitute was agreed to.

On motion of Senator Boysko, the bill was ordered to be engrossed and read by title the third time.

- S.B. 1390 (one thousand three hundred ninety), on motion of Senator Lewis, was passed by for the day.
- **S.B. 1402** (one thousand four hundred two) was read by title the second time and, on motion of Senator Suetterlein, was ordered to be engrossed and read by title the third time.
 - S.B. 1412 (one thousand four hundred twelve) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 3.2-6511.1 and 3.2-6511.2 of the Code of Virginia, relating to pet shops, dealers, and dog breeders; employees convicted of animal abuse; penalty.

The reading of the substitute was waived.

On motion of Senator Stanley, the substitute was agreed to.

On motion of Senator Stanley, the bill was ordered to be engrossed and read by title the third time.

S.B. 1422 (one thousand four hundred twenty-two) was read by title the second time and, on motion of Senator Kiggans, was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- **S.B.** 1176 (one thousand one hundred seventy-six).
- S.B. 1183 (one thousand one hundred eighty-three).
- **S.B.** 1184 (one thousand one hundred eighty-four).
- **S.B. 1225** (one thousand two hundred twenty-five).
- **S.B. 1237** (one thousand two hundred thirty-seven).
- **S.B. 1272** (one thousand two hundred seventy-two).
- **S.B. 1288** (one thousand two hundred eighty-eight).
- **S.B. 1304** (one thousand three hundred four).
- **S.B. 1327** (one thousand three hundred twenty-seven).

```
S.B. 1349 (one thousand three hundred forty-nine).
```

S.B. 1356 (one thousand three hundred fifty-six).

S.B. 1410 (one thousand four hundred ten).

S.B. 1431 (one thousand four hundred thirty-one).

S.B. 1437 (one thousand four hundred thirty-seven).

S.B. 1456 (one thousand four hundred fifty-six).

S.B. 1458 (one thousand four hundred fifty-eight).

S.B. 1464 (one thousand four hundred sixty-four).

S.B. 1138 (one thousand one hundred thirty-eight).

S.B. 1178 (one thousand one hundred seventy-eight).

S.B. 1180 (one thousand one hundred eighty). S.B. 1198 (one thousand one hundred ninety-eight).

S.B. 1215 (one thousand two hundred fifteen).

S.B. 1235 (one thousand two hundred thirty-five).

S.B. 1266 (one thousand two hundred sixty-six).

S.B. 1287 (one thousand two hundred eighty-seven).

S.B. 1303 (one thousand three hundred three).

S.B. 1317 (one thousand three hundred seventeen).

S.B. 1325 (one thousand three hundred twenty-five).

S.B. 1381 (one thousand three hundred eighty-one).

S.B. 1384 (one thousand three hundred eighty-four).

S.B. 1439 (one thousand four hundred thirty-nine).

S.B. 1468 (one thousand four hundred sixty-eight).

The motion was agreed to.

The recorded vote is as follows:

YEAS--35. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--35.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

```
S.B. 1176 (one thousand one hundred seventy-six).
```

S.B. 1183 (one thousand one hundred eighty-three).

S.B. 1184 (one thousand one hundred eighty-four).

S.B. 1225 (one thousand two hundred twenty-five).

S.B. 1237 (one thousand two hundred thirty-seven). **S.B. 1272** (one thousand two hundred seventy-two).

S.B. 1288 (one thousand two hundred eighty-eight).

S.B. 1304 (one thousand three hundred four).

S.B. 1327 (one thousand three hundred twenty-seven).

S.B. 1349 (one thousand three hundred forty-nine).

S.B. 1356 (one thousand three hundred fifty-six). S.B. 1410 (one thousand four hundred ten).

S.B. 1431 (one thousand four hundred thirty-one).

S.B. 1437 (one thousand four hundred thirty-seven).

- S.B. 1456 (one thousand four hundred fifty-six).
- S.B. 1458 (one thousand four hundred fifty-eight).
- S.B. 1464 (one thousand four hundred sixty-four).
- S.B. 1138 (one thousand one hundred thirty-eight).
- S.B. 1178 (one thousand one hundred seventy-eight).
- S.B. 1180 (one thousand one hundred eighty).
- **S.B.** 1198 (one thousand one hundred ninety-eight).
- S.B. 1215 (one thousand two hundred fifteen).
- S.B. 1235 (one thousand two hundred thirty-five).
- S.B. 1266 (one thousand two hundred sixty-six).
- S.B. 1287 (one thousand two hundred eighty-seven).
- **S.B. 1303** (one thousand three hundred three).
- **S.B.** 1317 (one thousand three hundred seventeen).
- **S.B. 1325** (one thousand three hundred twenty-five).
- **S.B. 1381** (one thousand three hundred eighty-one).
- S.B. 1384 (one thousand three hundred eighty-four).
- **S.B. 1439** (one thousand four hundred thirty-nine).
- **S.B.** 1468 (one thousand four hundred sixty-eight).

SENATE JOINT RESOLUTION ON SECOND READING

S.J.R. 327 (three hundred twenty-seven) was read by title the second time and, on motion of Senator Deeds, was ordered to be engrossed and read by title the third time.

SENATE BILL ON THIRD READING RECONSIDERATION

Senator Peake moved to reconsider the vote by which **S.B. 1413** (one thousand four hundred thirteen) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--35. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell--35.

NAYS--0.

RULE 36--0.

S.B. 1413, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--28. NAYS--7. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Obenshain, Peake, Ruff, Saslaw, Spruill, Stanley, Surovell--28.

NAYS--DeSteph, Hanger, Kiggans, Newman, Norment, Reeves, Suetterlein--7.

RULE 36--0.

On motion of Senator Lucas, the Senate adjourned until Monday, February 1, 2021, at 12 m.

Jacke Ochson

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

MONDAY, FEBRUARY 1, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable George L. Barker, Thirty-ninth Senatorial District, offered the following prayer:

O God, I thank You for each and every one of us who are here today. I appreciate greatly my colleagues in the Senate and we collectively appreciate the opportunity to work together on the great responsibility entrusted to us to make our Commonwealth good for all Virginians. Please help us to reach agreements to solutions and solutions to challenges we face so that we can help value the future of all Virginians and rectify damage from our past so that all Virginians are one. We have weeks to go before we leave. Please help us as we work hard and that we leave proud of what we have done, who we have helped, and how we have done it for all the people of Virginia. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Nathan A. Hatfield, Assistant Clerk-LINCS, Senate Clerk's Office.

On motion of Senator McDougle, the Rules were suspended to allow Senators Norment and Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--37.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Stanley, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, De Steph, McDougle, Petersen, Reeves--5.

RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates January 29, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1737.** A BILL to amend and reenact § 54.1-2957 of the Code of Virginia, relating to nurse practitioners; practice without a practice agreement.
- **H.B. 1832.** A BILL to amend and reenact §§ 56-539 and 56-542 of the Code of Virginia, relating to Virginia Highway Corporation Act; alteration of certificate of authority; powers and duties of the State Corporation Commission.
- **H.B. 1834.** A BILL to amend and reenact § 56-599 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 26 of Title 45.1 a section numbered 45.1-394.1, relating to public disclosure of electric generating facility closures; integrated resource plans.
- **H.B. 1854.** A BILL to amend the Code of Virginia by adding a section numbered 15.2-719.1, relating to naming U.S. Route 29; county manager plan of government.
- **H.B. 1864.** A BILL to amend and reenact § 2.2-3905 of the Code of Virginia, relating to Virginia Human Rights Act; definition of employer; person employing one or more domestic workers.
- **H.B. 1876.** A BILL to amend and reenact § 2.2-435.8 of the Code of Virginia, relating to workforce development; data sharing.
- **H.B. 1877.** A BILL to amend and reenact § 59.1-441.2 of the Code of Virginia, relating to legal service plans; seller registration.
- **H.B. 1889.** A BILL to repeal the second enactment of Chapter 46 of the Acts of Assembly of 2020, Special Session I, relating to the Virginia Residential Landlord and Tenant Act; landlord remedies; noncompliance with rental agreement; payment plan; removal of sunset.
- **H.B. 1891.** A BILL to amend and reenact § 2.2-1201 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 12 of Title 2.2 a section numbered 2.2-1212, relating to the Department of Human Resource Management; duties of the Department; annual safety and disaster awareness training.
- H.B. 1907. A BILL to amend and reenact § 56-585.5 of the Code of Virginia, relating to electric utilities; Renewable Energy Certificates; contracts with accelerated renewable energy buyers; exemption from certain costs.
- **H.B. 1926.** A BILL to amend and reenact § 33.2-3703 of the Code of Virginia, relating to Central Virginia Transportation Authority; membership.
- **H.B. 1971.** A BILL to amend and reenact § 36-96.3:2 of the Code of Virginia, relating to the Virginia Fair Housing Law; reasonable accommodations; disability-related requests for parking.

- **H.B. 1993.** A BILL to amend and reenact § 2.2-602 of the Code of Virginia, relating to duties of agencies and their appointing authorities; diversity, equity, and inclusion strategic plans.
- **H.B. 2008.** A BILL to amend and reenact § 38.2-3407.15:2 of the Code of Virginia, relating to health insurance; authorization of drug prescribed for the treatment of a mental disorder.
- **H.B. 2009.** A BILL to revert certain property upon which the Chamberlin Hotel at Fort Monroe is located to the Commonwealth and to repeal § 1 of Chapter 809 of the Acts of Assembly of 1998, as amended by Chapter 713 of the Acts of Assembly of 2004.
- H.B. 2029. A BILL to amend the Code of Virginia by adding a section numbered 9.1-207.2, relating to Department of Fire Programs; prohibition on the use of certain oriented strand board in fire training activities.
- **H.B. 2036.** A BILL to amend the Code of Virginia by adding in Article 2 of Chapter 1 of Title 60.2 a section numbered 60.2-121.1, relating to Virginia Employment Commission; communications with parties; report.
- **H.B. 2046.** A BILL to amend and reenact §§ 36-96.3 and 36-96.17 of the Code of Virginia, relating to the Virginia Fair Housing Law; unlawful discriminatory housing practices.
- **H.B. 2071.** A BILL to amend and reenact §§ 33.2-214.2 and 33.2-353 of the Code of Virginia, relating to transportation projects; resiliency.
- **H.B. 2072.** A BILL to direct the Virginia Housing Development Authority to report on recommendations for the creation of a Virginia Good Neighbor Next Door Program.
- H.B. 2140. A BILL to amend and reenact § 2.2-2901.1 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 12 of Title 2.2 a section numbered 2.2-1212, relating to Department of Human Resource Management; alternative application for employment for persons with a disability; report.
- **H.B. 2191.** A BILL to amend and reenact § 63.2-104 of the Code of Virginia, relating to local department of social services; location of child in local department's custody.
- **H.B. 2229.** A BILL to amend and reenact §§ 54.1-2108.1 and 55.1-1237 of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; responsibilities of real estate brokers; foreclosure of single-family residential dwelling units.
- **H.B. 2282.** A BILL to direct the State Corporation Commission to recommend policy proposals to accelerate transportation electrification in the Commonwealth; report.
- **H.B. 2304.** A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to provision of broadband services by investor-owned electric utilities.
- H.B. 2307. A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-581, relating to Consumer Data Protection Act
- THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTIONS:

- **H.J.R. 522.** Continuing the Joint Committee of the House Committee on Health, Welfare and Institutions; the House Committee on Public Safety; the Senate Committee on the Judiciary; and the Senate Committee on Rehabilitation and Social Services Studying Staffing Levels, Employment Conditions, and Compensation at the Virginia Department of Corrections. Report.
- H.J.R. 526. Establishing a joint subcommittee to study comprehensive campaign finance reform. Report.
- **H.J.R. 563.** Directing the Division of Legislative Services, in conjunction with the Department of Taxation, to establish a work group to assess the feasibility of transitioning to a unitary combined reporting system for corporate income tax purposes.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, DeSteph, McDougle, Petersen--4.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1737 was referred to the Committee on Education and Health.
- H.B. 1832, H.B. 1834, H.B. 1877, H.B. 1907, H.B. 2008, H.B. 2036, H.B. 2282, and H.B. 2304 were referred to the Committee on Commerce and Labor.
 - H.B. 1854, H.B. 1926, and H.B. 2071 were referred to the Committee on Transportation.
- H.B. 1864, H.B. 1876, H.B. 1889, H.B. 1891, H.B. 1971, H.B. 1993, H.B. 2029, H.B. 2046, H.B. 2072, H.B. 2140, H.B. 2229, and H.B. 2307 were referred to the Committee on General Laws and Technology.
 - H.B. 2009 was referred to the Committee on the Judiciary.
 - H.B. 2191 was referred to the Committee on Rehabilitation and Social Services.

The House joint resolutions, communicated as agreed to by the House of Delegates, the first reading of their titles having been waived, were referred as follows:

H.J.R. 522, H.J.R. 526, and H.J.R. 563 were referred to the Committee on Rules.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- **S.B.** 1105 (one thousand one hundred five) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1113** (one thousand one hundred thirteen) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1129 (one thousand one hundred twenty-nine) with substitute.
 - **S.B. 1234** (one thousand two hundred thirty-four) with substitute.
 - S.B. 1306 (one thousand three hundred six) with substitute.
- **S.B. 1339** (one thousand three hundred thirty-nine) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **S.B. 1406** (one thousand four hundred six) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1415 (one thousand four hundred fifteen) with amendments.
 - **S.B. 1426** (one thousand four hundred twenty-six) with amendments.
 - **S.B. 1446** (one thousand four hundred forty-six) with amendment.
- **S.B. 1465** (one thousand four hundred sixty-five) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1475 (one thousand four hundred seventy-five) with substitute.

The following bills, having been considered by the committee in session, were reported by Senator Lewis from the Committee on Local Government:

- S.B. 1267 (one thousand two hundred sixty-seven).
- S.B. 1385 (one thousand three hundred eighty-five) with substitute.
- H.B. 1743 (one thousand seven hundred forty-three).
- **H.B.** 1768 (one thousand seven hundred sixty-eight).
- H.B. 1825 (one thousand eight hundred twenty-five).
- H.B. 1859 (one thousand eight hundred fifty-nine).
- H.B. 1927 (one thousand nine hundred twenty-seven).
- **H.B. 1949** (one thousand nine hundred forty-nine).
- S.B. 1105, S.B. 1113, S.B. 1339, S.B. 1406, and S.B. 1465 were rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 335. Celebrating the life of Roy L. Pearson.

Patron--Mason

S.J.R. 336. Commending Frank Shatz.

Patron--Mason

S.J.R. 337. Celebrating the life of Mary Esterine Hundley Moyler.

Patron--Mason

S.J.R. 338. Celebrating the life of Jane Carey Gardner. Patron--Favola

S.J.R. 339. Celebrating the life of Edward A. Chappell, Jr.

Patron--Mason

S.J.R. 340. Commending Edwin C. Roessler, Jr.

Patrons--Surovell; Delegates: Helmer, Krizek, Simon and Watts

S.J.R. 341. Commending Carl Hershner.

Patron--Mason

S.J.R. 342. Celebrating the life of Wendell Harding Butler.

Patron--Edwards

S.J.R. 343. Celebrating the life of Charles Wallace Nuttycombe, Sr.

Patron--Mason

S.J.R. 344. Commending Raymond O. Anderson.

Patron--Mason

S.J.R. 345. Celebrating the life of Sigmund Edward Davidson.

Patron--Edwards

S.J.R. 346. Commending John T. Wells.

Patrons--Norment and Mason; Delegate: Hodges

S.R. 106. Celebrating the life of Avicia Beatrice Hooper Thorpe.

Patron--Stanley

S.R. 107. Celebrating the life of Elizabeth Bateman Stone.

Patron--Stanley

S.R. 108. Celebrating the life of Bryan Wayne Galentine.

Patrons--Stanley and Ruff

S.R. 109. Commending telehealth care providers in Virginia.

Patron--Stanley

S.R. 110. Celebrating the life of Thomas Henry Francis.

Patrons--Hashmi and McClellan

S.R. 111. Celebrating the life of Pauline Allen Mitchell.

Patron--Hashmi

CALENDAR

HOUSE BILL ON THIRD READING

H.B. 2333 (two thousand three hundred thirty-three) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to facilitate the administration of the COVID-19 vaccine; emergency.

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

Senator Dunnavant offered the following amendment to the substitute:

1. Line 110, substitute, after *squads* strike

the remainder of line 110 and through registered nurses on line 111

On motion of Senator Dunnavant, the reading of amendment No. 1 was waived.

Senator Dunnavant moved that amendment No. 1 be agreed to.

H.B. 2333, on motion of Senator McPike, was passed by temporarily.

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- **S.B.** 1119 (one thousand one hundred nineteen).
- **S.B.** 1125 (one thousand one hundred twenty-five).
- **S.B.** 1135 (one thousand one hundred thirty-five).
- S.B. 1143 (one thousand one hundred forty-three).
- S.B. 1188 (one thousand one hundred eighty-eight).
- S.B. 1193 (one thousand one hundred ninety-three).
- **S.B. 1279** (one thousand two hundred seventy-nine). **S.B. 1331** (one thousand three hundred thirty-one).
- S.B. 1417 (one thousand four hundred seventeen).
- S.B. 1423 (one thousand four hundred twenty-three).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- S.B. 1119 (one thousand one hundred nineteen).
- S.B. 1125 (one thousand one hundred twenty-five).
- **S.B.** 1135 (one thousand one hundred thirty-five).
- **S.B.** 1143 (one thousand one hundred forty-three).

S.B. 1188 (one thousand one hundred eighty-eight).

S.B. 1193 (one thousand one hundred ninety-three).

S.B. 1279 (one thousand two hundred seventy-nine).

S.B. 1331 (one thousand three hundred thirty-one).

S.B. 1417 (one thousand four hundred seventeen).

S.B. 1423 (one thousand four hundred twenty-three).

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1453 (one thousand four hundred fifty-three) was taken up.

RECONSIDERATION

Senator Edwards moved to reconsider the vote by which **S.B. 1453** (one thousand four hundred fifty-three) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Edwards offered the following amendments:

1. Line 713, introduced, after mine

strike

the remainder of line 713, all of lines 714 and 715, and through *person* on line 716

2. Line 6234, introduced, after mine

strike

the remainder of line 6234, all of lines 6235 and 6236, and through *person* on line 6237

On motion of Senator Edwards, the reading of the amendments was waived.

On motion of Senator Edwards, the amendments were agreed to.

On motion of Senator Edwards, the bill was ordered to be engrossed and read by title the third time.

Senator Edwards moved that the Rules be suspended and the third reading of the title of **S.B. 1453** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1453, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

RECONSIDERATION

Senator Mason moved to reconsider the vote by which the following Senate bills were passed en bloc with their titles:

- S.B. 1119 (one thousand one hundred nineteen).
- S.B. 1125 (one thousand one hundred twenty-five).
- **S.B.** 1135 (one thousand one hundred thirty-five).
- **S.B.** 1143 (one thousand one hundred forty-three).
- **S.B.** 1188 (one thousand one hundred eighty-eight).
- **S.B.** 1193 (one thousand one hundred ninety-three).
- S.B. 1279 (one thousand two hundred seventy-nine).
- **S.B. 1331** (one thousand three hundred thirty-one).
- S.B. 1417 (one thousand four hundred seventeen).
- S.B. 1423 (one thousand four hundred twenty-three).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Ebbin, the following Senate bills were passed en bloc with their titles:

S.B. 1119 (one thousand one hundred nineteen).

S.B. 1125 (one thousand one hundred twenty-five).

S.B. 1135 (one thousand one hundred thirty-five).

S.B. 1143 (one thousand one hundred forty-three).

S.B. 1188 (one thousand one hundred eighty-eight).

S.B. 1193 (one thousand one hundred ninety-three).

S.B. 1279 (one thousand two hundred seventy-nine).

S.B. 1331 (one thousand three hundred thirty-one).

S.B. 1417 (one thousand four hundred seventeen).

S.B. 1423 (one thousand four hundred twenty-three).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1403 (one thousand four hundred three) was read by title the third time and, on motion of Senator Pillion, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--1.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--Stanley--1.

S.B. 1366 (one thousand three hundred sixty-six) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--23. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--16.

RULE 36--0.

STATEMENT ON VOTE

Senator Norment stated that he voted yea on the question of the passage of **S.B. 1366**, whereas he intended to vote nay.

S.B. 1393 (one thousand three hundred ninety-three) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--26.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Stuart, Suetterlein--13.

RULE 36--0.

S.B. 1103 (one thousand one hundred three) was read by title the third time and, on motion of Senator Suetterlein, was passed with its title.

The recorded vote is as follows:

YEAS--33. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Edwards, Favola, Hanger, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Vogel--33.

NAYS--Deeds, Ebbin, Hashmi, Locke, Spruill, Surovell--6.

RULE 36--0.

S.B. 1108 (one thousand one hundred eight) was read by title the third time and, on motion of Senator Stanley, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Petersen, Pillion, Reeves, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Chase, Norment, Peake, Ruff--4.

RULE 36--0.

S.B. 1115 (one thousand one hundred fifteen) was read by title the third time and, on motion of Senator Peake, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1148 (one thousand one hundred forty-eight) was read by title the third time and, on motion of Senator Kiggans, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--12. RULE 36--0.

YEAS--Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, Newman, Norment, Peake, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Vogel--27.

NAYS--Barker, Chase, Hanger, Marsden, McDougle, McPike, Morrissey, Obenshain, Petersen, Reeves, Suetterlein, Surovell--12.

RULE 36--0.

S.B. 1245 (one thousand two hundred forty-five) was read by title the third time and, on motion of Senator Deeds, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1254 (one thousand two hundred fifty-four) was read by title the third time and, on motion of Senator McPike, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1300 (one thousand three hundred) was read by title the third time and, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

- **S.B. 1305** (one thousand three hundred five), on motion of Senator McPike, was passed by for the day.
- **S.B. 1314** (one thousand three hundred fourteen) was read by title the third time and, on motion of Senator Hashmi, was passed with its title.

The recorded vote is as follows:

YEAS--28. NAYS--11. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Ruff, Saslaw, Spruill, Surovell, Vogel--28.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Obenshain, Peake, Reeves, Stanley, Stuart, Suetterlein--11.

RULE 36--0.

S.B. 1379 (one thousand three hundred seventy-nine) was read by title the third time and, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Stanley, Surovell, Vogel--24.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stuart, Suetterlein--15.

RULE 36--0.

S.B. 1387 (one thousand three hundred eighty-seven) was read by title the third time and, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows: YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1402 (one thousand four hundred two) was read by title the third time and, on motion of Senator Pillion, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--Stuart--1.

RULE 36--0.

S.B. 1412 (one thousand four hundred twelve) was read by title the third time and, on motion of Senator Stanley, was passed with its title.

The recorded vote is as follows:

YEAS--29. NAYS--10. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Reeves, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--29.

NAYS--Chase, DeSteph, Hanger, McDougle, Newman, Norment, Obenshain, Peake, Ruff, Stuart--10.

RULE 36--0.

S.B. 1422 (one thousand four hundred twenty-two) was read by title the third time and, on motion of Senator Kiggans, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Vogel--34.

NAYS--Deeds, Ebbin, Edwards, Locke, Surovell--5.

RULE 36--0.

RECONSIDERATION

Senator Boysko moved to reconsider the vote by which S.B. 1379 (one thousand three hundred seventy-nine) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1379, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Saslaw, Spruill, Stanley, Surovell, Vogel--27.

NAYS--Chase, DeSteph, Dunnavant, Hanger, McDougle, Newman, Obenshain, Peake, Reeves, Ruff, Stuart, Suetterlein--12.

RULE 36--0.

RECESS

At 1:35 p.m., Senator McDougle moved that the Senate recess until 2:00 p.m.

The motion was agreed to.

The hour of 2:00 p.m. having arrived, the Chair was resumed.

SENATE BILLS ON SECOND READING

S.B. 1246 (one thousand two hundred forty-six), on motion of Senator Deeds, was passed by for the day.

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

S.B. 1104 (one thousand one hundred four).

S.B. 1176 (one thousand one hundred seventy-six).

S.B. 1183 (one thousand one hundred eighty-three).

S.B. 1184 (one thousand one hundred eighty-four).

S.B. 1225 (one thousand two hundred twenty-five).

S.B. 1237 (one thousand two hundred thirty-seven).

S.B. 1272 (one thousand two hundred seventy-two).

- S.B. 1288 (one thousand two hundred eighty-eight).
- **S.B.** 1304 (one thousand three hundred four).
- **S.B.** 1327 (one thousand three hundred twenty-seven).
- S.B. 1349 (one thousand three hundred forty-nine).
- S.B. 1356 (one thousand three hundred fifty-six).
- S.B. 1410 (one thousand four hundred ten).
- S.B. 1431 (one thousand four hundred thirty-one).
- **S.B. 1437** (one thousand four hundred thirty-seven).
- S.B. 1456 (one thousand four hundred fifty-six).
- S.B. 1458 (one thousand four hundred fifty-eight).
- S.B. 1464 (one thousand four hundred sixty-four).

The motion was agreed to.

S.B. 1104 (one thousand one hundred four) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 53.1-136 and 53.1-155 of the Code of Virginia, relating to parole; notice and certification; monthly reports.

The reading of the substitute was waived.

On motion of Senator Obenshain, the substitute was agreed to.

Senator Morrissey offered the following amendment to the substitute:

1. Line 23, substitute, after *electronic* strike

and certified

Senator Morrissey withdrew the amendment.

S.B. 1176 (one thousand one hundred seventy-six) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 37.2-314, 37.2-408.1, 37.2-416, and 37.2-506 of the Code of Virginia, relating to the Department of Behavioral Health and Developmental Services; barrier crimes.

The reading of the substitute was waived.

On motion of Senator Ruff, the substitute was agreed to.

S.B. 1184 (one thousand one hundred eighty-four) was taken up.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 16.1-349, 16.1-350, 16.1-351, 16.1-352, and 16.1-353 of the Code of Virginia, relating to standby guardianship; triggering event.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

S.B. 1304 (one thousand three hundred four) was taken up.

The following amendments proposed by the Committee on Education and Health. were offered:

1. Line 49, introduced, after to

strike

the remainder of line 49 and through center on line 50

insert

prior to the individual's discharge

2. Line 56, introduced, after without

strike

completion

insert

initiation

3. After line 82, introduced

insert

2. That the Commissioner of Behavioral Health and Developmental Services shall establish a work group with representatives of the Virginia Association of Community Services Boards to (i) review the current process for discharging patients from state mental health hospitals, including the current assigned responsibilities of state hospital staff and community services board staff, as well as the barriers to timely discharge for patients clinically ready to discharge, and (ii) develop potential options to expedite the discharge process for individuals who can be safely discharged back into the community. The work group shall develop a plan that includes recommendations for expediting the discharge process and shall identify the necessary funding to ensure that individuals receive essential services upon discharge and that discharges are timely. The work group shall report its findings and conclusions and its plan to the Chairmen of the House Committee on Health, Welfare and Institutions and the Senate Committee on Education and Health by September 1, 2021.

The reading of the amendments was waived.

On motion of Senator McPike, amendments Nos. 1 and 3 were agreed to.

Senator McPike moved that amendment No. 2 be rejected.

The question was put on agreeing to amendment No. 2.

Amendment No. 2 was rejected.

S.B. 1327 (one thousand three hundred twenty-seven) was taken up.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-2223.5, relating to housing protections; foreclosures; manufactured housing.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

S.B. 1356 (one thousand three hundred fifty-six) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 32.1-127, 32.1-162.5, and 63.2-1732 of the Code of Virginia, relating to hospitals, nursing homes, certified nursing facilities, hospices, and assisted living facilities; visits by clergy; public health emergency.

The reading of the substitute was waived.

On motion of Senator Kiggans, the substitute was agreed to.

S.B. 1410 (one thousand four hundred ten) was taken up.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1226, and 55.1-1310 of the Code of Virginia, relating to public accommodations, employment, and housing; prohibited discrimination on the basis of status as active military or a military spouse.

The reading of the substitute was waived.

On motion of Senator Bell, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- **S.B.** 1104 (one thousand one hundred four) as amended.
- **S.B.** 1176 (one thousand one hundred seventy-six) as amended.
- **S.B.** 1183 (one thousand one hundred eighty-three).
- S.B. 1184 (one thousand one hundred eighty-four) as amended.
- **S.B. 1225** (one thousand two hundred twenty-five).
- **S.B. 1237** (one thousand two hundred thirty-seven).
- **S.B. 1272** (one thousand two hundred seventy-two).
- S.B. 1304 (one thousand three hundred four) as amended.
- S.B. 1327 (one thousand three hundred twenty-seven) as amended.
- S.B. 1349 (one thousand three hundred forty-nine).

- **S.B.** 1356 (one thousand three hundred fifty-six) as amended.
- S.B. 1410 (one thousand four hundred ten) as amended.
- S.B. 1431 (one thousand four hundred thirty-one).
- **S.B. 1437** (one thousand four hundred thirty-seven).
- S.B. 1456 (one thousand four hundred fifty-six).
- S.B. 1458 (one thousand four hundred fifty-eight).
- S.B. 1464 (one thousand four hundred sixty-four).
- S.B. 1288 (one thousand two hundred eighty-eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 22.1-214, 22.1-215, 22.1-253.13:4, and 22.1-298.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-214.4, relating to Department of Education; special education.

The reading of the substitute was waived.

On motion of Senator Dunnavant, the substitute was agreed to.

- S.B. 1288, on motion of Senator Dunnavant, was passed by for the day.
- **S.B.** 1165 (one thousand one hundred sixty-five), on motion of Senator Surovell, was passed by for the day.
- **S.B. 1170** (one thousand one hundred seventy) was read by title the second time and, on motion of Senator Norment, was ordered to be engrossed and read by title the third time.
 - S.B. 1106 (one thousand one hundred six) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Article 3 of Chapter 9 of Title 22.1 a section numbered 22.1-141.3, relating to the establishment of the Public School Assistance Fund and Program.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

Senator McClellan offered the following amendments to the substitute:

1. Line 28, substitute, after roofs.

insert

Grants from the Fund may also be provided for the purpose of funding the financing, including principal and interest payments, of any of the aforementioned repairs or replacements, so long as such financing costs were incurred on or after January 1, 2011.

2. Line 31, substitute, after renewals.

insert

The guidelines shall establish a minimum amount for a grand award, but the Department of Education may provide for variable minimums depending on the type of project to be funded by the grant.

On motion of Senator McClellan, the reading of the amendments was waived.

On motion of Senator McClellan, the amendments were agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

S.B. 1390 (one thousand three hundred ninety) was read by title the second time.

The following amendments proposed by the Committee on Agriculture, Conservation and Natural Resources were offered:

1. Line 441, introduced, after *Any* insert

set

2. Line 441, introduced, after *trap* strike

set

3. Line 443, introduced, after released

strike

from the trap

insert

at the location where trapped

The reading of the amendments was waived.

On motion of Senator Lewis, the amendments were agreed to.

Senator Stuart offered the following amendment:

1. Line 437, introduced, after *program* insert

, provided that the governing body of the locality in which the public or private animal shelter, releasing agency, or hospital or clinic is located, by ordinance, authorizes the operation of a TNR program

On motion of Senator Stuart, the reading of the amendment was waived.

Senator Stuart moved that the amendment be agreed to.

The question was put on agreeing to the amendment.

The amendment was rejected.

On motion of Senator Lewis, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1138** (one thousand one hundred thirty-eight), on motion of Senator McClellan, was passed by for the day.
- **S.B. 1178** (one thousand one hundred seventy-eight) was read by title the second time and, on motion of Senator Ebbin, was ordered to be engrossed and read by title the third time.
 - S.B. 1180 (one thousand one hundred eighty) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 8.01-267.1 of the Code of Virginia, relating to civil actions filed on behalf of multiple persons.

The reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

Senator Surovell offered the following amendments to the substitute:

1. Line 35, substitute, after transferring

insert

and settlement

2. Line 37, substitute, after Procedure.

insert

Such rules shall include provisions comparable to those provided in 28 U.S.C. §§ 1712 and 1713 and shall require notice in the event of settlement to the Attorney General in a manner comparable to that provided by 28 U.S.C. § 1715.

On motion of Senator Surovell, the reading of the amendments was waived.

On motion of Senator Surovell, the amendments were agreed to.

On motion of Senator Surovell, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1198** (one thousand one hundred ninety-eight), on motion of Senator Petersen, was passed by for the day.
 - S.B. 1215 (one thousand two hundred fifteen) was read by title the second time.

The following amendment proposed by the Committee on General Laws and Technology was offered:

1. Line 24, introduced, after C.

strike

the remainder of line 24 and through days. on line 25

insert

The initial hearing on the tenant's position shall be held within five calendar days from the date of the filing of the petition.

The reading of the amendment was waived.

On motion of Senator Ebbin, the amendment was agreed to.

On motion of Senator Ebbin, the bill was ordered to be engrossed and read by title the third time.

S.B. 1235 (one thousand two hundred thirty-five) was read by title the second time.

The following amendments proposed by the Committee on Education and Health were offered:

1. Line 16, introduced, after *regarding* strike

health-related

2. Line 16, introduced, after *matters* insert

set forth in subsection B of § 22.1-207.1

The reading of the amendments was waived.

On motion of Senator Peake, the amendments were agreed to.

On motion of Senator Peake, the bill was ordered to be engrossed and read by title the third time.

S.B. 1266 (one thousand two hundred sixty-six) was read by title the second time.

The following amendment proposed by the Committee on the Judiciary was offered:

1. Line 20, introduced, after himself insert

, family or household members as defined in § 16.1-228,

The reading of the amendment was waived.

On motion of Senator Deeds, the amendment was agreed to.

On motion of Senator Deeds, the bill was ordered to be engrossed and read by title the third time.

S.B. 1287 (one thousand two hundred eighty-seven) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-340.19 and 18.2-340.28 of the Code of Virginia, relating to the Charitable Gaming Board; regulations; electronic pull tabs.

The reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

On motion of Senator McPike, the bill was ordered to be engrossed and read by title the third time.

S.B. 1303 (one thousand three hundred three) was read by title the second time.

Senator Dunnavant offered the following amendments:

1. Line 11, introduced, after *make* insert

virtual and

2. Line 13, introduced strike

all of line 13

On motion of Senator Dunnavant, the reading of the amendments was waived.

On motion of Senator Dunnavant, the amendments were agreed to.

On motion of Senator Dunnavant, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1317** (one thousand three hundred seventeen) was read by title the second time and, on motion of Senator Dunnavant, was ordered to be engrossed and read by title the third time.
 - **S.B.** 1325 (one thousand three hundred twenty-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 20-124.2 of the Code of Virginia, relating to visitation; petition of grandparent.

The reading of the substitute was waived.

On motion of Senator Dunnavant, the substitute was agreed to.

On motion of Senator Dunnavant, the bill was ordered to be engrossed and read by title the third time.

S.B. 1381 (one thousand three hundred eighty-one) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding a section numbered 18.2-283.2, relating to possessing or transporting a weapon within Capitol Square or into building owned or leased by the Commonwealth; penalty.

The reading of the substitute was waived.

On motion of Senator Ebbin, the substitute was agreed to.

On motion of Senator Ebbin, the bill was ordered to be engrossed and read by title the third time.

- **S.B.** 1384 (one thousand three hundred eighty-four), on motion of Senator Surovell, was passed by for the day.
- **S.B. 1439** (one thousand four hundred thirty-nine), on motion of Senator McClellan, was passed by for the day.
 - S.B. 1468 (one thousand four hundred sixty-eight) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Title 9.1 a chapter numbered 15, consisting of sections numbered 9.1-1500, 9.1-1501, and 9.1-1502, relating to certifications for victims of qualifying criminal activity.

The reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

On motion of Senator Surovell, the bill was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

S.B. 1474 (one thousand four hundred seventy-four), on motion of Senator Newman, was rereferred to the Committee on Finance and Appropriations.

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1260 (one thousand two hundred sixty).
- S.B. 1273 (one thousand two hundred seventy-three).
- S.B. 1472 (one thousand four hundred seventy-two).
- **S.B. 1473** (one thousand four hundred seventy-three).
- **S.B.** 1144 (one thousand one hundred forty-four).
- S.B. 1160 (one thousand one hundred sixty).
- **S.B. 1259** (one thousand two hundred fifty-nine).
- **S.B.** 1397 (one thousand three hundred ninety-seven).
- S.B. 1470 (one thousand four hundred seventy).

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- **S.B. 1260** (one thousand two hundred sixty).
- S.B. 1273 (one thousand two hundred seventy-three).
- **S.B. 1472** (one thousand four hundred seventy-two).
- S.B. 1473 (one thousand four hundred seventy-three).
- S.B. 1144 (one thousand one hundred forty-four).
- S.B. 1160 (one thousand one hundred sixty).
- S.B. 1259 (one thousand two hundred fifty-nine).
- **S.B.** 1397 (one thousand three hundred ninety-seven).
- **S.B. 1470** (one thousand four hundred seventy).

SENATE JOINT RESOLUTION ON THIRD READING

S.J.R. 327 (three hundred twenty-seven) was read by title the third time and, on motion of Senator Deeds, was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

SENATE JOINT RESOLUTIONS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate joint resolutions be waived:

S.J.R. 288 (two hundred eighty-eight).

S.J.R. 293 (two hundred ninety-three).

S.J.R. 323 (three hundred twenty-three).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate joint resolutions were passed by for the day:

S.J.R. 288 (two hundred eighty-eight).

S.J.R. 293 (two hundred ninety-three).

S.J.R. 323 (three hundred twenty-three).

SENATE RESOLUTION ON FIRST READING

S.R. 95 (ninety-five) was read by title the first time.

HOUSE BILL ON THIRD READING

H.B. 2333 (two thousand three hundred thirty-three) was taken up.

Senator Dunnavant offered the following amendment No. 2 to the substitute:

2. Line 120, substitute, after (i)

insert

for

On motion of Senator Dunnavant, the reading of amendment No. 2 was waived.

On motion of Senator Dunnavant, amendments Nos. 1 and 2 were agreed to.

The substitute with amendments was ordered to be engrossed.

H.B. 2333, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Lucas, the Senate adjourned until tomorrow at $12\ m.$

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

TUESDAY, FEBRUARY 2, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Jen A. Kiggans, Seventh Senatorial District, offered the following prayer:

I'll start with a verse from First Corinthians 13, "if I speak in the tongues of men and of angels, but have not love, I am a noisy gong or a clanging cymbal. And if I have prophetic powers and understand all mysteries and all knowledge, and if I have all faith so as to move mountains but have not love, I am nothing. If I give away all I have, and if I deliver my body to be burned but have not love, I gain nothing. Love is patient and kind; love is not jealous or boastful; it is not arrogant or rude. Love does not insist on its own way; it is not irritable or resentful; it does not rejoice at wrong, but rejoices in the right. Love bears all things, believes all things, hopes all things, endures all things."

As British troops were descending upon Boston on September 6th, 1774, the Continental Congress's first official act was to request that Reverend Jacob Douché, Pastor of Philadelphia's Christ Church, open Congress in prayer for the fledgling nation seeking its independence from England, and I quote a small part of that prayer.

Be thou present, O God of wisdom, and direct the councils of this honorable assembly. Enable them to settle things on the best and surest foundation. That the scene of blood may be speedily closed. That order, harmony, and peace may be effectively restored and truth, justice, religion, and piety prevail and flourish amongst the people. Preserve the health of their bodies and vigor of their minds. Shower down on them and the millions here they represent such temporal blessings as thou see as expedient for them in this world and crown them with everlasting glory in the world to come. All this may we ask in the name and through the merits of Jesus Christ, Thy Son and our Savior. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Nathan A. Hatfield, Assistant Clerk-LINCS, Senate Clerk's Office.

On motion of Senator McDougle, the Rules were suspended to allow Senators Norment and Vogel, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--37.

NAYS--0. RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Ruff, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, DeSteph, McDougle, Petersen, Stanley--5. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 1, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1817.** A BILL to amend and reenact §§ 54.1-2957 and 54.1-2957.01 of the Code of Virginia, relating to practice of certified nurse midwives.
- **H.B. 1820.** A BILL to amend and reenact §§ 63.2-608 and 63.2-801 of the Code of Virginia, relating to SNAP benefits program.
- **H.B. 1867.** A BILL to amend and reenact § 19.2-368.10 of the Code of Virginia, relating to compensating victims of crime; reporting requirement; sexual abuse.
- **H.B. 1884.** A BILL to amend and reenact §§ 38.2-6505, 58.1-3, and 58.1-341.1 of the Code of Virginia, relating to facilitated enrollment program.
- **H.B. 1885.** A BILL to require the Department of Education to perform a comprehensive review of computer science standards, courses, and course pathways in public schools; report.
- H.B. 1890. A BILL to amend and reenact §§ 24.2-105, as it shall become effective, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 24.2-104.1, by adding a section numbered 24.2-1005.2, and by adding in Title 24.2 a chapter numbered 1.1, consisting of sections numbered 24.2-125 through 24.2-131; and to repeal § 24.2-124, as it shall become effective, of the Code of Virginia, relating to elections; prohibited discrimination in voting and elections administration; required process for enacting certain covered practices; civil causes of action; penalties.
- **H.B. 1899.** A BILL to amend and reenact §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1 of the Code of Virginia, relating to sunset of coal tax credits.
- **H.B. 1928.** A BILL to amend and reenact §§ 10.1-2202 and 10.1-2204 of the Code of Virginia, relating to historic resources; acquisition and lease of land.
- **H.B. 1936.** A BILL to amend and reenact § 18.2-58 of the Code of Virginia, relating to robbery; penalties.

H.B. 1952. A BILL to amend and reenact § 24.2-948.4 of the Code of Virginia and to amend the Code of Virginia by adding in Article 3 of Chapter 9.3 of Title 24.2 a section numbered 24.2-948.5, relating to campaign finance; prohibited personal use; child care exception.

-317-

- **H.B. 1953.** A BILL to amend and reenact § 54.1-2900 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 54.1-2957.04, relating to licensed certified midwives; licensure; practice.
- H.B. 1965. A BILL to amend and reenact § 10.1-1307 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 10.1-1307.04, relating to the State Air Pollution Control Board; low-emissions and zero-emissions vehicle standards.
- **H.B. 1969.** A BILL to amend and reenact §§ 58.1-3221.6 and 58.1-3970.1 of the Code of Virginia, relating to administration of blighted and derelict properties in certain localities.
- **H.B. 1978.** A BILL to amend and reenact § 2.2-424 of the Code of Virginia and to amend the Code of Virginia by adding in Title 30 a chapter numbered 1.2, consisting of a section numbered 30-19.21, relating to the Legislative Staff Development Fund.
- H.B. 1981. A BILL to amend and reenact § 55.1-1229 of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; access to dwelling unit during certain declared states of emergency.
- **H.B. 1990.** A BILL to amend the Code of Virginia by adding a section numbered 30-19.1:13, relating to racial and ethnic impact statements for criminal justice legislation.
- H.B. 1992. A BILL to amend and reenact §§ 18.2-308.09, 18.2-308.2:1, as it is currently effective and as it shall become effective, 18.2-308.2:2, as it is currently effective and as it shall become effective, 18.2-308.2:3, as it is currently effective and as it shall become effective, and 19.2-386.28 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-308.1:8, relating to purchase, possession, or transportation of firearms following conviction for assault and battery of a family or household member; penalties.
- **H.B. 2006.** A BILL to amend and reenact §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660, relating to tax exemptions for energy storage systems.
- **H.B. 2017.** A BILL to amend and reenact § 16.1-260 of the Code of Virginia and to amend the Code of Virginia by adding in Article 12.1 of Chapter 11 of Title 16.1 a section numbered 16.1-309.11, relating to juvenile offenders; youth justice diversion programs.
- **H.B. 2020.** A BILL to amend and reenact § 24.2-509 of the Code of Virginia, relating to nomination of candidates for elected offices; restrictions on nomination method selected by political party.
- **H.B. 2030.** A BILL to amend the Code of Virginia by adding a section numbered 3.2-3929.1, relating to neonicotinoid pesticides; civil penalty.
- **H.B. 2038.** A BILL to amend and reenact §§ 19.2-303, 19.2-303.1, and 19.2-306 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 19.2-306.1, relating to probation, revocation, and suspension of sentence; limitations.

- **H.B. 2039.** A BILL to amend and reenact §§ 54.1-2902, 54.1-2950.1, 54.1-2951.1, 54.1-2951.2, 54.1-2952, 54.1-2952.1, 54.1-2953, and 54.1-2972 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 54.1-2951.4, relating to practice as a physician assistant.
- **H.B. 2056.** A BILL to amend and reenact §§ 16.1-278.5, 16.1-291, and 16.1-292 of the Code of Virginia, relating to status offenders; willful and material violation of court order or terms of probation; notice; orders of disposition for violation.
- H.B. 2073. A BILL to amend and reenact § 8.01-244 of the Code of Virginia, relating to wrongful death statute of limitations; criminal investigations by law enforcement; emergency. EMERGENCY
- **H.B. 2079.** A BILL to amend and reenact §§ 54.1-3300 and 54.1-3303.1 of the Code of Virginia, relating to pharmacists; initiation of treatment; certain drugs and devices.
- H.B. 2098. A BILL to authorize the Commonwealth to lease a portion of property previously used by the Department of Behavioral Health and Developmental Services as the Southwestern Virginia Mental Health Institute and to amend and reenact §§ 1 and 2 of Chapter 678 of the Acts of Assembly of 2019.
- H.B. 2123. A BILL to amend and reenact § 23.1-506 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 23.1-505.1, relating to the State Council of Higher Education for Virginia; eligibility for in-state tuition.
- **H.B. 2124.** A BILL to require the Department of Medical Assistance Services to deem testing for, treatment of, and vaccination against COVID-19 to be emergency services.
- **H.B. 2125.** A BILL to amend and reenact § 24.2-404 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 4 of Title 24.2 a section numbered 24.2-403.1, relating to voter registration; preregistration of persons 16 years of age or older.
- **H.B. 2139.** A BILL to amend and reenact § 8.01-249 of the Code of Virginia, relating to accrual of cause of action; diagnosis of latent injury or disease.
- **H.B. 2147.** A BILL to amend and reenact §§ 2.2-520, as it is currently effective and as it shall become effective, 2.2-522, 2.2-523, 2.2-3902, 2.2-3907, 2.2-3909, and 15.2-1604 of the Code of Virginia, relating to the Division of Human Rights; renamed as Office of Civil Rights.
- **H.B. 2148.** A BILL to amend and reenact 10.1-1197.5 of the Code of Virginia, relating to small renewable energy projects; energy storage.
- **H.B. 2165.** A BILL to amend and reenact § 58.1-3965 of the Code of Virginia, relating to sale of land for delinquent taxes.
- **H.B. 2190.** A BILL to amend and reenact § 8.01-53 of the Code of Virginia, relating to wrongful death beneficiaries.
- **H.B. 2198.** A BILL to amend and reenact §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223 of the Code of Virginia, relating to local elections for governing bodies; elections for school boards; qualification of voters.

- **H.B. 2212.** A BILL to amend and reenact § 2.2-2649 of the Code of Virginia, relating to Children's Services Act; effective monitoring and implementation.
- **H.B. 2220.** A BILL to amend and reenact § 54.1-2956.12 of the Code of Virginia, relating to surgical technologist; certification; use of title.
- **H.B. 2234.** A BILL to amend the Code of Virginia by adding in Article 3 of Chapter 8 of Title 18.2 a section numbered 18.2-361.1, relating to victims of sex trafficking; affirmative defense to prosecution for certain offenses.
- **H.B. 2236.** A BILL to amend and reenact §§ 9.1-175 and 18.2-254.3 of the Code of Virginia, relating to behavioral health docket; transfer of supervision.
- **H.B. 2238.** A BILL to amend and reenact § 22.1-323 of the Code of Virginia, relating to licensed private schools for students with disabilities; accreditation.
- **H.B. 2269.** A BILL to amend and reenact § 58.1-2636 of the Code of Virginia, relating to revenue share for solar energy projects and energy storage systems.
- **H.B. 2275.** A BILL to amend and reenact §§ 3.2-5100, 3.2-5101, 3.2-5130, and 15.2-2288.6 of the Code of Virginia, relating to food and drink permit requirements.
- **H.B. 2290.** A BILL to repeal § 18.2-104 of the Code of Virginia, relating to punishment for conviction of second or subsequent misdemeanor larceny.
- **H.B. 2293.** A BILL to amend and reenact § 58.1-3713 of the Code of Virginia, relating to local gas severance tax; sunset date.
- **H.B. 2295.** A BILL to amend the Code of Virginia by adding a section numbered 18.2-283.2, relating to carrying a firearm within Capitol Square and the surrounding area, into building owned or leased by the Commonwealth, etc.; penalty.
- H.B. 2324. A BILL to amend and reenact § 30-392 of the Code of Virginia, relating to Virginia Redistricting Commission; removal of commissioners; emergency.
 EMERGENCY
- THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTIONS:
- H.J.R. 615. Commending the Williamsburg Veterans of Foreign Wars Post 4639.
- H.J.R. 617. Commending the River Basin Grand Winners of the Clean Water Farm Award.
- H.J.R. 619. Commending Little Oil Company, Inc.
- H.J.R. 620. Commending Wolf Trap Foundation for the Performing Arts.
- H.J.R. 621. Commending Colonial Behavioral Health.
- H.J.R. 622. Commending the Northern Virginia Association of REALTORS®.
- H.J.R. 623. Celebrating the life of Robert E. Mannion.

- H.J.R. 624. Celebrating the life of Erik Gutshall.
- H.J.R. 625. Celebrating the life of George H. Banks.
- H.J.R. 627. Celebrating the life of Alfred Ernest Moreau.
- H.J.R. 630. Commending Clenise and Alex White.
- H.J.R. 631. Celebrating the life of the Honorable Bernard S. Cohen.
- H.J.R. 632. Commending Redella S. Pepper.
- H.J.R. 633. Celebrating the life of Kelly Brice Walters.
- H.J.R. 634. Commending Western Fairfax Christian Ministries.
- H.J.R. 635. Commending Karen Campblin.
- H.J.R. 636. Commending Aquatech.
- H.J.R. 637. Commending Chris and Jeanne Mussig.
- H.J.R. 638. Commending Homework Helpers Organization.
- THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING SENATE JOINT RESOLUTIONS:
- S.J.R. 305. Celebrating the life of Billy Joe Roberts.
- S.J.R. 319. Celebrating the life of Michael Stephen Horwatt.
- S.J.R. 320. Celebrating the life of Dr. John E. Thomasson.
- S.J.R. 321. Commending Wolf Trap Foundation for the Performing Arts.
- S.J.R. 324. Celebrating the life of Edward S. Garcia.
- S.J.R. 325. Commending Harold J. Roesch II.
- S.J.R. 326. Celebrating the life of David Anthony Rice.
- S.J.R. 328. Celebrating the life of Roger Hamilton Brown.
- S.J.R. 329. Celebrating the life of Joseph Richard Stowers.
- S.J.R. 330. Celebrating the life of Robert Emmett Mannion, Sr.
- IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, McDougle, Norment, Petersen--4.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

H.B. 1817, H.B. 1885, H.B. 1953, H.B. 2039, H.B. 2079, H.B. 2123, H.B. 2124, H.B. 2220, and H.B. 2238 were referred to the Committee on Education and Health.

H.B. 1820, H.B. 2098, and H.B. 2212 were referred to the Committee on Rehabilitation and Social Services.

H.B. 1867, H.B. 1936, H.B. 1992, H.B. 2017, H.B. 2038, H.B. 2056, H.B. 2073, H.B. 2139, H.B. 2190, H.B. 2234, H.B. 2236, and H.B. 2290 were referred to the Committee on the Judiciary.

H.B. 1884, H.B. 1965, and H.B. 2269 were referred to the Committee on Commerce and Labor.

H.B. 1890, H.B. 1952, H.B. 2020, H.B. 2125, H.B. 2198, and H.B. 2324 were referred to the Committee on Privileges and Elections.

H.B. 1899, H.B. 1969, H.B. 2006, H.B. 2165, and H.B. 2293 were referred to the Committee on Finance and Appropriations.

H.B. 1928, H.B. 2030, H.B. 2148, and H.B. 2275 were referred to the Committee on Agriculture, Conservation and Natural Resources.

H.B. 1978, H.B. 1990, and H.B. 2295 were referred to the Committee on Rules.

H.B. 1981 and H.B. 2147 were referred to the Committee on General Laws and Technology.

The House joint resolutions, communicated as agreed to by the House of Delegates, were laid on the Clerk's Desk under Senate Rule 26 (g) as follows:

H.J.R. 615, H.J.R. 617, H.J.R. 619, H.J.R. 620, H.J.R. 621, H.J.R. 622, H.J.R. 623, H.J.R. 624, H.J.R. 625, H.J.R. 627, H.J.R. 630, H.J.R. 631, H.J.R. 632, H.J.R. 633, H.J.R. 634, H.J.R. 635, H.J.R. 636, H.J.R. 637, and H.J.R. 638.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Saslaw from the Committee on Commerce and Labor:

- S.B. 1247 (one thousand two hundred forty-seven) with substitute.
- **S.B. 1284** (one thousand two hundred eighty-four) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1289 (one thousand two hundred eighty-nine) with substitute.
 - **S.B.** 1295 (one thousand two hundred ninety-five) with substitute.
 - **S.B. 1341** (one thousand three hundred forty-one).
 - S.B. 1420 (one thousand four hundred twenty).

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- **S.B.** 1196 (one thousand one hundred ninety-six).
- **S.B.** 1197 (one thousand one hundred ninety-seven).
- S.B. 1201 (one thousand two hundred one) with substitute.
- S.B. 1212 (one thousand two hundred twelve) with amendment.
- **S.B. 1226** (one thousand two hundred twenty-six) with amendment.
- S.B. 1252 (one thousand two hundred fifty-two) with substitute.
- S.B. 1258 (one thousand two hundred fifty-eight) with amendment.
- **S.B. 1261** (one thousand two hundred sixty-one) with amendment.
- **S.B. 1262** (one thousand two hundred sixty-two).
- **S.B. 1275** (one thousand two hundred seventy-five) with amendment.
- **S.B.** 1310 (one thousand three hundred ten) with amendment.
- S.B. 1318 (one thousand three hundred eighteen) with substitute.
- **S.B. 1350** (one thousand three hundred fifty).
- **S.B. 1380** (one thousand three hundred eighty).
- **S.B. 1391** (one thousand three hundred ninety-one).
- S.B. 1398 (one thousand three hundred ninety-eight) with substitute.
- S.B. 1405 (one thousand four hundred five).
- S.B. 1425 (one thousand four hundred twenty-five) with substitute.
- **S.B. 1442** (one thousand four hundred forty-two) with amendment.
- S.B. 1443 (one thousand four hundred forty-three).
- S.B. 1284 was rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.J.R. 347. Commending Angeline Godwin.

Patrons--Stanley and Ruff

S.J.R. 348. Celebrating the life of Michael Wayne Woods.

Patrons--Stanley; Delegates: Fowler and O'Quinn

S.J.R. 349. Celebrating the life of William Henry Edwards, Sr.

Patrons--Stuart; Delegate: Ransone

S.J.R. 350. Commending Robert Pinkard, Jr.

Patrons--Stuart; Delegate: Cole, M.L.

S.J.R. 351. Celebrating the life of Sally G. Lamb.

Patron--Stuart

S.J.R. 352. Celebrating the life of John B. Davis.

Patrons--Hanger; Delegates: Avoli and Runion

S.J.R. 353. Celebrating the life of William Mantz.

Patrons--Obenshain; Delegate: Gilbert

S.J.R. 354. Celebrating the life of Karl David Stoltzfus, Sr.

Patrons--Obenshain and Hanger; Delegates: Gilbert, Runion and Wilt

S.J.R. 355. Celebrating the life of Joseph Brisco Dellinger.

Patrons--Obenshain and Hanger; Delegates: Runion and Wilt

S.J.R. 356. Celebrating the life of Lowell Robert Barb.

Patrons--Obenshain and Hanger; Delegates: Bell, Gilbert, Runion and Wilt

S.J.R. 357. Commending the Patrick Henry School of Science and Arts.

Patron--Morrissey

S.J.R. 358. Commending Ryan Janaske.

Patron--Bell

S.J.R. 359. Commending The Lady Chamberlains.

Patron--Bell

S.J.R. 360. Celebrating the life of Jean Smith Brown.

Patron--Bell

S.J.R. 361. Commending the Warrior Retreat at Bull Run.

Patron--Bell

S.J.R. 362. Commending Akshath Mahajan and Maneesh Vallurupalli.

Patron--Bell

S.J.R. 363. Commending Steven R. Cover.

Patron--DeSteph

S.J.R. 364. Celebrating the life of Stephen Johnson.

Patron--DeSteph

S.J.R. 365. Commending LoudounGo.

Patron--Bell

S.J.R. 366. Commending Kelly Lazzara.

Patron--Kiggans

S.J.R. 367. Commending the Virginia Beach Pandemic Relief Partnership.

Patron--Kiggans

S.J.R. 368. Celebrating the life of John Chatburn Stevens.

Patron--Kiggans

S.J.R. 369. Celebrating the life of C. Opal B. Abernathey. Patron--Kiggans

S.J.R. 370. Commending Senior Services of Southeastern Virginia.

Patron--Kiggans

S.J.R. 371. Commending Michael S. Rolband.

Patron--Petersen

S.J.R. 373. Commending Hero Homes, Inc.

Patron--Bell

S.J.R. 374. Commending Panorama Latino TV Show.

Patron--Bell

S.J.R. 375. Celebrating the life of Elaine Dolores Walker.

Patron--Bell

S.J.R. 376. Commending Loudoun Shops Black.

Patron--Bell

S.J.R. 377. Celebrating the life of Officer Brian David Sicknick.

Patrons--Barker, Deeds and Saslaw; Delegates: Watts, Delaney, Helmer and Scott

S.J.R. 378. Celebrating the life of the Honorable Robert Lathan Calhoun.

Patrons--Ebbin, Barker, Bell, Boysko, Deeds, Hashmi, Howell, Marsden, Petersen, Reeves, Ruff, Saslaw, Spruill and Stanley

S.J.R. 379. Commending Tom Hirst and Magaly Galdo-Hirst.

Patrons--Ebbin and Saslaw

S.J.R. 380. Celebrating the life of the Honorable Jerry M. Wood.

Patron--Deeds

S.J.R. 381. Commending the Loudoun County branch of the NAACP.

Patron--Boysko

S.J.R. 382. Commending Joshua Thiel.

Patron--Boysko

S.J.R. 383. Commending the Reverend Deborah Dodson Parsons.

Patron--Boysko

$\textbf{S.J.R.\ 384.}\ Commending\ the\ Loudoun\ Medical\ Reserve\ Corps.$

Patron--Boysko

S.J.R. 385. Commending Kirk Kincannon.

Patrons--Ebbin, Barker, Boysko, Howell, Marsden and Saslaw; Delegates: Bulova, Delaney, Helmer, Kory, Murphy, Reid and Watts

S.J.R. 386. Celebrating the life of Sara Lu P. Snyder.

Patron--Deeds

S.J.R. 387. Commending Ron Campbell.

Patron--Boysko

S.J.R. 388. Commending William McKenna.

Patron--Boysko

S.J.R. 389. Commending Kojo Nnamdi.

Patron--Ebbin

S.J.R. 390. Commending Jennifer K. Baker.

Patron--Boysko

S.J.R. 391. Celebrating the life of Lester Zidel.

Patron--Boysko

S.J.R. 392. Commending Darnell Dozier.

Patron--DeSteph

S.J.R. 393. Celebrating the life of the Honorable Robert S. Bloxom, Sr.

Patrons--Lewis, Deeds, Hanger, Mason, Reeves, Ruff, Saslaw and Spruill; Delegates: Adams, D.M., Austin, Avoli, Ayala, Bagby and Bell

S.J.R. 394. Celebrating the life of Carla Yvette Savage-Wells.

Patron--Lewis

S.J.R. 395. Commemorating the 150th anniversary of the enactment of the Civil Rights Act of 1871.

Patron--McClellan

S.J.R. 398. Commending the Honorable Clyde H. Perdue, Jr.

Patron--Stanley

S.R. 112. Celebrating the life of the Reverend Dr. Willie Woodson.

Patron--McClellan

S.R. 113. Commending the Robert Russa Moton Museum.

Patron--McClellan

S.R. 114. Commending Gilfield Baptist Church.

Patron--McClellan

S.R. 115. Celebrating the life of James Perry Wark.

Patron--McClellan

S.R. 116. Celebrating the life of Rhea K. Hale.

Patron--McClellan

S.R. 117. Commending Richmond Heritage Federal Credit Union.

Patron--McClellan

S.R. 118. Commending Richmond Raceway.

Patron--McClellan

S.R. 119. Commending the Reverend Doug Barber.

Patrons--Stanley and Ruff

S.R. 120. Commending James Smith.

Patron--McDougle

S.R. 121. Commending the Reverend Dr. Barbara L. Cain. Patron--McDougle

The following, by leave, was presented and referred under Senate Rule 26 (g):

S.J.R. 372. Commending the Nation of Greece.

Patron--Dunnavant Referred to Committee on Rules

CALENDAR

SENATE BILLS ON THIRD READING

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1104 (one thousand one hundred four).
- **S.B.** 1176 (one thousand one hundred seventy-six).
- S.B. 1183 (one thousand one hundred eighty-three).
- **S.B.** 1184 (one thousand one hundred eighty-four).
- S.B. 1225 (one thousand two hundred twenty-five).
- **S.B. 1237** (one thousand two hundred thirty-seven).
- S.B. 1272 (one thousand two hundred seventy-two).
- **S.B. 1304** (one thousand three hundred four).
- **S.B.** 1327 (one thousand three hundred twenty-seven).
- S.B. 1349 (one thousand three hundred forty-nine).
- S.B. 1356 (one thousand three hundred fifty-six).
- S.B. 1410 (one thousand four hundred ten).
- S.B. 1431 (one thousand four hundred thirty-one).
- **S.B.** 1437 (one thousand four hundred thirty-seven).
- S.B. 1456 (one thousand four hundred fifty-six).
- S.B. 1458 (one thousand four hundred fifty-eight).
- S.B. 1464 (one thousand four hundred sixty-four).

The motion was agreed to.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- S.B. 1104 (one thousand one hundred four).
- **S.B.** 1176 (one thousand one hundred seventy-six).
- S.B. 1183 (one thousand one hundred eighty-three).
- **S.B.** 1184 (one thousand one hundred eighty-four).
- S.B. 1272 (one thousand two hundred seventy-two).
- S.B. 1304 (one thousand three hundred four).
- S.B. 1327 (one thousand three hundred twenty-seven).
- **S.B.** 1349 (one thousand three hundred forty-nine).
- **S.B.** 1356 (one thousand three hundred fifty-six).

S.B. 1410 (one thousand four hundred ten).

S.B. 1431 (one thousand four hundred thirty-one).

S.B. 1437 (one thousand four hundred thirty-seven).

S.B. 1456 (one thousand four hundred fifty-six).

S.B. 1458 (one thousand four hundred fifty-eight).

S.B. 1464 (one thousand four hundred sixty-four).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1225 (one thousand two hundred twenty-five), on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--1.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--McClellan--1.

S.B. 1237 (one thousand two hundred thirty-seven), on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Locke, Lucas, Spruill--3.

RULE 36--0.

S.B. 1305 (one thousand three hundred five) was taken up.

RECONSIDERATION

Senator McPike moved to reconsider the vote by which **S.B. 1305** (one thousand three hundred five) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0. RULE 36--0.

Senator McPike offered the following amendments to the substitute:

1. Line 12, substitute, after bodies

insert

in a locality with a population in excess of 25,000

2. Line 23, substitute, after nature

strike

and

insert

,(b)

3. Line 24, substitute, after workforce

strike

or (b)

insert

, or(c)

On motion of Senator McPike, the reading of the amendments was waived.

On motion of Senator McPike, the amendments were agreed to.

On motion of Senator McPike, the bill was ordered to be engrossed and read by title the third time.

Senator McPike moved that the Rules be suspended and the third reading of the title of **S.B. 1305** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1305, on motion of Senator McPike, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

RECONSIDERATION

Senator Mason moved to reconsider the vote by which **S.B. 1237** (one thousand two hundred thirty-seven) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1237, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Hanger, Hashmi, Howell, Kiggans, Lewis, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Favola, Locke, Lucas, McClellan, Spruill--5.

RULE 36--0.

S.B. 1170 (one thousand one hundred seventy) was read by title the third time.

Senator Norment moved that **S.B. 1170** be passed with its title.

The question was put on passing S.B. 1170 with its title.

S.B. 1170 was defeated with its title, having failed to receive the necessary affirmative votes required by Article VII, Section 1, of the Constitution.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Ruff, Saslaw, Spruill, Stanley, Surovell--25.

NAYS--Chase, Cosgrove, Dunnavant, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Stuart, Suetterlein, Vogel--14.

RULE 36--0.

S.B. 1106 (one thousand one hundred six) was read by title the third time and, on motion of Senator Stanley, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1390 (one thousand three hundred ninety) was read by title the third time and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Boysko, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Peake, Petersen, Pillion, Saslaw, Spruill, Stanley, Surovell--26.

NAYS--Bell, Chase, Cosgrove, Hanger, McDougle, Newman, Norment, Obenshain, Reeves, Ruff, Stuart, Suetterlein, Vogel--13.

RULE 36--0.

S.B. 1178 (one thousand one hundred seventy-eight) was read by title the third time and, on motion of Senator Ebbin, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Spruill, Surovell, Vogel--24.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--14.

RULE 36--0.

STATEMENT ON VOTE

-331-

Senator Stuart stated that he was recorded as not voting on the question of the passage of **S.B. 1178**, whereas he intended to vote nay.

S.B. 1180 (one thousand one hundred eighty) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1215 (one thousand two hundred fifteen) was read by title the third time and, on motion of Senator Ebbin, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Peake, Petersen, Ruff, Saslaw, Spruill, Suetterlein, Surovell, Vogel--27.

NAYS--Chase, Cosgrove, DeSteph, Hanger, McDougle, Newman, Norment, Obenshain, Pillion, Reeves, Stanley, Stuart--12.

RULE 36--0.

- **S.B. 1235** (one thousand two hundred thirty-five), on motion of Senator Boysko, was passed by for the day.
- **S.B. 1266** (one thousand two hundred sixty-six) was read by title the third time and, on motion of Senator Deeds, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1287 (one thousand two hundred eighty-seven) was read by title the third time and, on motion of Senator McPike, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1303 (one thousand three hundred three) was read by title the third time.

Senator Dunnavant moved that S.B. 1303 be passed with its title.

RECESS

At 2:30 p.m., Senator Saslaw moved that the Senate recess until 3:10 p.m.

The motion was agreed to.

The hour of 3:10 p.m. having arrived, the Chair was resumed.

SENATE BILLS ON THIRD READING

S.B. 1303 was taken up and, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Bell, Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Howell, Kiggans, Lewis, Marsden, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--26.

NAYS--Barker, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Locke, Lucas, Mason, McClellan, McPike, Spruill--13.

RULE 36--0.

S.B. 1317 (one thousand three hundred seventeen) was read by title the third time.

Senator Dunnavant moved that **S.B. 1317** be passed with its title.

The question was put on passing S.B. 1317 with its title.

S.B. 1317 was defeated with its title.

The recorded vote is as follows:

YEAS--15. NAYS--24. RULE 36--0.

YEAS--Chase, Dunnavant, Kiggans, McDougle, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Ruff, Stanley, Stuart, Suetterlein, Vogel--15.

NAYS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Norment, Reeves, Saslaw, Spruill, Surovell--24.

RULE 36--0.

S.B. 1325 (one thousand three hundred twenty-five) was read by title the third time and, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--35.

NAYS--Chase, Deeds, Peake, Suetterlein--4.

RULE 36--0.

S.B. 1381 (one thousand three hundred eighty-one) was read by title the third time and, on motion of Senator Ebbin, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1468 (one thousand four hundred sixty-eight) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

RECONSIDERATION

Senator Newman moved to reconsider the vote by which S.B. 1170 (one thousand one hundred seventy) was defeated with its title.

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1170, on motion of Senator Newman, was passed by for the day.

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- H.B. 1743 (one thousand seven hundred forty-three).
- H.B. 1768 (one thousand seven hundred sixty-eight).
- H.B. 1825 (one thousand eight hundred twenty-five).
- H.B. 1927 (one thousand nine hundred twenty-seven).
- H.B. 1949 (one thousand nine hundred forty-nine).
- H.B. 1859 (one thousand eight hundred fifty-nine).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

- H.B. 1743 (one thousand seven hundred forty-three).
- H.B. 1768 (one thousand seven hundred sixty-eight).
- H.B. 1825 (one thousand eight hundred twenty-five).
- H.B. 1927 (one thousand nine hundred twenty-seven).
- H.B. 1949 (one thousand nine hundred forty-nine).
- H.B. 1859 (one thousand eight hundred fifty-nine).

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

```
S.B. 1246 (one thousand two hundred forty-six).
```

S.B. 1260 (one thousand two hundred sixty).

S.B. 1273 (one thousand two hundred seventy-three).

S.B. 1472 (one thousand four hundred seventy-two).

S.B. 1473 (one thousand four hundred seventy-three).

The motion was agreed to.

S.B. 1246 (one thousand two hundred forty-six) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Privileges and Elections was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 24.2-709.1 and 24.2-712 of the Code of Virginia, relating to absentee voting; mandatory processing of returned absentee ballots before election day; central absentee voter precinct in the office of the general registrar.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

S.B. 1288 (one thousand two hundred eighty-eight) was taken up, the committee substitute having been agreed to on February 1, 2021.

S.B. 1260 (one thousand two hundred sixty) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Transportation was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 25.1-203 and 33.2-1011 of the Code of Virginia, relating to entry onto land for inspection.

The reading of the substitute was waived.

On motion of Senator Bell, the substitute was agreed to.

S.B. 1273 (one thousand two hundred seventy-three) was taken up.

The following amendments proposed by the Committee on Rules were offered:

```
1. Line 39, introduced, after least strike

two
insert

one
```

2. Line 39, introduced, after be

strike

members

insert

a member

3. Line 40, introduced, after Health

insert

, at least one of whom shall be a member of the Senate Committee on Rehabilitation and Social Services,

The reading of the amendments was waived.

On motion of Senator Deeds, the amendments were agreed to.

S.B. 1472 (one thousand four hundred seventy-two) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to require the Department of Medical Assistance Services to establish a work group to study and develop recommendations for the permanent use of virtual supports and increasing access to virtual supports and services for individuals with intellectual and developmental disabilities.

The reading of the substitute was waived.

On motion of Senator Suetterlein, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1246 (one thousand two hundred forty-six) as amended.
- S.B. 1288 (one thousand two hundred eighty-eight) as amended.
- S.B. 1260 (one thousand two hundred sixty) as amended.
- S.B. 1273 (one thousand two hundred seventy-three) as amended.
- S.B. 1472 (one thousand four hundred seventy-two) as amended.
- S.B. 1473 (one thousand four hundred seventy-three).

Senator Saslaw moved that the Rules be suspended and the third reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- **S.B. 1246** (one thousand two hundred forty-six).
- S.B. 1288 (one thousand two hundred eighty-eight).
- S.B. 1260 (one thousand two hundred sixty).
- **S.B. 1273** (one thousand two hundred seventy-three).
- S.B. 1472 (one thousand four hundred seventy-two).
- S.B. 1473 (one thousand four hundred seventy-three).

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0. RULE 36--0.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

S.B. 1246 (one thousand two hundred forty-six).

S.B. 1288 (one thousand two hundred eighty-eight).

S.B. 1260 (one thousand two hundred sixty).

S.B. 1273 (one thousand two hundred seventy-three).

S.B. 1472 (one thousand four hundred seventy-two).

S.B. 1473 (one thousand four hundred seventy-three).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1165 (one thousand one hundred sixty-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, as it is currently effective and as it shall become effective, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307 of the Code of Virginia and to repeal §§ 8.01-654.1, 8.01-654.2, 17.1-313, and 18.2-17, Article 4.1 (§§ 19.2-163.7 and 19.2-163.8) of Chapter 10 of Title 19.2, Article 4.1 (§§ 19.2-264.2 through 19.2-264.5) of Chapter 15 of Title 19.2, § 19.2-163.8, and Chapter 13 (§§ 19.2-264.2 through 19.2-264.5) of Chapter 15 of Title 19.2, § 19.2-163.8, and Chapter 13 (§§ 19.2-264.2 through 19.2-264.5) of Chapter 15 of Title 19.2, § 19.2-163.80, and Chapter 130 (§§ 19.2-264.21) through 19.2-264.51 of the Code of Virginia, relating to abolition of the death penalty.

The reading of the substitute was waived.

Senator Surovell moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

Senator Stanley offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, as it is currently effective and as it shall become effective, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307 of the Code of Virginia and to repeal §§ 8.01-654.1, 8.01-654.2, 17.1-313, and 18.2-17, Article 4.1 (§§ 19.2-163.7 and 19.2-163.8) of Chapter 10 of Title 19.2, Article 4.1 (§§ 19.2-264.2 through 19.2-264.5) of Chapter 15 of Title 19.2, § 53.1-230, and Chapter 13 (§§ 53.1-232 through 53.1-236) of Title 53.1 of the Code of Virginia, relating to abolition of the death penalty.

On motion of Senator Stanley, the reading of the substitute was waived.

Senator Stanley moved that the substitute be agreed to.

The question was put on agreeing to the substitute.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--17. NAYS--22. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Spruill, Surovell--22.

RULE 36--0.

The substitute was rejected.

Senator Norment offered an amendment in the nature of a substitute No. 1, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 19.2-71, 19.2-76.1, 19.2-152.2, 19.2-163, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-299, 19.2-311, 19.2-400, and 37.2-900 of the Code of Virginia, relating to abolition of the death penalty.

Senator Norment withdrew substitute No. 1.

Senator Surovell offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, as it is currently effective and as it shall become effective, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307 of the Code of Virginia and to repeal §§ 8.01-654.1, 8.01-654.2, 17.1-313, and 18.2-17, Article 4.1 (§§ 19.2-163.7 and 19.2-163.8) of Chapter 10 of Title 19.2, Article 4.1 (§§ 19.2-264.2 through 19.2-264.5) of Chapter 15 of Title 19.2, § 53.1-230, and Chapter 13 (§§ 53.1-232 through 53.1-236) of Title 53.1 of the Code of Virginia, relating to abolition of the death penalty.

On motion of Senator Surovell, his substitute was passed by temporarily.

Senator Norment offered an amendment in the nature of a substitute No. 2, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 19.2-71, 19.2-76.1, 19.2-152.2, 19.2-163, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-299, 19.2-311, 19.2-400, and 37.2-900 of the Code of Virginia, relating to abolition of the death penalty.

On motion of Senator Norment, substitute No. 2 was passed by temporarily.

Senator Norment offered an amendment in the nature of a substitute No. 3, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 19.2-71, 19.2-76.1, 19.2-152.2, 19.2-163, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-299, 19.2-311, 19.2-400, and 37.2-900 of the Code of Virginia, relating to abolition of the death penalty.

On motion of Senator Norment, the reading of substitute No. 3 was waived.

Senator Norment moved that substitute No. 3 be agreed to.

The question was put on agreeing to substitute No. 3.

The yeas and nays were called for by the Chair.

The recorded vote is as follows:

YEAS--16. NAYS--23. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart--16.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Suetterlein, Surovell, Vogel--23.

RULE 36--0.

Substitute No. 3 was rejected.

On motion of Senator Norment, the reading of substitute No. 2 was waived.

Senator Norment moved that substitute No. 2 be agreed to.

The question was put on agreeing to substitute No. 2.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows: YEAS--17. NAYS--22. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--17.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--22. RULE 36--0.

Substitute No. 2 was rejected.

On motion of Senator Surovell, the reading of his substitute was waived.

On motion of Senator Surovell, his substitute was agreed to.

On motion of Senator Surovell, the bill was ordered to be engrossed and read by title the third time.

S.B. 1138 (one thousand one hundred thirty-eight) was read by title the second time.

Senator McClellan offered the following amendment:

1. Line 196, introduced, after C.

strike

the remainder of line 196, all of lines 197 through 210, and through D. on line 211

insert

Upon receiving a report of a positive test for hepatitis C, the State Health Commissioner may share protected health information relating to such positive test with relevant sheriffs' offices, the state police, local police departments, adult or youth correctional facilities, salaried or volunteer firefighters, paramedics or emergency medical technicians, officers of the court, and regional or local jails (i) to the extent necessary to advise exposed individuals of the risk of infection and to enable exposed individuals to seek appropriate testing and treatment, and (ii) as may be needed to prevent and control disease and is deemed necessary to prevent serious harm and serious threats to the health and safety of individuals and the public.

The disclosed protected health information shall be held confidential; no person to whom such information is disclosed shall redisclose or otherwise reveal the protected health information without first obtaining the specific authorization from the individual who was the subject of the test for such redisclosure.

Such protected health information shall only be used to protect the health and safety of individuals and the public in conformance with the regulations

concerning patient privacy promulgated by the federal Department of Health and Human Services, as such regulations may be amended.

D.

On motion of Senator McClellan, the reading of the amendment was waived.

On motion of Senator McClellan, the amendment was agreed to.

On motion of Senator Locke, the bill was ordered to be engrossed and read by title the third time.

- **S.B.** 1198 (one thousand one hundred ninety-eight), on motion of Senator Petersen, was passed by for the day.
 - **S.B.** 1384 (one thousand three hundred eighty-four) was read by title the second time.

Senator Surovell offered the following amendment No. 1:

1. Line 51, introduced, after *and* insert

nonprofessional

Senator Surovell withdrew amendment No. 1.

Senator Surovell offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Chapter 43 of Title 2.2 an article numbered 7, consisting of sections numbered 2.2-4377.1 through 2.2-4377.5, relating to the Virginia Public Procurement Act; local arbitration agreements.

On motion of Senator Surovell, the reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

Senator Surovell offered the following amendments Nos. 2 and 3 to the substitute:

2. At the beginning of line 89, substitute strike

A.

3. Line 93, substitute strike

all of lines 93 through 97

On motion of Senator Surovell, the reading of amendments Nos. 2 and 3 was waived.

On motion of Senator Surovell, amendments Nos. 2 and 3 were agreed to.

RECONSIDERATION

Senator Surveell moved to reconsider the vote by which his amendments Nos. 2 and 3 to the substitute to **S.B. 1384** (one thousand three hundred eighty-four) were agreed to.

The motion was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

- S.B. 1384, on motion of Senator Surovell, was passed by for the day.
- S.B. 1439 (one thousand four hundred thirty-nine) was read by title the second time.

The following amendment proposed by the Committee on Education and Health was offered:

1. Line 147, introduced, after *civic* strike

or political

The reading of the amendment was waived.

Senator McClellan moved that the amendment be rejected.

The question was put on agreeing to the amendment.

The amendment was rejected.

Senator McClellan offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 22.1-254 of the Code of Virginia, relating to the Department of Education; guidelines on excused student absences; civic engagement.

On motion of Senator McClellan, the reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

S.B. 1144 (one thousand one hundred forty-four) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Transportation was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 5.1-5 of the Code of Virginia, relating to aircraft; registration and licensing.

The reading of the substitute was waived.

On motion of Senator Stuart, the substitute was agreed to.

On motion of Senator Stuart, the bill was ordered to be engrossed and read by title the third time.

S.B. 1160 (one thousand one hundred sixty) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Transportation was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 46.2-644.01, 46.2-644.02, 46.2-644.03, 46.2-1200.2, 46.2-1202, 46.2-1202.1, 46.2-1203, 46.2-1209, and 46.2-1212.1 of the Code of Virginia and to amend the Code of Virginia by adding in Article 2 of Chapter 6 of Title 46.2 a section numbered 46.2-644.04 and by adding sections numbered 46.2-1200.3 and 46.2-1202.2, relating to vehicles; liens; abandoned vehicles; removing vehicles involved in accidents.

The reading of the substitute was waived.

On motion of Senator Suetterlein, the substitute was agreed to.

On motion of Senator Suetterlein, the bill was ordered to be engrossed and read by title the third time.

S.B. 1259 (one thousand two hundred fifty-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Transportation was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 56-539 and 56-542 of the Code of Virginia, relating to Virginia Highway Corporation Act; alteration of certificate of authority; powers and duties of the State Corporation Commission.

The reading of the substitute was waived.

On motion of Senator Bell, the substitute was agreed to.

Senator Marsden offered the following amendment to the substitute:

1. At the beginning of line 91, substitute

strike

one year

insert

three years

On motion of Senator Marsden, the reading of the amendment was waived.

Senator Marsden moved that the amendment be agreed to.

The question was put on agreeing to the amendment.

The amendment was rejected.

On motion of Senator Bell, the bill was ordered to be engrossed and read by title the third time.

S.B. 1397 (one thousand three hundred ninety-seven) was read by title the second time.

The following amendments proposed by the Committee on Rehabilitation and Social Services were offered:

1. Line 25, introduced, after mail

insert

or verifiable electronic means

2. Line 53, introduced, after mail

insert

or verifiable electronic means

The reading of the amendments was waived.

On motion of Senator Norment, the amendments were agreed to.

On motion of Senator Norment, the bill was ordered to be engrossed and read by title the third time.

S.B. 1470 (one thousand four hundred seventy) was read by title the second time.

The following amendment proposed by the Committee on Transportation was offered:

1. Line 18, introduced, after communication.

strike

the remainder of line 18 and all of lines 19, 20, and 21

insert

On any application on which the applicant indicates that a person who will regularly occupy his vehicle has such a disability, the Department may require the applicant to certify that he has the consent of the regular occupant of the vehicle to release information pursuant to subsection D.

The reading of the amendment was waived.

On motion of Senator Barker, the amendment was agreed to.

On motion of Senator Barker, the bill was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1234 (one thousand two hundred thirty-four).
- **S.B.** 1267 (one thousand two hundred sixty-seven).
- S.B. 1415 (one thousand four hundred fifteen).
- **S.B.** 1426 (one thousand four hundred twenty-six).
- **S.B.** 1475 (one thousand four hundred seventy-five).
- S.B. 1129 (one thousand one hundred twenty-nine).
- S.B. 1306 (one thousand three hundred six).
- S.B. 1385 (one thousand three hundred eighty-five).
- **S.B.** 1446 (one thousand four hundred forty-six).

The motion was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- **S.B.** 1234 (one thousand two hundred thirty-four).
- S.B. 1267 (one thousand two hundred sixty-seven).
- S.B. 1415 (one thousand four hundred fifteen).
- **S.B. 1426** (one thousand four hundred twenty-six).
- **S.B.** 1475 (one thousand four hundred seventy-five).
- S.B. 1129 (one thousand one hundred twenty-nine).
- **S.B. 1306** (one thousand three hundred six).
- S.B. 1385 (one thousand three hundred eighty-five).
- S.B. 1446 (one thousand four hundred forty-six).

SENATE JOINT RESOLUTIONS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate joint resolutions that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate joint resolutions, the titles of the joint resolutions having been printed in the Calendar for their second reading, be placed before the Senate by number only:

- S.J.R. 288 (two hundred eighty-eight).
- S.J.R. 293 (two hundred ninety-three).
- S.J.R. 323 (three hundred twenty-three).

The motion was agreed to.

S.J.R. 288 (two hundred eighty-eight) was taken up.

Senator Lucas offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

Authorizing, and directing the submission to the Joint Committee of Congress on the Library, that the vacant spot of the Commonwealth in the National Statuary Hall Collection at the United States Capitol be filled with a statue to commemorate Barbara Rose Johns.

On motion of Senator Lucas, the reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

S.J.R. 293 (two hundred ninety-three) was taken up.

The following amendments proposed by the Committee on Rules were offered:

1. At the beginning of line 4, introduced, Title

strike

Directing

insert

Requesting

2. Line 30, introduced, after be

strike

directed

insert

requested

The reading of the amendments was waived.

On motion of Senator Spruill, the amendments were agreed to.

On motion of Senator Saslaw, the following Senate joint resolutions were ordered en bloc to be engrossed and read by title the third time:

S.J.R. 288 (two hundred eighty-eight) as amended.

S.J.R. 293 (two hundred ninety-three) as amended.

S.J.R. 323 (three hundred twenty-three).

SENATE RESOLUTION ON SECOND READING

S.R. 95 (ninety-five) was read by title the second time.

The following amendment proposed by the Committee on Rules was offered:

1. Line 36, introduced, after Washington.

insert

The Capitol Square Preservation Council shall review the Commission's recommendation prior to the recommendation being made to the Senate.

The reading of the amendment was waived.

On motion of Senator Suetterlein, the amendment was agreed to.

On motion of Senator Suetterlein, the resolution was ordered to be engrossed and read by title the third time.

HOUSE BILL ON THIRD READING RECONSIDERATION

Senator Dunnavant moved to reconsider the vote by which **H.B. 2333** (two thousand three hundred thirty-three) was passed with its title on February 1, 2021.

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

RECONSIDERATION

Senator Dunnavant moved to reconsider the vote by which amendment No. 2 to the substitute to **H.B. 2333** (two thousand three hundred thirty-three) was agreed to on February 1, 2021.

The motion was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Deeds--1.

RULE 36--0.

Senator Dunnavant moved that amendment No. 2 be rejected.

The question was put on agreeing to amendment No. 2.

Amendment No. 2 was rejected.

Senator Dunnavant offered the following amendment No. 3 to the substitute:

3. Line 120, substitute, after liable

insert

for

On motion of Senator Dunnavant, the reading of amendment No. 3 was waived.

On motion of Senator Dunnavant, amendment No. 3 was agreed to.

The substitute with amendments was ordered to be engrossed.

H.B. 2333, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0. RULE 36--0.

On motion of Senator Lucas, the Senate adjourned until tomorrow at 1:00 p.m.

wan Carle Solman

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

WEDNESDAY, FEBRUARY 3, 2021

The Senate met at 1:00 p.m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable T. Montgomery Mason, First Senatorial District, offered the following prayer:

First, a reading from Psalms 46, verse one through five, "God is our refuge and strength, a very present help in trouble. Therefore we will not fear though the Earth gives way, though the mountains be moved into the heart of the sea, though its waters roar and foam, though the mountains tremble at its swelling. There is a river whose streams make good the city of God, the holy habitation of the most high. God is in the midst of her, she shall not be moved. God will help her when morning dawns."

God, help us find strength and courage to face each day, strength and courage to tackle the pressing issues that we all face, strength and courage to do the best we can for our families, our friends, and our constituents. Help us understand that we are always better when we work together—even through and with our differences. Help us remember that we have far more in common than we have that divides us. God bless those connected with this body, this Commonwealth, and this nation. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Nathan A. Hatfield, Assistant Clerk-LINCS, Senate Clerk's Office.

On motion of Senator McDougle, the Rules were suspended to allow Senator Norment, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Vogel, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--30. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Pillion, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--30.

NAYS--Deeds, DeSteph, Dunnavant, McDougle, Peake, Petersen, Reeves, Stanley, Suetterlein--9. RULE 36--0.

HOUSE COMMUNICATIONS

The following communications were received:

In the House of Delegates February 2, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1786.** A BILL to amend and reenact § 40.1-28.9 of the Code of Virginia, relating to the Virginia Minimum Wage Act; farm laborers and farm employees.
- **H.B. 1862.** A BILL to amend the Code of Virginia by adding a section numbered 40.1-27.4, relating to employee protections; medicinal use of cannabis oil.
- **H.B. 2002.** A BILL to amend and reenact §§ 20-60.3, 20-108.2, and 63.2-1903 of the Code of Virginia, relating to child support; health care coverage.
- **H.B. 2032.** A BILL to amend and reenact §§ 40.1-2, 40.1-49.3, 40.1-49.8, 65.2-101, and 65.2-305 of the Code of Virginia, relating to the employees providing domestic service; application of laws applicable to employee safety and workers' compensation.
- **H.B. 2037.** A BILL to amend and reenact §§ 60.2-528 and 60.2-618 of the Code of Virginia, relating to unemployment compensation; benefits; suitable work; benefits charges.
- **H.B. 2062.** A BILL to amend and reenact § 59.1-200 of the Code of Virginia and to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571, 59.1-572, and 59.1-573, relating to food delivery platforms; agreements with restaurants required; penalty.
- **H.B. 2063.** A BILL to amend and reenact §§ 40.1-29 and 40.1-29.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 40.1-29.2, relating to the Virginia Overtime Wage Act; penalties.
- **H.B. 2134.** A BILL to amend and reenact §§ 40.1-28.7:7 and 60.2-212, as it is currently effective and as it shall become effective, of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 65.2-301.2, relating to employee classification: disaster; personal protective equipment.
- **H.B. 2162.** A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 5 of Title 32.1 a section numbered 32.1-137.08, relating to medical care facilities; persons with disabilities; designated support persons.

EMERGENCY

- **H.B. 2176.** A BILL to amend and reenact § 22.1-291.4 of the Code of Virginia, relating to school board policies; abusive work environments; definitions.
- H.B. 2206. A BILL to temporarily expand the Child Care Subsidy Program to provide financial assistance for child care to families in need during the public health emergency; emergency. EMERGENCY

- **H.B. 2250.** A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-574, relating to Humane Cosmetics Act; civil penalties.
- **H.B. 2254.** A BILL to amend the Code of Virginia by adding a section numbered 18.2-386.3, relating to unlawful dissemination of unsolicited, obscene image of self to another; penalty.
- **H.B. 2262.** A BILL to amend and reenact §§ 46.2-839 and 46.2-905 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 46.2-903.1, relating to traffic regulation; bicycles.
- **H.B. 2284.** A BILL to direct the Commissioner of the Department of Motor Vehicles to reinstate certain driving privileges suspended prior to July 1, 2019.
- **H.B. 2294.** A BILL to amend and reenact § 46.2-629 of the Code of Virginia, relating to odometer disclosure exemption.
- **H.B. 2305.** A BILL to require the Board of Education to issue guidance on the governance of academic year Governor's Schools.
- **H.B. 2318.** A BILL to amend the Code of Virginia by adding a section numbered 46.2-1533.1, relating to test driving vehicles; residence districts.

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTION:

H.J.R. 542. Requesting the Department of Rail and Public Transportation to study transit equity and modernization in the Commonwealth. Report.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

In the House of Delegates February 1, 2021

THE HOUSE OF DELEGATES HAS PASSED WITH A SUBSTITUTE WITH AMENDMENT THE FOLLOWING SENATE BILL:

S.B. 1445. A BILL to facilitate the administration of the COVID-19 vaccine; emergency. EMERGENCY

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates On motion of Senator Saslaw, the Rules were suspended and the reading of the communications from the House of Delegates was waived.

The recorded vote is as follows: YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--McDougle, Petersen--2. RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1786, H.B. 1862, H.B. 2032, H.B. 2037, H.B. 2062, H.B. 2063, H.B. 2134, and H.B. 2250 were referred to the Committee on Commerce and Labor.
 - H.B. 2002 and H.B. 2254 were referred to the Committee on the Judiciary.
- H.B. 2162, H.B. 2176, H.B. 2206, and H.B. 2305 were referred to the Committee on Education and Health.
 - H.B. 2262, H.B. 2284, H.B. 2294, and H.B. 2318 were referred to the Committee on Transportation.

The House joint resolution, communicated as agreed to by the House of Delegates, the first reading of its title having been waived, was referred as follows:

H.J.R. 542 was referred to the Committee on Rules.

COMMITTEE REPORT

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- S.B. 1105 (one thousand one hundred five) with amendment.
- **S.B. 1113** (one thousand one hundred thirteen).
- S.B. 1127 (one thousand one hundred twenty-seven) with substitute.
- **S.B.** 1146 (one thousand one hundred forty-six) with substitute.
- S.B. 1162 (one thousand one hundred sixty-two) with substitute.
- **S.B.** 1163 (one thousand one hundred sixty-three) with amendments.
- S.B. 1210 (one thousand two hundred ten).
- S.B. 1250 (one thousand two hundred fifty) with substitute.
- S.B. 1257 (one thousand two hundred fifty-seven) with substitute.
- **S.B. 1278** (one thousand two hundred seventy-eight) with amendment.
- S.B. 1282 (one thousand two hundred eighty-two).
- **S.B.** 1284 (one thousand two hundred eighty-four).
- **S.B.** 1301 (one thousand three hundred one) with amendments.
- **S.B. 1307** (one thousand three hundred seven).
- **S.B. 1313** (one thousand three hundred thirteen) with substitute.
- **S.B.** 1315 (one thousand three hundred fifteen).

- **S.B.** 1316 (one thousand three hundred sixteen) with amendments.
- S.B. 1326 (one thousand three hundred twenty-six) with amendment.
- **S.B.** 1333 (one thousand three hundred thirty-three) with substitute.
- **S.B.** 1336 (one thousand three hundred thirty-six).
- **S.B.** 1338 (one thousand three hundred thirty-eight) with amendments.
- **S.B.** 1339 (one thousand three hundred thirty-nine) with substitute.
- **S.B.** 1357 (one thousand three hundred fifty-seven) with amendments.
- **S.B. 1360** (one thousand three hundred sixty) with amendment with the recommendation that it be rereferred to the Committee on the Judiciary.
 - S.B. 1362 (one thousand three hundred sixty-two) with amendments.
 - S.B. 1363 (one thousand three hundred sixty-three) with substitute.
 - **S.B.** 1365 (one thousand three hundred sixty-five) with substitute.
 - **S.B.** 1375 (one thousand three hundred seventy-five) with substitute.
 - S.B. 1382 (one thousand three hundred eighty-two).
 - S.B. 1389 (one thousand three hundred eighty-nine) with amendment.
 - S.B. 1392 (one thousand three hundred ninety-two) with amendments.
 - S.B. 1395 (one thousand three hundred ninety-five) with substitute.
 - **S.B. 1396** (one thousand three hundred ninety-six).
 - **S.B. 1401** (one thousand four hundred one).
 - **S.B. 1406** (one thousand four hundred six) with amendments.
 - S.B. 1418 (one thousand four hundred eighteen) with substitute.
 - S.B. 1428 (one thousand four hundred twenty-eight).
 - S.B. 1429 (one thousand four hundred twenty-nine) with substitute.
 - **S.B. 1436** (one thousand four hundred thirty-six) with amendment.
 - **S.B. 1461** (one thousand four hundred sixty-one).
 - **S.B. 1462** (one thousand four hundred sixty-two) with amendment.
 - **S.B. 1465** (one thousand four hundred sixty-five).
 - S.B. 1469 (one thousand four hundred sixty-nine) with substitute.
 - S.B. 1471 (one thousand four hundred seventy-one).
 - **S.B.** 1360 was rereferred to the Committee on the Judiciary.

INTRODUCTION OF LEGISLATION

Pursuant to the provisions of House Joint Resolution No. 575 and Senate Rule 26 (g), Senator Peake requested and was granted unanimous consent to introduce joint resolutions; subsequently, the following were presented and laid on the Clerk's Desk:

S.J.R. 396. Commending Rebecca Carter.

Patron--Peake

S.J.R. 397. Celebrating the life of Pamela Sprouse Palmore.

Patron--Peake

CALENDAR

HOUSE BILLS ON THIRD READING

Senator Saslaw moved that the following House bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

H.B. 1743 (one thousand seven hundred forty-three).

H.B. 1768 (one thousand seven hundred sixty-eight).

H.B. 1825 (one thousand eight hundred twenty-five).

H.B. 1927 (one thousand nine hundred twenty-seven).

H.B. 1949 (one thousand nine hundred forty-nine).

The motion was agreed to.

Senator Saslaw moved that the passage of the House bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following House bills were passed en bloc with their titles:

H.B. 1743 (one thousand seven hundred forty-three).

H.B. 1768 (one thousand seven hundred sixty-eight).

H.B. 1949 (one thousand nine hundred forty-nine).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- **H.B. 1825** (one thousand eight hundred twenty-five), on motion of Senator Lewis, was passed by temporarily.
- **H.B. 1927** (one thousand nine hundred twenty-seven), on motion of Senator Lewis, was passed by temporarily.
- **H.B. 1859** (one thousand eight hundred fifty-nine) was read by title the third time and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--12. RULE 36--1.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, Morrissey, Petersen, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--26.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Suetterlein--12.

RULE 36--McPike--1.

H.B. 1825 (one thousand eight hundred twenty-five) was taken up and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--3. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--35.

NAYS--Chase, Obenshain, Suetterlein--3.

RULE 36--Norment--1.

H.B. 1927 (one thousand nine hundred twenty-seven) was taken up and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1.

RULE 36--0.

SENATE BILLS ON THIRD READING

S.B. 1170 (one thousand one hundred seventy) was taken up.

Senator Norment moved that S.B. 1170 be passed with its title.

S.B. 1170, on motion of Senator Chase, was passed by temporarily.

S.B. 1235 (one thousand two hundred thirty-five) was taken up.

RECONSIDERATION

Senator Peake moved to reconsider the vote by which S.B. 1235 (one thousand two hundred thirty-five) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Peake offered the following amendment:

1. Line 17, engrossed, after parent.

insert

This provision shall not apply to school nurses, physicians, or Virginia Department of Health employees or agents who are inquiring about medical conditions, outbreaks, pandemics, or any other declared state of emergency relating to a communicable disease of public health threat.

On motion of Senator Peake, the reading of amendment No. 1 was waived.

Senator Peake moved that amendment No. 1 be agreed to.

- **S.B. 1235,** on motion of Senator Peake, was passed by temporarily.
- **S.B.** 1165 (one thousand one hundred sixty-five) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

RECESS

At 3:25 p.m., Senator Saslaw moved that the Senate recess until 4:50 p.m.

The motion was agreed to.

The hour of 4:50 p.m. having arrived, the Chair was resumed.

COMMITTEE REPORT

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- S.B. 1209 (one thousand two hundred nine) with amendment.
- $\pmb{S.B.}\ 1360$ (one thousand three hundred sixty) with amendments.

SENATE BILLS ON THIRD READING

 $S.B.\ 1170$ (one thousand one hundred seventy) was taken up and, on motion of Senator Norment, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--31.

NAYS--Chase, Hanger, McDougle, Newman, Obenshain, Peake, Reeves, Suetterlein--8. RULE 36--0.

S.B. 1235 (one thousand two hundred thirty-five) was taken up.

Senator Peake withdrew amendment No. 1.

Senator Peake offered the following amendment No. 2:

2. Line 17, engrossed, after parent

insert

or guardian or person serving in loco parentis. This provision shall not apply to school nurses, physicians, or Virginia Department of Health employees or agents who are inquiring about medical conditions, outbreaks, pandemics, or any other declared state of emergency relating to a communicable disease or public health threat

On motion of Senator Peake, the reading of amendment No. 2 was waived.

On motion of Senator Peake, amendment No. 2 was agreed to.

On motion of Senator Peake, the bill was ordered to be engrossed and read by title the third time.

Senator Peake moved that the Rules be suspended and the third reading of the title of **S.B. 1235** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1235, on motion of Senator Peake, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--19. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Lewis, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--20.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Locke, Lucas, Marsden, Mason, McClellan, McPike, Petersen, Saslaw, Spruill, Surovell--19.

RULE 36--0.

S.B. 1138 (one thousand one hundred thirty-eight) was read by title the third time.

Senator Locke moved that S.B. 1138 be passed with its title.

S.B. 1138, on motion of Senator Morrissey, was passed by for the day.

S.B. 1439 (one thousand four hundred thirty-nine) was read by title the third time and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Reeves, Saslaw, Spruill, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Ruff, Stanley, Stuart, Suetterlein--14.

RULE 36--0.

S.B. 1144 (one thousand one hundred forty-four) was read by title the third time and, on motion of Senator Stuart, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1160 (one thousand one hundred sixty) was read by title the third time and, on motion of Senator Suetterlein, was passed with its title.

The recorded vote is as follows: YEAS--36. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Suetterlein, Surovell, Vogel--36.

NAYS--Reeves, Stanley, Stuart--3. RULE 36--0.

S.B. 1259 (one thousand two hundred fifty-nine) was read by title the third time and, on motion of Senator Bell, was passed with its title.

The recorded vote is as follows: YEAS--32. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--32.

NAYS--Chase, Cosgrove, DeSteph, Norment, Peake--5. RULE 36--0.

S.B. 1397 (one thousand three hundred ninety-seven) was read by title the third time and, on motion of Senator Norment, was passed with its title.

The recorded vote is as follows: YEAS--36, NAYS--3, RULE 36--0.

YEAS--Barker, Bell, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Boysko, Locke, McPike--3. RULE 36--0.

S.B. 1470 (one thousand four hundred seventy) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

RECONSIDERATION

Senator Cosgrove moved to reconsider the vote by which **S.B. 1160** (one thousand one hundred sixty) was passed with its title.

The motion was agreed to.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

S.B. 1160, on motion of Senator Suetterlein, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--7. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Suetterlein, Vogel--31.

NAYS--McDougle, Morrissey, Reeves, Spruill, Stanley, Stuart, Surovell--7. RULE 36--0.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- S.B. 1234 (one thousand two hundred thirty-four).
- **S.B.** 1267 (one thousand two hundred sixty-seven).
- S.B. 1415 (one thousand four hundred fifteen).
- S.B. 1426 (one thousand four hundred twenty-six).
- **S.B.** 1475 (one thousand four hundred seventy-five).

The motion was agreed to.

S.B. 1234 (one thousand two hundred thirty-four) was taken up.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 54.1-3926 of the Code of Virginia, relating to foreign applicants for Virginia Bar examination.

The reading of the substitute was waived.

On motion of Senator Petersen, the substitute was agreed to.

S.B. 1415 (one thousand four hundred fifteen) was taken up.

The following amendments proposed by the Committee on the Judiciary were offered:

```
1. Line 117, introduced, after that
strike

endangers
insert

endanger

2. Line 118, introduced, after development or
strike

results
insert

result
```

The reading of the amendments was waived.

On motion of Senator Stanley, the amendments were agreed to.

S.B. 1426 (one thousand four hundred twenty-six) was taken up.

The following amendments proposed by the Committee on the Judiciary were offered:

```
    Line 191, introduced, after defendant
insert
        , unless an order for restitution is docketed in the name of the victim or it is
ordered that an assignment of the judgment to the victim be docketed
```

```
2. Line 216, introduced, after a strike judgement insert judgment
```

3. Line 309, introduced, after *assignment of the* strike

judgement

insert

judgment

The reading of the amendments was waived.

On motion of Senator Stanley, the amendments were agreed to.

S.B. 1475 (one thousand four hundred seventy-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 19.2-56 of the Code of Virginia, relating to execution of search warrants; emergency.

The reading of the substitute was waived.

On motion of Senator Stuart, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1234 (one thousand two hundred thirty-four) as amended.
- **S.B. 1267** (one thousand two hundred sixty-seven).
- **S.B.** 1415 (one thousand four hundred fifteen) as amended.
- S.B. 1426 (one thousand four hundred twenty-six) as amended.
- S.B. 1475 (one thousand four hundred seventy-five) as amended.

Senator Saslaw moved that the Rules be suspended and the third reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- **S.B. 1234** (one thousand two hundred thirty-four).
- S.B. 1267 (one thousand two hundred sixty-seven).
- **S.B. 1415** (one thousand four hundred fifteen).
- **S.B. 1426** (one thousand four hundred twenty-six).
- S.B. 1475 (one thousand four hundred seventy-five).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

- **S.B. 1234** (one thousand two hundred thirty-four).
- **S.B. 1267** (one thousand two hundred sixty-seven).
- S.B. 1415 (one thousand four hundred fifteen).
- S.B. 1426 (one thousand four hundred twenty-six).

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1475 (one thousand four hundred seventy-five), on motion of Senator Stuart, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Spruill--1.

RULE 36--0.

S.B. 1198 (one thousand one hundred ninety-eight) was read by title the second time.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 46, introduced, after person

insert

, or in support of the emissions inspection program authorized by § 46.2-1177 et seq

2. Line 49, introduced, after than

strike

seven

insert

30

3. Line 52, introduced, after plate

insert

unless such license plate reader technology is being utilized as part of the emissions inspection program authorized by § 46.2-1177 et seq

4. Line 72, introduced, after manual,

strike

the remainder of line 72 and through subject on line 73

insert

and regardless of any separation of databases within an agency, or among or between cooperating agencies, that collects, stores, or disseminates personal information The reading of the amendments was waived.

Senator Petersen moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

Senator Petersen offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-3800, 2.2-3801, 2.2-3802, and 52-48 of the Code of Virginia, relating to Government Data Collection and Dissemination Practices Act; license plate readers.

On motion of Senator Petersen, the reading of the substitute was waived.

Senator Petersen moved that the substitute be agreed to.

- **S.B.** 1198, on motion of Senator Vogel, was passed by temporarily.
- **S.B. 1384** (one thousand three hundred eighty-four) was taken up, the substitute offered by Senator Surovell having been agreed to and amendments Nos. 2 and 3 to the substitute having been offered by Senator Surovell on February 2, 2021.

Senator Surovell withdrew amendments Nos. 2 and 3 to the substitute.

Senator Survoell offered an amendment in the nature of a substitute No. 2, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Chapter 43 of Title 2.2 an article numbered 7, consisting of sections numbered 2.2-4377.1 through 2.2-4377.5, relating to the Virginia Public Procurement Act; local arbitration agreements.

Senator Surovell withdrew substitute No. 2.

- S.B. 1384, on motion of Senator Surovell, was passed by for the day.
- S.B. 1129 (one thousand one hundred twenty-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-282 and 18.2-433.2 of the Code of Virginia, relating to military honor guards and veterans service organizations; paramilitary activities.

The reading of the substitute was waived.

On motion of Senator Reeves, the substitute was agreed to.

On motion of Senator Reeves, the bill was ordered to be engrossed and read by title the third time.

S.B. 1306 (one thousand three hundred six) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 18.2-57 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 19.2-9.2, relating to assault and battery; penalty.

The reading of the substitute was waived.

On motion of Senator Morrissey, the substitute was agreed to.

Senator Edwards offered the following amendments to the substitute:

1. Line 102, substitute strike

Before insert

In the event of

2. Line 105, substitute, after (ii)

strike

the rest of line 105 and through Commonwealth. on line 106

insert

such investigative information shall be provided in discovery to the defendant prior to trial.

On motion of Senator Edwards, the reading of the amendments was waived.

Senator Morrissey requested that, pursuant to Senate Rule 31, there be a division of the amendments for consideration.

On motion of Senator Edwards, amendment No. 1 was agreed to.

On motion of Senator Edwards, amendment No. 2 was agreed to.

On motion of Senator Morrissey, the bill was ordered to be engrossed and read by title the third time.

S.B. 1385 (one thousand three hundred eighty-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Local Government was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 15.2-816.1 of the Code of Virginia, relating to underground utility facilities; Fairfax County.

The reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

On motion of Senator Surovell, the bill was ordered to be engrossed and read by title the third time.

S.B. 1446 (one thousand four hundred forty-six), on motion of Senator Surovell, was passed by for the day.

S.B. 1198 (one thousand one hundred ninety-eight) was taken up.

On motion of Senator Petersen, the substitute was agreed to.

On motion of Senator Petersen, the bill was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1197 (one thousand one hundred ninety-seven).
- S.B. 1212 (one thousand two hundred twelve).
- **S.B. 1247** (one thousand two hundred forty-seven).
- S.B. 1258 (one thousand two hundred fifty-eight).
- S.B. 1275 (one thousand two hundred seventy-five).
- S.B. 1295 (one thousand two hundred ninety-five).
- S.B. 1391 (one thousand three hundred ninety-one).
- S.B. 1398 (one thousand three hundred ninety-eight).
- S.B. 1405 (one thousand four hundred five).
- S.B. 1420 (one thousand four hundred twenty).
- **S.B. 1425** (one thousand four hundred twenty-five).
- S.B. 1196 (one thousand one hundred ninety-six).
- S.B. 1201 (one thousand two hundred one).
- **S.B. 1226** (one thousand two hundred twenty-six).
- S.B. 1252 (one thousand two hundred fifty-two).
- **S.B. 1261** (one thousand two hundred sixty-one).
- S.B. 1262 (one thousand two hundred sixty-two).
- S.B. 1289 (one thousand two hundred eighty-nine).
- **S.B.** 1310 (one thousand three hundred ten).
- S.B. 1318 (one thousand three hundred eighteen).
- S.B. 1341 (one thousand three hundred forty-one).
- S.B. 1350 (one thousand three hundred fifty).
- **S.B. 1380** (one thousand three hundred eighty).
- S.B. 1442 (one thousand four hundred forty-two).
- **S.B.** 1443 (one thousand four hundred forty-three).

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- **S.B.** 1197 (one thousand one hundred ninety-seven).
- S.B. 1212 (one thousand two hundred twelve).
- **S.B. 1247** (one thousand two hundred forty-seven).
- S.B. 1258 (one thousand two hundred fifty-eight).
- S.B. 1275 (one thousand two hundred seventy-five).
- S.B. 1295 (one thousand two hundred ninety-five).
- S.B. 1391 (one thousand three hundred ninety-one).
- S.B. 1398 (one thousand three hundred ninety-eight).
- S.B. 1405 (one thousand four hundred five).
- S.B. 1420 (one thousand four hundred twenty).
- S.B. 1425 (one thousand four hundred twenty-five).
- **S.B.** 1196 (one thousand one hundred ninety-six).
- S.B. 1201 (one thousand two hundred one).
- S.B. 1226 (one thousand two hundred twenty-six).
- S.B. 1252 (one thousand two hundred fifty-two).
- S.B. 1261 (one thousand two hundred sixty-one).
- **S.B. 1262** (one thousand two hundred sixty-two).
- S.B. 1289 (one thousand two hundred eighty-nine).
- S.B. 1310 (one thousand three hundred ten).
- S.B. 1318 (one thousand three hundred eighteen).
- **S.B. 1341** (one thousand three hundred forty-one).
- S.B. 1350 (one thousand three hundred fifty).
- **S.B. 1380** (one thousand three hundred eighty).
- S.B. 1442 (one thousand four hundred forty-two).
- **S.B.** 1443 (one thousand four hundred forty-three).

SENATE JOINT RESOLUTIONS ON THIRD READING

Senator Saslaw moved that the following Senate joint resolutions, the titles of the joint resolutions having been printed in the Calendar for their third reading, be placed before the Senate by number only:

- S.J.R. 288 (two hundred eighty-eight).
- S.J.R. 293 (two hundred ninety-three).
- S.J.R. 323 (three hundred twenty-three).

The motion was agreed to.

Senator Saslaw moved that the questions on agreeing to the Senate joint resolutions that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate joint resolutions were agreed to en bloc:

- S.J.R. 288 (two hundred eighty-eight).
- S.J.R. 293 (two hundred ninety-three).
- S.J.R. 323 (three hundred twenty-three).

SENATE RESOLUTION ON THIRD READING

S.R. 95 (ninety-five), on motion of Senator Suetterlein, was passed by for the day.

SENATE BILL ON THIRD READING RECONSIDERATION

Senator Dunnavant moved to reconsider the vote by which **S.B. 1288** (one thousand two hundred eighty-eight) was passed with its title on February 2, 2021.

The motion was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Vogel--38.

NAYS--0. RULE 36--0.

RECONSIDERATION

Senator Dunnavant moved to reconsider the vote by which **S.B. 1288** (one thousand two hundred eighty-eight) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Vogel--38.

NAYS--0. RULE 36--0.

Senator Dunnavant offered the following amendments to the substitute:

1. Line 70 substitute, after disabilities.

strike

the remainder of line 70, all of lines 71 and 72, and through meeting. on line 73

2. At the beginning of line 85, substitute

strike

and are of high quality

insert

governing IEP content

3. Line 85, substitute, after reviews and

insert

recommendations regarding

4. Line 92, substitute, after members.

insert

Nothing in this section shall be construed to (i) direct the Department to make determinations regarding whether a particular IEP provides a free appropriate public education to any individual student or (ii) authorize the Department to override a parent's consent to proposed revisions to an individual student's IEP. In determining corrective actions, the Department shall make recommendations to the relevant school division regarding, among other things, those individual IEPs for which the IEP team should convene to consider revisions necessary to incorporate content required by special education regulations.

5. After line 590, substitute

insert

14. That the Board of Education shall amend its Regulations Governing Special Education Programs for Children with Disabilities in Virginia (8VAC20-81-10 et seq.) to include a provision requiring that, if a school division develops a draft individualized education program (IEP) prior to a scheduled IEP meeting, it shall provide such draft IEP to the parents at least two business days in advance of such IEP meeting.

On motion of Senator Dunnavant, the reading of the amendments was waived.

On motion of Senator Dunnavant, the amendments were agreed to.

On motion of Senator Dunnavant, the bill was ordered to be engrossed and read by title the third time.

Senator Dunnavant moved that the Rules be suspended and the third reading of the title of **S.B. 1288** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1288, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SUPPLEMENTAL CALENDAR NO. 1

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- **S.B.** 1162 (one thousand one hundred sixty-two).
- S.B. 1163 (one thousand one hundred sixty-three).
- S.B. 1210 (one thousand two hundred ten).
- S.B. 1307 (one thousand three hundred seven).
- **S.B.** 1313 (one thousand three hundred thirteen).
- **S.B.** 1336 (one thousand three hundred thirty-six).
- S.B. 1357 (one thousand three hundred fifty-seven).
- **S.B.** 1365 (one thousand three hundred sixty-five).
- S.B. 1375 (one thousand three hundred seventy-five).
- **S.B.** 1389 (one thousand three hundred eighty-nine).
- S.B. 1429 (one thousand four hundred twenty-nine).
- **S.B. 1436** (one thousand four hundred thirty-six).
- **S.B. 1461** (one thousand four hundred sixty-one).
- **S.B.** 1465 (one thousand four hundred sixty-five).
- **S.B.** 1105 (one thousand one hundred five).
- **S.B.** 1113 (one thousand one hundred thirteen).
- **S.B.** 1127 (one thousand one hundred twenty-seven).
- S.B. 1146 (one thousand one hundred forty-six).
- S.B. 1250 (one thousand two hundred fifty).
- S.B. 1257 (one thousand two hundred fifty-seven).
- S.B. 1278 (one thousand two hundred seventy-eight).
- **S.B. 1282** (one thousand two hundred eighty-two).
- **S.B. 1284** (one thousand two hundred eighty-four).
- S.B. 1301 (one thousand three hundred one).
- S.B. 1315 (one thousand three hundred fifteen).
- **S.B.** 1316 (one thousand three hundred sixteen).
- **S.B. 1326** (one thousand three hundred twenty-six).
- S.B. 1333 (one thousand three hundred thirty-three).
- **S.B. 1339** (one thousand three hundred thirty-nine).
- **S.B. 1362** (one thousand three hundred sixty-two).
- **S.B. 1363** (one thousand three hundred sixty-three). **S.B. 1382** (one thousand three hundred eighty-two).
- **S.B.** 1392 (one thousand three hundred ninety-two).
- S.B. 1395 (one thousand three hundred ninety-five).
- S.B. 1396 (one thousand three hundred ninety-six).
- S.B. 1401 (one thousand four hundred one).
- **S.B. 1406** (one thousand four hundred six).
- S.B. 1418 (one thousand four hundred eighteen).
- S.B. 1428 (one thousand four hundred twenty-eight).
- **S.B. 1462** (one thousand four hundred sixty-two).
- S.B. 1469 (one thousand four hundred sixty-nine).
- S.B. 1471 (one thousand four hundred seventy-one).

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- **S.B.** 1162 (one thousand one hundred sixty-two).
- **S.B.** 1163 (one thousand one hundred sixty-three).
- S.B. 1210 (one thousand two hundred ten).
- S.B. 1307 (one thousand three hundred seven).
- **S.B. 1313** (one thousand three hundred thirteen).
- **S.B. 1336** (one thousand three hundred thirty-six).
- **S.B.** 1357 (one thousand three hundred fifty-seven).
- **S.B. 1365** (one thousand three hundred sixty-five).
- S.B. 1375 (one thousand three hundred seventy-five).
- **S.B.** 1389 (one thousand three hundred eighty-nine).
- **S.B. 1429** (one thousand four hundred twenty-nine).
- **S.B. 1436** (one thousand four hundred thirty-six).
- **S.B. 1461** (one thousand four hundred sixty-one).
- **S.B.** 1465 (one thousand four hundred sixty-five).
- S.B. 1105 (one thousand one hundred five).
- **S.B.** 1113 (one thousand one hundred thirteen).
- **S.B.** 1127 (one thousand one hundred twenty-seven).
- **S.B.** 1146 (one thousand one hundred forty-six).
- **S.B. 1250** (one thousand two hundred fifty).
- **S.B. 1257** (one thousand two hundred fifty-seven). **S.B. 1278** (one thousand two hundred seventy-eight).
- S.B. 1282 (one thousand two hundred eighty-two). S.B. 1284 (one thousand two hundred eighty-four).
- **S.B. 1301** (one thousand three hundred one).
- S.B. 1315 (one thousand three hundred fifteen).
- **S.B.** 1316 (one thousand three hundred sixteen).
- S.B. 1326 (one thousand three hundred twenty-six).
- S.B. 1333 (one thousand three hundred thirty-three).
- **S.B.** 1339 (one thousand three hundred thirty-nine).
- **S.B. 1362** (one thousand three hundred sixty-two).
- S.B. 1363 (one thousand three hundred sixty-three).
- **S.B.** 1382 (one thousand three hundred eighty-two).
- **S.B. 1392** (one thousand three hundred ninety-two).
- S.B. 1395 (one thousand three hundred ninety-five).
- S.B. 1396 (one thousand three hundred ninety-six).
- S.B. 1401 (one thousand four hundred one).
- **S.B. 1406** (one thousand four hundred six).
- **S.B.** 1418 (one thousand four hundred eighteen).
- S.B. 1428 (one thousand four hundred twenty-eight).

- **S.B. 1462** (one thousand four hundred sixty-two).
- S.B. 1469 (one thousand four hundred sixty-nine).
- S.B. 1471 (one thousand four hundred seventy-one).

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

Jacke Solman

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

THURSDAY, FEBRUARY 4, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Emmett W. Hanger, Jr., Twenty-fourth Senatorial District, offered the following prayer:

There's a scripture that I like in Corinthians that states, "where the Spirit of the Lord is, there is liberty." We're all, for the most part and most of the time, proud to be Americans, proud to live in a country that values and protects our freedoms. What we sometimes overlook is that real freedom cannot be guaranteed by a Constitution like ours, but rather must be accompanied by a citizenry that possesses a sense of decency, fair play, and respect for their fellow man. In other words, "where the Spirit of the Lord is, there is liberty."

Dear Lord, we are blessed to live in what we sometimes regard as the greatest state in the greatest nation in the world. Please help us by our words and deeds this day to exhibit such a sense of decency, fair play, and respect for others, that we will confirm by our actions: liberty is assured; for the spirit of the Lord is with us. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Nathan A. Hatfield, Assistant Clerk-LINCS, Senate Clerk's Office.

On motion of Senator McDougle, the Rules were suspended to allow Senator Norment, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Spruill, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, DeSteph, McDougle, Petersen--4.

RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 3, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1750.** A BILL to amend the Code of Virginia by adding in Title 3.2 a chapter numbered 33.1, consisting of sections numbered 3.2-3304 through 3.2-3307, relating to Dairy Producer Margin Coverage Premium Assistance Program; report.
- **H.B. 1763.** A BILL to amend and reenact §§ 58.1-339.3 and 58.1-439.5 of the Code of Virginia, relating to tax credit; agricultural best management practices.
- H.B. 1776. A BILL to require the Board of Education to temporarily extend certain teachers' licenses.
- **H.B. 1778.** A BILL to amend and reenact § 15.2-901 of the Code of Virginia, relating to removal of clutter from property.
- **H.B. 1796.** A BILL to amend and reenact § 46.2-746.5 of the Code of Virginia, relating to special license plates; Virginia National Guard retirees.
- **H.B. 1801.** A BILL to amend and reenact § 33.2-802 of the Code of Virginia, relating to disposing of litter; penalty.
- **H.B. 1843.** A BILL to amend and reenact § 18.2-340.33 of the Code of Virginia, relating to charitable gaming; increase in certain maximum allowable prize amounts.
- H.B. 1856. A BILL to amend and reenact §§ 32.1-291.7, 54.1-2982, 54.1-2983, 54.1-2984, 59.1-481, 64.2-100, 64.2-403, 64.2-404, 64.2-407, 64.2-450, and 64.2-701 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 4 of Title 64.2 an article numbered 7, consisting of sections numbered 64.2-459 through 64.2-468, relating to electronic execution of estate planning documents; Uniform Electronic Wills Act.
- **H.B. 1879.** A BILL to amend and reenact §§ 4.1-119, as it is currently effective and as it shall become effective, 4.1-204, as it is currently effective and as it shall become effective, 4.1-206.1, as it shall become effective, 4.1-207, 4.1-210, 4.1-212.1, as it is currently effective and as it shall become effective, and 4.1-221 of the Code of Virginia, relating to alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption.
- **H.B. 1895.** A BILL to amend and reenact §§ 19.2-353.5 through 19.2-355 of the Code of Virginia, relating to fines and costs; accrual of interest; deferral or installment payment agreements.
- **H.B. 1912.** A BILL to amend and reenact §§ 16.1-263, 16.1-286, and 16.1-290 of the Code of Virginia, relating to child support payments; juvenile in custody of or committed to the Department of Juvenile Justice.
- **H.B. 1919.** A BILL to amend the Code of Virginia by adding a section numbered 15.2-958.3:1, relating to local green banks.

- H.B. 1932. A BILL to repeal § 63.2-1709.3 of the Code of Virginia, relating to child-placing agencies; conscience clause; repeal.
- **H.B. 1962.** A BILL to amend and reenact §§ 16.1-281, 16.1-283, 63.2-906, and 63.2-910.2 of the Code of Virginia, relating to foster care; termination of parental rights; relatives and fictive kin.
- **H.B. 1963.** A BILL to amend the Code of Virginia by adding in Article 5 of Chapter 1 of Title 32.1 a section numbered 32.1-34.3, relating to cooperative local health budget; report.
- **H.B. 1973.** A BILL to amend and reenact §§ 4.1-206.3, as it shall become effective, and 4.1-209, as it is currently effective, of the Code of Virginia, relating to alcoholic beverage control; privileges of banquet licensees.
- **H.B. 1985.** A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability of health care providers from COVID-19.
- **H.B. 1987.** A BILL to amend and reenact §§ 32.1-325, 38.2-3418.16, and 54.1-3303 of the Code of Virginia, relating to telemedicine.
- **H.B. 1988.** A BILL to amend and reenact §§ 54.1-3408.3, 54.1-3442.5, 54.1-3442.6, and 54.1-3442.7 of the Code of Virginia, relating to Board of Pharmacy; pharmaceutical processors; processing and dispensing cannabis oil.
- **H.B. 1989.** A BILL to amend the Code of Virginia by adding in Article 3 of Chapter 2 of Title 32.1 a section numbered 32.1-48.001, relating to public health emergency; emergency medical services agencies; real-time access to information.
- **H.B. 2005.** A BILL to amend and reenact §§ 32.1-309.1, 54.1-2800, 54.1-2807, 54.1-2825, and 57-27.3 of the Code of Virginia; to amend the Code of Virginia by adding in Article 1 of Chapter 3 of Title 57 sections numbered 57-27.4, 57-27.5, and 57-27.6; and to repeal §§ 54.1-2807.01 and 54.1-2807.02 of the Code of Virginia, relating to disposition of the remains of a decedent; persons to make arrangements for funeral and disposition of remains.
- H.B. 2007. A BILL to amend and reenact § 2.2-3705.6 of the Code of Virginia and to amend the Code of Virginia by adding in Article 3 of Chapter 1 of Title 32.1 a section numbered 32.1-23.3, by adding a section numbered 38.2-3407.15:6, by adding in Article 1 of Chapter 34 of Title 38.2 a section numbered 38.2-3407.22, by adding in Article 3 of Chapter 34 of Title 54.1 a section numbered 54.1-3436.1, and by adding in Article 4 of Chapter 34 of Title 54.1 a section numbered 54.1-3442.02, relating to prescription drug price transparency.
- **H.B. 2010.** A BILL to amend and reenact § 53.1-202.3, as it shall become effective, of the Code of Virginia, relating to earned sentence credits; revocation of suspended sentence.
- **H.B. 2035.** A BILL to amend and reenact § 63.2-608 of the Code of Virginia, relating to Virginia Initiative for Education and Work; Full Employment Program.
- H.B. 2047. A BILL to amend and reenact §§ 19.2-120, 19.2-163.03, and 19.2-299 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 16 of Title 19.2 a section numbered 19.2-271.6, relating to criminal proceedings; consideration of mental condition and intellectual and developmental disabilities.

- H.B. 2064. A BILL to amend and reenact §§ 17.1-223, 47.1-2, 47.1-16, and 55.1-606 of the Code of Virginia, relating to electronic notary; remote notarization; emergency.
 EMERGENCY
- **H.B. 2069.** A BILL to amend and reenact §§ 46.2-742.1, 46.2-742.2, 46.2-745.1, and 46.2-745.2 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 46.2-725.3, 46.2-745.4, and 46.2-745.5, relating to special license plates; military decorations.
- H.B. 2075. A BILL to rename certain sections of U.S. Route 1 in Virginia the "Emancipation Highway" and to repeal Chapter 286 of the Acts of Assembly of 1922.
- **H.B. 2086.** A BILL to amend and reenact §§ 19.2-389, as it is currently effective and as it shall become effective, 22.1-289.035, as it shall become effective, 22.1-289.039, as it shall become effective, 63.2-1720.1, and 63.2-1724 of the Code of Virginia, relating to child care providers; background check portability; subsidy pilot program; report.
- H.B. 2091. A BILL to amend and reenact §§ 2.02, 5.021, and 6.02, § 12.01, as amended, and § 15.03 of Chapter 227, as amended, of the Acts of Assembly of 1954, which provided a charter for the City of Covington, and to amend and reenact § 22.1-32 of the Code of Virginia, relating to consolidated school board of Alleghany County and the City of Covington; school board salaries.
- **H.B. 2101.** A BILL to repeal the second enactment of Chapter 525 of the Acts of Assembly of 2020, relating to GO Virginia Grants; matching funds; sunset.
- **H.B. 2110.** A BILL to amend and reenact § 2.2-3802 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 9 of Title 19.2 a section numbered 19.2-134.1, relating to pretrial data collection.
- H.B. 2113. A BILL to amend and reenact §§ 9.1-101, as it is currently effective and as it shall become effective, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-340, 19.2-389.3, and 19.2-390 of the Code of Virginia and to amend the Code of Virginia by adding in Title 19.2 a chapter numbered 23.2, consisting of sections numbered 19.2-392.5 through 19.2-392.16, relating to automatic expungement of criminal records; penalties.
- **H.B. 2117.** A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; special education programs.
- H.B. 2129. A BILL to amend and reenact §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14 of the Code of Virginia, relating to Chesapeake Bay Phase III Watershed Improvement Plan; nutrient removal; regulations.
- **H.B. 2131.** A BILL to amend and reenact §§ 4.1-230, as it is currently effective and as it shall become effective, and 15.2-907 of the Code of Virginia, relating to alcoholic beverage control; license application; locality input; corrective action.
- **H.B. 2146.** A BILL to amend and reenact § 46.2-889 of the Code of Virginia, relating to parked cars; VDOT right-of-way.
- **H.B. 2155.** A BILL to amend and reenact §§ 2.2-522, 2.2-3905, 2.2-3907, and 2.2-3908 of the Code of Virginia, relating to the Virginia Human Rights Act; nondiscrimination in employment; sexual harassment and workplace harassment.

- **H.B. 2161.** A BILL to amend and reenact §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310 of the Code of Virginia, relating to public accommodations, employment, and housing; prohibited discrimination on the basis of status as active military or a military spouse.
- **H.B. 2163.** A BILL to amend and reenact §§ 46.2-100, 46.2-208, 46.2-209, 46.2-209.1, and 46.2-328.3 of the Code of Virginia, relating to Department of Motor Vehicles; privileged information.
- **H.B. 2168.** A BILL to amend and reenact § 18.2-325, as it is currently effective and as it shall become effective, of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-331.1, relating to illegal gambling; skills games; civil penalty; enforcement by localities and Attorney General.
- **H.B. 2193.** A BILL to amend the Code of Virginia by adding in Chapter 16 of Title 8.01 a section numbered 8.01-425.2, relating to settlement agreements; staying of dismissal.
- H.B. 2202. A BILL to amend and reenact § 54.1-1141 of the Code of Virginia, relating to professions and occupations; Board for Contractors; exemption from licensure as an elevator mechanic or accessibility mechanic.
- H.B. 2217. A BILL to amend and reenact § 15.2-1809 of the Code of Virginia, relating to liability of public access authorities.
- **H.B. 2218.** A BILL to amend and reenact §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8 of the Code of Virginia, relating to pharmaceutical processors; cannabis products.
- **H.B. 2222.** A BILL to amend and reenact § 2.2-2001.4 of the Code of Virginia, relating to the military medical personnel program.
- H.B. 2249. A BILL to amend and reenact §§ 17.1-275, 55.1-1200, 55.1-1204, 55.1-1206, 55.1-1208, 55.1-1211, 55.1-1226, 64.2-2008, and 64.2-2012 of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; landlord charges for security deposits, insurance premiums for damage insurance, and insurance premiums for renter's insurance; filing of information regarding resident agent appointed by nonresident property owner.
- **H.B. 2252.** A BILL to authorize a quitclaim and release of interest and the conveyance of an easement by the Board of Wildlife Resources in Tazewell County.
- **H.B. 2257.** A BILL to amend and reenact §§ 1, 2, 8, 10, 11, 12, 21, and 22, as amended, §§ 25, 26, 28, and 31, §§ 32 and 35, as amended, and § 38 of Chapter 66 of the Acts of Assembly of 1960, relating to Hampton Roads Sanitation District.
- **H.B. 2258.** A BILL to amend and reenact § 19.2-387.3 of the Code of Virginia, relating to Substantial Risk Order Registry; maintenance and access.
- **H.B. 2261.** A BILL to amend and reenact § 46.2-744 of the Code of Virginia, relating to special license plates; member of the Virginia National Guard.

- **H.B. 2266.** A BILL to amend and reenact §§ 4.1-206, 4.1-206.3, as it shall become effective, 4.1-231, 4.1-231.1, as it shall become effective, 4.1-233, 4.1-233.1, as it shall become effective, and 4.1-308 of the Code of Virginia, relating to alcoholic beverage control; outdoor refreshment areas.
- **H.B. 2291.** A BILL to amend and reenact § 9.1-601, as it shall become effective, of the Code of Virginia, relating to law-enforcement civilian oversight bodies; deputy sheriffs.
- **H.B. 2299.** A BILL to amend the Code of Virginia by adding a section numbered 22.1-214.4, relating to the Department of Education; duties; special education.
- H.B. 2300. A BILL to amend and reenact § 32.1-127 of the Code of Virginia, relating to State Board of Health; hospitals; emergency treatment for substance use-related emergencies; services.
- H.B. 2310. A BILL relating to concealed handgun permits; demonstration of competence; emergency. EMERGENCY
- **H.B. 2319.** A BILL to amend the Code of Virginia by adding a section numbered 18.2-308.2:02, relating to allowing access to firearms; cohabitants; civil penalty.
- **H.B. 2323.** A BILL to amend and reenact § 15.2-4116 of the Code of Virginia, relating to library aid; former regional library system.
- **H.B. 2326.** A BILL to amend and reenact § 15.2-914, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to regulation of child care services in localities.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--35. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, McDougle, Petersen--3.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1750 and H.B. 2129 were referred to the Committee on Agriculture, Conservation and Natural Resources.
 - **H.B. 1763** and **H.B. 2101** were referred to the Committee on Finance and Appropriations.

- H.B. 1776, H.B. 1963, H.B. 1987, H.B. 1988, H.B. 1989, H.B. 2007, H.B. 2035, H.B. 2117, H.B. 2218, H.B. 2299, and H.B. 2300 were referred to the Committee on Education and Health.
- H.B. 1778, H.B. 1919, H.B. 2091, H.B. 2217, H.B. 2257, H.B. 2323, and H.B. 2326 were referred to the Committee on Local Government.
- H.B. 1796, H.B. 1801, H.B. 2069, H.B. 2075, H.B. 2146, H.B. 2163, and H.B. 2261 were referred to the Committee on Transportation.
- H.B. 1843, H.B. 2005, H.B. 2155, H.B. 2161, H.B. 2202, H.B. 2222, and H.B. 2249 were referred to the Committee on General Laws and Technology.
- H.B. 1856, H.B. 1895, H.B. 1912, H.B. 2047, H.B. 2064, H.B. 2110, H.B. 2113, H.B. 2168, H.B. 2193, H.B. 2252, H.B. 2258, H.B. 2291, H.B. 2310, and H.B. 2319 were referred to the Committee on the Judiciary.
- H.B. 1879, H.B. 1932, H.B. 1962, H.B. 1973, H.B. 2010, H.B. 2086, H.B. 2131, and H.B. 2266 were referred to the Committee on Rehabilitation and Social Services.
 - H.B. 1985 was referred to the Committee on Commerce and Labor.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Petersen from the Committee on Agriculture, Conservation and Natural Resources:

- S.B. 1164 (one thousand one hundred sixty-four) with substitute.
- **S.B.** 1265 (one thousand two hundred sixty-five) with substitute.
- **S.B. 1311** (one thousand three hundred eleven) with substitute.
- S.B. 1319 (one thousand three hundred nineteen) with substitute.
- **S.B. 1354** (one thousand three hundred fifty-four) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - S.B. 1374 (one thousand three hundred seventy-four) with amendment.
 - S.B. 1404 (one thousand four hundred four).
- **S.B. 1411** (one thousand four hundred eleven) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills and joint resolutions, having been considered by the committee in session, were reported by Senator Deeds from the Committee on Privileges and Elections:

- **S.B.** 1109 (one thousand one hundred nine) with amendments.
- S.B. 1153 (one thousand one hundred fifty-three).
- S.B. 1444 (one thousand four hundred forty-four) with substitute.
- S.J.R. 270 (two hundred seventy) with substitute.
- S.J.R. 271 (two hundred seventy-one).
- S.J.R. 272 (two hundred seventy-two).
- S.J.R. 275 (two hundred seventy-five) with amendments.
- S.J.R. 289 (two hundred eighty-nine).
- S.J.R. 290 (two hundred ninety).
- S.J.R. 310 (three hundred ten) with amendment.
- S.J.R. 322 (three hundred twenty-two).

S.B. 1354 and S.B. 1411 were rereferred to the Committee on Finance and Appropriations.

CALENDAR

SENATE BILLS ON THIRD READING

- **S.B. 1138** (one thousand one hundred thirty-eight), on motion of Senator Locke, was passed by for the day.
- **S.B.** 1198 (one thousand one hundred ninety-eight) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--28. NAYS--11. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Howell, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Reeves, Spruill, Stuart, Suetterlein, Surovell--28.

NAYS--Cosgrove, DeSteph, Hashmi, Kiggans, Lewis, Norment, Pillion, Ruff, Saslaw, Stanley, Vogel--11.

RULE 36--0.

S.B. 1129 (one thousand one hundred twenty-nine) was read by title the third time and, on motion of Senator Reeves, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, Ebbin, Edwards, Locke, Petersen--5.

RULE 36--0.

S.B. 1306 (one thousand three hundred six) was read by title the third time and, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1385 (one thousand three hundred eighty-five) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows: YEAS--33. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--33.

NAYS--Chase, Cosgrove, DeSteph, Reeves, Stanley, Suetterlein--6. RULE 36--0.

RECESS

At 1:05 p.m., Senator Saslaw moved that the Senate recess until 2:10 p.m.

The motion was agreed to.

The hour of 2:10 p.m. having arrived, the Chair was resumed.

COMMITTEE REPORT

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- S.B. 1354 (one thousand three hundred fifty-four) with amendments.
- S.B. 1411 (one thousand four hundred eleven).

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- **S.B. 1338** (one thousand three hundred thirty-eight).
- S.B. 1162 (one thousand one hundred sixty-two).
- S.B. 1163 (one thousand one hundred sixty-three).
- S.B. 1197 (one thousand one hundred ninety-seven).
- S.B. 1210 (one thousand two hundred ten).
- S.B. 1212 (one thousand two hundred twelve).
- S.B. 1247 (one thousand two hundred forty-seven).
- S.B. 1258 (one thousand two hundred fifty-eight).
- S.B. 1275 (one thousand two hundred seventy-five).
- S.B. 1295 (one thousand two hundred ninety-five).
- **S.B. 1307** (one thousand three hundred seven).
- S.B. 1313 (one thousand three hundred thirteen).
- **S.B.** 1336 (one thousand three hundred thirty-six).
- S.B. 1357 (one thousand three hundred fifty-seven).
- S.B. 1365 (one thousand three hundred sixty-five).
- S.B. 1375 (one thousand three hundred seventy-five).
- **S.B. 1389** (one thousand three hundred eighty-nine).

```
S.B. 1391 (one thousand three hundred ninety-one).
```

S.B. 1398 (one thousand three hundred ninety-eight).

S.B. 1405 (one thousand four hundred five).

S.B. 1420 (one thousand four hundred twenty).

S.B. 1425 (one thousand four hundred twenty-five).

S.B. 1429 (one thousand four hundred twenty-nine).

S.B. 1436 (one thousand four hundred thirty-six).

S.B. 1461 (one thousand four hundred sixty-one).

S.B. 1465 (one thousand four hundred sixty-five).

The motion was agreed to.

S.B. 1338 (one thousand three hundred thirty-eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 32.1-325 and 38.2-3418.16 of the Code of Virginia, relating to telemedicine services; remote patient monitoring services.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

1. Line 174, substitute, after surgery;

insert

and

2. Line 176, substitute, after months

strike

; and (vi) patients at risk for hospitalization

3. After line 374, substitute

insert

4. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendments was waived.

On motion of Senator Barker, the amendments were agreed to.

S.B. 1162 (one thousand one hundred sixty-two) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-339.3 and 58.1-439.5 of the Code of Virginia, relating to tax credit; agricultural best management practices.

The reading of the substitute was waived.

On motion of Senator Hanger, the substitute was agreed to.

S.B. 1163 (one thousand one hundred sixty-three) was taken up.

The following amendments proposed by the Committee on Finance and Appropriations were offered:

The reading of the amendments was waived.

On motion of Senator Hanger, the amendments were agreed to.

but before January 1, 2025,

S.B. 1210 (one thousand two hundred ten) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to direct the Department of Environmental Quality to convene working groups to revise permit fee schedules.

The reading of the substitute was waived.

On motion of Senator Petersen, the substitute was agreed to.

S.B. 1212 (one thousand two hundred twelve) was taken up.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

```
1. Line 140, introduced, after higher strike 
educaton 
insert 
education
```

The reading of the amendment was waived.

On motion of Senator Edwards, the amendment was agreed to.

S.B. 1247 (one thousand two hundred forty-seven) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 56-599 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 26 of Title 45.1 a section numbered 45.1-394.1, relating to public disclosure of electric generating facility closures; integrated resource plans.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

S.B. 1258 (one thousand two hundred fifty-eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding a section numbered 62.1-44.15:55.1, relating to solar projects; erosion and sediment control.

The reading of the substitute was waived.

On motion of Senator Marsden, the substitute was agreed to.

The following amendment proposed by the Committee on Finance and Appropriations to the substitute was offered:

1. Line 32, substitute

strike

all of lines 32 through 35

insert

2. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendment was waived.

On motion of Senator Marsden, the amendment was agreed to.

S.B. 1275 (one thousand two hundred seventy-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 65.2-402 of the Code of Virginia, relating to workers' compensation; presumption of compensation for certain diseases; applicable to salaried and volunteer emergency medical services personnel.

The reading of the substitute was waived.

On motion of Senator Marsden, the substitute was agreed to.

The following amendment proposed by the Committee on Finance and Appropriations to the substitute was offered:

1. After line 76, substitute

insert

2. That the provisions of this act shall not apply to any individual who was diagnosed with hypertension or heart disease before July 1, 2021.

The reading of the amendment was waived.

On motion of Senator Marsden, the amendment was agreed to.

S.B. 1295 (one thousand two hundred ninety-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 56-585.1:11 and 56-585.5 of the Code of Virginia, relating to electric utilities; procurement.

The reading of the substitute was waived.

On motion of Senator DeSteph, the substitute was agreed to.

S.B. 1307 (one thousand three hundred seven) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 32.1-325 and 32.1-326.3 of the Code of Virginia, relating to Department of Medical Assistance Services; school-based health services; telemedicine.

The reading of the substitute was waived.

On motion of Senator Dunnavant, the substitute was agreed to.

S.B. 1313 (one thousand three hundred thirteen) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; special education programs.

The reading of the substitute was waived.

Senator Mason moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; special education programs.

The reading of the substitute was waived.

On motion of Senator Mason, the substitute was agreed to.

S.B. 1336 (one thousand three hundred thirty-six) was taken up.

Senator Surovell offered the following amendment:

1. Line 13, introduced, after *case* insert

for any violation of Article 7 (§ 46.2-852 et seq.) of Chapter 8 of Title 46.2

On motion of Senator Surovell, the reading of the amendment was waived.

On motion of Senator Surovell, the amendment was agreed to.

S.B. 1365 (one thousand three hundred sixty-five) was taken up.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 180, introduced, after membership of

strike

27

insert

28

2. Line 181, introduced, after and

strike

15

insert

16

3. Line 182, introduced, after follows:

strike

two

insert

three

The reading of the amendments was waived.

Senator Barker moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 2.2-203.2:4 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 25 of Title 2.2 an article numbered 13, consisting of sections numbered 2.2-2558 through 2.2-2564, relating to data governance; Office of Data Governance and Analytics; Chief Data Officer; Virginia Data Commission; report.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

S.B. 1375 (one thousand three hundred seventy-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability from COVID-19.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

S.B. 1389 (one thousand three hundred eighty-nine) was taken up.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 55.1-703 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 55.1-708.2, relating to property; required disclosures for buyer to exercise due diligence; flood risk report.

The reading of the substitute was waived.

On motion of Senator Lewis, the substitute was agreed to.

The following amendment proposed by the Committee on Finance and Appropriations to the substitute was offered:

```
1. Line 171, substitute strike
all of lines 171, 172, and 173
```

The reading of the amendment was waived.

On motion of Senator Lewis, the amendment was agreed to.

S.B. 1391 (one thousand three hundred ninety-one) was taken up.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 2.2-3802 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 9 of Title 19.2 a section numbered 19.2-134.1, relating to pretrial data collection.

-388-

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

S.B. 1398 (one thousand three hundred ninety-eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-602, 58.1-603, as it is currently effective and as it may become effective, 58.1-3819, as it shall become effective, 58.1-3823, as it shall become effective, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, as it shall become effective, 58.1-3826, 58.1-3842, and 58.1-3843 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 2.2-2320.2 and 58.1-612.2 and by adding in Article 6 of Chapter 38 of Title 58.1 a section numbered 58.1-3818.8, relating to retail sales and transient occupancy taxes on room rentals.

The reading of the substitute was waived.

Senator Norment moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

Senator Norment offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-602, 58.1-603, as it is currently effective and as it may become effective, 58.1-3819, as it shall become effective, 58.1-3823, as it shall become effective, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, as it shall become effective, 58.1-3826, 58.1-3842, and 58.1-3843 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 2.2-2320.2 and 58.1-612.2 and by adding in Article 6 of Chapter 38 of Title 58.1 a section numbered 58.1-3818.8, relating to retail sales and transient occupancy taxes on room rentals.

On motion of Senator Norment, the reading of the substitute was waived.

On motion of Senator Norment, the substitute was agreed to.

S.B. 1425 (one thousand four hundred twenty-five) taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

The reading of the substitute was waived.

On motion of Senator Ruff, the substitute was agreed to.

S.B. 1429 (one thousand four hundred twenty-nine) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to authorize the Commonwealth to lease a portion of property previously used by the Department of Behavioral Health and Developmental Services as the Southwestern Virginia Mental Health Institute and to amend and reenact §§ 1 and 2 of Chapter 678 of the Acts of Assembly of 2019.

The reading of the substitute was waived.

On motion of Senator Pillion, the substitute was agreed to.

S.B. 1436 (one thousand four hundred thirty-six) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Article 2 of Chapter 1 of Title 23.1 a section numbered 23.1-110, by adding in Article 3 of Chapter 1 of Title 32.1 a section numbered 32.1-23.3, and by adding a section numbered 54.1-2400.03, relating to Department of Health; Eligible Health Care Provider Reserve Directory.

The reading of the substitute was waived.

On motion of Senator Hanger, the substitute was agreed to.

The following amendment proposed by the Committee on Finance and Appropriations to the substitute was offered:

1. Line 53, substitute, after *Directory*.

However, a health regulatory board shall not report information for any such person who has notified the health regulatory board in writing that he does not want his information included in the Eligible Health Care Provider Reserve Directory.

The reading of the amendment was waived.

On motion of Senator Hanger, the amendment was agreed to.

S.B. 1465 (one thousand four hundred sixty-five) was taken up.

The following amendments proposed by the Committee on the Judiciary were offered:

1. Line 103, introduced, after *of* insert

up to

2. Line 108, introduced, after *penalty of* insert

up to

The reading of the amendments was waived.

On motion of Senator Reeves, the amendments were agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

- S.B. 1338 (one thousand three hundred thirty-eight) as amended.
- S.B. 1162 (one thousand one hundred sixty-two) as amended.
- S.B. 1163 (one thousand one hundred sixty-three) as amended.
- **S.B.** 1197 (one thousand one hundred ninety-seven).
- S.B. 1210 (one thousand two hundred ten) as amended.
- S.B. 1212 (one thousand two hundred twelve) as amended.
- S.B. 1247 (one thousand two hundred forty-seven) as amended.
- **S.B. 1258** (one thousand two hundred fifty-eight) as amended.
- S.B. 1275 (one thousand two hundred seventy-five) as amended.
- **S.B.** 1295 (one thousand two hundred ninety-five) as amended.
- S.B. 1307 (one thousand three hundred seven) as amended.
- **S.B. 1313** (one thousand three hundred thirteen) as amended.
- **S.B.** 1336 (one thousand three hundred thirty-six) as amended.
- **S.B.** 1365 (one thousand three hundred sixty-five) as amended.
- S.B. 1375 (one thousand three hundred seventy-five) as amended.
- S.B. 1389 (one thousand three hundred eighty-nine) as amended.
- **S.B. 1391** (one thousand three hundred ninety-one) as amended.
- S.B. 1398 (one thousand three hundred ninety-eight) as amended.
- S.B. 1405 (one thousand four hundred five).
- S.B. 1420 (one thousand four hundred twenty).
- S.B. 1425 (one thousand four hundred twenty-five) as amended.
- S.B. 1429 (one thousand four hundred twenty-nine) as amended.
- S.B. 1436 (one thousand four hundred thirty-six) as amended.
- S.B. 1461 (one thousand four hundred sixty-one).
- S.B. 1465 (one thousand four hundred sixty-five) as amended.
- **S.B. 1357** (one thousand three hundred fifty-seven), on motion of Senator Dunnavant, was passed by temporarily.

Senator Saslaw moved that the Rules be suspended and the third reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1338 (one thousand three hundred thirty-eight).
- S.B. 1162 (one thousand one hundred sixty-two).
- S.B. 1163 (one thousand one hundred sixty-three).
- S.B. 1197 (one thousand one hundred ninety-seven).
- S.B. 1210 (one thousand two hundred ten).
- S.B. 1212 (one thousand two hundred twelve).
- S.B. 1247 (one thousand two hundred forty-seven).
- S.B. 1258 (one thousand two hundred fifty-eight).
- S.B. 1275 (one thousand two hundred seventy-five).

```
S.B. 1295 (one thousand two hundred ninety-five).
```

S.B. 1307 (one thousand three hundred seven).

S.B. 1313 (one thousand three hundred thirteen).

S.B. 1336 (one thousand three hundred thirty-six).

S.B. 1365 (one thousand three hundred sixty-five).

S.B. 1375 (one thousand three hundred seventy-five).

S.B. 1389 (one thousand three hundred eighty-nine).

S.B. 1391 (one thousand three hundred ninety-one).

S.B. 1398 (one thousand three hundred ninety-eight).

S.B. 1405 (one thousand four hundred five).

S.B. 1420 (one thousand four hundred twenty).

S.B. 1425 (one thousand four hundred twenty-five).

S.B. 1429 (one thousand four hundred twenty-nine).

S.B. 1436 (one thousand four hundred thirty-six). S.B. 1461 (one thousand four hundred sixty-one).

S.B. 1465 (one thousand four hundred sixty-five).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

```
S.B. 1338 (one thousand three hundred thirty-eight).
```

S.B. 1162 (one thousand one hundred sixty-two).

S.B. 1163 (one thousand one hundred sixty-three).

S.B. 1197 (one thousand one hundred ninety-seven).

S.B. 1247 (one thousand two hundred forty-seven).

S.B. 1258 (one thousand two hundred fifty-eight).

S.B. 1275 (one thousand two hundred seventy-five). **S.B. 1307** (one thousand three hundred seven).

S.B. 1313 (one thousand three hundred thirteen).

S.B. 1336 (one thousand three hundred thirty-six).

S.B. 1365 (one thousand three hundred sixty-five).

S.B. 1375 (one thousand three hundred seventy-five).

S.B. 1389 (one thousand three hundred eighty-nine). **S.B. 1391** (one thousand three hundred ninety-one).

S.B. 1405 (one thousand four hundred five).

S.B. 1420 (one thousand four hundred twenty).

S.B. 1429 (one thousand four hundred twenty-nine).

S.B. 1436 (one thousand four hundred thirty-six).

S.B. 1461 (one thousand four hundred sixty-one).

S.B. 1465 (one thousand four hundred sixty-five).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1210 (one thousand two hundred ten), on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1212 (one thousand two hundred twelve), on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1295 (one thousand two hundred ninety-five), on motion of Senator DeSteph, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--2.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

NAYS--0.

RULE 36--McPike, Suetterlein--2.

S.B. 1398 (one thousand three hundred ninety-eight), on motion of Senator Norment, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--34.

NAYS--Chase, McDougle, Obenshain, Reeves, Suetterlein--5.

RULE 36--0.

S.B. 1425 (one thousand four hundred twenty-five), on motion of Senator Ruff, was passed with its title.

The recorded vote is as follows:

YEAS--28. NAYS--10. RULE 36--0.

YEAS--Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Edwards, Hanger, Kiggans, Locke, Lucas, Mason, McClellan, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart--28.

NAYS--Barker, Ebbin, Favola, Hashmi, Howell, Lewis, Marsden, McPike, Suetterlein, Surovell--10.

RULE 36--0.

S.B. 1384 (one thousand three hundred eighty-four) was taken up, substitute No. 1 offered by Senator Surovell having been agreed to on February 2, 2021.

RECONSIDERATION

Senator Survoell moved to reconsider the vote by which his substitute No. 1 to **S.B. 1384** (one thousand three hundred eighty-four) was agreed to.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

Senator Surovell withdrew substitute No. 1.

Senator Survivell offered an amendment in the nature of a substitute No. 3, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Chapter 43 of Title 2.2 an article numbered 7, consisting of sections numbered 2.2-4377.1 through 2.2-4377.5, relating to the Virginia Public Procurement Act; local arbitration agreements.

On motion of Senator Surovell, the reading of substitute No. 3 was waived.

On motion of Senator Surovell, substitute No. 3 was agreed to.

On motion of Senator Surovell, the bill was ordered to be engrossed and read by title the third time.

S.B. 1446 (one thousand four hundred forty-six) was read by title the second time and, on motion of Senator Surovell, was recommitted to the Committee on the Judiciary.

S.B. 1105 (one thousand one hundred five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Title 19.2 a chapter numbered 19.4, consisting of a section numbered 19.2-327.15, relating to post-conviction relief; previously admitted forensic scientific evidence.

The reading of the substitute was waived.

On motion of Senator Stanley, the substitute was agreed to.

The following amendment proposed by the Committee on Finance and Appropriations to the substitute was offered:

1. After line 109, substitute

insert

5. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendment was waived.

On motion of Senator Stanley, the amendment was agreed to.

On motion of Senator Stanley, the bill was ordered to be engrossed and read by title the third time.

S.B. 1113 (one thousand one hundred thirteen) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 18.2-60 of the Code of Virginia, relating to communicating threats of death or bodily injury to a person with intent to intimidate; penalty.

The reading of the substitute was waived.

On motion of Senator Spruill, the substitute was agreed to.

On motion of Senator Spruill, the bill was ordered to be engrossed and read by title the third time.

S.B. 1127 (one thousand one hundred twenty-seven) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-334.2, 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.31 of the Code of Virginia, relating to charitable gaming; conduct of instant bingo, network bingo, pull tabs, and seal cards.

The reading of the substitute was waived.

Senator Reeves moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-334.2, 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.34 of the Code of Virginia, relating to charitable gaming; conduct of instant bingo, network bingo, pull tabs, and seal cards.

The reading of the substitute was waived.

On motion of Senator Reeves, the substitute was agreed to.

Senator Reeves offered the following amendments to the substitute:

1. Line 145, substitute, after use

strike

the remainder of line 145 and through percent on line 146

insert

a predetermined percentage

2. After line 414, substitute

insert

3. That, except as otherwise provided in subdivision A 1 of § 18.2-340.19 of the Code of Virginia, as amended by this act, the Charitable Gaming Board shall not adjust the percentage of gross receipts, as set forth in 11VAC15-40-20 as it was in effect on January 1, 2021, that an organization with a charitable gaming permit must use for religious, charitable, community, or educational purposes or for certain real property expenses until a study is completed by a joint subcommittee of the Senate Committee on General Laws and Technology, the Senate Committee on Finance and Appropriations, the House Committee on General Laws, and the House Committee on Appropriations. The joint subcommittee shall have a total membership of eight members that shall consist of two members from each committee, to be appointed by the respective committee chairmen. In

conducting its study, the joint subcommittee shall analyze and make recommendations, as appropriate, regarding (i) the percentage of an organization's gross receipts that should be used for religious, charitable, community, or educational purposes for which the organization was chartered or organized and certain real property expenses; (ii) the regulatory oversight of charitable gaming in the Commonwealth, including the membership, structure, and necessity of the Charitable Gaming Board; and (iii) necessary safeguards and conflict of interest prohibitions on the Charitable Gaming Board. Administrative staff support for the joint subcommittee shall be provided by the Office of the Clerk of the Senate. Legal, research, policy analysis, and other services as requested by the joint subcommittee shall be provided by the Division of Legislative Services. All agencies of the Commonwealth shall provide assistance to the joint subcommittee for this study, upon request. The joint subcommittee shall complete its meetings by November 1, 2021, and report its findings to the General Assembly no later than the first day of the 2022 Regular Session of the General Assembly.

On motion of Senator Reeves, the reading of the amendments was waived.

On motion of Senator Reeves, the amendments were agreed to.

On motion of Senator Reeves, the bill was ordered to be engrossed and read by title the third time.

S.B. 1146 (one thousand one hundred forty-six) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.

The reading of the substitute was waived.

On motion of Senator Howell, the substitute was agreed to.

Senator Newman offered the following amendments to the substitute:

1. At the beginning of line 255, substitute

strike

\$50,000

insert

\$100,000

2. At the beginning of line 388, substitute

strike

\$50,000

insert

\$100,000

3. At the beginning of line 721 substitute

```
strike
                    $50,000
          insert
                    $100,000
     4. Line 752, substitute, after up to
          strike
                    $50,000
          insert
                    $100,000
On motion of Senator Newman, the amendments were passed by temporarily.
Senator McDougle offered the following amendments to the substitute:
     1. Line 6, substitute, Title, after reenact
          strike
                    şş
          insert
                    ş
     2. Line 6, substitute, Title, after 58.1-301
          strike
                    , 58.1-322.02, 58.1-322.03, and 58.1-402
     3. Line 10, substitute, after That
          strike
                    §§
          insert
                    §
     4. Line 10, substitute, after 58.1-301
          strike
                    , 58.1-322.02, 58.1-322.03, and 58.1-402
     5. Line 10, substitute, after Virginia
          strike
                    are
          insert
                    is
     6. Line 52, substitute, after corporations;
          insert
                    and
```

7. Line 54, substitute, after interest

strike

the remainder of line 54, all of lines 55, 56, 57, and through *assistance* on line 58

8. Line 62, substitute

strike

all of lines 62 through 754

On motion of Senator McDougle, the reading of the amendments was waived.

Senator McDougle moved that the amendments be agreed to.

The question was put on agreeing to the amendments.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--17. NAYS--22. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22. RULE 36--0.

The amendments were rejected.

On motion of Senator Newman, the reading of his amendments was waived.

On motion of Senator Newman, his amendments were agreed to.

On motion of Senator Newman, the bill was ordered to be engrossed and read by title the third time.

Senator Newman moved that the Rules be suspended and the third reading of the title of **S.B. 1146** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1146, on motion of Senator Newman, was passed with its title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- **S.B. 1196** (one thousand one hundred ninety-six) was read by title the second time and, on motion of Senator Locke, was ordered to be engrossed and read by title the third time.
 - S.B. 1201 (one thousand two hundred one) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660, relating to tax exemptions for energy storage systems.

The reading of the substitute was waived.

On motion of Senator Petersen, the substitute was agreed to.

On motion of Senator Petersen, the bill was ordered to be engrossed and read by title the third time.

S.B. 1226 (one thousand two hundred twenty-six) was read by title the second time.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. After line 66, introduced

insert

2. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendment was waived.

On motion of Senator Boysko, the amendment was agreed to.

On motion of Senator Boysko, the bill was ordered to be engrossed and read by title the third time.

S.B. 1250 (one thousand two hundred fifty) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 18.2-308.2:2, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to criminal history record information check required for firearm rentals; penalty.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

On motion of Senator Deeds, the bill was ordered to be engrossed and read by title the third time.

S.B. 1252 (one thousand two hundred fifty-two) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1 of the Code of Virginia, relating to sunset of coal tax credits.

The reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

On motion of Senator McPike, the bill was ordered to be engrossed and read by title the third time.

S.B. 1257 (one thousand two hundred fifty-seven) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 22.1-253.13:2 and 22.1-274 of the Code of Virginia, relating to Standards of Quality.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

S.B. 1261 (one thousand two hundred sixty-one) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-511, 8.01-36, 8.01-267.8, 8.01-383.1, 8.01-555, 8.01-626, 8.01-670, 8.01-671, 8.01-675.3, 8.01-676.1, 9.1-909, 15.2-1627, 15.2-1643, 15.2-2139, 15.2-2140, 15.2-2656, 15.2-3104, 15.2-3217, 15.2-3221, 15.2-3222, 15.2-3227, 15.2-3244, 15.2-3308, 15.2-3528, 15.2-3605, 15.2-3809, 15.2-3909, 15.2-4108, 15.2-4120, 15.2-5218, 15.2-5367, 15.2-6606, 15.2-6632, 15.2-7406, 16.1-279.1, 17.1-309, 17.1-400 through 17.1-403, 17.1-405 through 17.1-408, 17.1-410, 17.1-413, 17.1-503, 17.1-513, 18.2-308.08, 18.2-384, 19.2-152.10, 19.2-165, 19.2-321.1, 19.2-321.2, 19.2-322.1, 19.2-386.13, 19.2-402, 19.2-403, 19.2-404, 22.1-97, 22.1-289.024, as it shall become effective, 24.2-237, 24.2-422, 24.2-433, 25.1-239, 32.1-48.010, 32.1-48.013, 33.2-928, 33.2-2917, 37.2-920, 45.1-161.322, 55.1-1833, 55.1-1966, 55.1-2211, 57-2.02, 58.1-527, 58.1-1828, 58.1-2282, 58.1-3147, 58.1-3992, and 63.2-1710 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 26.1 of Title 8.01 sections numbered 8.01-675.5 and 8.01-675.6; and to repeal §§ 8.01-670.1 and 8.01-672 of the Code of Virginia, relating to the Court of Appeals; jurisdiction; number of judges.

The reading of the substitute was waived.

On motion of Senator Edwards, the substitute was agreed to.

The following amendment proposed by the Committee on Finance and Appropriations to the substitute was offered:

1. After line 2702, substitute

insert

5. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendment was waived.

On motion of Senator Edwards, the amendment was agreed to.

On motion of Senator Edwards, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1262** (one thousand two hundred sixty-two) was read by title the second time and, on motion of Senator Morrissey, was ordered to be engrossed and read by title the third time.
 - S.B. 1278 (one thousand two hundred seventy-eight) was read by title the second time.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 112, introduced, after of poker

strike

the remainder of line 112 and through tables, on line 113

2. Line 122, introduced, after competition

strike

the remainder of line 122 and through tables, on line 123

3. Line 257, introduced, after tournaments

insert

but at no time shall conduct more than three Texas Hold'em poker games or Texas Hold'em poker tournaments at a time

The reading of the amendments was waived.

On motion of Senator Bell, the amendments were agreed to.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. After line 334, introduced

insert

2. That the provisions of this act shall not become effective unless reenacted by the 2022 Session of the General Assembly.

The reading of the amendment was waived.

On motion of Senator Bell, the amendment was agreed to.

On motion of Senator Bell, the bill was ordered to be engrossed and read by title the third time.

S.B. 1282 (one thousand two hundred eighty-two) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding a section numbered 10.1-1307.04, relating to greenhouse gas emissions inventory.

The reading of the substitute was waived.

On motion of Senator Morrissey, the substitute was agreed to.

On motion of Senator Morrissey, the bill was ordered to be engrossed and read by title the third time.

S.B. 1284 (one thousand two hundred eighty-four) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 56-46.1, 56-585.1, 56-598, 56-601, 62.1-199, 67-103, 67-104, and 67-201 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 67-101.1; and to repeal §§ 67-101 and 67-102 of the Code of Virginia, relating to the Commonwealth Energy Policy.

The reading of the substitute was waived.

On motion of Senator Favola, the substitute was agreed to.

On motion of Senator Favola, the bill was ordered to be engrossed and read by title the third time.

S.B. 1289 (one thousand two hundred eighty-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 38.2-3407.15 of the Code of Virginia, relating to health insurance; carrier business practices; provider contracts.

The reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

On motion of Senator Surovell, the bill was ordered to be engrossed and read by title the third time.

S.B. 1301 (one thousand three hundred one) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding sections numbered 53.1-39.2 and 66-20.1, relating to correctional facilities; use of isolated confinement.

The reading of the substitute was waived.

On motion of Senator Morrissey, the substitute was agreed to.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

1. Line 36, substitute, after *confinement* strike

the remainder of line 36 and through hours on line 37

2. Line 43, substitute, after for

strike

no more than

3. Line 44, substitute, after hours,

insert

or longer if necessary to complete the investigation of an incident,

4. Line 54, substitute, after confinement

strike

the remainder of line 54 and through indicated on line 55

5. Line 91, substitute, after D.

strike

the remainder of line 91, all of lines 92 through 97, and through E. on line 98

6. At the beginning of line 100, substitute

strike

F

insert

Е

7. At the beginning of line 103, substitute

strike

G

insert

F

8. After line 176, substitute

insert

3. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendments was waived.

On motion of Senator Morrissey, the amendments were agreed to.

On motion of Senator Morrissey, the bill was ordered to be engrossed and read by title the third time.

S.B. 1310 (one thousand three hundred ten) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-3905, 40.1-2, 40.1-29, 40.1-49.3, and 40.1-49.8 of the Code of Virginia, relating to the employees providing domestic service; the Virginia Human Rights Act; application of laws applicable to employee safety and payment of wages.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

The following amendment proposed by the Committee on Finance and Appropriations to the substitute was offered:

1. After line 315, substitute insert

2. That the provisions of this act may result in a net increase in periods of imprisonment or commitment. Pursuant to § 30-19.1:4 of the Code of Virginia, the estimated amount of the necessary appropriation cannot be determined for periods of imprisonment in state adult correctional facilities; therefore, Chapter 1289 of the Acts of Assembly of 2020 requires the Virginia Criminal Sentencing Commission to assign a minimum fiscal impact of \$50,000. Pursuant to § 30-19.1:4 of the Code of Virginia, the estimated amount of the necessary appropriation cannot be determined for periods of commitment to the custody of the Department of Juvenile Justice.

The reading of the amendment was waived.

On motion of Senator McClellan, the amendment was agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

S.B. 1315 (one thousand three hundred fifteen) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 19.2-120, 19.2-163.03, and 19.2-299 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 16 of Title 19.2 a section numbered 19.2-271.6, relating to criminal proceedings; consideration of mental condition and intellectual and developmental disabilities.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

S.B. 1316 (one thousand three hundred sixteen) was read by title the second time.

The following amendments proposed by the Committee on Education and Health were offered:

1. Line 333, introduced, after dissemination;

insert

however, nothing in this subdivision shall be construed to prohibit the Commissioner of Social Services' representative from issuing written certifications regarding the results of a background check that was conducted before July 1, 2021, in accordance with subsection J of § 22.1-289.035 or § 22.1-289.039;

2. Line 585, introduced, after Department

insert

or its designee

3. Line 679, introduced, after Department

insert

or its designee

The reading of the amendments was waived.

On motion of Senator McClellan, the amendments were agreed to.

The following amendments proposed by the Committee on Finance and Appropriations were offered:

1. Line 705, introduced, after 3.

insert

That the provisions of §§ 19.2-389, as it shall become effective, 22.1-289.035, as it shall become effective, and 22.1-289.039, as it shall become effective, of the Code of Virginia, as amended by this act, shall become effective on January 1, 2022.

4.

2. At the beginning of line 738, introduced

strike

4

insert

5

The reading of the amendments was waived.

On motion of Senator McClellan, the amendments were agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

S.B. 1318 (one thousand three hundred eighteen) was read by title the second time.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 27, introduced, after the impact

insert

, both positive and negative,

2. Line 90, introduced, after least

strike

13

insert

15

3. Line 90, introduced, after consist of

strike

12

insert

14

4. Line 94, introduced, after Council

insert

, one representative of an environmental justice community or environmental justice advocacy group to be appointed by the Governor, one representative from the regulated community to be appointed by the Governor,

The reading of the amendments was waived.

Senator Hashmi moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 2.2-234 of the Code of Virginia and to amend the Code of Virginia by adding in Article 12 of Chapter 2 of Title 2.2 a section numbered 2.2-236, relating to environmental justice; interagency working group.

The reading of the substitute was waived.

On motion of Senator Hashmi, the substitute was agreed to.

On motion of Senator Hashmi, the bill was ordered to be engrossed and read by title the third time.

- $S.B.\ 1326$ (one thousand three hundred twenty-six), on motion of Senator Hanger, was passed by temporarily.
 - S.B. 1333 (one thousand three hundred thirty-three) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8 of the Code of Virginia, relating to pharmaceutical processors; cannabis products.

The reading of the substitute was waived.

Senator Lucas moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8 of the Code of Virginia, relating to pharmaceutical processors; cannabis products.

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

On motion of Senator Lucas, the bill was ordered to be engrossed and read by title the third time.

S.B. 1339 (one thousand three hundred thirty-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 9.1-128, 9.1-134, 17.1-502, and 19.2-392.1 through 19.2-392.4 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 1-229.1, by adding in Article 1 of Chapter 2 of Title 17.1 a section numbered 17.1-205.1, by adding sections numbered 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, and 19.2-392.4:1, and by adding in Chapter 23.1 of Title 19.2 sections numbered 19.2-392.5, 19.2-392.6, and 19.2-392.7, relating to expungement and sealing of police and court records; Expungement Fee Fund created; protection of public record information; penalties.

The reading of the substitute was waived.

Senator Surovell moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 9.1-128, 9.1-134, 17.1-502, and 19.2-392.1 through 19.2-392.4 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 1-229.1, by adding in Article 1 of Chapter 2 of Title 17.1 a section numbered 17.1-205.1, by adding sections numbered 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, and 19.2-392.4:1, and by adding in Chapter 23.1 of Title 19.2 sections numbered 19.2-392.5, 19.2-392.6, and 19.2-392.7, relating to expungement and sealing of police and court records; Expungement Fee Fund created; protection of public record information; penalties.

The reading of the substitute was waived.

Senator Surovell moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

Senator Surovell offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 9.1-128, 9.1-134, 17.1-502, and 19.2-392.1 through 19.2-392.4 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 1-229.1, by adding in Article 1 of Chapter 2 of Title 17.1 a section numbered 17.1-205.1, by adding sections numbered 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, and 19.2-392.4:1, and by adding in Chapter 23.1 of Title 19.2 sections numbered 19.2-392.5, 19.2-392.6, and 19.2-392.7, relating to expungement and sealing of police and court records; Expungement Fee Fund created; protection of public record information; penalties.

On motion of Senator Surovell, the reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

On motion of Senator Surovell, the bill was ordered to be engrossed and read by title the third time.

S.B. 1341 (one thousand three hundred forty-one) was read by title the second time.

Senator Barker offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 38.2-3521.1 of the Code of Virginia, relating to health insurance; association health plan for real estate salespersons.

On motion of Senator Barker, the reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

On motion of Senator Barker, the bill was ordered to be engrossed and read by title the third time.

S.B. 1350 (one thousand three hundred fifty) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Transportation was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 33.2-214.2 and 33.2-353 of the Code of Virginia, relating to transportation projects; resiliency.

The reading of the substitute was waived.

On motion of Senator Lewis, the substitute was agreed to.

On motion of Senator Lewis, the bill was ordered to be engrossed and read by title the third time.

S.B. 1362 (one thousand three hundred sixty-two) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to require employers to report outbreaks of COVID-19; emergency.

The reading of the substitute was waived.

On motion of Senator Lewis, the substitute was agreed to.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

1. Line 18, substitute, after section,

insert

"employer" has the same meaning as defined in § 40.1-2 of the Code of Virginia,

2. After line 27, substitute

insert

4. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendments was waived.

On motion of Senator Lewis, the amendments were agreed to.

On motion of Senator Lewis, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1363** (one thousand three hundred sixty-three), on motion of Senator Marsden, was passed by temporarily.
- **S.B. 1380** (one thousand three hundred eighty) was read by title the second time and, on motion of Senator Lucas, was ordered to be engrossed and read by title the third time.
 - S.B. 1382 (one thousand three hundred eighty-two) was read by title the second time.

Senator Favola offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-308.09, 18.2-308.2:1, as it is currently effective and as it shall become effective, 18.2-308.2:2, as it is currently effective and as it shall become effective, 18.2-308.2:3, as it is currently effective and as it shall become effective, and 19.2-386.28 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-308.1:8, relating to purchase, possession, or transportation of firearms following conviction for assault and battery of a family or household member; penalties.

On motion of Senator Favola, the reading of the substitute was waived.

On motion of Senator Favola, the substitute was agreed to.

On motion of Senator Favola, the bill was ordered to be engrossed and read by title the third time.

S.B. 1392 (one thousand three hundred ninety-two) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-581, relating to Consumer Data Protection Act.

The reading of the substitute was waived.

On motion of Senator Marsden, the substitute was agreed to.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

```
1. Line 467, substitute strike
all of lines 467 and 468 and through 3. on line 469 insert
2.
```

2. At the beginning of line 472, substitute

strike

4.

insert

3.

The reading of the amendments was waived.

On motion of Senator Marsden, the amendments were agreed to.

On motion of Senator Marsden, the bill was ordered to be engrossed and read by title the third time.

S.B. 1395 (one thousand three hundred ninety-five) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Privileges and Elections was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 24.2-105, as it shall become effective, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 24.2-104.1, by adding a section numbered 24.2-1005.2, by adding in Article 5 of Chapter 2 of Title 24.2 a section numbered 24.2-224.1, and by adding in Title 24.2 a chapter numbered 1.1, consisting of sections numbered 24.2-125 through 24.2-130; and to repeal § 24.2-124, as it shall become effective, of the Code of Virginia, relating to elections; prohibited discrimination in voting and elections administration; required process for enacting certain covered practices; civil causes of action; penalties.

The reading of the substitute was waived.

Senator McClellan moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 24.2-105, as it shall become effective, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 24.2-104.1, by adding a section numbered 24.2-1005.2, and by adding in Title 24.2 a chapter numbered 1.1, consisting of sections numbered 24.2-125 through 24.2-131; and to repeal § 24.2-124, as it shall become effective, of the Code of Virginia, relating to elections; prohibited discrimination in voting and elections administration; required process for enacting certain covered practices; civil causes of action; penalties.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1396** (one thousand three hundred ninety-six) was read by title the second time and, on motion of Senator Hashmi, was ordered to be engrossed and read by title the third time.
- **S.B. 1401** (one thousand four hundred one) was read by title the second time and, on motion of Senator Pillion, was ordered to be engrossed and read by title the third time.
 - S.B. 1406 (one thousand four hundred six) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101, 4.1-101.01, 4.1-101.04, 101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-102, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-122, 4.1-124, as it is currently effective and as it shall become effective,

4.1-128, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-201.1, as it is currently effective and as it shall become effective, 4.1-202, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29 consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6 consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, consisting of chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding a section numbered 19.2-392.2:1, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

The reading of the substitute was waived.

Senator Lucas moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 101.07, 4.1-101.09, 4.1-101.01, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118,

4.1-119, as it is currently effective and as it shall become effective, 4.1-122, 4.1-124, as it is currently effective and as it shall become effective, 4.1-128, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-202, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29, consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6, consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, containing chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Chapter 7 of Title 18.2 an article numbered 1.4, consisting of sections numbered 18.2-265.22 through 18.2-265.28, by adding a section numbered 19.2-392.2:1, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

1. At the beginning of line 1985, Judiciary substitute strike

"Intoxicated" means a condition in which a person has drunk enough alcoholic beverages to

insert

"Intoxicated" means a condition in which a person has drunk enough alcoholic beverages to

2. Line 6643, Judiciary substitute, after *to* insert

the Department of Behavioral Health and Developmental Services, which shall distribute such appropriated funds to community services boards for the purpose of administering

3. Line 6646, Judiciary substitute, after under §

strike

4.1-1003

insert

4.1-1004

4. Line 7528, Judiciary substitute, after *locality* strike

that has by referendum authorized the operation of retail marijuana stores

5. After line 13858, Judiciary substitute insert

23. That there shall be established a Cannabis Oversight Commission (the Commission), which shall consist of 18 members of the General Assembly as follows: three members appointed by the Chairman of the Senate Committee on Finance and Appropriations; three members appointed by the Chairman of the Senate Committee on the Judiciary; three members appointed by the Chairman of the Senate Committee on Rehabilitation and Social Services; three members appointed by the Chairman of the House Committee on Appropriations; three members appointed by the Chairman of the House Committee for Courts of Justice; and three members appointed by the Chairman of the House Committee on General Laws. The Commission shall exercise the function of overseeing the implementation of the provisions of this act and shall convene regularly in the exercise of that function. The Virginia Cannabis Control Authority (the Authority) shall report to the Commission at the Commission's request. The Commission shall expire on January 1, 2024. The provisions of this enactment shall become effective in due course.

24. That it shall be the duty of the officers conducting the November 2021 general election, at the places appointed for holding the same, to open a poll and take the sense of the qualified voters on the question stated below.

QUESTION: "Should the state legalize the sale of recreational marijuana from privately licensed retailers, wholesalers, and growers for use by adults?"

The ballots shall be prepared, distributed, and voted, and the results thereof ascertained and certified, in the manner prescribed by Title 24.2 of the Code of Virginia, and information on the proposed question shall be prepared and distributed in accordance with the provisions of § 30-19.10 of the Code of Virginia.

The expenses incurred in conducting this referendum shall be defrayed as in the case of other November general elections.

The results of the referendum shall be advisory only.

The provisions of this enactment shall become effective in due course.

The reading of the amendments was waived.

On motion of Senator Lucas, the amendments were agreed to.

On motion of Senator Lucas, the bill was ordered to be engrossed and read by title the third time.

-415-

S.B. 1418 (one thousand four hundred eighteen) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 2.2-115 of the Code of Virginia, relating to grants from the Commonwealth's Development Opportunity Fund; waiver or reduction of capital investment and local match requirements.

The reading of the substitute was waived.

On motion of Senator McPike, the substitute was agreed to.

On motion of Senator McPike, the bill was ordered to be engrossed and read by title the third time.

S.B. 1428 (one thousand four hundred twenty-eight) was read by title the second time.

The following amendments proposed by the Committee on Rehabilitation and Social Services were offered:

```
1. Line 15, introduced, after of
     strike
               alcoholic beverages, other than beer and
     insert
               spirits,
2. Line 15, introduced, after wine
     strike
               not
3. Line 16, introduced, after wineries
     strike
               , and
     insert
               , [the comma]
4. Line 16, introduced, after coolers
     strike
               not
5. Line 110, introduced, after of
     strike
               alcoholic beverages, other than beer and
    insert
               spirits,
6. Line 110, introduced, after wine
```

strike

not

```
7. Line 111, introduced, after wineries
     strike
               , and
     insert
               , [the comma]
8. Line 111, introduced, after coolers
     strike
9. Line 205, introduced, after of
     strike
               alcoholic beverages, other than beer and
     insert
               spirits,
10. Line 205, introduced, after wine
     strike
11. Line 206, introduced, after wineries
     strike
               , and
     insert
               , [the comma]
12. Line 206, introduced, after coolers
     strike
```

The reading of the amendments was waived.

not

On motion of Senator Locke, the amendments were agreed to.

On motion of Senator Locke, the bill was ordered to be engrossed and read by title the third time.

S.B. 1442 (one thousand four hundred forty-two) was read by title the second time.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

- 1. After line 40, introduced insert
 - 2. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendment was waived.

On motion of Senator Morrissey, the amendment was agreed to.

On motion of Senator Morrissey, the bill was ordered to be engrossed and read by title the third time.

S.B. 1443 (one thousand four hundred forty-three) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 3.2-4212, 4.1-302, 16.1-253.2, 18.2-36.1, 18.2-36.2, 18.2-46.3:3, 18.2-51.1, 18.2-53.1,18.2-57, 18.2-60.4, 18.2-61, 18.2-67.1, 18.2-67.2, 18.2-67.5:2, 18.2-67.5:3, 18.2-154, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5, 18.2-255, 18.2-255.2, 18.2-270, 18.2-308.1, 18.2-308.2, 18.2-308.2:2, as it is currently effective and as it shall become effective, 18.2-308.4, 18.2-374.1, 18.2-374.1:1, 18.2-374.3, 19.2-297.1, 46.2-341.28, 46.2-357, 46.2-391, 46.2-865.1, and 53.1-203 of the Code of Virginia, relating to elimination of mandatory minimum sentences; modification of sentence to mandatory minimum term of confinement for felony offenses.

The reading of the substitute was waived.

On motion of Senator Edwards, the substitute was agreed to.

Senator Edwards offered the following amendment to the substitute:

1. Line 122, substitute, after 18.2-46.3 strike

upon any property listed in this section

insert

(i) upon the property, including buildings and grounds, of any public or private elementary, secondary, or postsecondary school or institution of higher education; (ii) upon public property or any property open to public use within 1,000 feet of such school property; (iii) on any school bus as defined in § 46.2-100; or (iv) upon the property, including buildings and grounds, of any publicly owned or operated community center or any publicly owned or operated recreation center

On motion of Senator Edwards, the reading of the amendment was waived.

On motion of Senator Edwards, the amendment was agreed to.

On motion of Senator Edwards, the bill was ordered to be engrossed and read by title the third time.

S.B. 1462 (one thousand four hundred sixty-two) was read by title the second time.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. After line 46, introduced

insert

3. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendment was waived.

On motion of Senator Mason, the amendment was agreed to.

On motion of Senator Mason, the bill was ordered to be engrossed and read by title the third time.

S.B. 1469 (one thousand four hundred sixty-nine) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 2.2-212 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-507.3, by adding in Chapter 22 of Title 2.2 an article numbered 12, consisting of sections numbered 2.2-2365 through 2.2-2376, and by adding in Article 3.1 of Chapter 1 of Title 51.1 a section numbered 51.1-124.40, relating to establishing an Opioid Abatement Authority.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

Senator Barker offered the following amendment to the substitute:

1. Line 104, substitute, after *term*. insert

The members identified in clause (ii) shall serve without voting privileges.

On motion of Senator Barker, the reading of the amendment was waived.

On motion of Senator Barker, the amendment was agreed to.

On motion of Senator Barker, the bill was ordered to be engrossed and read by title the third time.

S.B. 1471 (one thousand four hundred seventy-one) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 4.1-206, 4.1-206.3, as it shall become effective, 4.1-231, 4.1-231.1, as it shall become effective, 4.1-233, and 4.1-233.1, as it shall become effective, of the Code of Virginia, relating to alcoholic beverage control; local special events license.

The reading of the substitute was waived.

On motion of Senator Dunnavant, the substitute was agreed to.

On motion of Senator Dunnavant, the bill was ordered to be engrossed and read by title the third time.

S.B. 1326 (one thousand three hundred twenty-six) was taken up and read by title the second time.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. Line 66, introduced, after board.

insert

However, if such locality, on or before January 31, 2021, enacted a cigarette tax ordinance or adopted a resolution indicating its intent to impose a cigarette tax, such locality may impose a tax under this article without being a member of a regional cigarette tax board until January 1, 2026, at which time such locality may impose a tax under this article only if it is a member of a regional cigarette tax board.

The reading of the amendment was waived.

Senator Hanger moved that the amendment be rejected.

The question was put on agreeing to the amendment.

The amendment was rejected.

Senator Hanger offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 58.1-3830 of the Code of Virginia and to amend the Code of Virginia by adding in Article 7 of Chapter 38 of Title 58.1 a section numbered 58.1-3832.1, relating to local cigarette taxes; regional cigarette tax boards.

On motion of Senator Hanger, the reading of the substitute was waived.

On motion of Senator Hanger, the substitute was agreed to.

On motion of Senator Hanger, the bill was ordered to be engrossed and read by title the third time.

S.B. 1363 (one thousand three hundred sixty-three) was taken up and read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 53.1-2, 53.1-4, and 53.1-5 of the Code of Virginia and to amend the Code of Virginia by adding in Article 2 of Chapter 1 of Title 53.1 a section numbered 53.1-7.1, relating to State Board of Local and Regional Jails; membership; powers and duties.

The reading of the substitute was waived.

Senator Marsden moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 53.1-2, 53.1-4, and 53.1-5 of the Code of Virginia, relating to State Board of Local and Regional Jails; membership; powers and duties.

The reading of the substitute was waived.

On motion of Senator Marsden, the substitute was agreed to.

On motion of Senator Marsden, the bill was ordered to be engrossed and read by title the third time.

RECONSIDERATION

Senator Dunnavant moved to reconsider the vote by which **S.B. 1250** (one thousand two hundred fifty) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

Senator Deeds moved that **S.B. 1250** be ordered to be engrossed and read by title the third time.

Senator Dunnavant moved, as a substitute motion, that **S.B. 1250** be recommitted to the Committee on the Judiciary.

The question was put on recommitting S.B. 1250 to the Committee on the Judiciary.

The yeas and nays were called for by the Chair.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Hashmi, Kiggans, Lewis, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--21.

NAYS--Barker, Bell, Deeds, Ebbin, Edwards, Favola, Howell, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--18.

RULE 36--0.

- S.B. 1250 was recommitted to the Committee on the Judiciary.
- **S.B.** 1357 (one thousand three hundred fifty-seven) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to direct the Board of Education to establish a through-year growth assessment system.

The reading of the substitute was waived.

Senator Dunnavant moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

1. Line 10, substitute, after in

strike

each core subject

insert

English and mathematics

2. After line 13, substitute

insert

2. That the provisions of this act shall not become effective unless an appropriation effectuating the purposes of this act is included in a general appropriation act passed in 2021 by the General Assembly that becomes law.

The reading of the amendments was waived.

Senator Dunnavant moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

Senator Dunnavant offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to direct the Board of Education to establish a through-year growth assessment system.

On motion of Senator Dunnavant, the reading of the substitute was waived.

On motion of Senator Dunnavant, the substitute was agreed to.

On motion of Senator Dunnavant, the bill was ordered to be engrossed and read by title the third time.

SENATE BILLS ON FIRST READING

- **S.B. 1209** (one thousand two hundred nine) was read by title the first time.
- S.B. 1360 (one thousand three hundred sixty) was read by title the first time.

SENATE RESOLUTION ON THIRD READING

S.R. 95 (ninety-five) was read by title the third time and, on motion of Senator Suetterlein, was agreed to.

MEMORIAL RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following House joint resolutions were taken up and agreed to en bloc by a unanimous standing vote:

H.J.R. 623 (six hundred twenty-three).

H.J.R. 624 (six hundred twenty-four).

H.J.R. 625 (six hundred twenty-five).

H.J.R. 627 (six hundred twenty-seven).

H.J.R. 633 (six hundred thirty-three).

H.J.R. 631 (six hundred thirty-one) was taken up, as follows:

HOUSE JOINT RESOLUTION NO. 631

Celebrating the life of the Honorable Bernard S. Cohen.

WHEREAS, the Honorable Bernard S. Cohen, a former member of the Virginia House of Delegates who had previously served as co-counsel for the appellants in the historic *Loving v. Virginia* case, in which the Supreme Court of the United States struck down laws banning interracial marriage, died on October 12, 2020; and

WHEREAS, the child of Jewish immigrants from Eastern Europe, Bernard "Bernie" S. Cohen grew up in Brooklyn and graduated from the City College of New York; he subsequently worked for the United States Department of Labor while studying to earn a law degree from Georgetown University; and

WHEREAS, Bernie Cohen was working as a volunteer attorney for the American Civil Liberties Union when he met Richard and Mildred Loving, who had been arrested under Virginia's anti-miscegenation laws and been given a suspended jail sentence on the condition that they leave the Commonwealth and not return for a period of 25 years; and

WHEREAS, during the *Loving v. Virginia* case, Bernie Cohen and co-counsel Philip Hirschkop argued that laws prohibiting interracial marriage violated the due process and equal protection clauses of the Fourteenth Amendment to the Constitution of the United States, and in 1967, the Supreme Court of the United States ruled unanimously in favor of the Lovings; and

WHEREAS, the landmark case was a significant victory for the civil rights movement in the United States, and Bernie Cohen's contributions have been immortalized in numerous books, documentaries, and dramatizations, including the 2016 major motion picture *Loving*; and

WHEREAS, desirous to be of further service to the Commonwealth, Bernie Cohen ran for and was elected to the Virginia House of Delegates in 1979; he represented residents of the 21st District until 1983, when he took office as the delegate for the 46th District, representing parts of Alexandria; and

WHEREAS, during his tenure as a state lawmaker, Bernie Cohen offered his expertise to several standing committees and introduced many important pieces of legislation, including a bill to impose restrictions on indoor smoking and measures to ensure that terminally ill patients would have the right to decline life-prolonging treatments; and

WHEREAS, Bernie Cohen introduced a bill to decriminalize same-sex marriage in the Commonwealth, which was unsuccessful, but the decision in *Loving v. Virginia* was ultimately cited as a precedent in the Supreme Court's decision in another landmark case, *Obergefell v. Hodges* (2015), which guaranteed same-sex couples the right to marry throughout the nation; and

WHEREAS, Bernie Cohen continued to practice law in Alexandria for many years, specializing in environmental and employment cases until his well-earned retirement in 2006; and

WHEREAS, a man of great integrity, Bernie Cohen served the Commonwealth with the utmost dedication and distinction as both an attorney and a state legislator; and

WHEREAS, Bernie Cohen will be fondly remembered and greatly missed by his wife of 61 years, Rae Rose; his children, Bennett and Karen, and their families; and numerous other family members and friends; now, therefore, be it

RESOLVED by the House of Delegates, the Senate concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Bernard S. Cohen, a highly admired attorney and a dedicated public servant who touched the lives of generations of Americans through his contributions to civil rights; and, be it

RESOLVED FURTHER, That the Clerk of the House of Delegates prepare a copy of this resolution for presentation to the family of the Honorable Bernard S. Cohen as an expression of the General Assembly's respect for his memory.

H.J.R. 631, on motion of Senator Ebbin, was agreed to by a unanimous standing vote.

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions and resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions and resolutions were taken up, ordered to be engrossed, and agreed to en bloc by a unanimous standing vote:

```
S.J.R. 333 (three hundred thirty-three).
```

S.J.R. 335 (three hundred thirty-five).

S.J.R. 337 (three hundred thirty-seven).

S.J.R. 338 (three hundred thirty-eight).

S.J.R. 339 (three hundred thirty-nine).

S.J.R. 342 (three hundred forty-two).

S.J.R. 343 (three hundred forty-three).

S.J.R. 345 (three hundred forty-five).

S.J.R. 348 (three hundred forty-eight).

S.J.R. 349 (three hundred forty-nine).

S.J.R. 351 (three hundred fifty-one).

S.J.R. 352 (three hundred fifty-two).

S.J.R. 353 (three hundred fifty-three).

S.J.R. 354 (three hundred fifty-four). S.J.R. 355 (three hundred fifty-five).

S.J.R. 356 (three hundred fifty-six).

S.J.R. 360 (three hundred sixty).

S.J.R. 364 (three hundred sixty-four).

S.J.R. 368 (three hundred sixty-eight).

S.J.R. 369 (three hundred sixty-nine).

- S.J.R. 375 (three hundred seventy-five).
- S.J.R. 377 (three hundred seventy-seven).
- S.J.R. 386 (three hundred eighty-six).
- S.J.R. 391 (three hundred ninety-one).
- S.J.R. 394 (three hundred ninety-four).
- **S.J.R. 397** (three hundred ninety-seven).
- S.R. 106 (one hundred six).
- S.R. 107 (one hundred seven).
- S.R. 108 (one hundred eight).
- S.R. 110 (one hundred ten).
- **S.R. 112** (one hundred twelve).
- S.R. 115 (one hundred fifteen).
- **S.R. 116** (one hundred sixteen).
- S.J.R. 316 (three hundred sixteen) was taken up.

Senator Deeds offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

Celebrating the lives of Walter Lowrie Martin III and Edith Luke Martin.

On motion of Senator Deeds, the reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

- **S.J.R. 316**, on motion of Senator Deeds, was ordered to be engrossed and was agreed to by a unanimous standing vote.
 - **S.J.R. 378** (three hundred seventy-eight) was taken up, as follows:

SENATE JOINT RESOLUTION NO. 378

Celebrating the life of the Honorable Robert Lathan Calhoun.

WHEREAS, the Honorable Robert Lathan Calhoun, esteemed attorney, member of the Senate of Virginia from 1988 to 1996, and beloved member of the Alexandria community, died on August 6, 2020; and

WHEREAS, Robert "Bob" Calhoun was born in Illinois into a family that valued political engagement; he was inspired throughout his life by his grandmother, who was active in state issues, and followed in his father's footsteps by attending Yale Law School; and

WHEREAS, Bob Calhoun practiced law with various firms and in 1973 made his first foray into politics; naturally shy, but with an aptitude for public speaking, he lost his first race to Jim Thomson, majority leader in the House of Delegates, but he would be undeterred in his pursuit of public office; and

WHEREAS, Bob Calhoun's intelligence, practicality, and passion soon won him a seat on the Alexandria City Council in 1976; despite being the lone Republican on the council at the time, he would become vice mayor only three years later and help more Republicans win seats to the council; and

WHEREAS, during his time on the Alexandria City Council, Bob Calhoun promoted the development of the Torpedo Factory Art Center, now a cornerstone of Alexandria, facilitated the creation of the Covanta Waste to Energy Plan, and was involved in a preliminary study of the DASH bus system; and

WHEREAS, from the Alexandria City Council, Bob Calhoun went on to serve in the Senate of Virginia, where he was known for both his sharp wit and his willingness to work with colleagues from across the aisle, representing the 30th District from 1988 to 1996; and

WHEREAS, keen on transportation issues, one of Bob Calhoun's many accomplishments was his co-sponsorship of the Public-Private Transportation Act of 1995, which supported development of the Washington Metropolitan Area Transit Authority's Silver Line to Dulles International Airport; and

WHEREAS, an avid reader with a passion for the arts and the outdoors, Bob Calhoun charmed innumerable friends and colleagues with the wisdom and experience he had acquired over the years; and

WHEREAS, Bob Calhoun's involvement in the community built on his interests and extended beyond politics; he co-chaired the fundraising campaign to found the Rachel M. Schlesinger Concert Hall and Arts Center, home of the Alexandria Symphony Orchestra, and assisted in establishing the Scholarship Fund of Alexandria; and

WHEREAS, Bob Calhoun reminded all to think critically and take an active role in bringing about the world we want to see, encouraging everyone to get involved, strive for goals, and remain active in the community; and

WHEREAS, guided throughout his life by his deep and abiding faith, Bob Calhoun enjoyed worship and fellowship with his community at Westminster Presbyterian Church in Alexandria, where he was a member for many years and popular speaker in the church's congregational life programs; and

WHEREAS, Bob Calhoun will be fondly remembered and dearly missed by his loving wife, Sandra; his children, Andrew, Rob, and Gordon, and their families; and numerous other family members, friends, and colleagues on both sides of the aisle; now, therefore, be it

RESOLVED by the Senate, the House of Delegates concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Robert Lathan Calhoun, former member of the Senate of Virginia, whose unwavering dedication to his community and the Commonwealth touched many lives; and, be it

RESOLVED FURTHER, That the Clerk of the Senate prepare a copy of this resolution for presentation to the family of the Honorable Robert Lathan Calhoun as an expression of the General Assembly's respect for his memory.

S.J.R. 378, on motion of Senator Ebbin, was ordered to be engrossed and was agreed to by a unanimous standing vote.

S.J.R. 380 (three hundred eighty) was taken up as follows:

SENATE JOINT RESOLUTION NO. 380

Celebrating the life of the Honorable Jerry M. Wood.

WHEREAS, the Honorable Jerry M. Wood, a trusted pharmacist who touched countless lives throughout his career and a respected public servant who was elected to the Warrenton Town Council and the House of Delegates, died on January 8, 2021; and

WHEREAS, a native of Roanoke, Jerry Wood attended Andrew Lewis High School in Salem and subsequently graduated from what is now the Virginia Commonwealth University (VCU) School of Pharmacy; he also served his country for eight years as a member of the United States Navy Reserve; and

WHEREAS, Jerry Wood worked at pharmacies in Fredericksburg and Culpeper, then joined the staff of Rhodes Drug Store in Warrenton in 1968; and

WHEREAS, Jerry Wood established Fauquier Pharmacy in Warrenton in 1972 and served the community from its Main Street location for 20 years, earning a reputation for his commitment to customer service and his devotion to serving young people in the community; and

WHEREAS, Jerry Wood sold Fauquier Pharmacy to Rite Aid in 1992 and continued to work as a family pharmacist for the company until his retirement in 2005; and

WHEREAS, desirous to be of further service to the Commonwealth, Jerry Wood ran for and was elected to the House of Delegates in 1991 and served the residents of the 31st District with dedication and integrity for one term; and

WHEREAS, during his time as a state lawmaker, Jerry Wood helped established a Fauquier County campus of Lord Fairfax Community College and supported the development of pari-mutuel horse racing in the Commonwealth; and

WHEREAS, Jerry Wood continued his career in public service as a member of the Warrenton Town Council from 2014 to 2020 and represented the Warrenton community as a member of the Rappahannock-Rapidan Regional Commission; and

WHEREAS, Jerry Wood offered his wisdom and expertise to the Virginia Board of Pharmacy and the Virginia Board of Health Professions, and he volunteered his time and leadership with the local American Legion post, Rotary club, and Optimist club and several other civic organizations; and

WHEREAS, among many awards and accolades for his personal and professional achievements, Jerry Wood received a Pharmacy Alumni Award from VCU in 2013; and

WHEREAS, guided by his deep and abiding faith throughout his life, Jerry Wood enjoyed fellowship and worship with the congregation of St. James' Episcopal Church; and

WHEREAS, Jerry Wood will be fondly remembered and greatly missed by his wife of 29 years, Coleen; his children, Gregory, Christian, Julie, Laura, Brandon, and Ashley, and their families; and numerous other family members and friends; now, therefore, be it

RESOLVED by the Senate, the House of Delegates concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Jerry M. Wood, a highly admired medical professional and public servant in Warrenton; and, be it

RESOLVED FURTHER, That the Clerk of the Senate prepare a copy of this resolution for presentation to the family of the Honorable Jerry M. Wood as an expression of the General Assembly's respect for his memory.

- S.J.R. 380, on motion of Senator Deeds, was ordered to be engrossed and was agreed to by a unanimous standing vote.
 - S.J.R. 393 (three hundred ninety-three), on motion of Senator Lewis, was passed by for the day.
 - **S.R. 104** (one hundred four) was taken up as follows:

SENATE RESOLUTION NO. 104

Celebrating the life of the Honorable Leslie D. Campbell, Jr.

WHEREAS, Leslie D. Campbell, Jr., a respected attorney and public servant who represented the residents of the 29th District in the Senate of Virginia for three terms, died on December 26, 2020; and

WHEREAS, a proud, lifelong resident of Hanover County, Leslie "Les" Campbell learned the value of hard work and responsibility at a young age, helping to tend to his family's farm and care for his siblings; and

WHEREAS, Les Campbell joined many of the other young men of his generation in service to the nation as a member of the United States Navy during World War II; after his honorable military service, he graduated from Randolph-Macon College and the T.C. Williams School of Law at the University of Richmond; and

WHEREAS, Les Campbell went on to serve the community as an attorney for more than 50 years and provided his expertise as a substitute judge of the 15th Judicial Circuit of Virginia; he was elected as Hanover County's attorney for the Commonwealth in 1955 and served in that capacity until 1963; and

WHEREAS, desirous to be of further service, Les Campbell ran for and was elected to the Senate of Virginia and ably represented the residents of the 29th District with the utmost integrity and dedication from 1964 to 1976; and

WHEREAS, during his tenure as a state lawmaker, Les Campbell introduced and supported many important pieces of legislation to benefit all Virginians and offered his leadership and insights to the Transportation Committee and the Finance Committee; he also helped young people engage with state government as chair of the YMCA Model General Assembly program; and

WHEREAS, Les Campbell served on the boards of directors of Hanover National Bank and First Virginia Bank and was a member of local Kiwanis and Ruritan clubs, as well as many other civic and service organizations; he also served on a committee to revitalize the Camptown Races held on Meadow Farm near Doswell; and

WHEREAS, Les Campbell was a talented athlete throughout his life; he played football at Randolph-Macon College and for a semi-professional team in Richmond, and he also took up tennis, sailing, and catamaran racing, winning a national catamaran championship in 1969; and

WHEREAS, Les Campbell was proud of his Scottish heritage and built his home "Lochland" overlooking Campbell Lake; he hosted community ice skating parties and was an avid gardener who imparted his love of horticulture to his children; and

WHEREAS, Les Campbell will be fondly remembered and greatly missed by his children, Sally, Mary Scott, Dee, and Jenny, and their families; his former wife, Eleanor; and numerous other family members, friends, and colleagues on both sides of the aisle; now, therefore, be it

RESOLVED, That the Senate of Virginia hereby note with great sadness the loss of the Honorable Leslie D. Campbell, Jr., a highly admired attorney and public servant who made many contributions to Hanover County and the Commonwealth; and, be it

RESOLVED FURTHER, That the Clerk of the Senate prepare a copy of this resolution for presentation to the family of the Honorable Leslie D. Campbell, Jr., as an expression of the Senate of Virginia's respect for his memory.

S.R. 104, on motion of Senator McDougle, was ordered to be engrossed and was agreed to by a unanimous standing vote.

S.R. 111 (one hundred eleven) was taken up.

Senator Hashmi offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

Celebrating the life of Pauline Allen Mitchell.

On motion of Senator Hashmi, the reading of the substitute was waived.

On motion of Senator Hashmi, the substitute was agreed to.

S.R. 111, on motion of Senator Hashmi, was ordered to be engrossed and was agreed to by a unanimous standing vote.

COMMENDING RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following House joint resolutions were taken up and agreed to en bloc:

H.J.R. 615 (six hundred fifteen).

H.J.R. 617 (six hundred seventeen).

H.J.R. 619 (six hundred nineteen).

H.J.R. 620 (six hundred twenty).

H.J.R. 621 (six hundred twenty-one).

H.J.R. 622 (six hundred twenty-two).

H.J.R. 630 (six hundred thirty).

H.J.R. 632 (six hundred thirty-two).

H.J.R. 634 (six hundred thirty-four).

H.J.R. 635 (six hundred thirty-five).

```
H.J.R. 636 (six hundred thirty-six).
```

H.J.R. 637 (six hundred thirty-seven).

H.J.R. 638 (six hundred thirty-eight).

On motion of Senator Saslaw, the questions on agreeing to the Senate joint resolutions and resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate joint resolutions and resolutions were taken up, ordered to be engrossed, and agreed to en bloc:

```
S.J.R. 332 (three hundred thirty-two).
```

S.J.R. 334 (three hundred thirty-four).

S.J.R. 336 (three hundred thirty-six).

S.J.R. 340 (three hundred forty).

S.J.R. 341 (three hundred forty-one).

S.J.R. 344 (three hundred forty-four).

S.J.R. 346 (three hundred forty-six).

S.J.R. 347 (three hundred forty-seven).

S.J.R. 350 (three hundred fifty).

S.J.R. 357 (three hundred fifty-seven).

S.J.R. 358 (three hundred fifty-eight).

S.J.R. 359 (three hundred fifty-nine).

S.J.R. 361 (three hundred sixty-one).

S.J.R. 362 (three hundred sixty-two).

S.J.R. 363 (three hundred sixty-three).

S.J.R. 365 (three hundred sixty-five).

S.J.R. 366 (three hundred sixty-six).

S.J.R. 367 (three hundred sixty-seven).

S.J.R. 370 (three hundred seventy).

S.J.R. 371 (three hundred seventy-one).

S.J.R. 373 (three hundred seventy-three).

S.J.R. 374 (three hundred seventy-four).

S.J.R. 376 (three hundred seventy-six).

S.J.R. 379 (three hundred seventy-nine).

S.J.R. 381 (three hundred eighty-one).

S.J.R. 382 (three hundred eighty-two).

S.J.R. 383 (three hundred eighty-three).

S.J.R. 384 (three hundred eighty-four).

S.J.R. 385 (three hundred eighty-five).

S.J.R. 387 (three hundred eighty-seven).

S.J.R. 388 (three hundred eighty-eight).

S.J.R. 389 (three hundred eighty-nine).

S.J.R. 390 (three hundred ninety).

S.J.R. 392 (three hundred ninety-two).

S.J.R. 395 (three hundred ninety-five).

S.J.R. 396 (three hundred ninety-six).

S.J.R. 398 (three hundred ninety-eight).

S.R. 105 (one hundred five).

S.R. 109 (one hundred nine).

S.R. 113 (one hundred thirteen).

S.R. 114 (one hundred fourteen).

S.R. 117 (one hundred seventeen).

- S.R. 118 (one hundred eighteen).
- S.R. 119 (one hundred nineteen).
- S.R. 120 (one hundred twenty).
- S.R. 121 (one hundred twenty-one).
- S.J.R. 331 (three hundred thirty-one) was taken up and, on motion of Senator Newman, was referred to the Committee on Rules.

SUPPLEMENTAL CALENDAR NO. 1

UNFINISHED BUSINESS—SENATE

S.B. 1445 (one thousand four hundred forty-five) was taken up with the amendment in the nature of a substitute with amendment proposed by the House of Delegates, the title of the substitute, printed separately, and amendment being as follows:

A BILL to facilitate the administration of the COVID-19 vaccine; emergency.

1. Line 119, substitute, after *liable*

insert

for

On motion of Senator Dunnavant, the substitute with amendment was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

SENATE BILLS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- **S.B.** 1265 (one thousand two hundred sixty-five).
- **S.B. 1404** (one thousand four hundred four).
- S.B. 1444 (one thousand four hundred forty-four).
- S.B. 1109 (one thousand one hundred nine).
- S.B. 1153 (one thousand one hundred fifty-three).
- **S.B.** 1164 (one thousand one hundred sixty-four).
- S.B. 1311 (one thousand three hundred eleven).
- **S.B.** 1319 (one thousand three hundred nineteen).
- S.B. 1354 (one thousand three hundred fifty-four).
- **S.B.** 1374 (one thousand three hundred seventy-four).
- **S.B. 1411** (one thousand four hundred eleven).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate bills were passed by for the day:

- **S.B. 1265** (one thousand two hundred sixty-five).
- S.B. 1404 (one thousand four hundred four).
- S.B. 1444 (one thousand four hundred forty-four).
- S.B. 1109 (one thousand one hundred nine).
- S.B. 1153 (one thousand one hundred fifty-three).
- **S.B.** 1164 (one thousand one hundred sixty-four).
- **S.B. 1311** (one thousand three hundred eleven).
- S.B. 1319 (one thousand three hundred nineteen).
- S.B. 1354 (one thousand three hundred fifty-four).
- **S.B. 1374** (one thousand three hundred seventy-four).
- S.B. 1411 (one thousand four hundred eleven).

SENATE JOINT RESOLUTIONS ON FIRST READING

Senator Saslaw moved that the Rules be suspended and the first reading of the titles of the following Senate joint resolutions be waived:

- S.J.R. 275 (two hundred seventy-five).
- S.J.R. 270 (two hundred seventy).
- S.J.R. 271 (two hundred seventy-one).
- S.J.R. 272 (two hundred seventy-two).
- S.J.R. 289 (two hundred eighty-nine).
- S.J.R. 290 (two hundred ninety).
- S.J.R. 310 (three hundred ten).
- S.J.R. 322 (three hundred twenty-two).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following Senate joint resolutions were passed by for the day:

S.J.R. 275 (two hundred seventy-five).

S.J.R. 270 (two hundred seventy).

S.J.R. 271 (two hundred seventy-one).

S.J.R. 272 (two hundred seventy-two).

S.J.R. 289 (two hundred eighty-nine).

S.J.R. 290 (two hundred ninety).

S.J.R. 310 (three hundred ten).

S.J.R. 322 (three hundred twenty-two).

HONORARY ADJOURNMENT

Senator Barker addressed the Senate in memory of Officer Brian David Sicknick.

wan Carle Schoon

Senator Barker requested that when the Senate adjourns today, it adjourn in memory of Officer Brian David Sicknick.

On motion of Senator Lucas, the Senate, in memory of Officer Brian David Sicknick, adjourned until tomorrow at 10:00 a.m.

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

FRIDAY, FEBRUARY 5, 2021

The Senate met at 10:00 a.m. and was called to order by the President pro tempore, Senator L. Louise Lucas.

The Honorable Lionell Spruill, Sr., Fifth Senatorial District, offered the following prayer:

God, my Father, Father of Abraham, Isaac, Jacob. We come to You this morning dear God, my Father, and say thank You because You have been so good to us. Thank You, dear Father, because You woke us up this morning. Allow us to see another day. Bless You, dear Father please, to forgive us for all our many sins. But we are here today, dear Father, first, to give You all the honor, all the glory, give You all the praise. I ask, dear Father please, to touch each and everyone here today. Help us, dear Father, to work together, and we will please You first, dear Father, then we will do the business of this Commonwealth. Help us, dear Father, to love one another because love soothes things. Help us, dear Father, to touch one another. Let us know, dear Father, we can do whatever is on Your mind to please this Commonwealth. Touch us, dear Father. Let us stop fussing. Let us stop arguing. Let us do what is best for the Commonwealth of Virginia. We will thank You, dear Father, for all that You have done for us. We are so careful to give You all the honor, give You all the praise, and all the glory. In Jesus' name we ask these things. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Nathan A. Hatfield, Assistant Clerk-LINCS, Senate Clerk's Office.

On motion of Senator McDougle, the Rules were suspended to allow Senators Chase and Norment, impacted by COVID-19, to participate remotely and electronically in the daily session and committees.

The recorded vote is as follows:

YEAS--32. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--32.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell.

A quorum was present.

After the roll call, Senators Chase, Dunnavant, Obenshain, and Vogel notified the Clerk of their presence.

On motion of Senator Locke, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--31. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--31.

NAYS--Deeds, DeSteph, McDougle, Petersen--4. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 4, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1789.** A BILL to amend and reenact § 30-140 of the Code of Virginia, relating to Auditor of Public Accounts; audits of certain political subdivisions.
- **H.B. 1818.** A BILL to amend and reenact § 65.2-402 of the Code of Virginia, relating to workers' compensation; presumption of compensation for certain diseases; applicable to salaried and volunteer emergency medical services personnel.
- **H.B. 1822.** A BILL to amend the Code of Virginia by adding a section numbered 38.2-3407.15:6, relating to health insurance; cost-sharing payments for prescription asthma inhalers.
- H.B. 1849. A BILL related to apprenticeship training programs; report.
- **H.B. 1980.** A BILL to amend the Code of Virginia by adding a section numbered 23.1-615.1, relating to the establishment of the Enslaved Ancestors College Access Scholarship and Memorial Program.
- **H.B. 2004.** A BILL to amend and reenact §§ 2.2-3706, 2.2-3711, 19.2-174.1, and 19.2-368.3 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-3706.1, relating to the Virginia Freedom of Information Act; law-enforcement criminal incident information; criminal investigative files.
- **H.B. 2065.** A BILL to direct the Department of Social Services to establish a work group to develop a plan for a three-year pilot Produce Rx Plan.
- **H.B. 2118.** A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 13 of Title 10.1 a section numbered 10.1-1322.5, relating to Virginia Electric Vehicle Grant Fund and Program; creation; work group report.
- **H.B. 2120.** A BILL to amend and reenact §§ 23.1-409, 23.1-802, and 23.1-1303 of the Code of Virginia, relating to public institutions of higher education; governing boards; meetings, input, and disclosures.
- **H.B. 2137.** A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 40.1 an article numbered 2.1, consisting of sections numbered 40.1-33.3 through 40.1-33.9, relating to employees; paid sick leave.

- **H.B. 2166.** A BILL to amend and reenact §§ 37.2-817, 37.2-817.1, 37.2-817.2, and 37.2-817.4 of the Code of Virginia and to repeal § 37.2-817.3 of the Code of Virginia, relating to involuntary admission.
- **H.B. 2207.** A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability from COVID-19.
- **H.B. 2230.** A BILL to amend and reenact §§ 64.2-2000, 64.2-2003, and 64.2-2007 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 3 of Title 37.2 a section numbered 37.2-314.3, relating to supported decision-making agreements. Report.
- **H.B. 2273.** A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.
- H.B. 2288. A BILL to amend the Code of Virginia by adding a section numbered 2.2-4303.02, relating to the Virginia Public Procurement Act; construction contracts; requirement to submit list of subcontractors.
- H.B. 2322. A BILL to amend and reenact § 2.2-212 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-507.3, by adding in Chapter 22 of Title 2.2 an article numbered 12, consisting of sections numbered 2.2-2365 through 2.2-2376, and by adding in Article 3.1 of Chapter 1 of Title 51.1 a section numbered 51.1-124.40, relating to establishing an Opioid Abatement Authority.
- THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTIONS:
- **H.J.R.** 555. Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters; felon disenfranchisement; automatic restoration of political rights.
- **H.J.R. 582.** Proposing an amendment to Section 15-A of Article I of the Constitution of Virginia, relating to marriage; repeal of same-sex marriage prohibition; affirmative right to marry.
- **H.J.R. 629.** Confirming the appointment by the Chief Justice of the Supreme Court of Virginia of the Chairman of the Virginia Criminal Sentencing Commission.
- H.J.R. 639. Commending Linda Y. Kelleher.
- H.J.R. 640. Commending Nancy Van Doren.
- H.J.R. 641. Commending Charlene Bickford.
- H.J.R. 642. Celebrating the life of Ann Bisson.
- **H.J.R. 643.** Celebrating the life of Nancy Todd Renfro.
- H.J.R. 644. Celebrating the life of Earl Mark Ferguson.
- H.J.R. 645. Celebrating the life of Bob Colyer, Sr.
- H.J.R. 646. Celebrating the life of Dolson Barnett Anderson, Jr.

- H.J.R. 647. Celebrating the life of Herbert Allen Dabney III.
- H.J.R. 648. Celebrating the life of Edward Andrews.
- H.J.R. 649. Celebrating the life of Alexander Hoke Slaughter.
- H.J.R. 650. Celebrating the life of the Honorable Thomas Overton Jones.
- H.J.R. 651. Celebrating the life of Katherine Bridgforth Hooker.
- H.J.R. 652. Celebrating the life of Charles Lincoln Garner.
- H.J.R. 653. Celebrating the life of Alfred Jerome Denney.
- H.J.R. 654. Celebrating the life of Arthur Warrington Gosling.
- H.J.R. 655. Commending the staff at Parham Doctors' Hospital.
- H.J.R. 656. Commending Murray Jay Farr.
- H.J.R. 657. Commending James B. Cole.
- H.J.R. 658. Commending Cintia Johnson.
- H.J.R. 659. Commending Randall R. Silber.
- H.J.R. 660. Commending Danny TK Avula, M.D.
- H.J.R. 661. Commending the Henrico County Registrar's Office.
- H.J.R. 662. Celebrating the life of Joseph Maurice Tarantino.
- H.J.R. 663. Commending Richard G. Johnstone, Jr.
- H.J.R. 664. Commending the Honorable James S. Yoffy.
- H.J.R. 665. Commending Bobbie Jean Meriwether.
- H.J.R. 666. Commending the City of Fredericksburg.
- H.J.R. 667. Commending the Fredericksburg Dog Mart.
- H.J.R. 668. Celebrating the life of Franklin Delano Robertson.
- H.J.R. 669. Celebrating the life of Isabel Gallimore Shelor.
- H.J.R. 670. Celebrating the life of Robert Preston Midgett II.
- H.J.R. 671. Commending Humberto Cardounel, Jr.
- H.J.R. 672. Commending Darrell W. Warren, Jr.

- H.J.R. 673. Commending John Hutchison Anderson.
- H.J.R. 674. Commending Gregory Garfield Harris.
- H.J.R. 675. Commending the Virginia Academy of Science, Engineering and Medicine.
- H.J.R. 676. Commending the T.C. Williams School of Law at the University of Richmond.
- H.J.R. 677. Commending the New River Valley Public Health Task Force.
- H.J.R. 678. Celebrating the life of Elizabeth Ann Kerr Ledgerton.
- H.J.R. 679. Celebrating the life of Perry Anthony Hodge.
- H.J.R. 680. Celebrating the life of the Honorable Joan Hardie Munford.
- H.J.R. 681. Commending Samuel Allen.
- H.J.R. 682. Commending Bobby Hill.
- H.J.R. 683. Commending the Loudoun County Office of Elections and Voter Registration.
- H.J.R. 684. Commending Ethel L. Grandy.
- H.J.R. 685. Celebrating the life of Larry Jerome Bland.
- H.J.R. 686. Celebrating the life of Charles Lindsay McDowell, M.D.
- H.J.R. 687. Celebrating the life of Sheila Kavanagh Mandt.
- H.J.R. 688. Celebrating the life of Mozelle Willis Minor.
- H.J.R. 689. Celebrating the life of Jamile J. Hill.
- H.J.R. 690. Celebrating the life of William Archer Royall, Jr.
- H.J.R. 691. Celebrating the life of the Reverend Dr. James Alfred Carey.
- H.J.R. 692. Commending Josh Sweat.
- H.J.R. 693. Commending Marty L. Miller.
- H.J.R. 694. Commending Adrienne Warren.
- H.J.R. 695. Commending Annie Laura Downing.
- H.J.R. 696. Commending Dr. Ronald A. Crutcher.
- H.J.R. 697. Commending Barry R. Lawrence.
- H.J.R. 698. Commending O. R. Singleton, Jr.

-438-

- H.J.R. 699. Commending Joseph P. Rapisarda, Jr.
- H.J.R. 700. Commending Tim Foster.
- **H.J.R.** 701. Celebrating the life of Helen Dukas.
- H.J.R. 702. Celebrating the life of George B. Vaughan.
- H.J.R. 703. Celebrating the life of William Walker Byers, Jr.
- H.J.R. 704. Celebrating the life of Ida Johnson Lewis.
- H.J.R. 705. Celebrating the life of the Reverend James William Wright, Sr.
- H.J.R. 706. Celebrating the life of John F. Merchant.
- H.J.R. 707. Celebrating the life of Joseph William Teague, Sr.
- H.J.R. 708. Celebrating the life of Lovell Louis Coleman, Sr.
- H.J.R. 709. Celebrating the life of Stuart Wallace Connock.
- **H.J.R.** 710. Celebrating the life of Virginia Leonard Plotnick.
- H.J.R. 711. Celebrating the life of Oakley W. Hogg III.
- H.J.R. 712. Commending the Virginia State Police.
- H.J.R. 713. Commending the Virginia National Guard.
- H.J.R. 714. Commending John R. Broderick.
- H.J.R. 715. Commending Valerie Braxton-Williams.
- H.J.R. 716. Celebrating the life of Thomas Henry Francis.
- H.J.R. 717. Commending the Center for Excellence in Education.
- H.J.R. 718. Commending the Asian Pacific American Bar Association of the Greater Washington, D.C., Area, Inc.
- H.J.R. 719. Commending Joe Szakos.
- H.J.R. 720. Commending Lieutenant Junior Grade Madeline Swegle, USN.
- H.J.R. 721. Commending the Fairfax County Park Authority.
- H.J.R. 722. Celebrating the life of Ae Ja Kang.
- H.J.R. 723. Celebrating the life of Arthur William Walker.
- H.J.R. 724. Celebrating the life of Tony E. Colden, Jr.

- H.J.R. 725. Celebrating the life of Connie Weldon Edwards.
- H.J.R. 726. Celebrating the life of the Reverend Bobby Eugene Holmes.
- H.J.R. 727. Celebrating the life of Algenon L. Brown.
- H.J.R. 728. Celebrating the life of Ena Ampy Logan.
- H.J.R. 729. Celebrating the life of Willie Makently Andrews.
- H.J.R. 730. Celebrating the life of the Honorable Edna Elizabeth Keys-Chavis.
- H.J.R. 731. Celebrating the life of Rudolph E. Ford, Jr.
- H.J.R. 732. Celebrating the life of Larry Jerome Bland.
- H.J.R. 733. Celebrating the life of the Reverend Dr. Willie Woodson.
- **H.J.R. 734.** Celebrating the life of Thelonius Leander Wood.
- H.J.R. 735. Celebrating the life of William Henry Womack, Jr.
- H.J.R. 736. Celebrating the life of Josephine Johnson Bigger.
- **H.J.R.** 737. Celebrating the life of Thomas Henry Francis.
- H.J.R. 738. Celebrating the life of Javier J. Smith.
- H.J.R. 739. Celebrating the life of Carolyn Louise Johnson.
- H.J.R. 740. Celebrating the life of Andrew D. Washington.
- H.J.R. 741. Celebrating the life of Mary Louise Tuell.
- H.J.R. 742. Celebrating the life of Jean Smith Brown.
- H.J.R. 743. Celebrating the life of Paul Wendell Dick.
- H.J.R. 744. Commending The Chatham Garden Club.
- H.J.R. 745. Commending Joann Grant Luck.
- H.J.R. 746. Commending William S. Feasenmyer, Jr.
- H.J.R. 747. Commending Pleasant View, Inc.
- H.J.R. 748. Commending Saint Joseph Catholic School.
- H.J.R. 749. Commending Newport News Public Library.
- H.J.R. 750. Commending The Virginian-Pilot.

- H.J.R. 751. Commending the Williamsburg Health Foundation.
- H.J.R. 752. Commending Patty Ortiz.
- H.J.R. 753. Commending Katie Gaylord.
- H.J.R. 754. Commending Sergeant Steven Pebler.
- H.J.R. 755. Commending Williamsburg Faith in Action and the Arc of Greater Williamsburg.
- H.J.R. 756. Commending Tammy Williams Guido.
- H.J.R. 757. Celebrating the life of George S. Genakos.
- H.J.R. 758. Commending the authors of Stir Crazy in Williamsburg.
- H.J.R. 759. Commending John Unger.
- H.J.R. 760. Commending Dennis Linaburg.
- H.J.R. 761. Commending Sergeant Brian Rosenberry.
- H.J.R. 762. Commending Clifton Presbyterian Church.
- H.J.R. 763. Commending the Korean Central Presbyterian Church.
- H.J.R. 764. Commending the Northern Virginia Therapeutic Riding Program.
- H.J.R. 765. Commending Rebecca Suerdieck and Julia Oxrieder.
- H.J.R. 766. Celebrating the life of Lucille Minchin Zaleski.
- H.J.R. 767. Celebrating the life of Flournoy A. Keatts.
- H.J.R. 768. Celebrating the life of Avicia Beatrice Hooper Thorpe.
- H.J.R. 769. Celebrating the life of Joe Louis Hairston.
- H.J.R. 770. Celebrating the life of Robert George Wingfield.
- H.J.R. 771. Celebrating the life of Mary Elene Williams Farlow.
- H.J.R. 772. Celebrating the life of Michael Wayne Woods.
- **H.J.R.** 773. Celebrating the life of Grover Harold Plaster.
- H.J.R. 774. Commending the Danville Life Saving and First Aid Crew, Inc.
- H.J.R. 775. Commending Daniel Banister.

- H.J.R. 776. Commending T. David Luther.
- H.J.R. 777. Commending Fred O. Shanks III.
- **H.J.R.** 778. Commending the Reverend Doug Barber.
- H.J.R. 779. Commending Hardy Petroleum.
- H.J.R. 780. Commending Dr. Lewis R. Brown.
- H.J.R. 781. Commending Dr. Shamira A. Brown.
- H.J.R. 782. Celebrating the life of Frederick Cameron.
- H.J.R. 783. Commending Zion Baptist Church.
- H.J.R. 784. Commending the Reverend Dr. Oretha P. Cross.
- **H.J.R.** 785. Commending the Community Free Clinic of Newport News.
- H.J.R. 786. Commending the C. Waldo Scott Center for H.O.P.E.
- H.J.R. 787. Commending the Newsome House Museum and Cultural Center.
- THE HOUSE OF DELEGATES HAS AGREED TO THE SUBSTITUTE WITH AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILL:
- **H.B. 2333.** A BILL to facilitate the administration of the COVID-19 vaccine; emergency. EMERGENCY
- THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING SENATE JOINT RESOLUTIONS:
- **S.J.R. 298.** Confirming appointments by the Governor of certain persons communicated to the General Assembly June 1, 2020.
- **S.J.R. 299.** Confirming appointments by the Governor of certain persons communicated to the General Assembly August 1, 2020.
- **S.J.R. 300.** Confirming appointments by the Governor of certain persons communicated to the General Assembly October 1, 2020.
- **S.J.R. 301.** Confirming appointments by the Governor of certain persons communicated to the General Assembly April 1, 2020.
- **S.J.R. 302.** Confirming appointments by the Governor of certain persons communicated to the General Assembly December 1, 2020.
- **S.J.R. 303.** Confirming appointments by the Governor of certain persons communicated to the General Assembly January 1, 2021.

S.J.R. 327. Confirming appointments by the Governor of certain persons communicated to the General Assembly January 24, 2021.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows: YEAS--30. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell--30.

NAYS--Deeds, DeSteph, Edwards, McDougle, Petersen, Stanley--6. RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1789 was referred to the Committee on Rules.
- H.B. 1818, H.B. 1822, H.B. 2137, and H.B. 2207 were referred to the Committee on Commerce and Labor.
- H.B. 1849, H.B. 2004, H.B. 2288, and H.B. 2322 were referred to the Committee on General Laws and Technology.
 - H.B. 1980, H.B. 2120, and H.B. 2230 were referred to the Committee on Education and Health.
 - H.B. 2065 and H.B. 2166 were referred to the Committee on Rehabilitation and Social Services.
 - H.B. 2118 and H.B. 2273 were referred to the Committee on Finance and Appropriations.

The House joint resolutions, communicated as agreed to by the House of Delegates, the first reading of their titles having been waived, were referred as follows:

- H.J.R. 555 and H.J.R. 582 were referred to the Committee on Privileges and Elections.
- H.J.R. 629 was referred to the Committee on the Judiciary.

The House joint resolutions, communicated as agreed to by the House of Delegates, were laid on the Clerk's Desk under Senate Rule 26 (g) as follows:

H.J.R. 639, H.J.R. 640, H.J.R. 641, H.J.R. 642, H.J.R. 643, H.J.R. 644, H.J.R. 645, H.J.R. 646, H.J.R. 647, H.J.R. 648, H.J.R. 649, H.J.R. 650, H.J.R. 651, H.J.R. 652, H.J.R. 653, H.J.R. 654, H.J.R. 655, H.J.R. 656, H.J.R. 657, H.J.R. 658, H.J.R. 659, H.J.R. 660, H.J.R. 661, H.J.R. 662, H.J.R. 663, H.J.R. 664, H.J.R. 665, H.J.R. 666, H.J.R. 667, H.J.R. 669, H.J.R. 670, H.J.R. 671, H.J.R. 672, H.J.R. 673, H.J.R. 674, H.J.R. 675, H.J.R. 676, H.J.R. 677, H.J.R. 678, H.J.R. 679, H.J.R. 680,

H.J.R. 681, H.J.R. 682, H.J.R. 683, H.J.R. 684, H.J.R. 685, H.J.R. 686, H.J.R. 687, H.J.R. 688, H.J.R. 689, H.J.R. 690, H.J.R. 691, H.J.R. 692, H.J.R. 693, H.J.R. 694, H.J.R. 695, H.J.R. 696, H.J.R. 697, H.J.R. 698, H.J.R. 699, H.J.R. 700, H.J.R. 701, H.J.R. 702, H.J.R. 703, H.J.R. 704, H.J.R. 705, H.J.R. 706, H.J.R. 707, H.J.R. 708, H.J.R. 709, H.J.R. 710, H.J.R. 711, H.J.R. 712, H.J.R. 713, H.J.R. 714, H.J.R. 715, H.J.R. 716, H.J.R. 717, H.J.R. 718, H.J.R. 719, H.J.R. 720, H.J.R. 721, H.J.R. 722, H.J.R. 723, H.J.R. 724, H.J.R. 725, H.J.R. 726, H.J.R. 727, H.J.R. 728, H.J.R. 729, H.J.R. 730, H.J.R. 731, H.J.R. 732, H.J.R. 733, H.J.R. 734, H.J.R. 735, H.J.R. 736, H.J.R. 737, H.J.R. 738, H.J.R. 739, H.J.R. 740, H.J.R. 741, H.J.R. 742, H.J.R. 743, H.J.R. 744, H.J.R. 745, H.J.R. 746, H.J.R. 747, H.J.R. 748, H.J.R. 749, H.J.R. 750, H.J.R. 751, H.J.R. 752, H.J.R. 753, H.J.R. 754, H.J.R. 755, H.J.R. 756, H.J.R. 757, H.J.R. 758, H.J.R. 768, H.J.R. 760, H.J.R. 761, H.J.R. 761, H.J.R. 762, H.J.R. 763, H.J.R. 764, H.J.R. 765, H.J.R. 766, H.J.R. 766, H.J.R. 766, H.J.R. 767, H.J.R. 768, H.J.R. 779, H.J.R. 771, H.J.R. 772, H.J.R. 773, H.J.R. 774, H.J.R. 775, H.J.R. 776, H.J.R. 777, H.J.R. 778, H.J.R. 779, H.J.R. 780, H.J.R. 781, H.J.R. 782, H.J.R. 783, H.J.R. 784, H.J.R. 785, H.J.R. 786, and H.J.R. 787.

CALENDAR

SENATE BILL ON THIRD READING

S.B. 1357 (one thousand three hundred fifty-seven) was read by title the third time and, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--36.

NAYS--0.

RULE 36--0.

STATEMENTS ON VOTE

Senator Chase stated that she was recorded as not voting on the question of the passage of **S.B. 1357**, whereas she intended to vote yea.

Senator Newman stated that he was recorded as not voting on the question of the passage of **S.B.** 1357, whereas he intended to vote yea.

RECESS

At 10:30 a.m., Senator McDougle moved that the Senate recess until 11:30 a.m.

The motion was agreed to.

The hour of 11:30 a.m. having arrived, the Chair was resumed.

THE PRESIDENT PRESIDING

The President assumed the Chair.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 5, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 2024.** A BILL to approve a construction plan for a replica of the Bob White Covered Bridge in Patrick County.
- **H.B. 2167.** A BILL to amend and reenact §§ 53.1-136 and 53.1-155 of the Code of Virginia, relating to parole; notice and certification; monthly reports; discretionary early consideration.
- H.B. 2185. A BILL to amend the Code of Virginia by adding a section numbered 58.1-609.14, relating to sales tax; exemption for personal protective equipment; emergency.
 EMERGENCY
- **H.B. 2194.** A BILL to amend and reenact § 18.2-60 of the Code of Virginia, relating to communicating threats of death or bodily injury to a person with intent to intimidate; penalty.
- **H.B. 2219.** A BILL to amend and reenact §§ 38.2-3407.7, 38.2-4209.1, and 38.2-4312.1 of the Code of Virginia, relating to pharmacies; freedom of choice.
- **H.B. 2311.** A BILL to amend and reenact §§ 10.1-2300 and 10.1-2306 of the Code of Virginia, relating to state archaeological sites; battlefields.
- **H.B. 2314.** A BILL to require the Board of Education to amend a certain regulation relating to special education.
- **H.B. 2316.** A BILL to require the Department of Education and Board of Education to take certain actions relating to special education and related services for students with disabilities.
- **H.B. 2317.** A BILL to amend and reenact §§ 9.1-116.2 and 9.1-116.3 of the Code of Virginia, relating to the Advisory Committee on Sexual and Domestic Violence; membership; duties.
- **H.B. 2337.** A BILL to amend and reenact § 58.1-322.02 of the Code of Virginia, relating to income tax; subtraction for low-income military veterans with a permanent service-connected disability.
- IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows: YEAS--35. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, McDougle--2.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 2024 was referred to the Committee on Transportation.
- H.B. 2167 was referred to the Committee on Rehabilitation and Social Services.
- H.B. 2185 and H.B. 2337 were referred to the Committee on Finance and Appropriations.
- H.B. 2194 and H.B. 2317 were referred to the Committee on the Judiciary.
- H.B. 2219 was referred to the Committee on Commerce and Labor.
- H.B. 2311 was referred to the Committee on Agriculture, Conservation and Natural Resources.
- H.B. 2314 and H.B. 2316 were referred to the Committee on Education and Health.

SENATE BILLS ON THIRD READING

S.B. 1138 (one thousand one hundred thirty-eight) was taken up.

RECONSIDERATION

Senator McClellan moved to reconsider the vote by which **S.B.** 1138 (one thousand one hundred thirty-eight) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Vogel--36.

NAYS--0.

RULE 36--0.

RECONSIDERATION

Senator McClellan moved to reconsider the vote by which the amendment offered by Senator McClellan to S.B. 1138 (one thousand one hundred thirty-eight) was agreed to.

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

Senator McClellan moved that the amendment be rejected.

The question was put on agreeing to the amendment.

The amendment was rejected.

Senator McClellan offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-52.1, 18.2-67.4:1, 18.2-346.1, 32.1-291.16, 54.1-2982, 54.1-2983, and 57-48 of the Code of Virginia and to repeal §§ 18.2-62 and 32.1-289.2 of the Code of Virginia, relating to sexually transmitted infections, infected sexual battery.

On motion of Senator McClellan, the reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

Senator McClellan moved that the Rules be suspended and the third reading of the title of **S.B. 1138** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

S.B. 1138, on motion of Senator McClellan, was passed by temporarily.

S.B. 1384 (one thousand three hundred eighty-four) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows: YEAS--20. NAYS--17. RULE 36--1.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

-447-

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--Stuart--1.

STATEMENT ON VOTE

Senator Lewis stated that he was recorded as not voting on the question of the passage of **S.B. 1384**, whereas he intended to vote yea.

S.B. 1105 (one thousand one hundred five) was read by title the third time and, on motion of Senator Stanley, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator Lewis stated that he was recorded as not voting on the question of the passage of **S.B. 1105**, whereas he intended to vote yea.

S.B. 1113 (one thousand one hundred thirteen) was read by title the third time and, on motion of Senator Spruill, was passed with its title.

The recorded vote is as follows:

YEAS--32. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--32.

NAYS--Chase, Newman, Obenshain, Peake, Stanley, Suetterlein--6.

RULE 36--0.

STATEMENT ON VOTE

Senator Lewis stated that he was recorded as not voting on the question of the passage of **S.B. 1113**, whereas he intended to vote yea.

S.B. 1127 (one thousand one hundred twenty-seven) was read by title the third time and, on motion of Senator Reeves, was passed with its title.

The recorded vote is as follows:

YEAS--29. NAYS--7. RULE 36--3.

YEAS--Barker, Bell, Boysko, Cosgrove, Dunnavant, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Locke, Lucas, Marsden, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein--29.

NAYS--Chase, Deeds, Lewis, Mason, Stuart, Surovell, Vogel--7.

RULE 36--DeSteph, Ebbin, Petersen--3.

S.B. 1196 (one thousand one hundred ninety-six) was read by title the third time and, on motion of Senator Locke, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--16.

RULE 36--0.

S.B. 1201 (one thousand two hundred one) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--1.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--38.

NAYS--0.

RULE 36--Suetterlein--1.

S.B. 1226 (one thousand two hundred twenty-six) was read by title the third time and, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Peake, Petersen, Pillion, Reeves, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Newman, Obenshain, Ruff, Stanley--4.

RULE 36--0.

S.B. 1252 (one thousand two hundred fifty-two) was read by title the third time and, on motion of Senator McPike, was passed with its title.

The recorded vote is as follows: YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

S.B. 1257 (one thousand two hundred fifty-seven) was taken up.

RECONSIDERATION

Senator McClellan moved to reconsider the vote by which **S.B. 1257** (one thousand two hundred fifty-seven) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--DeSteph--1. RULE 36--0.

Senator McClellan offered the following amendment to the substitute:

1. Line 159, substitute, after *at least* strike

two
insert

three

On motion of Senator McClellan, the reading of the amendment was waived.

On motion of Senator McClellan, the amendment was agreed to.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

Senator McClellan moved that the Rules be suspended and the third reading of the title of **S.B. 1257** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1257, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Chase, Peake--2.

RULE 36--0.

S.B. 1261 (one thousand two hundred sixty-one) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1262 (one thousand two hundred sixty-two) was read by title the third time and, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--28. NAYS--11. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Reeves, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--28.

NAYS--Chase, DeSteph, Hanger, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Stuart--11.

RULE 36--0.

S.B. 1278 (one thousand two hundred seventy-eight) was read by title the third time and, on motion of Senator Bell, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--4. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Chase, DeSteph, McDougle, Obenshain--4.

RULE 36--Petersen--1.

S.B. 1282 (one thousand two hundred eighty-two) was read by title the third time and, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--16.

RULE 36--0.

STATEMENT ON VOTE

Senator McDougle stated that he was recorded as not voting on the question of the passage of **S.B.** 1282, whereas he intended to vote nay.

S.B. 1284 (one thousand two hundred eighty-four) was read by title the third time and, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1289 (one thousand two hundred eighty-nine) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1301 (one thousand three hundred one) was read by title the third time and, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--23. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--15.

RULE 36--0.

S.B. 1310 (one thousand three hundred ten) was read by title the third time and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1315 (one thousand three hundred fifteen) was read by title the third time.

Senator McClellan moved that **S.B. 1315** be passed with its title.

- S.B. 1315, on motion of Senator Dunnavant, was passed by temporarily.
- **S.B. 1316** (one thousand three hundred sixteen) was read by title the third time and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator Lucas stated that she was recorded as not voting on the question of the passage of **S.B. 1316**, whereas she intended to vote yea.

RECONSIDERATION

Senator Vogel moved to reconsider the vote by which **S.B. 1301** (one thousand three hundred one) was passed with its title.

The motion was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw,

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator Lucas stated that she was recorded as not voting on the question of the reconsideration of the passage of **S.B. 1301**, whereas she intended to vote yea.

S.B. 1301, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--23. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--16.

RULE 36--0.

S.B. 1318 (one thousand three hundred eighteen) was read by title the third time and, on motion of Senator Hashmi, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--17.

RULE 36--0.

S.B. 1326 (one thousand three hundred twenty-six) was taken up.

RECONSIDERATION

Senator Hanger moved to reconsider the vote by which **S.B. 1326** (one thousand three hundred twenty-six) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- S.B. 1326, on motion of Senator Hanger, was passed by temporarily.
- **S.B. 1333** (one thousand three hundred thirty-three), on motion of Senator Lucas, was passed by temporarily.
- **S.B. 1339** (one thousand three hundred thirty-nine) was read by title the third time and, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

RECONSIDERATION

Senator Petersen moved to reconsider the vote by which S.B. 1339 (one thousand three hundred thirty-nine) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1339, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows: YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

S.B. 1341 (one thousand three hundred forty-one) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Favola--1. RULE 36--0.

HOUSE COMMUNICATIONS

The following communications were received:

In the House of Delegates February 5, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1914.** A BILL to amend and reenact § 56-585.1 of the Code of Virginia, relating to electric utilities; triennial review; period costs; rate reductions.
- **H.B. 1935.** A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, and 58.1-322.03 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.
- H.B. 1984. A BILL to authorize the State Corporation Commission, in triennial review proceedings for a Phase II Utility, to determine fair rates of return on common equity for the utility's generation and distribution services and to order increases or decreases to the utility's rates for generation and distribution.
- **H.B. 2003.** A BILL to amend and reenact § 59.1-200 of the Code of Virginia, relating to the Consumer Protection Act; prohibited practices; certain advertising related to school quality.
- **H.B. 2014.** A BILL to amend and reenact §§ 36-139 and 55.1-1250 of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; landlord remedies; landlord's acceptance of rent with reservation; tenant's right of redemption.
- **H.B. 2048.** A BILL to amend and reenact § 56-577 of the Code of Virginia, relating to electric utility regulation; purchasing from competitive suppliers.

- H.B. 2049. A BILL to amend and reenact § 56-585.1 of the Code of Virginia, relating to electric utilities; customer credit reinvestment offsets.
- **H.B. 2160.** A BILL to amend and reenact § 56-585.1 of the Code of Virginia, relating to electric utilities; triennial review; fair rate of return; customer bill credits.
- **H.B. 2175.** A BILL to amend and reenact §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-2223.5, relating to housing protections; foreclosures; manufactured housing.
- **H.B. 2200.** A BILL to amend and reenact § 56-585.1 of the Code of Virginia, relating to electric utilities; triennial review.
- H.B. 2227. A BILL to direct the Board of Housing and Community Development to consider adopting amendments to the Uniform Statewide Building Code relating to energy efficiency and conservation upon each publication of a new version of the International Code Council's International Energy Conservation Code.
- **H.B. 2332.** A BILL to amend and reenact §§ 38.2-4214 and 38.2-4319 of the Code of Virginia and to amend the Code of Virginia by adding in Title 38.2 a chapter numbered 66, consisting of sections numbered 38.2-6600 through 38.2-6607, relating to the Commonwealth Health Reinsurance Program; established; special fund established; assessment; federal waiver application.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

> In the House of Delegates February 5, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

- **H.B. 1827.** A BILL to amend and reenact § 22.1-9 of the Code of Virginia, relating to the Board of Education; membership; geographic representation.
- **H.B. 1979.** A BILL to amend the Code of Virginia by adding in Title 67 a chapter numbered 18, consisting of sections numbered 67-1800 through 67-1806, relating to electric vehicle rebate program; creation and funding; report.
- **H.B. 2027.** A BILL to amend and reenact § 22.1-253.13:3 of the Code of Virginia, relating to Standards of Learning assessments; reading and mathematics; grades three through eight; individual student growth.
- H.B. 2040. A BILL to amend and reenact §§ 60.2-528.1, 60.2-619, and 60.2-633 of the Code of Virginia, relating to unemployment compensation; failure to respond; continuation of benefits; repayment of overpayments.
- H.B. 2053. A BILL to direct the Department of Housing and Community Development to convene a stakeholder advisory group to evaluate the construction of internal, attached, and detached accessory dwelling units as a strategy to address the Commonwealth's growing demand for affordable and market-rate housing.

- **H.B. 2074.** A BILL to amend and reenact § 2.2-234 of the Code of Virginia and to amend the Code of Virginia by adding in Article 12 of Chapter 2 of Title 2.2 sections numbered 2.2-236 and 2.2-237 and by adding a section numbered 15.2-2223.5, relating to environmental justice; interagency working group.
- **H.B. 2082.** A BILL to amend and reenact §§ 30-396 and 30-399 of the Code of Virginia, relating to Virginia Redistricting Commission; transparency in the redistricting process.
- **H.B. 2132.** A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 4 of Title 18.2 a section numbered 18.2-37.1 and by adding in Article 4 of Chapter 4 of Title 18.2 a section numbered 18.2-57.5, relating to homicides and assaults and bodily woundings; certain matters not to constitute defenses.
- **H.B. 2308.** A BILL to amend and reenact § 57-20 of the Code of Virginia, relating to religious and charitable matters; quantity of land certain associations may hold.
- **H.B. 2313.** A BILL to amend and reenact § 2.2-3705.7 of the Code of Virginia, relating to Freedom of Information Act; exclusions; definition of "Office of the Governor."
- **H.B. 2320.** A BILL to amend and reenact § 55.1-703 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 55.1-708.2, relating to property; required disclosures for buyer to exercise due diligence; flood risk report.
- **H.B. 2327.** A BILL to amend and reenact § 2.2-4321.3 of the Code of Virginia, relating to the Virginia Public Procurement Act; payment of prevailing wage; transportation infrastructure projects.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

> In the House of Delegates February 5, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILLS:

H.B. 2078. A BILL to amend and reenact §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4118, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446 of the Code of Virginia, relating to industrial hemp; emergency.
EMERGENCY

- **H.B. 2104.** A BILL to amend and reenact § 33.2-613 of the Code of Virginia and to repeal § 33.2-1823 of the Code of Virginia and Chapter 20 (§§ 56-535 through 56-552) of Title 56 of the Code of Virginia, relating to repeal of the Virginia Highway Corporation Act of 1988; roadways operating under the Highway Corporation Act of 1988 to operate under the Public-Private Transportation Act of 1995.
- **H.B. 2213.** A BILL to require the establishment of a work group to study the mining and processing of gold in the Commonwealth, and to prohibit the issuance of certain mining permits; report.

- **H.B. 2221.** A BILL to amend and reenact § 10.1-1408.1 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 10.1-1321.2 and 10.1-1437.1, by adding in Article 1 of Chapter 3.1 of Title 62.1 a section numbered 62.1-44.6:1, and by adding a section numbered 62.1-266.1, relating to environmental permits; community and environmental justice outreach.
- **H.B. 2298.** A BILL to amend and reenact § 29.1-100 of the Code of Virginia, relating to muzzleloading rifle and shotgun; definitions.
- **H.B. 2302.** A BILL to require that farmers market food and beverage sales be considered essential during a declared state of emergency.
- **H.B. 2312.** A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-122, 4.1-124, as it is currently effective and as it shall become effective, 4.1-128, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-202, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-390, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29 consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6 consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, consisting of chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Title 19.2 a

chapter numbered 23.2, consisting of sections numbered 19.2-392.5 through 19.2-392.13, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

- H.B. 2321. A BILL to amend and reenact §§ 2.2-200, 2.2-204, 2.2-205, 2.2-205.1, 2.2-435.6, 2.2-435.8, 2.2-435.9, 2.2-435.10, 2.2-2471, 2.2-2471.1, 2.2-2472, 2.2-2472.2, 2.2-2472.3, and 30-377 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 2 of Title 2.2 an article numbered 6.1, consisting of sections numbered 2.2-214.2 and 2.2-214.3; and to repeal § 2.2-435.7 of the Code of Virginia, relating to Governor's Secretaries; Secretary of Labor created.
- **H.B. 2330.** A BILL to amend and reenact §§ 56-576 and 56-585.6 of the Code of Virginia, relating to electric utilities; Percentage of Income Payment Program.
- **H.B. 2331.** A BILL to amend and reenact §§ 3.2-4212, 4.1-302, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5, 18.2-255, 18.2-255.2, 46.2-357, and 46.2-391 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 19.2-303.03, relating to elimination of mandatory minimum sentences; modification of sentence to mandatory minimum term of confinement for felony offenses.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

> In the House of Delegates February 5, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILL:

H.B. 2263. A BILL to amend and reenact §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, as it is currently effective and as it shall become effective, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307 of the Code of Virginia and to repeal §§ 8.01-654.1, 8.01-654.2, 17.1-313, and 18.2-17, Article 4.1 (§§ 19.2-163.7 and 19.2-163.8) of Chapter 10 of Title 19.2, Article 4.1 (§§ 19.2-264.2 through 19.2-264.5) of Chapter 15 of Title 19.2, § 53.1-230, and Chapter 13 (§§ 53.1-232 through 53.1-236) of Title 53.1 of the Code of Virginia, relating to abolition of the death penalty.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

Senator Saslaw moved that the Rules be suspended and the reading of the communications from the House of Delegates be waived.

The question was put on suspending the Rules and waiving the reading of the communications from the House of Delegates.

The motion was rejected, having failed to receive the necessary affirmative votes required by Article IV, Section 11, of the Constitution.

The recorded vote is as follows:

YEAS--30. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Pillion, Ruff, Saslaw, Stuart, Suetterlein, Surovell, Vogel--30.

NAYS--Deeds, DeSteph, Marsden, McDougle, Peake, Petersen, Reeves, Spruill, Stanley--9. RULE 36--0.

RECONSIDERATION

Senator Petersen moved to reconsider the vote by which the motion to suspend the Rules and waive the reading of the communications from the House of Delegates was rejected.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Deeds--1.

RULE 36--0.

On motion of Senator Saslaw, the Rules were suspended and the reading of the communications from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, DeSteph, McDougle, Petersen--4.

RULE 36--0.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

H.B. 1914, H.B. 1984, H.B. 2048, H.B. 2049, H.B. 2160, H.B. 2200, and H.B. 2332 were referred to the Committee on Commerce and Labor.

- H.B. 1935 was referred to the Committee on Finance and Appropriations.
- **H.B. 2003**, **H.B. 2014**, **H.B. 2175**, and **H.B. 2227** were referred to the Committee on General Laws and Technology.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 1827 and H.B. 2027 were referred to the Committee on Education and Health.
- H.B. 1979 was referred to the Committee on Finance and Appropriations.
- H.B. 2040 was referred to the Committee on Commerce and Labor.
- H.B. 2053, H.B. 2074, H.B. 2308, H.B. 2313, H.B. 2320, and H.B. 2327 were referred to the Committee on General Laws and Technology.
 - H.B. 2082 was referred to the Committee on Privileges and Elections.
 - H.B. 2132 was referred to the Committee on the Judiciary.

The House bills communicated as passed by the House of Delegates, the first reading of their titles required by the Constitution having been dispensed with, were referred as follows:

- H.B. 2078, H.B. 2221, and H.B. 2302 were referred to the Committee on Agriculture, Conservation and Natural Resources.
 - H.B. 2104 was referred to the Committee on Transportation.
 - H.B. 2213 was referred to the Committee on Rules.
 - H.B. 2298 and H.B. 2331 were referred to the Committee on the Judiciary.
 - H.B. 2312 was referred to the Committee on Rehabilitation and Social Services.
 - H.B. 2321 was referred to the Committee on General Laws and Technology.
 - H.B. 2330 was referred to the Committee on Commerce and Labor.

The House bill communicated as passed by the House of Delegates, the first reading of its title required by the Constitution having been dispensed with, was referred as follows:

H.B. 2263 was referred to the Committee on the Judiciary.

SENATE BILLS ON THIRD READING

S.B. 1350 (one thousand three hundred fifty) was read by title the third time and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1362 (one thousand three hundred sixty-two) was read by title the third time.

Senator Lewis moved that **S.B. 1362** be passed with its title.

The question was put on passing S.B. 1362 with its title.

S.B. 1362 was defeated with its title, having failed to receive the necessary affirmative votes required by Article IV, Section 13, of the Constitution.

The recorded vote is as follows:

YEAS--29. NAYS--10. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Saslaw, Spruill, Stanley, Surovell--29.

NAYS--Chase, Cosgrove, DeSteph, Hanger, McDougle, Obenshain, Ruff, Stuart, Suetterlein, Vogel--10.

RULE 36--0.

RECONSIDERATION

Senator Cosgrove moved to reconsider the vote by which S.B. 1362 was defeated with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- **S.B. 1362,** on motion of Senator Lewis, was passed by temporarily.
- **S.B. 1363** (one thousand three hundred sixty-three) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows: YEAS--28. NAYS--11. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Ruff, Saslaw, Spruill, Surovell, Vogel--28.

NAYS--Chase, DeSteph, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Stanley, Stuart, Suetterlein--11.

RULE 36--0.

STATEMENT ON VOTE

Senator Suetterlein stated that he was recorded as voting nay on the question of the passage of **S.B.** 1363, whereas he intended to vote yea.

S.B. 1380 (one thousand three hundred eighty) was read by title the third time and, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows: YEAS--33. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, Morrissey, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--33.

NAYS--McDougle, Obenshain, Stuart, Suetterlein--4. RULE 36--0.

STATEMENT ON VOTE

Senator McPike stated that he was recorded as not voting on the question of the passage of **S.B. 1380**, whereas he intended to vote yea.

S.B. 1382 (one thousand three hundred eighty-two) was read by title the third time.

Senator Favola moved that S.B. 1382 be passed with its title.

The question was put on passing S.B. 1382 with its title.

S.B. 1382 was defeated with its title.

The recorded vote is as follows:

YEAS--16. NAYS--22. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Locke, Lucas, Marsden, McClellan, Petersen, Saslaw, Surovell--16.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Lewis, Mason, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Vogel--22.

RULE 36--0.

S.B. 1392 (one thousand three hundred ninety-two) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--1.

YEAS--Barker, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--36.

NAYS--0.

RULE 36--Bell--1.

STATEMENT ON VOTE

Senator McPike stated that he was recorded as not voting on the question of the passage of **S.B. 1392**, whereas he intended to vote yea.

S.B. 1395 (one thousand three hundred ninety-five) was read by title the third time and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--0.

S.B. 1396 (one thousand three hundred ninety-six) was read by title the third time and, on motion of Senator Hashmi, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--16.

RULE 36--0.

S.B. 1401 (one thousand four hundred one) was read by title the third time and, on motion of Senator Pillion, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--12. RULE 36--0.

YEAS--Bell, Boysko, Chase, Deeds, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McPike, Morrissey, Norment, Pillion, Reeves, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--26.

NAYS--Barker, Cosgrove, DeSteph, Dunnavant, Ebbin, Marsden, McDougle, Newman, Obenshain, Peake, Petersen, Ruff--12.

RULE 36--0.

S.B. 1406 (one thousand four hundred six) was read by title the third time and, on motion of Senator Ebbin, was passed with its title.

The recorded vote is as follows:

YEAS--23. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--15.

RULE 36--0.

S.B. 1418 (one thousand four hundred eighteen) was read by title the third time and, on motion of Senator McPike, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1428 (one thousand four hundred twenty-eight) was read by title the third time and, on motion of Senator Locke, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--36.

NAYS--Mason, Suetterlein--2.

RULE 36--0.

S.B. 1442 (one thousand four hundred forty-two) was read by title the third time and, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--37.

NAYS--McDougle--1.

RULE 36--0.

S.B. 1443 (one thousand four hundred forty-three) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--0.

S.B. 1462 (one thousand four hundred sixty-two) was read by title the third time and, on motion of Senator Mason, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

S.B. 1469 (one thousand four hundred sixty-nine) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--2.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--36.

NAYS--0.

RULE 36--Edwards, Petersen--2.

STATEMENT ON VOTE

Senator Norment stated that he voted yea on the question of the passage of **S.B. 1469**, whereas he intended to abstain pursuant to Rule 36.

S.B. 1471 (one thousand four hundred seventy-one) was read by title the third time and, on motion of Senator Dunnavant, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1138 (one thousand one hundred thirty-eight) was taken up and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--0.

S.B. 1315 (one thousand three hundred fifteen) was taken up and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, Mason, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Suetterlein--13.

RULE 36--0.

S.B. 1326 (one thousand three hundred twenty-six) was taken up.

Senator Hanger offered the following amendments to the substitute:

1. Line 14, substitute, after cigarettes

strike

the remainder of line 14 through 58.1-3832.1 on line 15

2. Line 62, substitute

strike

all of lines 62 through 67

3. After line 69, substitute

insert

3. That it is the policy of the Commonwealth that, where practical, local cigarette stamping and tax collection is encouraged to be accomplished through regional cigarette tax boards modeled on the Northern Virginia

Cigarette Tax Board. Recognizing that the current system of stamping and tax collection is antiquated and places a burden on wholesalers and distributors, the Department of Taxation shall establish a task force to develop methods for modernizing the system and shall provide assistance as appropriate to localities seeking new regional cigarette tax boards. The task force shall include local government representatives, local commissioners of the revenue, cigarette wholesalers and distributors, and representatives of the Northern Virginia Cigarette Tax Board. The task force shall submit its recommendations to the Virginia General Assembly by November 1, 2021.

On motion of Senator Hanger, the reading of the amendments was waived.

On motion of Senator Hanger, the amendments were agreed to.

On motion of Senator Hanger, the bill was ordered to be engrossed and read by title the third time.

Senator Hanger moved that the Rules be suspended and the third reading of the title of **S.B. 1326** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1326, on motion of Senator Hanger, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Petersen, Ruff, Saslaw, Spruill, Surovell--26.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Norment, Obenshain, Peake, Pillion, Reeves, Stanley, Suetterlein, Vogel--12.

RULE 36--0.

S.B. 1333 (one thousand three hundred thirty-three) was taken up, read by title the third time and, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--32. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Suetterlein, Surovell, Vogel--32.

NAYS--Chase, McDougle, Newman, Reeves, Stanley--5.

RULE 36--0.

S.B. 1362 (one thousand three hundred sixty-two) was taken up.

RECONSIDERATION

Senator Lewis moved to reconsider the vote by which **S.B. 1362** (one thousand three hundred sixty-two) was ordered to be engrossed and read by title the third time.

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

Senator Lewis offered the following amendment to the substitute:

1. Line 25, substitute

strike

all of line 25

On motion of Senator Lewis, the reading of the amendment was waived.

On motion of Senator Lewis, the amendment was agreed to.

On motion of Senator Lewis, the bill was ordered to be engrossed and read by title the third time.

Senator Lewis moved that the Rules be suspended and the third reading of the title of **S.B. 1362** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1362, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--11. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Petersen, Saslaw, Spruill, Stanley, Surovell--27.

NAYS--Chase, DeSteph, Dunnavant, Hanger, McDougle, Obenshain, Pillion, Reeves, Ruff, Suetterlein, Vogel--11.

RULE 36--0.

SENATE BILLS ON SECOND READING

Senator Saslaw moved that the engrossment of the Senate bills that follow be considered en bloc.

The motion was agreed to.

Senator Saslaw moved that the following Senate bills, the titles of the bills having been printed in the Calendar for their second reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

S.B. 1265 (one thousand two hundred sixty-five).

S.B. 1404 (one thousand four hundred four).

S.B. 1444 (one thousand four hundred forty-four).

The motion was agreed to.

S.B. 1265 (one thousand two hundred sixty-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 62.1-44.15:37.1 and 62.1-44.15:58.1 of the Code of Virginia, relating to natural gas pipelines; stop work orders.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

S.B. 1444 (one thousand four hundred forty-four) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Privileges and Elections was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 24.2-947.11 of the Code of Virginia, relating to filing of campaign finance reports; special report of pre-legislative session contributions.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

On motion of Senator Saslaw, the following Senate bills were ordered en bloc to be engrossed and read by title the third time:

S.B. 1265 (one thousand two hundred sixty-five) as amended.

S.B. 1404 (one thousand four hundred four).

S.B. 1444 (one thousand four hundred forty-four) as amended.

Senator Saslaw moved that the Rules be suspended and the third reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

S.B. 1265 (one thousand two hundred sixty-five).

S.B. 1404 (one thousand four hundred four).

S.B. 1444 (one thousand four hundred forty-four).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

Senator Saslaw moved that the passage of the Senate bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following Senate bills were passed en bloc with their titles:

S.B. 1265 (one thousand two hundred sixty-five).

S.B. 1404 (one thousand four hundred four).

S.B. 1444 (one thousand four hundred forty-four).

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1109 (one thousand one hundred nine) was read by title the second time.

The following amendments proposed by the Committee on Privileges and Elections were offered:

1. Line 11, introduced, after November

strike

2021

insert

2022

2. After line 34, introduced

insert

2. That the provisions of this act shall not become effective unless reenacted by the 2022 Session of the General Assembly.

The reading of the amendments was waived.

On motion of Senator Stanley, the amendments were agreed to.

On motion of Senator Stanley, the bill was ordered to be engrossed and read by title the third time.

- **S.B. 1153** (one thousand one hundred fifty-three) was read by title the second time and, on motion of Senator Suetterlein, was ordered to be engrossed and read by title the third time.
 - S.B. 1164 (one thousand one hundred sixty-four) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 10.1-1400 of the Code of Virginia, relating to advanced recycling; definition.

The reading of the substitute was waived.

On motion of Senator Hanger, the substitute was agreed to.

On motion of Senator Hanger, the bill was ordered to be engrossed and read by title the third time.

S.B. 1209 (one thousand two hundred nine) was read by title the second time.

The following amendment proposed by the Committee on the Judiciary was offered:

1. Line 44, introduced, after \$500,000.

strike

It shall be a valid defense if the general contractor obtains a written certification

insert

As evidence a general contractor may offer a written certification, under oath,

The reading of the amendment was waived.

On motion of Senator Petersen, the amendment was agreed to.

On motion of Senator Petersen, the bill was ordered to be engrossed and read by title the third time.

S.B. 1311 (one thousand three hundred eleven) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 62.1-44.15:81 of the Code of Virginia, relating to water quality standards; modification of permits and certifications.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

Senator Mason offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 62.1-44.15:81 of the Code of Virginia, relating to water quality standards; modification of permits and certifications.

Senator Mason withdrew the substitute.

On motion of Senator McClellan, the bill was ordered to be engrossed and read by title the third time.

S.B. 1319 (one thousand three hundred nineteen) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to study waste control and recycling; permits.

The reading of the substitute was waived.

On motion of Senator Hashmi, the substitute was agreed to.

Senator Hanger offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to study waste control and recycling; permits.

Senator Hanger withdrew the substitute.

On motion of Senator Hashmi, the bill was ordered to be engrossed and read by title the third time.

S.B. 1354 (one thousand three hundred fifty-four) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14 of the Code of Virginia, relating to Chesapeake Bay Phase III Watershed Improvement Plan; nutrient removal; regulations.

The reading of the substitute was waived.

On motion of Senator Hanger, the substitute was agreed to.

The following amendments proposed by the Committee on Finance and Appropriations to the substitute were offered:

1. Line 18, substitute, after as

strike

the remainder of line 18 and through reach on line 19

2. Line 20, substitute, after I

strike

the remainder of line 20 and all of lines 21 through 26

The reading of the amendments was waived.

On motion of Senator Hanger, the amendments were agreed to.

On motion of Senator Hanger, the bill was ordered to be engrossed and read by title the third time.

S.B. 1360 (one thousand three hundred sixty) was read by title the second time.

Senator Obenshain moved that S.B. 1360 be recommitted to the Committee on the Judiciary.

The question was put on recommitting **S.B. 1360** to the Committee on the Judiciary.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--20. NAYS--18. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Lewis, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--20.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Locke, Lucas, Marsden, Mason, McClellan, McPike, Saslaw, Spruill, Surovell--18.

RULE 36--0.

- S.B. 1360 was recommitted to the Committee on the Judiciary.
- S.B. 1374 (one thousand three hundred seventy-four) was read by title the second time.

The following amendment proposed by the Committee on Agriculture, Conservation and Natural Resources was offered:

1. Line 21, introduced, after *Districts*, insert

the Virginia Forestry Association,

The reading of the amendment was waived.

On motion of Senator Lewis, the amendment was agreed to.

On motion of Senator Lewis, the bill was ordered to be engrossed and read by title the third time.

S.B. 1411 (one thousand four hundred eleven) was read by title the second time and, on motion of Senator Lucas, was ordered to be engrossed and read by title the third time.

Senator Saslaw moved that the Rules be suspended and the third reading of the titles of the following Senate bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- S.B. 1109 (one thousand one hundred nine).
- S.B. 1153 (one thousand one hundred fifty-three).
- **S.B.** 1164 (one thousand one hundred sixty-four).
- S.B. 1209 (one thousand two hundred nine).
- **S.B. 1311** (one thousand three hundred eleven).
- **S.B.** 1319 (one thousand three hundred nineteen).
- S.B. 1354 (one thousand three hundred fifty-four).
- **S.B.** 1374 (one thousand three hundred seventy-four).
- **S.B. 1411** (one thousand four hundred eleven).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1109 (one thousand one hundred nine), on motion of Senator Stanley, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--7. RULE 36--0.

YEAS--Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Edwards, Favola, Hanger, Hashmi, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--31.

NAYS--Barker, Deeds, Ebbin, Howell, Locke, McPike, Petersen--7.

RULE 36--0.

S.B. 1153 (one thousand one hundred fifty-three), on motion of Senator Suetterlein, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--37.

NAYS--Ebbin--1.

RULE 36--0.

S.B. 1164 (one thousand one hundred sixty-four), on motion of Senator Hanger, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Hanger, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--34.

NAYS--Favola, Hashmi, Locke, McClellan--4.

RULE 36--0.

S.B. 1209 (one thousand two hundred nine), on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Edwards, Hanger, Kiggans, Lewis, Marsden, McDougle, Morrissey, Newman, Obenshain, Peake, Petersen, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell--24.

NAYS--Boysko, Ebbin, Favola, Hashmi, Howell, Locke, Lucas, Mason, McClellan, McPike, Norment, Pillion, Vogel--13.

RULE 36--0.

S.B. 1311 (one thousand three hundred eleven), on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows: YEAS--20. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--0.

STATEMENT ON VOTE

Senator Deeds stated that he was recorded as not voting on the question of the passage of S.B. 1311, whereas he intended to vote yea.

S.B. 1319 (one thousand three hundred nineteen), on motion of Senator Hashmi, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--16.

RULE 36--0.

S.B. 1354 (one thousand three hundred fifty-four), on motion of Senator Hanger, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1374 (one thousand three hundred seventy-four), on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1411 (one thousand four hundred eleven), on motion of Senator Lucas, was passed with its title.

-478-

The recorded vote is as follows: YEAS--35. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--35.

NAYS--Chase, Obenshain, Suetterlein--3. RULE 36--0.

SENATE JOINT RESOLUTIONS ON SECOND READING

S.J.R. 275 (two hundred seventy-five) was read by title the second time.

The following amendments proposed by the Committee on Privileges and Elections were offered:

1. Line 20, introduced, after with

strike

equal

insert

equitable

2. Line 28, introduced, after with

strike

equal

insert

equitable

The reading of the amendments was waived.

On motion of Senator Stanley, the amendments were agreed to.

On motion of Senator Stanley, the joint resolution was ordered to be engrossed and read by title the third time.

Senator Stanley moved that the Rules be suspended and the third reading of the title of **S.J.R. 275** be waived.

The motion was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

SENATE JOINT RESOLUTION NO. 275

Proposing amendments to Section 1 and Section 2 of Article VIII of the Constitution of Virginia, relating to public schools of the Commonwealth; equal educational opportunities.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendments to the Constitution of Virginia be, and the same hereby are, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 1 and Section 2 of Article VIII of the Constitution of Virginia as follows:

ARTICLE VIII EDUCATION

Section 1. Public schools of high quality to be maintained.

The General Assembly shall provide for a system of free public elementary and secondary schools with [equal equitable] educational opportunities for all children of school age throughout the Commonwealth, and shall seek to ensure that an educational program of high quality is established and continually maintained.

Section 2. Standards of quality; State and local support of public schools.

Standards of quality for the several school divisions shall be determined and prescribed from time to time by the Board of Education, subject to revision only by the General Assembly.

The General Assembly shall determine the manner in which funds are to be provided for the cost of maintaining an educational program meeting the prescribed standards of quality, shall ensure that all children of school age are provided with [equal equitable] educational opportunities, and shall provide for the apportionment of the cost of such program between the Commonwealth and the local units of government comprising such school divisions. Each unit of local government shall provide its portion of such cost by local taxes or from other available funds.

S.J.R. 275, on motion of Senator Stanley, was agreed to.

The recorded vote is as follows: YEAS--34. NAYS--1. RULE 36--1.

YEAS--Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--34.

NAYS--Barker--1.

RULE 36--Petersen--1.

S.J.R. 270 (two hundred seventy) was read by title the second time.

The amendment in the nature of a substitute proposed by the Committee on Privileges and Elections was offered, having been printed separately, with its title reading as follows:

Proposing an amendment to Section 15-A of Article I of the Constitution of Virginia, relating to marriage; repeal of same-sex marriage prohibition; affirmative right to marry.

The reading of the substitute was waived.

Senator Ebbin moved that the substitute be agreed to.

The question was put on agreeing to the substitute.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--19. NAYS--19. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Saslaw, Surovell--19.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Suetterlein, Vogel--19.

RULE 36--0.

There being an equal division, the President cast his vote in accordance with Article V, Section 14, of the Constitution.

The President voted yea.

The final vote is as follows:

YEAS--20. NAYS--19. RULE 36--0.

The substitute was agreed to.

On motion of Senator Ebbin, the joint resolution was ordered to be engrossed and read by title the third time.

S.J.R. 271 (two hundred seventy-one) was read by title the second time and, on motion of Senator Ebbin, was ordered to be engrossed and read by title the third time.

S.J.R. 272 (two hundred seventy-two) was read by title the second time.

Senator Surovell offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters and the right to vote; persons not entitled to vote.

On motion of Senator Surovell, the reading of the substitute was waived.

On motion of Senator Surovell, the substitute was agreed to.

On motion of Senator Locke, the joint resolution was ordered to be engrossed and read by title the third time.

S.J.R. 289 (two hundred eighty-nine) was read by title the second time and, on motion of Senator Mason, was ordered to be engrossed and read by title the third time.

S.J.R. 290 (two hundred ninety) was read by title the second time.

Senator McPike offered the following amendments:

1. Line 29, introduced, after *Assembly* insert

and the Governor jointly

2. Line 30, introduced, after *Assembly* insert

and the Governor jointly

On motion of Senator McPike, the reading of the amendments was waived.

On motion of Senator McPike, the amendments were agreed to.

On motion of Senator McPike, the joint resolution was ordered to be engrossed and read by title the third time.

S.J.R. 310 (three hundred ten) was passed by temporarily.

S.J.R. 322 (three hundred twenty-two) was read by title the second time and, on motion of Senator DeSteph, was ordered to be engrossed and read by title the third time.

MEMORIAL RESOLUTION

S.J.R. 393 (three hundred ninety-three) was taken up as follows:

SENATE JOINT RESOLUTION NO. 393

Celebrating the life of the Honorable Robert S. Bloxom, Sr.

WHEREAS, the Honorable Robert S. Bloxom, Sr., a consummate public servant who was known for his commitment to the restoration of the Chesapeake Bay, represented the residents of the Eastern Shore as a longtime member of the House of Delegates and became Virginia's first Secretary of Agriculture and Forestry, died on December 13, 2020; and

WHEREAS, Robert "Bob" Bloxom ran the family business, Bloxom Auto Supply, for many years, serving the members of the Mappsville community and the Eastern Shore with fairness and integrity; and

WHEREAS, desirous to be of further service, Bob Bloxom ran for and was elected to the House of Delegates and represented the residents of the 100th District for 13 terms from 1978 through 2003; and

WHEREAS, during his tenure as a state legislator, Bob Bloxom introduced or supported legislation to preserve the health of the Commonwealth's waterways, establish the commercial spaceport on Wallops Island, and create a dental program for school children in need; and

WHEREAS, Bob Bloxom supported the creation of Kiptopeke State Park, the Commonwealth's first new state park in two decades, and helped provide state funds for the opening of the Eastern Shore Farmer's Market; and

WHEREAS, Bob Bloxom offered his expertise to the committees on Appropriations, Commerce and Labor, and Agriculture, Chesapeake and Natural Resources and served as chair of the multistate Chesapeake Bay Commission; and

WHEREAS, admired for his honesty and dedication to his constituents, Bob Bloxom worked to build bipartisan respect and foster a culture of mutual respect between delegates from both parties; and

WHEREAS, in December 2004, Bob Bloxom was appointed by Governor Mark Warner to the newly created cabinet position of Secretary of Agriculture and Forestry; he continued to serve in that capacity under Governor Tim Kaine until his well-earned retirement from public life in January 2010; and

WHEREAS, as Secretary of Agriculture and Forestry, Bob Bloxom visited every county in Virginia and promoted the Commonwealth's exports in other countries; he also supported farmland preservation efforts and oversaw the completion of a study that estimated the value of the Commonwealth's agricultural industries at \$79 billion; and

WHEREAS, at the local level, Bob Bloxom volunteered his time and leadership to Eastern Shore Community College, Eastern Shore Public Library, the Eastern Shore Chamber of Commerce, and the Chesapeake Bay Bridge-Tunnel Commission, but his greatest joy in life was his family, and he relished every opportunity to support his beloved grandchildren at sporting events and activities; and

WHEREAS, Bob Bloxom will be fondly remembered and greatly missed by his wife, Pat; his children Lee and Robert, Jr., and their families; and numerous other family members, friends, and colleagues on both sides of the aisle; now, therefore, be it

RESOLVED by the Senate, the House of Delegates concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Robert S. Bloxom, Sr., a respected public servant who strengthened the Eastern Shore; and, be it

RESOLVED FURTHER, That the Clerk of the Senate prepare a copy of this resolution for presentation to the family of the Honorable Robert S. Bloxom, Sr., as an expression of the General Assembly's admiration for his achievements on behalf of all residents of the Commonwealth and respect for his memory.

S.J.R. 393, on motion of Senator Lewis, was ordered to be engrossed and was agreed to by a unanimous standing vote.

SENATE JOINT RESOLUTIONS ON SECOND READING

S.J.R. 310 (three hundred ten) was taken up and read by title the second time.

The following amendment proposed by the Committee on Privileges and Elections was offered:

Line 23, introduced, after exceeding strike
 thirty
 insert
 fourteen

The reading of the amendment was waived.

Senator Saslaw moved that the amendment be rejected.

The question was put on agreeing to the amendment.

The amendment was rejected.

Senator Deeds offered the following amendment:

1. Line 23, introduced, after days

insert

, provided that in an odd-numbered year the session may be extended for a period not exceeding 14 days

On motion of Senator Saslaw, the reading of the amendment was waived.

On motion of Senator Saslaw, the amendment was agreed to.

On motion of Senator Saslaw, the joint resolution was ordered to be engrossed and read by title the third time.

Senator Saslaw moved that the Rules be suspended and the third reading of the titles of the following Senate joint resolutions be waived:

S.J.R. 270 (two hundred seventy).

S.J.R. 271 (two hundred seventy-one).

S.J.R. 272 (two hundred seventy-two).

S.J.R. 289 (two hundred eighty-nine).

S.J.R. 290 (two hundred ninety).

S.J.R. 310 (three hundred ten).

S.J.R. 322 (three hundred twenty-two).

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

S.J.R. 270 (two hundred seventy) was taken up.

SENATE JOINT RESOLUTION NO. 270

Proposing an amendment to Section 15-A of Article I of the Constitution of Virginia, relating to marriage; repeal of same-sex marriage prohibition; affirmative right to marry.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 15-A of Article I of the Constitution of Virginia as follows:

ARTICLE I BILL OF RIGHTS

Section 15-A. Marriage Fundamental right to marry.

That only a union between one man and one woman may be a marriage valid in or recognized by this Commonwealth and its political subdivisions the right to marry is a fundamental right, inherent in the liberty of persons, and marriage is one of the vital personal rights essential to the orderly pursuit of happiness.

This Commonwealth and its political subdivisions shall not create or recognize a legal status for relationships of unmarried individuals that intends to approximate the design, qualities, significance, or effects of marriage. Nor shall this Commonwealth or its political subdivisions create or recognize another union, partnership, or other legal status to which is assigned the rights, benefits, obligations, qualities, or effects of marriage and agents shall issue marriage licenses, recognize marriages, and treat all marriages equally under the law regardless of the sex or gender of the parties to the marriage.

Religious organizations and clergy acting in their religious capacity shall have the right to refuse to perform any marriage.

S.J.R. 270, on motion of Senator Ebbin, was agreed to.

The recorded vote is as follows:

YEAS--24. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Surovell, Vogel--24.

NAYS--Chase, Cosgrove, DeSteph, Hanger, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley--12.

RULE 36--0.

S.J.R. 271 (two hundred seventy-one) was taken up.

SENATE JOINT RESOLUTION NO. 271

Proposing an amendment to Section 1 of Article V of the Constitution of Virginia, relating to executive power; Governor's term of office.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 1 of Article V of the Constitution of Virginia as follows:

ARTICLE V EXECUTIVE

Section 1. Executive power; Governor's term of office.

The chief executive power of the Commonwealth shall be vested in a Governor. He shall hold office for a term commencing upon his inauguration on the Saturday after the second Wednesday in January, next succeeding his election, and ending in the fourth year thereafter immediately upon the inauguration of his successor. He shall be ineligible to the same office for the term next succeeding that for which he was elected, and to any other office during his term of service. No person shall be elected to the office of Governor more than twice, and no person who has held the office of Governor, or acted as Governor, for more than two years of a term to which some other person was elected Governor shall be elected to the office of Governor more than once. The authorization to serve two terms in succession shall be applicable to persons elected to serve as Governor in 2025 and thereafter.

Senator Ebbin moved that S.J.R. 271 be agreed to.

The question was put on agreeing to S.J.R. 271.

S.J.R. 271 was rejected.

The recorded vote is as follows:

YEAS--12. NAYS--26. RULE 36--0.

YEAS--Bell, Deeds, Ebbin, Edwards, Hashmi, Howell, Locke, Lucas, McClellan, Saslaw, Surovell, Vogel--12.

NAYS--Barker, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Favola, Hanger, Kiggans, Lewis, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Suetterlein--26.

RULE 36--0.

S.J.R. 272 (two hundred seventy-two) was taken up.

SENATE JOINT RESOLUTION NO. 272

Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters and the right to vote; persons not entitled to vote.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 1 of Article II of the Constitution of Virginia as follows:

ARTICLE II FRANCHISE AND OFFICERS

Section 1. Qualifications of voters.

In elections by the people, the qualifications of voters shall be as follows: Each voter shall be a citizen of the United States, shall be eighteen years of age, shall fulfill the residence requirements set forth in this section, and shall be registered to vote pursuant to this article. No Every person who meets these qualifications shall have the fundamental right to vote in the Commonwealth, and such right shall not be abridged by law, except that no person who has been convicted of a felony shall be qualified entitled to vote unless his civil rights have been restored by the Governor or other appropriate authority. As prescribed by law during any period of imprisonment, no person adjudicated to be mentally incompetent shall be qualified to vote until his competency has been reestablished and no person who has been adjudicated by a court of competent jurisdiction to lack the capacity to understand the act of voting shall be entitled to vote until his capacity has been reestablished as prescribed by law.

The residence requirements shall be that each voter shall be a resident of the Commonwealth and of the precinct where he votes. Residence, for all purposes of qualification to vote, requires both domicile and a place of abode. The General Assembly may provide for persons who are employed overseas, and their spouses and dependents residing with them, and who are qualified to vote except for relinquishing their place of abode in the Commonwealth while overseas, to vote in the Commonwealth subject to conditions and time limits defined by law. The General Assembly may provide for persons who are qualified to vote except for having moved their residence from one precinct to another within the Commonwealth to continue to vote in a former precinct subject to conditions and time limits defined by law. The General Assembly may also provide, in elections for President and Vice President of the United States, alternatives to registration for new residents of the Commonwealth.

Any person who will be qualified with respect to age to vote at the next general election shall be permitted to register in advance and also to vote in any intervening primary or special election.

S.J.R. 272, on motion of Senator Locke, was agreed to.

The recorded vote is as follows:

YEAS--22. NAYS--15. RULE 36--1.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--15.

RULE 36--Norment--1.

S.J.R. 289 (two hundred eighty-nine) was taken up.

SENATE JOINT RESOLUTION NO. 289

Proposing an amendment to Section 9 of Article VII of the Constitution of Virginia, relating to local government; sale of property; authority to grant perpetual easements to units of government.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 9 of Article VII of the Constitution of Virginia as follows:

ARTICLE VII LOCAL GOVERNMENT

Section 9. Sale of property and granting of franchises by cities and towns.

No rights of a city or town in and to its waterfront, wharf property, public landings, wharves, docks, streets, avenues, parks, bridges, or other public places, or its gas, water, or electric works shall be sold except by an ordinance or resolution passed by a recorded affirmative vote of three fourths of all members elected to the governing body.

No franchise, lease, or right of any kind to use any such public property or any other public property or easement of any description in a manner not permitted to the general public shall be granted for a longer period than forty years, except for air rights together with easements for columns of support, which may be granted for a period not exceeding sixty years. Before granting any such franchise or privilege for a term in excess of five years, except for a trunk railway, the city or town shall, after due advertisement, publicly receive bids therefor. Such grant, and any contract in pursuance thereof, may provide that upon the termination of the grant, the plant as well as the property, if any, of the grantee in the streets, avenues, and other public places shall thereupon, without compensation to the grantee, or upon the payment of a fair valuation therefor, become the property of the said city or town; but the grantee shall be entitled to no payment by reason of the value of the franchise. Any such plant or property acquired by a city or town may be sold or leased or, unless prohibited by general law, maintained, controlled, and operated by such city or town. Every such grant shall specify the mode of determining any valuation therein provided for and shall make adequate provisions by way of forfeiture of the grant, or otherwise, to secure efficiency of public service at reasonable rates and the maintenance of the property in good order throughout the term of the grant.

Notwithstanding any other provision of this section, an easement on public property may be granted in perpetuity to a public body, political subdivision, or authority of the Commonwealth of Virginia or to the United States of America or any of its departments or agencies. The requirement to advertise and publicly receive bids shall not apply to easements conveyed to any such governmental entity.

S.J.R. 289, on motion of Senator Mason, was agreed to.

The recorded vote is as follows: YEAS--21, NAYS--17, RULE 36--0.

YEAS--Barker, Bell, Boysko, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Newman, Norment, Peake, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Morrissey, Obenshain, Petersen, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--0.

S.J.R. 290 (two hundred ninety) was taken up.

SENATE JOINT RESOLUTION NO. 290

Proposing an amendment to Section 5 of Article IV of the Constitution of Virginia and proposing an amendment to the Constitution of Virginia by adding in Article IV a section numbered 5-A, relating to legislative compensation; Legislative Compensation Commission.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendments to the Constitution of Virginia be, and the same hereby are, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 5 of Article IV of the Constitution of Virginia and amend the Constitution of Virginia by adding in Article IV a section numbered 5-A as follows:

ARTICLE IV LEGISLATURE

Section § 5. Compensation; election to civil office of profit.

The members of the General Assembly shall receive such salary and allowances as may be prescribed by law compensation as determined by the Legislative Compensation Commission established under § 5-A of Article IV, but no increase in salary shall take effect for a given member until after the end of the term for which he was elected. Other than the compensation determined by the Legislative Compensation Commission, the members of the General Assembly shall receive no other compensation from the Commonwealth. No member during the term for which he shall have been elected shall be elected by the General Assembly to any civil office of profit in the Commonwealth.

Section 5-A. Legislative Compensation Commission.

- (a) Beginning on July 1, 2026, and quadrennially thereafter, the General Assembly [and the Governor jointly] shall appoint a temporary Legislative Compensation Commission (the Commission). The General Assembly [and the Governor jointly] shall determine the membership of the Commission. No member or employee of the General Assembly shall serve as a member of the Commission. The Commission shall be charged with determining compensation for members of the General Assembly, which shall include only the following:
 - (1) Regular salary;
 - (2) Additional salary for attending meetings;
 - (3) Expense allowance; and
 - (4) Retirement benefits.
 - (b) In determining compensation, the Commission shall:
 - (1) Review the current compensation of the members of the General Assembly;
 - (2) Examine the Commonwealth's history of legislative compensation;
 - (3) Review compensation for legislative service in other states;

- (4) Collect and analyze data on comparative compensation in the private and public sectors;
- (5) Assess various state methodologies in determining reasonable legislative compensation, including the tying of compensation to certain indexes or economic indicators; and
- (6) Seek the assistance and input of members of the General Assembly and citizens of the Commonwealth by conducting surveys and holding public hearings, as deemed appropriate by the Commission.
- (c) The Commission may determine a different compensation for a member of the General Assembly based on the duties of his office.
 - (d) The Commission shall not have the power to determine the compensation of:
 - (1) Any elected official other than the members of the General Assembly;
 - (2) Legislative staff; or
 - (3) Any other employee of the Commonwealth.
- (e) The Commission shall determine whether any adjustments to compensation are needed and shall submit its determinations to the Governor and General Assembly on or before December 1 of the year preceding the election of all members of the General Assembly. The adjustments, if any, shall be effective for the next legislative term in office following submission of the report. The compensation, as determined by the Commission, shall be appropriated in a general appropriation law.
- (f) The General Assembly shall enact such laws and make such appropriations as may be necessary to effectuate the provisions of this section and the determinations of the Commission.
- (g) Upon the submission of its report pursuant to subsection (e), the Commission shall cease to exist until a new Commission is appointed pursuant to subsection (a).

Senator McPike moved that **S.J.R. 290** be agreed to.

The question was put on agreeing to S.J.R. 290.

S.J.R. 290 was rejected, having failed to receive the necessary affirmative votes required by Article XII, Section 1 of the Constitution.

The recorded vote is as follows:

YEAS--19. NAYS--19. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Saslaw, Spruill, Surovell--19.

NAYS--Chase, Deeds, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--19.

RULE 36--0.

S.J.R. 310 (three hundred ten) was taken up.

SENATE JOINT RESOLUTION NO. 310

Proposing an amendment to Section 6 of Article IV of the Constitution of Virginia, relating to legislative sessions; length of regular sessions convened in odd-numbered years.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 6 of Article IV of the Constitution of Virginia as follows:

ARTICLE IV LEGISLATURE

Section 6. Legislative sessions.

The General Assembly shall meet once each year on the second Wednesday in January. Except as herein provided for reconvened sessions, no regular session of the General Assembly convened in an even-numbered year shall continue longer than sixty days; no regular session of the General Assembly convened in an odd-numbered year shall continue longer than thirty forty-six days; but with the concurrence of two-thirds of the members elected to each house, any regular session may be extended for a period not exceeding thirty days [, provided that in an odd-numbered year the session may be extended for a period not exceeding fourteen days]. Neither house shall, without the consent of the other, adjourn to another place, nor for more than three days.

The Governor may convene a special session of the General Assembly when, in his opinion, the interest of the Commonwealth may require and shall convene a special session upon the application of two-thirds of the members elected to each house.

The General Assembly shall reconvene on the sixth Wednesday after adjournment of each regular or special session for the purpose of considering bills which may have been returned by the Governor with recommendations for their amendment and bills and items of appropriation bills which may have been returned by the Governor with his objections. No other business shall be considered at a reconvened session. Such reconvened session shall not continue longer than three days unless the session be extended, for a period not exceeding seven additional days, upon the vote of the majority of the members elected to each house. The General Assembly may provide, by a joint resolution approved during a regular or special session by the vote of the majority of the members elected to each house, that it shall reconvene on a date after the sixth Wednesday after adjournment of the regular or special session but no later than the seventh Wednesday after adjournment.

S.J.R. 310, on motion of Senator Saslaw, was agreed to.

The recorded vote is as follows:

YEAS--23. NAYS--14. RULE 36--1.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--14.

RULE 36--Norment--1.

S.J.R. 322 (three hundred twenty-two) was taken up.

Senator DeSteph moved that S.J.R. 322 be agreed to.

The question was put on agreeing to S.J.R. 322.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Lewis, Mason, McClellan, McDougle, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--21.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Locke, Lucas, Marsden, McPike, Norment, Saslaw, Spruill, Surovell--17.

RULE 36--0.

S.J.R. 322 was agreed to.

MEMORIAL AND COMMENDING RESOLUTIONS

On motion of Senator Saslaw, the following memorial and commending resolutions were referred to the Committee on Rules and continued to the 2021 Special Session:

```
H.J.R. 639 (six hundred thirty-nine).
```

H.J.R. 640 (six hundred forty).

H.J.R. 641 (six hundred forty-one).

H.J.R. 642 (six hundred forty-two).

H.J.R. 643 (six hundred forty-three).

H.J.R. 644 (six hundred forty-four).

H.J.R. 645 (six hundred forty-five). H.J.R. 646 (six hundred forty-six).

H.J.R. 647 (six hundred forty-seven).

H.J.R. 648 (six hundred forty-eight).

H.J.R. 649 (six hundred forty-nine).

H.J.R. 650 (six hundred fifty).

H.J.R. 651 (six hundred fifty-one).

H.J.R. 652 (six hundred fifty-two).

H.J.R. 653 (six hundred fifty-three).

H.J.R. 654 (six hundred fifty-four).

- H.J.R. 655 (six hundred fifty-five).
- H.J.R. 656 (six hundred fifty-six).
- H.J.R. 657 (six hundred fifty-seven).
- H.J.R. 658 (six hundred fifty-eight).
- H.J.R. 659 (six hundred fifty-nine).
- H.J.R. 660 (six hundred sixty).
- H.J.R. 661 (six hundred sixty-one).
- H.J.R. 662 (six hundred sixty-two).
- H.J.R. 663 (six hundred sixty-three).
- H.J.R. 664 (six hundred sixty-four).
- H.J.R. 665 (six hundred sixty-five).
- H.J.R. 666 (six hundred sixty-six).
- H.J.R. 667 (six hundred sixty-seven).
- H.J.R. 668 (six hundred sixty-eight).
- H.J.R. 669 (six hundred sixty-nine).
- H.J.R. 670 (six hundred seventy).
- H.J.R. 671 (six hundred seventy-one).
- H.J.R. 672 (six hundred seventy-two).
- H.J.R. 673 (six hundred seventy-three).
- H.J.R. 674 (six hundred seventy-four).
- H.J.R. 675 (six hundred seventy-five).
- **H.J.R.** 676 (six hundred seventy-six).
- H.J.R. 677 (six hundred seventy-seven).
- **H.J.R.** 678 (six hundred seventy-eight).
- H.J.R. 679 (six hundred seventy-nine).
- H.J.R. 680 (six hundred eighty).
- H.J.R. 681 (six hundred eighty-one).
- H.J.R. 682 (six hundred eighty-two).
- H.J.R. 683 (six hundred eighty-three).
- H.J.R. 684 (six hundred eighty-four).
- H.J.R. 685 (six hundred eighty-five).
- **H.J.R. 686** (six hundred eighty-six). **H.J.R. 687** (six hundred eighty-seven).
- H.J.R. 688 (six hundred eighty-eight).
- H.J.R. 689 (six hundred eighty-rine).
- **H.J.R. 690** (six hundred ninety).
- H.J.R. 691 (six hundred ninety-one).
- H.J.R. 692 (six hundred ninety-two).
- H.J.R. 693 (six hundred ninety-three).
- H.J.R. 694 (six hundred ninety-four).
- H.J.R. 695 (six hundred ninety-five).
- H.J.R. 696 (six hundred ninety-six).
- H.J.R. 697 (six hundred ninety-seven).
- H.J.R. 698 (six hundred ninety-eight).
- H.J.R. 699 (six hundred ninety-nine).
- H.J.R. 700 (seven hundred).
- H.J.R. 701 (seven hundred one).
- H.J.R. 702 (seven hundred two).
- H.J.R. 703 (seven hundred three).
- H.J.R. 704 (seven hundred four).
- H.J.R. 705 (seven hundred five).
- H.J.R. 706 (seven hundred six).

- H.J.R. 707 (seven hundred seven).
- H.J.R. 708 (seven hundred eight).
- H.J.R. 709 (seven hundred nine).
- H.J.R. 710 (seven hundred ten).
- H.J.R. 711 (seven hundred eleven).
- H.J.R. 712 (seven hundred twelve).
- H.J.R. 713 (seven hundred thirteen).
- H.J.R. 714 (seven hundred fourteen).
- **H.J.R.** 715 (seven hundred fifteen).
- H.J.R. 716 (seven hundred sixteen).
- H.J.R. 717 (seven hundred seventeen).
- H.J.R. 718 (seven hundred eighteen).
- H.J.R. 719 (seven hundred nineteen).
- **H.J.R.** 720 (seven hundred twenty).
- H.J.R. 721 (seven hundred twenty-one).
- H.J.R. 722 (seven hundred twenty-two).
- H.J.R. 723 (seven hundred twenty-three).
- H.J.R. 724 (seven hundred twenty-four).
- H.J.R. 725 (seven hundred twenty-five).
- **H.J.R.** 726 (seven hundred twenty-six).
- H.J.R. 727 (seven hundred twenty-seven).
- H.J.R. 728 (seven hundred twenty-eight).
- H.J.R. 729 (seven hundred twenty-nine).
- **H.J.R.** 730 (seven hundred thirty).
- H.J.R. 731 (seven hundred thirty-one).
- H.J.R. 732 (seven hundred thirty-two).
- H.J.R. 733 (seven hundred thirty-three).
- **H.J.R. 734** (seven hundred thirty-four). H.J.R. 735 (seven hundred thirty-five).
- H.J.R. 736 (seven hundred thirty-six).
- H.J.R. 737 (seven hundred thirty-seven).
- H.J.R. 738 (seven hundred thirty-eight).
- H.J.R. 739 (seven hundred thirty-nine).
- H.J.R. 740 (seven hundred forty).
- H.J.R. 741 (seven hundred forty-one).
- H.J.R. 742 (seven hundred forty-two).
- H.J.R. 743 (seven hundred forty-three).
- H.J.R. 744 (seven hundred forty-four).
- H.J.R. 745 (seven hundred forty-five).
- H.J.R. 746 (seven hundred forty-six).
- H.J.R. 747 (seven hundred forty-seven).
- H.J.R. 748 (seven hundred forty-eight).
- H.J.R. 749 (seven hundred forty-nine).
- H.J.R. 750 (seven hundred fifty).
- H.J.R. 751 (seven hundred fifty-one).
- H.J.R. 752 (seven hundred fifty-two).
- H.J.R. 753 (seven hundred fifty-three).
- H.J.R. 754 (seven hundred fifty-four).
- H.J.R. 755 (seven hundred fifty-five). H.J.R. 756 (seven hundred fifty-six).
- H.J.R. 757 (seven hundred fifty-seven).
- H.J.R. 758 (seven hundred fifty-eight).

- H.J.R. 759 (seven hundred fifty-nine).
- H.J.R. 760 (seven hundred sixty).
- H.J.R. 761 (seven hundred sixty-one).
- H.J.R. 762 (seven hundred sixty-two).
- H.J.R. 763 (seven hundred sixty-three).
- H.J.R. 764 (seven hundred sixty-four).
- **H.J.R. 765** (seven hundred sixty-five).
- H.J.R. 766 (seven hundred sixty-six).
- H.J.R. 767 (seven hundred sixty-seven).
- H.J.R. 768 (seven hundred sixty-eight).
- H.J.R. 769 (seven hundred sixty-nine).
- H.J.R. 770 (seven hundred seventy).
- H.J.R. 771 (seven hundred seventy-one).
- H.J.R. 772 (seven hundred seventy-two).
- H.J.R. 773 (seven hundred seventy-three).
- H.J.R. 774 (seven hundred seventy-four).
- H.J.R. 775 (seven hundred seventy-five).
- H.J.R. 776 (seven hundred seventy-six).
- **H.J.R.** 777 (seven hundred seventy-seven).
- **H.J.R.** 778 (seven hundred seventy-eight).
- H.J.R. 779 (seven hundred seventy-nine).
- H.J.R. 780 (seven hundred eighty).
- H.J.R. 781 (seven hundred eighty-one).
- **H.J.R.** 782 (seven hundred eighty-two).
- H.J.R. 783 (seven hundred eighty-three).
- H.J.R. 784 (seven hundred eighty-four).
- **H.J.R.** 785 (seven hundred eighty-five).
- H.J.R. 786 (seven hundred eighty-six).
- H.J.R. 787 (seven hundred eighty-seven).

LEGISLATION CONTINUED

The following report was received from the Clerk:

TO THE MEMBERS OF THE SENATE:

Pursuant to the provisions of Senate Rules 20 (f) and 20 (g) and House Joint Resolution No. 575 of the 2021 Regular Session, certain bills and joint resolutions have been continued to the 2021 Special Session I of the General Assembly in the several Senate committees, as follows:

COMMITTEE ON AGRICULTURE, CONSERVATION AND NATURAL RESOURCES

H.B. 1750, H.B. 1751, H.B. 1760, H.B. 1804, H.B. 1819, H.B. 1833, H.B. 1836, H.B. 1837, H.B. 1855, H.B. 1902, H.B. 1928, H.B. 1958, H.B. 1982, H.B. 1983, H.B. 2030, H.B. 2068, H.B. 2078, H.B. 2129, H.B. 2148, H.B. 2159, H.B. 2187, H.B. 2203, H.B. 2221, H.B. 2275, H.B. 2302, H.B. 2311.

COMMITTEE ON COMMERCE AND LABOR

H.B. 1754, H.B. 1775, H.B. 1786, H.B. 1807, H.B. 1818, H.B. 1822, H.B. 1829, H.B. 1832, H.B. 1834, H.B. 1862, H.B. 1877, H.B. 1881, H.B. 1884, H.B. 1892, H.B. 1896, H.B. 1907, H.B. 1914, H.B. 1923, H.B. 1925, H.B. 1942, H.B. 1964, H.B. 1965, H.B. 1984, H.B. 1985, H.B. 1994, H.B. 2008, H.B. 2032, H.B. 2034, H.B. 2036, H.B. 2037, H.B. 2040, H.B. 2048, H.B. 2049, H.B. 2062, H.B. 2063, H.B. 2121, H.B. 2134, H.B. 2137, H.B. 2160, H.B. 2200, H.B. 2207, H.B. 2219, H.B. 2250, H.B. 2269, H.B. 2282, H.B. 2304, H.B. 2330, H.B. 2332.

COMMITTEE ON EDUCATION AND HEALTH

H.B. 1736, H.B. 1737, H.B. 1747, H.B. 1776, H.B. 1790, H.B. 1798, H.B. 1808, H.B. 1817, H.B. 1823, H.B. 1827, H.B. 1838, H.B. 1865, H.B. 1873, H.B. 1885, H.B. 1894, H.B. 1904, H.B. 1905, H.B. 1909, H.B. 1913, H.B. 1918, H.B. 1930, H.B. 1940, H.B. 1950, H.B. 1953, H.B. 1963, H.B. 1976, H.B. 1980, H.B. 1986, H.B. 1987, H.B. 1988, H.B. 1989, H.B. 1995, H.B. 1998, H.B. 2007, H.B. 2013, H.B. 2019, H.B. 2027, H.B. 2035, H.B. 2039, H.B. 2058, H.B. 2061, H.B. 2079, H.B. 2105, H.B. 2111, H.B. 2116, H.B. 2117, H.B. 2119, H.B. 2120, H.B. 2123, H.B. 2124, H.B. 2135, H.B. 2162, H.B. 2176, H.B. 2182, H.B. 2204, H.B. 2206, H.B. 2218, H.B. 2220, H.B. 2230, H.B. 2238, H.B. 2299, H.B. 2300, H.B. 2305, H.B. 2314, H.B. 2316.

COMMITTEE ON FINANCE AND APPROPRIATIONS

S.B. 1100.

H.B. 1763, H.B. 1774, H.B. 1899, H.B. 1916, H.B. 1935, H.B. 1969, H.B. 1979, H.B. 1999, H.B. 2006, H.B. 2059, H.B. 2060, H.B. 2101, H.B. 2118, H.B. 2165, H.B. 2174, H.B. 2177, H.B. 2178, H.B. 2179, H.B. 2181, H.B. 2185, H.B. 2223, H.B. 2273, H.B. 2293, H.B. 2337.

COMMITTEE ON GENERAL LAWS AND TECHNOLOGY

H.B. 1811, H.B. 1812, H.B. 1816, H.B. 1824, H.B. 1830, H.B. 1842, H.B. 1843, H.B. 1847, H.B. 1848, H.B. 1849, H.B. 1864, H.B. 1876, H.B. 1882, H.B. 1889, H.B. 1891, H.B. 1900, H.B. 1908, H.B. 1931, H.B. 1943, H.B. 1944, H.B. 1967, H.B. 1971, H.B. 1981, H.B. 1993, H.B. 1996, H.B. 2001, H.B. 2003, H.B. 2004, H.B. 2005, H.B. 2014, H.B. 2025, H.B. 2029, H.B. 2031, H.B. 2046, H.B. 2053, H.B. 2072, H.B. 2074, H.B. 2085, H.B. 2130, H.B. 2140, H.B. 2147, H.B. 2155, H.B. 2161, H.B. 2170, H.B. 2171, H.B. 2172, H.B. 2175, H.B. 2202, H.B. 2222, H.B. 2227, H.B. 2229, H.B. 2249, H.B. 2288, H.B. 2307, H.B. 2308, H.B. 2313, H.B. 2320, H.B. 2321, H.B. 2322, H.B. 2327.

COMMITTEE ON THE JUDICIARY

H.B. 1806, H.B. 1814, H.B. 1821, H.B. 1852, H.B. 1853, H.B. 1856, H.B. 1866, H.B. 1867, H.B. 1878, H.B. 1895, H.B. 1897, H.B. 1911, H.B. 1912, H.B. 1936, H.B. 1948, H.B. 1951, H.B. 1991, H.B. 1992, H.B. 2002, H.B. 2009, H.B. 2012, H.B. 2017, H.B. 2018, H.B. 2038, H.B. 2047, H.B. 2055, H.B. 2056, H.B. 2064, H.B. 2073, H.B. 2081, H.B. 2099, H.B. 2110, H.B. 2113, H.B. 2128, H.B. 2132, H.B. 2133, H.B. 2139, H.B. 2150, H.B. 2168, H.B. 2169, H.B. 2190, H.B. 2192, H.B. 2193, H.B. 2194, H.B. 2233, H.B. 2234, H.B. 2236, H.B. 2252, H.B. 2254, H.B. 2258, H.B. 2263, H.B. 2276, H.B. 2290, H.B. 2291, H.B. 2298, H.B. 2310, H.B. 2317, H.B. 2319, H.B. 2331.

H.J.R. 629.

COMMITTEE ON LOCAL GOVERNMENT

H.B. 1749, H.B. 1764, H.B. 1778, H.B. 1783, H.B. 1858, H.B. 1898, H.B. 1919, H.B. 2042, H.B. 2054, H.B. 2091, H.B. 2095, H.B. 2180, H.B. 2186, H.B. 2201, H.B. 2217, H.B. 2257, H.B. 2287, H.B. 2323, H.B. 2326.

COMMITTEE ON PRIVILEGES AND ELECTIONS

H.B. 1810, H.B. 1888, H.B. 1890, H.B. 1921, H.B. 1952, H.B. 1968, H.B. 2020, H.B. 2082, H.B. 2125, H.B. 2198, H.B. 2324.

H.J.R. 555, H.J.R. 582.

COMMITTEE ON REHABILITATION AND SOCIAL SERVICES

H.B. 1805, H.B. 1820, H.B. 1831, H.B. 1845, H.B. 1874, H.B. 1879, H.B. 1932, H.B. 1957, H.B. 1962, H.B. 1973, H.B. 2010, H.B. 2065, H.B. 2070, H.B. 2086, H.B. 2092, H.B. 2098, H.B. 2131, H.B. 2154, H.B. 2166, H.B. 2167, H.B. 2191, H.B. 2197, H.B. 2212, H.B. 2266, H.B. 2312.

COMMITTEE ON TRANSPORTATION

H.B. 1796, H.B. 1801, H.B. 1813, H.B. 1828, H.B. 1841, H.B. 1846, H.B. 1850, H.B. 1851, H.B. 1854, H.B. 1868, H.B. 1887, H.B. 1893, H.B. 1901, H.B. 1903, H.B. 1926, H.B. 1960, H.B. 1961, H.B. 2024, H.B. 2069, H.B. 2071, H.B. 2075, H.B. 2104, H.B. 2138, H.B. 2146, H.B. 2163, H.B. 2216, H.B. 2261, H.B. 2262, H.B. 2284, H.B. 2294, H.B. 2318.

COMMITTEE ON RULES

H.B. 1789, H.B. 1978, H.B. 1990, H.B. 2208, H.B. 2213, H.B. 2271, H.B. 2295.

H.J.R. 522, H.J.R. 525, H.J.R. 526, H.J.R. 527, H.J.R. 537, H.J.R. 538, H.J.R. 542, H.J.R. 549, H.J.R. 562, H.J.R. 563, H.J.R. 567, H.J.R. 572, H.J.R. 578, H.J.R. 579, H.J.R. 583, H.J.R. 596, H.J.R. 604, H.J.R. 605, H.J.R. 606, H.J.R. 639, H.J.R. 640, H.J.R. 641, H.J.R. 642, H.J.R. 643, H.J.R. 644, H.J.R. 645, H.J.R. 646, H.J.R. 647, H.J.R. 648, H.J.R. 649, H.J.R. 650, H.J.R. 651, H.J.R. 652, H.J.R. 653, H.J.R. 654, H.J.R. 655, H.J.R. 656, H.J.R. 657, H.J.R. 658, H.J.R. 659, H.J.R. 660, H.J.R. 661, H.J.R. 662, H.J.R. 663, H.J.R. 664, H.J.R. 665, H.J.R. 666, H.J.R. 667, H.J.R. 668, H.J.R. 669, H.J.R. 670, H.J.R. 671, H.J.R. 672, H.J.R. 673, H.J.R. 674, H.J.R. 675, H.J.R. 676, H.J.R. 677, H.J.R. 678, H.J.R. 679, H.J.R. 680, H.J.R. 681, H.J.R. 682, H.J.R. 683, H.J.R. 684, H.J.R. 685, H.J.R. 686, H.J.R. 687, H.J.R. 688, H.J.R. 689, H.J.R. 690, H.J.R. 691, H.J.R. 692, H.J.R. 693, H.J.R. 694, H.J.R. 695, H.J.R. 696, H.J.R. 697, H.J.R. 698, H.J.R. 699, H.J.R. 700, H.J.R. 701, H.J.R. 702, H.J.R. 703, H.J.R. 704, H.J.R. 705, H.J.R. 706, H.J.R. 707, H.J.R. 708, H.J.R. 709, H.J.R. 710, H.J.R. 711, H.J.R. 712, H.J.R. 713, H.J.R. 714, H.J.R. 715, H.J.R. 716, H.J.R. 717, H.J.R. 718, H.J.R. 719, H.J.R. 720, H.J.R. 721, H.J.R. 722, H.J.R. 723, H.J.R. 724, H.J.R. 725, H.J.R. 726, H.J.R. 727, H.J.R. 728, H.J.R. 729, H.J.R. 730, H.J.R. 731, H.J.R. 732, H.J.R. 733, H.J.R. 734, H.J.R. 735, H.J.R. 736, H.J.R. 737, H.J.R. 738, H.J.R. 739, H.J.R. 740, H.J.R. 741, H.J.R. 742, H.J.R. 743, H.J.R. 744, H.J.R. 745, H.J.R. 746, H.J.R. 747, H.J.R. 748, H.J.R. 749, H.J.R. 750, H.J.R. 751, H.J.R. 752, H.J.R. 753, H.J.R. 754, H.J.R. 755, H.J.R. 756, H.J.R. 757, H.J.R. 758, H.J.R. 759, H.J.R. 760, H.J.R. 761, H.J.R. 762, H.J.R. 763, H.J.R. 764, H.J.R. 765, H.J.R. 766, H.J.R. 767, H.J.R. 768, H.J.R. 769, H.J.R. 770, H.J.R. 771, H.J.R. 772, H.J.R. 773, H.J.R. 774, H.J.R. 775, H.J.R. 776, H.J.R. 777, H.J.R. 778, H.J.R. 779, H.J.R. 780, H.J.R. 781, H.J.R. 782, H.J.R. 783, H.J.R. 784, H.J.R. 785, H.J.R. 786, H.J.R. 787.

Pursuant to the provisions of House Joint Resolution No. 575 of the 2021 Regular Session, certain bills and resolutions have been continued to the 2021 Special Session I of the General Assembly in the several House committees, as follows:

COMMITTEE ON AGRICULTURE, CHESAPEAKE AND NATURAL RESOURCES

S.B. 1115, S.B. 1135, S.B. 1143, S.B. 1161, S.B. 1164, S.B. 1188, S.B. 1193, S.B. 1194, S.B. 1199, S.B. 1210, S.B. 1258, S.B. 1265, S.B. 1274, S.B. 1280, S.B. 1282, S.B. 1290, S.B. 1291, S.B. 1311, S.B. 1319, S.B. 1354, S.B. 1374, S.B. 1390, S.B. 1393, S.B. 1396, S.B. 1402, S.B. 1404, S.B. 1411, S.B. 1412, S.B. 1417, S.B. 1453.

COMMITTEE ON APPROPRIATIONS

S.B. 1134, S.B. 1145, S.B. 1155, S.B. 1156, S.B. 1251, S.B. 1367. H.B. 1800.

COMMITTEE ON COMMUNICATIONS, TECHNOLOGY AND INNOVATION

S.B. 1098, S.B. 1349, S.B. 1365, S.B. 1392, S.B. 1458, S.B. 1462.

COMMITTEE ON COUNTIES, CITIES AND TOWNS

S.B. 1120, S.B. 1128, S.B. 1141, S.B. 1152, S.B. 1207, S.B. 1208, S.B. 1216, S.B. 1267, S.B. 1285, S.B. 1298, S.B. 1309, S.B. 1385, S.B. 1399, S.B. 1400, S.B. 1447, S.B. 1457.

COMMITTEE ON COURTS OF JUSTICE

S.B. 1103, S.B. 1104, S.B. 1105, S.B. 1108, S.B. 1113, S.B. 1122, S.B. 1123, S.B. 1125, S.B. 1138, S.B. 1142, S.B. 1165, S.B. 1168, S.B. 1180, S.B. 1181, S.B. 1184, S.B. 1206, S.B. 1209, S.B. 1213, S.B. 1226, S.B. 1234, S.B. 1241, S.B. 1242, S.B. 1248, S.B. 1261, S.B. 1262, S.B. 1266, S.B. 1270, S.B. 1272, S.B. 1297, S.B. 1306, S.B. 1315, S.B. 1324, S.B. 1325, S.B. 1329, S.B. 1336, S.B. 1339, S.B. 1391, S.B. 1397, S.B. 1415, S.B. 1426, S.B. 1431, S.B. 1437, S.B. 1442, S.B. 1443, S.B. 1456, S.B. 1461, S.B. 1465, S.B. 1468, S.B. 1475.

COMMITTEE ON EDUCATION

S.B. 1106, S.B. 1132, S.B. 1169, S.B. 1175, S.B. 1190, S.B. 1196, S.B. 1204, S.B. 1225, S.B. 1257, S.B. 1288, S.B. 1303, S.B. 1313, S.B. 1322, S.B. 1357, S.B. 1387, S.B. 1401, S.B. 1405, S.B. 1439.

COMMITTEE ON FINANCE

S.B. 1112, S.B. 1130, S.B. 1146, S.B. 1158, S.B. 1162, S.B. 1163, S.B. 1170, S.B. 1197, S.B. 1201, S.B. 1252, S.B. 1326, S.B. 1398, S.B. 1403, S.B. 1423, S.B. 1438.

COMMITTEE ON GENERAL LAWS

S.B. 1110, S.B. 1127, S.B. 1150, S.B. 1171, S.B. 1183, S.B. 1215, S.B. 1254, S.B. 1271, S.B. 1278, S.B. 1279, S.B. 1287, S.B. 1299, S.B. 1305, S.B. 1314, S.B. 1318, S.B. 1327, S.B. 1343, S.B. 1369, S.B. 1384, S.B. 1389, S.B. 1406, S.B. 1410, S.B. 1418, S.B. 1428, S.B. 1429, S.B. 1469, S.B. 1471.

COMMITTEE ON HEALTH, WELFARE AND INSTITUTIONS

S.B. 1102, S.B. 1121, S.B. 1147, S.B. 1154, S.B. 1176, S.B. 1178, S.B. 1187, S.B. 1189, S.B. 1205, S.B. 1220, S.B. 1221, S.B. 1227, S.B. 1235, S.B. 1237, S.B. 1302, S.B. 1304, S.B. 1307, S.B. 1316, S.B. 1320, S.B. 1321, S.B. 1328, S.B. 1333, S.B. 1338, S.B. 1356, S.B. 1362, S.B. 1366, S.B. 1421, S.B. 1427, S.B. 1436, S.B. 1464, S.B. 1472.

COMMITTEE ON LABOR AND COMMERCE

S.B. 1182, S.B. 1195, S.B. 1202, S.B. 1219, S.B. 1223, S.B. 1247, S.B. 1255, S.B. 1269, S.B. 1275, S.B. 1276, S.B. 1284, S.B. 1289, S.B. 1295, S.B. 1310, S.B. 1334, S.B. 1341, S.B. 1351, S.B. 1375, S.B. 1379, S.B. 1380, S.B. 1413, S.B. 1420, S.B. 1425.

COMMITTEE ON PRIVILEGES AND ELECTIONS

S.B. 1097, S.B. 1109, S.B. 1111, S.B. 1148, S.B. 1153, S.B. 1157, S.B. 1239, S.B. 1245, S.B. 1246, S.B. 1281, S.B. 1331, S.B. 1395, S.B. 1422, S.B. 1444.

S.J.R. 270, S.J.R. 272, S.J.R. 275, S.J.R. 289, S.J.R. 310, S.J.R. 322.

COMMITTEE ON PUBLIC SAFETY

S.B. 1119, S.B. 1129, S.B. 1198, S.B. 1256, S.B. 1296, S.B. 1300, S.B. 1301, S.B. 1363, S.B. 1381.

COMMITTEE ON RULES

S.B. 1172, S.B. 1273, S.B. 1408, S.B. 1414, S.B. 1473.

S.J.R. 276, S.J.R. 285, S.J.R. 286, S.J.R. 288, S.J.R. 292, S.J.R. 293, S.J.R. 294, S.J.R. 308, S.J.R. 323, S.J.R. 395.

COMMITTEE ON TRANSPORTATION

S.B. 1126, S.B. 1136, S.B. 1144, S.B. 1160, S.B. 1211, S.B. 1212, S.B. 1214, S.B. 1229, S.B. 1253, S.B. 1259, S.B. 1260, S.B. 1277, S.B. 1335, S.B. 1350, S.B. 1470.

Pursuant to the provisions of Senate Rule 2 (c), the President pro tempore named Senator Jennifer L. McClellan to perform the duties of the Presiding Officer in her absence on Monday, February 8, 2021.

On motion of Senator Lucas, the Senate adjourned until Monday, February 8, 2021, at 9:00 a.m.

Susan Chile Gelson

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

MONDAY, FEBRUARY 8, 2021

The Senate met at 9:00 a.m. and was called to order by the Acting President pro tempore, Senator Jennifer L. McClellan.

The Senate observed a moment of silent prayer.

The Pledge of Allegiance to the Flag of the United States of America was led by Corinne Sloan, Assistant Journal Clerk, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

McClellan, McDougle.

On motion of Senator McDougle, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--2. NAYS--0. RULE 36--0.

YEAS--McClellan, McDougle--2.

NAYS--0.

RULE 36--0.

Pursuant to the provisions of Senate Rule 2 (c), the President pro tempore named Senator Ryan T. McDougle to perform the duties of the Presiding Officer in her absence on Wednesday, March 17, 2021.

ADJOURNMENT SINE DIE

Senator McDougle moved that the Senate adjourn sine die.

The motion was agreed to.

The Acting President pro tempore declared the Senate adjourned sine die.

Senator McDougle was ordered to inform the House of Delegates thereof.

HOUSE COMMUNICATIONS SUBSEQUENT TO ADJOURNMENT SINE DIE

Subsequent to adjournment sine die, the following communications were received and read:

In the House of Delegates February 8, 2021

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING SENATE JOINT RESOLUTIONS:

S.J.R. 316. Celebrating the lives of Walter Lowrie Martin III and Edith Luke Martin.

S.J.R. 332. Commending the Honorable Mark C. Christie.

- S.J.R. 333. Celebrating the life of Frederick Cameron.
- S.J.R. 334. Commending Major Wayne R. Lee, USA, Ret.
- S.J.R. 335. Celebrating the life of Roy L. Pearson.
- S.J.R. 336. Commending Frank Shatz.
- S.J.R. 337. Celebrating the life of Mary Esterine Hundley Moyler.
- S.J.R. 338. Celebrating the life of Jane Carey Gardner.
- S.J.R. 339. Celebrating the life of Edward A. Chappell, Jr.
- S.J.R. 340. Commending Edwin C. Roessler, Jr.
- S.J.R. 341. Commending Carl Hershner.
- S.J.R. 342. Celebrating the life of Wendell Harding Butler.
- S.J.R. 343. Celebrating the life of Charles Wallace Nuttycombe, Sr.
- S.J.R. 344. Commending Raymond O. Anderson.
- S.J.R. 345. Celebrating the life of Sigmund Edward Davidson.
- S.J.R. 346. Commending John T. Wells.
- S.J.R. 347. Commending Angeline Godwin.
- **S.J.R. 348.** Celebrating the life of Michael Wayne Woods.
- S.J.R. 349. Celebrating the life of William Henry Edwards, Sr.
- S.J.R. 350. Commending Robert Pinkard, Jr.
- S.J.R. 351. Celebrating the life of Sally G. Lamb.
- **S.J.R. 352.** Celebrating the life of John B. Davis.
- **S.J.R. 353.** Celebrating the life of William Mantz.
- S.J.R. 354. Celebrating the life of Karl David Stoltzfus, Sr.
- S.J.R. 355. Celebrating the life of Joseph Brisco Dellinger.
- **S.J.R. 356.** Celebrating the life of Lowell Robert Barb.
- S.J.R. 357. Commending the Patrick Henry School of Science and Arts.
- S.J.R. 358. Commending Ryan Janaske.

- S.J.R. 359. Commending The Lady Chamberlains.
- **S.J.R. 360.** Celebrating the life of Jean Smith Brown.
- S.J.R. 361. Commending the Warrior Retreat at Bull Run.
- S.J.R. 362. Commending Akshath Mahajan and Maneesh Vallurupalli.
- **S.J.R. 363.** Commending Steven R. Cover.
- **S.J.R. 364.** Celebrating the life of Stephen Johnson.
- S.J.R. 365. Commending LoudounGo.
- S.J.R. 366. Commending Kelly Lazzara.
- S.J.R. 367. Commending the Virginia Beach Pandemic Relief Partnership.
- S.J.R. 368. Celebrating the life of John Chatburn Stevens.
- **S.J.R. 369.** Celebrating the life of C. Opal B. Abernathey.
- S.J.R. 370. Commending Senior Services of Southeastern Virginia.
- S.J.R. 371. Commending Michael S. Rolband.
- **S.J.R. 373.** Commending Hero Homes, Inc.
- S.J.R. 374. Commending Panorama Latino TV Show.
- **S.J.R. 375.** Celebrating the life of Elaine Dolores Walker.
- S.J.R. 376. Commending Loudoun Shops Black.
- **S.J.R. 377.** Celebrating the life of Officer Brian David Sicknick.
- **S.J.R. 378.** Celebrating the life of the Honorable Robert Lathan Calhoun.
- S.J.R. 379. Commending Tom Hirst and Magaly Galdo-Hirst.
- **S.J.R. 380.** Celebrating the life of the Honorable Jerry M. Wood.
- S.J.R. 381. Commending the Loudoun County branch of the NAACP.
- S.J.R. 382. Commending Joshua Thiel.
- **S.J.R. 383.** Commending the Reverend Deborah Dodson Parsons.
- S.J.R. 384. Commending the Loudoun Medical Reserve Corps.
- S.J.R. 385. Commending Kirk Kincannon.

- S.J.R. 386. Celebrating the life of Sara Lu P. Snyder.
- S.J.R. 387. Commending Ron Campbell.
- S.J.R. 388. Commending William McKenna.
- S.J.R. 389. Commending Kojo Nnamdi.
- S.J.R. 390. Commending Jennifer K. Baker.
- S.J.R. 391. Celebrating the life of Lester Zidel.
- S.J.R. 392. Commending Darnell Dozier.
- S.J.R. 393. Celebrating the life of the Honorable Robert S. Bloxom, Sr.
- S.J.R. 394. Celebrating the life of Carla Yvette Savage-Wells.
- S.J.R. 396. Commending Rebecca Carter.
- **S.J.R. 397.** Celebrating the life of Pamela Sprouse Palmore.
- S.J.R. 398. Commending the Honorable Clyde H. Perdue, Jr.

/s/ Suzette Denslow Clerk of the House of Delegates

In the House of Delegates February 8, 2021

THE HOUSE OF DELEGATES HAS ADJOURNED SINE DIE.

/s/ Suzette Denslow Clerk of the House of Delegates

LEGISLATION SIGNED BY THE PRESIDING OFFICER SUBSEQUENT TO ADJOURNMENT SINE DIE

Subsequent to adjournment sine die of the 2021 Regular Session, the President of the Senate, as required by Article IV, Section 11, of the Constitution, on the date recorded below, signed the following bills that had been passed by both houses and duly enrolled:

February 10, 2021

- **H.B. 1743.** An Act to amend and reenact § 15.2-4904 of the Code of Virginia, relating to industrial development authorities; size of board in certain towns; quorum.
- **H.B. 1768.** An Act to amend and reenact § 5, as amended, of Chapter 76 of the Acts of Assembly of 1974, which created the Lynchburg Parking Authority, relating to terms of members.
- **H.B. 1825.** An Act to amend and reenact § 15.2-5931 of the Code of Virginia, relating to Virginia Beach Sports or Entertainment Project; bond issuance.

- **H.B. 1859.** An Act to amend and reenact § 15.2-958.3 of the Code of Virginia, relating to financing clean energy and other programs; when owner costs incurred.
- **H.B. 1927.** An Act to amend and reenact §§ 3, 4, and 5 of Chapter 643 of the Acts of Assembly of 1964, as amended by Chapter 882 of the Acts of Assembly of 2003, relating to economic development authorities.
- **H.B. 1949.** An Act to amend and reenact §§ 15.2-520 and 15.2-2506 of the Code of Virginia, relating to county executive form of government; local budgets.
- **H.B. 2333.** An Act to facilitate the administration of the COVID-19 vaccine; emergency. EMERGENCY

S.B. 1445. An Act to facilitate the administration of the COVID-19 vaccine; emergency. EMERGENCY

Susan Carle Gelman

Jennifer L. McClellan

Acting President pro tempore of the Senate

Susan Clarke Schaar Clerk of the Senate

WEDNESDAY, MARCH 17, 2021

The Senate met at 2:20 p.m. in Reconvened Session of the 2021 Regular Session and was called to order by the Acting President pro tempore, Senator Ryan T. McDougle.

The Senate observed a moment of silent prayer.

The Pledge of Allegiance to the Flag of the United States of America was led by Lindley Griffin, Deputy Director of Committee Operations and Calendar Clerk, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Locke, McDougle, McPike.

On motion of Senator Locke, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--3. NAYS--0. RULE 36--0.

YEAS--Locke, McDougle, McPike--3.

NAYS--0.

RULE 36--0.

ADJOURNMENT SINE DIE

Senator Locke moved that the Senate adjourn sine die.

The motion was agreed to.

The Acting President pro tempore, Senator Ryan T. McDougle, declared the Senate adjourned sine die.

Senator Locke was ordered to inform the House of Delegates thereof.

HOUSE COMMUNICATION SUBSEQUENT TO ADJOURNMENT SINE DIE

Subsequent to adjournment sine die, the following communication was received and read:

In the House of Delegates
March 17, 2021

THE HOUSE OF DELEGATES HAS ADJOURNED SINE DIE.

/s/ Suzette Denslow Clerk of the House of Delegates

Ryan T. Redough

Susan Carle Delsan

Acting President pro tempore of the Senate

Susan Clarke Schaar Clerk of the Senate

INDEX

TABLE OF CODE TITLES OF THE CODE OF VIRGINIA.

NUMERICAL INDEX--Bills, Resolutions and Documents showing legislative history.

SUBJECT INDEX--Titles of Bills, Resolutions and Documents listed alphabetically under headings of subject matter; all other business transacted by the Senate listed by subject or individual names.

TABLE OF CODE TITLES

TITLE	
1.	GENERAL PROVISIONS.
2.	ADMINISTRATION OF THE GOVERNMENT GENERALLY [Repealed].
2.1.	ADMINISTRATION OF THE GOVERNMENT GENERALLY [Repealed].
2.2.	ADMINISTRATION OF GOVERNMENT.
3.	AGRICULTURE, HORTICULTURE AND FOOD [Repealed].
3.1.	AGRICULTURE, HORTICULTURE AND FOOD [Repealed].
3.2.	AGRICULTURE, ANIMAL CARE, AND FOOD.
4.	ALCOHOLIC BEVERAGES AND INDUSTRIAL ALCOHOL [Repealed].
4.1.	ALCOHOLIC BEVERAGE CONTROL ACT.
5.	AVIATION [Repealed].
5.1.	AVIATION.
6.	BANKING AND FINANCE [Repealed].
6.1.	BANKING AND FINANCE [Repealed].
6.2.	FINANCIAL INSTITUTIONS AND SERVICES.
7.	BOUNDARIES, JURISDICTION AND EMBLEMS OF THE COMMONWEALTH
	[Repealed].
7.1.	BOUNDARIES, JURISDICTION AND EMBLEMS OF THE COMMONWEALTH
	[Repealed].
8.	CIVIL REMEDIES AND PROCEDURE; EVIDENCE GENERALLY [Repealed].
8.01.	CIVIL REMEDIES AND PROCEDURE.
8.1.	COMMERCIAL CODE - GENERAL PROVISIONS [Repealed].
8.1A.	UNIFORM COMMERCIAL CODE - GENERAL PROVISIONS.
8.2.	COMMERCIAL CODE - SALES.
8.2A.	COMMERCIAL CODE - LEASES.
8.3.	COMMERCIAL CODE - COMMERCIAL PAPER [Repealed].
8.3A.	COMMERCIAL CODE - NEGOTIABLE INSTRUMENTS.
8.4.	COMMERCIAL CODE - BANK DEPOSITS AND COLLECTIONS.
8.4A.	COMMERCIAL CODE - FUNDS TRANSFERS.
8.5.	COMMERCIAL CODE - LETTERS OF CREDIT [Repealed].
8.5A.	UNIFORM COMMERCIAL CODE - LETTERS OF CREDIT.
8.6.	COMMERCIAL CODE - BULK TRANSFERS [Repealed].
8.6A.	COMMERCIAL CODE - BULK TRANSFERS [Repealed].
8.7.	COMMERCIAL CODE - WAREHOUSE RECEIPTS, BILLS OF LADING AND OTHER
	DOCUMENTS OF TITLE.
8.8.	COMMERCIAL CODE - INVESTMENT SECURITIES [Repealed].
8.8A.	COMMERCIAL CODE - INVESTMENT SECURITIES.
8.9.	COMMERCIAL CODE - SECURED TRANSACTIONS; SALES OF ACCOUNTS,
	CONTRACT RIGHTS AND CHATTEL PAPER [Repealed].
8.9A.	COMMERCIAL CODE - SECURED TRANSACTIONS.
8.10.	COMMERCIAL CODE - EFFECTIVE DATE - TRANSITIONAL PROVISIONS.
8.11.	1973 AMENDATORY ACT - EFFECTIVE DATE AND TRANSITION PROVISIONS.
9.	COMMISSIONS, BOARDS AND INSTITUTIONS GENERALLY [Repealed].
9.1.	COMMONWEALTH PUBLIC SAFETY.

CONSERVATION GENERALLY [Repealed].

10.

- 10.1. CONSERVATION.
- CONTRACTS.
- 12. CORPORATION COMMISSION [Repealed].
- 12.1. STATE CORPORATION COMMISSION.
- 13. CORPORATIONS GENERALLY [Repealed].
- 13.1. CORPORATIONS.
- 14. COSTS, FEES, SALARIES AND ALLOWANCES [Repealed].
- 14.1. COSTS, FEES, SALARIES AND ALLOWANCES [Repealed].
- 15. COUNTIES, CITIES AND TOWNS [Repealed].
- 15.1. COUNTIES, CITIES AND TOWNS [Repealed].
- 15.2. COUNTIES, CITIES AND TOWNS.
- 16. COURTS NOT OF RECORD [Repealed].
- 16.1. COURTS NOT OF RECORD.
- 17. COURTS OF RECORD [Repealed].
- 17.1. COURTS OF RECORD.
- 18. CRIMES AND OFFENSES GENERALLY [Repealed].
- 18.1. CRIMES AND OFFENSES GENERALLY [Repealed].
- 18.2. CRIMES AND OFFENSES GENERALLY.
- 19. CRIMINAL PROCEDURE [Repealed].
- 19.1. CRIMINAL PROCEDURE [Repealed].
- 19.2. CRIMINAL PROCEDURE.
- 20. DOMESTIC RELATIONS.
- 21. DRAINAGE, SOIL CONSERVATION, SANITATION AND PUBLIC FACILITIES DISTRICTS.
- 22. EDUCATION [Repealed].
- 22.1. EDUCATION.
- 23. EDUCATIONAL INSTITUTIONS [Repealed].
- 23.1. INSTITUTIONS OF HIGHER EDUCATION; OTHER EDUCATIONAL AND CULTURAL INSTITUTIONS.
- 24. ELECTIONS [Repealed].
- 24.1. ELECTIONS [Repealed].
- 24.2. ELECTIONS.
- 25. EMINENT DOMAIN. [Repealed].
- 25.1. EMINENT DOMAIN.
- 26. FIDUCIARIES GENERALLY. [Repealed].
- 27. FIRE PROTECTION.
- 28. FISH, OYSTERS AND SHELLFISH [Repealed].
- 28.1. FISH, OYSTERS, SHELLFISH AND OTHER MARINE LIFE [Repealed].
- 28.2. FISHERIES AND HABITAT OF THE TIDAL WATERS.
- 29. GAME, INLAND FISHERIES AND DOGS [Repealed].
- 29.1. GAME, INLAND FISHERIES AND BOATING [Repealed].
- 29.1 WILDLIFE, INLAND FISHERIES AND BOATING.
- GENERAL ASSEMBLY.
- 31. GUARDIAN AND WARD. [Repealed].
- 32. HEALTH [Repealed].
- 32.1. HEALTH.
- 33. HIGHWAYS, BRIDGES AND FERRIES [Repealed].
- 33.1. HIGHWAYS, BRIDGES AND FERRIES [Repealed effective 10/1/14].
- 33.2. HIGHWAYS AND OTHER SURFACE TRANSPORTATION SYSTEMS.
- 34. HOMESTEAD AND OTHER EXEMPTIONS.
- HOTELS, RESTAURANTS AND CAMPS [Repealed].
- 35.1. HOTELS, RESTAURANTS, SUMMER CAMPS, AND CAMPGROUNDS.
- 36. HOUSING.
- 37. INSANE, EPILEPTIC, FEEBLE-MINDED AND INEBRIATE PERSONS [Repealed].
- 37.1. INSTITUTIONS FOR THE MENTALLY ILL; MENTAL HEALTH GENERALLY [Repealed].
- 37.2 BEHAVIORAL HEALTH AND DEVELOPMENTAL SERVICES.

- 38. INSURANCE [Repealed].
- 38.1. INSURANCE [Repealed].
- 38.2. INSURANCE.
- 39. JUSTICES OF THE PEACE [Repealed].
- 39.1. JUSTICES OF THE PEACE [Repealed].
- 40. LABOR AND EMPLOYMENT [Repealed].
- 40.1. LABOR AND EMPLOYMENT.
- 41. LAND OFFICE [Repealed].
- 41.1. LAND OFFICE.
- 42. LIBRARIES [Repealed].
- 42.1. LIBRARIES.
- 43. MECHANICS' AND CERTAIN OTHER LIENS.
- 44. MILITARY AND EMERGENCY LAWS.
- 45. MINES AND MINING [Repealed].
- 45.1. MINES AND MINING.
- 45.2. MINES, MINERALS, AND ENERGY.
- 46. MOTOR VEHICLES [Repealed].
- 46.1. MOTOR VEHICLES [Repealed].
- 46.2. MOTOR VEHICLES.
- 47. NOTARIES AND OUT-OF-STATE COMMISSIONERS [Repealed].
- 47.1. NOTARIES AND OUT-OF-STATE COMMISSIONERS.
- 48. NUISANCES.
- 49. OATHS, AFFIRMATIONS AND BONDS.
- 50. PARTNERSHIPS.
- 51. PENSIONS AND RETIREMENT [Repealed].
- 51.01. PERSONS WITH DISABILITIES [Recodified].
- 51.1. PENSIONS, BENEFITS, AND RETIREMENT.
- 51.5. PERSONS WITH DISABILITIES.
- 52. POLICE (STATE).
- 53. PRISONS AND OTHER METHODS OF CORRECTION [Repealed].
- 53.1. PRISONS AND OTHER METHODS OF CORRECTION.
- 54. PROFESSIONS AND OCCUPATIONS [Repealed].
- 54.1. PROFESSIONS AND OCCUPATIONS.
- 55. PROPERTY AND CONVEYANCES. [Repealed]
- 55.1. PROPERTY AND CONVEYANCES.
- PUBLIC SERVICE COMPANIES.
- 57. RELIGIOUS AND CHARITABLE MATTERS; CEMETERIES.
- 58. TAXATION [Repealed].
- 58.1. TAXATION.
- 59. TRADE AND COMMERCE [Repealed].
- 59.1. TRADE AND COMMERCE.
- 60. UNEMPLOYMENT COMPENSATION [Repealed].
- 60.1. UNEMPLOYMENT COMPENSATION [Repealed].
- 60.2. UNEMPLOYMENT COMPENSATION.
- 61. WAREHOUSES, COLD STORAGE AND REFRIGERATED LOCKER PLANTS [Repealed].
- 61.1. WAREHOUSES, COLD STORAGE AND REFRIGERATED LOCKER PLANTS.
- 62. WATERS OF THE STATE, PORTS AND HARBORS [Repealed].
- 62.1. WATERS OF THE STATE, PORTS AND HARBORS.
- 63. WELFARE [Repealed].
- 63.1. WELFARE (SOCIAL SERVICES) [Repealed].
- 63.2. WELFARE (SOCIAL SERVICES).
- 64. WILLS AND DECEDENTS' ESTATES [Repealed].
- 64.1. WILLS AND DECEDENTS' ESTATES [Repealed].
- 64.2. WILLS, TRUSTS, AND FIDICUARIES.
- 65. WORKMEN'S COMPENSATION [Repealed].
- 65.1. WORKERS' COMPENSATION [Repealed].
- 65.2. WORKERS' COMPENSATION.

- JUVENILE JUSTICE.
- 66. 67. VIRGINIA ENERGY PLAN.

NUMERICAL INDEX

S.B.	13. Capitol Square; possessing or transporting a weapon within Square, penalty. Adding § 18.2-283.2.	
	 -	
	Patrons: Ebbin, et al.	,
c D	Continued from 2020 Regular Session in Senate Committee on the Judiciary	O
э.в.	15. Weapons; carrying into building owned or leased by the Commonwealth, penalty.	
	Adding § 18.2-283.2.	
	Patrons: Ebbin, et al.	,
C D	Continued from 2020 Regular Session in Senate Committee on the Judiciary	6
S.B.	27. Uninsured and underinsured motorist insurance policies; bad faith.	
	Amending §§ 8.01-66.1 and 38.2-2206.	
	Patrons: Petersen, et al.	_
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor	6
S.B.	32. Corporal punishment of a child with an object; penalty. Adding § 18.2-371.1:1.	
	Patrons: Petersen, et al.	
	Continued from 2020 Regular Session in Senate Committee on the Judiciary	6
S.B.	80. Student journalists; freedom of speech and the press. Adding §§ 22.1-203.4 and	
	23.1-401.2.	
	Patrons: Marsden, et al.	
	Continued from 2020 Regular Session in Senate Committee on Education and Health	6
S.B.	91. Parole; Virginia Parole Board shall establish procedures for consideration of parole	
	for persons who were previously ineligible, repeals code in reference to limitation on	
	application of parole statutes. Adding § 53.1-165.2; repealing § 53.1-165.1.	
	Patrons: Edwards, et al.	
	Continued from 2020 Regular Session in Senate Committee on the Judiciary	6
S.B.	97. Virginia Fair Housing Law; it is an unlawful discriminatory housing practice for any	
	political jurisdiction or its employees or appointed commissions to discriminate in the	
	application of local land use ordinances or guidelines, etc., effective clause.	
	Amending §§ 36-96.3 and 36-96.17.	
	Patrons: McClellan, et al.	
	Continued from 2020 Regular Session in House Committee on General Laws	7
S.B.	128. Education, Department of; pilot program, feasibility of educational placement	
	transition of certain students with disabilities.	
	Patrons: Suetterlein, et al.	
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
S.B.	135. Children's Services Act; special education programs, expands eligibility for services	
	under Act to students who transfer from an approved private school program to a public	
	school program. Amending §§ 2.2-5211 and 2.2-5212.	
	Patrons: Stuart, et al.	
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
S.B.	143. Disabled veterans and surviving spouses; state subsidy of property tax exemptions,	
	defines "eligible locality," etc. Adding §§ 58.1-3537 and 58.1-3538.	
	Patrons: Stuart, et al.	
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
S.B.	148. Driving under the influence; provisions regarding driving or operating a motor	
	vehicle on his residential curtilage or his adjoining property while intoxicated and	
	regarding operating a motor vehicle by a person under the age of 21.	
	Amending §§ 18.2-266 and 18.2-266.1.	
	Patron: Stuart	
	Continued from 2020 Regular Session in House Committee for Courts of Justice	7

SR	150. Criminal fiscal impact statements; bills resulting in a net increase of period of
у.р.	imprisonment or commitment. Amending §§ 30-19.1:4 and 30-28.18.
	Patron: Howell
	Continued from 2020 Regular Session in Senate Committee on Rules
S.B.	166. Campaign finance; disbursement of surplus funds, prohibited conversion to personal
S.D.	use. Amending §§ 24.2-948.4, 24.2-949.9, 24.2-950.9, 24.2-951.9, and 24.2-952.7;
	adding §§ 24.2-948.5, 24.2-949.9:01, 24.2-950.10, 24.2-951.10, and 24.2-952.8.
	Patrons: Saslaw, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections
SR	190. Children's Services Act; community policy and management teams, use of funds.
у.р.	Amending §§ 2.2-5211 and 2.2-5212.
	Patron: Peake
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
Ç D	191. Income tax, state; subtraction for active duty military income, etc., or veteran
у.р.	retirement compensation. Amending § 58.1-322.02.
	Patron: Peake
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
c D	218. Income tax, state and corporate; tax credit for employers of National Guard
3.D.	members and self-employed National Guard members, a taxpayer allowed to carry unused
	credits over for up to five taxable years. Adding § 58.1-339.13.
	Patrons: Suetterlein, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
C D	223. Juvenile records; expungement of records for offenses that would be felony larceny
э.р.	if committed by an adult. Amending §§ 16.1-305 and 16.1-306.
	Patrons: Favola, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
Ç D	234. Human Resource Management, Department of; health insurance for local school
э.р.	board employees. Amending § 2.2-1204.
	Patron: Chafin
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	** *
э.р.	Amending § 22.1-253.13:3; adding § 22.1-212.33.
	Patron: Favola
	Continued from 2020 Regular Session in Senate Committee on Education and Health
SR	253. Deeds of trust; fiduciary duties, trustee shall comply with all restrictive covenants
у.р.	regarding affordability that affect the property secured by the trust. Amending § 55.1-320.
	Patron: Chafin
	Continued from 2020 Regular Session in House Committee for Courts of Justice
SR	274. Motion picture theaters; required open-captioned showings. Adding §§ 59.1-261.1,
S.D.	59.1-261.2, and 59.1-261.3.
	Patron: Barker
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor
S R	285. Intentional or negligent infliction of injury or death; bystander claims for
у.р.	emotional distress. Adding § 8.01-42.6.
	Patron: Deeds
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S R	291. Mortgage brokers; prohibition on dual compensation. Amending § 6.2-1616.
.	Patron: Deeds
	Continued from 2020 Regular Session in House Committee on Labor and Commerce
SR	295. Employment; disclosure of terms to employees. Adding § 40.1-29.1.
.	Patron: Marsden
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor
	Commission from Bobb regular Debbion in Denate Committee on Commission and Education of the Commission

S.B.	304. Animal shelters; public animal shelter shall annually file with State Veterinarian a copy of its intake policy, number of animals on which a euthanasia procedure was
	performed, etc. Amending § 3.2-6546.
	Patron: Stanley
	Continued from 2020 Regular Session in House Committee on Agriculture, Chesapeake and Natural
C D	Resources
э.Б.	convictions, effective clause. Amending §§ 4.1-305, 16.1-69.55, and 18.2-251.
	Patrons: Stanley, et al.
	Continued from 2020 Regular Session in House Committee for Courts of Justice
C D	317. Licensed inpatient nursing services; in-home follow-up care upon discharge.
э.Б.	Adding § 37.2-422.1.
	Patron: Kiggans
	Continued from 2020 Regular Session in House Committee on Health, Welfare and
	Institutions
C D	318. Balloons; reduces the number per hour that a person may release.
у.р.	Amending § 29.1-556.1.
	Patron: Kiggans
	Continued from 2020 Regular Session in House Committee on Agriculture, Chesapeake and Natural
	Resources
S R	326. Sentencing proceeding by the jury after conviction; recommendation of leniency,
э.р.	suspension of sentence imposed. Amending § 19.2-295.1.
	Patrons: Deeds, et al.
	Continued from 2020 Regular Session in House Committee for Courts of Justice
S R	327. Accomack County and Northampton County School Boards; eligibility to receive
.	cost of competing adjustment to salaries, etc.
	Patron: Lewis
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	346. Contractors, on-site superintendents; boiler and pressure vessel operator license,
~ 	continuing education, penalty. Adding §§ 54.1-1149 through 54.1-1156.
	Patron: Bell
	Continued from 2020 Regular Session in House Committee on General Laws
S.B.	352. Guardianship and conservatorship; supported decision-making alternative.
	Amending §§ 64.2-2000 and 64.2-2003.
	Patron: Lucas
	Continued from 2020 Regular Session in House Committee on Health, Welfare and
	Institutions
S.B.	353. Outdoor shooting ranges; prohibited adjacent to residential areas, exceptions, civil
	penalty. Adding § 18.2-511.2.
	Patron: Bell
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	359. Gifts of real estate; title search required for recordation. Adding § 55.1-604.1.
	Patron: Cosgrove
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	362. Hyperbaric oxygen therapy; Department of Veterans Services to contract with any
	hospital that furnishes the treatment option of such therapy to provide to any veteran who
	has been certified as having post-traumatic stress disorder or traumatic brain injury.
	Patrons: Dunnavant, et al.
	Continued from 2020 Regular Session in House Committee on General Laws
S.B.	363. Virginia Works Portal; created, report. Adding § 2.2-435.10:1.
	Patrons: Dunnavant, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6

S.B.	366. Learning management system; Department of Education to obtain a statewide
	system for use in public schools.
	Patron: Dunnavant
S.B.	Continued from 2020 Regular Session in House Committee on Appropriations
S.B.	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6 382. Health insurance; coverage for prosthetic devices, repeals existing requirement for coverage. Amending § 32.1-325; adding § 38.2-3418.15:1; repealing § 38.2-3418.15. Patrons: McPike, et al.
	Continued from 2020 Regular Session in House Committee on Labor and Commerce
S.B.	387. Charitable gaming; creates special permit for the play of electronic versions of
	instant bingo, pull tabs, or seal cards on certain premises. Amending §§ 18.2-340.27:1 and 18.2-340.28. Patron: McPike
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	390. Public schools; reduces total number and type of required Standards of Learning assessments. Amending § 22.1-253.13:3.
	Patron: McPike
C D	Continued from 2020 Regular Session in Senate Committee on Education and Health
Э.Б.	Patron: Hashmi
	Continued from 2020 Regular Session in Senate Committee on Agriculture, Conservation and
	Natural Resources
S.B.	411. Employment health and safety standards; heat illness prevention, definition.
	Adding § 40.1-44.2.
	Patron: Hashmi
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor
S.B.	419. Trees; Town of Vienna, by ordinance, allowed to require that a subdivision or
	development provide for the preservation or replacement of trees on the development site.
	Amending § 15.2-961.1. Patron: Petersen
	Continued from 2020 Regular Session in Senate Committee on Local Government
S.B.	
ь.ь.	Amending § 8.01-225; adding § 22.1-274.5.
	Patrons: DeSteph, et al.
	Continued from 2020 Regular Session in House Committee on Education
S.B.	427. Employee protection; discharge for protective order prohibited. Adding § 40.1-27.3.
	Patron: Saslaw
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor
S.B.	446. Constitutional amendment; personal property tax exemption for one motor vehicle
	owned and used primarily by or for a disabled veteran, "motor vehicle" shall include only
	automobiles and pickup trucks, exception (submitting to qualified voter).
	Amending Section 6 of Article X.
	Patrons: Reeves, et al. Continued from 2020 Regular Session in Sanata Committee on Finance and Appropriations.
Ç D	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6 449. Death penalty; abolishes penalty, including those persons currently under a death
у.В.	sentence. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8,
	18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32,
	18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2,
	19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175,

S.B.	. 449 (continued) 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.2, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-400,
	53.1-204, and 53.1-229; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236.
	Patrons: Surovell, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	. 456. Income tax, state; subtraction for low-income military veterans with a permanent service-connected disability. Amending § 58.1-322.02.
	Patrons: Reeves, et al.
S.B.	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6. 457. Income tax, state and corporate; tax credit for employers of National Guard members and self-employed National Guard members, includes active or reserve members. Adding § 58.1-339.13.
	Patrons: Reeves, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	. 460. Income tax, state and corporate; tax credit for employers of military spouses. Adding § 58.1-339.13.
	Patrons: Reeves, et al.
~ -	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	. 461. Veterans and active duty members of the Armed Forces; local school board to give any veteran, etc., who it employs as a teacher credit for time served in determining school division's teacher salary scale. Adding § 22.1-289.3.
	Patrons: Reeves, et al.
~ ~	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	. 464. Higher educational institutions; intercollegiate athletics, student-athletes, compensation, representation, and injury. Adding § 23.1-408.1.
	Patrons: Reeves, et al.
S.B.	Continued from 2020 Regular Session in Senate Committee on Education and Health
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor
S.B.	486. Food waste; Virginia Department of Agriculture and Consumer Services, et al., to study sources, scale, and prevention in the Commonwealth, effective clause. Patron: Favola
	Continued from 2020 Regular Session in House Committee on Rules
S.B.	490. Firearms; purchase, possession, etc., following conviction for assault and battery of a family or household member, permit to restore rights, penalties. Amending §§ 18.2-308.09, 18.2-308.2, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:6.
	Patron: Favola
S.B.	Continued from 2020 Regular Session in Senate Committee on the Judiciary
	Continued from 2020 Regular Session in Senate Committee on Rehabilitation and Social
	Services

S.B.	510. Milk; definition, misbranding product, prohibition. Amending §§ 3.2-5120, 3.2-5121, and 3.2-5123.
	Patrons: Reeves, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	533. Pari-mutuel wagering; breakage, distribution for problem gambling treatment and support, creation of Problem Gambling Treatment and Support Fund. Amending §§ 37.2-304 and 59.1-392; adding § 37.2-314.1. Patron: Reeves
	Continued from 2020 Regular Session in Senate Committee on General Laws and Technology \dots 6
S.B.	536. Wireless broadband services; Tobacco Region Revitalization Commission to award at least \$50 million per year in grants to cover expenditures for the purchase and installation of wireless and broadband equipment to rural service areas in the Commonwealth. Adding § 3.2-3108.1. Patron: Edwards
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	537. Mandatory minimum punishment; limitation. Adding § 18.2-12.2.
	Patron: Edwards
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	550. Dating relationship abuse; expands the crime of assault and battery against a family or household member to include persons in a dating relationship. Amending §§ 16.1-228, 16.1-243, 16.1-253.1, 16.1-253.2, 16.1-253.4, 16.1-260, 16.1-266.2, 16.1-279.1, 16.1-296, 16.1-298, 18.2-57.2, 19.2-11.2, and 19.2-81.3. Patrons: Bell and Ebbin
C D	Continued from 2020 Regular Session in Senate Committee on the Judiciary
э.в.	569. Behavioral Health and Developmental Services and State Police, Departments of; required to develop or obtain mobile applications for mental health and public safety, effective clause. Adding §§ 37.2-312.3 and 52-50.
	Patrons: Dunnavant, et al.
	Continued from 2020 Regular Session in House Committee on Health, Welfare and
	Institutions
S.B.	606. Virginia Retirement System; increased retirement allowance for certain judges. Amending § 51.1-169.
	Patron: McDougle
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
S.B.	618. Specialty dockets; any court may refer a defendant to a local docket if docket exists within that jurisdiction. Adding § 19.2-266.4.
	Patron: Deeds
a -	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	624. Geriatric prisoners; petitioning Parole Board for conditional release. Amending §§ 19.2-297.1 and 53.1-40.01.
	Patrons: Spruill, et al.
c D	Continued from 2020 Regular Session in House Committee for Courts of Justice
З.В.	Control Board to regulate aboveground storage tanks, etc. Adding §§ 62.1-44.34:29 through 62.1-44.34:39.
	Patrons: Surovell, et al.
	Continued from 2020 Regular Session in Senate Committee on Agriculture, Conservation and
	Natural Resources

S.B.	627. Contractors to subcontractors ; contracts under Virginia Public Procurement Act shall include a payment clause requiring contractor to pay a subcontractor within seven days after receipt of amounts paid by the state agency or locality. Amending § 2.2-4354; adding § 11-4.6. Patron: Vogel	
S.B.	Continued from 2020 Regular Session in Senate Committee on General Laws and Technology 637. Estate tax; reinstates the tax for persons dying on and after July 1, 2020. Amending §§ 58.1-901, 58.1-902, 58.1-905, and 58.1-912. Patron: Survovell	5
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
S.B.	641. Civil action; sale of personal data. Adding § 8.01-40.5.	
	Patron: Surovell	_
S.B.	Continued from 2020 Regular Session in Senate Committee on the Judiciary)
S.B.	Continued from 2020 Regular Session in Senate Committee on the Judiciary	5
	Patrons: Boysko, et al. Continued from 2020 Regular Session in Senate Committee on Agriculture, Conservation and Natural Resources	6
S.B.	681. Criminal cases; compensation of experts, effective clause. Amending § 19.2-175.	,
	Patron: Mason	
S.B.	Continued from 2020 Regular Session in House Committee for Courts of Justice	7
S.B.	Patron: Obenshain Continued from 2020 Regular Session in Senate Committee on the Judiciary	5
S.B.	intellectual disability. Amending § 40.1-28.9. Patron: McClellan Continued from 2020 Regular Session in Senate Committee on Commerce and Labor	5
S.B.	Patron: McClellan Continued from 2020 Regular Session in Senate Committee on the Judiciary	6
S.B.	Patron: McClellan Continued from 2020 Regular Session in House Committee on Appropriations	7
S.B.	Patron: Marsden Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
	Patron: Barker Continued from 2020 Regular Session in Senate Committee on Commerce and Labor	6

	777. Virginia Security for Public Deposits Act; collateral for public deposits. Amending § 2.2-4402.
	Patrons: Lewis, et al.
	Continued from 2020 Regular Session in Senate Committee on General Laws and Technology 6
S.B.	779. Sixth grade science curriculum; Department of Education to coordinate with the Department of Environmental Quality to update the "Window into a Green Virginia" curriculum. Patron: Lewis
S.B.	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6 780. Campgrounds; inherent risks, liability. Adding § 35.1-17.1. Patron: Lewis
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	785. Mortgage loan originators; exempts retailers of manufactured or modular homes, etc. Amending § 6.2-1701. Patron: Lewis
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor 6
S.B.	802. Death penalty executions; imposes a moratorium.
	Patrons: Morrissey, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
	803. Attorneys for the Commonwealth; compensation and collection of fees. Amending §§ 15.2-1626, 15.2-1627.2, 15.2-1627.3, and 15.2-1636.8. Patrons: Morrissey, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
	805. Robbery; definition, degrees of punishment, penalty. Amending § 18.2-58. Patrons: Morrissey, et al.
S.B.	Continued from 2020 Regular Session in House Committee for Courts of Justice
	Patrons: Morrissey, et al.
S.B.	Continued from 2020 Regular Session in Senate Committee on the Judiciary
	Continued from 2020 Regular Session in House Committee for Courts of Justice
	punishment, etc. Amending §§ 19.2-264.3, 19.2-288, 19.2-295, 19.2-295.1, and 19.2-295.3. Patrons: Morrissey, et al.
	Continued from 2020 Regular Session in House Committee for Courts of Justice
S.B.	813. Virginia High Speed Chase Alert Program; created. Adding §§ 52-34.13, 52-34.14, and 52-34.15.
	Patron: Morrissey
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
	819. Drug Treatment Court Act; authorization and availability. Amending § 18.2-254.1.
	Patrons: Morrissey, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
	824. Alcoholic beverage control; distiller licenses, Internet orders and shipments. Amending § 4.1-119. Patron: Ruff
	Continued from 2020 Regular Session in House Committee on General Laws

S.B.	826. Water and sewer service charges; reduces maximum potential responsibility of a property owner for tenant's unpaid charges. Amending § 15.2-2119.4.	
	Patron: McDougle	
	Continued from 2020 Regular Session in House Committee on General Laws	7
c D	842. Electric energy; customer choice. Adding § 56-585.1:11.	′
э.Б.	Patron: Petersen	
		_
S.B.	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor	O
э.в.	854. Virginia Urban Agriculture Advisory Council; created, report. Adding §§ 3.2-3122 through 3.2-3127.	
	Patron: Petersen	
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
C D	863. Southwestern Virginia Mental Health Institute; Governor to lease a portion of	U
э.р.	property to Smyth County. Amending Chapter 678, 2019 Acts.	
	Patron: Pillion	
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
C R	867. Health care provider panels; vertically integrated carriers, providers.	Ů
э.р.	Amending §§ 38.2-3407.10 and 38.2-4319.	
	Patrons: Petersen, et al.	
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor	6
S.B.	887. Voter registration; automatic electronic transmission by DMV to the Department of	Ĭ
	Elections of certain information for any individual. Amending §§ 24.2-410.1, 24.2-411.1,	
	24.2-412, 24.2-413, 24.2-418, 24.2-418.1, 24.2-428.2, 24.2-653, and 24.2-1016;	
	adding § 24.2-411.3.	
	Patron: Ebbin	
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections	6
S.B.	898. Business advisory committee; established. Adding § 23.1-201.1.	
	Patron: DeSteph	
	Continued from 2020 Regular Session in Senate Committee on Education and Health	6
S.B.	906. Landlord and tenant; noncompliance as defense to action for possession for	
	nonpayment of rent. Amending § 55.1-1241.	
	Patrons: Stanley, et al.	
	Continued from 2020 Regular Session in Senate Committee on General Laws and Technology	6
S.B.	914. Victim of human trafficking; petition for vacatur and expungement of convictions	
	and police and court records. Amending § 19.2-392.4; adding § 19.2-392.2:1.	
	Patron: Locke	
~ ~	Continued from 2020 Regular Session in Senate Committee on the Judiciary	6
S.B.	917. Reproductive health services; health benefit plans to cover costs of specified health	
	care services, drugs, devices, products, and procedures, pharmacy claims for reimbursement of all contraceptives approved for over-the-counter sale.	
	Amending §§ 32.1-325, 38.2-3407.5:1, 38.2-3451, and 38.2-4319; adding § 38.2-3418.18.	
	Patrons: Locke, et al.	
	Continued from 2020 Regular Session in Senate Committee on Education and Health	6
S R	946. Medical assistance services; state plan to include doulas. Amending § 32.1-325.	U
э.р.	Patrons: Locke, et al.	
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations	6
S.B.	* * *	U
J.D.	Amending §§ 19.2-392.1 and 19.2-392.2.	
	Patrons: Saslaw, et al.	
	Continued from 2020 Regular Session in Senate Committee on the Judiciary	6
		-

	965. Income tax, state; subtraction for certain active duty military income. Amending § 58.1-322.02.
	Patron: Peake
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations
S.B.	973. Judges; maximum number in each judicial circuit and district. Amending §§ 16.1-69.6:1 and 17.1-507. Patron: Edwards
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	983. Certificate of public need; definition of "medical care facility," facilities subject to review. Amending § 32.1-102.1.
	Patron: Lucas
	Continued from 2020 Regular Session in House Committee on Health, Welfare and
	Institutions
S.B.	993. State Health Commissioner; local health directors, qualifications. Amending §§ 32.1-17 and 32.1-30.
	Patron: Locke
	Continued from 2020 Regular Session in House Committee on Health, Welfare and
	Institutions
	1011. Transportation, Department of; application of design standards, Residency Engineer of the Department, etc., shall have the authority to approve modifications. Adding § 33.2-269.1.
	Patron: McDougle
S.B.	Continued from 2020 Regular Session in House Committee on Transportation
	Patrons: Cosgrove, et al.
	Continued from 2020 Regular Session in Senate Committee on Transportation
S.B.	1033. Juvenile community correctional centers and facilities; establishment, placement of juveniles. Amending §§ 2.2-1837, 2.2-3007, 8.01-195.10, 9.1-801, 9.1-903, 16.1-249, 16.1-269.1, 16.1-285.1, 16.1-285.2, 16.1-309.4, 16.1-322.5, 16.1-322.6, 18.2-48.1, 18.2-431.1, 18.2-473, 18.2-473.1, 18.2-474, 18.2-474.1, 18.2-477.2, 22.1-209.1:2, 22.1-289, 29.1-317, 51.1-212, 66-3, 66-10, 66-13, 66-13.1, 66-18, 66-22.1, 66-25.1:1, 66-25.1:3, 66-25.2:1, 66-25.3, 66-25.4, 66-25.6, and 66-25.7.
	Patrons: Locke, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations
S.B.	1042. Wills; presumption of undue influence. Adding § 64.2-454.1.
	Patron: Obenshain
	Continued from 2020 Regular Session in House Committee for Courts of Justice
	1049. Involuntary commitment; notice and participation, discharge plans. Amending §§ 37.2-505, 37.2-814, 37.2-817 through 37.2-817.4, and 37.2-838.
	Patron: Deeds
	Continued from 2020 Regular Session in House Committee on Health, Welfare and
	Institutions
	1050. Hospitals; custody of person subject to emergency custody order, regulations. Amending §§ 32.1-127 and 37.2-808.
	Patron: Deeds
	Continued from 2020 Regular Session in Senate Committee on Education and Health

S.B.	1052. Virginia Charitable Gaming Board; regulations of the Board, electronic or mechanical equipment used in the conduct of charitable gaming, increases the number of electronic pull tabs that may be placed in the social quarters of a qualified organization to 28. Amending § 18.2-340.19.
	Patron: Reeves
S.B.	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6 1060. For good cause shown or upon agreement of all parties; court may dismiss an action without prejudice. Adding § 8.01-380.1. Patron: Petersen
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
S.B.	1061. C-PACE loans; removes an exclusion for residential dwellings with fewer than five dwelling units and condominium projects from certain requirements related to a voluntary special assessment lien that secures a loan for the initial acquisition and installation of clean energy, etc. Amending § 15.2-958.3.
	Patrons: Petersen, et al.
S.B.	Continued from 2020 Regular Session in Senate Committee on General Laws and Technology 6 1062. Virginia Higher Education Funding Review Commission; established, duties, report. Amending § 2.2-2101; adding § 2.2-208.2.
	Patron: Petersen
C D	Continued from 2020 Regular Session in Senate Committee on Rules
у.р.	January 1 through December 31 of the year immediately preceding affected academic year, is less than or equal to zero, no governing board of a baccalaureate public institution shall increase the in-state tuition rate charged to undergraduate students for such academic year. Amending §§ 23.1-307 and 23.1-1301. Patron: Kiggans
	Continued from 2020 Regular Session in Senate Committee on Education and Health
S.B.	1078. Education, Department of; to amend its state plan pursuant to the Every Student Succeeds Act to include the Preliminary SAT/National Merit Scholarship Qualifying Test and the PreACT in the next federal accountability indicators application. Patron: Suetterlein
	Continued from 2020 Regular Session in Senate Committee on Education and Health
S.B.	1079. Medically unnecessary chaperones; Board of Medicine to amend its regulations to require that patients be notified that they have the right to opt out of the presence of a chaperone during medical examinations. Patron: Suetterlein
	Continued from 2020 Regular Session in Senate Committee on Education and Health
S.B.	1080. Urban Teacher Fund and Program; established. Adding § 22.1-23.3.
	Patrons: Morrissey, et al.
S.B.	Continued from 2020 Regular Session in House Committee on Appropriations
S.B.	Continued from 2020 Regular Session in Senate Committee on Local Government
	Patron: Morrissey Continued from 2020 Regular Session in Senate Committee on Rehabilitation and Social

S.B.	1097. Absentee voting; witness signature not required. Amending §§ 24.2-702.1,
	24.2-704, 24.2-706, and 24.2-707.
	Patrons: Favola, et al. Profiled and referred to Committee on Privileges and Elections
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported
	Read second time and engrossed
	Read third time and passed
C D	1098. Unmanned aircraft; exempts an owner from the requirement to register.
э.Б.	Amending § 5.1-5.
	Patrons: Favola, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Communications, Technology and
	Innovation
S.B.	1099. Children's Services Act; special education programs, expands eligibility for
э.Б.	services under Act to students who transfer from an approved private school program to a
	public school program. Amending §§ 2.2-5211 and 2.2-5212.
	Patrons: Stuart, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
S.B.	1100. Budget bill; appropriations for 2020-2022 biennium. Amending Chapter 56, 2020
	Sp. I Acts.
	Patron: Howell
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
S.B.	1101. Presidential electors; enters Virginia into an interstate compact known as the
	Agreement Among the States to Elect the President by National Popular Vote.
	Adding §§ 24.2-209.1 and 24.2-209.2.
	Patrons: Ebbin and Lucas, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
S.B.	1102. Personal care aides; Department of Medical Assistance Services shall establish an
	orientation program for all aides who provide self-directed services through the Medicaid
	program. Adding § 32.1-331.04.
	Patrons: Locke, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Read third time and passed 271
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions 40°

c D	1102 Year to Frank with Constitution And Trib Land Constitution And Year	
S.B.	1103. Virginia Freedom of Information Act; individual votes of members of the Virginia	
	Parole Board shall be public records and subject to provisions of the Act.	
	Amending § 2.2-3703.	
	Patrons: Suetterlein, et al.	2
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	
	Reported	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	497
S.B.	1104. Parole; Department of Corrections shall release a prisoner no sooner than 21	
	business days after the date of notification by the Virginia Parole Board to the appropriate	
	attorney for the Commonwealth of the decision to grant parole, etc.	
	Amending §§ 53.1-136 and 53.1-155.	
	Patrons: Obenshain, et al.	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported	237
	Constitutional reading dispensed, passed by for day	261
	Passed by for day	
	Read second time	302
	Reading of substitute waived	303
	Committee substitute agreed to	303
	Amendment by Senator Morrissey withdrawn	303
	Engrossed	305
	Read third time and passed	327
	Continued to 2021 Special Session I in House Committee for Courts of Justice	497
S.B.	1105. Post-conviction relief; previously admitted forensic scientific evidence, effective	
	date, report, and effective clause. Adding § 19.2-327.15.	
	Patrons: Stanley, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	. 25
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	292
	Reported with amendment	
	Constitutional reading dispensed, passed by for day	
	Read second time	394
	Reading of substitute waived	
	Committee substitute agreed to	
	Reading of amendment waived	
	Committee amendment agreed to	
	Engrossed	
	Read third time and passed	
	Statement on vote	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
S.B.	1106. Public School Assistance Fund and Program; created. Adding § 22.1-141.3.	. , ,
•	Patrons: Stanley, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	. 25
	Reported	
	Rereferred to Committee on Finance and Appropriations	

S.B.	1106 (continued)
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendments waived
	Amendments by Senator McClellan agreed to
	Engrossed
	Read third time and passed
c n	Continued to 2021 Special Session I in House Committee on Education
5.В.	1107. Medical malpractice; eliminates the cap on the recovery in actions against health
	care providers where the act or acts occurred on or after July 1, 2021.
	Amending § 8.01-581.15.
	Patrons: Stanley, et al.
c D	Prefiled, ordered printed, and referred to Committee on the Judiciary
5.Б.	1108. General district courts; increases to \$50,000 the maximum civil jurisdictional limit
	of courts, appeal bond. Amending §§ 8.01-195.4, 16.1-77, and 16.1-107.
	Patron: Stanley
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Reading of substitute waived
	Committee substitute rejected
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
c D	1109. Voter referendum; issuance of state general obligation bonds for school facility
э.р.	modernization, November 2022 general election, effective clause for provisions.
	Patrons: Stanley, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Privileges and Elections
S R	1110. Real estate settlement agents; SCC may share information collected from an agent
J. .	or agency regarding any errors, etc., with any party to the real estate transaction in
	connection with the actions arising out of a settlement. Amending § 55.1-1004.
	Patron: Spruill
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 25
	Reported with amendment
	Constitutional reading dispensed passed by for day

S.B.	1110 (continued) Read second time
	Reading of amendment waived.
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on General Laws
C D	1111. Elections; preservation of order at the polls, powers of officers of election.
у.р.	Amending § 24.2-606.
	Patrons: Spruill, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Privileges and Elections
C D	1112. Research and development expenses; tax credit available against the bank
э.Б.	franchise tax for taxable years beginning on and after January 1, 2021.
	Amending §§ 58.1-439.12:08 and 58.1-439.12:11.
	Patrons: Locke, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Finance
S.B.	1113. Communicating threats of death or bodily injury to a person with intent to
	intimidate, etc.; threats to another in writing, penalties. Amending § 18.2-60.
	Patrons: Spruill, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1114. Children's Services Act; expands eligibility for use of the state pool of funds under
•	the Act to services that are provided in a public school setting. Amending §§ 2.2-5211 and
	2.2-5212.
	Patron: Peake
	Prefiled, ordered printed, and referred to Committee on Education and Health

S.B.	1115. Industrial hemp; increases the maximum tetrahydrocannabinol (THC) concentration, expands definition of "hemp product", registration renewal fee not to exceed \$50, etc. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. Patrons: Peake, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1116. State Health Commissioner; powers during an epidemic, vaccinations or
	immunizations for a child, religious tenets or practices. Amending §§ 32.1-46 and
	32.1-48.
	Patrons: Peake, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
S.B.	1117. Immunizations; a parent or guardian is allowed to object to a vaccination or
	immunization of a child on the grounds that it conflicts with his religious tenets or
	practices. Amending § 32.1-46.
	Patrons: Peake, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
S.B.	1118. Voter registration; verification of social security numbers, etc., provisional
	registration status. Amending § 24.2-653; adding §§ 24.2-418.01 and 24.2-653.4.
	Patrons: Peake, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
S.B.	1119. Law-enforcement agencies; creates a special nonreverting fund to be known as the
	Body-Worn Camera System Fund to assist state or local agencies with costs of purchasing,
	etc., body-worn camera systems. Adding § 9.1-116.7.
	Patrons: Reeves, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported 86
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Public Safety
S.B.	1120. Albemarle and Prince William Counties; counties that have adopted the county
	executive form of government may carry over unspent funds from year to year for
	multiyear capital projects and outstanding grants. Amending §§ 15.2-520 and 15.2-2506.
	Patrons: Reeves, et al.
	Prefiled, ordered printed, and referred to Committee on Local Government
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed

S.B.	1120 (continued) Read third time and passed
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 496
S R	1121. Birth certificates; an amendment of a certificate shall be evaluated by the State
э.р.	Registrar through an administrative process. Amending § 32.1-269.
	Patrons: Locke, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
c D	1122. Habitual offenders; required that the Commissioner of DMV reinstate a person's
э.Б.	privilege to drive a motor vehicle that was suspended or revoked solely on the basis that
	such person was determined to be or adjudicated a habitual offender, repeals remaining
	provisions of the Habitual Offender Act. Amending §§ 8.01-9, 8.01-407, 16.1-77,
	16.1-305, 17.1-213, 19.2-389, 46.2-301, 46.2-301.1, 46.2-411, and 53.1-21;
	repealing §§ 46.2-355.1 through 46.2-363.
	Patrons: Stanley, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Passed by for day
	Engrossment reconsidered
	Reading of amendment waived
	Amendment by Senator Stanley agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee for Courts of Justice
C D	1123. Will contest; presumption of undue influence. Adding § 64.2-454.1.
э.в.	Patron: Obenshain
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
a =	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1124. Execution of wills; witnesses to a will required to be disinterested, definition.
	Amending § 64.2-403.
	Patron: Obenshain
	Prefiled, ordered printed, and referred to Committee on the Judiciary

S.B.	1125. Parole Board; Board, within seven days of making any decision regarding the parole of a prisoner, to provide written or electronic notice of such decision to the victim	
	of crime. Amending §§ 53.1-136 and 53.1-155. Patrons: Obenshain, et al.	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	
	Reported with amendments	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Reading of amendments waived	
	Committee amendments agreed to	. 277
	Engrossed	
	Read third time and passed	
	Reconsideration of vote on passage	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
S.B.	1126. Transportation District Commission of Hampton Roads; change in membership.	
	Amending § 33.2-1907.	
	Patron: Spruill	2.5
	Prefiled, ordered printed, and referred to Committee on Rules	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	. 238
	Continued to 2021 Special Session I in House Committee on Transportation	. 497
S.B.	. 1127. Charitable gaming; definitions, regulations, conduct of instant bingo, network bingo, pull tabs, and seal cards, report. Amending §§ 18.2-334.2, 18.2-340.16,	
	18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28,	
	18.2-340.28:1, and 18.2-340.34.	
	Patron: Reeves	
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Constitutional reading dispensed, passed by for day	
	Read second time	-
	Reading of substitute waived	
	Committee substitute rejected.	
	Reading of substitute waived	
	Reading of amendments waived.	
	Amendments by Senator Reeves agreed to	
	Engrossed	
	Read third time and passed	
c D	Continued to 2021 Special Session I in House Committee on General Laws	. 497
S.B.	. 1128. Norfolk, City of; amending charter, general updates. Patrons: Spruill, et al.	
	Prefiled, ordered printed, and referred to Committee on Local Government	2.7
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	. 103
	Read second time	113

S.B.	1133. Children's Services Act; eligibility for state pool of funds, pilot program related to
	educational placement transition for certain students with disabilities, report.
	Amending §§ 2.2-5211 and 2.2-5212.
	Patrons: Suetterlein, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
S.B.	1134. Refunding bonds; alters the principal and interest requirements, maturity date, and
	allowable discount for bonds previously issued, sunset clause. Amending first enactment
	of Chapters 265 and 408, 1992 Acts.
	Patrons: Howell, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations 28
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee on Appropriations
S.B.	•
у.р.	Amending §§ 3.2-6540 and 3.2-6542; adding §§ 3.2-6540.01 through 3.2-6540.04,
	3.2-6541.1, 3.2-6542.1, 3.2-6542.2, 3.2-6543.1, 3.2-6562.2, and 18.2-52.2.
	Patrons: Marsden, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute 237
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute rejected. 277 Reading of substitute waived . 277
	Substitute by Senator Marsden agreed to
	·
	Engrossed
	Read third time and passed
	Reconsideration of vote on passage
	Passed Senate
	Resources
C D	1136. License plates, special; range of bumper heights for any street rod bearing a street
э.Б.	rod license plate, repeals the authorizations for issuance of certain plates no longer issued
	due to low plate sales, expired authorizations, etc. Amending § 46.2-1063;
	repealing §§ 46.2-746.6, 46.2-746.9, 46.2-746.12, 46.2-747, 46.2-748, 46.2-749.10,
	46.2-749.69:1, and Section 1 of Chapter 776, 2010 Acts.
	Patrons: Marsden, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Passed by for day 211 Read second time 241
	Reading of amendment waived
	· ·
	Committee amendment agreed to
	Amendment by Senator Marsden agreed to 242
	ATTICIONED DY ACTUAL VIGINICA ASSECT TO

-531-

INDEX

S.B.	1142. Marriage ; persons who may celebrate rites, authorizes current members of the General Assembly, Governor, Lieutenant Governor, and Attorney General. Amending § 20-25.
	Patron: Cosgrove
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Engrossment reconsidered
	Passed by for day
	Reading of substitute waived
	Substitute by Senator Ebbin agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1143. Wetlands; extension of certain permits through 2021.
	Patron: Cosgrove
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1144. Aircraft civil; registration and licensing. Amending § 5.1-5.
	Patron: Stuart
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
~	Continued to 2021 Special Session I in House Committee on Transportation
	1145. Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021;
	created.
	Patron: Howell
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee on Appropriations

S.B.	1146. Income tax, state; conformity of the Commonwealth's taxation system with the Internal Revenue Code, taxable income, subtractions. Amending §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402.
	Patrons: Howell, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations 30
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Amendments by Senator Newman passed by temporarily
	Reading of amendments waived
	Amendments by Senator McDougle rejected
	Reading of amendments waived
	Amendments by Senator Newman agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
c D	Continued to 2021 Special Session I in House Committee on Finance
S.B.	1147. Nurse Loan Repayment Program; expands eligibility for Program to include
	certified nurse aids. Amending § 32.1-122.6:04.
	Patrons: Kiggans, et al. Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S.B.	1148. Primary election; changes date of election held in June from second Tuesday in
. .	June to third Tuesday in June, also changes candidate filing deadlines to reflect change of
	date. Amending §§ 24.2-311, 24.2-503, 24.2-507, 24.2-510, 24.2-515, and 24.2-515.1.
	Patrons: Kiggans, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Privileges and Elections
S.B.	1149. Nursing homes; standards of care and staff requirements, regulations.
	Amending § 32.1-127.
	Patrons: Kiggans, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
S.B.	1150. Military Spouse Liaison; position created in Department of Veterans Services,
	report, effective clause. Adding § 2.2-2002.2.
	Patrons: Kiggans, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology $\dots 30$
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with amendment

S.B.	1150 (continued) Constitutional reading dispensed, passed by for day
	Read second time
	Committee amendment agreed to
	Reading of amendment waived
	Amendment by Senator Kiggans agreed to
	Engrossed
	Read third time and passed
C D	Continued to 2021 Special Session I in House Committee on General Laws
S.B.	
	Amending § 58.1-322.02.
	Patron: Kiggans
C D	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
5.B.	1152. Appomattox, Town of; amending charter, shifts local elections from May to
	November, etc.
	Patron: Peake
	Prefiled, ordered printed, and referred to Committee on Local Government
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
~ ~	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 496
S.B.	1153. Absentee voting; ballots to be sorted and results to be reported by precinct.
	Amending § 24.2-712.
	Patrons: Suetterlein, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Constitutional reading dispensed
	Passed Senate
C D	Continued to 2021 Special Session I in House Committee on Privileges and Elections
S.B.	1154. Behavioral Health and Developmental Services, Commissioner of; reports to
	designated protection and advocacy system, incident reporting system.
	Amending § 37.2-304.
	Patron: Favola
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
~ -	Institutions
S.B.	1155. Capital outlay plan; projects to be funded from general fund-supported resources,
	repeals existing six-year capital outlay for projects to be funded. Repealing Chapter 1134,
	2020 Acts.
	Patron: Howell
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Kenoriea wun amenament

S.B.	1155 (continued) Constitutional reading dispensed, passed by for day
	Read second time 125
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee on Appropriations
S.B.	1156. Technology Development Grant Fund; created. Adding § 59.1-284.38.
	Patron: Howell
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee on Appropriations
S.B.	1157. Municipal elections; shifting elections to November. Amending § 15.2-1400.
	Patrons: Spruill, et al.
	Prefiled, ordered printed, and referred to Committee on Local Government
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Amendment by Senator Petersen withdrawn
	Engrossed
	Read third time
	Tie vote, Chair votes Yea
	Passed Senate
c n	Continued to 2021 Special Session I in House Committee on Privileges and Elections
5. B.	1158. Port of Virginia tax credits; extends the sunset date. Amending §§ 58.1-439.12:06, 58.1-439.12:09, and 58.1-439.12:10.
	Patrons: Spruill, et al. Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Finance
S R	1159. Sick leave; use for the care of immediate family members. Adding § 40.1-28.6:1.
J. .	Patrons: Favola, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
S.B.	1160. Removal of vehicles involved in accidents; lien of keeper of vehicles, limitation on
	removal and sale of abandoned vehicles, notice of intent to auction and sell vehicle,
	posting requirements. Amending §§ 46.2-644.01, 46.2-644.02, 46.2-644.03, 46.2-1200.2,
	46.2-1202, 46.2-1202.1, 46.2-1203, 46.2-1209, and 46.2-1212.1; adding §§ 46.2-644.04,
	46.2-1200.3, and 46.2-1202.2.
	Patrons: Suetterlein, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported with substitute
	Constitutional reading dispensed, passed by for day

S.B.	1160 (continued) Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Transportation
S.B.	1161. Virginia Soil and Water Conservation Board; clarifies membership.
	Amending § 10.1-502.
	Patrons: Hanger, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1162. Agricultural best management practices; creates an enhanced individual and
	corporate income tax credit beginning in taxable year 2021 but before January 1, 2025, for
	the implementation of certain practices by the taxpayer that are required as part of a
	certified resource management plan. Amending §§ 58.1-339.3 and 58.1-439.5.
	Patrons: Hanger, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Finance
R	1163. Agricultural equipment; establishes a refundable individual and corporate income
	tax credit. Amending §§ 58.1-334, 58.1-337, 58.1-432, and 58.1-436.
	Patrons: Hanger, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with amendments
	•
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Finance

S.B.	1164. Advanced recycling, etc.; definitions. Amending § 10.1-1400.	
	Patrons: Hanger, et al.	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural	
	Resources	32
	Reported with substitute	. 379
	Constitutional reading dispensed, passed by for day), 431
	Read second time	. 472
	Reading of substitute waived	. 472
	Committee substitute agreed to	. 472
	Engrossed	. 472
	Constitutional reading dispensed	. 475
	Passed Senate	. 476
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Nat	
	Resources	
S.B.	1165. Death penalty; abolition of current penalty, sentences changed to life	
	imprisonment, repeals provisions referring to trial of capital cases, etc.	
	Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10,	
	18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01,	
	19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157,	
	19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1,	
	19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2,	
	19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204,	
	53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17,	
	19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236.	
	Patrons: Surovell, et al.	22
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute rejected.	
	Reading of substitute waived	
	Substitute by Senator Stanley rejected	
	Substitute No. 1 by Senator Norment withdrawn	
	Substitute by Senator Surovell passed by temporarily	
	Substitute No. 2 by Senator Norment passed by temporarily	
	Reading of substitute waived	
	Substitute No. 3 by Senator Norment rejected	
	Reading of substitute waived	
	Substitute No. 2 by Senator Norment rejected	. 340
	Reading of substitute waived	. 340
	Substitute by Senator Surovell agreed to	. 340
	Engrossed	. 340
	Read third time and passed	. 356
	Continued to 2021 Special Session I in House Committee for Courts of Justice	. 497

S.B.	1166. Nursing professional; establishes a nonrefundable individual income tax credit for certified nurse aides, etc. Adding § 58.1-339.13.
	Patron: Kiggans
S.B.	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	United States military. Amending §§ 54.1-3018, 54.1-3021, and 54.1-3024.
	Patron: Kiggans
	Prefiled, ordered printed, and referred to Committee on Education and Health
S.B.	1168. "Abused or neglected child;" definition. Amending § 16.1-228.
	Patrons: Lucas, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1169. High school student parking passes; valid driver's license or driver privilege card
	required. Adding § 22.1-205.1.
	Patron: Norment
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Education
S.B.	1170. Isle of Wight County; authorized to impose an additional local sales and use tax to
S.12.	support schools. Amending §§ 58.1-602 and 58.1-605.
	Patron: Norment
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time and engrossed
	Read third time
	Defeated by Senate
	Reconsideration of defeat
	Passed by for day
	Passed by temporarily
	Passed Senate
~ ~	Continued to 2021 Special Session I in House Committee on Finance
S.B.	1171. Conflict of Interests Act, State and Local Government; disclosure requirements,
	industrial development and economic development authorities. Amending § 2.2-3115.
	Patron: Norment
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 33
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment rejected
	Reading of amendments waived
	Amendments by Senator Barker agreed to

	Lead third time and passed
	172. Capitol Square Preservation Council; powers and duties, review and approval of
	lans for changes to artifacts contained within the Capitol Building. Amending § 30-194.
_	lation: Norment
	refiled, ordered printed, and referred to Committee on Rules
	Reported
	Constitutional reading dispensed, passed by for day
	Lead second time and engrossed
	Lead third time and passed
	Continued to 2021 Special Session I in House Committee on Rules
S.B.	· · · · · · · · · · · · · · · · · · ·
	Amending §§ 3.2-3100 and 3.2-3102.
	atton: Norment
	refiled, ordered printed, and referred to Committee on Rules
	1174. Diversity, Equity, and Inclusion, Director of; reporting requirements.
	Amending § 2.2-435.12.
	atron: Norment
	refiled, ordered printed, and referred to Committee on General Laws and Technology
	175. Brunswick County school board; removes school board from the list of approved
	nember salaries for appointed school boards. Amending § 22.1-32.
	atron: Ruff
	refiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Constitutional reading dispensed, passed by for day
	Lead second time and engrossed
	the and third time and passed
	Continued to 2021 Special Session I in House Committee on Education
	176. Barrier crimes; amends the current requirements for the Department of Behavioral
	Health and Developmental Services to provide, etc. Amending §§ 37.2-314, 37.2-408.1,
	7.2-416, and 37.2-506.
	atron: Ruff
P	refiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
C	Constitutional reading dispensed, passed by for day
R	Lead second time
R	Reading of substitute waived
C	Committee substitute agreed to
E	Engrossed
R	tead third time and passed
C	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S.B. 1	177. Public weighmasters; allows a business entity to apply to the Commissioner of
Α	agriculture and Consumer Services for a license. Amending § 3.2-5802.
P	atron: Ruff
P	refiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
S.B. 1	178. Genetic counseling; repeals conscience clause. Repealing § 54.1-2957.21.
	atron: Ebbin
P	refiled, ordered printed, and referred to Committee on Education and Health

B. 1178 (continued) Reported	251
Constitutional reading dispensed, passed by for day	285, 286
Read second time and engrossed	308
Read third time and passed	330
Statement on vote	331
Continued to 2021 Special Session I in House Committee on Health, Welfare and Institutions	401
B. 1179. Corrections Private Management Act; name change, private management	7/
prohibited, repeals provisions relating to authority of security employees, etc.	
Amending §§ 2.2-1837, 2.2-3703, 8.01-195.10, 8.01-690, 53.1-1, 53.1-31.1, 53.1-261,	
53.1-262, and 53.1-265; repealing §§ 53.1-264, and 53.1-266.	
Patrons: Ebbin, et al.	
Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services.	34
B. 1180. Civil actions; actions filed on behalf of multiple similarly situated persons.	
Amending § 8.01-267.1.	
Patron: Surovell	
Prefiled, ordered printed, and referred to Committee on the Judiciary	34
Reported with substitute	
Constitutional reading dispensed, passed by for day	
Read second time	
Reading of substitute waived	
Committee substitute agreed to	
Reading of amendments waived	308
Amendments by Senator Surovell agreed to	
Engrossed	308
Read third time and passed	331
Continued to 2021 Special Session I in House Committee for Courts of Justice	497
B. 1181. Special immigrant juvenile status; permits the juvenile and domestic relations	
district court to retain jurisdiction in certain cases, etc. Amending § 16.1-241.	
Patrons: Surovell, et al.	
Prefiled, ordered printed, and referred to Committee on the Judiciary	
Reported	
Constitutional reading dispensed, passed by for day	
Read second time and engrossed	
Read third time and passed	
Continued to 2021 Special Session I in House Committee for Courts of Justice	497
B. 1182. Motor vehicle liability insurance; increases coverage amounts.	
Amending §§ 46.2-419, 46.2-472, and 46.2-2057.	
Patrons: Surovell, et al.	
Prefiled, ordered printed, and referred to Committee on Commerce and Labor	
Reported	
Constitutional reading dispensed, passed by for day	
Passed by for day	
Read second time and engrossed	
Read third time and passed	
Continued to 2021 Special Session I in House Committee on Labor and Commerce	497
B. 1183. Property owners' associations, boards of directors, unit owners' associations,	
etc.; meetings to be held entirely or partially by electronic means, provided guidelines	
have been adopted. Amending §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832,	
55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953.	
Patrons: Dunnavant, et al. Prefiled, ordered printed, and referred to Committee on General Laws and Technology	35
Premen progred printed and reterred to Lammittee on Laneral Laws and Lechnology	4.

S.B.	1183 (continued) Reported	52
	Constitutional reading dispensed, passed by for day	35
	Read second time and engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on General Laws	
S.B.	1184. Standby guardianship; triggering event, definitions, court approval of guardian.	
	Amending §§ 16.1-349 through 16.1-353.	
	Patrons: Deeds, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	35
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
C D	1185. Assisted living facilities; residents that are auxiliary grant recipients.	,,
э.р.	Amending § 51.5-160.	
	Patrons: Dunnavant, et al.	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 3	, 5
	Reported with amendments	
c D	Rereferred to Committee on Finance and Appropriations) /
э.Б.	1186. Landfill siting; prohibits the construction of any new municipal solid waste landfill	
	within three miles of any designated historic district, etc. Amending § 10.1-1408.4.	
	Patrons: Hashmi, et al. Profiled and and referred to Committee on Assignifican Consequentian and Natural	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural Resources	, 5
c D	1187. Physical therapy; extends time allowed for a therapist to evaluate and treat patients.	,,
э.Б.	Amending § 54.1-3482.	
	Patrons: Hashmi, et al. Prefiled, ordered printed, and referred to Committee on Education and Health) E
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	Jō
	Continued to 2021 Special Session I in House Committee on Health, Welfare and	77
c n	Institutions) /
S.B.	1188. Virginia Agriculture Food Assistance Program and Fund; established and	
	created, guidelines and regulations. Adding §§ 3.2-4780 through 3.2-4783.	
	Patrons: Hashmi, et al.	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural	
	Resources	
	Reported	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendment waived	
	Amendment by Senator Norment agreed to	
	Engrossed	
	Read third time and passed) 5

S.B.	1188 (continued)
	Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1189. Occupational therapists; licensure, authorizes Virginia to become a signatory to
	the Occupational Therapy Interjurisdictional Licensure Compact. Adding § 54.1-2956.7:1.
	Patrons: Hashmi, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S.B.	1190. Health Standards of Learning; advanced directive education for high school
	students.
	Patrons: Kiggans, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Education
S.B.	1191. School nurses; excludes nurse positions from requirements for student support
	positions and instead requires each local school board to employ at least one full-time
	equivalent school nurse position in each elementary school, middle school, and high
	school, Department of Education to establish and administer a waiver process for local
	school boards for which the requirements create an undue hardship.
	Amending §§ 22.1-253.13:2, 22.1-274, and 22.1-274.01:1.
	Patrons: Kiggans, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with amendment
	Rereferred to Committee on Finance and Appropriations
S.B.	1192. Naturopathic doctors; Department of Health Professions to amend its regulations.
	Patron: Kiggans
	Prefiled, ordered printed, and referred to Committee on Education and Health
S.B.	1193. Dairy Producer Margin Coverage Premium Assistance Fund and Program;
	created and established, eligible dairy producer shall apply to the Department of
	Conservation and Recreation by February 1 of each year to participate, effective clause.
	Adding §§ 3.2-3304 through 3.2-3307.
	Patron: Obenshain
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time 276

S.B.	1193 (continued) Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Reconsideration of vote on passage
	Passed Senate
	Resources
SR	1194. Produce safety; removes the sunset date.
у.Б.	Patron: Obenshain
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1195. Motor vehicle insurance; underinsured motor vehicle, uninsured motorist
	coverage, policies issued or renewed on or after January 1, 2022. Amending § 38.2-2206.
	Patrons: Obenshain, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1196. Teachers and other licensed school board employees; cultural competency.
	Amending §§ 22.1-253.13:5 and 22.1-298.1; adding § 22.1-298.7.
	Patrons: Locke, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Rereferred to Committee on Finance and Appropriations 75 Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Education
S R	1197. Virginia housing opportunity; tax credit established starting in taxable year 2021,
э.р.	which is equal to the amount of the federal low-income housing tax credit allocated.
	Adding §§ 58.1-439.29 and 58.1-439.30.
	Patrons: Locke, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations 36
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Engrossed
	Constitutional reading dispensed

S.B.	1197 (continued)
	Passed Senate
~ -	Continued to 2021 Special Session I in House Committee on Finance
S.B.	1198. Government Data Collection and Dissemination Practices Act; license plate
	readers shall not be used to collect or maintain personal information, etc., definitions.
	Amending §§ 2.2-3800, 2.2-3801, 2.2-3802, and 52-48.
	Patrons: Petersen, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 37
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments rejected
	Reading of substitute waived
	Passed by temporarily
	Substitute by Senator Petersen agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Public Safety
S.B.	1199. Conservation easements; construction. Adding §§ 10.1-1016.1 and 10.1-1705.1.
	Patron: Petersen
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Parliamentary inquiry
	Passed by for day
	Read second time
	Reading of amendments waived
	Passed by for day
	Committee amendments agreed to
	Ruling of the Chair
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
c n	Resources
5.В.	1200. Waste disposal; requires any application to store, provide treatment for, or dispose
	of hazardous waste or for a new solid waste management facility permit.
	Amending §§ 10.1-1408.1 and 10.1-1426.
	Patrons: Hashmi, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
~ ~	Resources
S.B.	1201. Energy storage systems; definitions, tax exemption, revenue share for systems.
	Amending §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660.
	Patron: Petersen
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations 37
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed

S.B.	1201 (continued) Read third time and passed	48
	Continued to 2021 Special Session I in House Committee on Finance	
C D	1202. Uninsured and underinsured motorist insurance policies; bad faith.	7 1
э.р.	Amending §§ 8.01-66.1 and 38.2-2206.	
	Patron: Petersen	
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor	37
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Labor and Commerce	
c D	1203. Hate crimes; associational relationships, penalty. Amending §§ 8.01-42.1,	11
5.Б.		
	8.01-49.1, 18.2-51, 18.2-57, 18.2-121, and 52-8.5.	
	Patrons: Hashmi, et al.	٠ -
c n	Prefiled, ordered printed, and referred to Committee on the Judiciary) ,
5.В.	1204. George Mason University; management agreement with the Commonwealth.	
	Patrons: Barker, et al.	
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations	
	Reported with amendments	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendments waived	
	Committee amendments agreed to	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Education	97
S.B.	1205. Career fatigue and wellness in certain health care providers; programs to	
	address, civil immunity. Amending §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909.	
	Patron: Barker	
	Prefiled, ordered printed, and referred to Committee on Education and Health	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	96
	Continued to 2021 Special Session I in House Committee on Health, Welfare and	
	Institutions	97
S.B.	1206. Juvenile records; confidentiality of records relevant to treatment, services, etc.,	
	exceptions. Amending § 16.1-300.	
	Patrons: Barker, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to	
	Reading of amendments waived	
	Amendments by Senator Barker agreed to	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	97

S.B.	1207. Solar and energy storage projects ; siting agreements throughout the Commonwealth. Amending §§ 15.2-2288.8 and 15.2-2316.6 through 15.2-2316.9.	
	Patrons: Barker, et al.	
	Prefiled, ordered printed, and referred to Committee on Local Government	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	59
	Committee substitute agreed to	59
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 4	96
S.B.	1208. Continuity of government; extends to 12 months the period of time after an enemy	
	attack or other disaster that a locality may, by ordinance, provide. Amending § 15.2-1413.	
	Patron: Barker	
	Prefiled, ordered printed, and referred to Committee on Local Government	38
	Reported	9(
	Constitutional reading dispensed, passed by for day	03
	Read second time and engrossed	13
	Read third time and passed	24
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 4	96
S.B.	1209. Subcontractor's employees; liability of general contractor for wages.	
	Amending § 11-4.6.	
	Patron: Petersen	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	38
	Reported with amendment	56
	Read first time4	21
	Read second time	72
	Reading of amendment waived	73
	Committee amendment agreed to	73
	Engrossed	73
	Constitutional reading dispensed	75
	Passed Senate	76
	Continued to 2021 Special Session I in House Committee for Courts of Justice	97
S.B.	1210. Permit fee schedules; Department of Environmental Quality shall convene a work	
	group to revise schedule for nonhazardous solid waste management facilities.	
	Patron: Petersen	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural	
	Resources	38
	Reported with substitute	37
	Rereferred to Committee on Finance and Appropriations	38
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to	
	Engrossed	
	Constitutional reading dispensed	

S.B.	1210 (continued)
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
C D	Resources
э.в.	Safety Trust Fund. Amending §§ 46.2-686 and 46.2-694; adding § 46.2-694.2.
	Patrons: Edwards, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations 38
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed.
	Continued to 2021 Special Session I in House Committee on Transportation
SR	1212. New River Valley Passenger Rail Station Authority; creation of authority in
у.р.	Planning District 4 (New River Valley RC). Adding §§ 33.2-3800 through 33.2-3816.
	Patrons: Edwards, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Transportation
S.B.	1213. Restricted licenses; authorizes DMV to issue restricted credentials to individuals
	with driver's license suspensions resulting from drug-related offenses.
	Amending § 18.2-271.1.
	Patrons: Edwards, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
C P	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1214. Metropolitan Washington Airports Authority; repeals effective date for creation
	of Authority. Repealing § 5.1-178.
	Patron: Edwards Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Transportation
S R	1215. Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion
э.р.	from dwelling unit, interruption of services, or actions taken to make premises unsafe,
	repeals provision relating to tenant's remedies for landlord's unlawful ouster.
	Adding § 55.1-1243.1; repealing § 55.1-1243.
	Patrons: Ebbin, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 39

C D	1015 ()	
S.B	. 1215 (continued) Reported with amendment	250
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendment waived.	
	Committee amendment agreed to	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on General Laws	
S B	. 1216. Crewe, Town of; amending charter, various changes to the charter including	т) /
э.Б	staggering town council elections, etc.	
	Patron: Ruff	
	Prefiled, ordered printed, and referred to Committee on Local Government	30
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns	
S R	. 1217. Secondary state highway system; taking certain private roads into the system.	T)(
у.п	Amending § 33.2-335.	
	Patron: Ruff	
	Prefiled, ordered printed, and referred to Committee on Transportation	30
SB	. 1218. Naturopathic doctors; Board of Medicine to license and regulate.	5,
S.D	Amending §§ 54.1-2900, 54.1-2901, 54.1-2914, 54.1-2973.1, and 54.1-3401;	
	adding §§ 54.1-2956.15 through 54.1-2956.18.	
	Patron: Petersen	
	Prefiled, ordered printed, and referred to Committee on Education and Health	. 39
S.B	. 1219. Paid family leave; State Corporation Commission's Bureau of Insurance to review	
	and make recommendations, report.	
	Patrons: Favola, et al.	
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor	. 39
	Reported with amendment	97
	Constitutional reading dispensed, passed by for day	115
	Read second time	127
	Reading of amendment waived	127
	Committee amendment agreed to	
	Engrossed	127
	Read third time and passed	142
	Statement on vote	142
	Continued to 2021 Special Session I in House Committee on Labor and Commerce	
S.B	. 1220. State facilities; repeals requirements that the Commissioner of Behavioral Health	
	and Developmental Services determine the nationality of each person admitted to a state	
	facility. Repealing § 37.2-827.	
	Patrons: Favola, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	. 39
	Reported	119
	Constitutional reading dispensed, passed by for day	147
	Read second time and engrossed	162
	Read third time and passed	208
	Continued to 2021 Special Session I in House Committee on Health, Welfare and	
	Institutions	10

S.B.	1221. Loudoun County; County to enter into a contract with the Department of Health for the local administration of local health services.
	Patrons: Favola, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S.B.	1222. Virginia Public Procurement Act; a state agency, whenever purchasing personal
	protective equipment (PPE) for public use, to purchase from a Virginia-based company or
	manufacturer. Adding § 2.2-4328.1.
	Patron: DeSteph
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 40
	Reported
~ _	Rereferred to Committee on Finance and Appropriations
S.B.	1223. Virginia Energy Plan; amends Plan to include an analysis of electric vehicle
	charging infrastructure. Amending §§ 67-102, 67-201, and 67-202.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
c D	Continued to 2021 Special Session I in House Committee on Labor and Commerce
э.р.	conservation. Amending § 36-99.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 40
S R	1225. Broadband services; authorizes school boards to appropriate funds for the purposes
у.р.	of promoting, facilitating, and encouraging the expansion and operation of services for
	educational purposes. Adding §§ 15.2-986 and 22.1-79.9.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Rereferred to Committee on Education and Health
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Education
S.B.	1226. Compensation Board; determining staffing and salaries for an attorney for the
	Commonwealth, effective clause. Amending §§ 15.2-1626 and 15.2-1636.8.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on Local Government
	Rereferred to Committee on the Judiciary
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with amendment

S.B. 1	1226 (continued) Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
	1227. Hormonal contraceptives; payment of medical assistance for 12-month supply.
	Amending § 32.1-325.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with amendment
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
`	Institutions
SR 1	1228. Virginia Equal Pay Act; civil penalties, repealing provision relating to equal pay
	respective of sex. Adding §§ 40.1-28.13 through 40.1-28.17; repealing § 40.1-28.6.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	1229. License plates, special; issuance for supporters of Ducks Unlimited, fees.
	Amending § 46.2-749.7.
	Patron: Stuart
I	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported with substitute
(Constitutional reading dispensed, passed by for day
F	Read second time
I	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Transportation
	1230. Criminal cases; any criminal proceeding pending in a juvenile and domestic
	relations district court may be transferred to the general district court in the same territorial
j	urisdiction for disposition. Amending § 16.1-241.
I	Patron: Chase
I	Prefiled, ordered printed, and referred to Committee on the Judiciary
	1231. General district courts; filing an order of disposition from a criminal case.
A	Adding § 16.1-305.3.
F	Patron: Chase
F	Prefiled, ordered printed, and referred to Committee on the Judiciary

S.B.	1232. Geriatric prisoners; prohibits persons serving a sentence imposed upon a conviction of murder in the first or second degree or murder of a pregnant woman, etc.
	Amending § 53.1-40.01.
S.B.	Patron: Chase Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 41 1233. Campaign contribution limits; prohibits persons from making any single contribution, or any combination of contributions, that exceeds \$20,000 to any one candidate for Governor, Lieutenant Governor, Attorney General, or the General Assembly
	in any one election cycle. Adding §§ 24.2-948.5 through 24.2-948.8 and 24.2-953.6.
	Patron: Petersen
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections $\dots \dots \dots$
S.B.	1234. Virginia State Bar examination; foreign applicants. Amending § 54.1-3926.
	Patron: Petersen
	Prefiled, ordered printed, and referred to Committee on the Judiciary $\dots \dots \dots$
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
C D	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1235. Health, Department of; certain communication prohibited regarding matters with a minor, exception. Adding § 32.1-4.1.
	Patrons: Peake, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Passed by for day
	Engrossment reconsidered
	Reading of amendment waived
	Passed by temporarily
	Amendment No. 1 by Senator Peake withdrawn
	Reading of amendment waived
	Amendment No. 2 by Senator Peake agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S.B.	1236. Campaign finance; prohibits candidates, campaign committees, and political
	committees from soliciting or accepting contributions from any public utility, etc.
	Adding § 24.2-947.4:2.
	Patrons: Petersen, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections

	1237. Emergency and quarantine orders, certain; additional procedural requirements. Amending § 32.1-48.010; adding § 44-146.17:1.1.
	Patron: Petersen
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time
	Passed Senate
	Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
	1238. School security officers; carrying of firearms by United States Armed Forces veterans and certain employees of local school boards. Amending § 22.1-280.2:1. Patron: Chase
	Prefiled, ordered printed, and referred to Committee on Education and Health
S.B.	1239. Absentee voting; third-party absentee ballot assembly and distribution. Amending § 24.2-706.
	Patrons: Bell, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Privileges and Elections 49
S.B.	1240. Criminal sexual assault; penalty when committed by parents, etc., person is 13
	years of age, etc. Amending §§ 18.2-67.1, 18.2-67.2, and 18.2-366.
	Patron: Stuart
S.B.	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Patron: Stuart
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Constitutional reading dispensed, passed by for day
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
э.в.	1242. Personal appearance by two-way electronic video and audio communication; entry of plea or nolle prosequi, adjudication of probation violations. Amending § 19.2-3.1.
	Patron: Edwards
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	•
	Constitutional reading dispensed, passed by for day
	Constitutional reading dispensed, passed by for day
	Constitutional reading dispensed, passed by for day
	Reported
S.B.	Constitutional reading dispensed, passed by for day
S.B.	Constitutional reading dispensed, passed by for day 14 Read second time and engrossed 159, 16 Read third time and passed 207-0 Continued to 2021 Special Session I in House Committee for Courts of Justice 49

S.B. 1243 (continued)	
17.1-275, 17.1-275.8, 17.1-805, 18.2-46.1, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251,	
18.2-251.02, 18.2-251.03, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258	
through 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.09,	
18.2-308.1:5, 18.2-308.4, 18.2-460, 18.2-474.1, 18.2-513, 19.2-11.2, 19.2-66, 19.2-81.1,	
19.2-83.1, 19.2-120, 19.2-120.1, 19.2-188.1, 19.2-215.1, 19.2-291.1, 19.2-299,	
19.2-299.2, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-386.28, 19.2-389,	
19.2-392.02, 19.2-392.2, 22.1-277.08, 22.1-315, 24.2-233, 37.2-314, 37.2-416, 37.2-506,	
48-17, 52-8.1:1, 52-35, 53.1-220.1, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, and	
54.1-3442.8; adding §§ 3.2-4122 through 3.2-4199.6 and 19.2-392.2:1;	
repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1 through 18.2-251.1:3, and 19.2-389.3.	
Patron: Morrissey	
Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	42
S.B. 1244. Sexually violent predators; repeals sections authorizing civil commitment.	
Amending §§ 2.2-3703, 2.2-3705.2, 2.2-3711, 9.1-101, 9.1-177.1, 9.1-907, 9.1-908,	
10.1-104.7, 16.1-69.55, 16.1-300, 16.1-305, 17.1-213, 17.1-805, 19.2-11.01, 19.2-169.3,	
19.2-174.1, 19.2-299, 19.2-301, 19.2-388, 19.2-389, 19.2-389.1, 19.2-392.02, 37.2-844,	
37.2-845, 37.2-846, 37.2-1102, 44-146.18:4, 44-146.22, 53.1-136, 53.1-145, and	
63.2-105; repealing §§ 37.2-900 through 37.2-921.	
Patrons: Morrissey, et al.	
Prefiled, ordered printed, and referred to Committee on the Judiciary	43
S.B. 1245. Absentee voting; establishment of drop-off locations for return of absentee ballots,	
ballot defects, cure process. Amending §§ 24.2-706, 24.2-707, 24.2-709, 24.2-709.1, and	
24.2-710; adding § 24.2-707.1.	
Patrons: Deeds, et al.	
Prefiled, ordered printed, and referred to Committee on Privileges and Elections	43
Reported with substitute	
Constitutional reading dispensed, passed by for day	
Read second time	
Reading of substitute waived	
Committee substitute agreed to.	
Engrossed	
Read third time and passed	
· · · · · · · · · · · · · · · · · · ·	
Continued to 2021 Special Session I in House Committee on Privileges and Elections	49/
S.B. 1246. Absentee voting; mandatory processing of returned absentee ballots before election	
day, central absentee voter precinct in the office of the general registrar.	
Amending §§ 24.2-709.1 and 24.2-712.	
Patrons: Deeds, et al.	
Prefiled, ordered printed, and referred to Committee on Privileges and Elections	
Reported with substitute	
Constitutional reading dispensed, passed by for day	
Passed by for day	276, 302
Read second time	335
Reading of substitute waived	335
Committee substitute agreed to	335
Engrossed	336
Constitutional reading dispensed	. 336-37
Passed Senate	
Continued to 2021 Special Session Lin House Committee on Privileges and Elections	

S.B.	1247. Electric generating facility closures; public closure, integrated resource plans, provisions apply to facility owned or operated by an electric utility. Amending § 56-599; adding § 45.1-394.1.
	Patron: Deeds
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
S.B.	1248. Juveniles; competency evaluation, appointed evaluator or director of community
	services board, etc., shall acknowledge receipt of court order to clerk of court.
	Amending § 16.1-356.
	Patron: Deeds
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1249. Local planning commissions; adds rezoning or generalized development plans,
	special use permits, etc., to various review deadlines and requirements.
	Amending § 15.2-2259.
	Patron: Stuart
	Prefiled, ordered printed, and referred to Committee on Local Government
S.B.	1250. Firearm rentals; criminal history record information check required, penalty,
	effective date. Amending § 18.2-308.2:2.
	Patrons: Deeds, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Engrossment reconsidered
~ ~	Recommitted to Committee on the Judiciary
S.B.	1251. Virginia Retirement System; amendments to reflect recent changes to federal law.
	Amending §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308.
	Patron: Newman
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported
	CONSTITUTION AT TEAUTIFY UISDENSEU, DASSEU DV TOT UAV

S.B.	1251 (continued) Read second time and engrossed
	Read third time and passed
S.B.	Continued to 2021 Special Session I in House Committee on Appropriations
	Enhancement Tax Credits; sunset date, credits earned prior to January 1, 2022.
	Amending §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1.
	Patrons: McPike, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
c D	Continued to 2021 Special Session I in House Committee on Finance
5.В.	1253. Access roads to economic development sites; criteria for use of funds. Amending § 33.2-1509.
	Patrons: McPike, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Transportation
S.B.	1254. Sports betting; clarifies the procedures by which the Virginia Lottery determines
	whether an event is considered youth sports, on which betting is prohibited, an
	international athletic event organized by the International Olympic Committee shall not be
	considered to be youth sports, regardless of the age of the participants.
	Amending §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100.
	Patron: McPike
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
c D	Continued to 2021 Special Session I in House Committee on General Laws
э.Б.	registrations during an emergency. Amending § 38.2-200.
	Patron: Mason
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived. 126
	Committee amendment agreed to
	Engrossed
	Read third time

S.B.	1255 (continued) Passed Senate
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
C D	1256. Criminal Justice Services Board and Committee on Training; membership,
э.р.	law-enforcement training. Amending §§ 9.1-102, 9.1-108, and 9.1-112.
	Patron: Marsden
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Public Safety
C D	1257. Standards of Quality; each school board shall provide at least three specialized
э.р.	student support positions per 1,000 students, effective clause. Amending §§ 22.1-253.13:2
	and 22.1-274.
	Patrons: McClellan, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Read second time
	Reading of substitute waived
	Engrossed
	Engrossed 400 Engrossment reconsidered 449
	Reading of amendment waived
	Amendment by Senator McClellan agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Education
c D	1258. Solar projects; State Water Control Board to administer a Virginia Erosion and
э.р.	Sediment Control Program (VESCP) on behalf of any locality that notifies the Department
	of Environmental Quality that it has chosen not to administer a VESCP, effective clause.
	Adding § 62.1-44.15:55.1.
	Patrons: Marsden, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources

S.B.	1259. Virginia Highway Corporation Act; alteration of certificate of authority, powers and duties of State Corporation Commission, agreements between toll operator and	
	Department. Amending §§ 56-539 and 56-542.	
	Patrons: Bell, et al.	
	Prefiled, ordered printed, and referred to Committee on Transportation	45
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to	343
	Reading of amendment waived	343
	Amendment by Senator Marsden rejected	343
	Engrossed	343
	Read third time and passed	359
	Continued to 2021 Special Session I in House Committee on Transportation	497
S.B.	1260. Transportation purposes; inspection of property to ascertain suitability of the	
	property for highway and other transportation purposes. Amending §§ 25.1-203 and	
	33.2-1011.	
	Patron: Bell	
	Prefiled, ordered printed, and referred to Committee on Transportation	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to.	
	Engrossed	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Transportation	
SR	1261. Court of Appeals; expands jurisdiction, increases from 11 to 17 number of judges	
J.D.	on Court, effective clause. Amending §§ 2.2-511, 8.01-36, 8.01-267.8, 8.01-383.1,	
	8.01-555, 8.01-626, 8.01-670, 8.01-671, 8.01-675.3, 8.01-676.1, 9.1-909, 15.2-1627,	
	15.2-1643, 15.2-2139, 15.2-2140, 15.2-2656, 15.2-3104, 15.2-3217, 15.2-3221,	
	15.2-3222, 15.2-3227, 15.2-3244, 15.2-3308, 15.2-3528, 15.2-3605, 15.2-3809,	
	15.2-3909, 15.2-4108, 15.2-4120, 15.2-5218, 15.2-5367, 15.2-6606, 15.2-6632,	
	15.2-7406, 16.1-279.1, 17.1-309, 17.1-400 through 17.1-403, 17.1-405 through 17.1-408,	
	$17.1-410,\ 17.1-413,\ 17.1-503,\ 17.1-513,\ 18.2-308.08,\ 18.2-384,\ 19.2-152.10,\ 19.2-165,$	
	19.2-321.1, 19.2-321.2, 19.2-322.1, 19.2-386.13, 19.2-402, 19.2-403, 19.2-404, 22.1-97,	
	22.1-289.024, 24.2-237, 24.2-422, 24.2-433, 25.1-239, 32.1-48.010, 32.1-48.013,	
	33.2-928, 33.2-2917, 37.2-920, 45.1-161.322, 55.1-1833, 55.1-1966, 55.1-2211, 57-2.02,	
	58.1-527, 58.1-1828, 58.1-2282, 58.1-3147, 58.1-3992, and 63.2-1710;	
	adding §§ 8.01-675.5 and 8.01-675.6; repealing §§ 8.01-670.1 and 8.01-672.	
	Patrons: Edwards, et al.	4.6
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported with amendment	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Committee substitute agreed to.	
	Reading of amendment waived	401

S.B.	1261 (continued)
	Committee amendment agreed to
	Engrossed
	Read third time and passed
C D	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1262. Restricted permit; prepayment of fines and costs. Amending § 18.2-271.1.
	Patron: Morrissey
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
c D	Continued to 2021 Special Session I in House Committee for Courts of Justice
5.B.	1263. Bicycles; permits operators to treat a stop sign as a yield sign in certain situations.
	Amending §§ 46.2-839 and 46.2-905; adding § 46.2-903.1.
	Patrons: Morrissey, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
c D	1264. Emergency and preliminary protective orders; expungement of orders.
S.D.	Amending § 19.2-392.4; adding § 19.2-392.2:1.
	Patron: Morrissey
	Prefiled, ordered printed, and referred to Committee on the Judiciary
S R	1265. Natural gas pipelines; Department of Environmental Quality to conduct
. .	inspections of the land-disturbing activities related to construction of any transmission
	pipeline, where substantial adverse impacts or likely adverse impacts are found on a
	repeated, etc., basis, the Department may issue a stop work instruction.
	Amending §§ 62.1-44.15:37.1 and 62.1-44.15:58.1.
	Patrons: Deeds, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1266. Admission to bail; rebuttable presumptions against bail, repeals provision relating
	to presumption of no bail for illegal aliens charged with certain crimes.
	Amending §§ 19.2-120 and 19.2-124; repealing § 19.2-120.1.
	Patrons: Deeds, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time

 Read second time and engrossed
 159, 161

 Read third time and passed
 207-08

 Continued to 2021 Special Session I in House Committee on General Laws
 497

	1272. Unrestorably incompetent defendant; disposition, capital murder charge, inpatient custody of the Commissioner of the Department of Behavioral Health and Developmental Services. Adding § 19.2-169.3:1. Patron: Mason
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
	1273. Behavioral Health Commission; created, report. Adding §§ 30-401 through
o	30-408.
	Patrons: Deeds, et al.
	Prefiled, ordered printed, and referred to Committee on Rules
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Rules
	1274. Wildlife corridors; various agencies to consider and incorporate, where applicable,
	government planning, Department of Wildlife Resources shall publish plans and any
	updates on its website. Amending §§ 10.1-207, 10.1-1105, 29.1-579, and 33.2-353.
	Patrons: Marsden, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Constitutional reading dispensed, passed by for day
	Read second time
	Read second time
	Read second time144Reading of amendments waived144Committee amendments agreed to144
	Read second time144Reading of amendments waived144Committee amendments agreed to144Engrossed144
	Read second time144Reading of amendments waived144Committee amendments agreed to144Engrossed144Read third time and passed158
	Read second time
	Read second time
S.B.	Read second time

S.B.	1275 (continued) Committee amendment agreed to
	Engrossed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
c D	1276. Health insurance; essential benefits, removes the prohibition on the provisions of
э.Б.	coverage for abortions. Amending § 38.2-3451.
	Patrons: McClellan, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Rereferred to Committee on Commerce and Labor
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
S R	1277. Motor Vehicles, Department of, and Supreme Court of Virginia; repeals
э.р.	reporting requirement.
	Patron: Newman
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Transportation
S.B.	1278. Charitable gaming; regulations of operators of Texas Hold'em poker games,
	conduct of games, effective clause. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.18,
	18.2-340.19, 18.2-340.22, 18.2-340.28:2, and 18.2-340.34:2.
	Patron: Bell
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 47
	Reported with amendments
	Rereferred to Committee on Finance and Appropriations
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on General Laws
S.B.	1279. Veterans Services, Department of; initiatives to reduce unemployment among
	veterans, comprehensive transition program. Amending § 2.2-2001.2.
	Patrons: Bell, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 48
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Reconsideration of vote on passage 296-97

S.B.	1279 (continued) Passed Senate	007
	Continued to 2021 Special Session I in House Committee on General Laws	
S R	1280. Dams; Virginia Soil and Water Conservation Board to enter into a negotiated	17/
э.р.	settlement with the owners of certain impounding structures. Adding § 10.1-613.6.	
	Patron: Bell	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural	
	Resources	10
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natura	
	Resources	
S R	1281. General registrar; qualifications, residency. Amending § 24.2-110.	r
у.р.	Patron: Morrissey	
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections	48
	Reported with amendments	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendments waived	
	Committee amendments agreed to	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Privileges and Elections	
S.B.	1282. Greenhouse gas emissions inventory; regulations, proprietary information.	.,
	Adding § 10.1-1307.04.	
	Patrons: Morrissey, et al.	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural	
	Resources	48
	Reported with substitute	237
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	102
	Committee substitute agreed to	
	Engrossed	102
	Read third time and passed	ļ 51
	Statement on vote	I 51
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natura	ıl
	Resources	
S.B.	1283. Criminal records, certain; establishes a process for the automatic expungement,	
	report. Amending §§ 9.1-101, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74,	
	19.2-310.7, 19.2-389.3, and 19.2-390; adding §§ 19.2-392.5 through 19.2-392.14.	
	Patrons: Morrissey, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	48
S.B.	1284. Commonwealth Clean Energy Policy; established, repeals provisions relating to	
	energy objectives and Commonwealth Energy Policy. Amending §§ 56-46.1, 56-585.1,	
	56-598, 56-601, 62.1-199, 67-103, 67-104, and 67-201; adding § 67-101.1;	
	repealing §§ 67-101 and 67-102.	
	Patrons: Favola, et al.	
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor	48

S.B.	1284 (continued) Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1285. Vacant buildings; removes requirement that a building meet the definition of
	"derelict building" for Town of Clifton Forge, Town of Pulaski, Town of Timberville, and
	any city to require, by ordinance, the owner or owners of buildings to register such
	buildings on an annual basis. Amending § 15.2-1127.
	Patron: Locke
	Prefiled, ordered printed, and referred to Committee on Local Government
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed. 124-25
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 496
S R	1286. Income tax, state; rate increase, funding for schools and law-enforcement officer
у.р.	salaries. Amending § 58.1-320; adding § 58.1-327.
	Patron: Deeds
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations 49
C D	1287. Charitable Gaming Board; prohibits the Board from promulgating regulations that
у.р.	prohibit the use of multiple video monitors or touchscreens on an electronic pull tab
	device, use of devices shall be limited to one player at a time. Amending §§ 18.2-340.19
	and 18.2-340.28.
	Patron: McPike
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 49
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on General Laws
S.B.	1288. Special education; Department of and the Board of Education to develop new
	policies and procedures, individualized education program (IEP), duty of Department to
	provide training and guidance documents to local school divisions on development of
	IEPs. Amending §§ 22.1-214, 22.1-215, 22.1-253.13:4, and 22.1-298.1;
	adding § 22.1-214.4.
	Patron: Dunnavant
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time

S.B.	1288 (continued)
	Reading of substitute waived
	Committee substitute agreed to
	Passed by for day
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Reconsideration of vote on passage
	Engrossment reconsidered
	Reading of amendments waived
	Amendments by Senator Dunnavant agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Education
S.B.	1289. Health insurance ; carrier business practices, every carrier shall include in provider
	contracts a provision that prohibits a provider from discriminating against any enrollee.
	Amending § 38.2-3407.15.
	Patron: Surovell
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
SR	1290. ConserveVirginia program; established. Amending §§ 10.1-1018.1 and
5.D .	10.1-1021; adding § 10.1-104.6:1.
	Patron: Mason
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	<u> </u>
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
G P	Resources
S.B.	1291. Virginia Water Protection Permit; withdrawal of surface water or ground water,
	plans for water auditing and leak detection and repair. Amending §§ 62.1-44.15:22 and
	62.1-262.
	Patron: Mason
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to

S R	1291 (continued)
э.р.	Engrossed
	Read third time
	Passed by for day
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1292. Electric utilities; State Corporation Commission required to direct that 100 percent
	of the amount of a utility's earnings above a certain threshold be credited to customers'
	bills. Amending § 56-585.1.
	Patron: McClellan
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
S.B.	1293. Parole Board; requires that statements issued by the Board regarding actions taken
	on the parole of prisoners to be published within 30 days of such actions.
	Amending § 53.1-136.
	Patrons: DeSteph, et al.
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 49
S.B.	1294. Police and court records; fee for filing a petition for expungement.
	Amending §§ 17.1-275 and 19.2-392.2.
	Patron: DeSteph
~ ~	Prefiled, ordered printed, and referred to Committee on the Judiciary
S.B.	1295. Electric utilities; procurement of certain equipment from Virginia-based or United
	States-based manufacturer using materials or product components made in Virginia or the
	United States, if reasonably available and competitively priced. Amending §§ 56-585.1:11
	and 56-585.5.
	Patron: DeSteph
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
S.B.	-
J.D.	Amending § 44-146.18.
	Patron: Spruill
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 50
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Public Safety
S.B.	1297. Emergency order for adult protective services; acts of violence, etc., or financial
	exploitation. Amending §§ 63.2-1603 and 63.2-1609.
	Patrons: Obenshain, et al.
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 50
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed

S.B.	1297 (continued) Read third time and passed	25
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
c D	1298. Tourism improvement districts; authorizes any locality to create.	:71
э.р.	Adding §§ 15.2-2413.11 through 15.2-2413.11.	
	Patrons: Bell, et al.	
	Prefiled, ordered printed, and referred to Committee on Local Government	5(
	Reported	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 4	
c D	1299. Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed	.90
э.р.	wine for off-premises consumption, report, sunset provision. Amending §§ 4.1-119,	
	4.1-204, 4.1-206.1, 4.1-206.3, 4.1-207, 4.1-210, 4.1-212.1, and 4.1-221.	
	Patrons: Bell, et al.	
	,	51
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services Reported with substitute	
	Read first time.	
	Read second time	
	Reading of substitute waived	
	Engrossed	
	Read third time and passed	
C D	1300. Inmates; Board of Local and Regional Jails to review services provided during	17
э.р.	pregnancy, etc., report.	
	Patrons: Favola, et al.	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	50
	Reported	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Public Safety	
S.R.	1301. Correctional facilities; restrictions on use of isolated confinement, effective clause.	,
·.D.	Adding §§ 53.1-39.2 and 66-20.1.	
	Patrons: Morrissey, et al.	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	51
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported with amendments	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to.	
	Reading of amendments waived	
	Committee amendments agreed to	
	Engrossed	
	Read third time and passed	

Prefiled, ordered printed, and referred to Committee on General Laws and Technology 51

Adding § 2.2-4303.02. Patron: McPike

S.B.	1305 (continued)	
	Reported with amendments	
	Rereferred to Committee on Finance and Appropriations	
	Reported with amendment	
	Constitutional reading dispensed, passed by for day	261
	Read second time	
	Reading of amendments waived.	
	Committee amendments rejected	283
	Reading of amendment waived	283
	Committee amendment rejected	283
	Reading of substitute waived	283
	Substitute by Senator McPike agreed to	283
	Engrossed	283
	Passed by for day	
	Engrossment reconsidered	328
	Reading of amendments waived	
	Amendments by Senator McPike agreed to	
	Engrossed	
	Constitutional reading dispensed	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on General Laws	
S.B.	1306. Assault and battery; eliminates the mandatory minimum term of confinement of	
S.25.	six months if committed against a judge, magistrate, law-enforcement officer, etc., persons	
	with physical or mental disability. Amending § 18.2-57; adding § 19.2-9.2.	
	Patrons: Morrissey, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	51
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived .	
	Committee substitute agreed to.	
	Reading of amendments waived .	
	Amendment No. 1 by Senator Edwards agreed to	
	Amendment No. 2 by Senator Edwards agreed to	
	Engrossed	
	Read third time and passed.	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
C D	1307. School-based health services; Board of Medical Assistance Services to amend the	1 フ/
э.в.	state plan for medical assistance services to provide for payment, no health care provider	
	who provides services through telemedicine shall be required to use proprietary	
	technology or applications in order to be reimbursed. Amending §§ 32.1-325 and	
	32.1-326.3.	
	Patron: Dunnavant	
		<i>-</i> 1
	Prefiled, ordered printed, and referred to Committee on Education and Health	
	Reported with substitute	251 251
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to.	
	Engrossed	
	Constitutional reading dispensed	391

S.B.	1307 (continued) Passed Senate
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S R	1308. Pedestrians; interference with traffic, penalty. Amending § 46.2-923.
э.р.	Patron: DeSteph
	Prefiled, ordered printed, and referred to Committee on Transportation
c D	1309. Local stormwater assistance; flood mitigation and protection measures.
э.Б.	Amending § 15.2-2114.01.
	- ·
	Patrons: Ebbin, et al.
	Prefiled, ordered printed, and referred to Committee on Local Government
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 496
S.B.	1310. Virginia Human Rights Act; application of laws applicable to employee safety and
	payment of wages. Amending §§ 2.2-3905, 40.1-2, 40.1-29, 40.1-49.3, and 40.1-49.8.
	Patrons: McClellan, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1311. Water quality standards; modification of permits and certifications.
	Amending § 62.1-44.15:81.
	Patron: McClellan
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Substitute by Senator Mason withdrawn
	Engrossed
	Constitutional reading dispensed
	Passed Senate 476-77
	Statement on vote
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
	11000 010 00 1 1 1 1 1 1 1 1 1 1 1 1 1

S.B.	1312. Virginia Spirits Promotion Fund; assessment on distillers. Amending § 3.2-3011; adding §§ 3.2-3012.1 and 3.2-3012.2.	
	Patron: Mason	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	50
	Reported	
	Rereferred to Committee on Finance and Appropriations	
c D	1313. Children's Services Act; funds expended special education programs, definition of	. 01
э.Б.		
	"transitional services." Amending §§ 2.2-5211 and 2.2-5212.	
	Patrons: Mason, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	385
	Committee substitute rejected.	
	Reading of substitute waived	386
	Committee substitute agreed to	386
	Engrossed	390
	Constitutional reading dispensed	391
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Education	
S.B.	1314. Education and Labor Market Alignment, Office of; established.	
	Amending § 2.2-2238.	
	Patrons: Hashmi, et al.	
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	52
	Reported	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
c B	Continued to 2021 Special Session I in House Committee on General Laws	49
5.B.	1315. Criminal proceedings; consideration of mental condition and intellectual and	
	developmental disabilities, evidence offered by the defendant. Amending §§ 19.2-120,	
	19.2-163.03, and 19.2-299; adding § 19.2-271.6.	
	Patrons: McClellan, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	371
	Read second time	
	Reading of substitute waived	404
	Committee substitute agreed to	404
	Engrossed	405
	Read third time	452
	Passed by temporarily	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	

S.B.	1316. Child care providers; background checks portability, check shall not be required if	
	individual completed a background check within previous five years, pilot program, report. Amending §§ 19.2-389, 22.1-289.035, 22.1-289.039, 63.2-1720.1, and 63.2-1724.	
	Patrons: McClellan, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	53
	Rereferred to Committee on Education and Health	
	Reported with amendments	
	Rereferred to Committee on Finance and Appropriations	
	Reported with amendments	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendments waived	
	Committee amendments agreed to	
	Reading of amendments waived	
	Committee amendments agreed to	
	Engrossed	
	Read third time and passed	
	Statement on vote	
	Continued to 2021 Special Session I in House Committee on Health, Welfare and	: 32
	Institutions	0.7
c D	1317. Local school boards; open enrollment policy required. Amending § 22.1-7.1.	91
э.Б.	Patron: Dunnayant	
	Prefiled, ordered printed, and referred to Committee on Education and Health	50
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
c D	Read third time and defeated by Senate	33
э.в.	1318. Interagency Environmental Justice Working Group; established, definitions,	
	report, sunset date. Amending § 2.2-234; adding § 2.2-236.	
	Patrons: Hashmi, et al. Prefiled, ordered printed, and referred to Committee on General Laws and Technology	50
	Reported with amendments	
	•	
	Rereferred to Committee on Finance and Appropriations	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendments waived	
	Committee amendments rejected	
	Reading of substitute waived	
	Committee substitute agreed to	
	Engrossed	
	Read third time and passed	
~ -	Continued to 2021 Special Session I in House Committee on General Laws	.97
S.B.	1319. Waste Diversion and Recycling Task Force; Department of Environmental	
	Quality to continue Task Force.	
	Patrons: Hashmi, et al.	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural	
	Resources	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived 4	173

S.B. 1	319 (continued) Committee substitute agreed to	17:
	Substitute by Senator Hanger withdrawn	
	Engrossed	
	Constitutional reading dispensed	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natura	al
	Resources	190
1	consultation with a licensed physician with a practice agreement, report. Amending § 54.1-2900; adding § 54.1-2957.04.	
	Patrons: Lucas, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	
	Reported with amendment	
(Constitutional reading dispensed, passed by for day	46
J	Read second time	59
]	Reading of amendment waived	60
(Committee amendment agreed to	60
	Engrossed	
	Read third time and passed	-08
(Continued to 2021 Special Session I in House Committee on Health, Welfare and	
	Institutions	197
S.B. 1	321. Confirmatory adoption ; expands the stepparent adoption provisions, person with a	
1	egitimate interest. Amending § 63.2-1241.	
]	Patrons: Boysko, et al.	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	
]	Reported with amendments	37
]	Rereferred to Committee on Finance and Appropriations	37
	Reported	
(Constitutional reading dispensed, passed by for day	244
]	Read second time	259
]	Reading of amendments waived	259
(Committee amendments agreed to	259
1	Engrossed	260
I	Read third time and passed	275
5	Statement on vote	275
(Continued to 2021 Special Session I in House Committee on Health, Welfare and	
	Institutions	197
	1322. Public schools; seizure management and action plans, biennial training, effective date. Amending § 8.01-225; adding § 22.1-274.6.	
	Patrons: DeSteph, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	53
	Reported with substitute	
	•	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to	
	Engrossed	
	Read third time and passed	
(Continued to 2021 Special Session Lin House Committee on Education	197

S.B.	1323. Worker classification; qualification of independent contractors. Amending	
	§ 40.1-28.7:7.	
	Patron: Dunnavant	
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor	
	Rereferred to Committee on the Judiciary	.74
S.B.	1324. Actions against real estate appraisers or appraisal management companies;	
	statute of limitations on actions for damages or other relief. Adding §§ 54.1-2019.1 and	
	54.1-2024.	
	Patron: Dunnavant	
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	
	Reported with amendments	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendments waived	
	Committee amendments agreed to	
	Engrossed	
	Read third time and passed	
G D	Continued to 2021 Special Session I in House Committee for Courts of Justice	.97
S.B.	1325. Court-ordered custody and visitation arrangements; petition of grandparent for	
	visitation with minor grandchild. Amending § 20-124.2.	
	Patrons: Dunnavant, et al.	<i>-</i> 1
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Committee substitute agreed to.	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
SR	1326. Local cigarette taxes; any locality is authorized to levy taxes upon sale of	
D.D.	cigarettes, regional cigarette tax boards, definitions, report. Amending § 58.1-3830;	
	adding § 58.1-3832.1.	
	Patrons: Hanger, et al.	
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations	54
	Reported with amendment	
	Constitutional reading dispensed, passed by for day	
	Passed by temporarily	
	Read second time	
	Reading of amendment waived	
	Committee amendment rejected	
	Reading of substitute waived	
	Substitute by Senator Hanger agreed to	
	Engrossed	
	Engrossment reconsidered	
	Passed by temporarily	154
	Reading of amendments waived	68
	Amendments by Senator Hanger agreed to	68
	Engrossed	68
	Constitutional reading dispensed	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Finance	97

S.B	. 1327. Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, etc. Amending §§ 8.01-463, 36-139, 55.1-320,
	55.1-321, and 55.1-1303; adding § 15.2-2223.5.
	Patrons: McClellan, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 54
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to.
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on General Laws
c D	. 1328. State-Funded Kinship Guardianship Assistance program; created, clarifies
э.Б	definition of "foster care services." Amending §§ 16.1-228, 16.1-282.1, 63.2-100,
	63.2-905, 63.2-906, and 63.2-1305; adding § 63.2-1306.
	Patron: Mason
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 54
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
c D	Institutions
S.B	. 1329. Summons; promises to appear after issuance. Amending §§ 46.2-936 and 46.2-940.
	Patrons: Mason, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
c D	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B	. 1330. Paid family and medical leave program; Virginia Employment Commission to
	establish, benefits beginning January 1, 2024. Adding §§ 60.2-800 through 60.2-820.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
S.B	. 1331. Absentee voting; accessibility for voters with a visual impairment or print
	disability. Amending § 24.2-704; adding § 24.2-103.2.
	Patrons: Reeves, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Reconsideration of vote on passage

S.B.	1331 (continued) Passed Senate
	Continued to 2021 Special Session I in House Committee on Privileges and Elections
S.B.	1332. Law-enforcement officer; use of deadly force during an arrest or detention.
	Amending § 19.2-83.5.
	Patron: Reeves
	Prefiled, ordered printed, and referred to Committee on the Judiciary
S.B.	1333. Pharmaceutical processors; permits processors to produce and distribute cannabis
	products, dispensing botanical cannabis to a minor, certification.
	Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5
	through 54.1-3442.8.
	Patrons: Lucas, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute rejected
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Passed by temporarily
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S.B.	1334. Broadband capacity; expands existing pilot program, municipal broadband
	authorities. Amending § 56-585.1:9.
	Patrons: Edwards, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1335. Learner's permits; use of personal communication devices. Amending § 46.2-335.
	Patrons: Stuart, et al.
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Substitute by Senator Stuart agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Transportation
S.B.	1336. Restricted permits to operate a motor vehicle; ignition interlock systems.
	Adding § 18.2-271.5.
	Patrons: Stuart, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Rereferred to Committee on Finance and Appropriations

S.B.	. 1336 (continued)	2.52
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendment waived.	
	Amendment by Senator Surovell agreed to	
	Engrossed	
	Constitutional reading dispensed	
	Passed Senate	
c D	Continued to 2021 Special Session I in House Committee for Courts of Justice	49/
Э.Д.	ordinance, authorize a vehicle to be operated on a designated public highway where the	
	posted speed limit is 40 miles per hour or less. Amending § 46.2-916.3.	
	Patron: Stuart	
	Prefiled, ordered printed, and referred to Committee on Transportation	55
S R	. 1338. Telemedicine services; payment for medical assistance for remote patient	
э.р.	monitoring services, effective clause. Amending §§ 32.1-325 and 38.2-3418.16.	
	Patrons: Barker, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	55
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Rereferred to Committee on Finance and Appropriations	
	Reported with amendments	
	Read second time	
	Reading of substitute waived	382
	Committee substitute agreed to	
	Reading of amendments waived	
	Committee amendments agreed to	382
	Engrossed	390
	Constitutional reading dispensed	390
	Passed Senate	391-92
	Continued to 2021 Special Session I in House Committee on Health, Welfare and	
	Institutions	497
S.B.	. 1339. Police and court records; expungement and sealing of records, Expungement Fee	
	Fund created, protection of public record information, penalties, report, effective clause.	
	Amending §§ 9.1-128, 9.1-134, 17.1-502, 19.2-392.1, 19.2-392.2, 19.2-392.3, and	
	19.2-392.4; adding §§ 1-229.1, 17.1-205.1, 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2,	
	19.2-392.2:3, 19.2-392.3:1, 19.2-392.4:1, 19.2-392.5, 19.2-392.6, and 19.2-392.7.	
	Patrons: Surovell, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	e	
	Committee substitute rejected	
	Committee substitute rejected.	
	Reading of substitute waived	
	Substitute by Senator Surovell agreed to	
	Engrossed	
	Read third time and passed	454

S.B.	1339 (continued) Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1340. Medical debt; statute of limitations to collect. Amending § 8.01-246.
	Patrons: Hashmi, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
S.B.	1341. Health insurance; association health plan for real estate salespersons, policy for group accident and sickness insurance. Amending § 38.2-3521.1.
	Patrons: Barker, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Substitute by Senator Barker agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1342. Workers' compensation; establishes a presumption that COVID-19 causing the
	death or disability of firefighters, emergency medical services personnel, law-enforcement
	officers, and correctional officers is an occupational disease. Amending § 65.2-402.1.
	Patron: Vogel
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor 56
	Reported
	Rereferred to Committee on Finance and Appropriations
S.B.	1343. Virginia Freedom of Information Act; proprietary records and trade secrets,
	carbon sequestration agreements. Amending § 2.2-3705.6.
	Patrons: Vogel, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 56
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
c D	Continued to 2021 Special Session I in House Committee on General Laws
S.D.	1344. Overdoses; arrest and prosecution when experiencing or reporting, first offender deferred disposition. Amending § 18.2-251.03.
	Patron: Vogel
	Prefiled, ordered printed, and referred to Committee on the Judiciary
S R	1345. Alcoholic beverage control; license applications, notice requirements.
у.р.	Amending § 4.1-230.
	Patron: Vogel
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 56
S.B.	1346. Alcoholic beverage control; creates a mobile retailer license, trucks or trailers equipped with beverage taps. Amending §§ 4.1-206.3, 4.1-209, 4.1-233, and 4.1-233.1.
	Patron: Stuart
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 57
S.B.	1347. Motorcycles; authorizes the use of any color auxiliary lighting, other than blue.
	Amending § 46.2-1012.
	Patron: Reeves
	Prefiled, ordered printed, and referred to Committee on Transportation

S.B.	1348. Local advisory referenda; exception to the rule that prohibits a referendum from being placed on the ballot unless specifically authorized by statute or by charter, etc. Amending § 24.2-684.
	Patron: Newman
S.B.	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Patron: Newman
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 57
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Communications, Technology and Innovation
S.B.	
	Amending §§ 33.2-214.2 and 33.2-353.
	Patron: Lewis
	Prefiled, ordered printed, and referred to Committee on Transportation
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Transportation
S.B.	1351. Workers' compensation; claims not barred. Adding § 65.2-706.2.
	Patrons: Lewis, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1352. Flood Control and Commonwealth Flood Control Board, Department of; established, report. Amending §§ 2.2-215, 10.1-603.25, 10.1-659, and 10.1-1330; adding §§ 10.1-658.1 through 10.1-658.7.
	Patron: Lewis
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural Resources
S.B.	1353. Income tax, corporate; combined reporting requirements. Amending §§ 58.1-400.1, 58.1-406, 58.1-442, and 58.1-443; adding §§ 58.1-424 through 58.1-427.
	Patron: Marsden
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations 57

S.B.	1354. Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment,
	Enhanced Nutrient Removal Certainty Program established, regulations.
	Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14.
	Patrons: Hanger, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
c D	Resources
S.B.	1355. Rapidan Service Authority; withdrawal of a member.
	Patron: Hanger
	Prefiled, ordered printed, and referred to Committee on Local Government
	Reported with substitute
	Read second time
	Reading of substitute waived
	· ·
	Passed by for day
C D	1356. Hospitals, nursing homes, etc.; visits by clergy, priest, etc., during a declared
э.Б.	public health emergency. Amending §§ 32.1-127, 32.1-162.5, and 63.2-1732.
	Patrons: Kiggans, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S.B.	1357. Through-year growth assessment system; Board of Education shall establish,
	report, effective clause.
	Patron: Dunnavant
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time

S.B.	1357 (continued)	200
	Passed by temporarily	
	Reading of substitute waived	
	Committee substitute rejected.	
	Reading of amendments waived	
	Committee amendments rejected	
	Reading of substitute waived	
	Substitute by Senator Dunnavant agreed to	
	Engrossed	
	Read third time and passed	
	Statements on vote	
~ ~	Continued to 2021 Special Session I in House Committee on Education	497
S.B.	1358. Employment health and safety standards; Safety and Health Codes Board to	
	adopt regulations establishing standards designed to protect employees from heat illness.	
	Adding § 40.1-44.2.	
	Patron: Hashmi	
с Б	Prefiled, ordered printed, and referred to Committee on Commerce and Labor	. 58
S.B.	1359. Higher educational institutions, public; any fees levied for the purpose of	
	collegiate athletics shall be made optional for students. Amending § 23.1-1301.	
	Patron: DeSteph	
~ ~	Prefiled, ordered printed, and referred to Committee on Education and Health	. 58
S.B.	1360. Virginia Human Rights Act; nondiscrimination in employment, sexual harassment	
	and workplace harassment. Amending §§ 2.2-522, 2.2-3902, 2.2-3905, 2.2-3906,	
	2.2-3907, and 2.2-3908; adding § 2.2-3905.1.	
	Patrons: McClellan, et al.	
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported with amendment	
	Rereferred to Committee on the Judiciary	
	Reported with amendments	
	Read first time.	
	Read second time	
C D	Recommitted to Committee on the Judiciary	4/3
5.D.	1361. Law-enforcement civilian oversight bodies; person appointed shall be U.S. citizen, etc. Amending § 9.1-601.	
	Patron: Reeves	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	50
S R	1362. Employers; reporting outbreaks of COVID-19, effective clause.	. 50
ь.р.	Patrons: Lewis, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	50
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	
	Reported with amendments	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to.	
	Reading of amendments waived	
	Committee amendments agreed to	
	Engrossed	
	Read third time and defeated by Senate	
		462

	Passed by temporarily
	Engrossment reconsidered
	Reading of amendment waived
	Amendment by Senator Lewis agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
S R	1363. Local and Regional Jails, State Board of; membership, powers and duties,
O.D.	effective clause. Amending §§ 53.1-2, 53.1-4, and 53.1-5.
	Patrons: Marsden, et al.
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 59
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations 270
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Passed by temporarily
	Read second time
	Reading of substitute waived
	Committee substitute rejected
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee on Public Safety
S.B.	. 1364. Commonwealth Transit Ridership Fund; created. Amending § 33.2-1526.3;
	adding § 33.2-1526.3:1.
	Patron: Cosgrove
	Prefiled, ordered printed, and referred to Committee on Transportation
S.B.	1365. Data Governance and Analytics, Office of; created, establishes the advisory
	Virginia Data Commission, membership, report. Amending § 2.2-203.2:4;
	adding §§ 2.2-2558 through 2.2-2564.
	Patrons: Barker, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology 59
	Reported with amendments
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time 381
	Reading of amendments waived
	Committee amendments rejected
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Communications, Technology and
	· · · · · · · · · · · · · · · · · · ·
	Innovation

S.B.	1366. Aging services; Department for Aging and Rehabilitative Services shall use available resources to provide services to older persons with the greatest economic and social needs, definitions. Amending §§ 51.5-134 and 51.5-135. Patrons: Barker, et al.	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	5 9
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute rejected	
	Reading of substitute waived	
	Substitute by Senator Barker agreed to	
	Engrossed	
	Read third time and passed	
	Statement on vote	
	Continued to 2021 Special Session I in House Committee on Health, Welfare and Institutions	
C D	1367. Line of Duty Act; Virginia licensed health practitioners required to conduct medical	71
у.Б.	reviews, persons issued a comparable license, as determined by Virginia Retirement System, by the District of Columbia or a state that is contiguous to Virginia.	
	Amending §§ 9.1-404 and 9.1-405. Patron: DeSteph	
	Prefiled, ordered printed, and referred to Committee on the Judiciary Rereferred to Committee on Finance and Appropriations	
S.B.	Reported	44 60 75
S.B.	Prefiled, ordered printed, and referred to Committee on the Judiciary	59
	business." Amending §§ 2.2-1604 and 2.2-4310. Patrons: Obenshain, et al.	
		50
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	
	Reported	
	Read second time and engrossed	
	*	
C D	Continued to 2021 Special Session I in House Committee on General Laws	9/
5.В.	1370. Parole statutes; application of statutes for juveniles and persons committed upon felony offenses committed on or after January 1, 1995. Amending § 53.1-165.1.	
	Patrons: Edwards, et al. Profiled ordered minted and referred to Committee on Rehabilitation and Social Services.	۲۰
S.B.	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	ЭU
	Patron: Ruff	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	60

S.B.	1372. Criminal records ; establishes a process for automatic expungement for certain convictions, report. Amending §§ 9.1-101, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-389.3, and 19.2-390; adding §§ 19.2-392.5 through
	19.2-392.14.
	Patrons: Lucas, et al.
~ -	Prefiled, ordered printed, and referred to Committee on the Judiciary
S.B.	1373. Environmental permits; community and environmental justice outreach. Amending § 10.1-1408.1; adding §§ 10.1-1321.2, 10.1-1437.1, 62.1-44.6:1, and 62.1-266.1.
	Patrons: McClellan, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural Resources
S.B.	1374. Carbon Sequestration Task Force; established by Secretary of Natural Resources, jointly with the Secretary of Agriculture and Consumer Services, to convene a task force, report.
	Patrons: Lewis, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural Resources
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural Resources
S.B.	1375. Workers' compensation; establishes a presumption that COVID-19 causing the
	death or disability of firefighters, law-enforcement officers or correctional officers is an occupational disease. Amending § 65.2-402.1.
	Patrons: Saslaw, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time 381
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1376. Absentee voting; processing of absentee ballots before election day, reporting of
	absentee ballot vote totals, central absentee voter precincts. Amending §§ 24.2-101, 24.2-603, 24.2-709.1, 24.2-710, 24.2-711, and 24.2-712; adding § 24.2-667.1.
	Patrons: Saslaw, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections

S.	B. 1377. Campaign finance; special report for large pre-legislative session contributions, contributions in aggregate. Amending § 24.2-947.11.	
	Patron: Suetterlein	
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections	61
S.	B. 1378. Emergency Services and Disaster Law; limits the duration of any rule, etc., issued by the Governor pursuant to his powers to no more than 30 days from the date of issuance. Amending § 44-146.17.	. 01
	Patron: Newman	
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	. 61
S.	B. 1379. Humane Cosmetics Act; prohibits a cosmetics manufacturer from testing cosmetics on animals in the Commonwealth, civil penalties. Adding §§ 59.1-571 through 59.1-574.	
	Patrons: Boysko, et al.	
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural	
	Resources	. 61
	Reported with substitute	237
	Constitutional reading dispensed, passed by for day	
	Read second time	283
	Reading of substitute waived	283
	Committee substitute agreed to	283
	Engrossed	283
	Read third time and passed	300
	Reconsideration of vote on passage	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Labor and Commerce	497
S.	B. 1380. Electric utilities; authorizes electric utilities to partner with school divisions to	
	implement projects designed to encourage the proliferation of school buses that are fueled	
	in whole or in part by electricity. Amending § 58.1-3660; adding § 56-585.1:13.	
	Patron: Lucas Prefiled, ordered printed, and referred to Committee on Commerce and Labor	61
	Reported	
	Rereferred to Committee on Finance and Appropriations	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed.	
	Statement on vote	
	Continued to 2021 Special Session I in House Committee on Labor and Commerce	
S.	B. 1381. Weapons; possessing or transporting within Capitol Square or into a building	
	owned or leased by the Commonwealth, provisions shall not apply to certain	
	law-enforcement officers, etc., penalty. Adding § 18.2-283.2.	
	Patrons: Ebbin, et al.	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on Public Safety	497

S.B.	1382. Firearms; purchase, possession, or transportation following conviction for assault and battery of a family or household member, penalties, report. Amending §§ 18.2-308.09, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:8.
	Patrons: Favola, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Substitute by Senator Favola agreed to
	Engrossed
	Read third time and defeated by Senate
S.B.	1383. Criminal proceedings; consideration of mental condition and intellectual and
	developmental disabilities, permits the admission of evidence. Amending §§ 19.2-120 and
	19.2-299; adding § 19.2-271.6.
	Patrons: Favola, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
S.B.	1384. Virginia Public Procurement Act; local arbitration agreements, definitions,
	procurement procedures, compliance for post-award actions, penalty.
	Adding §§ 2.2-4377.1 through 2.2-4377.5.
	Patron: Surovell
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology $\dots 62$
	Reported
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time
	Amendment No. 1 by Senator Surovell withdrawn
	Reading of substitute waived
	Substitute by Senator Surovell agreed to
	Reading of amendments waived
	Amendments Nos. 2 and 3 by Senator Surovell agreed to
	Amendments Nos. 2 and 3 by Senator Surovell reconsidered
	Passed by for day
	Substitute No. 2 by Senator Surovell withdrawn
	Passed by for day
	Substitute No. 1 by Senator Surovell reconsidered
	Substitute No. 1 by Senator Surovell withdrawn
	Reading of substitute waived
	Substitute No. 3 by Senator Surovell agreed to
	Engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee on General Laws
S.B.	1385. Underground utility facilities; removes the sunset on a pilot program allowing a
	locality that has adopted the urban county executive form of government (Fairfax County)
	to request an electric utility to place underground electric distribution lines, written
	agreement. Amending § 15.2-816.1.
	Patrons: Surovell, et al.
	Prefiled, ordered printed, and referred to Committee on Local Government

C D	1205 ()	
S.B.	. 1385 (continued) Reported with substitute	293
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived .	
	Committee substitute agreed to.	
	Engrossed	
	Read third time and passed	
G P	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns	. 496
S.B.	. 1386. Vehicle registration; prohibits Commissioner of DMV from refusing to issue or renew solely because of unpaid tolls, etc., owed to the private operator of a toll facility. Adding § 46.2-819.11.	
	Patron: Spruill	
	Prefiled, ordered printed, and referred to Committee on Transportation	6'
S.B.	. 1387. Students; eligibility for in-state tuition and state financial assistance program. Amending § 23.1-506; adding § 23.1-505.1.	02
	Patrons: Boysko, et al.	
	Prefiled, ordered printed, and referred to Committee on Education and Health	
	Reported with substitute	
	Rereferred to Committee on Finance and Appropriations	. 120
	Reported	. 237
	Constitutional reading dispensed, passed by for day	. 261
	Read second time	. 284
	Reading of substitute waived	. 284
	Committee substitute agreed to	. 284
	Engrossed	. 284
	Read third time and passed	
C D	. 1388. Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed	. 49
у.р.	wine for off-premises consumption. Amending §§ 4.1-119, 4.1-204, 4.1-206.1, 4.1-206.3, 4.1-207, 4.1-210, 4.1-212.1, and 4.1-221.	
	Patron: Reeves	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	62
S.B.	. 1389. Real property; required disclosures for buyer to exercise due diligence, flood risk report, effective date. Amending § 55.1-703; adding § 55.1-708.2.	
	Patrons: Lewis, et al.	
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology	62
	Reported with substitute	. 252
	Rereferred to Committee on Finance and Appropriations	. 253
	Reported with amendment	. 353
	Constitutional reading dispensed, passed by for day	, 371
	Read second time	. 381
	Reading of substitute waived	. 387
	Committee substitute agreed to	. 387
	Reading of amendment waived.	
	Committee amendment agreed to	
	Engrossed	
	Constitutional reading dispensed	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on General Laws	

S.B.	1390. Cats; trap, neuter, and return programs, civil penalty. Amending §§ 3.2-6500,
. .	3.2-6524, 3.2-6546, and 3.2-6551; adding § 3.2-6579.1.
	Patron: Lewis
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Reading of amendment waived
	Amendment by Senator Stuart rejected
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
C D	1391. Pretrial data collection; Virginia Criminal Sentencing Commission to collect and
э.Б.	disseminate on an annual basis, report. Amending § 2.2-3802; adding § 19.2-134.1.
	Patrons: Lucas, et al.
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
~ -	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1392. Consumer Data Protection Act; establishes a framework for controlling and
	processing personal data, effective date. Adding §§ 59.1-571 through 59.1-581.
	Patrons: Marsden, et al.
	Prefiled, ordered printed, and referred to Committee on General Laws and Technology
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Read third time and passed
	Statement on vote
	Continued to 2021 Special Session I in House Committee on Communications, Technology and
	Innovation

S.B.	1393. Trees; replacement and conservation during development, projects located in a
	Chesapeake Bay Preservation Area to address recurrent flooding, effective date, report,
	effective clause. Amending §§ 15.2-961 and 15.2-961.1.
	Patrons: Marsden, et al.
	Prefiled, ordered printed, and referred to Committee on Local Government
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Substitute by Senator Marsden withdrawn
	Reading of amendments waived
	Amendments by Senator Marsden agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S R	1394. Income tax, state; an exclusion for Paycheck Protection Plan loan forgiveness,
	deductibility. Amending § 58.1-321.
	Patrons: Petersen and Pillion
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
R	1395. Discrimination; prohibited in voting and elections administration, required process
) .	for enacting certain covered practices, civil causes of action, penalties, repeals provision
	relating to minority language accessibility, etc. Amending §§ 24.2-105, 24.2-306,
	24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1; adding §§ 24.2-104.1, 24.2-125
	through 24.2-131, and 24.2-1005.2; repealing § 24.2-124.
	Patrons: McClellan, et al.
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute rejected
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Privileges and Elections
z D	1396. Onsite Sewage Indemnification Fund; authorizes the State Board of Health to use
э.Б.	Fund to provide grants and loans to property owners with income at or below 200 percent
	of the federal poverty guidelines to repair failing onsite sewage systems or install onsite
	sewage systems on properties that lack adequate sewage disposal. Amending §§ 32.1-164
	and 32.1-164.1:01; adding §§ 62.1-223.1, 62.1-223.2, and 62.1-223.3.
	and 32.1-104.1:01; adding §§ 62.1-223.1, 62.1-223.2, and 62.1-223.3. Patron: Hashmi
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported 353

S.B.	1396 (continued)	71
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	1
~ -	Resources	96
	1397. Geriatric or terminally ill inmates; conditional release, notify by certified or	
	verifiable electronic means to attorney for the Commonwealth. Amending § 53.1-136.	
	Patron: Norment	
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services	
	Reported with amendments	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of amendments waived	
	Committee amendments agreed to	
	Engrossed	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	97
	1398. Retail sales and transient occupancy taxes; definitions, taxes on transient room	
	rentals shall be computed on the basis of the total charges or the total price paid for the use	
	or possession of the room, etc. Amending §§ 58.1-602, 58.1-603, 58.1-3819, 58.1-3819.1,	
	58.1-3823, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, 58.1-3826, 58.1-3842, and	
	58.1-3843; adding §§ 2.2-2320.2, 58.1-612.2, and 58.1-3818.8.	
	Patrons: Norment, et al.	
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute rejected	
	Reading of substitute waived	
	Substitute by Senator Norment agreed to	
	Engrossed	
	Constitutional reading dispensed	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Finance	91
	1399. Tourism Development Authority; changes name of Authority to the Heart of	
	Appalachia Tourism Authority. Amending §§ 15.2-5500, 15.2-5501, 15.2-5505, 15.2-5506, and 45.1-246.	
	Patron: Pillion	
	Prefiled, ordered printed, and referred to Committee on Local Government	
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time and passed	
	*	90
	1400. Tazewell County; authorizes a quitclaim and release of interest and conveyance of	
	an easement by Board of Wildlife Resources.	
	Patron: Pillion	
	Prefiled, ordered printed, and referred to Committee on the Judiciary	
	Reported 1 Constitutional reading dispensed, passed by for day 2	24 12
	Constitutional reading dispensed, passed by for day	

S.B.	1400 (continued)Read third time and passed253-54
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 496
c D	•
5.Б.	1401. Standards of Learning; reduces total number and type of required assessments to minimum requirements, report. Amending § 22.1-253.13:3.
	Patrons: Pillion, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Education
S.B.	1402. Trout fishing in stocked waters; equalizes for residents and nonresidents
	requirements to fish in designated waters. Amending § 29.1-311.
	Patron: Pillion
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1403. Retail Sales and Use Tax; exemption for personal protective equipment.
	Adding § 58.1-609.14.
	Patrons: Pillion, et al.
	Prefiled, ordered printed, and referred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Passed by for day
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Finance
S.B.	1404. Stormwater Local Assistance Fund; grants awarded for projects related to
	Chesapeake Bay total maximum daily load (TMDL) requirements.
	Amending § 62.1-44.15:29.1.
	Patron: Lewis
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources

S.B.	. 1405. Get Skilled, Get a Job, Give Back (G3) Fund and Program; established.
	Adding § 23.1-2911.2.
	Patrons: Saslaw, et al.
	Prefiled, ordered printed, and referred to Committee on Education and Health
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Education
S.B.	. 1406. Marijuana; unlawful possession in certain cases, using or consuming while in a
	motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119,
	2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010,
	3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01,
	4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105,
	4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119,
	4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1,
	4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216,
	4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300,
	4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2,
	4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353,
	4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820,
	16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9,
	17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02,
	18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255,
	18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2,
	18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016,
	18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1,
	19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389,
	19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609,
	23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2,
	54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107,
	65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1,
	3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503,
	6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7;
	repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3.
	Patrons: Ebbin and Lucas, et al.
	Prefiled, ordered printed, and referred to Committee on Rehabilitation and Social Services 65
	Reported with substitute
	Rereferred to Committee on the Judiciary
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute rejected
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendments waived

S.B. 1406 (continued)	414
•	415
-	
	House Committee on General Laws
0 0 , 0	rulation, penalties. Amending §§ 2.2-419, 2.2-2905,
	3705.7, 2.2-3711, 2.2-4002, 8.01-195.11, 9.1-101, -334.3, 18.2-340.22, 19.2-389, 37.2-314.2, 58.1-3,
	-534.5, 16.2-540.22, 19.2-589, 57.2-514.2, 58.1-5, 4000, 58.1-4002, 58.1-4003, 58.1-4006, 58.1-4007,
	58.1-4012, 58.1-4020.1, 58.1-4025, 58.1-4027,
	11-16.3 and 58.1-4048 through 58.1-4075.
Patron: Bell	11 1010 und cont 10 to une ugu cont 10 to
	to Committee on General Laws and Technology 65
	non; repeals sunset provision. Repealing § 30-170.
Patrons: Barker, et al.	ton, repeate sumber previous responsing 3 5 5 17 5.
,	to Committee on Rules
_	
-	ed by for day
-	
-	House Committee on Rules
	; retired law-enforcement officers employed as
fingerprint examiners. Amending § 51.	
Patron: Cosgrove	
•	to Committee on Finance and Appropriations 66
-	ary spouse; prohibits discrimination in public
	ousing, landlords may not charge a higher security
deposit. Amending §§ 2.2-2901.1, 2.2	2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904,
	5, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2,
	-96.4, 36-96.6, 55.1-1226, and 55.1-1310.
Patrons: Bell, et al.	
_	to Committee on General Laws and Technology 66
•	
	ed by for day
Reading of substitute waived	
Committee substitute agreed to	
Engrossed	
Read third time and passed	
Continued to 2021 Special Session I in	House Committee on General Laws 497
S.B. 1411. Peanuts; extends sunset date	of excise tax on all peanuts grown in Virginia.
Amending § 3.2-1905.	
Patron: Lucas	
Prefiled, ordered printed, and referred	to Committee on Agriculture, Conservation and Natural
	nd Appropriations
	ed by for day
C	

S.B.	1411 (continued) Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S R	1412. Pet shops, dealers, and dog breeders; employees convicted of animal abuse,
э.р.	penalty. Amending §§ 3.2-6511.1 and 3.2-6511.2.
	Patron: Stanley
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute 237
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed 301
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
c D	
э.в.	1413. Phase I or Phase II electric utilities; provision of broadband capacity, State Corporation Commission shall condition any approval of a petition on the requirement
	that construction shall commence within 18 months of such approval.
	Amending § 56-585.1:9.
	Patrons: Boysko, et al.
	Prefiled, ordered printed, and referred to Committee on Commerce and Labor
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	· ·
	Engrossed
	•
	Statement on vote
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
C D	1414. Henrietta Lacks Commission; extends sunset provision. Amending § 2.2-2543.
э.Б.	Patrons: Stanley, et al.
	Prefiled, ordered printed, and referred to Committee on Rules
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Rules
C D	1415. Protective orders; violations of preliminary child protective order, violation
э.р.	involves an act that endangers the child's normal development, changes punishment, etc.
	Amending §§ 16.1-253 and 16.1-253.2.
	Patron: Stanley
	Prefiled, ordered printed, and referred to Committee on the Judiciary
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to

S.B.	1415 (continued) Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1416. Remote patient monitoring; amends State Plan for Medical Assistance Services to
	require payment of assistance for rural and underserved populations, report.
	Amending § 32.1-325.
	Patron: Stanley
	Prefiled, ordered printed, and referred to Committee on Education and Health 67
S.B.	1417. Healthcare Regulatory Sandbox Program; established, report, sunset date. Amending § 3.2-6591; adding § 3.2-6593.1.
	Patrons: Stanley, et al.
	Prefiled, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported with substitute 237
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
	Resources
S.B.	1418. Commonwealth's Development Opportunity Fund; grants from Fund, waiver or
	reduction of capital investment and local match requirements. Amending § 2.2-115.
	Patron: McPike
	Presented, ordered printed, and referred to Committee on General Laws and Technology 70
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	C
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on General Laws
S.B.	1419. Project labor agreements; public body to determine if agreement advances public
	interest. Amending § 2.2-4321.2.
	Patron: Marsden
	Presented, ordered printed, and referred to Committee on General Laws and Technology 70
	Rereferred to Committee on Commerce and Labor
S.B.	1420. Electric utilities; nonjurisdictional customers, third party power purchase
	agreements. Amending Chapters 1187, 1188, 1189, 1193, 1194, and 1239, 2020 Acts.
	Patrons: Edwards, et al.
	Presented, ordered printed, and referred to Committee on Commerce and Labor
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Constitutional reading dispensed
	Constitutional reading dispensed

S.B.	1420 (continued)
	Passed Senate
~ ~	Continued to 2021 Special Session I in House Committee on Labor and Commerce 497
S.B.	1421. Brain injury ; clarifies definition. Amending § 37.2-403.
	Patrons: Edwards, et al.
	Presented, ordered printed, and referred to Committee on Education and Health
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
c D	Institutions
5.В.	1422. Voter registration; list maintenance, lists of decedents transmitted by State
	Registrar of Vital Records to Department of Elections on a weekly basis.
	Amending §§ 24.2-408 and 24.2-427.
	Patrons: Kiggans, et al. Presented, ordered printed, and referred to Committee on Privileges and Elections
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Privileges and Elections
C D	1423. Data centers; sales and use tax exemption, identifying a "distressed locality."
э.Б.	Amending § 58.1-609.3.
	Patrons: McPike and Pillion
	Presented, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute rejected
	Reading of substitute waived
	Substitute by Senator McPike agreed to
	Engrossed
	Read third time and passed
	Reconsideration of vote on passage
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Finance
S R	1424. Funeral service establishments; defines "manager of record."
э.р.	Amending §§ 54.1-2800 and 54.1-2810.
	Patron: Cosgrove
	Presented, ordered printed, and referred to Committee on General Laws and Technology 71
SR	1425. Data centers; expands sales and use tax exemption. Amending § 58.1-609.3.
э.Б.	Patrons: Ruff, et al.
	Presented, ordered printed, and referred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Constitutional reading dispensed 391

S.B.	1425 (continued) Passed Senate 393
	Continued to 2021 Special Session I in House Committee on Labor and Commerce
C D	1426. Orders of restitution; docketed on behalf of victim, assignment of judgment,
э.Б.	enforcement. Amending §§ 19.2-305.1, 19.2-305.2, 19.2-349, and 19.2-354.
	Patrons: Stanley, et al.
	Presented, ordered printed, and referred to Committee on the Judiciary
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
σъ	Continued to 2021 Special Session I in House Committee for Courts of Justice
э.Б.	1427. Early Psychosis Intervention and Coordinated Specialty Care Program
	Advisory Board; established. Adding § 37.2-313.2.
	Patron: Morrissey
	Presented, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and Institutions
S.B.	1428. Alcoholic beverage control; operation of government stores, sale of low alcohol
	beverage coolers, effective date. Amending § 4.1-119.
	Patrons: Locke, et al.
	Presented, ordered printed, and referred to Committee on Rehabilitation and Social Services 76
	Reported with amendments
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on General Laws
e D	
3.D.	1429. Southwestern Virginia Mental Health Institute; Governor to lease a portion of property to Smyth County for a term of three years, responsibility of County, corrects tax
	map references. Amending Chapter 678, 2019 Acts.
	Patrons: Pillion, et al. Presented and referred to Committee on Consul Legys and Technology.
	Presented, ordered printed, and referred to Committee on General Laws and Technology 8
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time

concerning fourth-year medical students, etc. Adding §§ 23.1-110, 32.1-23.3, and

Presented, ordered printed, and referred to Committee on Education and Health 82
Reported with substitute 251
Rereferred to Committee on Finance and Appropriations 253
Reported with amendment 353
Constitutional reading dispensed, passed by for day 370, 371
Read second time 382
Reading of substitute waived 389
Committee substitute agreed to 389
Reading of amendment waived 389
Committee amendment agreed to 389
Engrossed 390

54.1-2400.03. Patron: Hanger

S.B.	1436 (continued) Constitutional reading dispensed	91
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Health, Welfare and	
C D	Institutions	9,
S.B.	1437. Summons; eliminates the requirement that a promise to appear be completed after	
	issuance for a misdemeanor offense or an administrative violation. Amending §§ 2.2-311, 19.2-74, 28.2-901, 29.1-210, 46.2-936, 46.2-940, 54.1-306, 54.1-2506, and 54.1-4407.	
	Patrons: Hanger, et al. Presented, ordered printed, and referred to Committee on the Judiciary	0-
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read second time and engrossed	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
c D	1438. Combined transient occupancy and food and beverage tax; for purposes of taxes	91
э.Б.	that are currently authorized for Rappahannock and Madison Counties, the rate limit for	
	such tax shall be the same as if the two taxes were imposed separately.	
	Amending § 58.1-3842.	
	Patron: Hanger	
	Presented, ordered printed, and referred to Committee on Finance and Appropriations	82
	Reported	
	Constitutional reading dispensed, passed by for day	
	Read second time and engrossed	
	Read third time	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee on Finance	
S.B.	1439. Students; guidelines on excused student absences, each school board shall permit	-
	one school day-long excused absence per school year for any middle or high school	
	student who is absent from school to engage in a civic event. Amending § 22.1-254.	
	Patrons: McClellan, et al.	
	Presented, ordered printed, and referred to Committee on Education and Health	83
	Reported with amendment	
	Constitutional reading dispensed, passed by for day	86
	Passed by for day	11
	Read second time	42
	Reading of amendment waived	42
	Committee amendment rejected	42
	Reading of substitute waived	42
	Substitute by Senator McClellan agreed to	42
	Engrossed	
	Read third time and passed	58
	Continued to 2021 Special Session I in House Committee on Education	97
S.B.	1440. Law-enforcement officer or correctional officer; civil action for unlawful acts of	
	force or failure to intervene. Adding § 8.01-42.6.	
	Patron: Surovell	
	Presented, ordered printed, and referred to Committee on the Judiciary	91
S.B.	1441. Commissioner of accounts; prohibits appointment of General Assembly members.	
	Amending §§ 30-105, 64.2-1200, and 64.2-1202.	
	Patron: Lucas	
	Presented, ordered printed, and referred to Committee on Rules	91

INDEX

S.B.	1442. Public defender office; establishes an office for the County of Chesterfield, effective clause. Amending § 19.2-163.04.		
	Patrons: Morrissey, et al. Presented, ordered printed, and referred to Committee on the Judiciary		
	•		
	Reported		
	Rereferred to Committee on Finance and Appropriations		
	Reported with amendment		
	Constitutional reading dispensed, passed by for day		
	Read second time		
	Reading of amendment waived		
	Committee amendment agreed to		
	Engrossed		
	Read third time and passed		
	Continued to 2021 Special Session I in House Committee for Courts of Justice		
S.B.	1443. Mandatory minimum sentences; elimination, modification of sentence to		
	mandatory minimum term of confinement for felony offenses, report.		
	Amending §§ 3.2-4212, 4.1-302, 16.1-253.2, 18.2-36.1, 18.2-36.2, 18.2-46.3:3, 18.2-51.1,		
	18.2-53.1, 18.2-57, 18.2-60.4, 18.2-61, 18.2-67.1, 18.2-67.2, 18.2-67.5:2, 18.2-67.5:3,		
	18.2-154, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5,		
	18.2-255, 18.2-255.2, 18.2-270, 18.2-308.1, 18.2-308.2, 18.2-308.2:2, 18.2-308.4,		
	18.2-374.1, 18.2-374.1:1, 18.2-374.3, 19.2-297.1, 46.2-341.28, 46.2-357, 46.2-391,		
	46.2-865.1, and 53.1-203.		
	Patrons: Edwards, et al.		
	Presented, ordered printed, and referred to Committee on the Judiciary		
	Reported with substitute		
	Rereferred to Committee on Finance and Appropriations		
	Reported		
	Constitutional reading dispensed, passed by for day		
	Read second time		
	Reading of substitute waived		
	Committee substitute agreed to		
	Reading of amendment waived		
	Amendment by Senator Edwards agreed to		
	Engrossed		
	Read third time and passed		
	Continued to 2021 Special Session I in House Committee for Courts of Justice		
S.B.	1444. Campaign finance reports; special report of pre-legislative session contributions.		
	Amending § 24.2-947.11.		
	Patrons: Saslaw, et al.		
	Presented, ordered printed, and referred to Committee on Privileges and Elections		
	Reported with substitute		
	Constitutional reading dispensed, passed by for day		
	Read second time		
	Reading of substitute waived		
	Committee substitute agreed to.		
	Engrossed		
	Constitutional reading dispensed		
	Passed Senate		
	Continued to 2021 Special Session Lin House Committee on Privileges and Elections 497		
	A ADMINISTRAÇÃO ZAZZE ODECTAL DESSIONEE IN FROUSE VADIMINISEE ON FITVIESES AND EXECUONS 497		

S.B.	1445. COVID-19 ; facilitates vaccine administration, localities with fire departments, emergency medical services departments, and volunteer rescue squads, etc., may establish and staff clinics for administering vaccines.
	Patrons: Dunnavant, et al.
	Presented, ordered printed, and referred to Committee on Education and Health
	Reported with substitute
	Constitutional reading dispensed
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendments waived
	Amendments by Senator Mason agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Passed House with substitute with amendment
	House substitute with amendment agreed to
	Signed by President
	Approved by Governor-Chapter 2 (effective 2/15/21)
S.B.	1446. Medicine and other healing arts; practice, provision of litigation assistance.
J.D.	Adding § 8.01-401.4.
	Patron: Surovell
	Presented, ordered printed, and referred to Committee on Education and Health
	Rereferred to Committee on the Judiciary
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Passed by for day
	Read second time
	Recommitted to Committee on the Judiciary
S.B.	1447. Buckingham County; fees for disposal of solid waste. Amending § 15.2-2159.
	Patron: Peake
	Presented, ordered printed, and referred to Committee on Local Government
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Passed by for day
	Read third time
	Engrossment reconsidered
	Reading of amendment waived
	Amendment by Senator Peake agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 496
S.B.	1448. Income tax, state; subtraction for military retirement income and survivor benefits. Amending § 58.1-322.02.
	Patron: Chase
	Presented, ordered printed, and referred to Committee on Finance and Appropriations
	1 reserved, ordered printed, and referred to committee our rinaries and Appropriations

S.B.	1449. COVID-19 immunization; prohibition on requirement, employment discrimination prohibited. Amending §§ 2.2-2901.1, 2.2-3004, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 32.1-127, and 65.2-402.1; adding §§ 32.1-15.2, 37.2-205,
	40.1-27.4, and 63.2-221.1.
	Patron: Chase
S.B.	1 J 1
	Amending §§ 2.2-2203.3, 2.2-2901.1, 2.2-3004, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 23.1-2415, and 33.2-290.
	Patron: Chase Presented, ordered printed, and referred to Committee on General Laws and Technology 98
S.R.	1451. COVID-19 vaccine; equal access to state and local public property by persons who
·	choose not to receive. Adding § 2.2-1147.4.
	Patron: Chase
	Presented, ordered printed, and referred to Committee on General Laws and Technology 99
S.B.	1452. License plates, special; issuance for supporters of the Southwest Regional
	Recreation Authority bearing legend SPEARHEAD TRAILS.
	Patron: Pillion
	Presented, ordered printed, and referred to Committee on Transportation
S.B.	1453. Mines and Mining and Virginia Energy Plan; revision of Titles 45.1 and 67.
	Adding §§ 10.1-1332, 10.1-1333, 33.2-120, 33.2-221.1, 45.2-100 through 45.2-402,
	45.2-500 through 45.2-1051, 45.2-1100 through 45.2-1505, 45.2-1600 through 45.2-1649,
	45.2-1700 through 45.2-2119, 55.1-1820.1, 55.1-1951.1, 55.1-2133.1, and 56-614 through 56-624; repealing §§ 11-34.1 through 11-34.4, 45.1-161.1 through 45.1-399, 62.1-195.1,
	62.1-195.3, and 67-100 through 67-1700.
	Patron: Edwards
	Presented, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Engrossment reconsidered
	Reading of amendments waived
	Amendments by Senator Edwards agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee on Agriculture, Chesapeake and Natural
C D	Resources
э.Б.	Patron: Ruff
	Presented, ordered printed, and referred to Committee on the Judiciary
S.B.	1455. Absentee ballots; witness requirement, printed name and residence address.
	Amending §§ 24.2-702.1, 24.2-706, and 24.2-707.
	Patron: Ruff
	Presented, ordered printed, and referred to Committee on Privileges and Elections 109
S.B.	1456. Juveniles; eligibility for commitment to the Department of Juvenile Justice,
	eligibility for predispositional confinement in a secure facility. Amending §§ 16.1-248.1,
	16.1-249, 16.1-278.7, and 16.1-278.8.
	Patron: Marsden
	Presented, ordered printed, and referred to Committee on the Judiciary
	Reported

S.B.	1456 (continued) Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S B	1457. Historic sites; urban county executive form of government (Fairfax County),
S.B.	provisions in its preservation ordinance, etc. Amending § 15.2-2306.
	Patrons: Surovell, et al.
	Presented, ordered printed, and referred to Committee on Local Government
	Reported with amendments
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Read third time and passed
~ ~	Continued to 2021 Special Session I in House Committee on Counties, Cities and Towns 496
S.B.	1458. Identity Management Standards Advisory Council; transfers management of
	Council. Amending §§ 2.2-436 and 2.2-437.
	Patron: Boysko
	Presented, ordered printed, and referred to Committee on General Laws and Technology 110
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Communications, Technology and
~ ~	Innovation
S.B.	1459. Elections; voter identification and absentee voting, process for applying for,
	casting, and counting absentee ballots. Amending §§ 24.2-101, 24.2-404, 24.2-411.3,
	24.2-643, 24.2-653, 24.2-653.01, 24.2-701, 24.2-701.1, 24.2-702.1, 24.2-706, 24.2-707,
	24.2-709.1, and 24.2-712.
	Patron: Norment
a D	Presented, ordered printed, and referred to Committee on Privileges and Elections
S.B.	1460. Deer hunting ; authorizes the use of a .223 caliber centerfire rifle. Amending
	§ 29.1-519.
	Patron: Lewis
	Presented, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
c D	Resources
5.В.	1461. Bribery in correctional facilities; penalty. Adding § 18.2-474.2.
	Patron: Lewis
	Presented, ordered printed, and referred to Committee on the Judiciary
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Constitutional reading dispensed
	Passed Senate
a -	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1462. Virginia Digital Equity Pilot Program and Fund; established, report, sunset date,
	effective clause. Adding § 63.2-806.
	Patrons: Mason, et al.
	Presented, ordered printed, and referred to Committee on Rehabilitation and Social Services 121

S.B.	1462 (continued) Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Communications, Technology and
	Innovation
S.B.	1463. Covenants regarding solar power; planned unit developments.
	Amending § 67-701.
	Patron: Cosgrove
	Presented, ordered printed, and referred to Committee on General Laws and Technology 121
S.B.	1464. Drug Control Act; adds certain chemicals to Schedules I, II, IV, and V.
	Amending § 54.1-3446.
	Patron: Newman
	Presented, ordered printed, and referred to Committee on Education and Health
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time and engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Health, Welfare and
	Institutions
C D	1465. Illegal gambling; skill games, definitions, enforcement by localities and Attorney
э.р.	General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1.
	Patrons: Reeves, et al.
	Presented, ordered printed, and referred to Committee on the Judiciary
	Reported with amendments
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendments waived
	Committee amendments agreed to
	Engrossed
	Constitutional reading dispensed
	Passed Senate
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1466. Concealed handgun permit; local control of firearms, certain ordinances shall not
	apply to a person who has a valid permit issued by another state. Amending § 15.2-915.
	Patrons: Obenshain, et al.
	Presented, ordered printed, and referred to Committee on the Judiciary
S.B.	1467. Menhaden fishing; prohibits catching with purse nets in the territorial waters of the
J.17.	Commonwealth lying west or within three miles of the Chesapeake Bay Bridge-Tunnel.
	Amending § 28.2-409.
	Patrons: DeSteph, et al.
	Presented, ordered printed, and referred to Committee on Agriculture, Conservation and Natural
	Resources

S.B.	1468. Victims of crime; definitions, certifications for victims of qualifying criminal activity, disclosure of exculpatory information to a defendant in a criminal case. Adding §§ 9.1-1500, 9.1-1501, and 9.1-1502.
	Patrons: Surovell, et al.
	Presented, ordered printed, and referred to Committee on the Judiciary
	Reported with substitute
	Constitutional reading dispensed, passed by for day $\dots \dots 285, 286$
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee for Courts of Justice
S.B.	1469. Opioid Abatement Authority; established, Fund created, report, membership.
	Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40.
	Patrons: Barker, et al.
	Presented, ordered printed, and referred to Committee on General Laws and Technology 138
	Reported
	Rereferred to Committee on Finance and Appropriations
	Reported with substitute
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived
	Committee substitute agreed to
	Reading of amendment waived
	Amendment by Senator Barker agreed to
	Engrossed
	Read third time and passed
	Statement on vote
C D	1470. Vehicle registration; expands eligibility for special communication needs indicator.
у.р.	Amending § 46.2-600.1.
	Patrons: Barker, et al.
	Presented, ordered printed, and referred to Committee on Transportation
	Reported with amendment
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of amendment waived
	Committee amendment agreed to
	Engrossed
	Read third time and passed
	Continued to 2021 Special Session I in House Committee on Transportation
S.B.	1471. Alcoholic beverage control; local special events license, taxes and fees.
	Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, and 4.1-233.1.
	Patrons: Dunnavant, et al.
	Presented, ordered printed, and referred to Committee on Rehabilitation and Social Services 138
	Reported with substitute
	Rereferred to Committee on Finance and Appropriations
	Reported
	Constitutional reading dispensed, passed by for day
	Read second time
	Reading of substitute waived

S.B.	1471 (continued) Committee substitute agreed to	1 (
	Engrossed 4	
	Read third time and passed	
	Continued to 2021 Special Session I in House Committee on General Laws	
C D	1472. Individuals with intellectual and developmental disabilities; Department of	71
э.р.	Medical Assistance Services shall establish a work group to study and develop	
	recommendations for permanent use of virtual supports and increasing access to virtual	
	supports and services.	
	Patrons: Suetterlein, et al.	
	Presented, ordered printed, and referred to Committee on Rehabilitation and Social Services 13	2 (
	•	
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to	
	Engrossed	
	Constitutional reading dispensed	
	Passed Senate	37
	Continued to 2021 Special Session I in House Committee on Health, Welfare and	
	Institutions	97
S.B.	1473. Health Insurance Reform Commission; mandated health insurance benefit or	
	provider legislation to be referred to Commission. Amending § 30-343.	
	Patron: Saslaw	
	Presented, ordered printed, and referred to Committee on Rules	
	Reported	7(
	Constitutional reading dispensed, passed by for day	12
	Read second time and engrossed	36
	Constitutional reading dispensed	37
	Passed Senate	37
	Continued to 2021 Special Session I in House Committee on Rules	97
S.B.	1474. Nonrepairable and rebuilt vehicles; extends sunset provision relating to certain	
	requirements. Amending second enactment of Chapter 342, 2017 Acts and third	
	enactment of Chapter 362, 2017 Acts.	
	Patron: Newman	
	Presented, ordered printed, and referred to Committee on Transportation	39
	Reported with substitute	7(
	Rereferred to Committee on Finance and Appropriations	11
S.B.	1475. Search warrants; date and time of issuance, law-enforcement officer to be	
	recognizable and identifiable, exceptions. Amending § 19.2-56.	
	Patron: Stuart	
	Presented, ordered printed, and referred to Committee on the Judiciary	39
	Reported with substitute	
	Constitutional reading dispensed, passed by for day	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to.	
	Engrossed 36	
	Constitutional reading dispensed	
	Passed Senate	
	Continued to 2021 Special Session I in House Committee for Courts of Justice	
	Communica to 2021 Special Session 1 in House Committee for Courts of Justice	11

	2. Constitutional amendment; personal reproductive liberty (first reference). Adding Section 18 in Article I.
	Patron: Saslaw
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections
S.J.R	2. 3. Constitutional amendment; repeals provision dealing with marriage that was approved by referendum at the November 2006 election (first reference). Repealing Section 15-A of Article I.
	Patrons: Ebbin, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections 6
	4. 6. Constitutional amendment; Governor's term of office (first reference). Amending Section 1 of Article V.
	Patron: Ebbin
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections 6
	7. Constitutional amendment; repeals provision dealing with marriage that was approved by referendum at the November 2006 election (first reference). Repealing Section 15-A of Article I.
	Patrons: Edwards, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections 6
	8. 8. Constitutional amendment; qualifications of voters and the right to vote (first reference). Amending Section 1 of Article II.
	Patrons: Locke, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections 6
	1. 11. Constitutional amendment; qualifications of Governor and Lieutenant Governor, residency requirement (first reference). Amending Section 3 of Article V.
	Patrons: Chase, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections 6
	1. 14. Constitutional amendment; qualifications of voters, restoration of civil rights by general law (first reference). Amending Section 1 of Article II.
	Patron: Deeds
S.J.F	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections
	Patrons: Chase, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections 6
	1. 19. Constitutional amendment; prohibits the establishment of electoral districts that intentionally or unduly favor or disfavor any political party, etc. (first reference). Amending Section 6 of Article II and Section 14 of Article IV.
	Patrons: Chase, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections 6
	2. 25. Virginia Tech Carilion School of Medicine; Virginia Polytechnic Institute and State University to study the requirements to be designated as a teaching hospital.
	Patron: Stanley
	Continued from 2020 Regular Session in Senate Committee on Education and Health 6
	3. 39. Justice system; Virginia State Crime Commission to study effect of abolishing jury sentencing.
	Patrons: Edwards, et al.
	Continued from 2020 Regular Session in House Committee on Rules

S.J.R. 58. Constitutional amendment; personal property tax exemption for one motor vehicle owned and used primarily by or for a disabled veteran, "motor vehicle" shall include only automobiles and pickup trucks, exception (second reference), Chapters 822 and 823, 2019	
Acts (first reference). Amending Section 6 of Article X.	
Patrons: Morrissey, et al.	
Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations S.J.R. 59. Constitutional amendment; automatic restoration of civil rights for person	6
convicted of a felony upon completion of his sentence (first reference). Amending Section 1 of Article II.	
Patrons: Morrissey, et al.	
Continued from 2020 Regular Session in Senate Committee on Privileges and Elections	6
S.J.R. 63. Constitutional amendment; uniform schedule of elections for members of the General Assembly and statewide offices (first reference). Amending Section 4 of Article	
II, Sections 2 and 3 of Article IV, and Section 1 of Article V.	
Patron: Ebbin	,
Continued from 2020 Regular Session in Senate Committee on Privileges and Elections S.J.R. 75. Constitutional amendment; public schools in the Commonwealth, equal	(
educational opportunities (first reference). Amending Sections 1 and 2 of Article VIII. Patron: Stanley	
Continued from 2020 Regular Session in Senate Committee on Privileges and Elections	6
S.J.R. 270. Constitutional amendment; fundamental right to marry, removes same-sex	
marriage prohibition (first reference). Amending Section 15-A of Article I.	
Patrons: Ebbin, et al.	
Prefiled, ordered printed, and referred to Committee on Privileges and Elections	67
Reported with substitute	
Reading waived, passed by for day	
Read second time	
Reading of substitute waived	. 480
Tie vote, Chair votes Yea	. 480
Committee substitute agreed to	. 480
Engrossed	. 480
Reading waived	. 483
Agreed to by Senate	
Continued to 2021 Special Session I in House Committee on Privileges and Elections	. 497
S.J.R. 271. Constitutional amendment; Governor's term of office (first reference).	
Amending Section 1 of Article V.	
Patrons: Ebbin, et al.	
Prefiled, ordered printed, and referred to Committee on Privileges and Elections	
Reported	
Reading waived, passed by for day	
Read second time and engrossed	
Reading waived	
Rejected by Senate	. 485
S.J.R. 272. Constitutional amendment; qualifications of voters, fundamental right to vote,	
persons not entitled to vote (first reference). Amending Section 1 of Article II.	
Patrons: Locke, et al.	
Prefiled, ordered printed, and referred to Committee on Privileges and Elections	
Reported	
Reading waived, passed by for day	
Reading of substitute waived	
Substitute by Senator Surovell agreed to	. 480

S.J.R. 272 (continued)	
•	
e e e e e e e e e e e e e e e e e e e	
•	ommittee on Privileges and Elections 49°
S.J.R. 273. Edwards, Bruce Winston; recording sorn	
Patrons: DeSteph, et al.	tow upon death.
• '	69
	· · · · · · · · · · · · · · · · · · ·
S.J.R. 274. Constitutional amendment; limits in Amending Section 9 of Article IV.	
Patron: Reeves	
Prefiled, ordered printed, and referred to Commit S.J.R. 275. Constitutional amendment; equitable schools in the Commonwealth (first reference) VIII. Patrons: Stanley, et al.	
Prefiled, ordered printed, and referred to Commit	ttee on Privileges and Elections
•	
_	
· ·	
Agreed to by Senate	
S.J.R. 276. Brain Aneurysm Awareness Month; d	esignating as September 2021 and each
succeeding year thereafter.	
Patrons: Favola, et al.	
-	ttee on Rules
•	ommittee on Rules
S.J.R. 277. Marymount University; commemorating Patrons: Favola, Howell and Ebbin, et al.	g its /0th anniversary.
	69
Prefiled, presented, and laid on Clerk's Desk	
S.J.R. 278. White, Bruce D.; commending.	
Patrons: Surovell, et al.	
	6!
S.J.R. 279. Smith, Steve Selwyn; commending.	
Patrons: Surovell, et al.	
	90

S.J.R. 280. Daniel, Kimberly J.; commending.	
Patrons: Surovell, et al.	
Prefiled, presented, and laid on Clerk's Desk.	69
Engrossed and agreed to by Senate.	
Agreed to by House	96
S.J.R. 281. Saxe, Janine M.; commending.	
Patrons: Surovell, et al.	
Prefiled, presented, and laid on Clerk's Desk	69
Engrossed and agreed to by Senate.	78
Agreed to by House	96
S.J.R. 282. Crater Planning District Commission; commemorating its 50th anniversary.	
Patrons: Ruff, et al.	
Prefiled, presented, and laid on Clerk's Desk	69
Engrossed and agreed to by Senate.	78
Agreed to by House	96
S.J.R. 283. Fleck, Marisa L.; recording sorrow upon death.	
Patrons: Surovell, et al.	
Prefiled, presented, and laid on Clerk's Desk	69
Engrossed and agreed to by Senate.	77
Agreed to by House	96
S.J.R. 284. Chafin, Augustus Benton, Jr.; recording sorrow upon death.	
Patrons: Pillion, et al.	
Prefiled, presented, and laid on Clerk's Desk	69
Engrossed and agreed to by Senate.	9
Agreed to by House	96
S.J.R. 285. Barrier Crimes and Criminal History Records Checks, Joint Subcommit	tee
Studying; continued, appropriation.	
Patrons: Edwards, et al.	
Prefiled, ordered printed, and referred to Committee on Rules	68
Reported	137
Reading waived, passed by for day	165
Read second time and engrossed	213
Read third time and agreed to by Senate	244
Continued to 2021 Special Session I in House Committee on Rules	497
S.J.R. 286. Emergency Management Professionals Week; designating as third week	in
March 2021 and each succeeding year thereafter.	
Patron: McPike	
Prefiled, ordered printed, and referred to Committee on Rules	
Reported	137
Reading waived, passed by for day	
Read second time and engrossed	212, 213
Read third time and agreed to by Senate	244
Continued to 2021 Special Session I in House Committee on Rules	497
S.J.R. 287. Cohen, Bernard S.; recording sorrow upon death.	
Patrons: Saslaw, et al.	
Prefiled, presented, and laid on Clerk's Desk.	69
Engrossed and agreed to by Senate.	78
Agreed to by House	96

S.J.	R. 288. Johns, Barbara Rose; Joint Committee of Congress on the Library requested to	
	fill the Commonwealth's vacant spot in the National Statuary Hall Collection at the United	
	States Capitol with a statue to commemorate.	
	Patrons: Lucas, et al.	
	Prefiled, ordered printed, and referred to Committee on Rules	
	Reported	
	Reading waived, passed by for day	
	Read second time	
	Reading of substitute waived	
	Substitute by Senator Lucas agreed to	
	Engrossed	
	Read third time and agreed to by Senate	
	Continued to 2021 Special Session I in House Committee on Rules	197
S.J.	R. 289. Constitutional amendment; authority to grant perpetual easements to units of	
	government (first reference). Amending Section 9 of Article VII.	
	Patron: Mason	
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections	68
	Reported 3	379
	Reading waived, passed by for day	132
	Read second time and engrossed	180
	Reading waived.	
	Agreed to by Senate	
	Continued to 2021 Special Session I in House Committee on Privileges and Elections	197
S.J.	R. 290. Constitutional amendment; legislative compensation, Legislative Compensation	
	Commission established (first reference). Amending Section 5 of Article IV; adding 5-A	
	in Article IV.	
	Patrons: McPike, et al.	
	Prefiled, ordered printed, and referred to Committee on Privileges and Elections	68
	Reported	
	Reading waived, passed by for day	
	Read second time	
	Reading of amendments waived	
	Amendments by Senator McPike agreed to	
	Engrossed	
	Reading waived.	
	Tie vote	
	Rejected by Senate	
СТ	R. 291. Washington, Booker T.; General Assembly to support establishment of a	102
	commemorative commission to honor with a statue in the State Capitol.	
	Patron: Suetterlein	
	Prefiled, ordered printed, and referred to Committee on Rules	65
СТ	R. 292. Women's Suffrage Month; designating as August 2020 and each succeeding year	U
5. J.	thereafter.	
	Patrons: Vogel, et al.	
		۷.
	Prefiled, ordered printed, and referred to Committee on Rules	
	Reported	
	Reading waived, passed by for day	
	Read second time and engrossed	
	Read third time and agreed to by Senate	
	Continued to 2021 Special Session I in House Committee on Rules	197

S.J.R. 293. Assisted living and auxiliary grants; Joint Commission on Health Care to study
available data regarding, etc.
Patron: Spruill
Prefiled, ordered printed, and referred to Committee on Rules
Reported with amendments
Reading waived, passed by for day
Read second time
Reading of amendments waived
Committee amendments agreed to
Engrossed
Read third time and agreed to by Senate
Continued to 2021 Special Session I in House Committee on Rules
S.J.R. 294. Standards of Quality; Joint Legislative Audit and Review Commission to study
the true cost of education in the Commonwealth and provide an accurate assessment of the
costs to implement, meetings shall be completed for the first year by November 30, 2022.
Patrons: Lewis, et al.
Presented, ordered printed, and referred to Committee on Rules
Reported with amendments
Reading waived, passed by for day
Read second time
Reading of amendments waived
Committee amendments agreed to
Engrossed
Read third time and agreed to by Senate
Continued to 2021 Special Session I in House Committee on Rules
S.J.R. 295. Spotsylvania County; commemorating its 300th anniversary.
Patrons: Reeves, et al.
Presented and laid on Clerk's Desk
Engrossed and agreed to by Senate.
Agreed to by House
S.J.R. 296. Barnard, Barry M.; commending.
Patrons: Reeves, et al.
Presented and laid on Clerk's Desk
Engrossed and agreed to by Senate.
Agreed to by House
S.J.R. 297. Constitutional amendment; authority of General Assembly to exempt from
taxation property owned by certain veterans' and service members' services organizations
(first reference). Amending Section 6-A of Article X. Patron: Bell
Presented, ordered printed, and referred to Committee on Finance and Appropriations
S.J.R. 298. Governor; confirming appointments. Patron: Deeds
T WILL D TO US
Presented, ordered printed, and referred to Committee on Privileges and Elections
Reported
Reading waived, passed by for day
Read second time and engrossed
Read third time and agreed to by Senate
Agreed to by House
S.J.R. 299. Governor; confirming appointments.
Patron: Deeds
Presented, ordered printed, and referred to Committee on Privileges and Elections
Reported with amendment

S.J.R. 299 (continued) Reading waived, passed by for day	120 120
Read second time	
Reading of amendment waived.	
Committee amendment agreed to	
Engrossed	
Read third time	
Agreed to by Senate	
Statement on vote	
Agreed to by House	441
S.J.R. 300. Governor; confirming appointments.	
Patron: Deeds	
Presented, ordered printed, and referred to Committee on Privileges and Elections	
Reported	
Reading waived, passed by for day	
Read second time and engrossed	. 148, 149
Read third time	163
Agreed to by Senate	164
Statement on vote	164
Agreed to by House	441
S.J.R. 301. Governor; confirming appointments.	
Patron: Deeds	
Presented, ordered printed, and referred to Committee on Privileges and Elections	76
Reported	
Reading waived, passed by for day	
Read second time and engrossed	
Read third time	
Agreed to by Senate	
Statement on vote	
Agreed to by House	
S.J.R. 302. Governor; confirming appointments.	
Patron: Deeds	
Presented, ordered printed, and referred to Committee on Privileges and Elections	76
Reported	
Reading waived, passed by for day	
Read second time and engrossed	
Read third time and agreed to by Senate	
Agreed to by House	44
S.J.R. 303. Governor; confirming appointments.	
Patron: Deeds	
Presented, ordered printed, and referred to Committee on Privileges and Elections	
Reported	
Reading waived, passed by for day	. 128, 129
Read second time and engrossed	
Read third time and agreed to by Senate	. 163, 164
Agreed to by House	441
S.J.R. 304. BaCote, Mamye E.; recording sorrow upon death.	
Patrons: Locke, et al.	
Presented and laid on Clerk's Desk	83
Engrossed and agreed to by Senate	
Agreed to by House	

S.J.R. 305. Roberts, Billy Joe; recording sorrow upon death.	
Patrons: Locke, et al.	
Presented and laid on Clerk's Desk	83
Passed by for day	132
Engrossed and agreed to by Senate	263
Agreed to by House	320
S.J.R. 306. Moneta Elementary School; commemorating its 50th anniversary.	
Patron: Newman	
Presented and laid on Clerk's Desk	83
Engrossed and agreed to by Senate.	133
Agreed to by House	171
S.J.R. 307. Pete's Pizza; commemorating its 45th anniversary.	
Patron: Reeves	
Presented and laid on Clerk's Desk	83
Engrossed and agreed to by Senate	133
Agreed to by House	171
S.J.R. 308. COVID-19; Joint Legislative Audit and Review Commission to study the impact	
on Virginia's public schools, students, and school employees, meetings shall be completed	
by November 30, 2022.	
Patrons: Lucas, et al.	
Presented, ordered printed, and referred to Committee on Rules	83
Reported with amendments	137
Reading waived, passed by for day	165
Read second time	212
Reading of amendments waived	213
Committee amendments agreed to	213
Engrossed	213
Read third time and agreed to by Senate	
Continued to 2021 Special Session I in House Committee on Rules	
S.J.R. 309. Constitutional amendment; General Assembly, limits members of the Senate to	
three full terms and members of the House of Delegates to six full terms (first reference).	
Amending Section 4 of Article IV.	
Patron: Chase	
Presented, ordered printed, and referred to Committee on Rules	83
S.J.R. 310. Constitutional amendment; length of regular sessions convened in odd-numbered	
years (first reference). Amending Section 6 of Article IV.	
Patrons: Saslaw, et al.	
Presented, ordered printed, and referred to Committee on Privileges and Elections	92
Reported with amendment	
Reading waived, passed by for day	1, 432
Passed by temporarily	481
Read second time	482
Reading of amendment waived	482
Committee amendment rejected	483
Reading of amendment waived	483
Amendment by Senator Deeds agreed to	483
Engrossed	483
Reading waived	483
Agreed to by Senate	190-91
Continued to 2021 Special Session I in House Committee on Privileges and Elections	497

S.J.R. 311. Mann, Robert Edward; recording sorrow upon death.	
Patron: Locke	
Presented and laid on Clerk's Desk	92
Engrossed and agreed to by Senate	31
Agreed to by House	71
S.J.R. 312. Singleton, O.R., Jr.; commending.	
Patrons: Hashmi, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	71
S.J.R. 313. Virginia's Crossroads; commending.	
Patrons: Lucas, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	71
S.J.R. 314. American Promise and Take Back Our Republic, Virginia Chapter of;	
commending.	
Patrons: Deeds, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	71
S.J.R. 315. DeLaney, Theodore Carter, Jr.; recording sorrow upon death.	
Patrons: Deeds, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	72
S.J.R. 316. Martin, Walter Lowrie, III, and Edith Luke Martin; recording sorrow upon	
death.	
Patron: Deeds	
Presented and laid on Clerk's Desk	
Passed by for day	
Reading of substitute waived	
Substitute by Senator Deeds agreed to	
Engrossed and agreed to by Senate	
Agreed to by House	99
Patron: Deeds	
Presented and laid on Clerk's Desk	00
D J l, £ J	32
Passed by for day	
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death.	
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al.	20
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk	
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate.	31
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate. 1 Agreed to by House 1	31
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate. 1. Agreed to by House 1. S.J.R. 319. Horwatt, Michael Stephen; recording sorrow upon death.	31
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate. 1. Agreed to by House 1. S.J.R. 319. Horwatt, Michael Stephen; recording sorrow upon death. Patrons: Howell, et al.	31 72
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate. 1. Agreed to by House 1. S.J.R. 319. Horwatt, Michael Stephen; recording sorrow upon death. Patrons: Howell, et al. Presented and laid on Clerk's Desk 1	31 72 10
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate. 1 Agreed to by House 1 S.J.R. 319. Horwatt, Michael Stephen; recording sorrow upon death. Patrons: Howell, et al. Presented and laid on Clerk's Desk 1 Engrossed and agreed to by Senate. 2	31 72 10 63
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate. 1 Agreed to by House 1 S.J.R. 319. Horwatt, Michael Stephen; recording sorrow upon death. Patrons: Howell, et al. Presented and laid on Clerk's Desk 1 Engrossed and agreed to by Senate. 2 Agreed to by House 3	31 72 10 63
S.J.R. 318. Goodwin, Stephen Teel; recording sorrow upon death. Patrons: Reeves, et al. Presented and laid on Clerk's Desk Engrossed and agreed to by Senate. 1 Agreed to by House 1 S.J.R. 319. Horwatt, Michael Stephen; recording sorrow upon death. Patrons: Howell, et al. Presented and laid on Clerk's Desk 1 Engrossed and agreed to by Senate. 2	31 72 10 63

S.J.R. 328. Brown, Roger Hamilton; recording sorrow upon death.	
Patrons: Howell, et al.	
Presented and laid on Clerk's Desk	155
Engrossed and agreed to by Senate.	264
Agreed to by House	320
S.J.R. 329. Stowers, Joseph Richard; recording sorrow upon death.	
Patrons: Howell, et al.	
Presented and laid on Clerk's Desk	155
Engrossed and agreed to by Senate.	264
Agreed to by House	320
S.J.R. 330. Mannion, Robert Emmett, Sr.; recording sorrow upon death.	
Patron: Petersen	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate.	
Agreed to by House	320
S.J.R. 331. Pillow, Richard D.; commending.	
Patron: Newman	
Presented and laid on Clerk's Desk	
Passed by for day	
Rereferred to Committee on Rules	430
S.J.R. 332. Christie, Mark C.; commending.	
Patrons: McDougle, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate.	
Agreed to by House	499
S.J.R. 333. Cameron, Frederick; recording sorrow upon death.	
Patrons: DeSteph, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate.	
Agreed to by House	500
S.J.R. 334. Lee, Wayne R.; commending.	
Patrons: Reeves, et al. Presented and laid on Clerk's Desk	270
Engrossed and agreed to by Senate.	
Agreed to by House	500
Patrons: Mason, et al.	
Presented and laid on Clerk's Desk	202
Engrossed and agreed to by Senate.	
Agreed to by House	
S.J.R. 336. Shatz, Frank; commending.	500
Patrons: Mason, et al.	
Presented and laid on Clerk's Desk	292
Engrossed and agreed to by Senate.	
Agreed to by House	
S.J.R. 337. Moyler, Mary Esterine Hundley; recording sorrow upon death.	500
Patrons: Mason, et al.	
Presented and laid on Clerk's Desk	292
Engrossed and agreed to by Senate.	
Agreed to by House	

S.J.R. 338. Gardner, Jane Carey; recording sorrow upon death.	
Patrons: Favola, et al.	
Presented and laid on Clerk's Desk):
Engrossed and agreed to by Senate	
Agreed to by House	
S.J.R. 339. Chappell, Edward A., Jr.; recording sorrow upon death.	
Patrons: Mason, et al.	
Presented and laid on Clerk's Desk)3
Engrossed and agreed to by Senate	23
Agreed to by House)(
S.J.R. 340. Roessler, Edwin C., Jr.; commending.	
Patrons: Surovell, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	20
Agreed to by House)(
S.J.R. 341. Hershner, Carl; commending.	
Patrons: Mason, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate. 42	
Agreed to by House)(
S.J.R. 342. Butler, Wendell Harding; recording sorrow upon death.	
Patrons: Edwards, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate. 42	
Agreed to by House)(
S.J.R. 343. Nuttycombe, Charles Wallace, Sr.; recording sorrow upon death.	
Patrons: Mason, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House)(
S.J.R. 344. Anderson, Raymond O.; commending.	
Patron: Mason Presented and laid on Clerk's Desk	١,
Engrossed and agreed to by Senate. 42 Agreed to by House	
•	J(
S.J.R. 345. Davidson, Sigmund Edward; recording sorrow upon death. Patrons: Edwards, et al.	
Presented and laid on Clerk's Desk	12
Engrossed and agreed to by Senate	
Agreed to by House	
S.J.R. 346. Wells, John T.; commending.	,
Patrons: Norment, et al.	
Presented and laid on Clerk's Desk):
Engrossed and agreed to by Senate	
Agreed to by House	
S.J.R. 347. Godwin, Angeline; commending.	,
Patrons: Stanley, et al.	
Presented and laid on Clerk's Desk	22
Engrossed and agreed to by Senate	
Agreed to by House	

S.J.R. 348. Woods, Michael Wayne; recording sorrow upon death.	
Patrons: Stanley, et al.	
Presented and laid on Clerk's Desk	322
Engrossed and agreed to by Senate.	423
Agreed to by House	500
S.J.R. 349. Edwards, William Henry, Sr.; recording sorrow upon death.	
Patrons: Stuart, et al.	
Presented and laid on Clerk's Desk	322
Engrossed and agreed to by Senate.	423
Agreed to by House	500
S.J.R. 350. Pinkard, Robert, Jr.; commending.	
Patrons: Stuart, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	500
S.J.R. 351. Lamb, Sally G.; recording sorrow upon death.	
Patrons: Stuart, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	500
S.J.R. 352. Davis, John B.; recording sorrow upon death.	
Patrons: Hanger, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	500
S.J.R. 353. Mantz, William; recording sorrow upon death.	
Patrons: Obenshain, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate.	
Agreed to by House	500
S.J.R. 354. Stoltzfus, Karl David, Sr.; recording sorrow upon death.	
Patrons: Obenshain, et al. Presented and laid on Clerk's Desk	222
Engrossed and agreed to by Senate.	
Agreed to by House	500
Patrons: Obenshain, et al.	
Presented and laid on Clerk's Desk	222
Engrossed and agreed to by Senate.	
Agreed to by House	
S.J.R. 356. Barb, Lowell Robert; recording sorrow upon death.	500
Patrons: Obenshain, et al.	
Presented and laid on Clerk's Desk	323
Engrossed and agreed to by Senate.	
Agreed to by House	
S.J.R. 357. Patrick Henry School of Science and Arts; commemorating its 10th anniversary.	500
Patron: Morrissey	
Presented and laid on Clerk's Desk	323
Engrossed and agreed to by Senate.	
Agreed to by House	

S.J.R. 358. Janaske, Ryan; commending.	
Patrons: Bell, et al.	
Presented and laid on Clerk's Desk	323
Engrossed and agreed to by Senate	429
Agreed to by House	500
S.J.R. 359. The Lady Chamberlains; commending.	
Patrons: Bell, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	429
Agreed to by House	501
S.J.R. 360. Brown, Jean Smith; recording sorrow upon death.	
Patrons: Bell, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	423
Agreed to by House	501
S.J.R. 361. Warrior Retreat at Bull Run; commending.	
Patrons: Bell, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	501
S.J.R. 362. Mahajan, Akshath, and Maneesh Vallurupalli; commending.	
Patrons: Bell, et al.	
Presented and laid on Clerk's Desk	323
Engrossed and agreed to by Senate	429
Agreed to by House	501
S.J.R. 363. Cover, Steven R.; commending.	
Patrons: DeSteph, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	429
Agreed to by House	501
S.J.R. 364. Johnson, Stephen; recording sorrow upon death.	
Patrons: DeSteph, et al.	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	501
S.J.R. 365. LoudounGo; commending.	
Patron: Bell	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate	
Agreed to by House	501
S.J.R. 366. Lazzara, Kelly; commending.	
Patron: Kiggans	
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate.	
Agreed to by House	501
S.J.R. 367. Virginia Beach Pandemic Relief Partnership; commending.	
Patron: Kiggans	<u>.</u>
Presented and laid on Clerk's Desk	
Engrossed and agreed to by Senate.	
Agreed to by House	501

S.J	R. 368. Stevens, John Chatburn; recording sorrow upon death.
	Patron: Kiggans
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
S.J	R. 369. Abernathey, C. Opal B.; recording sorrow upon death.
	Patron: Kiggans
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
S.J	R. 370. Senior Services of Southeastern Virginia; commending.
	Patrons: Kiggans, et al.
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
S.J	R. 371. Rolband, Michael S.; commending.
	Patron: Petersen
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
S.J	R. 372. Greece, Nation of; commending.
	Patron: Dunnavant
	Presented, ordered printed, and referred to Committee on Rules
S.J	R. 373. Hero Homes, Inc.; commending.
	Patron: Bell
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
S.J	R. 374. Panorama Latino TV Show; commemorating its 20th anniversary.
	Patron: Bell
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
S.J	R. 375. Walker, Elaine Dolores; recording sorrow upon death.
	Patrons: Bell, et al.
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
	R. 376. Loudoun Shops Black; commending.
	Patron: Bell
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
	Agreed to by House
S.J	R. 377. Sicknick, Brian David; recording sorrow upon death.
	Patrons: Barker, et al.
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
~ .	Agreed to by House
S.J	R. 378. Calhoun, Robert Lathan; recording sorrow upon death.
	Patrons: Ebbin, et al.
	Presented and laid on Clerk's Desk

S.J.R. 389. Nnamdi, Kojo; commending.	
Patrons: Ebbin, et al.	
Presented and laid on Clerk's Desk .	
Engrossed and agreed to by Senate	
S.J.R. 390. Baker, Jennifer K.; commend	ling.
Patron: Boysko	
Presented and laid on Clerk's Desk .	
Engrossed and agreed to by Senate	
S.J.R. 391. Zidel, Lester; recording sorro	w upon death.
Patrons: Boysko, et al.	
-	
S.J.R. 392. Dozier, Darnell; commending	j .
Patrons: DeSteph, et al.	
	325
•	
S.J.R. 393. Bloxom, Robert S., Sr.; recor	ding sorrow upon death.
Patrons: Lewis, et al.	
	325
• •	42
	502
S.J.R. 394. Savage-Wells, Carla Yvette; Patron: Lewis	recording sorrow upon death.
	325
	nmemorating the 150th anniversary of the enactment.
Patrons: McClellan, et al.	illicinorating the 130th anniversary of the chaethicht.
· · · · · · · · · · · · · · · · · · ·	
	in House Committee on Rules
S.J.R. 396. Carter, Rebecca; commending	
Patron: Peake	ъ.
S.J.R. 397. Palmore, Pamela Sprouse; re	
Patron: Peake	
Unanimous consent to introduce	
S.J.R. 398. Perdue, Clyde H., Jr.; comm	
Patrons: Stanley, et al.	
Presented and laid on Clerk's Desk.	324

S.R.	91 (continued) Appeal from Ruling of the Chair, Ruling of the Chair not sustained	16
	Agreed to by Senate	
c D	92. Foster, Luther Hilton; recording sorrow upon death.	Ю
5. K.	Patron: Ruff	
	Presented and laid on Clerk's Desk	۱۵
c D	Engrossed and agreed to by Senate.	1
5.K.	93. Auditor of Public Accounts; nomination for election.	
	Patron: Deeds	^
	Presented, ordered printed, and referred to Committee on Privileges and Elections	
	Rules suspended	
	Committee discharged	
	Readings waived	
	Taken up for immediate consideration	
	Engrossed	
	Agreed to by Senate	
~ ~	Election by Senate	.9
S.R.	94. Boyd, Madison, Jr.; recording sorrow upon death.	
	Patron: McDougle	
	Presented and laid on Clerk's Desk	
	Engrossed and agreed to by Senate	,4
S.R.	95. Washington, Booker T.; Senate of Virginia to support the establishment of a	
	commemorative commission to honor with a statue in the Old Senate Chamber.	
	Patron: Suetterlein	
	Presented, ordered printed, and referred to Committee on Rules	9
	Reported with amendment	
	Read first time	
	Read second time	
	Reading of amendment waived	
	Committee amendment agreed to	6
	Engrossed	6
	Passed by for day	7
	Read third time	1
	Agreed to by Senate	1
S.R.	96. Via-Gossman, Elizabeth Stamoulis; commending.	
	Patron: McPike	
	Presented and laid on Clerk's Desk	
	Engrossed and agreed to by Senate	4
S.R.	97. Walker, Gloria Mae Benton; recording sorrow upon death.	
	Patron: Lucas	
	Presented and laid on Clerk's Desk	9
	Engrossed and agreed to by Senate	54
S.R.	98. Holley, Mary; recording sorrow upon death.	
	Patron: Lucas	
	Presented and laid on Clerk's Desk	9
	Engrossed and agreed to by Senate	54
S.R.	99. Judges; nominations for election to circuit court.	
	Patron: Edwards	
	Presented, ordered printed, and referred to Committee on the Judiciary)4
	Rules suspended	
	Committee discharged	
	Readings waived	
	Taken up for immediate consideration	
	*	

S.R.	104. Campbell, Leslie D., Jr.; recording sorrow upon death.
	Patron: McDougle
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R.	105. Virginia Piedmont Heritage Area Association; commending.
	Patron: Vogel
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R.	106. Thorpe, Avicia Beatrice Hooper; recording sorrow upon death.
	Patron: Stanley
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R.	107. Stone, Elizabeth Bateman; recording sorrow upon death.
	Patron: Stanley
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R.	108. Galentine, Bryan Wayne; recording sorrow upon death.
	Patrons: Stanley, et al.
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R.	109. Telehealth care providers in Virginia; commending.
	Patron: Stanley
	Presented and laid on Clerk's Desk
~ -	Engrossed and agreed to by Senate
S.R.	110. Francis, Thomas Henry; recording sorrow upon death.
	Patrons: Hashmi, et al.
	Presented and laid on Clerk's Desk
~ -	Engrossed and agreed to by Senate
S.R.	111. Mitchell, Pauline Allen; recording sorrow upon death.
	Patron: Hashmi
	Presented and laid on Clerk's Desk
	Reading of substitute waived
	Substitute by Senator Hashmi agreed to
~ ~	Engrossed and agreed to by Senate
S.R.	112. Woodson, Willie; recording sorrow upon death.
	Patron: McClellan
	Presented and laid on Clerk's Desk
c D	Engrossed and agreed to by Senate
5.K.	113. Robert Russa Moton Museum; commemorating its 20th anniversary and 70th
	anniversary of R. R. Moton High School student strike. Patron: McClellan
	Presented and laid on Clerk's Desk
c D	Engrossed and agreed to by Senate
5.K.	Patron: McClellan
	Presented and laid on Clerk's Desk
Ç D	Engrossed and agreed to by Senate. 429
S.K.	115. Wark, James Perry; recording sorrow upon death. Patron: McClellan
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate

S.R	. 116. Hale, Rhea K.; recording sorrow upon death.
	Patron: McClellan
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R	. 117. Richmond Heritage Federal Credit Union; commemorating its 85th anniversary.
	Patron: McClellan
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R	. 118. Richmond Raceway; commemorating its 75th anniversary.
	Patron: McClellan
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R	. 119. Barber, Doug; commending.
	Patrons: Stanley, et al.
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R	. 120. Smith, James; commending.
	Patron: McDougle
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
S.R	. 121. Cain, Barbara L.; commending.
	Patron: McDougle
	Presented and laid on Clerk's Desk
	Engrossed and agreed to by Senate
H.B	7. Virginia Fair Housing Law; unlawful discriminatory housing practices, prohibits any
	locality, its employees, or its appointed commissions from discriminating in the
	application of local land use ordinances or guidelines, etc., on the basis of sexual
	orientation, gender identity, etc. Amending §§ 36-96.3 and 36-96.17.
	Patrons: Bourne, et al.
	Continued from 2020 Regular Session in Senate Committee on General Laws and Technology 6
H.B	. 157. Drivers of vehicles approaching stationary vehicles displaying certain warning
	lights; duties, penalty. Amending § 46.2-861.1.
	Patron: Fowler
	Continued from 2020 Regular Session in Senate Committee on Transportation
H.B	. 177. Presidential electors; enters Virginia into an interstate compact known as the
	Agreement Among the States to Elect the President by National Popular Vote.
	Adding §§ 24.2-209.1 and 24.2-209.2.
	Patrons: Levine, et al.
	Continued from 2020 Regular Session in Senate Committee on Privileges and Elections
H.B	251. Prostitution-related crimes; minors, increases penalties. Amending §§ 9.1-902,
	17.1-805, 18.2-46.1, 18.2-49, 18.2-347, 18.2-368, 18.2-513, and 19.2-215.1.
	Patrons: Watts, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
H.B	281. Prisoners; eliminates Department of Corrections prisoner co-payment program for
	nonemergency health care services. Amending § 53.1-32.
	Patrons: Hope, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
H.B	3. 304. Guardianship and conservatorship petitions; identifying characteristics of the
	respondent, petition contents. Amending §§ 64.2-2002 and 64.2-2014.
	Patron: Hope
	Continued from 2020 Regular Session in Senate Committee on the Judiciary

H.B	. 332. Reading diagnostic tests; Department of Education shall develop and implement a
	geographically representative two-year pilot program to administer, report.
	Patrons: Hope, et al.
	Continued from 2020 Regular Session in Senate Committee on Education and Health
H.B	. 333. Virginia Minimum Wage Act; persons whose employment is covered by the
	federal Fair Labor Standards Act of 1938 (FLSA), etc. Amending § 40.1-28.9.
	Patrons: Krizek, et al.
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor
H.B	. 416. Wage or salary history; inquiries of history of prospective employees prohibited,
	civil penalty, provisions shall not apply to an employer with fewer than 25 employees.
	Adding § 40.1-28.7:7.
	Patrons: Cole, J.G., et al.
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor
H.B	. 547. Virginia Energy and Economy Transition Council; established, membership,
	report, sunset provision. Adding § 67-1700.
	Patrons: Delaney, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6
H.B	. 553. Workplace harassment; policies for legislative branch, effective date for certain
	provision. Amending §§ 30-129.4, 30-129.5, and 30-129.6; adding § 30-129.5:1.
	Patrons: Watts, et al.
	Continued from 2020 Regular Session in Senate Committee on Rules
H.B	. 610. Limitations period; previously time-barred actions, sexual abuse, two-year time
	period to file. Amending § 8.01-243.
	Patron: Miyares
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
H.B	. 634. School Divisions of Innovation; local assessments. Amending § 22.1-253.13:3;
	adding § 22.1-212.33.
	Patrons: LaRock, et al.
	Continued from 2020 Regular Session in Senate Committee on Education and Health
H.B	. 672. State, regional, and local planning; establishes a policy of the Commonwealth to
	prevent and to minimize actions that contribute to the detrimental effects of climate
	change, designation of areas for implementation of reasonable measures to prevent and
	minimize the impact of climate change. Amending §§ 15.2-2200, 15.2-2223, 15.2-2283,
	15.2-2284, and 15.2-4209; adding § 2.2-5515.
	Patrons: Willett, et al.
	Continued from 2020 Regular Session in Senate Committee on Agriculture, Conservation and
	Natural Resources
H.B	. 673. Cruelty to children; increases penalty to a Class 4 felony, etc.
	Amending §§ 8.01-226.5:2, 17.1-805, 19.2-392.02, 40.1-103, and 63.2-1727.
	Patron: Mullin
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
H.B	. 707. Trees; Town of Vienna, by ordinance, allowed to require that a subdivision or
	development provide for the preservation or replacement of trees on the development site.
	Amending § 15.2-961.1.
	Patron: Keam
	Continued from 2020 Regular Session in Senate Committee on Local Government
H.B	. 729. Transit funding; raises the existing regional transportation fee, grantor's tax for
	localities in the Northern Virginia Transportation Authority, regional congestion relief fee,
	etc. Amending §§ 33.2-2509, 58.1-802.3, 58.1-811, 58.1-816, 58.1-1743, and 58.1-1744;
	adding § 58.1-802.4.
	Patrons: Watts, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6

H.B.	734. Income tax, state; rolling conformity with the Internal Revenue Code, nonconformance with certain amendments. Amending § 58.1-301.
	Patrons: Watts, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations \dots 6
	800. Employment; employer shall furnish each employee a written disclosure of information regarding terms of employment. Adding § 40.1-29.1.
	Patrons: Delaney, et al.
	Continued from 2020 Regular Session in Senate Committee on Commerce and Labor 6
H.B.	845. Washington Metropolitan Area Transit Authority; certain employees of Authority added to definition of law-enforcement officers, law-enforcement authority of certain agents. Amending §§ 9.1-101 and 9.1-400; adding § 33.2-3100.2.
	Patrons: Krizek, et al.
	Continued from 2020 Regular Session in Senate Committee on Transportation
H.B.	869. Public defender or any deputies or employees; proportionally supplementing compensation. Amending § 19.2-163.01:1.
	Patrons: Bourne, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary 6
H.B.	961. Assault firearms, certain firearm magazines, etc.; prohibiting sale, transport, etc.,
	unlawful for any person to possess any large-capacity firearms magazine, penalties. Amending §§ 16.1-278.9, 18.2-287.4, 18.2-308.2:01, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.7, and 19.2-386.28; adding §§ 18.2-308.9 through 18.2-308.13.
	Patrons: Levine, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
H.B.	1010. Criminal fiscal impact statements; bills resulting in a net increase of period of imprisonment or commitment. Amending §§ 30-19.1:4 and 30-28.18.
	Patron: Herring
	Continued from 2020 Regular Session in Senate Committee on Rules
H.B.	1063. Suicide; abolishes the common-law crime. Adding § 18.2-16.1.
	Patrons: Kory, et al.
	Continued from 2020 Regular Session in Senate Committee on the Judiciary
	1134. Small Business and Supplier Diversity, Department of; redefines "small business" for purposes of programs and the Virginia Public Procurement Act to mean a business that together with its affiliates has both 250 or fewer employees and average annual gross receipts of \$10 million or less averaged over the previous three years. Amending §§ 2.2-1604, 2.2-1617, and 2.2-4310.
	Patrons: Lopez, et al.
	Continued from 2020 Regular Session in Senate Committee on General Laws and Technology 6 $$
H.B.	1187. Alcoholic beverage control; distiller licenses, Internet orders and shipments.
	Amending § 4.1-119.
	Patrons: Wampler, et al.
	Continued from 2020 Regular Session in Senate Committee on Rehabilitation and Social
	Services
	1194. Dairy Producer Margin Coverage Premium Assistance Program; established, eligible dairy producer shall apply to the Department of Conservation and Recreation by February 1 to participate. Adding §§ 3.2-3304, 3.2-3305, and 3.2-3306.
	Patrons: Lopez, et al.
	Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations \dots 6

H.B. 1288. Firearms; purchase, possession, or transportation following conviction for assault	
and battery of a family or household member, permit to restore rights, penalties.	
Amending §§ 18.2-308.09, 18.2-308.2, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and	
19.2-386.28; adding § 18.2-308.1:6.	
Patrons: Murphy, et al.	
Continued from 2020 Regular Session in Senate Committee on the Judiciary	
H.B. 1329. Chesapeake Bay Preservation Areas; local governments to designate Areas,	
incorporate into local plans and ordinances certain penalties. Amending § 62.1-44.15:74.	
Patron: Kory	
Continued from 2020 Regular Session in Senate Committee on Agriculture, Conservation and	
Natural Resources	
H.B. 1619. Income tax, state; subtraction for low-income military veterans with a permanent	
service-connected disability. Amending § 58.1-322.02.	
Patrons: Helmer, et al.	
Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6	
H.B. 1621. Virginia Urban Agriculture Advisory Council; created, report.	
Adding §§ 3.2-3122 through 3.2-3127.	
Patrons: Plum, et al.	
Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6	
H.B. 1633. Education, Board of; school modernization loan interest rate subsidy payments,	
eligibility. Amending § 22.1-146.1.	
Patron: Edmunds	
Continued from 2020 Regular Session in Senate Committee on Finance and Appropriations 6	
H.B. 1650. Small Business Procurement Enhancement Program; established, report.	
Amending §§ 2.2-1604, 2.2-1605, 2.2-4310, 2.2-4310.3, and 2.2-4343;	
adding §§ 2.2-1618 through 2.2-1623.	
Patron: Ward	
Continued from 2020 Regular Session in Senate Committee on General Laws and Technology 6	
H.B. 1674. Eastern Virginia Groundwater Management Area; provisional surface water	
withdrawal permit. Adding § 62.1-44.15:22.1. Patrons: Hodges, et al.	
Continued from 2020 Regular Session in Senate Committee on Agriculture, Conservation and	
Natural Resources	
H.B. 1736. School nurses; no individual who provides nursing services in a public elementary	
or secondary school shall use title unless individual is a registered nurse.	
Amending § 22.1-274.	
Patrons: Adams, D.M., et al.	
Passed House	
Constitutional reading dispensed, referred to Committee on Education and Health 172	
Continued to 2021 Special Session I in Senate Committee on Education and Health	
H.B. 1737. Nurse practitioners; reduces the number of years of full-time clinical experience a	
practitioner must have to be eligible to practice without a written or electronic practice	
agreement, sunset provision. Amending § 54.1-2957.	
Patrons: Adams, D.M., et al.	
Passed House	
Constitutional reading dispensed, referred to Committee on Education and Health	
Continued to 2021 Special Session I in Senate Committee on Education and Health	
H.B. 1743. Industrial development authorities; reduces number of members that may be	
appointed to authority boards by the Towns of Kenbridge and Victoria and reduces the	
quorum requirement. Amending § 15.2-4904.	
Patron: Wright	
Passed House	

H.B	. 1743 (continued) Constitutional reading dispensed, referred to Committee on Local Government
	Reported
	Constitutional reading dispensed, passed by for day
	Read third time and passed Senate
	Signed by President 502
D	Approved by Governor-Chapter 3 (effective 7/1/21)
н.в	. 1747. Clinical nurse specialist; licensure of nurse practitioners as specialists by the Boards of Medicine and Nursing and a practitioner licensed as a specialist shall practice pursuant to a practice agreement between the specialist and a licensed physician, etc., prescriptive authority, repeals provision relating to registered nurse's licensure by endorsement. Amending §§ 54.1-2900, 54.1-2901, 54.1-2957, 54.1-2957.01, and 54.1-3000; repealing § 54.1-3018.1.
	Patrons: Adams, D.M., et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 108
	Continued to 2021 Special Session I in Senate Committee on Education and Health 495
H.B	. 1749. Nassawadox, Town of; amending charter, updates charter to reflect the town's shift of municipal elections from May to November.
	Patron: Bloxom
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session I in Senate Committee on Local Government
	. 1750. Dairy Producer Margin Coverage Premium Assistance Program; established, eligible dairy producer shall apply to the Department of Conservation and Recreation by February 1 of each year to participate, report, sunset date. Adding §§ 3.2-3304 through 3.2-3307.
	Patrons: Gooditis, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
H.B	. 1751. Peanuts; extends sunset date of excise tax on all peanuts grown in Virginia. Amending § 3.2-1905.
	Patrons: Brewer, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
H.B	. 1754. Employer or other person; retaliatory discharge of employee prohibited. Amending §§ 40.1-27.3 and 65.2-308.
	Patrons: Carter, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494

H.B	1. 1760. Conservation easements; an easement held pursuant to the Virginia Conservation Easement Act or the Open-Space Land Act, etc., be liberally construed in favor of
	achieving the purposes for which it was created. Adding §§ 10.1-1016.1 and 10.1-1705.1. Patrons: Webert, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
H.B	3. 1763. Agricultural best management practices; creates an enhanced individual and
	corporate income tax credit beginning in taxable year 2021 for the implementation of
	certain practices by the taxpayer that are required as part of a certified resource
	management plan, definition of "liability after such cost-share." Amending §§ 58.1-339.3
	and 58.1-439.5.
	Patrons: Wilt, et al.
	Passed House
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B	1. 1764. Crewe, Town of; amending charter, various changes to the charter including
	staggering town council elections, etc.
	Patron: Wright
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
H.B	1. 1768. Lynchburg Parking Authority; shortens the term of office of appointees.
	Amending Chapter 76, 1974 Acts.
	Patrons: Walker, et al. Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Reported
	Constitutional reading dispensed, passed by for day
	Read third time and passed Senate
	Signed by President
	Approved by Governor-Chapter 4 (effective 7/1/21)
H.B	2. 1774. Tangible personal property taxes; classification of certain motor vehicles,
	trailers, and semitrailers. Amending § 58.1-3506.
	Patrons: Walker, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 136
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B	. 1775. State Corporation Commission; adds the Commission to the list of agencies that
	are exempt from paying fees for remote access to local land records.
	Amending § 17.1-276.
	Patron: Kilgore
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
H.B	. 1776. Education, Board of; temporary extension of the license of any individual licensed
	by Board whose license expires on June 30, 2021.
	Patrons: Ward, et al.
	Passed House

H.B	. 1776 (continued) Constitutional reading dispensed, referred to Committee on Education and Health 378, 379
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B	•
	Amending § 15.2-901.
	Patrons: Ward, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government 378, 379
	Continued to 2021 Special Session I in Senate Committee on Local Government
H.B	. 1783. Glasgow, Town of; new charter (previous charter repealed).
	Patron: Campbell, R.R.
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session I in Senate Committee on Local Government
H.B	. 1786. Minimum wage; eliminates the exemption from Virginia's requirements for
	persons employed as farm laborers or farm employees and any individual employed as a
	temporary foreign worker. Amending § 40.1-28.9.
	Patrons: Ward, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B	. 1789. Auditor of Public Accounts; audits of certain political subdivisions.
	Amending § 30-140.
	Patron: McNamara
	Passed House
	Constitutional reading dispensed, referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.B	. 1790. Public schools; severe weather conditions and other emergency situations,
	unscheduled remote learning days, school provides instruction and student services, etc.
	Amending § 22.1-98.
	Patrons: McNamara, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 172
	Continued to 2021 Special Session I in Senate Committee on Education and Health \ldots 495
H.B	. 1796. License plates, special; removes the fee for the issuance of a license plate for
	retired members of the Virginia National Guard. Amending § 46.2-746.5.
	Patrons: Walker, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation 378, 379
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B	. 1798. Brunswick County school board; removes school board from the list of approved
	member salaries for appointed school boards. Amending § 22.1-32.
	Patron: Tyler
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 136
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B	. 1800. Budget bill; appropriations for 2020-2022 biennium. Amending Chapter 56, 2020
	Sp. I Acts.
	Patron: Torian
	Continued to 2021 Special Session I in House Committee on Appropriations

	1801. Litter; increases fine for dumping or disposing of trash or other unsightly matter on
	public or private property. Amending § 33.2-802.
	Patrons: Edmunds, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B.	1804. State parks; Department of Conservation and Recreation to develop
	recommendations for funding, report.
	Patrons: Orrock, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
	Resources
H.B.	1805. Aging services; Department for Aging and Rehabilitative Services shall prioritize
	providing services to older persons with greatest economic or social need, definition.
	Amending §§ 51.5-134 and 51.5-135.
	Patrons: Adams, D.M., et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
	Services
H.B.	1806. Suspension or modification of sentence; transfer to the Department of
	Corrections. Amending § 19.2-303.
	Patron: Kilgore
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B.	1807. Health maintenance organizations; updates provisions of the Code of Virginia
	related to insolvency procedures for health maintenance organizations (HMOs).
	Amending § 38.2-4310.
	Patron: Kilgore
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B.	1808. Behavioral Health and Developmental Services, Commissioner of; reports to
	designated protection and advocacy system, incident reporting system.
	Amending § 37.2-304.
	Patron: Orrock
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 108
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B.	1810. Voter registration ; failure of online voter registration system, deadline extension.
	Amending § 24.2-416.
	Patrons: VanValkenburg, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Privileges and Elections
	Continued to 2021 Special Session Lin Senate Committee on Privileges and Elections

H.B. 1811. Virginia Public Procurement Act; preference for energy-efficient and water-efficient goods. Adding § 2.2-4328.1. Patrons: Helmer, et al.	
Passed House	233
Constitutional reading dispensed, referred to Committee on General Laws and Technology Continued to 2021 Special Session I in Senate Committee on General Laws and Technology . H.B. 1812. Casino gaming; technical amendments to the gaming law related to its interaction with sports betting law, the capital investment required of an applicant for a license, etc. Amending §§ 2.2-3711, 58.1-4100, 58.1-4109, 58.1-4110, 58.1-4114, 58.1-4122, 58.1-4124, and 58.1-4125.	. 236
Patrons: Krizek, et al. Passed House	240
Constitutional reading dispensed, referred to Committee on General Laws and Technology	
Continued to 2021 Special Session I in Senate Committee on General Laws and Technology.	
H.B. 1813. Highway construction; increases the value of highway maintenance and construction projects eligible to be performed by state or local employees. Amending § 33.2-234. Patron: Krizek	
Passed House	
Constitutional reading dispensed, referred to Committee on Transportation	
Continued to 2021 Special Session I in Senate Committee on Transportation H.B. 1814. Garnishment of wages; protected portion of disposable earnings. Amending § 34-29. Patrons: Krizek, et al.	. 496
Passed House	. 168
Constitutional reading dispensed, referred to Committee on the Judiciary	. 172
etc. ; meetings to be held entirely or partially by electronic means, provided guidelines have been adopted. Amending §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953.	
Patrons: Bulova, et al. Passed House	106
Constitutional reading dispensed, referred to Committee on General Laws and Technology Continued to 2021 Special Session I in Senate Committee on General Laws and Technology . H.B. 1817. Certified nurse midwives; eliminates the requirement that midwives practice pursuant to a practice agreement and shall practice in accordance with regulations of the Boards of Medicine and Nursing. Amending §§ 54.1-2957 and 54.1-2957.01. Patrons: Adams, D.M., et al.	. 108
Passed House	316
Constitutional reading dispensed, referred to Committee on Education and Health	. 321
H.B. 1818. Workers' compensation; adds salaried or volunteer emergency medical services personnel to the list of persons to whom, after five years of service, the occupational disease presumption for death caused by hypertension or heart disease applies, personnel operating in a locality that has legally adopted a resolution declaring that it will provide one or more of the presumptions. Amending § 65.2-402. Patrons: Heretick, et al.	
Passed House	. 434
Constitutional reading dispensed, referred to Committee on Commerce and Labor	. 442

H.B. 1819. Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia	
Scenic Rivers System. Amending § 10.1-415.	
Patrons: Cole, J.G., et al.	
Passed House	233
Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and N Resources	
Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and	Natural
Resources	494
H.B. 1820. SNAP benefits program; eligibility for benefits, postsecondary education. Amending §§ 63.2-608 and 63.2-801.	
Patrons: Helmer, et al.	
Passed House	316
Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	
Services	
Services	495
H.B. 1821. Experiencing or reporting overdoses; prohibits arrest and prosecution.	
Amending § 18.2-251.03.	
Patrons: Bulova, et al.	1.60
Passed House	
Constitutional reading dispensed, referred to Committee on the Judiciary	
Continued to 2021 Special Session I in Senate Committee on the Judiciary	495
H.B. 1822. Health insurance; cost-sharing payments for covered tier one or tier two	
prescription asthma inhalers. Adding § 38.2-3407.15:6.	
Patrons: Askew, et al.	121
Passed House	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	
H.B. 1823. Public schools, child day programs, and certain other programs; carbon monoxide detectors required. Adding §§ 22.1-138.2, 22.1-289.058, and 63.2-1705.2.	434
Patrons: Askew, et al.	
Passed House	136
Constitutional reading dispensed, referred to Committee on Education and Health	
Continued to 2021 Special Session I in Senate Committee on Education and Health	
H.B. 1824. Virginia Residential Property Disclosure Act; required disclosures for buyer to	
beware, mold assessment obtained by purchaser. Amending § 55.1-703.	
Patrons: Askew, et al.	
Passed House	152
Constitutional reading dispensed, referred to Committee on General Laws and Technology.	
Continued to 2021 Special Session I in Senate Committee on General Laws and Technology	
H.B. 1825. Virginia Beach Sports or Entertainment Project; amends provisions related to	
bond issuance. Amending § 15.2-5931.	
Patron: Askew	
Passed House	118
Constitutional reading dispensed, referred to Committee on Local Government	119
Reported	
Constitutional reading dispensed, passed by for day	334
Read third time	354
Passed by temporarily	354
Passed Senate	354-55

515	ned by Tresident
App	proved by Governor-Chapter 5 (effective 7/1/21)
H.B. 182	27. Education, Board of; geographic representation of members. Amending § 22.1-9.
Pat	rons: Austin, et al.
Pas	sed House
Cor	nstitutional reading dispensed, referred to Committee on Education and Health
Cor	ntinued to 2021 Special Session I in Senate Committee on Education and Health
	28. Commissioner of the Department of Motor Vehicles; powers and duties during a clared state of emergency. Amending § 46.2-223.
Patr	rons: Roem, et al.
Pas	sed House
Cor	nstitutional reading dispensed, referred to Committee on Transportation
Cor	ntinued to 2021 Special Session I in Senate Committee on Transportation
reas pro	829. Health insurance; requiring health insurers and other carriers to establish sonable protocols and procedures for reimbursing a health professional for services wided while such professional's credentialing application is pending. nending §§ 38.2-4319 and 38.2-4509.
Patr	ron: Head
Pas	sed House
Cor	nstitutional reading dispensed, referred to Committee on Commerce and Labor
Cor	ntinued to 2021 Special Session I in Senate Committee on Commerce and Labor
	30. Virginia Small Business Financing Authority; members to have small business ding experience. Amending § 2.2-2282.
Patr	ron: Head
Pas	sed House
Cor	nstitutional reading dispensed, referred to Committee on General Laws and Technology 153
Cor	ntinued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
org	31. Home care organizations; Board of Health to include in regulations governing anizations a provision for supervision of home care attendants providing personal care vices by a licensed nurse through use of interactive audio or video technology. mending § 32.1-162.12.
	rons: Head, et al.
	ssed House
	Services
Cor	ntinued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
	Services
and	32. Virginia Highway Corporation Act; alteration of certificate of authority, powers I duties of State Corporation Commission, agreements between toll operator and partment. Amending §§ 56-539 and 56-542.
Pat	rons: Subramanyam, et al.
Pas	sed House
Cor	nstitutional reading dispensed, referred to Committee on Commerce and Labor
Cor	ntinued to 2021 Special Session I in Senate Committee on Commerce and Labor

H.B	. 1833. Conservation and Recreation, Department of; removes the authority from the Department to sell or demise certain lands over which it has control, etc., report.
	Amending § 10.1-109.
	Patrons: Subramanyam, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
uр	. 1834. Electric generating facility closures; public disclosure, integrated resource plans.
п.в	Amending § 56-599; adding § 45.1-394.1.
	Patrons: Subramanyam, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
н.в	. 1836. Natural Resources, Secretary of; changes name to the Secretary of Natural and Cultural Resources. Amending §§ 2.2-200, 2.2-205, 2.2-206, 2.2-215, 2.2-220.1, 2.2-220.2, 2.2-220.3, 2.2-1156, 2.2-1176, 2.2-2316, 2.2-2338, 2.2-2481, 2.2-2699.10, 2.2-2699.11, 2.2-2699.13, 10.1-603.25, 10.1-704, 10.1-1018, 10.1-1181.15, 10.1-1188, 10.1-1329, 10.1-1402.03, 10.1-1402.04, 10.1-1405, 10.1-2129, 10.1-2202.3, 15.2-2295.1, 28.2-207, 29.1-102, 29.1-573, 29.1-579, 30-377, 56-596.2, 58.1-344.3, 62.1-44.15:68, 62.1-44.34:25, 62.1-44.34:28, 62.1-44.117, 62.1-44.118, 62.1-44.119:1, 62.1-69.31, 62.1-69.32, 62.1-69.33, 62.1-69.41, 62.1-69.52, and 62.1-195.1.
	Patron: Plum
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
H.B	. 1837. Virginia Soil and Water Conservation Board; clarifies membership.
	Amending § 10.1-502.
	Patrons: Plum, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
H.B	. 1838. Loudoun County school board; staggered terms of its members.
	Amending § 22.1-57.3:1.1.
	Patrons: Reid, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 172
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.R	. 1841. Crosswalk design; Commissioner of Highways shall convene a work group to
11,12	determine whether there should be model policies for design and installation, and, if so,
	establish recommendations.
	Patrons: Keam, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation

H.B	. 1842. Property owners' associations and unit owners' associations; rulemaking authority concerning smoking. Amending §§ 55.1-1819 and 55.1-1959; adding §§ 55.1-1819.1 and 55.1-1960.1.
	Patrons: Keam, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology \dots 108 Continued to 2021 Special Session I in Senate Committee on General Laws and Technology \dots 495
H.B	. 1843. Charitable gaming; increase in certain maximum allowable prize amounts.
	Amending § 18.2-340.33.
	Patrons: Keam, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and
	Technology
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
н.в	. 1845. Alcoholic beverage control; delays the effective date of the 2020 alcoholic beverage control license and fee reform. Amending §§ 4.1-230, 4.1-233.1, and third, fifth, and eighth enactments of Chapters 1113 and 1114, 2020 Acts. Patrons: VanValkenburg, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Services
H.B	. 1846. License restrictions for minors; prohibition on use of handheld personal
	communications devices. Amending § 46.2-334.01.
	Patron: Robinson
	Passed House 267 Constitutional reading dispensed, referred to Committee on Transportation 269 Continued to 2021 Special Session I in Senate Committee on Transportation 496
H.B	. 1847. Sports betting; clarifies the procedures by which the Virginia Lottery determines whether an event is considered youth sports, on which betting is prohibited, an international athletic event organized by the International Olympic Committee shall not be considered to be youth sports, issuance of permits to operate sports betting platforms. Amending §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100.
	Patron: Sickles
	Passed House
	Technology
H R	. 1848. Virginia Human Rights Acts; adds discrimination on the basis of disability as an
11.1	unlawful employment practice, reasonable accommodations for persons with disabilities. Amending §§ 2.2-3902, 2.2-3905, and 51.5-41; adding § 2.2-3905.1.
	Patrons: Sickles, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 153
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	. 1849. Apprenticeship training programs; Virginia Board of Workforce Development, et al., shall review availability of programs, etc., report.
	Patron: Simonds
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 442 Continued to 2021 Special Session Lin Senate Committee on General Laws and Technology 495

H.B	. 1850. Motor vehicle weight limits; vehicles powered primarily by electric battery power or fueled primarily by natural gas. Amending § 46.2-1129.2.
	Patrons: Reid, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
H.B	. 1851. Unmanned aircraft; exempts an owner from the requirement to register.
	Amending § 5.1-5.
	Patrons: Delaney, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B	. 1852. Uniform Collaborative Law Act; created. Adding §§ 20-168 through 20-187.
	Patrons: Sullivan, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 1853. Lawyers; client accounts, repeals the provision prohibiting the Supreme Court of Virginia from adopting a disciplinary rule requiring that lawyers deposit client funds in an interest-bearing account. Amending § 54.1-3916; repealing § 54.1-3915.1.
	Patrons: Sullivan, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 1854. U.S. Route 29; authorizes the board of any locality that has adopted the county manager plan of government (Arlington County) to name any section located within the boundaries of the locality. Adding § 15.2-719.1.
	Patrons: Sullivan, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
н.в	1855. Mines, Minerals and Energy, Department of; renamed the Department of Energy. Amending §§ 2.2-204, 2.2-604.2, 2.2-1157, 2.2-1176.1, 2.2-3705.6, 2.2-4006, 10.1-606.3, 10.1-659, 10.1-1194, 10.1-1329, 10.1-1330, 10.1-1406.2, 11-34.3, 15.2-958.3, 15.2-980, 15.2-2224, 23.1-2626, 23.1-2627, 28.2-1208, 30-275, 33.2-236, 45.1-161.1, 45.1-161.2, 45.1-161.5, 45.1-161.15, 45.1-161.179, 45.1-161.292:2, 45.1-180, 45.1-229, 45.1-230, 45.1-270.4:1, 45.1-361.28, 45.1-361.41, 45.1-383, 45.1-390, 56-265.15:1, 56-576, 56-585.5, 56-594.3, 56-596.2, 58.1-439.2, 58.1-439.12:02, 58.1-3660, 58.1-3706, 58.1-3745, 62.1-44.15:21, 62.1-44.15:66, 62.1-195.1, 62.1-243, 62.1-256, 62.1-259, 63.2-805, 67-200, 67-202.1, 67-602, 67-900, 67-1000, 67-1206, 67-1208, 67-1209, 67-1403, and 67-1506; adding § 45.1-161.4:1. Patron: Sullivan
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources

H.B	. 1856. Estate planning documents; electronic execution, validity of will, codifies Uniform Electronic Wills Act. Amending §§ 32.1-291.7, 54.1-2982, 54.1-2983, 54.1-2984, 59.1-481, 64.2-100, 64.2-403, 64.2-404, 64.2-407, 64.2-450, and 64.2-701; adding §§ 64.2-459 through 64.2-468.
	Patrons: Sullivan, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
н.в	Continued to 2021 Special Session I in Senate Committee on the Judiciary
	November, etc.
	Patron: Fariss
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
н.в	Continued to 2021 Special Session I in Senate Committee on Local Government
	incurred. Amending § 15.2-958.3.
	Patrons: Guy, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Reported
	Constitutional reading dispensed, passed by for day
	Read third time and passed Senate
	Signed by President
	Approved by Governor-Chapter 6 (effective 7/1/21)
H.B	. 1862. Employee protections; medicinal use of cannabis oil. Adding § 40.1-27.4.
	Patrons: Helmer, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494
H.B	. 1864. Virginia Human Rights Act; expands definition of employer to include person
	employing one or more domestic workers. Amending § 2.2-3905.
	Patrons: Price, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 291 Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	. 1865. Kindergarten through grade 3; reading intervention services for certain students.
	Amending § 22.1-253.13:1.
	Patrons: Delaney, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 172
	Continued to 2021 Special Session I in Senate Committee on Education and Health
н.в	. 1866. Court-appointed special advocates; information sharing when advocate is
	participating in family partnership meetings. Amending §§ 9.1-153 and 9.1-156.
	Patrons: Delaney, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.R	. 1867. Victims of crime; compensation, reporting requirement. Amending § 19.2-368.10.
	Patrons: Delaney, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary

H.B	. 1868. Commercial driver's licenses; disqualifies for life from holding a license persons convicted of a felony involving an act or practice of severe forms of trafficking, etc.	
	Amending §§ 46.2-341.18, 46.2-382, and 46.2-1702.	
	Patrons: Delaney, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Transportation	
H.B	. 1873. Brain injury; clarifies definition. Amending § 37.2-403.	
	Patrons: Coyner, et al.	
	Passed House)7
	Constitutional reading dispensed, referred to Committee on Education and Health	8
	Continued to 2021 Special Session I in Senate Committee on Education and Health)5
H.B	. 1874. Behavioral health; assessments in local correctional facilities, report.	
	Amending § 53.1-68.	
	Patrons: Coyner, et al.	
	Passed House	3
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	
	Services	6
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	
	Services)5
H.B	. 1876. Workforce development; expands type of data sharing. Amending § 2.2-435.8.	
	Patrons: Subramanyam, et al.	
	Passed House	;9
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 29	
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 49	15
H.B	. 1877. Legal service plans; legal services organization shall submit registration	
	information and fees to the Commissioner. Amending § 59.1-441.2.	
	Patron: Jenkins	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Commerce and Labor	
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	14
H.B	. 1878. Juvenile intake and petition; appeal to a magistrate on a finding of no probable	
	cause. Amending §§ 16.1-256 and 16.1-260.	
	Patrons: Jenkins, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on the Judiciary	
	Continued to 2021 Special Session I in Senate Committee on the Judiciary	15
H.B	. 1879. Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed	
	wine for off-premises consumption, report, sunset provision. Amending §§ 4.1-119,	
	4.1-204, 4.1-206.1, 4.1-206.3, 4.1-207, 4.1-210, 4.1-212.1, and 4.1-221.	
	Patrons: Bulova, et al.	
	Passed House	!4
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	
	Services	g
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	
	Services	15
H.B	. 1881. Enterprise zone job creation grants; for purposes of wage requirements, the	
	minimum wage shall be the higher of the state minimum wage or the federal minimum	
	wage, delayed effective date. Amending § 59.1-547.	
	Patrons: Heretick, et al.	
	Passed House 10	ſ.

H.B	. 1881 (continued)
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
II D	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
п.р	refinanced mortgage. Amending § 55.1-319; adding § 55.1-318.1.
	Patron: Heretick
	Passed House
	Technology
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
ИD	. 1884. Income tax, state; Department of Taxation to include space on the appropriate
п.р	forms for voluntary inclusion of personal and contact information, facilitated enrollment
	program. Amending §§ 38.2-6505, 58.1-3, and 58.1-341.1.
	Patrons: Sickles, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B	. 1885. Comprehensive review of computer science standards, courses, and pathways
	in public schools; Department of Education to perform, report.
	Patrons: Simonds, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 321
	Continued to 2021 Special Session I in Senate Committee on Education and Health 495
H.B	. 1887. Foreign market vehicles; titling and registration. Amending § 46.2-602.
	Patrons: Jenkins, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B	. 1888. Absentee voting; procedural and process reforms, availability and accessibility
	reforms, penalty. Amending §§ 24.2-101, 24.2-416.1, 24.2-603, 24.2-653.1, 24.2-704,
	24.2-705, 24.2-706, 24.2-707, 24.2-708 through 24.2-711, and 24.2-712;
	adding §§ 24.2-103.2, 24.2-667.1, and 24.2-707.1.
	Patrons: VanValkenburg, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Privileges and Elections
	Continued to 2021 Special Session I in Senate Committee on Privileges and Elections 495
H.B	. 1889. Virginia Residential Landlord and Tenant Act; landlord remedies,
	noncompliance with rental agreement, payment plan, removal of sunset. Repealing second
	enactment of Chapter 46, 2020 Sp. I Acts.
	Patrons: Price, et al.
	Passed House
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
пD	. 1890. Discrimination; prohibited in voting and elections administration, required process
п.Б	for enacting certain covered practices, civil causes of action, penalties, repeals provision
	relating to minority language accessibility, etc. Amending §§ 24.2-105, 24.2-306,
	24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1; adding §§ 24.2-104.1, 24.2-125
	through 24.2-131, and 24.2-1005.2; repealing § 24.2-124.
	Patrons: Price, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Privileges and Elections
	Continued to 2021 Special Session Lin Senate Committee on Privileges and Elections

H.B	. 1891. Annual safety and disaster awareness training; Department of Human Resource
	Management, et al., to develop an online training module addressing safety and disaster awareness. Amending § 2.2-1201; adding § 2.2-1212.
	Patron: Ayala
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 29 Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 49:
H.B	. 1892. Property and casualty insurance form; approval of form by State Corporation
	Commission. Amending § 38.2-317.
	Patron: Ayala
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B	. 1893. New River Valley Passenger Rail Station Authority; creation of authority in Planning District 4 (New River Valley RC). Adding §§ 33.2-3800 through 33.2-3816.
	Patrons: Hurst, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B	. 1894. Naloxone or other opioid antagonist; certain employees of Department of
	Juvenile Justice authorized to administer. Amending § 54.1-3408.
	Patrons: Kory, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health 49:
H.B	. 1895. Fines and costs; accrual of interest, deferral or installment payment agreements. Amending §§ 19.2-353.5 through 19.2-355.
	Patrons: Hudson, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 1896. Health insurance; essential benefits, removes the prohibition on the provisions of
	coverage for abortions. Amending § 38.2-3451.
	Patrons: Hudson, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494
H.B	. 1897. Summons for unlawful detainer; notice to tenant, adverse employment actions
	prohibited. Amending § 8.01-126.
	Patrons: Jenkins, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary 172
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 1898. Zoning appeals, board of; any elected official of an incorporated town may serve
	on board of county in which member also resides. Amending § 15.2-2308.
	Patrons: Roem, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session Lin Senate Committee on Local Government

H.B	Enhancement Tax Credits; sunset date, credits earned prior to January 1, 2022. Amending §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1.
	Patrons: Hudson, et al.
	Passed House
H.B	. 1900. Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion
	from dwelling unit, interruption of services, or actions taken to make premises unsafe,
	repeals provision relating to tenant's remedies for landlord's unlawful ouster. Adding § 55.1-1243.1; repealing § 55.1-1243.
	Patrons: Hudson, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 153
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	3. 1901. Online Virginia Driver's Manual course; training school. Amending § 46.2-325.
	Patrons: Carr, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B	. 1902. Expanded polystyrene food service containers; definition, prohibits dispensing
	by a food vendor of prepared foods to a customer, civil penalty. Amending §§ 10.1-1414
	and 10.1-1422.01; adding § 10.1-1424.3.
	Patrons: Carr, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
	Resources
H.B	5. 1903. Local government; authority to reduce the speed limit in a business district or residence district. Amending § 46.2-1300.
	Patrons: Carr, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
H.B	1904. Teachers and other licensed school board employees; cultural competency.
	Amending §§ 22.1-253.13:5 and 22.1-298.1; adding § 22.1-298.7.
	Patrons: Jenkins, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 136
	Continued to 2021 Special Session I in Senate Committee on Education and Health 495
H.B	. 1905. Economic education and financial literacy required in middle and high school
	grades; employment arrangements. Amending § 22.1-200.03.
	Patrons: Cole, J.G., et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B	1. 1907. Electric utilities ; advanced renewable energy buyers. Amending § 56-585.5.
	Patrons: Sullivan, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Communication / U/ L Special Session L in Senale Committee on Commerce and Labor 494

H.B.	. 1908. Virginia Residential Landlord and Tenant Act; noncompliance with rental agreement, etc., prohibition on using negative credit information that arose during a closure of the United States Government against certain applications for tenancy, penalty. Amending § 55.1-1245. Patrons: Helmer, et al.
	Passed House
H.B.	. 1909. School board property, certain; establishment of gun-free zone permitted.
	Adding § 22.1-131.1.
	Patrons: Subramanyam, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 250, 251
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B.	. 1911. No-fault divorce; corroboration requirement. Amending §§ 20-99 and 20-106.
	Patrons: Hope, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
H.B.	1912. Child support payments; juvenile in custody of or committed to the Department
	of Juvenile Justice. Amending §§ 16.1-263, 16.1-286, and 16.1-290.
	Patrons: Hope, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
H.B.	1913. Career fatigue and wellness in certain health care providers; programs to
	address, civil immunity. Amending §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909.
	Patrons: Hope, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
D	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B.	
	Amending § 56-585.1.
	Patrons: Helmer, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
II D	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
н.в.	. 1916. Research and development tax credits; availability against the bank franchise tax for taxable years beginning on and after January 1, 2021. Amending §§ 58.1-439.12:08 and 58.1-439.12:11.
	Patrons: Mugler, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 172, 173
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations
HВ	1918. Student driver safety; driver education program shall include the dangers of
11,13	distracted driving and speeding, high school student parking passes, valid driver's license
	required. Amending § 22.1-205; adding § 22.1-205.1.
	Patrons: Mugler, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health

pro	19. Local green banks; authorizes a locality, by ordinance, to establish a bank to mote the investment in clean energy technologies in its locality, etc. ding § 15.2-958.3:1.
	rons: Kory, et al.
	sed House
	nstitutional reading dispensed, referred to Committee on Local Government
	ntinued to 2021 Special Session I in Senate Committee on Local Government
H.B. 1	921. Assistance for certain voters; curbside voting. Amending §§ 24.2-638, 2-646.1, and 24.2-649; adding § 24.2-649.1.
Pat	rons: Price, et al.
Pas	sed House
Con	nstitutional reading dispensed, referred to Committee on Privileges and Elections 172, 173
Con	ntinued to 2021 Special Session I in Senate Committee on Privileges and Elections 495
H.B. 1	923. Electric utilities; expands existing broadband capacity pilot program. nending § 56-585.1:9.
Pat	rons: Ayala, et al.
Pas	sed House
Con	nstitutional reading dispensed, referred to Committee on Commerce and Labor 153, 154
Con	ntinued to 2021 Special Session I in Senate Committee on Commerce and Labor
	25. Virginia Brownfield and Coal Mine Renewable Energy Grant Fund and ogram; established, report. Adding § 67-1800.
Pat	rons: Kilgore, et al.
Pas	sed House
Con	nstitutional reading dispensed, referred to Committee on Commerce and Labor
Con	ntinued to 2021 Special Session I in Senate Committee on Commerce and Labor
Vir	26. Central Virginia Transportation Authority; adds the Executive Director of the ginia Port Authority, or his designee, as a nonvoting ex officio member. aending § 33.2-3703.
Pat	rons: McQuinn, et al.
Pas	sed House
Con	nstitutional reading dispensed, referred to Committee on Transportation
Con	ntinued to 2021 Special Session I in Senate Committee on Transportation
	27. Economic development authorities; allows Fairfax County to appoint nine, rather n seven, commissioners to the authority. Amending Chapter 882, 2003 Acts.
Pat	rons: Sickles, et al.
Pas	sed House
Con	nstitutional reading dispensed, referred to Committee on Local Government
Rep	ported
Cor	nstitutional reading dispensed, passed by for day
Rea	ad third time
Pas	sed by temporarily
Pas	sed Senate
Sig	ned by President
_	proved by Governor-Chapter 7 (effective 7/1/21)

H.B	. 1928. Historic resources; land acquisition activities of the Department of Historic Resources and the Board of Historic Resources, etc. Amending §§ 10.1-2202 and
	10.1-2204.
	Patrons: Aird, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
H.B	. 1930. Higher educational institutions, public; admissions applications that contain
	questions about criminal history. Adding § 23.1-407.1.
	Patrons: Aird, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 269
H.B	Continued to 2021 Special Session I in Senate Committee on Education and Health 495 1931. Virginia Freedom of Information Act; public body authorized to conduct
	electronic meetings. Amending § 2.2-3708.2.
	Patrons: Levine, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 236 $$
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology $\dots 495$
H.B	. 1932. Child-placing agencies; repeals provisions that allowed agencies to refuse to
	perform, assist with, etc., in any child placements when the proposed placement would
	violate the agency's written religious or moral convictions or policies, any agency that
	operated prior to February 1, 2021, may continue to operate until January 1, 2022, etc.
	Repealing § 63.2-1709.3.
	Patrons: Levine, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Services
H.B	. 1935. Income tax, state; conformity of the Commonwealth's taxation system with the
	Internal Revenue Code, deduction relating to payroll expenses. Amending §§ 58.1-301, 58.1-322.02, and 58.1-322.03.
	Patrons: Watts, et al.
	Passed House
	$Constitutional\ reading\ dispensed,\ referred\ to\ Committee\ on\ Finance\ and\ Appropriations\ \dots\ 460,\ 461$
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B	. 1936. Robbery; definition, creates degrees of punishment corresponding to the severity
	of a robbery offense. Amending § 18.2-58.
	Patrons: Watts, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 1940. Students; guidelines established by the Department of Education on excused
	student absences, local school boards may require that student provide advance notice of
	intended absence, etc., civic engagement. Amending § 22.1-254.
	Patrons: Rasoul, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 153, 154 Continued to 2021 Special Session I in Senate Committee on Education and Health 495

H.B. 1942. Public adjusters; continuing education requirements, repeals provision relating to continuing education, approval of credits, etc. Amending §§ 38.2-1866, 38.2-1867, 38.2-1868.1, 38.2-1869, 38.2-1871, and 38.2-1873; repealing § 38.2-1845.9. Patron: Kilgore	
Passed House	152
Constitutional reading dispensed, referred to Committee on Commerce and Labor Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	
device, use of devices shall be limited to one player at a time. Amending §§ 18.2-340.19 and 18.2-340.28. Patron: Willett	222
Passed House	236
H.B. 1944. Casino gaming; requirements for issuance of operator's license, human trafficking training. Amending § 58.1-4110.Patrons: Simonds, et al.	
Passed House	234
Constitutional reading dispensed, referred to Committee on General Laws and Technology Continued to 2021 Special Session I in Senate Committee on General Laws and Technolog	236
H.B. 1948. Law-enforcement officer; duty to render aid, duty to report wrongdoing by another officer. Amending § 52-30.1; adding §§ 15.2-1704.1 and 15.2-1704.2. Patrons: Levine, et al.	
Passed House	118
Constitutional reading dispensed, referred to Committee on the Judiciary	
H.B. 1949. Albemarle and Prince William Counties; counties that have adopted the county executive form of government may carry over unspent funds from year to year for multiyear capital projects and outstanding grants. Amending §§ 15.2-520 and 15.2-2506. Patrons: Runion, et al.	110
Passed House	
Constitutional reading dispensed, referred to Committee on Local Government	
Constitutional reading dispensed, passed by for day	
Read third time and passed Senate	
Signed by President	
Approved by Governor-Chapter 8 (effective 7/1/21)	
H.B. 1950. Fetal and Infant Mortality Review Team ; Office of the Chief Medical Examiner of Department of Health shall convene a work group to develop a plan for the	
establishment of Team, report.	
Patrons: Ayala, et al. Passed House	267
Constitutional reading dispensed, referred to Committee on Education and Health Continued to 2021 Special Session I in Senate Committee on Education and Health	269
H.B. 1951. Suicide; abolishes the common-law crime. Adding § 18.2-16.1. Patrons: Simon, et al.	
Passed House	
Constitutional reading dispensed, referred to Committee on the Judiciary	

	1952. Campaign finance; prohibited personal use, child care exception. Amending § 24.2-948.4; adding § 24.2-948.5.
	Patrons: Simon, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Privileges and Elections
	consultation with a licensed physician with a practice agreement, report. Amending § 54.1-2900; adding § 54.1-2957.04. Patrons: Gooditis, et al.
	Passed House
	1957. Adult adoption; investigation and report. Amending § 63.2-1244.
	Patrons: Avoli, et al.
	Passed House
	Services
D	Services
	1958. South River; designates a 6.5-mile segment in the City of Waynesboro as a
	component of the Virginia Scenic Rivers System. Adding § 10.1-417.1.
	Patrons: Avoli, et al.
	Passed House
	Resources
	Resources
HВ	1960. Vehicle registration; expands eligibility for special communication needs
	indicator. Amending § 46.2-600.1.
	Patrons: Bell, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
	1961. Special identification cards; application by guardian. Amending § 46.2-345.
	Patrons: Bell, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
	1962. Foster care; termination of parental rights, relatives and fictive kin.
	Amending §§ 16.1-281, 16.1-283, 63.2-906, and 63.2-910.2.
	Patrons: Gooditis, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social Services
H.B.	1963. Funding local health departments; cooperative local health budget, report.
	Adding § 32.1-34.3.
	Patrons: Bagby, et al.
	Passed House 375

H.B	. 1963 (continued)
	Constitutional reading dispensed, referred to Committee on Education and Health 378, 379
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B	. 1964. State Corporation Commission; supervisory merger or transfer of assets of
	financially unstable credit unions, repeals provision relating to consolidation or merger.
	Amending § 6.2-1317; repealing § 6.2-1318.
	Patron: Bagby
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor 153, 154
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B	. 1965. State Air Pollution Control Board; implementation of a low-emissions and
	zero-emissions vehicle program, energy jurisdictional retail sales. Amending § 10.1-1307;
	adding § 10.1-1307.04.
	Patrons: Bagby, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
II D	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B	. 1967. Virginia Jobs Investment Program and Fund; minimum wage requirements.
	Amending § 2.2-2240.3.
	Patrons: Bagby, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 153
rr n	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	. 1968. Absentee voting; availability on Sundays in office of general registrar or voter
	satellite office. Amending § 24.2-701.1.
	Patrons: Bagby, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Privileges and Elections 172, 173
II D	Continued to 2021 Special Session I in Senate Committee on Privileges and Elections
н.в	. 1969. Administration of blighted and derelict properties; modifies definition of
	"qualifying locality." Amending §§ 58.1-3221.6 and 58.1-3970.1.
	Patrons: Carr, et al.
	Passed House
II D	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
п.в	. 1971. Virginia Fair Housing Law; reasonable accommodations, disability-related requests for parking. Amending § 36-96.3:2.
	Patrons: Carr, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 291
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	•
11.D	Amending §§ 4.1-206.3 and 4.1-209.
	Patrons: Rush, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
	Services
II D	. 1976. Virginia Health Workforce Development Authority; mission of Authority,
11.D	membership. Amending §§ 32.1-122.7 and 32.1-122.7:1.
	Patrons: Willett, et al.
	Passed House 234

H.B. 1	976 (continued)
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health 495
H.B.	1978. Legislative Staff Development Fund; created, effective date.
	mending § 2.2-424; adding § 30-19.21.
	atrons: Reid, et al.
P	assed House
C	Constitutional reading dispensed, referred to Committee on Rules
C	Continued to 2021 Special Session I in Senate Committee on Rules
H.B. 1	1979. Electric vehicle rebate program; creation and funding, report, sunset date.
A	dding §§ 67-1800 through 67-1806.
P	atrons: Reid, et al.
P	assed House
C	Constitutional reading dispensed, referred to Committee on Finance
	and Appropriations
C	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
	1980. Enslaved Ancestors College Access Scholarship and Memorial Program;
	stablished, report. Adding § 23.1-615.1.
	atrons: Reid, et al.
	assed House
C	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
	1981. Virginia Residential Landlord and Tenant Act; access to dwelling unit during
	ertain declared states of emergency, nonemergency repairs, employees and agents sent by
	andlord for maintenance or inspection are to wear all appropriate personal protective
	quipment as required by state law. Amending § 55.1-1229.
	atrons: Carr, et al.
	assed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 321
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
	1982. Nutrient credits; use by facility with certain stormwater discharge permit.
	mending § 62.1-44.19:21.
	atrons: Bulova, et al.
	assed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
C	Resources
C	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
	Resources
HВ	1983. Wetland and stream mitigation banks; proximity of impacted site.
	mending §§ 62.1-44.15:23 and 62.1-44.15:23.1.
	atrons: Bulova, et al.
	assed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
C	Resources
C	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
C	Resources
пр 1	984. Electric utilities; triennial review proceeding by SCC, fair rates of return.
	atrons: Hudson, et al.
	assed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
C	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494

H.B.	. 1985. Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of certain health care providers is an occupational disease. Amending § 65.2-402.1. Patrons: Hurst, et al.
	Passed House
H.B.	. 1986. George Mason University; management agreement with the Commonwealth.
	Patrons: Bulova, et al.
	Passed House
H.B.	. 1987. Telemedicine service; coverage of telehealth services by an insurer, etc., services
	delivered through real-time audio-only telephone, report. Amending §§ 32.1-325, 38.2-3418.16, and 54.1-3303. Patrons: Adams, D.M., et al.
H.B.	Passed House
	Patrons: Adams, D.M., et al.
	Passed House
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B.	. 1989. Public health emergency; emergency medical services agencies, real-time access to information regarding confirmed and suspected cases of communicable disease of public health threat. Adding § 32.1-48.001. Patrons: Aird, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 378, 379. Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B.	. 1990. Criminal justice legislation; racial and ethnic impact statements.
	Adding § 30-19.1:13.
	Patrons: Aird, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.B.	. 1991. Juveniles; release and review hearing for serious offender, plea agreement.
	Amending §§ 16.1-285.1 and 16.1-285.2.
	Patrons: Jones, et al.
	Passed House
11 D	Constitutional reading dispensed, referred to Committee on the Judiciary
H.B.	. 1992. Firearms; purchase, possession, etc., following conviction for assault and battery
	of a family or household member, definition, penalties. Amending §§ 18.2-308.09, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:8.
	Patrons: Murphy, et al. Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary

H.B	. 1993. State agencies and their appointing authorities; diversity, equity, and inclusion strategic plans. Amending § 2.2-602.
	Patrons: Askew, et al.
	Passed House
II D	Constitutional reading dispensed, referred to Committee on General Laws and Technology 29 Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
н.в	. 1994. Small agricultural generators; expands definition. Amending § 56-594.2.
	Patrons: Murphy, et al. Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor 153, 154 Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B	. 1995. Rare Disease Council and Rare Disease Council Fund; created, report.
	Adding §§ 32.1-73.14 through 32.1-73.17.
	Patrons: Murphy, et al.
H.B	Passed House
	Patron: Murphy
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 236
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	. 1998. Public schools; lock-down drills, annual requirement. Amending § 22.1-137.2.
	Patrons: Murphy, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
H.B	. 1999. Tax Commissioner; waiver of accrual of interest in the event that Governor
	declares state of emergency. Amending § 58.1-112.
	Patrons: Murphy, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 172, 173
H.B	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495 2001. State and local buildings, certain; definitions, building standards, high
	performance building certification program, sufficient ZEV charging and fueling
	infrastructure, etc. Amending §§ 2.2-1182 and 2.2-1183; adding § 15.2-1804.1.
	Patrons: Helmer, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 269 Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	. 2002. Child support; availability of medical assistance through Family Access to
	Medical Insurance Security (FAMIS) plan or other government-sponsored coverage,
	eligibility requirements. Amending §§ 16.1-260 and 63.2-1903.
	Patrons: Samirah, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 2003. Consumer Protection Act; prohibited practices, certain advertising related to any
	public or private school quality. Amending § 59.1-200.
	Patrons: Samirah, et al.
	Passed House

H.B.	2003 (continued) Constitutional reading dispensed, referred to Committee on General Laws and	
	Technology	
H.B.	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 4 2004. Virginia Freedom of Information Act; law-enforcement criminal incident	95
	information, criminal investigative files. Amending §§ 2.2-3706, 2.2-3711, 19.2-174.1, and 19.2-368.3; adding § 2.2-3706.1.	
	Patrons: Hurst, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 4 Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 4	
	2005. Disposition of the remains of a decedent; persons to make arrangements for funeral and disposition of remains, person may designate in a signed and notarized writing, repeals provisions relating to when next of kin disagree and absence of next of kin. Amending §§ 32.1-309.1, 54.1-2800, 54.1-2807, 54.1-2825, and 57-27.3; adding §§ 54.1-2825.1, 57-27.4, 57-27.5, and 57-27.6; repealing §§ 54.1-2807.01 and 54.1-2807.02.	
	Patron: Sickles	
	Passed House	
	Technology	
	2006. Energy storage systems ; definitions, tax exemption, revenue share for systems. Amending §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660.	
	Patrons: Heretick, et al.	
	Passed House	21
H.B.	2007. Prescription drugs; price transparency, definitions. Amending § 2.2-3705.6; adding §§ 32.1-23.3, 38.2-3407.15:6, 38.2-3407.22, 54.1-3436.1, and 54.1-3442.02.	
	Patrons: Sickles, et al.	
	Passed House	79
	2008. Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization issued by carrier for drug, certain provisions shall not apply to the state employee health insurance plan, etc. Amending § 38.2-3407.15:2.	
	Patrons: Heretick, et al.	00
	Passed House	91
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	94
	2009. Chamberlin Hotel at Fort Monroe; reverts certain property to the Commonwealth. Repealing Section 1 of Chapter 809, 1998 Acts.	
	Patron: Mugler	
	Passed House	91
	Continued to 2021 Special Session I in Senate Committee on the Judiciary	95
	2010. Earned sentence credits; rate at which sentence rates may be earned, prerequisites. Amending § 53.1-202.3.	
	Patrons: Campbell, J.L., et al.	75
	Pacced House	. 17

H.B.	2010 (continued) Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social Services
HВ	2012. Protective orders; violations of preliminary child protective order, violation
11.0	involves an act that endangers the child's normal development, changes punishment, etc. Amending §§ 16.1-253 and 16.1-253.2.
	Patron: Campbell, J.L.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
нв	2013. School boards; each board to adopt a policy that prohibits a lawsuit against a
п.р.	student or the student's parent because student cannot pay for a meal at school, etc.
	Amending § 22.1-79.7.
	Patrons: Roem, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B.	2014. Virginia Residential Landlord and Tenant Act; sample termination notice,
	landlord's acceptance of rent with reservation, tenant's right of redemption.
	Amending §§ 36-139 and 55.1-1250.
	Patrons: Price, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and
	Technology
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B.	, , ,
	Amending § 16.1-260; adding § 16.1-309.11.
	Patrons: Mullin, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B.	2018. Emergency order for adult protective services; definitions, acts of violence, etc.,
	or financial exploitation. Amending §§ 63.2-1603, 63.2-1606, and 63.2-1609.
	Patrons: Mullin, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B.	2019. Public elementary and secondary schools; possession and administration of
	undesignated stock albuterol inhalers and valved holding chambers, Department of
	Health, in conjunction with the Department of Education, shall develop and implement
	policies. Amending §§ 8.01-225, 22.1-274.2, and 54.1-3408.
	Patrons: McQuinn, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 269
	Continued to 2021 Special Session I in Senate Committee on Education and Health 495
H.B.	2020. Nomination of candidates for elected offices; restrictions on nomination method
	selected by political party. Amending § 24.2-509.
	Patrons: Helmer, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Privileges and Elections 321
	Continued to 2021 Special Session I in Senate Committee on Privileges and Elections 495

	2024. Bob White Covered Bridge; Department of Transportation to work with the
	governing body of Patrick County and community groups interested in constructing a
	replica of Bridge.
	Patron: Poindexter
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
	2025. Virginia Freedom of Information Act; record exclusion for personal contact
	information provided to a public body. Amending § 2.2-3705.1.
	Patrons: Gooditis, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 236
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
	2027. Standards of Learning; reading and mathematics assessments for grades three
	through eight, individual student growth, through-year growth assessments.
	Amending § 22.1-253.13:3.
	Patrons: Coyner, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	2029. Fire training activities; definitions, prohibition on the use of certain oriented
	strand board. Adding § 9.1-207.2.
	Patrons: Krizek, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 291
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
	2030. Neonicotinoid pesticides; application in residentially zoned outdoor areas, civil
	penalty. Adding § 3.2-3929.1.
	Patrons: Krizek, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
	Resources
H.B.	2031. Facial recognition technology; definition, authorization of use by local
	law-enforcement agencies and public institutions of higher education.
	Adding §§ 15.2-1723.2 and 23.1-815.1.
	Patrons: Aird, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 136
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
	2032. Employees providing domestic service; application of laws applicable to
	employee safety and workers' compensation. Amending §§ 40.1-2, 40.1-49.3, 40.1-49.8,
	65.2-101, and 65.2-305.
	Patrons: Gooditis, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494
	2034. Electric utilities; nonjurisdictional customers, third party power purchase
	agreements. Amending Chapters 1187, 1188, 1189, 1193, 1194, and 1239, 2020 Acts.
	Patrons: Hurst, et al. Passed House 15
	Passed mouse 15:

H.B. 2024 (continued)	
H.B. 2034 (continued) Constitutional reading dispensed, referred to Committee on Commerce and Labor 153,	154
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	
H.B. 2035. Virginia Initiative for Education and Work; participants, modifies Full	171
Employment Program (FEP), wages earned by FEP employee during period for which his	
employer receives a subsidy. Amending § 63.2-608.	
Patrons: Tran, et al.	
Passed House	375
Constitutional reading dispensed, referred to Committee on Education and Health 378,	
Continued to 2021 Special Session I in Senate Committee on Education and Health	
H.B. 2036. Virginia Employment Commission; communications with parties, use of	173
electronic means, report. Adding § 60.2-121.1.	
Patrons: Tran, et al.	
Passed House	290
Constitutional reading dispensed, referred to Committee on Commerce and Labor	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	
H.B. 2037. Unemployment compensation; under specific conditions related to the COVID-19	171
virus, work will not be deemed suitable and benefits will not be denied to any otherwise	
eligible individual for refusing to accept new work. Amending §§ 60.2-528 and 60.2-618.	
Patrons: Tran, et al.	
Passed House	350
Constitutional reading dispensed, referred to Committee on Commerce and Labor	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	
H.B. 2038. Probation, revocation, and suspension of sentence; limitations on sentence,	
exceptions, technical violation. Amending §§ 19.2-303, 19.2-303.1, and 19.2-306;	
adding § 19.2-306.1.	
Patrons: Scott, et al.	
Passed House	317
Constitutional reading dispensed, referred to Committee on the Judiciary	
Continued to 2021 Special Session I in Senate Committee on the Judiciary	
H.B. 2039. Physician assistant; eliminates requirement that an assistant enter into a practice	
agreement with one or more patient care team physicians, etc. Amending §§ 54.1-2902,	
54.1-2950.1, 54.1-2951.1, 54.1-2951.2, 54.1-2952, 54.1-2952.1, 54.1-2953, and	
54.1-2972; adding § 54.1-2951.4.	
Patrons: Rasoul, et al.	
Passed House	318
Constitutional reading dispensed, referred to Committee on Education and Health	321
Continued to 2021 Special Session I in Senate Committee on Education and Health	495
H.B. 2040. Unemployment compensation; failure to respond, continuation of benefits,	
repayment of overpayments. Amending §§ 60.2-528.1, 60.2-619, and 60.2-633.	
Patrons: Hudson, et al.	
Passed House	456
Constitutional reading dispensed, referred to Committee on Commerce and Labor	461
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	494
H.B. 2042. Trees; replacement and conservation during development, effective date.	
Amending §§ 15.2-961 and 15.2-961.1.	
Patrons: Guy, et al.	
Passed House	249
Constitutional reading dispensed, referred to Committee on Local Government	250
Continued to 2021 Special Session Lin Senate Committee on Local Government	495

H.B. 2046. Virginia Fair Housing Law; unlawful discriminatory housing practices. Amending §§ 36-96.3 and 36-96.17.	
Patrons: Bourne, et al. Passed House	91
 H.B. 2047. Criminal proceedings; consideration of mental condition and intellectual and developmental disabilities, qualifications for court-appointed council. Amending §§ 19.2-120, 19.2-163.03, and 19.2-299; adding § 19.2-271.6. Patrons: Bourne, et al. 	
Passed House	79
 H.B. 2048. Electric utility regulation; purchasing from competitive suppliers. Amending § 56-577. Patrons: Bourne, et al. 	
Passed House	55
Constitutional reading dispensed, referred to Committee on Commerce and Labor	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 49	
H.B. 2049. Electric utilities; eliminates customer credit reinvestment offsets.	
Amending § 56-585.1.	
Patrons: Bourne, et al.	
Passed House	
Constitutional reading dispensed, referred to Committee on Commerce and Labor	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor) 4
H.B. 2053. Affordable and market-rate housing; Department of Housing and Community	
Development to convene an advisory group to evaluate construction of internal, etc., dwelling units.	
Patrons: Samirah, et al.	
Passed House	56
Constitutional reading dispensed, referred to Committee on General Laws and Technology 40	
Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 49	
H.B. 2054. Comprehensive plan; provision for transit-oriented development.	
Amending § 15.2-2223.4.	
Patrons: Samirah, et al.	
Passed House	19
Constitutional reading dispensed, referred to Committee on Local Government	50
Continued to 2021 Special Session I in Senate Committee on Local Government)5
H.B. 2055. Child support obligations; party's incarceration not deemed voluntary	
unemployment or underemployment. Amending §§ 20-108.1 and 63.2-1918.	
Patrons: Scott, et al.	
Passed House	
Constitutional reading dispensed, referred to Committee on the Judiciary	
Continued to 2021 Special Session I in Senate Committee on the Judiciary)5
H.B. 2056. Status offenders; willful and material violation of court order or terms of	
probation, notice, orders of disposition for violation. Amending §§ 16.1-278.5, 16.1-291,	
and 16.1-292.	
Patrons: Scott, et al.	10
Passed House	
Constitutional reading dispensed, referred to Committee on the Judiciary	

H.B.	2058. Virginia Science, Technology, Engineering, and Mathematics (STEM) Advisory Board; established, report. Adding §§ 22.1-364 through 22.1-368.
	Patrons: Simonds, et al.
	Passed House
H.B.	Constitutional reading dispensed, referred to Committee on Education and Health
	to file tax returns when due to prepare and file such returns, exception. Adding § 58.1-1802.2. Patrons: Coyner, et al. Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 236 Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B.	2060. Online portal for tax practitioners; Department of Taxation shall analyze
	prospect of establishing, report.
	Patrons: Coyner, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 172, 173 Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B.	2061. Virginia Immunization Information System; any health care provider in the
	Commonwealth that administers immunizations to participate in System. Amending § 32.1-46.01.
	Patrons: Willett, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B	2062. Food delivery platforms; agreements required, penalty. Amending § 59.1-200; adding §§ 59.1-571, 59.1-572, and 59.1-573.
	Patrons: Willett, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494
H.B.	2063. Virginia Overtime Wage Act; overtime compensation employees, definitions, penalties. Amending §§ 40.1-29 and 40.1-29.1; adding § 40.1-29.2.
	Patrons: Mullin, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494
H.B.	2064. Electronic notarial certificate; clerk's office to record a paper copy of an electronic document, provided that such copy otherwise meets the requirements for
	recordation and is certified to be a true and correct copy. Amending §§ 17.1-223, 47.1-2,
	47.1-16, and 55.1-606.
	Patrons: Simon, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
нρ	Continued to 2021 Special Session I in Senate Committee on the Judiciary
11.D	a three-year pilot Program, report.
	Patrons: McQuinn, et al.
	Passed House 434

H.B. 2065 (continued) Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	
Services	
H.B. 2068. Local Food and Farming Infrastructure Grant Program; established. Amending §§ 3.2-303, 3.2-304, and 3.2-310; adding § 3.2-311. Patrons: Rasoul, et al.	
Passed House	
Resources	
Resources	
H.B. 2069. License plates, special; establishes a system for determining cost of plates for recipients of a military decoration. Amending §§ 46.2-742.1, 46.2-742.2, 46.2-745.1, and 46.2-745.2; adding §§ 46.2-725.3, 46.2-745.4, and 46.2-745.5. Patrons: McQuinn, et al.	
Passed House	. 376
Constitutional reading dispensed, referred to Committee on Transportation	, 379
Patrons: Willett, et al. Passed House	22/
Constitutional reading dispensed, referred to Committee on Rehabilitation and Social Services	
Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	
Services	. 495
Patrons: Convirs-Fowler, et al.	
Passed House	. 290
Constitutional reading dispensed, referred to Committee on Transportation	
H.B. 2072. Virginia Good Neighbor Next Door program; Virginia Housing Development	
Authority shall report recommendations for creating Program.	
Patrons: Convirs-Fowler, et al.	
Passed House	
Constitutional reading dispensed, referred to Committee on General Laws and Technology	
Continued to 2021 Special Session I in Senate Committee on General Laws and Technology . H.B. 2073. Wrongful death statute of limitations; criminal investigations by law	
enforcement. Amending § 8.01-244.	
Patrons: Convirs-Fowler, et al.	
Passed House	. 318
Constitutional reading dispensed, referred to Committee on the Judiciary	
H.B. 2074. Interagency Environmental Justice Working Group; established as an advisory	
council in executive branch of state government to further environmental justice. Amending § 2.2-234; adding §§ 2.2-236, 2.2-237, and 15.2-2223.5.	
Patrons: Simonds, et al.	
Passed House Constitutional reading dispensed referred to Committee on General Laws and Technology	
Constitutional reading dispensed, referred to Committee on General Laws and Technology Continued to 2021 Special Session Lin Senate Committee on General Laws and Technology	

H.B. 2075. Jefferson Davis Highway; renames any section of U.S. Route 1 to "Emancipation Highway." Repealing Chapter 286, 1922 Acts.	
Patrons: Cole, J.G., et al.	
Passed House	78, 379
H.B. 2078. Industrial hemp; definitions, updates laws to address the new hemp producer license issued by the U.S. Department of Agriculture. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4118, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446.	
Patrons: Marshall, et al.	
Passed House	457
Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and N	
Resources	
Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and I Resources	Natural
H.B. 2079. Pharmacists; initiation of treatment with and dispensing and administering of drugs and devices. Amending §§ 54.1-3300 and 54.1-3303.1.	
Patrons: Rasoul, et al.	
Passed House	318
Constitutional reading dispensed, referred to Committee on Education and Health Continued to 2021 Special Session I in Senate Committee on Education and Health	321
H.B. 2081. Polling places; prohibited activities, unlawful possession of a firearm, penalty. Amending §§ 24.2-604, 24.2-671, and 24.2-802.1.	
Patrons: Levine, et al.	170
Passed House Constitutional reading dispensed, referred to Committee on the Judiciary Continued to 2021 Special Session I in Senate Committee on the Judiciary H.B. 2082. Virginia Redistricting Commission; transparency in redistricting process.	172
Amending §§ 30-396 and 30-399.	
Patrons: Levine, et al.	4.55
Passed House	461
H.B. 2085. Emergency Services and Disaster Law; local and interjurisdictional emergency operations plans. Amending § 44-146.19.	
Patrons: Askew, et al.	110
Passed House	
Constitutional reading dispensed, referred to Committee on General Laws and Technology. Continued to 2021 Special Session I in Senate Committee on General Laws and Technology H.B. 2086. Child care providers; background checks portability, check shall not be required if	
individual completed a background check within previous five years, pilot program, report. Amending §§ 19.2-389, 22.1-289.035, 22.1-289.039, 63.2-1720.1, and 63.2-1724. Patron: McGuire	
Passed House	376
Constitutional reading dispensed, referred to Committee on Rehabilitation and Social Services	
Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	, 0, 317
Services	105

	2091. Covington, City of; amending charter, consolidation of school divisions of the City of Covington and Alleghany County, salaries. Amending § 22.1-32.	
	Patron: Austin	
	Passed House	76
	Constitutional reading dispensed, referred to Committee on Local Government	
	Continued to 2021 Special Session I in Senate Committee on Local Government	
	2092. Behavioral Health and Developmental Services, Department of; background	, .
	checks, persons providing contractual services. Amending §§ 19.2-389, 37.2-416, and	
	37.2-506.	
	Patrons: Willett, et al.	_
	Passed House	68
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	_
	Services	69
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	
	Services	95
	2095. Bristol, City of; amending charter, changes to powers and organization.	
	Patron: O'Quinn	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Local Government	
	Continued to 2021 Special Session I in Senate Committee on Local Government	92
	2098. Southwestern Virginia Mental Health Institute; Governor to lease a portion of	
	property to Smyth County for a term of three years, responsibility of County, corrects tax map references. Amending Chapter 678, 2019 Acts.	
	Patrons: O'Quinn, et al.	
	Passed House	15
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	1(
	Services	21
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	
	Services	94
H.R.	2099. Judgments; limitations on enforcement, judgment liens, settlement agents,	, .
	effective date. Amending §§ 8.01-251, 8.01-458, and 55.1-339.	
	Patrons: Coyner, et al.	
	Passed House	49
	Constitutional reading dispensed, referred to Committee on the Judiciary	51
	Continued to 2021 Special Session I in Senate Committee on the Judiciary	
H.B.	2101. GO Virginia Grants; matching funds, repeals sunset provision.	
	Patron: Wampler	
	Passed House	76
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 3	78
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 4	95
H.B.	. 2104. Virginia Highway Corporation Act of 1988; repeals Act, roadways to operate	
	under the Public-Private Transportation Act of 1995, implementation of distance-based	
	tolling on certain roadways, etc. Amending § 33.2-613; repealing §§ 33.2-1823 and	
	56-535 through 56-552.	
	Patrons: Reid, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Transportation	
	Continued to 2021 Special Session I in Senate Committee on Transportation	96

H.B	2. 2105. Early childhood education; quality rating and improvement system participation, School Readiness Committee reinstated. Amending § 2.2-208.1 and second and eleventh
	enactments of Chapters 860 and 861, 2020 Acts.
	Patrons: Bulova, et al. Passed House 268
	Constitutional reading dispensed, referred to Committee on Education and Health
H.B	. 2110. Pretrial data collection; Virginia Criminal Sentencing Commission to collect and
	disseminate on an annual basis, report. Amending § 2.2-3802; adding § 19.2-134.1.
	Patrons: Herring, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
H.B	2111. Maternal Health Data and Quality Measures, Task Force on; established,
	report.
	Patrons: Herring, et al.
	Passed House
H.B	2. 2113. Criminal records; establishes a process for automatic expungement, etc., definitions, penalties, report. Amending §§ 9.1-101, 17.1-293.1, 17.1-323, 17.1-413,
	17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-340, 19.2-389.3, and 19.2-390;
	adding §§ 19.2-392.5 through 19.2-392.16.
	Patrons: Herring, et al.
	Passed House 376
	Constitutional reading dispensed, referred to Committee on the Judiciary
II D	Continued to 2021 Special Session I in Senate Committee on the Judiciary
п.р	employed by a funeral service establishment shall be considered essential workers, access
	to personal protective equipment and immunization. Adding §§ 32.1-42.2 and
	44-146.17:1.1.
	Patrons: Mugler, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
H.B	2. 2117. Children's Services Act; funds expended special education programs, state pool
	funds shall be allocated for no longer than 12 months for transitional services, report.
	Amending §§ 2.2-5211 and 2.2-5212.
	Patrons: VanValkenburg, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 378, 379
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B	2. 2118. Virginia Electric Vehicle Grant Fund and Program; created, report.
	Adding § 10.1-1322.5.
	Patrons: Keam, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations
11 P	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
н.В	3. 2119. Student driver education program; parent/student component exemption.
	Amending § 22.1-205.
	Patrons: Keam, et al. Passed House

H.B. 2119 (continued) Constitutional reading dispensed, referred to Committee on Education and Health	60
Continued to 2021 Special Session I in Senate Committee on Education and Health	13
H.B. 2120. Higher educational institutions, public; governing boards, meetings, input, and disclosures. Amending §§ 23.1-409, 23.1-802, and 23.1-1303.	
Patrons: Keam, et al.	
Passed House	2 /
Constitutional reading dispensed, referred to Committee on Education and Health	
Continued to 2021 Special Session I in Senate Committee on Education and Health 49 H.B. 2121. State Corporation Commission; business entities filings, amends various	"
provisions of Va. Stock Corporation Act. Amending §§ 13.1-609, 13.1-610, 13.1-615,	
13.1-615.1, 13.1-625, 13.1-628, 13.1-630, 13.1-639, 13.1-639, 13.1-658, 13.1-661,	
13.1-710, 13.1-711, 13.1-716, 13.1-718, 13.1-721.1, 13.1-722.6, 13.1-722.12:1, 13.1-759,	
13.1-765, 13.1-775.1, 13.1-803, 13.1-806, 13.1-807, 13.1-809, 13.1-815, 13.1-815.1,	
13.1-816, 13.1-829, 13.1-830, 13.1-831, 13.1-835, 13.1-894, 13.1-897.1, 13.1-898.7,	
13.1-921, 13.1-927, 13.1-936.1, 13.1-944.7, 13.1-1002, 13.1-1004, 13.1-1005, 13.1-1012,	
13.1-1017, 13.1-1052, 13.1-1054, 13.1-1062, 13.1-1065, 13.1-1073.1, 13.1-1074,	
13.1-1075, 13.1-1080, 13.1-1087, 13.1-1096, 13.1-1099.14, 13.1-1099.26, 13.1-1201,	
13.1-1203, 13.1-1212, 13.1-1214, 13.1-1222, 13.1-1242, 13.1-1252, 13.1-1255,	
13.1-1264, 13.1-1265, 13.1-1271, 13.1-1277, 15.2-5112, 15.2-5431.9, 50-73.1, 50-73.2,	
50-73.6, 50-73.17, 50-73.54, 50-73.67, 50-73.70, 50-73.83, and 50-73.135;	
adding §§ 13.1-898.1:1, 13.1-1263.1, 15.2-5431.8:1, 15.2-5431.9:1, 15.2-5431.35:1, and	
50-73.48:5; repealing §§ 13.1-941.01 through 13.1-944.	
Patrons: Keam, et al.	. 4
Passed House	
Constitutional reading dispensed, referred to Committee on Commerce and Labor	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	14
H.B. 2123. Students; eligibility for in-state tuition and state financial assistance program.	
Amending § 23.1-506; adding § 23.1-505.1.	
Patrons: Lopez, et al.	10
Passed House 31	
Constitutional reading dispensed, referred to Committee on Education and Health	
Continued to 2021 Special Session I in Senate Committee on Education and Health	10
H.B. 2124. COVID-19; Department of Medical Assistance Services shall deem testing,	
treatment, and vaccination to be emergency services.	
Patrons: Lopez, et al. Passed House 31	10
Constitutional reading dispensed, referred to Committee on Education and Health	
Continued to 2021 Special Session I in Senate Committee on Education and Health	13
H.B. 2125. Voter registration; preregistration for persons 16 years of age or older, effective date. Amending § 24.2-404; adding § 24.2-403.1.	
Patrons: Lopez, et al.	
Passed House	18
Constitutional reading dispensed, referred to Committee on Privileges and Elections	
Continued to 2021 Special Session I in Senate Committee on Privileges and Elections 49	
H.B. 2128. Firearms; criminal history record information check delay increased to five days.	, ,
Amending § 18.2-308.2:2.	
Patrons: Lopez, et al.	
Passed House	49
Constitutional reading dispensed, referred to Committee on the Judiciary	
Continued to 2021 Special Session Lin Senate Committee on the Judiciary 49	

	2129. Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment, Enhanced Nutrient Removal Certainty Program established.
	Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14.
	Patrons: Lopez, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
пъ	Resources
	2130. Virginia LGBTQ+ Advisory Board; established, report. Adding §§ 2.2-2499.1 through 2.2-2499.4.
	Patrons: Lopez, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 153
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
	2131. Alcoholic beverage control; adds the chief administrative officer of a locality to
	the list of persons who may be sent notice of certain license applications by Board of
	Directors of the Virginia Alcoholic Beverage Control Authority. Amending §§ 4.1-230
	and 15.2-907.
	Patrons: Lopez, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
II D	Services 495
	2132. Homicides and assaults and bodily woundings; certain matters not to constitute defenses. Adding §§ 18.2-37.1 and 18.2-57.5.
	Patrons: Roem, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
	2133. Commercial sex trafficking; issuance of writ of vacatur for victims. Adding §§ 19.2-327.15 through 19.2-327.20.
	Patrons: Delaney, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
	2134. Employee classification; provision of personal protective equipment in response to
	a disaster. Amending §§ 40.1-28.7:7 and 60.2-212; adding § 65.2-301.2.
	Patron: Batten
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494
	2135. School boards, certain; participation in the Afterschool Meal Program.
	Adding § 22.1-207.4:2.
	Patrons: Roem, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session Lin Senate Committee on Education and Health

н.в.	. 2137. Paid sick leave; employers to provide to certain employees, essential workers that work an average of 20 hours weekly, etc., provisions shall not apply to any retail business with fewer than 25 employees. Adding §§ 40.1-33.3 through 40.1-33.9. Patrons: Guzman, et al.	
	Passed House	434
	Constitutional reading dispensed, referred to Committee on Commerce and Labor	
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	
HВ	2138. Identification privilege cards; authorizes DMV to issue, fee, confidentiality,	דעד
11.0	penalties. Amending §§ 24.2-411.3, 24.2-643, 46.2-203.2, 46.2-216.1, 46.2-323.01, 46.2-323.1, 46.2-346, 46.2-600.1, and 58.1-3; adding § 46.2-345.3.	
	Patrons: Guzman, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Transportation	236
	Continued to 2021 Special Session I in Senate Committee on Transportation	496
H.B	. 2139. Accrual of cause of action; diagnosis of latent injury or disease. Amending § 8.01-249. Patrons: Guzman, et al.	
	·	210
	Passed House	
	Constitutional reading dispensed, referred to Committee on the Judiciary	
н.в	Continued to 2021 Special Session I in Senate Committee on the Judiciary 2140. Alternative application for employment for persons with a disability; Department of Hyman Passayana Management to proceed a process.	493
	Department of Human Resource Management to create a process. Amending § 2.2-2901.1; adding § 2.2-1212.	
	Patrons: Guzman, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on General Laws and Technology	
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology	495
н.в	. 2146. Parked cars; authorizes vehicles to be stopped or parked in an unoccupied Department of Transportation right-of-way adjacent to a highway in case of an emergency, etc. Amending § 46.2-889.	
	Patrons: Miyares, et al.	
	Passed House	376
	Constitutional reading dispensed, referred to Committee on Transportation	
	Continued to 2021 Special Session I in Senate Committee on Transportation	
н.в	. 2147. Human Rights, Division of; renamed as Office of Civil Rights. Amending §§ 2.2-520, 2.2-522, 2.2-523, 2.2-3902, 2.2-3907, 2.2-3909, and 15.2-1604.	170
	Patrons: Herring, et al.	210
	Passed House	
	Constitutional reading dispensed, referred to Committee on General Laws and Technology	
II D	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology	493
н.в	. 2148. Small renewable energy projects; includes in the definition certain energy storage facilities and projects that include storage facility components. Amending § 10.1-1197.5.	
	Patrons: Willett, et al.	210
	Passed House	
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Nature Resources	321
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Nature Resources	ıral 101

	150. Jurisdiction over criminal cases; certification by general district court of felony
	ancillary misdemeanor charges, jurisdiction to such charges shall vest in circuit court.
	mending §§ 16.1-123.1, 16.1-241, and 17.1-513.
	tron: Adams, L.R.
	ssed House
	onstitutional reading dispensed, referred to Committee on the Judiciary
Co	ontinued to 2021 Special Session I in Senate Committee on the Judiciary
H.B. 2	154. Hospitals, nursing homes, and certified nursing facilities; regulations, policies
to	ensure the permissible access to and use of an intelligent personal assistant provided by
a j	patient while receiving inpatient services. Amending § 32.1-127.
Pa	trons: Adams, L.R., et al.
Pa	ssed House
Co	onstitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
Co	ontinued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
	Services
H.B. 2	155. Virginia Human Rights Act; nondiscrimination in employment, definitions,
	xual and workplace harassment. Amending §§ 2.2-522, 2.2-3905, 2.2-3907, and
	2-3908.
	trons: Watts, et al.
	ssed House
	onstitutional reading dispensed, referred to Committee on General Laws and
Ο.	Technology
C	ontinued to 2021 Special Session I in Senate Committee on General Laws and Technology 49
	159. Balloons; release of nonbiodegradable balloons outdoors prohibited, civil penalty.
	mending § 29.1-556.1.
	trons: Guy, et al.
	ssed House
	onstitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
C	Resources
C	ontinued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natura
C	Resources
11 D 1	160. Electric utilities; triennial review, fair rate of return, customer bill credits.
	nending § 56-585.1.
	 -
	trons: Tran, et al.
	ssed House
	onstitutional reading dispensed, referred to Committee on Commerce and Labor
	ontinued to 2021 Special Session I in Senate Committee on Commerce and Labor
	2161. Active military or a military spouse; definition of "military status,"
dı	scrimination in public accommodations, employment, and housing.
	mending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905,
	.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306,
	-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310.
	trons: Tran, et al.
	ssed House
Co	onstitutional reading dispensed, referred to Committee on General Laws and
	Technology
Co	ontinued to 2021 Special Session I in Senate Committee on General Laws and Technology 49
	162. Medical care facilities; facility shall establish protocols, definitions, designated
	pport persons for persons with disabilities. Adding § 32.1-137.08.
	trons: Tran, et al.
	ssed House 35

H.B.	2162 (continued) Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health 495
	2163. Motor Vehicles, Department of; limits the release of privileged information to
	government entities and law-enforcement agencies. Amending §§ 46.2-100, 46.2-208,
	46.2-209, 46.2-209.1, and 46.2-328.3.
	Patrons: Tran, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B.	1 1 1 1 1
	Amending § 58.1-3965.
	Patrons: Hope, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 321
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
	2166. Involuntary admission; amends provisions governing involuntary inpatient and
	mandatory outpatient treatment, effective date. Amending §§ 37.2-817, 37.2-817.1,
	37.2-817.2, and 37.2-817.4; repealing § 37.2-817.3.
	Patrons: Hope, et al.
	Passed House
	Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
	Services
HВ	2167. Parole; notice and certification, monthly reports, conditional release, discretionary
	early consideration. Amending §§ 53.1-136 and 53.1-155.
	Patrons: Scott, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
	Services
H.B.	2168. Illegal gambling; skill games, enforcement by localities and Attorney General,
	civil penalty. Amending § 18.2-325; adding § 18.2-331.1.
	Patron: Scott
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B.	2169. Prostitution; reorganizes the statute penalizing into two distinct sections.
	Amending §§ 8.01-42.4, 9.1-116.5, 9.1-902, 16.1-69.48:6, 16.1-69.55, 17.1-275.13,
	17.1-805, 18.2-46.1, 18.2-346, 18.2-346.1, 18.2-350, 18.2-357.1, 18.2-513, 19.2-10.2,
	19.2-215.1, 19.2-268.3, 19.2-386.16, 19.2-386.35, 19.2-392.02, 32.1-58, 37.2-314,
	37.2-416, and 37.2-506; adding § 18.2-346.01.
	Patrons: Mundon King, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
	2170. Virginia Small Business Financing Authority; risk-based review of outstanding
	loans. Adding § 2.2-2312.1.
	Patrons: Mundon King, et al.
	Passed House

H.B. 2	2170 (continued)	
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 1	
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 4	95
	2171. Virginia Small Business Financing Authority; utilization or award of loan and	
	grant program funds. Amending § 2.2-2312.	
	Patrons: Mundon King, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 1 Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 4	
	2172. Small, women-owned, and minority-owned businesses; right to appeal denial of	
	nitial certification. Amending § 2.2-1606.	
	Patrons: Mundon King, et al.	
F	Passed House	53
(Constitutional reading dispensed, referred to Committee on General Laws and Technology 1	53
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 4	95
H.B.	2174. VirginiaSaves Program; established, membership. Amending § 23.1-701;	
	adding §§ 2.2-2744 through 2.2-2756.	
	Patrons: Torian, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 2	
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 4	95
	2175. Homeowners and tenants of manufactured home parks; housing protections,	
	Foreclosures, enforcement of lien, cases of a deed of trust conveying owner-occupied	
	residential real estate, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and	
	55.1-1303; adding § 15.2-2223.5.	
	Patrons: Torian, et al.	<i>5 (</i>
	Passed House	30
(Technology	61
(Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 4	
н.в.		
	Amending § 22.1-291.4.	
	Patrons: Torian, et al.	
	Passed House	50
	Constitutional reading dispensed, referred to Committee on Education and Health	
	Continued to 2021 Special Session I in Senate Committee on Education and Health 4	
	2177. Capital outlay plan; repeals existing six-year capital outlay for projects to be	
f	funded. Repealing Chapter 1134, 2020 Acts.	
I	Patron: Torian	
I	Passed House	68
(Constitutional reading dispensed, referred to Committee on Finance and Appropriations $\dots 2$	69
(Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 4	95
H.B. 1	2178. Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021;	
C	created.	
	Patrons: Torian, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 172, 1	
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 4	95
	2179. Refunding bonds; alters the principal and interest requirements, maturity date, and	
	allowable discount for bonds previously issued, sunset clause. Amending first enactment	
	of Chapters 265 and 408, 1992 Acts.	
	Patron: Torian	70

H.B	. 2179 (continued) Constitutional and displayed and amount of Committee on Finance and Amazonaictions 172, 177
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 172, 173
D	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B	. 2180. Lynchburg, City of; amending charter, salaries of members of City Council.
	Patrons: Walker, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session I in Senate Committee on Local Government
H.B	. 2181. Virginia Retirement System; amendments to reflect recent changes to federal law.
	Amending §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308.
	Patrons: Mundon King, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 172, 173
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B	. 2182. Traumatic brain injury; definition.
	Patron: Wilt
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 269
	Continued to 2021 Special Session I in Senate Committee on Education and Health 495
H.B	. 2185. Retail Sales and Use Tax; exemption for personal protective equipment.
	Adding § 58.1-609.14.
	Patrons: Byron, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 445
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495
H.B	. 2186. Mathews County; board of supervisors may appoint one employee of the locality
	onto the board of directors for the Economic Development Authority of the County.
	Amending § 15.2-4904.
	Patron: Hodges
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session I in Senate Committee on Local Government
H.B	. 2187. Recurrent Flooding Resiliency, Commonwealth Center for; Center shall
	evaluate development of Flood Resiliency Clearinghouse Program for coordinating flood
	mitigation solutions, report.
	Patrons: Hodges, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
D	Resources
н.в	2190. Wrongful death; beneficiaries. Amending § 8.01-53.
	Patron: Leftwich
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 2191. Social services, local department of; location of child in local department's
	custody. Amending § 63.2-104.
	Patron: Leftwich
	Passed House

H.B.	. 2191 (continued) Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	201
	Services	
	2192. Support orders; contents of orders, change in employment status, unemployment benefits. Amending §§ 20-60.3 and 63.2-1916.	473
	Patron: Leftwich	
	Passed House	249
	Constitutional reading dispensed, referred to Committee on the Judiciary	
	2193. Settlement agreements; staying of dismissal. Adding § 8.01-425.2.	.,,
	Patrons: Leftwich, et al.	
	Passed House	377
	Constitutional reading dispensed, referred to Committee on the Judiciary	
	Continued to 2021 Special Session I in Senate Committee on the Judiciary	
	2194. Communicating threats of death or bodily injury to a person with intent to	7))
	intimidate, etc.; threats in writing, including electronically transmitted communications,	
	penalties. Amending § 18.2-60.	
	Patron: Leftwich	
	Passed House	444
	Constitutional reading dispensed, referred to Committee on the Judiciary	
	Continued to 2021 Special Session I in Senate Committee on the Judiciary	
	2197. Individuals with intellectual and developmental disabilities; Department of	473
	Medical Assistance Services to study and develop recommendations for use of virtual	
	support, etc.	
	Patrons: Runion, et al.	
	Passed House	235
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	
	Services	236
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	
	Services	495
	. 2198. Local elections for governing bodies; elections for school boards, qualification of voters. Amending §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223.	
	Patrons: Convirs-Fowler, et al.	
	Passed House	318
	Constitutional reading dispensed, referred to Committee on Privileges and Elections	321
	Continued to 2021 Special Session I in Senate Committee on Privileges and Elections	495
H.B.	. 2200. Electric utilities; procedures under which the State Corporation Commission	
	reviews the earnings and sets the rates of investor-owned incumbent electric utilities,	
	triennial review. Amending § 56-585.1.	
	Patrons: Jones, et al.	
	Passed House	456
	Constitutional reading dispensed, referred to Committee on Commerce and Labor	460
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	494
	. 2201. Solar and energy storage projects; siting agreements throughout the	
	Commonwealth. Amending §§ 15.2-2288.8 and 15.2-2316.6 through 15.2-2316.9.	
	Patrons: Jones, et al.	
	Passed House	249
	Constitutional reading dispensed, referred to Committee on Local Government	250
	- ·	105

	2202. Elevator mechanic or accessibility mechanic, certain; exemption from
	certification. Amending § 54.1-1141. Patrons: Jones, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and
H.B.	Technology
	Patrons: Filler-Corn, et al.
	Passed House
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural Resources
H.B.	2204. Get Skilled, Get a Job, Give Back (G3) Fund and Program; established.
	Adding § 23.1-2911.2.
	Patrons: Filler-Corn, et al.
	Passed House 269 Constitutional reading dispensed, referred to Committee on Education and Health 269 Continued to 2021 Special Session I in Senate Committee on Education and Health 495
	2206. Child Care Subsidy Program; temporarily expanding Program to provide
	financial assistance for child care to families in need during public health emergency. Patrons: Filler-Corn, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	death or disability of firefighters, emergency medical services personnel, law-enforcement officers, and correctional officers is an occupational disease. Amending § 65.2-402.1.
	Patrons: Jones, et al.
	Passed House
	2208. Harry F. Byrd, Sr., statue; Department of General Services to remove statue of former Virginia Governor and U.S. Senator from Capitol Square.
	Patrons: Jones, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rules
H.B.	
	Amending § 2.2-2649. Patrons: Plum, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
пъ	Services
	2213. Gold; Secretary of Natural Resources, et al., shall establish a work group to study mining and processing, issuance of certain mining permits.
	Patrons: Guzman, et al. 457 Passed House 457
	1 ubbou 110ubo

	2213 (continued) Constitutional reading dispensed, referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	2216. Virginia Missing Child with Autism Alert Program; renames the Virginia
	Missing Person with Autism Alert Program, Amending §§ 52-34.13, 52-34.14, and
	52-34.15.
	Patrons: Hayes, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B.	
	Amending § 15.2-1809.
	Patron: Hodges
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	÷ *
	Continued to 2021 Special Session I in Senate Committee on Local Government
	2218. Pharmaceutical processors; permits processors to produce and distribute cannabis products, testing standards for botanical cannabis and products. Amending §§ 18.2-250.1,
	54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8.
	Patrons: Hayes, et al.
	• •
	Constitutional reading dispensed, referred to Committee on Education and Health 378, 379
	Continued to 2021 Special Session I in Senate Committee on Education and Health
	2219. Pharmacies; freedom of choice by covered individual. Amending §§ 38.2-3407.7,
	38.2-4209.1, and 38.2-4312.1.
	Patron: Hodges
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
	2220. Surgical technologist; certification, use of title. Amending § 54.1-2956.12.
	Patron: Hayes
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
	2221. Environmental permits; community and environmental justice outreach, new
	individual permit application. Amending § 10.1-1408.1; adding §§ 10.1-1321.2,
	10.1-1437.1, 62.1-44.6:1, and 62.1-266.1.
	Patrons: Hayes, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
	Resources
	2222. Military medical personnel program; adds any facility that offers medical
	services to the public and that is supervised by one or more physicians or podiatrists to the
	list of entities that may participate in the program. Amending § 2.2-2001.4.
	Patrons: Hayes, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and
	Technology
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 493

	2223. Treasury and State Treasurer, Department of the; surety bonds. Amending §§ 2.2-2809, 5.1-1.3, 10.1-2006, 21-163, 30-131, 33.2-205, 36-111, 42.1-16, 44-21, 46.2-202, 52-3, 53.1-11, 54.1-305, 58.1-201, and 60.2-109.
	Patron: Hayes
	Passed House
	Constitutional reading dispensed, referred to Committee on Finance and Appropriations 172, 173
	Continued to 2021 Special Session I in Senate Committee on Finance and Appropriations 495 2227. Uniform Statewide Building Code; Board of Housing and Community
	Development shall consider amendments to Code to address changes in the IECC relating
	to energy efficiency and conservation.
	Patrons: Kory, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and
	Technology
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.R.	2229. Virginia Residential Landlord and Tenant Act; responsibilities of real estate
	brokers, foreclosure of single-family residential dwelling units. Amending §§ 54.1-2108.1
	and 55.1-1237.
	Patron: Simon
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 291
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B.	2230. Supported decision-making agreements; Department of Behavioral Health and
	Developmental Services to develop and implement a program to educate certain
	individuals. Amending §§ 64.2-2000, 64.2-2003, and 64.2-2007; adding § 37.2-314.3.
	Patrons: Bell, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
	2233. Orders of restitution; docketed on behalf of victim, assignment of judgment,
	enforcement. Amending §§ 19.2-305.1, 19.2-305.2, 19.2-349, and 19.2-354.
	Patrons: Bell, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
	2234. Victims of sex trafficking; clarifies definition, affirmative defense to prosecution
	for certain offenses. Adding § 18.2-361.1.
	Patrons: Brewer, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B.	2236. Behavioral health docket; standards for transfer of supervision between local
	community-based probation agencies. Amending §§ 9.1-175 and 18.2-254.3.
	Patron: Bell
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
	2238. Licensed private schools for students with disabilities; accreditation.
	Amending § 22.1-323.
	Patron: Kory
	Passed House 319

H.B	. 2238 (continued) Constitutional reading dispensed, referred to Committee on Education and Health
II D	Continued to 2021 Special Session I in Senate Committee on Education and Health
н.в	. 2249. Virginia Residential Landlord and Tenant Act; landlord charges for security deposits, insurance premiums for damage insurance, etc., clerk of SCC shall provide a
	downloadable form for the filing of a resident agent appointment, etc.
	Amending §§ 17.1-275, 55.1-1200, 55.1-1204, 55.1-1206, 55.1-1208, 55.1-1211,
	55.1-1226, 64.2-2008, and 64.2-2012.
	Patrons: McQuinn, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and
	Technology
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
II D	•
п.р	. 2250. Humane Cosmetics Act; prohibits a cosmetics manufacturer from testing cosmetics on animals in the Commonwealth, civil penalties. Adding §§ 59.1-571 through
	59.1-574.
	Patrons: Kory, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor 494
H.B	. 2252. Tazewell County; authorizes a quitclaim and release of interest and conveyance of
	an easement by Board of Wildlife Resources.
	Patrons: Wampler, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 2254. Unlawful dissemination of unsolicited obscene image of self to another; penalty.
	Adding § 18.2-386.3.
	Patrons: Convirs-Fowler, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 2257. Hampton Roads Sanitation District; changes to the enabling act related to the
	addition of Northampton and Accomack Counties. Amending Chapter 66, 1960 Acts.
	Patron: Bloxom
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session I in Senate Committee on Local Government
H.B	. 2258. Substantial Risk Order Registry; Department of State Police shall keep and
	maintain a computerized Registry, etc. Amending § 19.2-387.3.
	Patrons: Simonds, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	. 2261. License plates, special; removes fee for issuance to a member of the Virginia
	National Guard. Amending § 46.2-744.
	Patrons: Wyatt, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation 378, 379
	Continued to 2021 Special Session Lin Senate Committee on Transportation 496

H.B. 2262. Bicycles; permits operators to treat a stop sign as a yield sign in certain situatio Amending §§ 46.2-839 and 46.2-905; adding § 46.2-903.1.	ns.
Patrons: Hurst, et al.	
Passed House	351
Constitutional reading dispensed, referred to Committee on Transportation	
Continued to 2021 Special Session I in Senate Committee on Transportation	
H.B. 2263. Death penalty; abolishes current penalty, including for those persons curren under a death sentence, repeals provisions relating to trial of capital cas Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.0 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-205, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 19.2-264.5.	es. 10, 01, 57, 11, .2, 04,
Patrons: Mullin, et al.	
Passed House	
Constitutional reading dispensed, referred to Committee on the Judiciary	
Continued to 2021 Special Session I in Senate Committee on the Judiciary	
H.B. 2266. Alcoholic beverage control; outdoor refreshment area license, fees for state a local licenses. Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, 4.1-233 and 4.1-308.	
Patrons: Ayala, et al.	
Passed House	378
Constitutional reading dispensed, referred to Committee on Rehabilitation and Social	
Services	378, 379
Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social	
Services	
H.B. 2269. Solar energy projects and energy storage systems; revenue share for projects a systems. Amending § 58.1-2636.	nd
Patrons: Heretick, et al.	210
Passed House	
Constitutional reading dispensed, referred to Committee on Commerce and Labor	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor H.B. 2271. Universal health care; Joint Commission on Health Care shall enter into a contra with a qualified entity to study options for financing.	
Patrons: Samirah, et al.	
Passed House	260
Constitutional reading dispensed, referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.B. 2273. Data centers; sales and use tax exemption, criteria, report. Amending § 58.1-609.	
Patrons: Morefield, et al.	٥.
Passed House	435
Constitutional reading dispensed, referred to Committee on Finance and Appropriations	
Continued to 2021 Special Session Lin Senate Committee on Finance and Appropriation	

H.B	2. 2275. Food and drink permit; inspections required to operate food establishment, Commissioner of Agriculture and Consumer Services to issue permit for the operation of any food manufacturer, food storage warehouse, etc. Amending §§ 3.2-5100, 3.2-5101, 3.2-5130, and 15.2-2288.6.
	Patron: Gooditis
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
	Resources
H.B	2. 2276. Plastic firearms and unfinished frames, etc.; manufacture, import, sale, transfer,
	etc., penalty. Amending § 18.2-308.5; adding § 18.2-308.5:2.
	Patrons: Simon, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary
H.B	2. 2282. State Corporation Commission; transportation electrification, utility recovery of
	certain costs, report.
	Patrons: Sullivan, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Commerce and Labor
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor
H.B	2284. Driving privileges, certain; Commissioner of DMV to reinstate privileges, and waive fees for individuals whose privileges were suspended for failure to pay court fines
	and costs in other jurisdictions.
	Patrons: Williams Graves, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B	2. 2287. Economic development authorities; size of board in Powhatan County, reduces
	quorum requirement. Amending § 15.2-4904.
	Patron: Ware
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session I in Senate Committee on Local Government
H.B	2288. Virginia Public Procurement Act; bids or offers on construction contracts,
	requirement to submit list of subcontractors, effective clause. Adding § 2.2-4303.02.
	Patrons: Williams Graves, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 442
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	2. 2290. Larceny; repeals punishment for conviction of second or subsequent misdemeanor.
	Repealing § 18.2-104.
	Patron: Plum
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session Lin Senate Committee on the Judiciary 405

H.B	. 2291. Law-enforcement civilian oversight bodies; adds a sheriff's office to those law-enforcement agencies that may be overseen by a body created by a locality and adds a nonprobationary deputy sheriff to those officers who are subject to such body, disciplinary
	determinations. Amending § 9.1-601.
	Patrons: Williams Graves, et al.
	Passed House
H.B	. 2293. Local gas severance tax; extends sunset date. Amending § 58.1-3713.
	Patron: Morefield
	Passed House
H.B	. 2294. Vehicle's odometer; disclosure exemption. Amending § 46.2-629.
	Patron: Cole, J.G.
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
ИR	Continued to 2021 Special Session I in Senate Committee on Transportation
11.D	owned or leased by the Commonwealth, etc., penalty. Adding § 18.2-283.2.
	Patrons: Levine, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.B	. 2298. Muzzleloading rifle and shotgun; clarifies definitions. Amending § 29.1-100.
	Patron: Fariss
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
H.B	. 2299. Special education; Department of Education to provide training and guidance
	documents to local school divisions on the development of Individualized Education Programs (IEPs) for children with disabilities, etc. Adding § 22.1-214.4.
	Patrons: Carr, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health 378, 379. Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B	. 2300. Hospitals; emergency treatment for substance use-related emergencies,
	Department of Health, et al., to develop recommendations for best practices for treatment
	and discharging of patients in emergency departments, etc. Amending § 32.1-127.
	Patrons: Delaney, et al.
	Passed House
н.в	. 2302. Farmers market food and beverage products; sales considered essential during
	state of emergency.
	Patrons: Wyatt, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natural
	Resources
	Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natural
	Resources

H.B. 2304. Phase I or Phase II electric utilities; provision of broadband capacity, State Corporation Commission shall condition any approval of a petition on the requirement	
7 11 1	
that construction shall commence within 18 months of such approval.	
Amending § 56-585.1:9.	
Patrons: Tyler, et al.	200
Passed House	
Constitutional reading dispensed, referred to Committee on Commerce and Labor	
Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	494
H.B. 2305. Governor's Schools; Board of Education shall issue guidance on the governance of	
academic year, including admissions policies, and guidelines on diversity, etc.	
Patron: Tyler	
Passed House	
Constitutional reading dispensed, referred to Committee on Education and Health	352
Continued to 2021 Special Session I in Senate Committee on Education and Health	495
H.B. 2307. Consumer Data Protection Act; definitions, establishes a framework for	
controlling and processing personal data, applies to all persons that conduct business in	
the Commonwealth, etc. Adding §§ 59.1-571 through 59.1-581.	
Patrons: Hayes, et al.	
Passed House	290
Constitutional reading dispensed, referred to Committee on General Laws and Technology	
Continued to 2021 Special Session I in Senate Committee on General Laws and Technology	
H.B. 2308. Veterans of Foreign Wars, American Legion, etc.; quantity of land certain	7/3
associations may hold. Amending § 57-20.	
Patron: Brewer	
	157
Passed House	
Constitutional reading dispensed, referred to Committee on General Laws and Technology	
Continued to 2021 Special Session I in Senate Committee on General Laws and Technology	495
H.B. 2310. Concealed handgun permits; demonstration of competence, eligibility to apply	
for permit due to restrictions of COVID-19.	
Patrons: Runion, et al.	
Passed House	378
Constitutional reading dispensed, referred to Committee on the Judiciary	379
Continued to 2021 Special Session I in Senate Committee on the Judiciary	495
H.B. 2311. Objects of antiquity; unlawful to remove from battlefield, penalty.	
Amending §§ 10.1-2300 and 10.1-2306.	
Patrons: Runion, et al.	
Passed House	444
Constitutional reading dispensed, referred to Committee on Agriculture, Conservation and Natur	al
Resources	
Continued to 2021 Special Session I in Senate Committee on Agriculture, Conservation and Natu	
Resources	
H.B. 2312. Marijuana; legalization of simple possession, etc., expungement of criminal	., .
records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818,	
2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906,	
3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-101, 4.1-101.01, 4.1-101.02,	
4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106,	
4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122,	
4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2,	
4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1,	
4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303,	
4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336,	
4 1-337 4 1-338 4 1-348 4 1-349 4 1-350 4 1-351 4 1-352 4 1-353 4 1-354 5 1-13	

	2312 (continued)
	9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1,
	16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276,
	17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248,
	18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2,
	18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258,
	18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03,
	18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460,
	18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1,
	19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389,
	19.2-390, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08,
	23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3,
	65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through
	4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9,
	19.2-392.10, 19.2-392.11, 19.2-392.12, 19.2-392.13, and 46.2-341.20:7;
	repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3.
	Patrons: Herring, et al.
	5 .
	Passed House
	Constitutional reading dispensed, referred to Committee on Rehabilitation and Social
	Services
	Continued to 2021 Special Session I in Senate Committee on Rehabilitation and Social
	Services
	2313. Freedom of Information Act; adds Chief Workforce Development Advisor and
	Director of Diversity, Equity, and Inclusion to definition of "Office of the Governor" for
	purposes of excluding the working papers and correspondence of the Office of the
	Governor from mandatory disclosure. Amending § 2.2-3705.7.
	Patron: McQuinn
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 461
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
	2314. Special education; Board of Education to amend a certain regulation to remove the
	word "component" following the word "evaluation," thereby ensuring compliance with
	the relevant federal regulation.
	Patron: Mugler
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
H.B.	2316. Students with disabilities; Department of Education to update its special education
	and related services, etc.
	Patron: Mundon King
	Passed House
	Constitutional reading dispensed, referred to Committee on Education and Health
	Continued to 2021 Special Session I in Senate Committee on Education and Health
	2317. Sexual and Domestic Violence, Advisory Committee on; increases membership,
	duties. Amending §§ 9.1-116.2 and 9.1-116.3.
	Patron: Robinson
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
	Continued to 2021 Special Session I in Senate Committee on the Judiciary 495

H.B	. 2318. Test driving vehicles; residence districts, civil penalty. Adding § 46.2-1533.1.
	Patron: Roem
	Passed House
	Constitutional reading dispensed, referred to Committee on Transportation
	Continued to 2021 Special Session I in Senate Committee on Transportation
H.B	. 2319. Firearms; access to any individual who resides or cohabitates in a shared dwelling,
	civil penalty. Adding § 18.2-308.2:02.
	Patrons: Guzman, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on the Judiciary
H.B	. 2320. Virginia Residential Property Disclosure Act; required disclosures for buyer to
	exercise due diligence, flood risk report, effective date and clause. Amending § 55.1-703;
	adding § 55.1-708.2.
	Patrons: Convirs-Fowler, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 46
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	. 2321. Labor, Secretary of; position created in Governor's Cabinet, removes position of
	Chief Workforce Development Advisor. Amending §§ 2.2-200, 2.2-204, 2.2-205,
	2.2-205.1, 2.2-435.6, 2.2-435.8, 2.2-435.9, 2.2-435.10, 2.2-2471, 2.2-2471.1, 2.2-2472,
	2.2-2472.2, 2.2-2472.3, and 30-377; adding §§ 2.2-214.2 and 2.2-214.3;
	repealing § 2.2-435.7.
	Patrons: Simonds, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 46
n	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 495
H.B	. 2322. Opioid Abatement Authority; established, Opioid Abatement Fund created,
	investment of assets of Fund, report. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365
	through 2.2-2376, and 51.1-124.40. Patrons: Herring, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on General Laws and Technology 442
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology 49:
нR	2323. Library aid; former regional library systems that receive state aid notwithstanding
11.D	a loss of regional library status due to a city reversion, shall receive such aid regardless.
	Amending § 15.2-4116.
	Patron: Edmunds
	Passed House
	Constitutional reading dispensed, referred to Committee on Local Government
	Continued to 2021 Special Session I in Senate Committee on Local Government
H R	2324. Virginia Redistricting Commission; removing a citizen commissioner of the
11,15	Commission for neglect of duty or gross misconduct by other commissioners in a public
	meeting. Amending § 30-392.
	Patrons: Watts, et al.
	Passed House
	Constitutional reading dispensed, referred to Committee on Privileges and Elections
	Continued to 2021 Special Session I in Senate Committee on Privileges and Elections 495
H.B	. 2326. Child care; expands to include all cities the authority related to the regulation of
	services. Amending § 15.2-914.
	Patrons: Williams Graves, et al.
	Passed House

H.B	2. 2326 (continued)	
	Constitutional reading dispensed, referred to Committee on Local Government	
	Continued to 2021 Special Session I in Senate Committee on Local Government	495
H.B	2. 2327. Prevailing wage rate; clarifies that public works includes transportation	
	infrastructure projects. Amending § 2.2-4321.3.	
	Patrons: Krizek, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on General Laws and Technology .	
	Continued to 2021 Special Session I in Senate Committee on General Laws and Technology	495
H.B	2330. Percentage of Income Payment Program and Fund; Department of Housing and	
	Community Development and the Department of Social Services to adopt rules or	
	establish guidelines for the adoption, etc., of Program and Fund. Amending §§ 56-576 and	
	56-585.6.	
	Patron: Kory	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Commerce and Labor	
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	494
H.B	2. 2331. Mandatory minimum sentences; elimination, modification of sentence to	
	mandatory minimum term of confinement for felony offenses. Amending §§ 3.2-4212,	
	4.1-302, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5,	
	18.2-255, 18.2-255.2, 46.2-357, and 46.2-391; adding § 19.2-303.03.	
	Patrons: Mullin, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on the Judiciary	
	Continued to 2021 Special Session I in Senate Committee on the Judiciary	495
H.B	2332. Commonwealth Health Reinsurance Program; established, report.	
	Amending §§ 38.2-4214 and 38.2-4319; adding §§ 38.2-6600 through 38.2-6607.	
	Patrons: Sickles, et al.	
	Passed House	
	Constitutional reading dispensed, referred to Committee on Commerce and Labor	
	Continued to 2021 Special Session I in Senate Committee on Commerce and Labor	494
H.B	2. 2333. COVID-19; Department of Health shall establish a process whereby any health	
	care provider may administer the vaccine, etc., health care providers may also be	
	employees of localities, pharmacies, or hospitals, localities with paramedics, licensed	
	practical nurse, etc., may establish and staff clinics to administer.	
	Patrons: Bagby, et al.	225
	Passed House	
	Constitutional reading dispensed, referred to Committee on Education and Health	
	Reported with substitute	
	Read second time	
	Reading of substitute waived	
	Committee substitute agreed to.	
	Reading of amendment waived	
	Passed by temporarily	
	Reading of amendment waived.	
	Amendments Nos. 1 and 2 by Senator Dunnavant agreed to	
	Engrossed	
	Passed Senate	
	Reconsideration of vote on Senate passage	
	Amendment No. 2 by Senator Dunnavant reconsidered	
	Amendment No. 2 by Senator Dunnavant rejected	
	Reading of amendment waived	54/

H.B. 2333 (continued) Amendment No. 3 by Senator Dunnavant agreed to
Engrossed
Passed Senate
Senate substitute with amendments agreed to by House
Signed by President
Approved by Governor-Chapter 1 (effective 2/15/21)
H.B. 2337. Income tax, state; subtraction for low-income military veterans with a permanent
service-connected disability. Amending § 58.1-322.02.
Patrons: Wiley, et al.
Passed House
Constitutional reading dispensed, referred to Committee on Finance and Appropriations 445
Continued to 2021 Session in Special Session I Committee on Finance and Appropriations 495
H.J.R. 512. Mann, William; commending.
Patrons: Batten, et al.
Agreed to by House
Laid on Clerk's Desk
Agreed to by Senate
H.J.R. 517. Carson, Lue Ethel Lapsley; recording sorrow upon death.
Patrons: Kilgore, et al.
Agreed to by House
Laid on Clerk's Desk
Agreed to by Senate
H.J.R. 518. Cridlin, Nathaniel; recording sorrow upon death.
Patrons: Kilgore, et al.
Agreed to by House
Laid on Clerk's Desk
Agreed to by Senate
H.J.R. 519. Barger, Frederick, Jr.; recording sorrow upon death.
Patrons: Kilgore, et al.
Agreed to by House
Laid on Clerk's Desk
Agreed to by Senate
H.J.R. 520. Osborne, Phillip G.; recording sorrow upon death.
Patrons: Kilgore, et al.
Agreed to by House
Laid on Clerk's Desk
Agreed to by Senate
H.J.R. 521. Crater Planning District Commission; commemorating its 50th anniversary.
Patron: Tyler
Agreed to by House95
Laid on Clerk's Desk
Agreed to by Senate
H.J.R. 522. Staffing levels, employment conditions, and compensation at the Virginia
Department of Corrections, joint committee of various House and Senate
Committees Studying; continued, appropriations.
Patrons: Tyler, et al.
Agreed to by House
Referred to Committee on Rules

H.J.R. 525. Johns, Barbara Rose; Joint Committee of Congress on the Library requested to	
fill the Commonwealth's vacant spot in the National Statuary Hall Collection at the United	
States Capitol with a statue to commemorate.	
Patrons: Ward, et al.	
Agreed to by House	. 235
Referred to Committee on Rules	. 237
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 526. Comprehensive campaign finance reform; joint subcommittee to study.	
Patrons: Bulova, et al.	
Agreed to by House	. 291
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 527. Invasive plant species; Department of Conservation and Recreation, et al., to	
study the sale and use of species.	
Patrons: Bulova, et al.	
Agreed to by House	. 235
Referred to Committee on Rules	. 237
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 532. McLean Volunteer Fire Department; commemorating its 100th anniversary.	
Patrons: Sullivan, et al.	
Agreed to by House	95
Laid on Clerk's Desk	
Agreed to by Senate	. 133
H.J.R. 533. Keating, George Markey; recording sorrow upon death.	
Patron: Sullivan	
Agreed to by House	95
Laid on Clerk's Desk	
Agreed to by Senate	. 129
H.J.R. 534. Richards, Lilla Dunovant McCutchen; recording sorrow upon death.	
Patrons: Sullivan, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Agreed to by Senate	. 129
H.J.R. 535. Stripe, Mark; commending.	
Patrons: Sullivan, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Agreed to by Senate	. 133
H.J.R. 537. Racism ; General Assembly to recognize as a public health crisis.	
Patrons: Aird, et al.	227
Agreed to by House	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 538. Access to water; General Assembly to recognize that access to clean, potable, and	
affordable water is a necessary human right.	
Patrons: Aird, et al.	227
Agreed to by House	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 540. Tamberrino, Frank; commending.	
Patrons: Runion, et al. Agreed to by House	04
Agreed to by House	

Н.,	J.R. 540 (continued)
	Laid on Clerk's Desk
	Agreed to by Senate
н.,	J.R. 542. Transit equity and modernization; Department of Rail and Public Transportation
	to study.
	Patrons: McQuinn, et al. Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
п.	I.R. 544. Clark, Peter; commending.
п.,	Patrons: Simon, et al.
	Agreed to by House95
	Laid on Clerk's Desk
	Agreed to by Senate
н	J.R. 547. Norfolk and Western Railway Section Foreman's House; commending.
11.0	Patron: Morefield
	Agreed to by House95
	Laid on Clerk's Desk
	Agreed to by Senate
н.	I.R. 549. COVID-19; Joint Legislative Audit and Review Commission to study the impact
11.0	on Virginia's public schools, students, and school employees, meetings shall be completed
	by November 30, 2022.
	Patrons: Guy, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
Н.,	J.R. 553. Flanagan, William Joseph, Jr.; recording sorrow upon death.
	Patrons: Cox, et al.
	Agreed to by House95
	Laid on Clerk's Desk
	Agreed to by Senate
Н.,	J.R. 554. Stigall, Kent; commending.
	Patron: Cox
	Agreed to by House
	Laid on Clerk's Desk96
	Agreed to by Senate
Н.,	J.R. 555. Constitutional amendment; felon disenfranchisement, automatic restoration of
	political rights (first reference). Amending Section 1 of Article II.
	Patrons: Herring, et al.
	Agreed to by House
	Referred to Committee on Privileges and Elections
	Continued to 2021 Special Session I in Senate Committee on Privileges and Elections 495
Н.,	J.R. 562. International Overdose Awareness Day; designating as August 31, 2021, and
	each succeeding year thereafter.
	Patrons: Miyares, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
Н.,	I.R. 563. Corporate income tax; Division of Legislative Services, et al., to establish a work
	group to assess the feasibility of transitioning to a unitary combined reporting system.
	Patrons: Watts, et al.
	Agreed to by House

H.J	.R. 563 (continued) Referred to Committee on Rules
	Continued to 2021 Session in Special Session I Committee on Rules
H.J	.R. 565. Chafin, Augustus Benton, Jr.; recording sorrow upon death.
	Patrons: Kilgore, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
H.J	.R. 567. Income tax, state; Joint Legislative Audit and Review Commission to study
	increasing the progressivity of Virginia's system, meetings shall be completed by
	November 30, 2022.
	Patrons: Watts, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 570. Harrell, Jill; commending.
	Patron: VanValkenburg
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
H.J	.R. 571. Zimm, Alan; recording sorrow upon death.
	Patrons: VanValkenburg, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
H.J	.R. 572. Wolf Creek Cherokee Tribe of Virginia; General Assembly to extend state
	recognition.
	Patrons: McQuinn, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 575. General Assembly; providing for a Joint Assembly, establishing a schedule for the
	conduct of business for 2021 Regular Session and providing for legislative continuity
	between 2021 Regular Session and a special session.
	Patron: Herring
	Agreed to by House
	Referred to Committee on Rules
	Rules suspended
	Committee discharged
	Reading waived
	Taken up for immediate consideration
	Read third time
	Reading of substitute waived
	Substitute by Senator Surovell agreed to
	Reading of amendment waived
	Amendment by Senator Locke agreed to
	Engrossed
	Rejected to by Senate
	Reconsideration of vote by which resolution was rejected
	Amendment by Senator Locke reconsidered
	Amendment by Senator Locke withdrawn
	Substitute by Senator Surovell reconsidered
	Substitute by Senator Surovell withdrawn

H.J.	R. 575 (continued)
	Reading of amendment waived
	Amendment by Senator Locke agreed to
	Engrossed
	Agreed to by Senate
	Senate amendment agreed to by House
H.J.	R. 576. General Assembly; establishing a prefiling schedule for the 2022 Regular
	Session.
	Patron: Herring
	Agreed to by House
	Referred to Committee on Rules
	Rules suspended
	Committee discharged
	Readings waived
	Taken up for immediate consideration
	Agreed to by Senate
	R. 578. Criminal justice, behavioral health, and other human services records;
	Department of Behavioral Health and Developmental Services to establish a work group
	to study the feasibility of developing a secure, de-identified, etc., database.
	Patrons: Price, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J.	R. 579. Speaker of the House of Delegates; confirming appointment to Virginia
	Commonwealth University Health System Authority Board of Directors.
	Patron: Filler-Corn
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J.	R. 580. Lake Ridge Occoquan Coles Civic Association; commemorating its 50th
	anniversary.
	Patron: Ayala
	Agreed to by House95
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 581. Stoltzfus, Karl David, Sr.; recording sorrow upon death.
	Patrons: Runion, et al.
	Agreed to by House95
	Laid on Clerk's Desk
	Agreed to by Senate
H.J.	R. 582. Constitutional amendment; fundamental right to marry, removes same-sex
	marriage prohibition (first reference). Amending Section 15-A of Article I.
	Patrons: Sickles, et al.
	Agreed to by House
	Referred to Committee on Privileges and Elections
	Continued to 2021 Session in Special Session I Committee on Privileges and Elections 495
H.J.	R. 583. Liberty Amendments Month; designating as June 19 through the third Monday
	in July 2021 and each succeeding year thereafter.
	Patrons: Bagby, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules

H.J.	R. 584. Eagle, Edwin; commending.
	Patrons: Avoli, et al.
	Agreed to by House
	Laid on Clerk's Desk96
	Agreed to by Senate
H.J.	R. 585. Reid, Harold; recording sorrow upon death.
	Patrons: Avoli, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
H.J.	R. 586. Hamilton, Alphonso; recording sorrow upon death.
	Patrons: Avoli, et al.
	Agreed to by House95
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 587. Ewing, Sim; commending.
	Patrons: Kilgore, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 588. Weeks, Dave Andrew, Jr.; recording sorrow upon death.
	Patrons: Walker, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 589. BaCote, Mamye E.; recording sorrow upon death.
	Patrons: Ward, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 590. Goodwin, Stephen Teel; recording sorrow upon death.
	Patrons: Freitas, et al. Agreed to by House
	Agreed to by House
	Agreed to by Senate
	R. 591. Roberts, Billy Joe; recording sorrow upon death. Patrons: Mugler, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 592. Noll, Charles F., Sr.; recording sorrow upon death.
	Patrons: Batten, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 593. Warrick, William Burford; recording sorrow upon death.
	Patrons: Batten, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate

	R. 594. Pierce's Pitt Bar-B-Que; commemorating its 50th anniversary.
	Patrons: Batten, et al.
1	Agreed to by House
I	Laid on Clerk's Desk
	Agreed to by Senate
	R. 595. Melton, Charlotte; recording sorrow upon death.
	Patrons: Bagby, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 596. Brain Aneurysm Awareness Month; designating as September 2021 and each
	succeeding year thereafter.
	Patrons: Gooditis, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 597. Virginia's Crossroads; commending.
	Patrons: Tyler, et al. Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 598. Richmond Raceway; commemorating its 75th anniversary.
	Patrons: Bagby, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
	R. 599. American Promise and Take Back Our Republic, Virginia Chapter of;
	commending.
]	Patron: Mullin
1	Agreed to by House
]	Laid on Clerk's Desk
	Agreed to by Senate
H.J.F	R. 600. Rode, Lisa; commending.
	Patrons: Watts, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
H.J.F	R. 601. Wagner, Marvin Howard; recording sorrow upon death.
H.J.F	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al.
H.J.F	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F 1 2	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F 1 1 1	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F 1 1 1 1 H.J.F	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F 1 1 1 H.J.F	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F]]] H.J.F]	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F 1 1 4 H.J.F 1	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F. 1	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House 171 Laid on Clerk's Desk 173 Agreed to by Senate 262 R. 602. Alfred, Tiffany J.; commending. Patron: Hodges Agreed to by House 171 Laid on Clerk's Desk 173 Agreed to by Senate 264
H.J.F.	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House 171 Laid on Clerk's Desk 262 R. 602. Alfred, Tiffany J.; commending. Patron: Hodges Agreed to by House 171 Laid on Clerk's Desk 173 Agreed to by House 171 Laid on Clerk's Desk 264 R. 603. Christchurch School; commemorating its 100th anniversary.
H.J.F 1 2 H.J.F 1 4 H.J.F	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House
H.J.F. 1 1 1 1 1 1 1 1 1 1 1 1 1	R. 601. Wagner, Marvin Howard; recording sorrow upon death. Patrons: Watts, et al. Agreed to by House 171 Laid on Clerk's Desk 262 R. 602. Alfred, Tiffany J.; commending. Patron: Hodges Agreed to by House 171 Laid on Clerk's Desk 173 Agreed to by House 171 Laid on Clerk's Desk 264 R. 603. Christchurch School; commemorating its 100th anniversary.

H.J.R. 604. General Assembly; confirming appointment to Joint Legislative Audit and	
Review Commission.	
Patron: Plum	
Agreed to by House	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	496
each succeeding year thereafter.	
Patrons: McQuinn, et al.	
Agreed to by House	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 606. Equal Citizens Month; designating as November 2021 and each succeeding year	
thereafter.	
Patrons: Convirs-Fowler, et al.	
Agreed to by House	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 608. Sykes, Robert; commending.	
Patron: Tyler	
Agreed to by House	171
Laid on Clerk's Desk	173
Agreed to by Senate	264
H.J.R. 609. Morris, Dennis K.; commending.	
Patrons: Aird, et al.	
Agreed to by House	171
Laid on Clerk's Desk	173
Agreed to by Senate	264
H.J.R. 610. Pillow, Richard D.; commending.	
Patrons: Aird, et al.	
Agreed to by House	171
Laid on Clerk's Desk	173
Agreed to by Senate	264
H.J.R. 611. Andrew's Grill; commending.	
Patrons: Aird, et al.	
Agreed to by House	171
Laid on Clerk's Desk	173
Agreed to by Senate	264
H.J.R. 612. Thomas, Beulah E.; commending.	
Patrons: Aird, et al.	
Agreed to by House	171
Laid on Clerk's Desk	173
Agreed to by Senate	264
H.J.R. 613. Cherry, Leroy A.; recording sorrow upon death.	
Patrons: Aird, et al.	
Agreed to by House	171
Laid on Clerk's Desk	
Agreed to by Senate	
H.J.R. 615. Williamsburg Veterans of Foreign Wars Post 4639; commemorating its 75th	
anniversary.	
Patrons: Batten, et al.	
Agreed to by House	319

H.	J.R. 615 (continued) Laid on Clerk's Desk 3	21
тт	Agreed to by Senate	:20
п.	Patron: Plum	
	Agreed to by House	10
	•	
	Laid on Clerk's Desk	
	Agreed to by Senate	-28
н.	J.R. 619. Little Oil Company, Inc.; commemorating its 100th anniversary. Patron: Willett	
	Agreed to by House	10
	Laid on Clerk's Desk	
	Agreed to by Senate	-28
н.	J.R. 620. Wolf Trap Foundation for the Performing Arts; commending.	
	Patron: Kory Agreed to by House 3	10
	•	
	Laid on Clerk's Desk	
	Agreed to by Senate	-28
н.	J.R. 621. Colonial Behavioral Health; commending.	
	Patrons: Batten, et al. Agreed to by House	10
	· ·	
	Laid on Clerk's Desk	
	Agreed to by Senate	-28
н.	J.R. 622. Northern Virginia Association of REALTORS®; commemorating its 100th anniversary.	
	•	
	Patrons: Bulova, et al. Agreed to by House	10
	Laid on Clerk's Desk	
	Agreed to by Senate	
тт	•	:20
н.	J.R. 623. Mannion, Robert E.; recording sorrow upon death.	
	Patron: Bulova Agreed to by House	10
	•	
	Laid on Clerk's Desk	
	Agreed to by Senate	:22
н.	J.R. 624. Gutshall, Erik; recording sorrow upon death.	
	Patrons: Hope, et al.	20
	Agreed to by House	
	Laid on Clerk's Desk	
	Agreed to by Senate	-22
н.	J.R. 625. Banks, George H.; recording sorrow upon death.	
	Patron: Jones	
	Agreed to by House	
	Laid on Clerk's Desk	
	Agreed to by Senate	-22
Н.	J.R. 627. Moreau, Alfred Ernest; recording sorrow upon death.	
	Patrons: Hope, et al.	
	Agreed to by House	
	Laid on Clerk's Desk	
	Agreed to by Senate 4	122

I.J.R. 628. Judges; election in circuit court, general district court, juvenile and domestic relations district court, members of the Judicial Inquiry and Review Commission,	
members of the State Corporation Commission, and the Auditor of Public Accounts. Patron: Sullivan	
Agreed to by House	2
Reading waived, referred to Committee on the Judiciary	
Rules suspended	
Committee discharged	
Readings waived	
Taken up for immediate consideration	
Agreed to by Senate	
I.J.R. 629. Virginia Criminal Sentencing Commission; confirms the appointment by the	•
Chief Justice of the Supreme Court of Chairman of Commission.	
Patron: Herring	
Agreed to by House	5
Referred to Committee on the Judiciary	
Continued to 2021 Special Session I in Senate Committee on the Judiciary	
I.J.R. 630. White, Clenise and Alex; commending.	J
Patron: VanValkenburg	
Agreed to by House	'n
Laid on Clerk's Desk	
Agreed to by Senate	
H.J.R. 631. Cohen, Bernard S.; recording sorrow upon death.	·O
Patrons: Herring, et al.	
Agreed to by House	20
Laid on Clerk's Desk	
Agreed to by Senate	,3
Patrons: Herring, et al.	
Agreed to by House	00
Laid on Clerk's Desk	
Agreed to by Senate	
Agreed to by Senate	.0
Patron: Rush	
Agreed to by House	20
Laid on Clerk's Desk	
Agreed to by Senate	
	.∠
H.J.R. 634. Western Fairfax Christian Ministries; commending.	
Patrons: Delaney, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Agreed to by Senate	.8
I.J.R. 635. Campblin, Karen; commending.	
Patrons: Delaney, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Agreed to by Senate	8
H.J.R. 636. Aquatech; commending.	
Patrons: Delaney, et al.	
Agreed to by House	
Laid on Clerk's Desk	
0 (TEMPO 10 DV NOTO)	ч

H.J	.R. 637. Mussig, Chris and Jeanne; commending.
	Patrons: Delaney, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
H.J	.R. 638. Homework Helpers Organization; commending.
	Patrons: Delaney, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Agreed to by Senate
H.J	.R. 639. Kelleher, Linda Y.; commending.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 640. Van Doren, Nancy; commending.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 641. Bickford, Charlene; commending.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 642. Bisson, Ann; recording sorrow upon death.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 643. Renfro, Nancy Todd; recording sorrow upon death.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 644. Ferguson, Earl Mark; recording sorrow upon death.
	Patrons: VanValkenburg, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
TT T	Continued to 2021 Special Session I in Special Session I Committee on Rules
H.J	.R. 645. Colyer, Bob, Sr.; recording sorrow upon death.
	Patrons: Kilgore, et al. Agreed to by House
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Special Session I Committee on Rules
	Commune to 2021 opecial ocssion i in opecial ocssion i Commune on Rules 490

H.J.R. 646. Anderson, Dolson Barnett; recording sorrow upon death.	
Patrons: Carr, et al.	
Agreed to by House	435
Laid on Clerk's Desk	442
Referred to Committee on Rules	491
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 647. Dabney, Herbert Allen, III; recording sorrow upon death.	
Patrons: Carr, et al.	
Agreed to by House	436
Laid on Clerk's Desk	442
Referred to Committee on Rules	491
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 648. Andrews, Edward; recording sorrow upon death.	
Patrons: Carr, et al.	
Agreed to by House	436
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 649. Slaughter, Alexander Hoke; recording sorrow upon death.	
Patrons: Carr. et al.	
Agreed to by House	436
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 650. Jones, Thomas Overton; recording sorrow upon death.	
Patrons: Carr, et al.	
Agreed to by House	436
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 651. Hooker, Katherine Bridgforth; recording sorrow upon death.	·····
Patrons: Carr, et al.	
Agreed to by House	436
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 652. Garner, Charles Lincoln; recording sorrow upon death.	490
Patrons: Carr. et al.	
Agreed to by House	124
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
•	490
H.J.R. 653. Denny, Alfred Jerome; recording sorrow upon death.	
Patrons: Hope, et al. Agreed to by House	124
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 654. Gosling, Arthur Warrington; recording sorrow upon death.	
Patrons: Hope, et al. Agreed to by House	42.4
Agreed to by House	
Laid on Clerk's Desk	44,

H.J.I	R. 654 (continued)
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 655. Parham Doctors' Hospital; commending.
	Patrons: VanValkenburg, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 656. Farr, Murray Jay; commending.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 657. Cole, James B.; commending.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 658. Johnson, Cintia; commending.
	Patrons: Hope, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 659. Silber, Randall R.; commending.
	Patrons: VanValkenburg, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 660. Avula, Danny TK; commending.
	Patrons: VanValkenburg, et al. Agreed to by House
	· ·
	Laid on Clerk's Desk
	Continued to 2021 Special Session I in Senate Committee on Rules
	1
	R. 661. Henrico County Registrar's Office; commending.
	Patrons: VanValkenburg, et al. Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 662. Tarantino, Joseph Maurice; recording sorrow upon death.
	Patrons: Robinson, et al.
	Agreed to by House
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	Continued to 2021 Special Session I in Schate Confinitite on Rules

H.J.R. 663. Johnstone, Richard G., Jr.; commending.	
Patron: VanValkenburg	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rul	es 496
H.J.R. 664. Yoffy, James S.; commending.	
Patrons: VanValkenburg, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rul	es
H.J.R. 665. Meriwether, Bobbie Jean; commending.	
Patrons: Cole, M.L., et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rul	
H.J.R. 666. Fredericksburg, City of; commending.	
Patrons: Cole, M.L., et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rul	
H.J.R. 667. Fredericksburg Dog Mart; commending.	
Patrons: Cole, M.L., et al.	
Agreed to by House	436
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rul	
H.J.R. 668. Robertson, Franklin Delano; recording sorrow upon dea	
Patron: Morefield	••••
Agreed to by House	436
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rul	
H.J.R. 669. Shelor, Isabel Gallimore; recording sorrow upon death.	
Patrons: Rush, et al.	
Agreed to by House	436
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rul	
H.J.R. 670. Midgett, Robert Preston, II; recording sorrow upon dea	
Patrons: Knight, et al.	ui.
Agreed to by House	136
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rule	
H.J.R. 671. Cardounel, Humberto, Jr.; commending.	490
Patrons: VanValkenburg, et al.	
Agreed to by House	124
Laid on Clerk's Desk	
Daire Off Civin & Dook	

H.J	.R. 671 (continued)
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 672. Warren, Darrell W., Jr.; commending.
	Patrons: Hodges, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 673. Anderson, John Hutchison; commending.
	Patron: Webert
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 674. Harris, Gregory Garfield; commending.
	Patron: Webert
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 675. Virginia Academy of Science, Engineering and Medicine; commending.
	Patrons: Hayes, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
** *	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 676. T.C. Williams School of Law at the University of Richmond; commemorating
	its 150th anniversary.
	Patrons: Adams, D.M., et al. Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
и т	R. 677. New River Valley Public Health Task Force; commending.
п.ј	Patron: Hurst
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules 492
	Continued to 2021 Special Session I in Senate Committee on Rules
ні	R. 678. Ledgerton, Elizabeth Ann Kerr; recording sorrow upon death.
11.0	Patron: Webert
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
н.і	R. 679. Hodge, Perry Anthony; recording sorrow upon death.
	Patron: Hurst
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	*

H.J.R. 680. Munford, Joan Hardie; recording sorrow upon death.	
Patron: Hurst	
Agreed to by House	437
Laid on Clerk's Desk	442
Referred to Committee on Rules	492
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 681. Allen, Samuel; commending.	
Patrons: Hayes, et al.	
Agreed to by House	437
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 682. Hill, Bobby; commending.	490
Patrons: Convirs-Fowler, et al.	125
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 683. Loudoun County Office of Elections and Voter Registration; commending.	
Patrons: Subramanyam, et al.	
Agreed to by House	437
Laid on Clerk's Desk	443
Referred to Committee on Rules	492
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 684. Grandy, Ethel L.; commending.	
Patrons: Hayes, et al.	
Agreed to by House	437
Laid on Clerk's Desk	
Referred to Committee on Rules	492
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 685. Bland, Larry Jerome; recording sorrow upon death.	170
Patrons: Carr, et al.	
Agreed to by House	137
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
*	490
H.J.R. 686. McDowell, Charles Lindsay; recording sorrow upon death.	
Patrons: Carr, et al.	125
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 687. Mandt, Sheila Kavanagh; recording sorrow upon death.	
Patrons: Carr, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 688. Minor, Mozelle Willis; recording sorrow upon death.	
Patrons: Carr, et al.	
Agreed to by House	437
Laid on Clerk's Desk	443

H.J.Ŗ	. 688 (continued)
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	. 689. Hill, Jamile J.; recording sorrow upon death.
	Patrons: Hayes, et al.
	Agreed to by House
_	aid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J.R	. 690. Royall, William Archer, Jr.; recording sorrow upon death.
	Patrons: Carr, et al.
A	Agreed to by House
	aid on Clerk's Desk
F	Referred to Committee on Rules
(Continued to 2021 Special Session I in Senate Committee on Rules
H.J.R	. 691. Carey, James Alfred; recording sorrow upon death.
F	Patron: Tyler
A	Agreed to by House
I	aid on Clerk's Desk
F	Referred to Committee on Rules
(Continued to 2021 Special Session I in Senate Committee on Rules
	. 692. Sweat, Josh; commending.
	Patrons: Hayes, et al.
	Agreed to by House
	aid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	a. 693. Miller, Marty L.; commending.
	Patrons: Hayes, et al.
	Agreed to by House
	aid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	a. 694. Warren, Adrienne ; commending.
	Patrons: Hayes, et al.
	Agreed to by House
	agired to by House 437 aid on Clerk's Desk 443
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	. 695. Downing, Annie Laura; commending.
	Patrons: Wyatt, et al.
	Agreed to by House
	aid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	. 696. Crutcher, Ronald A.; commending.
	Patrons: VanValkenburg, et al.
	Agreed to by House
	aid on Clerk's Desk
	Referred to Committee on Rules
(Continued to 2021 Special Session I in Senate Committee on Rules

H.J.R. 697. Lawrence, Barry R.; commending.	
Patrons: VanValkenburg, et al.	
Agreed to by House	437
Laid on Clerk's Desk	443
Referred to Committee on Rules	492
Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.R. 698. Singleton, O.R., Jr.; commending.	
Patron: VanValkenburg	
Agreed to by House	437
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 699. Rapisarda, Joseph P., Jr.; commending.	
Patrons: VanValkenburg, et al.	
Agreed to by House	438
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 700. Foster, Tim; commending.	., .
Patrons: VanValkenburg, et al.	
Agreed to by House	438
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 701. Dukas, Helen; recording sorrow upon death.	.,,
Patrons: Keam, et al.	
Agreed to by House	438
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 702. Vaughan, George B.; recording sorrow upon death.	170
Patrons: Bell, et al.	
Agreed to by House	438
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 703. Byers, William Walker, Jr.; recording sorrow upon death.	170
Patrons: Hudson, et al.	
Agreed to by House	438
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 704. Lewis, Ida Johnson; recording sorrow upon death.	170
Patrons: Hudson, et al.	
Agreed to by House	438
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 705. Wright, James William, Sr.; recording sorrow upon death.	T
Patrons: Hudson, et al.	
Agreed to by House	439
	1/13

	5 (continued)
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules
	6. Merchant, John F.; recording sorrow upon death.
	ns: Hudson, et al.
_	ed to by House
	on Clerk's Desk
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules
	7. Teague, Joseph William, Sr.; recording sorrow upon death.
	ns: Hudson, et al. ed to by House
	on Clerk's Desk
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules
	3. Coleman, Lovell Louis, Sr.; recording sorrow upon death.
	n: Hudson
_	ed to by House
	on Clerk's Desk
	nued to 2021 Special Session I in Senate Committee on Rules
	•
	O. Connock, Stuart Wallace; recording sorrow upon death. as: Hudson, et al.
	als: Hudson, et al. 2d to by House
_	on Clerk's Desk
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules
	D. Plotnick, Virginia Leonard; recording sorrow upon death.
	ns: Hudson, et al.
	ed to by House
_	on Clerk's Desk
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules
	. Hogg, Oakley W., III; recording sorrow upon death.
	ns: Wyatt, et al.
	ed to by House
_	on Clerk's Desk
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules
	2. Virginia State Police; commending.
	ns: Filler-Corn, et al.
	ed to by House
_	on Clerk's Desk
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules
	B. Virginia National Guard; commending.
	ns: Filler-Corn, et al.
	ed to by House
	on Clerk's Desk
	red to Committee on Rules
	nued to 2021 Special Session I in Senate Committee on Rules

	R. 714. Broderick, John R.; commending.	
	Patrons: Filler-Corn, et al.	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	496
	R. 715. Braxton-Williams, Valerie; commending.	
	Patrons: Bagby, et al.	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	496
	R. 716. Francis, Thomas Henry; recording sorrow upon death.	
	Patrons: Bagby, et al.	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	496
	R. 717. Center for Excellence in Education; commending.	
	Patrons: Keam, et al.	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	496
	R. 718. Asian Pacific American Bar Association of the Greater Washington, D.C.,	
	Area, Inc.; commemorating its 40th anniversary.	
	Patrons: Keam, et al.	
	Agreed to by House	438
	Laid on Clerk's Desk	443
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.	R. 719. Szakos, Joe; commending.	
	Patrons: Hudson, et al.	
	Agreed to by House	438
	Laid on Clerk's Desk	443
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.	R. 720. Swegle, Madeline; commending.	
	Patrons: Filler-Corn, et al.	
	Agreed to by House	438
	Laid on Clerk's Desk	443
	Referred to Committee on Rules	493
	Continued to 2021 Special Session I in Senate Committee on Rules	496
H.J.	R. 721. Fairfax County Park Authority; commending.	
	Patrons: Filler-Corn, et al.	
	Agreed to by House	438
	Laid on Clerk's Desk	443
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	496
	R. 722. Kang, Ae Ja; recording sorrow upon death.	
	Patrons: Keam, et al.	
	Agreed to by House	438

H.J.	R. 722 (continued) Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 723. Walker, Arthur William; recording sorrow upon death.
	Patrons: Hudson, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 724. Colden, Tony E., Jr.; recording sorrow upon death.
	Patrons: Bourne, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 725. Edwards, Connie Weldon; recording sorrow upon death.
	Patrons: McQuinn, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 726. Holmes, Bobby Eugene; recording sorrow upon death.
	Patrons: McQuinn, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 727. Brown, Algenon L.; recording sorrow upon death.
	*
	Patrons: McQuinn, et al. Agreed to by House
	Laid on Clerk's Desk
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 728. Logan, Ena Ampy; recording sorrow upon death.
	Patrons: McQuinn, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 729. Andrews, Willie Makently; recording sorrow upon death.
	Patrons: McQuinn, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
	R. 730. Keys-Chavis, Edna Elizabeth; recording sorrow upon death.
	Patrons: McQuinn, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules

H.J.R. 731. Ford, Rudolph E., Jr.; recording sorrow u	pon death.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con-	
H.J.R. 732. Bland, Larry Jerome; recording sorrow u	pon death.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con-	nmittee on Rules
H.J.R. 733. Woodson, Willie; recording sorrow upon d	leath.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con-	nmittee on Rules
H.J.R. 734. Wood, Thelonius Leander; recording sorr	ow upon death.
Patrons: McQuinn, et al.	
Agreed to by House	439
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con	nmittee on Rules496
H.J.R. 735. Womack, William Henry, Jr.; recording s	orrow upon death.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con	nmittee on Rules496
H.J.R. 736. Bigger, Josephine Johnson; recording sort	row upon death.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con	nmittee on Rules496
H.J.R. 737. Francis, Thomas Henry; recording sorrow	v upon death.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con	
H.J.R. 738. Smith, Javier J.; recording sorrow upon de	eath.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Con	
H.J.R. 739. Johnson, Carolyn Louise; recording sorro	w upon death.
Patrons: McQuinn, et al.	
Agreed to by House	
Laid on Clerk's Desk	

H.J.	R. 739 (continued)	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	196
H.J.	R. 740. Washington, Andrew D.; recording sorrow upon death.	
	Patrons: Krizek, et al.	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	196
H.J.	R. 741. Tuell, Mary Louise; recording sorrow upon death.	
	Patron: Torian	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	1 96
H.J.	R. 742. Brown, Jean Smith; recording sorrow upon death.	
	Patron: LaRock	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	196
H.J.	R. 743. Dick, Paul Wendell; recording sorrow upon death.	
	Patrons: LaRock, et al.	
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	196
H.J.	R. 744. The Chatham Garden Club; commemorating its 100th anniversary.	
	Patrons: Adams, L.R., et al.	420
	Agreed to by House	
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
TT T	Continued to 2021 Special Session I in Senate Committee on Rules	190
H.J.	R. 745. Luck, Joann Grant; commending.	
	Patrons: Adams, L.R., et al. Agreed to by House	120
	•	
	Laid on Clerk's Desk	
	Continued to 2021 Special Session I in Senate Committee on Rules	
TT T	R. 746. Feasenmyer, William S., Jr.; commending.	+90
п.ј.	Patrons: Cox, et al.	
	Agreed to by House	120
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
	Continued to 2021 Special Session I in Senate Committee on Rules	
тт т	*	+90
п.J.	R. 747. Pleasant View, Inc.; commemorating its 50th anniversary. Patron: Wilt	
	Agreed to by House	120
	Laid on Clerk's Desk	
	Referred to Committee on Rules	
		496

II I D. 749 Saint Jasanh Cathalia Sahaali sammamanatina ita 145th anniversari	
H.J.R. 748. Saint Joseph Catholic School; commemorating its 145th anniversary.	
Patron: Aird Agreed to by House	120
•	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 490
H.J.R. 749. Newport News Public Library; commending.	
Patrons: Mullin, et al.	426
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 750. The Virginian-Pilot; commending.	
Patron: Mullin	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 751. Williamsburg Health Foundation; commending.	
Patrons: Mullin, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 752. Ortiz, Patty; commending.	
Patron: LaRock	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 753. Gaylord, Katie; commending.	
Patrons: Mullin, et al.	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 754. Pebler, Steven; commending.	
Patron: LaRock	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 755. Williamsburg Faith in Action and the Arc of Greater Williamsburg;	
commending.	
Patron: Mullin	
Agreed to by House	
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	. 496
H.J.R. 756. Guido, Tammy Williams; commending.	
Patron: Mullin	
Agreed to by House	. 44(

H.J	.R. 756 (continued) Laid on Clerk's Desk
	Referred to Committee on Rules 49:
	Continued to 2021 Special Session I in Senate Committee on Rules
и 1	.R. 757. Genakos, George S.; recording sorrow upon death.
п.ј	Patron: Mullin
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules 49:
	Continued to 2021 Special Session I in Senate Committee on Rules
H 1	I.R. 758. "Stir Crazy in Williamsburg"; commending the authors.
11.5	Patron: Mullin
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules 49:
	Continued to 2021 Special Session I in Senate Committee on Rules
н.і	S.R. 759. Unger, John; commending.
	Patron: LaRock
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 760. Linaburg, Dennis; commending.
	Patrons: LaRock, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	R. 761. Rosenberry, Brian; commending.
	Patron: LaRock
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 762. Clifton Presbyterian Church; commemorating its 150th anniversary.
	Patron: Helmer
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	I.R. 763. Korean Central Presbyterian Church; commending.
	Patrons: Helmer, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 764. Northern Virginia Therapeutic Riding Program; commemorating its 40th
	anniversary.
	Patrons: Helmer, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session Lin Senate Committee on Rules 490

H.J.R. 765. Suerdieck, Rebecca and Julia Oxrieder; commending.	
Patron: Mullin	
Agreed to by House)
Laid on Clerk's Desk	;
Referred to Committee on Rules	ļ
Continued to 2021 Special Session I in Senate Committee on Rules	í
H.J.R. 766. Zaleski, Lucille Minchin; recording sorrow upon death.	
Patron: Jones	
Agreed to by House)
Laid on Clerk's Desk	;
Referred to Committee on Rules	ļ
Continued to 2021 Special Session I in Senate Committee on Rules	í
H.J.R. 767. Keatts, Flournoy A.; recording sorrow upon death.	
Patrons: Marshall, et al.	
Agreed to by House)
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 768. Thorpe, Avicia Beatrice Hooper; recording sorrow upon death.	
Patrons: Marshall, et al.	
Agreed to by House)
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 769. Hairston, Joe Louis; recording sorrow upon death.	
Patrons: Marshall, et al.	
Agreed to by House)
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 770. Wingfield, Robert George; recording sorrow upon death.	
Patrons: McNamara, et al.	
Agreed to by House)
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 771. Farlow, Mary Elene Williams; recording sorrow upon death.	
Patrons: Marshall, et al.	
Agreed to by House)
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 772. Woods, Michael Wayne; recording sorrow upon death.	,
Patrons: Marshall, et al.	
Agreed to by House)
Laid on Clerk's Desk	
Referred to Committee on Rules	
Continued to 2021 Special Session I in Senate Committee on Rules	
H.J.R. 773. Plaster, Grover Harold; recording sorrow upon death.	
Patrons: Marshall, et al.	
Agreed to by House)
Laid on Clerk's Desk	

H.J	.R. 773 (continued)
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 774. Danville Life Saving and First Aid Crew, Inc.; commemorating its 75th
	anniversary.
	Patrons: Marshall, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	R. 775. Banister, Daniel; commending.
	Patron: Jones
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules
** *	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 776. Luther, T. David; commending.
	Patrons: Marshall, et al. Agreed to by House
	•
	Laid on Clerk's Desk
	Referred to Committee on Rules
** *	Continued to 2021 Special Session I in Senate Committee on Rules
H.J	.R. 777. Shanks, Fred O., III; commending.
	Patrons: Marshall, et al. Agreed to by House
	Laid on Clerk's Desk 44
	Referred to Committee on Rules 494
	Continued to 2021 Special Session I in Senate Committee on Rules
и т	.R. 778. Barber, Doug; commending.
п.ј	Patrons: Marshall, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules 494
	Continued to 2021 Special Session I in Senate Committee on Rules
ні	.R. 779. Hardy Petroleum; commemorating its 100th anniversary.
11.5	Patrons: Marshall, et al.
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules 494
	Continued to 2021 Special Session I in Senate Committee on Rules
ні	.R. 780. Brown, Lewis R.; commending.
11.0	Patron: Guzman
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules 494
	Continued to 2021 Special Session I in Senate Committee on Rules
нл	.R. 781. Brown, Shamira A.; commending.
-1.0	Patron: Guzman
	Agreed to by House
	Laid on Clerk's Desk
	Referred to Committee on Rules 494
	Continued to 2021 Special Session Lin Senate Committee on Rules 496

H.J.R. 782. Cameron, Frederick; recording sorrow upon death.
Patrons: Tran, et al.
Agreed to by House
Laid on Clerk's Desk
Referred to Committee on Rules
Continued to 2021 Special Session I in Senate Committee on Rules
H.J.R. 783. Zion Baptist Church; commemorating its 125th anniversary.
Patrons: Price, et al.
Agreed to by House
Laid on Clerk's Desk
Referred to Committee on Rules
Continued to 2021 Special Session I in Senate Committee on Rules
H.J.R. 784. Cross, Oretha P.; commending.
Patrons: Price, et al.
Agreed to by House
Laid on Clerk's Desk
Referred to Committee on Rules
Continued to 2021 Special Session I in Senate Committee on Rules
H.J.R. 785. Community Free Clinic of Newport News; commemorating its 10th anniversary.
Patrons: Price, et al.
Agreed to by House
Laid on Clerk's Desk
Referred to Committee on Rules
Continued to 2021 Special Session I in Senate Committee on Rules
H.J.R. 786. C. Waldo Scott Center for H.O.P.E.; commemorating its 30th anniversary.
Patrons: Price, et al.
Agreed to by House
Laid on Clerk's Desk
Referred to Committee on Rules
Continued to 2021 Special Session I in Senate Committee on Rules
H.J.R. 787. Newsome House Museum and Cultural Center; commemorating its 30th
anniversary.
Patrons: Price, et al.
Agreed to by House
Laid on Clerk's Desk
Referred to Committee on Rules
Continued to 2021 Special Session I in Senate Committee on Rules

SUBJECT INDEX

ABORTION

Health insurance; essential benefits, removes the prohibition on the provisions of coverage for abortions. Amending § 38.2-3451. (Patron–Hudson, HB 1896; McClellan, SB 1276)

ABSENTEE BALLOTS

- Absentee voting; accessibility for voters with a visual impairment or print disability. Amending § 24.2-704; adding § 24.2-103.2. (Patron–Reeves, SB 1331)
- Absentee voting; ballots to be sorted and results to be reported by precinct. Amending § 24.2-712. (Patron–Suetterlein, SB 1153)
- Absentee voting; establishment of drop-off locations for return of absentee ballots, ballot defects, cure process. Amending §§ 24.2-706, 24.2-707, 24.2-709, 24.2-709.1, and 24.2-710; adding § 24.2-707.1. (Patron–Deeds, SB 1245)
- Absentee voting; mandatory processing of returned absentee ballots before election day, central absentee voter precinct in the office of the general registrar. Amending §§ 24.2-709.1 and 24.2-712. (Patron–Deeds, SB 1246)
- Absentee voting; procedural and process reforms, availability and accessibility reforms, penalty. Amending §§ 24.2-101, 24.2-416.1, 24.2-603, 24.2-653.1, 24.2-704, 24.2-705, 24.2-706, 24.2-707, 24.2-708 through 24.2-711, and 24.2-712; adding §§ 24.2-103.2, 24.2-667.1, and 24.2-707.1. (Patron–VanValkenburg, HB 1888)
- Absentee voting; third-party absentee ballot assembly and distribution. Amending § 24.2-706. (Patron–Bell, SB 1239)
- Absentee voting; witness signature not required. Amending §§ 24.2-702.1, 24.2-704, 24.2-706, and 24.2-707. (Patron–Favola, SB 1097)

ACCIDENT AND SICKNESS INSURANCE

Health insurance; association health plan for real estate salespersons, policy for group accident and sickness insurance. Amending § 38.2-3521.1. (Patron–Barker, SB 1341)

ACCOMACK COUNTY

Hampton Roads Sanitation District; changes to the enabling act related to the addition of Northampton and Accomack Counties. Amending Chapter 66, 1960 Acts. (Patron–Bloxom, HB 2257)

ADMINISTRATION OF GOVERNMENT

- Active military or a military spouse; definition of "military status," discrimination in public accommodations, employment, and housing. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310. (Patron–Tran, HB 2161)
- Active military or a military spouse; prohibits discrimination in public accommodations, employment, and housing, landlords may not charge a higher security deposit. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1226, and 55.1-1310. (Patron–Bell, SB 1410)
- Alternative application for employment for persons with a disability; Department of Human Resource Management to create a process. Amending § 2.2-2901.1; adding § 2.2-1212. (Patron–Guzman, HB 2140)
- Annual safety and disaster awareness training; Department of Human Resource Management, et al., to develop an online training module addressing safety and disaster awareness. Amending § 2.2-1201; adding § 2.2-1212. (Patron–Ayala, HB 1891)
- Budget bill; appropriations for 2020-2022 biennium. Amending Chapter 56, 2020 Sp. I Acts. (Patron-Torian, HB 1800; Howell, SB 1100)
- Capital outlay plan; projects to be funded from general fund-supported resources, repeals existing six-year capital outlay for projects to be funded. Repealing Chapter 1134, 2020 Acts. (Patron-Howell, SB 1155)

ADMINISTRATION OF GOVERNMENT (continued)

- Capital outlay plan; repeals existing six-year capital outlay for projects to be funded. Repealing Chapter 1134, 2020 Acts. (Patron-Torian, HB 2177)
- Carbon Sequestration Task Force; established by Secretary of Natural Resources, jointly with the Secretary of Agriculture and Consumer Services, to convene a task force, report. (Patron–Lewis, SB 1374)
- Children's Services Act; effective monitoring and implementation. Amending § 2.2-2649. (Patron-Plum, HB 2212)
- Children's Services Act; funds expended special education programs, definition of "transitional services." Amending §§ 2.2-5211 and 2.2-5212. (Patron–Mason, SB 1313)
- Children's Services Act; funds expended special education programs, state pool funds shall be allocated for no longer than 12 months for transitional services, report. Amending §§ 2.2-5211 and 2.2-5212. (Patron–VanValkenburg, HB 2117)
- Commonwealth's Development Opportunity Fund; grants from Fund, waiver or reduction of capital investment and local match requirements. Amending § 2.2-115. (Patron–McPike, SB 1418)
- Conflict of Interests Act, State and Local Government; disclosure requirements, industrial development and economic development authorities. Amending § 2.2-3115. (Patron–Norment, SB 1171)
- Data Governance and Analytics, Office of; created, establishes the advisory Virginia Data Commission, membership, report. Amending § 2.2-203.2:4; adding §§ 2.2-2558 through 2.2-2564. (Patron–Barker, SB 1365)
- Early childhood education; quality rating and improvement system participation, School Readiness Committee reinstated. Amending § 2.2-208.1 and second and eleventh enactments of Chapters 860 and 861, 2020 Acts. (Patron–Bulova, HB 2105)
- Education and Labor Market Alignment, Office of; established. Amending § 2.2-2238. (Patron-Hashmi, SB 1314)
- Freedom of Information Act; adds Chief Workforce Development Advisor and Director of Diversity, Equity, and Inclusion to definition of "Office of the Governor" for purposes of excluding the working papers and correspondence of the Office of the Governor from mandatory disclosure. Amending § 2.2-3705.7. (Patron–McQuinn, HB 2313)
- Government Data Collection and Dissemination Practices Act; exemptions, email addresses of licensed professionals. Amending § 2.2-3802. (Patron–Newman, SB 1349)
- Government Data Collection and Dissemination Practices Act; license plate readers shall not be used to collect or maintain personal information, etc., definitions. Amending §§ 2.2-3800, 2.2-3801, 2.2-3802, and 52-48. (Patron–Petersen, SB 1198)
- Henrietta Lacks Commission; extends sunset provision. Amending § 2.2-2543. (Patron-Stanley, SB 1414)
- Human Rights, Division of; renamed as Office of Civil Rights. Amending §§ 2.2-520, 2.2-522, 2.2-523, 2.2-3902, 2.2-3907, 2.2-3909, and 15.2-1604. (Patron–Herring, HB 2147)
- Identity Management Standards Advisory Council; transfers management of Council. Amending §§ 2.2-436 and 2.2-437. (Patron–Boysko, SB 1458)
- Interagency Environmental Justice Working Group; established as an advisory council in executive branch of state government to further environmental justice. Amending § 2.2-234; adding §§ 2.2-236, 2.2-237, and 15.2-2223.5. (Patron–Simonds, HB 2074)
- Interagency Environmental Justice Working Group; established, definitions, report, sunset date. Amending § 2.2-234; adding § 2.2-236. (Patron–Hashmi, SB 1318)
- Labor, Secretary of; position created in Governor's Cabinet, removes position of Chief Workforce Development Advisor. Amending §§ 2.2-200, 2.2-204, 2.2-205, 2.2-205.1, 2.2-435.6, 2.2-435.8, 2.2-435.9, 2.2-435.10, 2.2-2471, 2.2-2471.1, 2.2-2472, 2.2-2472.2, 2.2-2472.3, and 30-377; adding §§ 2.2-214.2 and 2.2-214.3; repealing § 2.2-435.7. (Patron–Simonds, HB 2321)
- Legislative Staff Development Fund; created, effective date. Amending § 2.2-424; adding § 30-19.21. (Patron–Reid, HB 1978)
- Marijuana; legalization of simple possession, etc., expungement of criminal records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115,

ADMINISTRATION OF GOVERNMENT (continued)

4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, $4.1-329,\ 4.1-336,\ 4.1-337,\ 4.1-338,\ 4.1-348,\ 4.1-349,\ 4.1-350,\ 4.1-351,\ 4.1-352,\ 4.1-353,$ 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, $18.2 - 252, \ 18.2 - 254, \ 18.2 - 255, \ 18.2 - 255.1, \ 18.2 - 255.2, \ 18.2 - 258, \ 18.2 - 258.02, \ 18.2 - 258.1,$ 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-390, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9, 19.2-392.11, 19.2-392.12, 19.2-392.13, 19.2-392.10, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Herring, HB 2312)

Marijuana; unlawful possession in certain cases, using or consuming while in a motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, $54.1-3442.6, \quad 54.1-3442.8, \quad 58.1-3, \quad 59.1-148.3, \quad 65.2-107, \quad 65.2-402, \quad \text{and} \quad 65.2-402.1; \\ \text{adding } \S\S\ 2.2-2499.1 \quad \text{through} \quad 2.2-2499.4, \quad 3.2-4117.1, \quad 3.2-4117.2, \quad 3.2-4122, \quad 3.2-5145.6$ through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Ebbin, SB 1406)

Military medical personnel program; adds any facility that offers medical services to the public and that is supervised by one or more physicians or podiatrists to the list of entities that may participate in the program. Amending § 2.2-2001.4. (Patron–Hayes, HB 2222)

Military Spouse Liaison; position created in Department of Veterans Services, report, effective clause. Adding § 2.2-2002.2. (Patron–Kiggans, SB 1150)

Mines, Minerals and Energy, Department of; renamed the Department of Energy. Amending §§ 2.2-204, 2.2-604.2, 2.2-1157, 2.2-1176.1, 2.2-3705.6, 2.2-4006, 10.1-606.3, 10.1-659, 10.1-1194, 10.1-1329, 10.1-1330, 10.1-1406.2, 11-34.3, 15.2-958.3, 15.2-980, 15.2-2224, 23.1-2626, 23.1-2627, 28.2-1208, 30-275, 33.2-236, 45.1-161.1, 45.1-161.2, 45.1-161.5, 45.1-161.15, 45.1-161.179, 45.1-161.292:2, 45.1-180, 45.1-229, 45.1-230, 45.1-270.4:1, 45.1-361.28, 45.1-361.41, 45.1-383, 45.1-390, 56-265.15:1, 56-576, 56-585.5, 56-594.3, 56-596.2, 58.1-439.2, 58.1-439.12:02, 58.1-3600, 58.1-3706, 58.1-3745,

ADMINISTRATION OF GOVERNMENT (continued)

- 62.1-44.15:21, 62.1-44.15:66, 62.1-195.1, 62.1-243, 62.1-256, 62.1-259, 63.2-805, 67-200, 67-202.1, 67-602, 67-900, 67-1000, 67-1206, 67-1208, 67-1209, 67-1403, and 67-1506; adding § 45.1-161.4:1. (Patron–Sullivan, HB 1855)
- Motor Vehicles, Department of, and Supreme Court of Virginia; repeals reporting requirement. (Patron-Newman, SB 1277)
- Natural Resources, Secretary of; changes name to the Secretary of Natural and Cultural Resources. Amending §§ 2.2-200, 2.2-205, 2.2-206, 2.2-215, 2.2-220.1, 2.2-220.2, 2.2-220.3, 2.2-1156, 2.2-1176, 2.2-2316, 2.2-2338, 2.2-2481, 2.2-2699.10, 2.2-2699.11, 2.2-2699.13, 10.1-603.25, 10.1-704, 10.1-1018, 10.1-1181.15, 10.1-1188, 10.1-1329, 10.1-1402.03, 10.1-1402.04, 10.1-1405, 10.1-2129, 10.1-2202.3, 15.2-2295.1, 28.2-207, 29.1-102, 29.1-573, 29.1-579, 30-377, 56-596.2, 58.1-344.3, 62.1-44.15:68, 62.1-44.34:25, 62.1-44.34:28, 62.1-44.117, 62.1-44.118, 62.1-44.119:1, 62.1-69.31, 62.1-69.32, 62.1-69.33, 62.1-69.41, 62.1-69.52, and 62.1-195.1. (Patron-Plum, HB 1836)
- Opioid Abatement Authority; established, Fund created, report, membership. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Barker, SB 1469)
- Opioid Abatement Authority; established, Opioid Abatement Fund created, investment of assets of Fund, report. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Herring, HB 2322)
- Prescription drugs; price transparency, definitions. Amending § 2.2-3705.6; adding §§ 32.1-23.3, 38.2-3407.15:6, 38.2-3407.22, 54.1-3436.1, and 54.1-3442.02. (Patron–Sickles, HB 2007)
- Pretrial data collection; Virginia Criminal Sentencing Commission to collect and disseminate on an annual basis, report. Amending § 2.2-3802; adding § 19.2-134.1. (Patron–Herring, HB 2110; Lucas, SB 1391)
- Prevailing wage rate; clarifies that public works includes transportation infrastructure projects. Amending § 2.2-4321.3. (Patron–Krizek, HB 2327)
- Small Business and Supplier Diversity, Department of; redefines "small business." Amending §§ 2.2-1604 and 2.2-4310. (Patron–Obenshain, SB 1369)
- Small, women-owned, and minority-owned businesses; right to appeal denial of initial certification. Amending § 2.2-1606. (Patron–Mundon King, HB 2172)
- State agencies and their appointing authorities; diversity, equity, and inclusion strategic plans. Amending § 2.2-602. (Patron–Askew, HB 1993)
- State and local buildings, certain; definitions, building standards, high performance building certification program, sufficient ZEV charging and fueling infrastructure, etc. Amending §§ 2.2-1182 and 2.2-1183; adding § 15.2-1804.1. (Patron–Helmer, HB 2001)
- Summons; eliminates the requirement that a promise to appear be completed after issuance for a misdemeanor offense or an administrative violation. Amending §§ 2.2-311, 19.2-74, 28.2-901, 29.1-210, 46.2-936, 46.2-940, 54.1-306, 54.1-2506, and 54.1-4407. (Patron–Hanger, SB 1437)
- Treasury and State Treasurer, Department of the; surety bonds. Amending §§ 2.2-2809, 5.1-1.3, 10.1-2006, 21-163, 30-131, 33.2-205, 36-111, 42.1-16, 44-21, 46.2-202, 52-3, 53.1-11, 54.1-305, 58.1-201, and 60.2-109. (Patron–Hayes, HB 2223)
- Veterans Services, Department of; initiatives to reduce unemployment among veterans, comprehensive transition program. Amending § 2.2-2001.2. (Patron–Bell, SB 1279)
- Virginia Freedom of Information Act; individual votes of members of the Virginia Parole Board shall be public records and subject to provisions of the Act. Amending § 2.2-3703. (Patron–Suetterlein, SB 1103)
- Virginia Freedom of Information Act; law-enforcement criminal incident information, criminal investigative files. Amending §§ 2.2-3706, 2.2-3711, 19.2-174.1, and 19.2-368.3; adding § 2.2-3706.1. (Patron–Hurst, HB 2004)
- Virginia Freedom of Information Act; meetings held through electronic communication means during a state of emergency. Amending § 2.2-3708.2. (Patron–McPike, SB 1271)
- Virginia Freedom of Information Act; proprietary records and trade secrets, carbon sequestration agreements. Amending § 2.2-3705.6. (Patron-Vogel, SB 1343)
- Virginia Freedom of Information Act; public body authorized to conduct electronic meetings. Amending § 2.2-3708.2. (Patron–Levine, HB 1931)
- Virginia Freedom of Information Act; record exclusion for personal contact information provided to a public body. Amending § 2.2-3705.1. (Patron–Gooditis, HB 2025)

ADMINISTRATION OF GOVERNMENT (continued)

- Virginia Human Rights Act; expands definition of employer to include person employing one or more domestic workers. Amending § 2.2-3905. (Patron–Price, HB 1864)
- Virginia Human Rights Act; nondiscrimination in employment, definitions, sexual and workplace harassment. Amending §§ 2.2-522, 2.2-3905, 2.2-3907, and 2.2-3908. (Patron–Watts, HB 2155)
- Virginia Human Rights Acts; adds discrimination on the basis of disability as an unlawful employment practice, reasonable accommodations for persons with disabilities. Amending §§ 2.2-3902, 2.2-3905, and 51.5-41; adding § 2.2-3905.1. (Patron–Sickles, HB 1848)
- Virginia Jobs Investment Program and Fund; minimum wage requirements. Amending § 2.2-2240.3. (Patron-Bagby, HB 1967)
- Virginia LGBTQ+ Advisory Board; established, report. Adding §§ 2.2-2499.1 through 2.2-2499.4. (Patron-Lopez, HB 2130)
- Virginia Public Procurement Act; bids or offers on construction contracts, requirement to submit list of subcontractors, effective clause. Adding § 2.2-4303.02. (Patron–Williams Graves, HB 2288)
- Virginia Public Procurement Act; construction contracts, localities with a population in excess of 25,000, subcontractor workforce requirements. Adding § 2.2-4303.02. (Patron–McPike, SB 1305)
- Virginia Public Procurement Act; determination of responsibility, local option to include criteria in Invitation to Bid. Amending § 2.2-4302.1. (Patron–Murphy, HB 1996)
- Virginia Public Procurement Act; local arbitration agreements, definitions, procurement procedures, compliance for post-award actions, penalty. Adding §§ 2.2-4377.1 through 2.2-4377.5. (Patron–Surovell, SB 1384)
- Virginia Public Procurement Act; preference for energy-efficient and water-efficient goods. Adding § 2.2-4328.1. (Patron-Helmer, HB 1811)
- Virginia Small Business Financing Authority; members to have small business lending experience. Amending § 2.2-2282. (Patron–Head, HB 1830)
- Virginia Small Business Financing Authority; risk-based review of outstanding loans. Adding § 2.2-2312.1. (Patron–Mundon King, HB 2170)
- Virginia Small Business Financing Authority; utilization or award of loan and grant program funds. Amending § 2.2-2312. (Patron–Mundon King, HB 2171)
- VirginiaSaves Program; established, membership. Amending § 23.1-701; adding §§ 2.2-2744 through 2.2-2756. (Patron–Torian, HB 2174)
- Workforce development; expands type of data sharing. Amending § 2.2-435.8. (Patron–Subramanyam, HB 1876)

ADMINISTRATION, SECRETARY OF

Military-overseas ballots; Secretary of Administration to oversee and develop a charter and directives for the State Board of Elections to form a working group to study implementation of electronic return of voted ballots. (Patron–DeSteph, SJR 322)

ADOPTION

- Adult adoption; investigation and report. Amending § 63.2-1244. (Patron–Avoli, HB 1957)
- Confirmatory adoption; expands the stepparent adoption provisions, person with a legitimate interest. Amending § 63.2-1241. (Patron–Boysko, SB 1321)

AGING AND REHABILITATIVE SERVICES, DEPARTMENT FOR

- Aging services; Department for Aging and Rehabilitative Services shall prioritize providing services to older persons with greatest economic or social need, definition. Amending §§ 51.5-134 and 51.5-135. (Patron–Adams, D.M., HB 1805)
- Aging services; Department for Aging and Rehabilitative Services shall use available resources to provide services to older persons with the greatest economic and social needs, definitions. Amending §§ 51.5-134 and 51.5-135. (Patron–Barker, SB 1366)

AGRICULTURE AND CONSUMER SERVICES, SECRETARY OF

Carbon Sequestration Task Force; established by Secretary of Natural Resources, jointly with the Secretary of Agriculture and Consumer Services, to convene a task force, report. (Patron–Lewis, SB 1374)

AGRICULTURE, ANIMAL CARE AND FOOD

- Agricultural best management practices; creates an enhanced individual and corporate income tax credit beginning in taxable year 2021 but before January 1, 2025, for the implementation of certain practices by the taxpayer that are required as part of a certified resource management plan. Amending §§ 58.1-339.3 and 58.1-439.5. (Patron–Hanger, SB 1162)
- Agricultural best management practices; creates an enhanced individual and corporate income tax credit beginning in taxable year 2021 for the implementation of certain practices by the taxpayer that are required as part of a certified resource management plan, definition of "liability after such cost-share." Amending §§ 58.1-339.3 and 58.1-439.5. (Patron–Wilt, HB 1763)
- Cats; trap, neuter, and return programs, civil penalty. Amending §§ 3.2-6500, 3.2-6524, 3.2-6546, and 3.2-6551; adding § 3.2-6579.1. (Patron–Lewis, SB 1390)
- Dairy Producer Margin Coverage Premium Assistance Fund and Program; created and established, eligible dairy producer shall apply to the Department of Conservation and Recreation by February 1 of each year to participate, effective clause. Adding §§ 3.2-3304 through 3.2-3307. (Patron–Obenshain, SB 1193)
- Dairy Producer Margin Coverage Premium Assistance Program; established, eligible dairy producer shall apply to the Department of Conservation and Recreation by February 1 of each year to participate, report, sunset date. Adding §§ 3.2-3304 through 3.2-3307. (Patron–Gooditis, HB 1750)
- Dangerous dogs; restructures procedure for adjudication, penalty. Amending §§ 3.2-6540 and 3.2-6542; adding §§ 3.2-6540.01 through 3.2-6540.04, 3.2-6541.1, 3.2-6542.1, 3.2-6542.2, 3.2-6543.1, 3.2-6562.2, and 18.2-52.2. (Patron–Marsden, SB 1135)
- Food and drink permit; inspections required to operate food establishment, Commissioner of Agriculture and Consumer Services to issue permit for the operation of any food manufacturer, food storage warehouse, etc. Amending §§ 3.2-5100, 3.2-5101, 3.2-5130, and 15.2-2288.6. (Patron–Gooditis, HB 2275)
- Healthcare Regulatory Sandbox Program; established, report, sunset date. Amending § 3.2-6591; adding § 3.2-6593.1. (Patron–Stanley, SB 1417)
- Industrial hemp; definitions, updates laws to address the new hemp producer license issued by the U.S. Department of Agriculture. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4118, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. (Patron–Marshall, HB 2078)
- Industrial hemp; increases the maximum tetrahydrocannabinol (THC) concentration, expands definition of "hemp product", registration renewal fee not to exceed \$50, etc. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. (Patron–Peake, SB 1115)
- Invasive plant species; Department of Conservation and Recreation, et al., to study the sale and use of species. (Patron–Bulova, HJR 527)
- Local Food and Farming Infrastructure Grant Program; established. Amending §§ 3.2-303, 3.2-304, and 3.2-310; adding § 3.2-311. (Patron–Rasoul, HB 2068)
- Marijuana; legalization of simple possession, etc., expungement of criminal records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, $3.2-4114.2,\ 3.2-4116,\ 4.1-100,\ 4.1-101.01,\ 4.1-101.02,\ 4.1-101.07,\ 4.1-101.09,\ 4.1-101.010,$ 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72,

AGRICULTURE, ANIMAL CARE AND FOOD (continued)

19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-390, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9, 19.2-392.10, 19.2-392.11, 19.2-392.12, 19.2-392.13, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Herring, HB 2312)

Marijuana; unlawful possession in certain cases, using or consuming while in a motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Ebbin, SB 1406)

Neonicotinoid pesticides; application in residentially zoned outdoor areas, civil penalty. Adding § 3.2-3929.1. (Patron-Krizek, HB 2030)

Peanuts; extends sunset date of excise tax on all peanuts grown in Virginia. Amending § 3.2-1905. (Patron–Brewer, HB 1751; Lucas, SB 1411)

Pet shops, dealers, and dog breeders; employees convicted of animal abuse, penalty. Amending §§ 3.2-6511.1 and 3.2-6511.2. (Patron–Stanley, SB 1412)

Produce safety; removes the sunset date. (Patron-Obenshain, SB 1194)

Virginia Agriculture Food Assistance Program and Fund; established and created. Adding §§ 3.2-4780 through 3.2-4783. (Patron–Filler-Corn, HB 2203)

Virginia Agriculture Food Assistance Program and Fund; established and created, guidelines and regulations. Adding §§ 3.2-4780 through 3.2-4783. (Patron–Hashmi, SB 1188)

AIR POLLUTION

State Air Pollution Control Board; implementation of a low-emissions and zero-emissions vehicle program, energy jurisdictional retail sales. Amending § 10.1-1307; adding § 10.1-1307.04. (Patron–Bagby, HB 1965)

AIRCRAFT AND AIRPORTS

Aircraft civil; registration and licensing. Amending § 5.1-5. (Patron-Stuart, SB 1144)

Metropolitan Washington Airports Authority; repeals effective date for creation of Authority. Repealing § 5.1-178. (Patron–Edwards, SB 1214)

Unmanned aircraft; exempts an owner from the requirement to register. Amending § 5.1-5. (Patron–Delaney, HB 1851; Favola, SB 1098)

ALBEMARLE COUNTY

Albemarle and Prince William Counties; counties that have adopted the county executive form of government may carry over unspent funds from year to year for multiyear capital projects and outstanding grants. Amending §§ 15.2-520 and 15.2-2506. (Patron–Reeves, SB 1120)

ALCOHOLIC BEVERAGE CONTROL ACT

- Alcoholic beverage control; adds the chief administrative officer of a locality to the list of persons who may be sent notice of certain license applications by Board of Directors of the Virginia Alcoholic Beverage Control Authority. Amending §§ 4.1-230 and 15.2-907. (Patron–Lopez, HB 2131)
- Alcoholic beverage control; delays the effective date of the 2020 alcoholic beverage control license and fee reform. Amending §§ 4.1-230, 4.1-233.1, and third, fifth, and eighth enactments of Chapters 1113 and 1114, 2020 Acts. (Patron–VanValkenburg, HB 1845)
- Alcoholic beverage control; local special events license, taxes and fees. Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, and 4.1-233.1. (Patron–Dunnavant, SB 1471)
- Alcoholic beverage control; operation of government stores, sale of low alcohol beverage coolers, effective date. Amending § 4.1-119. (Patron–Locke, SB 1428)
- Alcoholic beverage control; outdoor refreshment area license, fees for state and local licenses. Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, 4.1-233.1, and 4.1-308. (Patron–Ayala, HB 2266)
- Alcoholic beverage control; privileges of banquet licensees. Amending §§ 4.1-206.3 and 4.1-209. (Patron–Rush, HB 1973)
- Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption, report, sunset provision. Amending §§ 4.1-119, 4.1-204, 4.1-206.1, 4.1-206.3, 4.1-207, 4.1-210, 4.1-212.1, and 4.1-221. (Patron–Bulova, HB 1879; Bell, SB 1299)

ALLEGHANY COUNTY

Covington, City of; amending charter, consolidation of school divisions of the City of Covington and Alleghany County, salaries. Amending § 22.1-32. (Patron–Austin, HB 2091; Deeds, SB 1267)

ALLEN, SAMUEL

Allen, Samuel; commending. (Patron-Hayes, HJR 681)

ANDERSON, DOLSON BARNETT

Anderson, Dolson Barnett; recording sorrow upon death. (Patron-Carr, HJR 646)

ANDERSON, JOHN HUTCHISON

Anderson, John Hutchison; commending. (Patron-Webert, HJR 673)

ANDREWS, EDWARD

Andrews, Edward; recording sorrow upon death. (Patron–Carr, HJR 648)

ANDREWS, WILLIE MAKENTLY

Andrews, Willie Makently; recording sorrow upon death. (Patron-McQuinn, HJR 729)

APPOINTMENTS

General Assembly; confirming appointment to Joint Legislative Audit and Review Commission. (Patron-Plum, HJR 604)

Speaker of the House of Delegates; confirming appointment to Virginia Commonwealth University Health System Authority Board of Directors. (Patron–Filler-Corn, HJR 579)

Virginia Criminal Sentencing Commission; confirms the appointment by the Chief Justice of the Supreme Court of Chairman of Commission. (Patron–Herring, HJR 629)

APPOMATTOX, TOWN OF

Appomattox, Town of; amending charter, shifts local elections from May to November, etc. (Patron-Fariss, HB 1858; Peake, SB 1152)

APPROPRIATIONS

Barrier Crimes and Criminal History Records Checks, Joint Subcommittee Studying; continued, appropriation. (Patron–Edwards, SJR 285)

Bribery in correctional facilities; penalty. Adding § 18.2-474.2. (Patron–Lewis, SB 1461)

Budget bill; appropriations for 2020-2022 biennium. Amending Chapter 56, 2020 Sp. I Acts. (Patron-Torian, HB 1800; Howell, SB 1100)

Communicating threats of death or bodily injury to a person with intent to intimidate, etc.; threats in writing, including electronically transmitted communications, penalties. Amending § 18.2-60. (Patron–Leftwich, HB 2194)

Communicating threats of death or bodily injury to a person with intent to intimidate, etc.; threats to another in writing, penalties. Amending § 18.2-60. (Patron–Spruill, SB 1113)

Death penalty; abolishes current penalty, including for those persons currently under a death sentence, repeals provisions relating to trial of capital cases. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Mullin, HB 2263)

Death penalty; abolition of current penalty, sentences changed to life imprisonment, repeals provisions referring to trial of capital cases, etc. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Surovell, SB 1165)

Discrimination; prohibited in voting and elections administration, required process for enacting certain covered practices, civil causes of action, penalties, repeals provision relating to minority language accessibility, etc. Amending §§ 24.2-105, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1; adding §§ 24.2-104.1, 24.2-125 through 24.2-131, and 24.2-1005.2; repealing § 24.2-124. (Patron–Price, HB 1890; McClellan, SB 1395)

Firearms; purchase, possession, etc., following conviction for assault and battery of a family or household member, definition, penalties. Amending §§ 18.2-308.09, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:8. (Patron–Murphy, HB 1992)

Homicides and assaults and bodily woundings; certain matters not to constitute defenses. Adding §§ 18.2-37.1 and 18.2-57.5. (Patron-Roem, HB 2132)

Identification privilege cards; authorizes DMV to issue, fee, confidentiality, penalties. Amending §§ 24.2-411.3, 24.2-643, 46.2-203.2, 46.2-216.1, 46.2-323.01, 46.2-323.1, 46.2-346, 46.2-600.1, and 58.1-3; adding § 46.2-345.3. (Patron–Guzman, HB 2138)

Marijuana; legalization of simple possession, etc., expungement of criminal records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-261.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-251.1:3, 18.2-251.1:2, 18.2-251.1:3,

APPROPRIATIONS (continued)

18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-390, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9, 19.2-392.10, 19.2-392.11, 19.2-392.12, 19.2-392.13, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Herring, HB 2312)

Marijuana; unlawful possession in certain cases, using or consuming while in a motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, $4.1-231,\ 4.1-240,\ 4.1-300,\ 4.1-302,\ 4.1-303,\ 4.1-310,\ 4.1-310.1,\ 4.1-320,\ 4.1-323,\ 4.1-324,$ 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Ebbin, SB 1406)

Plastic firearms and unfinished frames, etc.; manufacture, import, sale, transfer, etc., penalty. Amending § 18.2-308.5; adding § 18.2-308.5:2. (Patron–Simon, HB 2276)

Police and court records; expungement and sealing of records, Expungement Fee Fund created, protection of public record information, penalties, report, effective clause. Amending §§ 9.1-128, 9.1-134, 17.1-502, 19.2-392.1, 19.2-392.2, 19.2-392.3, and 19.2-392.4; adding §§ 1-229.1, 17.1-205.1, 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, 19.2-392.4:1, 19.2-392.5, 19.2-392.6, and 19.2-392.7. (Patron–Surovell, SB 1339)

Restricted permits to operate a motor vehicle; ignition interlock systems. Adding § 18.2-271.5. (Patron-Stuart, SB 1336)

Staffing levels, employment conditions, and compensation at the Virginia Department of Corrections, joint committee of various House and Senate Committees Studying; continued, appropriations. (Patron–Tyler, HJR 522)

Virginia Overtime Wage Act; overtime compensation employees, definitions, penalties. Amending §§ 40.1-29 and 40.1-29.1; adding § 40.1-29.2. (Patron–Mullin, HB 2063)

ARC OF GREATER WILLIAMSBURG

Williamsburg Faith in Action and the Arc of Greater Williamsburg; commending. (Patron-Mullin, HJR 755)

ARLINGTON COUNTY

U.S. Route 29; authorizes the board of any locality that has adopted the county manager plan of government (Arlington County) to name any section located within the boundaries of the locality. Adding § 15.2-719.1. (Patron–Sullivan, HB 1854)

ARMED FORCES

- Active military or a military spouse; definition of "military status," discrimination in public accommodations, employment, and housing. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310. (Patron–Tran, HB 2161)
- Active military or a military spouse; prohibits discrimination in public accommodations, employment, and housing, landlords may not charge a higher security deposit. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1226, and 55.1-1310. (Patron–Bell, SB 1410)
- Income tax, state; subtraction for low-income military veterans with a permanent service-connected disability. Amending § 58.1-322.02. (Patron–Wiley, HB 2337)
- Military medical personnel program; adds any facility that offers medical services to the public and that is supervised by one or more physicians or podiatrists to the list of entities that may participate in the program. Amending § 2.2-2001.4. (Patron–Hayes, HB 2222)
- Military Spouse Liaison; position created in Department of Veterans Services, report, effective clause. Adding § 2.2-2002.2. (Patron–Kiggans, SB 1150)
- Military-overseas ballots; Secretary of Administration to oversee and develop a charter and directives for the State Board of Elections to form a working group to study implementation of electronic return of voted ballots. (Patron–DeSteph, SJR 322)

ARREST

Experiencing or reporting overdoses; prohibits arrest and prosecution. Amending § 18.2-251.03. (Patron-Bulova, HB 1821)

ASIAN PACIFIC AMERICAN BAR ASSOCIATION OF THE GREATER WASHINGTON, D.C., AREA, INC.

Asian Pacific American Bar Association of the Greater Washington, D.C., Area, Inc.; commemorating its 40th anniversary. (Patron–Keam, HJR 718)

ASSAULT AND BATTERY

- Assault and battery; eliminates the mandatory minimum term of confinement of six months if committed against a judge, magistrate, law-enforcement officer, etc., persons with physical or mental disability. Amending § 18.2-57; adding § 19.2-9.2. (Patron–Morrissey, SB 1306)
- Firearms; purchase, possession, etc., following conviction for assault and battery of a family or household member, definition, penalties. Amending §§ 18.2-308.09, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:8. (Patron–Murphy, HB 1992)

ASSISTED LIVING FACILITIES

Assisted living and auxiliary grants; Joint Commission on Health Care to study available data regarding, etc. (Patron–Spruill, SJR 293)

ATTORNEY GENERAL

- Illegal gambling; skill games, definitions, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron–Reeves, SB 1465)
- Illegal gambling; skill games, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron–Scott, HB 2168)
- Marriage; persons who may celebrate rites, authorizes current members of the General Assembly, Governor, Lieutenant Governor, and Attorney General. Amending § 20-25. (Patron–Cosgrove, SB 1142)

AUDITOR OF PUBLIC ACCOUNTS

Auditor of Public Accounts; audits of certain political subdivisions. Amending § 30-140. (Patron–McNamara, HB 1789)

AUTHORITIES

- Alcoholic beverage control; adds the chief administrative officer of a locality to the list of persons who may be sent notice of certain license applications by Board of Directors of the Virginia Alcoholic Beverage Control Authority. Amending §§ 4.1-230 and 15.2-907. (Patron–Lopez, HB 2131)
- Central Virginia Transportation Authority; adds the Executive Director of the Virginia Port Authority, or his designee, as a nonvoting ex officio member. Amending § 33.2-3703. (Patron–McQuinn, HB 1926)
- Hampton Roads area refuse collection authority; changes various requirements of the Southeastern Public Service Authority when budgeting or incurring debt, etc. Amending § 15.2-5102.1. (Patron–Cosgrove, SB 1141)
- Mathews County; board of supervisors may appoint one employee of the locality onto the board of directors for the Economic Development Authority of the County. Amending § 15.2-4904. (Patron–Hodges, HB 2186)
- Metropolitan Washington Airports Authority; repeals effective date for creation of Authority. Repealing § 5.1-178. (Patron–Edwards, SB 1214)
- New River Valley Passenger Rail Station Authority; creation of authority in Planning District 4 (New River Valley RC). Adding §§ 33.2-3800 through 33.2-3816. (Patron–Hurst, HB 1893; Edwards, SB 1212)
- Opioid Abatement Authority; established, Fund created, report, membership. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Barker, SB 1469)
- Opioid Abatement Authority; established, Opioid Abatement Fund created, investment of assets of Fund, report. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Herring, HB 2322)
- Tourism Development Authority; changes name of Authority to the Heart of Appalachia Tourism Authority. Amending §§ 15.2-5500, 15.2-5501, 15.2-5505, 15.2-5506, and 45.1-246. (Patron–Pillion, SB 1399)
- Virginia Good Neighbor Next Door program; Virginia Housing Development Authority shall report recommendations for creating Program. (Patron-Convirs-Fowler, HB 2072)
- Virginia Health Workforce Development Authority; mission of Authority, membership. Amending §§ 32.1-122.7 and 32.1-122.7:1. (Patron–Willett, HB 1976)
- Virginia Small Business Financing Authority; members to have small business lending experience. Amending § 2.2-2282. (Patron–Head, HB 1830)
- Virginia Small Business Financing Authority; risk-based review of outstanding loans. Adding § 2.2-2312.1. (Patron–Mundon King, HB 2170)
- Virginia Small Business Financing Authority; utilization or award of loan and grant program funds. Amending § 2.2-2312. (Patron–Mundon King, HB 2171)

AUTISM

Virginia Missing Child with Autism Alert Program; renames the Virginia Missing Person with Autism Alert Program. Amending §§ 52-34.13, 52-34.14, and 52-34.15. (Patron–Hayes, HB 2216)

AVIATION

Aircraft civil; registration and licensing. Amending § 5.1-5. (Patron-Stuart, SB 1144)

Metropolitan Washington Airports Authority; repeals effective date for creation of Authority. Repealing § 5.1-178. (Patron–Edwards, SB 1214)

Unmanned aircraft; exempts an owner from the requirement to register. Amending § 5.1-5. (Patron–Delaney, HB 1851; Favola, SB 1098)

AVULA, DANNY TK

Avula, Danny TK; commending. (Patron-VanValkenburg, HJR 660)

BALLOTS AND BALLOTING

Military-overseas ballots; Secretary of Administration to oversee and develop a charter and directives for the State Board of Elections to form a working group to study implementation of electronic return of voted ballots. (Patron–DeSteph, SJR 322)

BANISTER, DANIEL

Banister, Daniel; commending. (Patron–Jones, HJR 775)

BAR, VIRGINIA STATE

Virginia State Bar examination; foreign applicants. Amending § 54.1-3926. (Patron-Petersen, SB 1234)

BARBER, DOUG

Barber, Doug; commending. (Patron-Marshall, HJR 778)

BARKER, GEORGE L.

million, OE	ONGE	₽.									
Addressed	the Se	enate	in	memory	of	Officer	Brian	David	Sicknick;	requested	adjournment in
memo	ory										432
Prayer offe	ered										288

BEHAVIORAL HEALTH AND DEVELOPMENTAL SERVICES

- Barrier crimes; amends the current requirements for the Department of Behavioral Health and Developmental Services to provide, etc. Amending §§ 37.2-314, 37.2-408.1, 37.2-416, and 37.2-506. (Patron–Ruff, SB 1176)
- Behavioral health; assessments in local correctional facilities, report. Amending § 53.1-68. (Patron-Coyner, HB 1874)
- Behavioral Health and Developmental Services, Commissioner of; reports to designated protection and advocacy system, incident reporting system. Amending § 37.2-304. (Patron–Orrock, HB 1808; Favola, SB 1154)
- Behavioral Health and Developmental Services, Department of; background checks, persons providing contractual services. Amending §§ 19.2-389, 37.2-416, and 37.2-506. (Patron–Willett, HB 2092)
- Behavioral Health Commission; created, report. Adding §§ 30-401 through 30-408. (Patron–Deeds, SB 1273)
- Behavioral health docket; standards for transfer of supervision between local community-based probation agencies. Amending §§ 9.1-175 and 18.2-254.3. (Patron–Bell, HB 2236)
- Brain injury; clarifies definition. Amending § 37.2-403. (Patron-Coyner, HB 1873; Edwards, SB 1421)
- Community services boards; contracts with private providers. Amending § 37.2-500. (Patron–Willett, HB 2070)
- Community services boards; discharge plan shall be completed prior to the individual's discharge, report. Amending § 37.2-505. (Patron–McPike, SB 1304)
- Criminal justice, behavioral health, and other human services records; Department of Behavioral Health and Developmental Services to establish a work group to study the feasibility of developing a secure, de-identified, etc., database. (Patron–Price, HJR 578)
- Criminal proceedings; consideration of mental condition and intellectual and developmental disabilities, evidence offered by the defendant. Amending §§ 19.2-120, 19.2-163.03, and 19.2-299; adding § 19.2-271.6. (Patron–McClellan, SB 1315)
- Criminal proceedings; consideration of mental condition and intellectual and developmental disabilities, qualifications for court-appointed council. Amending §§ 19.2-120, 19.2-163.03, and 19.2-299; adding § 19.2-271.6. (Patron–Bourne, HB 2047)
- Crisis Call Center Fund; created, consistency with federal guidelines, liability for emergency calls to the National Suicide Prevention Lifeline. Amending §§ 37.2-311.1, 56-484.12, 56-484.17, and 56-484.17:1; adding §§ 37.2-311.2, 37.2-311.3, 37.2-311.4, and 56-484.18:1. (Patron–McPike, SB 1302)
- Early Psychosis Intervention and Coordinated Specialty Care Program Advisory Board; established. Adding § 37.2-313.2. (Patron–Morrissey, SB 1427)
- Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization for drug by carrier. Amending § 38.2-3407.15:2. (Patron–McPike, SB 1269)

BEHAVIORAL HEALTH AND DEVELOPMENTAL SERVICES (continued)

- Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization issued by carrier for drug, certain provisions shall not apply to the state employee health insurance plan, etc. Amending § 38.2-3407.15:2. (Patron–Heretick, HB 2008)
- Individuals with intellectual and developmental disabilities; Department of Medical Assistance Services shall establish a work group to study and develop recommendations for permanent use of virtual supports and increasing access to virtual supports and services. (Patron–Suetterlein, SB 1472)
- Individuals with intellectual and developmental disabilities; Department of Medical Assistance Services to study and develop recommendations for use of virtual support, etc. (Patron–Runion, HB 2197)
- Involuntary admission; amends provisions governing involuntary inpatient and mandatory outpatient treatment, effective date. Amending §§ 37.2-817, 37.2-817.1, 37.2-817.2, and 37.2-817.4; repealing § 37.2-817.3. (Patron–Hope, HB 2166)
- Southwestern Virginia Mental Health Institute; Governor to lease a portion of property to Smyth County for a term of three years, responsibility of County, corrects tax map references. Amending Chapter 678, 2019 Acts. (Patron–O'Quinn, HB 2098; Pillion, SB 1429)
- State facilities; repeals requirements that the Commissioner of Behavioral Health and Developmental Services determine the nationality of each person admitted to a state facility. Repealing § 37.2-827. (Patron–Favola, SB 1220)
- Supported decision-making agreements; Department of Behavioral Health and Developmental Services to develop and implement a program to educate certain individuals. Amending §§ 64.2-2000, 64.2-2003, and 64.2-2007; adding § 37.2-314.3. (Patron–Bell, HB 2230)
- Unrestorably incompetent defendant; disposition, capital murder charge, inpatient custody of the Commissioner of the Department of Behavioral Health and Developmental Services. Adding § 19.2-169.3:1. (Patron–Mason, SB 1272)

BICKFORD, CHARLENE

Bickford, Charlene; commending. (Patron-Hope, HJR 641)

BICYCLES

Bicycles; permits operators to treat a stop sign as a yield sign in certain situations. Amending §§ 46.2-839 and 46.2-905; adding § 46.2-903.1. (Patron–Hurst, HB 2262)

BIGGER, JOSEPHINE JOHNSON

Bigger, Josephine Johnson; recording sorrow upon death. (Patron-McQuinn, HJR 736)

BINGO

Charitable gaming; definitions, regulations, conduct of instant bingo, network bingo, pull tabs, and seal cards, report. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.34. (Patron–Reeves, SB 1127)

BIRTH AND DEATH RECORDS

Birth certificates; an amendment of a certificate shall be evaluated by the State Registrar through an administrative process. Amending § 32.1-269. (Patron–Locke, SB 1121)

BISSON, ANN

Bisson, Ann; recording sorrow upon death. (Patron-Hope, HJR 642)

BLAND, LARRY JEROME

Bland, Larry Jerome; recording sorrow upon death. (Patron-Carr, HJR 685; McQuinn, HJR 732)

BOND ISSUES

Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021; created. (Patron-Torian, HB 2178; Howell, SB 1145)

Refunding bonds; alters the principal and interest requirements, maturity date, and allowable discount for bonds previously issued, sunset clause. Amending first enactment of Chapters 265 and 408, 1992 Acts. (Patron–Torian, HB 2179; Howell, SB 1134)

BRAIN ANEURYSM AWARENESS MONTH

Brain Aneurysm Awareness Month; designating as September 2021 and each succeeding year thereafter. (Patron-Gooditis, HJR 596)

BRAXTON-WILLIAMS, VALERIE

Braxton-Williams, Valerie; commending. (Patron-Bagby, HJR 715)

BRISTOL, CITY OF

Bristol, City of; amending charter, changes to powers and organization. (Patron-O'Quinn, HB 2095)

BRODERICK, JOHN R.

Broderick, John R.; commending. (Patron-Filler-Corn, HJR 714)

BROWN, ALGENON L.

Brown, Algenon L.; recording sorrow upon death. (Patron–McQuinn, HJR 727)

BROWN, JEAN SMITH

Brown, Jean Smith; recording sorrow upon death. (Patron-LaRock, HJR 742)

BROWN, LEWIS R.

Brown, Lewis R.; commending. (Patron-Guzman, HJR 780)

BROWN, SHAMIRA A.

Brown, Shamira A.; commending. (Patron-Guzman, HJR 781)

BRUNSWICK COUNTY

Brunswick County school board; removes school board from the list of approved member salaries for appointed school boards. Amending § 22.1-32. (Patron–Tyler, HB 1798; Ruff, SB 1175)

BUCKINGHAM COUNTY

Buckingham County; fees for disposal of solid waste. Amending § 15.2-2159. (Patron-Peake, SB 1447)

BUDGET, STATE

Budget bill; appropriations for 2020-2022 biennium. Amending Chapter 56, 2020 Sp. I Acts. (Patron-Torian, HB 1800; Howell, SB 1100)

BUILDING CODE

Uniform Statewide Building Code; Board of Housing and Community Development shall consider amendments to Code to address changes in the IECC relating to energy efficiency and conservation. (Patron-Kory, HB 2227)

BUSINESSES

Paid sick leave; employers to provide to certain employees, essential workers that work an average of 20 hours weekly, etc., provisions shall not apply to any retail business with fewer than 25 employees. Adding §§ 40.1-33.3 through 40.1-33.9. (Patron–Guzman, HB 2137)

BYERS, WILLIAM WALKER, JR.

Byers, William Walker, Jr.; recording sorrow upon death. (Patron-Hudson, HJR 703)

C. WALDO SCOTT CENTER FOR H.O.P.E.

C. Waldo Scott Center for H.O.P.E.; commemorating its 30th anniversary. (Patron-Price, HJR 786)

CAMERON, FREDERICK

Cameron, Frederick; recording sorrow upon death. (Patron-Tran, HJR 782)

CAMPAIGN PRACTICES

Campaign finance; prohibited personal use, child care exception. Amending § 24.2-948.4; adding § 24.2-948.5. (Patron–Simon, HB 1952)

Campaign finance reports; special report of pre-legislative session contributions. Amending § 24.2-947.11. (Patron–Saslaw, SB 1444)

Comprehensive campaign finance reform; joint subcommittee to study. (Patron-Bulova, HJR 526)

CANDIDATES IN ELECTIONS

Nomination of candidates for elected offices; restrictions on nomination method selected by political party. Amending § 24.2-509. (Patron–Helmer, HB 2020)

Primary election; changes date of election held in June from second Tuesday in June to third Tuesday in June, also changes candidate filing deadlines to reflect change of date. Amending §§ 24.2-311, 24.2-503, 24.2-507, 24.2-510, 24.2-515, and 24.2-515.1. (Patron–Kiggans, SB 1148)

CANNABIS

Cannabis oil; processing and dispensing by pharmaceutical processors, certification for use of oil for treatment, report. Amending §§ 54.1-3408.3, 54.1-3442.5, 54.1-3442.6, and 54.1-3442.7. (Patron–Adams, D.M., HB 1988)

Employee protections; medicinal use of cannabis oil. Adding § 40.1-27.4. (Patron-Helmer, HB 1862)

Pharmaceutical processors; permits processors to produce and distribute cannabis products, dispensing botanical cannabis to a minor, certification. Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8. (Patron–Lucas, SB 1333)

Pharmaceutical processors; permits processors to produce and distribute cannabis products, testing standards for botanical cannabis and products. Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8. (Patron–Hayes, HB 2218)

CAPITAL OUTLAY

Capital outlay plan; projects to be funded from general fund-supported resources, repeals existing six-year capital outlay for projects to be funded. Repealing Chapter 1134, 2020 Acts. (Patron–Howell, SB 1155)

Capital outlay plan; repeals existing six-year capital outlay for projects to be funded. Repealing Chapter 1134, 2020 Acts. (Patron-Torian, HB 2177)

CAPITOL SOUARE

Capitol Square Preservation Council; powers and duties, review and approval of plans for changes to artifacts contained within the Capitol Building. Amending § 30-194. (Patron–Norment, SB 1172)

Firearm; carrying within Capitol Square and the surrounding area, into building owned or leased by the Commonwealth, etc., penalty. Adding § 18.2-283.2. (Patron–Levine, HB 2295)

Harry F. Byrd, Sr., statue; Department of General Services to remove statue of former Virginia Governor and U.S. Senator from Capitol Square. (Patron–Jones, HB 2208)

Weapons; possessing or transporting within Capitol Square or into a building owned or leased by the Commonwealth, provisions shall not apply to certain law-enforcement officers, etc., penalty. Adding § 18.2-283.2. (Patron–Ebbin, SB 1381)

CARDOUNEL, HUMBERTO, JR.

Cardounel, Humberto, Jr.; commending. (Patron-VanValkenburg, HJR 671)

CAREY, JAMES ALFRED

Carey, James Alfred; recording sorrow upon death. (Patron-Tyler, HJR 691)

CAROLINE COUNTY

Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron–Cole, J.G., HB 1819)

CASINO GAMING

Casino gaming; requirements for issuance of operator's license, human trafficking training. Amending § 58.1-4110. (Patron–Simonds, HB 1944)

Casino gaming; technical amendments to the gaming law related to its interaction with sports betting law, the capital investment required of an applicant for a license, etc. Amending §§ 2.2-3711, 58.1-4100, 58.1-4109, 58.1-4110, 58.1-4114, 58.1-4122, 58.1-4124, and 58.1-4125. (Patron–Krizek, HB 1812)

CATS

Cats; trap, neuter, and return programs, civil penalty. Amending §§ 3.2-6500, 3.2-6524, 3.2-6546, and 3.2-6551; adding § 3.2-6579.1. (Patron–Lewis, SB 1390)

Healthcare Regulatory Sandbox Program; established, report, sunset date. Amending § 3.2-6591; adding § 3.2-6593.1. (Patron–Stanley, SB 1417)

CELL PHONES

Learner's permits; use of personal communication devices. Amending § 46.2-335. (Patron-Stuart, SB 1335)

License restrictions for minors; prohibition on use of handheld personal communications devices. Amending § 46.2-334.01. (Patron–Robinson, HB 1846)

CENTER FOR EXCELLENCE IN EDUCATION

Center for Excellence in Education; commending. (Patron-Keam, HJR 717)

CHAMBERLIN HOTEL

Chamberlin Hotel at Fort Monroe; reverts certain property to the Commonwealth. Repealing Section 1 of Chapter 809, 1998 Acts. (Patron–Mugler, HB 2009)

CHARITABLE GAMING

- Charitable gaming; definitions, regulations, conduct of instant bingo, network bingo, pull tabs, and seal cards, report. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.34. (Patron–Reeves, SB 1127)
- Charitable gaming; increase in certain maximum allowable prize amounts. Amending § 18.2-340.33. (Patron–Keam, HB 1843)
- Charitable gaming; regulations of operators of Texas Hold'em poker games, conduct of games, effective clause. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.18, 18.2-340.19, 18.2-340.22, 18.2-340.28:2, and 18.2-340.34:2. (Patron–Bell, SB 1278)
- Charitable Gaming Board; prohibits the Board from promulgating regulations that prohibit the use of multiple video monitors or touchscreens on an electronic pull tab device, use of devices shall be limited to one player at a time. Amending §§ 18.2-340.19 and 18.2-340.28. (Patron–Willett, HB 1943; McPike, SB 1287)

CHARTERS

Appomattox, Town of; amending charter, shifts local elections from May to November, etc. (Patron-Fariss, HB 1858; Peake, SB 1152)

Bristol, City of; amending charter, changes to powers and organization. (Patron-O'Quinn, HB 2095)

Covington, City of; amending charter, consolidation of school divisions of the City of Covington and Alleghany County, salaries. Amending § 22.1-32. (Patron–Austin, HB 2091; Deeds, SB 1267)

Crewe, Town of; amending charter, various changes to the charter including staggering town council elections, etc. (Patron–Wright, HB 1764; Ruff, SB 1216)

Glasgow, Town of; new charter (previous charter repealed). (Patron-Campbell, R.R., HB 1783)

Lynchburg, City of; amending charter, salaries of members of City Council. (Patron-Walker, HB 2180)

Nassawadox, Town of; amending charter, updates charter to reflect the town's shift of municipal elections from May to November. (Patron–Bloxom, HB 1749)

Norfolk, City of; amending charter, general updates. (Patron-Spruill, SB 1128)

CHASE, AMANDA F.

102911111111111111111111111111111111111
Notified Clerk of presence
Participated remotely
Statements on votes:
S.B. 1357
S.R. 1413

CHESAPEAKE BAY

- Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment, Enhanced Nutrient Removal Certainty Program established. Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14. (Patron–Lopez, HB 2129)
- Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment, Enhanced Nutrient Removal Certainty Program established, regulations. Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14. (Patron–Hanger, SB 1354)
- Stormwater Local Assistance Fund; grants awarded for projects related to Chesapeake Bay total maximum daily load (TMDL) requirements. Amending § 62.1-44.15:29.1. (Patron-Lewis, SB 1404)
- Trees; replacement and conservation during development, projects located in a Chesapeake Bay Preservation Area to address recurrent flooding, effective date, report, effective clause. Amending §§ 15.2-961 and 15.2-961.1. (Patron–Marsden, SB 1393)

CHESTERFIELD COUNTY

Public defender office; establishes an office for the County of Chesterfield, effective clause. Amending § 19.2-163.04. (Patron–Morrissey, SB 1442)

CHILD CARE

- Child care; expands to include all cities the authority related to the regulation of services. Amending § 15.2-914. (Patron–Williams Graves, HB 2326)
- Child care providers; background checks portability, check shall not be required if individual completed a background check within previous five years, pilot program, report. Amending §§ 19.2-389, 22.1-289.035, 22.1-289.039, 63.2-1720.1, and 63.2-1724. (Patron–McGuire, HB 2086; McClellan, SB 1316)
- Child Care Subsidy Program; temporarily expanding Program to provide financial assistance for child care to families in need during public health emergency. (Patron–Filler-Corn, HB 2206)

CHILD SUPPORT

- Child support; availability of medical assistance through Family Access to Medical Insurance Security (FAMIS) plan or other government-sponsored coverage, eligibility requirements. Amending §§ 16.1-260 and 63.2-1903. (Patron–Samirah, HB 2002)
- Child support obligations; party's incarceration not deemed voluntary unemployment or underemployment. Amending §§ 20-108.1 and 63.2-1918. (Patron–Scott, HB 2055)
- Child support payments; juvenile in custody of or committed to the Department of Juvenile Justice. Amending §§ 16.1-263, 16.1-286, and 16.1-290. (Patron–Hope, HB 1912)

CHILDREN

- "Abused or neglected child;" definition. Amending § 16.1-228. (Patron-Lucas, SB 1168)
- Child-placing agencies; repeals provisions that allowed agencies to refuse to perform, assist with, etc., in any child placements when the proposed placement would violate the agency's written religious or moral convictions or policies, any agency that operated prior to February 1, 2021, may continue to operate until January 1, 2022, etc. Repealing § 63.2-1709.3. (Patron–Levine, HB 1932)
- Children's Services Act; effective monitoring and implementation. Amending § 2.2-2649. (Patron-Plum, HB 2212)
- Protective orders; violations of preliminary child protective order, violation involves an act that endangers the child's normal development, changes punishment, etc. Amending §§ 16.1-253 and 16.1-253.2. (Patron–Campbell, J.L., HB 2012; Stanley, SB 1415)
- Social services, local department of; location of child in local department's custody. Amending § 63.2-104. (Patron–Leftwich, HB 2191)
- Special education; Department of Education to provide training and guidance documents to local school divisions on the development of Individualized Education Programs (IEPs) for children with disabilities, etc. Adding § 22.1-214.4. (Patron–Carr, HB 2299)

CIGARETTES

Local cigarette taxes; any locality is authorized to levy taxes upon sale of cigarettes, regional cigarette tax boards, definitions, report. Amending § 58.1-3830; adding § 58.1-3832.1. (Patron–Hanger, SB 1326)

CIRCUIT COURTS

Jurisdiction over criminal cases; certification by general district court of felony or ancillary misdemeanor charges, jurisdiction to such charges shall vest in circuit court. Amending §§ 16.1-123.1, 16.1-241, and 17.1-513. (Patron–Adams, L.R., HB 2150)

CIVIL REMEDIES AND PROCEDURE

- Accrual of cause of action; diagnosis of latent injury or disease. Amending § 8.01-249. (Patron-Guzman, HB 2139)
- Career fatigue and wellness in certain health care providers; programs to address, civil immunity. Amending §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909. (Patron-Hope, HB 1913; Barker, SB 1205)
- Civil actions; actions filed on behalf of multiple similarly situated persons. Amending § 8.01-267.1. (Patron–Surovell, SB 1180)
- Court of Appeals; expands jurisdiction, increases from 11 to 17 number of judges on Court, effective clause. Amending §§ 2.2-511, 8.01-36, 8.01-267.8, 8.01-383.1, 8.01-555, 8.01-626, 8.01-670, 8.01-671, 8.01-675.3, 8.01-676.1, 9.1-909, 15.2-1627, 15.2-1643, 15.2-2139, 15.2-2140, 15.2-2656, 15.2-3104, 15.2-3217, 15.2-3221, 15.2-3222, 15.2-3227, 15.2-3244, 15.2-3308, 15.2-3528, 15.2-3605, 15.2-3809, 15.2-3909, 15.2-4108, 15.2-4120, 15.2-5218, 15.2-5367, 15.2-6606, 15.2-6632, 15.2-7406, 16.1-279.1, 17.1-309, 17.1-400 through 17.1-403, 17.1-405 through 17.1-408, 17.1-410, 17.1-413, 17.1-503, 17.1-513, 18.2-308.08, 18.2-384, 19.2-152.10, 19.2-165, 19.2-321.1, 19.2-321.2, 19.2-322.1, 19.2-386.13, 19.2-402, 19.2-403, 19.2-404, 22.1-97, 22.1-289.024, 24.2-237, 24.2-422, 24.2-433, 25.1-239, 32.1-48.010, 32.1-48.013, 33.2-928, 33.2-2917, 37.2-920, 45.1-161.322, 55.1-1833, 55.1-1966, 55.1-2211, 57-2.02, 58.1-527, 58.1-1828, 58.1-2282, 58.1-3147, 58.1-3992, and 63.2-1710; adding §§ 8.01-675.5 and 8.01-675.6; repealing §§ 8.01-670.1 and 8.01-672. (Patron–Edwards, SB 1261)
- Death penalty; abolition of current penalty, sentences changed to life imprisonment, repeals provisions referring to trial of capital cases, etc. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Surovell, SB 1165)
- General district courts; increases to \$50,000 the maximum civil jurisdictional limit of courts, appeal bond. Amending §§ 8.01-195.4, 16.1-77, and 16.1-107. (Patron–Stanley, SB 1108)
- Habitual offenders; required that the Commissioner of DMV reinstate a person's privilege to drive a motor vehicle that was suspended or revoked solely on the basis that such person was determined to be or adjudicated a habitual offender, repeals remaining provisions of the Habitual Offender Act. Amending §§ 8.01-9, 8.01-407, 16.1-77, 16.1-305, 17.1-213, 19.2-389, 46.2-301, 46.2-301.1, 46.2-411, and 53.1-21; repealing §§ 46.2-355.1 through 46.2-363. (Patron–Stanley, SB 1122)
- Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, cases of a deed of trust conveying owner-occupied residential real estate, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–Torian, HB 2175)
- Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–McClellan, SB 1327)
- Judgments; limitations on enforcement, judgment liens, settlement agents, effective date. Amending §§ 8.01-251, 8.01-458, and 55.1-339. (Patron–Coyner, HB 2099)
- Personal injury claim; disclosure of insurance policy limits. Amending § 8.01-417. (Patron-Stuart, SB 1241)
- Public elementary and secondary schools; possession and administration of undesignated stock albuterol inhalers and valved holding chambers, Department of Health, in conjunction with the Department of Education, shall develop and implement policies. Amending §§ 8.01-225, 22.1-274.2, and 54.1-3408. (Patron–McQuinn, HB 2019)

CIVIL REMEDIES AND PROCEDURE (continued)

Public schools; seizure management and action plans, biennial training, effective date. Amending § 8.01-225; adding § 22.1-274.6. (Patron–DeSteph, SB 1322)

Settlement agreements; staying of dismissal. Adding § 8.01-425.2. (Patron–Leftwich, HB 2193)

Summons for unlawful detainer; notice to tenant, adverse employment actions prohibited. Amending § 8.01-126. (Patron–Jenkins, HB 1897)

Uninsured and underinsured motorist insurance policies; bad faith. Amending §§ 8.01-66.1 and 38.2-2206. (Patron-Petersen, SB 1202)

Wrongful death; beneficiaries. Amending § 8.01-53. (Patron–Leftwich, HB 2190)

Wrongful death statute of limitations; criminal investigations by law enforcement. Amending § 8.01-244. (Patron–Convirs-Fowler, HB 2073)

CIVIL RIGHTS ACT OF 1871

Civil Rights Act of 1871; commemorating the 150th anniversary of the enactment. (Patron–McClellan, SJR 395)

CLERK OF THE SENATE

Susan Clarke Schaar

Report of continued legislation. 5-7	
Report of bills and resolutions continued to 2021 Special Session I	
Report on disclosure forms	

CLERKS OF COURTS

Juveniles; competency evaluation, appointed evaluator or director of community services board, etc., shall acknowledge receipt of court order to clerk of court. Amending § 16.1-356. (Patron–Deeds, SB 1248)

CLIFTON FORGE, TOWN OF

Vacant buildings; removes requirement that a building meet the definition of "derelict building" for Town of Clifton Forge, Town of Pulaski, Town of Timberville, and any city to require, by ordinance, the owner or owners of buildings to register such buildings on an annual basis. Amending § 15.2-1127. (Patron–Locke, SB 1285)

CLIFTON PRESBYTERIAN CHURCH

Clifton Presbyterian Church; commemorating its 150th anniversary. (Patron-Helmer, HJR 762)

COAL MINING

Coal Employment and Production Incentive and Coalfield Employment Enhancement Tax Credits; sunset date, credits earned prior to January 1, 2022. Amending §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1. (Patron–Hudson, HB 1899; McPike, SB 1252)

CODE OF VIRGINIA

§ 1-229.1, adding	SB 1339
§ 2.2-115, amending	SB 1418
§ 2.2-200, amending	НВ 1836, НВ 2321
§ 2.2-203.2:4, amending	SB 1365
§ 2.2-204, amending	НВ 1855, НВ 2321
§ 2.2-205, amending	НВ 1836
§ 2.2-205, amending	HB 2321
§ 2.2-205.1, amending	HB 2321
§ 2.2-206, amending	НВ 1836
§ 2.2-208.1, amending	HB 2105

CODE OF VIRGINIA (continued)	
§ 2.2-208.2, adding	SB 1062
§ 2.2-212, amending	HB 2322, SB 1469
§ 2.2-214.2, adding	
§ 2.2-214.3, adding	
§ 2.2-215, amending	HB 1836, SB 1352
§ 2.2-220.1, amending	НВ 1836
§ 2.2-220.2, amending	НВ 1836
§ 2.2-220.3, amending	НВ 1836
§ 2.2-221, amending	HB 2312, SB 1406
§ 2.2-234, amending	HB 2074, SB 1318
§ 2.2-236, adding	HB 2074, SB 1318
§ 2.2-237, adding	HB 2074
§ 2.2-311, amending	SB 1437
§ 2.2-419, amending	SB 1407
§ 2.2-424, amending	HB 1978
§ 2.2-435.6, amending	HB 2321
§ 2.2-435.7, repealing	HB 2321
§ 2.2-435.8, amending	HB 1876, HB 2321
§ 2.2-435.9, amending	HB 2321
§ 2.2-435.10, amending	HB 2321
§ 2.2-435.10:1, adding	SB 363
§ 2.2-435.12, amending	SB 1174
§ 2.2-436, amending	SB 1458
§ 2.2-437, amending	SB 1458
§ 2.2-507, amending	HB 2312, SB 1406
§ 2.2-507.3, adding	HB 2322, SB 1469
§ 2.2-511, amending	HB 2312, SB 1261, SB 1406
§ 2.2-520, amending	HB 2147
§ 2.2-522, amending	HB 2147, HB 2155, SB 1360
§ 2.2-523, amending	HB 2147
§ 2.2-602, amending	НВ 1993
8.2.2-604.2. amending	HR 1854

CODE OF VIRGINIA (continued)	TVD 2212 GD 1406
	HB 2312, SB 1406
	SB 1451
	НВ 1836
§ 2.2-1176, amending	НВ 1836
§ 2.2-1176.1, amending	HB 1855
§ 2.2-1182, amending	HB 2001
§ 2.2-1183, amending	HB 2001
§ 2.2-1201, amending	HB 1891
§ 2.2-1204, amending	
§ 2.2-1212, adding	HB 1891, HB 2140
§ 2.2-1604, amending	HB 1134, HB 1650, SB 1369
§ 2.2-1605, amending	HB 1650
§ 2.2-1606, amending	HB 2172
§ 2.2-1617, amending	
§ 2.2-1618, adding	
§ 2.2-1619, adding	
§ 2.2-1620, adding	HB 1650
§ 2.2-1621, adding	HB 1650
§ 2.2-1622, adding	HB 1650
§ 2.2-1623, adding	HB 1650
§ 2.2-1837, amending	SB 1033, SB 1179
§ 2.2-2001.2, amending	SB 1279
§ 2.2-2001.4, amending	НВ 2222
§ 2.2-2002.2, adding	SB 1150
§ 2.2-2101, amending	SB 1062
§ 2.2-2203.3, amending	SB 1450
§ 2.2-2238, amending	SB 1314
§ 2.2-2240.3, amending	НВ 1967
	НВ 1830
	HB 2171
	HB 2170

-735-

INDEX

2021 SENATE JOURNAL

ĊO	DE OF VIRGINIA (continued)	
	§ 2.2-2481, amending	. HB 1836
	§ 2.2-2499.1, adding	. HB 2130
	§ 2.2-2499.2, adding	. HB 2130
	§ 2.2-2499.3, adding	. HB 2130
	§ 2.2-2499.4, adding	. HB 2130
	§ 2.2-2499.1, adding	. HB 2312
	§ 2.2-2499.2, adding	. HB 2312
	§ 2.2-2499.3, adding	. HB 2312
	§ 2.2-2499.4, adding	. HB 2312
	§ 2.2-2499.1, adding	SB 1406
	§ 2.2-2499.2, adding	SB 1406
	§ 2.2-2499.3, adding	SB 1406
	§ 2.2-2499.4, adding	SB 1406
	§ 2.2-2543, amending	SB 1414
	§ 2.2-2558, adding	SB 1365
	§ 2.2-2559, adding	SB 1365
	§ 2.2-2560, adding	SB 1365
	§ 2.2-2561, adding	SB 1365
	§ 2.2-2562, adding	SB 1365
	§ 2.2-2563, adding	SB 1365
	§ 2.2-2564, adding	SB 1365
	§ 2.2-2649, amending	. HB 2212
	§ 2.2-2699.10, amending	. HB 1836
	§ 2.2-2699.11, amending	. HB 1836
	§ 2.2-2699.13, amending	. HB 1836
	§ 2.2-2744, adding	. HB 2174
	§ 2.2-2745, adding	. HB 2174
	§ 2.2-2746, adding	. HB 2174
	§ 2.2-2747, adding	. HB 2174
	§ 2.2-2748, adding	. HB 2174
	§ 2.2-2749, adding	. HB 2174
	§ 2.2-2750, adding	. HB 2174

-737-

INDEX

2021 SENATE JOURNAL

CO	DE OF VIRGINIA (continued) 8 2 2-3802 amending	SB 1406
		HB 2161, SB 1410
	-	HB 2161, SB 1410
		HB 1848, HB 2147, HB 2161, SB 1360, SB 1410
		HB 2161, SB 1410
		НВ 1848, НВ 1864
	-	НВ 2155
		HB 2161, SB 1310
		SB 1360
		SB 1410
		HB 1848, SB 1360
		SB 1360
		HB 2147
		HB 2155, SB 1360
		HB 2155, SB 1360
		HB 2147
	-	SB 1407
	§ 2.2-4006, amending	HB 1855
	§ 2.2-4024, amending	HB 2312, SB 1406
	§ 2.2-4302.1, amending	HB 1996
	§ 2.2-4303.02, adding	HB 2288, SB 1305
	§ 2.2-4310, amending	HB 1134, HB 1650, SB 1369
	§ 2.2-4310.3, amending	HB 1650
	§ 2.2-4321.2, amending	SB 1419
	§ 2.2-4321.3, amending	HB 2327
	§ 2.2-4328.1, adding	HB 1811, SB 1222
	§ 2.2-4343, amending	НВ 1650
	§ 2.2-4354, amending	SB 627
	§ 2.2-4377.1, adding	SB 1384
	§ 2.2-4377.2, adding	SB 1384
	§ 2.2-4377.3, adding	SB 1384
	8 2.2-4377.4, adding	SB 1384

CODE OF VIRGINIA (continued)	
§ 2.2-4377.5, adding	
§ 2.2-4402, amending	
§ 2.2-5211, amending I	HB 2117, SB 135, SB 190, SB 1099, SB 1114, SB 1133, SB 1313
§ 2.2-5212, amending I	HB 2117, SB 135, SB 190, SB 1099, SB 1114, SB 1133, SB 1313
§ 2.2-5515, adding	
§ 3.2-303, amending	
§ 3.2-304, amending	
§ 3.2-310, amending	
§ 3.2-311, adding	
§ 3.2-1010, amending	
§ 3.2-1905, amending	
§ 3.2-3011, amending	SB 1312
§ 3.2-3012.1, adding	SB 1312
§ 3.2-3012.2, adding	SB 1312
§ 3.2-3100, amending	SB 1173
§ 3.2-3102, amending	SB 1173
§ 3.2-3108.1, adding	
§ 3.2-3122, adding	HB 1621
§ 3.2-3123, adding	НВ 1621
§ 3.2-3124, adding	НВ 1621
§ 3.2-3125, adding	НВ 1621
§ 3.2-3126, adding	НВ 1621
§ 3.2-3127, adding	НВ 1621
§ 3.2-3122, adding	SB 854
§ 3.2-3123, adding	SB 854
§ 3.2-3124, adding	SB 854
§ 3.2-3125, adding	SB 854
§ 3.2-3126, adding	SB 854
§ 3.2-3127, adding	SB 854
§ 3.2-3304, adding	НВ 1194, НВ 1750
§ 3.2-3305, adding	НВ 1750
8 3 2-3306 adding	HR 1750

О	DDE OF VIRGINIA (continued) § 3.2-3307, adding	HB 1750
	§ 3.2-3304, adding	SB 1193
	§ 3.2-3305, adding	SB 1193
	§ 3.2-3306, adding	SB 1193
	§ 3.2-3307, adding	SB 1193
	§ 3.2-3305, adding	HB 1194
	§ 3.2-3306, adding	HB 1194
	§ 3.2-3906, amending	HB 2312, SB 1406
	§ 3.2-3929.1, adding	HB 2030
	§ 3.2-4112, amending	HB 2078, HB 2312, SB 1115, SB 1406
	§ 3.2-4113, amending	HB 2078, HB 2312
	§ 3.2-4113, amending	SB 1115
	§ 3.2-4113, amending	SB 1243, SB 1406
	§ 3.2-4114, amending	HB 2312, SB 1406
	§ 3.2-4114.2, amending	HB 2078, HB 2312
	§ 3.2-4114.2, amending	
	§ 3.2-4114.2, amending	SB 1406
	§ 3.2-4115, amending	HB 2078
	§ 3.2-4115, amending	SB 1115
	§ 3.2-4116, amending	HB 2078, HB 2312
	§ 3.2-4116, amending	
	§ 3.2-4116, amending	
	§ 3.2-4117.1, adding	HB 2312, SB 1406
	§ 3.2-4117.2, adding	
	§ 3.2-4118, amending	
	§ 3.2-4119, amending	
	§ 3.2-4119, amending	
	§ 3.2-4122, adding	
	§ 3.2-4123, adding	
	§ 3.2-4124, adding	
	§ 3.2-4125, adding	
	8 3.2-4126. adding	

CO	DE OF VIRGINIA (continued)		
	§ 3.2-4159, adding	SB	1243
	§ 3.2-4160, adding	SB	1243
	§ 3.2-4161, adding	SB	1243
	§ 3.2-4162, adding	SB	1243
	§ 3.2-4163, adding	SB	1243
	§ 3.2-4164, adding	SB	1243
	§ 3.2-4165, adding	SB	1243
	§ 3.2-4166, adding	SB	1243
	§ 3.2-4167, adding	SB	1243
	§ 3.2-4168, adding	SB	1243
	§ 3.2-4169, adding	SB	1243
	§ 3.2-4170, adding	SB	1243
	§ 3.2-4171, adding	SB	1243
	§ 3.2-4172, adding	SB	1243
	§ 3.2-4173, adding	SB	1243
	§ 3.2-4174, adding	SB	1243
	§ 3.2-4175, adding	SB	1243
	§ 3.2-4176, adding	SB	1243
	§ 3.2-4177, adding	SB	1243
	§ 3.2-4178, adding	SB	1243
	§ 3.2-4179, adding	SB	1243
	§ 3.2-4180, adding	SB	1243
	§ 3.2-4181, adding	SB	1243
	§ 3.2-4182, adding	SB	1243
	§ 3.2-4183, adding	SB	1243
	§ 3.2-4184, adding	SB	1243
	§ 3.2-4185, adding	SB	1243
	§ 3.2-4186, adding	SB	1243
	§ 3.2-4187, adding	SB	1243
	§ 3.2-4188, adding	SB	1243
	§ 3.2-4189, adding	SB	1243
	§ 3.2-4190, adding	SB	1243

-743-

INDEX

O	DE OF VIRGINIA (continued) § 3.2-5145.7, adding	HB 2312
	§ 3.2-5145.8, adding	НВ 2312
	§ 3.2-5145.9, adding	HB 2312
	§ 3.2-5145.6, adding	SB 1406
	§ 3.2-5145.7, adding	SB 1406
	§ 3.2-5145.8, adding	SB 1406
	§ 3.2-5145.9, adding	SB 1406
	§ 3.2-5802, amending	SB 1177
	§ 3.2-6500, amending	SB 1390
	§ 3.2-6511.1, amending	SB 1412
	§ 3.2-6511.2, amending	SB 1412
	§ 3.2-6524, amending	SB 1390
	§ 3.2-6540, amending	SB 1135
	§ 3.2-6540.01, adding	SB 1135
	§ 3.2-6540.02, adding	SB 1135
	§ 3.2-6540.03, adding	SB 1135
	§ 3.2-6540.04, adding	SB 1135
	§ 3.2-6541.1, adding	SB 1135
	§ 3.2-6542, amending	SB 1135
	§ 3.2-6542.1, adding	SB 1135
	§ 3.2-6542.2, adding	SB 1135
	§ 3.2-6543.1, adding	SB 1135
	§ 3.2-6546, amending	SB 304, SB 1390
	§ 3.2-6551, amending	SB 1390
	§ 3.2-6562.2, adding	SB 1135
	§ 3.2-6579.1, adding	SB 1390
	§ 3.2-6591, amending	SB 1417
	§ 3.2-6592.1, adding	SB 669
	§ 3.2-6593.1, adding	SB 1417
	§ 4.1-100, amending	HB 2312, SB 1406
	§ 4.1-101.01, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-101.02 amending	HB 2312. HB 2312. SB 1406. SB 1406

CO	DE OF VIRGINIA (continued)	
CO	§ 4.1-101.07, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-101.09, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-101.010, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-101.1, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-103, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-104, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-105, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-106, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-107, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-111, amending	HB 2312, HB 2312
	§ 4.1-111, amending	SB 1095
	§ 4.1-111, amending	SB 1406, SB 1406
	§ 4.1-112.2, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-113, amending	SB 1095
	§ 4.1-113.1, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-115, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-116, amending	
	§ 4.1-118, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-119, amending	НВ 1187, НВ 1879
	§ 4.1-119, amending	HB 2312, HB 2312
	§ 4.1-119, amending	SB 824, SB 1299, SB 1388
	§ 4.1-119, amending	
	§ 4.1-119, amending	
	§ 4.1-122, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-124, amending	
	§ 4.1-128, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-200, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-201, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-202, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-204, amending	
	§ 4.1-205, amending	HB 2312, HB 2312, SB 1406, SB 1406
	§ 4.1-206, amending	. HB 2266, HB 2312, HB 2312, SB 1406, SB 1406, SB 1471

INDEX

CODE OF VIRGINIA (continued)	
§ 4.1-230, amending	SB 1345
§ 4.1-230, amending	SB 1406, SB 1406
§ 4.1-231, amending	HB 2266, HB 2312, HB 2312, SB 1406, SB 1406, SB 1471
§ 4.1-231.1, amending	
§ 4.1-233, amending	НВ 2266
§ 4.1-233, amending	SB 1346
§ 4.1-233, amending	SB 1471
§ 4.1-233.1, amending	НВ 1845
§ 4.1-233.1, amending	НВ 2266
§ 4.1-233.1, amending	SB 1346
§ 4.1-233.1, amending	SB 1471
§ 4.1-240, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-300, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-302, amending	НВ 2312, НВ 2312
§ 4.1-302, amending	НВ 233
§ 4.1-302, amending	SB 1406, SB 1406
§ 4.1-302, amending	SB 1443
§ 4.1-303, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-305, amending	SB 306
§ 4.1-308, amending	НВ 2266
§ 4.1-310, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-310.1, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-320, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-323, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-324, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-325, amending	HB 2312, HB 2312
§ 4.1-325, amending	SB 1095
§ 4.1-325, amending	SB 1406, SB 1406
§ 4.1-325.2, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-329, amending	HB 2312, HB 2312, SB 1406, SB 1406
§ 4.1-336, amending	HB 2312, HB 2312, SB 1406, SB 1406
8 11-337 amending	HR 2312 HR 2312 SR 1406 SR 1406

CODE	F VIRGINIA (continued)				
	1-338, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-348, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-349, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-350, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-351, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-352, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-353, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-354, amending	HB 2312	2, HB 2312,	SB 1406, S	B 1406
§ 4.	1-600, adding		•••••	Н	B 2312
§ 4.	1-601, adding		•••••	Н	B 2312
§ 4.	1-602, adding			Н	B 2312
§ 4.	1-603, adding		•••••	Н	B 2312
§ 4.	1-604, adding		•••••	Н	B 2312
§ 4.	1-605, adding		•••••	Н	B 2312
§ 4.	1-606, adding		•••••	Н	B 2312
§ 4.	1-607, adding		•••••	Н	B 2312
§ 4.	1-608, adding	•••••		Н	B 2312
§ 4.	1-609, adding		•••••	Н	B 2312
§ 4.	1-610, adding		•••••	Н	B 2312
§ 4.	1-611, adding		•••••	Н	B 2312
§ 4.	1-612, adding		•••••	Н	B 2312
§ 4.	1-613, adding		•••••	Н	B 2312
§ 4.	1-614, adding	•••••		Н	B 2312
§ 4.	1-615, adding	•••••		Н	B 2312
§ 4.	1-616, adding	•••••		Н	B 2312
§ 4.	1-617, adding			Н	B 2312
§ 4.	1-618, adding			Н	B 2312
§ 4.	1-619, adding			Н	B 2312
§ 4.	1-620, adding			Н	B 2312
§ 4.	1-621, adding			Н	B 2312
§ 4.	1-622, adding			Н	B 2312
8 1	1-623 adding			п	B 2312

-749-

INDEX

2021 SENATE JOURNAL

ODI	F OF VIDCINIA (continued)	
	E OF VIRGINIA (continued) 4.1-1004, adding	. HB 2312
§	4.1-1005, adding	. HB 2312
§	4.1-1006, adding	. HB 2312
§	4.1-1007, adding	. HB 2312
§	4.1-1008, adding	. HB 2312
§	4.1-1009, adding	. HB 2312
§	4.1-1100, adding	. HB 2312
§	4.1-1101, adding	. HB 2312
§	4.1-1102, adding	. HB 2312
§	4.1-1103, adding	. HB 2312
§	4.1-1104, adding	. HB 2312
§	4.1-1105, adding	. HB 2312
§	4.1-1106, adding	. HB 2312
§	4.1-1107, adding	. HB 2312
§	4.1-1108, adding	. HB 2312
§	4.1-1109, adding	. HB 2312
§	4.1-1110, adding	. HB 2312
§	4.1-1111, adding	. HB 2312
§	4.1-1112, adding	. HB 2312
§	4.1-1113, adding	. HB 2312
§	4.1-1114, adding	. HB 2312
§	4.1-1115, adding	. HB 2312
§	4.1-1116, adding	. HB 2312
§	4.1-1117, adding	. HB 2312
§	4.1-1118, adding	. HB 2312
§	4.1-1119, adding	. HB 2312
§	4.1-1120, adding	. HB 2312
§	4.1-1121, adding	. HB 2312
§	4.1-1200, adding	. HB 2312
§	4.1-1201, adding	. HB 2312
§	4.1-1202, adding	. HB 2312
8	4.1-1203. adding	. HB 2312

-751-

INDEX

2021 SENATE JOURNAL

	E OF VIRGINIA (continued)	CD	1406
_	4.1-609, adding		
_	4.1-610, adding		
	4.1-611, adding		
	4.1-612, adding		
	4.1-613, adding		
	4.1-614, adding		
_	4.1-615, adding		
§	4.1-616, adding	.SB	1406
§	4.1-617, adding	.SB	1406
§	4.1-618, adding	.SB	1406
§	4.1-619, adding	.SB	1406
§	4.1-620, adding	.SB	1406
§	4.1-621, adding	.SB	1406
§	4.1-622, adding	.SB	1406
§	4.1-623, adding	.SB	1406
§	4.1-624, adding	.SB	1406
§	4.1-625, adding	.SB	1406
§	4.1-626, adding	.SB	1406
§	4.1-627, adding	.SB	1406
§	4.1-628, adding	.SB	1406
§	4.1-629, adding	.SB	1406
§	4.1-630, adding	.SB	1406
§	4.1-631, adding	.SB	1406
§	4.1-632, adding	.SB	1406
§	4.1-633, adding	.SB	1406
§	4.1-700, adding	.SB	1406
§	4.1-701, adding	.SB	1406
§	4.1-702, adding	.SB	1406
§	4.1-703, adding	.SB	1406
§	4.1-800, adding	.SB	1406
§	4.1-801, adding	.SB	1406
	4.1-802. adding		1406

-753-

INDEX

2021 SENATE JOURNAL

COD	E OF VIRGINIA (continued)		
§	4.1-1108, adding	SB	1406
§	4.1-1109, adding	SB	1406
§	4.1-1110, adding	SB	1406
§	4.1-1111, adding	SB	1406
§	4.1-1112, adding	SB	1406
§	4.1-1113, adding	SB	1406
§	4.1-1114, adding	SB	1406
§	4.1-1115, adding	SB	1406
§	4.1-1116, adding	SB	1406
§	4.1-1117, adding	SB	1406
§	4.1-1118, adding	SB	1406
§	4.1-1119, adding	SB	1406
§	4.1-1120, adding	SB	1406
§	4.1-1200, adding	SB	1406
§	4.1-1201, adding	SB	1406
§	4.1-1202, adding	SB	1406
§	4.1-1203, adding	SB	1406
§	4.1-1204, adding	SB	1406
§	4.1-1205, adding	SB	1406
§	4.1-1206, adding	SB	1406
§	4.1-1207, adding	SB	1406
§	4.1-1300, adding	SB	1406
§	4.1-1301, adding	SB	1406
§	4.1-1302, adding	SB	1406
§	4.1-1303, adding	SB	1406
§	4.1-1304, adding	SB	1406
§	4.1-1305, adding	SB	1406
§	4.1-1306, adding	SB	1406
8	4.1-1307, adding	SB	1406
§	4.1-1308, adding	SB	1406
§	4.1-1309, adding	SB	1406
Ş	4.1-1310, adding	SB	1406

-755-

INDEX

2021 SENATE JOURNAL

CODE OF VIRGINIA (continued)	
§ 8.01-53, amending	HB 2190
§ 8.01-66.1, amending	
§ 8.01-126, amending	НВ 1897
§ 8.01-195.4, amending	SB 1108
§ 8.01-195.10, amending	.HB 2263, SB 449, SB 1033, SB 1165, SB 1179
§ 8.01-195.11, amending	SB 1407
§ 8.01-225, amending	HB 2019, SB 420, SB 1322
§ 8.01-226.5:2, amending	HB 673, SB 1371
§ 8.01-243, amending	HB 610
§ 8.01-244, amending	HB 2073
§ 8.01-246, amending	SB 1340
§ 8.01-249, amending	HB 2139
§ 8.01-251, amending	HB 2099
§ 8.01-267.1, amending	SB 1180
§ 8.01-267.8, amending	SB 1261
§ 8.01-352.1, adding	SB 699
§ 8.01-380.1, adding	SB 1060
§ 8.01-383.1, amending	SB 1261
§ 8.01-401.4, adding	SB 1446
§ 8.01-407, amending	SB 1122
§ 8.01-413.03, adding	SB 1454
§ 8.01-417, amending	SB 1241
§ 8.01-425.2, adding	НВ 2193
§ 8.01-458, amending	HB 2099
§ 8.01-463, amending	HB 2175, SB 1327
§ 8.01-555, amending	SB 1261
§ 8.01-581.15, amending	SB 1107
§ 8.01-581.16, amending	HB 1913, SB 1205
§ 8.01-626, amending	SB 1261
§ 8.01-654, amending	HB 2263, SB 449, SB 1165
§ 8.01-654.1, repealing	HB 2263, SB 449, SB 1165
8 8.01-654.2, repealing	HB 2263, SB 449, SB 1165

-757-

INDEX

2021 SENATE JOURNAL

CODE OF VIRGINIA (continued) § 9.1-801, amending	SB 1406, SB 1406, SB 1406
§ 9.1-801, amending	
§ 9.1-902, amending	
§ 9.1-903, amending	
§ 9.1-907, amending	
§ 9.1-908, amending	
§ 9.1-909, amending	
§ 9.1-1101, amending	HB 2312
§ 9.1-1101, amending	SB 1406, SB 1406, SB 1406
§ 9.1-1500, adding	SB 1468
§ 9.1-1501, adding	SB 1468
§ 9.1-1502, adding	SB 1468
§ 10.1-104.6:1, adding	SB 1290
§ 10.1-104.7, amending	SB 1244
§ 10.1-109, amending	HB 1833
§ 10.1-207, amending	SB 1274
§ 10.1-415, amending	HB 1819
§ 10.1-417.1, adding	НВ 1958
§ 10.1-502, amending	HB 1837, SB 1161
§ 10.1-603.25, amending	HB 1836, SB 1352
§ 10.1-606.3, amending	НВ 1855
§ 10.1-613.6, adding	SB 1280
§ 10.1-658.1, adding	SB 1352
§ 10.1-658.2, adding	SB 1352
§ 10.1-658.3, adding	SB 1352
§ 10.1-658.4, adding	SB 1352
§ 10.1-658.5, adding	SB 1352
§ 10.1-658.6, adding	SB 1352
§ 10.1-658.7, adding	SB 1352
§ 10.1-659, amending	HB 1855, SB 1352
§ 10.1-704, amending	HB 1836
§ 10.1-1016.1, adding	HB 1760, SB 1199

CO	DDE OF VIRGINIA (continued)	IID 1027
	§ 10.1-2129, amending	
	§ 10.1-2202, amending	
	§ 10.1-2202.3, amending	
	§ 10.1-2204, amending	
	§ 10.1-2300, amending	
	§ 10.1-2306, amending	HB 2311
	§ 11-4.6, adding	627, SB 1209
	§ 11-16.3, adding	SB 1407
	§ 11-34.1, repealing	SB 1453
	§ 11-34.2, repealing	SB 1453
	§ 11-34.3, repealing	SB 1453
	§ 11-34.4, repealing	SB 1453
	§ 11-34.3, amending	HB 1855
	§ 13.1-609, amending	HB 2121
	§ 13.1-610, amending	HB 2121
	§ 13.1-615, amending	HB 2121
	§ 13.1-615.1, amending	HB 2121
	§ 13.1-625, amending	HB 2121
	§ 13.1-628, amending	HB 2121
	§ 13.1-630, amending	HB 2121
	§ 13.1-636, amending	HB 2121
	§ 13.1-639, amending	HB 2121
	§ 13.1-658, amending	HB 2121
	§ 13.1-661, amending	HB 2121
	§ 13.1-710, amending	HB 2121
	§ 13.1-711, amending	HB 2121
	§ 13.1-716, amending	HB 2121
	§ 13.1-718, amending	HB 2121
	§ 13.1-721.1, amending	HB 2121
	§ 13.1-722.6, amending	HB 2121
	§ 13.1-722.12:1, amending	HB 2121
	8 13.1-759, amending	HB 2121

-761-

INDEX

CO	DE OF VIRGINIA (continued)	
CO	§ 13.1-1062, amending	HB 2121
	§ 13.1-1065, amending	HB 2121
	§ 13.1-1073.1, amending	HB 2121
	§ 13.1-1074, amending	HB 2121
	§ 13.1-1075, amending	HB 2121
	§ 13.1-1080, amending	HB 2121
	§ 13.1-1087, amending	HB 2121
	§ 13.1-1096, amending	HB 2121
	§ 13.1-1099.14, amending	HB 2121
	§ 13.1-1099.26, amending	НВ 2121
	§ 13.1-1201, amending	НВ 2121
	§ 13.1-1203, amending	НВ 2121
	§ 13.1-1212, amending	НВ 2121
	§ 13.1-1214, amending	НВ 2121
	§ 13.1-1222, amending	НВ 2121
	§ 13.1-1242, amending	НВ 2121
	§ 13.1-1252, amending	НВ 2121
	§ 13.1-1255, amending	НВ 2121
	§ 13.1-1263.1, adding	НВ 2121
	§ 13.1-1264, amending	НВ 2121
	§ 13.1-1265, amending	НВ 2121
	§ 13.1-1271, amending	НВ 2121
	§ 13.1-1277, amending	HB 2121
	§ 15.2-520, amending	HB 1949, SB 1120
	§ 15.2-719.1, adding	HB 1854
	§ 15.2-816.1, amending	SB 1385
	§ 15.2-853, amending	HB 2161, SB 1410
	§ 15.2-854, amending	HB 2161, SB 1410
	§ 15.2-901, amending	НВ 1778
	§ 15.2-907, amending	HB 2131
	§ 15.2-914, amending	НВ 2326
	8 15 2.015 amending	SR 1466

COL	DE OF VIRGINIA (continued)	VID 1055 VID 1050 GD 1061
	·	HB 1855, HB 1859, SB 1061
	-	НВ 1919
	·	HB 707, HB 2042, SB 419, SB 1393
	-	
		HB 1855
	§ 15.2-986, adding	
	§ 15.2-1127, amending	
	§ 15.2-1400, amending	
	§ 15.2-1413, amending	SB 1208
	§ 15.2-1500.1, amending	HB 2161, SB 1410, SB 1449, SB 1450
	§ 15.2-1507, amending	HB 2161, SB 1410, SB 1449, SB 1450
	§ 15.2-1604, amending	HB 2147, HB 2161, SB 1410, SB 1449, SB 1450
	§ 15.2-1626, amending	
	§ 15.2-1627, amending	HB 2312, SB 1243, SB 1261
	§ 15.2-1627, amending	SB 1406, SB 1406, SB 1406
	§ 15.2-1627.2, amending	SB 803
	§ 15.2-1627.3, amending	
	§ 15.2-1636.8, amending	SB 803, SB 1226
	§ 15.2-1643, amending	SB 1261
	§ 15.2-1704.1, adding	HB 1948
	§ 15.2-1704.2, adding	HB 1948
	§ 15.2-1723.2, adding	HB 2031
	§ 15.2-1804.1, adding	HB 2001
	§ 15.2-1809, amending	НВ 2217
	§ 15.2-2114.01, amending	SB 1309
	§ 15.2-2119.4, amending	SB 826
	§ 15.2-2139, amending	SB 1261
	§ 15.2-2140, amending	SB 1261
	§ 15.2-2159, amending	SB 1447
	§ 15.2-2200, amending	НВ 672
	8 15 2-2223 amending	HR 672

ODE OF VIRGINIA (continued) § 15.2-2223.4, amending	НВ 2054
§ 15.2-2223.5, adding	
§ 15.2-2224, amending	
§ 15.2-2245, amending	
§ 15.2-2259, amending	
§ 15.2-2283, amending	HB 672
§ 15.2-2284, amending	
§ 15.2-2288.6, amending	HB 2275
§ 15.2-2288.8, amending	НВ 2201
§ 15.2-2288.8, amending	SB 1207
§ 15.2-2295.1, amending	НВ 1836
§ 15.2-2306, amending	SB 1457
§ 15.2-2308, amending	HB 1898
§ 15.2-2316.6, amending	HB 2201
§ 15.2-2316.7, amending	HB 2201
§ 15.2-2316.8, amending	HB 2201
§ 15.2-2316.9, amending	HB 2201
§ 15.2-2316.6, amending	SB 1207
§ 15.2-2316.7, amending	SB 1207
§ 15.2-2316.8, amending	SB 1207
§ 15.2-2316.9, amending	SB 1207
§ 15.2-2413.1, adding	SB 1298
§ 15.2-2413.2, adding	SB 1298
§ 15.2-2413.3, adding	SB 1298
§ 15.2-2413.4, adding	SB 1298
§ 15.2-2413.5, adding	SB 1298
§ 15.2-2413.6, adding	SB 1298
§ 15.2-2413.7, adding	SB 1298
§ 15.2-2413.8, adding	SB 1298
§ 15.2-2413.9, adding	SB 1298
§ 15.2-2413.10, adding	SB 1298
8 15.2-2413.11, adding	SB 1298

CODE OF VIRGINIA (continued)	
§ 15.2-2506, amending	HB 1949, SB 1120
§ 15.2-2656, amending	SB 1261
§ 15.2-2820, amending	HB 2312
§ 15.2-2820, amending	SB 1406, SB 1406, SB 1406
§ 15.2-3104, amending	SB 1261
§ 15.2-3217, amending	SB 1261
§ 15.2-3221, amending	SB 1261
§ 15.2-3222, amending	SB 1261
§ 15.2-3227, amending	SB 1261
§ 15.2-3244, amending	SB 1261
§ 15.2-3308, amending	SB 1261
§ 15.2-3528, amending	SB 1261
§ 15.2-3605, amending	SB 1261
§ 15.2-3809, amending	SB 1261
§ 15.2-3909, amending	SB 1261
§ 15.2-4108, amending	SB 1261
§ 15.2-4116, amending	HB 2323
§ 15.2-4120, amending	SB 1261
§ 15.2-4209, amending	HB 672
§ 15.2-4904, amending	HB 1743, HB 2186, HB 2287
§ 15.2-5102.1, amending	SB 1141
§ 15.2-5112, amending	HB 2121
§ 15.2-5218, amending	SB 1261
§ 15.2-5367, amending	SB 1261
§ 15.2-5431.8:1, adding	HB 2121
§ 15.2-5431.9, amending	HB 2121
§ 15.2-5431.9:1, adding	HB 2121
§ 15.2-5431.35:1, adding	HB 2121
§ 15.2-5500, amending	SB 1399
§ 15.2-5501, amending	SB 1399
§ 15.2-5505, amending	SB 1399
§ 15.2-5506, amending	SB 1399

ODE OF VIRGINIA (continued)	
§ 15.2-5931, amending	
§ 15.2-6606, amending	SB 1261
§ 15.2-6632, amending	SB 1261
§ 15.2-7406, amending	SB 1261
§ 16.1-69.6:1, amending	
§ 16.1-69.40:1, amending	SB 1406, SB 1406, SB 1406
§ 16.1-69.40:1, amending	
§ 16.1-69.48:1, amending	SB 1406, SB 1406, SB 1406
§ 16.1-69.48:1, amending	HB 2312, SB 1243
§ 16.1-69.48:3, amending	SB 1243
§ 16.1-69.48:6, amending	HB 2169
§ 16.1-69.55, amending	HB 2169, SB 306, SB 1244
§ 16.1-77, amending	SB 1108, SB 1122
§ 16.1-107, amending	SB 1108
§ 16.1-123.1, amending	HB 2150
§ 16.1-228, amending	HB 2312, SB 550, SB 1168, SB 1243
§ 16.1-228, amending	SB 1328
§ 16.1-228, amending	SB 1371
§ 16.1-228, amending	SB 1406, SB 1406, SB 1406
§ 16.1-241, amending	HB 2150, SB 1181, SB 1230
§ 16.1-243, amending	SB 550
§ 16.1-248.1, amending	SB 1456
§ 16.1-249, amending	SB 1033, SB 1456
§ 16.1-253, amending	HB 2012, SB 1415
§ 16.1-253.1, amending	SB 550
§ 16.1-253.2, amending	HB 2012, SB 550, SB 1415, SB 1443
§ 16.1-253.4, amending	SB 550
§ 16.1-256, amending	HB 1878
§ 16.1-260, amending	HB 1878
§ 16.1-260, amending	HB 2002
§ 16.1-260, amending	HB 2017, HB 2312, SB 550, SB 1243
§ 16.1-260, amending	SB 1406, SB 1406, SB 1406

CODE OF VIRGINIA (continued)	
§ 16.1-263, amending	НВ 1912
§ 16.1-266.2, amending	SB 550
§ 16.1-269.1, amending	SB 1033
§ 16.1-273, amending	HB 2312, SB 1243
§ 16.1-273, amending	SB 1406, SB 1406, SB 1406
§ 16.1-278.5, amending	HB 2056
§ 16.1-278.7, amending	SB 1456
§ 16.1-278.8, amending	SB 1456
§ 16.1-278.8:01, amending	HB 2312, SB 1243
§ 16.1-278.8:01, amending	SB 1406, SB 1406, SB 1406
§ 16.1-278.9, amending	HB 961, HB 2312, SB 1243
§ 16.1-278.9, amending	SB 1406, SB 1406, SB 1406
§ 16.1-279.1, amending	SB 550, SB 1261
§ 16.1-281, amending	HB 1962
§ 16.1-282.1, amending	SB 1328
§ 16.1-283, amending	HB 1962
§ 16.1-285.1, amending	HB 1991, SB 1033
§ 16.1-285.2, amending	HB 1991, SB 1033
§ 16.1-286, amending	HB 1912
§ 16.1-290, amending	HB 1912
§ 16.1-291, amending	HB 2056
§ 16.1-292, amending	HB 2056
§ 16.1-296, amending	SB 550
§ 16.1-298, amending	SB 550
§ 16.1-300, amending	SB 1206
§ 16.1-300, amending	SB 1244
§ 16.1-305, amending	SB 223, SB 1122, SB 1244
§ 16.1-305.3, adding	SB 1231
§ 16.1-306, amending	SB 223
§ 16.1-309.1, amending	SB 1243
§ 16.1-309.4, amending	SB 1033
8 16.1-309.11 adding	НВ 2017

CO	DE OF VIRGINIA (continued)	
	§ 16.1-322.5, amending	
	§ 16.1-322.6, amending	SB 1033
	§ 16.1-351, amending	SB 1184
	§ 16.1-352, amending	SB 1184
	§ 16.1-353, amending	SB 1184
	§ 16.1-349, amending	SB 1184
	§ 16.1-350, amending	SB 1184
	§ 16.1-356, amending	SB 1248
	§ 17.1-205.1, adding	SB 1339
	§ 17.1-213, amending	SB 1122, SB 1244
	§ 17.1-223, amending	HB 2064
	§ 17.1-275, amending	HB 2249
	§ 17.1-275, amending	SB 1243, SB 1294
	§ 17.1-275.8, amending	SB 1243
	§ 17.1-275.13, amending	HB 2169
	§ 17.1-276, amending	HB 1775, HB 2312
	§ 17.1-276, amending	SB 1406, SB 1406, SB 1406
	§ 17.1-293.1, amending	НВ 2113
	§ 17.1-293.1, amending	HB 2312, HB 2312
	§ 17.1-293.1, amending	SB 1283, SB 1372
	§ 17.1-309, amending	SB 1261
	§ 17.1-310, amending	HB 2263, SB 449, SB 1165
	§ 17.1-313, repealing	HB 2263, SB 449, SB 1165
	§ 17.1-323, amending	НВ 2113
	§ 17.1-323, amending	НВ 2312, НВ 2312
	§ 17.1-323, amending	SB 1283, SB 1372
	§ 17.1-400, amending	SB 1261
	§ 17.1-401, amending	SB 1261
	§ 17.1-402, amending	SB 1261
	§ 17.1-403, amending	SB 1261
	§ 17.1-405, amending	SB 1261
	§ 17.1-406, amending	SB 1261

CODE OF VIRGINIA (continued)	
	SB 126
§ 17.1-408, amending	SB 126
§ 17.1-406, amending	HB 2263, SB 449, SB 1165
§ 17.1-410, amending	
§ 17.1-413, amending	HB 2113
§ 17.1-413, amending	HB 2312, HB 2312
§ 17.1-413, amending	SB 1261, SB 1283, SB 1372
§ 17.1-502, amending	
§ 17.1-502, amending	HB 2312, HB 2312
§ 17.1-502, amending	SB 1283, SB 1339, SB 1372
§ 17.1-503, amending	SB 126
§ 17.1-507, amending	SB 973
§ 17.1-513, amending	HB 2150, SB 126
§ 17.1-805, amending	HB 251, HB 673, HB 2169, SB 1243, SB 1244
§ 18.2-8, amending	HB 2263, SB 449, SB 1165
§ 18.2-10, amending	HB 2263, SB 449, SB 1165
§ 18.2-12.2, adding	SB 537
§ 18.2-16.1, adding	НВ 1063, НВ 195
§ 18.2-17, repealing	HB 2263, SB 449, SB 1165
§ 18.2-18, amending	HB 2263, SB 449, SB 1165
§ 18.2-19, amending	HB 2263, SB 449, SB 1165
§ 18.2-22, amending	HB 2263, SB 449, SB 1165
§ 18.2-25, amending	HB 2263, SB 449, SB 1165
§ 18.2-26, amending	HB 2263, SB 449, SB 1165
§ 18.2-30, amending	HB 2263, SB 449, SB 1165
§ 18.2-31, amending	HB 2263, SB 449, SB 1165
§ 18.2-32, amending	HB 2263, SB 449, SB 1165
§ 18.2-36.1, amending	SB 1443
§ 18.2-36.2, amending	SB 1443
§ 18.2-37.1, adding	НВ 2132
§ 18.2-46.1, amending	HB 251, HB 2169, HB 2312, SB 1243
8 18 2-46 1 amending	SR 1406 SR 1406 SR 1406

CODE OF VIRGINIA (continued)	
§ 18.2-46.3:3, amending	SB 1443
§ 18.2-48.1, amending	SB 1033
§ 18.2-49, amending	HB 251
§ 18.2-51, amending	SB 1203
§ 18.2-51.1, amending	SB 1443
§ 18.2-52.1, amending	SB 1138
§ 18.2-52.2, adding	SB 1135
§ 18.2-53.1, amending	SB 1443
§ 18.2-57, amending	HB 2312, SB 1203, SB 1306
§ 18.2-57, amending	SB 1406, SB 1406, SB 1406
§ 18.2-57, amending	SB 1443
§ 18.2-57.2, amending	SB 550
§ 18.2-57.5, adding	HB 2132
§ 18.2-58, amending	HB 1936, SB 805
§ 18.2-60, amending	HB 2194, SB 1113
§ 18.2-60.4, amending	SB 1443
§ 18.2-61, amending	SB 1443
§ 18.2-62, repealing	SB 1138
§ 18.2-67.1, amending	SB 1240, SB 1443
§ 18.2-67.2, amending	SB 1240, SB 1443
§ 18.2-67.4:1, amending	SB 1138
§ 18.2-67.5:2, amending	SB 1443
§ 18.2-67.5:3, amending	SB 1443
§ 18.2-104, repealing	HB 2290
§ 18.2-121, amending	SB 1203
§ 18.2-154, amending	SB 1443
§ 18.2-186.4, amending	HB 2331, SB 1443
§ 18.2-247, amending	HB 2078, HB 2312
§ 18.2-247, amending	SB 1115
§ 18.2-247, amending	SB 1243
§ 18.2-247, amending	SB 1406, SB 1406, SB 1406
§ 18.2-248, amending	HB 2312, HB 2331, SB 1243

CODE OF VIRGINIA (continued)	SB 1406, SB 1406, SB 1406
	SB 1400, SB 1400, SB 1400
	HB 2312, HB 2331, SB 1243
•	SB 1406, SB 1406, SB 1406
	SB 1443
	HB 2331, SB 1443
	HB 2312, HB 2331, SB 1243, SB 1406, SB 1443
-	HB 2331, SB 1443
	HB 2218, HB 2312, SB 1243, SB 1333, SB 1406
•	HB 2312, SB 306, SB 1243
	SB 1406, SB 1406, SB 1406
· ·	HB 2263, SB 449, SB 1165
§ 18.2-251.02, amending	
§ 18.2-251.02, amending	SB 1406, SB 1406, SB 1406
§ 18.2-251.03, amending	HB 1821, HB 2312, SB 1243, SB 1344
§ 18.2-251.03, amending	SB 1406, SB 1406, SB 1406
§ 18.2-251.1, repealing	HB 2312, SB 1243
§ 18.2-251.1:1, repealing	SB 1243
§ 18.2-251.1:2, repealing	SB 1243
§ 18.2-251.1:3, repealing	SB 1243
§ 18.2-251.1, repealing	SB 1406
§ 18.2-251.1:1, amending	НВ 2312
§ 18.2-251.1:1, amending	SB 1406, SB 1406, SB 1406
§ 18.2-251.1:2, amending	НВ 2312
§ 18.2-251.1:2, amending	SB 1406, SB 1406, SB 1406
§ 18.2-251.1:3, amending	НВ 2078, НВ 2312
§ 18.2-251.1:3, amending	SB 1115, SB 1406, SB 1406, SB 1406
§ 18.2-252, amending	HB 2312, SB 1243
	SB 1406, SB 1406, SB 1406
	HB 2312, SB 1243
	SB 1406, SB 1406, SB 1406, SB 1406
8 18 2-254 1 amending	SB 816

CODE OF VIDCINIA (continued)	
CODE OF VIRGINIA (continued) § 18.2-254.3, amending	НВ 2236
§ 18.2-255, amending	HB 2312, HB 2331, SB 1243
§ 18.2-255, amending	SB 1406, SB 1406, SB 1406
§ 18.2-255, amending	SB 1443
§ 18.2-255.1, amending	HB 2312, SB 1243
§ 18.2-255.1, amending	SB 1406, SB 1406, SB 1406
§ 18.2-255.2, amending	HB 2312, HB 2331, SB 1243
§ 18.2-255.2, amending	SB 1406, SB 1406, SB 1406
§ 18.2-255.2, amending	SB 1443
§ 18.2-258, amending	HB 2312, SB 1243
§ 18.2-258.01, amending	SB 1243
§ 18.2-258.02, amending	SB 1243
§ 18.2-258.1, amending	SB 1243
§ 18.2-258, amending	SB 1406, SB 1406, SB 1406
§ 18.2-258.02, amending	НВ 2312
§ 18.2-258.02, amending	SB 1406, SB 1406, SB 1406
§ 18.2-258.1, amending	HB 2312
§ 18.2-258.1, amending	SB 1406, SB 1406, SB 1406
§ 18.2-265.1, amending	HB 2312, SB 1243
§ 18.2-265.1, amending	SB 1406, SB 1406, SB 1406
§ 18.2-265.2, amending	HB 2312, SB 1243
§ 18.2-265.2, amending	SB 1406, SB 1406, SB 1406
§ 18.2-265.3, amending	HB 2312, SB 1243
§ 18.2-265.3, amending	SB 1406, SB 1406, SB 1406
§ 18.2-265.22, adding	SB 1406
§ 18.2-265.23, adding	SB 1406
§ 18.2-265.24, adding	SB 1406
§ 18.2-265.25, adding	SB 1406
§ 18.2-265.26, adding	SB 1406
§ 18.2-265.27, adding	SB 1406
§ 18.2-265.28, adding	SB 1406
8 18 2-266 amending	SR 148

CODE OF VIRGINIA (continued)	
	SB 148
§ 18.2-270, amending	
§ 18.2-271.5, adding	SB 1336
§ 18.2-282, amending	
§ 18.2-283.2, adding	HB 2295, SB 13, SB 15, SB 1381
§ 18.2-287.2, amending	
§ 18.2-287.2, amending	
§ 18.2-287.4, amending	HB 961
§ 18.2-308.03, amending	НВ 2312
§ 18.2-308.03, amending	
§ 18.2-308.08, amending	SB 1261
§ 18.2-308.09, amending	HB 1288, HB 1992, HB 2312, SB 490, SB 1243, SB 1382
§ 18.2-308.09, amending	
§ 18.2-308.012, amending	
§ 18.2-308.012, amending	SB 1406, SB 1406, SB 1406
§ 18.2-308.016, amending	HB 2312
§ 18.2-308.016, amending	
§ 18.2-308.016, amending	SB 1407
§ 18.2-308.1, amending	
§ 18.2-308.1:5, amending	
§ 18.2-308.1:5, amending	SB 1406, SB 1406, SB 1406
§ 18.2-308.1:6, adding	HB 1288, SB 490
§ 18.2-308.1:8, adding	HB 1992, SB 1382
§ 18.2-308.2, amending	HB 1288, SB 490, SB 1443
§ 18.2-308.2:01, amending	НВ 961
§ 18.2-308.2:1, amending	HB 961, HB 1288, HB 1992, SB 490, SB 1382
§ 18.2-308.2:2, amending	HB 961, HB 1288, HB 1992, HB 2128
	HB 2319
	SB 490, SB 1250, SB 1382, SB 1443
	HB 1288, HB 1992, SB 490, SB 1382
8 18 2-308 2:5 amending	SB 1430

CODE OF VIRGINIA (continued) § 18.2-308.4, amending	HB 2312, SB 1243
§ 18.2-308.4, amending	SB 1406, SB 1406, SB 1406
§ 18.2-308.4, amending	SB 1443
§ 18.2-308.5, amending	HB 2276
§ 18.2-308.5:2, adding	HB 2276
§ 18.2-308.7, amending	HB 961
§ 18.2-308.9, adding	HB 961
§ 18.2-308.10, adding	HB 961
§ 18.2-308.11, adding	HB 961
§ 18.2-308.12, adding	HB 961
§ 18.2-308.13, adding	HB 961
§ 18.2-325, amending	HB 2168, SB 1407, SB 1465
§ 18.2-331.1, adding	HB 2168, SB 1465
§ 18.2-334.2, amending	SB 1127
§ 18.2-334.2, amending	SB 1278
§ 18.2-334.3, amending	SB 1407
§ 18.2-340.16, amending	SB 1127
§ 18.2-340.16, amending	SB 1278
§ 18.2-340.18, amending	SB 1278
§ 18.2-340.19, amending	HB 1943, SB 1052
§ 18.2-340.19, amending	SB 1127
§ 18.2-340.19, amending	SB 1278, SB 1287
§ 18.2-340.22, amending	SB 1278, SB 1407
§ 18.2-340.23, amending	SB 1127
§ 18.2-340.25, amending	SB 1127
§ 18.2-340.26:1, amending	SB 1127
§ 18.2-340.27, amending	SB 1127
§ 18.2-340.27:1, amending	SB 387
§ 18.2-340.28, amending	НВ 1943
§ 18.2-340.28, amending	SB 387, SB 1127
§ 18.2-340.28, amending	SB 1287
§ 18.2-340.28:1, amending	SB 1127

CODE OF VIRGINIA (continued) § 18.2-474.2, adding	SB 1461
§ 18.2-477.2, amending	SB 1033
§ 18.2-511.2, adding	SB 353
§ 18.2-513, amending	HB 251, HB 2169, SB 1243
§ 19.2-3.1, amending	SB 1242
§ 19.2-9.2, adding	SB 1306
§ 19.2-10.2, amending	НВ 2169
§ 19.2-11.01, amending	HB 2263, SB 449, SB 1165, SB 1244
§ 19.2-11.2, amending	SB 550, SB 1243
§ 19.2-56, amending	SB 1475
§ 19.2-66, amending	HB 2312, SB 1243
§ 19.2-66, amending	SB 1406, SB 1406, SB 1406
§ 19.2-71, amending	HB 2263, SB 449, SB 1165
§ 19.2-72, amending	НВ 2113
§ 19.2-72, amending	НВ 2312, НВ 2312
§ 19.2-72, amending	SB 1283, SB 1372
§ 19.2-74, amending	НВ 2113
§ 19.2-74, amending	НВ 2312, НВ 2312
§ 19.2-74, amending	SB 1283, SB 1372, SB 1437
§ 19.2-76.1, amending	HB 2263, SB 449, SB 1165
§ 19.2-81, amending	HB 2312
§ 19.2-81, amending	SB 1406, SB 1406, SB 1406
§ 19.2-81.1, amending	HB 2312, SB 1243
§ 19.2-81.1, amending	SB 1406, SB 1406, SB 1406
§ 19.2-81.3, amending	SB 550
§ 19.2-83.1, amending	HB 2312, SB 1243
§ 19.2-83.1, amending	SB 1406, SB 1406, SB 1406
§ 19.2-83.5, amending	SB 1332
§ 19.2-100, amending	HB 2263, SB 449, SB 1165
§ 19.2-102, amending	HB 2263, SB 449, SB 1165
§ 19.2-120, amending HB 2047, HB 2263, SB 449, SB 1165	, SB 1243, SB 1266, SB 1315, SB 1383
§ 19.2-120.1, amending	SB 1243, SB 1266

CODE OF VIRGINIA (continued)	
	SB 1266
§ 19.2-134.1, adding	
§ 19.2-134.1, adding	
§ 19.2-152.2, amending	HB 2263, SB 449, SB 1165
§ 19.2-152.10, amending	SB 1261
§ 19.2-157, amending	HB 2263, SB 449, SB 1165
§ 19.2-159, amending	HB 2263, SB 449, SB 1165
§ 19.2-163, amending	HB 2263, SB 449, SB 1165
§ 19.2-163.01, amending	HB 2263, SB 449, SB 1165
§ 19.2-163.03, amending	HB 2047
§ 19.2-163.03, amending	SB 1315
§ 19.2-163.04, amending	SB 1442
§ 19.2-163.01:1, amending	
§ 19.2-163.4:1, amending	HB 2263, SB 449, SB 1165
§ 19.2-163.7, repealing	HB 2263, SB 449, SB 1165
§ 19.2-163.8, repealing	HB 2263, SB 449, SB 1165
§ 19.2-165, amending	SB 1261
§ 19.2-169.1, amending	SB 1431
§ 19.2-169.3, amending	HB 2263, SB 449, SB 1165, SB 1244
§ 19.2-169.3:1, adding	SB 1272
§ 19.2-174.1, amending	
§ 19.2-175, amending	HB 2263, SB 449, SB 681, SB 1165
§ 19.2-188.1, amending	
§ 19.2-188.1, amending	SB 1406, SB 1406, SB 1406
§ 19.2-215.1, amending	HB 251, HB 2169, SB 1243
§ 19.2-217.1, amending	HB 2263, SB 449, SB 1165
§ 19.2-247, amending	HB 2263, SB 449, SB 1165
§ 19.2-264.2, repealing	НВ 2263
§ 19.2-264.3, repealing	НВ 2263
§ 19.2-264.3:1, repealing	НВ 2263
§ 19.2-264.3:1.1, repealing	НВ 2263
§ 19.2-264.3:1.2, repealing	HB 2263

DE OF VIRGINIA (continued)	
§ 19.2-264.3:1.3, repealing	HB 2263
§ 19.2-264.3:2, repealing	HB 2263
§ 19.2-264.3:3, repealing	HB 2263
§ 19.2-264.3:4, repealing	HB 2263
§ 19.2-264.4, repealing	HB 2263
§ 19.2-264.5, repealing	НВ 2263
§ 19.2-264.2, repealing	SB 449
§ 19.2-264.3, repealing	SB 449
§ 19.2-264.3:1, repealing	SB 449
§ 19.2-264.3:1.1, repealing	SB 449
§ 19.2-264.3:1.2, repealing	SB 449
§ 19.2-264.3:1.3, repealing	SB 449
§ 19.2-264.3:2, repealing	SB 449
§ 19.2-264.3:3, repealing	SB 449
§ 19.2-264.3:4, repealing	SB 449
§ 19.2-264.4, repealing	SB 449
§ 19.2-264.5, repealing	SB 449
§ 19.2-264.2, repealing	SB 1165
§ 19.2-264.3, repealing	SB 1165
§ 19.2-264.3:1, repealing	SB 1165
§ 19.2-264.3:1.1, repealing	SB 1165
§ 19.2-264.3:1.2, repealing	SB 1165
§ 19.2-264.3:1.3, repealing	SB 1165
§ 19.2-264.3:2, repealing	SB 1165
§ 19.2-264.3:3, repealing	SB 1165
§ 19.2-264.3:4, repealing	SB 1165
§ 19.2-264.4, repealing	SB 1165
§ 19.2-264.5, repealing	SB 1165
§ 19.2-264.3, amending	SB 811
§ 19.2-266.4, adding	SB 618
§ 19.2-268.3, amending	НВ 2169
§ 19.2-270.4:1, amending	HB 2263, SB 449, SB 1165

CODE OF VIRGINIA (continued)
§ 19.2-271.6, adding
§ 19.2-288, amending
§ 19.2-291.1, amending
§ 19.2-295, amending
§ 19.2-295.1, amending
§ 19.2-295.3, amending
§ 19.2-297.1, amending
§ 19.2-298.01, amending
§ 19.2-299, amending HB 2047, HB 2263, SB 449, SB 1165, SB 1243, SB 1244, SB 1315, SB 1383
§ 19.2-299.1, amending
§ 19.2-299.2, amending
§ 19.2-301, amending
§ 19.2-303, amending
§ 19.2-303, amending
§ 19.2-303.01, amending
§ 19.2-303.01, amending
§ 19.2-303.03, adding
§ 19.2-303.1, amending
§ 19.2-305.1, amending
§ 19.2-305.2, amending
§ 19.2-306, amending
§ 19.2-306.1, adding
§ 19.2-310.7, amending
§ 19.2-310.7, amending
§ 19.2-310.7, amending
§ 19.2-311, amending
§ 19.2-319, amending
§ 19.2-321.1, amending
§ 19.2-321.2, amending
§ 19.2-322.1, amending
§ 19.2-327.1, amending
§ 19.2-327.2, amending

CODE OF VIRGINIA (continued) § 19.2-327.3, amending	HB 2263, SB 449, SB 1165
§ 19.2-327.11, amending	HB 2263, SB 449, SB 1165
§ 19.2-327.15, adding	НВ 2133
§ 19.2-327.16, adding	НВ 2133
§ 19.2-327.17, adding	НВ 2133
§ 19.2-327.18, adding	HB 2133
§ 19.2-327.19, adding	HB 2133
§ 19.2-327.20, adding	HB 2133
§ 19.2-327.15, adding	SB 1105
§ 19.2-340, amending	HB 2113
§ 19.2-340, amending	HB 2312, HB 2312
§ 19.2-349, amending	HB 2233, SB 1426
§ 19.2-353.5, amending	HB 1895
§ 19.2-354, amending	HB 1895
§ 19.2-354.1, amending	HB 1895
§ 19.2-355, amending	HB 1895
§ 19.2-354, amending	HB 2233, SB 1426
§ 19.2-368.3, amending	HB 2004
§ 19.2-368.10, amending	HB 1867
§ 19.2-386.13, amending	SB 1261
§ 19.2-386.16, amending	HB 2169
§ 19.2-386.22, amending	НВ 2312
§ 19.2-386.23, amending	HB 2312
§ 19.2-386.24, amending	HB 2312
§ 19.2-386.25, amending	HB 2312
§ 19.2-386.22, amending	SB 1243
§ 19.2-386.23, amending	SB 1243
§ 19.2-386.24, amending	SB 1243
§ 19.2-386.25, amending	SB 1243
§ 19.2-386.22, amending	SB 1406
§ 19.2-386.23, amending	SB 1406
§ 19.2-386.24, amending	SB 1406

CO	DE OF VIRGINIA (contin	ued)
	-	SB 1406
	-	HB 961, HB 1288, HB 1992, SB 490, SB 1243, SB 1382
	§ 19.2-386.35, amending	
	§ 19.2-387.3, amending	HB 2258
	§ 19.2-388, amending	
	§ 19.2-389, amending	HB 2086, HB 2092, HB 2312, SB 1122, SB 1243, SB 1244, SB 1316
	§ 19.2-389, amending	
	§ 19.2-389, amending	
	§ 19.2-389.1, amending	HB 2263, SB 449, SB 1165, SB 1244
	§ 19.2-389.3, amending SB 1406	HB 2113, HB 2263, HB 2312, SB 1165, SB 1243, SB 1283, SB 1372,
	§ 19.2-390, amending	
	§ 19.2-390, amending	
	§ 19.2-390, amending	
	§ 19.2-392.02, amending	HB 673, HB 2169, HB 2312, SB 1243, SB 1244
	§ 19.2-392.02, amending	
	§ 19.2-392.1, amending	
	§ 19.2-392.1, amending	
	§ 19.2-392.1:1, adding	SB 1339
	§ 19.2-392.2, amending	HB 2312, SB 808, SB 947, SB 1243, SB 1294
	§ 19.2-392.2, amending	
	§ 19.2-392.2:1, adding	
	§ 19.2-392.2:1, adding	SB 1339, SB 1406, SB 1406, SB 1406
	§ 19.2-392.2:2, adding	SB 1339
	§ 19.2-392.2:3, adding	SB 1339
	§ 19.2-392.3, amending	
	§ 19.2-392.3:1, adding	
	§ 19.2-392.4, amending	HB 2312, SB 808, SB 914, SB 1264
	§ 19.2-392.4, amending	SB 1339, SB 1406, SB 1406, SB 1406
	§ 19.2-392.4:1, adding	
	§ 19.2-392.5, adding	НВ 2113
	§ 19.2-392.6, adding	НВ 2113
	\$ 10.2-302.7 adding	HR 2113

CODE OF VIRGINIA (continued)	
§ 19.2-392.8, adding	HB 2113
§ 19.2-392.9, adding	HB 2113
§ 19.2-392.10, adding	HB 2113
§ 19.2-392.11, adding	HB 2113
§ 19.2-392.12, adding	HB 2113
§ 19.2-392.13, adding	HB 2113
§ 19.2-392.14, adding	HB 2113
§ 19.2-392.15, adding	HB 2113
§ 19.2-392.16, adding	HB 2113
§ 19.2-392.5, adding	НВ 2312, НВ 2312
§ 19.2-392.5, adding	SB 1283
§ 19.2-392.6, adding	SB 1283
§ 19.2-392.7, adding	SB 1283
§ 19.2-392.8, adding	SB 1283
§ 19.2-392.9, adding	SB 1283
§ 19.2-392.10, adding	SB 1283
§ 19.2-392.11, adding	SB 1283
§ 19.2-392.12, adding	SB 1283
§ 19.2-392.13, adding	SB 1283
§ 19.2-392.14, adding	SB 1283
§ 19.2-392.5, adding	SB 1339, SB 1372
§ 19.2-392.6, adding	SB 1372
§ 19.2-392.7, adding	SB 1372
§ 19.2-392.8, adding	SB 1372
§ 19.2-392.9, adding	SB 1372
§ 19.2-392.10, adding	SB 1372
§ 19.2-392.11, adding	SB 1372
§ 19.2-392.12, adding	SB 1372
§ 19.2-392.13, adding	SB 1372
§ 19.2-392.14, adding	SB 1372
§ 19.2-392.6, adding	HB 2312, HB 2312
§ 19.2-392.6, adding	SB 1339

INDEX

CO	DDE OF VIRGINIA (continued) § 20-183, adding	HB 1852
	§ 20-184, adding	
	§ 20-185, adding	HB 1852
	§ 20-186, adding	НВ 1852
	§ 20-187, adding	НВ 1852
	§ 21-163, amending	НВ 2223
	§ 22.1-7.1, amending	SB 1317
	§ 22.1-9, amending	НВ 1827
	§ 22.1-23.3, adding	SB 1080
	§ 22.1-29, amending	НВ 2198
	§ 22.1-32, amending	. HB 1798, HB 2091, SB 1175, SB 1267
	§ 22.1-57.3:1.1, amending	HB 1838
	§ 22.1-79.7, amending	HB 2013
	§ 22.1-79.9, adding	SB 1225
	§ 22.1-97, amending	SB 1261
	§ 22.1-98, amending	HB 1790, SB 1132
	§ 22.1-124.1, adding	SB 1433
	§ 22.1-124.2, adding	SB 1433
	§ 22.1-124.3, adding	SB 1433
	§ 22.1-124.4, adding	SB 1433
	§ 22.1-124.5, adding	SB 1433
	§ 22.1-124.6, adding	SB 1433
	§ 22.1-131.1, adding	HB 1909
	§ 22.1-137.2, amending	HB 1998
	§ 22.1-138.2, adding	HB 1823
	§ 22.1-141.3, adding	SB 1106
	§ 22.1-146.1, amending	НВ 1633
	§ 22.1-200.03, amending	НВ 1905
	§ 22.1-203.4, adding	SB 80
	§ 22.1-205, amending	НВ 1918
	§ 22.1-205, amending	НВ 2119
	§ 22.1-205.1, adding	HB 1918. SB 1169

CODE OF VIRGINIA (continued)				
§ 22.1-206, amending	HB 2312			
§ 22.1-206, amending	SB 1406, SB 1406, SB 1406			
§ 22.1-207.4:2, adding	HB 2135			
§ 22.1-209.1:2, amending	SB 1033			
§ 22.1-212.1, amending	SB 1434			
§ 22.1-212.33, adding	HB 634, SB 249			
§ 22.1-214, amending	SB 1288			
§ 22.1-214.4, adding	HB 2299, SB 1288			
§ 22.1-215, amending	SB 1288			
§ 22.1-253.13:3, amending	HB 634, SB 249			
§ 22.1-253.13:1, amending	HB 1865			
§ 22.1-253.13:2, amending	SB 1191, SB 1257			
§ 22.1-253.13:3, amending	HB 2027, SB 390, SB 1401			
§ 22.1-253.13:4, amending	SB 1288			
§ 22.1-253.13:5, amending	HB 1904, SB 1196			
§ 22.1-254, amending	HB 1940, SB 1439			
§ 22.1-274, amending	HB 1736, SB 1191, SB 1257			
§ 22.1-274.01:1, amending	SB 1191			
§ 22.1-274.2, amending	HB 2019			
§ 22.1-274.5, adding	SB 420			
§ 22.1-274.6, adding	SB 1322			
§ 22.1-277.08, amending	HB 2312, SB 1243			
§ 22.1-277.08, amending	SB 1406, SB 1406, SB 1406			
§ 22.1-280.2:1, amending	SB 1238			
§ 22.1-289, amending	SB 1033			
§ 22.1-289.024, amending	SB 1261			
§ 22.1-289.035, amending	HB 2086, SB 1316			
§ 22.1-289.039, amending	HB 2086, SB 1316			
§ 22.1-289.058, adding	HB 1823			
§ 22.1-289.3, adding	SB 461			
§ 22.1-291.4, amending	HB 2176			
§ 22.1-295.2, amending	HB 2161, SB 1410, SB 1449, SB 1450			

CODE OF VIRGINIA (continued) § 22.1-298.1, amending	HB 1904, SB 1196, SB 1288
§ 22.1-298.7, adding	HB 1904, SB 1196
§ 22.1-306, amending	HB 2161, SB 1410, SB 1449, SB 1450
§ 22.1-315, amending	SB 1243
§ 22.1-323, amending	НВ 2238
§ 22.1-364, adding	НВ 2058
§ 22.1-365, adding	НВ 2058
§ 22.1-366, adding	НВ 2058
§ 22.1-367, adding	НВ 2058
§ 22.1-368, adding	НВ 2058
§ 23.1-110, adding	SB 1436
§ 23.1-201.1, adding	SB 898
§ 23.1-307, amending	SB 1068
§ 23.1-401.2, adding	SB 80
§ 23.1-407.1, adding	НВ 1930
§ 23.1-408.1, adding	SB 464
§ 23.1-409, amending	НВ 2120
§ 23.1-505.1, adding	HB 2123, SB 1387
§ 23.1-506, amending	HB 2123, SB 1387
§ 23.1-609, amending	НВ 2312
§ 23.1-609, amending	SB 1406, SB 1406, SB 1406
§ 23.1-615.1, adding	НВ 1980
§ 23.1-701, amending	НВ 2174
§ 23.1-802, amending	НВ 2120
§ 23.1-815.1, adding	HB 2031
§ 23.1-1301, amending	HB 2312, SB 1068, SB 1359
§ 23.1-1301, amending	SB 1406, SB 1406, SB 1406
§ 23.1-1303, amending	HB 2120
§ 23.1-2415, amending	SB 1450
§ 23.1-2626, amending	HB 1855
§ 23.1-2627, amending	HB 1855
§ 23.1-2911.2, adding	HB 2204, SB 1405

INDEX

CODE OF VIRGINIA (continued) § 24.2-101, amending	IID 1000 CD 1274 CD 1450
§ 24.2-103.2, adding	
§ 24.2-104.1, adding	
§ 24.2-105, amending	
§ 24.2-110, amending	
§ 24.2-124, repealing	
§ 24.2-125, adding	
§ 24.2-126, adding	HB 1890
§ 24.2-127, adding	HB 1890
§ 24.2-128, adding	HB 1890
§ 24.2-129, adding	HB 1890
§ 24.2-130, adding	HB 1890
§ 24.2-131, adding	HB 1890
§ 24.2-125, adding	SB 1395
§ 24.2-126, adding	SB 1395
§ 24.2-127, adding	SB 1395
§ 24.2-128, adding	SB 1395
§ 24.2-129, adding	SB 1395
§ 24.2-130, adding	SB 1395
§ 24.2-131, adding	SB 1395
§ 24.2-202, amending	SB 1432
§ 24.2-203, amending	SB 1432
§ 24.2-209.1, adding	HB 177, SB 1101
§ 24.2-209.2, adding	HB 177, SB 1101
§ 24.2-218, amending	HB 2198
§ 24.2-222, amending	HB 2198
§ 24.2-223, amending	HB 2198
§ 24.2-233, amending	HB 2312, SB 1243
§ 24.2-233, amending	SB 1406, SB 1406, SB 1406
§ 24.2-237, amending	SB 1261
§ 24.2-306, amending	HB 1890, SB 1395
§ 24.2-311, amending	SB 1148

ODE OF VIRGINIA (continued) § 24.2-403.1, adding	НВ 2125
§ 24.2-404, amending	
§ 24.2-408, amending	
§ 24.2-410.1, amending	SB 887
§ 24.2-411.1, amending	SB 887
§ 24.2-411.3, amending	HB 2138, SB 887, SB 1459
§ 24.2-412, amending	SB 887
§ 24.2-413, amending	SB 887
§ 24.2-416, amending	HB 1810
§ 24.2-416.1, amending	HB 1888
§ 24.2-418, amending	SB 887
§ 24.2-418.01, adding	SB 1118
§ 24.2-418.1, amending	SB 887
§ 24.2-422, amending	SB 1261
§ 24.2-427, amending	SB 1422
§ 24.2-428.2, amending	SB 887
§ 24.2-433, amending	SB 1261
§ 24.2-503, amending	SB 1148
§ 24.2-507, amending	SB 1148
§ 24.2-509, amending	HB 2020
§ 24.2-510, amending	SB 1148
§ 24.2-515, amending	SB 1148
§ 24.2-515.1, amending	SB 1148
§ 24.2-542, amending	SB 1432
§ 24.2-542.1, amending	SB 1432
§ 24.2-603, amending	HB 1888, SB 1376
§ 24.2-604, amending	HB 2081
§ 24.2-606, amending	SB 1111
§ 24.2-638, amending	НВ 1921
§ 24.2-643, amending	HB 2138, SB 1459
§ 24.2-646.1, amending	НВ 1921
8 24.2-649, amending	HB 1890. HB 1921. SR 1395

CODE OF VIRGINIA (continued)	
§ 24.2-649.1, adding	НВ 1921
§ 24.2-653, amending	SB 887, SB 1118, SB 1459
§ 24.2-653.01, amending	SB 1459
§ 24.2-653.1, amending	НВ 1888
§ 24.2-653.4, adding	
§ 24.2-667.1, adding	HB 1888, SB 1376
§ 24.2-671, amending	НВ 2081
§ 24.2-673, amending	SB 1432
§ 24.2-684, amending	SB 1348
§ 24.2-701, amending	SB 1459
§ 24.2-701.1, amending	
§ 24.2-702.1, amending	SB 1097, SB 1455, SB 1459
§ 24.2-704, amending	HB 1888, SB 1097, SB 1331
§ 24.2-705, amending	НВ 1888
§ 24.2-706, amending	НВ 1888
§ 24.2-706, amending	SB 1097, SB 1239, SB 1245, SB 1455, SB 1459
§ 24.2-707, amending	НВ 1888
§ 24.2-707, amending	SB 1097, SB 1245, SB 1455, SB 1459
§ 24.2-707.1, adding	
§ 24.2-708, amending	НВ 1888
§ 24.2-709, amending	НВ 1888
§ 24.2-709.1, amending	НВ 1888
§ 24.2-710, amending	НВ 1888
§ 24.2-711, amending	НВ 1888
§ 24.2-709, amending	SB 1245
§ 24.2-709.1, amending	SB 1245, SB 1246, SB 1376, SB 1459
§ 24.2-710, amending	SB 1245, SB 1376
§ 24.2-711, amending	SB 1376
§ 24.2-712, amending	HB 1888, SB 1153, SB 1246, SB 1376, SB 1459
§ 24.2-802.1, amending	НВ 2081
§ 24.2-947.4:2, adding	SB 1236
8 24 2-047 11 amending	SB 1377

SB 1444	CODE OF VIRGINIA (continued) § 24.2-947.11, amending
	§ 24.2-948.4, amending
HB 1952, SB 166, SB 1233	§ 24.2-948.5, adding
	§ 24.2-948.6, adding
	§ 24.2-948.7, adding
SB 1233	§ 24.2-948.8, adding
SB 166	§ 24.2-949.9, amending
SB 166	§ 24.2-949.9:01, adding
SB 166	§ 24.2-950.9, amending
SB 166	§ 24.2-950.10, adding
SB 166	§ 24.2-951.9, amending
SB 166	§ 24.2-951.10, adding
SB 166	§ 24.2-952.7, amending
SB 166	§ 24.2-952.8, adding
	§ 24.2-953.6, adding
HB 1890, SB 1395	§ 24.2-1000, amending
HB 1890, SB 1395	§ 24.2-1005, amending
HB 1890, SB 1395	§ 24.2-1005.1, amending
HB 1890, SB 1395	§ 24.2-1005.2, adding
SB 887	§ 24.2-1016, amending
SB 1260	§ 25.1-203, amending
SB 1261	§ 25.1-239, amending
SB 1270	§ 25.1-306, amending
НВ 1836	§ 28.2-207, amending
SB 1467	§ 28.2-409, amending
SB 1437	§ 28.2-901, amending
HB 1855	§ 28.2-1208, amending
НВ 2298, НВ 2298	§ 29.1-100, amending
НВ 1836	§ 29.1-102, amending
SB 1437	§ 29.1-210, amending
SB 1402	§ 29.1-311, amending
SB 1033	§ 29.1-317, amending

CODE OF VIRGINIA (continued) § 29.1-519, amending	SB 1460
§ 29.1-556.1, amending	
§ 29.1-573, amending	
§ 29.1-579, amending	
§ 30-19.1:4, amending	
§ 30-19.1:13, adding	
§ 30-19.21, adding	
§ 30-28.18, amending	
§ 30-105, amending	
§ 30-129.4, amending	
§ 30-129.5, amending	
§ 30-129.5:1, adding	
§ 30-129.6, amending	
§ 30-131, amending	
§ 30-140, amending	
§ 30-170, repealing	
§ 30-194, amending	SB 1172
§ 30-275, amending	HB 1855
§ 30-343, amending	SB 1473
§ 30-377, amending	НВ 1836, НВ 2321
§ 30-392, amending	НВ 2324
§ 30-396, amending	HB 2082
§ 30-399, amending	HB 2082
§ 30-401, adding	SB 1273
§ 30-402, adding	SB 1273
§ 30-403, adding	SB 1273
§ 30-404, adding	SB 1273
§ 30-405, adding	SB 1273
§ 30-406, adding	SB 1273
§ 30-407, adding	SB 1273
§ 30-408, adding	SB 1273
8 32.1-4.1. adding	SB 1235

Ol	DE OF VIRGINIA (continued) § 32.1-15.2, adding	SB 1449
	§ 32.1-17, amending	SB 993
	§ 32.1-23.3, adding	HB 2007, SB 1436
	§ 32.1-30, amending	SB 993
	§ 32.1-34.3, adding	НВ 1963
	§ 32.1-42.2, adding	НВ 2116
	§ 32.1-46, amending	SB 1116, SB 1117
	§ 32.1-46.01, amending	НВ 2061
	§ 32.1-48, amending	SB 1116
	§ 32.1-48.001, adding	НВ 1989
	§ 32.1-48.010, amending	SB 1237, SB 1261
	§ 32.1-48.013, amending	SB 1261
	§ 32.1-58, amending	НВ 2169
	§ 32.1-73.14, adding	НВ 1995
	§ 32.1-73.15, adding	НВ 1995
	§ 32.1-73.16, adding	НВ 1995
	§ 32.1-73.17, adding	HB 1995
	§ 32.1-102.1, amending	SB 983
	§ 32.1-122.6:04, amending	SB 1147
	§ 32.1-122.7, amending	НВ 1976
	§ 32.1-122.7:1, amending	НВ 1976
	§ 32.1-127, amending	. HB 2154, HB 2300, SB 1050, SB 1149, SB 1356, SB 1449
	§ 32.1-137.08, adding	HB 2162
	§ 32.1-162.5, amending	
	§ 32.1-162.12, amending	
	§ 32.1-164, amending	
	§ 32.1-164.1:01, amending	
	§ 32.1-269, amending	SB 1121
	§ 32.1-289.2, repealing	SB 1138
	§ 32.1-291.7, amending	
	§ 32.1-291.16, amending	SB 1138
	§ 32.1-309.1, amending	HB 2005, SB 1268

COI	DE OF VIRGINIA (continued)	
	§ 32.1-325, amending HB 1987, SB 382, SB 917, SB 946, SB 1227, S	SB 1307, SB 1338, SB 1416
	§ 32.1-326.3, amending	SB 1307
	§ 32.1-331.04, adding	SB 1102
	§ 33.2-120, adding	SB 1453
	§ 33.2-205, amending	НВ 2223
	§ 33.2-214.2, amending	HB 2071, SB 1350
	§ 33.2-221.1, adding	SB 1453
	§ 33.2-234, amending	HB 1813
	§ 33.2-236, amending	HB 1855
	§ 33.2-269.1, adding	SB 1011
	§ 33.2-290, amending	SB 1450
	§ 33.2-335, amending	SB 1217
	§ 33.2-353, amending	HB 2071
	§ 33.2-353, amending	SB 1274
	§ 33.2-353, amending	SB 1350
	§ 33.2-613, amending	HB 2104, HB 2312
	§ 33.2-613, amending	SB 1406, SB 1406, SB 1406
	§ 33.2-802, amending	HB 1801
	§ 33.2-928, amending	SB 1261
	§ 33.2-1011, amending	SB 1260
	§ 33.2-1509, amending	SB 1253
	§ 33.2-1526.3, amending	SB 1364
	§ 33.2-1526.3:1, adding	SB 1364
	§ 33.2-1823, repealing	HB 2104
	§ 33.2-1907, amending	SB 1126
	§ 33.2-2509, amending	НВ 729, НВ 729
	§ 33.2-2917, amending	SB 1261
	§ 33.2-3100.2, adding	HB 845
	§ 33.2-3703, amending	НВ 1926
	§ 33.2-3800, adding	НВ 1893
	§ 33.2-3801, adding	НВ 1893
	§ 33.2-3802, adding	НВ 1893

O	DE OF VIRGINIA (continued) § 33.2-3803, adding	HB	1803
	§ 33.2-3804, adding		
	§ 33.2-3805, adding		
	§ 33.2-3806, adding		
	§ 33.2-3807, adding		
	§ 33.2-3808, adding		
	§ 33.2-3809, adding		
	§ 33.2-3810, adding		
	§ 33.2-3811, adding		
	§ 33.2-3812, adding		
	§ 33.2-3813, adding		
	§ 33.2-3814, adding	HB	1893
	§ 33.2-3815, adding	HB	1893
	§ 33.2-3816, adding	HB	1893
	§ 33.2-3800, adding	SB	1212
	§ 33.2-3801, adding	SB	1212
	§ 33.2-3802, adding	SB	1212
	§ 33.2-3803, adding	SB	1212
	§ 33.2-3804, adding	SB	1212
	§ 33.2-3805, adding	SB	1212
	§ 33.2-3806, adding	SB	1212
	§ 33.2-3807, adding	SB	1212
	§ 33.2-3808, adding	SB	1212
	§ 33.2-3809, adding	SB	1212
	§ 33.2-3810, adding	SB	1212
	§ 33.2-3811, adding	SB	1212
	§ 33.2-3812, adding	SB	1212
	§ 33.2-3813, adding	SB	1212
	§ 33.2-3814, adding	SB	1212
	§ 33.2-3815, adding		
	§ 33.2-3816, adding		
	8 34-29, amending		

CODE OF VIRGINIA (continued) § 35.1-17.1, adding	SB 780
§ 36-96.1, amending	HB 2161
§ 36-96.1:1, amending	HB 2161
§ 36-96.2, amending	НВ 2161
§ 36-96.3, amending	НВ 2161
§ 36-96.1, amending	SB 1410
§ 36-96.1:1, amending	SB 1410
§ 36-96.2, amending	SB 1410
§ 36-96.3, amending	SB 1410, HB 7, HB 2046, HB 2046, SB 97
§ 36-96.3:2, amending	НВ 1971
§ 36-96.4, amending	HB 2161, SB 1410
§ 36-96.6, amending	HB 2161, SB 1410
§ 36-96.17, amending	HB 7, HB 2046, HB 2046, SB 97
§ 36-99, amending	SB 1224
§ 36-111, amending	HB 2223
§ 36-139, amending	HB 2014
§ 36-139, amending	
§ 37.2-205, adding	SB 1449
§ 37.2-304, amending	HB 1808, SB 533, SB 1154
§ 37.2-311.1, amending	SB 1302
§ 37.2-311.2, adding	SB 1302
§ 37.2-311.3, adding	SB 1302
§ 37.2-311.4, adding	SB 1302
§ 37.2-312.3, adding	SB 569
§ 37.2-313.2, adding	SB 1427, SB 1427
§ 37.2-314, amending	HB 2169, SB 1176, SB 1243
§ 37.2-314.1, adding	
§ 37.2-314.2, amending	SB 1407
§ 37.2-314.3, adding	НВ 2230
§ 37.2-403, amending	
§ 37.2-408.1, amending	SB 1176
§ 37.2-416, amending	HB 2092, HB 2169, SB 1176, SB 1243

ODE OF VIRGINIA (continued) § 37.2-422.1, adding	SB 317
§ 37.2-500, amending	НВ 2070
§ 37.2-505, amending	SB 1049, SB 1304
§ 37.2-506, amending	HB 2092, HB 2169, SB 1176, SB 1243
§ 37.2-808, amending	SB 1050
§ 37.2-814, amending	SB 1049
§ 37.2-817, amending	HB 2166, SB 1049
§ 37.2-817.1, amending	SB 1049
§ 37.2-817.2, amending	SB 1049
§ 37.2-817.3, amending	SB 1049
§ 37.2-817.4, amending	SB 1049
§ 37.2-817.1, amending	HB 2166
§ 37.2-817.2, amending	HB 2166
§ 37.2-817.3, repealing	HB 2166
§ 37.2-817.4, amending	HB 2166
§ 37.2-827, repealing	SB 1220
§ 37.2-838, amending	SB 1049
§ 37.2-844, amending	SB 1244
§ 37.2-845, amending	SB 1244
§ 37.2-846, amending	SB 1244
§ 37.2-900, repealing	SB 1244
§ 37.2-900.1, repealing	SB 1244
§ 37.2-901, repealing	SB 1244
§ 37.2-902, repealing	SB 1244
§ 37.2-903, repealing	SB 1244
§ 37.2-904, repealing	SB 1244
§ 37.2-905, repealing	SB 1244
§ 37.2-905.1, repealing	SB 1244
§ 37.2-905.2, repealing	SB 1244
§ 37.2-906, repealing	SB 1244
§ 37.2-907, repealing	SB 1244
§ 37.2-908, repealing	SB 1244

l) SB 382	ODE OF VIRGINIA (continued § 38.2-3418.15, repealing	O
SB 382	§ 38.2-3418.15:1, adding	
HB 1987, SB 1338		
SB 917	§ 38.2-3418.18, adding	
HB 1896, SB 917, SB 1276	§ 38.2-3451, amending	
SB 1341	§ 38.2-3521.1, amending	
HB 2219	§ 38.2-4209.1, amending	
НВ 2332	§ 38.2-4214, amending	
НВ 1807	§ 38.2-4310, amending	
НВ 2219	§ 38.2-4312.1, amending	
HB 1829, HB 2332, SB 867, SB 917	§ 38.2-4319, amending	
НВ 1829	§ 38.2-4509, amending	
НВ 1884	§ 38.2-6505, amending	
НВ 2332	§ 38.2-6600, adding	
НВ 2332	§ 38.2-6601, adding	
НВ 2332	§ 38.2-6602, adding	
НВ 2332	§ 38.2-6603, adding	
НВ 2332	§ 38.2-6604, adding	
НВ 2332	§ 38.2-6605, adding	
НВ 2332	§ 38.2-6606, adding	
НВ 2332	§ 38.2-6607, adding	
	§ 40.1-2, amending	
	§ 40.1-27.3, amending	
	§ 40.1-27.4, adding	
	§ 40.1-28.6, repealing	
	§ 40.1-28.6:1, adding	
HB 416, HB 2134, SB 1323	§ 40.1-28.7:7, adding	
HB 333, HB 1786, SB 719	§ 40.1-28.9, amending	
	§ 40.1-28.13, adding	
SB 660	§ 40.1-28.14, adding	
SB 660	§ 40.1-28.15, adding	
SB 660	8 40.1-28.16, adding	

§ 45.1-161.3, repealing SB 1452 § 45.1-161.5, repealing SB 1453 § 45.1-161.5, repealing SB 1453 § 45.1-161.6, repealing SB 1453 § 45.1-161.7, repealing SB 1453 § 45.1-161.8, repealing SB 1453 § 45.1-161.9, repealing SB 1453 § 45.1-161.10, repealing SB 1453 § 45.1-161.11, repealing SB 1453 § 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.27, repealing SB 1453	CODE	OF VIRGINIA (continued)	
§ 45.1-161.5, repealing SB 1453 § 45.1-161.6, repealing SB 1453 § 45.1-161.7, repealing SB 1453 § 45.1-161.8, repealing SB 1453 § 45.1-161.8, repealing SB 1453 § 45.1-161.0, repealing SB 1453 § 45.1-161.11, repealing SB 1453 § 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.16, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453	§ 4	15.1-161.3, repealing	SB 1453
§ 45.1-161.5:1, repealing SB 1453 § 45.1-161.6, repealing SB 1453 § 45.1-161.8, repealing SB 1453 § 45.1-161.9, repealing SB 1453 § 45.1-161.10, repealing SB 1453 § 45.1-161.11, repealing SB 1453 § 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.16, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 </td <td>§ 4</td> <td>45.1-161.4, repealing</td> <td>SB 1453</td>	§ 4	45.1-161.4, repealing	SB 1453
§ 45.1-161.6, repealing SB 1453 § 45.1-161.8, repealing SB 1453 § 45.1-161.9, repealing SB 1453 § 45.1-161.10, repealing SB 1453 § 45.1-161.11, repealing SB 1453 § 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.16, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 <td>§ 4</td> <td>45.1-161.5, repealing</td> <td>SB 1453</td>	§ 4	45.1-161.5, repealing	SB 1453
§ 45.1-161.7, repealing SB 1453 § 45.1-161.8, repealing SB 1453 § 45.1-161.10, repealing SB 1453 § 45.1-161.11, repealing SB 1453 § 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.16, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.29, repealing SB 1453 </td <td>§ 4</td> <td>45.1-161.5:1, repealing</td> <td>SB 1453</td>	§ 4	45.1-161.5:1, repealing	SB 1453
§ 45.1-161.8, repealing SB 1453 § 45.1-161.10, repealing SB 1453 § 45.1-161.11, repealing SB 1453 § 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.16, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453<	§ 4	45.1-161.6, repealing	SB 1453
§ 45.1-161.9, repealing SB 1453 § 45.1-161.10, repealing SB 1453 § 45.1-161.11, repealing SB 1453 § 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.16, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453<	§ 4	15.1-161.7, repealing	SB 1453
\$ 45.1-161.10, repealing	§ 4	15.1-161.8, repealing	SB 1453
\$ 45.1-161.11, repealing	§ 4	15.1-161.9, repealing	SB 1453
§ 45.1-161.12, repealing SB 1453 § 45.1-161.13, repealing SB 1453 § 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.10, repealing	SB 1453
\$ 45.1-161.13, repealing	§ 4	15.1-161.11, repealing	SB 1453
§ 45.1-161.14, repealing SB 1453 § 45.1-161.15, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.12, repealing	SB 1453
§ 45.1-161.15, repealing SB 1453 § 45.1-161.16, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.13, repealing	SB 1453
§ 45.1-161.16, repealing SB 1453 § 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.14, repealing	SB 1453
§ 45.1-161.17, repealing SB 1453 § 45.1-161.18, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.15, repealing	SB 1453
§ 45.1-161.18, repealing SB 1453 § 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.16, repealing	SB 1453
§ 45.1-161.19, repealing SB 1453 § 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.17, repealing	SB 1453
§ 45.1-161.20, repealing SB 1453 § 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.18, repealing	SB 1453
§ 45.1-161.21, repealing SB 1453 § 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.19, repealing	SB 1453
§ 45.1-161.22, repealing SB 1453 § 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	45.1-161.20, repealing	SB 1453
§ 45.1-161.23, repealing SB 1453 § 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	45.1-161.21, repealing	SB 1453
§ 45.1-161.24, repealing SB 1453 § 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.22, repealing	SB 1453
§ 45.1-161.25, repealing SB 1453 § 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	45.1-161.23, repealing	SB 1453
§ 45.1-161.26, repealing SB 1453 § 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	45.1-161.24, repealing	SB 1453
§ 45.1-161.27, repealing SB 1453 § 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.25, repealing	SB 1453
§ 45.1-161.28, repealing SB 1453 § 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.26, repealing	SB 1453
§ 45.1-161.29, repealing SB 1453 § 45.1-161.30, repealing SB 1453 § 45.1-161.31, repealing SB 1453 § 45.1-161.32, repealing SB 1453	§ 4	15.1-161.27, repealing	SB 1453
\$ 45.1-161.30, repealing	§ 4	15.1-161.28, repealing	SB 1453
\$ 45.1-161.31, repealing	§ 4	15.1-161.29, repealing	SB 1453
§ 45.1-161.32, repealing	§ 4	45.1-161.30, repealing	SB 1453
	§ 4	15.1-161.31, repealing	SB 1453
§ 45.1-161.33, repealing	§ 4	15.1-161.32, repealing	SB 1453
	§ 4	45.1-161.33, repealing	SB 1453

CC	DDE OF VIRGINIA (continued)		
	§ 45.1-161.80, repealing	SB	1453
	§ 45.1-161.81, repealing	SB	1453
	§ 45.1-161.82, repealing	SB	1453
	§ 45.1-161.83, repealing	SB	1453
	§ 45.1-161.84, repealing	SB	1453
	§ 45.1-161.85, repealing	SB	1453
	§ 45.1-161.86, repealing	SB	1453
	§ 45.1-161.87, repealing	SB	1453
	§ 45.1-161.88, repealing	SB	1453
	§ 45.1-161.89, repealing	SB	1453
	§ 45.1-161.90, repealing	SB	1453
	§ 45.1-161.91, repealing	SB	1453
	§ 45.1-161.92, repealing	SB	1453
	§ 45.1-161.93, repealing	SB	1453
	§ 45.1-161.94, repealing	SB	1453
	§ 45.1-161.95, repealing	SB	1453
	§ 45.1-161.96, repealing	SB	1453
	§ 45.1-161.97, repealing	SB	1453
	§ 45.1-161.98, repealing	SB	1453
	§ 45.1-161.99, repealing	SB	1453
	§ 45.1-161.100, repealing	SB	1453
	§ 45.1-161.101, repealing	SB	1453
	§ 45.1-161.102, repealing	SB	1453
	§ 45.1-161.103, repealing	SB	1453
	§ 45.1-161.104, repealing	SB	1453
	§ 45.1-161.105, repealing	SB	1453
	§ 45.1-161.106, repealing	SB	1453
	§ 45.1-161.107, repealing	SB	1453
	§ 45.1-161.108, repealing	SB	1453
	§ 45.1-161.109, repealing	SB	1453
	§ 45.1-161.110, repealing	SB	1453
	§ 45.1-161.111, repealing	SB	1453

CODE OF MINCIPAL (
CODE OF VIRGINIA (continued) § 45.1-161.145, repealing	SB 1453
§ 45.1-161.146, repealing	SB 1453
§ 45.1-161.147, repealing	SB 1453
§ 45.1-161.148, repealing	SB 1453
§ 45.1-161.149, repealing	SB 1453
§ 45.1-161.150, repealing	SB 1453
§ 45.1-161.151, repealing	SB 1453
§ 45.1-161.152, repealing	SB 1453
§ 45.1-161.153, repealing	SB 1453
§ 45.1-161.154, repealing	SB 1453
§ 45.1-161.155, repealing	SB 1453
§ 45.1-161.156, repealing	SB 1453
§ 45.1-161.157, repealing	SB 1453
§ 45.1-161.158, repealing	SB 1453
§ 45.1-161.159, repealing	SB 1453
§ 45.1-161.160, repealing	SB 1453
§ 45.1-161.161, repealing	SB 1453
§ 45.1-161.162, repealing	SB 1453
§ 45.1-161.163, repealing	SB 1453
§ 45.1-161.164, repealing	SB 1453
§ 45.1-161.165, repealing	SB 1453
§ 45.1-161.166, repealing	SB 1453
§ 45.1-161.167, repealing	SB 1453
§ 45.1-161.168, repealing	SB 1453
§ 45.1-161.169, repealing	SB 1453
§ 45.1-161.170, repealing	SB 1453
§ 45.1-161.171, repealing	SB 1453
§ 45.1-161.172, repealing	SB 1453
§ 45.1-161.173, repealing	SB 1453
§ 45.1-161.174, repealing	SB 1453
§ 45.1-161.175, repealing	SB 1453
§ 45.1-161.176, repealing	SB 1453

O	DE OF VIRGINIA (continued) § 45.1-161.209, repealing	SB	1453
	§ 45.1-161.210, repealing	SB	1453
	§ 45.1-161.211, repealing	SB	1453
	§ 45.1-161.212, repealing	SB	1453
	§ 45.1-161.213, repealing	SB	1453
	§ 45.1-399, repealing	SB	1453
	§ 45.1-161.2, amending	. HB	1855
	§ 45.1-161.4:1, adding	. HB	1855
	§ 45.1-161.5, amending	. HB	1855
	§ 45.1-161.15, amending	. HB	1855
	§ 45.1-161.179, amending	. HB	1855
	§ 45.1-161.292:2, amending	. HB	1855
	§ 45.1-161.322, amending	SB	1261
	§ 45.1-180, amending	. HB	1855
	§ 45.1-229, amending	. HB	1855
	§ 45.1-230, amending	. HB	1855
	§ 45.1-246, amending	SB	1399
	§ 45.1-270.4:1, amending	. HB	1855
	§ 45.1-361.28, amending	. HB	1855
	§ 45.1-361.41, amending	. HB	1855
	§ 45.1-383, amending		
	§ 45.1-390, amending		
	§ 45.1-394.1, adding	, SB	1247
	§ 45.2-100, adding	SB	1453
	§ 45.2-101, adding		
	§ 45.2-102, adding		
	§ 45.2-103, adding		
	§ 45.2-104, adding		
	§ 45.2-105, adding		
	§ 45.2-106, adding		
	§ 45.2-107, adding		
	8 45 2-108. adding		1453

CODE OF LIBORAL (
CODE OF VIRGINIA (continued) § 45.2-519, adding	SB 1453
§ 45.2-520, adding	
§ 45.2-521, adding	
§ 45.2-522, adding	
§ 45.2-523, adding	
§ 45.2-524, adding	
§ 45.2-525, adding	
§ 45.2-526, adding	
§ 45.2-527, adding	
§ 45.2-528, adding	
§ 45.2-529, adding	
§ 45.2-530, adding	
§ 45.2-531, adding	
§ 45.2-532, adding	
§ 45.2-533, adding	
§ 45.2-534, adding	
§ 45.2-535, adding	
§ 45.2-536, adding	
§ 45.2-537, adding	
§ 45.2-538, adding	
§ 45.2-539, adding	
§ 45.2-540, adding	
§ 45.2-541, adding	
§ 45.2-542, adding	
§ 45.2-543, adding	
§ 45.2-544, adding	
§ 45.2-545, adding	
§ 45.2-546, adding	
§ 45.2-547, adding	
§ 45.2-548, adding	
§ 45.2-549, adding	
§ 45.2-550, adding	SB 1453

conn on the contract of the co		
§ 45.2-601, addir	gSB 145	3
§ 45.2-602, addir	g	3
§ 45.2-603, addir	g	3
§ 45.2-604, addir	g	3
§ 45.2-605, addir	g	3
§ 45.2-606, addir	g	3
§ 45.2-607, addir	g	3
§ 45.2-608, addir	g	3
§ 45.2-609, addir	g	3
§ 45.2-610, addir	g	3
§ 45.2-611, addir	g	3
§ 45.2-612, addir	g	3
§ 45.2-613, addir	g	3
§ 45.2-614, addir	g	3
§ 45.2-615, addir	g	3
§ 45.2-616, addir	g	3
§ 45.2-617, addir	g	3
§ 45.2-618, addir	g	3
§ 45.2-619, addir	g	3
§ 45.2-620, addir	g	3
§ 45.2-700, addir	g	3
§ 45.2-701, addir	g	3
§ 45.2-702, addir	g	3
§ 45.2-703, addir	g	3
§ 45.2-704, addir	g	3
§ 45.2-705, addir	g	3
§ 45.2-706, addir	g	3
§ 45.2-707, addir	g	3
§ 45.2-708, addir	g	3
§ 45.2-709, addir	g	3
§ 45.2-710, addir	g	3
§ 45.2-711, addin	g	3

CO	DDE OF VIRGINIA (continued)	CD	1450
	§ 45.2-744, adding		
	§ 45.2-745, adding		
	§ 45.2-746, adding		
	§ 45.2-747, adding	SB	1453
	§ 45.2-748, adding	SB	1453
	§ 45.2-749, adding	SB	1453
	§ 45.2-750, adding	SB	1453
	§ 45.2-751, adding	SB	1453
	§ 45.2-752, adding	SB	1453
	§ 45.2-753, adding	SB	1453
	§ 45.2-754, adding	SB	1453
	§ 45.2-755, adding	SB	1453
	§ 45.2-756, adding	SB	1453
	§ 45.2-757, adding	SB	1453
	§ 45.2-758, adding	SB	1453
	§ 45.2-759, adding	SB	1453
	§ 45.2-760, adding	SB	1453
	§ 45.2-761, adding	SB	1453
	§ 45.2-762, adding	SB	1453
	§ 45.2-763, adding	SB	1453
	§ 45.2-764, adding	SB	1453
	§ 45.2-765, adding	SB	1453
	§ 45.2-766, adding	SB	1453
	§ 45.2-767, adding	SB	1453
	§ 45.2-768, adding	SB	1453
	§ 45.2-769, adding	SB	1453
	§ 45.2-770, adding	SB	1453
	§ 45.2-771, adding	SB	1453
	§ 45.2-772, adding	SB	1453
	§ 45.2-773, adding	SB	1453
	§ 45.2-774, adding	SB	1453
	8 45.2-775, adding	SB	1453

CODE OF VIII	DCDHA (
	RGINIA (continued) 1, adding	SB	1453
§ 45.2-832	2, adding	SB	1453
§ 45.2-833	3, adding	SB	1453
§ 45.2-83 ⁴	4, adding	SB	1453
§ 45.2-835	5, adding	SB	1453
§ 45.2-836	6, adding	SB	1453
§ 45.2-837	7, adding	SB	1453
§ 45.2-838	8, adding	SB	1453
§ 45.2-839	9, adding	SB	1453
§ 45.2-840	0, adding	SB	1453
§ 45.2-841	1, adding	SB	1453
§ 45.2-842	2, adding	SB	1453
§ 45.2-843	3, adding	SB	1453
§ 45.2-84 ⁴	4, adding	SB	1453
§ 45.2-845	5, adding	SB	1453
§ 45.2-846	6, adding	SB	1453
§ 45.2-847	7, adding	SB	1453
§ 45.2-848	8, adding	SB	1453
§ 45.2-849	9, adding	SB	1453
§ 45.2-850	0, adding	SB	1453
§ 45.2-851	1, adding	SB	1453
§ 45.2-852	2, adding	SB	1453
§ 45.2-853	3, adding	SB	1453
§ 45.2-85 ⁴	4, adding	SB	1453
§ 45.2-855	5, adding	SB	1453
§ 45.2-856	6, adding	SB	1453
§ 45.2-857	7, adding	SB	1453
§ 45.2-858	8, adding	SB	1453
§ 45.2-859	9, adding	SB	1453
§ 45.2-860	0, adding	SB	1453
§ 45.2-861	1, adding	SB	1453
§ 45.2-862	2, adding	SB	1453

CODE OF VIRGINIA (continued)	ap 4.450
§ 45.2-931, adding	
§ 45.2-932, adding	SB 1453
§ 45.2-933, adding	SB 1453
§ 45.2-934, adding	SB 1453
§ 45.2-935, adding	SB 1453
§ 45.2-936, adding	SB 1453
§ 45.2-937, adding	SB 1453
§ 45.2-938, adding	SB 1453
§ 45.2-939, adding	SB 1453
§ 45.2-1000, adding	SB 1453
§ 45.2-1001, adding	SB 1453
§ 45.2-1002, adding	SB 1453
§ 45.2-1003, adding	SB 1453
§ 45.2-1004, adding	SB 1453
§ 45.2-1005, adding	SB 1453
§ 45.2-1006, adding	SB 1453
§ 45.2-1007, adding	SB 1453
§ 45.2-1008, adding	SB 1453
§ 45.2-1009, adding	SB 1453
§ 45.2-1010, adding	SB 1453
§ 45.2-1011, adding	SB 1453
§ 45.2-1012, adding	SB 1453
§ 45.2-1013, adding	SB 1453
§ 45.2-1014, adding	SB 1453
§ 45.2-1015, adding	SB 1453
§ 45.2-1016, adding	SB 1453
§ 45.2-1017, adding	SB 1453
§ 45.2-1018, adding	SB 1453
§ 45.2-1019, adding	SB 1453
§ 45.2-1020, adding	SB 1453
§ 45.2-1021, adding	SB 1453
§ 45.2-1022, adding	SB 1453

-817-

INDEX

CO	DDE OF VIRGINIA (continued)	
	§ 45.2-1103, adding	SB 1453
	§ 45.2-1104, adding	SB 1453
	§ 45.2-1105, adding	SB 1453
	§ 45.2-1106, adding	SB 1453
	§ 45.2-1107, adding	SB 1453
	§ 45.2-1108, adding	SB 1453
	§ 45.2-1109, adding	SB 1453
	§ 45.2-1110, adding	SB 1453
	§ 45.2-1111, adding	SB 1453
	§ 45.2-1112, adding	SB 1453
	§ 45.2-1113, adding	SB 1453
	§ 45.2-1114, adding	SB 1453
	§ 45.2-1115, adding	SB 1453
	§ 45.2-1116, adding	SB 1453
	§ 45.2-1117, adding	SB 1453
	§ 45.2-1118, adding	SB 1453
	§ 45.2-1119, adding	SB 1453
	§ 45.2-1120, adding	SB 1453
	§ 45.2-1121, adding	SB 1453
	§ 45.2-1122, adding	SB 1453
	§ 45.2-1123, adding	SB 1453
	§ 45.2-1124, adding	SB 1453
	§ 45.2-1125, adding	SB 1453
	§ 45.2-1126, adding	SB 1453
	§ 45.2-1127, adding	SB 1453
	§ 45.2-1128, adding	SB 1453
	§ 45.2-1129, adding	SB 1453
	§ 45.2-1130, adding	SB 1453
	§ 45.2-1131, adding	SB 1453
	§ 45.2-1132, adding	SB 1453
	§ 45.2-1133, adding	SB 1453
	8 45.2-1134, adding	SB 1453

CC	DDE OF VIRGINIA (continued)	
	§ 45.2-1167, adding	SB 1453
	§ 45.2-1200, adding	SB 1453
	§ 45.2-1201, adding	SB 1453
	§ 45.2-1202, adding	SB 1453
	§ 45.2-1203, adding	SB 1453
	§ 45.2-1204, adding	SB 1453
	§ 45.2-1205, adding	SB 1453
	§ 45.2-1206, adding	SB 1453
	§ 45.2-1207, adding	SB 1453
	§ 45.2-1208, adding	SB 1453
	§ 45.2-1209, adding	SB 1453
	§ 45.2-1210, adding	SB 1453
	§ 45.2-1211, adding	SB 1453
	§ 45.2-1212, adding	SB 1453
	§ 45.2-1213, adding	SB 1453
	§ 45.2-1214, adding	SB 1453
	§ 45.2-1215, adding	SB 1453
	§ 45.2-1216, adding	SB 1453
	§ 45.2-1217, adding	SB 1453
	§ 45.2-1218, adding	SB 1453
	§ 45.2-1219, adding	SB 1453
	§ 45.2-1220, adding	SB 1453
	§ 45.2-1221, adding	SB 1453
	§ 45.2-1222, adding	SB 1453
	§ 45.2-1223, adding	SB 1453
	§ 45.2-1224, adding	SB 1453
	§ 45.2-1225, adding	SB 1453
	§ 45.2-1226, adding	SB 1453
	§ 45.2-1227, adding	SB 1453
	§ 45.2-1228, adding	SB 1453
	§ 45.2-1229, adding	SB 1453
	§ 45.2-1230, adding	SB 1453

O	DE OF VIRGINIA (continued) § 45.2-1602, adding	CD	1/153
	§ 45.2-1603, adding		
	§ 45.2-1604, adding		
	§ 45.2-1605, adding		
	§ 45.2-1606, adding		
	§ 45.2-1607, adding		
	§ 45.2-1608, adding		
	§ 45.2-1609, adding		
	§ 45.2-1610, adding		
	§ 45.2-1611, adding		
	§ 45.2-1612, adding		
	§ 45.2-1613, adding	SB	1453
	§ 45.2-1614, adding	SB	1453
	§ 45.2-1615, adding	SB	1453
	§ 45.2-1616, adding	SB	1453
	§ 45.2-1617, adding	SB	1453
	§ 45.2-1618, adding	SB	1453
	§ 45.2-1619, adding	SB	1453
	§ 45.2-1620, adding	SB	1453
	§ 45.2-1621, adding	SB	1453
	§ 45.2-1622, adding	SB	1453
	§ 45.2-1623, adding	SB	1453
	§ 45.2-1624, adding	SB	1453
	§ 45.2-1625, adding	SB	1453
	§ 45.2-1626, adding	SB	1453
	§ 45.2-1627, adding	SB	1453
	§ 45.2-1628, adding	SB	1453
	§ 45.2-1629, adding	SB	1453
	§ 45.2-1630, adding		
	§ 45.2-1631, adding		
	§ 45.2-1632, adding		
	8 45.2-1633, adding		1453

CO	DE OF VIRGINIA (continued)	CD	1.450
	§ 45.2-1716, adding		
	§ 45.2-1717, adding		
	§ 45.2-1718, adding	SB	1453
	§ 45.2-1719, adding	SB	1453
	§ 45.2-1720, adding	SB	1453
	§ 45.2-1721, adding	SB	1453
	§ 45.2-1722, adding	SB	1453
	§ 45.2-1723, adding	SB	1453
	§ 45.2-1724, adding	SB	1453
	§ 45.2-1800, adding	SB	1453
	§ 45.2-1801, adding	SB	1453
	§ 45.2-1802, adding	SB	1453
	§ 45.2-1803, adding	SB	1453
	§ 45.2-1804, adding	SB	1453
	§ 45.2-1805, adding	SB	1453
	§ 45.2-1806, adding	SB	1453
	§ 45.2-1807, adding	SB	1453
	§ 45.2-1808, adding	SB	1453
	§ 45.2-1809, adding	SB	1453
	§ 45.2-1810, adding	SB	1453
	§ 45.2-1811, adding	SB	1453
	§ 45.2-1812, adding	SB	1453
	§ 45.2-1813, adding	SB	1453
	§ 45.2-1900, adding	SB	1453
	§ 45.2-1901, adding	SB	1453
	§ 45.2-1902, adding	SB	1453
	§ 45.2-1903, adding	SB	1453
	§ 45.2-1904, adding		
	§ 45.2-1905, adding		
	§ 45.2-1906, adding		
	§ 45.2-1907, adding		
	8 45.2-1908, adding		1453

INDEX

CO	PDE OF VIRGINIA (continued) § 45.2-2114, adding	SB 1453
	§ 45.2-2115, adding	
	§ 45.2-2116, adding	
	§ 45.2-2117, adding	
	§ 45.2-2118, adding	
	§ 45.2-2119, adding	
	§ 46.2-100, amending	
	§ 46.2-105.2, amending	
	§ 46.2-105.2, amending	
	§ 46.2-202, amending	
	§ 46.2-203.2, amending	
	§ 46.2-208, amending	
	§ 46.2-209, amending	
	§ 46.2-209.1, amending	
	§ 46.2-216.1, amending	
	§ 46.2-223, amending	
	§ 46.2-301, amending	
	§ 46.2-301.1, amending	
	§ 46.2-323.01, amending	
	§ 46.2-323.1, amending	
	§ 46.2-325, amending	
	§ 46.2-328.3, amending	
	§ 46.2-334.01, amending	НВ 1846
	§ 46.2-335, amending	SB 1335
	§ 46.2-341.18, amending	НВ 1868
	§ 46.2-341.20:7, adding	
	§ 46.2-341.20:7, adding	SB 1406, SB 1406, SB 1406
	§ 46.2-341.28, amending	SB 1443
	§ 46.2-345, amending	НВ 1961
	§ 46.2-345.3, adding	
	§ 46.2-346, amending	HB 2138
	8 46.2-347, amending	HB 2312

CODE OF VIRGINIA (continued)	
§ 46.2-347, amending	SB 1406, SB 1406, SB 1406
§ 46.2-355.1, repealing	SB 1122
§ 46.2-356, repealing	SB 1122
§ 46.2-357, repealing	SB 1122
§ 46.2-358, repealing	SB 1122
§ 46.2-359, repealing	SB 1122
§ 46.2-360, repealing	SB 1122
§ 46.2-361, repealing	SB 1122
§ 46.2-362, repealing	SB 1122
§ 46.2-363, repealing	SB 1122
§ 46.2-357, amending	HB 2331, SB 1443
§ 46.2-382, amending	HB 1868
§ 46.2-391, amending	HB 2331, SB 1443
§ 46.2-411, amending	SB 1122
§ 46.2-419, amending	SB 1182
§ 46.2-472, amending	SB 1182
§ 46.2-600.1, amending	HB 1960, HB 2138, SB 1470
§ 46.2-602, amending	
§ 46.2-629, amending	HB 2294
§ 46.2-644.01, amending	SB 1160
§ 46.2-644.02, amending	SB 1160
§ 46.2-644.03, amending	SB 1160
§ 46.2-644.04, adding	SB 1160
§ 46.2-686, amending	SB 1211
§ 46.2-694, amending	SB 1211
§ 46.2-694.2, adding	SB 1211
§ 46.2-725.3, adding	HB 2069
§ 46.2-742.1, amending	HB 2069
§ 46.2-742.2, amending	HB 2069
§ 46.2-744, amending	HB 2261
§ 46.2-745.1, amending	HB 2069
§ 46.2-745.2, amending	HB 2069

CO	DDE OF VIRGINIA (continued) § 46.2-745.4, adding	HB 2060
	§ 46.2-745.5, adding	
	§ 46.2-746.5, amending	
	§ 46.2-746.6, repealing	
	§ 46.2-746.9, repealing	
	§ 46.2-746.12, repealing	
	§ 46.2-747, repealing	
	§ 46.2-748, repealing	
	§ 46.2-749.7, amending	
	§ 46.2-749.10, repealing	
	§ 46.2-749.69:1, repealing	
	§ 46.2-819.11, adding	
	§ 46.2-839, amending	
	§ 46.2-861.1, amending	
	§ 46.2-865.1, amending	
	§ 46.2-889, amending	HB 2146
	§ 46.2-903.1, adding	HB 2262, SB 1263
	§ 46.2-905, amending	HB 2262, SB 1263
	§ 46.2-916.3, amending	SB 1337
	§ 46.2-923, amending	SB 1308
	§ 46.2-936, amending	SB 1329, SB 1437
	§ 46.2-940, amending	SB 1329, SB 1437
	§ 46.2-1012, amending	SB 1347
	§ 46.2-1063, amending	SB 1136
	§ 46.2-1129.2, amending	НВ 1850
	§ 46.2-1193, adding	SB 1139
	§ 46.2-1200.2, amending	SB 1160
	§ 46.2-1200.3, adding	SB 1160
	§ 46.2-1202, amending	SB 1160
	§ 46.2-1202.1, amending	SB 1160
	§ 46.2-1202.2, adding	SB 1160
	8 46.2-1203, amending	SB 1160

-829-

INDEX

2021 SENATE JOURNAL

CODE OF MECHANICAL AND A	
§ 51.5-134, amending	
§ 51.5-135, amending	
§ 51.5-160, amending	SB 1185
§ 52-3, amending	
§ 52-8.1:1, amending	SB 1243
§ 52-8.5, amending	SB 1203
§ 52-30.1, amending	HB 1948
§ 52-34.13, amending	
§ 52-34.14, amending	
§ 52-34.15, amending	
§ 52-35, amending	SB 1243
§ 52-48, amending	SB 1198
§ 52-50, adding	SB 569
§ 53.1-1, amending	SB 1179
§ 53.1-2, amending	SB 1363
§ 53.1-4, amending	SB 1363
§ 53.1-5, amending	SB 1363
§ 53.1-11, amending	
§ 53.1-21, amending	SB 1122
§ 53.1-31.1, amending	SB 1179
§ 53.1-32, amending	HB 281
§ 53.1-39.2, adding	SB 1301
§ 53.1-40.01, amending	
§ 53.1-68, amending	НВ 1874
§ 53.1-136, amending	HB 2167, SB 1104, SB 1125, SB 1244, SB 1293, SB 1397
§ 53.1-145, amending	SB 1244
§ 53.1-155, amending	HB 2167, SB 1104, SB 1125
§ 53.1-165.1, repealing	SB 91, SB 1370
§ 53.1-165.2, adding	SB 91
§ 53.1-202.3, amending	НВ 2010
§ 53.1-203, amending	
§ 53.1-204, amending	HB 2263, SB 449, SB 1165

CODE OF VIRGINIA (continued) § 53.1-220.1, amending	SB 1243
§ 53.1-229, amending	HB 2263, SB 449, SB 1165
§ 53.1-230, repealing	HB 2263, SB 449, SB 1165
§ 53.1-231.2, amending	HB 2312, SB 1243
§ 53.1-231.2, amending	SB 1406, SB 1406, SB 1406
§ 53.1-232, repealing	НВ 2263
§ 53.1-232.1, repealing	НВ 2263
§ 53.1-233, repealing	НВ 2263
§ 53.1-234, repealing	НВ 2263
§ 53.1-235, repealing	НВ 2263
§ 53.1-236, repealing	НВ 2263
§ 53.1-232, repealing	SB 449
§ 53.1-232.1, repealing	SB 449
§ 53.1-233, repealing	SB 449
§ 53.1-234, repealing	SB 449
§ 53.1-235, repealing	SB 449
§ 53.1-236, repealing	SB 449
§ 53.1-232, repealing	SB 1165
§ 53.1-232.1, repealing	SB 1165
§ 53.1-233, repealing	SB 1165
§ 53.1-234, repealing	SB 1165
§ 53.1-235, repealing	SB 1165
§ 53.1-236, repealing	SB 1165
§ 53.1-261, amending	SB 1179
§ 53.1-262, amending	SB 1179
§ 53.1-263, repealing	SB 1179
§ 53.1-264, repealing	SB 1179
§ 53.1-265, amending	SB 1179
§ 53.1-266, repealing	SB 1179
§ 54.1-305, amending	НВ 2223
§ 54.1-306, amending	SB 1437
8 54.1-1141, amending	НВ 2202

CODE OF VIRGINIA (continued)	
	SB 346
§ 54.1-1150, adding	SB 346
§ 54.1-1156, adding	SB 346
§ 54.1-2019.1, adding	SB 1324
§ 54.1-2024, adding	SB 1324
§ 54.1-2108.1, amending	
§ 54.1-2400.03, adding	
§ 54.1-2400.6, amending	
§ 54.1-2506, amending	
§ 54.1-2519, amending	
§ 54.1-2521, amending	
§ 54.1-2800, amending	HB 2005, SB 1268, SB 1424
§ 54.1-2807, amending	
§ 54.1-2807.01, repealing	
§ 54.1-2807.02, repealing	
§ 54.1-2810, amending	SB 1424
§ 54.1-2825, amending	
§ 54.1-2825.1, adding	
§ 54.1-2900, amending	HB 1747, HB 1953, SB 1218, SB 1320
§ 54.1-2901, amending	
§ 54.1-2902, amending	НВ 2039
§ 54.1-2903, amending	HB 2218, HB 2312, SB 1243, SB 1333
§ 54.1-2903, amending	SB 1406, SB 1406, SB 1406
§ 54.1-2909, amending	
§ 54.1-2914, amending	SB 1218
§ 54.1-2950.1, amending	НВ 2039
§ 54.1-2951.1, amending	НВ 2039
§ 54.1-2951.2, amending	НВ 2039
§ 54.1-2951.4, adding	НВ 2039
§ 54.1-2952, amending	НВ 2039
§ 54.1-2952.1, amending	НВ 2039
§ 54.1-2953, amending	HB 2039

CODE OF VIRGINIA (continued)	HD 2210
§ 54.1-3442.7, amending	
§ 54.1-3442.8, amending	
§ 54.1-3442.5, amending	
§ 54.1-3442.6, amending	SB 1333
§ 54.1-3442.7, amending	SB 1333
§ 54.1-3442.8, amending	SB 1333
§ 54.1-3442.6, amending	
§ 54.1-3442.6, amending	SB 1406, SB 1406, SB 1406
§ 54.1-3442.7, amending	
§ 54.1-3442.8, amending	HB 2312, SB 1243
§ 54.1-3442.8, amending	SB 1406, SB 1406, SB 1406
§ 54.1-3446, amending	НВ 2078
§ 54.1-3446, amending	
§ 54.1-3446, amending	SB 1464
§ 54.1-3482, amending	SB 1187
§ 54.1-3915.1, repealing	HB 1853
§ 54.1-3916, amending	HB 1853
§ 54.1-3926, amending	SB 1234
§ 54.1-4407, amending	SB 1437
§ 55.1-318.1, adding	НВ 1882
§ 55.1-319, amending	НВ 1882
§ 55.1-320, amending	HB 2175, SB 253, SB 1327
§ 55.1-321, amending	HB 2175, SB 1327
§ 55.1-339, amending	НВ 2099
§ 55.1-604.1, adding	SB 359, SB 1140
§ 55.1-606, amending	НВ 2064
§ 55.1-703, amending	НВ 1824
§ 55.1-703, amending	HB 2320, SB 1389
§ 55.1-708.2, adding	HB 2320, SB 1389
§ 55.1-1004, amending	SB 1110
§ 55.1-1200, amending	HB 2249
8 55.1-1204, amending	НВ 2249

O.	DE OF VIRGINIA (continued) § 55.1-2211, amending	SB	1261
	§ 56-46.1, amending		
	§ 56-265.15:1, amending		
	§ 56-484.12, amending		
	§ 56-484.17;1, amending		
	§ 56-484.18:1, adding		
	§ 56-535, repealing		
	§ 56-536, repealing		
	§ 56-537, repealing		
	§ 56-538, repealing		
	§ 56-539, repealing		
	§ 56-540, repealing		
	§ 56-541, repealing	HB 2	2104
	§ 56-542, repealing	HB 2	2104
	§ 56-543, repealing	HB 2	2104
	§ 56-544, repealing	HB 2	2104
	§ 56-545, repealing	HB 2	2104
	§ 56-546, repealing	HB 2	2104
	§ 56-547, repealing	HB 2	2104
	§ 56-548, repealing	HB 2	2104
	§ 56-549, repealing	HB 2	2104
	§ 56-550, repealing	HB 2	2104
	§ 56-551, repealing	HB 2	2104
	§ 56-552, repealing	HB 2	2104
	§ 56-539, amending	2, SB	1259
	§ 56-542, amending	2, SB	1259
	§ 56-576, amending	, HB 2	2330
	§ 56-577, amending	HB 2	2048
	§ 56-585.1, amendingHB 1914, HB 2049, HB 2160, HB 2200, SB 128-	l, SB	1292
	§ 56-585.1:9, amending	l, SB	1413
	8 56-585 1:11, adding	2. SB	1295

ODE OF VIRGINIA (continued) § 56-585.1:13, adding	CD 1290
§ 56-585.5, amending	
§ 56-585.6, amending	
§ 56-594.2, amending	
§ 56-594.3, amending	
§ 56-596.2, amending	
§ 56-598, amending	
§ 56-599, amending	
§ 56-601, amending	SB 1284
§ 56-614, adding	SB 1453
§ 56-615, adding	
§ 56-616, adding	
§ 56-617, adding	
§ 56-618, adding	SB 1453
§ 56-619, adding	SB 1453
§ 56-620, adding	SB 1453
§ 56-621, adding	SB 1453
§ 56-622, adding	SB 1453
§ 56-623, adding	SB 1453
§ 56-624, adding	SB 1453
§ 57-2.02, amending	SB 1261
§ 57-20, amending	НВ 2308
§ 57-27.3, amending	HB 2005
§ 57-27.4, adding	HB 2005
§ 57-27.5, adding	HB 2005
§ 57-27.6, adding	HB 2005
§ 57-48, amending	SB 1138
§ 58.1-3, amending	
§ 58.1-3, amending	
§ 58.1-3, amending	
§ 58.1-112, amending	
8 58 1-201, amending	

CODE OF VIRGINIA (continued) § 58.1-301, amending	HB 734, HB 1935, SB 1146
§ 58.1-302, amending	SB 1407
§ 58.1-320, amending	SB 1286
§ 58.1-321, amending	SB 1394
§ 58.1-322.02, amending	НВ 1619
§ 58.1-322.02, amending	НВ 1935
§ 58.1-322.02, amending	HB 2337, SB 191, SB 456, SB 965
§ 58.1-322.02, amending	SB 1146
§ 58.1-322.02, amending	SB 1151, SB 1407, SB 1448
§ 58.1-322.03, amending	HB 1935, SB 1146
§ 58.1-327, adding	SB 1286
§ 58.1-334, amending	SB 1163
§ 58.1-337, amending	SB 1163
§ 58.1-339.3, amending	HB 1763, SB 1162
§ 58.1-339.13, adding	SB 218, SB 457, SB 460, SB 1166
§ 58.1-341.1, amending	HB 1884
§ 58.1-344.3, amending	НВ 1836
§ 58.1-400.1, amending	SB 1353
§ 58.1-402, amending	SB 1146
§ 58.1-406, amending	SB 756, SB 1353
§ 58.1-424, adding	SB 756
§ 58.1-425, adding	SB 756
§ 58.1-426, adding	SB 756
§ 58.1-427, adding	SB 756
§ 58.1-428, adding	SB 756
§ 58.1-429, adding	SB 756
§ 58.1-424, adding	SB 1353
§ 58.1-425, adding	SB 1353
§ 58.1-426, adding	SB 1353
§ 58.1-427, adding	SB 1353
§ 58.1-432, amending	SB 1163
8 58 1-433 1, amending	HB 1899, SB 1252

CODE OF VIRGINIA (continued)	
§ 58.1-436, amending	SB 1163
§ 58.1-439.2, amending	HB 1855, HB 1899, SB 1252
§ 58.1-439.5, amending	HB 1763, SB 1162
§ 58.1-439.12:02, amending	HB 1855
§ 58.1-439.12:06, amending	SB 1158
§ 58.1-439.12:08, amending	HB 1916, SB 1112
§ 58.1-439.12:09, amending	SB 1158
§ 58.1-439.12:10, amending	SB 1158
§ 58.1-439.12:11, amending	HB 1916, SB 1112
§ 58.1-439.29, adding	SB 1197
§ 58.1-439.30, adding	SB 1197
§ 58.1-442, amending	SB 1353
§ 58.1-443, amending	SB 756, SB 1353
§ 58.1-460, amending	SB 1407
§ 58.1-527, amending	SB 1261
§ 58.1-602, amending	SB 1170, SB 1398
§ 58.1-603, amending	SB 1398
§ 58.1-605, amending	SB 1170
§ 58.1-609.3, amending	HB 2273, SB 1423, SB 1425
§ 58.1-609.14, adding	HB 2185, SB 1403
§ 58.1-612.2, adding	SB 1398
§ 58.1-802.3, amending	НВ 729, НВ 729
§ 58.1-802.4, adding	НВ 729
§ 58.1-811, amending	НВ 729
§ 58.1-816, amending	НВ 729
§ 58.1-901, amending	SB 637
§ 58.1-902, amending	SB 637
§ 58.1-905, amending	SB 637
§ 58.1-912, amending	SB 637
§ 58.1-1743, amending	НВ 729, НВ 729
§ 58.1-1744, amending	НВ 729
8 58 1-1802 2 adding	HR 2050

§ 58.1-1828, amending	SB 1261
§ 58.1-2282, amending	SB 1261
§ 58.1-2600, amending	HB 2006, SB 1201
§ 58.1-2626.1, amending	HB 1899, SB 1252
§ 58.1-2628, amending	HB 2006, SB 1201
§ 58.1-2636, amending	HB 2006, HB 2269, SB 1201
§ 58.1-3147, amending	SB 1261
§ 58.1-3221.6, amending	НВ 1969
§ 58.1-3506, amending	НВ 1774
§ 58.1-3537, adding	SB 143
§ 58.1-3538, adding	SB 143
§ 58.1-3660, amending	HB 1855
§ 58.1-3660, amending	HB 2006, SB 1201
§ 58.1-3660, amending	SB 1380
§ 58.1-3668, adding	SB 1130
§ 58.1-3706, amending	HB 1855
§ 58.1-3713, amending	HB 2293
§ 58.1-3745, amending	HB 1855
§ 58.1-3818.8, adding	SB 1398
§ 58.1-3819, amending	SB 1398
§ 58.1-3819.1, amending	SB 1398
§ 58.1-3823, amending	SB 1398
§ 58.1-3824, amending	SB 1398
§ 58.1-3825, amending	SB 1398
§ 58.1-3825.2, amending	SB 1398
§ 58.1-3825.3, amending	SB 1398
§ 58.1-3826, amending	SB 1398
§ 58.1-3830, amending	SB 1326
§ 58.1-3832.1, adding	SB 1326
§ 58.1-3842, amending	SB 1398, SB 1438
§ 58.1-3843, amending	SB 1398
§ 58.1-3965, amending	HB 2165

Ol	DE OF VIRGINIA (continued) § 58.1-4062, adding	SB	1407
	§ 58.1-4063, adding		
	§ 58.1-4064, adding		
	§ 58.1-4065, adding		
	§ 58.1-4066, adding		
	§ 58.1-4067, adding		
	§ 58.1-4068, adding		
	§ 58.1-4069, adding		
	§ 58.1-4070, adding		
	§ 58.1-4071, adding		
	§ 58.1-4072, adding		
	§ 58.1-4073, adding		
	§ 58.1-4074, adding	SB	1407
	§ 58.1-4075, adding		
	§ 58.1-4100, amending		
	§ 58.1-4109, amending		
	§ 58.1-4110, amending		
	§ 58.1-4114, amending	HB	1812
	§ 58.1-4122, amending		
	§ 58.1-4124, amending	HB	1812
	§ 58.1-4125, amending	HB	1812
	§ 59.1-148.3, amending		
	§ 59.1-148.3, amending	6, SB	1406
	§ 59.1-148.3, amending	SB	1407
	§ 59.1-200, amending	3, HB	2062
	§ 59.1-261.1, adding	SI	3 274
	§ 59.1-261.2, adding	SI	3 274
	§ 59.1-261.3, adding	SI	3 274
	§ 59.1-284.38, adding	SB	1156
	§ 59.1-392, amending	SI	3 533
	§ 59.1-441.2, amending	HB	1877
	8 59.1-481, amending	6. SB	1435

-843-

INDEX

2021 SENATE JOURNAL

O.	DE OF VIRGINIA (continued)		
	§ 59.1-572, adding	HB	2062
	§ 59.1-573, adding	HB	2062
	§ 60.2-109, amending	HB	2223
	§ 60.2-121.1, adding	HB	2036
	§ 60.2-212, amending	HB	2134
	§ 60.2-528, amending	HB	2037
	§ 60.2-528.1, amending	HB	2040
	§ 60.2-618, amending	HB	2037
	§ 60.2-619, amending	HB	2040
	§ 60.2-633, amending	HB	2040
	§ 60.2-800, adding	SB	1330
	§ 60.2-801, adding	SB	1330
	§ 60.2-802, adding	SB	1330
	§ 60.2-803, adding	SB	1330
	§ 60.2-804, adding	SB	1330
	§ 60.2-805, adding	SB	1330
	§ 60.2-806, adding	SB	1330
	§ 60.2-807, adding	SB	1330
	§ 60.2-808, adding	SB	1330
	§ 60.2-809, adding	SB	1330
	§ 60.2-810, adding	SB	1330
	§ 60.2-811, adding	SB	1330
	§ 60.2-812, adding	SB	1330
	§ 60.2-813, adding	SB	1330
	§ 60.2-814, adding	SB	1330
	§ 60.2-815, adding	SB	1330
	§ 60.2-816, adding	SB	1330
	§ 60.2-817, adding	SB	1330
	§ 60.2-818, adding	SB	1330
	§ 60.2-819, adding	SB	1330
	§ 60.2-820, adding	SB	1330
	8 62 1-44 6:1. adding HB 222	L SB	1373

 § 62.1-44.34:28, amending
 HB 1836

 § 62.1-44.117, amending
 HB 1836

 § 62.1-44.118, amending
 HB 1836

 § 62.1-44.119:1, amending
 HB 1836

DE OF VIRGINIA (continued)	HD 1936
§ 62.1-195.1, amending	HB 1836, HB 1855, SB 1453
§ 62.1-195.3, repealing	SB 1453
§ 62.1-199, amending	SB 1284
§ 62.1-223.1, adding	SB 1396
§ 62.1-223.2, adding	SB 1396
§ 62.1-223.3, adding	SB 1396
§ 62.1-243, amending	НВ 1855
§ 62.1-256, amending	НВ 1855
§ 62.1-259, amending	НВ 1855
§ 62.1-262, amending	SB 1291
§ 62.1-266.1, adding	HB 2221, SB 1373
§ 63.2-100, amending	SB 1328, SB 1371
§ 63.2-104, amending	НВ 2191
§ 63.2-105, amending	SB 1244
§ 63.2-221.1, adding	SB 1449
§ 63.2-608, amending	НВ 1820
§ 63.2-608, amending	НВ 2035
§ 63.2-801, amending	HB 1820
§ 63.2-805, amending	HB 1855
§ 63.2-806, adding	SB 1462
§ 63.2-905, amending	SB 1328
§ 63.2-906, amending	HB 1962, SB 1328
§ 63.2-910.2, amending	HB 1962
8 63.2-1306. adding	
	§ 62.1-69.31, amending

INDEX

CODE OF VIDCINIA (continued)	
CODE OF VIRGINIA (continued) § 64.2-460, adding	SB 1435
§ 64.2-461, adding	SB 1435
§ 64.2-462, adding	SB 1435
§ 64.2-463, adding	SB 1435
§ 64.2-464, adding	SB 1435
§ 64.2-465, adding	SB 1435
§ 64.2-466, adding	SB 1435
§ 64.2-467, adding	SB 1435
§ 64.2-468, adding	SB 1435
§ 64.2-701, amending	HB 1856, SB 1435
§ 64.2-1200, amending	SB 1441
§ 64.2-1202, amending	SB 1441
§ 64.2-2000, amending	HB 2230, SB 352
§ 64.2-2002, amending	HB 304
§ 64.2-2003, amending	HB 2230, SB 352
§ 64.2-2007, amending	HB 2230
§ 64.2-2008, amending	HB 2249
§ 64.2-2012, amending	HB 2249
§ 64.2-2014, amending	HB 304
§ 65.2-101, amending	HB 2032
§ 65.2-107, amending	HB 2312
§ 65.2-107, amending	SB 1406, SB 1406, SB 1406
§ 65.2-301.2, adding	HB 2134
§ 65.2-305, amending	HB 2032
§ 65.2-308, amending	HB 1754
§ 65.2-402, amending	HB 1818
§ 65.2-402, amending	HB 2312, SB 1275
§ 65.2-402, amending	SB 1406, SB 1406, SB 1406
§ 65.2-402.1, amending HB 1985, HI	B 2207, HB 2312, SB 1342, SB 1375
§ 65.2-402.1, amending	SB 1406, SB 1406, SB 1406
§ 65.2-402.1, amending	SB 1449
§ 65.2-706.2, adding	SB 1351

-849-

\mathbf{C}	DDE OF VIRGINIA (continued)	
	§ 67-601, repealing	SB 1453
	§ 67-602, repealing	SB 1453
	§ 67-603, repealing	SB 1453
	§ 67-604, repealing	SB 1453
	§ 67-700, repealing	SB 1453
	§ 67-701, repealing	SB 1453
	§ 67-800, repealing	SB 1453
	§ 67-801, repealing	SB 1453
	§ 67-900, repealing	SB 1453
	§ 67-901, repealing	SB 1453
	§ 67-902, repealing	SB 1453
	§ 67-903, repealing	SB 1453
	§ 67-1000, repealing	SB 1453
	§ 67-1001, repealing	SB 1453
	§ 67-1002, repealing	SB 1453
	§ 67-1003, repealing	SB 1453
	§ 67-1100, repealing	SB 1453
	§ 67-1101, repealing	SB 1453
	§ 67-1102, repealing	SB 1453
	§ 67-1103, repealing	SB 1453
	§ 67-1104, repealing	SB 1453
	§ 67-1105, repealing	SB 1453
	§ 67-1106, repealing	SB 1453
	§ 67-1107, repealing	SB 1453
	§ 67-1108, repealing	SB 1453
	§ 67-1109, repealing	SB 1453
	§ 67-1110, repealing	SB 1453
	§ 67-1200, repealing	SB 1453
	§ 67-1201, repealing	SB 1453
	§ 67-1202, repealing	SB 1453
	§ 67-1203, repealing	SB 1453
	8 67-1204, renealing	SB 1453

CODE OF VIRGINIA (continued) § 67-1607, repealing	
§ 67-1700, repealing	SB 1453
§ 67-101, repealing	SB 1284
§ 67-101.1, adding	SB 1284
§ 67-102, amending	SB 1223, SB 1284
§ 67-103, amending	SB 1284
§ 67-104, amending	SB 1284
§ 67-200, amending	
§ 67-201, amending	SB 1223, SB 1284
§ 67-202, amending	SB 1223
§ 67-202.1, amending	
§ 67-602, amending	
§ 67-701, amending	SB 1463
§ 67-900, amending	
§ 67-1000, amending	
§ 67-1206, amending	
§ 67-1208, amending	
§ 67-1209, amending	
§ 67-1403, amending	
§ 67-1506, amending	
§ 67-1700, adding	HB 547
§ 67-1800, adding	
§ 67-1800, adding	НВ 1979
§ 67-1801, adding	НВ 1979
§ 67-1802, adding	НВ 1979
§ 67-1803, adding	НВ 1979
§ 67-1804, adding	НВ 1979
§ 67-1805, adding	НВ 1979
§ 67-1806, adding	НВ 1979

COLDEN, TONY E., JR.
Colden, Tony E., Jr.; recording sorrow upon death. (Patron–Bourne, HJR 724)

COLE, JAMES B. Cole, James B.; commending. (Patron–Hope, HJR 657)

COLEMAN, LOVELL LOUIS, SR.

Coleman, Lovell Louis, Sr.; recording sorrow upon death. (Patron-Hudson, HJR 708)

COLYER, BOB, SR.

Colver, Bob, Sr.; recording sorrow upon death. (Patron–Kilgore, HJR 645)

COMMENDING RESOLUTIONS

Allen, Samuel; commending. (Patron-Hayes, HJR 681)

Anderson, John Hutchison; commending. (Patron-Webert, HJR 673)

Asian Pacific American Bar Association of the Greater Washington, D.C., Area, Inc.; commemorating its 40th anniversary. (Patron–Keam, HJR 718)

Avula, Danny TK; commending. (Patron-VanValkenburg, HJR 660)

Banister, Daniel; commending. (Patron-Jones, HJR 775)

Barber, Doug; commending. (Patron-Marshall, HJR 778)

Bickford, Charlene; commending. (Patron-Hope, HJR 641)

Braxton-Williams, Valerie; commending. (Patron-Bagby, HJR 715)

Broderick, John R.; commending. (Patron-Filler-Corn, HJR 714)

Brown, Lewis R.; commending. (Patron-Guzman, HJR 780)

Brown, Shamira A.; commending. (Patron-Guzman, HJR 781)

C. Waldo Scott Center for H.O.P.E.; commemorating its 30th anniversary. (Patron-Price, HJR 786)

Cardounel, Humberto, Jr.; commending. (Patron-VanValkenburg, HJR 671)

Center for Excellence in Education; commending. (Patron-Keam, HJR 717)

Civil Rights Act of 1871; commemorating the 150th anniversary of the enactment. (Patron–McClellan, SJR 395)

Clifton Presbyterian Church; commemorating its 150th anniversary. (Patron-Helmer, HJR 762)

Cole, James B.; commending. (Patron-Hope, HJR 657)

Community Free Clinic of Newport News; commemorating its 10th anniversary. (Patron-Price, HJR 785)

Cross, Oretha P.; commending. (Patron-Price, HJR 784)

Crutcher, Ronald A.; commending. (Patron-VanValkenburg, HJR 696)

Danville Life Saving and First Aid Crew, Inc.; commemorating its 75th anniversary. (Patron–Marshall, HJR 774)

Downing, Annie Laura; commending. (Patron-Wyatt, HJR 695)

Fairfax County Park Authority; commending. (Patron-Filler-Corn, HJR 721)

Farr, Murray Jay; commending. (Patron-Hope, HJR 656)

Feasenmyer, William S., Jr.; commending. (Patron-Cox, HJR 746)

Foster, Tim; commending. (Patron-VanValkenburg, HJR 700)

Fredericksburg, City of; commending. (Patron-Cole, M.L., HJR 666)

Fredericksburg Dog Mart; commending. (Patron-Cole, M.L., HJR 667)

Gaylord, Katie; commending. (Patron-Mullin, HJR 753)

Grandy, Ethel L.; commending. (Patron–Hayes, HJR 684)

Guido, Tammy Williams; commending. (Patron-Mullin, HJR 756)

Hardy Petroleum; commemorating its 100th anniversary. (Patron-Marshall, HJR 779)

Harris, Gregory Garfield; commending. (Patron-Webert, HJR 674)

Henrico County Registrar's Office; commending. (Patron-VanValkenburg, HJR 661)

Hill, Bobby; commending. (Patron-Convirs-Fowler, HJR 682)

Johnson, Cintia; commending. (Patron-Hope, HJR 658)

Johnstone, Richard G., Jr.; commending. (Patron-VanValkenburg, HJR 663)

Kelleher, Linda Y.; commending. (Patron-Hope, HJR 639)

Korean Central Presbyterian Church; commending. (Patron-Helmer, HJR 763)

Lawrence, Barry R.; commending. (Patron-VanValkenburg, HJR 697)

Linaburg, Dennis; commending. (Patron-LaRock, HJR 760)

Loudoun County Office of Elections and Voter Registration; commending. (Patron-Subramanyam, HJR 683)

COMMENDING RESOLUTIONS (continued)

Luck, Joann Grant; commending. (Patron-Adams, L.R., HJR 745)

Luther, T. David; commending. (Patron-Marshall, HJR 776)

Meriwether, Bobbie Jean; commending. (Patron-Cole, M.L., HJR 665)

Miller, Marty L.; commending. (Patron-Hayes, HJR 693)

New River Valley Public Health Task Force; commending. (Patron-Hurst, HJR 677)

Newport News Public Library; commending. (Patron–Mullin, HJR 749)

Newsome House Museum and Cultural Center; commemorating its 30th anniversary. (Patron-Price, HJR 787)

Northern Virginia Therapeutic Riding Program; commemorating its 40th anniversary. (Patron–Helmer, HJR 764)

Ortiz, Patty; commending. (Patron-LaRock, HJR 752)

Parham Doctors' Hospital; commending. (Patron-VanValkenburg, HJR 655)

Pebler, Steven; commending. (Patron-LaRock, HJR 754)

Pleasant View, Inc.; commemorating its 50th anniversary. (Patron-Wilt, HJR 747)

Rapisarda, Joseph P., Jr.; commending. (Patron-VanValkenburg, HJR 699)

Rosenberry, Brian; commending. (Patron-LaRock, HJR 761)

Saint Joseph Catholic School; commemorating its 145th anniversary. (Patron-Aird, HJR 748)

Shanks, Fred O., III; commending. (Patron–Marshall, HJR 777)

Silber, Randall R.; commending. (Patron–VanValkenburg, HJR 659)

Singleton, O.R., Jr.; commending. (Patron-VanValkenburg, HJR 698)

"Stir Crazy in Williamsburg"; commending the authors. (Patron-Mullin, HJR 758)

Suerdieck, Rebecca and Julia Oxrieder; commending. (Patron-Mullin, HJR 765)

Sweat, Josh; commending. (Patron-Hayes, HJR 692)

Swegle, Madeline; commending. (Patron-Filler-Corn, HJR 720)

Szakos, Joe; commending. (Patron–Hudson, HJR 719)

T.C. Williams School of Law at the University of Richmond; commemorating its 150th anniversary. (Patron–Adams, D.M., HJR 676)

The Chatham Garden Club; commemorating its 100th anniversary. (Patron-Adams, L.R., HJR 744)

The Virginian-Pilot; commending. (Patron–Mullin, HJR 750)

Unger, John; commending. (Patron-LaRock, HJR 759)

Van Doren, Nancy; commending. (Patron-Hope, HJR 640)

Virginia Academy of Science, Engineering and Medicine; commending. (Patron-Hayes, HJR 675)

Virginia National Guard; commending. (Patron–Filler-Corn, HJR 713)

Virginia State Police; commending. (Patron–Filler-Corn, HJR 712)

Warren, Adrienne; commending. (Patron-Hayes, HJR 694)

Warren, Darrell W., Jr.; commending. (Patron–Hodges, HJR 672)

Williamsburg Faith in Action and the Arc of Greater Williamsburg; commending. (Patron–Mullin, HJR 755)

Williamsburg Health Foundation; commending. (Patron-Mullin, HJR 751)

Yoffy, James S.; commending. (Patron-VanValkenburg, HJR 664)

Zion Baptist Church; commemorating its 125th anniversary. (Patron-Price, HJR 783)

COMMERCIAL VEHICLES

Commercial driver's licenses; disqualifies for life from holding a license persons convicted of a felony involving an act or practice of severe forms of trafficking, etc. Amending §§ 46.2-341.18, 46.2-382, and 46.2-1702. (Patron–Delaney, HB 1868)

COMMISSIONS, BOARDS, AND INSTITUTIONS GENERALLY

Apprenticeship training programs; Virginia Board of Workforce Development, et al., shall review availability of programs, etc., report. (Patron–Simonds, HB 1849)

Assisted living and auxiliary grants; Joint Commission on Health Care to study available data regarding, etc. (Patron-Spruill, SJR 293)

Behavioral Health Commission; created, report. Adding §§ 30-401 through 30-408. (Patron–Deeds, SB 1273)

COMMISSIONS, BOARDS, AND INSTITUTIONS GENERALLY (continued)

- Certified nurse midwives; eliminates the requirement that midwives practice pursuant to a practice agreement and shall practice in accordance with regulations of the Boards of Medicine and Nursing. Amending §§ 54.1-2957 and 54.1-2957.01. (Patron–Adams, D.M., HB 1817)
- Charitable Gaming Board; prohibits the Board from promulgating regulations that prohibit the use of multiple video monitors or touchscreens on an electronic pull tab device, use of devices shall be limited to one player at a time. Amending §§ 18.2-340.19 and 18.2-340.28. (Patron–Willett, HB 1943; McPike, SB 1287)
- Clinical nurse specialist; licensure of nurse practitioners as specialists by the Boards of Medicine and Nursing and a practitioner licensed as a specialist shall practice pursuant to a practice agreement between the specialist and a licensed physician, etc., prescriptive authority, repeals provision relating to registered nurse's licensure by endorsement. Amending §§ 54.1-2900, 54.1-2901, 54.1-2957, 54.1-2957.01, and 54.1-3000; repealing § 54.1-3018.1. (Patron–Adams, D.M., HB 1747)
- Compensation Board; determining staffing and salaries for an attorney for the Commonwealth, effective clause. Amending §§ 15.2-1626 and 15.2-1636.8. (Patron–Boysko, SB 1226)
- Criminal Justice Services Board and Committee on Training; membership, law-enforcement training. Amending §§ 9.1-102, 9.1-108, and 9.1-112. (Patron–Marsden, SB 1256)
- Dams; Virginia Soil and Water Conservation Board to enter into a negotiated settlement with the owners of certain impounding structures. Adding § 10.1-613.6. (Patron–Bell, SB 1280)
- Data Governance and Analytics, Office of; created, establishes the advisory Virginia Data Commission, membership, report. Amending § 2.2-203.2:4; adding §§ 2.2-2558 through 2.2-2564. (Patron–Barker, SB 1365)
- Early Psychosis Intervention and Coordinated Specialty Care Program Advisory Board; established. Adding § 37.2-313.2. (Patron–Morrissey, SB 1427)
- Education, Board of; geographic representation of members. Amending § 22.1-9. (Patron-Austin, HB 1827)
- Education, Board of; temporary extension of the license of any individual licensed by Board whose license expires on June 30, 2021. (Patron-Ward, HB 1776)
- Governor's Schools; Board of Education shall issue guidance on the governance of academic year, including admissions policies, and guidelines on diversity, etc. (Patron–Tyler, HB 2305)
- Health Care, Joint Commission on; repeals sunset provision. Repealing § 30-170. (Patron–Barker, SB 1408)
- Health Insurance Reform Commission; mandated health insurance benefit or provider legislation to be referred to Commission. Amending § 30-343. (Patron–Saslaw, SB 1473)
- Henrietta Lacks Commission; extends sunset provision. Amending § 2.2-2543. (Patron-Stanley, SB 1414)
- Historic resources; land acquisition activities of the Department of Historic Resources and the Board of Historic Resources, etc. Amending §§ 10.1-2202 and 10.1-2204. (Patron–Aird, HB 1928)
- Home care organizations; Board of Health to include in regulations governing organizations a provision for supervision of home care attendants providing personal care services by a licensed nurse through use of interactive audio or video technology. Amending § 32.1-162.12. (Patron–Head, HB 1831)
- Inmates; Board of Local and Regional Jails to review services provided during pregnancy, etc., report. (Patron-Favola, SB 1300)
- Local and Regional Jails, State Board of; membership, powers and duties, effective clause. Amending §§ 53.1-2, 53.1-4, and 53.1-5. (Patron–Marsden, SB 1363)
- Military-overseas ballots; Secretary of Administration to oversee and develop a charter and directives for the State Board of Elections to form a working group to study implementation of electronic return of voted ballots. (Patron–DeSteph, SJR 322)
- Parole; Department of Corrections shall release a prisoner no sooner than 21 business days after the date of notification by the Virginia Parole Board to the appropriate attorney for the Commonwealth of the decision to grant parole, etc. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1104)
- Parole Board; Board, within seven days of making any decision regarding the parole of a prisoner, to provide written or electronic notice of such decision to the victim of crime. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1125)

COMMISSIONS, BOARDS, AND INSTITUTIONS GENERALLY (continued)

INDEX

- Pretrial data collection; Virginia Criminal Sentencing Commission to collect and disseminate on an annual basis, report. Amending § 2.2-3802; adding § 19.2-134.1. (Patron–Herring, HB 2110; Lucas, SB 1391)
- School-based health services; Board of Medical Assistance Services to amend the state plan for medical assistance services to provide for payment, no health care provider who provides services through telemedicine shall be required to use proprietary technology or applications in order to be reimbursed. Amending §§ 32.1-325 and 32.1-326.3. (Patron–Dunnavant, SB 1307)
- Solar projects; State Water Control Board to administer a Virginia Erosion and Sediment Control Program (VESCP) on behalf of any locality that notifies the Department of Environmental Quality that it has chosen not to administer a VESCP, effective clause. Adding § 62.1-44.15:55.1. (Patron–Marsden, SB 1258)
- Special education; Board of Education to amend a certain regulation to remove the word "component" following the word "evaluation," thereby ensuring compliance with the relevant federal regulation. (Patron–Mugler, HB 2314)
- Special education; Department of and the Board of Education to develop new policies and procedures, individualized education program (IEP), duty of Department to provide training and guidance documents to local school divisions on development of IEPs. Amending §§ 22.1-214, 22.1-215, 22.1-253.13:4, and 22.1-298.1; adding § 22.1-214.4. (Patron–Dunnavant, SB 1288)
- Standards of Quality; Joint Legislative Audit and Review Commission to study the true cost of education in the Commonwealth and provide an accurate assessment of the costs to implement, meetings shall be completed for the first year by November 30, 2022. (Patron–Lewis, SJR 294)
- State Air Pollution Control Board; implementation of a low-emissions and zero-emissions vehicle program, energy jurisdictional retail sales. Amending § 10.1-1307; adding § 10.1-1307.04. (Patron–Bagby, HB 1965)
- State Corporation Commission; adds the Commission to the list of agencies that are exempt from paying fees for remote access to local land records. Amending § 17.1-276. (Patron–Kilgore, HB 1775)
- Tazewell County; authorizes a quitclaim and release of interest and conveyance of an easement by Board of Wildlife Resources. (Patron–Wampler, HB 2252; Pillion, SB 1400)
- Transportation District Commission of Hampton Roads; change in membership. Amending § 33.2-1907. (Patron–Spruill, SB 1126)
- Uniform Statewide Building Code; Board of Housing and Community Development shall consider amendments to Code to address changes in the IECC relating to energy efficiency and conservation. (Patron–Kory, HB 2227)
- Universal health care; Joint Commission on Health Care shall enter into a contract with a qualified entity to study options for financing. (Patron–Samirah, HB 2271)
- Virginia Criminal Sentencing Commission; confirms the appointment by the Chief Justice of the Supreme Court of Chairman of Commission. (Patron–Herring, HJR 629)
- Virginia Employment Commission; communications with parties, use of electronic means, report. Adding § 60.2-121.1. (Patron–Tran, HB 2036)
- Virginia Freedom of Information Act; individual votes of members of the Virginia Parole Board shall be public records and subject to provisions of the Act. Amending § 2.2-3703. (Patron–Suetterlein, SB 1103)
- Virginia LGBTQ+ Advisory Board; established, report. Adding §§ 2.2-2499.1 through 2.2-2499.4. (Patron-Lopez, HB 2130)
- Virginia Redistricting Commission; removing a citizen commissioner of the Commission for neglect of duty or gross misconduct by other commissioners in a public meeting. Amending § 30-392. (Patron–Watts, HB 2324)
- Virginia Redistricting Commission; transparency in redistricting process. Amending §§ 30-396 and 30-399. (Patron-Levine, HB 2082)
- Virginia Science, Technology, Engineering, and Mathematics (STEM) Advisory Board; established, report. Adding §§ 22.1-364 through 22.1-368. (Patron–Simonds, HB 2058)
- Virginia Soil and Water Conservation Board; clarifies membership. Amending § 10.1-502. (Patron-Plum, HB 1837; Hanger, SB 1161)

COMMONWEALTH PUBLIC SAFETY

- Behavioral health docket; standards for transfer of supervision between local community-based probation agencies. Amending §§ 9.1-175 and 18.2-254.3. (Patron–Bell, HB 2236)
- Court-appointed special advocates; information sharing when advocate is participating in family partnership meetings. Amending §§ 9.1-153 and 9.1-156. (Patron–Delaney, HB 1866)
- Criminal Justice Services Board and Committee on Training; membership, law-enforcement training. Amending §§ 9.1-102, 9.1-108, and 9.1-112. (Patron–Marsden, SB 1256)
- Criminal records; establishes a process for automatic expungement, etc., definitions, penalties, report. Amending §§ 9.1-101, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-340, 19.2-389.3, and 19.2-390; adding §§ 19.2-392.5 through 19.2-392.16. (Patron–Herring, HB 2113)
- Fire training activities; definitions, prohibition on the use of certain oriented strand board. Adding § 9.1-207.2. (Patron–Krizek, HB 2029)
- Law-enforcement agencies; creates a special nonreverting fund to be known as the Body-Worn Camera System Fund to assist state or local agencies with costs of purchasing, etc., body-worn camera systems. Adding § 9.1-116.7. (Patron–Reeves, SB 1119)
- Law-enforcement civilian oversight bodies; adds a sheriff's office to those law-enforcement agencies that may be overseen by a body created by a locality and adds a nonprobationary deputy sheriff to those officers who are subject to such body, disciplinary determinations. Amending § 9.1-601. (Patron–Williams Graves, HB 2291)
- Line of Duty Act; Virginia licensed health practitioners required to conduct medical reviews, persons issued a comparable license, as determined by Virginia Retirement System, by the District of Columbia or a state that is contiguous to Virginia. Amending §§ 9.1-404 and 9.1-405. (Patron–DeSteph, SB 1367)
- Sexual and Domestic Violence, Advisory Committee on; increases membership, duties. Amending §§ 9.1-116.2 and 9.1-116.3. (Patron–Robinson, HB 2317)
- Victims of crime; definitions, certifications for victims of qualifying criminal activity, disclosure of exculpatory information to a defendant in a criminal case. Adding §§ 9.1-1500, 9.1-1501, and 9.1-1502. (Patron–Surovell, SB 1468)

COMMONWEALTH'S ATTORNEYS

- Compensation Board; determining staffing and salaries for an attorney for the Commonwealth, effective clause. Amending §§ 15.2-1626 and 15.2-1636.8. (Patron–Boysko, SB 1226)
- Geriatric or terminally ill inmates; conditional release, notify by certified or verifiable electronic means to attorney for the Commonwealth. Amending § 53.1-136. (Patron-Norment, SB 1397)
- Parole; Department of Corrections shall release a prisoner no sooner than 21 business days after the date of notification by the Virginia Parole Board to the appropriate attorney for the Commonwealth of the decision to grant parole, etc. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1104)

COMMUNICABLE DISEASES

Public health emergency; emergency medical services agencies, real-time access to information regarding confirmed and suspected cases of communicable disease of public health threat. Adding § 32.1-48.001. (Patron–Aird, HB 1989)

COMMUNITY FREE CLINIC OF NEWPORT NEWS

Community Free Clinic of Newport News; commemorating its 10th anniversary. (Patron-Price, HJR 785)

COMPACTS

Occupational therapists; licensure, authorizes Virginia to become a signatory to the Occupational Therapy Interjurisdictional Licensure Compact. Adding § 54.1-2956.7:1. (Patron-Hashmi, SB 1189)

COMPUTER SERVICES AND USES

Broadband capacity; expands existing pilot program, municipal broadband authorities. Amending § 56-585.1:9. (Patron–Edwards, SB 1334)

COMPUTER SERVICES AND USES (continued)

- Broadband services; authorizes school boards to appropriate funds for the purposes of promoting, facilitating, and encouraging the expansion and operation of services for educational purposes. Adding §§ 15.2-986 and 22.1-79.9. (Patron–Boysko, SB 1225)
- Data centers; expands sales and use tax exemption. Amending § 58.1-609.3. (Patron–Ruff, SB 1425)
- Data centers; sales and use tax exemption, criteria, report. Amending § 58.1-609.3. (Patron–Morefield, HB 2273)
- Data centers; sales and use tax exemption, identifying a "distressed locality." Amending § 58.1-609.3. (Patron–McPike, SB 1423)
- Online portal for tax practitioners; Department of Taxation shall analyze prospect of establishing, report. (Patron-Coyner, HB 2060)
- Phase I or Phase II electric utilities; provision of broadband capacity, State Corporation Commission shall condition any approval of a petition on the requirement that construction shall commence within 18 months of such approval. Amending § 56-585.1:9. (Patron–Tyler, HB 2304; Boysko, SB 1413)

CONFLICT OF INTERESTS

Conflict of Interests Act, State and Local Government; disclosure requirements, industrial development and economic development authorities. Amending § 2.2-3115. (Patron–Norment, SB 1171)

CONNOCK, STUART WALLACE

Connock, Stuart Wallace; recording sorrow upon death. (Patron–Hudson, HJR 709)

CONSERVATION

- Advanced recycling, etc.; definitions. Amending § 10.1-1400. (Patron-Hanger, SB 1164)
- Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment, Enhanced Nutrient Removal Certainty Program established. Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14. (Patron–Lopez, HB 2129)
- Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment, Enhanced Nutrient Removal Certainty Program established, regulations. Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14. (Patron–Hanger, SB 1354)
- Conservation and Recreation, Department of; removes the authority from the Department to sell or demise certain lands over which it has control, etc., report. Amending § 10.1-109. (Patron–Subramanyam, HB 1833)
- Conservation easements; an easement held pursuant to the Virginia Conservation Easement Act or the Open-Space Land Act, etc., be liberally construed in favor of achieving the purposes for which it was created. Adding §§ 10.1-1016.1 and 10.1-1705.1. (Patron–Webert, HB 1760)
- Conservation easements; construction. Adding §§ 10.1-1016.1 and 10.1-1705.1. (Patron-Petersen, SB 1199)
- ConserveVirginia program; established. Amending §§ 10.1-1018.1 and 10.1-1021; adding § 10.1-104.6:1. (Patron–Mason, SB 1290)
- Dams; Virginia Soil and Water Conservation Board to enter into a negotiated settlement with the owners of certain impounding structures. Adding § 10.1-613.6. (Patron–Bell, SB 1280)
- Environmental permits; community and environmental justice outreach, new individual permit application. Amending § 10.1-1408.1; adding §§ 10.1-1321.2, 10.1-1437.1, 62.1-44.6:1, and 62.1-266.1. (Patron–Hayes, HB 2221)
- Expanded polystyrene food service containers; definition, prohibits dispensing by a food vendor of prepared foods to a customer, civil penalty. Amending §§ 10.1-1414 and 10.1-1422.01; adding § 10.1-1424.3. (Patron–Carr, HB 1902)
- Greenhouse gas emissions inventory; regulations, proprietary information. Adding § 10.1-1307.04. (Patron–Morrissey, SB 1282)
- Historic resources; land acquisition activities of the Department of Historic Resources and the Board of Historic Resources, etc. Amending §§ 10.1-2202 and 10.1-2204. (Patron–Aird, HB 1928)
- Natural Resources, Secretary of; changes name to the Secretary of Natural and Cultural Resources. Amending §§ 2.2-200, 2.2-205, 2.2-206, 2.2-215, 2.2-220.1, 2.2-220.2, 2.2-220.3, 2.2-1156, 2.2-1176, 2.2-2316, 2.2-2338, 2.2-2481, 2.2-2699.10, 2.2-2699.11, 2.2-2699.13, 10.1-603.25, 10.1-704, 10.1-1018, 10.1-1181.15, 10.1-1188, 10.1-1329, 10.1-1402.03, 10.1-1402.04,

CONSERVATION (continued)

- 10.1-1405, 10.1-2129, 10.1-2202.3, 15.2-2295.1, 28.2-207, 29.1-102, 29.1-573, 29.1-579, 30-377, 56-596.2, 58.1-344.3, 62.1-44.15:68, 62.1-44.34:25, 62.1-44.34:28, 62.1-44.117, 62.1-44.118, 62.1-44.119:1, 62.1-69.31, 62.1-69.32, 62.1-69.33, 62.1-69.41, 62.1-69.52, and 62.1-195.1. (Patron–Plum, HB 1836)
- Objects of antiquity; unlawful to remove from battlefield, penalty. Amending §§ 10.1-2300 and 10.1-2306. (Patron-Runion, HB 2311)
- Permit fee schedules; Department of Environmental Quality shall convene a work group to revise schedule for nonhazardous solid waste management facilities. (Patron-Petersen, SB 1210)
- Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron–Cole, J.G., HB 1819)
- Small renewable energy projects; includes in the definition certain energy storage facilities and projects that include storage facility components. Amending § 10.1-1197.5. (Patron–Willett, HB 2148)
- South River; designates a 6.5-mile segment in the City of Waynesboro as a component of the Virginia Scenic Rivers System. Adding § 10.1-417.1. (Patron–Avoli, HB 1958)
- State Air Pollution Control Board; implementation of a low-emissions and zero-emissions vehicle program, energy jurisdictional retail sales. Amending § 10.1-1307; adding § 10.1-1307.04. (Patron–Bagby, HB 1965)
- Virginia Electric Vehicle Grant Fund and Program; created, report. Adding § 10.1-1322.5. (Patron-Keam, HB 2118)
- Virginia Soil and Water Conservation Board; clarifies membership. Amending § 10.1-502. (Patron-Plum, HB 1837; Hanger, SB 1161)
- Wildlife corridors; various agencies to consider and incorporate, where applicable, government planning, Department of Wildlife Resources shall publish plans and any updates on its website. Amending §§ 10.1-207, 10.1-1105, 29.1-579, and 33.2-353. (Patron–Marsden, SB 1274)

CONSTITUTIONAL AMENDMENTS

- Constitutional amendment; authority to grant perpetual easements to units of government (first reference). Amending Section 9 of Article VII. (Patron–Mason, SJR 289)
- Constitutional amendment; equitable educational opportunities in all public schools in the Commonwealth (first reference). Amending Sections 1 and 2 of Article VIII. (Patron–Stanley, SJR 275)
- Constitutional amendment; felon disenfranchisement, automatic restoration of political rights (first reference). Amending Section 1 of Article II. (Patron-Herring, HJR 555)
- Constitutional amendment; fundamental right to marry, removes same-sex marriage prohibition (first reference). Amending Section 15-A of Article I. (Patron–Sickles, HJR 582; Ebbin, SJR 270)
- Constitutional amendment; length of regular sessions convened in odd-numbered years (first reference). Amending Section 6 of Article IV. (Patron–Saslaw, SJR 310)
- Constitutional amendment; qualifications of voters, fundamental right to vote, persons not entitled to vote (first reference). Amending Section 1 of Article II. (Patron–Locke, SJR 272)

CONSUMER PROTECTION

- Consumer Data Protection Act; definitions, establishes a framework for controlling and processing personal data, applies to all persons that conduct business in the Commonwealth, etc. Adding §§ 59.1-571 through 59.1-581. (Patron–Hayes, HB 2307)
- Consumer Data Protection Act; establishes a framework for controlling and processing personal data, effective date. Adding §§ 59.1-571 through 59.1-581. (Patron–Marsden, SB 1392)
- Consumer Protection Act; prohibited practices, certain advertising related to any public or private school quality. Amending § 59.1-200. (Patron–Samirah, HB 2003)

CONTINUING EDUCATION

Public adjusters; continuing education requirements, repeals provision relating to continuing education, approval of credits, etc. Amending §§ 38.2-1866, 38.2-1867, 38.2-1868.1, 38.2-1869, 38.2-1871, and 38.2-1873; repealing § 38.2-1845.9. (Patron–Kilgore, HB 1942)

CONTRACTORS AND SUBCONTRACTORS

Subcontractor's employees; liability of general contractor for wages. Amending § 11-4.6. (Patron–Petersen, SB 1209)

Virginia Public Procurement Act; bids or offers on construction contracts, requirement to submit list of subcontractors, effective clause. Adding § 2.2-4303.02. (Patron–Williams Graves, HB 2288)

Virginia Public Procurement Act; construction contracts, localities with a population in excess of 25,000, subcontractor workforce requirements. Adding § 2.2-4303.02. (Patron–McPike, SB 1305)

CONTRACTS

Subcontractor's employees; liability of general contractor for wages. Amending § 11-4.6. (Patron–Petersen, SB 1209)

Virginia Public Procurement Act; local arbitration agreements, definitions, procurement procedures, compliance for post-award actions, penalty. Adding §§ 2.2-4377.1 through 2.2-4377.5. (Patron–Surovell, SB 1384)

Virginia Public Procurement Act; preference for energy-efficient and water-efficient goods. Adding § 2.2-4328.1. (Patron-Helmer, HB 1811)

CORPORATIONS

Agricultural best management practices; creates an enhanced individual and corporate income tax credit beginning in taxable year 2021 for the implementation of certain practices by the taxpayer that are required as part of a certified resource management plan, definition of "liability after such cost-share." Amending §§ 58.1-339.3 and 58.1-439.5. (Patron–Wilt, HB 1763)

Agricultural equipment; establishes a refundable individual and corporate income tax credit. Amending §§ 58.1-334, 58.1-337, 58.1-432, and 58.1-436. (Patron–Hanger, SB 1163)

Corporate income tax; Division of Legislative Services, et al., to establish a work group to assess the feasibility of transitioning to a unitary combined reporting system. (Patron–Watts, HJR 563)

State Corporation Commission; business entities filings, amends various provisions of Va. Stock Corporation Act. Amending §§ 13.1-609, 13.1-610, 13.1-615, 13.1-615.1, 13.1-625, 13.1-628, 13.1-630, 13.1-636, 13.1-639, 13.1-658, 13.1-661, 13.1-710, 13.1-711, 13.1-716, 13.1-718, 13.1-721.1, 13.1-722.6, 13.1-722.12:1, 13.1-759, 13.1-765, 13.1-775.1, 13.1-803, 13.1-806, 13.1-807, 13.1-809, 13.1-815, 13.1-815.1, 13.1-816, 13.1-829, 13.1-830, 13.1-831, 13.1-835, 13.1-894, 13.1-897.1, 13.1-898.7, 13.1-921, 13.1-927, 13.1-936.1, 13.1-944.7, 13.1-1002, 13.1-1004, 13.1-1005, 13.1-1012, 13.1-1017, 13.1-1052, 13.1-1054, 13.1-1062, 13.1-1065, 13.1-1073.1, 13.1-1075, 13.1-1080, 13.1-1087, 13.1-1096, 13.1-10914, 13.1-1099.26, 13.1-1201, 13.1-1203, 13.1-1212, 13.1-1214, 13.1-1222, 13.1-1242, 13.1-1252, 13.1-1255, 13.1-1264, 13.1-1265, 13.1-1271, 13.1-1277, 15.2-5112, 15.2-5431.9, 50-73.1, 50-73.2, 50-73.6, 50-73.17, 50-73.54, 50-73.67, 50-73.70, 50-73.83, and 50-73.135; adding §§ 13.1-898.1:1, 13.1-1263.1, 15.2-5431.8:1, 15.2-5431.9:1, 15.2-5431.35:1, and 50-73.48:5; repealing §§ 13.1-941.01 through 13.1-944. (Patron–Keam, HB 2121)

CORRECTIONAL ENTERPRISES

Behavioral health; assessments in local correctional facilities, report. Amending § 53.1-68. (Patron-Coyner, HB 1874)

Bribery in correctional facilities; penalty. Adding § 18.2-474.2. (Patron–Lewis, SB 1461)

Correctional facilities; restrictions on use of isolated confinement, effective clause. Adding §§ 53.1-39.2 and 66-20.1. (Patron–Morrissey, SB 1301)

CORRECTIONS, BOARD OF OR DEPARTMENT OF

Staffing levels, employment conditions, and compensation at the Virginia Department of Corrections, joint committee of various House and Senate Committees Studying; continued, appropriations. (Patron–Tyler, HJR 522)

Suspension or modification of sentence; transfer to the Department of Corrections. Amending § 19.2-303. (Patron–Kilgore, HB 1806)

COSGROVE, JOHN A., JR.

Addressed the Senate in memory of	of former Delegate	Mamye E. BaCote; r	requested adjournment in
memory			133

COUNTIES, CITIES, AND TOWNS

- Active military or a military spouse; definition of "military status," discrimination in public accommodations, employment, and housing. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310. (Patron—Tran, HB 2161)
- Active military or a military spouse; prohibits discrimination in public accommodations, employment, and housing, landlords may not charge a higher security deposit. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1226, and 55.1-1310. (Patron–Bell, SB 1410)
- Administration of blighted and derelict properties; modifies definition of "qualifying locality." Amending §§ 58.1-3221.6 and 58.1-3970.1. (Patron–Carr, HB 1969)
- Albemarle and Prince William Counties; counties that have adopted the county executive form of government may carry over unspent funds from year to year for multiyear capital projects and outstanding grants. Amending §§ 15.2-520 and 15.2-2506. (Patron–Reeves, SB 1120)
- Alcoholic beverage control; adds the chief administrative officer of a locality to the list of persons who may be sent notice of certain license applications by Board of Directors of the Virginia Alcoholic Beverage Control Authority. Amending §§ 4.1-230 and 15.2-907. (Patron–Lopez, HB 2131)
- Alcoholic beverage control; local special events license, taxes and fees. Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, and 4.1-233.1. (Patron–Dunnavant, SB 1471)
- Alcoholic beverage control; outdoor refreshment area license, fees for state and local licenses. Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, 4.1-233.1, and 4.1-308. (Patron–Ayala, HB 2266)
- Broadband services; authorizes school boards to appropriate funds for the purposes of promoting, facilitating, and encouraging the expansion and operation of services for educational purposes. Adding §§ 15.2-986 and 22.1-79.9. (Patron–Boysko, SB 1225)
- Brunswick County school board; removes school board from the list of approved member salaries for appointed school boards. Amending § 22.1-32. (Patron–Tyler, HB 1798; Ruff, SB 1175)
- Buckingham County; fees for disposal of solid waste. Amending § 15.2-2159. (Patron-Peake, SB 1447)
- Child care; expands to include all cities the authority related to the regulation of services. Amending § 15.2-914. (Patron–Williams Graves, HB 2326)
- Compensation Board; determining staffing and salaries for an attorney for the Commonwealth, effective clause. Amending §§ 15.2-1626 and 15.2-1636.8. (Patron–Boysko, SB 1226)
- Comprehensive plan; provision for transit-oriented development. Amending § 15.2-2223.4. (Patron–Samirah, HB 2054)
- Continuity of government; extends to 12 months the period of time after an enemy attack or other disaster that a locality may, by ordinance, provide. Amending § 15.2-1413. (Patron–Barker, SB 1208)
- Economic development authorities; size of board in Powhatan County, reduces quorum requirement. Amending § 15.2-4904. (Patron–Ware, HB 2287)
- Facial recognition technology; definition, authorization of use by local law-enforcement agencies and public institutions of higher education. Adding §§ 15.2-1723.2 and 23.1-815.1. (Patron–Aird, HB 2031)
- GO Virginia Grants; matching funds, repeals sunset provision. (Patron-Wampler, HB 2101)
- Hampton Roads area refuse collection authority; changes various requirements of the Southeastern Public Service Authority when budgeting or incurring debt, etc. Amending § 15.2-5102.1. (Patron-Cosgrove, SB 1141)
- Highway construction; increases the value of highway maintenance and construction projects eligible to be performed by state or local employees. Amending § 33.2-234. (Patron–Krizek, HB 1813)
- Historic sites; urban county executive form of government (Fairfax County), provisions in its preservation ordinance, etc. Amending § 15.2-2306. (Patron–Surovell, SB 1457)
- Illegal gambling; skill games, definitions, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron–Reeves, SB 1465)

COUNTIES, CITIES, AND TOWNS (continued)

- Illegal gambling; skill games, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron-Scott, HB 2168)
- Law-enforcement civilian oversight bodies; adds a sheriff's office to those law-enforcement agencies that may be overseen by a body created by a locality and adds a nonprobationary deputy sheriff to those officers who are subject to such body, disciplinary determinations. Amending § 9.1-601. (Patron–Williams Graves, HB 2291)
- Law-enforcement officer; duty to render aid, duty to report wrongdoing by another officer. Amending § 52-30.1; adding §§ 15.2-1704.1 and 15.2-1704.2. (Patron-Levine, HB 1948)
- Library aid; former regional library systems that receive state aid notwithstanding a loss of regional library status due to a city reversion, shall receive such aid regardless. Amending § 15.2-4116. (Patron-Edmunds, HB 2323)
- Local cigarette taxes; any locality is authorized to levy taxes upon sale of cigarettes, regional cigarette tax boards, definitions, report. Amending § 58.1-3830; adding § 58.1-3832.1. (Patron-Hanger, SB 1326)
- Local elections for governing bodies; elections for school boards, qualification of voters. Amending §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223. (Patron-Convirs-Fowler,
- Local government; authority to reduce the speed limit in a business district or residence district. Amending § 46.2-1300. (Patron–Carr, HB 1903)
- Local green banks; authorizes a locality, by ordinance, to establish a bank to promote the investment in clean energy technologies in its locality, etc. Adding § 15.2-958.3:1. (Patron-Kory, HB 1919)
- Local stormwater assistance; flood mitigation and protection measures. Amending § 15.2-2114.01. (Patron-Ebbin, SB 1309)
- Mathews County; board of supervisors may appoint one employee of the locality onto the board of directors for the Economic Development Authority of the County. Amending § 15.2-4904. (Patron–Hodges, HB 2186)
- Municipal elections; shifting elections to November. Amending § 15.2-1400. (Patron-Spruill, SB 1157)
- Public access authorities; granted certain liability protections. Amending § 15.2-1809. (Patron–Hodges, HB 2217)
- Removal of clutter from property; definition, civil penalty. Amending § 15.2-901. (Patron-Ward, HB 1778)
- Solar and energy storage projects; siting agreements throughout the Commonwealth. Amending §§ 15.2-2288.8 and 15.2-2316.6 through 15.2-2316.9. (Patron–Jones, HB 2201; Barker, SB 1207)
- Solar projects; State Water Control Board to administer a Virginia Erosion and Sediment Control Program (VESCP) on behalf of any locality that notifies the Department of Environmental Quality that it has chosen not to administer a VESCP, effective clause. Adding § 62.1-44.15:55.1. (Patron–Marsden, SB 1258)
- State and local buildings, certain; definitions, building standards, high performance building certification program, sufficient ZEV charging and fueling infrastructure, Amending §§ 2.2-1182 and 2.2-1183; adding § 15.2-1804.1. (Patron–Helmer, HB 2001)
- Test driving vehicles; residence districts, civil penalty. Adding § 46.2-1533.1. (Patron-Roem, HB 2318)
- Tourism Development Authority; changes name of Authority to the Heart of Appalachia Tourism Authority. Amending §§ 15.2-5500, 15.2-5501, 15.2-5505, 15.2-5506, and 45.1-246. (Patron–Pillion, SB 1399)
- Tourism improvement districts; authorizes any locality to create. Adding §§ 15.2-2413.1 through 15.2-2413.11. (Patron-Bell, SB 1298)
- Trees; replacement and conservation during development, effective date. Amending §§ 15.2-961 and 15.2-961.1. (Patron-Guy, HB 2042)
- Trees; replacement and conservation during development, projects located in a Chesapeake Bay Preservation Area to address recurrent flooding, effective date, report, effective clause. Amending §§ 15.2-961 and 15.2-961.1. (Patron–Marsden, SB 1393)

COUNTIES, CITIES, AND TOWNS (continued)

- Underground utility facilities; removes the sunset on a pilot program allowing a locality that has adopted the urban county executive form of government (Fairfax County) to request an electric utility to place underground electric distribution lines, written agreement. Amending § 15.2-816.1. (Patron–Surovell, SB 1385)
- U.S. Route 29; authorizes the board of any locality that has adopted the county manager plan of government (Arlington County) to name any section located within the boundaries of the locality. Adding § 15.2-719.1. (Patron–Sullivan, HB 1854)
- Vacant buildings; removes requirement that a building meet the definition of "derelict building" for Town of Clifton Forge, Town of Pulaski, Town of Timberville, and any city to require, by ordinance, the owner or owners of buildings to register such buildings on an annual basis. Amending § 15.2-1127. (Patron–Locke, SB 1285)
- Virginia Public Procurement Act; determination of responsibility, local option to include criteria in Invitation to Bid. Amending § 2.2-4302.1. (Patron–Murphy, HB 1996)
- Workers' compensation; adds salaried or volunteer emergency medical services personnel to the list of persons to whom, after five years of service, the occupational disease presumption for death caused by hypertension or heart disease applies, personnel operating in a locality that has legally adopted a resolution declaring that it will provide one or more of the presumptions. Amending § 65.2-402. (Patron–Heretick, HB 1818)
- Zoning appeals, board of; any elected official of an incorporated town may serve on board of county in which member also resides. Amending § 15.2-2308. (Patron–Roem, HB 1898)

COURT-APPOINTED SPECIAL ADVOCATE (CASA) PROGRAMS

Court-appointed special advocates; information sharing when advocate is participating in family partnership meetings. Amending §§ 9.1-153 and 9.1-156. (Patron–Delaney, HB 1866)

COURT OF APPEALS OF VIRGINIA

Court of Appeals; expands jurisdiction, increases from 11 to 17 number of judges on Court, effective clause. Amending §§ 2.2-511, 8.01-36, 8.01-267.8, 8.01-383.1, 8.01-555, 8.01-626, 8.01-670, 8.01-671, 8.01-675.3, 8.01-676.1, 9.1-909, 15.2-1627, 15.2-1643, 15.2-2139, 15.2-2140, 15.2-2656, 15.2-3104, 15.2-3217, 15.2-3221, 15.2-3222, 15.2-3227, 15.2-3244, 15.2-3308, 15.2-3528, 15.2-3605, 15.2-3809, 15.2-3909, 15.2-4108, 15.2-4120, 15.2-5218, 15.2-5367, 15.2-6606, 15.2-6632, 15.2-7406, 16.1-279.1, 17.1-309, 17.1-400 through 17.1-403, 17.1-405 through 17.1-408, 17.1-410, 17.1-413, 17.1-503, 17.1-513, 18.2-308.08, 18.2-384, 19.2-152.10, 19.2-165, 19.2-321.1, 19.2-321.2, 19.2-322.1, 19.2-386.13, 19.2-402, 19.2-403, 19.2-404, 22.1-97, 22.1-289.024, 24.2-237, 24.2-422, 24.2-433, 25.1-239, 32.1-48.010, 32.1-48.013, 33.2-928, 33.2-2917, 37.2-920, 45.1-161.322, 55.1-1833, 55.1-1966, 55.1-2211, 57-2.02, 58.1-527, 58.1-1828, 58.1-2282, 58.1-3147, 58.1-3992, and 63.2-1710; adding §§ 8.01-675.5 and 8.01-675.6; repealing §§ 8.01-670.1 and 8.01-672. (Patron-Edwards, SB 1261)

COURTS NOT OF RECORD

- "Abused or neglected child;" definition. Amending § 16.1-228. (Patron-Lucas, SB 1168)
- Child support; availability of medical assistance through Family Access to Medical Insurance Security (FAMIS) plan or other government-sponsored coverage, eligibility requirements. Amending §§ 16.1-260 and 63.2-1903. (Patron–Samirah, HB 2002)
- Child support payments; juvenile in custody of or committed to the Department of Juvenile Justice. Amending §§ 16.1-263, 16.1-286, and 16.1-290. (Patron–Hope, HB 1912)
- Court of Appeals; expands jurisdiction, increases from 11 to 17 number of judges on Court, effective clause. Amending §§ 2.2-511, 8.01-36, 8.01-267.8, 8.01-383.1, 8.01-555, 8.01-626, 8.01-670, 8.01-671, 8.01-675.3, 8.01-676.1, 9.1-909, 15.2-1627, 15.2-1643, 15.2-2139, 15.2-2140, 15.2-2656, 15.2-3104, 15.2-3217, 15.2-3221, 15.2-3222, 15.2-3227, 15.2-3244, 15.2-3308, 15.2-3528, 15.2-3605, 15.2-3809, 15.2-3909, 15.2-4108, 15.2-4120, 15.2-5218, 15.2-5367, 15.2-6606, 15.2-6632, 15.2-7406, 16.1-279.1, 17.1-309, 17.1-400 through 17.1-403, 17.1-405 through 17.1-408, 17.1-410, 17.1-413, 17.1-503, 17.1-513, 18.2-308.08, 18.2-384, 19.2-152.10, 19.2-165, 19.2-321.1, 19.2-321.2, 19.2-322.1, 19.2-386.13, 19.2-402, 19.2-403, 19.2-404, 22.1-97, 22.1-289.024, 24.2-237, 24.2-422, 24.2-433, 25.1-239, 32.1-48.010, 32.1-48.013, 33.2-928, 33.2-2917, 37.2-920, 45.1-161.322, 55.1-1833, 55.1-1966, 55.1-2211,

COURTS NOT OF RECORD (continued)

57-2.02, 58.1-527, 58.1-1828, 58.1-2282, 58.1-3147, 58.1-3992, and 63.2-1710; adding §§ 8.01-675.5 and 8.01-675.6; repealing §§ 8.01-670.1 and 8.01-672. (Patron–Edwards, SB 1261)

Foster care; termination of parental rights, relatives and fictive kin. Amending §§ 16.1-281, 16.1-283, 63.2-906, and 63.2-910.2. (Patron–Gooditis, HB 1962)

General district courts; increases to \$50,000 the maximum civil jurisdictional limit of courts, appeal bond. Amending §§ 8.01-195.4, 16.1-77, and 16.1-107. (Patron–Stanley, SB 1108)

Jurisdiction over criminal cases; certification by general district court of felony or ancillary misdemeanor charges, jurisdiction to such charges shall vest in circuit court. Amending §§ 16.1-123.1, 16.1-241, and 17.1-513. (Patron–Adams, L.R., HB 2150)

Juvenile intake and petition; appeal to a magistrate on a finding of no probable cause. Amending §§ 16.1-256 and 16.1-260. (Patron–Jenkins, HB 1878)

Juvenile offenders; youth justice diversion programs, report. Amending § 16.1-260; adding § 16.1-309.11. (Patron–Mullin, HB 2017)

Juvenile records; confidentiality of records relevant to treatment, services, etc., exceptions. Amending § 16.1-300. (Patron–Barker, SB 1206)

Juveniles; competency evaluation, appointed evaluator or director of community services board, etc., shall acknowledge receipt of court order to clerk of court. Amending § 16.1-356. (Patron–Deeds, SB 1248)

Juveniles; eligibility for commitment to the Department of Juvenile Justice, eligibility for predispositional confinement in a secure facility. Amending §§ 16.1-248.1, 16.1-249, 16.1-278.7, and 16.1-278.8. (Patron–Marsden, SB 1456)

Juveniles; release and review hearing for serious offender, plea agreement. Amending §§ 16.1-285.1 and 16.1-285.2. (Patron–Jones, HB 1991)

Protective orders; violations of preliminary child protective order, violation involves an act that endangers the child's normal development, changes punishment, etc. Amending §§ 16.1-253 and 16.1-253.2. (Patron–Campbell, J.L., HB 2012; Stanley, SB 1415)

Special immigrant juvenile status; permits the juvenile and domestic relations district court to retain jurisdiction in certain cases, etc. Amending § 16.1-241. (Patron–Surovell, SB 1181)

Standby guardianship; triggering event, definitions, court approval of guardian. Amending §§ 16.1-349 through 16.1-353. (Patron–Deeds, SB 1184)

State-Funded Kinship Guardianship Assistance program; created, clarifies definition of "foster care services." Amending §§ 16.1-228, 16.1-282.1, 63.2-100, 63.2-905, 63.2-906, and 63.2-1305; adding § 63.2-1306. (Patron–Mason, SB 1328)

Status offenders; willful and material violation of court order or terms of probation, notice, orders of disposition for violation. Amending §§ 16.1-278.5, 16.1-291, and 16.1-292. (Patron–Scott, HB 2056)

COURTS OF RECORD

Criminal records; establishes a process for automatic expungement, etc., definitions, penalties, report. Amending §§ 9.1-101, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-340, 19.2-389.3, and 19.2-390; adding §§ 19.2-392.5 through 19.2-392.16. (Patron–Herring, HB 2113)

Jurisdiction over criminal cases; certification by general district court of felony or ancillary misdemeanor charges, jurisdiction to such charges shall vest in circuit court. Amending §§ 16.1-123.1, 16.1-241, and 17.1-513. (Patron–Adams, L.R., HB 2150)

State Corporation Commission; adds the Commission to the list of agencies that are exempt from paying fees for remote access to local land records. Amending § 17.1-276. (Patron-Kilgore, HB 1775)

COVID-19

Concealed handgun permits; demonstration of competence, eligibility to apply for permit due to restrictions of COVID-19. (Patron–Runion, HB 2310)

COVID-19; Department of Medical Assistance Services shall deem testing, treatment, and vaccination to be emergency services. (Patron-Lopez, HB 2124)

COVID-19; Joint Legislative Audit and Review Commission to study the impact on Virginia's public schools, students, and school employees, meetings shall be completed by November 30, 2022. (Patron–Guy, HJR 549; Lucas, SJR 308)

COVID-19 (continued)

Employers; reporting outbreaks of COVID-19, effective clause. (Patron-Lewis, SB 1362)

- Unemployment compensation; under specific conditions related to the COVID-19 virus, work will not be deemed suitable and benefits will not be denied to any otherwise eligible individual for refusing to accept new work. Amending §§ 60.2-528 and 60.2-618. (Patron–Tran, HB 2037)
- Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of certain health care providers is an occupational disease. Amending § 65.2-402.1. (Patron–Hurst, HB 1985)
- Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, emergency medical services personnel, law-enforcement officers, and correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Jones, HB 2207)
- Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, law-enforcement officers or correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Saslaw, SB 1375)

COVINGTON, CITY OF

Covington, City of; amending charter, consolidation of school divisions of the City of Covington and Alleghany County, salaries. Amending § 22.1-32. (Patron–Austin, HB 2091; Deeds, SB 1267)

CREDIT CARDS, CREDIT SERVICES, AND CREDIT UNIONS

State Corporation Commission; supervisory merger or transfer of assets of financially unstable credit unions, repeals provision relating to consolidation or merger. Amending § 6.2-1317; repealing § 6.2-1318. (Patron–Bagby, HB 1964)

CREWE, TOWN OF

Crewe, Town of; amending charter, various changes to the charter including staggering town council elections, etc. (Patron–Wright, HB 1764; Ruff, SB 1216)

CRIMES AND OFFENSES GENERALLY

- Assault and battery; eliminates the mandatory minimum term of confinement of six months if committed against a judge, magistrate, law-enforcement officer, etc., persons with physical or mental disability. Amending § 18.2-57; adding § 19.2-9.2. (Patron–Morrissey, SB 1306)
- Barrier Crimes and Criminal History Records Checks, Joint Subcommittee Studying; continued, appropriation. (Patron-Edwards, SJR 285)
- Behavioral health docket; standards for transfer of supervision between local community-based probation agencies. Amending §§ 9.1-175 and 18.2-254.3. (Patron–Bell, HB 2236)
- Bribery in correctional facilities; penalty. Adding § 18.2-474.2. (Patron-Lewis, SB 1461)
- Charitable gaming; definitions, regulations, conduct of instant bingo, network bingo, pull tabs, and seal cards, report. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.34. (Patron–Reeves, SB 1127)
- Charitable gaming; increase in certain maximum allowable prize amounts. Amending § 18.2-340.33. (Patron–Keam, HB 1843)
- Charitable gaming; regulations of operators of Texas Hold'em poker games, conduct of games, effective clause. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.18, 18.2-340.19, 18.2-340.22, 18.2-340.28:2, and 18.2-340.34:2. (Patron–Bell, SB 1278)
- Charitable Gaming Board; prohibits the Board from promulgating regulations that prohibit the use of multiple video monitors or touchscreens on an electronic pull tab device, use of devices shall be limited to one player at a time. Amending §§ 18.2-340.19 and 18.2-340.28. (Patron–Willett, HB 1943; McPike, SB 1287)
- Communicating threats of death or bodily injury to a person with intent to intimidate, etc.; threats in writing, including electronically transmitted communications, penalties. Amending § 18.2-60. (Patron–Leftwich, HB 2194)
- Communicating threats of death or bodily injury to a person with intent to intimidate, etc.; threats to another in writing, penalties. Amending § 18.2-60. (Patron–Spruill, SB 1113)
- Death penalty; abolishes current penalty, including for those persons currently under a death sentence, repeals provisions relating to trial of capital cases. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22,

CRIMES AND OFFENSES GENERALLY (continued)

18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Mullin, HB 2263)

Death penalty; abolition of current penalty, sentences changed to life imprisonment, repeals provisions referring to trial of capital cases, etc. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Surovell, SB 1165)

Experiencing or reporting overdoses; prohibits arrest and prosecution. Amending § 18.2-251.03. (Patron–Bulova, HB 1821)

Firearm; carrying within Capitol Square and the surrounding area, into building owned or leased by the Commonwealth, etc., penalty. Adding § 18.2-283.2. (Patron–Levine, HB 2295)

Firearms; access to any individual who resides or cohabitates in a shared dwelling, civil penalty. Adding § 18.2-308.2:02. (Patron–Guzman, HB 2319)

Firearms; criminal history record information check delay increased to five days. Amending § 18.2-308.2:2. (Patron–Lopez, HB 2128)

Firearms; purchase, possession, etc., following conviction for assault and battery of a family or household member, definition, penalties. Amending §§ 18.2-308.09, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:8. (Patron–Murphy, HB 1992)

Homicides and assaults and bodily woundings; certain matters not to constitute defenses. Adding §§ 18.2-37.1 and 18.2-57.5. (Patron–Roem, HB 2132)

Illegal gambling; skill games, definitions, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron–Reeves, SB 1465)

Illegal gambling; skill games, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron–Scott, HB 2168)

Larceny; repeals punishment for conviction of second or subsequent misdemeanor. Repealing § 18.2-104. (Patron–Plum, HB 2290)

Mandatory minimum sentences; elimination, modification of sentence to mandatory minimum term of confinement for felony offenses. Amending §§ 3.2-4212, 4.1-302, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5, 18.2-255, 18.2-255.2, 46.2-357, and 46.2-391; adding § 19.2-303.03. (Patron–Mullin, HB 2331)

Mandatory minimum sentences; elimination, modification of sentence to mandatory minimum term of confinement for felony offenses, report. Amending §§ 3.2-4212, 4.1-302, 16.1-253.2, 18.2-36.1, 18.2-36.2, 18.2-46.3:3, 18.2-51.1, 18.2-53.1, 18.2-57, 18.2-60.4, 18.2-61, 18.2-67.1, 18.2-67.2, 18.2-67.5:2, 18.2-67.5:3, 18.2-154, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5, 18.2-255, 18.2-255.2, 18.2-270, 18.2-308.1, 18.2-308.2, 18.2-308.2:2, 18.2-308.4, 18.2-374.1, 18.2-374.1:1, 18.2-374.3, 19.2-297.1, 46.2-341.28, 46.2-357, 46.2-391, 46.2-865.1, and 53.1-203. (Patron–Edwards, SB 1443)

Marijuana; legalization of simple possession, etc., expungement of criminal records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.07, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-352, 4.1-353,

CRIMES AND OFFENSES GENERALLY (continued)

4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-390, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9, 19.2-392.11, 19.2-392.12, 19.2-392.10, 19.2-392.13, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Herring, HB 2312)

Marijuana; unlawful possession in certain cases, using or consuming while in a motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, $16.1\text{-}278.9, \quad 17.1\text{-}276, \quad 18.2\text{-}46.1, \quad 18.2\text{-}57, \quad 18.2\text{-}247, \quad 18.2\text{-}248, \quad 18.2\text{-}248.01, \quad 18.2\text{-}251, \quad 18.2\text{-}248.01, \quad 18.2\text{-}248.01, \quad 18.2\text{-}248.01, \quad 18.2\text{-}251, \quad 18.2\text{-}248.01, \quad 18.2\text{-}$ 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, $33.2-613,\ 46.2-105.2,\ 46.2-347,\ 48-17.1,\ 51.1-212,\ 53.1-231.2,\ 54.1-2903,\ 54.1-3408.3,$ 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Ebbin, SB 1406)

- Military honor guards and veterans service organizations; paramilitary activities. Amending §§ 18.2-282 and 18.2-433.2. (Patron–Reeves, SB 1129)
- Pharmaceutical processors; permits processors to produce and distribute cannabis products, dispensing botanical cannabis to a minor, certification. Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8. (Patron–Lucas, SB 1333)
- Pharmaceutical processors; permits processors to produce and distribute cannabis products, testing standards for botanical cannabis and products. Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8. (Patron–Hayes, HB 2218)
- Plastic firearms and unfinished frames, etc.; manufacture, import, sale, transfer, etc., penalty. Amending § 18.2-308.5; adding § 18.2-308.5:2. (Patron–Simon, HB 2276)
- Prostitution; reorganizes the statute penalizing into two distinct sections. Amending §§ 8.01-42.4, 9.1-116.5, 9.1-902, 16.1-69.48:6, 16.1-69.55, 17.1-275.13, 17.1-805, 18.2-46.1, 18.2-346, 18.2-346.1, 18.2-350, 18.2-357.1, 18.2-513, 19.2-10.2, 19.2-215.1, 19.2-268.3, 19.2-386.16, 19.2-386.35, 19.2-392.02, 32.1-58, 37.2-314, 37.2-416, and 37.2-506; adding § 18.2-346.01. (Patron–Mundon King, HB 2169)

CRIMES AND OFFENSES GENERALLY (continued)

Restricted licenses; authorizes DMV to issue restricted credentials to individuals with driver's license suspensions resulting from drug-related offenses. Amending § 18.2-271.1. (Patron–Edwards, SB 1213)

Restricted permit; prepayment of fines and costs. Amending § 18.2-271.1. (Patron-Morrissey, SB 1262)

Restricted permits to operate a motor vehicle; ignition interlock systems. Adding § 18.2-271.5. (Patron-Stuart, SB 1336)

Robbery; definition, creates degrees of punishment corresponding to the severity of a robbery offense. Amending § 18.2-58. (Patron–Watts, HB 1936)

Sexually transmitted infections; infected sexual battery, penalty, repeals the crime of donating or selling blood, body fluids, etc., by persons infected with human immunodeficiency virus (HIV), etc. Amending §§ 18.2-52.1, 18.2-67.4:1, 18.2-346.1, 32.1-291.16, 54.1-2982, 54.1-2983, and 57-48; repealing §§ 18.2-62 and 32.1-289.2. (Patron–Locke, SB 1138)

Suicide; abolishes the common-law crime. Adding § 18.2-16.1. (Patron–Simon, HB 1951)

Unlawful dissemination of unsolicited obscene image of self to another; penalty. Adding § 18.2-386.3. (Patron–Convirs-Fowler, HB 2254)

Victims of sex trafficking; clarifies definition, affirmative defense to prosecution for certain offenses. Adding § 18.2-361.1. (Patron–Brewer, HB 2234)

Weapons; possessing or transporting within Capitol Square or into a building owned or leased by the Commonwealth, provisions shall not apply to certain law-enforcement officers, etc., penalty. Adding § 18.2-283.2. (Patron–Ebbin, SB 1381)

CRIMINAL HISTORY INFORMATION

Barrier Crimes and Criminal History Records Checks, Joint Subcommittee Studying; continued, appropriation. (Patron–Edwards, SJR 285)

Behavioral Health and Developmental Services, Department of; background checks, persons providing contractual services. Amending §§ 19.2-389, 37.2-416, and 37.2-506. (Patron–Willett, HB 2092)

Child care providers; background checks portability, check shall not be required if individual completed a background check within previous five years, pilot program, report. Amending §§ 19.2-389, 22.1-289.035, 22.1-289.039, 63.2-1720.1, and 63.2-1724. (Patron–McGuire, HB 2086; McClellan, SB 1316)

Firearms; criminal history record information check delay increased to five days. Amending § 18.2-308.2:2. (Patron–Lopez, HB 2128)

Higher educational institutions, public; admissions applications that contain questions about criminal history. Adding § 23.1-407.1. (Patron–Aird, HB 1930)

CRIMINAL JUSTICE SERVICES

Criminal justice, behavioral health, and other human services records; Department of Behavioral Health and Developmental Services to establish a work group to study the feasibility of developing a secure, de-identified, etc., database. (Patron–Price, HJR 578)

Criminal Justice Services Board and Committee on Training; membership, law-enforcement training. Amending §§ 9.1-102, 9.1-108, and 9.1-112. (Patron–Marsden, SB 1256)

CRIMINAL PROCEDURE

Admission to bail; rebuttable presumptions against bail, repeals provision relating to presumption of no bail for illegal aliens charged with certain crimes. Amending §§ 19.2-120 and 19.2-124; repealing § 19.2-120.1. (Patron–Deeds, SB 1266)

Assault and battery; eliminates the mandatory minimum term of confinement of six months if committed against a judge, magistrate, law-enforcement officer, etc., persons with physical or mental disability. Amending § 18.2-57; adding § 19.2-9.2. (Patron–Morrissey, SB 1306)

Behavioral Health and Developmental Services, Department of; background checks, persons providing contractual services. Amending §§ 19.2-389, 37.2-416, and 37.2-506. (Patron–Willett, HB 2092)

Child care providers; background checks portability, check shall not be required if individual completed a background check within previous five years, pilot program, report. Amending §§ 19.2-389, 22.1-289.035, 22.1-289.039, 63.2-1720.1, and 63.2-1724. (Patron–McGuire, HB 2086; McClellan, SB 1316)

CRIMINAL PROCEDURE (continued)

- Commercial sex trafficking; issuance of writ of vacatur for victims. Adding §§ 19.2-327.15 through 19.2-327.20. (Patron–Delaney, HB 2133)
- Criminal proceedings; consideration of mental condition and intellectual and developmental disabilities, evidence offered by the defendant. Amending §§ 19.2-120, 19.2-163.03, and 19.2-299; adding § 19.2-271.6. (Patron–McClellan, SB 1315)
- Criminal proceedings; consideration of mental condition and intellectual and developmental disabilities, qualifications for court-appointed council. Amending §§ 19.2-120, 19.2-163.03, and 19.2-299; adding § 19.2-271.6. (Patron–Bourne, HB 2047)
- Criminal records; establishes a process for automatic expungement, etc., definitions, penalties, report. Amending §§ 9.1-101, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-340, 19.2-389.3, and 19.2-390; adding §§ 19.2-392.5 through 19.2-392.16. (Patron–Herring, HB 2113)
- Death penalty; abolishes current penalty, including for those persons currently under a death sentence, repeals provisions relating to trial of capital cases. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Mullin, HB 2263)
- Death penalty; abolition of current penalty, sentences changed to life imprisonment, repeals provisions referring to trial of capital cases, etc. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-152, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Surovell, SB 1165)
- Fines and costs; accrual of interest, deferral or installment payment agreements. Amending §§ 19.2-353.5 through 19.2-355. (Patron–Hudson, HB 1895)
- Firearms; purchase, possession, etc., following conviction for assault and battery of a family or household member, definition, penalties. Amending §§ 18.2-308.09, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:8. (Patron–Murphy, HB 1992)
- Habitual offenders; required that the Commissioner of DMV reinstate a person's privilege to drive a motor vehicle that was suspended or revoked solely on the basis that such person was determined to be or adjudicated a habitual offender, repeals remaining provisions of the Habitual Offender Act. Amending §§ 8.01-9, 8.01-407, 16.1-77, 16.1-305, 17.1-213, 19.2-389, 46.2-301, 46.2-301.1, 46.2-411, and 53.1-21; repealing §§ 46.2-355.1 through 46.2-363. (Patron–Stanley, SB 1122)
- Mandatory minimum sentences; elimination, modification of sentence to mandatory minimum term of confinement for felony offenses. Amending §§ 3.2-4212, 4.1-302, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5, 18.2-255, 18.2-255.2, 46.2-357, and 46.2-391; adding § 19.2-303.03. (Patron–Mullin, HB 2331)
- Marijuana; legalization of simple possession, etc., expungement of criminal records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325, 4.1-325, 4.1-326, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820,

CRIMINAL PROCEDURE (continued)

16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-390, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9, 19.2-392.11, 19.2-392.10, 19.2-392.12, 19.2-392.13, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Herring, HB 2312)

Marijuana; unlawful possession in certain cases, using or consuming while in a motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, $4.1-101.010,\ 4.1-101.1,\ 4.1-103,\ 4.1-104,\ 4.1-105,\ 4.1-106,\ 4.1-107,\ 4.1-111,\ 4.1-112.2,$ 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Ebbin, SB 1406)

- Orders of restitution; docketed on behalf of victim, assignment of judgment, enforcement. Amending §§ 19.2-305.1, 19.2-305.2, 19.2-349, and 19.2-354. (Patron–Bell, HB 2233; Stanley, SB 1426)
- Personal appearance by two-way electronic video and audio communication; entry of plea or nolle prosequi, adjudication of probation violations. Amending § 19.2-3.1. (Patron–Edwards, SB 1242)
- Police and court records; expungement and sealing of records, Expungement Fee Fund created, protection of public record information, penalties, report, effective clause. Amending §§ 9.1-128, 9.1-134, 17.1-502, 19.2-392.1, 19.2-392.2, 19.2-392.3, and 19.2-392.4; adding §§ 1-229.1, 17.1-205.1, 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, 19.2-392.4:1, 19.2-392.5, 19.2-392.6, and 19.2-392.7. (Patron–Surovell, SB 1339)
- Post-conviction relief; previously admitted forensic scientific evidence, effective date, report, and effective clause. Adding § 19.2-327.15. (Patron-Stanley, SB 1105)
- Pretrial data collection; Virginia Criminal Sentencing Commission to collect and disseminate on an annual basis, report. Amending § 2.2-3802; adding § 19.2-134.1. (Patron–Herring, HB 2110; Lucas, SB 1391)

CRIMINAL PROCEDURE (continued)

- Probation, revocation, and suspension of sentence; limitations on sentence, exceptions, technical violation. Amending §§ 19.2-303, 19.2-303.1, and 19.2-306; adding § 19.2-306.1. (Patron–Scott, HB 2038)
- Prostitution; reorganizes the statute penalizing into two distinct sections. Amending §§ 8.01-42.4, 9.1-116.5, 9.1-902, 16.1-69.48:6, 16.1-69.55, 17.1-275.13, 17.1-805, 18.2-46.1, 18.2-346, 18.2-346.1, 18.2-350, 18.2-357.1, 18.2-513, 19.2-10.2, 19.2-215.1, 19.2-268.3, 19.2-386.16, 19.2-386.35, 19.2-392.02, 32.1-58, 37.2-314, 37.2-416, and 37.2-506; adding § 18.2-346.01. (Patron–Mundon King, HB 2169)
- Public defender office; establishes an office for the County of Chesterfield, effective clause. Amending § 19.2-163.04. (Patron–Morrissey, SB 1442)
- Search warrants; date and time of issuance, law-enforcement officer to be recognizable and identifiable, exceptions. Amending § 19.2-56. (Patron–Stuart, SB 1475)
- Substantial Risk Order Registry; Department of State Police shall keep and maintain a computerized Registry, etc. Amending § 19.2-387.3. (Patron–Simonds, HB 2258)
- Summons; eliminates the requirement that a promise to appear be completed after issuance for a misdemeanor offense or an administrative violation. Amending §§ 2.2-311, 19.2-74, 28.2-901, 29.1-210, 46.2-936, 46.2-940, 54.1-306, 54.1-2506, and 54.1-4407. (Patron–Hanger, SB 1437)
- Suspension or modification of sentence; transfer to the Department of Corrections. Amending § 19.2-303. (Patron–Kilgore, HB 1806)
- Unrestorably incompetent defendant; competency report. Amending § 19.2-169.1. (Patron–Mason, SB 1431)
- Unrestorably incompetent defendant; disposition, capital murder charge, inpatient custody of the Commissioner of the Department of Behavioral Health and Developmental Services. Adding § 19.2-169.3:1. (Patron–Mason, SB 1272)
- Victims of crime; compensation, reporting requirement. Amending § 19.2-368.10. (Patron–Delaney, HB 1867)
- Virginia Freedom of Information Act; law-enforcement criminal incident information, criminal investigative files. Amending §§ 2.2-3706, 2.2-3711, 19.2-174.1, and 19.2-368.3; adding § 2.2-3706.1. (Patron–Hurst, HB 2004)

CROSS, ORETHA P.

Cross, Oretha P.; commending. (Patron-Price, HJR 784)

CRUTCHER, RONALD A.

Crutcher, Ronald A.; commending. (Patron-VanValkenburg, HJR 696)

DABNEY, HERBERT ALLEN, III

Dabney, Herbert Allen, III; recording sorrow upon death. (Patron–Carr, HJR 647)

DAIRY PRODUCTS

- Dairy Producer Margin Coverage Premium Assistance Fund and Program; created and established, eligible dairy producer shall apply to the Department of Conservation and Recreation by February 1 of each year to participate, effective clause. Adding §§ 3.2-3304 through 3.2-3307. (Patron-Obenshain, SB 1193)
- Dairy Producer Margin Coverage Premium Assistance Program; established, eligible dairy producer shall apply to the Department of Conservation and Recreation by February 1 of each year to participate, report, sunset date. Adding §§ 3.2-3304 through 3.2-3307. (Patron–Gooditis, HB 1750)

DAMS

Dams; Virginia Soil and Water Conservation Board to enter into a negotiated settlement with the owners of certain impounding structures. Adding § 10.1-613.6. (Patron–Bell, SB 1280)

DANVILLE LIFE SAVING AND FIRST AID CREW, INC.

Danville Life Saving and First Aid Crew, Inc.; commemorating its 75th anniversary. (Patron-Marshall, HJR 774)

DEATH PENALTY

Death penalty; abolishes current penalty, including for those persons currently under a death sentence, repeals provisions relating to trial of capital cases. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Mullin, HB 2263)

Death penalty; abolition of current penalty, sentences changed to life imprisonment, repeals provisions referring to trial of capital cases, etc. Amending §§ 2.2-3705.7, 8.01-195.10, 8.01-654, 17.1-310, 17.1-406, 18.2-8, 18.2-10, 18.2-18, 18.2-19, 18.2-22, 18.2-25, 18.2-26, 18.2-30, 18.2-31, 18.2-32, 18.2-251.01, 19.2-11.01, 19.2-71, 19.2-76.1, 19.2-100, 19.2-102, 19.2-120, 19.2-152.2, 19.2-157, 19.2-159, 19.2-163, 19.2-163.01, 19.2-163.4:1, 19.2-169.3, 19.2-175, 19.2-217.1, 19.2-247, 19.2-270.4:1, 19.2-295.3, 19.2-299, 19.2-299.1, 19.2-311, 19.2-319, 19.2-321.2, 19.2-327.1, 19.2-327.3, 19.2-327.11, 19.2-389.1, 19.2-389.3, 19.2-400, 53.1-204, 53.1-229, and 54.1-3307; repealing §§ 8.01-654.1, 8.01-654.2, 17.1-313, 18.2-17, 19.2-163.7, 19.2-163.8, 19.2-264.2 through 19.2-264.5, 53.1-230, and 53.1-232 through 53.1-236. (Patron–Surovell, SB 1165)

DEEDS AND DEEDS OF TRUST

Deeds of trust; amendment to loan document, statement of interest rate of a refinanced mortgage. Amending § 55.1-319; adding § 55.1-318.1. (Patron–Heretick, HB 1882)

DEEDS, R. CREIGH

Leave of absence.	264
Prayer offered	135
Statements on votes:	
S.B. 1311	477
S.J.R. 299	164

DEFENDANTS

Criminal proceedings; consideration of mental condition and intellectual and developmental disabilities, evidence offered by the defendant. Amending §§ 19.2-120, 19.2-163.03, and 19.2-299; adding § 19.2-271.6. (Patron–McClellan, SB 1315)

Victims of crime; definitions, certifications for victims of qualifying criminal activity, disclosure of exculpatory information to a defendant in a criminal case. Adding §§ 9.1-1500, 9.1-1501, and 9.1-1502. (Patron–Surovell, SB 1468)

DENNY, ALFRED JEROME

Denny, Alfred Jerome; recording sorrow upon death. (Patron–Hope, HJR 653)

DESTEPH, WILLIAM R., JR.

DICK, PAUL WENDELL

Dick, Paul Wendell; recording sorrow upon death. (Patron-LaRock, HJR 743)

DISASTER

Annual safety and disaster awareness training; Department of Human Resource Management, et al., to develop an online training module addressing safety and disaster awareness. Amending § 2.2-1201; adding § 2.2-1212. (Patron–Ayala, HB 1891)

Continuity of government; extends to 12 months the period of time after an enemy attack or other disaster that a locality may, by ordinance, provide. Amending § 15.2-1413. (Patron–Barker, SB 1208)

Employee classification; provision of personal protective equipment in response to a disaster. Amending §§ 40.1-28.7:7 and 60.2-212; adding § 65.2-301.2. (Patron–Batten, HB 2134)

DISCRIMINATION

- Active military or a military spouse; definition of "military status," discrimination in public accommodations, employment, and housing. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310. (Patron—Tran, HB 2161)
- Active military or a military spouse; prohibits discrimination in public accommodations, employment, and housing, landlords may not charge a higher security deposit. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1226, and 55.1-1310. (Patron–Bell, SB 1410)
- Discrimination; prohibited in voting and elections administration, required process for enacting certain covered practices, civil causes of action, penalties, repeals provision relating to minority language accessibility, etc. Amending §§ 24.2-105, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1; adding §§ 24.2-104.1, 24.2-125 through 24.2-131, and 24.2-1005.2; repealing § 24.2-124. (Patron–Price, HB 1890; McClellan, SB 1395)
- Health insurance; carrier business practices, every carrier shall include in provider contracts a provision that prohibits a provider from discriminating against any enrollee. Amending § 38.2-3407.15. (Patron–Surovell, SB 1289)
- Virginia Fair Housing Law; unlawful discriminatory housing practices. Amending §§ 36-96.3 and 36-96.17. (Patron–Bourne, HB 2046)
- Virginia Human Rights Acts; adds discrimination on the basis of disability as an unlawful employment practice, reasonable accommodations for persons with disabilities. Amending §§ 2.2-3902, 2.2-3905, and 51.5-41; adding § 2.2-3905.1. (Patron–Sickles, HB 1848)

DISTRICT COURTS

- General district courts; increases to \$50,000 the maximum civil jurisdictional limit of courts, appeal bond. Amending §§ 8.01-195.4, 16.1-77, and 16.1-107. (Patron–Stanley, SB 1108)
- Jurisdiction over criminal cases; certification by general district court of felony or ancillary misdemeanor charges, jurisdiction to such charges shall vest in circuit court. Amending §§ 16.1-123.1, 16.1-241, and 17.1-513. (Patron–Adams, L.R., HB 2150)

DIVORCE

No-fault divorce; corroboration requirement. Amending §§ 20-99 and 20-106. (Patron-Hope, HB 1911)

DOCUMENTS

- SD1 The State of the Commonwealth Address to the Joint Assembly 2021 Governor Ralph S. Northam
- SD2 List of Pardons, Commutations, Reprieves and Other Forms of Clemency January 11, 2020 to January 15, 2021
- SD3 Report of the Virginia Facilitated Enrollment Work Group (Chapter 973, 2020)
- SD4 Aviation Workforce Development Study (SJR 30, 2020)
- SD5 Interim Report of the Joint Subcommittee to Study Barrier Crimes and Criminal History Records Checks in the Commonwealth in the 21st Century (SJR 35, 2020)
- SD6 Audio-Visual Recording in Assisted Living Facilities (Chapter 848, 2020)
- SD7 Reductions of Average Time for Certificate of Public Need Review (Chapter 1271, 2020)
- SD8 Report and Work Plan of the American Revolution 250 Commission (Chapter 915, 2020)
- HD1 Budget Bill
- HD2 The Virginia Plan for Higher Education 2020 Annual Report (HJR 555, 2015)
- HD3 Feasibility Study of Developing an Early Childhood Mental Health Consultation Program (HJR 51, 2020)
- HD4 Ransomware Study Report (HJR 64, 2020)
- HD5 E-911 Border Response Workgroup Report (2020 Appropriation Act, Item 391.F.)
- HD6 Report of the Virginia Freedom of Information Advisory Council December 2020

DOCUMENTS (continued)

- HD7 Report on a Protocol to Determine Whether an Unpermitted Discharge Has a De Minimis Impact on the Beneficial Uses of State Waters and a Proposed Implementation Procedure if § 62.1-44.19:6 of the Code of Virginia Were Amended to Require Dissemination of all Unpermitted Discharges Reported Pursuant to § 62.1-44.5 of the Code of Virginia Except for Those Determined to Have a De Minimis Impact on the Beneficial Uses of State Waters (Chapter 1182, 2020)
- HD8 Proactive Enforcement of Virginia's Equal Pay Law A Report by the Division of Human Rights, Virginia Office of the Attorney General
- HD9 Report of the Small Business Commission Pursuant to HJ 687 (Regular Session, 2019)
- HD10 Earned Sentence Credit Legislation Workgroup Report (Chapter 50, 2020 SSI)

DOGS AND DOG LAWS

- Dangerous dogs; restructures procedure for adjudication, penalty. Amending §§ 3.2-6540 and 3.2-6542; adding §§ 3.2-6540.01 through 3.2-6540.04, 3.2-6541.1, 3.2-6542.1, 3.2-6542.2, 3.2-6543.1, 3.2-6562.2, and 18.2-52.2. (Patron–Marsden, SB 1135)
- Healthcare Regulatory Sandbox Program; established, report, sunset date. Amending § 3.2-6591; adding § 3.2-6593.1. (Patron–Stanley, SB 1417)

DOMESTIC RELATIONS

- Child support; availability of medical assistance through Family Access to Medical Insurance Security (FAMIS) plan or other government-sponsored coverage, eligibility requirements. Amending §§ 16.1-260 and 63.2-1903. (Patron–Samirah, HB 2002)
- Child support obligations; party's incarceration not deemed voluntary unemployment or underemployment. Amending §§ 20-108.1 and 63.2-1918. (Patron–Scott, HB 2055)
- Constitutional amendment; fundamental right to marry, removes same-sex marriage prohibition (first reference). Amending Section 15-A of Article I. (Patron–Sickles, HJR 582; Ebbin, SJR 270)
- Court-ordered custody and visitation arrangements; petition of grandparent for visitation with minor grandchild. Amending § 20-124.2. (Patron–Dunnavant, SB 1325)
- Marriage; persons who may celebrate rites, authorizes current members of the General Assembly, Governor, Lieutenant Governor, and Attorney General. Amending § 20-25. (Patron–Cosgrove, SB 1142)
- No-fault divorce; corroboration requirement. Amending §§ 20-99 and 20-106. (Patron-Hope, HB 1911)
- Sexual and Domestic Violence, Advisory Committee on; increases membership, duties. Amending §§ 9.1-116.2 and 9.1-116.3. (Patron–Robinson, HB 2317)
- Support orders; contents of orders, change in employment status, unemployment benefits. Amending §§ 20-60.3 and 63.2-1916. (Patron–Leftwich, HB 2192)
- Uniform Collaborative Law Act; created. Adding §§ 20-168 through 20-187. (Patron-Sullivan, HB 1852)

DOWNING, ANNIE LAURA

Downing, Annie Laura; commending. (Patron-Wyatt, HJR 695)

DRIVER EDUCATION PROGRAM

- Student driver education program; parent/student component exemption. Amending § 22.1-205. (Patron–Keam, HB 2119)
- Student driver safety; driver education program shall include the dangers of distracted driving and speeding, high school student parking passes, valid driver's license required. Amending § 22.1-205; adding § 22.1-205.1. (Patron–Mugler, HB 1918)

DRIVERS' LICENSES

- Commercial driver's licenses; disqualifies for life from holding a license persons convicted of a felony involving an act or practice of severe forms of trafficking, etc. Amending §§ 46.2-341.18, 46.2-382, and 46.2-1702. (Patron–Delaney, HB 1868)
- High school student parking passes; valid driver's license or driver privilege card required. Adding § 22.1-205.1. (Patron–Norment, SB 1169)

DRIVERS' LICENSES (continued)

Learner's permits; use of personal communication devices. Amending § 46.2-335. (Patron-Stuart, SB 1335)

Restricted licenses; authorizes DMV to issue restricted credentials to individuals with driver's license suspensions resulting from drug-related offenses. Amending § 18.2-271.1. (Patron–Edwards, SB 1213)

Restricted permit; prepayment of fines and costs. Amending § 18.2-271.1. (Patron-Morrissey, SB 1262)

DRUG ABUSE

Hospitals; emergency treatment for substance use-related emergencies, Department of Health, et al., to develop recommendations for best practices for treatment and discharging of patients in emergency departments, etc. Amending § 32.1-127. (Patron–Delaney, HB 2300)

DUKAS, HELEN

Dukas, Helen; recording sorrow upon death. (Patron-Keam, HJR 701)

DUNNAVANT, SIOBHAN S.

Notified Clerk of presence4	433
Prayer offered	80

EASEMENTS

Conservation easements; an easement held pursuant to the Virginia Conservation Easement Act or the Open-Space Land Act, etc., be liberally construed in favor of achieving the purposes for which it was created. Adding §§ 10.1-1016.1 and 10.1-1705.1. (Patron–Webert, HB 1760)

Constitutional amendment; authority to grant perpetual easements to units of government (first reference). Amending Section 9 of Article VII. (Patron–Mason, SJR 289)

ECONOMIC DEVELOPMENT

Access roads to economic development sites; criteria for use of funds. Amending § 33.2-1509. (Patron-McPike, SB 1253)

Conflict of Interests Act, State and Local Government; disclosure requirements, industrial development and economic development authorities. Amending § 2.2-3115. (Patron–Norment, SB 1171)

Economic development authorities; size of board in Powhatan County, reduces quorum requirement. Amending § 15.2-4904. (Patron–Ware, HB 2287)

Mathews County; board of supervisors may appoint one employee of the locality onto the board of directors for the Economic Development Authority of the County. Amending § 15.2-4904. (Patron–Hodges, HB 2186)

EDUCATION

Broadband services; authorizes school boards to appropriate funds for the purposes of promoting, facilitating, and encouraging the expansion and operation of services for educational purposes. Adding §§ 15.2-986 and 22.1-79.9. (Patron–Boysko, SB 1225)

Brunswick County school board; removes school board from the list of approved member salaries for appointed school boards. Amending § 22.1-32. (Patron–Tyler, HB 1798; Ruff, SB 1175)

Comprehensive review of computer science standards, courses, and pathways in public schools; Department of Education to perform, report. (Patron–Simonds, HB 1885)

Constitutional amendment; equitable educational opportunities in all public schools in the Commonwealth (first reference). Amending Sections 1 and 2 of Article VIII. (Patron–Stanley, SJR 275)

Consumer Protection Act; prohibited practices, certain advertising related to any public or private school quality. Amending § 59.1-200. (Patron–Samirah, HB 2003)

Covington, City of; amending charter, consolidation of school divisions of the City of Covington and Alleghany County, salaries. Amending § 22.1-32. (Patron–Austin, HB 2091; Deeds, SB 1267)

Early childhood education; quality rating and improvement system participation, School Readiness Committee reinstated. Amending § 2.2-208.1 and second and eleventh enactments of Chapters 860 and 861, 2020 Acts. (Patron–Bulova, HB 2105)

EDUCATION (continued)

- Economic education and financial literacy required in middle and high school grades; employment arrangements. Amending § 22.1-200.03. (Patron–Cole, J.G., HB 1905)
- Education, Board of; geographic representation of members. Amending § 22.1-9. (Patron-Austin, HB 1827)
- Education, Board of; temporary extension of the license of any individual licensed by Board whose license expires on June 30, 2021. (Patron-Ward, HB 1776)
- Governor's Schools; Board of Education shall issue guidance on the governance of academic year, including admissions policies, and guidelines on diversity, etc. (Patron–Tyler, HB 2305)
- Health Standards of Learning; advanced directive education for high school students. (Patron-Kiggans, SB 1190)
- High school student parking passes; valid driver's license or driver privilege card required. Adding § 22.1-205.1. (Patron-Norment, SB 1169)
- Isle of Wight County; authorized to impose an additional local sales and use tax to support schools. Amending §§ 58.1-602 and 58.1-605. (Patron–Norment, SB 1170)
- Kindergarten through grade 3; reading intervention services for certain students. Amending § 22.1-253.13:1. (Patron–Delaney, HB 1865)
- Licensed private schools for students with disabilities; accreditation. Amending § 22.1-323. (Patron-Kory, HB 2238)
- Local elections for governing bodies; elections for school boards, qualification of voters. Amending §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223. (Patron–Convirs-Fowler, HB 2198)
- Local school divisions; availability of virtual and in-person learning to all students. (Patron-Dunnavant, SB 1303)
- Loudoun County school board; staggered terms of its members. Amending § 22.1-57.3:1.1. (Patron-Reid, HB 1838)
- Public elementary and secondary schools; possession and administration of undesignated stock albuterol inhalers and valved holding chambers, Department of Health, in conjunction with the Department of Education, shall develop and implement policies. Amending §§ 8.01-225, 22.1-274.2, and 54.1-3408. (Patron–McQuinn, HB 2019)
- Public School Assistance Fund and Program; created. Adding § 22.1-141.3. (Patron-Stanley, SB 1106)
- Public schools; lock-down drills, annual requirement. Amending § 22.1-137.2. (Patron–Murphy, HB 1998)
- Public schools; seizure management and action plans, biennial training, effective date. Amending § 8.01-225; adding § 22.1-274.6. (Patron–DeSteph, SB 1322)
- Public schools; severe weather conditions and other emergency situations, unscheduled remote learning days, school provides instruction and student services, etc. Amending § 22.1-98. (Patron–McNamara, HB 1790; Suetterlein, SB 1132)
- Public schools, child day programs, and certain other programs; carbon monoxide detectors required. Adding §§ 22.1-138.2, 22.1-289.058, and 63.2-1705.2. (Patron–Askew, HB 1823)
- School board policies; abusive work environments, definitions. Amending § 22.1-291.4. (Patron-Torian, HB 2176)
- School board property, certain; establishment of gun-free zone permitted. Adding § 22.1-131.1. (Patron–Subramanyam, HB 1909)
- School boards; each board to adopt a policy that prohibits a lawsuit against a student or the student's parent because student cannot pay for a meal at school, etc. Amending § 22.1-79.7. (Patron–Roem, HB 2013)
- School boards, certain; participation in the Afterschool Meal Program. Adding § 22.1-207.4:2. (Patron-Roem, HB 2135)
- School nurses; no individual who provides nursing services in a public elementary or secondary school shall use title unless individual is a registered nurse. Amending § 22.1-274. (Patron–Adams, D.M., HB 1736)
- Special education; Board of Education to amend a certain regulation to remove the word "component" following the word "evaluation," thereby ensuring compliance with the relevant federal regulation. (Patron–Mugler, HB 2314)

EDUCATION (continued)

- Special education; Department of and the Board of Education to develop new policies and procedures, individualized education program (IEP), duty of Department to provide training and guidance documents to local school divisions on development of IEPs. Amending §§ 22.1-214, 22.1-215, 22.1-253.13:4, and 22.1-298.1; adding § 22.1-214.4. (Patron–Dunnavant, SB 1288)
- Special education; Department of Education to provide training and guidance documents to local school divisions on the development of Individualized Education Programs (IEPs) for children with disabilities, etc. Adding § 22.1-214.4. (Patron–Carr, HB 2299)
- Standards of Learning; reading and mathematics assessments for grades three through eight, individual student growth, through-year growth assessments. Amending § 22.1-253.13:3. (Patron-Coyner, HB 2027)
- Standards of Learning; reduces total number and type of required assessments to minimum requirements, report. Amending § 22.1-253.13:3. (Patron–Pillion, SB 1401)
- Standards of Quality; each school board shall provide at least three specialized student support positions per 1,000 students, effective clause. Amending §§ 22.1-253.13:2 and 22.1-274. (Patron–McClellan, SB 1257)
- Standards of Quality; Joint Legislative Audit and Review Commission to study the true cost of education in the Commonwealth and provide an accurate assessment of the costs to implement, meetings shall be completed for the first year by November 30, 2022. (Patron–Lewis, SJR 294)
- Student driver education program; parent/student component exemption. Amending § 22.1-205. (Patron–Keam, HB 2119)
- Student driver safety; driver education program shall include the dangers of distracted driving and speeding, high school student parking passes, valid driver's license required. Amending § 22.1-205; adding § 22.1-205.1. (Patron–Mugler, HB 1918)
- Students; guidelines established by the Department of Education on excused student absences, local school boards may require that student provide advance notice of intended absence, etc., civic engagement. Amending § 22.1-254. (Patron–Rasoul, HB 1940)
- Students; guidelines on excused student absences, each school board shall permit one school day-long excused absence per school year for any middle or high school student who is absent from school to engage in a civic event. Amending § 22.1-254. (Patron–McClellan, SB 1439)
- Students with disabilities; Department of Education to update its special education and related services, etc. (Patron–Mundon King, HB 2316)
- Teachers and other licensed school board employees; cultural competency. Amending §§ 22.1-253.13:5 and 22.1-298.1; adding § 22.1-298.7. (Patron–Jenkins, HB 1904; Locke, SB 1196)
- Through-year growth assessment system; Board of Education shall establish, report, effective clause. (Patron-Dunnavant, SB 1357)
- Traumatic brain injury; definition. (Patron-Wilt, HB 2182)
- Virginia Science, Technology, Engineering, and Mathematics (STEM) Advisory Board; established, report. Adding §§ 22.1-364 through 22.1-368. (Patron–Simonds, HB 2058)
- Voter referendum; issuance of state general obligation bonds for school facility modernization, November 2022 general election, effective clause for provisions. (Patron–Stanley, SB 1109)

ELECTIONSAbsentee voting; accessibility for voters with a visual impairment or print disability.

Amending § 24.2-704; adding § 24.2-103.2. (Patron–Reeves, SB 1331)

Absentee voting; availability on Sundays in office of general registrar or voter satellite office. Amending § 24.2-701.1. (Patron–Bagby, HB 1968)

ELECTIONS (continued)

- Absentee voting; ballots to be sorted and results to be reported by precinct. Amending § 24.2-712. (Patron-Suetterlein, SB 1153)
- Absentee voting; establishment of drop-off locations for return of absentee ballots, ballot defects, cure process. Amending §§ 24.2-706, 24.2-707, 24.2-709, 24.2-709.1, and 24.2-710; adding § 24.2-707.1. (Patron–Deeds, SB 1245)
- Absentee voting; mandatory processing of returned absentee ballots before election day, central absentee voter precinct in the office of the general registrar. Amending §§ 24.2-709.1 and 24.2-712. (Patron–Deeds, SB 1246)
- Absentee voting; procedural and process reforms, availability and accessibility reforms, penalty. Amending §§ 24.2-101, 24.2-416.1, 24.2-603, 24.2-653.1, 24.2-704, 24.2-705, 24.2-706, 24.2-707, 24.2-708 through 24.2-711, and 24.2-712; adding §§ 24.2-103.2, 24.2-667.1, and 24.2-707.1. (Patron-VanValkenburg, HB 1888)
- Absentee voting; third-party absentee ballot assembly and distribution. Amending § 24.2-706. (Patron-Bell, SB 1239)
- Absentee voting; witness signature not required. Amending §§ 24.2-702.1, 24.2-704, 24.2-706, and 24.2-707. (Patron–Favola, SB 1097)
- Assistance for certain voters; curbside voting. Amending §§ 24.2-638, 24.2-646.1, and 24.2-649; adding § 24.2-649.1. (Patron–Price, HB 1921)
- Campaign finance; prohibited personal use, child care exception. Amending § 24.2-948.4; adding § 24.2-948.5. (Patron–Simon, HB 1952)
- Campaign finance reports; special report of pre-legislative session contributions. Amending § 24.2-947.11. (Patron–Saslaw, SB 1444)
- Comprehensive campaign finance reform; joint subcommittee to study. (Patron-Bulova, HJR 526)
- Constitutional amendment; qualifications of voters, fundamental right to vote, persons not entitled to vote (first reference). Amending Section 1 of Article II. (Patron–Locke, SJR 272)
- Discrimination; prohibited in voting and elections administration, required process for enacting certain covered practices, civil causes of action, penalties, repeals provision relating to minority language accessibility, etc. Amending §§ 24.2-105, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1; adding §§ 24.2-104.1, 24.2-125 through 24.2-131, and 24.2-1005.2; repealing § 24.2-124. (Patron–Price, HB 1890; McClellan, SB 1395)
- Elections; preservation of order at the polls, powers of officers of election. Amending § 24.2-606. (Patron-Spruill, SB 1111)
- General registrar; qualifications, residency. Amending § 24.2-110. (Patron-Morrissey, SB 1281)
- Identification privilege cards; authorizes DMV to issue, fee, confidentiality, penalties. Amending §§ 24.2-411.3, 24.2-643, 46.2-203.2, 46.2-216.1, 46.2-323.01, 46.2-323.1, 46.2-346, 46.2-600.1, and 58.1-3; adding § 46.2-345.3. (Patron–Guzman, HB 2138)
- Local elections for governing bodies; elections for school boards, qualification of voters. Amending §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223. (Patron–Convirs-Fowler,
- Military-overseas ballots; Secretary of Administration to oversee and develop a charter and directives for the State Board of Elections to form a working group to study implementation of electronic return of voted ballots. (Patron-DeSteph, SJR 322)
- Nomination of candidates for elected offices; restrictions on nomination method selected by political party. Amending § 24.2-509. (Patron–Helmer, HB 2020)
- Polling places; prohibited activities, unlawful possession of a firearm, penalty. Amending §§ 24.2-604, 24.2-671, and 24.2-802.1. (Patron–Levine, HB 2081)
- Primary election; changes date of election held in June from second Tuesday in June to third Tuesday in June, also changes candidate filing deadlines to reflect change of date. Amending §§ 24.2-311, 24.2-503, 24.2-507, 24.2-510, 24.2-515, and 24.2-515.1. (Patron-Kiggans, SB 1148)
- Voter referendum; issuance of state general obligation bonds for school facility modernization, November 2022 general election, effective clause for provisions. (Patron-Stanley, SB 1109)
- Voter registration; failure of online voter registration system, deadline extension. Amending § 24.2-416. (Patron-VanValkenburg, HB 1810)
- Voter registration; list maintenance, lists of decedents transmitted by State Registrar of Vital Records to Department of Elections on a weekly basis. Amending §§ 24.2-408 and 24.2-427. (Patron-Kiggans, SB 1422)

ELECTIONS (continued)

Voter registration; preregistration for persons 16 years of age or older, effective date. Amending § 24.2-404; adding § 24.2-403.1. (Patron–Lopez, HB 2125)

ELECTRIC COMPANIES

- Electric generating facility closures; public closure, integrated resource plans, provisions apply to facility owned or operated by an electric utility. Amending § 56-599; adding § 45.1-394.1. (Patron–Deeds, SB 1247)
- Electric utilities; advanced renewable energy buyers. Amending § 56-585.5. (Patron-Sullivan, HB 1907)
- Electric utilities; eliminates customer credit reinvestment offsets. Amending § 56-585.1. (Patron-Bourne, HB 2049)
- Electric utilities; nonjurisdictional customers, third party power purchase agreements. Amending Chapters 1187, 1188, 1189, 1193, 1194, and 1239, 2020 Acts. (Patron–Hurst, HB 2034; Edwards, SB 1420)
- Electric utilities; procedures under which the State Corporation Commission reviews the earnings and sets the rates of investor-owned incumbent electric utilities, triennial review. Amending § 56-585.1. (Patron–Jones, HB 2200)
- Electric utilities; procurement of certain equipment from Virginia-based or United States-based manufacturer using materials or product components made in Virginia or the United States, if reasonably available and competitively priced. Amending §§ 56-585.1:11 and 56-585.5. (Patron–DeSteph, SB 1295)
- Electric utilities; triennial review, fair rate of return, customer bill credits. Amending § 56-585.1. (Patron-Tran, HB 2160)
- Electric utilities; triennial review, period costs, rate reductions. Amending § 56-585.1. (Patron-Helmer, HB 1914)
- Electric utility regulation; purchasing from competitive suppliers. Amending § 56-577. (Patron-Bourne, HB 2048)
- Phase I or Phase II electric utilities; provision of broadband capacity, State Corporation Commission shall condition any approval of a petition on the requirement that construction shall commence within 18 months of such approval. Amending § 56-585.1:9. (Patron–Tyler, HB 2304; Boysko, SB 1413)

ELECTRONIC PROCESSES

- Estate planning documents; electronic execution, validity of will, codifies Uniform Electronic Wills Act. Amending §§ 32.1-291.7, 54.1-2982, 54.1-2983, 54.1-2984, 59.1-481, 64.2-100, 64.2-403, 64.2-404, 64.2-407, 64.2-450, and 64.2-701; adding §§ 64.2-459 through 64.2-468. (Patron–Sullivan, HB 1856)
- Geriatric or terminally ill inmates; conditional release, notify by certified or verifiable electronic means to attorney for the Commonwealth. Amending § 53.1-136. (Patron-Norment, SB 1397)
- Property owners' associations, boards of directors, unit owners' associations, etc.; meetings to be held entirely or partially by electronic means, provided guidelines have been adopted. Amending §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953. (Patron–Bulova, HB 1816; Dunnavant, SB 1183)
- Virginia Employment Commission; communications with parties, use of electronic means, report. Adding § 60.2-121.1. (Patron–Tran, HB 2036)
- Virginia Freedom of Information Act; meetings held through electronic communication means during a state of emergency. Amending § 2.2-3708.2. (Patron–McPike, SB 1271)
- Virginia Freedom of Information Act; public body authorized to conduct electronic meetings. Amending § 2.2-3708.2. (Patron–Levine, HB 1931)

ELEMENTARY SCHOOLS

- Public elementary and secondary schools; possession and administration of undesignated stock albuterol inhalers and valved holding chambers, Department of Health, in conjunction with the Department of Education, shall develop and implement policies. Amending §§ 8.01-225, 22.1-274.2, and 54.1-3408. (Patron–McQuinn, HB 2019)
- School nurses; no individual who provides nursing services in a public elementary or secondary school shall use title unless individual is a registered nurse. Amending § 22.1-274. (Patron–Adams, D.M., HB 1736)

EMANCIPATION HIGHWAY

Jefferson Davis Highway; renames any section of U.S. Route 1 to "Emancipation Highway." Repealing Chapter 286, 1922 Acts. (Patron-Cole, J.G., HB 2075)

EMERGENCY LEGISLATION

- Alcoholic beverage control; delays the effective date of the 2020 alcoholic beverage control license and fee reform. Amending §§ 4.1-230, 4.1-233.1, and third, fifth, and eighth enactments of Chapters 1113 and 1114, 2020 Acts. (Patron–VanValkenburg, HB 1845)
- Career fatigue and wellness in certain health care providers; programs to address, civil immunity. Amending §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909. (Patron–Hope, HB 1913; Barker, SB 1205)
- Child Care Subsidy Program; temporarily expanding Program to provide financial assistance for child care to families in need during public health emergency. (Patron–Filler-Corn, HB 2206)
- Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021; created. (Patron-Torian, HB 2178; Howell, SB 1145)
- Concealed handgun permits; demonstration of competence, eligibility to apply for permit due to restrictions of COVID-19. (Patron–Runion, HB 2310)
- Declared states of emergency; funeral service licensees and any other person employed by a funeral service establishment shall be considered essential workers, access to personal protective equipment and immunization. Adding §§ 32.1-42.2 and 44-146.17:1.1. (Patron–Mugler, HB 2116)
- Electronic notarial certificate; clerk's office to record a paper copy of an electronic document, provided that such copy otherwise meets the requirements for recordation and is certified to be a true and correct copy. Amending §§ 17.1-223, 47.1-2, 47.1-16, and 55.1-606. (Patron–Simon, HB 2064)
- General district courts; increases to \$50,000 the maximum civil jurisdictional limit of courts, appeal bond. Amending §§ 8.01-195.4, 16.1-77, and 16.1-107. (Patron–Stanley, SB 1108)
- Industrial hemp; definitions, updates laws to address the new hemp producer license issued by the U.S. Department of Agriculture. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4118, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. (Patron–Marshall, HB 2078)
- Industrial hemp; increases the maximum tetrahydrocannabinol (THC) concentration, expands definition of "hemp product", registration renewal fee not to exceed \$50, etc. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. (Patron–Peake, SB 1115)
- Medical care facilities; facility shall establish protocols, definitions, designated support persons for persons with disabilities. Adding § 32.1-137.08. (Patron–Tran, HB 2162)
- Refunding bonds; alters the principal and interest requirements, maturity date, and allowable discount for bonds previously issued, sunset clause. Amending first enactment of Chapters 265 and 408, 1992 Acts. (Patron–Torian, HB 2179; Howell, SB 1134)
- Retail Sales and Use Tax; exemption for personal protective equipment. Adding § 58.1-609.14. (Patron–Byron, HB 2185; Pillion, SB 1403)
- Search warrants; date and time of issuance, law-enforcement officer to be recognizable and identifiable, exceptions. Amending § 19.2-56. (Patron–Stuart, SB 1475)
- Virginia Redistricting Commission; removing a citizen commissioner of the Commission for neglect of duty or gross misconduct by other commissioners in a public meeting. Amending § 30-392. (Patron–Watts, HB 2324)
- Wrongful death statute of limitations; criminal investigations by law enforcement. Amending § 8.01-244. (Patron–Convirs-Fowler, HB 2073)

EMERGENCY MANAGEMENT PROFESSIONALS WEEK

Emergency Management Professionals Week; designating as third week in March 2021 and each succeeding year thereafter. (Patron–McPike, SJR 286)

EMERGENCY SERVICES AND VEHICLES

- Workers' compensation; adds salaried or volunteer emergency medical services personnel to the list of persons to whom, after five years of service, the occupational disease presumption for death caused by hypertension or heart disease applies, personnel operating in a locality that has legally adopted a resolution declaring that it will provide one or more of the presumptions. Amending § 65.2-402. (Patron–Heretick, HB 1818)
- Workers' compensation; adds salaried or volunteer emergency medical services personnel to the list of persons to whom occupational disease presumption for death caused by hypertension or heart disease applies, provisions shall not apply to any individual who was diagnosed before July 1, 2021. Amending § 65.2-402. (Patron–Marsden, SB 1275)
- Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, emergency medical services personnel, law-enforcement officers, and correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Jones, HB 2207)

EMINENT DOMAIN

- Eminent domain; notice of intent to file certificate. Amending § 25.1-306. (Patron-Cosgrove, SB 1270)
- Transportation purposes; inspection of property to ascertain suitability of the property for highway and other transportation purposes. Amending §§ 25.1-203 and 33.2-1011. (Patron–Bell, SB 1260)

EMISSIONS STANDARDS

State Air Pollution Control Board; implementation of a low-emissions and zero-emissions vehicle program, energy jurisdictional retail sales. Amending § 10.1-1307; adding § 10.1-1307.04. (Patron–Bagby, HB 1965)

ENERGY CONSERVATION AND RESOURCES

- Energy storage systems; definitions, tax exemption, revenue share for systems. Amending §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660. (Patron–Heretick, HB 2006; Petersen, SB 1201)
- Small renewable energy projects; includes in the definition certain energy storage facilities and projects that include storage facility components. Amending § 10.1-1197.5. (Patron–Willett, HB 2148)
- Solar and energy storage projects; siting agreements throughout the Commonwealth. Amending §§ 15.2-2288.8 and 15.2-2316.6 through 15.2-2316.9. (Patron–Jones, HB 2201; Barker, SB 1207)
- Solar energy projects and energy storage systems; revenue share for projects and systems. Amending § 58.1-2636. (Patron–Heretick, HB 2269)
- Uniform Statewide Building Code; Board of Housing and Community Development shall consider amendments to Code to address changes in the IECC relating to energy efficiency and conservation. (Patron–Kory, HB 2227)

ENERGY PLAN

Mines, Minerals and Energy, Department of; renamed the Department of Energy. Amending §§ 2.2-204, 2.2-604.2, 2.2-1157, 2.2-1176.1, 2.2-3705.6, 2.2-4006, 10.1-606.3, 10.1-659, 10.1-1194, 10.1-1329, 10.1-1330, 10.1-1406.2, 11-34.3, 15.2-958.3, 15.2-980, 15.2-2224, 23.1-2626, 23.1-2627, 28.2-1208, 30-275, 33.2-236, 45.1-161.1, 45.1-161.2, 45.1-161.5, 45.1-161.15, 45.1-161.179, 45.1-161.292:2, 45.1-180, 45.1-229, 45.1-230, 45.1-270.4:1, 45.1-361.28, 45.1-361.41, 45.1-383, 45.1-390, 56-265.15:1, 56-576, 56-585.5, 56-594.3, 56-596.2, 58.1-439.2, 58.1-439.12:02, 58.1-3600, 58.1-3706, 58.1-3745, 62.1-44.15:21, 62.1-44.15:66, 62.1-195.1, 62.1-243, 62.1-256, 62.1-259, 63.2-805, 67-200, 67-202.1, 67-602, 67-900, 67-1000, 67-1206, 67-1208, 67-1209, 67-1403, and 67-1506; adding § 45.1-161.4:1. (Patron–Sullivan, HB 1855)

ENTERPRISE ZONES

Enterprise zone job creation grants; for purposes of wage requirements, the minimum wage shall be the higher of the state minimum wage or the federal minimum wage, delayed effective date. Amending § 59.1-547. (Patron–Heretick, HB 1881)

EQUAL CITIZENS MONTH

Equal Citizens Month; designating as November 2021 and each succeeding year thereafter. (Patron-Convirs-Fowler, HJR 606)

EROSION AND SEDIMENT CONTROL

Solar projects; State Water Control Board to administer a Virginia Erosion and Sediment Control Program (VESCP) on behalf of any locality that notifies the Department of Environmental Quality that it has chosen not to administer a VESCP, effective clause. Adding § 62.1-44.15:55.1. (Patron–Marsden, SB 1258)

ESSEX COUNTY

Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron–Cole, J.G., HB 1819)

ETHNIC GROUPS

Criminal justice legislation; racial and ethnic impact statements. Adding § 30-19.1:13. (Patron–Aird, HB 1990)

Enslaved Ancestors College Access Scholarship and Memorial Program; established, report. Adding § 23.1-615.1. (Patron–Reid, HB 1980)

Racism; General Assembly to recognize as a public health crisis. (Patron-Aird, HJR 537)

EVIDENCE

Criminal proceedings; consideration of mental condition and intellectual and developmental disabilities, evidence offered by the defendant. Amending §§ 19.2-120, 19.2-163.03, and 19.2-299; adding § 19.2-271.6. (Patron–McClellan, SB 1315)

EXCISE TAX

Peanuts; extends sunset date of excise tax on all peanuts grown in Virginia. Amending § 3.2-1905. (Patron–Brewer, HB 1751; Lucas, SB 1411)

FAIR HOUSING LAW

Virginia Fair Housing Law; reasonable accommodations, disability-related requests for parking. Amending § 36-96.3:2. (Patron–Carr, HB 1971)

FAIRFAX COUNTY

Fairfax County Park Authority; commending. (Patron-Filler-Corn, HJR 721)

Historic sites; urban county executive form of government (Fairfax County), provisions in its preservation ordinance, etc. Amending § 15.2-2306. (Patron–Surovell, SB 1457)

Underground utility facilities; removes the sunset on a pilot program allowing a locality that has adopted the urban county executive form of government (Fairfax County) to request an electric utility to place underground electric distribution lines, written agreement. Amending § 15.2-816.1. (Patron–Surovell, SB 1385)

FAIRFAX, JUSTIN E., LIEUTENANT GOVERNOR OF VIRGINIA AND PRESIDENT OF THE SENATE

SENATE
Chair assumed
Chair resumed
Tie Votes:
S.B. 1157
S.J.R. 270
S.J.R. 290. Chair did not vote. 489

FARLOW, MARY ELENE WILLIAMS

Farlow, Mary Elene Williams; recording sorrow upon death. (Patron–Marshall, HJR 771)

FARMERS, FARM PRODUCE, AND EQUIPMENT

Farmers market food and beverage products; sales considered essential during state of emergency. (Patron-Wyatt, HB 2302)

FARMERS, FARM PRODUCE, AND EQUIPMENT (continued)

Minimum wage; eliminates the exemption from Virginia's requirements for persons employed as farm laborers or farm employees and any individual employed as a temporary foreign worker. Amending § 40.1-28.9. (Patron–Ward, HB 1786)

FARR, MURRAY JAY

Farr, Murray Jay; commending. (Patron-Hope, HJR 656)

FAVOLA, BARBARA A.

FEASENMYER, WILLIAM S., JR.

Feasenmyer, William S., Jr.; commending. (Patron-Cox, HJR 746)

FELONS AND FELONIES

Constitutional amendment; felon disenfranchisement, automatic restoration of political rights (first reference). Amending Section 1 of Article II. (Patron–Herring, HJR 555)

FERGUSON, EARL MARK

Ferguson, Earl Mark; recording sorrow upon death. (Patron-VanValkenburg, HJR 644)

FINANCIAL INSTITUTIONS AND SERVICES

State Corporation Commission; supervisory merger or transfer of assets of financially unstable credit unions, repeals provision relating to consolidation or merger. Amending § 6.2-1317; repealing § 6.2-1318. (Patron–Bagby, HB 1964)

FINES AND COSTS

Fines and costs; accrual of interest, deferral or installment payment agreements. Amending §§ 19.2-353.5 through 19.2-355. (Patron–Hudson, HB 1895)

FIREARMS

Firearm; carrying within Capitol Square and the surrounding area, into building owned or leased by the Commonwealth, etc., penalty. Adding § 18.2-283.2. (Patron–Levine, HB 2295)

Firearms; access to any individual who resides or cohabitates in a shared dwelling, civil penalty. Adding § 18.2-308.2:02. (Patron–Guzman, HB 2319)

Firearms; criminal history record information check delay increased to five days. Amending § 18.2-308.2:2. (Patron–Lopez, HB 2128)

Firearms; purchase, possession, etc., following conviction for assault and battery of a family or household member, definition, penalties. Amending §§ 18.2-308.09, 18.2-308.2:1, 18.2-308.2:2, 18.2-308.2:3, and 19.2-386.28; adding § 18.2-308.1:8. (Patron–Murphy, HB 1992)

Plastic firearms and unfinished frames, etc.; manufacture, import, sale, transfer, etc., penalty. Amending § 18.2-308.5; adding § 18.2-308.5:2. (Patron–Simon, HB 2276)

Polling places; prohibited activities, unlawful possession of a firearm, penalty. Amending §§ 24.2-604, 24.2-671, and 24.2-802.1. (Patron–Levine, HB 2081)

FIREFIGHTERS AND FIRE MARSHALS

Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, emergency medical services personnel, law-enforcement officers, and correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Jones, HB 2207)

Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, law-enforcement officers or correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Saslaw, SB 1375)

FISHING LAWS AND LICENSES

Trout fishing in stocked waters; equalizes for residents and nonresidents requirements to fish in designated waters. Amending § 29.1-311. (Patron–Pillion, SB 1402)

FLOODS AND FLOOD CONTROL

Local stormwater assistance; flood mitigation and protection measures. Amending § 15.2-2114.01. (Patron–Ebbin, SB 1309)

FLOODS AND FLOOD CONTROL (continued)

Recurrent Flooding Resiliency, Commonwealth Center for; Center shall evaluate development of Flood Resiliency Clearinghouse Program for coordinating flood mitigation solutions, report. (Patron–Hodges, HB 2187)

Virginia Residential Property Disclosure Act; required disclosures for buyer to exercise due diligence, flood risk report, effective date and clause. Amending § 55.1-703; adding § 55.1-708.2. (Patron–Convirs-Fowler, HB 2320)

FOOD AND BEVERAGE PRODUCTS, AND CONTAINERS

Combined transient occupancy and food and beverage tax; for purposes of taxes that are currently authorized for Rappahannock and Madison Counties, the rate limit for such tax shall be the same as if the two taxes were imposed separately. Amending § 58.1-3842. (Patron–Hanger, SB 1438)

Expanded polystyrene food service containers; definition, prohibits dispensing by a food vendor of prepared foods to a customer, civil penalty. Amending §§ 10.1-1414 and 10.1-1422.01; adding § 10.1-1424.3. (Patron–Carr, HB 1902)

Food and drink permit; inspections required to operate food establishment, Commissioner of Agriculture and Consumer Services to issue permit for the operation of any food manufacturer, food storage warehouse, etc. Amending §§ 3.2-5100, 3.2-5101, 3.2-5130, and 15.2-2288.6. (Patron–Gooditis, HB 2275)

FORD, RUDOLPH E., JR.

Ford, Rudolph E., Jr.; recording sorrow upon death. (Patron–McQuinn, HJR 731)

FORENSIC SCIENCE

Post-conviction relief; previously admitted forensic scientific evidence, effective date, report, and effective clause. Adding § 19.2-327.15. (Patron–Stanley, SB 1105)

FOSTER CARE

Foster care; termination of parental rights, relatives and fictive kin. Amending §§ 16.1-281, 16.1-283, 63.2-906, and 63.2-910.2. (Patron–Gooditis, HB 1962)

State-Funded Kinship Guardianship Assistance program; created, clarifies definition of "foster care services." Amending §§ 16.1-228, 16.1-282.1, 63.2-100, 63.2-905, 63.2-906, and 63.2-1305; adding § 63.2-1306. (Patron–Mason, SB 1328)

FOSTER, TIM

Foster, Tim; commending. (Patron-VanValkenburg, HJR 700)

FRANCHISE TAX AND FRANCHISES

Research and development expenses; tax credit available against the bank franchise tax for taxable years beginning on and after January 1, 2021. Amending §§ 58.1-439.12:08 and 58.1-439.12:11. (Patron–Locke, SB 1112)

FRANCIS, THOMAS HENRY

Francis, Thomas Henry; recording sorrow upon death. (Patron-Bagby, HJR 716; McQuinn, HJR 737)

FREDERICKSBURG, CITY OF

Fredericksburg, City of; commending. (Patron-Cole, M.L., HJR 666)

FREDERICKSBURG DOG MART

Fredericksburg Dog Mart; commending. (Patron-Cole, M.L., HJR 667)

FREEDOM OF INFORMATION

Freedom of Information Act; adds Chief Workforce Development Advisor and Director of Diversity, Equity, and Inclusion to definition of "Office of the Governor" for purposes of excluding the working papers and correspondence of the Office of the Governor from mandatory disclosure. Amending § 2.2-3705.7. (Patron–McQuinn, HB 2313)

Virginia Freedom of Information Act; individual votes of members of the Virginia Parole Board shall be public records and subject to provisions of the Act. Amending § 2.2-3703. (Patron–Suetterlein, SB 1103)

FREEDOM OF INFORMATION (continued)

- Virginia Freedom of Information Act; law-enforcement criminal incident information, criminal investigative files. Amending §§ 2.2-3706, 2.2-3711, 19.2-174.1, and 19.2-368.3; adding § 2.2-3706.1. (Patron–Hurst, HB 2004)
- Virginia Freedom of Information Act; meetings held through electronic communication means during a state of emergency. Amending § 2.2-3708.2. (Patron–McPike, SB 1271)
- Virginia Freedom of Information Act; proprietary records and trade secrets, carbon sequestration agreements. Amending § 2.2-3705.6. (Patron-Vogel, SB 1343)
- Virginia Freedom of Information Act; public body authorized to conduct electronic meetings. Amending § 2.2-3708.2. (Patron–Levine, HB 1931)
- Virginia Freedom of Information Act; record exclusion for personal contact information provided to a public body. Amending § 2.2-3705.1. (Patron–Gooditis, HB 2025)

FUNERAL HOME DIRECTORS AND SERVICES

Declared states of emergency; funeral service licensees and any other person employed by a funeral service establishment shall be considered essential workers, access to personal protective equipment and immunization. Adding §§ 32.1-42.2 and 44-146.17:1.1. (Patron–Mugler, HB 2116)

GAMBLING, LOTTERIES, ETC.

- Casino gaming; requirements for issuance of operator's license, human trafficking training. Amending § 58.1-4110. (Patron–Simonds, HB 1944)
- Casino gaming; technical amendments to the gaming law related to its interaction with sports betting law, the capital investment required of an applicant for a license, etc. Amending §§ 2.2-3711, 58.1-4100, 58.1-4109, 58.1-4110, 58.1-4114, 58.1-4122, 58.1-4124, and 58.1-4125. (Patron–Krizek, HB 1812)
- Charitable gaming; regulations of operators of Texas Hold'em poker games, conduct of games, effective clause. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.18, 18.2-340.19, 18.2-340.22, 18.2-340.28:2, and 18.2-340.34:2. (Patron–Bell, SB 1278)
- Illegal gambling; skill games, definitions, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron–Reeves, SB 1465)
- Illegal gambling; skill games, enforcement by localities and Attorney General, civil penalty. Amending § 18.2-325; adding § 18.2-331.1. (Patron–Scott, HB 2168)
- Sports betting; clarifies the procedures by which the Virginia Lottery determines whether an event is considered youth sports, on which betting is prohibited, an international athletic event organized by the International Olympic Committee shall not be considered to be youth sports, issuance of permits to operate sports betting platforms. Amending §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100. (Patron–Sickles, HB 1847)
- Sports betting; clarifies the procedures by which the Virginia Lottery determines whether an event is considered youth sports, on which betting is prohibited, an international athletic event organized by the International Olympic Committee shall not be considered to be youth sports, regardless of the age of the participants. Amending §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100. (Patron–McPike, SB 1254)

GARNER, CHARLES LINCOLN

Garner, Charles Lincoln; recording sorrow upon death. (Patron-Carr, HJR 652)

GAS AND OIL MINING

Local gas severance tax; extends sunset date. Amending § 58.1-3713. (Patron-Morefield, HB 2293)

GAYLORD, KATIE

Gaylord, Katie; commending. (Patron-Mullin, HJR 753)

GENAKOS, GEORGE S.

Genakos, George S.; recording sorrow upon death. (Patron–Mullin, HJR 757)

GENERAL ASSEMBLY

Access to water; General Assembly to recognize that access to clean, potable, and affordable water is a necessary human right. (Patron-Aird, HJR 538)

GENERAL ASSEMBLY (continued)

- Auditor of Public Accounts; audits of certain political subdivisions. Amending § 30-140. (Patron–McNamara, HB 1789)
- Behavioral Health Commission; created, report. Adding §§ 30-401 through 30-408. (Patron–Deeds, SB 1273)
- Capitol Square Preservation Council; powers and duties, review and approval of plans for changes to artifacts contained within the Capitol Building. Amending § 30-194. (Patron–Norment, SB 1172)
- Constitutional amendment; length of regular sessions convened in odd-numbered years (first reference). Amending Section 6 of Article IV. (Patron–Saslaw, SJR 310)
- Criminal justice legislation; racial and ethnic impact statements. Adding § 30-19.1:13. (Patron–Aird, HB 1990)
- General Assembly; confirming appointment to Joint Legislative Audit and Review Commission. (Patron-Plum, HJR 604)
- Health Care, Joint Commission on; repeals sunset provision. Repealing § 30-170. (Patron–Barker, SB 1408)
- Health Insurance Reform Commission; mandated health insurance benefit or provider legislation to be referred to Commission. Amending § 30-343. (Patron–Saslaw, SB 1473)
- Legislative Staff Development Fund; created, effective date. Amending § 2.2-424; adding § 30-19.21. (Patron–Reid, HB 1978)
- Marriage; persons who may celebrate rites, authorizes current members of the General Assembly, Governor, Lieutenant Governor, and Attorney General. Amending § 20-25. (Patron–Cosgrove, SB 1142)
- Racism; General Assembly to recognize as a public health crisis. (Patron-Aird, HJR 537)
- Virginia Redistricting Commission; removing a citizen commissioner of the Commission for neglect of duty or gross misconduct by other commissioners in a public meeting. Amending § 30-392. (Patron–Watts, HB 2324)
- Virginia Redistricting Commission; transparency in redistricting process. Amending §§ 30-396 and 30-399. (Patron-Levine, HB 2082)
- Wolf Creek Cherokee Tribe of Virginia; General Assembly to extend state recognition. (Patron–McQuinn, HJR 572)

GENERAL SERVICES, DEPARTMENT OF

Harry F. Byrd, Sr., statue; Department of General Services to remove statue of former Virginia Governor and U.S. Senator from Capitol Square. (Patron–Jones, HB 2208)

GEORGE MASON UNIVERSITY

George Mason University; management agreement with the Commonwealth. (Patron-Bulova, HB 1986; Barker, SB 1204)

GLASGOW, TOWN OF

Glasgow, Town of; new charter (previous charter repealed). (Patron-Campbell, R.R., HB 1783)

GOSLING, ARTHUR WARRINGTON

Gosling, Arthur Warrington; recording sorrow upon death. (Patron-Hope, HJR 654)

GOVERNOR

- Freedom of Information Act; adds Chief Workforce Development Advisor and Director of Diversity, Equity, and Inclusion to definition of "Office of the Governor" for purposes of excluding the working papers and correspondence of the Office of the Governor from mandatory disclosure. Amending § 2.2-3705.7. (Patron–McQuinn, HB 2313)
- Labor, Secretary of; position created in Governor's Cabinet, removes position of Chief Workforce Development Advisor. Amending §§ 2.2-200, 2.2-204, 2.2-205, 2.2-205.1, 2.2-435.6, 2.2-435.8, 2.2-435.9, 2.2-435.10, 2.2-2471, 2.2-2471.1, 2.2-2472, 2.2-2472.2, 2.2-2472.3, and 30-377; adding §§ 2.2-214.2 and 2.2-214.3; repealing § 2.2-435.7. (Patron–Simonds, HB 2321)
- Marriage; persons who may celebrate rites, authorizes current members of the General Assembly, Governor, Lieutenant Governor, and Attorney General. Amending § 20-25. (Patron–Cosgrove, SB 1142)

GOVERNOR (continued)

Southwestern Virginia Mental Health Institute; Governor to lease a portion of property to Smyth County for a term of three years, responsibility of County, corrects tax map references. Amending Chapter 678, 2019 Acts. (Patron–O'Quinn, HB 2098; Pillion, SB 1429)

Tax Commissioner; waiver of accrual of interest in the event that Governor declares state of emergency. Amending § 58.1-112. (Patron–Murphy, HB 1999)

GRANDY, ETHEL L.

Grandy, Ethel L.; commending. (Patron-Hayes, HJR 684)

GREEK WAR OF INDEPENDENCE

Greece, Nation of; commending. (Patron-Dunnavant SJR 372)

GROUNDWATER

Virginia Water Protection Permit; withdrawal of surface water or ground water, plans for water auditing and leak detection and repair. Amending §§ 62.1-44.15:22 and 62.1-262. (Patron–Mason, SB 1291)

GUIDO, TAMMY WILLIAMS

Guido, Tammy Williams; commending. (Patron-Mullin, HJR 756)

GUNS

School board property, certain; establishment of gun-free zone permitted. Adding § 22.1-131.1. (Patron–Subramanyam, HB 1909)

HAIRSTON, JOE LOUIS

Hairston, Joe Louis; recording sorrow upon death. (Patron-Marshall, HJR 769)

HAMPTON ROADS AREA

Hampton Roads area refuse collection authority; changes various requirements of the Southeastern Public Service Authority when budgeting or incurring debt, etc. Amending § 15.2-5102.1. (Patron–Cosgrove, SB 1141)

Hampton Roads Sanitation District; changes to the enabling act related to the addition of Northampton and Accomack Counties. Amending Chapter 66, 1960 Acts. (Patron–Bloxom, HB 2257)

Transportation District Commission of Hampton Roads; change in membership. Amending § 33.2-1907. (Patron–Spruill, SB 1126)

HANDGUNS

Concealed handgun permits; demonstration of competence, eligibility to apply for permit due to restrictions of COVID-19. (Patron–Runion, HB 2310)

HANGER, EMMETT W., JR.

HARASSMENT

Virginia Human Rights Act; nondiscrimination in employment, definitions, sexual and workplace harassment. Amending §§ 2.2-522, 2.2-3905, 2.2-3907, and 2.2-3908. (Patron–Watts, HB 2155)

HARDY PETROLEUM

Hardy Petroleum; commemorating its 100th anniversary. (Patron-Marshall, HJR 779)

HARRIS, GREGORY GARFIELD

Harris, Gregory Garfield; commending. (Patron-Webert, HJR 674)

HASHMI, GHAZALA F.

HEALTH

- Birth certificates; an amendment of a certificate shall be evaluated by the State Registrar through an administrative process. Amending § 32.1-269. (Patron–Locke, SB 1121)
- Child Care Subsidy Program; temporarily expanding Program to provide financial assistance for child care to families in need during public health emergency. (Patron–Filler-Corn, HB 2206)
- COVID-19; Department of Medical Assistance Services shall deem testing, treatment, and vaccination to be emergency services. (Patron–Lopez, HB 2124)
- Declared states of emergency; funeral service licensees and any other person employed by a funeral service establishment shall be considered essential workers, access to personal protective equipment and immunization. Adding §§ 32.1-42.2 and 44-146.17:1.1. (Patron–Mugler, HB 2116)
- Disposition of the remains of a decedent; persons to make arrangements for funeral and disposition of remains, person may designate in a signed and notarized writing, repeals provisions relating to when next of kin disagree and absence of next of kin. Amending §§ 32.1-309.1, 54.1-2800, 54.1-2807, 54.1-2825, and 57-27.3; adding §§ 54.1-2825.1, 57-27.4, 57-27.5, and 57-27.6; repealing §§ 54.1-2807.01 and 54.1-2807.02. (Patron–Sickles, HB 2005)
- Eligible Health Care Provider Reserve Directory; established, information concerning fourth-year medical students, etc. Adding §§ 23.1-110, 32.1-23.3, and 54.1-2400.03. (Patron-Hanger, SB 1436)
- Emergency and quarantine orders, certain; additional procedural requirements. Amending § 32.1-48.010; adding § 44-146.17:1.1. (Patron–Petersen, SB 1237)
- Employers; reporting outbreaks of COVID-19, effective clause. (Patron-Lewis, SB 1362)
- Fetal and Infant Mortality Review Team; Office of the Chief Medical Examiner of Department of Health shall convene a work group to develop a plan for the establishment of Team, report. (Patron–Ayala, HB 1950)
- Funding local health departments; cooperative local health budget, report. Adding § 32.1-34.3. (Patron-Bagby, HB 1963)
- Health, Department of; certain communication prohibited regarding matters with a minor, exception. Adding § 32.1-4.1. (Patron–Peake, SB 1235)
- Home care organizations; Board of Health to include in regulations governing organizations a provision for supervision of home care attendants providing personal care services by a licensed nurse through use of interactive audio or video technology. Amending § 32.1-162.12. (Patron–Head, HB 1831)
- Hormonal contraceptives; payment of medical assistance for 12-month supply. Amending § 32.1-325. (Patron–Boysko, SB 1227)
- Hospitals; emergency treatment for substance use-related emergencies, Department of Health, et al., to develop recommendations for best practices for treatment and discharging of patients in emergency departments, etc. Amending § 32.1-127. (Patron–Delaney, HB 2300)
- Hospitals, nursing homes, and certified nursing facilities; regulations, policies to ensure the permissible access to and use of an intelligent personal assistant provided by a patient while receiving inpatient services. Amending § 32.1-127. (Patron–Adams, L.R., HB 2154)
- Hospitals, nursing homes, etc.; visits by clergy, priest, etc., during a declared public health emergency. Amending §§ 32.1-127, 32.1-162.5, and 63.2-1732. (Patron–Kiggans, SB 1356)
- Line of Duty Act; Virginia licensed health practitioners required to conduct medical reviews, persons issued a comparable license, as determined by Virginia Retirement System, by the District of Columbia or a state that is contiguous to Virginia. Amending §§ 9.1-404 and 9.1-405. (Patron–DeSteph, SB 1367)
- Loudoun County; County to enter into a contract with the Department of Health for the local administration of local health services. (Patron-Favola, SB 1221)
- Maternal Health Data and Quality Measures, Task Force on; established, report. (Patron-Herring, HB 2111)
- Medical care facilities; facility shall establish protocols, definitions, designated support persons for persons with disabilities. Adding § 32.1-137.08. (Patron–Tran, HB 2162)
- Nurse Loan Repayment Program; expands eligibility for Program to include certified nurse aids. Amending § 32.1-122.6:04. (Patron-Kiggans, SB 1147)

HEALTH (continued)

- Onsite Sewage Indemnification Fund; authorizes the State Board of Health to use Fund to provide grants and loans to property owners with income at or below 200 percent of the federal poverty guidelines to repair failing onsite sewage systems or install onsite sewage systems on properties that lack adequate sewage disposal. Amending §§ 32.1-164 and 32.1-164.1:01; adding §§ 62.1-223.1, 62.1-223.2, and 62.1-223.3. (Patron–Hashmi, SB 1396)
- Personal care aides; Department of Medical Assistance Services shall establish an orientation program for all aides who provide self-directed services through the Medicaid program. Adding § 32.1-331.04. (Patron–Locke, SB 1102)
- Prescription drugs; price transparency, definitions. Amending § 2.2-3705.6; adding §§ 32.1-23.3, 38.2-3407.15:6, 38.2-3407.22, 54.1-3436.1, and 54.1-3442.02. (Patron–Sickles, HB 2007)
- Produce Rx Program; Department of Social Services, et al., to develop a plan for a three-year pilot Program, report. (Patron–McQuinn, HB 2065)
- Public health emergency; emergency medical services agencies, real-time access to information regarding confirmed and suspected cases of communicable disease of public health threat. Adding § 32.1-48.001. (Patron–Aird, HB 1989)
- Racism; General Assembly to recognize as a public health crisis. (Patron-Aird, HJR 537)
- Rare Disease Council and Rare Disease Council Fund; created, report. Adding §§ 32.1-73.14 through 32.1-73.17. (Patron–Murphy, HB 1995)
- School-based health services; Board of Medical Assistance Services to amend the state plan for medical assistance services to provide for payment, no health care provider who provides services through telemedicine shall be required to use proprietary technology or applications in order to be reimbursed. Amending §§ 32.1-325 and 32.1-326.3. (Patron–Dunnavant, SB 1307)
- Sexually transmitted infections; infected sexual battery, penalty, repeals the crime of donating or selling blood, body fluids, etc., by persons infected with human immunodeficiency virus (HIV), etc. Amending §§ 18.2-52.1, 18.2-67.4:1, 18.2-346.1, 32.1-291.16, 54.1-2982, 54.1-2983, and 57-48; repealing §§ 18.2-62 and 32.1-289.2. (Patron–Locke, SB 1138)
- Telemedicine service; coverage of telehealth services by an insurer, etc., services delivered through real-time audio-only telephone, report. Amending §§ 32.1-325, 38.2-3418.16, and 54.1-3303. (Patron–Adams, D.M., HB 1987)
- Telemedicine services; payment for medical assistance for remote patient monitoring services, effective clause. Amending §§ 32.1-325 and 38.2-3418.16. (Patron–Barker, SB 1338)
- Traumatic brain injury; definition. (Patron-Wilt, HB 2182)
- Universal health care; Joint Commission on Health Care shall enter into a contract with a qualified entity to study options for financing. (Patron–Samirah, HB 2271)
- Virginia Health Workforce Development Authority; mission of Authority, membership. Amending §§ 32.1-122.7 and 32.1-122.7:1. (Patron–Willett, HB 1976)
- Virginia Immunization Information System; any health care provider in the Commonwealth that administers immunizations to participate in System. Amending § 32.1-46.01. (Patron-Willett, HB 2061)

HEALTH INSURANCE

- Commonwealth Health Reinsurance Program; established, report. Amending §§ 38.2-4214 and 38.2-4319; adding §§ 38.2-6600 through 38.2-6607. (Patron–Sickles, HB 2332)
- Health insurance; association health plan for real estate salespersons, policy for group accident and sickness insurance. Amending § 38.2-3521.1. (Patron–Barker, SB 1341)
- Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization for drug by carrier. Amending § 38.2-3407.15:2. (Patron–McPike, SB 1269)
- Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization issued by carrier for drug, certain provisions shall not apply to the state employee health insurance plan, etc. Amending § 38.2-3407.15:2. (Patron–Heretick, HB 2008)
- Health insurance; carrier business practices, every carrier shall include in provider contracts a provision that prohibits a provider from discriminating against any enrollee. Amending § 38.2-3407.15. (Patron–Surovell, SB 1289)
- Health insurance; cost-sharing payments for covered tier one or tier two prescription asthma inhalers. Adding § 38.2-3407.15:6. (Patron–Askew, HB 1822)
- Health insurance; essential benefits, removes the prohibition on the provisions of coverage for abortions. Amending § 38.2-3451. (Patron–Hudson, HB 1896; McClellan, SB 1276)

HEALTH INSURANCE (continued)

- Health insurance; requiring health insurers and other carriers to establish reasonable protocols and procedures for reimbursing a health professional for services provided while such professional's credentialing application is pending. Amending §§ 38.2-4319 and 38.2-4509. (Patron-Head, HB 1829)
- Health Insurance Reform Commission; mandated health insurance benefit or provider legislation to be referred to Commission. Amending § 30-343. (Patron–Saslaw, SB 1473)
- Health maintenance organizations; updates provisions of the Code of Virginia related to insolvency procedures for health maintenance organizations (HMOs). Amending § 38.2-4310. (Patron–Kilgore, HB 1807)

HEMP PRODUCTS

- Industrial hemp; definitions, updates laws to address the new hemp producer license issued by the U.S. Department of Agriculture. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4118, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. (Patron–Marshall, HB 2078)
- Industrial hemp; increases the maximum tetrahydrocannabinol (THC) concentration, expands definition of "hemp product", registration renewal fee not to exceed \$50, etc. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. (Patron-Peake, SB 1115)

HENRICO COUNTY

Henrico County Registrar's Office; commending. (Patron-VanValkenburg, HJR 661)

HIGH SCHOOLS

- Economic education and financial literacy required in middle and high school grades; employment arrangements. Amending § 22.1-200.03. (Patron–Cole, J.G., HB 1905)
- Health Standards of Learning; advanced directive education for high school students. (Patron-Kiggans, SB 1190)
- High school student parking passes; valid driver's license or driver privilege card required. Adding § 22.1-205.1. (Patron-Norment, SB 1169)
- Student driver safety; driver education program shall include the dangers of distracted driving and speeding, high school student parking passes, valid driver's license required. Amending § 22.1-205; adding § 22.1-205.1. (Patron–Mugler, HB 1918)
- Students; guidelines on excused student absences, each school board shall permit one school day-long excused absence per school year for any middle or high school student who is absent from school to engage in a civic event. Amending § 22.1-254. (Patron-McClellan, SB 1439)

HIGHER EDUCATION

- Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021; created. (Patron-Torian, HB 2178; Howell, SB 1145)
- Facial recognition technology; definition, authorization of use by local law-enforcement agencies and public institutions of higher education. Adding §§ 15.2-1723.2 and 23.1-815.1. (Patron-Aird, HB 2031)
- Higher educational institutions, public; admissions applications that contain questions about criminal history. Adding § 23.1-407.1. (Patron–Aird, HB 1930)
- Higher educational institutions, public; governing boards, meetings, input, and disclosures. Amending §§ 23.1-409, 23.1-802, and 23.1-1303. (Patron–Keam, HB 2120)

HIGHWAYS AND OTHER SURFACE TRANSPORTATION SYSTEMS

- Access roads to economic development sites; criteria for use of funds. Amending § 33.2-1509. (Patron-McPike, SB 1253)
- Bob White Covered Bridge; Department of Transportation to work with the governing body of Patrick County and community groups interested in constructing a replica of Bridge. (Patron-Poindexter, HB 2024)
- Central Virginia Transportation Authority; adds the Executive Director of the Virginia Port Authority, or his designee, as a nonvoting ex officio member. Amending § 33.2-3703. (Patron-McQuinn, HB 1926)

HIGHWAYS AND OTHER SURFACE TRANSPORTATION SYSTEMS (continued)

- Highway construction; increases the value of highway maintenance and construction projects eligible to be performed by state or local employees. Amending § 33.2-234. (Patron–Krizek, HB 1813)
- Jefferson Davis Highway; renames any section of U.S. Route 1 to "Emancipation Highway." Repealing Chapter 286, 1922 Acts. (Patron-Cole, J.G., HB 2075)
- Litter; increases fine for dumping or disposing of trash or other unsightly matter on public or private property. Amending § 33.2-802. (Patron–Edmunds, HB 1801)
- New River Valley Passenger Rail Station Authority; creation of authority in Planning District 4 (New River Valley RC). Adding §§ 33.2-3800 through 33.2-3816. (Patron–Hurst, HB 1893; Edwards, SB 1212)
- Parked cars; authorizes vehicles to be stopped or parked in an unoccupied Department of Transportation right-of-way adjacent to a highway in case of an emergency, etc. Amending § 46.2-889. (Patron–Miyares, HB 2146)
- Transportation District Commission of Hampton Roads; change in membership. Amending § 33.2-1907. (Patron–Spruill, SB 1126)
- Transportation funding; statewide prioritization process, resiliency. Amending §§ 33.2-214.2 and 33.2-353. (Patron–Convirs-Fowler, HB 2071; Lewis, SB 1350)
- Transportation purposes; inspection of property to ascertain suitability of the property for highway and other transportation purposes. Amending §§ 25.1-203 and 33.2-1011. (Patron–Bell, SB 1260)
- Virginia Highway Corporation Act; alteration of certificate of authority, powers and duties of State Corporation Commission, agreements between toll operator and Department. Amending §§ 56-539 and 56-542. (Patron–Subramanyam, HB 1832; Bell, SB 1259)
- Virginia Highway Corporation Act of 1988; repeals Act, roadways to operate under the Public-Private Transportation Act of 1995, implementation of distance-based tolling on certain roadways, etc. Amending § 33.2-613; repealing §§ 33.2-1823 and 56-535 through 56-552. (Patron–Reid, HB 2104)
- Wildlife corridors; various agencies to consider and incorporate, where applicable, government planning, Department of Wildlife Resources shall publish plans and any updates on its website. Amending §§ 10.1-207, 10.1-1105, 29.1-579, and 33.2-353. (Patron–Marsden, SB 1274)

HILL, BOBBY

Hill, Bobby; commending. (Patron-Convirs-Fowler, HJR 682)

HILL, JAMILE J.

Hill, Jamile J.; recording sorrow upon death. (Patron-Hayes, HJR 689)

HISTORIC AREAS, LANDMARKS, AND MONUMENTS

- Harry F. Byrd, Sr., statue; Department of General Services to remove statue of former Virginia Governor and U.S. Senator from Capitol Square. (Patron–Jones, HB 2208)
- Historic resources; land acquisition activities of the Department of Historic Resources and the Board of Historic Resources, etc. Amending §§ 10.1-2202 and 10.1-2204. (Patron–Aird, HB 1928)
- Historic sites; urban county executive form of government (Fairfax County), provisions in its preservation ordinance, etc. Amending § 15.2-2306. (Patron–Surovell, SB 1457)
- Johns, Barbara Rose; Joint Committee of Congress on the Library requested to fill the Commonwealth's vacant spot in the National Statuary Hall Collection at the United States Capitol with a statue to commemorate. (Patron–Ward, HJR 525; Lucas, SJR 288)
- Objects of antiquity; unlawful to remove from battlefield, penalty. Amending §§ 10.1-2300 and 10.1-2306. (Patron-Runion, HB 2311)

HODGE, PERRY ANTHONY

Hodge, Perry Anthony; recording sorrow upon death. (Patron-Hurst, HJR 679)

HOGG, OAKLEY W., III

Hogg, Oakley W., III; recording sorrow upon death. (Patron-Wyatt, HJR 711)

HOLIDAYS, SPECIAL DAYS, ETC.

Brain Aneurysm Awareness Month; designating as September 2021 and each succeeding year thereafter. (Patron-Gooditis, HJR 596)

HOLIDAYS, SPECIAL DAYS, ETC. (continued)

Emergency Management Professionals Week; designating as third week in March 2021 and each succeeding year thereafter. (Patron–McPike, SJR 286)

Equal Citizens Month; designating as November 2021 and each succeeding year thereafter. (Patron-Convirs-Fowler, HJR 606)

International Overdose Awareness Day; designating as August 31, 2021, and each succeeding year thereafter. (Patron–Miyares, HJR 562)

Liberty Amendments Month; designating as June 19 through the third Monday in July 2021 and each succeeding year thereafter. (Patron–Bagby, HJR 583; Locke, SJR 323)

Victims of COVID-19 Remembrance Day; designating as March 14, 2021, and each succeeding year thereafter. (Patron–McQuinn, HJR 605)

Women's Suffrage Month; designating as August 2020 and each succeeding year thereafter. (Patron-Vogel, SJR 292)

HOLMES, BOBBY EUGENE

Holmes, Bobby Eugene; recording sorrow upon death. (Patron-McQuinn, HJR 726)

HOMESTEAD AND OTHER EXEMPTIONS

Garnishment of wages; protected portion of disposable earnings. Amending § 34-29. (Patron-Krizek, HB 1814)

HOMICIDE

Homicides and assaults and bodily woundings; certain matters not to constitute defenses. Adding §§ 18.2-37.1 and 18.2-57.5. (Patron–Roem, HB 2132)

HOOKER, KATHERINE BRIDGFORTH

Hooker, Katherine Bridgforth; recording sorrow upon death. (Patron-Carr, HJR 651)

HOSPITALS AND HOSPITALIZATION

Hospitals; emergency treatment for substance use-related emergencies, Department of Health, et al., to develop recommendations for best practices for treatment and discharging of patients in emergency departments, etc. Amending § 32.1-127. (Patron–Delaney, HB 2300)

Hospitals, nursing homes, and certified nursing facilities; regulations, policies to ensure the permissible access to and use of an intelligent personal assistant provided by a patient while receiving inpatient services. Amending § 32.1-127. (Patron–Adams, L.R., HB 2154)

Hospitals, nursing homes, etc.; visits by clergy, priest, etc., during a declared public health emergency. Amending §§ 32.1-127, 32.1-162.5, and 63.2-1732. (Patron–Kiggans, SB 1356)

HOTELS, RESTAURANTS, SUMMER CAMPS, AND CAMPGROUNDS

Retail sales and transient occupancy taxes; definitions, taxes on transient room rentals shall be computed on the basis of the total charges or the total price paid for the use or possession of the room, etc. Amending §§ 58.1-602, 58.1-603, 58.1-3819, 58.1-3819.1, 58.1-3823, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, 58.1-3826, 58.1-3842, and 58.1-3843; adding §§ 2.2-2320.2, 58.1-612.2, and 58.1-3818.8. (Patron–Norment, SB 1398)

HOUSING

Active military or a military spouse; definition of "military status," discrimination in public accommodations, employment, and housing. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310. (Patron–Tran, HB 2161)

Active military or a military spouse; prohibits discrimination in public accommodations, employment, and housing, landlords may not charge a higher security deposit. Amending §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1226, and 55.1-1310. (Patron–Bell, SB 1410)

Affordable and market-rate housing; Department of Housing and Community Development to convene an advisory group to evaluate construction of internal, etc., dwelling units. (Patron–Samirah, HB 2053)

HOUSING (continued)

- Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, cases of a deed of trust conveying owner-occupied residential real estate, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–Torian, HB 2175)
- Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–McClellan, SB 1327)
- Uniform Statewide Building Code; Board of Housing and Community Development shall consider amendments to Code to address changes in the IECC relating to energy efficiency and conservation. (Patron–Kory, HB 2227)
- Virginia Fair Housing Law; reasonable accommodations, disability-related requests for parking. Amending § 36-96.3:2. (Patron–Carr, HB 1971)
- Virginia Fair Housing Law; unlawful discriminatory housing practices. Amending §§ 36-96.3 and 36-96.17. (Patron–Bourne, HB 2046)
- Virginia Good Neighbor Next Door program; Virginia Housing Development Authority shall report recommendations for creating Program. (Patron–Convirs-Fowler, HB 2072)
- Virginia housing opportunity; tax credit established starting in taxable year 2021, which is equal to the amount of the federal low-income housing tax credit allocated. Adding §§ 58.1-439.29 and 58.1-439.30. (Patron–Locke, SB 1197)

HUMAN IMMUNODEFICIENCY VIRUS (HIV)

Sexually transmitted infections; infected sexual battery, penalty, repeals the crime of donating or selling blood, body fluids, etc., by persons infected with human immunodeficiency virus (HIV), etc. Amending §§ 18.2-52.1, 18.2-67.4:1, 18.2-346.1, 32.1-291.16, 54.1-2982, 54.1-2983, and 57-48; repealing §§ 18.2-62 and 32.1-289.2. (Patron–Locke, SB 1138)

HUMAN RIGHTS

- Access to water; General Assembly to recognize that access to clean, potable, and affordable water is a necessary human right. (Patron-Aird, HJR 538)
- Virginia Human Rights Act; application of laws applicable to employee safety and payment of wages. Amending §§ 2.2-3905, 40.1-2, 40.1-29, 40.1-49.3, and 40.1-49.8. (Patron–McClellan, SB 1310)
- Virginia Human Rights Act; expands definition of employer to include person employing one or more domestic workers. Amending § 2.2-3905. (Patron–Price, HB 1864)
- Virginia Human Rights Act; nondiscrimination in employment, definitions, sexual and workplace harassment. Amending §§ 2.2-522, 2.2-3905, 2.2-3907, and 2.2-3908. (Patron–Watts, HB 2155)
- Virginia Human Rights Acts; adds discrimination on the basis of disability as an unlawful employment practice, reasonable accommodations for persons with disabilities. Amending §§ 2.2-3902, 2.2-3905, and 51.5-41; adding § 2.2-3905.1. (Patron–Sickles, HB 1848)

HUMAN TRAFFICKING

- Casino gaming; requirements for issuance of operator's license, human trafficking training. Amending § 58.1-4110. (Patron–Simonds, HB 1944)
- Commercial driver's licenses; disqualifies for life from holding a license persons convicted of a felony involving an act or practice of severe forms of trafficking, etc. Amending §§ 46.2-341.18, 46.2-382, and 46.2-1702. (Patron–Delaney, HB 1868)
- Commercial sex trafficking; issuance of writ of vacatur for victims. Adding §§ 19.2-327.15 through 19.2-327.20. (Patron–Delaney, HB 2133)

IGNITION INTERLOCK DEVICES

Restricted permits to operate a motor vehicle; ignition interlock systems. Adding § 18.2-271.5. (Patron–Stuart, SB 1336)

IMMUNIZATIONS

- Declared states of emergency; funeral service licensees and any other person employed by a funeral service establishment shall be considered essential workers, access to personal protective equipment and immunization. Adding §§ 32.1-42.2 and 44-146.17:1.1. (Patron–Mugler, HB 2116)
- Virginia Immunization Information System; any health care provider in the Commonwealth that administers immunizations to participate in System. Amending § 32.1-46.01. (Patron–Willett, HB 2061)

INCOME TAX

- Agricultural best management practices; creates an enhanced individual and corporate income tax credit beginning in taxable year 2021 but before January 1, 2025, for the implementation of certain practices by the taxpayer that are required as part of a certified resource management plan. Amending §§ 58.1-339.3 and 58.1-439.5. (Patron–Hanger, SB 1162)
- Agricultural best management practices; creates an enhanced individual and corporate income tax credit beginning in taxable year 2021 for the implementation of certain practices by the taxpayer that are required as part of a certified resource management plan, definition of "liability after such cost-share." Amending §§ 58.1-339.3 and 58.1-439.5. (Patron–Wilt, HB 1763)
- Agricultural equipment; establishes a refundable individual and corporate income tax credit. Amending §§ 58.1-334, 58.1-337, 58.1-432, and 58.1-436. (Patron–Hanger, SB 1163)
- Corporate income tax; Division of Legislative Services, et al., to establish a work group to assess the feasibility of transitioning to a unitary combined reporting system. (Patron–Watts, HJR 563)
- Income tax, state; conformity of the Commonwealth's taxation system with the Internal Revenue Code, deduction relating to payroll expenses. Amending §§ 58.1-301, 58.1-322.02, and 58.1-322.03. (Patron–Watts, HB 1935)
- Income tax, state; conformity of the Commonwealth's taxation system with the Internal Revenue Code, taxable income, subtractions. Amending §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402. (Patron–Howell, SB 1146)
- Income tax, state; Department of Taxation to include space on the appropriate forms for voluntary inclusion of personal and contact information, facilitated enrollment program. Amending §§ 38.2-6505, 58.1-3, and 58.1-341.1. (Patron–Sickles, HB 1884)
- Income tax, state; Joint Legislative Audit and Review Commission to study increasing the progressivity of Virginia's system, meetings shall be completed by November 30, 2022. (Patron-Watts, HJR 567)
- Income tax, state; subtraction for low-income military veterans with a permanent service-connected disability. Amending § 58.1-322.02. (Patron–Wiley, HB 2337)

INDIAN TRIBES

Wolf Creek Cherokee Tribe of Virginia; General Assembly to extend state recognition. (Patron–McQuinn, HJR 572)

INDUSTRIAL DEVELOPMENT

Conflict of Interests Act, State and Local Government; disclosure requirements, industrial development and economic development authorities. Amending § 2.2-3115. (Patron–Norment, SB 1171)

INFANTS

Fetal and Infant Mortality Review Team; Office of the Chief Medical Examiner of Department of Health shall convene a work group to develop a plan for the establishment of Team, report. (Patron-Ayala, HB 1950)

INMATES

Geriatric or terminally ill inmates; conditional release, notify by certified or verifiable electronic means to attorney for the Commonwealth. Amending § 53.1-136. (Patron-Norment, SB 1397)

Inmates; Board of Local and Regional Jails to review services provided during pregnancy, etc., report. (Patron-Favola, SB 1300)

INSTITUTIONS OF HIGHER EDUCATION; OTHER EDUCATIONAL AND CULTURAL INSTITUTIONS

- Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021; created. (Patron-Torian, HB 2178; Howell, SB 1145)
- Eligible Health Care Provider Reserve Directory; established, information concerning fourth-year medical students, etc. Adding §§ 23.1-110, 32.1-23.3, and 54.1-2400.03. (Patron–Hanger, SB 1436)
- Enslaved Ancestors College Access Scholarship and Memorial Program; established, report. Adding § 23.1-615.1. (Patron–Reid, HB 1980)
- Facial recognition technology; definition, authorization of use by local law-enforcement agencies and public institutions of higher education. Adding §§ 15.2-1723.2 and 23.1-815.1. (Patron–Aird, HB 2031)
- George Mason University; management agreement with the Commonwealth. (Patron-Bulova, HB 1986; Barker, SB 1204)
- Get Skilled, Get a Job, Give Back (G3) Fund and Program; established. Adding § 23.1-2911.2. (Patron–Filler-Corn, HB 2204; Saslaw, SB 1405)
- Higher educational institutions, public; admissions applications that contain questions about criminal history. Adding § 23.1-407.1. (Patron–Aird, HB 1930)
- Higher educational institutions, public; governing boards, meetings, input, and disclosures. Amending §§ 23.1-409, 23.1-802, and 23.1-1303. (Patron–Keam, HB 2120)
- SNAP benefits program; eligibility for benefits, postsecondary education. Amending §§ 63.2-608 and 63.2-801. (Patron–Helmer, HB 1820)
- Students; eligibility for in-state tuition and state financial assistance program. Amending § 23.1-506; adding § 23.1-505.1. (Patron–Lopez, HB 2123; Boysko, SB 1387)
- VirginiaSaves Program; established, membership. Amending § 23.1-701; adding §§ 2.2-2744 through 2.2-2756. (Patron–Torian, HB 2174)

INSURANCE

- Commonwealth Health Reinsurance Program; established, report. Amending §§ 38.2-4214 and 38.2-4319; adding §§ 38.2-6600 through 38.2-6607. (Patron–Sickles, HB 2332)
- Health insurance; association health plan for real estate salespersons, policy for group accident and sickness insurance. Amending § 38.2-3521.1. (Patron–Barker, SB 1341)
- Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization for drug by carrier. Amending § 38.2-3407.15:2. (Patron–McPike, SB 1269)
- Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization issued by carrier for drug, certain provisions shall not apply to the state employee health insurance plan, etc. Amending § 38.2-3407.15:2. (Patron–Heretick, HB 2008)
- Health insurance; carrier business practices, every carrier shall include in provider contracts a provision that prohibits a provider from discriminating against any enrollee. Amending § 38.2-3407.15. (Patron–Surovell, SB 1289)
- Health insurance; cost-sharing payments for covered tier one or tier two prescription asthma inhalers. Adding § 38.2-3407.15:6. (Patron–Askew, HB 1822)
- Health insurance; essential benefits, removes the prohibition on the provisions of coverage for abortions. Amending § 38.2-3451. (Patron–Hudson, HB 1896; McClellan, SB 1276)
- Health insurance; requiring health insurers and other carriers to establish reasonable protocols and procedures for reimbursing a health professional for services provided while such professional's credentialing application is pending. Amending §§ 38.2-4319 and 38.2-4509. (Patron–Head, HB 1829)
- Health maintenance organizations; updates provisions of the Code of Virginia related to insolvency procedures for health maintenance organizations (HMOs). Amending § 38.2-4310. (Patron–Kilgore, HB 1807)
- Income tax, state; Department of Taxation to include space on the appropriate forms for voluntary inclusion of personal and contact information, facilitated enrollment program. Amending §§ 38.2-6505, 58.1-3, and 58.1-341.1. (Patron–Sickles, HB 1884)
- Motor vehicle insurance; underinsured motor vehicle, uninsured motorist coverage, policies issued or renewed on or after January 1, 2022. Amending § 38.2-2206. (Patron–Obenshain, SB 1195)

INSURANCE (continued)

- Paid family leave; State Corporation Commission's Bureau of Insurance to review and make recommendations, report. (Patron-Favola, SB 1219)
- Personal injury claim; disclosure of insurance policy limits. Amending § 8.01-417. (Patron–Stuart, SB 1241)
- Pharmacies; freedom of choice by covered individual. Amending §§ 38.2-3407.7, 38.2-4209.1, and 38.2-4312.1. (Patron–Hodges, HB 2219)
- Prescription drugs; price transparency, definitions. Amending § 2.2-3705.6; adding §§ 32.1-23.3, 38.2-3407.15:6, 38.2-3407.22, 54.1-3436.1, and 54.1-3442.02. (Patron–Sickles, HB 2007)
- Property and casualty insurance form; approval of form by State Corporation Commission. Amending § 38.2-317. (Patron–Ayala, HB 1892)
- Public adjusters; continuing education requirements, repeals provision relating to continuing education, approval of credits, etc. Amending §§ 38.2-1866, 38.2-1867, 38.2-1868.1, 38.2-1869, 38.2-1871, and 38.2-1873; repealing § 38.2-1845.9. (Patron–Kilgore, HB 1942)
- State Corporation Commission; issuance or renewal of insurance licenses or registrations during an emergency. Amending § 38.2-200. (Patron–Mason, SB 1255)
- Telemedicine service; coverage of telehealth services by an insurer, etc., services delivered through real-time audio-only telephone, report. Amending §§ 32.1-325, 38.2-3418.16, and 54.1-3303. (Patron–Adams, D.M., HB 1987)
- Telemedicine services; payment for medical assistance for remote patient monitoring services, effective clause. Amending §§ 32.1-325 and 38.2-3418.16. (Patron–Barker, SB 1338)
- Uninsured and underinsured motorist insurance policies; bad faith. Amending §§ 8.01-66.1 and 38.2-2206. (Patron–Petersen, SB 1202)

INTERNATIONAL OVERDOSE AWARENESS DAY

International Overdose Awareness Day; designating as August 31, 2021, and each succeeding year thereafter. (Patron–Miyares, HJR 562)

INTERNET

Electric utilities; expands existing broadband capacity pilot program. Amending § 56-585.1:9. (Patron-Ayala, HB 1923)

ISLE OF WIGHT COUNTY

Isle of Wight County; authorized to impose an additional local sales and use tax to support schools. Amending §§ 58.1-602 and 58.1-605. (Patron–Norment, SB 1170)

JEFFERSON DAVIS HIGHWAY

Jefferson Davis Highway; renames any section of U.S. Route 1 to "Emancipation Highway." Repealing Chapter 286, 1922 Acts. (Patron-Cole, J.G., HB 2075)

JOHNS, BARBARA ROSE

Johns, Barbara Rose; Joint Committee of Congress on the Library requested to fill the Commonwealth's vacant spot in the National Statuary Hall Collection at the United States Capitol with a statue to commemorate. (Patron–Ward, HJR 525; Lucas, SJR 288)

JOHNSON, CAROLYN LOUISE

Johnson, Carolyn Louise; recording sorrow upon death. (Patron-McQuinn, HJR 739)

JOHNSON, CINTIA

Johnson, Cintia; commending. (Patron-Hope, HJR 658)

JOHNSTONE, RICHARD G., JR.

Johnstone, Richard G., Jr.; commending. (Patron-VanValkenburg, HJR 663)

JOINT ASSEMBLY

Journal of the House of Delegates; resolution	n to receive	Governor	Northam,	roll call,	guests
presented					. 71-73
Address of Governor Ralph S. Northam (printed	l as S.D. 1) .				72

JOINT LEGISLATIVE AUDIT AND REVIEW COMMISSION (JLARC)

- COVID-19; Joint Legislative Audit and Review Commission to study the impact on Virginia's public schools, students, and school employees, meetings shall be completed by November 30, 2022. (Patron–Guy, HJR 549; Lucas, SJR 308)
- General Assembly; confirming appointment to Joint Legislative Audit and Review Commission. (Patron-Plum, HJR 604)
- Income tax, state; Joint Legislative Audit and Review Commission to study increasing the progressivity of Virginia's system, meetings shall be completed by November 30, 2022. (Patron-Watts, HJR 567)
- Standards of Quality; Joint Legislative Audit and Review Commission to study the true cost of education in the Commonwealth and provide an accurate assessment of the costs to implement, meetings shall be completed for the first year by November 30, 2022. (Patron–Lewis, SJR 294)

JONES, THOMAS OVERTON

Jones, Thomas Overton; recording sorrow upon death. (Patron-Carr, HJR 650)

JUDGES, JUSTICES, AND OTHER ELECTIVE OFFICERS

Assault and battery; eliminates the mandatory minimum term of confinement of six months if committed against a judge, magistrate, law-enforcement officer, etc., persons with physical or mental disability. Amending § 18.2-57; adding § 19.2-9.2. (Patron–Morrissey, SB 1306)

JUDGES, JUSTICES, AND OTHER ELECTIVE OFFICERS

DUGES, JUSTICES, AND OTHER ELECTIVE OFFICERS
ALBERTSON, BRUCE D., Judge, Twenty-sixth Judicial Circuit
Certified
Nominated by District Senators
Nominated
Roll Call
Elected
ALLEN, TIMOTHY W., Judge, Juvenile and Domestic Relations District Court, Twenty-second Judicial District
Certified
Nominated by District Senators
Nominated
Roll Call
Elected
BENNETT, JEFFREY P., Judge, Juvenile and Domestic Relations District Court, Twenty-fourth
Judicial District
Certified
Nominated by District Senators
Nominated
Roll Call
Elected
BUSH, MICHAEL J., Judge, Juvenile and Domestic Relations District Court, Twenty-ninth Judicial
District
Certified
Nominated by Senate
Nominated
Roll Call
Elected
CAPSALIS, MANUEL A., Judge, General District Court, Nineteenth Judicial District
Certified
Nominated by District Senators
Nominated by District Schators 164-65 Nominated
Roll Call
-, -
Elected

JUDGI	CS, JUSTICES, AND OTHER ELECTIVE OFFICERS (continued) RSON, DAVID B., Judge, Twenty-third Judicial Circuit
C1	Certified
	Nominated by District Senators
	Nominated
	Roll Call
	Elected
C	DLEMAN, ROBERT P., Judge, General District Court, Thirty-first Judicial District
	Certified
	Nominated by District Senators
	Nominated by District Schators
	Roll Call
	Elected
DI	EHART, ERIN J., Judge, General District Court, Twenty-seventh Judicial District
וט	Certified
	Nominated by District Senators
	Nominated by District Schators 201, 21
	Roll Call
	Elected
DI	
וע	JNKUM, B. CRAIG, Judge, General District Court, Fourteenth Judicial District Certified
	Nominated by District Senators 18
	Nominated
	Roll Call
	Elected
FL	YTHE, TYNEKA L. D., Judge, General District Court, Seventh Judicial District
	Certified
	Nominated by District Senators
	Nominated
	Roll Call
	Elected
FF	ANKLIN, JOHN E., Judge, Juvenile and Domestic Relations District Court, Fifteenth Judicia
	District Certified
	Nominated by District Senators 19
	Nominated
	Roll Call
FF	Elected
FF	IEDMAN, MELISSA W., Judge, Juvenile and Domestic Relations District Court, Twenty-thir Judicial District
	Certified
	Nominated by District Senators
	Nominated by District Schators 202, 203, 21
	Roll Call
ET	Elected
FU	LTON, JUNIUS P., III, Judge, Fourth Judicial Circuit Certified
	Nominated by District Senators
	Nominated
	Roll Call
	Elected
Gl	EER, G. CARTER, Judge, Twenty-first Judicial Circuit
	Certified
	Nominated by District Senators
	Nominated. 198, 199, 21

JUDGES, JUSTICES, AND OTHER ELECTIVE OFFICERS (continued) Roll Call	221
Elected	
GRIFFITH, HILARY D., Judge, Juvenile and Domestic Relations District Court, Twenty-t	
Judicial District	
Certified	. 89
Nominated by District Senators	192
Nominated	217
Roll Call	226
Elected	230
HENDERSON, VIVIAN F., Judge, General District Court, Second Judicial District	
Certified	
Nominated by District Senators	
Nominated	
Roll Call	
Elected	229
HENDERSON-STITH, TONYA, Judge, General District Court, Eighth Judicial District	
Certified	
Nominated by District Senators	182
Nominated	217
Roll Call	222
Elected	229
HENSHAW, STACI A., Auditor of Public Accounts	
Certified	
Nominated	
Roll Call	228
Elected	231
HICKS, DAVID M., Judge, General District Court, Thirteenth Judicial District	
Certified	. 88
Nominated by District Senators	2-83
Nominated	217
Roll Call	223
Elected	229
HONEY, JOHN K., JR., Judge, General District Court, Fourteenth Judicial District	
Certified	. 88
Nominated by District Senators	184
Nominated	217
Roll Call	223
Elected	229
HUDSON, JEHMAL T., Member, State Corporation Commission	
Certified	. 97
Nominated	
Roll Call	7-28
Elected	231
HURLEY, JACK S., JR., Judge, Twenty-ninth Judicial Circuit	
Certified	. 87
Nominated by Senate	199
Nominated	
Roll Call	
Elected	
KELLY, PATRICIA, Judge, Fifteenth Judicial Circuit	
Certified	. 87
Nominated by District Senators	
Nominated 198 199	

JUDGES, JUSTICES, AND OTHER ELECTIVE OFFICERS (continued)	10.00
Roll Call	
Elected	228
KEMLER, LISA BONDAREFF, Judge, Eighteenth Judicial Circuit	0.5
Certified	8
Nominated by District Senators	
Nominated	,
Roll Call	
Elected	
KETRON, MARTHA P., Judge, Juvenile and Domestic Relations District Court, Twent Judicial District	ty-nintl
Certified	90
Nominated by Senate	
Nominated	
Roll Call	
Elected	-
LEVY, MICHAEL E., Judge, Fifteenth Judicial Circuit	25
Certified	8′
Nominated by District Senators	
Nominated	
Roll Call	
Elected	
LINDNER, MICHAEL J., Judge, General District Court, Nineteenth Judicial District	22
Certified	88
Nominated by District Senators	
Nominated	
Roll Call	-
Elected	
LYLE, JOSEPH B., Judge, Juvenile and Domestic Relations District Court, Twenty-eighth.	
District	
Certified	90
Nominated by District Senators	194
Nominated	04, 21
Roll Call	19, 22
Elected	230
MADDOX, STEPHANIE S., Judge, General District Court, Twenty-fourth Judicial District	
Certified	88
Nominated by District Senators	
Nominated	01, 21
Roll Call	18, 224
Elected	230
MCCLENNEY, JACQUELINE S., Judge, General District Court, Thirteenth Judicial District	t
Certified	88
Nominated by District Senators	183
Nominated	01, 21
Roll Call	18, 223
Elected	229
MCELYEA, TAMMY S., Judge, Thirtieth Judicial Circuit	
Certified	87
Nominated by District Senators	180
Nominated	00, 217
Roll Call	18, 222
Elected	229
MINOR, WILLIAM J., JR., Judge, General District Court, Nineteenth Judicial District	
Certified	89

JUDGES, JUSTICES, AND OTHER ELECTIVE OFFICERS (co	ontinued)
Nominated	
Roll Call	
	· · · · · · · · · · · · · · · · · · ·
Elected	
NAVARRO, ANGELA L., Member, State Corporation Commiss	ion
Certified	
Nominated	
Elected	
O'DONNELL, HUGH DAVID, Judge, Juvenile and Don	
Twenty-sixth Judicial District	
Certified	
Nominated by District Senators	
Nominated	
Roll Call	
Elected	
REYNOLDS, JAMES J., Judge, Twenty-second Judicial Circuit	
Certified	
Nominated by District Senators	178
Nominated	
Roll Call	
Elected	
SCOTT, TASHA D., Judge, General District Court, Fourth Judic	
Certified	iai District
Nominated by District Senators	
Nominated	
Roll Call	
Elected	
SHAH, MANSI J., Judge, General District Court, Thirteenth Jud	
Certified	iiciai District
Nominated by District Senators	
Nominated by District Senators	
Roll Call	
Elected	
SIMMONS, DEANIS L., Judge, Twenty-eighth Judicial Circuit Certified	or
Nominated by District Senators	
	, ,
Roll Call	
Elected	
SIMMONS, LYN M., Judge, Juvenile and Domestic Relation	ons District Court, Fourth Judicia
District	90
Certified	
Nominated by District Senators	
Nominated	
Roll Call	,
Elected	
SNEE, TINA L., Judge, General District Court, Nineteenth Judic	
Certified	
Nominated by District Senators	
Nominated	,
Roll Call	
Flected	220

JUDGES, JUSTICES, AND OTHER ELECTIVE	VE OFFICERS (continued)
District	Domestic Relations District Court, Nineteenth Judicial
	89
SPRY, BRENDA C., Judge, Third Judicial C	
Certified	
STEIN, M. SCOTT, Judge, General District	
Certified	
	General District Court, Twenty-third Judicial District
THOMPSON, TERRIE N., Member, Judicia	
Certified	
TISINGER, AMY B., Judge, General Distric	
Certified	
•	
TRAN, JOHN M., Judge, Nineteenth Judicia	
Certified	
•	
KOII Căli	
UVANNI, FRANK G., Judge, Juvenile and District	Domestic Relations District Court, Fifteenth Judicial
	89
•	
	nd Domestic Relations District Court, Twenty-seventh
Judicial District Certified	90
Nominated Nominated	

JUDGES, JUSTICES, AND OTHER ELECTIVE OFFICERS (continued) Roll Call
Elected
WAYMACK, JACQUELINE R., Judge, Juvenile and Domestic Relations District Court, Sixth Judicial District
Certified89
Nominated by District Senators
Nominated
Elected
WEIMER, CARROLL A., JR., Judge, Thirty-first Judicial Circuit
Certified87
Nominated by District Senators
Nominated
Roll Call
Elected
Certified87
Nominated by District Senators
Nominated
Roll Call 218, 220 Elected 228
WILLIAMS, GINO W., Judge, General District Court, Twenty-seventh Judicial District
Certified89
Nominated by District Senators
Nominated
Roll Call
Elected
NOT ELECTED: O'BERRY, PAMELA Y., General District Court, Twelfth Judicial District
Nominated
Roll Call
PLOWMAN, JAMES E., Member, Judicial Inquiry and Review Commission
Certified90
Nominated
NO ELECTION RESULTED:
General District Court Twelfth Judicial District
Judicial Inquiry and Review Commission
JUDGMENT Judgments; limitations on enforcement, judgment liens, settlement agents, effective date Amending §§ 8.01-251, 8.01-458, and 55.1-339. (Patron–Coyner, HB 2099)
Orders of restitution; docketed on behalf of victim, assignment of judgment, enforcement Amending §§ 19.2-305.1, 19.2-305.2, 19.2-349, and 19.2-354. (Patron–Bell, HB 2233 Stanley, SB 1426)
JUDICIARY, COMMITTEE ON THE Members listed
JUVENILE AND DOMESTIC RELATIONS DISTRICT COURTS Special immigrant juvenile status; permits the juvenile and domestic relations district court to retain jurisdiction in certain cases, etc. Amending § 16.1-241. (Patron–Surovell, SB 1181)
HIVENH E HISTICE

JUVENILE JUSTICE

Child support payments; juvenile in custody of or committed to the Department of Juvenile Justice.

Amending §§ 16.1-263, 16.1-286, and 16.1-290. (Patron–Hope, HB 1912)

JUVENILE JUSTICE (continued)

Correctional facilities; restrictions on use of isolated confinement, effective clause. Adding §§ 53.1-39.2 and 66-20.1. (Patron–Morrissey, SB 1301)

Juveniles; eligibility for commitment to the Department of Juvenile Justice, eligibility for predispositional confinement in a secure facility. Amending §§ 16.1-248.1, 16.1-249, 16.1-278.7, and 16.1-278.8. (Patron–Marsden, SB 1456)

JUVENILES

Child support payments; juvenile in custody of or committed to the Department of Juvenile Justice. Amending §§ 16.1-263, 16.1-286, and 16.1-290. (Patron–Hope, HB 1912)

Juvenile intake and petition; appeal to a magistrate on a finding of no probable cause. Amending §§ 16.1-256 and 16.1-260. (Patron–Jenkins, HB 1878)

Juvenile offenders; youth justice diversion programs, report. Amending § 16.1-260; adding § 16.1-309.11. (Patron–Mullin, HB 2017)

Juvenile records; confidentiality of records relevant to treatment, services, etc., exceptions. Amending § 16.1-300. (Patron–Barker, SB 1206)

Juveniles; competency evaluation, appointed evaluator or director of community services board, etc., shall acknowledge receipt of court order to clerk of court. Amending § 16.1-356. (Patron–Deeds, SB 1248)

Juveniles; eligibility for commitment to the Department of Juvenile Justice, eligibility for predispositional confinement in a secure facility. Amending §§ 16.1-248.1, 16.1-249, 16.1-278.7, and 16.1-278.8. (Patron–Marsden, SB 1456)

Juveniles; release and review hearing for serious offender, plea agreement. Amending §§ 16.1-285.1 and 16.1-285.2. (Patron–Jones, HB 1991)

Special immigrant juvenile status; permits the juvenile and domestic relations district court to retain jurisdiction in certain cases, etc. Amending § 16.1-241. (Patron–Surovell, SB 1181)

KANG, AE JA

Kang, Ae Ja; recording sorrow upon death. (Patron-Keam, HJR 722)

KEATTS, FLOURNOY A.

Keatts, Flournoy A.; recording sorrow upon death. (Patron-Marshall, HJR 767)

KELLEHER, LINDA Y.

Kelleher, Linda Y.; commending. (Patron-Hope, HJR 639)

KEYS-CHAVIS, EDNA ELIZABETH

Keys-Chavis, Edna Elizabeth; recording sorrow upon death. (Patron-McQuinn, HJR 730)

KIGGANS, JENNIFER A.

Prayer offered
Statement on vote:
S.B. 1321

KINDERGARTENS AND PRESCHOOLS

Kindergarten through grade 3; reading intervention services for certain students. Amending § 22.1-253.13:1. (Patron–Delaney, HB 1865)

KING GEORGE COUNTY

Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron-Cole, J.G., HB 1819)

KOREAN CENTRAL PRESBYTERIAN CHURCH

Korean Central Presbyterian Church; commending. (Patron-Helmer, HJR 763)

LABOR AND EMPLOYMENT

Employee classification; provision of personal protective equipment in response to a disaster. Amending §§ 40.1-28.7:7 and 60.2-212; adding § 65.2-301.2. (Patron–Batten, HB 2134)

Employee protections; medicinal use of cannabis oil. Adding § 40.1-27.4. (Patron-Helmer, HB 1862)

LABOR AND EMPLOYMENT (continued)

Employees providing domestic service; application of laws applicable to employee safety and workers' compensation. Amending §§ 40.1-2, 40.1-49.3, 40.1-49.8, 65.2-101, and 65.2-305. (Patron–Gooditis, HB 2032)

Employer or other person; retaliatory discharge of employee prohibited. Amending §§ 40.1-27.3 and 65.2-308. (Patron–Carter, HB 1754)

Employers; reporting outbreaks of COVID-19, effective clause. (Patron-Lewis, SB 1362)

Minimum wage; eliminates the exemption from Virginia's requirements for persons employed as farm laborers or farm employees and any individual employed as a temporary foreign worker. Amending § 40.1-28.9. (Patron–Ward, HB 1786)

Paid sick leave; employers to provide to certain employees, essential workers that work an average of 20 hours weekly, etc., provisions shall not apply to any retail business with fewer than 25 employees. Adding §§ 40.1-33.3 through 40.1-33.9. (Patron–Guzman, HB 2137)

Virginia Human Rights Act; application of laws applicable to employee safety and payment of wages. Amending §§ 2.2-3905, 40.1-2, 40.1-29, 40.1-49.3, and 40.1-49.8. (Patron–McClellan, SB 1310)

Virginia Jobs Investment Program and Fund; minimum wage requirements. Amending § 2.2-2240.3. (Patron–Bagby, HB 1967)

Virginia Overtime Wage Act; overtime compensation employees, definitions, penalties. Amending §§ 40.1-29 and 40.1-29.1; adding § 40.1-29.2. (Patron–Mullin, HB 2063)

LABOR, SECRETARY OF

Labor, Secretary of; position created in Governor's Cabinet, removes position of Chief Workforce Development Advisor. Amending §§ 2.2-200, 2.2-204, 2.2-205, 2.2-205.1, 2.2-435.6, 2.2-435.8, 2.2-435.9, 2.2-435.10, 2.2-2471, 2.2-2471.1, 2.2-2472, 2.2-2472.2, 2.2-2472.3, and 30-377; adding §§ 2.2-214.2 and 2.2-214.3; repealing § 2.2-435.7. (Patron–Simonds, HB 2321)

LANDFILLS

Waste Diversion and Recycling Task Force; Department of Environmental Quality to continue Task Force. (Patron–Hashmi, SB 1319)

LANDLORD AND TENANT

Virginia Residential Landlord and Tenant Act; landlord remedies, noncompliance with rental agreement, payment plan, removal of sunset. Repealing second enactment of Chapter 46, 2020 Sp. I Acts. (Patron-Price, HB 1889)

Virginia Residential Landlord and Tenant Act; noncompliance with rental agreement, etc., prohibition on using negative credit information that arose during a closure of the United States Government against certain applications for tenancy, penalty. Amending § 55.1-1245. (Patron–Helmer, HB 1908)

Virginia Residential Landlord and Tenant Act; sample termination notice, landlord's acceptance of rent with reservation, tenant's right of redemption. Amending §§ 36-139 and 55.1-1250. (Patron–Price, HB 2014)

LARCENY

Larceny; repeals punishment for conviction of second or subsequent misdemeanor. Repealing § 18.2-104. (Patron-Plum, HB 2290)

LAW-ENFORCEMENT OFFICERS

Assault and battery; eliminates the mandatory minimum term of confinement of six months if committed against a judge, magistrate, law-enforcement officer, etc., persons with physical or mental disability. Amending § 18.2-57; adding § 19.2-9.2. (Patron–Morrissey, SB 1306)

Law-enforcement civilian oversight bodies; adds a sheriff's office to those law-enforcement agencies that may be overseen by a body created by a locality and adds a nonprobationary deputy sheriff to those officers who are subject to such body, disciplinary determinations. Amending § 9.1-601. (Patron–Williams Graves, HB 2291)

Law-enforcement officer; duty to render aid, duty to report wrongdoing by another officer. Amending § 52-30.1; adding §§ 15.2-1704.1 and 15.2-1704.2. (Patron–Levine, HB 1948)

Search warrants; date and time of issuance, law-enforcement officer to be recognizable and identifiable, exceptions. Amending § 19.2-56. (Patron–Stuart, SB 1475)

LAW-ENFORCEMENT OFFICERS (continued)

Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, emergency medical services personnel, law-enforcement officers, and correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Jones, HB 2207)

Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, law-enforcement officers or correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Saslaw, SB 1375)

LAWRENCE, BARRY R.

Lawrence, Barry R.; commending. (Patron-VanValkenburg, HJR 697)

LAWYERS

Lawyers; client accounts, repeals the provision prohibiting the Supreme Court of Virginia from adopting a disciplinary rule requiring that lawyers deposit client funds in an interest-bearing account. Amending § 54.1-3916; repealing § 54.1-3915.1. (Patron–Sullivan, HB 1853)

LEAVES OF ABSENCE

Deeds, R. Creigh	4
Lewis, Lynwood W., Jr	4
Spruill, Lionell, Sr	4
Stuart, Richard H	4
Vogel, Jill Holtzman	9

LEDGERTON, ELIZABETH ANN KERR

Ledgerton, Elizabeth Ann Kerr; recording sorrow upon death. (Patron-Webert, HJR 678)

LEGISLATIVE SERVICES, DIVISION OF

Corporate income tax; Division of Legislative Services, et al., to establish a work group to assess the feasibility of transitioning to a unitary combined reporting system. (Patron–Watts, HJR 563)

LEWIS, IDA JOHNSON

Lewis, Ida Johnson; recording sorrow upon death. (Patron-Hudson, HJR 704)

LEWIS, LYNWOOD W., JR.

Leave of absence
Participated remotely
Prayer offered
Statements on votes:
S.B. 1105
S.B. 1113
S.B. 1384

LIBERTY AMENDMENTS MONTH

Liberty Amendments Month; designating as June 19 through the third Monday in July 2021 and each succeeding year thereafter. (Patron–Bagby, HJR 583; Locke, SJR 323)

LIBRARIES

Library aid; former regional library systems that receive state aid notwithstanding a loss of regional library status due to a city reversion, shall receive such aid regardless. Amending § 15.2-4116. (Patron–Edmunds, HB 2323)

LICENSE PLATES AND REGISTRATION

License plates, special; establishes a system for determining cost of plates for recipients of a military decoration. Amending §§ 46.2-742.1, 46.2-742.2, 46.2-745.1, and 46.2-745.2; adding §§ 46.2-725.3, 46.2-745.4, and 46.2-745.5. (Patron–McQuinn, HB 2069)

License plates, special; issuance for supporters of Ducks Unlimited, fees. Amending § 46.2-749.7. (Patron–Stuart, SB 1229)

LICENSE PLATES AND REGISTRATION (continued)

License plates, special; range of bumper heights for any street rod bearing a street rod license plate, repeals the authorizations for issuance of certain plates no longer issued due to low plate sales, expired authorizations, etc. Amending § 46.2-1063; repealing §§ 46.2-746.6, 46.2-746.9, 46.2-746.12, 46.2-747, 46.2-748, 46.2-749.10, 46.2-749.69:1, and Section 1 of Chapter 776, 2010 Acts. (Patron–Marsden, SB 1136)

License plates, special; removes fee for issuance to a member of the Virginia National Guard. Amending § 46.2-744. (Patron–Wyatt, HB 2261)

License plates, special; removes the fee for the issuance of a license plate for retired members of the Virginia National Guard. Amending § 46.2-746.5. (Patron–Walker, HB 1796)

Vehicle registration; expands eligibility for special communication needs indicator. Amending § 46.2-600.1. (Patron–Bell, HB 1960; Barker, SB 1470)

Vehicle registration fees; imposes an additional fee to be deposited into Public Safety Trust Fund. Amending §§ 46.2-686 and 46.2-694; adding § 46.2-694.2. (Patron–Edwards, SB 1211)

LIEUTENANT GOVERNOR

Marriage; persons who may celebrate rites, authorizes current members of the General Assembly, Governor, Lieutenant Governor, and Attorney General. Amending § 20-25. (Patron–Cosgrove, SB 1142)

LINABURG, DENNIS

Linaburg, Dennis; commending. (Patron-LaRock, HJR 760)

LINE OF DUTY ACT

Line of Duty Act; Virginia licensed health practitioners required to conduct medical reviews, persons issued a comparable license, as determined by Virginia Retirement System, by the District of Columbia or a state that is contiguous to Virginia. Amending §§ 9.1-404 and 9.1-405. (Patron–DeSteph, SB 1367)

LITTER CONTROL

Litter; increases fine for dumping or disposing of trash or other unsightly matter on public or private property. Amending § 33.2-802. (Patron–Edmunds, HB 1801)

LOCAL GOVERNMENT AND OFFICIALS

Conflict of Interests Act, State and Local Government; disclosure requirements, industrial development and economic development authorities. Amending § 2.2-3115. (Patron–Norment, SB 1171)

Local elections for governing bodies; elections for school boards, qualification of voters. Amending §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223. (Patron–Convirs-Fowler, HB 2198)

Local government; authority to reduce the speed limit in a business district or residence district. Amending § 46.2-1300. (Patron-Carr, HB 1903)

LOCAL GOVERNMENT, COMMITTEE ON

LOCKE, MAMIE E.

LOGAN, ENA AMPY

Logan, Ena Ampy; recording sorrow upon death. (Patron-McQuinn, HJR 728)

LOUDOUN COUNTY

Loudoun County; County to enter into a contract with the Department of Health for the local administration of local health services. (Patron-Favola, SB 1221)

Loudoun County Office of Elections and Voter Registration; commending. (Patron-Subramanyam, HJR 683)

Loudoun County school board; staggered terms of its members. Amending § 22.1-57.3:1.1. (Patron–Reid, HB 1838)

LUCAS, L. LOUISE

President pro tempore:	
Named Senator McClellan to perform duties	497
Named Senator McDougle to perform duties	499
Presiding	433
Statements on votes:	
S.B. 1301	453
S.B. 1316.	452

LUCK, JOANN GRANT

Luck, Joann Grant; commending. (Patron-Adams, L.R., HJR 745)

LUTHER, T. DAVID

Luther, T. David; commending. (Patron–Marshall, HJR 776)

LYNCHBURG, CITY OF

Lynchburg, City of; amending charter, salaries of members of City Council. (Patron-Walker, HB 2180)

MADISON COUNTY

Combined transient occupancy and food and beverage tax; for purposes of taxes that are currently authorized for Rappahannock and Madison Counties, the rate limit for such tax shall be the same as if the two taxes were imposed separately. Amending § 58.1-3842. (Patron–Hanger, SB 1438)

MAGISTRATES

Juvenile intake and petition; appeal to a magistrate on a finding of no probable cause. Amending §§ 16.1-256 and 16.1-260. (Patron–Jenkins, HB 1878)

MANDT, SHEILA KAVANAGH

Mandt, Sheila Kavanagh; recording sorrow upon death. (Patron-Carr, HJR 687)

MANUFACTURED HOUSING

Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, cases of a deed of trust conveying owner-occupied residential real estate, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–Torian, HB 2175)

Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–McClellan, SB 1327)

MARIJUANA

Marijuana; legalization of simple possession, etc., expungement of criminal records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-251.13, 18.2-251.13, 18.2-251.13, 18.2-251.13, 18.2-251.13, 18.2-251.13, 18.2-251.1, 18.2-251.13, 18.2-251.1, 18.2-251.13, 18.2-265.2, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258.0, 18.2-258.1, 18.2-308.016, 18.2-308.15, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-390, 19.2-390.2, 19.2-392.1,

MARIJUANA (continued)

19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9, 19.2-392.10, 19.2-392.11, 19.2-392.12, 19.2-392.13, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Herring, HB 2312)

Marijuana; unlawful possession in certain cases, using or consuming while in a motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325,2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, $15.2 - 2820, \quad 16.1 - 69.40:1, \quad 16.1 - 69.48:1, \quad 16.1 - 228, \quad 16.1 - 260, \quad 16.1 - 273, \quad 16.1 - 278.8:01, \quad 16.1 - 278.8:$ 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Ebbin, SB 1406)

MARRIAGE

Constitutional amendment; fundamental right to marry, removes same-sex marriage prohibition (first reference). Amending Section 15-A of Article I. (Patron–Sickles, HJR 582; Ebbin, SJR 270)

Marriage; persons who may celebrate rites, authorizes current members of the General Assembly, Governor, Lieutenant Governor, and Attorney General. Amending § 20-25. (Patron–Cosgrove, SB 1142)

MASON, T. MONTGOMERY

MATHEWS COUNTY

Mathews County; board of supervisors may appoint one employee of the locality onto the board of directors for the Economic Development Authority of the County. Amending § 15.2-4904. (Patron–Hodges, HB 2186)

MCCLELLAN, JENNIFER L.

Addressed the Senate in memory of the Reverend Dr. Martin Luther King, Jr.; requested
adjournment in memory
Named by President pro tempore to perform duties
Prayer offered
Presiding
Statement on vote:
S.J.R. 300

N P P	OUGLE, RYAN T. Named by President pro tempore to perform duties
	S.B. 1219
	OWELL, CHARLES LINDSAY McDowell, Charles Lindsay; recording sorrow upon death. (Patron–Carr, HJR 686)
	IKE, JEREMY S. Statements on votes: S.B. 1380. 463 S.B. 1392. 464
	ICAID AND MEDICARE PROGRAMS Personal care aides; Department of Medical Assistance Services shall establish an orientation program for all aides who provide self-directed services through the Medicaid program. Adding § 32.1-331.04. (Patron–Locke, SB 1102)
H	HCAL TREATMENT, CARE, AND ASSISTANCE Hormonal contraceptives; payment of medical assistance for 12-month supply. Amending § 32.1-325. (Patron–Boysko, SB 1227) School-based health services; Board of Medical Assistance Services to amend the state plan for medical assistance services to provide for payment, no health care provider who provides services through telemedicine shall be required to use proprietary technology or applications in order to be reimbursed. Amending §§ 32.1-325 and 32.1-326.3. (Patron–Dunnavant, SB 1307)
AAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAAA	Anderson, Dolson Barnett; recording sorrow upon death. (Patron—Carr, HJR 646) Andrews, Edward; recording sorrow upon death. (Patron—McQuinn, HJR 729) Bigger, Josephine Johnson; recording sorrow upon death. (Patron—McQuinn, HJR 736) Bisson, Ann; recording sorrow upon death. (Patron—Hope, HJR 642) Bland, Larry Jerome; recording sorrow upon death. (Patron—Carr, HJR 685; McQuinn, HJR 732) Brown, Algenon L.; recording sorrow upon death. (Patron—LaRock, HJR 742) Brown, Jean Smith; recording sorrow upon death. (Patron—LaRock, HJR 742) Byers, William Walker, Jr.; recording sorrow upon death. (Patron—Hudson, HJR 703) Cameron, Frederick; recording sorrow upon death. (Patron—Tran, HJR 782) Carey, James Alfred; recording sorrow upon death. (Patron—Tyler, HJR 691) Colden, Tony E., Jr.; recording sorrow upon death. (Patron—Bourne, HJR 724) Coleman, Lovell Louis, Sr.; recording sorrow upon death. (Patron—Hudson, HJR 708) Colyer, Bob, Sr.; recording sorrow upon death. (Patron—Hudson, HJR 709) Dabney, Herbert Allen, III; recording sorrow upon death. (Patron—Hudson, HJR 709) Dabney, Herbert Allen, III; recording sorrow upon death. (Patron—Hudson, HJR 743) Denny, Alfred Jerome; recording sorrow upon death. (Patron—Hope, HJR 653) Dick, Paul Wendell; recording sorrow upon death. (Patron—LaRock, HJR 743) Dukas, Helen; recording sorrow upon death. (Patron—McQuinn, HJR 725) Farlow, Mary Elene Williams; recording sorrow upon death. (Patron—McQuinn, HJR 771) Ferguson, Earl Mark; recording sorrow upon death. (Patron—McQuinn, HJR 731) Ferguson, Earl Mark; recording sorrow upon death. (Patron—Bagby, HJR 644) Ford, Rudolph E., Jr.; recording sorrow upon death. (Patron—Bagby, HJR 644) Ford, Rudolph E., Jr.; recording sorrow upon death. (Patron—Bagby, HJR 716; McQuinn, HJR 737) Gerner, Charles Lincoln; recording sorrow upon death. (Patron—Hope, HJR 652)

MEMORIAL RESOLUTIONS (continued)

Hairston, Joe Louis; recording sorrow upon death. (Patron-Marshall, HJR 769)

Hill, Jamile J.; recording sorrow upon death. (Patron-Hayes, HJR 689)

Hodge, Perry Anthony; recording sorrow upon death. (Patron–Hurst, HJR 679)

Hogg, Oakley W., III; recording sorrow upon death. (Patron-Wyatt, HJR 711)

Holmes, Bobby Eugene; recording sorrow upon death. (Patron–McQuinn, HJR 726)

Hooker, Katherine Bridgforth; recording sorrow upon death. (Patron-Carr, HJR 651)

Johnson, Carolyn Louise; recording sorrow upon death. (Patron-McQuinn, HJR 739)

Jones, Thomas Overton; recording sorrow upon death. (Patron-Carr, HJR 650)

Kang, Ae Ja; recording sorrow upon death. (Patron-Keam, HJR 722)

Keatts, Flournoy A.; recording sorrow upon death. (Patron-Marshall, HJR 767)

Keys-Chavis, Edna Elizabeth; recording sorrow upon death. (Patron-McQuinn, HJR 730)

Ledgerton, Elizabeth Ann Kerr; recording sorrow upon death. (Patron-Webert, HJR 678)

Lewis, Ida Johnson; recording sorrow upon death. (Patron-Hudson, HJR 704)

Logan, Ena Ampy; recording sorrow upon death. (Patron–McQuinn, HJR 728)

Mandt, Sheila Kavanagh; recording sorrow upon death. (Patron-Carr, HJR 687)

McDowell, Charles Lindsay; recording sorrow upon death. (Patron-Carr, HJR 686)

Merchant, John F.; recording sorrow upon death. (Patron-Hudson, HJR 706)

Midgett, Robert Preston, II; recording sorrow upon death. (Patron-Knight, HJR 670)

Minor, Mozelle Willis; recording sorrow upon death. (Patron-Carr, HJR 688)

Munford, Joan Hardie; recording sorrow upon death. (Patron-Hurst, HJR 680)

Plaster, Grover Harold; recording sorrow upon death. (Patron-Marshall, HJR 773)

Plotnick, Virginia Leonard; recording sorrow upon death. (Patron-Hudson, HJR 710)

Renfro, Nancy Todd; recording sorrow upon death. (Patron-Hope, HJR 643)

Robertson, Franklin Delano; recording sorrow upon death. (Patron-Morefield, HJR 668)

Royall, William Archer, Jr.; recording sorrow upon death. (Patron-Carr, HJR 690)

Shelor, Isabel Gallimore; recording sorrow upon death. (Patron–Rush, HJR 669)

Slaughter, Alexander Hoke; recording sorrow upon death. (Patron-Carr, HJR 649)

Smith, Javier J.; recording sorrow upon death. (Patron-McQuinn, HJR 738)

Tarantino, Joseph Maurice; recording sorrow upon death. (Patron-Robinson, HJR 662)

Teague, Joseph William, Sr.; recording sorrow upon death. (Patron-Hudson, HJR 707)

Thorpe, Avicia Beatrice Hooper; recording sorrow upon death. (Patron-Marshall, HJR 768)

Tuell, Mary Louise; recording sorrow upon death. (Patron-Torian, HJR 741)

Vaughan, George B.; recording sorrow upon death. (Patron-Bell, HJR 702)

Walker, Arthur William; recording sorrow upon death. (Patron-Hudson, HJR 723)

Washington, Andrew D.; recording sorrow upon death. (Patron-Krizek, HJR 740)

Wingfield, Robert George; recording sorrow upon death. (Patron-McNamara, HJR 770)

Womack, William Henry, Jr.; recording sorrow upon death. (Patron-McQuinn, HJR 735)

Wood, Thelonius Leander; recording sorrow upon death. (Patron-McQuinn, HJR 734)

Woods, Michael Wayne; recording sorrow upon death. (Patron–Marshall, HJR 772)

Woodson, Willie; recording sorrow upon death. (Patron-McQuinn, HJR 733)

Wright, James William, Sr.; recording sorrow upon death. (Patron-Hudson, HJR 705)

Zaleski, Lucille Minchin; recording sorrow upon death. (Patron-Jones, HJR 766)

MERCHANT, JOHN F.

Merchant, John F.; recording sorrow upon death. (Patron-Hudson, HJR 706)

MERIWETHER, BOBBIE JEAN

Meriwether, Bobbie Jean; commending. (Patron-Cole, M.L., HJR 665)

MIDDLE SCHOOLS

Economic education and financial literacy required in middle and high school grades; employment arrangements. Amending § 22.1-200.03. (Patron–Cole, J.G., HB 1905)

MIDDLE SCHOOLS (continued)

Students; guidelines on excused student absences, each school board shall permit one school day-long excused absence per school year for any middle or high school student who is absent from school to engage in a civic event. Amending § 22.1-254. (Patron–McClellan, SB 1439)

MIDGETT, ROBERT PRESTON, II

Midgett, Robert Preston, II; recording sorrow upon death. (Patron-Knight, HJR 670)

MIDWIVES

Certified nurse midwives; eliminates the requirement that midwives practice pursuant to a practice agreement and shall practice in accordance with regulations of the Boards of Medicine and Nursing. Amending §§ 54.1-2957 and 54.1-2957.01. (Patron–Adams, D.M., HB 1817)

Licensed certified midwives; clarifies definition, licensure, practice shall be in consultation with a licensed physician with a practice agreement, report. Amending § 54.1-2900; adding § 54.1-2957.04. (Patron–Gooditis, HB 1953; Lucas, SB 1320)

MILITARY AND EMERGENCY LAWS

Emergency and quarantine orders, certain; additional procedural requirements. Amending § 32.1-48.010; adding § 44-146.17:1.1. (Patron–Petersen, SB 1237)

Emergency Management Equity Working Group; established. Amending § 44-146.18. (Patron-Spruill, SB 1296)

Emergency Services and Disaster Law; local and interjurisdictional emergency operations plans. Amending § 44-146.19. (Patron–Askew, HB 2085)

MILLER, MARTY L.

Miller, Marty L.; commending. (Patron-Hayes, HJR 693)

MINES AND MINING

Electric generating facility closures; public closure, integrated resource plans, provisions apply to facility owned or operated by an electric utility. Amending § 56-599; adding § 45.1-394.1. (Patron–Deeds, SB 1247)

Electric generating facility closures; public disclosure, integrated resource plans. Amending § 56-599; adding § 45.1-394.1. (Patron–Subramanyam, HB 1834)

Gold; Secretary of Natural Resources, et al., shall establish a work group to study mining and processing, issuance of certain mining permits. (Patron–Guzman, HB 2213)

Mines and Mining and Virginia Energy Plan; revision of Titles 45.1 and 67. Adding §§ 10.1-1332, 10.1-1333, 33.2-120, 33.2-221.1, 45.2-100 through 45.2-402, 45.2-500 through 45.2-1051, 45.2-1100 through 45.2-1505, 45.2-1600 through 45.2-1649, 45.2-1700 through 45.2-2119, 55.1-1820.1, 55.1-1951.1, 55.1-2133.1, and 56-614 through 56-624; repealing §§ 11-34.1 through 11-34.4, 45.1-161.1 through 45.1-399, 62.1-195.1, 62.1-195.3, and 67-100 through 67-1700. (Patron–Edwards, SB 1453)

Mines, Minerals and Energy, Department of; renamed the Department of Energy. Amending §§ 2.2-204, 2.2-604.2, 2.2-1157, 2.2-1176.1, 2.2-3705.6, 2.2-4006, 10.1-606.3, 10.1-659, 10.1-1194, 10.1-1329, 10.1-1330, 10.1-1406.2, 11-34.3, 15.2-958.3, 15.2-980, 15.2-2224, 23.1-2626, 23.1-2627, 28.2-1208, 30-275, 33.2-236, 45.1-161.1, 45.1-161.2, 45.1-161.5, 45.1-161.15, 45.1-161.179, 45.1-161.292:2, 45.1-180, 45.1-229, 45.1-230, 45.1-270.4:1, 45.1-361.28, 45.1-361.41, 45.1-383, 45.1-390, 56-265.15:1, 56-576, 56-585.5, 56-594.3, 56-596.2, 58.1-439.2, 58.1-439.12:02, 58.1-360, 58.1-3706, 58.1-3745, 62.1-44.15:21, 62.1-44.15:66, 62.1-195.1, 62.1-243, 62.1-256, 62.1-259, 63.2-805, 67-200, 67-202.1, 67-602, 67-900, 67-1000, 67-1206, 67-1208, 67-1209, 67-1403, and 67-1506; adding § 45.1-161.4:1. (Patron–Sullivan, HB 1855)

Tourism Development Authority; changes name of Authority to the Heart of Appalachia Tourism Authority. Amending §§ 15.2-5500, 15.2-5501, 15.2-5505, 15.2-5506, and 45.1-246. (Patron–Pillion, SB 1399)

MINOR, MOZELLE WILLIS

Minor, Mozelle Willis; recording sorrow upon death. (Patron-Carr, HJR 688)

MINORITY BUSINESSES

Small, women-owned, and minority-owned businesses; right to appeal denial of initial certification. Amending § 2.2-1606. (Patron–Mundon King, HB 2172)

MINORS

Health, Department of; certain communication prohibited regarding matters with a minor, exception. Adding § 32.1-4.1. (Patron–Peake, SB 1235)

License restrictions for minors; prohibition on use of handheld personal communications devices. Amending § 46.2-334.01. (Patron–Robinson, HB 1846)

Pharmaceutical processors; permits processors to produce and distribute cannabis products, dispensing botanical cannabis to a minor, certification. Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8. (Patron–Lucas, SB 1333)

MISDEMEANORS

Summons; eliminates the requirement that a promise to appear be completed after issuance for a misdemeanor offense or an administrative violation. Amending §§ 2.2-311, 19.2-74, 28.2-901, 29.1-210, 46.2-936, 46.2-940, 54.1-306, 54.1-2506, and 54.1-4407. (Patron–Hanger, SB 1437)

MIXED BEVERAGES, ALCOHOLIC

Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption, report, sunset provision. Amending §§ 4.1-119, 4.1-204, 4.1-206.1, 4.1-206.3, 4.1-207, 4.1-210, 4.1-212.1, and 4.1-221. (Patron–Bulova, HB 1879; Bell, SB 1299)

MORTGAGES

Deeds of trust; amendment to loan document, statement of interest rate of a refinanced mortgage. Amending § 55.1-319; adding § 55.1-318.1. (Patron–Heretick, HB 1882)

MOTOR VEHICLE INSURANCE

Motor vehicle insurance; underinsured motor vehicle, uninsured motorist coverage, policies issued or renewed on or after January 1, 2022. Amending § 38.2-2206. (Patron–Obenshain, SB 1195)

Motor vehicle liability insurance; increases coverage amounts. Amending §§ 46.2-419, 46.2-472, and 46.2-2057. (Patron–Surovell, SB 1182)

Uninsured and underinsured motorist insurance policies; bad faith. Amending §§ 8.01-66.1 and 38.2-2206. (Patron-Petersen, SB 1202)

MOTOR VEHICLES

Bicycles; permits operators to treat a stop sign as a yield sign in certain situations. Amending §§ 46.2-839 and 46.2-905; adding § 46.2-903.1. (Patron–Hurst, HB 2262)

Commercial driver's licenses; disqualifies for life from holding a license persons convicted of a felony involving an act or practice of severe forms of trafficking, etc. Amending §§ 46.2-341.18, 46.2-382, and 46.2-1702. (Patron–Delaney, HB 1868)

Commissioner of the Department of Motor Vehicles; powers and duties during a declared state of emergency. Amending § 46.2-223. (Patron-Roem, HB 1828)

Crosswalk design; Commissioner of Highways shall convene a work group to determine whether there should be model policies for design and installation, and, if so, establish recommendations. (Patron–Keam, HB 1841)

Driving privileges, certain; Commissioner of DMV to reinstate privileges, and waive fees for individuals whose privileges were suspended for failure to pay court fines and costs in other jurisdictions. (Patron–Williams Graves, HB 2284)

Electric vehicle rebate program; creation and funding, report, sunset date. Adding §§ 67-1800 through 67-1806. (Patron–Reid, HB 1979)

Foreign market vehicles; titling and registration. Amending § 46.2-602. (Patron-Jenkins, HB 1887)

Habitual offenders; required that the Commissioner of DMV reinstate a person's privilege to drive a motor vehicle that was suspended or revoked solely on the basis that such person was determined to be or adjudicated a habitual offender, repeals remaining provisions of the Habitual Offender Act. Amending §§ 8.01-9, 8.01-407, 16.1-77, 16.1-305, 17.1-213, 19.2-389, 46.2-301, 46.2-301.1, 46.2-411, and 53.1-21; repealing §§ 46.2-355.1 through 46.2-363. (Patron–Stanley, SB 1122)

Identification privilege cards; authorizes DMV to issue, fee, confidentiality, penalties. Amending §§ 24.2-411.3, 24.2-643, 46.2-203.2, 46.2-216.1, 46.2-323.01, 46.2-323.1, 46.2-346, 46.2-600.1, and 58.1-3; adding § 46.2-345.3. (Patron–Guzman, HB 2138)

MOTOR VEHICLES (continued)

- Learner's permits; use of personal communication devices. Amending § 46.2-335. (Patron–Stuart, SB 1335)
- License plates, special; establishes a system for determining cost of plates for recipients of a military decoration. Amending §§ 46.2-742.1, 46.2-742.2, 46.2-745.1, and 46.2-745.2; adding §§ 46.2-725.3, 46.2-745.4, and 46.2-745.5. (Patron–McQuinn, HB 2069)
- License plates, special; issuance for supporters of Ducks Unlimited, fees. Amending § 46.2-749.7. (Patron–Stuart, SB 1229)
- License plates, special; range of bumper heights for any street rod bearing a street rod license plate, repeals the authorizations for issuance of certain plates no longer issued due to low plate sales, expired authorizations, etc. Amending § 46.2-1063; repealing §§ 46.2-746.6, 46.2-746.9, 46.2-746.12, 46.2-747, 46.2-748, 46.2-749.10, 46.2-749.69:1, and Section 1 of Chapter 776, 2010 Acts. (Patron–Marsden, SB 1136)
- License plates, special; removes fee for issuance to a member of the Virginia National Guard. Amending § 46.2-744. (Patron–Wyatt, HB 2261)
- License plates, special; removes the fee for the issuance of a license plate for retired members of the Virginia National Guard. Amending § 46.2-746.5. (Patron–Walker, HB 1796)
- License restrictions for minors; prohibition on use of handheld personal communications devices. Amending § 46.2-334.01. (Patron–Robinson, HB 1846)
- Local government; authority to reduce the speed limit in a business district or residence district. Amending § 46.2-1300. (Patron–Carr, HB 1903)
- Motor vehicle insurance; underinsured motor vehicle, uninsured motorist coverage, policies issued or renewed on or after January 1, 2022. Amending § 38.2-2206. (Patron–Obenshain, SB 1195)
- Motor vehicle liability insurance; increases coverage amounts. Amending §§ 46.2-419, 46.2-472, and 46.2-2057. (Patron–Surovell, SB 1182)
- Motor vehicle weight limits; vehicles powered primarily by electric battery power or fueled primarily by natural gas. Amending § 46.2-1129.2. (Patron–Reid, HB 1850)
- Motor Vehicles, Department of; limits the release of privileged information to government entities and law-enforcement agencies. Amending §§ 46.2-100, 46.2-208, 46.2-209, 46.2-209.1, and 46.2-328.3. (Patron–Tran, HB 2163)
- Online Virginia Driver's Manual course; training school. Amending § 46.2-325. (Patron-Carr, HB 1901)
- Parked cars; authorizes vehicles to be stopped or parked in an unoccupied Department of Transportation right-of-way adjacent to a highway in case of an emergency, etc. Amending § 46.2-889. (Patron–Miyares, HB 2146)
- Personal property tax; exemption for motor vehicle of a 100 percent disabled veteran. Adding § 58.1-3668. (Patron–Reeves, SB 1130)
- Removal of vehicles involved in accidents; lien of keeper of vehicles, limitation on removal and sale of abandoned vehicles, notice of intent to auction and sell vehicle, posting requirements. Amending §§ 46.2-644.01, 46.2-644.02, 46.2-644.03, 46.2-1200.2, 46.2-1202, 46.2-1202.1, 46.2-1203, 46.2-1209, and 46.2-1212.1; adding §§ 46.2-644.04, 46.2-1200.3, and 46.2-1202.2. (Patron–Suetterlein, SB 1160)
- Restricted permits to operate a motor vehicle; ignition interlock systems. Adding § 18.2-271.5. (Patron-Stuart, SB 1336)
- Special identification cards; application by guardian. Amending § 46.2-345. (Patron–Bell, HB 1961)
- Summons; promises to appear after issuance. Amending §§ 46.2-936 and 46.2-940. (Patron–Mason, SB 1329)
- Tangible personal property taxes; classification of certain motor vehicles, trailers, and semitrailers. Amending § 58.1-3506. (Patron–Walker, HB 1774)
- Test driving vehicles; residence districts, civil penalty. Adding § 46.2-1533.1. (Patron-Roem, HB 2318)
- Vehicle registration; expands eligibility for special communication needs indicator. Amending § 46.2-600.1. (Patron–Bell, HB 1960; Barker, SB 1470)
- Vehicle registration fees; imposes an additional fee to be deposited into Public Safety Trust Fund. Amending §§ 46.2-686 and 46.2-694; adding § 46.2-694.2. (Patron–Edwards, SB 1211)
- Vehicle's odometer; disclosure exemption. Amending § 46.2-629. (Patron-Cole, J.G., HB 2294)

MOTOR VEHICLES (continued)

Virginia Electric Vehicle Grant Fund and Program; created, report. Adding § 10.1-1322.5. (Patron-Keam, HB 2118)

Virginia Energy Plan; amends Plan to include an analysis of electric vehicle charging infrastructure. Amending §§ 67-102, 67-201, and 67-202. (Patron–Boysko, SB 1223)

MUNFORD, JOAN HARDIE

Munford, Joan Hardie; recording sorrow upon death. (Patron-Hurst, HJR 680)

NALOXONE

Naloxone or other opioid antagonist; certain employees of Department of Juvenile Justice authorized to administer. Amending § 54.1-3408. (Patron–Kory, HB 1894)

NARCOTICS AND DRUGS

Cannabis oil; processing and dispensing by pharmaceutical processors, certification for use of oil for treatment, report. Amending §§ 54.1-3408.3, 54.1-3442.5, 54.1-3442.6, and 54.1-3442.7. (Patron–Adams, D.M., HB 1988)

Drug Control Act; adds certain chemicals to Schedules I, II, IV, and V. Amending § 54.1-3446. (Patron-Newman, SB 1464)

Naloxone or other opioid antagonist; certain employees of Department of Juvenile Justice authorized to administer. Amending § 54.1-3408. (Patron–Kory, HB 1894)

Restricted licenses; authorizes DMV to issue restricted credentials to individuals with driver's license suspensions resulting from drug-related offenses. Amending § 18.2-271.1. (Patron–Edwards, SB 1213)

NASSAWADOX, TOWN OF

Nassawadox, Town of; amending charter, updates charter to reflect the town's shift of municipal elections from May to November. (Patron–Bloxom, HB 1749)

NATURAL GAS

Motor vehicle weight limits; vehicles powered primarily by electric battery power or fueled primarily by natural gas. Amending § 46.2-1129.2. (Patron–Reid, HB 1850)

Natural gas pipelines; Department of Environmental Quality to conduct inspections of the land-disturbing activities related to construction of any transmission pipeline, where substantial adverse impacts or likely adverse impacts are found on a repeated, etc., basis, the Department may issue a stop work instruction. Amending §§ 62.1-44.15:37.1 and 62.1-44.15:58.1. (Patron–Deeds, SB 1265)

NATURAL AND HISTORIC RESOURCES, SECRETARY OF

Natural Resources, Secretary of; changes name to the Secretary of Natural and Cultural Resources. Amending §§ 2.2-200, 2.2-205, 2.2-206, 2.2-215, 2.2-220.1, 2.2-220.2, 2.2-220.3, 2.2-1156, 2.2-1176, 2.2-2316, 2.2-2338, 2.2-2481, 2.2-2699.10, 2.2-2699.11, 2.2-2699.13, 10.1-603.25, 10.1-704, 10.1-1018, 10.1-1181.15, 10.1-1188, 10.1-1329, 10.1-1402.03, 10.1-1402.04, 10.1-1405, 10.1-2129, 10.1-2202.3, 15.2-2295.1, 28.2-207, 29.1-102, 29.1-573, 29.1-579, 30-377, 56-596.2, 58.1-344.3, 62.1-44.15:68, 62.1-44.34:25, 62.1-44.34:28, 62.1-44.117, 62.1-44.118, 62.1-44.119:1, 62.1-69.31, 62.1-69.32, 62.1-69.33, 62.1-69.41, 62.1-69.52, and 62.1-195.1. (Patron—Plum, HB 1836)

NATURAL RESOURCES, SECRETARY OF

Carbon Sequestration Task Force; established by Secretary of Natural Resources, jointly with the Secretary of Agriculture and Consumer Services, to convene a task force, report. (Patron–Lewis, SB 1374)

Gold; Secretary of Natural Resources, et al., shall establish a work group to study mining and processing, issuance of certain mining permits. (Patron–Guzman, HB 2213)

Natural Resources, Secretary of; changes name to the Secretary of Natural and Cultural Resources. Amending §§ 2.2-200, 2.2-205, 2.2-206, 2.2-215, 2.2-220.1, 2.2-220.2, 2.2-220.3, 2.2-1156, 2.2-1176, 2.2-2316, 2.2-2338, 2.2-2481, 2.2-2699.10, 2.2-2699.11, 2.2-2699.13, 10.1-603.25, 10.1-704, 10.1-1018, 10.1-1181.15, 10.1-1188, 10.1-1329, 10.1-1402.03, 10.1-1402.04, 10.1-1405, 10.1-2129, 10.1-2202.3, 15.2-2295.1, 28.2-207, 29.1-102, 29.1-573, 29.1-579,

NATURAL RESOURCES, SECRETARY OF (continued)

30-377, 56-596.2, 58.1-344.3, 62.1-44.15:68, 62.1-44.34:25, 62.1-44.34:28, 62.1-44.117, 62.1-44.118, 62.1-44.119:1, 62.1-69.31, 62.1-69.32, 62.1-69.33, 62.1-69.41, 62.1-69.52, and 62.1-195.1. (Patron–Plum, HB 1836)

NEW RIVER VALLEY PUBLIC HEALTH TASK FORCE

New River Valley Public Health Task Force; commending. (Patron-Hurst, HJR 677)

NEWMAN, STEPHEN D.

tatement	1 vote:	
S.B.	57	443

NEWPORT NEWS PUBLIC LIBRARY

Newport News Public Library; commending. (Patron-Mullin, HJR 749)

NEWSOME HOUSE MUSEUM AND CULTURAL CENTER

Newsome House Museum and Cultural Center; commemorating its 30th anniversary. (Patron-Price, HJR 787)

NORFOLK, CITY OF

Norfolk, City of; amending charter, general updates. (Patron-Spruill, SB 1128)

NORMENT, THOMAS K., JR.

Participated remotely	
Statement	
Statements on votes:	
S.B. 1366	
S.B. 1469	
NORTHAM, RALPH S., GOVERNOR OF VIRGINIA	
Address (Joint Assembly), printed as S.D. 1	

NORTHAMPTON COUNTY

Hampton Roads Sanitation District; changes to the enabling act related to the addition of Northampton and Accomack Counties. Amending Chapter 66, 1960 Acts. (Patron–Bloxom, HB 2257)

NORTHERN VIRGINIA

Northern Virginia Therapeutic Riding Program; commemorating its 40th anniversary. (Patron–Helmer, HJR 764)

NOTARIES AND OUT-OF-STATE COMMISSIONERS

Electronic notarial certificate; clerk's office to record a paper copy of an electronic document, provided that such copy otherwise meets the requirements for recordation and is certified to be a true and correct copy. Amending §§ 17.1-223, 47.1-2, 47.1-16, and 55.1-606. (Patron–Simon, HB 2064)

NURSES

Clinical nurse specialist; licensure of nurse practitioners as specialists by the Boards of Medicine and Nursing and a practitioner licensed as a specialist shall practice pursuant to a practice agreement between the specialist and a licensed physician, etc., prescriptive authority, repeals provision relating to registered nurse's licensure by endorsement. Amending §§ 54.1-2900, 54.1-2901, 54.1-2957, 54.1-2957.01, and 54.1-3000; repealing § 54.1-3018.1. (Patron–Adams, D.M., HB 1747)

Home care organizations; Board of Health to include in regulations governing organizations a provision for supervision of home care attendants providing personal care services by a licensed nurse through use of interactive audio or video technology. Amending § 32.1-162.12. (Patron–Head, HB 1831)

Nurse Loan Repayment Program; expands eligibility for Program to include certified nurse aids. Amending § 32.1-122.6:04. (Patron-Kiggans, SB 1147)

NURSES (continued)

Nurse practitioners; reduces the number of years of full-time clinical experience a practitioner must have to be eligible to practice without a written or electronic practice agreement, sunset provision. Amending § 54.1-2957. (Patron–Adams, D.M., HB 1737)

School nurses; no individual who provides nursing services in a public elementary or secondary school shall use title unless individual is a registered nurse. Amending § 22.1-274. (Patron–Adams, D.M., HB 1736)

NURSING HOMES

Hospitals, nursing homes, and certified nursing facilities; regulations, policies to ensure the permissible access to and use of an intelligent personal assistant provided by a patient while receiving inpatient services. Amending § 32.1-127. (Patron–Adams, L.R., HB 2154)

Hospitals, nursing homes, etc.; visits by clergy, priest, etc., during a declared public health emergency. Amending §§ 32.1-127, 32.1-162.5, and 63.2-1732. (Patron–Kiggans, SB 1356)

OBENSHAIN, MARK D.

OCCUPATIONAL THERAPY

Occupational therapists; licensure, authorizes Virginia to become a signatory to the Occupational Therapy Interjurisdictional Licensure Compact. Adding § 54.1-2956.7:1. (Patron-Hashmi, SB 1189)

OPIOIDS

Naloxone or other opioid antagonist; certain employees of Department of Juvenile Justice authorized to administer. Amending § 54.1-3408. (Patron–Kory, HB 1894)

Opioid Abatement Authority; established, Fund created, report, membership. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Barker, SB 1469)

Opioid Abatement Authority; established, Opioid Abatement Fund created, investment of assets of Fund, report. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Herring, HB 2322)

ORDINANCES

Continuity of government; extends to 12 months the period of time after an enemy attack or other disaster that a locality may, by ordinance, provide. Amending § 15.2-1413. (Patron-Barker, SB 1208)

Historic sites; urban county executive form of government (Fairfax County), provisions in its preservation ordinance, etc. Amending § 15.2-2306. (Patron–Surovell, SB 1457)

Local green banks; authorizes a locality, by ordinance, to establish a bank to promote the investment in clean energy technologies in its locality, etc. Adding § 15.2-958.3:1. (Patron-Kory, HB 1919)

Vacant buildings; removes requirement that a building meet the definition of "derelict building" for Town of Clifton Forge, Town of Pulaski, Town of Timberville, and any city to require, by ordinance, the owner or owners of buildings to register such buildings on an annual basis. Amending § 15.2-1127. (Patron–Locke, SB 1285)

ORTIZ, PATTY

Ortiz, Patty; commending. (Patron-LaRock, HJR 752)

OXRIEDER, JULIA

Suerdieck, Rebecca and Julia Oxrieder; commending. (Patron-Mullin, HJR 765)

PARHAM DOCTORS' HOSPITAL

Parham Doctors' Hospital; commending. (Patron-VanValkenburg, HJR 655)

PAROLE AND PROBATION

Parole; notice and certification, monthly reports, conditional release, discretionary early consideration. Amending §§ 53.1-136 and 53.1-155. (Patron–Scott, HB 2167)

Status offenders; willful and material violation of court order or terms of probation, notice, orders of disposition for violation. Amending §§ 16.1-278.5, 16.1-291, and 16.1-292. (Patron–Scott, HB 2056)

PARTICIPATED REMOTELY

Chase, Amanda F
Lewis, Lynwood W., Jr
Norment, Thomas K., Jr
Vogel, Jill Holtzman

PARTNERSHIPS

State Corporation Commission; business entities filings, amends various provisions of Va. Stock Corporation Act. Amending §§ 13.1-609, 13.1-610, 13.1-615, 13.1-615.1, 13.1-625, 13.1-628, 13.1-630, 13.1-636, 13.1-639, 13.1-658, 13.1-661, 13.1-710, 13.1-711, 13.1-716, 13.1-718, 13.1-721.1, 13.1-722.6, 13.1-722.12:1, 13.1-759, 13.1-765, 13.1-775.1, 13.1-803, 13.1-806, 13.1-807, 13.1-809, 13.1-815, 13.1-815.1, 13.1-816, 13.1-829, 13.1-830, 13.1-831, 13.1-835, 13.1-894, 13.1-897.1, 13.1-898.7, 13.1-921, 13.1-927, 13.1-936.1, 13.1-944.7, 13.1-1002, 13.1-1004, 13.1-1005, 13.1-1012, 13.1-1017, 13.1-1052, 13.1-1054, 13.1-1062, 13.1-1065, 13.1-1073.1, 13.1-1074, 13.1-1075, 13.1-1080, 13.1-1087, 13.1-1096, 13.1-1099.14, 13.1-1099.26, 13.1-1201, 13.1-1203, 13.1-1212, 13.1-1214, 13.1-1222, 13.1-1242, 13.1-1252, 13.1-1255, 13.1-1264, 13.1-1265, 13.1-1271, 13.1-1277, 15.2-5112, 15.2-5431.9, 50-73.1, 50-73.2, 50-73.6, 50-73.17, 50-73.54, 50-73.67, 50-73.70, 50-73.83, and 50-73.135; adding §§ 13.1-898.1:1, 13.1-1263.1, 15.2-5431.8:1, 15.2-5431.9:1, 15.2-5431.35:1, and 50-73.48:5; repealing §§ 13.1-941.01 through 13.1-944. (Patron–Keam, HB 2121)

PATRICK COUNTY

Bob White Covered Bridge; Department of Transportation to work with the governing body of Patrick County and community groups interested in constructing a replica of Bridge. (Patron-Poindexter, HB 2024)

PEANUTS

Peanuts; extends sunset date of excise tax on all peanuts grown in Virginia. Amending § 3.2-1905. (Patron–Brewer, HB 1751; Lucas, SB 1411)

PEBLER, STEVEN

Pebler, Steven; commending. (Patron-LaRock, HJR 754)

PENSIONS, BENEFITS, AND RETIREMENT

Virginia Retirement System; amendments to reflect recent changes to federal law. Amending §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308. (Patron–Mundon King, HB 2181; Newman, SB 1251)

PERSONAL PROPERTY AND PERSONAL PROPERTY TAX

Personal property tax; exemption for motor vehicle of a 100 percent disabled veteran. Adding § 58.1-3668. (Patron–Reeves, SB 1130)

Tangible personal property taxes; classification of certain motor vehicles, trailers, and semitrailers. Amending § 58.1-3506. (Patron–Walker, HB 1774)

PERSONS WITH DISABILITIES

Absentee voting; accessibility for voters with a visual impairment or print disability. Amending § 24.2-704; adding § 24.2-103.2. (Patron–Reeves, SB 1331)

Aging services; Department for Aging and Rehabilitative Services shall prioritize providing services to older persons with greatest economic or social need, definition. Amending §§ 51.5-134 and 51.5-135. (Patron–Adams, D.M., HB 1805)

Aging services; Department for Aging and Rehabilitative Services shall use available resources to provide services to older persons with the greatest economic and social needs, definitions. Amending §§ 51.5-134 and 51.5-135. (Patron–Barker, SB 1366)

Alternative application for employment for persons with a disability; Department of Human Resource Management to create a process. Amending § 2.2-2901.1; adding § 2.2-1212. (Patron–Guzman, HB 2140)

Licensed private schools for students with disabilities; accreditation. Amending § 22.1-323. (Patron-Kory, HB 2238)

Medical care facilities; facility shall establish protocols, definitions, designated support persons for persons with disabilities. Adding § 32.1-137.08. (Patron–Tran, HB 2162)

PERSONS WITH DISABILITIES (continued)

Personal property tax; exemption for motor vehicle of a 100 percent disabled veteran. Adding § 58.1-3668. (Patron–Reeves, SB 1130)

Special education; Department of Education to provide training and guidance documents to local school divisions on the development of Individualized Education Programs (IEPs) for children with disabilities, etc. Adding § 22.1-214.4. (Patron–Carr, HB 2299)

Students with disabilities; Department of Education to update its special education and related services, etc. (Patron–Mundon King, HB 2316)

Virginia Fair Housing Law; reasonable accommodations, disability-related requests for parking. Amending § 36-96.3:2. (Patron–Carr, HB 1971)

Virginia Human Rights Acts; adds discrimination on the basis of disability as an unlawful employment practice, reasonable accommodations for persons with disabilities. Amending §§ 2.2-3902, 2.2-3905, and 51.5-41; adding § 2.2-3905.1. (Patron–Sickles, HB 1848)

PESTS AND PESTICIDES

Neonicotinoid pesticides; application in residentially zoned outdoor areas, civil penalty. Adding § 3.2-3929.1. (Patron–Krizek, HB 2030)

PETS, PET DEALERS, AND SUPPLIES

Pet shops, dealers, and dog breeders; employees convicted of animal abuse, penalty. Amending §§ 3.2-6511.1 and 3.2-6511.2. (Patron–Stanley, SB 1412)

PHARMACIES

Pharmacies; freedom of choice by covered individual. Amending §§ 38.2-3407.7, 38.2-4209.1, and 38.2-4312.1. (Patron–Hodges, HB 2219)

PHARMACISTS

Pharmacists; initiation of treatment with and dispensing and administering of drugs and devices. Amending §§ 54.1-3300 and 54.1-3303.1. (Patron–Rasoul, HB 2079)

PHYSICAL THERAPISTS

Physical therapy; extends time allowed for a therapist to evaluate and treat patients. Amending § 54.1-3482. (Patron–Hashmi, SB 1187)

PHYSICIANS AND SURGEONS

Clinical nurse specialist; licensure of nurse practitioners as specialists by the Boards of Medicine and Nursing and a practitioner licensed as a specialist shall practice pursuant to a practice agreement between the specialist and a licensed physician, etc., prescriptive authority, repeals provision relating to registered nurse's licensure by endorsement. Amending §§ 54.1-2900, 54.1-2901, 54.1-2957, 54.1-2957.01, and 54.1-3000; repealing § 54.1-3018.1. (Patron–Adams, D.M., HB 1747)

Licensed certified midwives; clarifies definition, licensure, practice shall be in consultation with a licensed physician with a practice agreement, report. Amending § 54.1-2900; adding § 54.1-2957.04. (Patron–Gooditis, HB 1953; Lucas, SB 1320)

Physician assistant; eliminates requirement that an assistant enter into a practice agreement with one or more patient care team physicians, etc. Amending §§ 54.1-2902, 54.1-2950.1, 54.1-2951.1, 54.1-2951.2, 54.1-2952, 54.1-2952.1, 54.1-2953, and 54.1-2972; adding § 54.1-2951.4. (Patron–Rasoul, HB 2039)

PLASTER, GROVER HAROLD

Plaster, Grover Harold; recording sorrow upon death. (Patron-Marshall, HJR 773)

PLEASANT VIEW, INC.

Pleasant View, Inc.; commemorating its 50th anniversary. (Patron–Wilt, HJR 747)

PLEDGE OF ALLEGIANCE

Adams, Michael, Director of Human Resources	. 151, 167, 232, 247
Garrett, John McE	266
Griffin, Lindley, Deputy Director of Committee Operations and Calendar Clerk.	504
Hatfield, Nathan, Assistant Clerk-LINCS	, 315, 349, 373, 433

PLEDGE OF ALLEGIANCE (continued)

ADGE OF ALLEGIANCE (continued)		
Perkinson, Tara H., Chief Deputy Clerk	85, 94,	105, 117, 135
Schaar, Susan Clarke, Clerk of the Senate		2, 74, 80
Sloan, Corinne, Assistant Journal Clerk		499

PLOTNICK, VIRGINIA LEONARD

Plotnick, Virginia Leonard; recording sorrow upon death. (Patron–Hudson, HJR 710)

POLICE

- Facial recognition technology; definition, authorization of use by local law-enforcement agencies and public institutions of higher education. Adding §§ 15.2-1723.2 and 23.1-815.1. (Patron–Aird, HB 2031)
- Law-enforcement agencies; creates a special nonreverting fund to be known as the Body-Worn Camera System Fund to assist state or local agencies with costs of purchasing, etc., body-worn camera systems. Adding § 9.1-116.7. (Patron–Reeves, SB 1119)
- Motor Vehicles, Department of; limits the release of privileged information to government entities and law-enforcement agencies. Amending §§ 46.2-100, 46.2-208, 46.2-209, 46.2-209.1, and 46.2-328.3. (Patron-Tran, HB 2163)
- Police and court records; expungement and sealing of records, Expungement Fee Fund created, protection of public record information, penalties, report, effective clause. Amending §§ 9.1-128, 9.1-134, 17.1-502, 19.2-392.1, 19.2-392.2, 19.2-392.3, and 19.2-392.4; adding §§ 1-229.1, 17.1-205.1, 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, 19.2-392.4:1, 19.2-392.5, 19.2-392.6, and 19.2-392.7. (Patron-Surovell, SB 1339)
- Wrongful death statute of limitations; criminal investigations by law enforcement. Amending § 8.01-244. (Patron–Convirs-Fowler, HB 2073)

POLICE, STATE

- Government Data Collection and Dissemination Practices Act; license plate readers shall not be used to collect or maintain personal information, etc., definitions. Amending §§ 2.2-3800, 2.2-3801, 2.2-3802, and 52-48. (Patron–Petersen, SB 1198)
- Law-enforcement officer; duty to render aid, duty to report wrongdoing by another officer. Amending § 52-30.1; adding §§ 15.2-1704.1 and 15.2-1704.2. (Patron–Levine, HB 1948)
- Substantial Risk Order Registry; Department of State Police shall keep and maintain a computerized Registry, etc. Amending § 19.2-387.3. (Patron–Simonds, HB 2258)
- Virginia Missing Child with Autism Alert Program; renames the Virginia Missing Person with Autism Alert Program. Amending §§ 52-34.13, 52-34.14, and 52-34.15. (Patron-Hayes, HB 2216)

POLLING PLACES

- Elections; preservation of order at the polls, powers of officers of election. Amending § 24.2-606. (Patron–Spruill, SB 1111)
- Polling places; prohibited activities, unlawful possession of a firearm, penalty. Amending §§ 24.2-604, 24.2-671, and 24.2-802.1. (Patron–Levine, HB 2081)

POWHATAN COUNTY

Economic development authorities; size of board in Powhatan County, reduces quorum requirement. Amending § 15.2-4904. (Patron-Ware, HB 2287)

PRAYERS OFFERED

MENS OFFERED
Barker, George L
Deeds, R. Creigh
DeSteph, William R., Jr
Dunnavant, Siobhan S
Edwards, John S
Favola, Barbara A
Hanger, Emmett W., Jr
Hatfield, Nathan, Assistant Clerk-LINCS.
Kiggans, Jennifer A
Lewis, Lynwood W., Jr. 247
Locke, Mamie E

PRAYERS OFFERED (continued)

Mason, I. Montgomery	49
McClellan, Jennifer L	05
McDougle, Ryan T	17
Moment of silent prayer	04
Reeves, Bryce E	51
Ruff, Frank M., Jr	32
Spruill, Lionell, Sr	33
Stuart, Richard H.	94

PREGNANCY

Inmates; Board of Local and Regional Jails to review services provided during pregnancy, etc., report. (Patron-Favola, SB 1300)

PRESCRIPTION MEDICINES

Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization for drug by carrier. Amending § 38.2-3407.15:2. (Patron–McPike, SB 1269)

Health insurance; authorization of drug prescribed for the treatment of a mental disorder, prior authorization issued by carrier for drug, certain provisions shall not apply to the state employee health insurance plan, etc. Amending § 38.2-3407.15:2. (Patron–Heretick, HB 2008)

Health insurance; cost-sharing payments for covered tier one or tier two prescription asthma inhalers. Adding § 38.2-3407.15:6. (Patron–Askew, HB 1822)

Prescription drugs; price transparency, definitions. Amending § 2.2-3705.6; adding §§ 32.1-23.3, 38.2-3407.15:6, 38.2-3407.22, 54.1-3436.1, and 54.1-3442.02. (Patron–Sickles, HB 2007)

Public elementary and secondary schools; possession and administration of undesignated stock albuterol inhalers and valved holding chambers, Department of Health, in conjunction with the Department of Education, shall develop and implement policies. Amending §§ 8.01-225, 22.1-274.2, and 54.1-3408. (Patron–McQuinn, HB 2019)

PRIMARIES

Primary election; changes date of election held in June from second Tuesday in June to third Tuesday in June, also changes candidate filing deadlines to reflect change of date. Amending §§ 24.2-311, 24.2-503, 24.2-507, 24.2-510, 24.2-515, and 24.2-515.1. (Patron–Kiggans, SB 1148)

PRINCE WILLIAM COUNTY

Albemarle and Prince William Counties; counties that have adopted the county executive form of government may carry over unspent funds from year to year for multiyear capital projects and outstanding grants. Amending §§ 15.2-520 and 15.2-2506. (Patron–Reeves, SB 1120)

PRISONERS

Parole; Department of Corrections shall release a prisoner no sooner than 21 business days after the date of notification by the Virginia Parole Board to the appropriate attorney for the Commonwealth of the decision to grant parole, etc. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1104)

Parole Board; Board, within seven days of making any decision regarding the parole of a prisoner, to provide written or electronic notice of such decision to the victim of crime. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1125)

PRISONS AND OTHER METHODS OF CORRECTION

Behavioral health; assessments in local correctional facilities, report. Amending § 53.1-68. (Patron-Coyner, HB 1874)

Correctional facilities; restrictions on use of isolated confinement, effective clause. Adding §§ 53.1-39.2 and 66-20.1. (Patron–Morrissey, SB 1301)

Earned sentence credits; rate at which sentence rates may be earned, prerequisites. Amending § 53.1-202.3. (Patron-Campbell, J.L., HB 2010)

Geriatric or terminally ill inmates; conditional release, notify by certified or verifiable electronic means to attorney for the Commonwealth. Amending § 53.1-136. (Patron–Norment, SB 1397)

Local and Regional Jails, State Board of; membership, powers and duties, effective clause. Amending §§ 53.1-2, 53.1-4, and 53.1-5. (Patron–Marsden, SB 1363)

PRISONS AND OTHER METHODS OF CORRECTION (continued)

- Parole; Department of Corrections shall release a prisoner no sooner than 21 business days after the date of notification by the Virginia Parole Board to the appropriate attorney for the Commonwealth of the decision to grant parole, etc. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1104)
- Parole; notice and certification, monthly reports, conditional release, discretionary early consideration. Amending §§ 53.1-136 and 53.1-155. (Patron–Scott, HB 2167)
- Parole Board; Board, within seven days of making any decision regarding the parole of a prisoner, to provide written or electronic notice of such decision to the victim of crime. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1125)

PRIVILEGES AND ELECTIONS, COMMITTEE ON

PROFESSIONS AND OCCUPATIONS

INDEX

- Actions against real estate appraisers or appraisal management companies; statute of limitations on actions for damages or other relief. Adding §§ 54.1-2019.1 and 54.1-2024. (Patron–Dunnavant, SB 1324)
- Cannabis oil; processing and dispensing by pharmaceutical processors, certification for use of oil for treatment, report. Amending §§ 54.1-3408.3, 54.1-3442.5, 54.1-3442.6, and 54.1-3442.7. (Patron–Adams, D.M., HB 1988)
- Career fatigue and wellness in certain health care providers; programs to address, civil immunity. Amending §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909. (Patron–Hope, HB 1913; Barker, SB 1205)
- Certified nurse midwives; eliminates the requirement that midwives practice pursuant to a practice agreement and shall practice in accordance with regulations of the Boards of Medicine and Nursing. Amending §§ 54.1-2957 and 54.1-2957.01. (Patron–Adams, D.M., HB 1817)
- Clinical nurse specialist; licensure of nurse practitioners as specialists by the Boards of Medicine and Nursing and a practitioner licensed as a specialist shall practice pursuant to a practice agreement between the specialist and a licensed physician, etc., prescriptive authority, repeals provision relating to registered nurse's licensure by endorsement. Amending §§ 54.1-2900, 54.1-2901, 54.1-2957, 54.1-2957.01, and 54.1-3000; repealing § 54.1-3018.1. (Patron–Adams, D.M., HB 1747)
- Disposition of the remains of a decedent; persons to make arrangements for funeral and disposition of remains, person may designate in a signed and notarized writing, repeals provisions relating to when next of kin disagree and absence of next of kin. Amending §§ 32.1-309.1, 54.1-2800, 54.1-2807, 54.1-2825, and 57-27.3; adding §§ 54.1-2825.1, 57-27.4, 57-27.5, and 57-27.6; repealing §§ 54.1-2807.01 and 54.1-2807.02. (Patron–Sickles, HB 2005)
- Drug Control Act; adds certain chemicals to Schedules I, II, IV, and V. Amending § 54.1-3446. (Patron-Newman, SB 1464)
- Elevator mechanic or accessibility mechanic, certain; exemption from certification. Amending § 54.1-1141. (Patron–Jones, HB 2202)
- Genetic counseling; repeals conscience clause. Repealing § 54.1-2957.21. (Patron–Ebbin, SB 1178)
- Industrial hemp; increases the maximum tetrahydrocannabinol (THC) concentration, expands definition of "hemp product", registration renewal fee not to exceed \$50, etc. Amending §§ 3.2-4112, 3.2-4113, 3.2-4114.2, 3.2-4115, 3.2-4116, 3.2-4119, 18.2-247, 18.2-251.1:3, 54.1-3401, and 54.1-3446. (Patron–Peake, SB 1115)
- Lawyers; client accounts, repeals the provision prohibiting the Supreme Court of Virginia from adopting a disciplinary rule requiring that lawyers deposit client funds in an interest-bearing account. Amending § 54.1-3916; repealing § 54.1-3915.1. (Patron–Sullivan, HB 1853)
- Licensed certified midwives; clarifies definition, licensure, practice shall be in consultation with a licensed physician with a practice agreement, report. Amending § 54.1-2900; adding § 54.1-2957.04. (Patron–Gooditis, HB 1953; Lucas, SB 1320)
- Naloxone or other opioid antagonist; certain employees of Department of Juvenile Justice authorized to administer. Amending § 54.1-3408. (Patron–Kory, HB 1894)
- Nurse practitioners; reduces the number of years of full-time clinical experience a practitioner must have to be eligible to practice without a written or electronic practice agreement, sunset provision. Amending § 54.1-2957. (Patron–Adams, D.M., HB 1737)

PROFESSIONS AND OCCUPATIONS (continued)

- Occupational therapists; licensure, authorizes Virginia to become a signatory to the Occupational Therapy Interjurisdictional Licensure Compact. Adding § 54.1-2956.7:1. (Patron–Hashmi, SB 1189)
- Pharmaceutical processors; permits processors to produce and distribute cannabis products, dispensing botanical cannabis to a minor, certification. Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8. (Patron–Lucas, SB 1333)
- Pharmaceutical processors; permits processors to produce and distribute cannabis products, testing standards for botanical cannabis and products. Amending §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8. (Patron–Hayes, HB 2218)
- Pharmacists; initiation of treatment with and dispensing and administering of drugs and devices. Amending §§ 54.1-3300 and 54.1-3303.1. (Patron–Rasoul, HB 2079)
- Physical therapy; extends time allowed for a therapist to evaluate and treat patients. Amending § 54.1-3482. (Patron–Hashmi, SB 1187)
- Physician assistant; eliminates requirement that an assistant enter into a practice agreement with one or more patient care team physicians, etc. Amending §§ 54.1-2902, 54.1-2950.1, 54.1-2951.1, 54.1-2951.2, 54.1-2952, 54.1-2952.1, 54.1-2953, and 54.1-2972; adding § 54.1-2951.4. (Patron–Rasoul, HB 2039)
- Prescription drugs; price transparency, definitions. Amending § 2.2-3705.6; adding §§ 32.1-23.3, 38.2-3407.15:6, 38.2-3407.22, 54.1-3436.1, and 54.1-3442.02. (Patron–Sickles, HB 2007)
- Public elementary and secondary schools; possession and administration of undesignated stock albuterol inhalers and valved holding chambers, Department of Health, in conjunction with the Department of Education, shall develop and implement policies. Amending §§ 8.01-225, 22.1-274.2, and 54.1-3408. (Patron–McQuinn, HB 2019)
- Surgical technologist; certification, use of title. Amending § 54.1-2956.12. (Patron-Hayes, HB 2220)
- Virginia Residential Landlord and Tenant Act; responsibilities of real estate brokers, foreclosure of single-family residential dwelling units. Amending §§ 54.1-2108.1 and 55.1-1237. (Patron–Simon, HB 2229)
- Virginia State Bar examination; foreign applicants. Amending § 54.1-3926. (Patron–Petersen, SB 1234)

PROPERTY AND CONVEYANCES

- Chamberlin Hotel at Fort Monroe; reverts certain property to the Commonwealth. Repealing Section 1 of Chapter 809, 1998 Acts. (Patron–Mugler, HB 2009)
- Deeds of trust; amendment to loan document, statement of interest rate of a refinanced mortgage. Amending § 55.1-319; adding § 55.1-318.1. (Patron–Heretick, HB 1882)
- Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, cases of a deed of trust conveying owner-occupied residential real estate, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–Torian, HB 2175)
- Homeowners and tenants of manufactured home parks; housing protections, foreclosures, enforcement of lien, etc. Amending §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303; adding § 15.2-2223.5. (Patron–McClellan, SB 1327)
- Judgments; limitations on enforcement, judgment liens, settlement agents, effective date. Amending §§ 8.01-251, 8.01-458, and 55.1-339. (Patron–Coyner, HB 2099)
- Property owners' associations and unit owners' associations; rulemaking authority concerning smoking. Amending §§ 55.1-1819 and 55.1-1959; adding §§ 55.1-1819.1 and 55.1-1960.1. (Patron–Keam, HB 1842)
- Property owners' associations, boards of directors, unit owners' associations, etc.; meetings to be held entirely or partially by electronic means, provided guidelines have been adopted. Amending §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953. (Patron–Bulova, HB 1816; Dunnavant, SB 1183)
- Real estate settlement agents; SCC may share information collected from an agent or agency regarding any errors, etc., with any party to the real estate transaction in connection with the actions arising out of a settlement. Amending § 55.1-1004. (Patron–Spruill, SB 1110)

PROPERTY AND CONVEYANCES (continued)

- Real property; required disclosures for buyer to exercise due diligence, flood risk report, effective date. Amending § 55.1-703; adding § 55.1-708.2. (Patron–Lewis, SB 1389)
- Southwestern Virginia Mental Health Institute; Governor to lease a portion of property to Smyth County for a term of three years, responsibility of County, corrects tax map references. Amending Chapter 678, 2019 Acts. (Patron–O'Quinn, HB 2098; Pillion, SB 1429)
- Tazewell County; authorizes a quitclaim and release of interest and conveyance of an easement by Board of Wildlife Resources. (Patron–Wampler, HB 2252; Pillion, SB 1400)
- Virginia Residential Landlord and Tenant Act; access to dwelling unit during certain declared states of emergency, nonemergency repairs, employees and agents sent by landlord for maintenance or inspection are to wear all appropriate personal protective equipment as required by state law. Amending § 55.1-1229. (Patron–Carr, HB 1981)
- Virginia Residential Landlord and Tenant Act; landlord charges for security deposits, insurance premiums for damage insurance, etc., clerk of SCC shall provide a downloadable form for the filing of a resident agent appointment, etc. Amending §§ 17.1-275, 55.1-1200, 55.1-1204, 55.1-1206, 55.1-1208, 55.1-1211, 55.1-1226, 64.2-2008, and 64.2-2012. (Patron–McQuinn, HB 2249)
- Virginia Residential Landlord and Tenant Act; noncompliance with rental agreement, etc., prohibition on using negative credit information that arose during a closure of the United States Government against certain applications for tenancy, penalty. Amending § 55.1-1245. (Patron–Helmer, HB 1908)
- Virginia Residential Landlord and Tenant Act; responsibilities of real estate brokers, foreclosure of single-family residential dwelling units. Amending §§ 54.1-2108.1 and 55.1-1237. (Patron–Simon, HB 2229)
- Virginia Residential Landlord and Tenant Act; sample termination notice, landlord's acceptance of rent with reservation, tenant's right of redemption. Amending §§ 36-139 and 55.1-1250. (Patron–Price, HB 2014)
- Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion from dwelling unit, interruption of services, or actions taken to make premises unsafe, repeals provision relating to tenant's remedies for landlord's unlawful ouster. Adding § 55.1-1243.1; repealing § 55.1-1243. (Patron–Hudson, HB 1900; Ebbin, SB 1215)
- Virginia Residential Property Disclosure Act; required disclosures for buyer to beware, mold assessment obtained by purchaser. Amending § 55.1-703. (Patron–Askew, HB 1824)
- Virginia Residential Property Disclosure Act; required disclosures for buyer to exercise due diligence, flood risk report, effective date and clause. Amending § 55.1-703; adding § 55.1-708.2. (Patron–Convirs-Fowler, HB 2320)

PROPERTY, GROUNDS, AND BUILDINGS, STATE-OWNED

- Capitol Square Preservation Council; powers and duties, review and approval of plans for changes to artifacts contained within the Capitol Building. Amending § 30-194. (Patron–Norment, SB 1172)
- Firearm; carrying within Capitol Square and the surrounding area, into building owned or leased by the Commonwealth, etc., penalty. Adding § 18.2-283.2. (Patron–Levine, HB 2295)
- State parks; Department of Conservation and Recreation to develop recommendations for funding, report. (Patron–Orrock, HB 1804)
- Weapons; possessing or transporting within Capitol Square or into a building owned or leased by the Commonwealth, provisions shall not apply to certain law-enforcement officers, etc., penalty. Adding § 18.2-283.2. (Patron–Ebbin, SB 1381)

PROPERTY OWNERS

- Onsite Sewage Indemnification Fund; authorizes the State Board of Health to use Fund to provide grants and loans to property owners with income at or below 200 percent of the federal poverty guidelines to repair failing onsite sewage systems or install onsite sewage systems on properties that lack adequate sewage disposal. Amending §§ 32.1-164 and 32.1-164.1:01; adding §§ 62.1-223.1, 62.1-223.2, and 62.1-223.3. (Patron–Hashmi, SB 1396)
- Property owners' associations and unit owners' associations; rulemaking authority concerning smoking. Amending §§ 55.1-1819 and 55.1-1959; adding §§ 55.1-1819.1 and 55.1-1960.1. (Patron–Keam, HB 1842)

PROPERTY OWNERS (continued)

Property owners' associations, boards of directors, unit owners' associations, etc.; meetings to be held entirely or partially by electronic means, provided guidelines have been adopted. Amending §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953. (Patron–Bulova, HB 1816; Dunnavant, SB 1183)

PROSTITUTION

Prostitution; reorganizes the statute penalizing into two distinct sections. Amending §§ 8.01-42.4, 9.1-116.5, 9.1-902, 16.1-69.48:6, 16.1-69.55, 17.1-275.13, 17.1-805, 18.2-46.1, 18.2-346, 18.2-346.1, 18.2-350, 18.2-357.1, 18.2-513, 19.2-10.2, 19.2-215.1, 19.2-268.3, 19.2-386.16, 19.2-386.35, 19.2-392.02, 32.1-58, 37.2-314, 37.2-416, and 37.2-506; adding § 18.2-346.01. (Patron–Mundon King, HB 2169)

PROTECTIVE ORDERS

Protective orders; violations of preliminary child protective order, violation involves an act that endangers the child's normal development, changes punishment, etc. Amending §§ 16.1-253 and 16.1-253.2. (Patron-Campbell, J.L., HB 2012; Stanley, SB 1415)

PUBLIC BUILDINGS, FACILITIES, AND PROPERTY

Constitutional amendment; authority to grant perpetual easements to units of government (first reference). Amending Section 9 of Article VII. (Patron–Mason, SJR 289)

PUBLIC SCHOOLS

- Comprehensive review of computer science standards, courses, and pathways in public schools; Department of Education to perform, report. (Patron–Simonds, HB 1885)
- Constitutional amendment; equitable educational opportunities in all public schools in the Commonwealth (first reference). Amending Sections 1 and 2 of Article VIII. (Patron–Stanley, SJR 275)
- Consumer Protection Act; prohibited practices, certain advertising related to any public or private school quality. Amending § 59.1-200. (Patron–Samirah, HB 2003)
- COVID-19; Joint Legislative Audit and Review Commission to study the impact on Virginia's public schools, students, and school employees, meetings shall be completed by November 30, 2022. (Patron–Guy, HJR 549; Lucas, SJR 308)
- Public School Assistance Fund and Program; created. Adding § 22.1-141.3. (Patron-Stanley, SB 1106)
- Public schools; lock-down drills, annual requirement. Amending § 22.1-137.2. (Patron–Murphy, HB 1998)
- Public schools; seizure management and action plans, biennial training, effective date. Amending § 8.01-225; adding § 22.1-274.6. (Patron–DeSteph, SB 1322)
- Public schools; severe weather conditions and other emergency situations, unscheduled remote learning days, school provides instruction and student services, etc. Amending § 22.1-98. (Patron–McNamara, HB 1790; Suetterlein, SB 1132)
- Public schools, child day programs, and certain other programs; carbon monoxide detectors required. Adding §§ 22.1-138.2, 22.1-289.058, and 63.2-1705.2. (Patron–Askew, HB 1823)
- School nurses; no individual who provides nursing services in a public elementary or secondary school shall use title unless individual is a registered nurse. Amending § 22.1-274. (Patron–Adams, D.M., HB 1736)

PUBLIC SERVICE COMPANIES

- Broadband capacity; expands existing pilot program, municipal broadband authorities. Amending § 56-585.1:9. (Patron–Edwards, SB 1334)
- Commonwealth Clean Energy Policy; established, repeals provisions relating to energy objectives and Commonwealth Energy Policy. Amending §§ 56-46.1, 56-585.1, 56-598, 56-601, 62.1-199, 67-103, 67-104, and 67-201; adding § 67-101.1; repealing §§ 67-101 and 67-102. (Patron–Favola, SB 1284)
- Crisis Call Center Fund; created, consistency with federal guidelines, liability for emergency calls to the National Suicide Prevention Lifeline. Amending §§ 37.2-311.1, 56-484.12, 56-484.17, and 56-484.17:1; adding §§ 37.2-311.2, 37.2-311.3, 37.2-311.4, and 56-484.18:1. (Patron–McPike, SB 1302)

PUBLIC SERVICE COMPANIES (continued)

- Electric generating facility closures; public closure, integrated resource plans, provisions apply to facility owned or operated by an electric utility. Amending § 56-599; adding § 45.1-394.1. (Patron–Deeds, SB 1247)
- Electric generating facility closures; public disclosure, integrated resource plans. Amending § 56-599; adding § 45.1-394.1. (Patron–Subramanyam, HB 1834)
- Electric utilities; advanced renewable energy buyers. Amending § 56-585.5. (Patron-Sullivan, HB 1907)
- Electric utilities; authorizes electric utilities to partner with school divisions to implement projects designed to encourage the proliferation of school buses that are fueled in whole or in part by electricity. Amending § 58.1-3660; adding § 56-585.1:13. (Patron–Lucas, SB 1380)
- Electric utilities; eliminates customer credit reinvestment offsets. Amending § 56-585.1. (Patron-Bourne, HB 2049)
- Electric utilities; expands existing broadband capacity pilot program. Amending § 56-585.1:9. (Patron-Ayala, HB 1923)
- Electric utilities; nonjurisdictional customers, third party power purchase agreements. Amending Chapters 1187, 1188, 1189, 1193, 1194, and 1239, 2020 Acts. (Patron–Hurst, HB 2034; Edwards, SB 1420)
- Electric utilities; procedures under which the State Corporation Commission reviews the earnings and sets the rates of investor-owned incumbent electric utilities, triennial review. Amending § 56-585.1. (Patron–Jones, HB 2200)
- Electric utilities; procurement of certain equipment from Virginia-based or United States-based manufacturer using materials or product components made in Virginia or the United States, if reasonably available and competitively priced. Amending §§ 56-585.1:11 and 56-585.5. (Patron–DeSteph, SB 1295)
- Electric utilities; triennial review, fair rate of return, customer bill credits. Amending § 56-585.1. (Patron-Tran, HB 2160)
- Electric utilities; triennial review, period costs, rate reductions. Amending § 56-585.1. (Patron–Helmer, HB 1914)
- Electric utilities; triennial review proceeding by SCC, fair rates of return. (Patron-Hudson, HB 1984)
- Electric utility regulation; purchasing from competitive suppliers. Amending § 56-577. (Patron-Bourne, HB 2048)
- Percentage of Income Payment Program and Fund; Department of Housing and Community Development and the Department of Social Services to adopt rules or establish guidelines for the adoption, etc., of Program and Fund. Amending §§ 56-576 and 56-585.6. (Patron–Kory, HB 2330)
- Phase I or Phase II electric utilities; provision of broadband capacity, State Corporation Commission shall condition any approval of a petition on the requirement that construction shall commence within 18 months of such approval. Amending § 56-585.1:9. (Patron–Tyler, HB 2304; Boysko, SB 1413)
- Small agricultural generators; expands definition. Amending § 56-594.2. (Patron–Murphy, HB 1994)
- Underground utility facilities; removes the sunset on a pilot program allowing a locality that has adopted the urban county executive form of government (Fairfax County) to request an electric utility to place underground electric distribution lines, written agreement. Amending § 15.2-816.1. (Patron–Surovell, SB 1385)
- Virginia Highway Corporation Act; alteration of certificate of authority, powers and duties of State Corporation Commission, agreements between toll operator and Department. Amending §§ 56-539 and 56-542. (Patron–Subramanyam, HB 1832; Bell, SB 1259)

PULASKI, TOWN OF

Vacant buildings; removes requirement that a building meet the definition of "derelict building" for Town of Clifton Forge, Town of Pulaski, Town of Timberville, and any city to require, by ordinance, the owner or owners of buildings to register such buildings on an annual basis. Amending § 15.2-1127. (Patron–Locke, SB 1285)

RAPISARDA, JOSEPH P., JR.

Rapisarda, Joseph P., Jr.; commending. (Patron-VanValkenburg, HJR 699)

RAPPAHANNOCK COUNTY

Combined transient occupancy and food and beverage tax; for purposes of taxes that are currently authorized for Rappahannock and Madison Counties, the rate limit for such tax shall be the same as if the two taxes were imposed separately. Amending § 58.1-3842. (Patron–Hanger, SB 1438)

RAPPAHANNOCK RIVER

Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron–Cole, J.G., HB 1819)

RECYCLED PRODUCTS

Advanced recycling, etc.; definitions. Amending § 10.1-1400. (Patron-Hanger, SB 1164)

REDISTRICTING

Virginia Redistricting Commission; removing a citizen commissioner of the Commission for neglect of duty or gross misconduct by other commissioners in a public meeting. Amending § 30-392. (Patron–Watts, HB 2324)

Virginia Redistricting Commission; transparency in redistricting process. Amending §§ 30-396 and 30-399. (Patron–Levine, HB 2082)

REEVES, BRYCE E.

Prayer offered	151
Notified Clerk of presence	267

REFERENDUMS

Voter referendum; issuance of state general obligation bonds for school facility modernization, November 2022 general election, effective clause for provisions. (Patron–Stanley, SB 1109)

REGISTRARS

Absentee voting; availability on Sundays in office of general registrar or voter satellite office. Amending § 24.2-701.1. (Patron–Bagby, HB 1968)

Absentee voting; mandatory processing of returned absentee ballots before election day, central absentee voter precinct in the office of the general registrar. Amending §§ 24.2-709.1 and 24.2-712. (Patron–Deeds, SB 1246)

General registrar; qualifications, residency. Amending § 24.2-110. (Patron-Morrissey, SB 1281)

REHABILITATION AND SOCIAL SERVICES, COMMITTEE ON

Members listed	w
----------------	---

RELIGIOUS AND CHARITABLE MATTERS: CEMETERIES

Disposition of the remains of a decedent; persons to make arrangements for funeral and disposition of remains, person may designate in a signed and notarized writing, repeals provisions relating to when next of kin disagree and absence of next of kin. Amending §§ 32.1-309.1, 54.1-2800, 54.1-2807, 54.1-2825, and 57-27.3; adding §§ 54.1-2825.1, 57-27.4, 57-27.5, and 57-27.6; repealing §§ 54.1-2807.01 and 54.1-2807.02. (Patron–Sickles, HB 2005)

Veterans of Foreign Wars, American Legion, etc.; quantity of land certain associations may hold. Amending § 57-20. (Patron–Brewer, HB 2308)

RENFRO, NANCY TODD

Renfro, Nancy Todd; recording sorrow upon death. (Patron-Hope, HJR 643)

RENTAL PROPERTY

Virginia Residential Landlord and Tenant Act; landlord remedies, noncompliance with rental agreement, payment plan, removal of sunset. Repealing second enactment of Chapter 46, 2020 Sp. I Acts. (Patron–Price, HB 1889)

RETAIL SALES AND USE TAX

Retail sales and transient occupancy taxes; definitions, taxes on transient room rentals shall be computed on the basis of the total charges or the total price paid for the use or possession of the room, etc. Amending §§ 58.1-602, 58.1-603, 58.1-3819, 58.1-3819.1, 58.1-3823, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, 58.1-3826, 58.1-3842, and 58.1-3843; adding §§ 2.2-2320.2, 58.1-612.2, and 58.1-3818.8. (Patron–Norment, SB 1398)

Retail Sales and Use Tax; exemption for personal protective equipment. Adding § 58.1-609.14. (Patron-Byron, HB 2185; Pillion, SB 1403)

RETIREMENT SYSTEMS

Virginia Retirement System; amendments to reflect recent changes to federal law. Amending §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308. (Patron–Mundon King, HB 2181; Newman, SB 1251)

RICHMOND COUNTY

Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron–Cole, J.G., HB 1819)

ROADS

Access roads to economic development sites; criteria for use of funds. Amending § 33.2-1509. (Patron–McPike, SB 1253)

ROBERTSON, FRANKLIN DELANO

Robertson, Franklin Delano; recording sorrow upon death. (Patron-Morefield, HJR 668)

ROSENBERRY, BRIAN

Rosenberry, Brian; commending. (Patron-LaRock, HJR 761)

ROUTE 1

Jefferson Davis Highway; renames any section of U.S. Route 1 to "Emancipation Highway." Repealing Chapter 286, 1922 Acts. (Patron-Cole, J.G., HB 2075)

ROUTE 29

U.S. Route 29; authorizes the board of any locality that has adopted the county manager plan of government (Arlington County) to name any section located within the boundaries of the locality. Adding § 15.2-719.1. (Patron–Sullivan, HB 1854)

ROYALL, WILLIAM ARCHER, JR.

Royall, William Archer, Jr.; recording sorrow upon death. (Patron–Carr, HJR 690)

RUFF, FRANK M., JR.

RULES OF THE SENATE

RULINGS OF THE CHAIR AND PARLIAMENTARY INQUIRIES

- S.B. 1199. Chair ruled the substitute offered by Senator Stuart to S.B. 1199 was out of order . . . 211
- S.R. 91. Parliamentary inquiry as to whether the Senate would consider only S.R. 91, relating to the censure of a Senator, or other remarks made outside of S.R. 91. Statement by the Chair . . . 245

SAINT JOSEPH CATHOLIC SCHOOL

Saint Joseph Catholic School; commemorating its 145th anniversary. (Patron-Aird, HJR 748)

SALES AND USE TAX

- Data centers; expands sales and use tax exemption. Amending § 58.1-609.3. (Patron-Ruff, SB 1425)
- Data centers; sales and use tax exemption, criteria, report. Amending § 58.1-609.3. (Patron–Morefield, HB 2273)
- Data centers; sales and use tax exemption, identifying a "distressed locality." Amending § 58.1-609.3. (Patron–McPike, SB 1423)
- Isle of Wight County; authorized to impose an additional local sales and use tax to support schools. Amending §§ 58.1-602 and 58.1-605. (Patron–Norment, SB 1170)
- Retail sales and transient occupancy taxes; definitions, taxes on transient room rentals shall be computed on the basis of the total charges or the total price paid for the use or possession of the room, etc. Amending §§ 58.1-602, 58.1-603, 58.1-3819, 58.1-3819.1, 58.1-3823, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, 58.1-3826, 58.1-3842, and 58.1-3843; adding §§ 2.2-2320.2, 58.1-612.2, and 58.1-3818.8. (Patron–Norment, SB 1398)
- Retail Sales and Use Tax; exemption for personal protective equipment. Adding § 58.1-609.14. (Patron-Byron, HB 2185; Pillion, SB 1403)

SCENIC RIVERS

- Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron-Cole, J.G., HB 1819)
- South River; designates a 6.5-mile segment in the City of Waynesboro as a component of the Virginia Scenic Rivers System. Adding § 10.1-417.1. (Patron–Avoli, HB 1958)

SCHOOL BOARDS

- Broadband services; authorizes school boards to appropriate funds for the purposes of promoting, facilitating, and encouraging the expansion and operation of services for educational purposes. Adding §§ 15.2-986 and 22.1-79.9. (Patron–Boysko, SB 1225)
- Brunswick County school board; removes school board from the list of approved member salaries for appointed school boards. Amending § 22.1-32. (Patron–Tyler, HB 1798; Ruff, SB 1175)
- Local elections for governing bodies; elections for school boards, qualification of voters. Amending §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223. (Patron–Convirs-Fowler, HB 2198)
- Loudoun County school board; staggered terms of its members. Amending § 22.1-57.3:1.1. (Patron–Reid, HB 1838)
- School board policies; abusive work environments, definitions. Amending § 22.1-291.4. (Patron-Torian, HB 2176)
- School board property, certain; establishment of gun-free zone permitted. Adding § 22.1-131.1. (Patron–Subramanyam, HB 1909)
- School boards; each board to adopt a policy that prohibits a lawsuit against a student or the student's parent because student cannot pay for a meal at school, etc. Amending § 22.1-79.7. (Patron–Roem, HB 2013)
- School boards, certain; participation in the Afterschool Meal Program. Adding § 22.1-207.4:2. (Patron-Roem, HB 2135)
- Standards of Quality; each school board shall provide at least three specialized student support positions per 1,000 students, effective clause. Amending §§ 22.1-253.13:2 and 22.1-274. (Patron–McClellan, SB 1257)
- Students; guidelines established by the Department of Education on excused student absences, local school boards may require that student provide advance notice of intended absence, etc., civic engagement. Amending § 22.1-254. (Patron–Rasoul, HB 1940)
- Teachers and other licensed school board employees; cultural competency. Amending §§ 22.1-253.13:5 and 22.1-298.1; adding § 22.1-298.7. (Patron–Jenkins, HB 1904; Locke, SB 1196)

SCHOOL BUSES

Electric utilities; authorizes electric utilities to partner with school divisions to implement projects designed to encourage the proliferation of school buses that are fueled in whole or in part by electricity. Amending § 58.1-3660; adding § 56-585.1:13. (Patron–Lucas, SB 1380)

SEARCH WARRANTS

Search warrants; date and time of issuance, law-enforcement officer to be recognizable and identifiable, exceptions. Amending § 19.2-56. (Patron–Stuart, SB 1475)

SENATE OF VIRGINIA

Staffing levels, employment conditions, and compensation at the Virginia Department of Corrections, joint committee of various House and Senate Committees Studying; continued, appropriations. (Patron–Tyler, HJR 522)

SENATE OF VIRGINIA

Adjournments in memory:
BaCote, Mamye E., former Delegate
King, Reverend Dr. Martin Luther, Jr
Sicknick, Officer Brian David
Adjournment sine die at the Science Museum of Virginia in Richmond, Virginia
Reconvened session at the Science Museum of Virginia in Richmond, Virginia 504
Call to order at the Science Museum of Virginia in Richmond, Virginia
Reconvened session at the Science Museum of Virginia in Richmond, Virginia 504
Committees
Committee nominations report
Report of Committee to Review Disclosure Forms
Guests of Senate – See: Guests of Senate
Officers:
Clerk, Susan Clarke Schaar
Majority Leader, Richard L. Saslaw
Minority Leader, Thomas K. Norment, Jr.
President of the Senate, Justin E. Fairfax
President pro tempore, L. Louise Lucas
Sergeant-at-arms, D. Hobie Lehman
Organization, messages between Houses
Parliamentary Inquiries – See: Rulings of the Chair and Parliamentary Inquiries
President of the Senate
See: Justin E. Fairfax, Lieutenant Governor of Virginia and President of the Senate
President pro tempore
See: Lucas, L. Louise, President pro tempore
Proclamation of Governor
Roll Call
Rules of the Senate of Virginia
Senate operating under Rules adopted January 8, 2020
See also: Rules of the Senate
Rulings of the Chair – See: Rulings of the Chair and Parliamentary Inquiries Tie Votes – See: Tie Votes

SENIOR CITIZENS

Aging services; Department for Aging and Rehabilitative Services shall prioritize providing services to older persons with greatest economic or social need, definition. Amending §§ 51.5-134 and 51.5-135. (Patron–Adams, D.M., HB 1805)

Aging services; Department for Aging and Rehabilitative Services shall use available resources to provide services to older persons with the greatest economic and social needs, definitions. Amending §§ 51.5-134 and 51.5-135. (Patron–Barker, SB 1366)

SEWAGE DISPOSAL AND SEWERAGE SYSTEMS

Hampton Roads Sanitation District; changes to the enabling act related to the addition of Northampton and Accomack Counties. Amending Chapter 66, 1960 Acts. (Patron–Bloxom, HB 2257)

Onsite Sewage Indemnification Fund; authorizes the State Board of Health to use Fund to provide grants and loans to property owners with income at or below 200 percent of the federal poverty guidelines to repair failing onsite sewage systems or install onsite sewage systems on properties that lack adequate sewage disposal. Amending §§ 32.1-164 and 32.1-164.1:01; adding §§ 62.1-223.1, 62.1-223.2, and 62.1-223.3. (Patron–Hashmi, SB 1396)

SHANKS, FRED O., III

Shanks, Fred O., III; commending. (Patron-Marshall, HJR 777)

SHELOR, ISABEL GALLIMORE

Shelor, Isabel Gallimore; recording sorrow upon death. (Patron–Rush, HJR 669)

SHERIFFS

Law-enforcement civilian oversight bodies; adds a sheriff's office to those law-enforcement agencies that may be overseen by a body created by a locality and adds a nonprobationary deputy sheriff to those officers who are subject to such body, disciplinary determinations. Amending § 9.1-601. (Patron–Williams Graves, HB 2291)

SILBER, RANDALL R.

Silber, Randall R.; commending. (Patron-VanValkenburg, HJR 659)

SINGLETON, O.R., JR.

Singleton, O.R., Jr.; commending. (Patron-VanValkenburg, HJR 698)

SLAUGHTER, ALEXANDER HOKE

Slaughter, Alexander Hoke; recording sorrow upon death. (Patron-Carr, HJR 649)

SMALL BUSINESSES

Small Business and Supplier Diversity, Department of; redefines "small business." Amending §§ 2.2-1604 and 2.2-4310. (Patron–Obenshain, SB 1369)

Small, women-owned, and minority-owned businesses; right to appeal denial of initial certification. Amending \S 2.2-1606. (Patron–Mundon King, HB 2172)

Virginia Small Business Financing Authority; members to have small business lending experience. Amending § 2.2-2282. (Patron–Head, HB 1830)

Virginia Small Business Financing Authority; risk-based review of outstanding loans. Adding § 2.2-2312.1. (Patron–Mundon King, HB 2170)

Virginia Small Business Financing Authority; utilization or award of loan and grant program funds. Amending § 2.2-2312. (Patron–Mundon King, HB 2171)

SMITH, JAVIER J.

Smith, Javier J.; recording sorrow upon death. (Patron-McQuinn, HJR 738)

SMOKING

Property owners' associations and unit owners' associations; rulemaking authority concerning smoking. Amending §§ 55.1-1819 and 55.1-1959; adding §§ 55.1-1819.1 and 55.1-1960.1. (Patron–Keam, HB 1842)

SMYTH COUNTY

Southwestern Virginia Mental Health Institute; Governor to lease a portion of property to Smyth County for a term of three years, responsibility of County, corrects tax map references. Amending Chapter 678, 2019 Acts. (Patron–O'Quinn, HB 2098; Pillion, SB 1429)

SOCIAL SERVICES, BOARD OF OR DEPARTMENT OF

Produce Rx Program; Department of Social Services, et al., to develop a plan for a three-year pilot Program, report. (Patron–McQuinn, HB 2065)

SOLAR ENERGY

Solar and energy storage projects; siting agreements throughout the Commonwealth. Amending §§ 15.2-2288.8 and 15.2-2316.6 through 15.2-2316.9. (Patron–Jones, HB 2201; Barker, SB 1207)

Solar energy projects and energy storage systems; revenue share for projects and systems. Amending § 58.1-2636. (Patron–Heretick, HB 2269)

SOLID WASTE DISPOSAL

Buckingham County; fees for disposal of solid waste. Amending § 15.2-2159. (Patron-Peake, SB 1447)

Waste Diversion and Recycling Task Force; Department of Environmental Quality to continue Task Force. (Patron–Hashmi, SB 1319)

SOUTH RIVER

South River; designates a 6.5-mile segment in the City of Waynesboro as a component of the Virginia Scenic Rivers System. Adding § 10.1-417.1. (Patron–Avoli, HB 1958)

SOUTHWEST VIRGINIA

Southwestern Virginia Mental Health Institute; Governor to lease a portion of property to Smyth County for a term of three years, responsibility of County, corrects tax map references. Amending Chapter 678, 2019 Acts. (Patron–O'Quinn, HB 2098; Pillion, SB 1429)

SPEAKER OF THE VIRGINIA HOUSE OF DELEGATES

Speaker of the House of Delegates; confirming appointment to Virginia Commonwealth University Health System Authority Board of Directors. (Patron–Filler-Corn, HJR 579)

SPECIAL AND CONTINUING ORDERS: JOINT ORDERS

SPECIAL EDUCATION

- Children's Services Act; funds expended special education programs, definition of "transitional services." Amending §§ 2.2-5211 and 2.2-5212. (Patron–Mason, SB 1313)
- Children's Services Act; funds expended special education programs, state pool funds shall be allocated for no longer than 12 months for transitional services, report. Amending §§ 2.2-5211 and 2.2-5212. (Patron–VanValkenburg, HB 2117)
- Special education; Board of Education to amend a certain regulation to remove the word "component" following the word "evaluation," thereby ensuring compliance with the relevant federal regulation. (Patron–Mugler, HB 2314)
- Special education; Department of and the Board of Education to develop new policies and procedures, individualized education program (IEP), duty of Department to provide training and guidance documents to local school divisions on development of IEPs. Amending §§ 22.1-214, 22.1-215, 22.1-253.13:4, and 22.1-298.1; adding § 22.1-214.4. (Patron–Dunnavant, SB 1288)
- Special education; Department of Education to provide training and guidance documents to local school divisions on the development of Individualized Education Programs (IEPs) for children with disabilities, etc. Adding § 22.1-214.4. (Patron–Carr, HB 2299)
- Students with disabilities; Department of Education to update its special education and related services, etc. (Patron–Mundon King, HB 2316)

SPRUILL, LIONELL, SR.

Leave of absence.	34
Prayer offered 4	33

STANDARDS OF LEARNING

Health Standards of Learning; advanced directive education for high school students. (Patron-Kiggans, SB 1190)

Standards of Learning; reading and mathematics assessments for grades three through eight, individual student growth, through-year growth assessments. Amending § 22.1-253.13:3. (Patron-Coyner, HB 2027)

STANDARDS OF LEARNING (continued)

Standards of Learning; reduces total number and type of required assessments to minimum requirements, report. Amending § 22.1-253.13:3. (Patron–Pillion, SB 1401)

STANDARDS OF QUALITY

- Standards of Quality; each school board shall provide at least three specialized student support positions per 1,000 students, effective clause. Amending §§ 22.1-253.13:2 and 22.1-274. (Patron–McClellan, SB 1257)
- Standards of Quality; Joint Legislative Audit and Review Commission to study the true cost of education in the Commonwealth and provide an accurate assessment of the costs to implement, meetings shall be completed for the first year by November 30, 2022. (Patron–Lewis, SJR 294)

STATE AGENCIES

State agencies and their appointing authorities; diversity, equity, and inclusion strategic plans. Amending § 2.2-602. (Patron–Askew, HB 1993)

STATE CORPORATION COMMISSION

- Electric utilities; triennial review proceeding by SCC, fair rates of return. (Patron-Hudson, HB 1984)
- Paid family leave; State Corporation Commission's Bureau of Insurance to review and make recommendations, report. (Patron-Favola, SB 1219)
- Phase I or Phase II electric utilities; provision of broadband capacity, State Corporation Commission shall condition any approval of a petition on the requirement that construction shall commence within 18 months of such approval. Amending § 56-585.1:9. (Patron–Tyler, HB 2304; Boysko, SB 1413)
- Property and casualty insurance form; approval of form by State Corporation Commission. Amending § 38.2-317. (Patron–Ayala, HB 1892)
- Real estate settlement agents; SCC may share information collected from an agent or agency regarding any errors, etc., with any party to the real estate transaction in connection with the actions arising out of a settlement. Amending § 55.1-1004. (Patron–Spruill, SB 1110)
- State Corporation Commission; adds the Commission to the list of agencies that are exempt from paying fees for remote access to local land records. Amending § 17.1-276. (Patron–Kilgore, HB 1775)
- State Corporation Commission; business entities filings, amends various provisions of Va. Stock Corporation Act. Amending §§ 13.1-609, 13.1-610, 13.1-615, 13.1-615.1, 13.1-625, 13.1-628, 13.1-630, 13.1-636, 13.1-639, 13.1-658, 13.1-661, 13.1-710, 13.1-711, 13.1-716, 13.1-718, 13.1-721.1, 13.1-722.6, 13.1-722.12:1, 13.1-759, 13.1-765, 13.1-775.1, 13.1-803, 13.1-806, 13.1-807, 13.1-809, 13.1-815, 13.1-815.1, 13.1-816, 13.1-829, 13.1-830, 13.1-831, 13.1-835, 13.1-894, 13.1-897.1, 13.1-898.7, 13.1-921, 13.1-927, 13.1-936.1, 13.1-944.7, 13.1-1002, 13.1-1004, 13.1-1005, 13.1-1012, 13.1-1017, 13.1-1052, 13.1-1054, 13.1-1062, 13.1-1065, 13.1-1073.1, 13.1-1075, 13.1-1080, 13.1-1087, 13.1-1096, 13.1-1099.14, 13.1-1099.26, 13.1-1201, 13.1-1203, 13.1-1212, 13.1-1214, 13.1-1222, 13.1-1242, 13.1-1252, 13.1-1255, 13.1-1264, 13.1-1265, 13.1-1271, 13.1-1277, 15.2-5112, 15.2-5431.9, 50-73.1, 50-73.2, 50-73.6, 50-73.17, 50-73.54, 50-73.67, 50-73.70, 50-73.83, and 50-73.135; adding §§ 13.1-898.1:1, 13.1-1263.1, 15.2-5431.8:1, 15.2-5431.9:1, 15.2-5431.35:1, and 50-73.48:5; repealing §§ 13.1-941.01 through 13.1-944. (Patron–Keam, HB 2121)
- State Corporation Commission; issuance or renewal of insurance licenses or registrations during an emergency. Amending § 38.2-200. (Patron–Mason, SB 1255)
- State Corporation Commission; supervisory merger or transfer of assets of financially unstable credit unions, repeals provision relating to consolidation or merger. Amending § 6.2-1317; repealing § 6.2-1318. (Patron–Bagby, HB 1964)
- State Corporation Commission; transportation electrification, utility recovery of certain costs, report. (Patron–Sullivan, HB 2282)
- Virginia Highway Corporation Act; alteration of certificate of authority, powers and duties of State Corporation Commission, agreements between toll operator and Department. Amending §§ 56-539 and 56-542. (Patron–Subramanyam, HB 1832; Bell, SB 1259)

STATE CORPORATION COMMISSION (continued)

Virginia Residential Landlord and Tenant Act; landlord charges for security deposits, insurance premiums for damage insurance, etc., clerk of SCC shall provide a downloadable form for the filing of a resident agent appointment, etc. Amending §§ 17.1-275, 55.1-1200, 55.1-1204, 55.1-1206, 55.1-1208, 55.1-1211, 55.1-1226, 64.2-2008, and 64.2-2012. (Patron–McQuinn, HB 2249)

STATE EMPLOYEES

Highway construction; increases the value of highway maintenance and construction projects eligible to be performed by state or local employees. Amending § 33.2-234. (Patron–Krizek, HB 1813)

STATE PARKS

State parks; Department of Conservation and Recreation to develop recommendations for funding, report. (Patron–Orrock, HB 1804)

"STIR CRAZY IN WILLIAMSBURG"

"Stir Crazy in Williamsburg"; commending the authors. (Patron-Mullin, HJR 758)

STORMWATER MANAGEMENT

- Local stormwater assistance; flood mitigation and protection measures. Amending § 15.2-2114.01. (Patron–Ebbin, SB 1309)
- Nutrient credits; use by facility with certain stormwater discharge permit. Amending § 62.1-44.19:21. (Patron–Bulova, HB 1982)
- Stormwater Local Assistance Fund; grants awarded for projects related to Chesapeake Bay total maximum daily load (TMDL) requirements. Amending § 62.1-44.15:29.1. (Patron–Lewis, SB 1404)

STUART, RICHARD H.

Leave of absence	
Prayer offered	
Statement on vote:	
S.B. 1178	

STUDENTS

- Health Standards of Learning; advanced directive education for high school students. (Patron-Kiggans, SB 1190)
- High school student parking passes; valid driver's license or driver privilege card required. Adding § 22.1-205.1. (Patron-Norment, SB 1169)
- Kindergarten through grade 3; reading intervention services for certain students. Amending § 22.1-253.13:1. (Patron–Delaney, HB 1865)
- Licensed private schools for students with disabilities; accreditation. Amending § 22.1-323. (Patron-Kory, HB 2238)
- Local school divisions; availability of virtual and in-person learning to all students. (Patron-Dunnavant, SB 1303)
- School boards; each board to adopt a policy that prohibits a lawsuit against a student or the student's parent because student cannot pay for a meal at school, etc. Amending § 22.1-79.7. (Patron–Roem, HB 2013)
- Standards of Quality; each school board shall provide at least three specialized student support positions per 1,000 students, effective clause. Amending §§ 22.1-253.13:2 and 22.1-274. (Patron–McClellan, SB 1257)
- Student driver education program; parent/student component exemption. Amending § 22.1-205. (Patron–Keam, HB 2119)
- Student driver safety; driver education program shall include the dangers of distracted driving and speeding, high school student parking passes, valid driver's license required. Amending § 22.1-205; adding § 22.1-205.1. (Patron–Mugler, HB 1918)
- Students; eligibility for in-state tuition and state financial assistance program. Amending § 23.1-506; adding § 23.1-505.1. (Patron–Lopez, HB 2123; Boysko, SB 1387)

STUDENTS (continued)

- Students; guidelines established by the Department of Education on excused student absences, local school boards may require that student provide advance notice of intended absence, etc., civic engagement. Amending § 22.1-254. (Patron–Rasoul, HB 1940)
- Students; guidelines on excused student absences, each school board shall permit one school day-long excused absence per school year for any middle or high school student who is absent from school to engage in a civic event. Amending § 22.1-254. (Patron–McClellan, SB 1439)
- Students with disabilities; Department of Education to update its special education and related services, etc. (Patron–Mundon King, HB 2316)

STUDY COMMISSIONS, COMMITTEES, AND REPORTS

- Affordable and market-rate housing; Department of Housing and Community Development to convene an advisory group to evaluate construction of internal, etc., dwelling units. (Patron–Samirah, HB 2053)
- Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption, report, sunset provision. Amending §§ 4.1-119, 4.1-204, 4.1-206.1, 4.1-206.3, 4.1-207, 4.1-210, 4.1-212.1, and 4.1-221. (Patron–Bulova, HB 1879; Bell, SB 1299)
- Apprenticeship training programs; Virginia Board of Workforce Development, et al., shall review availability of programs, etc., report. (Patron–Simonds, HB 1849)
- Assisted living and auxiliary grants; Joint Commission on Health Care to study available data regarding, etc. (Patron-Spruill, SJR 293)
- Barrier Crimes and Criminal History Records Checks, Joint Subcommittee Studying; continued, appropriation. (Patron-Edwards, SJR 285)
- Behavioral health; assessments in local correctional facilities, report. Amending § 53.1-68. (Patron-Coyner, HB 1874)
- Behavioral Health Commission; created, report. Adding §§ 30-401 through 30-408. (Patron–Deeds, SB 1273)
- Cannabis oil; processing and dispensing by pharmaceutical processors, certification for use of oil for treatment, report. Amending §§ 54.1-3408.3, 54.1-3442.5, 54.1-3442.6, and 54.1-3442.7. (Patron–Adams, D.M., HB 1988)
- Carbon Sequestration Task Force; established by Secretary of Natural Resources, jointly with the Secretary of Agriculture and Consumer Services, to convene a task force, report. (Patron–Lewis, SB 1374)
- Charitable gaming; definitions, regulations, conduct of instant bingo, network bingo, pull tabs, and seal cards, report. Amending §§ 18.2-334.2, 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.34. (Patron–Reeves, SB 1127)
- Charitable gaming; increase in certain maximum allowable prize amounts. Amending § 18.2-340.33. (Patron–Keam, HB 1843)
- Child care providers; background checks portability, check shall not be required if individual completed a background check within previous five years, pilot program, report. Amending §§ 19.2-389, 22.1-289.035, 22.1-289.039, 63.2-1720.1, and 63.2-1724. (Patron–McGuire, HB 2086; McClellan, SB 1316)
- Children's Services Act; funds expended special education programs, definition of "transitional services." Amending §§ 2.2-5211 and 2.2-5212. (Patron–Mason, SB 1313)
- Children's Services Act; funds expended special education programs, state pool funds shall be allocated for no longer than 12 months for transitional services, report. Amending §§ 2.2-5211 and 2.2-5212. (Patron–VanValkenburg, HB 2117)
- Coal Employment and Production Incentive and Coalfield Employment Enhancement Tax Credits; sunset date, credits earned prior to January 1, 2022. Amending §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1. (Patron–Hudson, HB 1899; McPike, SB 1252)
- Commonwealth Health Reinsurance Program; established, report. Amending §§ 38.2-4214 and 38.2-4319; adding §§ 38.2-6600 through 38.2-6607. (Patron–Sickles, HB 2332)
- Community services boards; discharge plan shall be completed prior to the individual's discharge, report. Amending § 37.2-505. (Patron–McPike, SB 1304)
- Comprehensive review of computer science standards, courses, and pathways in public schools; Department of Education to perform, report. (Patron–Simonds, HB 1885)

- Conservation and Recreation, Department of; removes the authority from the Department to sell or demise certain lands over which it has control, etc., report. Amending § 10.1-109. (Patron–Subramanyam, HB 1833)
- Consumer Protection Act; prohibited practices, certain advertising related to any public or private school quality. Amending § 59.1-200. (Patron–Samirah, HB 2003)
- Corporate income tax; Division of Legislative Services, et al., to establish a work group to assess the feasibility of transitioning to a unitary combined reporting system. (Patron–Watts, HJR 563)
- COVID-19; Joint Legislative Audit and Review Commission to study the impact on Virginia's public schools, students, and school employees, meetings shall be completed by November 30, 2022. (Patron–Guy, HJR 549; Lucas, SJR 308)
- Criminal justice, behavioral health, and other human services records; Department of Behavioral Health and Developmental Services to establish a work group to study the feasibility of developing a secure, de-identified, etc., database. (Patron–Price, HJR 578)
- Criminal records; establishes a process for automatic expungement, etc., definitions, penalties, report. Amending §§ 9.1-101, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 19.2-72, 19.2-74, 19.2-310.7, 19.2-340, 19.2-389.3, and 19.2-390; adding §§ 19.2-392.5 through 19.2-392.16. (Patron–Herring, HB 2113)
- Crosswalk design; Commissioner of Highways shall convene a work group to determine whether there should be model policies for design and installation, and, if so, establish recommendations. (Patron–Keam, HB 1841)
- Dairy Producer Margin Coverage Premium Assistance Program; established, eligible dairy producer shall apply to the Department of Conservation and Recreation by February 1 of each year to participate, report, sunset date. Adding §§ 3.2-3304 through 3.2-3307. (Patron–Gooditis, HB 1750)
- Data centers; sales and use tax exemption, criteria, report. Amending § 58.1-609.3. (Patron–Morefield, HB 2273)
- Data Governance and Analytics, Office of; created, establishes the advisory Virginia Data Commission, membership, report. Amending § 2.2-203.2:4; adding §§ 2.2-2558 through 2.2-2564. (Patron–Barker, SB 1365)
- Early Psychosis Intervention and Coordinated Specialty Care Program Advisory Board; established. Adding § 37.2-313.2. (Patron–Morrissey, SB 1427)
- Electric utilities; authorizes electric utilities to partner with school divisions to implement projects designed to encourage the proliferation of school buses that are fueled in whole or in part by electricity. Amending § 58.1-3660; adding § 56-585.1:13. (Patron–Lucas, SB 1380)
- Electric vehicle rebate program; creation and funding, report, sunset date. Adding §§ 67-1800 through 67-1806. (Patron–Reid, HB 1979)
- Enslaved Ancestors College Access Scholarship and Memorial Program; established, report. Adding § 23.1-615.1. (Patron–Reid, HB 1980)
- Fetal and Infant Mortality Review Team; Office of the Chief Medical Examiner of Department of Health shall convene a work group to develop a plan for the establishment of Team, report. (Patron–Ayala, HB 1950)
- Funding local health departments; cooperative local health budget, report. Adding § 32.1-34.3. (Patron-Bagby, HB 1963)
- Gold; Secretary of Natural Resources, et al., shall establish a work group to study mining and processing, issuance of certain mining permits. (Patron–Guzman, HB 2213)
- Higher educational institutions, public; governing boards, meetings, input, and disclosures. Amending §§ 23.1-409, 23.1-802, and 23.1-1303. (Patron–Keam, HB 2120)
- Income tax, state; Joint Legislative Audit and Review Commission to study increasing the progressivity of Virginia's system, meetings shall be completed by November 30, 2022. (Patron-Watts, HJR 567)
- Individuals with intellectual and developmental disabilities; Department of Medical Assistance Services shall establish a work group to study and develop recommendations for permanent use of virtual supports and increasing access to virtual supports and services. (Patron–Suetterlein, SB 1472)
- Individuals with intellectual and developmental disabilities; Department of Medical Assistance Services to study and develop recommendations for use of virtual support, etc. (Patron–Runion, HB 2197)

Inmates; Board of Local and Regional Jails to review services provided during pregnancy, etc., report. (Patron-Favola, SB 1300)

Interagency Environmental Justice Working Group; established as an advisory council in executive branch of state government to further environmental justice. Amending § 2.2-234; adding §§ 2.2-236, 2.2-237, and 15.2-2223.5. (Patron–Simonds, HB 2074)

Interagency Environmental Justice Working Group; established, definitions, report, sunset date. Amending § 2.2-234; adding § 2.2-236. (Patron–Hashmi, SB 1318)

Invasive plant species; Department of Conservation and Recreation, et al., to study the sale and use of species. (Patron–Bulova, HJR 527)

Juvenile offenders; youth justice diversion programs, report. Amending § 16.1-260; adding § 16.1-309.11. (Patron–Mullin, HB 2017)

Juvenile records; confidentiality of records relevant to treatment, services, etc., exceptions. Amending § 16.1-300. (Patron–Barker, SB 1206)

Licensed certified midwives; clarifies definition, licensure, practice shall be in consultation with a licensed physician with a practice agreement, report. Amending § 54.1-2900; adding § 54.1-2957.04. (Patron–Gooditis, HB 1953; Lucas, SB 1320)

Local cigarette taxes; any locality is authorized to levy taxes upon sale of cigarettes, regional cigarette tax boards, definitions, report. Amending § 58.1-3830; adding § 58.1-3832.1. (Patron–Hanger, SB 1326)

Mandatory minimum sentences; elimination, modification of sentence to mandatory minimum term of confinement for felony offenses, report. Amending §§ 3.2-4212, 4.1-302, 16.1-253.2, 18.2-36.1, 18.2-36.2, 18.2-46.3:3, 18.2-51.1, 18.2-53.1, 18.2-57, 18.2-60.4, 18.2-61, 18.2-67.1, 18.2-67.2, 18.2-67.5:2, 18.2-67.5:3, 18.2-154, 18.2-186.4, 18.2-248, 18.2-248.01, 18.2-248.03, 18.2-248.1, 18.2-248.5, 18.2-255, 18.2-255.2, 18.2-270, 18.2-308.1, 18.2-308.2, 18.2-308.2; 18.2-308.4, 18.2-374.1, 18.2-374.1:1, 18.2-374.3, 19.2-297.1, 46.2-341.28, 46.2-357, 46.2-391, 46.2-865.1, and 53.1-203. (Patron–Edwards, SB 1443)

Marijuana; legalization of simple possession, etc., expungement of criminal records, reports, penalties. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010,4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, $16.1-69.40:1, \quad 16.1-69.48:1, \quad 16.1-228, \quad 16.1-260, \quad 16.1-273, \quad 16.1-278.8:01, \quad 16.1-278.9, \\ 17.1-276, \quad 17.1-293.1, \quad 17.1-323, \quad 17.1-413, \quad 17.1-502, \quad 18.2-46.1, \quad 18.2-57, \quad 18.2-247, \quad 18.2-248, \\$ 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-390, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 19.2-392.5, 19.2-392.6, 19.2-392.7, 19.2-392.8, 19.2-392.9, 19.2-392.12, 19.2-392.13, 19.2-392.11, 19.2-392.10, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Herring, HB 2312)

Marijuana; unlawful possession in certain cases, using or consuming while in a motor vehicle, etc., penalties, reports. Amending §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, 4.1-122, 4.1-124, 4.1-128, 4.1-200, 4.1-201, 4.1-202, 4.1-205, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208,

- 4.1-212, 4.1-213, 4.1-215, 4.1-216, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, 4.1-230, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, 4.1-310.1, 4.1-320, 4.1-323, 4.1-324, 4.1-325, 4.1-325.2, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, 19.2-392.02, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1; adding §§ 2.2-2499.1 through 2.2-2499.4, 3.2-4117.1, 3.2-4117.2, 3.2-4122, 3.2-5145.6 through 3.2-5145.9, 4.1-600 through 4.1-1503, 6.2-107.1, 18.2-265.22 through 18.2-265.28, 19.2-392.2:1, and 46.2-341.20:7; repealing §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3. (Patron-Ebbin, SB 1406)
- Maternal Health Data and Quality Measures, Task Force on; established, report. (Patron-Herring, HB 2111)
- Military Spouse Liaison; position created in Department of Veterans Services, report, effective clause. Adding § 2.2-2002.2. (Patron–Kiggans, SB 1150)
- Military-overseas ballots; Secretary of Administration to oversee and develop a charter and directives for the State Board of Elections to form a working group to study implementation of electronic return of voted ballots. (Patron–DeSteph, SJR 322)
- Motor Vehicles, Department of; limits the release of privileged information to government entities and law-enforcement agencies. Amending §§ 46.2-100, 46.2-208, 46.2-209, 46.2-209.1, and 46.2-328.3. (Patron–Tran, HB 2163)
- Online portal for tax practitioners; Department of Taxation shall analyze prospect of establishing, report. (Patron-Coyner, HB 2060)
- Opioid Abatement Authority; established, Fund created, report, membership. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Barker, SB 1469)
- Opioid Abatement Authority; established, Opioid Abatement Fund created, investment of assets of Fund, report. Amending § 2.2-212; adding §§ 2.2-507.3, 2.2-2365 through 2.2-2376, and 51.1-124.40. (Patron–Herring, HB 2322)
- Paid family leave; State Corporation Commission's Bureau of Insurance to review and make recommendations, report. (Patron–Favola, SB 1219)
- Permit fee schedules; Department of Environmental Quality shall convene a work group to revise schedule for nonhazardous solid waste management facilities. (Patron-Petersen, SB 1210)
- Police and court records; expungement and sealing of records, Expungement Fee Fund created, protection of public record information, penalties, report, effective clause. Amending §§ 9.1-128, 9.1-134, 17.1-502, 19.2-392.1, 19.2-392.2, 19.2-392.3, and 19.2-392.4; adding §§ 1-229.1, 17.1-205.1, 19.2-392.1:1, 19.2-392.2:1, 19.2-392.2:2, 19.2-392.2:3, 19.2-392.3:1, 19.2-392.4:1, 19.2-392.5, 19.2-392.6, and 19.2-392.7. (Patron–Surovell, SB 1339)
- Post-conviction relief; previously admitted forensic scientific evidence, effective date, report, and effective clause. Adding § 19.2-327.15. (Patron–Stanley, SB 1105)
- Pretrial data collection; Virginia Criminal Sentencing Commission to collect and disseminate on an annual basis, report. Amending § 2.2-3802; adding § 19.2-134.1. (Patron–Herring, HB 2110; Lucas, SB 1391)
- Produce Rx Program; Department of Social Services, et al., to develop a plan for a three-year pilot Program, report. (Patron–McQuinn, HB 2065)
- Rare Disease Council and Rare Disease Council Fund; created, report. Adding §§ 32.1-73.14 through 32.1-73.17. (Patron–Murphy, HB 1995)
- Recurrent Flooding Resiliency, Commonwealth Center for; Center shall evaluate development of Flood Resiliency Clearinghouse Program for coordinating flood mitigation solutions, report. (Patron–Hodges, HB 2187)

- Special education; Department of and the Board of Education to develop new policies and procedures, individualized education program (IEP), duty of Department to provide training and guidance documents to local school divisions on development of IEPs. Amending §§ 22.1-214, 22.1-215, 22.1-253.13:4, and 22.1-298.1; adding § 22.1-214.4. (Patron–Dunnavant, SB 1288)
- Staffing levels, employment conditions, and compensation at the Virginia Department of Corrections, joint committee of various House and Senate Committees Studying; continued, appropriations. (Patron–Tyler, HJR 522)
- Standards of Learning; reduces total number and type of required assessments to minimum requirements, report. Amending § 22.1-253.13:3. (Patron–Pillion, SB 1401)
- Standards of Quality; Joint Legislative Audit and Review Commission to study the true cost of education in the Commonwealth and provide an accurate assessment of the costs to implement, meetings shall be completed for the first year by November 30, 2022. (Patron–Lewis, SJR 294)
- State Corporation Commission; transportation electrification, utility recovery of certain costs, report. (Patron-Sullivan, HB 2282)
- State parks; Department of Conservation and Recreation to develop recommendations for funding, report. (Patron–Orrock, HB 1804)
- Telemedicine service; coverage of telehealth services by an insurer, etc., services delivered through real-time audio-only telephone, report. Amending §§ 32.1-325, 38.2-3418.16, and 54.1-3303. (Patron–Adams, D.M., HB 1987)
- Through-year growth assessment system; Board of Education shall establish, report, effective clause. (Patron-Dunnavant, SB 1357)
- Transit equity and modernization; Department of Rail and Public Transportation to study. (Patron-McQuinn, HJR 542)
- Trees; replacement and conservation during development, projects located in a Chesapeake Bay Preservation Area to address recurrent flooding, effective date, report, effective clause. Amending §§ 15.2-961 and 15.2-961.1. (Patron–Marsden, SB 1393)
- Universal health care; Joint Commission on Health Care shall enter into a contract with a qualified entity to study options for financing. (Patron–Samirah, HB 2271)
- Virginia Brownfield and Coal Mine Renewable Energy Grant Fund and Program; established, report. Adding § 67-1800. (Patron–Kilgore, HB 1925)
- Virginia Digital Equity Pilot Program and Fund; established, report, sunset date, effective clause. Adding § 63.2-806. (Patron–Mason, SB 1462)
- Virginia Electric Vehicle Grant Fund and Program; created, report. Adding § 10.1-1322.5. (Patron–Keam, HB 2118)
- Virginia Employment Commission; communications with parties, use of electronic means, report. Adding § 60.2-121.1. (Patron-Tran, HB 2036)
- Virginia Good Neighbor Next Door program; Virginia Housing Development Authority shall report recommendations for creating Program. (Patron–Convirs-Fowler, HB 2072)
- Virginia LGBTQ+ Advisory Board; established, report. Adding §§ 2.2-2499.1 through 2.2-2499.4. (Patron-Lopez, HB 2130)
- Virginia Science, Technology, Engineering, and Mathematics (STEM) Advisory Board; established, report. Adding §§ 22.1-364 through 22.1-368. (Patron–Simonds, HB 2058)
- Waste Diversion and Recycling Task Force; Department of Environmental Quality to continue Task Force. (Patron–Hashmi, SB 1319)

SUERDIECK, REBECCA

Suerdieck, Rebecca and Julia Oxrieder; commending. (Patron-Mullin, HJR 765)

SUETTERLEIN, DAVID R.

ements on votes:	
S.B. 1130	140
S.B. 1134	140
S.B. 1145	140
S.B. 1155	140
S.B. 1156	140
S.B. 1363	163

SUICIDE

Suicide; abolishes the common-law crime. Adding § 18.2-16.1. (Patron-Simon, HB 1951)

SUMMONS AND PROCESS

Summons; eliminates the requirement that a promise to appear be completed after issuance for a misdemeanor offense or an administrative violation. Amending §§ 2.2-311, 19.2-74, 28.2-901, 29.1-210, 46.2-936, 46.2-940, 54.1-306, 54.1-2506, and 54.1-4407. (Patron–Hanger, SB 1437)

Summons; promises to appear after issuance. Amending §§ 46.2-936 and 46.2-940. (Patron–Mason, SB 1329)

Summons for unlawful detainer; notice to tenant, adverse employment actions prohibited. Amending § 8.01-126. (Patron–Jenkins, HB 1897)

SUPPORT ORDERS

Support orders; contents of orders, change in employment status, unemployment benefits. Amending §§ 20-60.3 and 63.2-1916. (Patron–Leftwich, HB 2192)

SUPREME COURT OF VIRGINIA

Lawyers; client accounts, repeals the provision prohibiting the Supreme Court of Virginia from adopting a disciplinary rule requiring that lawyers deposit client funds in an interest-bearing account. Amending § 54.1-3916; repealing § 54.1-3915.1. (Patron–Sullivan, HB 1853)

Motor Vehicles, Department of, and Supreme Court of Virginia; repeals reporting requirement. (Patron–Newman, SB 1277)

Virginia Criminal Sentencing Commission; confirms the appointment by the Chief Justice of the Supreme Court of Chairman of Commission. (Patron–Herring, HJR 629)

SWEAT, JOSH

Sweat, Josh; commending. (Patron-Hayes, HJR 692)

SWEGLE, MADELINE

Swegle, Madeline; commending. (Patron-Filler-Corn, HJR 720)

SZAKOS, JOE

Szakos, Joe; commending. (Patron–Hudson, HJR 719)

TARANTINO, JOSEPH MAURICE

Tarantino, Joseph Maurice; recording sorrow upon death. (Patron-Robinson, HJR 662)

TAXATION

Administration of blighted and derelict properties; modifies definition of "qualifying locality." Amending §§ 58.1-3221.6 and 58.1-3970.1. (Patron–Carr, HB 1969)

Agricultural best management practices; creates an enhanced individual and corporate income tax credit beginning in taxable year 2021 but before January 1, 2025, for the implementation of certain practices by the taxpayer that are required as part of a certified resource management plan. Amending §§ 58.1-339.3 and 58.1-439.5. (Patron–Hanger, SB 1162)

Agricultural best management practices; creates an enhanced individual and corporate income tax credit beginning in taxable year 2021 for the implementation of certain practices by the taxpayer that are required as part of a certified resource management plan, definition of "liability after such cost-share." Amending §§ 58.1-339.3 and 58.1-439.5. (Patron–Wilt, HB 1763)

Agricultural equipment; establishes a refundable individual and corporate income tax credit. Amending §§ 58.1-334, 58.1-337, 58.1-432, and 58.1-436. (Patron–Hanger, SB 1163)

Alcoholic beverage control; local special events license, taxes and fees. Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, and 4.1-233.1. (Patron–Dunnavant, SB 1471)

Alcoholic beverage control; outdoor refreshment area license, fees for state and local licenses. Amending §§ 4.1-206, 4.1-206.3, 4.1-231, 4.1-231.1, 4.1-233, 4.1-233.1, and 4.1-308. (Patron–Ayala, HB 2266)

Casino gaming; requirements for issuance of operator's license, human trafficking training. Amending § 58.1-4110. (Patron–Simonds, HB 1944)

TAXATION (continued)

- Casino gaming; technical amendments to the gaming law related to its interaction with sports betting law, the capital investment required of an applicant for a license, etc. Amending §§ 2.2-3711, 58.1-4100, 58.1-4109, 58.1-4110, 58.1-4114, 58.1-4122, 58.1-4124, and 58.1-4125. (Patron–Krizek, HB 1812)
- Coal Employment and Production Incentive and Coalfield Employment Enhancement Tax Credits; sunset date, credits earned prior to January 1, 2022. Amending §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1. (Patron–Hudson, HB 1899; McPike, SB 1252)
- Combined transient occupancy and food and beverage tax; for purposes of taxes that are currently authorized for Rappahannock and Madison Counties, the rate limit for such tax shall be the same as if the two taxes were imposed separately. Amending § 58.1-3842. (Patron–Hanger, SB 1438)
- Corporate income tax; Division of Legislative Services, et al., to establish a work group to assess the feasibility of transitioning to a unitary combined reporting system. (Patron–Watts, HJR 563)
- Data centers; expands sales and use tax exemption. Amending § 58.1-609.3. (Patron-Ruff, SB 1425)
- Delinquent returns; Department of Taxation to request taxpayers who have failed to file tax returns when due to prepare and file such returns, exception. Adding § 58.1-1802.2. (Patron–Coyner, HB 2059)
- Electric utilities; authorizes electric utilities to partner with school divisions to implement projects designed to encourage the proliferation of school buses that are fueled in whole or in part by electricity. Amending § 58.1-3660; adding § 56-585.1:13. (Patron–Lucas, SB 1380)
- Energy storage systems; definitions, tax exemption, revenue share for systems. Amending §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660. (Patron–Heretick, HB 2006; Petersen, SB 1201)
- Income tax, state; conformity of the Commonwealth's taxation system with the Internal Revenue Code, deduction relating to payroll expenses. Amending §§ 58.1-301, 58.1-322.02, and 58.1-322.03. (Patron–Watts, HB 1935)
- Income tax, state; conformity of the Commonwealth's taxation system with the Internal Revenue Code, taxable income, subtractions. Amending §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402. (Patron–Howell, SB 1146)
- Income tax, state; Department of Taxation to include space on the appropriate forms for voluntary inclusion of personal and contact information, facilitated enrollment program. Amending §§ 38.2-6505, 58.1-3, and 58.1-341.1. (Patron–Sickles, HB 1884)
- Income tax, state; Joint Legislative Audit and Review Commission to study increasing the progressivity of Virginia's system, meetings shall be completed by November 30, 2022. (Patron-Watts, HJR 567)
- Income tax, state; subtraction for low-income military veterans with a permanent service-connected disability. Amending § 58.1-322.02. (Patron–Wiley, HB 2337)
- Isle of Wight County; authorized to impose an additional local sales and use tax to support schools. Amending §§ 58.1-602 and 58.1-605. (Patron–Norment, SB 1170)
- Local cigarette taxes; any locality is authorized to levy taxes upon sale of cigarettes, regional cigarette tax boards, definitions, report. Amending § 58.1-3830; adding § 58.1-3832.1. (Patron–Hanger, SB 1326)
- Local gas severance tax; extends sunset date. Amending § 58.1-3713. (Patron-Morefield, HB 2293)
- Online portal for tax practitioners; Department of Taxation shall analyze prospect of establishing, report. (Patron-Coyner, HB 2060)
- Peanuts; extends sunset date of excise tax on all peanuts grown in Virginia. Amending § 3.2-1905. (Patron–Brewer, HB 1751; Lucas, SB 1411)
- Personal property tax; exemption for motor vehicle of a 100 percent disabled veteran. Adding § 58.1-3668. (Patron–Reeves, SB 1130)
- Port of Virginia tax credits; extends the sunset date. Amending §§ 58.1-439.12:06, 58.1-439.12:09, and 58.1-439.12:10. (Patron-Spruill, SB 1158)
- Research and development expenses; tax credit available against the bank franchise tax for taxable years beginning on and after January 1, 2021. Amending §§ 58.1-439.12:08 and 58.1-439.12:11. (Patron–Locke, SB 1112)

TAXATION (continued)

- Research and development tax credits; availability against the bank franchise tax for taxable years beginning on and after January 1, 2021. Amending §§ 58.1-439.12:08 and 58.1-439.12:11. (Patron-Mugler, HB 1916)
- Retail sales and transient occupancy taxes; definitions, taxes on transient room rentals shall be computed on the basis of the total charges or the total price paid for the use or possession of the room, etc. Amending §§ 58.1-602, 58.1-603, 58.1-3819, 58.1-3819.1, 58.1-3823, 58.1-3824, 58.1-3825.2, 58.1-3825.3, 58.1-3842, 58.1-3825, 58.1-3826, adding §§ 2.2-2320.2, 58.1-612.2, and 58.1-3818.8. (Patron-Norment, SB 1398)
- Retail Sales and Use Tax; exemption for personal protective equipment. Adding § 58.1-609.14. (Patron–Byron, HB 2185; Pillion, SB 1403)
- Solar energy projects and energy storage systems; revenue share for projects and systems. Amending § 58.1-2636. (Patron–Heretick, HB 2269)
- Sports betting; clarifies the procedures by which the Virginia Lottery determines whether an event is considered youth sports, on which betting is prohibited, an international athletic event organized by the International Olympic Committee shall not be considered to be youth sports, issuance of permits to operate sports betting platforms. Amending §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100. (Patron–Sickles, HB 1847)
- Sports betting; clarifies the procedures by which the Virginia Lottery determines whether an event is considered youth sports, on which betting is prohibited, an international athletic event organized by the International Olympic Committee shall not be considered to be youth sports, regardless of the age of the participants. Amending §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100. (Patron-McPike, SB 1254)
- Tangible personal property taxes; classification of certain motor vehicles, trailers, and semitrailers. Amending § 58.1-3506. (Patron–Walker, HB 1774)
- Tax Commissioner; waiver of accrual of interest in the event that Governor declares state of emergency. Amending § 58.1-112. (Patron–Murphy, HB 1999)
- Tax delinquent property; sale of land for delinquent taxes. Amending § 58.1-3965. (Patron-Hope, HB 2165)
- Virginia housing opportunity; tax credit established starting in taxable year 2021, which is equal to the amount of the federal low-income housing tax credit allocated. Adding §§ 58.1-439.29 and 58.1-439.30. (Patron-Locke, SB 1197)

TAZEWELL COUNTY

Tazewell County; authorizes a quitclaim and release of interest and conveyance of an easement by Board of Wildlife Resources. (Patron–Wampler, HB 2252; Pillion, SB 1400)

T.C. WILLIAMS SCHOOL OF LAW AT THE UNIVERSITY OF RICHMOND

T.C. Williams School of Law at the University of Richmond; commemorating its 150th anniversary. (Patron-Adams, D.M., HJR 676)

TEACHERS

- Education, Board of; temporary extension of the license of any individual licensed by Board whose license expires on June 30, 2021. (Patron-Ward, HB 1776)
- other licensed school board employees; cultural Amending §§ 22.1-253.13:5 and 22.1-298.1; adding § 22.1-298.7. (Patron-Jenkins, HB 1904; Locke, SB 1196)

TEAGUE, JOSEPH WILLIAM, SR.

Teague, Joseph William, Sr.; recording sorrow upon death. (Patron-Hudson, HJR 707)

THE CHATHAM GARDEN CLUB

The Chatham Garden Club; commemorating its 100th anniversary. (Patron-Adams, L.R., HJR 744)

THE VIRGINIA-PILOT

The Virginian-Pilot; commending. (Patron-Mullin, HJR 750)

THORPE, AVICIA BEATRICE HOOPER

Thorpe, Avicia Beatrice Hooper; recording sorrow upon death. (Patron–Marshall, HJR 768)

TIE VOTES

S.B. 1157	124
S.J.R. 270	480
S.J.R. 290, Chair did not vote.	489

TIMBERVILLE, TOWN OF

Vacant buildings; removes requirement that a building meet the definition of "derelict building" for Town of Clifton Forge, Town of Pulaski, Town of Timberville, and any city to require, by ordinance, the owner or owners of buildings to register such buildings on an annual basis. Amending § 15.2-1127. (Patron–Locke, SB 1285)

TOBACCO AND TOBACCO PRODUCTS

Local cigarette taxes; any locality is authorized to levy taxes upon sale of cigarettes, regional cigarette tax boards, definitions, report. Amending § 58.1-3830; adding § 58.1-3832.1. (Patron–Hanger, SB 1326)

TOLLS

- Virginia Highway Corporation Act; alteration of certificate of authority, powers and duties of State Corporation Commission, agreements between toll operator and Department. Amending §§ 56-539 and 56-542. (Patron–Subramanyam, HB 1832; Bell, SB 1259)
- Virginia Highway Corporation Act of 1988; repeals Act, roadways to operate under the Public-Private Transportation Act of 1995, implementation of distance-based tolling on certain roadways, etc. Amending § 33.2-613; repealing §§ 33.2-1823 and 56-535 through 56-552. (Patron–Reid, HB 2104)

TOURISTS AND TOURIST INDUSTRY

Tourism improvement districts; authorizes any locality to create. Adding §§ 15.2-2413.1 through 15.2-2413.11. (Patron–Bell, SB 1298)

TRADE AND COMMERCE

- Consumer Data Protection Act; definitions, establishes a framework for controlling and processing personal data, applies to all persons that conduct business in the Commonwealth, etc. Adding §§ 59.1-571 through 59.1-581. (Patron–Hayes, HB 2307)
- Consumer Data Protection Act; establishes a framework for controlling and processing personal data, effective date. Adding §§ 59.1-571 through 59.1-581. (Patron–Marsden, SB 1392)
- Consumer Protection Act; prohibited practices, certain advertising related to any public or private school quality. Amending § 59.1-200. (Patron–Samirah, HB 2003)
- Enterprise zone job creation grants; for purposes of wage requirements, the minimum wage shall be the higher of the state minimum wage or the federal minimum wage, delayed effective date. Amending § 59.1-547. (Patron–Heretick, HB 1881)
- Food delivery platforms; agreements required, penalty. Amending § 59.1-200; adding §§ 59.1-571, 59.1-572, and 59.1-573. (Patron–Willett, HB 2062)
- Humane Cosmetics Act; prohibits a cosmetics manufacturer from testing cosmetics on animals in the Commonwealth, civil penalties. Adding §§ 59.1-571 through 59.1-574. (Patron–Kory, HB 2250; Boysko, SB 1379)
- Legal service plans; legal services organization shall submit registration information and fees to the Commissioner. Amending § 59.1-441.2. (Patron–Jenkins, HB 1877)
- Technology Development Grant Fund; created. Adding § 59.1-284.38. (Patron-Howell, SB 1156)

TRAFFIC REGULATIONS AND VIOLATIONS

Bicycles; permits operators to treat a stop sign as a yield sign in certain situations. Amending §§ 46.2-839 and 46.2-905; adding § 46.2-903.1. (Patron–Hurst, HB 2262)

TRANSIENT TAX

Combined transient occupancy and food and beverage tax; for purposes of taxes that are currently authorized for Rappahannock and Madison Counties, the rate limit for such tax shall be the same as if the two taxes were imposed separately. Amending § 58.1-3842. (Patron–Hanger, SB 1438)

TRANSIENT TAX (continued)

Retail sales and transient occupancy taxes; definitions, taxes on transient room rentals shall be computed on the basis of the total charges or the total price paid for the use or possession of the room, etc. Amending §§ 58.1-602, 58.1-603, 58.1-3819, 58.1-3819.1, 58.1-3823, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, 58.1-3826, 58.1-3842, and 58.1-3843; adding §§ 2.2-2320.2, 58.1-612.2, and 58.1-3818.8. (Patron–Norment, SB 1398)

TRANSPORTATION

- Prevailing wage rate; clarifies that public works includes transportation infrastructure projects. Amending § 2.2-4321.3. (Patron–Krizek, HB 2327)
- State Corporation Commission; transportation electrification, utility recovery of certain costs, report. (Patron-Sullivan, HB 2282)
- Transit equity and modernization; Department of Rail and Public Transportation to study. (Patron-McQuinn, HJR 542)
- Transportation purposes; inspection of property to ascertain suitability of the property for highway and other transportation purposes. Amending §§ 25.1-203 and 33.2-1011. (Patron–Bell, SB 1260)

TRANSPORTATION, COMMITTEE ON

TREASURY, TREASURY BOARD, AND TREASURER, STATE

Treasury and State Treasurer, Department of the; surety bonds. Amending §§ 2.2-2809, 5.1-1.3, 10.1-2006, 21-163, 30-131, 33.2-205, 36-111, 42.1-16, 44-21, 46.2-202, 52-3, 53.1-11, 54.1-305, 58.1-201, and 60.2-109. (Patron–Hayes, HB 2223)

TREES

Trees; replacement and conservation during development, effective date. Amending §§ 15.2-961 and 15.2-961.1. (Patron-Guy, HB 2042)

Trees; replacement and conservation during development, projects located in a Chesapeake Bay Preservation Area to address recurrent flooding, effective date, report, effective clause. Amending §§ 15.2-961 and 15.2-961.1. (Patron–Marsden, SB 1393)

TUELL, MARY LOUISE

Tuell, Mary Louise; recording sorrow upon death. (Patron–Torian, HJR 741)

TUITION

Students; eligibility for in-state tuition and state financial assistance program. Amending § 23.1-506; adding § 23.1-505.1. (Patron–Lopez, HB 2123; Boysko, SB 1387)

UNCODIFIED LEGISLATION

- Affordable and market-rate housing; Department of Housing and Community Development to convene an advisory group to evaluate construction of internal, etc., dwelling units. (Patron–Samirah, HB 2053)
- Bob White Covered Bridge; Department of Transportation to work with the governing body of Patrick County and community groups interested in constructing a replica of Bridge. (Patron-Poindexter, HB 2024)
- Budget bill; appropriations for 2020-2022 biennium. Amending Chapter 56, 2020 Sp. I Acts. (Patron-Torian, HB 1800; Howell, SB 1100)
- Capital outlay plan; projects to be funded from general fund-supported resources, repeals existing six-year capital outlay for projects to be funded. Repealing Chapter 1134, 2020 Acts. (Patron-Howell, SB 1155)
- Capital outlay plan; repeals existing six-year capital outlay for projects to be funded. Repealing Chapter 1134, 2020 Acts. (Patron-Torian, HB 2177)
- Carbon Sequestration Task Force; established by Secretary of Natural Resources, jointly with the Secretary of Agriculture and Consumer Services, to convene a task force, report. (Patron-Lewis, SB 1374)
- Chamberlin Hotel at Fort Monroe; reverts certain property to the Commonwealth. Repealing Section 1 of Chapter 809, 1998 Acts. (Patron–Mugler, HB 2009)

UNCODIFIED LEGISLATION (continued)

- Child Care Subsidy Program, temporarily expanding Program to provide financial assistance for child care to families in need during public health emergency. (Patron–Filler-Corn, HB 2206)
- Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021; created. (Patron-Torian, HB 2178; Howell, SB 1145)
- Comprehensive review of computer science standards, courses, and pathways in public schools; Department of Education to perform, report. (Patron–Simonds, HB 1885)
- Concealed handgun permits; demonstration of competence, eligibility to apply for permit due to restrictions of COVID-19. (Patron–Runion, HB 2310)
- COVID-19; Department of Medical Assistance Services shall deem testing, treatment, and vaccination to be emergency services. (Patron–Lopez, HB 2124)
- Crosswalk design; Commissioner of Highways shall convene a work group to determine whether there should be model policies for design and installation, and, if so, establish recommendations. (Patron–Keam, HB 1841)
- Driving privileges, certain; Commissioner of DMV to reinstate privileges, and waive fees for individuals whose privileges were suspended for failure to pay court fines and costs in other jurisdictions. (Patron–Williams Graves, HB 2284)
- Education, Board of; temporary extension of the license of any individual licensed by Board whose license expires on June 30, 2021. (Patron-Ward, HB 1776)
- Electric utilities; nonjurisdictional customers, third party power purchase agreements. Amending Chapters 1187, 1188, 1189, 1193, 1194, and 1239, 2020 Acts. (Patron–Hurst, HB 2034; Edwards, SB 1420)
- Electric utilities; triennial review proceeding by SCC, fair rates of return. (Patron-Hudson, HB 1984)
- Employers; reporting outbreaks of COVID-19, effective clause. (Patron-Lewis, SB 1362)
- Farmers market food and beverage products; sales considered essential during state of emergency. (Patron-Wyatt, HB 2302)
- Fetal and Infant Mortality Review Team; Office of the Chief Medical Examiner of Department of Health shall convene a work group to develop a plan for the establishment of Team, report. (Patron–Ayala, HB 1950)
- George Mason University; management agreement with the Commonwealth. (Patron–Bulova, HB 1986; Barker, SB 1204)
- GO Virginia Grants; matching funds, repeals sunset provision. (Patron-Wampler, HB 2101)
- Gold; Secretary of Natural Resources, et al., shall establish a work group to study mining and processing, issuance of certain mining permits. (Patron–Guzman, HB 2213)
- Governor's Schools; Board of Education shall issue guidance on the governance of academic year, including admissions policies, and guidelines on diversity, etc. (Patron–Tyler, HB 2305)
- Hampton Roads Sanitation District; changes to the enabling act related to the addition of Northampton and Accomack Counties. Amending Chapter 66, 1960 Acts. (Patron–Bloxom, HB 2257)
- Harry F. Byrd, Sr., statue; Department of General Services to remove statue of former Virginia Governor and U.S. Senator from Capitol Square. (Patron–Jones, HB 2208)
- Health Standards of Learning; advanced directive education for high school students. (Patron-Kiggans, SB 1190)
- Individuals with intellectual and developmental disabilities; Department of Medical Assistance Services shall establish a work group to study and develop recommendations for permanent use of virtual supports and increasing access to virtual supports and services. (Patron–Suetterlein, SB 1472)
- Individuals with intellectual and developmental disabilities; Department of Medical Assistance Services to study and develop recommendations for use of virtual support, etc. (Patron–Runion, HB 2197)
- Inmates; Board of Local and Regional Jails to review services provided during pregnancy, etc., report. (Patron-Favola, SB 1300)
- Local school divisions; availability of virtual and in-person learning to all students. (Patron-Dunnavant, SB 1303)
- Loudoun County; County to enter into a contract with the Department of Health for the local administration of local health services. (Patron-Favola, SB 1221)

UNCODIFIED LEGISLATION (continued)

Maternal Health Data and Quality Measures, Task Force on; established, report. (Patron-Herring, HB 2111)

Motor Vehicles, Department of, and Supreme Court of Virginia; repeals reporting requirement. (Patron-Newman, SB 1277)

Online portal for tax practitioners; Department of Taxation shall analyze prospect of establishing, report. (Patron-Coyner, HB 2060)

Paid family leave; State Corporation Commission's Bureau of Insurance to review and make recommendations, report. (Patron–Favola, SB 1219)

Permit fee schedules; Department of Environmental Quality shall convene a work group to revise schedule for nonhazardous solid waste management facilities. (Patron-Petersen, SB 1210)

Produce safety; removes the sunset date. (Patron-Obenshain, SB 1194)

Recurrent Flooding Resiliency, Commonwealth Center for; Center shall evaluate development of Flood Resiliency Clearinghouse Program for coordinating flood mitigation solutions, report. (Patron–Hodges, HB 2187)

Refunding bonds; alters the principal and interest requirements, maturity date, and allowable discount for bonds previously issued, sunset clause. Amending first enactment of Chapters 265 and 408, 1992 Acts. (Patron–Torian, HB 2179; Howell, SB 1134)

Southwestern Virginia Mental Health Institute; Governor to lease a portion of property to Smyth County for a term of three years, responsibility of County, corrects tax map references. Amending Chapter 678, 2019 Acts. (Patron–O'Quinn, HB 2098; Pillion, SB 1429)

Special education; Board of Education to amend a certain regulation to remove the word "component" following the word "evaluation," thereby ensuring compliance with the relevant federal regulation. (Patron–Mugler, HB 2314)

State Corporation Commission; transportation electrification, utility recovery of certain costs, report. (Patron–Sullivan, HB 2282)

State parks; Department of Conservation and Recreation to develop recommendations for funding, report. (Patron–Orrock, HB 1804)

Students with disabilities; Department of Education to update its special education and related services, etc. (Patron–Mundon King, HB 2316)

Tazewell County; authorizes a quitclaim and release of interest and conveyance of an easement by Board of Wildlife Resources. (Patron–Wampler, HB 2252; Pillion, SB 1400)

Through-year growth assessment system; Board of Education shall establish, report, effective clause. (Patron-Dunnavant, SB 1357)

Traumatic brain injury; definition. (Patron-Wilt, HB 2182)

Uniform Statewide Building Code; Board of Housing and Community Development shall consider amendments to Code to address changes in the IECC relating to energy efficiency and conservation. (Patron–Kory, HB 2227)

Universal health care; Joint Commission on Health Care shall enter into a contract with a qualified entity to study options for financing. (Patron–Samirah, HB 2271)

Virginia Good Neighbor Next Door program; Virginia Housing Development Authority shall report recommendations for creating Program. (Patron–Convirs-Fowler, HB 2072)

Virginia Residential Landlord and Tenant Act; landlord remedies, noncompliance with rental agreement, payment plan, removal of sunset. Repealing second enactment of Chapter 46, 2020 Sp. I Acts. (Patron–Price, HB 1889)

Voter referendum; issuance of state general obligation bonds for school facility modernization, November 2022 general election, effective clause for provisions. (Patron–Stanley, SB 1109)

Waste Diversion and Recycling Task Force; Department of Environmental Quality to continue Task Force. (Patron–Hashmi, SB 1319)

Wetlands; extension of certain permits through 2021. (Patron-Cosgrove, SB 1143)

UNEMPLOYMENT COMPENSATION

Employee classification; provision of personal protective equipment in response to a disaster. Amending §§ 40.1-28.7:7 and 60.2-212; adding § 65.2-301.2. (Patron–Batten, HB 2134)

Unemployment compensation; failure to respond, continuation of benefits, repayment of overpayments. Amending §§ 60.2-528.1, 60.2-619, and 60.2-633. (Patron–Hudson, HB 2040)

UNEMPLOYMENT COMPENSATION (continued)

Unemployment compensation; under specific conditions related to the COVID-19 virus, work will not be deemed suitable and benefits will not be denied to any otherwise eligible individual for refusing to accept new work. Amending §§ 60.2-528 and 60.2-618. (Patron–Tran, HB 2037)

Virginia Employment Commission; communications with parties, use of electronic means, report. Adding § 60.2-121.1. (Patron-Tran, HB 2036)

UNGER, JOHN

Unger, John; commending. (Patron-LaRock, HJR 759)

UNITED STATES GOVERNMENT

Johns, Barbara Rose; Joint Committee of Congress on the Library requested to fill the Commonwealth's vacant spot in the National Statuary Hall Collection at the United States Capitol with a statue to commemorate. (Patron–Ward, HJR 525; Lucas, SJR 288)

Virginia Residential Landlord and Tenant Act; noncompliance with rental agreement, etc., prohibition on using negative credit information that arose during a closure of the United States Government against certain applications for tenancy, penalty. Amending § 55.1-1245. (Patron–Helmer, HB 1908)

UTILITY SERVICES

Electric utilities; advanced renewable energy buyers. Amending § 56-585.5. (Patron-Sullivan, HB 1907)

VAN DOREN, NANCY

Van Doren, Nancy; commending. (Patron-Hope, HJR 640)

VAUGHAN, GEORGE B.

Vaughan, George B.; recording sorrow upon death. (Patron-Bell, HJR 702)

VETERANS

Income tax, state; subtraction for low-income military veterans with a permanent service-connected disability. Amending § 58.1-322.02. (Patron–Wiley, HB 2337)

Military honor guards and veterans service organizations; paramilitary activities. Amending §§ 18.2-282 and 18.2-433.2. (Patron–Reeves, SB 1129)

Personal property tax; exemption for motor vehicle of a 100 percent disabled veteran. Adding § 58.1-3668. (Patron–Reeves, SB 1130)

Veterans Services, Department of; initiatives to reduce unemployment among veterans, comprehensive transition program. Amending § 2.2-2001.2. (Patron–Bell, SB 1279)

VICTIMS OF COVID-19 REMEMBRANCE DAY

Victims of COVID-19 Remembrance Day; designating as March 14, 2021, and each succeeding year thereafter. (Patron–McQuinn, HJR 605)

VICTIMS OF CRIME

Parole Board; Board, within seven days of making any decision regarding the parole of a prisoner, to provide written or electronic notice of such decision to the victim of crime. Amending §§ 53.1-136 and 53.1-155. (Patron–Obenshain, SB 1125)

Victims of crime; compensation, reporting requirement. Amending § 19.2-368.10. (Patron–Delaney, HB 1867)

Victims of crime; definitions, certifications for victims of qualifying criminal activity, disclosure of exculpatory information to a defendant in a criminal case. Adding §§ 9.1-1500, 9.1-1501, and 9.1-1502. (Patron–Surovell, SB 1468)

Victims of sex trafficking; clarifies definition, affirmative defense to prosecution for certain offenses. Adding § 18.2-361.1. (Patron–Brewer, HB 2234)

VIDEO AND AUDIO COMMUNICATIONS

Home care organizations; Board of Health to include in regulations governing organizations a provision for supervision of home care attendants providing personal care services by a licensed nurse through use of interactive audio or video technology. Amending § 32.1-162.12. (Patron–Head, HB 1831)

VIDEO AND AUDIO COMMUNICATIONS (continued)

Personal appearance by two-way electronic video and audio communication; entry of plea or nolle prosequi, adjudication of probation violations. Amending § 19.2-3.1. (Patron–Edwards, SB 1242)

Telemedicine service; coverage of telehealth services by an insurer, etc., services delivered through real-time audio-only telephone, report. Amending §§ 32.1-325, 38.2-3418.16, and 54.1-3303. (Patron–Adams, D.M., HB 1987)

VIRGINIA ACADEMY OF SCIENCE, ENGINEERING AND MEDICINE

Virginia Academy of Science, Engineering and Medicine; commending. (Patron-Hayes, HJR 675)

VIRGINIA ENERGY PLAN

Commonwealth Clean Energy Policy; established, repeals provisions relating to energy objectives and Commonwealth Energy Policy. Amending §§ 56-46.1, 56-585.1, 56-598, 56-601, 62.1-199, 67-103, 67-104, and 67-201; adding § 67-101.1; repealing §§ 67-101 and 67-102. (Patron–Favola, SB 1284)

Electric vehicle rebate program; creation and funding, report, sunset date. Adding §§ 67-1800 through 67-1806. (Patron–Reid, HB 1979)

Mines and Mining and Virginia Energy Plan; revision of Titles 45.1 and 67. Adding §§ 10.1-1332, 10.1-1333, 33.2-120, 33.2-221.1, 45.2-100 through 45.2-402, 45.2-500 through 45.2-1051, 45.2-1100 through 45.2-1505, 45.2-1600 through 45.2-1649, 45.2-1700 through 45.2-2119, 55.1-1820.1, 55.1-1951.1, 55.1-2133.1, and 56-614 through 56-624; repealing §§ 11-34.1 through 11-34.4, 45.1-161.1 through 45.1-399, 62.1-195.1, 62.1-195.3, and 67-100 through 67-1700. (Patron–Edwards, SB 1453)

Virginia Brownfield and Coal Mine Renewable Energy Grant Fund and Program; established, report. Adding § 67-1800. (Patron–Kilgore, HB 1925)

Virginia Energy Plan; amends Plan to include an analysis of electric vehicle charging infrastructure. Amending §§ 67-102, 67-201, and 67-202. (Patron–Boysko, SB 1223)

VIRGINIA HOUSE OF DELEGATES

Staffing levels, employment conditions, and compensation at the Virginia Department of Corrections, joint committee of various House and Senate Committees Studying; continued, appropriations. (Patron–Tyler, HJR 522)

VIRGINIA NATIONAL GUARD

License plates, special; removes fee for issuance to a member of the Virginia National Guard. Amending § 46.2-744. (Patron–Wyatt, HB 2261)

License plates, special; removes the fee for the issuance of a license plate for retired members of the Virginia National Guard. Amending § 46.2-746.5. (Patron–Walker, HB 1796)

Virginia National Guard; commending. (Patron-Filler-Corn, HJR 713)

VIRGINIA PUBLIC PROCUREMENT ACT

Virginia Public Procurement Act; bids or offers on construction contracts, requirement to submit list of subcontractors, effective clause. Adding § 2.2-4303.02. (Patron–Williams Graves, HB 2288)

Virginia Public Procurement Act; construction contracts, localities with a population in excess of 25,000, subcontractor workforce requirements. Adding § 2.2-4303.02. (Patron–McPike, SB 1305)

Virginia Public Procurement Act; determination of responsibility, local option to include criteria in Invitation to Bid. Amending § 2.2-4302.1. (Patron–Murphy, HB 1996)

Virginia Public Procurement Act; local arbitration agreements, definitions, procurement procedures, compliance for post-award actions, penalty. Adding §§ 2.2-4377.1 through 2.2-4377.5. (Patron–Surovell, SB 1384)

Virginia Public Procurement Act; preference for energy-efficient and water-efficient goods. Adding § 2.2-4328.1. (Patron-Helmer, HB 1811)

VIRGINIA RESIDENTIAL LANDLORD AND TENANT ACT

Virginia Residential Landlord and Tenant Act; access to dwelling unit during certain declared states of emergency, nonemergency repairs, employees and agents sent by landlord for maintenance or inspection are to wear all appropriate personal protective equipment as required by state law. Amending § 55.1-1229. (Patron–Carr, HB 1981)

VIRGINIA RESIDENTIAL LANDLORD AND TENANT ACT (continued)

- Virginia Residential Landlord and Tenant Act; landlord charges for security deposits, insurance premiums for damage insurance, etc., clerk of SCC shall provide a downloadable form for the filing of a resident agent appointment, etc. Amending §§ 17.1-275, 55.1-1200, 55.1-1204, 55.1-1206, 55.1-1208, 55.1-1211, 55.1-1226, 64.2-2008, and 64.2-2012. (Patron–McQuinn, HB 2249)
- Virginia Residential Landlord and Tenant Act; landlord remedies, noncompliance with rental agreement, payment plan, removal of sunset. Repealing second enactment of Chapter 46, 2020 Sp. I Acts. (Patron-Price, HB 1889)
- Virginia Residential Landlord and Tenant Act; noncompliance with rental agreement, etc., prohibition on using negative credit information that arose during a closure of the United States Government against certain applications for tenancy, penalty. Amending § 55.1-1245. (Patron–Helmer, HB 1908)
- Virginia Residential Landlord and Tenant Act; responsibilities of real estate brokers, foreclosure of single-family residential dwelling units. Amending §§ 54.1-2108.1 and 55.1-1237. (Patron–Simon, HB 2229)
- Virginia Residential Landlord and Tenant Act; sample termination notice, landlord's acceptance of rent with reservation, tenant's right of redemption. Amending §§ 36-139 and 55.1-1250. (Patron–Price, HB 2014)
- Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion from dwelling unit, interruption of services, or actions taken to make premises unsafe, repeals provision relating to tenant's remedies for landlord's unlawful ouster. Adding § 55.1-1243.1; repealing § 55.1-1243. (Patron–Hudson, HB 1900; Ebbin, SB 1215)

VIRGINIA RESIDENTIAL PROPERTY DISCLOSURE ACT

- Virginia Residential Property Disclosure Act; required disclosures for buyer to beware, mold assessment obtained by purchaser. Amending § 55.1-703. (Patron–Askew, HB 1824)
- Virginia Residential Property Disclosure Act; required disclosures for buyer to exercise due diligence, flood risk report, effective date and clause. Amending § 55.1-703; adding § 55.1-708.2. (Patron–Convirs-Fowler, HB 2320)

VIRGINIA STATE POLICE

Virginia State Police; commending. (Patron-Filler-Corn, HJR 712)

VITAL STATISTICS

Voter registration; list maintenance, lists of decedents transmitted by State Registrar of Vital Records to Department of Elections on a weekly basis. Amending §§ 24.2-408 and 24.2-427. (Patron–Kiggans, SB 1422)

VOGEL, JILL HOLTZMAN

Leave of absence)
Notified Clerk of presence	;
Participated remotely	5

VOTER REGISTRATION

- $\label{thm:continuous} \begin{tabular}{lll} Voter & registration; & failure & of & online & voter & registration & system, & deadline & extension. \\ & Amending § 24.2-416. & (Patron-VanValkenburg, HB 1810) \\ \end{tabular}$
- Voter registration; list maintenance, lists of decedents transmitted by State Registrar of Vital Records to Department of Elections on a weekly basis. Amending §§ 24.2-408 and 24.2-427. (Patron–Kiggans, SB 1422)
- Voter registration; preregistration for persons 16 years of age or older, effective date. Amending § 24.2-404; adding § 24.2-403.1. (Patron–Lopez, HB 2125)

VOTERS AND VOTING

- Absentee voting; accessibility for voters with a visual impairment or print disability. Amending § 24.2-704; adding § 24.2-103.2. (Patron–Reeves, SB 1331)
- Absentee voting; availability on Sundays in office of general registrar or voter satellite office. Amending § 24.2-701.1. (Patron–Bagby, HB 1968)

VOTERS AND VOTING (continued)

- Absentee voting; procedural and process reforms, availability and accessibility reforms, penalty. Amending §§ 24.2-101, 24.2-416.1, 24.2-603, 24.2-653.1, 24.2-704, 24.2-705, 24.2-706, 24.2-707, 24.2-708 through 24.2-711, and 24.2-712; adding §§ 24.2-103.2, 24.2-667.1, and 24.2-707.1. (Patron–VanValkenburg, HB 1888)
- Assistance for certain voters; curbside voting. Amending §§ 24.2-638, 24.2-646.1, and 24.2-649; adding § 24.2-649.1. (Patron–Price, HB 1921)
- Constitutional amendment; qualifications of voters, fundamental right to vote, persons not entitled to vote (first reference). Amending Section 1 of Article II. (Patron–Locke, SJR 272)
- Discrimination; prohibited in voting and elections administration, required process for enacting certain covered practices, civil causes of action, penalties, repeals provision relating to minority language accessibility, etc. Amending §§ 24.2-105, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1; adding §§ 24.2-104.1, 24.2-125 through 24.2-131, and 24.2-1005.2; repealing § 24.2-124. (Patron–Price, HB 1890; McClellan, SB 1395)
- Local elections for governing bodies; elections for school boards, qualification of voters. Amending §§ 15.2-1400, 22.1-29, 24.2-218, 24.2-222, and 24.2-223. (Patron–Convirs-Fowler, HB 2198)

WAGES

- Enterprise zone job creation grants; for purposes of wage requirements, the minimum wage shall be the higher of the state minimum wage or the federal minimum wage, delayed effective date. Amending § 59.1-547. (Patron–Heretick, HB 1881)
- Garnishment of wages; protected portion of disposable earnings. Amending § 34-29. (Patron-Krizek, HB 1814)
- Minimum wage; eliminates the exemption from Virginia's requirements for persons employed as farm laborers or farm employees and any individual employed as a temporary foreign worker. Amending § 40.1-28.9. (Patron–Ward, HB 1786)
- Virginia Human Rights Act; application of laws applicable to employee safety and payment of wages. Amending §§ 2.2-3905, 40.1-2, 40.1-29, 40.1-49.3, and 40.1-49.8. (Patron–McClellan, SB 1310)
- Virginia Initiative for Education and Work; participants, modifies Full Employment Program (FEP), wages earned by FEP employee during period for which his employer receives a subsidy. Amending § 63.2-608. (Patron–Tran, HB 2035)
- Virginia Jobs Investment Program and Fund; minimum wage requirements. Amending § 2.2-2240.3. (Patron-Bagby, HB 1967)
- Virginia Overtime Wage Act; overtime compensation employees, definitions, penalties. Amending §§ 40.1-29 and 40.1-29.1; adding § 40.1-29.2. (Patron–Mullin, HB 2063)

WALKER, ARTHUR WILLIAM

Walker, Arthur William; recording sorrow upon death. (Patron-Hudson, HJR 723)

WARREN, ADRIENNE

Warren, Adrienne; commending. (Patron-Hayes, HJR 694)

WARREN, DARRELL W., JR.

Warren, Darrell W., Jr.; commending. (Patron-Hodges, HJR 672)

WASHINGTON, ANDREW D.

Washington, Andrew D.; recording sorrow upon death. (Patron-Krizek, HJR 740)

WASHINGTON, D.C. METROPOLITAN AREA

Metropolitan Washington Airports Authority; repeals effective date for creation of Authority. Repealing § 5.1-178. (Patron–Edwards, SB 1214)

WASTE DISPOSAL

Hampton Roads area refuse collection authority; changes various requirements of the Southeastern Public Service Authority when budgeting or incurring debt, etc. Amending § 15.2-5102.1. (Patron–Cosgrove, SB 1141)

WASTE MANAGEMENT

Permit fee schedules; Department of Environmental Quality shall convene a work group to revise schedule for nonhazardous solid waste management facilities. (Patron–Petersen, SB 1210)

WATERS OF THE STATE, PORTS, AND HARBORS

- Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment, Enhanced Nutrient Removal Certainty Program established. Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14. (Patron–Lopez, HB 2129)
- Chesapeake Bay Phase III Watershed Improvement Plan; wastewater treatment, Enhanced Nutrient Removal Certainty Program established, regulations. Amending §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14. (Patron–Hanger, SB 1354)
- Environmental permits; community and environmental justice outreach, new individual permit application. Amending § 10.1-1408.1; adding §§ 10.1-1321.2, 10.1-1437.1, 62.1-44.6:1, and 62.1-266.1. (Patron–Hayes, HB 2221)
- Natural gas pipelines; Department of Environmental Quality to conduct inspections of the land-disturbing activities related to construction of any transmission pipeline, where substantial adverse impacts or likely adverse impacts are found on a repeated, etc., basis, the Department may issue a stop work instruction. Amending §§ 62.1-44.15:37.1 and 62.1-44.15:58.1. (Patron–Deeds, SB 1265)
- Natural Resources, Secretary of; changes name to the Secretary of Natural and Cultural Resources. Amending §§ 2.2-200, 2.2-205, 2.2-206, 2.2-215, 2.2-220.1, 2.2-220.2, 2.2-220.3, 2.2-1156, 2.2-1176, 2.2-2316, 2.2-2338, 2.2-2481, 2.2-2699.10, 2.2-2699.11, 2.2-2699.13, 10.1-603.25, 10.1-704, 10.1-1018, 10.1-1181.15, 10.1-1188, 10.1-1329, 10.1-1402.03, 10.1-1402.04, 10.1-1405, 10.1-2129, 10.1-2202.3, 15.2-2295.1, 28.2-207, 29.1-102, 29.1-573, 29.1-579, 30-377, 56-596.2, 58.1-344.3, 62.1-44.15:68, 62.1-44.34:25, 62.1-44.34:28, 62.1-44.117, 62.1-44.118, 62.1-44.119:1, 62.1-69.31, 62.1-69.32, 62.1-69.33, 62.1-69.41, 62.1-69.52, and 62.1-195.1. (Patron—Plum, HB 1836)
- Nutrient credits; use by facility with certain stormwater discharge permit. Amending § 62.1-44.19:21. (Patron–Bulova, HB 1982)
- Onsite Sewage Indemnification Fund; authorizes the State Board of Health to use Fund to provide grants and loans to property owners with income at or below 200 percent of the federal poverty guidelines to repair failing onsite sewage systems or install onsite sewage systems on properties that lack adequate sewage disposal. Amending §§ 32.1-164 and 32.1-164.1:01; adding §§ 62.1-223.1, 62.1-223.2, and 62.1-223.3. (Patron–Hashmi, SB 1396)
- Permit fee schedules; Department of Environmental Quality shall convene a work group to revise schedule for nonhazardous solid waste management facilities. (Patron-Petersen, SB 1210)
- Port of Virginia tax credits; extends the sunset date. Amending §§ 58.1-439.12:06, 58.1-439.12:09, and 58.1-439.12:10. (Patron–Spruill, SB 1158)
- Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron–Cole, J.G., HB 1819)
- Solar projects; State Water Control Board to administer a Virginia Erosion and Sediment Control Program (VESCP) on behalf of any locality that notifies the Department of Environmental Quality that it has chosen not to administer a VESCP, effective clause. Adding § 62.1-44.15:55.1. (Patron–Marsden, SB 1258)
- South River; designates a 6.5-mile segment in the City of Waynesboro as a component of the Virginia Scenic Rivers System. Adding § 10.1-417.1. (Patron–Avoli, HB 1958)
- Stormwater Local Assistance Fund; grants awarded for projects related to Chesapeake Bay total maximum daily load (TMDL) requirements. Amending § 62.1-44.15:29.1. (Patron–Lewis, SB 1404)
- Virginia Water Protection Permit; withdrawal of surface water or ground water, plans for water auditing and leak detection and repair. Amending §§ 62.1-44.15:22 and 62.1-262. (Patron–Mason, SB 1291)
- Water quality standards; modification of permits and certifications. Amending § 62.1-44.15:81. (Patron–McClellan, SB 1311)
- Wetland and stream mitigation banks; proximity of impacted site. Amending §§ 62.1-44.15:23 and 62.1-44.15:23.1. (Patron–Bulova, HB 1983)

WAYNESBORO, CITY OF

South River; designates a 6.5-mile segment in the City of Waynesboro as a component of the Virginia Scenic Rivers System. Adding § 10.1-417.1. (Patron–Avoli, HB 1958)

WEAPONS

- Muzzleloading rifle and shotgun; clarifies definitions. Amending § 29.1-100. (Patron-Fariss, HB 2298)
- School board property, certain; establishment of gun-free zone permitted. Adding § 22.1-131.1. (Patron–Subramanyam, HB 1909)
- Weapons; possessing or transporting within Capitol Square or into a building owned or leased by the Commonwealth, provisions shall not apply to certain law-enforcement officers, etc., penalty. Adding § 18.2-283.2. (Patron–Ebbin, SB 1381)

WEATHER

Public schools; severe weather conditions and other emergency situations, unscheduled remote learning days, school provides instruction and student services, etc. Amending § 22.1-98. (Patron–McNamara, HB 1790; Suetterlein, SB 1132)

WELFARE (SOCIAL SERVICES)

- Adult adoption; investigation and report. Amending § 63.2-1244. (Patron-Avoli, HB 1957)
- Child care providers; background checks portability, check shall not be required if individual completed a background check within previous five years, pilot program, report. Amending §§ 19.2-389, 22.1-289.035, 22.1-289.039, 63.2-1720.1, and 63.2-1724. (Patron–McGuire, HB 2086; McClellan, SB 1316)
- Child support obligations; party's incarceration not deemed voluntary unemployment or underemployment. Amending §§ 20-108.1 and 63.2-1918. (Patron–Scott, HB 2055)
- Child-placing agencies; repeals provisions that allowed agencies to refuse to perform, assist with, etc., in any child placements when the proposed placement would violate the agency's written religious or moral convictions or policies, any agency that operated prior to February 1, 2021, may continue to operate until January 1, 2022, etc. Repealing § 63.2-1709.3. (Patron–Levine, HB 1932)
- Confirmatory adoption; expands the stepparent adoption provisions, person with a legitimate interest. Amending § 63.2-1241. (Patron–Boysko, SB 1321)
- Emergency order for adult protective services; acts of violence, etc., or financial exploitation. Amending §§ 63.2-1603 and 63.2-1609. (Patron–Obenshain, SB 1297)
- Emergency order for adult protective services; definitions, acts of violence, etc., or financial exploitation. Amending §§ 63.2-1603, 63.2-1606, and 63.2-1609. (Patron–Mullin, HB 2018)
- Foster care; termination of parental rights, relatives and fictive kin. Amending §§ 16.1-281, 16.1-283, 63.2-906, and 63.2-910.2. (Patron–Gooditis, HB 1962)
- Hospitals, nursing homes, etc.; visits by clergy, priest, etc., during a declared public health emergency. Amending §§ 32.1-127, 32.1-162.5, and 63.2-1732. (Patron–Kiggans, SB 1356)
- Public schools, child day programs, and certain other programs; carbon monoxide detectors required. Adding §§ 22.1-138.2, 22.1-289.058, and 63.2-1705.2. (Patron–Askew, HB 1823)
- SNAP benefits program; eligibility for benefits, postsecondary education. Amending §§ 63.2-608 and 63.2-801. (Patron–Helmer, HB 1820)
- Social services, local department of; location of child in local department's custody. Amending § 63.2-104. (Patron–Leftwich, HB 2191)
- State-Funded Kinship Guardianship Assistance program; created, clarifies definition of "foster care services." Amending §§ 16.1-228, 16.1-282.1, 63.2-100, 63.2-905, 63.2-906, and 63.2-1305; adding § 63.2-1306. (Patron–Mason, SB 1328)
- Support orders; contents of orders, change in employment status, unemployment benefits. Amending §§ 20-60.3 and 63.2-1916. (Patron–Leftwich, HB 2192)
- Virginia Digital Equity Pilot Program and Fund; established, report, sunset date, effective clause. Adding § 63.2-806. (Patron–Mason, SB 1462)
- Virginia Initiative for Education and Work; participants, modifies Full Employment Program (FEP), wages earned by FEP employee during period for which his employer receives a subsidy. Amending § 63.2-608. (Patron-Tran, HB 2035)

WESTMORELAND COUNTY

Rappahannock River; designating a 79-mile portion located in Caroline, King George, Westmoreland, Essex, and Richmond Counties as a component of the Virginia Scenic Rivers System. Amending § 10.1-415. (Patron–Cole, J.G., HB 1819)

WETLANDS

Wetland and stream mitigation banks; proximity of impacted site. Amending §§ 62.1-44.15:23 and 62.1-44.15:23.1. (Patron–Bulova, HB 1983)

Wetlands; extension of certain permits through 2021. (Patron-Cosgrove, SB 1143)

WILDLIFE, INLAND FISHERIES AND BOATING

Balloons; release of nonbiodegradable balloons outdoors prohibited, civil penalty. Amending § 29.1-556.1. (Patron–Guy, HB 2159)

Muzzleloading rifle and shotgun; clarifies definitions. Amending § 29.1-100. (Patron-Fariss, HB 2298)

Trout fishing in stocked waters; equalizes for residents and nonresidents requirements to fish in designated waters. Amending § 29.1-311. (Patron–Pillion, SB 1402)

Wildlife corridors; various agencies to consider and incorporate, where applicable, government planning, Department of Wildlife Resources shall publish plans and any updates on its website. Amending §§ 10.1-207, 10.1-1105, 29.1-579, and 33.2-353. (Patron–Marsden, SB 1274)

WILLIAMSBURG FAITH IN ACTION

Williamsburg Faith in Action and the Arc of Greater Williamsburg; commending. (Patron–Mullin, HJR 755)

WILLIAMSBURG HEALTH FOUNDATION

Williamsburg Health Foundation; commending. (Patron–Mullin, HJR 751)

WILLS, TRUSTS, AND FIDUCIARIES

Estate planning documents; electronic execution, validity of will, codifies Uniform Electronic Wills Act. Amending §§ 32.1-291.7, 54.1-2982, 54.1-2983, 54.1-2984, 59.1-481, 64.2-100, 64.2-403, 64.2-404, 64.2-407, 64.2-450, and 64.2-701; adding §§ 64.2-459 through 64.2-468. (Patron–Sullivan, HB 1856)

Supported decision-making agreements; Department of Behavioral Health and Developmental Services to develop and implement a program to educate certain individuals. Amending §§ 64.2-2000, 64.2-2003, and 64.2-2007; adding § 37.2-314.3. (Patron–Bell, HB 2230)

Will contest; presumption of undue influence. Adding § 64.2-454.1. (Patron-Obenshain, SB 1123)

WINE

Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption, report, sunset provision. Amending §§ 4.1-119, 4.1-204, 4.1-206.1, 4.1-206.3, 4.1-207, 4.1-210, 4.1-212.1, and 4.1-221. (Patron–Bulova, HB 1879; Bell, SB 1299)

WINGFIELD, ROBERT GEORGE

Wingfield, Robert George; recording sorrow upon death. (Patron-McNamara, HJR 770)

WIRELESS COMMUNICATIONS

License restrictions for minors; prohibition on use of handheld personal communications devices. Amending § 46.2-334.01. (Patron–Robinson, HB 1846)

WOLF CREEK CHEROKEE TRIBE OF VIRGINIA

Wolf Creek Cherokee Tribe of Virginia; General Assembly to extend state recognition. (Patron–McQuinn, HJR 572)

WOMACK, WILLIAM HENRY, JR.

Womack, William Henry, Jr.; recording sorrow upon death. (Patron-McQuinn, HJR 735)

WOMEN

Inmates; Board of Local and Regional Jails to review services provided during pregnancy, etc., report. (Patron-Favola, SB 1300)

Small, women-owned, and minority-owned businesses; right to appeal denial of initial certification. Amending § 2.2-1606. (Patron–Mundon King, HB 2172)

WOMEN'S SUFFRAGE MONTH

Women's Suffrage Month; designating as August 2020 and each succeeding year thereafter. (Patron-Vogel, SJR 292)

WOOD, THELONIUS LEANDER

Wood, Thelonius Leander; recording sorrow upon death. (Patron-McQuinn, HJR 734)

WOODS, MICHAEL WAYNE

Woods, Michael Wayne; recording sorrow upon death. (Patron–Marshall, HJR 772)

WOODSON, WILLIE

Woodson, Willie; recording sorrow upon death. (Patron-McQuinn, HJR 733)

WORKERS' COMPENSATION

Employee classification; provision of personal protective equipment in response to a disaster. Amending §§ 40.1-28.7:7 and 60.2-212; adding § 65.2-301.2. (Patron–Batten, HB 2134)

Employees providing domestic service; application of laws applicable to employee safety and workers' compensation. Amending §§ 40.1-2, 40.1-49.3, 40.1-49.8, 65.2-101, and 65.2-305. (Patron–Gooditis, HB 2032)

Employer or other person; retaliatory discharge of employee prohibited. Amending §§ 40.1-27.3 and 65.2-308. (Patron–Carter, HB 1754)

Workers' compensation; adds salaried or volunteer emergency medical services personnel to the list of persons to whom, after five years of service, the occupational disease presumption for death caused by hypertension or heart disease applies, personnel operating in a locality that has legally adopted a resolution declaring that it will provide one or more of the presumptions. Amending § 65.2-402. (Patron–Heretick, HB 1818)

Workers' compensation; adds salaried or volunteer emergency medical services personnel to the list of persons to whom occupational disease presumption for death caused by hypertension or heart disease applies, provisions shall not apply to any individual who was diagnosed before July 1, 2021. Amending § 65.2-402. (Patron–Marsden, SB 1275)

Workers' compensation; claims not barred. Adding § 65.2-706.2. (Patron-Lewis, SB 1351)

Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of certain health care providers is an occupational disease. Amending § 65.2-402.1. (Patron–Hurst, HB 1985)

Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, emergency medical services personnel, law-enforcement officers, and correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Jones, HB 2207)

Workers' compensation; establishes a presumption that COVID-19 causing the death or disability of firefighters, law-enforcement officers or correctional officers is an occupational disease. Amending § 65.2-402.1. (Patron–Saslaw, SB 1375)

WORKFORCE

Apprenticeship training programs; Virginia Board of Workforce Development, et al., shall review availability of programs, etc., report. (Patron-Simonds, HB 1849)

Virginia Health Workforce Development Authority; mission of Authority, membership. Amending §§ 32.1-122.7 and 32.1-122.7:1. (Patron–Willett, HB 1976)

Workforce development; expands type of data sharing. Amending § 2.2-435.8. (Patron–Subramanyam, HB 1876)

WRIGHT, JAMES WILLIAM, SR.

Wright, James William, Sr.; recording sorrow upon death. (Patron–Hudson, HJR 705)

YOFFY, JAMES S.

Yoffy, James S.; commending. (Patron-VanValkenburg, HJR 664)

ZALESKI, LUCILLE MINCHINZaleski, Lucille Minchin; recording sorrow upon death. (Patron–Jones, HJR 766)

ZION BAPTIST CHURCH

Zion Baptist Church; commemorating its 125th anniversary. (Patron-Price, HJR 783)

ZONING

Zoning appeals, board of; any elected official of an incorporated town may serve on board of county in which member also resides. Amending § 15.2-2308. (Patron–Roem, HB 1898)

PATRONAGE OF SENATE LEGISLATION

S.B. 13. Capitol Square; possessing or transporting a weapon within Square, penalty.

Patron(s): Ebbin

2021 SENATE JOURNAL

Co-Patron(s): Delegate(s): Kory

S.B. 15. Weapons; carrying into building owned or leased by the Commonwealth, penalty.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Morrissey; Delegate(s): Kory

S.B. 27. Uninsured and underinsured motorist insurance policies; bad faith.

Patron(s): Petersen

Co-Patron(s): Delegate(s): Samirah

S.B. 32. Corporal punishment of a child with an object; penalty.

Patron(s): Petersen

Co-Patron(s): Delegate(s): Kory

S.B. 80. Student journalists; freedom of speech and the press.

Patron(s): Marsden

Co-Patron(s): Delegate(s): Kory

S.B. 91. Parole; application of statutes.

Patron(s): Edwards

Chief Co-Patron(s): Senator(s): McClellan; Delegate(s): Guzman

Incorporated Chief Co-Patron(s): Senator(s): Morrissey

Co-Patron(s): Delegate(s): Kory

S.B. 97. Virginia Fair Housing Law; unlawful discriminatory housing practices, effective clause.

Patron(s): McClellan

Co-Patron(s): Delegate(s): Kory

S.B. 128. DOE; pilot program, placement transition of certain students.

Patron(s): Suetterlein

Chief Co-Patron(s): Senator(s): Vogel Co-Patron(s): Senator(s): Dunnavant

S.B. 135. Children's Services Act; special education programs.

Patron(s): Stuart

Co-Patron(s): Delegate(s): Cole, M.L.

S.B. 143. Disabled veterans and surviving spouses; state subsidy of property tax exemptions.

Patron(s): Stuart

Co-Patron(s): Delegate(s): Cole, M.L.

S.B. 148. Driving under the influence; provisions regarding driving or operating a motor vehicle, etc.

Patron(s): Stuart

S.B. 150. Criminal fiscal impact statement; bills resulting in a net increase of period of imprisonment.

Patron(s): Howell

S.B. 166. Campaign finance; disbursement of surplus funds, prohibited conversion to personal use.

Patron(s): Saslaw

Co-Patron(s): Senator(s): Vogel; Delegate(s): Kory

S.B. 190. Children's Services Act; community policy and management teams, use of funds.

Patron(s): Peake

S.B. 191. Income tax, state; subtraction for active duty military income, etc.

Patron(s): Peake

S.B. 218. Income tax, state and corporate; tax credit for employers of National Guard members.

Patron(s): Suetterlein

Co-Patron(s): Senator(s): Chase

S.B. 223. Juvenile records; expungement.

Patron(s): Favola

Co-Patron(s): Senator(s): Boysko

S.B. 234. Human Resource Management, Department of; health insurance for local school

board employees.

Patron(s): Chafin

S.B. 249. School Divisions of Innovation; performance-based assessments.

Patron(s): Favola

S.B. 253. Deeds of trust; fiduciary duties.

Patron(s): Chafin

S.B. 274. Motion picture theaters; required open-captioned showings.

Patron(s): Barker

S.B. 285. Intentional or negligent infliction of injury or death; bystander claims for emotional distress.

Patron(s): Deeds

S.B. 291. Mortgage brokers; prohibition on dual compensation.

Patron(s): Deeds

S.B. 295. Employment; disclosure of terms to employees.

Patron(s): Marsden

S.B. 304. Animal shelters; euthanasia rate.

Patron(s): Stanley

S.B. 306. Criminal history information; destruction of information for certain charges and convictions.

Patron(s): Stanley

Incorporated Chief Co-Patron(s): Senator(s): Deeds

S.B. 317. Licensed inpatient nursing services; in-home follow-up care upon discharge.

Patron(s): Kiggans

S.B. 318. Balloons; reduces the number per hour that a person may release.

Patron(s): Kiggans

S.B. 326. Sentencing proceeding by the jury after conviction; recommendation of leniency.

Patron(s): Deeds

Co-Patron(s): Senator(s): Morrissey

S.B. 327. School boards, certain; cost of competing adjustment, eligibility.

Patron(s): Lewis

S.B. 346. Contractors, on-site superintendents; boiler and pressure vessel operator license.

Patron(s): Bell

S.B. 352. Guardianship and conservatorship; supported decision-making alternative.

Patron(s): Lucas

S.B. 353. Outdoor shooting ranges; prohibited adjacent to residential areas, exceptions, civil penalty.

Patron(s): Bell

S.B. 359. Gifts of real estate; title search required for recordation.

Patron(s): Cosgrove

S.B. 362. Hyperbaric oxygen therapy; data collection.

Patron(s): Dunnavant

Co-Patron(s): Senator(s): Boysko, Cosgrove, Mason

S.B. 363. Virginia Works Portal; created, report.

Patron(s): Dunnavant

Co-Patron(s): Senator(s): Chase

S.B. 366. Learning management system; Department of Education to obtain for public schools.

Patron(s): Dunnavant

S.B. 367. Student growth measurement system; Department of Education to obtain an individualized system.

Patron(s): Dunnavant

S.B. 382. Health insurance; coverage for prosthetic devices.

Patron(s): McPike

Co-Patron(s): Senator(s): Hanger

S.B. 387. Charitable gaming; creates special permit for the play of electronic versions of instant bingo, etc.

Patron(s): McPike

S.B. 390. Public schools; Standards of Learning assessments.

Patron(s): McPike

S.B. 409. Landfills, large; local impact.

Patron(s): Hashmi

S.B. 411. Employment health and safety standards; heat illness prevention, definition.

Patron(s): Hashmi

S.B. 419. Conservation of trees; Town of Vienna.

Patron(s): Petersen

S.B. 420. Public schools; seizure management and action plan, training.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Marsden; Delegate(s): Avoli, Batten, Hope

S.B. 427. Employee protection; discharge for protective order prohibited.

Patron(s): Saslaw

S.B. 446. Constitutional amendment; personal property tax exemption for motor vehicle of a disabled veteran.

Patron(s): Reeves

Chief Co-Patron(s): Senator(s): DeSteph

Incorporated Chief Co-Patron(s): Senator(s): Morrissey

Co-Patron(s): Senator(s): Chase, Peake; Delegate(s): Avoli, Murphy, Simon

S.B. 449. Death penalty; abolishes penalty, including those persons currently under a death sentence.

Patron(s): Surovell

Co-Patron(s): Senator(s): Boysko

S.B. 456. Income tax, state; subtraction for low-income military veterans with a service-connected disability.

Patron(s): Reeves

 $\label{eq:Chief Co-Patron} \textbf{Chief Co-Patron(s):} \ DeSteph$

Co-Patron(s): Senator(s): Chase, Morrissey, Peake, Pillion; Delegate(s): Avoli, Cole, M.L., Murphy, Simon

S.B. 457. Income tax, state and corporate; tax credit for employers of National Guard members.

Patron(s): Reeves

Chief Co-Patron(s): DeSteph

Co-Patron(s): Senator(s): Chase, Peake, Pillion; Delegate(s): Austin, Avoli, Batten, Bloxom, Brewer, Byron, Campbell, J.L., Campbell, R.R., Cole, M.L., Collins, Cox, Coyner, Davis, Edmunds, Fariss, Fowler, Freitas, Head, Hodges, Kilgore, Leftwich, McGuire, Miyares,

Morefield, Murphy, Orrock, Robinson, Rush, Simon, Walker, Wampler, Webert, Wilt, Wright, Wyatt

S.B. 460. Income tax, state and corporate; tax credit for employers of military spouses.

Patron(s): Reeves

Co-Patron(s): Senator(s): Peake; Delegate(s): Avoli, Murphy, Simon

S.B. 461. Veterans and active duty members of the Armed Forces; teachers, credit for service.

Patron(s): Reeves

Co-Patron(s): Senator(s): Peake; Delegate(s): Avoli, Cole, M.L., Helmer, Murphy, Scott, Simon

S.B. 464. Higher educational institutions; intercollegiate athletics, student-athletes, compensation.

Patron(s): Reeves

Co-Patron(s): Senator(s): Boysko, Morrissey, Peake

S.B. 474. Health care provider panels; vertically integrated carriers, public hospitals.

Patron(s): Petersen

S.B. 486. Food waste; to study VDACS sources, scale, and prevention of food waste.

Patron(s): Favola

S.B. 490. Firearms; purchase, possession, etc., following a conviction.

Patron(s): Favola

S.B. 502. Child support enforcement; Department of Social Services to distribute support payments.

Patron(s): Reeves

S.B. 510. Milk; definition, misbranding product, prohibition.

Patron(s): Reeves

Co-Patron(s): Senator(s): Suetterlein; Delegate(s): Cole, M.L., Orrock, Runion

S.B. 533. Pari-mutuel wagering; breakage, distribution for problem gambling treatment and support.

Patron(s): Reeves

S.B. 536. Incentives for telecommunications companies; provision of wireless broadband services.

Patron(s): Edwards

S.B. 537. Mandatory minimum punishment; limitation.

Patron(s): Edwards

S.B. 550. Dating relationship abuse; penalty.

Patron(s): Bell, Ebbin

S.B. 569. DBHDS and VSP; mobile applications, mental health and public safety.

Patron(s): Dunnavant

Co-Patron(s): Senator(s): Barker, Mason

S.B. 606. Virginia Retirement System; increased retirement allowance for certain judges.

Patron(s): McDougle

S.B. 618. Specialty dockets; referrals.

Patron(s): Deeds

S.B. 624. Geriatric prisoners; conditional release.

Patron(s): Spruill

Co-Patron(s): Senator(s): Morrissey

S.B. 626. Hazardous Substance Aboveground Storage Tank Fund; created.

Patron(s): Surovell

Chief Co-Patron(s): Senator(s): Lewis Co-Patron(s): Delegate(s): Kory

S.B. 627. Contractors to subcontractors; prompt payment.

Patron(s): Vogel

S.B. 637. Estate tax; reinstatement.

Patron(s): Surovell

S.B. 641. Civil action; sale of personal data.

Patron(s): Surovell

S.B. 660. Virginia Equal Pay Act; civil penalties.

Patron(s): Boysko

Co-Patron(s): Senator(s): McClellan

S.B. 669. Animal testing; breeding.

Patron(s): Boysko

Incorporated Chief Co-Patron(s): Senator(s): Stanley

Co-Patron(s): Delegate(s): Hope, Kory

S.B. 681. Criminal cases; compensation of experts, effective clause.

Patron(s): Mason

S.B. 699. Juries; fine for failure to respond to questionnaire.

Patron(s): Obenshain

S.B. 719. Virginia Minimum Wage Act; exclusions.

Patron(s): McClellan

S.B. 721. Orders of restitution; enforcement.

Patron(s): McClellan

S.B. 723. Bail; data collection and reporting standards, report, effective clause.

Patron(s): McClellan

S.B. 756. Income tax, corporate; combined reporting requirements.

Patron(s): Marsden

S.B. 765. Health insurance; provider contracts, business practices, adverse changes, penalties.

Patron(s): Barker

S.B. 777. Virginia Security for Public Deposits Act; collateral for public deposits.

Patron(s): Lewis

Co-Patron(s): Delegate(s): Cole, M.L.

S.B. 779. Sixth grade science curriculum; DOE & DEQ to update.

Patron(s): Lewis

S.B. 780. Campgrounds; inherent risks, liability.

Patron(s): Lewis

S.B. 785. Mortgage loan originators; exempts retailers of manufactured or modular homes,

etc.

Patron(s): Lewis

S.B. 802. Death penalty executions; imposes a moratorium.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Boysko, Howell; Delegate(s): Hope, Rasoul, Samirah

S.B. 803. Attorneys for the Commonwealth; compensation and collection of fees.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Deeds; Delegate(s): Samirah

S.B. 805. Robbery; definition, penalty.

Patron(s): Morrissey

Co-Patron(s): Delegate(s): Bagby, Hope

S.B. 808. Police and court records; expungement of records, misdemeanor and felony

convictions.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Howell; Delegate(s): Cole, M.L., Hope, Samirah

S.B. 810. Discretionary sentencing guidelines worksheets; use by juries.

Patron(s): Morrissey

Co-Patron(s): Delegate(s): Bagby, Hope, Samirah

S.B. 811. Sentencing in a criminal case; procedure for trial by jury, ascertainment of punishment, etc.

Patron(s): Morrissey

Co-Patron(s): Delegate(s): Samirah

S.B. 813. Virginia High Speed Chase Alert Program; created.

Patron(s): Morrissey

S.B. 819. Drug Treatment Court Act; authorization and availability.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Spruill; Delegate(s): Bagby, Samirah

S.B. 824. Alcoholic beverage control; distiller licenses, Internet orders and shipments.

Patron(s): Ruff

S.B. 826. Water and sewer service charges; tenant or lessee.

Patron(s): McDougle

S.B. 842. Electric energy; customer choice.

Patron(s): Petersen

S.B. 854. Virginia Urban Agriculture Advisory Council; created, report.

Patron(s): Petersen

S.B. 863. Southwestern Va. Mental Health Institute; Governor to lease a portion of property

to Smyth County.

Patron(s): Pillion

S.B. 867. Health care provider panels; vertically integrated carriers, providers.

Patron(s): Petersen

Co-Patron(s): Senator(s): Bell, Chafin

S.B. 887. Voter registration; automatic voter registration.

Patron(s): Ebbin

S.B. 898. Business advisory committee; established.

Patron(s): DeSteph

S.B. 906. Landlord and tenant; noncompliance as defense to action for possession for nonpayment of rent.

Patron(s): Stanley

Chief Co-Patron(s): Senator(s): Locke Co-Patron(s): Senator(s): McClellan

S.B. 914. Victim of human trafficking; petition for vacatur/expungement of convictions & police/court records.

Patron(s): Locke

S.B. 917. Reproductive health services; health benefit plans to cover costs of specified health care services.

Patron(s): Locke

Co-Patron(s): Senator(s): Boysko

S.B. 946. Medical assistance services; state plan to include doulas.

Patron(s): Locke

Co-Patron(s): Senator(s): McClellan

S.B. 947. Police and court records; expungement of records, acquittals.

Patron(s): Saslaw

Co-Patron(s): Senator(s): Morrissey

S.B. 965. Income tax, state; subtraction for certain active duty military income.

Patron(s): Peake

S.B. 973. Judges; maximum number in each judicial circuit and district.

Patron(s): Edwards

S.B. 983. Certificate of public need; definition of "medical care facility," facilities subject to review.

Patron(s): Lucas

S.B. 993. State Health Commissioner; local health directors, qualifications.

Patron(s): Locke

S.B. 1011. Transportation, Department of; application of design standards, modifications.

Patron(s): McDougle

S.B. 1021. Kings Highway Bridge; located in the City of Suffolk, is eligible for state of good repair funds.

Patron(s): Cosgrove

Chief Co-Patron(s): Senator(s): Lucas

S.B. 1033. Juvenile community correctional centers and facilities; establishment, placement of juveniles.

Patron(s): Locke

Co-Patron(s): Senator(s): Boysko, McClellan

S.B. 1042. Wills; presumption of undue influence.

Patron(s): Obenshain

S.B. 1049. Involuntary commitment; notice and participation, discharge plans.

Patron(s): Deeds

S.B. 1050. Hospitals; custody of person subject to emergency custody order, regulations.

Patron(s): Deeds

S.B. 1052. Virginia Charitable Gaming Board; regulations of the Board.

Patron(s): Reeves

S.B. 1060. For good cause shown or upon agreement of all parties; court may dismiss action without prejudice.

Patron(s): Petersen

S.B. 1061. C-PACE loans; residential dwellings and condominiums.

Patron(s): Petersen

Co-Patron(s): Senator(s): Boysko

S.B. 1062. Virginia Higher Education Funding Review Commission; established, duties,

report.

Patron(s): Petersen

S.B. 1068. Higher education; in-state tuition.

Patron(s): Kiggans

S.B. 1078. Federal accountability indicators; PSAT/NMSQT, PreACT.

Patron(s): Suetterlein

S.B. 1079. Board of Medicine; medically unnecessary chaperones.

Patron(s): Suetterlein

S.B. 1080. Urban Teacher Fund and Program; established.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Mason, McClellan

S.B. 1092. Performance guarantees, certain; provisions for periodic partial and final release.

Patron(s): Surovell

S.B. 1095. Alcoholic beverage control; prohibition on mixed beverages at strip clubs.

Patron(s): Morrissey

S.B. 1097. Absentee voting; witness signature not required.

Patron(s): Favola

 $\textbf{\textbf{Co-Patron(s):}} \ Senator(s): McClellan, Surovell; Delegate(s): Hope, Kory, Willett$

S.B. 1098. Unmanned aircraft; exempts an owner from the requirement to register.

Patron(s): Favola

Co-Patron(s): Senator(s): Cosgrove; Delegate(s): Kory

S.B. 1099. Children's Services Act; special education programs.

Patron(s): Stuart

Co-Patron(s): Delegate(s): Cole, J.G., Cole, M.L.

S.B. 1100. Budget Bill.

Patron(s): Howell

S.B. 1101. Presidential electors; National Popular Vote Compact.

Patron(s): Ebbin, Lucas

Co-Patron(s): Senator(s): Boysko, Hashmi, Howell, Saslaw, Surovell; Delegate(s): Adams, D.M., Ayala, Carr, Carter, Cole, J.G., Guzman, Hope, Kory, Levine, McQuinn, Plum, Price, Rasoul, Samirah, Simon, Simonds, VanValkenburg, Watts

S.B. 1102. Personal care aides; DMAS shall establish an orientation program for certain aides.

Patron(s): Locke

Co-Patron(s): Senator(s): McClellan; Delegate(s): Hope

S.B. 1103. Virginia Freedom of Information Act; Virginia Parole Board member votes.

Patron(s): Suetterlein

Chief Co-Patron(s): Senator(s): DeSteph

Co-Patron(s): Senator(s): Newman, Norment, Obenshain, Peake, Reeves, Ruff

S.B. 1104. Parole; notice and certification, monthly reports.

Patron(s): Obenshain

Incorporated Chief Co-Patron(s): Senator(s): DeSteph

Co-Patron(s): Senator(s): Cosgrove, Newman, Suetterlein, Surveell; Delegate(s): Cole, M.L.

S.B. 1105. Post-conviction relief; previously admitted scientific evidence, report.

Patron(s): Stanley

Co-Patron(s): Senator(s): Morrissey

S.B. 1106. Public School Assistance Fund and Program; created.

Patron(s): Stanley

Chief Co-Patron(s): Senator(s): McClellan

Co-Patron(s): Delegate(s): Avoli

S.B. 1107. Medical malpractice; limitation on recovery.

Patron(s): Stanley

Co-Patron(s): Senator(s): Surovell

S.B. 1108. General district courts; jurisdictional limits.

Patron(s): Stanley

S.B. 1109. Voter referendum; issuance of state general obligation bonds for school facility modernization.

Patron(s): Stanley

Co-Patron(s): Senator(s): McDougle

S.B. 1110. Real estate settlement agents; duties.

Patron(s): Spruill

S.B. 1111. Elections; preservation of order at the polls, powers of officers of election.

Patron(s): Spruill

Co-Patron(s): Senator(s): Boysko, Locke

S.B. 1112. Research and development expenses; tax credit available against the bank

franchise tax.

Patron(s): Locke
Co-Patron(s): Senator(s): Lucas

S.B. 1113. Communicating threats of death or bodily injury to a person with intent to

intimidate; penalty.
Patron(s): Spruill

Co-Patron(s): Delegate(s): Heretick, Kory

S.B. 1114. Children's Services Act; expands eligibility for state pool of funds.

Patron(s): Peake

S.B. 1115. Industrial hemp; increases maximum THC concentration.

Patron(s): Peake

Co-Patron(s): Delegate(s): Edmunds

S.B. 1116. State Health Commissioner; powers during an epidemic, vaccinations, religious tenets or practices.

Patron(s): Peake

Co-Patron(s): Delegate(s): Cole, M.L.

S.B. 1117. Immunizations; religious tenets or practices.

Patron(s): Peake

Co-Patron(s): Delegate(s): Cole, M.L.

S.B. 1118. Voter registration; verification of social security numbers, provisional registration

status.

Patron(s): Peake

Co-Patron(s): Delegate(s): Cole, M.L.

S.B. 1119. Law-enforcement agencies; body-worn camera systems.

Patron(s): Reeves

Chief Co-Patron(s): Senator(s): DeSteph

Co-Patron(s): Senator(s): Cosgrove, Surovell; Delegate(s): Cole, M.L.

S.B. 1120. County executive form of government; local budgets.

Patron(s): Reeves

Co-Patron(s): Senator(s): Deeds; Delegate(s): Cole, M.L.

S.B. 1121. Birth certificates; amending certificate, review of request.

Patron(s): Locke

Co-Patron(s): Senator(s): McClellan, Surovell

S.B. 1122. Habitual offenders; repeals remaining provisions of Habitual Offender Act.

Patron(s): Stanley

Chief Co-Patron(s): Senator(s): McClellan

Co-Patron(s): Senator(s): Morrissey

S.B. 1123. Will contest; presumption of undue influence.

Patron(s): Obenshain

S.B. 1124. Execution of wills; witnesses to a will required to be disinterested, definition.

Patron(s): Obenshain

S.B. 1125. Parole Board; notice of parole of prisoner to victim.

Patron(s): Obenshain

Chief Co-Patron(s): Senator(s): DeSteph

Co-Patron(s): Senator(s): Cosgrove, Newman, Suetterlein; Delegate(s): Cole, M.L.

S.B. 1126. Transportation District Commission of Hampton Roads; change in membership. Patron(s): Spruill

S.B. 1127. Charitable gaming; conduct of instant bingo, network bingo, pull tabs, and seal cards.

Patron(s): Reeves

Removed: Senator(s): Lucas

S.B. 1128. Norfolk, City of; amending charter, general updates.

Patron(s): Spruill

Co-Patron(s): Delegate(s): Guy

S.B. 1129. Military honor guards and veterans service organizations; paramilitary activities.

Patron(s): Reeves

Co-Patron(s): Delegate(s): Cole, M.L.

S.B. 1130. Personal property tax; exemption for motor vehicle of a 100 percent disabled veteran

Patron(s): Reeves

Chief Co-Patron(s): DeSteph

Co-Patron(s): Senator(s): Kiggans, Morrissey; Delegate(s): Cole, M.L.

S.B. 1131. Emergency Services and Disaster Law; limitation on duration of executive orders.

Patron(s): Suetterlein

Chief Co-Patron(s): Senator(s): DeSteph

Co-Patron(s): Senator(s): Dunnavant, Kiggans, Peake, Pillion, Reeves, Ruff, Stanley; Delegate(s): Byron, Head, Runion

S.B. 1132. Public schools; severe weather conditions and other emergency situations.

Patron(s): Suetterlein

Co-Patron(s): Senator(s): Dunnavant; Delegate(s): McNamara

S.B. 1133. Children's Services Act; eligibility for state pool of funds, pilot program.

Patron(s): Suetterlein

Co-Patron(s): Senator(s): Edwards; Delegate(s): Fowler, Head

S.B. 1134. Refunding bonds; alters the principal and interest requirements.

Patron(s): Howell

Co-Patron(s): Senator(s): McClellan

S.B. 1135. Dangerous dogs; restructures procedure for adjudication, penalty.

Patron(s): Marsden

Co-Patron(s): Delegate(s): Kory

S.B. 1136. License plates, special; repeals issuance of certain plates.

Patron(s): Marsden

Co-Patron(s): Delegate(s): Kory

S.B. 1137. Virginia Retirement System; retired law-enforcement officers employed as school

security officers.

Patron(s): Cosgrove

S.B. 1138. Sexually transmitted infections; infected sexual battery, penalty.

Patron(s): Locke, McClellan

Co-Patron(s): Senator(s): Surovell; Delegate(s): Carr

S.B. 1139. Animal-drawn vehicles; establishes equipment requirements for vehicles operating on the highways.

Patron(s): Peake

S.B. 1140. Gifts of real estate; requirements.

Patron(s): Cosgrove

S.B. 1141. Hampton Roads area refuse collection authority; financial planning.

Patron(s): Cosgrove

S.B. 1142. Marriage; persons who may celebrate rites, authorizes current members of the General Assembly.

Patron(s): Cosgrove

S.B. 1143. Wetlands; extension of certain permits through 2021.

Patron(s): Cosgrove

S.B. 1144. Aircraft; registration and licensure.

Patron(s): Stuart

S.B. 1145. Commonwealth of Virginia Higher Educational Institutions Bond Act of 2021;

created.

Patron(s): Howell

S.B. 1146. Income tax, state; conformity with the Internal Revenue Code.

Patron(s): Howell

Incorporated Chief Co-Patron(s): Senator(s): Petersen

S.B. 1147. Nurse Loan Repayment Program; expands eligibility for Program to include

certified nurse aides. **Patron(s):** Kiggans

Co-Patron(s): Delegate(s): Kory

PATRONAGE

S.B. 1148. Elections; date of June primary election.

Patron(s): Kiggans

Co-Patron(s): Senator(s): Spruill

S.B. 1149. Nursing homes; standards of care and staff requirements, regulations.

Patron(s): Kiggans

Co-Patron(s): Senator(s): Hashmi

S.B. 1150. Military Spouse Liaison; position created in Department of Veterans Services,

report.

Patron(s): Kiggans

Co-Patron(s): Senator(s): Boysko, Ebbin, Hashmi, McClellan, Morrissey, Peake, Reeves, Stanley; Delegate(s): Avoli, Campbell, R.R., Coyner, Davis, Delaney, Fowler, Guy, Helmer, Kory, McQuinn, Mundon King, Scott, Simonds, Subramanyam, Walker, Wiley, Willett

S.B. 1151. Income tax, state; subtraction for veteran retirement income.

Patron(s): Kiggans

S.B. 1152. Appomattox, Town of; amending charter, shifts local elections from May to

November, etc. **Patron(s):** Peake

S.B. 1153. Absentee voting; ballots to be sorted and results to be reported by precinct.

Patron(s): Suetterlein

Co-Patron(s): Senator(s): DeSteph, Kiggans, Surovell

S.B. 1154. Behavioral Health and Developmental Services, Commissioner of; reports to designated protection.

Patron(s): Favola

S.B. 1155. Capital outlay plan; repeals existing six-year capital outlay for projects to be funded.

Patron(s): Howell

S.B. 1156. Technology Development Grant Fund; created.

Patron(s): Howell

S.B. 1157. Municipal elections; shifting elections to November.

Patron(s): Spruill

Co-Patron(s): Delegate(s): Bagby, Convirs-Fowler, Hayes, Jenkins, Scott, Simonds, Sullivan

Removed: Delegate(s): Guy, Jones, J.C.

S.B. 1158. Port of Virginia tax credits; extends the sunset date.

Patron(s): Spruill

Co-Patron(s): Senator(s): Mason

S.B. 1159. Sick leave; use for the care of immediate family members.

Patron(s): Favola

Co-Patron(s): Delegate(s): Guzman, Kory, Willett

S.B. 1160. Removal of vehicles involved in accidents; lien of keeper of vehicles.

Patron(s): Suetterlein

Co-Patron(s): Senator(s): Edwards

S.B. 1161. Virginia Soil and Water Conservation Board; clarifies membership.

Patron(s): Hanger

Co-Patron(s): Delegate(s): Avoli

S.B. 1162. Tax credit; agricultural best management practices.

Patron(s): Hanger

Co-Patron(s): Senator(s): Lewis, Mason; Delegate(s): Avoli

S.B. 1163. Agricultural equipment; establishes a refundable individual and corporate income

tax credit.

Patron(s): Hanger

Co-Patron(s): Delegate(s): Avoli

S.B. 1164. Advanced recycling, etc.; definitions.

Patron(s): Hanger

Co-Patron(s): Senator(s): Lewis; Delegate(s): Avoli S.B. 1165. Death penalty; abolition of current penalty.

Patron(s): Surovell

Chief Co-Patron(s): Senator(s): Stanley

Co-Patron(s): Senator(s): Bell, Boysko, Ebbin, Edwards, Favola, Hashmi, Locke, McClellan, Morrissey, Spruill; Delegate(s): Carr, Hope, Kory, Krizek, Lopez, Sickles

S.B. 1166. Nursing professional; establishes a nonrefundable individual income tax credit.

Patron(s): Kiggans

S.B. 1167. Nursing, Board of; licensure or certification by endorsement for members of the U.S. military.

Patron(s): Kiggans

S.B. 1168. "Abused or neglected child"; definition.

Patron(s): Lucas

Co-Patron(s): Senator(s): Locke

- **S.B. 1169. Student driver safety;** driver education program shall include dangers of speeding. **Patron(s):** Norment
- **S.B. 1170. Isle of Wight County;** authorized to impose an additional local sales and use tax to support schools.

Patron(s): Norment

S.B. 1171. Conflict of Interests Act, State and Local Government; disclosure requirements.

Patron(s): Norment

S.B. 1172. Capitol Square Preservation Council; powers and duties, review and approval of plans for changes.

Patron(s): Norment

S.B. 1173. Tobacco Region Revitalization Commission; membership.

Patron(s): Norment

S.B. 1174. Diversity, Equity, and Inclusion, Director of; reporting requirements.

Patron(s): Norment

S.B. 1175. Brunswick County school board; appointed school board salaries.

Patron(s): Ruff

S.B. 1176. Barrier crimes; amends current requirements for DBHDS to provide, etc.

Patron(s): Ruff

S.B. 1177. Public weighmasters; license.

Patron(s): Ruff

S.B. 1178. Genetic counseling; repeals conscience clause.

Patron(s): Ebbin

S.B. 1179. Corrections Private Management Act; name change, private management prohibited.

Patron(s): Ebbin

Chief Co-Patron(s): Delegate(s): Guzman

Co-Patron(s): Delegate(s): Kory

S.B. 1180. Civil actions; actions filed on behalf of multiple similarly situated persons.

Patron(s): Surovell

S.B. 1181. Special immigrant juvenile status; jurisdiction.

Patron(s): Surovell

Chief Co-Patron(s): Senator(s): Boysko Co-Patron(s): Senator(s): Morrissey

S.B. 1182. Motor vehicle liability insurance; increases coverage amounts.

Patron(s): Surovell

Co-Patron(s): Delegate(s): Bagby

S.B. 1183. Property Owners' Association Act/Condominium Act; use of electronic means

for meetings and voting. **Patron(s):** Dunnavant

Co-Patron(s): Senator(s): Boysko

S.B. 1184. Standby guardianship; triggering event.

Patron(s): Deeds

Chief Co-Patron(s): Delegate(s): Hudson

S.B. 1185. Assisted living facilities; residents that are auxiliary grant recipients.

Patron(s): Dunnavant

Chief Co-Patron(s): Senator(s): Morrissey S.B. 1186. Landfill siting; historic preservation.

Patron(s): Hashmi

Co-Patron(s): Delegate(s): Carr, Ware

S.B. 1187. Physical therapy; extends time allowed for a therapist to evaluate and treat patients.

Patron(s): Hashmi

Chief Co-Patron(s): Senator(s): Kiggans

S.B. 1188. Virginia Agriculture Food Assistance Program and Fund; established and created.

Patron(s): Hashmi

Co-Patron(s): Senator(s): McClellan, Morrissey; Delegate(s): Carr

S.B. 1189. Occupational therapists; licensure.

Patron(s): Hashmi

Co-Patron(s): Senator(s): Kiggans

S.B. 1190. Health Standards of Learning; advanced directive education for high school students.

Patron(s): Kiggans

Co-Patron(s): Delegate(s): Kory, Simonds, Wiley

S.B. 1191. School nurses; excludes positions from certain requirements, school board to employ in each school.

Patron(s): Kiggans

Chief Co-Patron(s): Senator(s): Hashmi

Co-Patron(s): Senator(s): Dunnavant

S.B. 1192. Naturopathic doctors; Department of Health Professions to amend its regulations.

Patron(s): Kiggans

S.B. 1193. Dairy Producer Margin Coverage Premium Assistance Program; established.

Patron(s): Obenshain

S.B. 1194. Produce safety; removes the sunset date.

Patron(s): Obenshain

S.B. 1195. Motor vehicle insurance; underinsured motor vehicle, uninsured motorist coverage.

Patron(s): Obenshain

Co-Patron(s): Senator(s): Surovell

S.B. 1196. Teachers and other licensed school board employees; cultural competency training.

Patron(s): Locke

Co-Patron(s): Senator(s): Bell, Boysko, McClellan

S.B. 1197. Virginia housing opportunity; tax credit established.

Patron(s): Locke

Co-Patron(s): Senator(s): Boysko, McClellan

S.B. 1198. Government Data Collection and Dissemination Practices Act; license plate readers.

Patron(s): Petersen

Co-Patron(s): Senator(s): Suetterlein, Surovell; Delegate(s): Rasoul

S.B. 1199. Conservation easements; construction.

Patron(s): Petersen

Removed: Senator(s): Mason

S.B. 1200. Waste disposal; local approval.

Patron(s): Hashmi

Chief Co-Patron(s): Delegate(s): Ware

S.B. 1201. Energy storage systems; definitions, tax exemption, revenue share for systems.

Patron(s): Petersen

S.B. 1202. Uninsured and underinsured motorist insurance policies; bad faith.

Patron(s): Petersen

S.B. 1203. Hate crimes; associational relationships, penalty.

Patron(s): Hashmi

Co-Patron(s): Delegate(s): Carr

S.B. 1204. George Mason University; management agreement with the Commonwealth.

Patron(s): Barker

Co-Patron(s): Senator(s): Boysko, Howell, McPike, Petersen, Saslaw, Surovell; Delegate(s): Bulova, Helmer

S.B. 1205. Career fatigue and wellness in certain health care providers; programs to address, civil immunity.

Patron(s): Barker

S.B. 1206. Confidentiality of juvenile court records; exceptions.

Patron(s): Barker

Co-Patron(s): Delegate(s): Cole, M.L., Herring

S.B. 1207. Solar and energy storage projects; siting agreements throughout the Commonwealth.

Patron(s): Barker

Co-Patron(s): Senator(s): McClellan

S.B. 1208. Continuity of government; extends period of time that locality may provide after disaster, etc.

Patron(s): Barker

S.B. 1209. Subcontractor's employees; liability of general contractor for wages.

Patron(s): Petersen

S.B. 1210. Permit fee schedules; DEQ to revise current schedule for nonhazardous solid waste mgmt. facilities.

Patron(s): Petersen

S.B. 1211. Vehicle registration fees; imposes an additional fee to be deposited into Public Safety Trust Fund.

Patron(s): Edwards

Co-Patron(s): Senator(s): Boysko

S.B. 1212. New River Valley Passenger Rail Station Authority; creation of authority in

Planning District 4.

Patron(s): Edwards

Co-Patron(s): Senator(s): McClellan

S.B. 1213. Restricted licenses; DMV authorized to issue.

Patron(s): Edwards

Chief Co-Patron(s): Senator(s): DeSteph Co-Patron(s): Senator(s): Surovell

S.B. 1214. Metropolitan Washington Airports Authority; repeals effective date for creation of Authority.

Patron(s): Edwards

S.B. 1215. Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion from dwelling unit.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Boysko, Hashmi, McClellan

S.B. 1216. Crewe, Town of; amending charter, changes to charter including town council, elections and powers.

Patron(s): Ruff

S.B. 1217. Secondary state highway system; taking certain private roads into the system.

Patron(s): Ruff

S.B. 1218. Naturopathic doctors; Board of Medicine to license and regulate.

Patron(s): Petersen

S.B. 1219. Paid family and medical leave; SCC's Bureau of Insurance to review and make recommendations, report.

Patron(s): Favola

Co-Patron(s): Delegate(s): Kory

S.B. 1220. State facilities; admission of certain aliens.

Patron(s): Favola

Chief Co-Patron(s): Senator(s): Barker

Co-Patron(s): Senator(s): Boysko, McClellan, Surovell; Delegate(s): Rasoul

S.B. 1221. Loudoun & Prince William Counties & Manassas & Manassas Park Cities; operation of local health dept.

Patron(s): Favola

Co-Patron(s): Senator(s): Bell, Boysko

S.B. 1222. Virginia Public Procurement Act; purchase of personal protective equipment.

Patron(s): DeSteph

S.B. 1223. Va. Energy Plan; amends Plan to include an analysis of electric vehicle charging infrastructure.

Patron(s): Boysko

Chief Co-Patron(s): Senator(s): McClellan; Delegate(s): Sullivan

Co-Patron(s): Senator(s): Hashmi, Howell, Mason, Surovell; Delegate(s): Ayala, Reid

S.B. 1224. Uniform Statewide Building Code; amendments, energy efficiency and conservation.

Patron(s): Boysko

Co-Patron(s): Senator(s): Howell, McClellan; Delegate(s): Sullivan

S.B. 1225. Broadband services; school boards to appropriate funds for expansion of services for education.

Patron(s): Boysko

Chief Co-Patron(s): Senator(s): Hashmi, Lewis

Co-Patron(s): Senator(s): Chase, Howell, Marsden, Mason, McPike, Morrissey; Delegate(s): Carr, Gooditis

S.B. 1226. Compensation Board; determining staffing and salaries for an attorney for the Commonwealth.

Patron(s): Boysko

Co-Patron(s): Senator(s): Morrissey; Delegate(s): Ayala, Helmer, Hope, Hudson, Kory, McQuinn, Mugler, Mundon King, Murphy, Plum, Rasoul, Reid, Samirah, Scott, Simon, Simonds, Subramanyam, Sullivan, Watts

S.B. 1227. Hormonal contraceptives; payment of medical assistance for 12-month supply.

Patron(s): Boysko

Co-Patron(s): Senator(s): Hashmi, Locke, Marsden, McClellan; Delegate(s): Adams, D.M., Aird, Ayala, Carr, Carter, Hope, Hudson, Keam, Kory, Mundon King, Plum, Price, Rasoul, Reid, Samirah, Sickles, Simon, Simonds, Subramanyam, Sullivan, Watts

S.B. 1228. Virginia Equal Pay Act; civil penalties.

Patron(s): Boysko

Co-Patron(s): Senator(s): Hashmi, Locke, McClellan

S.B. 1229. License plates, special; issuance for supporters of Ducks Unlimited, fees.

Patron(s): Stuart

S.B. 1230. Criminal cases; transfer to general district court.

Patron(s): Chase

S.B. 1231. General district courts; filing an order of disposition from a criminal case.

Patron(s): Chase

S.B. 1232. Conditional release of geriatric prisoners; exceptions.

Patron(s): Chase

S.B. 1233. Campaign contribution limits; contributions that exceed \$20,000.

Patron(s): Petersen

S.B. 1234. Virginia State Bar examination; foreign applicants, evidence required.

Patron(s): Petersen

S.B. 1235. Health, Department of; certain communication prohibited.

Patron(s): Peake

Co-Patron(s): Senator(s): Newman; Delegate(s): McGuire

S.B. 1236. Campaign finance; prohibited contributions to candidates.

Patron(s): Petersen

Co-Patron(s): Delegate(s): Subramanyam

S.B. 1237. Emergency and quarantine orders, certain; additional procedural requirements.

Patron(s): Petersen

S.B. 1238. School security officers; carrying of firearms by United States Armed Forces veterans.

Patron(s): Chase

S.B. 1239. Absentee voting; third-party absentee ballot assembly and distribution.

Patron(s): Bell

Co-Patron(s): Delegate(s): Simon

S.B. 1240. Criminal sexual assault; penalty when committed by parents, etc., person is 13 yrs. of age, etc.

Patron(s): Stuart

S.B. 1241. Personal injury claim; disclosure of insurance policy limits.

Patron(s): Stuart

S.B. 1242. Personal appearance by two-way electronic video and audio communication; entry of plea.

Patron(s): Edwards

S.B. 1243. Marijuana; legalization of cultivation, manufacture, sale, possession, and testing, penalties.

Patron(s): Morrissey

S.B. 1244. Sexually violent predators; repeals sections authorizing civil commitment.

Patron(s): Morrissey

Chief Co-Patron(s): Delegate(s): Hope

S.B. 1245. Absentee voting; establishment of drop-off locations preprocessing returned absentee ballots.

Patron(s): Deeds

Co-Patron(s): Senator(s): Boysko, Ebbin, McClellan, Surovell

S.B. 1246. Absentee voting; mandatory processing of returned absentee ballots before election

day.

Patron(s): Deeds

Incorporated Chief Co-Patron(s): Senator(s): Saslaw

Co-Patron(s): Senator(s): Boysko, Ebbin, McClellan, Surovell

S.B. 1247. Electric generating facility closures; public disclosure, integrated resource plans.

Patron(s): Deeds

S.B. 1248. Juveniles; competency evaluation, receipt of court order.

Patron(s): Deeds

S.B. 1249. Local planning commissions; review deadlines.

Patron(s): Stuart

S.B. 1250. Firearm rentals; criminal history record information check required, penalty.

Patron(s): Deeds

Co-Patron(s): Senator(s): McClellan

S.B. 1251. Virginia Retirement System; technical amendments.

Patron(s): Newman

S.B. 1252. Coal tax credits; sunset date.

Patron(s): McPike

Co-Patron(s): Delegate(s): Plum

S.B. 1253. Access roads to economic development sites; criteria for use of funds.

Patron(s): McPike

Co-Patron(s): Delegate(s): Plum

S.B. 1254. Sports betting; clarifies certain procedures.

Patron(s): McPike

S.B. 1255. SCC; issuance or renewal of insurance licenses or registrations during an emergency.

Patron(s): Mason

S.B. 1256. Criminal Justice Services Board and Committee on Training; membership.

Patron(s): Marsden

S.B. 1257. SOQ; school board to provide at least three specialized student support positions.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Hashmi; Delegate(s): Kory

S.B. 1258. Solar projects; erosion and sediment control.

Patron(s): Marsden

Co-Patron(s): Senator(s): McClellan

S.B. 1259. Virginia Highway Corporation Act; alteration of certificate of authority, powers

and duties of SCC.

Patron(s): Bell

Chief Co-Patron(s): Senator(s): Boysko

Co-Patron(s): Delegate(s): Reid, Subramanyam

S.B. 1260. Transportation purposes; entry onto land for inspection.

Patron(s): Bell

S.B. 1261. Court of Appeals; expands jurisdiction, increases from 11 to 17 number of judges

on Court.

Patron(s): Edwards

Co-Patron(s): Senator(s): Morrissey, Surovell

S.B. 1262. Restricted permit; prepayment of fines and costs.

Patron(s): Morrissey

S.B. 1263. Bicycles; permits operators to treat a stop sign as a yield sign in certain situations.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Boysko, Surovell; Delegate(s): Carter, Hope, Rasoul, Simonds

S.B. 1264. Emergency and preliminary protective orders; expungement of orders.

Patron(s): Morrissey

S.B. 1265. Natural gas pipelines; stop work orders.

Patron(s): Deeds

Co-Patron(s): Senator(s): Favola, McClellan, Suetterlein; Delegate(s): Kory

S.B. 1266. Admission to bail; rebuttable presumptions against bail.

Patron(s): Deeds

Chief Co-Patron(s): Senator(s): Morrissey Co-Patron(s): Senator(s): McClellan

S.B. 1267. Covington, City of; amending charter, consolidated school division, salaries.

Patron(s): Deeds

S.B. 1268. Disposition of the remains of a decedent; persons to make arrangements for funeral.

Patron(s): Deeds

S.B. 1269. Health insurance; authorization of drug prescribed for the treatment of a mental disorder.

Patron(s): McPike

Co-Patron(s): Senator(s): Boysko

S.B. 1270. Eminent domain; notice of intent to file certificate.

Patron(s): Cosgrove

S.B. 1271. Virginia Freedom of Information Act; meetings held through electronic communication means.

Patron(s): McPike

Co-Patron(s): Senator(s): Ebbin; Delegate(s): Gooditis, Sullivan

S.B. 1272. Unrestorably incompetent defendant; disposition, capital murder charge, inpatient custody.

Patron(s): Mason

S.B. 1273. Behavioral Health Commission; established, report.

Patron(s): Deeds

Co-Patron(s): Senator(s): Hanger, Mason; Delegate(s): Bell, Hope, Price, Ransone, Sickles, Watts

S.B. 1274. Wildlife corridors; various agencies to consider and incorporate.

Patron(s): Marsden

Co-Patron(s): Delegate(s): Edmunds

S.B. 1275. Workers' compensation; presumption of compensability for certain diseases.

Patron(s): Marsden

S.B. 1276. Essential health benefits; abortion coverage.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Favola, Hashmi, Locke; Delegate(s): Carr

S.B. 1277. Motor Vehicles, Department of, and Supreme Court of Virginia; repeals reporting requirement.

Patron(s): Newman

S.B. 1278. Charitable gaming; regulations of operators of Texas Hold'em poker games.

Patron(s): Bell

S.B. 1279. Veterans Services, Department of; initiatives to reduce unemployment among veterans.

Patron(s): Bell

Co-Patron(s): Delegate(s): Reid

S.B. 1280. Dams; negotiated settlement agreements.

Patron(s): Bell

S.B. 1281. General registrar; qualifications, residency.

Patron(s): Morrissey

S.B. 1282. Greenhouse gas emissions inventory; regulations.

Patron(s): Morrissey

Co-Patron(s): Delegate(s): Kory

S.B. 1283. Criminal records, certain; establishes a process for the automatic expungement,

report.

Patron(s): Morrissey

Co-Patron(s): Delegate(s): Helmer, Hope, Kory, Rasoul, Simon, Simonds

S.B. 1284. Commonwealth Clean Energy Policy; established.

Patron(s): Favola

Co-Patron(s): Senator(s): Morrissey; Delegate(s): Kory

S.B. 1285. Vacant buildings; registration.

Patron(s): Locke

S.B. 1286. Income tax, state; rate increase, funding for schools and law-enforcement officer salaries.

Patron(s): Deeds

S.B. 1287. Charitable Gaming Board; regulations, electronic pull tabs.

Patron(s): McPike

S.B. 1288. Special education; Department of and the Board of Education to develop new policies and procedures.

Patron(s): Dunnavant

S.B. 1289. Health insurance; carrier business practices, provider contracts.

Patron(s): Surovell

S.B. 1290. ConserveVirginia program; established.

Patron(s): Mason

S.B. 1291. Va. Water Protection Permit; withdrawal of surface water or ground water, plans for water auditing.

Patron(s): Mason

S.B. 1292. Electric utilities; overearnings, customer bill credits.

Patron(s): McClellan

S.B. 1293. Parole Board; monthly reports.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Reeves, Stanley

S.B. 1294. Police and court records; fee for filing a petition for expungement.

Patron(s): DeSteph

S.B. 1295. Electric utilities; procurement of certain equipment.

Patron(s): DeSteph

S.B. 1296. Emergency Management Equity Working Group; established.

Patron(s): Spruill

S.B. 1297. Emergency order for adult protective services; acts of violence, etc., or financial exploitation.

Patron(s): Obenshain

Co-Patron(s): Senator(s): Mason, Surovell

S.B. 1298. Tourism improvement districts; authorizes any locality to create.

Patron(s): Bell

Chief Co-Patron(s): Senator(s): Lewis

Co-Patron(s): Senator(s): Boysko, Lucas, Mason, Pillion; Delegate(s): Heretick, Mugler, Reid, Subramanyam

S.B. 1299. Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine.

Patron(s): Bell

Chief Co-Patron(s): Senator(s): DeSteph

Incorporated Chief Co-Patron(s): Senator(s): Reeves

Co-Patron(s): Senator(s): Mason

S.B. 1300. Inmates; Board of Local and Regional Jails to review services provided during pregnancy, etc.

Patron(s): Favola

Co-Patron(s): Senator(s): Boysko, Ebbin, Hashmi, McClellan; Delegate(s): Hope, Keam, Kory

S.B. 1301. Correctional facilities; use of isolated confinement.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Ebbin; Delegate(s): Cole, J.G., Kory

S.B. 1302. Crisis Call Center Fund; created, collection of 988 charges.

Patron(s): McPike

S.B. 1303. Local school divisions; availability of virtual and in-person learning to all students.

Patron(s): Dunnavant

Co-Patron(s): Senator(s): DeSteph, Kiggans, Morrissey, Petersen, Pillion, Reeves, Ruff, Stanley; Delegate(s): Miyares

S.B. 1304. Community services boards; discharge planning.

Patron(s): McPike

Co-Patron(s): Senator(s): Boysko

S.B. 1305. Virginia Public Procurement Act; construction contracts, subcontractor workforce requirements.

Patron(s): McPike

S.B. 1306. Assault and battery; penalty.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Surovell

S.B. 1307. School-based health services; Bd. of MAS to amend state plan for services to provide for payment.

Patron(s): Dunnavant

S.B. 1308. Pedestrians; interference with traffic, penalty.

Patron(s): DeSteph

S.B. 1309. Local stormwater assistance; flood mitigation and protection.

Patron(s): Ebbin

Co-Patron(s): Delegate(s): Levine

S.B. 1310. Virginia Human Rights Act; application of laws applicable to employee safety and payment of wages.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Hashmi, Locke, Lucas; Delegate(s): Bourne, Carr

S.B. 1311. Water quality standards; modification of permits and certifications.

Patron(s): McClellan

S.B. 1312. Virginia Spirits Promotion Fund; assessment on distillers.

Patron(s): Mason

S.B. 1313. Children's Services Act; funds expended special education programs.

Patron(s): Mason

Incorporated Chief Co-Patron(s): Senator(s): Peake, Stuart

Co-Patron(s): Delegate(s): Fowler, Kory

S.B. 1314. Education and Labor Market Alignment, Office of; established.

Patron(s): Hashmi

Co-Patron(s): Senator(s): Boysko, Ebbin, Mason, McClellan

S.B. 1315. Criminal proceedings; consideration of mental condition and intellectual, etc.

Patron(s): McClellan

Incorporated Chief Co-Patron(s): Senator(s): Favola

Co-Patron(s): Senator(s): Boysko, Hashmi, Locke, Lucas, Morrissey; Delegate(s): Carr, Keam, Willett

S.B. 1316. Child care providers; background checks, portability.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Hashmi, Locke, Mason

S.B. 1317. Local school boards; open enrollment policy required.

Patron(s): Dunnavant

S.B. 1318. Interagency Environmental Justice Working Group; established, report.

Patron(s): Hashmi

Co-Patron(s): Senator(s): Boysko, McClellan; Delegate(s): Aird, Simonds

S.B. 1319. Waste Diversion & Recycling Task Force; Department of Environmental Quality

to continue Task Force. Patron(s): Hashmi

Co-Patron(s): Delegate(s): Ware

S.B. 1320. Licensed certified midwives; clarifies definition, licensure, etc.

Patron(s): Lucas

Co-Patron(s): Senator(s): Hashmi, Kiggans, McClellan, Spruill; Delegate(s): Kory

S.B. 1321. Confirmatory adoption; expands the stepparent adoption provisions.

Patron(s): Boysko

Co-Patron(s): Senator(s): Barker, Mason, McClellan, Morrissey; Delegate(s): Carr

S.B. 1322. Public schools; seizure management and action plans, biennial training, effective date.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Marsden; Delegate(s): Askew, Avoli, Batten

S.B. 1323. Worker classification; qualification of independent contractors.

Patron(s): Dunnavant

S.B. 1324. Actions against real estate appraisers or appraisal management companies;

statute of limitations. Patron(s): Dunnavant

S.B. 1325. Visitation; petition of grandparent.

Patron(s): Dunnavant

Chief Co-Patron(s): Senator(s): Morrissey

S.B. 1326. Local cigarette taxes; regional cigarette tax boards.

Patron(s): Hanger

Co-Patron(s): Delegate(s): Watts

S.B. 1327. Homeowners and tenants of manufactured home parks; housing protections,

foreclosures, etc. Patron(s): McClellan

Co-Patron(s): Senator(s): Hashmi, Mason; Delegate(s): Bourne, Carr, Torian

S.B. 1328. State-Funded Kinship Guardianship Assistance program; created.

Patron(s): Mason

S.B. 1329. Summons; promises to appear after issuance.

Patron(s): Mason

Co-Patron(s): Senator(s): Suetterlein, Surovell

S.B. 1330. Paid family and medical leave program; Virginia Employment Commission to

establish.

Patron(s): Boysko

Chief Co-Patron(s): Senator(s): McClellan

Co-Patron(s): Senator(s): Hashmi, Locke, Morrissey; Delegate(s): Kory

S.B. 1331. Absentee voting; accessibility for voters with a visual impairment or print disability.

Patron(s): Reeves

Co-Patron(s): Delegate(s): Hayes, Kory

S.B. 1332. Law-enforcement officer; use of deadly force during an arrest or detention.

Patron(s): Reeves

S.B. 1333. Pharmaceutical processors; permits processors to produce & distribute cannabis products.

Patron(s): Lucas

Co-Patron(s): Senator(s): Dunnavant, Ebbin, McPike, Morrissey

S.B. 1334. Broadband capacity; expands existing pilot program, municipal broadband authorities.

Patron(s): Edwards

Co-Patron(s): Senator(s): Lewis, Surovell

S.B. 1335. Learner's permits; use of personal communication devices, restrictions.

Patron(s): Stuart

Co-Patron(s): Senator(s): Surovell

S.B. 1336. Ignition interlock systems; restricted permits to operate a motor vehicle.

Patron(s): Stuart

Co-Patron(s): Senator(s): Boysko, Surovell

S.B. 1337. Golf carts and utility vehicles; operation of vehicles on public highways in County of Westmoreland.

Patron(s): Stuart

S.B. 1338. Telemedicine; coverage of telehealth services by an insurer, etc.

Patron(s): Barker

Incorporated Chief Co-Patron(s): Senator(s): Stanley

Co-Patron(s): Senator(s): Mason; Delegate(s): Cole, M.L., Kory

S.B. 1339. Criminal records; sealing of records, Sealing Fee Fund created, penalties, report.

Patron(s): Surovell

Incorporated Chief Co-Patron(s): Senator(s): Lucas, Morrissey

Co-Patron(s): Delegate(s): Hope, Kory

S.B. 1340. Medical debt; statute of limitations to collect.

Patron(s): Hashmi

Co-Patron(s): Senator(s): Surovell

S.B. 1341. Health insurance; association health plan for real estate salespersons.

Patron(s): Barker

Chief Co-Patron(s): Senator(s): Vogel

Co-Patron(s): Senator(s): DeSteph, Dunnavant, Mason, Suetterlein; Delegate(s): Helmer

S.B. 1342. Workers' compensation; presumption of compensability for COVID-19.

Patron(s): Vogel

S.B. 1343. Virginia Freedom of Information Act; proprietary records and trade secrets.

Patron(s): Vogel

Co-Patron(s): Delegate(s): Guzman

S.B. 1344. Overdoses; arrest and prosecution when experiencing or reporting.

Patron(s): Vogel

S.B. 1345. Alcoholic beverage control; license applications, notice requirements.

Patron(s): Vogel

S.B. 1346. Alcoholic beverage control; creates a mobile retailer license.

Patron(s): Stuart

S.B. 1347. Motorcycles; authorizes the use of any color auxiliary lighting, other than blue.

Patron(s): Reeves

S.B. 1348. Local advisory referenda; exception to rule that prohibits referendum from being placed on ballot.

Patron(s): Newman

S.B. 1349. Government Data Collection and Dissemination Practices Act; exemptions, etc.

Patron(s): Newman

S.B. 1350. Transportation funding; statewide prioritization process, resiliency.

Patron(s): Lewis

S.B. 1351. Workers' compensation; claims not barred.

Patron(s): Lewis

Co-Patron(s): Senator(s): Surovell

S.B. 1352. Flood Control and Commonwealth Flood Control Board, Department of; established, report.

Patron(s): Lewis

S.B. 1353. Income tax, corporate; combined reporting requirements.

Patron(s): Marsden

S.B. 1354. Chesapeake Bay; wastewater treatment, Enhanced Nutrient Removal Certainty Program established.

Patron(s): Hanger

Chief Co-Patron(s): Senator(s): Morrissey

Co-Patron(s): Senator(s): Favola, Lewis, Mason, McPike

S.B. 1355. Rapidan Service Authority; withdrawal of a member.

Patron(s): Hanger

S.B. 1356. Hospitals, nursing homes, etc.; visits by clergy.

Patron(s): Kiggans

Co-Patron(s): Senator(s): Suetterlein

S.B. 1357. Standards of Learning; reading & mathematics assessments for students in grades three through eight.

Patron(s): Dunnavant

S.B. 1358. Employment health and safety standards; heat illness prevention.

Patron(s): Hashmi

S.B. 1359. Higher educational institutions, public; fees levied for collegiate athletics optional for students.

Patron(s): DeSteph

S.B. 1360. Virginia Human Rights Act; nondiscrimination in employment, sexual harassment.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Ebbin; Delegate(s): Carr, Watts

S.B. 1361. Law-enforcement civilian oversight bodies; person appointed shall be U.S. citizen, etc.

Patron(s): Reeves

S.B. 1362. Employers; reporting outbreaks of COVID-19, effective clause.

Patron(s): Lewis

Co-Patron(s): Delegate(s): Kory

S.B. 1363. Local and Regional Jails, State Board of; membership, powers and duties, effective clause.

Patron(s): Marsden

Co-Patron(s): Senator(s): Boysko, Mason, McClellan, Morrissey

S.B. 1364. Commonwealth Transit Ridership Fund; created.

Patron(s): Cosgrove

S.B. 1365. Data Governance and Analytics, Office of; created.

Patron(s): Barker

Co-Patron(s): Senator(s): Dunnavant

S.B. 1366. Aging services; economic and social needs.

Patron(s): Barker

Co-Patron(s): Senator(s): Mason; Delegate(s): Carr

S.B. 1367. Line of Duty Act; Virginia licensed health practitioners required to conduct medical reviews.

Patron(s): DeSteph

S.B. 1368. Abuse and neglect of children; contact with other parent, penalty.

Patron(s): DeSteph

S.B. 1369. Small Business and Supplier Diversity, Department of; redefines "small business."

D () O1 1

Patron(s): Obenshain

Chief Co-Patron(s): Senator(s): Deeds

Co-Patron(s): Senator(s): Hanger, Reeves, Stanley

S.B. 1370. Parole statutes; application for juveniles and persons committed upon certain

felony offenses.

Patron(s): Edwards

Chief Co-Patron(s): Senator(s): Morrissey

S.B. 1371. Safe haven protections; newborn safety device.

Patron(s): Ruff

S.B. 1372. Criminal records; establishes a process for automatic expungement for certain

convictions, report.

Patron(s): Lucas

Co-Patron(s): Senator(s): Spruill

S.B. 1373. Environmental permits; community and environmental justice outreach.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Locke; Delegate(s): Carr

S.B. 1374. Carbon Sequestration Task Force; established.

Patron(s): Lewis

Co-Patron(s): Senator(s): McClellan; Delegate(s): Willett

S.B. 1375. Workers' compensation; presumption of compensability for COVID-19.

Patron(s): Saslaw

Incorporated Chief Co-Patron(s): Senator(s): Vogel Co-Patron(s): Senator(s): Bell; Delegate(s): Kory

S.B. 1376. Absentee voting; processing of absentee ballots before election day.

Patron(s): Saslaw

Co-Patron(s): Senator(s): Surovell

S.B. 1377. Campaign finance; special report for large pre-legislative session contributions.

Patron(s): Suetterlein

S.B. 1378. Emergency Services and Disaster Law; limitation on duration of executive orders.

Patron(s): Newman

S.B. 1379. Humane Cosmetics Act; civil penalties.

Patron(s): Boysko

Chief Co-Patron(s): Senator(s): Stanley

S.B. 1380. Electric utilities; electric school bus projects.

Patron(s): Lucas

S.B. 1381. Firearm; carrying within Capitol Square and the surrounding area, state-owned bldgs.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Surovell

S.B. 1382. Firearms; purchase, etc., following conviction for assault and battery of a family member.

Patron(s): Favola

Co-Patron(s): Delegate(s): Hope, Kory, Levine, Murphy

S.B. 1383. Criminal proceedings; consideration of mental condition and intellectual.

Patron(s): Favola

Co-Patron(s): Delegate(s): Kory

S.B. 1384. Virginia Public Procurement Act; local arbitration agreements.

Patron(s): Surovell

S.B. 1385. Underground utility facilities; Fairfax County.

Patron(s): Surovell

Co-Patron(s): Senator(s): Ebbin; Delegate(s): Krizek, Sickles, Tran

S.B. 1386. Vehicle registration; enforcement of private tolls.

Patron(s): Spruill

S.B. 1387. Students; eligibility for in-state tuition.

Patron(s): Boysko

Chief Co-Patron(s): Senator(s): Hashmi
Co-Patron(s): Senator(s): McClellan, Surovell

S.B. 1388. Alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine.

Patron(s): Reeves

S.B. 1389. Real property; required disclosures for buyer to exercise due diligence, flood risk report.

Patron(s): Lewis

Co-Patron(s): Senator(s): Mason

S.B. 1390. Cats; trap, neuter, and return programs, civil penalty.

Patron(s): Lewis

S.B. 1391. Pretrial data collection; VCSC to collect and disseminate on an annual basis.

Patron(s): Lucas

Co-Patron(s): Senator(s): McClellan

S.B. 1392. Consumer Data Protection Act; personal data rights of consumer, etc.

Patron(s): Marsden

Chief Co-Patron(s): Delegate(s): Hayes

 $\textbf{Co-Patron(s):} \ Senator(s): Boysko, Dunnavant; \ Delegate(s): Hayes, Subramanyam$

S.B. 1393. Trees; replacement and conservation during development, effective clause.

Patron(s): Marsden

Co-Patron(s): Senator(s): Ebbin

S.B. 1394. Income tax, state; an exclusion for Paycheck Protection Plan loan forgiveness.

Patron(s): Petersen, Pillion

S.B. 1395. Discrimination; prohibited in voting and elections administration, etc.

Patron(s): McClellan

Co-Patron(s): Senator(s): Locke; Delegate(s): Bourne, Cole, J.G., McQuinn, Price, Rasoul, Scott

S.B. 1396. Onsite Sewage Indemnification Fund; use of Fund for grants to certain property owners.

Patron(s): Hashmi

S.B. 1397. Parole and conditional release; notice by electronic means and certification.

Patron(s): Norment

S.B. 1398. Retail sales and transient occupancy taxes; room rentals.

Patron(s): Norment

Co-Patron(s): Senator(s): Ebbin, Mason

S.B. 1399. Tourism Development Authority; name change.

Patron(s): Pillion

S.B. 1400. Tazewell County; quitclaim and conveyance of easement by Board of Wildlife Resources.

Patron(s): Pillion

S.B. 1401. Standards of Learning; reduces total number & type of required assessments to minimum requirements.

Patron(s): Pillion

Co-Patron(s): Senator(s): Boysko, Edwards, Mason, McPike, Suetterlein, Vogel; Delegate(s): Kilgore, O'Ouinn

S.B. 1402. Trout fishing in stocked waters; equalizes for residents and nonresidents requirements to fish.

Patron(s): Pillion

S.B. 1403. Retail Sales and Use Tax; exemption for personal protective equipment.

Patron(s): Pillion

Co-Patron(s): Senator(s): Bell, Ruff

S.B. 1404. Stormwater Local Assistance Fund; grants awarded for projects related to Chesapeake Bay.

Patron(s): Lewis

S.B. 1405. Get Skilled, Get a Job, Give Back (G3) Fund and Program; created and established, report.

Patron(s): Saslaw

Chief Co-Patron(s): Senator(s): Favola

Co-Patron(s): Senator(s): Bell, Boysko, Hashmi, Mason, McClellan, Surovell

S.B. 1406. Marijuana; legalization of simple possession, etc.

Patron(s): Ebbin, Lucas

Incorporated Chief Co-Patron(s): Senator(s): Morrissey Co-Patron(s): Senator(s): Spruill, Surovell; Delegate(s): Kory

S.B. 1407. Electronic gaming devices; regulation, penalties.

Patron(s): Bell

S.B. 1408. Health Care, Joint Commission on; repeals sunset provision.

Patron(s): Barker

Co-Patron(s): Delegate(s): Adams, D.M., Hodges, Hope, Rasoul, Samirah, Sickles

S.B. 1409. Virginia Retirement System; retired law-enforcement officers employed as fingerprint examiners.

Patron(s): Cosgrove

S.B. 1410. Active military or a military spouse; prohibits discrimination in public accommodations, etc.

Patron(s): Bell

Co-Patron(s): Senator(s): Boysko, Cosgrove, Hashmi, Peake, Reeves; Delegate(s): Adams, D.M., Ayala, Cole, M.L., Coyner, Davis, Delaney, Fowler, Hayes, Helmer, Heretick, Kory, Levine, McQuinn, Mugler, Mundon King, Murphy, Plum, Rasoul, Reid, Samirah, Scott, Sickles, Simonds, Subramanyam, Tran, Webert

S.B. 1411. Peanuts; extends sunset date of excise tax on all peanuts grown in Virginia.

Patron(s): Lucas

S.B. 1412. Pet shops, dealers, and dog breeders; employees convicted of animal abuse, penalty.

Patron(s): Stanley

S.B. 1413. Phase I or Phase II electric utilities; provision of broadband capacity.

Patron(s): Boysko

Co-Patron(s): Senator(s): Lewis

S.B. 1414. Henrietta Lacks Commission; extends sunset provision.

Patron(s): Stanley

Co-Patron(s): Delegate(s): Rasoul

S.B. 1415. Protective orders; violations of preliminary child protective order, changes punishment, etc.

Patron(s): Stanley

S.B. 1416. DMAS; remote patient monitoring, rural and underserved populations.

Patron(s): Stanley

S.B. 1417. Animal testing facilities; definitions, adoption of dogs and cats, civil penalty.

Patron(s): Stanley

Chief Co-Patron(s): Senator(s): Boysko

S.B. 1418. Commonwealth's Development Opportunity Fund; grants, waiver or reduction of capital investment.

Patron(s): McPike

S.B. 1419. Project labor agreements; public interest.

Patron(s): Marsden

S.B. 1420. Electric utilities; nonjurisdictional customers, third party power purchase agreements.

Patron(s): Edwards

Co-Patron(s): Senator(s): McClellan

S.B. 1421. Brain injury; clarifies definition.

Patron(s): Edwards

Co-Patron(s): Senator(s): Boysko

S.B. 1422. Voter registration; list of decedent transmitted by St. Reg. of Vital Records to Dept. of Elections.

Patron(s): Kiggans

Co-Patron(s): Senator(s): Suetterlein

S.B. 1423. Data centers; sales and use tax exemption, clarifies "distressed locality," report.

Patron(s): McPike, Pillion

S.B. 1424. Funeral service establishments; defines "manager of record."

Patron(s): Cosgrove

S.B. 1425. Data centers; expands sales and use tax exemption.

Patron(s): Ruff

Chief Co-Patron(s): Senator(s): Pillion Co-Patron(s): Delegate(s): Kilgore

S.B. 1426. Orders of restitution; docketed on behalf of victim, enforcement.

Patron(s): Stanley

Chief Co-Patron(s): Senator(s): McClellan

S.B. 1427. Early Psychosis Intervention and Coordinated Specialty Care Program Advisory Board; established.

Patron(s): Morrissey

S.B. 1428. Alcoholic beverage control; operation of government stores, sale of low alcohol beverage coolers.

Patron(s): Locke

Co-Patron(s): Delegate(s): Krizek

S.B. 1429. Southwestern Va. Mental Health Institute; Governor to lease a portion of property to Smyth County.

Patron(s): Pillion

Co-Patron(s): Senator(s): Mason

S.B. 1430. Firearms; criminal history record information check required to sell.

Patron(s): DeSteph

S.B. 1431. Unrestorably incompetent defendant; competency report.

Patron(s): Mason

S.B. 1432. Electoral college; revises process for allocation of electoral votes.

Patron(s): Chase

S.B. 1433. Public education; voucher program, policies and procedures, etc.

Patron(s): Chase

S.B. 1434. Public schools; mandatory virtual learning, provision of required technology and Internet service.

Patron(s): Chase

S.B. 1435. Estate planning documents; electronic execution, codifies Uniform Electronic Wills Act.

Patron(s): Hanger

S.B. 1436. Eligible Health Care Provider Reserve Directory; established.

Patron(s): Hanger

S.B. 1437. Summons; promises to appear after the issuance.

Patron(s): Hanger

Co-Patron(s): Senator(s): Surovell

S.B. 1438. Combined transient occupancy and food and beverage tax; technical amendments.

Patron(s): Hanger

S.B. 1439. Students; guidelines on excused student absences, civic engagement.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Mason; Delegate(s): Carr

S.B. 1440. Law-enforcement officer, etc.; civil action for unlawful acts of force or failure to intervene.

Patron(s): Surovell

S.B. 1441. Commissioner of accounts; prohibits appointment of General Assembly members.

Patron(s): Lucas

S.B. 1442. Public defender office; establishes an office for the County of Chesterfield.

Patron(s): Morrissey

Chief Co-Patron(s): Senator(s): Hashmi

Co-Patron(s): Senator(s): Surovell; Delegate(s): Ware, Willett

S.B. 1443. Mandatory minimum sentences; elimination, modification of sentence to mandatory minimum term.

Patron(s): Edwards

Chief Co-Patron(s): Senator(s): Morrissey

Co-Patron(s): Delegate(s): Kory

S.B. 1444. Campaign finance; special report for large pre-legislative session contributions.

Patron(s): Saslaw

Incorporated Chief Co-Patron(s): Senator(s): Suetterlein

Co-Patron(s): Senator(s): Surovell

S.B. 1445. COVID-19; facilitates vaccine administration.

Patron(s): Dunnavant

Chief Co-Patron(s): Senator(s): Kiggans

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin,

Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion,

Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel; Delegate(s): Adams,

D.M., Austin, Avoli, Batten, Cole, M.L., Davis, Edmunds, Fariss, Fowler, Hodges, Kilgore,

McGuire, Mugler, Simonds, Tyler, Walker, Watts, Wiley, Wilt, Wright, Wyatt

Removed: Delegate(s): Sullivan

S.B. 1446. Medicine and other healing arts; practice, provision of litigation assistance.

Patron(s): Surovell

S.B. 1447. Buckingham County; fees for disposal of solid waste.

Patron(s): Peake

S.B. 1448. Income tax, state; subtraction for military retirement income and survivor benefits.

Patron(s): Chase

S.B. 1449. COVID-19 immunization; prohibition on requirement, employment discrimination prohibited.

Patron(s): Chase

S.B. 1450. COVID-19 vaccination; discrimination in employment prohibited.

Patron(s): Chase

S.B. 1451. COVID-19 vaccine; access to state and local public property by persons who choose not to receive.

Patron(s): Chase

S.B. 1452. License plates, special; issuance for supporters of the Southwest Regional Recreation Authority.

Patron(s): Pillion

S.B. 1453. Mines and Mining and Virginia Energy Plan; revision of Titles 45.1 and 67.

Patron(s): Edwards

S.B. 1454. Evidence of mental competence; medical scans of brain.

Patron(s): Ruff

S.B. 1455. Absentee ballots; witness requirement, printed name and residence address.

Patron(s): Ruff

S.B. 1456. Juveniles; eligibility for commitment to the Department of Juvenile Justice.

Patron(s): Marsden

S.B. 1457. Historic sites; urban county executive form of gov't. (Fairfax County), provisions in its ordinance.

Patron(s): Surovell

Co-Patron(s): Senator(s): Ebbin; Delegate(s): Krizek

S.B. 1458. Identity Management Standards Advisory Council; transfers management of Council.

Patron(s): Boysko

S.B. 1459. Elections; voter identification & absentee voting, process for applying for absentee ballots.

Patron(s): Norment

S.B. 1460. Deer hunting; authorizes the use of a .223 caliber centerfire rifle.

Patron(s): Lewis

S.B. 1461. Bribery in correctional facilities; penalty.

Patron(s): Lewis

S.B. 1462. Virginia Digital Equity Pilot Program and Fund; established, report, sunset date.

Patron(s): Mason

Chief Co-Patron(s): Senator(s): Boysko

Co-Patron(s): Senator(s): Lewis

S.B. 1463. Covenants regarding solar power; planned unit developments.

Patron(s): Cosgrove

S.B. 1464. Drug Control Act; adds certain chemicals to Schedule I of Act.

Patron(s): Newman

S.B. 1465. Illegal gambling; skill games, definitions, enforcement by localities and Attorney General.

Patron(s): Reeves

Chief Co-Patron(s): Senator(s): Morrissey

Co-Patron(s): Senator(s): Mason

S.B. 1466. Concealed handgun permit; local control of firearms.

Patron(s): Obenshain

Co-Patron(s): Senator(s): Suetterlein

S.B. 1467. Menhaden fishing; prohibited in territorial waters.

Patron(s): DeSteph

Co-Patron(s): Delegate(s): Davis, Knight

S.B. 1468. Victims of crime; certifications for victims of qualifying criminal activity.

Patron(s): Surovell

Co-Patron(s): Senator(s): Boysko; Delegate(s): Guzman, Kory

S.B. 1469. Opioid Abatement Authority; established, Fund created, report, membership.

Patron(s): Barker

Co-Patron(s): Senator(s): Morrissey

S.B. 1470. Vehicle registration; special communication needs indicator.

Patron(s): Barker

Co-Patron(s): Senator(s): Boysko, McClellan

S.B. 1471. Alcoholic beverage control; designated outdoor refreshment area license.

Patron(s): Dunnavant

Co-Patron(s): Senator(s): Boysko, McPike

S.B. 1472. Individuals w/ intellectual & developmental disabilities; DMAS to study use of virtual support, etc.

Patron(s): Suetterlein

Co-Patron(s): Senator(s): Boysko

S.B. 1473. Health Insurance Reform Commission; mandated health insurance benefit or provider.

Patron(s): Saslaw

S.B. 1474. Nonrepairable and rebuilt vehicles; extends sunset provision relating to certain requirements.

Patron(s): Newman

S.B. 1475. Search warrants; date and time of issuance, exceptions.

Patron(s): Stuart

S.J.R. 2. Constitutional amendment; personal reproductive liberty (first reference).

Patron(s): Saslaw

S.J.R. 3. Constitutional amendment; marriage (first reference).

Patron(s): Ebbin

Co-Patron(s): Delegate(s): Kory

S.J.R. 6. Constitutional amendment; Governor's term of office, succession of terms (first reference).

Patron(s): Ebbin

S.J.R. 7. Constitutional amendment; marriage (first reference).

Patron(s): Edwards

Co-Patron(s): Delegate(s): Kory

S.J.R. 8. Constitutional amendment; qualifications of voters and the right to vote.

Patron(s): Locke

Co-Patron(s): Delegate(s): Kory

S.J.R. 11. Constitutional amendment; qualifications of Governor and Lieutenant Governor.

Patron(s): Chase

Co-Patron(s): Delegate(s): Cole, M.L.

S.J.R. 14. Constitutional amendment; qualifications of voters, restoration of civil rights (first reference).

Patron(s): Deeds

S.J.R. 16. Constitutional amendment; General Assembly, term limits.

Patron(s): Chase

Chief Co-Patron(s): Senator(s): Morrissey

S.J.R. 19. Constitutional amendment; political reform (first reference).

Patron(s): Chase

Chief Co-Patron(s): Senator(s): Morrissey

S.J.R. 25. Virginia Tech Carilion School of Medicine; VPSI & SU to study.

Patron(s): Stanley

S.J.R. 39. Justice system; Virginia State Crime Commission to study effect of abolishing jury sentencing.

PATRONAGE

Patron(s): Edwards

Co-Patron(s): Delegate(s): Hope, Rasoul, Samirah

S.J.R. 58. Constitutional amendment; personal property tax exemption for motor vehicle of a disabled veteran.

Patron(s): Morrissey

Incorporated Chief Co-Patron(s): Senator(s): Reeves

Co-Patron(s): Senator(s): Boysko, Chase; Delegate(s): Bagby, Samirah, Simonds

S.J.R. 59. Constitutional amendment; qualifications of voters, restoration of rights.

Patron(s): Morrissey

Co-Patron(s): Senator(s): Boysko, Ebbin; Delegate(s): Levine, Samirah

S.J.R. 63. Constitutional amendment; uniform schedule of elections (first reference).

Patron(s): Ebbin

S.J.R. 75. Constitutional amendment; public schools in the Commonwealth.

Patron(s): Stanley

S.J.R. 270. Constitutional amendment; fundamental right to marry, removes same-sex marriage prohibition.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Boysko, Favola, Surovell; Delegate(s): Carr, Kory

S.J.R. 271. Constitutional amendment; Governor's term of office (first reference).

Patron(s): Ebbin

Co-Patron(s): Senator(s): Surovell

S.J.R. 272. Constitutional amendment; qualifications of voters and the right to vote (first reference).

Patron(s): Locke

Co-Patron(s): Senator(s): Boysko, Ebbin, McClellan, Morrissey, Surovell

S.J.R. 273. Celebrating the life of Bruce Winston Edwards.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Cosgrove, Kiggans; Delegate(s): Davis

S.J.R. 274. Constitutional amendment; limits immunity of legislators (first reference).

Patron(s): Reeves

S.J.R. 275. Constitutional amendment; public schools in the Commonwealth (first reference).

Patron(s): Stanley

Co-Patron(s): Delegate(s): Avoli

S.J.R. 276. Brain Aneurysm Awareness Month; designating as September 2021 and each succeeding year thereafter.

Patron(s): Favola

Co-Patron(s): Senator(s): Barker, Boysko; Delegate(s): Hope, Kory, Rasoul

S.J.R. 277. Commending Marymount University.

Patron(s): Favola, Howell, Ebbin

Co-Patron(s): Delegate(s): Hope, Levine, Lopez, Sullivan, Tyler

S.J.R. 278. Commending the Honorable Bruce D. White.

Patron(s): Surovell

Co-Patron(s): Senator(s): Barker, Boysko, Ebbin, Favola, Howell, Marsden, Petersen, Saslaw; Delegate(s): Kory, Lopez

S.J.R. 279. Commending the Honorable Steven Selwyn Smith.

Patron(s): Surovell

Co-Patron(s): Senator(s): Barker, Bell, McPike, Stuart; Delegate(s): Ayala, Mundon King, Torian

S.J.R. 280. Commending the Honorable Kimberly J. Daniel.

Patron(s): Surovell

Co-Patron(s): Senator(s): Barker, Boysko, Ebbin, Favola, Howell, Marsden, Petersen, Saslaw; Delegate(s): Kory, Lopez

S.J.R. 281. Commending the Honorable Janine M. Saxe.

Patron(s): Surovell

Co-Patron(s): Senator(s): Barker, Boysko, Ebbin, Favola, Howell, Marsden, Petersen, Saslaw; Delegate(s): Kory, Lopez

S.J.R. 282. Commending the Crater Planning District Commission.

Patron(s): Ruff

Co-Patron(s): Senator(s): Morrissey

S.J.R. 283. Celebrating the life of Marisa L. Fleck.

Patron(s): Surovell

Co-Patron(s): Delegate(s): Krizek

S.J.R. 284. Celebrating the life of the Honorable Augustus Benton Chafin, Jr.

Patron(s): Pillion

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel; Delegate(s): Adams, D.M., Adams, L.R., Aird, Austin, Avoli, Ayala, Bagby, Batten, Bell, Bloxom, Bourne, Brewer, Bulova, Byron, Campbell, J.L., Campbell, R.R., Carr, Cole, J.G., Cole, M.L., Cox, Coyner, Davis, Delaney, Edmunds, Fariss, Filler-Corn, Fowler, Freitas, Gilbert, Gooditis, Hayes, Head, Helmer, Hodges, Hope, Hudson, Jones, J.C., Keam, Kilgore, Knight, Kory, Krizek, LaRock, Leftwich, Lopez, Marshall, McGuire, McNamara, McQuinn, Miyares, Morefield, Mullin, O'Quinn, Orrock, Plum, Poindexter, Price, Ransone, Reid, Robinson, Roem, Rush, Scott, Sickles, Simon, Simonds, Sullivan, Torian, Tran, Tyler, VanValkenburg, Walker, Wampler, Ware, Watts, Webert, Willett, Wilt, Wright, Wyatt

S.J.R. 285. Barrier Crimes and Criminal History Records Checks, Joint Subcommittee

Studying; continued. Patron(s): Edwards

Co-Patron(s): Senator(s): Mason

S.J.R. 286. Emergency Management Professionals Week; designating third week in March

2021 and each succeeding.

Patron(s): McPike

S.J.R. 287. Celebrating the life of the Honorable Bernard S. Cohen.

Patron(s): Saslaw

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Spruill, Stanley, Surovell; Delegate(s): Filler-Corn, Levine

S.J.R. 288. National Statuary Hall Collection at the United States Capitol; Barbara Rose

Johns.

Patron(s): Lucas

Co-Patron(s): Senator(s): Ebbin, Hashmi, Locke, McClellan, Spruill

S.J.R. 289. Constitutional amendment; authority to grant perpetual easements to units of government.

Patron(s): Mason

S.J.R. 290. Constitutional amendment; legislative compensation, Legislative Compensation Commission established.

Patron(s): McPike

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Cosgrove, Ebbin, Hashmi, Howell, Locke, Lucas, Mason, McClellan, Morrissey, Spruill, Surovell; Delegate(s): Carter, Cole, J.G., Hope, Kory, Mundon King, Plum, Rasoul, Reid, Samirah, Scott, Subramanyam

S.J.R. 291. Washington, Booker T.; establishes commemorative commission to honor w/ statue in State Capitol.

Patron(s): Suetterlein

S.J.R. 292. Women's Suffrage Month; designating as August 2020 and each succeeding year thereafter.

Patron(s): Vogel

Co-Patron(s): Senator(s): Ebbin, McClellan; Delegate(s): Cole, M.L.

S.J.R. 293. Assisted living and auxiliary grants; Joint Commission on Health Care to study available data.

Patron(s): Spruill

S.J.R. 294. JLARC; costs of education, report.

Patron(s): Lewis

Co-Patron(s): Senator(s): McClellan

S.J.R. 295. Commending Spotsylvania County.

Patron(s): Reeves

Co-Patron(s): Senator(s): Boysko, Howell, McDougle, Peake, Spruill, Stanley, Stuart; Delegate(s): Carr, Cole, J.G., Cole, M.L., Davis, Delaney, Fowler, Heretick, Hope, Kory, Mundon King, Orrock, Plum, Rasoul, Reid, Simonds, Ware, Webert, Wiley, Wyatt

S.J.R. 296. Commending Barry M. Barnard.

Patron(s): Reeves

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Howell, McPike, Peake, Spruill, Stanley, Surovell, Vogel; Delegate(s): Cole, J.G., Cole, M.L., Davis, Delaney, Fowler, Guzman, Heretick, Hope, Kory, Mundon King, Rasoul, Reid, Simonds, Ware, Webert, Wiley, Wyatt

S.J.R. 297. Constitutional amendment; authority of Gen. Assembly to tax exempt property for certain veterans.

Patron(s): Bell

S.J.R. 298. Governor; confirming appointments.

Patron(s): Deeds

S.J.R. 299. Governor; confirming appointments.

Patron(s): Deeds

S.J.R. 300. Governor; confirming appointments.

Patron(s): Deeds

S.J.R. 301. Governor; confirming appointments.

Patron(s): Deeds

S.J.R. 302. Governor; confirming appointments.

Patron(s): Deeds

S.J.R. 303. Governor; confirming appointments.

Patron(s): Deeds

S.J.R. 304. Celebrating the life of the Honorable Mamye E. BaCote.

Patron(s): Locke

Co-Patron(s): Senator(s): Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Mason, McClellan, Norment; Delegate(s): Carr, Filler-Corn, Levine

S.J.R. 305. Celebrating the life of Billy Joe Roberts.

Patron(s): Locke

Co-Patron(s): Senator(s): Deeds, Mason, Norment S.J.R. 306. Commending Moneta Elementary School.

Patron(s): Newman

S.J.R. 307. Commending Pete's Pizza.

Patron(s): Reeves

S.J.R. 308. COVID-19; JLARC to study the impact on Virginia's public schools, students, and

school employees.

Patron(s): Lucas

Chief Co-Patron(s): Senator(s): McClellan

Co-Patron(s): Senator(s): Bell, Boysko, Ebbin, Mason

S.J.R. 309. Constitutional amendment; General Assembly, term limits (first reference).

Patron(s): Chase

S.J.R. 310. Constitutional amendment; length of regular sessions convened in odd-numbered years.

Patron(s): Saslaw

Co-Patron(s): Senator(s): Surovell

S.J.R. 311. Celebrating the life of Robert Edward Mann.

Patron(s): Locke

S.J.R. 312. Commending O. R. Singleton, Jr.

Patron(s): Hashmi

Co-Patron(s): Delegate(s): Carr

S.J.R. 313. Commending Virginia's Crossroads.

Patron(s): Lucas

Co-Patron(s): Senator(s): Bell, Boysko, Howell, Ruff; Delegate(s): Aird, Ayala, Bagby, Carr, Edmunds, Guzman, Heretick, Hurst, Kory, Mugler, Rasoul, Reid, Simonds, Sullivan, Ware, Watts, Wright

S.J.R. 314. Commending the Virginia Chapter of American Promise and Take Back Our Republic.

Patron(s): Deeds

Co-Patron(s): Senator(s): Ebbin

S.J.R. 315. Celebrating the life of Theodore Carter DeLaney, Jr.

Patron(s): Deeds

Co-Patron(s): Senator(s): McClellan

S.J.R. 316. Celebrating the lives of Walter Lowrie Martin III and Edith Luke Martin.

Patron(s): Deeds

S.J.R. 317. Celebrating the life of Walter Lowrie Martin III.

Patron(s): Deeds

S.J.R. 318. Celebrating the life of Stephen Teel Goodwin.

Patron(s): Reeves

Co-Patron(s): Senator(s): Bell, Boysko, Howell, Peake; Delegate(s): Adams, D.M., Cole, J.G., Cole, M.L., Coyner, Davis, Delaney, Fowler, Freitas, Heretick, Hope, Kory, Mundon King, Rasoul, Reid, Simon, Simonds, Ware, Webert, Wiley

S.J.R. 319. Celebrating the life of Michael Stephen Horwatt.

Patron(s): Howell

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Marsden, Saslaw; Delegate(s): Bulova, Delaney, Hope, Plum, Reid, Samirah, Simon, Sullivan, Watts

S.J.R. 320. Celebrating the life of Dr. John E. Thomasson.

Patron(s): Reeves

Co-Patron(s): Senator(s): Bell, Boysko, Howell, Peake, Spruill; Delegate(s): Byron, Cole, J.G., Cole, M.L., Davis, Fowler, Heretick, Hope, Rasoul, Reid, Simon, Simonds, Ware, Watts

S.J.R. 321. Commending Wolf Trap Foundation for the Performing Arts.

Patron(s): Howell

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Ebbin, Marsden, McClellan, Petersen, Saslaw, Surovell, Vogel; Delegate(s): Ayala, Bulova, Delaney, Guzman, Helmer, Hope, Keam, Kory, Levine, Lopez, Murphy, Plum, Reid, Roem, Samirah, Sickles, Simon, Sullivan, Tran, Watts

S.J.R. 322. Military-overseas ballots; electronic return of voted ballots, report.

Patron(s): DeSteph

S.J.R. 323. Liberty Amendments Month; designating as June 19 through third Monday in July 2021.

Patron(s): Locke

Co-Patron(s): Delegate(s): Carr

S.J.R. 324. Celebrating the life of Edward S. Garcia.

Patron(s): Cosgrove

Chief Co-Patron(s): Senator(s): DeSteph

Co-Patron(s): Senator(s): Kiggans; Delegate(s): Knight

S.J.R. 325. Commending Harold J. Roesch II.

Patron(s): McPike

S.J.R. 326. Celebrating the life of David Anthony Rice.

Patron(s): Reeves

Chief Co-Patron(s): Senator(s): Stanley

Co-Patron(s): Senator(s): Hanger, Peake, Ruff; Delegate(s): Byron, Cole, M.L., Coyner, Fowler,

Morefield, Orrock, Runion, Ware **Removed:** Senator(s): Stanley

S.J.R. 327. Governor; confirming appointments.

Patron(s): Deeds

S.J.R. 328. Celebrating the life of Roger Hamilton Brown.

Patron(s): Howell

Co-Patron(s): Senator(s): Barker, Boysko, Surovell; Delegate(s): Helmer, Plum, Simon, Watts

S.J.R. 329. Celebrating the life of Joseph Richard Stowers.

Patron(s): Howell

Co-Patron(s): Senator(s): Barker, Boysko, Saslaw, Surovell; Delegate(s): Helmer, Plum, Simon, Sullivan, Watts

S.J.R. 330. Celebrating the life of Robert Emmett Mannion, Sr.

Patron(s): Petersen

S.J.R. 331. Commending Richard D. Pillow.

Patron(s): Newman

S.J.R. 332. Commending the Honorable Mark C. Christie.

Patron(s): McDougle

Co-Patron(s): Delegate(s): Fowler, Wyatt

S.J.R. 333. Celebrating the life of Frederick Cameron.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Barker, Boysko, Cosgrove, Dunnavant, Ebbin, Favola, Howell, Kiggans, Lewis, Marsden, McDougle, Newman, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel

S.J.R. 334. Commending Major Wayne R. Lee, USA, Ret.

Patron(s): Reeves

Co-Patron(s): Senator(s): Bell, Boysko, Cosgrove, DeSteph, Hashmi, Howell, Ruff, Stanley, Vogel; Delegate(s): Adams, D.M., Avoli, Byron, Cole, M.L., Coyner, Davis, Delaney, Fowler, Freitas, Helmer, Heretick, Hope, Mundon King, Orrock, Rasoul, Reid, Simonds, Ware, Webert, Wyatt

S.J.R. 335. Celebrating the life of Roy L. Pearson.

Patron(s): Mason

Co-Patron(s): Senator(s): Norment; Delegate(s): Mullin

S.J.R. 336. Commending Frank Shatz.

Patron(s): Mason

Co-Patron(s): Senator(s): Norment; Delegate(s): Mullin

S.J.R. 337. Celebrating the life of Mary Esterine Hundley Moyler.

Patron(s): Mason

Co-Patron(s): Senator(s): Norment; Delegate(s): Mullin **S.J.R. 338.** Celebrating the life of Jane Carey Gardner.

Patron(s): Favola

Chief Co-Patron(s): Delegate(s): Jones, J.C.

Co-Patron(s): Senator(s): Cosgrove, DeSteph, Lewis, Locke, Lucas, Mason, Spruill; Delegate(s): Guy, Heretick, Knight, Simonds, Williams Graves

S.J.R. 339. Celebrating the life of Edward A. Chappell, Jr.

Patron(s): Mason

Co-Patron(s): Senator(s): Norment; Delegate(s): Mullin

S.J.R. 340. Commending Edwin C. Roessler, Jr.

Patron(s): Surovell

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Ebbin, Howell, Petersen, Saslaw; Delegate(s): Bulova, Delaney, Filler-Corn, Helmer, Keam, Kory, Krizek, Levine, Plum, Sickles, Simon, Tran, Watts

S.J.R. 341. Commending Carl Hershner.

Patron(s): Mason

Co-Patron(s): Senator(s): Norment; Delegate(s): Mullin, Simonds

S.J.R. 342. Celebrating the life of Wendell Harding Butler.

Patron(s): Edwards

Co-Patron(s): Senator(s): Bell, Boysko, Howell, Spruill, Stanley; Delegate(s): Adams, D.M., Ayala, Cole, M.L., Convirs-Fowler, Delaney, Heretick, Hope, Kory, Levine, Rasoul, Reid, Simonds, Ware

S.J.R. 343. Celebrating the life of Charles Wallace Nuttycombe, Sr.

Patron(s): Mason

Co-Patron(s): Delegate(s): Mullin, Simonds S.J.R. 344. Commending Raymond O. Anderson.

Patron(s): Mason

S.J.R. 345. Celebrating the life of Sigmund Edward Davidson.

Patron(s): Edwards

Co-Patron(s): Senator(s): Boysko, Howell, Ruff, Spruill, Stanley; Delegate(s): Adams, D.M., Ayala, Cole, M.L., Convirs-Fowler, Delaney, Heretick, Hope, Kory, Levine, Rasoul, Reid, Simonds, Ware

S.J.R. 346. Commending John T. Wells.

Patron(s): Norment

Co-Patron(s): Senator(s): Mason; Delegate(s): Hodges

S.J.R. 347. Commending Angeline Godwin.

Patron(s): Stanley

Co-Patron(s): Senator(s): Ruff; Delegate(s): Adams, L.R.

S.J.R. 348. Celebrating the life of Michael Wayne Woods.

Patron(s): Stanley

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel; Delegate(s): Adams, L.R., Fowler, O'Quinn, Sickles

S.J.R. 349. Celebrating the life of William Henry Edwards, Sr.

Patron(s): Stuart

Co-Patron(s): Delegate(s): Ransone

S.J.R. 350. Commending Robert Pinkard, Jr.

Patron(s): Stuart

Co-Patron(s): Delegate(s): Cole, M.L.

S.J.R. 351. Celebrating the life of Sally G. Lamb.

Patron(s): Stuart

Co-Patron(s): Senator(s): Deeds, Hanger

S.J.R. 352. Celebrating the life of John B. Davis.

Patron(s): Hanger

Co-Patron(s): Senator(s): Deeds, Obenshain; Delegate(s): Avoli, Runion

S.J.R. 353. Celebrating the life of William Mantz.

Patron(s): Obenshain

Co-Patron(s): Delegate(s): Gilbert

S.J.R. 354. Celebrating the life of Karl David Stoltzfus, Sr.

Patron(s): Obenshain

Co-Patron(s): Senator(s): Hanger; Delegate(s): Gilbert, Runion, Wilt

S.J.R. 355. Celebrating the life of Joseph Brisco Dellinger.

Patron(s): Obenshain

Co-Patron(s): Senator(s): Hanger; Delegate(s): Runion, Wilt

S.J.R. 356. Celebrating the life of Lowell Robert Barb.

Patron(s): Obenshain

Co-Patron(s): Senator(s): Hanger; Delegate(s): Bell, Gilbert, Runion, Wilt

S.J.R. 357. Commending the Patrick Henry School of Science and Arts.

Patron(s): Morrissey

S.J.R. 358. Commending Ryan Janaske.

Patron(s): Bell

Chief Co-Patron(s): Senator(s): Boysko

Co-Patron(s): Delegate(s): Reid

S.J.R. 359. Commending The Lady Chamberlains.

Patron(s): Bell

Chief Co-Patron(s): Senator(s): Boysko

Co-Patron(s): Delegate(s): Reid, Subramanyam

S.J.R. 360. Celebrating the life of Jean Smith Brown.

Patron(s): Bell

Co-Patron(s): Senator(s): Boysko; Delegate(s): Reid, Subramanyam

S.J.R. 361. Commending the Warrior Retreat at Bull Run.

Patron(s): Bell

Chief Co-Patron(s): Senator(s): Reeves

S.J.R. 362. Commending Akshath Mahajan and Maneesh Vallurupalli.

Patron(s): Bell

Chief Co-Patron(s): Senator(s): Boysko

Co-Patron(s): Delegate(s): Reid, Subramanyam

S.J.R. 363. Commending Steven R. Cover.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Cosgrove, Dunnavant, Kiggans, Lewis, McDougle, Reeves; Delegate(s):

Askew, Convirs-Fowler, Davis, Guy, Knight, Miyares

S.J.R. 364. Celebrating the life of Stephen Johnson.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Cosgrove, Dunnavant, Kiggans, Lewis, McDougle, Newman, Reeves; Delegate(s): Davis, Miyares

S.J.R. 365. Commending LoudounGo.

Patron(s): Bell

S.J.R. 366. Commending Kelly Lazzara.

Patron(s): Kiggans

S.J.R. 367. Commending the Virginia Beach Pandemic Relief Partnership.

Patron(s): Kiggans

S.J.R. 368. Celebrating the life of John Chatburn Stevens.

Patron(s): Kiggans

S.J.R. 369. Celebrating the life of C. Opal B. Abernathey.

Patron(s): Kiggans

S.J.R. 370. Commending Senior Services of Southeastern Virginia.

Patron(s): Kiggans

Co-Patron(s): Senator(s): DeSteph

S.J.R. 371. Commending Michael S. Rolband.

Patron(s): Petersen

S.J.R. 372. Commending the Nation of Greece.

Patron(s): Dunnavant

S.J.R. 373. Commending Hero Homes, Inc.

Patron(s): Bell

S.J.R. 374. Commending Panorama Latino TV Show.

Patron(s): Bell

S.J.R. 375. Celebrating the life of Elaine Dolores Walker.

Patron(s): Bell

Co-Patron(s): Senator(s): Boysko; Delegate(s): Delaney, Reid, Subramanyam

S.J.R. 376. Commending Loudoun Shops Black.

Patron(s): Bell

S.J.R. 377. Celebrating the life of Officer Brian David Sicknick.

Patron(s): Barker

Chief Co-Patron(s): Delegate(s): Watts

Co-Patron(s): Senator(s): Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Petersen, Reeves, Saslaw, Spruill, Stuart, Surovell; Delegate(s): Adams, D.M., Avoli, Ayala, Bulova, Carr, Cole, J.G., Convirs-Fowler, Delaney, Fowler, Helmer, Heretick, Hodges, Hope, Keam, Knight, Kory, Levine, Murphy, Plum, Rasoul, Reid, Roem, Scott, Sickles, Simonds, Subramanyam, Webert, Willett, Williams Graves, Wilt

S.J.R. 378. Celebrating the life of the Honorable Robert Lathan Calhoun.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel; Delegate(s): Levine, Sickles

S.J.R. 379. Commending Tom Hirst and Magaly Galdo-Hirst.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Barker, Boysko, Saslaw; Delegate(s): Levine, Sickles, Sullivan

S.J.R. 380. Celebrating the life of the Honorable Jerry M. Wood.

Patron(s): Deeds

Co-Patron(s): Senator(s): Ruff

S.J.R. 381. Commending the Loudoun County branch of the NAACP.

Patron(s): Boysko

Chief Co-Patron(s): Senator(s): Bell

Co-Patron(s): Senator(s): Howell, McClellan; Delegate(s): Levine

S.J.R. 382. Commending Joshua Thiel.

Patron(s): Boysko

S.J.R. 383. Commending the Reverend Deborah Dodson Parsons.

Patron(s): Boysko

S.J.R. 384. Commending the Loudoun Medical Reserve Corps.

Patron(s): Boysko

S.J.R. 385. Commending Kirk Kincannon.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Barker, Boysko, Howell, Marsden, McPike, Saslaw; Delegate(s): Bulova, Delaney, Helmer, Kory, Levine, Murphy, Reid, Sickles, Sullivan, Watts

S.J.R. 386. Celebrating the life of Sara Lu P. Snyder.

Patron(s): Deeds

S.J.R. 387. Commending Ron Campbell.

Patron(s): Boysko

S.J.R. 388. Commending William McKenna.

Patron(s): Boysko

S.J.R. 389. Commending Kojo Nnamdi.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Howell; Delegate(s): Delaney, Hope, Keam, Plum, Reid, Subramanyam

S.J.R. 390. Commending Jennifer K. Baker.

Patron(s): Boysko

S.J.R. 391. Celebrating the life of Lester Zidel.

Patron(s): Boysko

Co-Patron(s): Senator(s): Mason

S.J.R. 392. Commending Darnell Dozier.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Cosgrove, Kiggans, Lewis, McDougle, Reeves; Delegate(s): Davis, Mivares

S.J.R. 393. Celebrating the life of the Honorable Robert S. Bloxom, Sr.

Patron(s): Lewis

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Favola, Hanger, Howell, Kiggans, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell; Delegate(s): Adams, D.M., Austin, Avoli, Ayala, Bagby, Bell, Bloxom, Carr, Coyner, Gilbert, Guzman, Heretick, Hodges, Kilgore, Marshall, Sickles, Simonds, Subramanyam, Ware, Watts

S.J.R. 394. Celebrating the life of Carla Yvette Savage-Wells.

Patron(s): Lewis

S.J.R. 395. Commemorating the 150th anniversary of the enactment of the Civil Rights

Act of 1871.

Patron(s): McClellan

Co-Patron(s): Senator(s): Ebbin; Delegate(s): Carr, McQuinn, Tyler

S.J.R. 396. Commending Rebecca Simpson Carter.

Patron(s): Peake

S.J.R. 397. Celebrating the life of Pamela Sprouse Palmore.

Patron(s): Peake

S.J.R. 398. Commending the Honorable Clyde H. Perdue, Jr.

Patron(s): Stanley

Co-Patron(s): Delegate(s): Adams, L.R.

S.J.R. 5001. Commending the East Coast Surfing Championship.

Patron(s): DeSteph

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Cosgrove, Dunnavant, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel; Delegate(s): Convirs-Fowler, Knight

S.R. 84. Celebrating the life of William C. Washington.

Patron(s): Lucas

S.R. 85. Celebrating the life of Jeffrey Peter Powell, M.D.

Patron(s): Cosgrove

Co-Patron(s): Senator(s): Spruill

S.R. 86. Celebrating the life of Dora McGlone Harris.

Patron(s): Lucas

S.R. 87. Celebrating the life of Ruth Ann Whitby Greene.

Patron(s): Hashmi

S.R. 88. Celebrating the life of Robert Gene LeRosen.

Patron(s): Hashmi

S.R. 89. Commending Tressie McMillan Cottom.

Patron(s): Hashmi

S.R. 90. Senate; 2021 operating resolution.

Patron(s): Locke

S.R. 91. Chase, Senator Amanda F.; expresses the censure of the Senate of Virginia.

Patron(s): Bell

Co-Patron(s): Senator(s): Barker, Boysko, Ebbin, Favola, Hashmi, Howell, Lucas, Marsden, Surovell

Removed: Senator(s): Spruill

S.R. 92. Celebrating the life of Luther Hilton Foster.

Patron(s): Ruff

S.R. 93. Auditor of Public Accounts; nomination for election.

Patron(s): Deeds

S.R. 94. Celebrating the life of Madison Boyd, Jr.

Patron(s): McDougle

S.R. 95. Washington, Booker T.; establishes commemorative commission to honor w/statue

in Old Senate Chamber. **Patron(s):** Suetterlein

S.R. 96. Commending Elizabeth Stamoulis Via-Gossman.

Patron(s): McPike

S.R. 97. Celebrating the life of Gloria Mae Benton Walker.

Patron(s): Lucas

S.R. 98. Celebrating the life of Mary Holley.

Patron(s): Lucas

S.R. 99. Judges; nominations for election to circuit court.

Patron(s): Edwards

S.R. 100. Judges; nominations for election to general district court.

Patron(s): Edwards

S.R. 101. Judges; nominations for election to juvenile and domestic relations district court.

Patron(s): Edwards

S.R. 102. Judicial Inquiry and Review Commission; nomination for election of member.

Patron(s): Edwards

S.R. 103. State Corporation Commission; nomination for election of members.

Patron(s): Saslaw

S.R. 104. Celebrating the life of the Honorable Leslie D. Campbell, Jr.

Patron(s): McDougle

S.R. 105. Commending the Virginia Piedmont Heritage Area Association.

Patron(s): Vogel

S.R. 106. Celebrating the life of Avicia Beatrice Hooper Thorpe.

Patron(s): Stanley

S.R. 107. Celebrating the life of Elizabeth Bateman Stone.

Patron(s): Stanley

S.R. 108. Celebrating the life of Bryan Wayne Galentine.

Patron(s): Stanley

Co-Patron(s): Senator(s): Ruff

S.R. 109. Commending telehealth care providers in Virginia.

Patron(s): Stanley

S.R. 110. Celebrating the life of Thomas Henry Francis.

Patron(s): Hashmi

Co-Patron(s): Senator(s): Deeds, McClellan

S.R. 111. Celebrating the life of Pauline Allen Mitchell.

Patron(s): Hashmi

S.R. 112. Celebrating the life of the Reverend Dr. Willie Woodson.

Patron(s): McClellan

S.R. 113. Commending the Robert Russa Moton Museum.

Patron(s): McClellan

S.R. 114. Commending Gilfield Baptist Church.

Patron(s): McClellan

S.R. 115. Celebrating the life of James Perry Wark.

Patron(s): McClellan

S.R. 116. Celebrating the life of Rhea K. Hale.

Patron(s): McClellan

S.R. 117. Commending Richmond Heritage Federal Credit Union.

Patron(s): McClellan

S.R. 118. Commending Richmond Raceway.

Patron(s): McClellan

S.R. 119. Commending the Reverend Doug Barber.

Patron(s): Stanley

Co-Patron(s): Senator(s): Ruff

S.R. 120. Commending James Smith.

Patron(s): McDougle

S.R. 121. Commending the Reverend Dr. Barbara L. Cain.

Patron(s): McDougle

S.R. 501. Senate; 2021 Special Session I operating resolution.

Patron(s): Locke

S.R. 502. Commending Danny TK Avula, M.D.

Patron(s): Dunnavant

S.R. 503. Celebrating the life of Franklin Delano Robertson.

Patron(s): Pillion

S.R. 504. Celebrating the life of Peter Michael Dolan, Jr.

Patron(s): McPike

S.R. 505. Commending Pamela Wooddy.

Patron(s): Hashmi

S.R. 506. Commending the Virginia State Police and the Virginia National Guard.

Patron(s): Surovell

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill

S.R. 507. Commending Danny TK Avula, M.D.

Patron(s): Hashmi

Co-Patron(s): Senator(s): McClellan

S.R. 508. Celebrating the life of Edward Paul Crapol.

Patron(s): Mason

Chief Co-Patron(s): Senator(s): Norment

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McDougle, McPike, Morrissey, Newman, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel

S.R. 509. Commending Walt Whitman Middle School.

Patron(s): Surovell

Co-Patron(s): Senator(s): Ebbin

S.R. 510. Commending Montclair Elementary School.

Patron(s): Surovell

Co-Patron(s): Senator(s): McPike

S.R. 511. Commending Swans Creek Elementary School.

Patron(s): Surovell

Co-Patron(s): Senator(s): McPike

S.R. 512. Commending Featherstone Elementary School.

Patron(s): Surovell

Co-Patron(s): Senator(s): McPike

S.R. 513. Celebrating the life of Charles R. Hooff III.

Patron(s): Surovell

S.R. 514. Commending the Science Museum of Virginia.

Patron(s): Hashmi

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel

S.R. 515. Commending Pocahontas County.

Patron(s): Deeds

S.R. 516. Commending Greek Orthodox churches in Virginia.

Patron(s): Dunnavant

S.R. 517. Commending the Virginia Women's Institute for Leadership at Mary Baldwin

University.

Patron(s): Hanger

S.R. 518. Commending Colonel Dean E. Gould.

Patron(s): Reeves

Chief Co-Patron(s): Senator(s): Bell

Co-Patron(s): Senator(s): Cosgrove, DeSteph, Hashmi, Howell, Kiggans, Peake, Ruff, Spruill, Stuart

S.R. 519. Commending the Virginia Wing of the Civil Air Patrol.

Patron(s): Reeves

Chief Co-Patron(s): Senator(s): Bell

Co-Patron(s): Senator(s): Cosgrove, DeSteph, Howell, Kiggans, Stuart

S.R. 520. Celebrating the life of Richard Blane Byrd.

Patron(s): Deeds

S.R. 521. Celebrating the life of Jamie Beletz.

Patron(s): McPike

S.R. 522. Celebrating the life of Thomas F. Cleary, M.D.

Patron(s): Surovell

Co-Patron(s): Senator(s): Ebbin

S.R. 523. Celebrating the life of Carlton Farquhar Andrus.

Patron(s): Surovell

Co-Patron(s): Senator(s): Ebbin

S.R. 524. Celebrating the life of the Honorable Clyde H. Perdue, Jr.

Patron(s): Stanley

Co-Patron(s): Senator(s): Edwards

S.R. 525. Commending Valley Health.

Patron(s): Vogel

S.R. 526. Commending Shenandoah University.

Patron(s): Vogel

Co-Patron(s): Senator(s): Boysko, Cosgrove

S.R. 527. Commending Freddie's Beach Bar.

Patron(s): Ebbin

Co-Patron(s): Senator(s): Favola, Howell

S.R. 528. Celebrating the life of Roger Allan Saunders, Jr.

Patron(s): Locke

S.R. 529. Commending John E. Smith.

Patron(s): Locke

S.R. 530. Celebrating the life of Claire Elizabeth Grainger.

Patron(s): Bell

S.R. 531. Commending Ayana Ahuja and Saanvi Paladugu.

Patron(s): Bell

S.R. 532. Judges; nominations for election to circuit court.

Patron(s): Edwards

S.R. 533. Judges; nominations for election to general district court.

Patron(s): Edwards

S.R. 534. Judges; nominations for election to juvenile and domestic relations district court.

Patron(s): Edwards

S.R. 535. Judicial Inquiry and Review Commission; nomination for election of a member.

Patron(s): Edwards

S.R. 536. Celebrating the life of Joseph Howard Tate.

Patron(s): Pillion

S.R. 537. Celebrating the life of Alvin Dwayne Harkleroad.

Patron(s): Deeds

Co-Patron(s): Senator(s): Hanger

S.R. 538. Commending Loudoun County Public Schools custodians and maintenance

technicians.
Patron(s): Favola

Co-Patron(s): Senator(s): Bell, Boysko

S.R. 539. Commending Olga L. Garrett.

Patron(s): Lucas

S.R. 540. Celebrating the life of Flossie Rebecca Branchcomb.

Patron(s): Lucas

S.R. 541. Celebrating the life of the Reverend Michael Duvall Green, Sr.

Patron(s): Lucas

S.R. 542. Commending John Hutchison Anderson.

Patron(s): Vogel

S.R. 543. Commending Gregory Garfield Harris.

Patron(s): Vogel

S.R. 544. Celebrating the life of Elizabeth Ann Kerr Ledgerton.

Patron(s): Vogel

S.R. 545. Celebrating the life of Maybelle Rutland Campbell.

Patron(s): Kiggans

S.R. 546. Commending Kim Jackson-Dinnall.

Patron(s): Kiggans

S.R. 547. Commending the Princess Anne High School girls' basketball team.

Patron(s): Kiggans

Co-Patron(s): Senator(s): DeSteph

S.R. 548. Commending David Hirn, Kate Williamson, and Courtni Pannell.

Patron(s): Morrissey

S.R. 549. Celebrating the life of Michael J. Weber, Ph.D.

Patron(s): Deeds

S.R. 550. Celebrating the life of John Richard Neese.

Patron(s): Obenshain

S.R. 551. Celebrating the life of Roy Kenneth Harris.

Patron(s): Obenshain

S.R. 552. Commending Kirk Dolson.

Patron(s): Boysko

Co-Patron(s): Senator(s): Bell, Favola

S.R. 553. Commending School Nutrition Services and Transportation Division of

Loudoun County Public Schools.

Patron(s): Boysko

Co-Patron(s): Senator(s): Bell, Favola

S.R. 554. Celebrating the life of Anjelica Marie Hoffer.

Patron(s): Morrissey

S.R. 555. Celebrating the life of Henry Lewis Livas, Jr.

Patron(s): Locke

S.R. 556. Celebrating the life of Frederick William Holland, Jr.

Patron(s): Lewis

S.R. 557. Celebrating the life of Sheila Kavanagh Mandt.

Patron(s): McClellan

Co-Patron(s): Senator(s): Barker, Bell, Boysko, Howell, Lucas, Marsden, Morrissey, Saslaw

S.R. 558. Commemorating the life and legacy of Giles Beecher Jackson.

Patron(s): McClellan

Co-Patron(s): Senator(s): Boysko, Lucas, Morrissey

JOURNAL

OF

THE

SENATE

2021 SPECIAL SESSION I

WEDNESDAY, FEBRUARY 10, 2021

The Senate met at 11:00 a.m. at the Science Museum of Virginia in Richmond, Virginia, and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Mark J. Peake, Twenty-second Senatorial District, offered the following prayer:

Our most gracious and heavenly Father, we give thanks to You for today and every day that You provide us. We give thanks to You for the opportunity You have given us to serve the citizens of the Commonwealth. We ask that You give us the wisdom, strength, and guidance to do Your will. We ask for blessings for our Governor and his administration and that You give us the temperament and discernment to work with them as we all work together to serve the people of the Commonwealth. We give thanks for the people that protect us, the staff that prepares us and informs us and enables us to do our job. We ask for special blessings for those people who have been terribly afflicted by this virus—financially, physically, and mentally. We ask that You give them hope and comfort and guidance. We ask this in Jesus' name. We give thanks for this day and every day. Thank You very much. In Jesus' name we pray. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by David Hessler, Senate Committee Operations Intern, Senate Clerk's Office.

On motion of Senator Norment, the Rules were suspended to allow Senator Boysko, impacted by COVID-19, to participate remotely and electronically in the daily session.

The recorded vote is as follows: YEAS--36. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--36.

NAYS--0.

RULE 36--0.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

COMMUNICATION

The following communication was received and read:

COMMONWEALTH OF VIRGINIA Executive Department

Proclamation

In accordance with the provisions of Article IV, Section 6, and Article V, Section 5, of the Constitution of Virginia and the powers thereby vested in the Governor to call a Special Session of the General Assembly;

I, Ralph S. Northam, Governor of Virginia, do hereby summon the members of the Senate and the House of Delegates, constituting the General Assembly of Virginia, to meet in Special Session commencing the tenth day of February, two thousand and twenty one, for the purpose of continuing the work of the 2021 regular session of the General Assembly.

Given under my hand as Governor of Virginia, and under the Lesser Seal of the Commonwealth, at Richmond, this fourth day of February, two thousand and twenty one, and in the two hundred and forty-fifth year of the Commonwealth.

/s/ Ralph S. Northam Governor of Virginia

[SEAL]

By the Governor of Virginia

/s/ Kelly Thomasson Secretary of the Commonwealth

RULES OF THE SENATE

The President announced that the Senate was operating under the Rules of the Senate as adopted at the 2020 Session, which state, "The Rules of the Senate shall be adopted at the commencement of the first regular session of the General Assembly after the election of the Senate, and shall be in force for the succeeding four years unless amended or suspended as provided by these Rules."

HOUSE COMMUNICATION

The following communication was received and read:

In the House of Delegates February 10, 2021

THE HOUSE OF DELEGATES IS DULY ORGANIZED AND READY TO PROCEED TO BUSINESS.

/s/ Suzette Denslow Clerk of the House of Delegates

IMMEDIATE CONSIDERATION

Senator Locke moved that the Rules be suspended, the Committee on Rules be discharged from further consideration of **S.R. 501** (five hundred one), the readings of the title and engrossment be waived, and the resolution be taken up for immediate consideration.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

SENATE RESOLUTION NO. 501

2021 Special Session I operating resolution.

RESOLVED by the Senate of Virginia, That the Comptroller be directed to issue his warrants on the Treasurer, payable from the contingent fund of the Senate, to accomplish the work of the Senate of Virginia as reported by the Clerk of the Senate to the Senate Rules Committee during the 2021 Special Session I. Necessary payments to cover salaries of temporary employees, per diem for legislative assistants who establish a temporary residence, and certain employees designated by the Clerk and reported to the Chair of the Senate Committee on Rules, as well as other contingent and incidental expenses, will be certified by the Clerk of the Senate or her designee; and, be it

RESOLVED FURTHER, That members of the Senate shall receive session per diem and mileage reimbursement for any day they attend a scheduled floor session at which an attendance roll call is taken. Session per diem shall not be allowed for members of the Senate during a recess of the Special Session. However, members may receive compensation while the General Assembly is in recess, as provided in § 30-19.12 of the Code of Virginia and in the 2020-2022 Appropriation Act, as follows: (i) members of any standing committee authorized by the Senate and the Committee on Rules; (ii) members of any committee of conference; and (iii) members of any legislative committee, commission, or council established by the General Assembly.

S.R. 501, on motion of Senator Locke, was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

Senator Saslaw was ordered to inform the House of Delegates that the Senate was duly organized and ready to proceed to business.

HOUSE COMMUNICATION

The following communication was received and read:

In the House of Delegates February 10, 2021

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING HOUSE JOINT RESOLUTION:

H.J.R. 5001. Limiting legislation to be considered by the 2021 Special Session I of the General Assembly and establishing a schedule for the conduct of business coming before such Special Session.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

The House joint resolution, communicated as agreed to by the House of Delegates, the first reading of its title having been read, was referred as follows:

H.J.R. 5001 was referred to the Committee on Rules.

IMMEDIATE CONSIDERATION

Senator Locke moved that the Rules be suspended, the Committee on Rules be discharged from further consideration of **H.J.R. 5001** (five thousand one), the readings of the title be waived, and the resolution be taken up for immediate consideration.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

HOUSE JOINT RESOLUTION NO. 5001

Limiting legislation to be considered by the 2021 Special Session I of the General Assembly and establishing a schedule for the conduct of business coming before such Special Session.

RESOLVED by the House of Delegates, the Senate concurring, That during the 2021 Special Session I of the General Assembly, summoned by proclamation of the Governor on February 4, 2021, to begin February 10, 2021, except with the unanimous consent of the house in which the legislation is offered, no bill, joint resolution, or resolution shall be offered or considered in either house during the Special Session other than (i) bills or joint resolutions continued from the 2021 Regular Session pursuant to Section 7 of Article IV of the Constitution of Virginia and House Joint Resolution No. 575 providing for legislative continuity; (ii) the Budget Bill; (iii) single-house commending or memorial resolutions; (iv) bills, joint resolutions, or resolutions affecting the rules of procedure or schedule of business of the General Assembly, either of its houses, or any of its committees; (v) joint resolutions or resolutions relating to appointments subject to the confirmation of the General Assembly or either house; (vi) bills, joint resolutions, or resolutions relating to the election of judges and other officials subject to the election of the General Assembly; or (vii) bills or joint resolutions requested in writing by the Governor; and, be it

RESOLVED FURTHER, That notwithstanding any other provision of this resolution and in accordance with the practices of each house, a request to be removed as a co-patron shall be received no later than 3:00 p.m., Friday, February 19, 2021; and, be it

RESOLVED FURTHER, That for purposes of the procedural deadlines established herein for the 2021 Regular Session of the General Assembly:

"Budget Bill" means the general appropriation bill introduced in each house continued from the 2021 Regular Session that authorizes the biennial expenditure of public revenues for the period from July 1, 2020, through June 30, 2022.

"Debt bill" means any bill that authorizes the issuance of debt.

"Legislative day" means the period of time that begins with the call to order by the presiding officer and ends when declared adjourned by the presiding officer. Unless another time is specified, any deadline established in this resolution shall expire at the end of the legislative day.

"Revenue bill" means any bill, except the Budget Bill and debt bills, that increases or decreases the total revenues available for appropriation.

"Unanimous consent" means the affirmation of all the members present in the house of origin. Any legislation intended to be offered for introduction with unanimous consent or with the written request of the Governor shall not require the consent of the house in order for the member to request the Division of Legislative Services to draft such legislation. The Division of Legislative Services shall return such legislation after the original introduction deadline; and, be it

RESOLVED FINALLY, That the 2021 Special Session I of the General Assembly shall be governed by the following procedural rules, which establish time limits for elections and all legislation continued from the 2021 Regular Session except:

(i) Bills, joint resolutions, or resolutions affecting the rules of procedure or the schedule of business of the General Assembly, either of its houses, or any of its committees, except for the time limitations established in Rule 13 for bills and joint resolutions;

- (ii) House and Senate resolutions, except for the limitations established in Rules 9, 11, and 13;
- (iii) Bills, joint resolutions, or resolutions introduced with unanimous consent including resolutions introduced with unanimous consent to exceed the time limitations established in Rules 9 and 11 but not the time limitations established in Rule 13:
- (iv) Joint resolutions or resolutions relating to appointments subject to the confirmation of the General Assembly or either house, except for the time limitations established in Rule 13;
- (v) Bills, joint resolutions, or resolutions relating to the election of judges and other officials subject to the election of the General Assembly, except for the time limitations established in Rule 13; or
 - (vi) Bills and joint resolutions requested in writing by the Governor.
- Rule 1. Except for the Budget Bill, beginning Wednesday, February 10, 2021, the House of Delegates shall consider only Senate bills, Senate joint resolutions, House bills with Senate amendments, and House joint resolutions with Senate amendments; the Senate shall consider only House bills, House joint resolutions, Senate bills with House amendments, and Senate joint resolutions with House amendments; and each house may consider conference reports and other privileged matters relating thereto to the end that the work of each house may be disposed of by the other.
- Rule 2. The committees responsible for the consideration of the Budget Bill in the houses of introduction shall complete their work on such bill no later than Wednesday, February 10, 2021, and any amendments proposed by such committees shall be made available to their respective houses on such day.
- Rule 3. The houses of introduction shall complete their consideration of the Budget Bill, except for conference reports and other privileged matters relating thereto, no later than Saturday, February 13, 2021.
- Rule 4. The committees responsible for consideration of revenue bills of the other house shall complete their consideration of such bills no later than Wednesday, February 17, 2021.
- Rule 5. No later than Wednesday, February 17, 2021, each house shall complete consideration of the Budget Bill of the other house, except for conference reports and other privileged matters relating thereto.
- Rule 6. No later than Wednesday, February 17, 2021, each house shall begin its consideration of any election to fill any judicial seat in the courts of the Commonwealth, or to fill a seat on any commission or office elected by the General Assembly. In the event that the houses cannot agree on such election before Thursday, February 18, 2021, such election shall become the subject in each house of a special and continuing joint order, and such special and continuing joint order shall have precedence over all other business of either house, until such time as both houses reach agreement on such election or either house votes to suspend or discharge the order. The Rules of each house, as far as applicable, shall be the rules governing any such election.
- Rule 7. No later than Friday, February 19, 2021, each house shall complete consideration of revenue bills of the other house, except for conference reports and other privileged matters relating thereto, and the appointing authority shall appoint the conferees to the Budget Bill and such revenue bills. Any conference committee on any revenue bills shall complete its deliberations and make the report of such conference available to the General Assembly as soon as practicable.
- Rule 8. Neither house shall receive from any committee any bill or joint resolution acted on by any committee later than midnight, Monday, February 22, 2021.

Rule 9. Requests for the drafting, redrafting, or correction of any single-house commending or memorial resolution shall be submitted to and received by the Division of Legislative Services no later than 5:00 p.m., Tuesday, February 23, 2021.

Rule 10. Any conference committee on the Budget Bill shall complete its deliberations and make the report of such conference available to the General Assembly as soon as practicable. Neither house shall receive, consider, or vote on any Budget Bill that is in conference unless it has been agreed to in writing or signed electronically by a majority of conferees from each house. Neither house shall consider such conference report earlier than 48 hours after receipt, unless both houses respectively determine to proceed earlier by a vote of two-thirds of the members voting in each house. No engrossment of the Budget Bill shall be required in either house, and any conference on the Budget Bill shall consider, as the basis of its deliberations, the Budget Bill as recommended by the Governor and introduced in the House and the amendments thereto proposed by each house. A report shall be issued concurrently with the report of the conference committee that identifies the following by item number, narrative description, and dollar amount: (i) any nonstate agency appropriation, (ii) any item in the conference report that was not included in a general appropriation bill as passed by either the House or the Senate, and (iii) any item that represents legislation that failed in either house during the regular or a special session.

Rule 11. No single-house commending or memorial resolution shall be offered in either house after 5:00 p.m., Thursday, February 25, 2021.

Rule 12. Except for joint resolutions affecting the rules of procedure or the schedule of business of the General Assembly, beginning Friday, February 26, 2021, the House shall consider only Senate joint resolutions and House joint resolutions with Senate amendments; the Senate shall consider only House joint resolutions and Senate joint resolutions with House amendments; and each house may consider conference reports or joint resolutions and other privileged matters relating thereto, to the end that the work of each house may be disposed of by the other.

Rule 13. Beginning Sunday, February 28, 2021, neither House shall consider any (i) conference reports, (ii) bills or joint resolutions, (iii) single-house commending or memorial resolutions, (iv) joint resolutions or resolutions relating to appointments subject to the confirmation of the General Assembly or either house, or (v) bills, joint resolutions, or resolutions relating to the election of judges and other officials subject to the election of the General Assembly.

Rule 14. This session of the General Assembly shall adjourn sine die no later than Monday March 1, 2021.

Rule 15. Pursuant to Section 6 of Article IV of the Constitution of Virginia, the General Assembly shall reconvene Wednesday, April 7, 2021, for the purpose of considering bills that may have been returned by the Governor with recommendations for their amendment and bills and items of appropriation bills, including the general appropriation act, that may have been returned by the Governor with his objections.

Rule 16. That members of the General Assembly and credentialed legislative staff attending floor sessions, committees, subcommittees, and any other meetings of the General Assembly in person or meeting in a legislative office in person shall be required to wear a facemask or face shield, and at the discretion of the Speaker of the House of Delegates for Delegates and House credentialed staff and the discretion of the chair of the Senate Committee on Rules for Senators and Senate credentialed staff, have their temperature taken daily when initially entering legislative space.

Rule 17. The conduct of the business of any subcommittee of any House committee, any joint subcommittee of House and Senate committees, and any interim study commission created pursuant to a House measure shall be governed by the Rules of the House of Delegates; the conduct of the business of any subcommittee of any Senate committee, any joint subcommittee of Senate and House committees, and any interim study commission created pursuant to a Senate measure shall be governed by the Rules of the Senate. If a House measure and a Senate measure create the same study, the conduct of business of the study shall be governed by the rules of the house of the chairman of the study, or in the case of co-chairmen, the rules of the house as agreed upon by the co-chairmen.

H.J.R. 5001, on motion of Senator Locke, was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--38.

NAYS--Suetterlein--1.

RULE 36--0.

Senator Locke was ordered to inform the House of Delegates thereof.

LEGISLATION CONTINUED

The following report was received from the Clerk:

TO THE MEMBERS OF THE SENATE:

Pursuant to the provisions of Senate Rules 20 (f) and 20 (g) and House Joint Resolution No. 575 of the 2021 Regular Session, certain bills and joint resolutions were continued from the 2021 Regular Session of the General Assembly in the several Senate committees, as follows:

COMMITTEE ON AGRICULTURE, CONSERVATION AND NATURAL RESOURCES

H.B. 1750, H.B. 1751, H.B. 1760, H.B. 1804, H.B. 1819, H.B. 1833, H.B. 1836, H.B. 1837, H.B. 1855, H.B. 1902, H.B. 1928, H.B. 1958, H.B. 1982, H.B. 1983, H.B. 2030, H.B. 2068, H.B. 2078, H.B. 2129, H.B. 2148, H.B. 2159, H.B. 2187, H.B. 2203, H.B. 2221, H.B. 2275, H.B. 2302, H.B. 2311.

COMMITTEE ON COMMERCE AND LABOR

H.B. 1754, H.B. 1775, H.B. 1786, H.B. 1807, H.B. 1818, H.B. 1822, H.B. 1829, H.B. 1832, H.B. 1834, H.B. 1862, H.B. 1877, H.B. 1881, H.B. 1884, H.B. 1892, H.B. 1896, H.B. 1907, H.B. 1914, H.B. 1923, H.B. 1925, H.B. 1942, H.B. 1964, H.B. 1965, H.B. 1984, H.B. 1985, H.B. 1994, H.B. 2008, H.B. 2032, H.B. 2034, H.B. 2036, H.B. 2037, H.B. 2040, H.B. 2048, H.B. 2049, H.B. 2062, H.B. 2063, H.B. 2121, H.B. 2134, H.B. 2137, H.B. 2160, H.B. 2200, H.B. 2207, H.B. 2219, H.B. 2250, H.B. 2269, H.B. 2282, H.B. 2304, H.B. 2330, H.B. 2332.

COMMITTEE ON EDUCATION AND HEALTH

H.B. 1736, H.B. 1737, H.B. 1747, H.B. 1776, H.B. 1790, H.B. 1798, H.B. 1808, H.B. 1817, H.B. 1823, H.B. 1827, H.B. 1838, H.B. 1865, H.B. 1873, H.B. 1885, H.B. 1894, H.B. 1904, H.B. 1905, H.B. 1909, H.B. 1913, H.B. 1918, H.B. 1930, H.B. 1940, H.B. 1950, H.B. 1953, H.B. 1963, H.B. 1976, H.B. 1980, H.B. 1986, H.B. 1987, H.B. 1988, H.B. 1989, H.B. 1995, H.B. 1998, H.B. 2007, H.B. 2013, H.B. 2019, H.B. 2027, H.B. 2035, H.B. 2039, H.B. 2058, H.B. 2061, H.B. 2079, H.B. 2105, H.B. 2111, H.B.

2116, H.B. 2117, H.B. 2119, H.B. 2120, H.B. 2123, H.B. 2124, H.B. 2135, H.B. 2162, H.B. 2176, H.B. 2182, H.B. 2204, H.B. 2206, H.B. 2218, H.B. 2220, H.B. 2230, H.B. 2238, H.B. 2299, H.B. 2300, H.B. 2305, H.B. 2314, H.B. 2316.

COMMITTEE ON FINANCE AND APPROPRIATIONS

S.B. 1100.

H.B. 1763, H.B. 1774, H.B. 1899, H.B. 1916, H.B. 1935, H.B. 1969, H.B. 1979, H.B. 1999, H.B. 2006, H.B. 2059, H.B. 2060, H.B. 2101, H.B. 2118, H.B. 2165, H.B. 2174, H.B. 2177, H.B. 2178, H.B. 2179, H.B. 2181, H.B. 2185, H.B. 2223, H.B. 2273, H.B. 2293, H.B. 2337.

COMMITTEE ON GENERAL LAWS AND TECHNOLOGY

H.B. 1811, H.B. 1812, H.B. 1816, H.B. 1824, H.B. 1830, H.B. 1842, H.B. 1843, H.B. 1847, H.B. 1848, H.B. 1849, H.B. 1864, H.B. 1876, H.B. 1882, H.B. 1889, H.B. 1891, H.B. 1900, H.B. 1908, H.B. 1931, H.B. 1943, H.B. 1944, H.B. 1967, H.B. 1971, H.B. 1981, H.B. 1993, H.B. 1996, H.B. 2001, H.B. 2003, H.B. 2004, H.B. 2005, H.B. 2014, H.B. 2025, H.B. 2029, H.B. 2031, H.B. 2046, H.B. 2053, H.B. 2072, H.B. 2074, H.B. 2085, H.B. 2130, H.B. 2140, H.B. 2147, H.B. 2155, H.B. 2161, H.B. 2170, H.B. 2171, H.B. 2172, H.B. 2175, H.B. 2202, H.B. 2222, H.B. 2227, H.B. 2229, H.B. 2249, H.B. 2288, H.B. 2307, H.B. 2308, H.B. 2313, H.B. 2320, H.B. 2321, H.B. 2322, H.B. 2327.

COMMITTEE ON THE JUDICIARY

H.B. 1806, H.B. 1814, H.B. 1821, H.B. 1852, H.B. 1853, H.B. 1856, H.B. 1866, H.B. 1867, H.B. 1878, H.B. 1895, H.B. 1897, H.B. 1911, H.B. 1912, H.B. 1936, H.B. 1948, H.B. 1951, H.B. 1991, H.B. 1992, H.B. 2002, H.B. 2009, H.B. 2012, H.B. 2017, H.B. 2018, H.B. 2038, H.B. 2047, H.B. 2055, H.B. 2056, H.B. 2064, H.B. 2073, H.B. 2081, H.B. 2099, H.B. 2110, H.B. 2113, H.B. 2128, H.B. 2132, H.B. 2133, H.B. 2139, H.B. 2150, H.B. 2168, H.B. 2169, H.B. 2190, H.B. 2192, H.B. 2193, H.B. 2194, H.B. 2233, H.B. 2234, H.B. 2236, H.B. 2252, H.B. 2254, H.B. 2258, H.B. 2263, H.B. 2276, H.B. 2290, H.B. 2291, H.B. 2298, H.B. 2310, H.B. 2317, H.B. 2319, H.B. 2331.

H.J.R. 629.

COMMITTEE ON LOCAL GOVERNMENT

H.B. 1749, H.B. 1764, H.B. 1778, H.B. 1783, H.B. 1858, H.B. 1898, H.B. 1919, H.B. 2042, H.B. 2054, H.B. 2091, H.B. 2095, H.B. 2180, H.B. 2186, H.B. 2201, H.B. 2217, H.B. 2257, H.B. 2287, H.B. 2323, H.B. 2326.

COMMITTEE ON PRIVILEGES AND ELECTIONS

H.B. 1810, H.B. 1888, H.B. 1890, H.B. 1921, H.B. 1952, H.B. 1968, H.B. 2020, H.B. 2082, H.B. 2125, H.B. 2198, H.B. 2324.

H.J.R. 555, H.J.R. 582.

COMMITTEE ON REHABILITATION AND SOCIAL SERVICES

H.B. 1805, H.B. 1820, H.B. 1831, H.B. 1845, H.B. 1874, H.B. 1879, H.B. 1932, H.B. 1957, H.B. 1962, H.B. 1973, H.B. 2010, H.B. 2065, H.B. 2070, H.B. 2086, H.B. 2092, H.B. 2098, H.B. 2131, H.B. 2154, H.B. 2166, H.B. 2167, H.B. 2191, H.B. 2197, H.B. 2212, H.B. 2266, H.B. 2312.

COMMITTEE ON TRANSPORTATION

H.B. 1796, H.B. 1801, H.B. 1813, H.B. 1828, H.B. 1841, H.B. 1846, H.B. 1850, H.B. 1851, H.B. 1854, H.B. 1868, H.B. 1887, H.B. 1893, H.B. 1901, H.B. 1903, H.B. 1926, H.B. 1960, H.B. 1961, H.B. 2024, H.B. 2069, H.B. 2071, H.B. 2075, H.B. 2104, H.B. 2138, H.B. 2146, H.B. 2163, H.B. 2216, H.B. 2261, H.B. 2262, H.B. 2284, H.B. 2294, H.B. 2318.

COMMITTEE ON RULES

H.B. 1789, H.B. 1978, H.B. 1990, H.B. 2208, H.B. 2213, H.B. 2271, H.B. 2295.

H.J.R. 522, H.J.R. 525, H.J.R. 526, H.J.R. 527, H.J.R. 537, H.J.R. 538, H.J.R. 542, H.J.R. 549, H.J.R. 562, H.J.R. 563, H.J.R. 567, H.J.R. 572, H.J.R. 578, H.J.R. 579, H.J.R. 583, H.J.R. 596, H.J.R. 604, H.J.R. 605, H.J.R. 606, H.J.R. 639, H.J.R. 640, H.J.R. 641, H.J.R. 642, H.J.R. 643, H.J.R. 644, H.J.R. 645, H.J.R. 646, H.J.R. 647, H.J.R. 648, H.J.R. 649, H.J.R. 650, H.J.R. 651, H.J.R. 652, H.J.R. 653, H.J.R. 654, H.J.R. 655, H.J.R. 656, H.J.R. 657, H.J.R. 658, H.J.R. 659, H.J.R. 660, H.J.R. 661, H.J.R. 662, H.J.R. 663, H.J.R. 664, H.J.R. 665, H.J.R. 666, H.J.R. 667, H.J.R. 668, H.J.R. 669, H.J.R. 670, H.J.R. 671, H.J.R. 672, H.J.R. 673, H.J.R. 674, H.J.R. 675, H.J.R. 676, H.J.R. 677, H.J.R. 678, H.J.R. 679, H.J.R. 680, H.J.R. 681, H.J.R. 682, H.J.R. 683, H.J.R. 684, H.J.R. 685, H.J.R. 686, H.J.R. 687, H.J.R. 688, H.J.R. 689, H.J.R. 690, H.J.R. 691, H.J.R. 692, H.J.R. 693, H.J.R. 694, H.J.R. 695, H.J.R. 696, H.J.R. 697, H.J.R. 698, H.J.R. 699, H.J.R. 700, H.J.R. 701, H.J.R. 702, H.J.R. 703, H.J.R. 704, H.J.R. 705, H.J.R. 706, H.J.R. 707, H.J.R. 708, H.J.R. 709, H.J.R. 710, H.J.R. 711, H.J.R. 712, H.J.R. 713, H.J.R. 714, H.J.R. 715, H.J.R. 716, H.J.R. 717, H.J.R. 718, H.J.R. 719, H.J.R. 720, H.J.R. 721, H.J.R. 722, H.J.R. 723, H.J.R. 724, H.J.R. 725, H.J.R. 726, H.J.R. 727, H.J.R. 728, H.J.R. 729, H.J.R. 730, H.J.R. 731, H.J.R. 732, H.J.R. 733, H.J.R. 734, H.J.R. 735, H.J.R. 736, H.J.R. 737, H.J.R. 738, H.J.R. 739, H.J.R. 740, H.J.R. 741, H.J.R. 742, H.J.R. 743, H.J.R. 744, H.J.R. 745, H.J.R. 746, H.J.R. 747, H.J.R. 748, H.J.R. 749, H.J.R. 750, H.J.R. 751, H.J.R. 752, H.J.R. 753, H.J.R. 754, H.J.R. 755, H.J.R. 756, H.J.R. 757, H.J.R. 758, H.J.R. 759, H.J.R. 760, H.J.R. 761, H.J.R. 762, H.J.R. 763, H.J.R. 764, H.J.R. 765, H.J.R. 766, H.J.R. 767, H.J.R. 768, H.J.R. 769, H.J.R. 770, H.J.R. 771, H.J.R. 772, H.J.R. 773, H.J.R. 774, H.J.R. 775, H.J.R. 776, H.J.R. 777, H.J.R. 778, H.J.R. 779, H.J.R. 780, H.J.R. 781, H.J.R. 782, H.J.R. 783, H.J.R. 784, H.J.R. 785, H.J.R. 786, H.J.R. 787.

Pursuant to the provisions of House Joint Resolution No. 575 of the 2021 Regular Session, certain bills and joint resolutions were continued from the 2021 Regular Session of the General Assembly in the several House committees, as follows:

COMMITTEE ON AGRICULTURE, CHESAPEAKE AND NATURAL RESOURCES

S.B. 1115, S.B. 1135, S.B. 1143, S.B. 1161, S.B. 1164, S.B. 1188, S.B. 1193, S.B. 1194, S.B. 1199, S.B. 1210, S.B. 1258, S.B. 1265, S.B. 1274, S.B. 1280, S.B. 1282, S.B. 1290, S.B. 1291, S.B. 1311, S.B. 1319, S.B. 1354, S.B. 1374, S.B. 1390, S.B. 1393, S.B. 1396, S.B. 1402, S.B. 1404, S.B. 1411, S.B. 1412, S.B. 1417, S.B. 1453.

COMMITTEE ON APPROPRIATIONS

S.B. 1134, S.B. 1145, S.B. 1155, S.B. 1156, S.B. 1251, S.B. 1367. H.B. 1800.

COMMITTEE ON COMMUNICATIONS, TECHNOLOGY AND INNOVATION

S.B. 1098, S.B. 1349, S.B. 1365, S.B. 1392, S.B. 1458, S.B. 1462.

COMMITTEE ON COUNTIES, CITIES AND TOWNS

S.B. 1120, S.B. 1128, S.B. 1141, S.B. 1152, S.B. 1207, S.B. 1208, S.B. 1216, S.B. 1267, S.B. 1285, S.B. 1298, S.B. 1309, S.B. 1385, S.B. 1399, S.B. 1400, S.B. 1447, S.B. 1457.

COMMITTEE ON COURTS OF JUSTICE

S.B. 1103, S.B. 1104, S.B. 1105, S.B. 1108, S.B. 1113, S.B. 1122, S.B. 1123, S.B. 1125, S.B. 1138, S.B. 1142, S.B. 1165, S.B. 1168, S.B. 1180, S.B. 1181, S.B. 1184, S.B. 1206, S.B. 1209, S.B. 1213, S.B. 1226, S.B. 1234, S.B. 1241, S.B. 1242, S.B. 1248, S.B. 1261, S.B. 1262, S.B. 1266, S.B. 1270, S.B. 1272, S.B. 1297, S.B. 1306, S.B. 1315, S.B. 1324, S.B. 1325, S.B. 1329, S.B. 1336, S.B. 1339, S.B. 1391, S.B. 1397, S.B. 1415, S.B. 1426, S.B. 1431, S.B. 1437, S.B. 1442, S.B. 1443, S.B. 1456, S.B. 1461, S.B. 1465, S.B. 1468, S.B. 1475.

COMMITTEE ON EDUCATION

S.B. 1106, S.B. 1132, S.B. 1169, S.B. 1175, S.B. 1190, S.B. 1196, S.B. 1204, S.B. 1225, S.B. 1257, S.B. 1288, S.B. 1303, S.B. 1313, S.B. 1322, S.B. 1357, S.B. 1387, S.B. 1401, S.B. 1405, S.B. 1439.

COMMITTEE ON FINANCE

S.B. 1112, S.B. 1130, S.B. 1146, S.B. 1158, S.B. 1162, S.B. 1163, S.B. 1170, S.B. 1197, S.B. 1201, S.B. 1252, S.B. 1326, S.B. 1398, S.B. 1403, S.B. 1423, S.B. 1438.

COMMITTEE ON GENERAL LAWS

S.B. 1110, S.B. 1127, S.B. 1150, S.B. 1171, S.B. 1183, S.B. 1215, S.B. 1254, S.B. 1271, S.B. 1278, S.B. 1279, S.B. 1287, S.B. 1299, S.B. 1305, S.B. 1314, S.B. 1318, S.B. 1327, S.B. 1343, S.B. 1369, S.B. 1384, S.B. 1389, S.B. 1406, S.B. 1410, S.B. 1418, S.B. 1428, S.B. 1429, S.B. 1469, S.B. 1471.

COMMITTEE ON HEALTH, WELFARE AND INSTITUTIONS

S.B. 1102, S.B. 1121, S.B. 1147, S.B. 1154, S.B. 1176, S.B. 1178, S.B. 1187, S.B. 1189, S.B. 1205, S.B. 1220, S.B. 1221, S.B. 1227, S.B. 1235, S.B. 1237, S.B. 1302, S.B. 1304, S.B. 1307, S.B. 1316, S.B. 1320, S.B. 1321, S.B. 1328, S.B. 1333, S.B. 1338, S.B. 1356, S.B. 1362, S.B. 1366, S.B. 1421, S.B. 1427, S.B. 1436, S.B. 1464, S.B. 1472.

COMMITTEE ON LABOR AND COMMERCE

S.B. 1182, S.B. 1195, S.B. 1202, S.B. 1219, S.B. 1223, S.B. 1247, S.B. 1255, S.B. 1269, S.B. 1275, S.B. 1276, S.B. 1284, S.B. 1289, S.B. 1295, S.B. 1310, S.B. 1334, S.B. 1341, S.B. 1351, S.B. 1375, S.B. 1379, S.B. 1380, S.B. 1413, S.B. 1420, S.B. 1425.

COMMITTEE ON PRIVILEGES AND ELECTIONS

S.B. 1097, S.B. 1109, S.B. 1111, S.B. 1148, S.B. 1153, S.B. 1157, S.B. 1239, S.B. 1245, S.B. 1246, S.B. 1281, S.B. 1331, S.B. 1395, S.B. 1422, S.B. 1444.

S.J.R. 270, S.J.R. 272, S.J.R. 275, S.J.R. 289, S.J.R. 310, S.J.R. 322.

COMMITTEE ON PUBLIC SAFETY

S.B. 1119, S.B. 1129, S.B. 1198, S.B. 1256, S.B. 1296, S.B. 1300, S.B. 1301, S.B. 1363, S.B. 1381.

COMMITTEE ON RULES

S.B. 1172, S.B. 1273, S.B. 1408, S.B. 1414, S.B. 1473.

S.J.R. 276, S.J.R. 285, S.J.R. 286, S.J.R. 288, S.J.R. 292, S.J.R. 293, S.J.R. 294, S.J.R. 308, S.J.R. 323, S.J.R. 395.

COMMITTEE ON TRANSPORTATION

S.B. 1126, S.B. 1136, S.B. 1144, S.B. 1160, S.B. 1211, S.B. 1212, S.B. 1214, S.B. 1229, S.B. 1253, S.B. 1259, S.B. 1260, S.B. 1277, S.B. 1335, S.B. 1350, S.B. 1470.

INTRODUCTION OF LEGISLATION

The following was prefiled on the date indicated, ordered to be printed, and referred pursuant to § 30-19.3 of the Code of Virginia:

S.R. 501. 2021 Special Session I operating resolution.

(Prefiled February 10, 2021)

Patron--Locke

Referred to Committee on Rules

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.R. 502. Commending Danny TK Avula, M.D.

Patron--Dunnavant

S.R. 503. Celebrating the life of Franklin Delano Robertson.

Patron--Pillion

S.R. 504. Celebrating the life of Peter Michael Dolan, Jr.

Patron--McPike

S.R. 505. Commending Pamela Wooddy.

Patron--Hashmi

Pursuant to the provisions of House Joint Resolution No. 5001 and Senate Rule 26 (g), Senator DeSteph requested and was granted unanimous consent to introduce a joint resolution; subsequently, the following was presented and laid on the Clerk's Desk:

S.J.R. 5001. Commending the East Coast Surfing Championship.

Patrons--DeSteph; Delegate: Knight

RECESS

At 12:45 p.m., Senator Saslaw moved that the Senate recess until 1:50 p.m.

The motion was agreed to.

The hour of 1:50 p.m. having arrived, the Chair was resumed.

COMMITTEE REPORT

The following bill, having been considered by the committee in session, was reported by Senator Howell from the Committee on Finance and Appropriations:

S.B. 1100 (one thousand one hundred) with amendments.

SUPPLEMENTAL CALENDAR NO. 1

SENATE BILL ON FIRST READING

S.B. 1100 (one thousand one hundred) was read by title the first time.

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m. The Clerk was ordered to receive the committee report.

COMMITTEE REPORT

The following bills, having been considered by the committee in session were recommended for rereferral by the Committee on Rehabilitation and Social Services:

- **H.B. 1831** (one thousand eight hundred thirty-one) with the recommendation that it be rereferred to the Committee on Education and Health.
- **H.B. 2086** (two thousand eighty-six) with the recommendation that it be rereferred to the Committee on Education and Health.
- **H.B. 2098** (two thousand ninety-eight) with the recommendation that it be rereferred to the Committee on General Laws and Technology.
- **H.B. 2154** (two thousand one hundred fifty-four) with the recommendation that it be rereferred to the Committee on Education and Health.

- **H.B. 2166** (two thousand one hundred sixty-six) with the recommendation that it be rereferred to the Committee on Education and Health.
- H.B. 1831, H.B. 2086, H.B. 2154, and H.B. 2166 were rereferred to the Committee on Education and Health.
 - H.B. 2098 was rereferred to the Committee on General Laws and Technology.

Susan Carle Gelman

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

THURSDAY, FEBRUARY 11, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable L. Louise Lucas, Eighteenth Senatorial District, offered the following prayer:

I'm going to say something I know most of you are going to identify with. Every now and then a song gets stuck in your mind and for days it rolls around like water in your head. Well, I've been humming this song over the last few days and it's the hymn "Lift Every Voice and Sing," written by James Weldon Johnson and set to music by his brother J. Rosamond Johnson in 1900. Often referred to as the black national anthem, it is a song of celebration, and my favorite verse is, "God of our weary years, God of our silent tears, Thou who has brought us thus far on the way, Thou who has by Thy might led us into the light, Keep us forever in the path, we pray."

Father we lift our hands to thee. No other help do we know. We pray that You encompass the President of the United States and the men and women who make their hearts and ears attentive to Godly counsel and do that which is right in Your sight. We lift up in prayer today the men and women of the Virginia Senate, the Lieutenant Governor, Madame Clerk and her staff, because we believe You cause them to be men and women of integrity, who are obedient concerning us that we may lead a quiet and peaceful life in all Godliness and honesty. Your word declares that, "blessed is the nation whose God is the Lord," Psalms 33:12. Standing on Your word we receive Your blessings. Father You are our refuge and stronghold in times of trouble. Thus we are more than conquerors through Christ Jesus. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Sean Sukol, Assistant Coordinator of Committee Operations, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel.

A quorum was present.

After the roll call, Senator Suetterlein notified the Clerk of his presence.

On motion of Senator Newman, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Deeds, DeSteph--2. RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Lucas from the Committee on Education and Health:

- H.B. 1737 (one thousand seven hundred thirty-seven).
- **H.B. 1776** (one thousand seven hundred seventy-six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1790 (one thousand seven hundred ninety).
 - H.B. 1798 (one thousand seven hundred ninety-eight).
 - H.B. 1808 (one thousand eight hundred eight) with substitute.
 - H.B. 1817 (one thousand eight hundred seventeen).
 - H.B. 1827 (one thousand eight hundred twenty-seven).
 - H.B. 1873 (one thousand eight hundred seventy-three).
 - H.B. 1885 (one thousand eight hundred eighty-five).
 - **H.B.** 1904 (one thousand nine hundred four).
 - H.B. 1905 (one thousand nine hundred five).
 - H.B. 1913 (one thousand nine hundred thirteen).
 - H.B. 1918 (one thousand nine hundred eighteen) with substitute.
 - **H.B. 1940** (one thousand nine hundred forty) with amendments.
- **H.B. 1953** (one thousand nine hundred fifty-three) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 1989** (one thousand nine hundred eighty-nine) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 1995** (one thousand nine hundred ninety-five) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1998 (one thousand nine hundred ninety-eight).
- **H.B. 2035** (two thousand thirty-five) with the recommendation that it be rereferred to the Committee on Rehabilitation and Social Services.
- **H.B. 2058** (two thousand fifty-eight) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2111** (two thousand one hundred eleven) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2117** (two thousand one hundred seventeen) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2119 (two thousand one hundred nineteen).
- **H.B. 2123** (two thousand one hundred twenty-three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2166 (two thousand one hundred sixty-six).
 - H.B. 2182 (two thousand one hundred eighty-two).
- **H.B. 2204** (two thousand two hundred four) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2218** (two thousand two hundred eighteen) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2230** (two thousand two hundred thirty) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2314 (two thousand three hundred fourteen).

The following bills, having been considered by the committee in session, were recommended for rereferral by the Committee on Education and Health:

- **H.B. 1986** (one thousand nine hundred eighty-six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2206** (two thousand two hundred six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills, having been considered by the committee in session, were reported by Senator Barker from the Committee on General Laws and Technology:

- H.B. 1811 (one thousand eight hundred eleven) with amendments.
- H.B. 1812 (one thousand eight hundred twelve).
- H.B. 1816 (one thousand eight hundred sixteen).
- H.B. 1824 (one thousand eight hundred twenty-four).
- H.B. 1830 (one thousand eight hundred thirty).
- H.B. 1842 (one thousand eight hundred forty-two) with amendment.
- H.B. 1847 (one thousand eight hundred forty-seven) with substitute.
- H.B. 1848 (one thousand eight hundred forty-eight).
- H.B. 1849 (one thousand eight hundred forty-nine).
- H.B. 1882 (one thousand eight hundred eighty-two).
- H.B. 1891 (one thousand eight hundred ninety-one).
- H.B. 1931 (one thousand nine hundred thirty-one).
- H.B. 1943 (one thousand nine hundred forty-three).
- H.B. 1944 (one thousand nine hundred forty-four).
- H.B. 1967 (one thousand nine hundred sixty-seven).
- **H.B.** 1971 (one thousand nine hundred seventy-one).
- H.B. 1981 (one thousand nine hundred eighty-one) with amendments.
- **H.B. 1993** (one thousand nine hundred ninety-three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2014 (two thousand fourteen).
 - H.B. 2031 (two thousand thirty-one) with amendment.
 - H.B. 2046 (two thousand forty-six) with amendment.
 - H.B. 2085 (two thousand eighty-five).
 - H.B. 2140 (two thousand one hundred forty).
 - H.B. 2161 (two thousand one hundred sixty-one).
 - H.B. 2170 (two thousand one hundred seventy).
 - H.B. 2171 (two thousand one hundred seventy-one).
 - H.B. 2172 (two thousand one hundred seventy-two).H.B. 2175 (two thousand one hundred seventy-five) with substitute.
 - H.B. 2202 (two thousand two hundred two).
 - H.B. 2222 (two thousand two hundred twenty-two).
 - H.B. 2229 (two thousand two hundred twenty-nine).
 - H.B. 2249 (two thousand two hundred forty-nine) with substitute.
 - H.B. 2308 (two thousand three hundred eight).
 - H.B. 2320 (two thousand three hundred twenty) with substitute.
- **H.B. 2321** (two thousand three hundred twenty-one) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

Senator Edwards, from the Committee on the Judiciary, presented the following report:

SENATE OF VIRGINIA

February 10, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following person is qualified to be elected to the respective juvenile and domestic relations district court judgeship as follows:

The Honorable Andrea M. Stewart, of Spotsylvania, as a judge of the Fifteenth Judicial District for a term of six years commencing July 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- H.B. 1814 (one thousand eight hundred fourteen).
- H.B. 1821 (one thousand eight hundred twenty-one).
- H.B. 1878 (one thousand eight hundred seventy-eight).
- **H.B. 1895** (one thousand eight hundred ninety-five) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2002 (two thousand two) with substitute.
 - H.B. 2009 (two thousand nine).
 - H.B. 2012 (two thousand twelve) with amendments.
- **H.B. 2047** (two thousand forty-seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2110** (two thousand one hundred ten) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **H.B. 2128** (two thousand one hundred twenty-eight).
 - H.B. 2168 (two thousand one hundred sixty-eight) with amendments.
 - H.B. 2169 (two thousand one hundred sixty-nine).
- **H.B. 2194** (two thousand one hundred ninety-four) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2233 (two thousand two hundred thirty-three).
- **H.B. 2252** (two thousand two hundred fifty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2258 (two thousand two hundred fifty-eight).
- **H.B. 2263** (two thousand two hundred sixty-three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2290 (two thousand two hundred ninety).
 - H.B. 2310 (two thousand three hundred ten).
- H.B. 1776, H.B. 1895, H.B. 1953, H.B. 1986, H.B. 1989, H.B. 1993, H.B. 1995, H.B. 2047, H.B. 2058, H.B. 2110, H.B. 2111, H.B. 2117, H.B. 2123, H.B. 2194, H.B. 2204, H.B. 2206, H.B. 2218, H.B. 2230, H.B. 2252, H.B. 2263, and H.B. 2321 were rereferred to the Committee on Finance and Appropriations.
 - H.B. 2035 was rereferred to the Committee on Rehabilitation and Social Services.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

 $\textbf{S.R. 506.} \ Commending \ the \ Virginia \ State \ Police \ and \ the \ Virginia \ National \ Guard.$

Patron--Surovell

S.R. 507. Commending Danny TK Avula, M.D. Patron--Hashmi

CALENDAR

SENATE BILL ON SECOND READING SPECIAL AND CONTINUING ORDER

Senator Howell moved that **S.B. 1100** (one thousand one hundred) be made a special and continuing order for Friday, February 12, 2021, upon completion of the Senate Calendar.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1100, on motion of Senator Howell, was passed by for the day.

MEMORIAL RESOLUTIONS

On motion of Senator Locke, the Rules were suspended and the Committee on Rules was discharged from further consideration of the House joint resolutions that follow:

```
H.J.R. 642 (six hundred forty-two).
```

H.J.R. 643 (six hundred forty-three).

H.J.R. 644 (six hundred forty-four).

H.J.R. 645 (six hundred forty-five).

H.J.R. 646 (six hundred forty-six).

H.J.R. 647 (six hundred forty-seven).

H.J.R. 648 (six hundred forty-eight).

H.J.R. 649 (six hundred forty-nine).

H.J.R. 650 (six hundred fifty).

H.J.R. 651 (six hundred fifty-one).

H.J.R. 652 (six hundred fifty-two).

H.J.R. 653 (six hundred fifty-three).

H.J.R. 654 (six hundred fifty-four).

H.J.R. 662 (six hundred sixty-two).

H.J.R. 668 (six hundred sixty-eight).

H.J.R. 669 (six hundred sixty-nine).

H.J.R. 670 (six hundred seventy).

H.J.R. 678 (six hundred seventy-eight).

H.J.R. 679 (six hundred seventy-nine).

H.J.R. 680 (six hundred eighty).

- H.J.R. 685 (six hundred eighty-five).
- H.J.R. 686 (six hundred eighty-six).
- H.J.R. 687 (six hundred eighty-seven).
- H.J.R. 688 (six hundred eighty-eight).
- H.J.R. 689 (six hundred eighty-nine).
- **H.J.R. 690** (six hundred ninety).
- **H.J.R. 691** (six hundred ninety-one).
- H.J.R. 701 (seven hundred one).
- H.J.R. 702 (seven hundred two).
- H.J.R. 703 (seven hundred three).
- H.J.R. 704 (seven hundred four).
- H.J.R. 705 (seven hundred five).
- H.J.R. 706 (seven hundred six).
- H.J.R. 707 (seven hundred seven).
- H.J.R. 708 (seven hundred eight).
- H.J.R. 709 (seven hundred nine).
- H.J.R. 710 (seven hundred ten).
- H.J.R. 711 (seven hundred eleven).
- H.J.R. 716 (seven hundred sixteen).
- H.J.R. 722 (seven hundred twenty-two).
- H.J.R. 723 (seven hundred twenty-three).
- H.J.R. 724 (seven hundred twenty-four).
- **H.J.R.** 725 (seven hundred twenty-five).
- H.J.R. 726 (seven hundred twenty-six).
- H.J.R. 727 (seven hundred twenty-seven).
- H.J.R. 728 (seven hundred twenty-eight).
- H.J.R. 729 (seven hundred twenty-nine).
- H.J.R. 730 (seven hundred thirty).
- H.J.R. 731 (seven hundred thirty-one).
- **H.J.R.** 732 (seven hundred thirty-two).
- H.J.R. 733 (seven hundred thirty-three).
- H.J.R. 734 (seven hundred thirty-four).
- H.J.R. 735 (seven hundred thirty-five).
- H.J.R. 736 (seven hundred thirty-six). H.J.R. 737 (seven hundred thirty-seven).
- H.J.R. 738 (seven hundred thirty-eight).
- H.J.R. 739 (seven hundred thirty-nine).
- H.J.R. 740 (seven hundred forty).
- H.J.R. 741 (seven hundred forty-one).
- H.J.R. 742 (seven hundred forty-two).
- H.J.R. 743 (seven hundred forty-three).
- H.J.R. 757 (seven hundred fifty-seven).
- H.J.R. 766 (seven hundred sixty-six).
- H.J.R. 767 (seven hundred sixty-seven).
- H.J.R. 768 (seven hundred sixty-eight).
- H.J.R. 769 (seven hundred sixty-nine).
- **H.J.R.** 770 (seven hundred seventy).
- H.J.R. 771 (seven hundred seventy-one).
- H.J.R. 772 (seven hundred seventy-two). H.J.R. 773 (seven hundred seventy-three).
- H.J.R. 782 (seven hundred eighty-two).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Locke, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Locke, the following House joint resolutions were taken up and agreed to en bloc by a unanimous standing vote:

- H.J.R. 642 (six hundred forty-two).
- H.J.R. 643 (six hundred forty-three).
- H.J.R. 644 (six hundred forty-four).
- H.J.R. 645 (six hundred forty-five).
- H.J.R. 646 (six hundred forty-six).
- H.J.R. 647 (six hundred forty-seven).
- H.J.R. 648 (six hundred forty-eight).
- H.J.R. 649 (six hundred forty-nine).
- H.J.R. 650 (six hundred fifty).
- H.J.R. 651 (six hundred fifty-one).
- H.J.R. 652 (six hundred fifty-two).
- $\textbf{H.J.R. 653} \ (six \ hundred \ fifty-three).$
- H.J.R. 654 (six hundred fifty-four). H.J.R. 662 (six hundred sixty-two).
- H.J.R. 668 (six hundred sixty-eight).
- H.J.R. 669 (six hundred sixty-nine).
- H.J.R. 670 (six hundred seventy).
- H.J.R. 678 (six hundred seventy-eight).
- H.J.R. 679 (six hundred seventy-nine).
- H.J.R. 685 (six hundred eighty-five).
- H.J.R. 686 (six hundred eighty-six).
- H.J.R. 687 (six hundred eighty-seven).
- H.J.R. 688 (six hundred eighty-eight).
- H.J.R. 689 (six hundred eighty-nine).
- H.J.R. 690 (six hundred ninety).
- H.J.R. 691 (six hundred ninety-one).
- H.J.R. 701 (seven hundred one).
- H.J.R. 702 (seven hundred two).
- H.J.R. 703 (seven hundred three).
- H.J.R. 704 (seven hundred four).
- **H.J.R.** 705 (seven hundred five).
- H.J.R. 707 (seven hundred seven).
- H.J.R. 708 (seven hundred eight).
- H.J.R. 709 (seven hundred nine).

```
H.J.R. 710 (seven hundred ten).
H.J.R. 711 (seven hundred eleven).
H.J.R. 716 (seven hundred sixteen).
H.J.R. 722 (seven hundred twenty-two).
H.J.R. 723 (seven hundred twenty-three).
H.J.R. 724 (seven hundred twenty-four).
H.J.R. 725 (seven hundred twenty-five).
H.J.R. 726 (seven hundred twenty-six).
H.J.R. 727 (seven hundred twenty-seven).
H.J.R. 728 (seven hundred twenty-eight).
H.J.R. 729 (seven hundred twenty-nine).
H.J.R. 730 (seven hundred thirty).
H.J.R. 731 (seven hundred thirty-one).
H.J.R. 732 (seven hundred thirty-two).
H.J.R. 733 (seven hundred thirty-three).
H.J.R. 734 (seven hundred thirty-four).
H.J.R. 735 (seven hundred thirty-five).
H.J.R. 736 (seven hundred thirty-six).
H.J.R. 737 (seven hundred thirty-seven).
H.J.R. 738 (seven hundred thirty-eight).
H.J.R. 739 (seven hundred thirty-nine).
H.J.R. 740 (seven hundred forty).
H.J.R. 741 (seven hundred forty-one).
H.J.R. 742 (seven hundred forty-two).
H.J.R. 743 (seven hundred forty-three).
H.J.R. 757 (seven hundred fifty-seven).
H.J.R. 766 (seven hundred sixty-six).
H.J.R. 767 (seven hundred sixty-seven).
H.J.R. 768 (seven hundred sixty-eight).
H.J.R. 769 (seven hundred sixty-nine).
H.J.R. 770 (seven hundred seventy).
H.J.R. 771 (seven hundred seventy-one).
H.J.R. 772 (seven hundred seventy-two).
H.J.R. 773 (seven hundred seventy-three).
H.J.R. 782 (seven hundred eighty-two).
```

H.J.R. 680 (six hundred eighty) was taken up, as follows:

HOUSE JOINT RESOLUTION NO. 680

Celebrating the life of the Honorable Joan Hardie Munford.

WHEREAS, the Honorable Joan Hardie Munford, an esteemed health care administrator and distinguished former member of the Virginia House of Delegates, died on November 11, 2020; and

WHEREAS, Joan Munford attended Blacksburg High School, where she met her loving husband, Thomas William Munford, and then Radford University, where she majored in education; and

WHEREAS, along with her brother, Joan Munford founded a nursing home development corporation in the 1970s; she became a licensed nursing home administrator, working with Heritage Hall in Blacksburg and rising to the position of vice president of operations; and

WHEREAS, Joan Munford's work in health care and with the elderly led her to get involved with policymaking, and in 1981, she was elected to represent the 13th District in the Virginia House of Delegates; and

WHEREAS, Joan Munford would continue to serve in the Virginia House of Delegates for 12 years, at a time when she was one of only eight women in the legislative body; and

WHEREAS, Joan Munford served on the Committee on Corporations, Insurance and Banking; the Committee on Labor and Commerce; and the Committee on Education, with which she played a vital role by crafting the legislation that introduced family life education into the Commonwealth's school curriculum; and

WHEREAS, Joan Munford became a major proponent for workers' rights and was known in the Committee on Labor and Commerce for working across the aisle on contentious issues, securing the passage of workplace safety regulations and revisions to child labor laws, namely policies on the working conditions for migrant children; and

WHEREAS, Joan Munford's legislative legacy includes strengthening nurse practitioners' responsibilities, leading studies on health insurance premiums, securing millions of dollars to fund capital projects and academic programs at Virginia Polytechnic Institute and State University, and advancing early childhood and preschool education; and

WHEREAS, in her personal life, Joan Munford was known as a woman of faith and a lover of music; she was especially fond of playing piano and was often asked to play during receptions at the Governor's Mansion and at events for the legislature; and

WHEREAS, despite suffering from dementia in recent years, Joan Munford continued to play for her family and friends at Commonwealth Senior Living at Christiansburg until the final days of her life; and

WHEREAS, guided throughout her life by her deep and abiding faith, Joan Munford enjoyed worship and fellowship with her communities at Blacksburg Baptist Church and Blacksburg Presbyterian Church; and

WHEREAS, preceded in death by her parents, her loving husband, Thomas, and her brother, Thomas D. Hardie, Joan Munford will be fondly remembered and dearly missed by her daughter and son-in-law Mary and David Harder; her grandsons, James Harder and Benjamin Harder and his wife, Jacquelyn Harder; great-grandchildren, Cole and Airlie Harder; and numerous other family members, friends, and colleagues on both sides of the aisle; now, therefore, be it

RESOLVED by the House of Delegates, the Senate concurring, That the General Assembly hereby note with great sadness the loss of the Honorable Joan Hardie Munford, a cherished former member of the Virginia House of Delegates whose unwavering dedication to the Commonwealth touched countless lives; and, be it

RESOLVED FURTHER, That the Clerk of the House of Delegates prepare a copy of this resolution for presentation to her daughter and son-in-law, Mary and David Harder; her grandsons, James Harder and

Benjamin Harder and his wife, Jacquelyn Harder; and great-grandchildren, Cole and Airlie Harder, as an expression of the General Assembly's respect for her memory.

- H.J.R. 680, on motion of Senator Obenshain, was agreed to by a unanimous standing vote.
- H.J.R. 706 (seven hundred six), on motion of Senator Petersen, was passed by for the day.

On motion of Senator Locke, the questions on agreeing to the Senate resolutions that follow were considered en bloc.

On motion of Senator Locke, the following Senate resolutions were taken up, ordered to be engrossed, and agreed to en bloc by a unanimous standing vote:

```
S.R. 503 (five hundred three). S.R. 504 (five hundred four).
```

COMMENDING RESOLUTIONS

On motion of Senator Locke, the Rules were suspended and the Committee on Rules was discharged from further consideration of the House joint resolutions that follow:

```
H.J.R. 639 (six hundred thirty-nine).
H.J.R. 640 (six hundred forty).
H.J.R. 641 (six hundred forty-one).
H.J.R. 655 (six hundred fifty-five).
H.J.R. 656 (six hundred fifty-six).
H.J.R. 657 (six hundred fifty-seven).
H.J.R. 658 (six hundred fifty-eight).
H.J.R. 659 (six hundred fifty-nine).
H.J.R. 660 (six hundred sixty).
H.J.R. 661 (six hundred sixty-one).
H.J.R. 663 (six hundred sixty-three).
H.J.R. 664 (six hundred sixty-four).
H.J.R. 665 (six hundred sixty-five).
H.J.R. 666 (six hundred sixty-six).
H.J.R. 667 (six hundred sixty-seven).
H.J.R. 671 (six hundred seventy-one).
H.J.R. 672 (six hundred seventy-two).
H.J.R. 673 (six hundred seventy-three).
H.J.R. 674 (six hundred seventy-four).
H.J.R. 675 (six hundred seventy-five).
H.J.R. 676 (six hundred seventy-six).
H.J.R. 677 (six hundred seventy-seven).
H.J.R. 681 (six hundred eighty-one).
H.J.R. 682 (six hundred eighty-two).
H.J.R. 683 (six hundred eighty-three).
H.J.R. 684 (six hundred eighty-four).
H.J.R. 692 (six hundred ninety-two).
H.J.R. 693 (six hundred ninety-three).
H.J.R. 694 (six hundred ninety-four).
H.J.R. 695 (six hundred ninety-five).
H.J.R. 696 (six hundred ninety-six).
```

- H.J.R. 697 (six hundred ninety-seven).
- H.J.R. 698 (six hundred ninety-eight).
- H.J.R. 699 (six hundred ninety-nine).
- H.J.R. 700 (seven hundred).
- H.J.R. 712 (seven hundred twelve).
- H.J.R. 713 (seven hundred thirteen).
- H.J.R. 714 (seven hundred fourteen).
- H.J.R. 715 (seven hundred fifteen).
- H.J.R. 717 (seven hundred seventeen).
- H.J.R. 718 (seven hundred eighteen).
- H.J.R. 719 (seven hundred nineteen).
- H.J.R. 720 (seven hundred twenty).
- H.J.R. 721 (seven hundred twenty-one).
- H.J.R. 744 (seven hundred forty-four).
- H.J.R. 745 (seven hundred forty-five).
- H.J.R. 746 (seven hundred forty-six).
- H.J.R. 747 (seven hundred forty-seven).
- H.J.R. 748 (seven hundred forty-eight).
- H.J.R. 749 (seven hundred forty-nine).
- H.J.R. 750 (seven hundred fifty).
- H.J.R. 751 (seven hundred fifty-one).
- $\textbf{H.J.R. 752} \ (seven \ hundred \ fifty-two).$
- **H.J.R.** 753 (seven hundred fifty-three).
- H.J.R. 754 (seven hundred fifty-four).
- H.J.R. 755 (seven hundred fifty-five).
- H.J.R. 756 (seven hundred fifty-six).
- H.J.R. 758 (seven hundred fifty-eight).
- H.J.R. 759 (seven hundred fifty-nine).
- H.J.R. 760 (seven hundred sixty).
- H.J.R. 761 (seven hundred sixty-one).
- H.J.R. 762 (seven hundred sixty-two).
- H.J.R. 763 (seven hundred sixty-three).
- H.J.R. 764 (seven hundred sixty-four).
- **H.J.R.** 765 (seven hundred sixty-five).
- H.J.R. 774 (seven hundred seventy-four).
- H.J.R. 775 (seven hundred seventy-five).
- H.J.R. 776 (seven hundred seventy-six).
- H.J.R. 777 (seven hundred seventy-seven).
- H.J.R. 778 (seven hundred seventy-eight).
- H.J.R. 779 (seven hundred seventy-nine).
- H.J.R. 780 (seven hundred eighty).
- H.J.R. 781 (seven hundred eighty-one).
- H.J.R. 783 (seven hundred eighty-three).
- H.J.R. 784 (seven hundred eighty-four).
- H.J.R. 785 (seven hundred eighty-five).
- H.J.R. 786 (seven hundred eighty-six).
- H.J.R. 787 (seven hundred eighty-seven).

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0. RULE 36--0.

On motion of Senator Locke, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Locke, the following House joint resolutions were taken up and agreed to en bloc:

```
H.J.R. 639 (six hundred thirty-nine).
H.J.R. 640 (six hundred forty).
H.J.R. 641 (six hundred forty-one).
H.J.R. 655 (six hundred fifty-five).
H.J.R. 656 (six hundred fifty-six).
H.J.R. 657 (six hundred fifty-seven).
H.J.R. 658 (six hundred fifty-eight).
H.J.R. 659 (six hundred fifty-nine).
H.J.R. 660 (six hundred sixty).
H.J.R. 661 (six hundred sixty-one).
H.J.R. 663 (six hundred sixty-three).
H.J.R. 664 (six hundred sixty-four).
H.J.R. 665 (six hundred sixty-five).
H.J.R. 666 (six hundred sixty-six).
H.J.R. 667 (six hundred sixty-seven).
H.J.R. 671 (six hundred seventy-one).
H.J.R. 672 (six hundred seventy-two).
H.J.R. 673 (six hundred seventy-three).
H.J.R. 674 (six hundred seventy-four).
H.J.R. 675 (six hundred seventy-five).
H.J.R. 676 (six hundred seventy-six).
H.J.R. 677 (six hundred seventy-seven).
H.J.R. 681 (six hundred eighty-one).
H.J.R. 682 (six hundred eighty-two).
H.J.R. 683 (six hundred eighty-three).
H.J.R. 684 (six hundred eighty-four).
H.J.R. 692 (six hundred ninety-two).
H.J.R. 693 (six hundred ninety-three).
H.J.R. 694 (six hundred ninety-four).
H.J.R. 695 (six hundred ninety-five).
H.J.R. 696 (six hundred ninety-six).
H.J.R. 697 (six hundred ninety-seven).
H.J.R. 698 (six hundred ninety-eight).
H.J.R. 699 (six hundred ninety-nine).
H.J.R. 700 (seven hundred).
```

H.J.R. 712 (seven hundred twelve).

- H.J.R. 713 (seven hundred thirteen).
- H.J.R. 714 (seven hundred fourteen).
- H.J.R. 715 (seven hundred fifteen).
- H.J.R. 717 (seven hundred seventeen).
- H.J.R. 719 (seven hundred nineteen).
- H.J.R. 720 (seven hundred twenty).
- H.J.R. 721 (seven hundred twenty-one).
- H.J.R. 744 (seven hundred forty-four).
- H.J.R. 745 (seven hundred forty-five).
- H.J.R. 746 (seven hundred forty-six).
- H.J.R. 747 (seven hundred forty-seven).
- H.J.R. 748 (seven hundred forty-eight).
- H.J.R. 749 (seven hundred forty-nine).
- H.J.R. 750 (seven hundred fifty).
- H.J.R. 751 (seven hundred fifty-one).
- H.J.R. 752 (seven hundred fifty-two).
- H.J.R. 753 (seven hundred fifty-three).
- H.J.R. 754 (seven hundred fifty-four).
- H.J.R. 755 (seven hundred fifty-five).
- H.J.R. 756 (seven hundred fifty-six).
- H.J.R. 758 (seven hundred fifty-eight).
- H.J.R. 759 (seven hundred fifty-nine).
- H.J.R. 760 (seven hundred sixty).
- H.J.R. 761 (seven hundred sixty-one).
- H.J.R. 762 (seven hundred sixty-two).
- H.J.R. 764 (seven hundred sixty-four).
- H.J.R. 765 (seven hundred sixty-five).
- H.J.R. 774 (seven hundred seventy-four).
- H.J.R. 775 (seven hundred seventy-five).
- H.J.R. 776 (seven hundred seventy-six).
- H.J.R. 777 (seven hundred seventy-seven).
- **H.J.R.** 778 (seven hundred seventy-eight).
- H.J.R. 779 (seven hundred seventy-nine).
- H.J.R. 780 (seven hundred eighty).
- H.J.R. 781 (seven hundred eighty-one).
- H.J.R. 783 (seven hundred eighty-three).
- **H.J.R. 784** (seven hundred eighty-four).
- H.J.R. 785 (seven hundred eighty-five).
- H.J.R. 786 (seven hundred eighty-six).
- H.J.R. 787 (seven hundred eighty-seven).
- H.J.R. 718 (seven hundred eighteen), on motion of Senator Petersen, was passed by for the day.
- H.J.R. 763 (seven hundred sixty-three), on motion of Senator Petersen, was passed by for the day.

On motion of Senator Locke, the questions on agreeing to the Senate joint resolution and resolutions that follow were considered en bloc.

On motion of Senator Locke, the following Senate joint resolution and resolutions were taken up, ordered to be engrossed, and agreed to en bloc:

S.J.R. 5001 (five thousand one). S.R. 502 (five hundred two). S.R. 505 (five hundred five).

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

Jacke Ochran

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

FRIDAY, FEBRUARY 12, 2021

The Senate met at 12 m. and was called to order by the President pro tempore, Senator L. Louise Lucas.

The Honorable Jill H. Vogel, Twenty-seventh Senatorial District, offered the following prayer:

Heavenly Father, we give thanks for the many blessings You have bestowed on us, this Commonwealth, and this body. We humbly seek Your continued blessings on all that we do and support a democracy and the people of this Commonwealth. May our hearts be open to one another and filled with Your Spirit as we seek to engage in our civic responsibilities in a manner that is respectful of one another and in the service of others. Help us to serve with humility, compassion, and love. Protect our families, our Senate staff, and our friends who support us so that we may do this work. Most especially, I remember our beloved Ben Chafin, today and every day, as he is not with us in body but in spirit in this Senate session, and I ask Your blessing on his family. We ask this in the Lord's name. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Abigail Smalley, Senate Committee Operations Intern, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

After the roll call, Senators Barker and Stanley notified the Clerk of their presence.

On motion of Senator Howell, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--31. NAYS--5. RULE 36--0.

YEAS--Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Pillion, Reeves, Ruff, Saslaw, Stuart, Suetterlein, Surovell, Vogel--31.

NAYS--Deeds, DeSteph, McDougle, Peake, Petersen--5. RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Favola from the Committee on Rehabilitation and Social Services:

- **H.B. 1805** (one thousand eight hundred five) with amendments.
- **H.B. 1820** (one thousand eight hundred twenty) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1845 (one thousand eight hundred forty-five).

- H.B. 1879 (one thousand eight hundred seventy-nine).
- H.B. 1932 (one thousand nine hundred thirty-two) with substitute.
- H.B. 1962 (one thousand nine hundred sixty-two).
- H.B. 1973 (one thousand nine hundred seventy-three).
- H.B. 2010 (two thousand ten).
- H.B. 2092 (two thousand ninety-two).
- H.B. 2131 (two thousand one hundred thirty-one).
- **H.B. 2167** (two thousand one hundred sixty-seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2197 (two thousand one hundred ninety-seven) with amendments.
- **H.B. 2212** (two thousand two hundred twelve) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2266 (two thousand two hundred sixty-six) with substitute.
- **H.B. 2312** (two thousand three hundred twelve) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills, having been considered by the committee in session, were reported by Senator Marsden from the Committee on Transportation:

- **H.B. 1796** (one thousand seven hundred ninety-six) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1801 (one thousand eight hundred one).
 - H.B. 1813 (one thousand eight hundred thirteen).
 - **H.B.** 1828 (one thousand eight hundred twenty-eight).
 - H.B. 1846 (one thousand eight hundred forty-six) with substitute.
 - H.B. 1850 (one thousand eight hundred fifty) with amendments.
 - H.B. 1851 (one thousand eight hundred fifty-one).
- **H.B. 1893** (one thousand eight hundred ninety-three) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1901 (one thousand nine hundred one).
 - H.B. 1903 (one thousand nine hundred three).
 - H.B. 1960 (one thousand nine hundred sixty).
 - H.B. 1961 (one thousand nine hundred sixty-one).
 - H.B. 2024 (two thousand twenty-four).
 - H.B. 2069 (two thousand sixty-nine).
 - H.B. 2071 (two thousand seventy-one) with amendment.
- **H.B. 2138** (two thousand one hundred thirty-eight) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2146 (two thousand one hundred forty-six) with amendment.
 - H.B. 2216 (two thousand two hundred sixteen).
- **H.B. 2261** (two thousand two hundred sixty-one) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2262 (two thousand two hundred sixty-two).
 - H.B. 2294 (two thousand two hundred ninety-four).
 - H.B. 2318 (two thousand three hundred eighteen).

The following bills and joint resolutions, having been considered by the committee in session, were reported by Senator Locke from the Committee on Rules:

- H.B. 2213 (two thousand two hundred thirteen) with amendments.
- **H.B. 2271** (two thousand two hundred seventy-one) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

- H.J.R. 522 (five hundred twenty-two).
- H.J.R. 526 (five hundred twenty-six).
- H.J.R. 527 (five hundred twenty-seven) with amendments.
- **H.J.R. 542** (five hundred forty-two) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.J.R. 567 (five hundred sixty-seven).

The following bill, having been considered by the committee in session, was recommended for rereferral by the Committee on Rules:

- **H.B. 2295** (two thousand two hundred ninety-five) with the recommendation that it be rereferred to the Committee on the Judiciary.
- H.B. 1796, H.B. 1820, H.B. 1893, H.B. 2138, H.B. 2167, H.B. 2212, H.B. 2261, H.B. 2271, H.B. 2312, and H.J.R. 542 were rereferred to the Committee on Finance and Appropriations.
 - H.B. 2295 was rereferred to the Committee on the Judiciary.

RECESS

At 12:15 p.m., Senator Saslaw moved that the Senate recess until 1:55 p.m.

The motion was agreed to.

The hour of 1:55 p.m. having arrived, the Chair was resumed.

THE PRESIDENT PRESIDING

The President assumed the Chair.

COMMITTEE REPORT

The following bill, having been considered by the committee in session, was reported by Senator Howell from the Committee on Finance and Appropriations:

H.B. 2312 (two thousand three hundred twelve).

CALENDAR

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- H.B. 1790 (one thousand seven hundred ninety).
- H.B. 1798 (one thousand seven hundred ninety-eight).
- H.B. 1808 (one thousand eight hundred eight).
- H.B. 1812 (one thousand eight hundred twelve).
- H.B. 1814 (one thousand eight hundred fourteen).
- H.B. 1816 (one thousand eight hundred sixteen).

- H.B. 1824 (one thousand eight hundred twenty-four).
- H.B. 1827 (one thousand eight hundred twenty-seven).
- **H.B. 1830** (one thousand eight hundred thirty).
- H.B. 1842 (one thousand eight hundred forty-two).
- H.B. 1848 (one thousand eight hundred forty-eight).
- H.B. 1873 (one thousand eight hundred seventy-three).
- H.B. 1885 (one thousand eight hundred eighty-five).
- H.B. 1891 (one thousand eight hundred ninety-one).
- H.B. 1905 (one thousand nine hundred five).
- H.B. 1913 (one thousand nine hundred thirteen).
- H.B. 1918 (one thousand nine hundred eighteen).
- H.B. 1943 (one thousand nine hundred forty-three).
- H.B. 1944 (one thousand nine hundred forty-four).
- H.B. 1967 (one thousand nine hundred sixty-seven).
- H.B. 1971 (one thousand nine hundred seventy-one).
- H.B. 1981 (one thousand nine hundred eighty-one).
- H.B. 1998 (one thousand nine hundred ninety-eight).
- H.B. 2009 (two thousand nine).
- H.B. 2012 (two thousand twelve).
- H.B. 2031 (two thousand thirty-one).
- H.B. 2046 (two thousand forty-six).
- H.B. 2119 (two thousand one hundred nineteen).
- H.B. 2140 (two thousand one hundred forty).
- H.B. 2161 (two thousand one hundred sixty-one).
- H.B. 2166 (two thousand one hundred sixty-six).
- H.B. 2169 (two thousand one hundred sixty-nine).
- H.B. 2170 (two thousand one hundred seventy).
- H.B. 2171 (two thousand one hundred seventy-one).
- H.B. 2182 (two thousand one hundred eighty-two).
- H.B. 2202 (two thousand two hundred two).
- H.B. 2222 (two thousand two hundred twenty-two).
- H.B. 2233 (two thousand two hundred thirty-three).
- H.B. 2308 (two thousand three hundred eight).
- **H.B. 2310** (two thousand three hundred ten).
- H.B. 2314 (two thousand three hundred fourteen).
- H.B. 2320 (two thousand three hundred twenty).
- H.B. 1737 (one thousand seven hundred thirty-seven).
- H.B. 1811 (one thousand eight hundred eleven).
- H.B. 1817 (one thousand eight hundred seventeen).
- H.B. 1821 (one thousand eight hundred twenty-one).
- H.B. 1847 (one thousand eight hundred forty-seven).
- H.B. 1849 (one thousand eight hundred forty-nine).
- H.B. 1878 (one thousand eight hundred seventy-eight).
- H.B. 1882 (one thousand eight hundred eighty-two).
- H.B. 1904 (one thousand nine hundred four).
- H.B. 1931 (one thousand nine hundred thirty-one).
- **H.B. 1940** (one thousand nine hundred forty).
- H.B. 2002 (two thousand two).
- H.B. 2014 (two thousand fourteen).
- H.B. 2085 (two thousand eighty-five).

```
H.B. 2128 (two thousand one hundred twenty-eight).
```

H.B. 2168 (two thousand one hundred sixty-eight).

H.B. 2172 (two thousand one hundred seventy-two).

H.B. 2175 (two thousand one hundred seventy-five).

H.B. 2229 (two thousand two hundred twenty-nine).

H.B. 2249 (two thousand two hundred forty-nine).

H.B. 2258 (two thousand two hundred fifty-eight).

H.B. 2290 (two thousand two hundred ninety).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

```
H.B. 1790 (one thousand seven hundred ninety).
```

H.B. 1798 (one thousand seven hundred ninety-eight).

H.B. 1808 (one thousand eight hundred eight).

H.B. 1812 (one thousand eight hundred twelve).

H.B. 1814 (one thousand eight hundred fourteen).

H.B. 1816 (one thousand eight hundred sixteen).

H.B. 1824 (one thousand eight hundred twenty-four).

H.B. 1827 (one thousand eight hundred twenty-seven).

H.B. 1830 (one thousand eight hundred thirty).

H.B. 1842 (one thousand eight hundred forty-two).

H.B. 1848 (one thousand eight hundred forty-eight).

H.B. 1873 (one thousand eight hundred seventy-three).

H.B. 1885 (one thousand eight hundred eighty-five).

H.B. 1891 (one thousand eight hundred ninety-one).

H.B. 1905 (one thousand nine hundred five).

H.B. 1913 (one thousand nine hundred thirteen).

H.B. 1918 (one thousand nine hundred eighteen).

H.B. 1943 (one thousand nine hundred forty-three).

H.B. 1944 (one thousand nine hundred forty-four).

H.B. 1967 (one thousand nine hundred sixty-seven).

H.B. 1971 (one thousand nine hundred seventy-one).

H.B. 1981 (one thousand nine hundred eighty-one).

H.B. 1998 (one thousand nine hundred ninety-eight).

H.B. 2009 (two thousand nine).

H.B. 2012 (two thousand twelve).

H.B. 2031 (two thousand thirty-one).

H.B. 2046 (two thousand forty-six).

H.B. 2119 (two thousand one hundred nineteen).

- H.B. 2140 (two thousand one hundred forty).
- H.B. 2161 (two thousand one hundred sixty-one).
- H.B. 2169 (two thousand one hundred sixty-nine).
- H.B. 2170 (two thousand one hundred seventy).
- H.B. 2171 (two thousand one hundred seventy-one).
- H.B. 2182 (two thousand one hundred eighty-two).
- H.B. 2202 (two thousand two hundred two).
- H.B. 2222 (two thousand two hundred twenty-two).
- H.B. 2233 (two thousand two hundred thirty-three).
- H.B. 2308 (two thousand three hundred eight).
- H.B. 2310 (two thousand three hundred ten).
- H.B. 2314 (two thousand three hundred fourteen).
- H.B. 2320 (two thousand three hundred twenty).
- H.B. 1737 (one thousand seven hundred thirty-seven).
- H.B. 1811 (one thousand eight hundred eleven).
- H.B. 1817 (one thousand eight hundred seventeen).
- H.B. 1821 (one thousand eight hundred twenty-one).
- H.B. 1847 (one thousand eight hundred forty-seven).
- H.B. 1849 (one thousand eight hundred forty-nine).
- H.B. 1878 (one thousand eight hundred seventy-eight).
- H.B. 1882 (one thousand eight hundred eighty-two).
- H.B. 1904 (one thousand nine hundred four).
- H.B. 1931 (one thousand nine hundred thirty-one).
- H.B. 1940 (one thousand nine hundred forty).
- H.B. 2002 (two thousand two).
- H.B. 2014 (two thousand fourteen).
- H.B. 2085 (two thousand eighty-five).
- H.B. 2128 (two thousand one hundred twenty-eight).
- H.B. 2168 (two thousand one hundred sixty-eight).
- H.B. 2172 (two thousand one hundred seventy-two).
- H.B. 2175 (two thousand one hundred seventy-five).
- **H.B. 2229** (two thousand two hundred twenty-nine). **H.B. 2249** (two thousand two hundred forty-nine).
- H.B. 2258 (two thousand two hundred fifty-eight).
- **H.B. 2290** (two thousand two hundred ninety).
- **H.B. 2166** (two thousand one hundred sixty-six), on motion of Senator Barker, was rereferred to the Committee on Finance and Appropriations.

SENATE BILL ON SECOND READING SPECIAL AND CONTINUING ORDER

Action upon the Senate Calendar having been completed, **S.B. 1100** (one thousand one hundred), under special and continuing order, was taken up.

S.B. 1100 (one thousand one hundred) was read by title the second time.

The following amendments proposed by the Committee on Finance and Appropriations were offered:

Item 0 #1s

Revenues

Revenues	Language
----------	----------

Language:

Page 1, strike lines 34 through 38, and insert:

	First Year	Second Year	Total
Unreserved Beginning Balance	\$2,874,058,799	\$0	\$2,874,058,799
Additions to Balance	(\$1,284,491,604)	\$29,850,000	(\$1,254,641,604)
Official Revenue Estimates	\$21,941,832,509	\$22,577,524,514	\$44,519,357,023
Transfer	\$659,277,131	\$643,217,349	\$1,302,494,480
Total General Fund Resources Available for Appropriation	\$24,190,676,835	\$23,250,591,863	\$47,441,268,698

The appropriations made in this act from nongeneral fund revenues are based upon the following:

	First Year	Second Year	Total
Balance, June 30, 2020	\$6,915,611,972		\$6,915,611,972
Official Revenue Estimates	\$39,115,808,362	\$39,883,177,181	\$78,998,985,543
Lottery Proceeds Fund	\$685,031,123	\$690,903,334	\$1,375,934,457
Internal Service Fund	\$2,127,455,883	\$2,293,917,698	\$4,421,373,581
Bond Proceeds	\$2,479,633,162	\$294,775,137	\$2,774,408,299
Revenues Available for			
Total Nongeneral Fund Revenues Available for Appropriation	\$51,323,540,502	\$43,162,773,350	\$94,486,313,852
TOTAL PROJECTED REVENUES	\$75,514,217,337	\$66,413,365,213	\$141,927,582,550

Page 2, strike lines 1 through 15.

Explanation:

(This amendment reflects the net total of general and nongeneral fund revenue, transfer and balance adjustments encompassed in the Committee amendment package to SB 1100.)

Item 1 #1s

Legislative Department

General Assembly of Virginia

Language

Language:

Page 14, after line 14, insert:

- "Z.1. The Chair of the Senate Finance and Appropriations Committee shall appoint six members from the Senate Committee on Finance and Appropriations and the Chairman of the House Appropriations Committee shall appoint three members from the House Committee on Appropriations and three members of the House Committee on Finance to a Joint Subcommittee on Tax Policy. The Joint Subcommittee shall elect a chairman and vice-chairman from among its membership.
- 2. The goals and objectives of the Joint Subcommittee shall include (i) evaluating the fiscal impact of amendments to tax brackets, tax rates, credits, deductions, and exemptions, as well as any other factors it deems relevant to making Virginia's individual income tax system more fair and equitable; (ii) giving consideration to the fairness, certainty, convenience of payment, economy in collection, simplicity, neutrality, and economic efficiency of the Commonwealth's tax policies and any changes thereto; and (iii) recommending whether the General Assembly should amend the Code of Virginia.
- 3. To assist the Joint Subcommittee, the Chair of the Joint Subcommittee may appoint a workgroup which includes the staff of the House Committee on Finance, the House Committee on Appropriations, the Senate Committee on Finance and Appropriations, and any other stakeholders deemed appropriate. All agencies of the Commonwealth shall provide technical assistance to the Joint Subcommittee, upon request."

Explanation:

(This amendment establishes a Joint Subcommittee on Tax Policy to evaluate and make recommendations on potential changes to Virginia's tax policies, including changes to tax brackets, tax rates, credits, deductions, and exemptions, and any other changes it deems necessary. The Joint Subcommittee will consider factors such as equity, certainty, convenience of payment, economy in collection, simplicity, neutrality, economic efficiency and any other factors it deems relevant to the Commonwealth's tax policies.)

Item 1 #2s

Legislative Department

General Assembly of Virginia

Language

Language:

Page 14, after line 14, insert:

"Z. The staff of the Senate Fiscal Office, in collaboration with the staff of the Senate Committee on Finance and Appropriations, shall perform a comprehensive review of the compensation and benefits currently provided to legislative assistants, including a review of what might be considered best practies in other states regarding the compensation and benefits, professional growth, and training opportunities provided to legislative assistants, and make recommendations on any changes to the compensation and benefits, professional growth and training opportunities as a result of such review, and deliver a report on the staffs' findings and recommendations to the Clerk of the Senate and Chairman of the Senate Committee on Finance and Appropriations by November 1, 2021."

Explanation:

(This amendment directs the staff of the Senate Fiscal Office and the staff of the Senate Finance and Appropriations Committee to perform a review of the compensation and benefits currently provided to legislative assistants, and to make any recommendations regarding the structure of compensation and benefits to the Clerk of the Senate and the Chairman of the Senate Finance and Appropriations Committee by November 1, 2021.)

		Item 4 #1s	
Legislative Department	FY20-21	FY21-22	
Division of Capitol Police	\$0	\$846,907	GF

Language:

Page 15, line 44, strike "\$13,270,924" and insert "\$14,117,831".

Explanation:

(This amendment provides funding of \$846,907 GF the second year to increase the starting salaries of Division of Capitol Police police officers following graduation, manage salary compression, increase the starting salary of communications officers, and increase the salary of support and wage employees.)

-		Item 9 #1s	
Legislative Department	FY20-21	FY21-22	
Dr. Martin Luther King, Jr. Memorial Commission	\$0	\$50,000	GF

Language:

Page 18, line 20, strike "\$50,643" and insert "\$100,643".

Page 18, after line 26, insert:

"Authority: Title 30, Chapter 27, Code of Virginia.

Included within the appropriation for this Item is \$50,000 the second year from the general fund for the Dr. Martin Luther King, Jr. Memorial Commission to complete a pre-planning study to locate a memorial tribute to the late Senator Yvonne Miller on Virginia's Capitol Square or another location. The Department of General Services shall consult with the Commission, if requested by the Commission, to provide its capital project pre-planning expertise and Capitol Square operation and maintenance knowledge to the Commission as it formulates its study findings. The Commission will complete its pre-planning study and report its findings to the Governor, and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than November 1, 2021. The Department of General Services shall be compensated for its services provided to the Commission from the funds authorized in this Item."

Explanation:

(This amendment provides \$50,000 GF the second year for the Dr. Martin Luther King, Jr. Memorial Commission, with the assistance of the Department of General Services, to begin the process of creating a memorial in tribute to the late Senator Yvonne Miller.)

			Item 29.1 #1s	
Legislative Depar	rtment	FY20-21	FY21-22	
Behavioral He	alth Commission	\$0 0.00	\$348,744 4.00	GF FTE
Language:				
Page 26, after line	31, insert:			
	"LEGISLA	TIVE DEPARTMENT		
	§ 1-8.5. Behav	vioral Health Commission		
29.1 Behavioral Health	Commission	\$0	\$348,774	
Fund Sources:	General	\$0	\$348,774 "	

Explanation:

(This amendment provides \$348,774 GF the second year for the establishment of a legislative Behavioral Health Commission as provided for in Senate Bill 1273. The requested amount is for four positions and associated office-related costs that are phased-in over the year to reflect the timing necessary to startup the Commission.)

Item 33 #1s

Legislative Department

FY20-21 \$42,397

FY21-22 \$66,377 GF

Virginia Commission on Intergovernmental Cooperation

Language:

Page 31, line 15, strike "\$780,935" and insert "\$823,332". Page 31, line 15, strike "\$780,935" and insert "\$847,312".

Explanation:

(This amendment provides \$42,397 GF the first year and \$66,377 GF the second year to fully fund the amount projected for dues specified in this Item.)

Judicial DepartmentFY20-21FY21-22Supreme Court\$0\$100,000GF

Language:

Page 33, line 4, strike "\$14,594,927" and insert "\$14,694,927".

Explanation:

(This amendment provides \$100,000 GF the second year to fund the potential increase in the use of court-appointed experts caused by the passage of SB 1315 of the 2021 General Assembly, which broadens defendants' ability to introduce evidence regarding their mental state at the time of an alleged defense, dependent upon its final passage.)

Item 39 #1s

Judicial Department

Supreme Court Language

Language:

Page 36, after line 9, insert:

"P. The Office of the Executive Secretary of the Supreme Court shall prepare and distribute evaluation forms in all Circuit Court cases that are overseen by a retired judge for the purpose of collecting information on the number and types of cases referred to retired judges, and use such information to prepare and annually publish a report to be distributed to the members of the House Committee on Courts of Justice and the Senate Committee on the Judiciary, on or

about January 1, each year."

Explanation:

(This amendment requires the Supreme Court of Virginia (SCV) to distribute evaluation forms in all Circuit Court cases that are overseen by a retired judge, to collect the results of the evaluation forms and publish the findings to the members of the House Courts of Justice Committee and Senate Committee on the Judiciary. As a part of that report, the SCV will also report back on the number and types of cases referred to retired judges.)

_		Item 40 #1s	
Judicial Department	FY20-21	FY21-22	
Court of Appeals of Virginia	\$1,064,609 27.00	\$2,736,885 25.00	GF FTE

Language:

Page 36, line 24, strike "\$10,183,547" and insert "\$11,248,156". Page 36, line 24, strike "\$15,460,379" and insert "\$18,197,264".

Explanation:

(This amendment provides \$1.1 million GF and 27 FTE positions in FY 2021 and \$2.7 million GF and 25 FTE positions in FY 2022 for fixed costs and the prorated portion for 27 judicial staff positions in the first year and funding for two additional judgeships and 23 support positions, including clerk, IT and support staff, additional transcripts, and office accommodations in the second year associated with SB 1261 of the 2020 General Assembly Session, which expands the jurisdiction of the Court of Appeals. This is in addition to the funding of \$235,419 GF the first year and \$4.9 million GF for four judgeships and 27 staff positions in the second year included in the Governor's Introduced Budget, SB 1100.)

		Item 41 #1s	
Judicial Department	FY20-21	FY21-22	
Circuit Courts	\$0	(\$421,117)	GF

Language:

Page 37, line 12, strike "\$112,595,520" and insert "\$112,174,403".

Explanation:

(This amendment reflects the estimated Criminal Fund savings (court-appointed attorney costs) from opening a public defender office in Chesterfield County. A companion amendment to Item 48 (Indigent Defense Commission), requests funding to establish a public defender office for

Chesterfield County. Savings reflect estimated first year cost savings prorated for six months.)

		Item 42 #1s	
Judicial Department	FY20-21	FY21-22	
General District Courts	\$0	(\$486,803)	GF

Language:

Page 39, line 18, strike "\$128,797,150" and insert "\$128,310,347".

Explanation:

(The amendment reflects the estimated Criminal Fund savings (court-appointed attorney costs) from opening a public defender office in Chesterfield County. A companion amendment to Item 48 (Indigent Defense Commission), requests funding to establish a public defender office for Chesterfield County. Savings reflect estimated first year cost savings prorated for six months.)

		Item 43 #1s	
Judicial Department	FY20-21	FY21-22	
Juvenile and Domestic Relations District Courts	\$0	(\$171,931)	GF

Language:

Page 40, line 20, strike "\$107,020,623" and insert "\$106,848,692".

Explanation:

(The amendment reflects the estimated Criminal Fund savings (court-appointed attorney costs) from opening a public defender office in Chesterfield County. A companion amendment to Item 48 (Indigent Defense Commission), requests funding to establish a public defender office for Chesterfield County. Savings reflect estimated first year cost savings prorated for six months.)

		Item 48 #1s	
Judicial Department	FY20-21	FY21-22	
Indigent Defense Commission	\$0	(\$2,909,010)	GF

Language:

Page 43, line 10, strike "\$63,148,850" and insert "\$60,239,840".

Page 43, after line 34, insert:

"E. Within the appropriation for this Item, \$1,019,506 in the second year from the general fund

for Capital Indigent Defense Services (32702), shall be reallocated to Criminal Indigent Defense Services (32701), and the remaining funding contained within (32702) shall revert to the general fund, pursuant to Senate Bill 1165, of the 2021 General Assembly Session, which repeals the death penalty."

Explanation:

(This amendment redirects \$1.0 million GF the second year for Capital Indigent Defense Services to Criminal Indigent Defense Services within the Indigent Defense Commission, and reverts remaining funding currently appropriated for Capital Indigent Defense Services, pursuant to SB 1165, of the 2021 General Assembly Session repealing the death penalty, contingent upon its final passage.)

		Item 48 #2s	
Judicial Department	FY20-21	FY21-22	
Indigent Defense Commission	\$0 0.00	\$3,164,584 33.00	GF FTE

Language:

Page 43, line 10, strike "\$63,148,850" and insert "\$66,313,434".

Explanation:

(The amendment provides \$3.2 million GF and 33 positions in FY 2022, which represents the cost and staffing necessary to establish a public defender office for Chesterfield County, pursuant to SB 1442, of the 2021 General Assembly Session, contingent upon its final passage. Companion amendments to Items 41, 42, and 43 reduce estimated Criminal Fund (GF) court-appointed attorney prorated costs based on projected first year savings from opening the office, and a companion amendment to Item 48 redirects a portion of funding currently appropriated for Capital Indigent Defense Services, due to the passage of SB 1165, of the 2021 General Assembly Session, which repealed the death penalty.)

		Item 48 #3s	
Judicial Department	FY20-21	FY21-22	
Indigent Defense Commission	\$0 0.00	\$824,277 8.00	GF FTE

Language:

Page 43, line 10, strike "\$63,148,850" and insert "\$63,973,127".

Explanation:

(This amendment provides \$824,277 GF and 8.0 FTE attorney positions in FY 2022 related to SB 1261 of the 2021 General Assembly Session, which expands the jurisdiction of the Court of Appeals, contingent on its final passage.)

		Item 49 #1s	
Judicial Department	FY20-21	FY21-22	
Virginia Criminal Sentencing Commission	\$0 0.00	\$333,200 2.00	GF FTE

Language:

Page 43, line 45, strike "\$1,240,651" and insert "\$1,573,851".

Explanation:

(This amendment provides \$333,200 GF and 2.0 FTE positions in FY 2022 for costs associated with SB 1391 of the 2021 General Assembly, which is a recommendation of the Virginia State Crime Commission, contingent upon its final passage.)

		Item 57 #1s	
Executive Offices	FY20-21	FY21-22	
Attorney General and Department of Law	\$0 0.00	\$7,547,029 61.00	GF FTE

Language:

Page 48, line 42, strike "\$38,488,923" and insert "\$46,035,952".

Explanation:

(This amendment provides \$7.5 million GF and 61.00 FTE positions, including 48 attorney positions and 13 administrative support staff positions, associated with SB 1261 of the 2021 General Assembly Session, which expands the jurisdiction of the Court of Appeals, contingent on its final passage.)

		Item 68 #1s	
Administration	FY20-21	FY21-22	
Compensation Board	\$0	\$600,000	GF

Language:

Page 55, line 45, strike "\$499,752,342" and insert "\$500,352,342".

Explanation:

(This amendment provides \$600,000 GF the second year for the Compensation Board to contract for services to be provided by the Virginia Center for Policing Innovation to provide automated protective order notification services as an enhancement to the Statewide Automated Victim Notification System (SAVIN).)

Item 73 #1s

Administration

Compensation Board

Language

Language:

Page 69, after line 6, insert:

"O. Notwithstanding § 17.1-275, Code of Virginia, and any other provision of law, the marriage license fee shall be increased from \$30 to \$50. All marriage license fee revenue collected by the circuit court clerks, less \$30 per marriage license, shall be used exclusively to support compensation increases for the clerk and deputy clerks of the circuit courts."

Explanation:

(This amendment increases the marriage license fee to \$50 and specifies that any additional revenues shall be used exclusively for compensation increases for the clerk and deputy clerks of the circuit courts.)

Item 73 #2s

Administration

Compensation Board

Language

Language:

Page 69, after line 6, insert:

"O. Notwithstanding § 17.1-275, Code of Virginia, and any other provision of law, the divorce filing fee shall be increased from \$86 to \$100. All divorce filing fee revenue collected by the circuit court clerks, less \$86 per divorce filing, and less any revenue directed to the Courts Technology Fund established under §17.1-132, Code of Virginia, shall be used exclusively to support compensation increases for the clerk and deputy clerks of the circuit courts."

Explanation:

(This amendment increases the divorce filing fee to \$100 and specifies that any additional revenues, less that already directed to the Courts Technology Fund, shall be used exclusively for compensation increases for the clerk and deputy clerks of the circuit courts.)

		Item 75 #2s	
Administration	FY20-21	FY21-22	
Compensation Board	\$0	\$250,000	GF

Language:

Page 70, line 16, strike "\$5,514,904" and insert "\$5,764,904". Page 73, after line 41, insert:

"V.1. Pursuant to SB 1226 of the 2021 General Assembly, the Compensation Board shall work with the Virginia Association of Commonwealth's Attorneys to examine the staffing standards used to determine the number of positions needed for allocation to Commonwealth's Attorneys' offices statewide and revise the duties and workload measures used as the basis for determining the allocation of new positions, such that the standards are not based solely on metrics related to felony charges and convictions. The examination shall identify funding needs to support staffing for statutorily prescribed duties while also identifying funding needs for participation in special programs, discretionary duties, and current local supplemental funds allocated. To assist in this goal, the Compensation Board shall contract with the National Center for State Courts to perform a time study as to the comprehensive duties and responsibilities of Commonwealth's Attorneys' offices including, but not limited to, "in-court" obligations, the use of diversion programs and specialty dockets, expungement/rights restoration volume as well as other obligations reflected in the *Code of Virginia* (e.g., duties prescribed under §15.2-1627, et seq). The study shall identify objective metrics for inclusion in the revised staffing standard. Included within this appropriation is \$250,000 in the second year from the general fund for the purpose of contracting with the Center to perform the study. All Commonwealth's Attorneys shall participate in the study as needed and identified by the Compensation Board and the National Center for State Courts.

2. The Compensation Board shall provide a status report on the progress of the study and participants to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021. The Compensation Board shall deliver a report containing the results of the study, anticipated costs, and staffing standards methodology revisions under review or approved by the Board to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2022."

Explanation:

(This amendment provides \$250,000 GF in FY 2022 for the Compensation Board to contract with the National Center for State Courts to assist the Compensation Board in a study, working in collaboration with the Virginia Association of Commonwealth's Attorneys, of the staffing standards for Commonwealth's Attorney's offices, to revise the workload measures used as the basis for the allocation of new positions so that they are not based solely on metrics related to felony charges and convictions. The budget language is supplemental to the provisions of SB 1226 of the 2021 General Assembly.)

Item 75 #3s

Administration

Compensation Board

Language

Language:

Page 73, after line 41, insert:

"V. The Compensation Board shall review the plan to be developed by the Department of Criminal Justice Services by July 1, 2021 outlining law enforcement agencies' roles and engagement with the development of the Mental Health Awareness Response and Community Understanding Services Alert System, established pursuant to House Bill 5043 and Senate Bill 5038 of the 2020 Special Session I of the General Assembly, and shall survey sheriffs' offices to determine anticipated costs to support staffing and training needs to meet the requirements established by the plan. The Compensation Board shall provide a report to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021 of the findings of the survey and estimated costs to meet the requirements established by the plan."

Explanation:

(This amendment adds language directing the Compensation Board to review the plan to be developed by the Department of Criminal Justice Services by July 1, 2021 outlining law enforcement agencies' roles and engagement with the development of the Mental Health Awareness Response and Community Understanding Services Alert System, established pursuant to House Bill 5043 and Senate Bill 5038 of the 2020 Special Session I of the General Assembly and to provide a report to the Chairs of the Money Committees by November 1, 2021.)

Item 83 #2s

Administration

Department of Human Resource Management

Language

Language:

Page 82, after line 28, insert:

"N.1. In order to ensure the continuity of state government operations during the COVID-19 pandemic, the Department of Human Resource Management (DHRM), and with the assistance of all other state agencies upon their request, shall establish the criteria and develop a return-towork policy for all state employees presently working remotely on either a partial or full-time basis, pursuant to applicable federal and state guidelines.

2. Furthermore, DHRM, in collaboration with the Department of General Services, the Virginia Information Technologies Agency, and any other state agency upon request, shall examine the Commonwealth's existing telework policies, and how agency program and service delivery tools and methodologies employed during the COVID-19 pandemic may inform future policy objectives regarding the use of telework and alternative work schedules as a means of achieving administrative efficiencies, and reducing cost, and sustaining the hiring and retention of a highly qualified workforce. DHRM shall report to the Governor and the Chairmen of the House Appropriations and Senate Finance and Appropriations Committees on its findings, and suggested policies, instructions and guidelines by September 1, 2021."

Explanation:

(This amendment directs the Department of Human Resource Management, in collaboration with other state agencies, to develop return-to-work policies for state government employees to ensure the continuity and performance of state government operations, to examine agency performance and sevice delivery, and to examine the Commonwealth's telework policies and how they may inform future policy objectives. A report on such findings and recommendations is due to the Governor and the Chairmen of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021.)

		Item 87 #1s	
Administration	FY20-21	FY21-22	
Department of Elections	\$0	\$2,534,575	GF

Language:

Page 85, line 51, strike "\$6,275,378" and insert "\$8,809,953".

Explanation:

(This amendment restores the full reimbursement for compensation for general registrars and electoral board members that was included in the budget as passed in March 2020, but unallotted in April and removed during the 2020 Special Session I.)

Item 92 #1s

Administration

Virginia Information Technologies Agency

Language

Language:

Page 90, after line 41, insert:

"F. The Virginia Information Technologies Agency shall provide a network infrastructure report to the House Appropriations Committee, Senate Finance and Appropriations Committee, and

Joint Legislative Audit and Review Commission by November 1 of each year. The report shall indicate whether the Commonwealth's network infrastructure is adequate to meet the needs of state agencies, and if not, identify any needed upgrades. For each network infrastructure upgrade identified, the report shall specify the estimated cost and whether the upgrade is to the portion of the network maintained by the Virginia Information Technologies Agency or another state agency."

Explanation:

(This amendment directs the Virginia Information Technologies Agency (VITA) to report, by November 1 of each year, on the adequacy of the Commonwealth's network infrastructure maintained by either VITA or another state agency, and any potential upgrades needed, and associated costs. This language is a recommendation of the Joint Legislative Audit and Review Commission.)

		Item 95 #1s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Agriculture and Consumer Services	\$0	(\$100,000)	GF

Language:

Page 94, line 13, strike "\$5,642,932" and insert "\$5,542,932".

Page 94, line 20, strike "\$600,000" and insert "\$500,000".

Page 94, line 23, strike "legislation to be", and insert "the provisions of Senate Bill 1188 of".

Page 94, line 24, strike "considered by".

Explanation:

(This amendment reduces by \$0.1 million GF the proposed deposit to the Virginia Agriculture Food Assistance Program established pursuant to SB 1188 of the 2020 General Assembly. The Committee amendments provide a total of \$500,000 for this new initiative beginning in FY 22.)

		Item 97 #1s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Agriculture and Consumer Services	\$250,000	\$0	GF

Language:

Page 94, line 40, strike "\$23,620,243" and insert "\$23,870,243". Page 96, unstrike line 15 through line 16.

Explanation:

(This amendment restores \$250,000 GF in the first year for the Holiday Lake 4-H Center that was approved in 2020 Regular Session and subsequently unallotted. Holiday Lake is an evacuation location for several public school systems in Central Virginia.)

		Item 97 #2s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Agriculture and Consumer Services	\$0	\$131,649	GF

Language:

Page 94, line 41, strike "\$21,892,069" and insert "\$22,023,718". Page 96, line 8, strike "\$1,120,226" and insert "\$1,620,226".

Explanation

(This amendment increases funding for the International Marketing activities of the Virginia Department of Agriculture and Consumer Services by \$0.1 million GF in the second year to restore reductions included in Chapter 1289 of the 2020 General Assembly.)

Item 105 #1s

Agriculture and Forestry

Department of Agriculture and Consumer Services

Language

Language:

Page 99, following line 18, insert:

"D. The Office of the State Inspector General shall, with the assistance of the Office of Charitable and Regulatory Programs, review the regulatory structure of charitable gaming in Virginia, to include, at a minimum: (i) current permitting requirements and exemptions, (ii) net revenue dedicated to charitable activities and which types of gaming revenue is excluded from this calculation, (iii) charitable gaming occurring in remote locations not located in the same jurisdiction as the registered address of the charitable organization, (iv) enforcement of the "social quarters" and "members and guests" limitation, (v) the structure of the Charitable Gaming Board including any changes needed to prevent conflicts of interest, (vi) the adequacy of enforcement and resources dedicated to oversight activities of the Office of Charitable and Regulatory Programs, and (vii) whether regulation of charitable gaming would be more appropriately vested with the Virginia Lottery. The Office of the State Inspector General shall report on their findings to the General Assembly no later than December 1, 2021."

Explanation:

(This amendment directs the Office of the State Inspector General to investigate the operations

of the Charitable Gaming Board and recommend any changes needed to prevent conflicts of interest, including an assessment of whether responsibility for the regulation of charitable gaming should be placed under the Virginia Lottery.)

		Item 107 #1s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Forestry	\$0	(\$316,146)	GF

Language:

Page 99, line 47, strike "\$37,431,710" and insert "\$37,115,564".

Explanation:

(This amendment removes funding for a proposed new initiative between the Department of Forestry and the Virginia Department of Emergency Management. A companion amendment reverses the proposed new initiative at VDEM.)

		Item 112 #1s	
Commerce and Trade	FY20-21	FY21-22	
Economic Development Incentive Payments	\$0	\$6,330,000	GF

Language:

Page 105, line 19, strike "\$58,585,483" and insert "\$64,915,483".

Page 107, after line 47, insert:

"O. Out of the appropriation in this Item, \$6,330,000 the second year from the general fund shall be deposited to a special, nonreverting fund for the award of grants to a qualified shipping and logistics company in a qualified locality in accordance with legislation to be considered by the 2022 General Assembly and subject to performance metrics agreed to in a memorandum of understanding with the Commonwealth."

Explanation:

(This amendment provides \$6.3 million the second year from the general fund for the award of grants to a qualified shipping and logistics company. This is part of an incentives package for an economic development project endorsed by the Major Employment and Investment (MEI) Project Approval Commission.)

Commerce and Trade	FY20-21	FY21-22	
Economic Development Incentive Payments	\$0	\$1,000,000	GF

Language:

Page 105, line 19, strike "\$58,585,483" and insert "\$59,585,483". Page 106, line 23, strike "\$3,000,000" and insert "\$4,000,000".

Explanation:

(This amendment provides \$1.0 million the second year from the general fund for the Governor's Motion Picture Opportunity Fund, restoring second year funding that was previously unallotted.)

		Item 113 #1s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	(\$15,700,000)	\$0	GF

Language:

Page 108, line 13, strike "\$168,760,089" and insert "\$153,060,089".

Page 109, line 5, strike "\$70,700,000" and insert "\$55,000,000".

Page 109, line 15, strike "\$28,200,000" and insert "\$12,500,000".

Page 109, after line 48, insert:

"5. It is the intent of the General Assembly that funds from Virginia's allocation of moneys distributed under the federal Consolidated Appropriations Act, P.L. 116-260 (2020) for rental assistance be designated to continue the Virginia Rent and Mortgage Relief Program, and that such funds be given priority over general fund appropriations provided in paragraph E.1. of this item in supporting the continuation of the Virginia Rent and Mortgage Relief Program."

Explanation:

(This amendment reduces the GF appropriation to the Virginia Housing Trust Fund intended to support the Virginia Rent and Mortgage Relief Program and directs that funding distributed under the federal Consolidated Appropriations Act, P.L. 116-260 (2020) for rental assistance be designated as the primary source of funding for the Commonwealth's rental assistance program.)

Ttem 114 #1s

Commerce and Trade FY20-21 FY21-22

Department of Housing and \$0 \$500,000 GF Community Development

Language:

Page 110, line 42, strike "\$128,538,362" and insert "\$129,038,362".

Page 112, line 32 strike "\$2,500,000" and insert "\$3,000,000".

Page 112, line 34, strike" each year" and insert "the first year and \$1,500,000 the second year.

Explanation:

(This amendment would provide an additional \$500,000 GF the second year for the Industrial Revitalization Fund and designate the increase for the removal, renovation or modernization of port-related buildings and facilities in the cities of Portsmouth, Norfolk, Newport News, Richmond, or Front Royal.)

		Item 114 #2s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0	\$500,000	GF

Language:

Page 110, line 42, strike "\$128,538,362" and insert "\$129,038,362".

Page 114, after line 46, insert:

"O. Out of the amounts in this Item, \$500,000 the second year from the general fund is provided for the Lenowisco Planning District Commission and Cumberland Plateau Planning District Commission designated for initiatives intended to expand education and telehealth access. Such funds for grants shall be managed by the Virginia Coalfield Economic Development Authority."

Explanation:

(This amendment would provide \$500,000 GF the second year for the Lenowisco Planning District Commission and Cumberland Plateau Planning District Commission designated for initiatives intended to expand education and telehealth access. Funding for grants would be managed by the Virginia Coalfield Economic Development Authority.)

		Item 114 #3s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0 0.00	\$10,000,000 3.00	GF FTE

Language:

Page 110, line 42, strike "\$128,538,362" and insert "\$138,538,362".

Page 114, after line 46, insert:

- "O.1. Out of the amounts in this Item, \$10,000,000 the second year from the general fund is provided to establish a special, non-reverting Virginia Community Development Financial Institutions (CDFI) Fund to provide grants to community development financial institutions (CDFIs), community development enterprises (CDE), or other such similar entities as permitted by law, whose primary purpose is to provide financing in the form of loans, grants or forgivable loans to small businesses or community revitalization real estate projects in Virginia. The Fund shall consist of any funds appropriated to it by the general appropriation act and revenue from any other source, public or private. The Fund shall be established on the books of the Comptroller, and any funds remaining in the Fund at the end of a biennium shall not revert to the general fund but shall remain in the Fund. Interest earned on the Fund shall be credited to the Fund. Of the amounts included in this paragraph, up to \$300,000 the second year from the general fund is provided to the Department for administrative costs.
- 2. The Department is hereby authorized to develop appropriate criteria and guidelines for the use of funding provided to the Virginia Community Development Financial Institution Fund. The Department shall award grant funding based on these criteria and guidelines and may enter into a contractual agreement with eligible CDFIs or similar private entities to make grants and lending to small businesses adversely impacted by the COVID pandemic. An eligible qualifying CDFI shall be a community development bank, community development credit union, or other similar private entity that the Department finds is (i) established to conduct business legally within the Commonwealth; (ii) subject to oversight by federal or state financial institutions or insurance regulatory agencies, as appropriate; and (iii) eligible for certification by the U.S. Department of Treasury as a community development financial institution or other similar charter or principles which require support of small businesses.
- 3. The community development bank, community development credit union, or other similar organization is intended to be a source of targeted lending and investment with the capacity to provide a high degree of leveraging for economic development and business support activities within communities throughout the Commonwealth. These activities may include loans and investments to start or expand small businesses, operating and working capital, property renovation or development, and financial services with a focus on small businesses impacted by the COVID pandemic. The entity may also provide services that help ensure that credit is used effectively, such as technical assistance to small businesses and credit counseling to consumers.
- 4. On or before December 1 of each year, the Department shall report to the Secretary of Commerce and Trade, the Governor, and the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations on such other matters regarding the Fund as the Department may deem appropriate, including the amount of funding committed to projects from the Fund, or other items as may be requested by any of the foregoing persons to whom such report is to be submitted."

Explanation:

(This amendment establishes the Virginia CDFI Fund (the Fund) within the Department of Housing and Community Development which may be comprised of appropriations, grants and loans from federal, state and private sources, and other types of financial assistance, to provide capital through grants to community development financial institutions (CDFIs), community development enterprises (CDE), or other such similar entities as permitted by law, whose primary purpose is to provide financing in the form of loans, grants or forgivable loans to small business or community revitalization real estate projects in Virginia. The amendment provides \$10.0 million GF the second year to capitalize the Fund.)

Item 114 #4s

Commerce and Trade

Department of Housing and Community Development

Language

Language:

Page 114, line 46, after "basis.", insert:

"O. The Commission on Local Government shall review the fiscal effects of mandatory property tax exemptions on the capacity of local governments to deliver essential services to the public. As part of the review, the Commission shall ascertain the impact of the exemptions on property tax collections, the shift to and reliance on other local revenues to compensate for exempted properties, the additional fiscal stress placed on non-exempted properties and non-exempted local taxpayers, and the ability of local governments to meet spending needs. In addition, the Commission shall develop and include in its review potential recommendations to mitigate the fiscal impacts on local governments tied to these state tax exemption initiatives. The Commission shall report its findings to the Governor, the Joint Subcommittee on Local Government Fiscal Stress, and the Chairs of the Senate Committee on Finance and Appropriations, the House Committee on Finance, and the House Committee on Appropriations by November 1, 2021."

Explanation:

(This amendment directs the Commission on Local Government to undertake a review of the effects of mandatory property tax exemptions on local governments and recommend potential options for mitigating their fiscal impacts.)

Item 114 #5s

Commerce and Trade

Department of Housing and Community Development

Language

Language:

Page 113, line 8, after "private sector", insert ", except as provided for in paragraph L.6. of this item. ".

Page 113, after line 36, insert:

"6. The Department shall create a pilot program within VATI, with awards not to exceed 10 percent of total available VATI funds in fiscal year 2022, to which public broadband authorities may apply without investment from the private sector."

Explanation:

(This amendment directs the Department of Housing and Community Development to create a pilot program within VATI, with awards not to exceed 10 percent of total available VATI funds in FY22, to which public broadband authorities may apply without investment from the private sector. This is the recommendation of the Broadband Advisory Council.)

		Item 114 #6s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0	\$294,000	GF
Language:			
Page 110, line 42, strike "\$128,538,362" and			

Page 111, line 11, strike "\$75,971" and insert "\$89,971".

Page 111, line 15, strike "\$75,971" and insert "\$89,971".

Page 111, line 20, strike "\$75,971" and insert "\$89,971".

Page 111, line 22, strike "\$75,971" and insert "\$89,971".

Page 111, line 24, strike "\$75,971" and insert "\$89,971".

Page 111, line 26, strike "\$75,971" and insert "\$89,971".

Page 111, line 28, strike "\$75,971" and insert "\$89,971".

Page 111, line 30, strike "\$151,943" and insert "\$165,943".

Page 111, line 32, strike "\$75,971" and insert "\$89,971".

Page 111, line 34, strike "\$75,971" and insert "\$89,971".

Page 111, line 36, strike "\$75,971" and insert "\$89,971".

Page 111, line 38, strike "\$75,971" and insert "\$89,971".

Page 111, line 40, strike "\$75,971" and insert "\$89,971".

Page 111, line 42, strike "\$75,971" and insert "\$89,971".

Page 111, line 44, strike "\$113,957" and insert "\$127,957".

Page 111, line 46, strike "\$75,971" and insert "\$89,971".

Page 112, line 1, strike "\$75,971" and insert "\$89,971".

Page 112, line 3, strike "\$75,971" and insert "\$89,971".

Page 112, line 5, strike "\$75,971" and insert "\$89,971".

Page 112, line 7, strike "\$75,971" and insert "\$89,971".

Page 112, line 9, strike "\$151,943" and insert "\$165,943".

Explanation:

(This amendment provides an additional \$294,000 the second year from the general fund to provide each of the 21 Planning District Commissions with an increase of \$14,000 per year. The 2020 Appropriation Act provided this amount in both years of the biennium; however, the increase was unallotted due to COVID-19 impacts to the state budget.)

		Item 115 #1s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0	\$250,000	GF

Language:

Page 114, line 48, strike "\$14,789,114" and insert "\$15,039,114". Page 114, line 56, strike "\$14,500,000" and insert "\$14,750,000".

Explanation:

(This amendment restores \$250,000 GF the second year that was unallotted and subsequently reduced in the Enterprise Zone program.)

		Item 120 #1s	
Commerce and Trade	FY20-21	FY21-22	
Department of Labor and Industry	(\$175,073)	(\$500,290)	GF

Language:

Page 116, line 14, strike "\$1,773,255" and insert "\$1,598,182".

Page 116, line 14, strike "\$2,520,193" and insert "\$2,019,903".

Page 116, line 18, strike "\$596,794" and insert "\$421,721".

Page 116, line 18, strike "\$1,343,732" and insert "\$843,442".

Explanation:

(This amendment removes funding in the amounts of \$175,073 the first year and \$500,290 the second year from the general fund for the Labor and Employment Law Division. This is funding that was included in Chapter 1289, 2020 Acts of Assembly, to support implementation and enforcement of the provisions of SB 481 (2020), which ultimately did not pass.)

Commerce and Trade

Department of Small Business and Supplier Diversity

Language

Language:

Page 122, after line 37, insert:

"I.1.Notwithstanding § 2.2-1604, Code of Virginia, "Small business" shall be defined as a business that is at least 51 percent independently owned and controlled by one or more individuals, or in the case of a cooperative association organized pursuant to Chapter 3 (§ 13.1-301 et seq.) of Title 13.1 as a nonstock corporation, is at least 51 percent independently controlled by one or more members, who are U.S. citizens or legal resident aliens and, together with affiliates, has 250 or fewer employees or average annual gross receipts of \$10 million or less averaged over the previous three years. One or more of the individual owners or members shall control both the management and daily business operations of the small business.

2. Notwithstanding § 2.2-4310, Code of Virginia, and for purposes of the Commonwealth's SWaM program certification, "Small business" shall mean (i) a business, independently owned and controlled by one or more individuals, or (ii) in the case of a cooperative association organized pursuant to Chapter 3 (§ 13.1-301 et seq.) of Title 13.1 as a nonstock corporation, controlled by one or more members, who are U.S. citizens or legal resident aliens, and together with affiliates, has 250 or fewer employees, or annual gross receipts of \$10 million or less averaged over the previous three years. One or more of the individual owners or members shall control both the management and daily business operations of the small business."

Explanation:

(This amendment amends the definition of small business to include certain co-ops.)

		Item 130 #1s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Economic Development Partnership	\$0	\$2,000,000	GF

Language:

Page 123, line 44, strike "\$40,302,309" and insert "\$42,302,309".

Page 124, line 49, after "the first year and", strike "\$5,020,387" and insert "\$7,020,387".

Explanation:

(This amendment provides an additional \$2.0 million GF the second year for the Talent Accelerator Program to provide custom recruitment and training services for high-value projects creating new jobs in Virginia. Funding supports staff compensation and benefits, contract trainers, program support, client recruitment, travel and other expenses.)

Item 130 #2s

Commerce and Trade	FY20-21	FY21-22
Virginia Economic Development Partnership	\$0	\$930,000 GF

Language:

Page 123, line 44, strike "\$40,302,309" and insert "\$41,232,309".

Page 125, after line 4, insert:

"M. Out of the amounts in this item, \$930,000 the second year from the general fund is provided to support implementation of Virginia's International Trade Plan."

Explanation:

(This amendment provides \$930,000 GF the second year to support initiatives identified in the International Trade Plan.)

		Item 131 #1s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Employment Commission	\$0	(\$6,752,701)	GF

Language:

Page 125, line 13, strike "\$574,596,796" and insert "\$567,844,095".

Page 126, line 11, strike "\$7,502,701" and insert "\$750,000".

Explanation:

(This amendment updates the estimated interest on federal cash advances for unemployment insurance benefits.)

Item 134 #1s

Commerce and Trade

Virginia Tourism Authority

Language

Language:

Page 127, line 27, strike "Coalfield Regional Tourism Authority" and insert "Heart of Appalachia Tourism Authority".

Explanation:

(This amendment updates a reference to the Coalfield Regional Tourism Authority to the Heart

of Appalachia Tourism Authority in accordance with the name change effectuated in SB 1399. This is a technical amendment.)

		Item 134 #2s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Tourism Authority	\$0	\$2,140,000	GF

Language:

Page 126, line 39, strike "\$21,093,272" and insert "\$23,233,272".

Page 128, after line 20, insert:

"M. Out of the amounts in this Item, \$2,140,000 the second year from the general fund is provided for grants to promote tourism in accordance with the provisions of Senate Bill 1398, as enacted during the 2021 Regular Session of the General Assembly."

Explanation:

(This amendment provides \$2,140,000 the second year from the general fund for grants to promote tourism pursuant to SB 1398. This is the amount of estimated GF revenue raised by sales tax imposed on accommodations fees.)

		Item 135 #1s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Innovation Partnership Authority	\$0	\$1,400,000	GF

Language:

Page 128, line 28, strike "\$40,450,000" and insert "\$41,850,000".

Page 132, line 16, strike "\$925,000" and insert "\$1,925,000".

Page 132, line 29, after "and" strike the remainder of the line.

Page 132, strike lines 30 through 31, and insert:

"(iv) grants to CCAM for seedling research project costs that enable CCAM to market new research programs to prospective and existing industry members. These funds shall not revert back to the general fund at the end of the fiscal year."

Page 132, line 39, after "facility." insert: "These funds shall not revert back to the general fund at the end of the fiscal year."

Page 132, after line 39, insert:

"4. Out of the appropriation in this item, VIPA shall provide \$400,000 the second year from the general fund to CCAM for the purposes of: (i) attracting federal funds for research projects to be conducted at CCAM, including marketing, travel, grant proposal writing, and business development costs; (ii) matching funds for federal research programs; and (iii) federal research

program costs not reimbursable on federal research awards. These funds shall not revert back to the general fund at the end of the fiscal year."

Page 132, line 40, strike "4" and insert "5".

Page 132, line 42, strike "5" and insert "6".

Page 132, line 48, after "disbursements;" insert:

"(iv) all efforts and costs associated with obtaining federal research grants".

Page 132, line 48, strike "iv" and insert "v".

Explanation:

(This amendment provides an increase of \$1.0 million GF in rent, operating support and maintenance for the Center for Advanced Manufacturing (CCAM), bringing the total to \$1.9 million in both FY 2021 and FY 2022. It includes increased flexibility on currently appropriated matching funds and provides an additional \$400,000 in GF resources for CCAM to attract federal funds.)

		Item 135 #2s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Innovation Partnership Authority	\$0	\$100,000	GF

Language:

Page 128, line 28, strike "\$40,450,000" and insert "\$40,550,000".

Page 134, after line 16, insert:

- "S.1. Out of the appropriation in this item, \$100,000 the second year from the general fund is provided for the Virginia Nuclear Energy Consortium Authority (VNECA) for the purpose of developing a proposal to create a nuclear research and innovation hub in Virginia. In creating this proposal, VNECA shall convene a workgroup that includes, but is not limited to, the Department of Minerals, Mines and Energy, the Virginia Economic Development Partnership, the Virginia Innovation Partnership Authority, Virginia public colleges and universities, and relevant industry representatives.
- 2. VNECA shall submit a report that includes planning activities and the final proposal to the Secretary of Commerce and Trade, Secretary of Education, Chairs of the House Appropriations Committee, the House Labor and Commerce Committee, the Senate Finance and Appropriations Committee, and the Senate Commerce and Labor Committee no later than November 1, 2021."

Explanation:

(This amendment provides \$100,000 GF the second year for the Virginia Nuclear Energy Consortium Authority as a one-time planning grant to develop a nuclear research and

innovation hub in Virginia.)

Item 137 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 139, after line 29, insert:

"Q. The Superintendent of Public Instruction shall convene a workgroup to make recommendations on the staffing standards for school nurses in local school divisions. The workgroup shall include recommendations on: (i) the appropriate staffing ratio of school nurses; and (ii) the desired qualifications and training for school nurses. The workgroup shall include at least (i) the Secretary of Education; (ii) two local school division representatives, including one superintendent; (iii) a member of a local school board; (iv) two school nurses; and (v) one member of the Board of Education. The recommendations shall be submitted to the Governor and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than October 1, 2021. Such recommendations shall detail any necessary legislative or budgetary changes to implement recommendations."

Explanation:

(This amendment directs the Superintendent of Public Instruction to convene a workgroup to make recommendations on the staffing standards for school nurses. The recommendations shall include the appropriate school nurse staffing ratio and the required qualifications and training for school nurses.)

Item 137 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 139, after line 29, insert:

"Q. To support the development and administration of a growth assessment system, \$10,000,000 in federal relief funds are provided from the federal funds reserved by the Department of Education from the Coronavirus Response and Relief Supplemental Appropriations Act, (P.L. 116-260) for such purpose. The system shall include diagnostic assessments in the fall and winter and a growth measurement assessment administered in the spring to measure student progress and competency in grades three through eight in English and mathematics for use in the 2021-2022 and 2022-2023 school year, pursuant to the provisions in Senate Bill 1357 of the 2021 General Assembly."

Explanation:

(This amendment provides \$10.0 million in federal relief funds for the Department of Education to provide and administer a diagnostic assessment tool for use in the 2021-22 and 2022-23 school year pursuant to the provisions of Senate Bill 1357.)

Item 138 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 11, insert:

"6. The Department of Education shall collect and publish data annually from each private special education day school on: (i) the number of teachers who are not fully endorsed in the content that they are teaching; (ii) the number of teachers who have less than one year of classroom experience; (iii) the number of teachers who are provisionally licensed; (iv) the type of academic credentials attained by each teacher and in what subjects; (v) the number of career and technical education credentials conferred by each school on its graduating students in each of the three prior academic years; (vi) each school's accreditation status, including the accrediting body; and (vii) the number of incidents of restraint and seclusion occurring in each of the previous three academic years."

Page 141, after line 14, insert:

"I. The Board of Education shall develop and promulgate regulations for private special education day schools on restraint and seclusion that establish the same requirements for restraint and seclusion as those for public schools."

Explanation:

(This amendment implements recommendation 8 and 9 from the 2020 JLARC report on the "Review of the Children's Services Act and Private Special Education Day School Costs".)

Item 138 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall develop and maintain a robust statewide plan for

improving (i) its ongoing oversight of local practices related to transition planning and services and (ii) technical assistance and guidance provided for post-secondary transition planning and services. The plan shall articulate how the Department will reliably and comprehensively assess the compliance and quality of transition plans for students with disabilities in Virginia on an ongoing basis and detail how the Department will communicate findings to local school division staff and local school boards. The plan shall be submitted to the Senate Education and Health and the House Education committees no later than December 1, 2022. The Department shall update those committees annually detailing its progress on implementing the plan."

Explanation:

(This amendment implements recommendation 9 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 138 #3s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall develop clear and simplified guidance, in multiple languages, for families of students with disabilities explaining (i) the limitations of the applied studies diploma; (ii) key curriculum and testing decisions that reduce the likelihood of a student being able to obtain a standard diploma; and (iii) that pursuit of an applied studies diploma may preclude a student's ability to pursue a standard diploma."

Explanation:

(This amendment implements recommendation 10 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 138 #4s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall revise the state's special education complaint procedures and practices to ensure the Department requires and enforces corrective actions that (i) achieve full and appropriate remedies for school divisions' non-compliance with special education laws and regulations, including, at a minimum, requiring school divisions to provide compensatory

services to students with disabilities when the Department determines divisions did not provide legally obligated services; and (ii) ensure that relevant personnel understand how to avoid similar non-compliance in the future."

Explanation:

(This amendment implements recommendation 21 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 138 #5s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall (i) conduct a one-time targeted review of the transition sections of a random sample of students' individualized education programs (IEPs) in each school division; (ii) communicate its findings to each local school division, school board, and local special education advisory committee; and (iii) ensure local school divisions correct any IEPs that are found out of compliance. The Department shall submit to the Senate Education and Health and the House Education Committees a letter certifying that school divisions have corrected all instances of non-compliance identified through these reviews, no later than May 1, 2022."

Explanation:

(This amendment implements recommendation 8 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

		Item 138 #6s	
Education	FY20-21	FY21-22	
Department of Education, Central Office Operations	\$0	\$300,000	GF

Language:

Page 139, line 30, strike "\$17,352,182" and insert "\$17,652,182".

Page 141. after line 14, insert:

"I. Out of this appropriation, \$300,000 the second year from the general fund is provided for the Department of Education to: (i) provide training and guidance documents to local school divisions on the development of individualized education programs (IEPs); (ii) develop a required training module for each individual who participates in an IEP meeting that

comprehensively addresses and explains in detail each IEP team member's respective role in the IEP meeting, the IEP development process, and components of effective IEPs; and (iii) conduct structured reviews of a sample of IEPs annually to verify that the IEPs are in compliance with state and federal laws and regulations."

Explanation:

(This amendment provides \$300,000 GF the second year for costs associated with implementation of Senate Bill 1288.)

		Item 139 #1s	
Education	FY20-21	FY21-22	
Department of Education, Central Office Operations	\$0	(\$1,812,000)	GF

Language:

Page 141, line 15, strike "\$39,750,487" and insert "\$37,938,487".

Explanation:

(This amendment captures savings in the amount of \$1.8 million GF from the reduction of Standards of Learning assessments to the minimum federal requirements, contingent on the passage of Senate Bill 1401.)

Item 141 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 143, after line 46, insert:

"F. To support a school division's needs for an analytics solution to evaluate student progress and determine instructional gaps, the Department of Education may provide funds out of this appropriation as one-time grants to divisions to support the costs of such analytics solution."

Explanation:

(This amendment allows the Department of Education to provide one-time grants to divisions to support the costs of an analytics solution to evaluate student progress and determine instructional gaps.)

Item 141 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 143, after line 46, insert:

"F. Virtual Education Supports

To continue the expansion of Virtual Virginia's Outreach Program initiated or expanded with federal funds from the Coronavirus Aid, Relief, and Economic Security (CARES) Act (P.L. 116-136), \$7,000,000 in Governor's Education Emergency Relief (GEER) funds from the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) are provided to support the continued expansion costs in the second year."

Explanation:

(This amendment provides \$7.0 million in Governor's Education Emergency Relief Funds from the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) to support the continued expansion of Virtual Virginia's learning management system and additional free course enrollment slots during the 2021-22 school year.)

Item 142 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 144, after line 28, insert:

"G. 1. The Board of Education shall review and update its regulations of general education K–12 teacher preparation programs to ensure graduates are required to demonstrate proficiency in: (i) differentiating instruction for students depending on their needs; (ii) understanding the role of general education teachers on the IEP team; (iii) implementing effective models of collaborative instruction, including co-teaching; and (iv) understanding the goals and benefits of inclusive education for all students.

2. The Board of Education shall review and update its regulations governing administrator preparation programs to ensure graduates are required to demonstrate comprehension of: (i) key special education laws and regulations; (ii) individualized education program (IEP) development; (iii) the roles and responsibilities of special education teachers; and (iv) appropriate behavior management practices."

Explanation:

(This amendment implements recommendation 15 and 17 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 142 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 144, after line 28, insert:

"G. The Department of Education shall develop and maintain a statewide strategic plan for recruiting and retaining special education teachers. At a minimum, the strategic plan shall (i) use data analyses to determine divisions' specific staffing needs on an ongoing basis; (ii) evaluate the potential effectiveness of strategies for addressing recruitment and retention challenges, including tuition assistance, differentiated pay for special education teachers, and the expansion of special education teacher mentorships; and (iii) estimate the costs of implementing each strategy, including the extent to which federal funds could be used to support implementation. The Department shall submit its plan to the Chairs of the House Education Committee and the Senate Education and Health Committee no later than November 1, 2021, and update those committees annually on its progress implementing the plan."

Explanation:

(This amendment implements recommendation 19 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

		Item 144 #1s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$300,000	GF

Language:

Page 146, line 23, strike "\$43,069,426" and insert "\$43,369,426".

Explanation:

(This amendment provides \$300,000 GF the second year to restore the VPI Minority Fellowship. The introduced budget included the language, but the appropriated amount was not included in the Item.)

Item 144 #2s

Education	FY20-21	FY21-22
Direct Aid to Public Education	\$0	\$250,000 GF

Language:

Page 146, line 22, strike "\$43,069,426" and insert "\$43,319,426".

Page 147, after line 4, insert: "Dual Enrollment Passport Pilots \$0 \$250,000"

Page 157, after line 31, insert:

"UU. Out of this appropriation, \$250,000 the second year from the general fund is provided for grants to support one-time pilot programs to school divisions to redesign dual enrollment course offerings to align/link to the Passport and Uniform Certificate of General Studies offered by Virginia's community colleges. Divisions awarded such grants shall collaborate with the local community college to effectively redesign the local school division's dual enrollment course offerings and increase the number of qualified teachers to teach dual enrollment courses. Divisions applying shall include: (i) an explanation of why such dual enrollment pilot program is warranted; (ii) the dual enrollment courses currently offered by the division; (iii) the projected student enrollment in dual enrollment courses; and (iv) the number of the division's employed staff qualified to teach dual enrollment and the number currently teaching a dual enrollment course. The Department of Education may consider in the awarding of a grant: (i) the division's local composite index; (ii) the level of misalignment in the division's dual enrollment course offerings to the Passport and Uniform Certificate of General Studies; and (iii) the division's level of dual enrollment course availability and current student enrollment in those courses. The Department of Education shall report, along with the divisions and community colleges, the components of the redesign and efforts to increase availability and participation in dual enrollment courses to the General Assembly by November 1, 2022. The Department of Education and the Virginia Community College System shall use these pilot programs to provide a comprehensive guide to every school division and community college to assist with aligning high school dual enrollment course offerings to the Passport and Uniform Certificate of General Studies."

Explanation:

(This amendment provides \$250,000 GF the second year for grants to support one-time pilot programs to redesign a division's dual enrollment course offerings to align/link to the Passport and Uniform Certificate of General Studies offered by Virginia's community colleges.)

		Item 144 #3s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$250,000	GF

Language:

Page 146, line 23, strike "\$43,069,426" and insert "\$43,319,426".

Page 146, after line 33, insert: "Active Learning Grants \$0 \$250,000".

Page 157, after line 31, insert:

"UU. Out of this appropriation, \$250,000 the second year from the general fund shall be provided for grants to school divisions for encouraging active in-class, remote and hybrid learning for students in pre-kindergarten through the second grade. School divisions seeking to apply for this grant shall submit a proposal to the Department of Education outlining the intended use of funds and a projected number of students to be served. The Department shall establish criteria for awarding these funds. The funds may be used to purchase a platform featuring on-demand activities that integrate math and English Standards of learning content into movement-rich activities that can be used at school, home and on all devices (i.e., computers, tablets and phones)."

Explanation:

(This amendment provides \$250,000 GF the second year to encourage active learning for students in pre-K through 2nd grade. The Department of Education will establish criteria by which to award these funds to school divisions. Local school divisions will use the grant funds to provide active learning curriculum that relate math and language Standard of Learning content into movement rich activities for students in pre-K through 2nd grade.)

		Item 145 #1s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	(\$15,897,461)	\$0	GF

Language:

Page 157, line 33, strike "\$7,722,644,551" and insert "\$7,706,747,090".

Page 159, line 12 strike "\$88,303,142" and insert "\$72,405,681".

Page 181, line 27 strike "\$71,703,142" and insert "\$55,805,681".

Explanation:

(This amendment captures \$15.9 million GF the first year from the undistributed balance in the Virginia Preschool Initiative program after updating final participation. The introduced budget retained \$21.0 million GF the first year in the undistributed balance to ensure that there were sufficient funds available to school divisions that exercised the enrollment flexibility contained in *Superintendent's Memo #282-20*.)

 Item 145 #2s

 Education
 FY20-21
 FY21-22

Direct Aid to Public Education

\$145,636

\$0 GF

Language:

Page 157, line 33, strike "\$7,722,644,551" and insert "\$7,722,790,187". Page 158, line 45, strike "\$18,993,450" and insert "\$19,139,086".

Explanation:

(This amendment updates enrollment for Governor's School in FY 2021.)

Item 145 #3s

Education FY20-21 FY21-22

Direct Aid to Public Education (\$2,500,000) \$0 GF

Language:

Page 157, line 33, strike "\$7,722,644,551" and insert "\$7,720,144,551".

Page 159, line 10, strike "\$2,500,000" and insert "\$0".

Page 196, line 28, strike "\$2,500,000 the first year and".

Explanation:

(This amendment captures savings of \$2.5 million GF the first year for the School Meals Expansion program. Under a federal waiver for the 2020-21 school year, all divisions are providing school meals at no cost to all students. These funds are no longer needed to eliminate the cost of reduced price school meals.)

Item 145 #4s

Education FY20-21 FY21-22
Direct Aid to Public Education \$0 (\$2,784,143) GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,855,855,978".

Page 170, line 9, after "Eight", strike "and the counties of Accomack and Northampton".

Page 170, line 17, after "Eight", strike "and the counties of Accomack and Northampton".

Page 170, line 21, after "Eight", strike "and the counties of Accomack and Northampton".

Explanation:

(This amendment removes Accomack and Northampton counties from the full cost of competing adjustment as included in the introduced budget.)

Item 145 #5s

Education	FY20-21	FY21-22
Direct Aid to Public Education	\$0	\$582,000 GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,859,222,121".

Page 159, after line 23, insert:

Explanation:

2022."

(This amendment provides \$582,000 GF the second year as an incentive for the Alleghany County and Covington City school divisions to consolidate. Such payments are intended to be set at this amount for five years, with no adjustments. The Board of Education approved the consolidation of these divisions to be effective July 1, 2022.)

		Item 145 #6s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$59,303,635	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,917,943,756".

Page 158, strike line 43 and insert: "Compensation Supplement \$0 \$139,160,538".

Page 175, strike lines 8 through 12, and insert:

"6) Out of the appropriation included in paragraph C.44. of this Item, \$455,787 the second year from the Lottery Proceeds Fund are provided for a compensation supplement payment equal to 3.0 percent of base pay on August 1, 2021, for Regional Alternative Education Program instructional and support positions, as referenced in paragraph C.44. of this Item."

Page 191, strike lines 26 through 30, and insert:

"j. Out of the appropriation included in paragraph C.44. of this Item, \$620,830 the second year from the general fund is provided in the Academic Year Governor's School funding allocation

[&]quot;Alleghany County - Covington City School Division Consolidation Incentive \$0 \$582,000" Page 198, after line 24, insert:

[&]quot;45. Alleghany County - Covington City School Division Consolidation Incentive Out of this appropriation, \$582,000 the second year from the general fund is provided as an incentive for the consolidation of the Alleghany County and Covington City school divisions. Such payments shall be provided for no more than five fiscal years, beginning in fiscal year

to increase the per pupil amount the second year as an add-on for a compensation supplement payment equal to 3.0 percent of base pay on August 1, 2021, for Academic Year Governor's School instructional and support positions."

Page 197, strike lines 33 through 52, and insert: "44. Compensation Supplement

"a. Out of this appropriation, \$139,781,368 the second year from the general fund and \$455,787 the second year from the Lottery Proceeds Fund is provided for the state share of a payment of the following salary increases for funded SOQ instructional and support positions. Funded SOQ instructional positions shall include the teacher, school counselor, librarian, instructional aide, principal, and assistant principal positions funded through the SOQ staffing standards for each school division in the biennium. This amount includes \$620,830 the second year from the general fund referenced in paragraph C. 27. j. for the Academic Year Governor's Schools for the state share of a payment of the following salary increases for instructional and support positions, and this amount includes \$455,787 the second year from the Lottery Proceeds Fund referenced in paragraph C. 9. f. 6) for Regional Alternative Education Programs for the state share of a payment equivalent to a 3.0 percent salary increase effective August 1, 2021, for SOQ instructional and support positions.

b. It is the intent that the instructional and support position salaries are increased in school divisions throughout the state by at least an average of 3.0 percent during the second year. Sufficient funds are appropriated in this act to finance, on a statewide basis, the state share of a 3.0 percent salary increase the second year for funded SOQ instructional and support positions, effective August 1, 2021, to school divisions that certify to the Department of Education that salary increases of a minimum average of 3.0 percent have been or will have been provided during the 2020-2022 biennium, either in the first year or in the second year or through a combination of the two years, to instructional and support personnel. The state funds for which the division is eligible to receive shall be matched by the local government, based on the composite index of local ability-to-pay, which shall be calculated using an effective date of August 1, 2021, as the basis for the local match requirement for both funded SOQ instructional and support positions.

c. This funding is not intended as a mandate to increase salaries."

Page 198, strike line 1.

Explanation:

(This amendment provides \$139.8 million the second year from the general fund and \$455,787 the second year from the Lottery Proceeds fund to provide a 3.0 percent salary increase for SOQ-recognized instructional and support positions, to become effective August 1, 2021. This represents an addition of \$59.3 million the second year to provide a 3.0 percent salary increase in lieu of the 2.0 percent bonus that was proposed in the budget as introduced.)

		Item 145 #7s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$29,997,718	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,888,637,839".

Page 158, line 47, strike "\$133,057,800" and insert "\$163,055,518".

Page 159, line 41, strike "\$97,331,981" and insert "\$67,334,263".

Page 160, after line 16, insert: "Learning Loss Instructional Supports \$0 \$29,997,717".

Page 198, after line 4, insert:

"45. Learning Loss Instructional Supports

An additional state payment estimated at \$29,997,717 the second year from the Lottery Proceeds Fund shall be disbursed to support the state share of \$117.18 per pupil the second year based on the estimated number of federal Free Lunch participants, in support of one-time programs and initiatives to address learning loss resulting from the COVID-19 pandemic. These funds shall cover expenses for: extended school year programs; summer learning programs; supplemental afterschool programs; remedial instructional supports, including activities to differentiate instruction to meet the needs of students; targeted supports and services for students with disabilities, students in grades pre-kindergarten through third, at-risk students, and English language learners; and modifications to facilities to assist with COVID-19 mitigation strategies for in-person learning. No local match is required to receive these state funds."

Explanation:

(This amendment provides \$30.0 million from the Lottery Proceeds Fund the second year to support one-time programs and initiatives to address learning loss experienced by students due to the COVID-19 pandemic.)

		Item 145 #8s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$49,230,162	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,907,870,283".

Page 161, line 7, after "pupils)", strike "school nurses,".

Page 161, strike lines 11 through 15.

Explanation:

(This amendment funds the state's share of three specialized student support positions per 1,000 students. Specialized student support positions, consistent with Senate Bill 1257, includes school social workers, school psychologists, school nurses, licensed behavior analysts, licensed assistant behavior analysts, and other licensed health and behavioral positions.)

		Item 145 #9s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$2,699,993	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,861,340,114".

Page 158, line 47, strike "\$133,057,800" and insert "\$135,757,793".

Page 159, line 41, strike "\$97,331,981" and insert "\$94,631,998".

Page 160, after line 16, insert: "Supplemental Support Accomack & Northampton \$0 \$2,700,000"

Page 198, after line 4, insert:

"45. Supplemental Support Accomack & Northampton

An additional state payment of \$2,700,000 the second year from the Lottery Proceeds Fund shall be disbursed to provide one-time support to Accomack and Northampton school divisions for teacher recruitment and retention efforts."

Explanation:

(This amendment provides \$2.7 million from the Lottery Proceeds Fund the second year to provide one-time support to Accomack and Northampton school divisions for recruitment and retention efforts.)

Item 145 #10s

Education

Direct Aid to Public Education

Language

Language:

Page 188, line 6, strike "c." and insert "c.1)". Page 188, after line 20, insert:

"2) The Board of Education shall make recommendations for: (i) appropriate staffing and funding levels necessary for State Operated Programs (SOP) in regional and local detention centers to provide a quality education program; (ii) implementation of appropriate efficiencies in staffing practices in such programs; (iii) statutory and regulatory changes needed to

implement the Board's findings; and (iv) appropriate programs to redirect any potential savings realized from implementation of the Board's findings.

In developing such recommendations, the Board shall consider: (i) the dramatic decrease in the Average Daily Population in detention centers over the course of two decades without a comparable decrease in state funding; (ii) establishing a system-wide staffing ratio that is comparable to those provided in Regional Alternative Education Programs and aligned with the staffing requirements provided in the federal Prison Rape Elimination Act; (iii) implementing efficiencies, such as sharing SOP instructional staff with participating school divisions, hiring part-time teachers and dually-certified teachers and principals, and utilizing lead teachers in lieu of full-time principals in programs with a low average daily population; (iv) changes to SOP operating agreements to facilitate more efficient staffing practices and to clarify the role of the state and school divisions in hiring and supervising SOP instructional staff; (v) increasing the use of enhanced distance learning; and (vi) the draft recommendations deliberated by the Commission on Youth from the 2020 study.

The Board shall convene a workgroup to assist in the development of findings and recommendations and shall include staff members from the Senate Finance and Appropriations Committee, House Appropriations Committee, Department of Planning and Budget, the Virginia Department of Education, the Department of Juvenile Justice, President of the Virginia Juvenile Detention Association or his/her designee, the chair of the Virginia Commission on Youth or his/her designee, and anyone else the Board deems as appropriate to serve on the workgroup. Findings and recommendations shall be reported to the House Appropriations Committee and the Senate Finance and Appropriations Committee by November 1, 2021."

Explanation:

(This amendment directs the Board of Education to develop recommendations for funding state operated programs in local and regional juvenile detention centers, which have experienced a significant decline in population in recent decades without an accompanying adjustment to state funding levels. This recommendation is from the Commission on Youth's 2020 study on "State Operated Programs Education in Juvenile Detention Centers".)

Item 145 #11s

Education

Direct Aid to Public Education

Language

Language:

Page 169, after line 27, insert:

"31. Each school division shall offer in-person instruction options to a student if requested by the parent or guardian in the 2021-2022 school year."

Explanation:

(This amendment requires all school divisions to offer in-person instruction options in the 2021-2022 school year to a student if the parent or guardian requests.)

		Item 150 #1s	
Education	FY20-21	FY21-22	
State Council of Higher Education for Virginia	\$150,000	\$300,000	GF

Language:

Page 200, line 25, strike "\$92,448,559" and insert "\$92,598,559".

Page 200, line 25, strike "\$105,568,934" and insert "\$105,868,934".

Page 202, line 26, strike the first instance of "\$1,980,000" and insert "\$2,130,000".

Page 202, line 26, strike the second instance of "\$1,980,000" and insert "\$2,280,000".

Explanation:

(This amendment provides additional funding based on estimated additional participation by qualifying veterans or their spouses or children for the Virginia Military Survivors and Dependent Education Program (VMSDEP), which provides assistance to Virginia military service members killed, missing in action, taken prisoner, or rated at least 90 percent permanently disabled as direct result of military service, by waiving tuition and required fees and providing a stipend to their dependents.)

		Item 152 #1s	
Education	FY20-21	FY21-22	
State Council of Higher Education for Virginia	\$0	\$250,000	GF

Language:

Page 205, line 6, strike "\$18,335,818" and insert "\$18,585,818".

Page 207, after line 55, insert:

"Q. Out of this appropriation, \$250,000 the second year from the general fund is designated for the Guidance to Postsecondary Success program. The program coordinates statewide efforts to increase college access and student success."

Explanation:

(This amendment restores \$250,000 GF the second year provided during the 2020 regular session for SCHEV to implement the Guidance to Postsecondary Success (GPS) program.

These funds were later unalloted. The purpose of this new initiative is to increase student transitions from high school to postsecondary education and also help students find the right fit for their future and in terms of costs. Each year, 25,000 high school graduates do not enroll in postsecondary education within 16 months, including significantly higher percentages for African-American, Hispanic and economically disadvantaged students. The pandemic and virtual learning environments in high schools are expected to exacerbate these inequitable outcomes in the coming year. Working closely with the Virginia Department of Education, SCHEV will seek to increase these enrollment rates through the development of materials and informational resources and the coordination of awareness events, professional development opportunities for secondary staff and partnerships with institutions, schools, school divisions and non-profit organizations. In addition, these funds will be used as a match for an upcoming SCHEV application for a federally-funded Gaining Early Awareness and Readiness for Undergraduate Programs (GEAR UP) grant that also will focus on school division-specific and statewide outreach programs, as well as, required scholarships for eligible students.)

		Item 152 #2s	
Education	FY20-21	FY21-22	
State Council of Higher Education for Virginia	\$0 0.00	\$300,000 1.00	GF FTE

Language:

Page 205, line 6, strike "\$18,335,818" and insert "\$18,635,818".

Page 208, after line 36, insert:

- "U. 1. Out of this appropriation, \$300,000 the second year from the general fund is designated to support related costs of undertaking a review of higher education costs, funding needs, appropriations and efficiencies.
- 2. The State Council of Higher Education, in consultation with representatives from House Appropriations Committee, Senate Finance and Appropriations Committee, Department of Planning and Budget, Secretary of Finance, and Secretary of Education, as well as representatives of public higher education institutions, shall review methodologies to determine higher education costs, funding needs, and appropriations in Virginia. The review shall identify and recommend: (1) methods to determine appropriate costs; (2) measures of efficiency and effectiveness; (3) provisions for any new reporting requirements; (4) strategies to allocate limited public resources based on outcomes that align with state needs related to affordability, access, completion, and workforce alignment, including with regard to nonresident pricing; (5) the impact of funding on underrepresented student populations; and (6) a timeline for implementation.

- 3. The review shall build on existing efforts including the assessment of base adequacy, recommendations provided through the Strategic Finance Plan, and peer institution comparisons to determine if existing funding models should be updated or replaced. It shall also build on promising practices and include input from Virginia's institutions, policy makers, and other education experts.
- 4. The Council shall present a proposed workplan to the Joint Subcommittee on the Future Competitiveness of Higher Education in Virginia by August 15, 2021. The Council shall submit a preliminary report and any related recommendations to the Governor and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by December 1, 2021 with a final report by July 1, 2022."

Explanation:

(This amendment restores funding for a review to benchmark costs for higher education using historic trends and comparisons to other state, institution, and national data and to identify methods to allocate additional funds to higher education when available.)

		Item 152 #3s	
Education	FY20-21	FY21-22	
State Council of Higher Education for Virginia	\$0	\$200,000	GF

Language:

Page 205, line 5, strike "\$18,335,818" and insert "\$18,535,818".

Explanation:

(This amendment provides \$200,000 GF for SCHEV to develop and maintain an application portal related to implementation of Senate Bill 1387, which provides that students who meet the criteria in the Code of Virginia that determine eligibility for in-state tuition regardless of their citizenship or immigration status shall be afforded the same educational benefits, including access to state financial assistance programs, as any other individual who is eligible for in-state tuition. The bill has a delayed effective date of August 1, 2022, but directs the State Council of Higher Education for Virginia, in coordination with institutions of higher education in the Commonwealth, to promulgate regulations to implement the provisions of the bill and develop the application portal in the interim. The portal will require additional privacy and security to collect financial aid data that SCHEV and institutions do not currently collect.)

Education

State Council of Higher Education for Virginia

Language

Language

Page 207, after line 21, insert:

"In addition, the office of the workforce development advisor shall also have access to wage records collected by the Council."

Explanation:

(This amendment allows the office of the workforce development advisor access to wage records under SCHEV's existing authority, as a result of language adopted in 2019 specifying certain needed data exchange with the U.S. Census in order to get wage outcomes for graduates working outside the Commonwealth.)

		Item 156 #1s	
Education	FY20-21	FY21-22	
Christopher Newport University	\$0 \$0 0.00	\$1,019,000 \$651,500 19.00	GF NGF FTE

Language:

Page 209, line 30, strike "\$81,019,468" and insert "\$82,689,968".

Explanation:

(This amendment provides Christopher Newport University with on-going support for operations and maintenance (O&M) costs related to the opening of the new Fine Arts Center. This request is also a part of the university's six year plan. Construction will be completed in spring 2021, the building will open to the public in summer 2021 and classes will be held starting in the fall 2021 semester.)

		Item 164 #1s	
Education	FY20-21	FY21-22	
Richard Bland College	\$0	(\$299,000)	GF

Language:

Page 214, line 22, strike "\$15,180,047" and insert "\$14,881,047".

Explanation:

(This amendment removes proposed new general funds that had been requested to cover debt payments from a previous renovation of classroom space to dorms, in order that Richard Bland College could revert the space back to classroom use.)

		Item 168 #1s	
Education	FY20-21	FY21-22	
Virginia Institute of Marine Science	\$0	\$225,000	GF
	0.00	2.20	FTE

Language:

Page 216, line 38, strike "\$27,010,448" and insert "\$27,235,448".

Explanation:

(This amendment provides funding for science-based guidance on the management of existing and emerging disease threats to critical fishery and aquaculture resources in the Commonwealth and Chesapeake Bay region. This initiative was funded in the 2020 General Assembly Session but subsequently unallotted. Recent outbreaks of disease have damaged economically important and ecologically sensitive marine resources in the Commonwealth and nation. The pathogens responsible for these outbreaks are not well known, their risks to marine life and potential to spread remain poorly understood, and their ecological impacts have been difficult to assess with existing resources. To meet the challenge of diseases in marine systems, this initiative will provide science-based guidance on the management of existing and emerging disease threats to critical fishery and aquaculture resources in the Commonwealth and Chesapeake Bay region.)

		Item 168 #2s	
Education	FY20-21	FY21-22	
Virginia Institute of Marine Science	\$0 0.00	\$250,000 2.70	GF FTE

Language:

Page 216, line 38, strike "\$27,010,448" and insert "\$27,260,448".

Explanation:

(This amendment provides funding to provide scientific data for the management of saltwater fisheries in the Commonwealth. This initiative was funded in the 2020 General Assembly Session but subsequently unalloted. Long-standing VIMS fisheries surveys have provided scientific data for the management of saltwater fisheries in the Commonwealth of Virginia. Recent reductions in state and federal funds resulted in reductions in five of the most crucial surveys.)

		Item 183 #1s	
Education	FY20-21	FY21-22	
Norfolk State University	\$0	\$1,250,000	GF

Language:

Page 227, line 13, strike "\$100,293,110" and insert "\$101,543,110".

Page 228, after line 34, insert:

"H. Out of this appropriation, \$1,250,000 in the second year from the general fund is designated for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation."

Explanation:

(This amendment provides \$1.25 million GF beginning in the second year for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation.)

		Item 184 #1s	
Education	FY20-21	FY21-22	
Norfolk State University	(\$1,729,795)	\$0	GF

Language:

Page 228, line 37, strike "\$21,647,706" and insert "\$19,917,911".

Page 229, line 3, strike "\$3,459,590" and insert "\$1,729,795".

Page 229, line 23, after "program." strike the remainder of the line.

Page 229, strike lines 24 and 25.

Explanation:

(This amendment adjusts first year funding for the new VCAN affordability initiative approved during the 2020 Regular Session. The funding was unalloted at the Reconvened Session, then restored, for FY 2021, during the Special Session. The required detailed budget and implementation plan was submitted in December 2020 and to be impactful beginning spring semester 2021. Funding remains for the spring semester and for the 2021-22 academic year. The amendment provides for the release of first year funding by removing existing language that required certain approvals prior to release of the funding.)

Education	FY20-21	FY21-22
Old Dominion University	\$0	\$1,250,000 GF

Language:

Page 230, line 6, strike "\$326,951,395" and insert "\$328,201,395".

Page 231, after line 47, insert:

"K. Out of this appropriation, \$1,250,000 in the second year from the general fund is designated for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation."

Explanation:

(This amendment provides \$1.25 million GF beginning in the second year from for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation.)

		Item 191 #1s	
Education	FY20-21	FY21-22	
Radford University	\$0	\$2,000,000	GF

Language:

Page 233, line 41, strike "\$135,081,721" and insert "\$137,081,721".

Explanation:

(This amendment provides \$2.0 million GF in additional funding for Radford University for operating support to further reduce tuition and fees for Virginians attending Radford University Carilion (RUC) in Roanoke to close the differential gap in tuition between the two campus sites, along with additional funding provided in FY 2021 and FY 2022 in another Item. The 2019 General Assembly authorized Radford University to establish RUC, a partnership with Carilion Clinic to strengthen the Roanoke region as a destination for health education and research, contributing to the overall growth and economic development of Southwest Virginia. Increased investment by the Commonwealth is intended to provide additional accessibility and affordability for Virginians enrolled in high-demand, high-paying health professions programs, and increase the number of health care professionals living and working in Southwest Virginia.)

		Item 196 #1s	
Education	FY20-21	FY21-22	
University of Mary Washington	\$0	\$568,000	GF

Language:

Page 236, line 10, strike "\$80,597,650" and insert "\$81,165,650".

Page 237, after line 15, insert:

"Out of this appropriation, \$568,000 the second year from the general fund is designated to support an educational partnership between regional K12; community college; University of Mary Washington and industry to develop a curriculum that accelerates time to degree; lowers cost; eliminates the skills gap and reduces reliance on student debt in the areas of Education, Healthcare, and Cybersecurity."

Explanation:

(This amendment provides \$568,000 GF for the University of Mary Washington's Workforce Development Initiative. This initiative partners three K-12 school districts, a community college, a public liberal arts university, and industry to develop a curriculum that accelerates time to degree, lowers cost of attendance, eliminates the skills gap that exists between educational experiences and works, reduces the reliance on student load debt to finance both degree competencies required to compete in the 21st century, and produces career-ready graduates whose experiences will empower them both socially and economically into their futures.)

		Item 205 #1s	
Education	FY20-21	FY21-22	
University of Virginia	\$0	(\$1,000,000)	GF

Language:

Page 241, line 48, strike "\$578,028,122" and insert "\$577,028,122". Page 242, line 14, strike "\$4,112,500" and insert "\$3,112,500".

Explanation:

(This amendment reallocates proposed additional funding for the Center for Focused Ultrasound to restorations of items from the 2020 Regular Session.)

		Item 214 #1s	
Education	FY20-21	FY21-22	
Virginia Commonwealth University	\$192,793	\$0	GF

Language:

Page 246, line 8, strike "\$661,732,918" and insert "\$661,925,711".

Page 248, after line 48, insert:

"Q. Out of this appropriation, \$192,753 the first year from the general fund shall be provided to

the L. Douglas Wilder School of Government and Public Affairs at Virginia Commonwealth University to support the Research Institute for Social Equity (RISE) addressing issues of racism and racial equity in public policy."

Explanation:

(This amendment provides \$192,753 GF the first year to the L. Douglas Wilder School of Government and Public Affairs at Virginia Commonwealth University to support the Research Institute for Social Equity (RISE) addressing issues of racism and racial equity in public policy. The L. Douglas Wilder School is a recognized national leader in advancing racial justice through multidisciplinary scholarship, research, training, and engagement. The creation of the Research Institute for Social Equity (RISE) within the Wilder School will serve as a vehicle to advance racial equity and social justice through multidisciplinary scholarship, research, training, and engagement. The goal of RISE is to be the national leader in advancing social justice to inform public policy, governance and practice to improve conditions for marginalized voices within our society, including communities of color, LGBTQIA individuals, people with disabilities, incarcerated populations, survivors of domestic violence, and women.)

		Item 214 #2s	
Education	FY20-21	FY21-22	
Virginia Commonwealth University	\$0	\$150,000	GF

Language:

Page 246, line 8, strike "\$661,732,918" and insert "\$661,882,918". Page 246, line 50, strike "\$386,685" and insert "\$536,685".

Page 247, after line 9, insert:

"All other funding support for the center shall be maintained by the university at least at the level provided in fiscal year 2019."

Explanation:

(This amendment restores funding in FY 2022 to the Virginia Center on Aging that was removed in the 2020 Special Session, and provides clarifying language.)

		Item 216 #1s	
Education	FY20-21	FY21-22	
Virginia Commonwealth University	(\$5,000,000)	(\$2,500,000)	GF

Language:

Page 249, line 26, strike "\$331,699,678" and insert "\$326,699,678".

Page 249, line 26, strike "\$331,699,678" and insert "\$329,199,678".

Page 249, line 39, strike "\$17,500,000" and insert "\$12,500,000."

Page 249, line 40, strike "\$17,500,000" and insert "\$15,000,000."

Page 249, line 41, after "support of cancer research" insert ", including pediatric cancer treatment".

Explanation:

(This amendment provides an additional \$2.5 million for Massey Cancer Center in FY 2022, consistent with the conference report during the 2020 Regular Session. The FY 2021 amount is level funded.)

		Item 220 #1s	
Education	FY20-21	FY21-22	
Virginia Community College System	\$0	\$385,177	GF

Language:

Page 251, line 14, strike "\$939,748,443" and insert "\$940,133,620".

Page 254, after line 1, insert:

"X. Out of this appropriation, \$385,177 the second year from the general fund is designated for costs of two associate degree programs in Physical Therapy Assistant and Surgical Technology that have transferred to Virginia Western Community College as a result of the merger of Radford University and the Jefferson College of Health Sciences authorized in Chapter 60 of the 2019 Acts of Assembly."

Explanation:

(This amendment provides funding for two programs that have transferred to Virginia Western Community College as a result of the merger of Radford University and the Jefferson College of Health Sciences authorized in Chapter 60 of the 2019 Acts of Assembly.)

		Item 220 #2s	
Education	FY20-21	FY21-22	
Virginia Community College System	\$0 0.00	\$5,000,000 60.00	GF FTE

Language:

Page 251, line 14, strike "\$939,748,443" and insert "\$944,748,443".

Page 254, line 9, strike "\$1,500,000" and insert "\$6,500,000".

Page 254, line 10, after "designated for" insert "advising,".

Explanation:

(This amendment provides \$5.0 million GF for Virginia community colleges to hire 60 additional advisors to assist students in identifying appropriate programs, resources and support, related to the G3 initiative.)

		Item 226 #1s	
Education	FY20-21	FY21-22	
Virginia Military Institute	\$0	\$103,048	GF
	0.00	1.00	FTE

Language:

Page 263, line 2, strike "\$44,354,698" and insert "\$44,457,746".

Explanation:

(This amendment restores funds for additional faculty positions to implement a redesigned required three-hour course in theory and practice of leadership, Leadership in Organizations. All cadets enroll in Leadership in Organizations, a required, 3-credit hour course in the theory and practice of leadership. This course will be redesigned to lessen the time commitment to leadership theory in order to enrich the application component of the course. In addition to curricular enhancements, this course will have a new course director who will oversee the development of: new training modules for current faculty and the training of Leadership Fellows who will teach the course.)

		Item 226 #2s	
Education	FY20-21	FY21-22	
Virginia Military Institute	\$0	\$126,000	GF

Language:

Page 263, line 2, strike "\$44,354,698" and insert "\$44,480,698".

Explanation:

(This amendment restores funds for the Math Education and Resource Center (MERC) and the Miller Academic Center (MAC) both of which were originally funded through private funds. The MERC was created to specifically address cadets' performance in mathematics with a particular emphasis on incoming freshmen and those in STEM majors. The MAC facilitates cadets' academic success and timely progress toward a degree by providing academic tutors and other academic support services.)

Item 236 #1s

Education	FY20-21	FY21-22	
Virginia Cooperative Extension and Agricultural Experiment Station	\$0	\$1,600,000	GF

Language:

Page 269, line 15, strike "\$93,864,832" and insert "\$95,464,832".

Explanation:

(This amendment supports the enhancement of the Virginia Cooperative Extension and Agricultural Experiment Station in order to support the resiliency and growth of the agricultural and forestry economy in the Commonwealth. Specific needs addressed in this multi-year request include improving internet connectivity, including one-time and on-going costs. Additional needs include modernizing research equipment, adding critical personnel, and supporting the market competitiveness of extension agent salaries.)

		Item 238 #1s	
Education	FY20-21	FY21-22	
Virginia State University	(\$1,886,745)	\$0	GF

Language:

Page 272, line 7, strike "\$19,677,956" and insert "\$17,791,211". Page 272, line 22, strike "\$3,773,490" and insert "\$1,886,745". Page 272, strike lines 42 and 43.

Explanation:

(This amendment adjusts first year funding for the new VCAN affordability initiative approved during the 2020 Regular Session. The funding was unalloted at the Reconvened Session, then restored, for FY 2021, during the Special Session. The required detailed budget and implementation plan included a timeline beginning with January 2021 for creating a pipeline for fall 2021. Funding remains for the spring semester and for the 2021-22 academic year. The amendment provides for the release of funding by removing existing language that required certain approvals prior to release of the funding.)

		Item 244 #1s	
Education	FY20-21	FY21-22	
Jamestown-Yorktown Foundation	\$0	\$412,484	GF

Language:

Page 275, line 8, strike "\$19,920,791" and insert "\$20,333,275".

Explanation:

(This amendment provides \$412,484 GF the second year to enhance digital marketing and social media capabilities to support key marketing strategies, and the development, promotion and maintenance of a new, re-focused website. Marketing funds support non-general fund revenue generation that has historically provided 50 percent of the agency's operating budget.)

		Item 244 #2s	
Education	FY20-21	FY21-22	
Jamestown-Yorktown Foundation	\$0	\$471,820	GF

Language:

Page 275, line 8, strike "\$19,920,791" and insert "\$20,392,611".

Explanation:

(This amendment provides \$471,820 GF the second year to support frontline personnel engaged in direct service delivery of museum and educational programming. These funds provide a competitive base salary and hourly rate for these employees relative to the local market and addresses retention challenges.)

		Item 247 #1s	
Education	FY20-21	FY21-22	
The Library Of Virginia	\$0	\$1,000,000	GF

Language:

Page 276, line 31, strike "\$17,233,584" and insert "\$18,233,584".

Explanation:

(This amendment restores \$1.0 million GF the second year to support additional statewide state aid to local public libraries that was approved at the 2020 Regular Session and then unallotted (Chapter 1289). State aid to local public libraries is currently funded in the second year at only 62 percent of the amount required. This amendment represents an additional step towards fully funding the state library aid formula over time.)

Education	FY20-21	FY21-22	
Virginia Museum of Fine Arts	\$0	(\$9,750,000)	GF

Language:

Page 279, line 23, strike "\$54,497,207" and insert "\$44,747,207".

Page 280, after line 4, insert:

"F. Out of this appropriation, \$1,000,000 in the second year from the general fund is provided to support the development of a plan for transforming Monument Avenue. The museum shall work with community stakeholders to develop the plan and utilize the recommendations from the *Report of the Monuments Work Group (2016)* on the best practices to foster constructive dialogues. The plan shall be reported to the Governor, Secretary of Education, and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2022."

Explanation:

(This amendment provides \$1.0 million GF the second year to support the museum's development of a plan for transforming Monument Avenue.)

		Item 256 #1s	
Education	FY20-21	FY21-22	
New College Institute	\$0	\$95,000	GF

Language:

Page 281, line 20, strike "\$4,318,700" and insert "\$4,413,700".

Explanation:

(This amendment restores funding in the second year for staffing at the New College Institute.)

		Item 257 #1s	
Education	FY20-21	FY21-22	
Institute for Advanced Learning and Research	\$0	\$95,000	GF

Language:

Page 282, line 14, strike "\$6,415,193" and insert "\$6,510,193".

Explanation:

(This amendment restores funding in the second year for staffing at the Institute for Advanced

Learning and Research.)

		Item 258 #1s	
Education	FY20-21	FY21-22	
Roanoke Higher Education Authority	\$0	\$98,817	GF

Language:

Page 282, line 43, strike "\$1,478,720" and insert "\$1,577,537".

Explanation:

(This amendment restores funding in the second year for the Roanoke Higher Education Center for one-time funding of \$50,873 for equipment and installation of blue light telephones, and a security camera system. Additionally, this budget amendment provides on-going funding of \$47,944 for additional hours of coverage from security officers.)

.		Item 258 #2s	
Education	FY20-21	FY21-22	
Roanoke Higher Education Authority	\$0	\$213,254	GF

Language:

Page 282, line 43, strike "\$1,478,720" and insert "\$1,691,974".

Explanation:

(This amendment restores funding in the second year for the Roanoke Higher Education Center for one-time funding of \$66,898 and on-going funding of \$146,356 for the development and maintenance of a student success center.)

		Item 259 #1s	
Education	FY20-21	FY21-22	
Southern Virginia Higher Education Center	\$0	\$388,972	GF

Language:

Page 283, line 8, strike "\$7,949,697" and insert "\$8,338,669".

Explanation:

(This amendment restores funding in the second year for one-time funding of equipment in the

amount of \$293,972 and on-going staffing funding of \$95,000 for personnel and technical training equipment to support high-demand workforce training programs. Equipment includes Zoom classroom equipment and software to allow students access to quality distance and virtual training courses and upgrades to Amatrol training equipment, including new electrical wiring learning system.)

		Item 260 #1s	
Education	FY20-21	FY21-22	
Southwest Virginia Higher Education Center	\$0	\$95,000	GF

Language:

Page 284, line 16, strike "\$3,386,650" and insert "\$3,481,650".

Explanation:

(This amendment restores funding for staffing at the Southwest Virginia Higher Education Center.)

Education	FY20-21	FY21-22	
Southwest Virginia Higher Education Center	\$0	\$1,000,000	GF

Language:

Page 284, line 16, strike "\$3,386,650" and insert "\$4,386,650".

Page 284, after line 32, insert:

"B. Out of the appropriation for this item, \$1,000,000 the second year from the general fund shall be deposited to the Virginia Rural Information Technology Apprenticeship Grant Fund, as established in § 23.1-3129.1 Code of Virginia, for the purpose of awarding grants on a competitive basis from the Fund to small, rural information technology businesses in qualifying localities to establish apprenticeship programs."

Explanation:

(This amendment restores funding for the Southwest Virginia Higher Education Center to develop and implement the Rural Information Technology Grant Apprenticeship Program. The Center will need to develop guidelines, criteria, an application process, and accountability reporting.)

Item 262.80 #1s

Education	FY20-21	FY21-22
Maintain Affordable Access	\$0	\$73,500,000 GF

Language:

Page 287, line 16, strike "\$0" and insert "\$73,500,000". Page 287, strike lines 16 through 43 and insert:

"Item 262.80

Educational and General Programs (10000)	\$60,000,000	\$73,500,000
Higher Education Instruction (10001)	\$60,000,000	\$73,500,000

Fund Sources:

General

Authority: Discretionary Inclusion

A. Out of this appropriation, \$60,000,000 the first year and \$73,500,000 the second year from the general fund is designated to maintain affordable access to public colleges and universities. Institutions may use these funds for operational support, to enhance financial aid, or to address the impacts of the COVID-19 pandemic. For purposes of base budget development, these appropriated funds shall be transferred to the individual institution. Allocations from this item are as follows:

Institution	FY 2021 Allocation	FY 2022 Allocation
Christopher Newport University	\$2,400,000	\$2,400,000
College of William and Mary	3,500,000	3,500,000
George Mason University	0	9,000,000
James Madison University	5,700,000	5,700,000
Longwood University	1,500,000	1,500,000
University of Mary Washington	3,300,000	3,300,000
Norfolk State University	2,000,000	2,000,000
Old Dominion University	0	4,500,000
Radford University	4,900,000	4,900,000
University of Virginia	3,000,000	3,000,000
University of Virginia's College at Wise	1,000,000	1,000,000

Virginia Commonwealth University	10,000,000	10,000,000
Virginia Military Institute	1,000,000	1,000,000
Virginia Polytechnic Institute & State University	4,000,000	4,000,000
Virginia State University	1,700,000	1,700,000
Richard Bland College	1,000,000	1,000,000
Virginia Community College System	15,000,000	15,000,000
Total	\$60,000,000	\$73,500,000"

- B. Pursuant to Item 479.10 of this Act, \$17,250,000 from the Coronavirus Relief Fund is provided for the costs of conducting COVID-19 tests at Virginia's public colleges and universities.
- C. To address student affordability, \$22,000,000 in Governor's Education Emergency Relief funds from the Coronavirus Response and Relief Supplemental Appropriations Act, (P.L. 116-260) shall be allocated to public institutions of higher education for need-based undergraduate financial aid in the second year.
- D. To provide additional operational relief to institutions of higher education, the following reporting and procurement policies shall be modified accordingly:
- 1. Pursuant to § 4-2.01.b.11 of this act, for future reporting on fiscal year 2023 and beyond required reporting on intercollegiate athletic revenues and expenses, specifically related to the share of athletic revenues from school funds and student fees, as set out in § 23.1-1309, Code of Virginia, fiscal years 2020, 2021, and 2022 shall be excluded from the calculated five-year rolling average of the change in generated revenue and student fees also outlined in § 23.1-1309, Code of Virginia.
- 2. Consistent with the 2019 updates to the Virginia Public Procurement Act, institutions of higher education that have entered into memoranda of understanding or management agreements with the state are permitted to conform their Request for Proposal advertising rules to that of § 2.2-4302.2.A.2."

Explanation:

(This amendment provides funding to Virginia's public colleges and universities to meet the challenges of the COVID-19 pandemic, support operations, and address access and affordability. Specifically, the amendment provides the following: 1) Continues first-year one-time support of \$60.0 million GF provided in Chapter 56, 2020 Special Session I, into the second year of the biennium, and includes funding for George Mason University and Old Dominion University, for operational support, to enhance financial aid, and to address the impacts of the COVID-19 pandemic; 2) Allocates an additional \$17.3 million from the Coronavirus Relief Fund to address the costs of testing for COVID-19; 3) Allocates \$22.0 million from "GEER2" for additional financial aid, and 4) Provides relief from certain

procurement and other requirements to provide institutions with additional flexibility and to generate cost savings.)

Item 266 #1s

Finance

Department of Accounts

Language

Language:

Page 292, after line 39, insert:

"E.1. There is hereby created in the state treasury a special, nonreverting fund to be known as the Opioid Abatement Fund. All funds appropriated to the Fund, all funds designated by the Attorney General under § 2.2-507.3 from settlements, judgments, verdicts, and other court orders relating to claims regarding the manufacturing, marketing, distribution, or sale of opioids, and any gifts, donations, grants, bequests, and other funds received on the Fund's behalf shall be paid into the state treasury and credited to the Fund. Interest earned on moneys in the Fund shall remain in the Fund and be credited to it. Any moneys remaining in the Fund at the end of each fiscal year, including interest thereon, shall not revert to the general fund but shall remain in the Fund.

2. The provisions contained in this paragraph shall be in effect until July 1, 2021, at which time any balances remaining in this Fund shall transfer to the Opioid Abatement Fund created pursuant to SB 1469 of the 2021 General Assembly, and subject to the provisions thereof."

Explanation:

(This amendment establishes the Opioid Abatement Fund, consistent with SB 1469 of the 2021 General Assembly, which shall become effective July 1, 2021. Establishment of this Fund will allow for the deposit of any opioid related litigation settlement moneys received by the Commonwealth until such time as the Fund and associated Authority established under SB 1469 are effectuated.)

		Item 282 #1s	
Finance	FY20-21	FY21-22	
Department of Taxation	\$0	(\$69,816) GF	

Language:

Page 301, line 17, strike "\$61,659,588" and insert "\$61,589,772".

Page 301, line 16, unstrike the stricken language.

Page 301, line 17, strike "\$61,659,588".

Page 301, line 18, unstrike the stricken language.

Page 301, line 21, strike "\$12,371,531".

Page 301, line 25, unstrike the striken language.

Page 301, line 26, strike "\$50,819,573".

Explanation:

(This amendment removes funding included in the Introduced Budget for the Department of Taxation to make systems changes related to the Virginia Facilitated Enrollment Program.)

		Item 284 #1s	
Finance	FY20-21	FY21-22	
Department of Taxation	\$0	(\$164,819)	GF
T			

Language:

Page 306, line 51, strike "\$52,406,035" and insert "\$52,241,216".

Page 306, line 50, unstrike the striken language.

Page 306, line 51, strike "\$52,406,035".

Page 306, line 52, unstrike the striken language.

Page 306, line 53, strike "\$31,261,776".

Page 307, line 1, unstrike the stricken language.

Page 307, line 2, strike "\$21,144,259".

Page 307, line 3, unstrike the stricken language.

Page 307, line 4, strike "\$52,252,581".

Page 307, line 31, unstrike the stricken language.

Page 307, line 32, strike "\$120,185,117".

Page 307, line 36, unstrike the stricken language.

Page 307, line 37, strike "\$107,702,426".

Explanation:

(This amendment removes funding included in the Introduced Budget for the Department of Taxation to make systems changes related to the Virginia Facilitated Enrollment Program.)

		Item 288 #1s	
Finance	FY20-21	FY21-22	
Treasury Board	\$4,000,000	\$0	GF

Language:

Page 310, line 34, strike "\$850,158,182" and insert "\$854,158,182".

Page 315, after line 41, insert:

"I. Out of this appropriation, \$4,000,000 the first year from the general fund is provided for the defeasance of the outstanding bonds on the Central Virginia Training Center."

Explanation:

(This amendment provides \$4 million GF in FY 2021 for the defeasance of the bonds of outstanding bonds on the Central Virginia Training Center.)

Item 291 #1s

Health and Human Resources

Secretary of Health and Human Resources

Language

Language:

Page 318, after line 53, insert:

"F. The Secretary of Health and Human Resources, or his designee, shall convene a workgroup of appropriate agencies within the secretariat and other stakeholders, as necessary, to research and recommend strategies for the financing of health care services for undocumented immigrant children. The workgroup shall: (i) identify the number of children who would qualify and their geographic location; (ii) demonstrate the impact a lack of health care coverage has on these children; (iii) determine the financial burden carried by hospital systems and other healthcare facilities that currently provide care for these children; (iv) identify the existing barriers these children face when trying to access essential medical services in a timely manner; (v) identify the long-term health impacts to children who do not have health care coverage and the future cost the Commonwealth will incur as a result; and (vi) recommend options for providing health care coverage to these children and the approximate cost to the Commonwealth."

Explanation:

(This amendment directs the Secretary of Health and Human Resources to convene a workgroup to research and recommend strategies for the financing of health care services for undocumented immigrant children. Approximately 13,000 immigrant children in Virginia lack health insurance and approximately 9,000 of those children live under 200 percent of the federal poverty level. They do not qualify for CHIP-funded (Children's Health Insurance Program) health insurance due to their immigration status.)

Item 291 #2s

Health and Human Resources

Secretary of Health and Human Resources

Language

Language:

Page 318, after line 53, insert:

- "F.1. It is the intent of the General Assembly that the Virginia Department for the Aging (VDA) be reestablished as an agency of the Commonwealth under the Office of the Secretary of Health and Human Resources beginning July 1, 2022. This agency shall oversee policies and programs impacting older Virginians and provide a leadership role across state government in evaluating the impact the aging population has on state services.
- 2. The Secretary of Health and Human Resources, or his designee, shall convene a workgroup that includes representatives from the Department for Aging and Rehabilitative Services, Area Agencies on Aging, the Department of Planning and Budget, the Division of Legislative Services, appropriate staff from the House Appropriations and Senate Finance and Appropriations Committees, and other appropriate stakeholders. The workgroup shall: (i) review other state aging departments and best practices for establishing an agency fully capable of leading across state government with regard to impacts from an aging population; (ii) review and develop an optimal organizational structure for the new agency; (iii) develop a transition plan for transferring staff, funding and making other operational changes to establish the new agency; (iv) draft legislation for consideration by the 2022 General Assembly; (v) determine potential costs to create the new agency; and (vi) develop draft changes to the Appropriation Act. The workgroup shall at a minimum evaluate transitioning aging services, adult services, adult protective services and auxiliary grant programs to the new agency. In addition, the workgroup shall examine any other aging-related programs in the Health and Human Resources Secretariat and make recommendations for inclusion in the new agency.
- 3. The workgroup shall provide all deliverables and report on its findings by October 1, 2021, to the Governor, the Department of Planning and Budget, and the Chairs of House Appropriations and Senate Finance and Appropriations Committees. The Department for Aging and Rehabilitative Services shall provide this information to the Governor to be considered for inclusion in the 2022-2024 introduced budget."

Explanation:

(This amendment provides that it is the intent of the General Assembly that a new agency for aging services be established as an agency of the Commonwealth under the Office of the Secretary of Health and Human Resources, beginning July 1, 2022. The language creates a workgroup to consider the details of such a transition and to make the appropriate recommendations to the General Assembly.)

Item 291 #3s

Health and Human Resources

Secretary of Health and Human Resources

Language

Language:

Page 324, after line 6, insert:

"D. The Office of Children's Services shall develop a plan to modify its staffing and operations to ensure effective local implementation of the Children's Services Act. The plan shall include

any new or different staff positions required, how those positions will be used to monitor and improve effectiveness, and the estimated cost of implementing these changes. The plan shall be submitted to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021.

E. The Office of Children's Services shall collect annually from each local Children's Services Act program the number of program staff by full- and part-time status and the administrative budget broken out by state and local funding to understand local program resources and target technical assistance to the most under-sourced local programs."

Explanation:

(This amendment requires the Office of Children's Services (OCS) to develop a plan to modify its staffing and operations to ensure effective local implementation of the Children's Services Act (CSA). It also directs OCS to collect annually from each local CSA program the number of program staff by full- and part-time status and information on local administrative budgets to better understand local program sources. These actions are based on recommendations from the Joint Legislative Audit and Review Commission's November 2020 study on CSA.)

Item 292 #1s

Health and Human Resources

Children's Services Act

Language

Language:

Page 323, strike lines 1 through 5.

Explanation:

(This amendment eliminates the annual two percent rate cap on increases that localities may pay for private day special services under the Children's Services Act. This rate cap was temporary until the completion of the rate setting study, which will be complete in fiscal year 2021.)

Item 292 #2s

Health and Human Resources

Children's Services Act

Language

Language:

Page 323, strike lines 6 through 14.

Explanation:

(This amendment removes language that allows localities to adjust daily or monthly rates for the 2020-2021 school year for virtual or distance learning provided by a private school serving

students with disabilities under the Children's Services Act.)

		Item 292 #3s	
Health and Human Resources	FY20-21	FY21-22	
Children's Services Act	\$0	\$305,357	GF

Language:

Page 319, line 4, strike "\$384,786,416" and insert "\$385,091,773".

Explanation:

(This amendment provides \$305,357 the second year from the general fund for the fiscal impact of Senate Bill 1338, which establishes the State Kinship Guardianship Assistance program, which allows payments to be made to relatives, including fictive kin, who receive custody of a child. A corresponding amendment in the Department of Social Services reduces funding through the Title IV-E program as some children will transition to the State-Funded Kinship Guardship Assistance Program.)

		Item 293 #1s	
Health and Human Resources	FY20-21	FY21-22	
Children's Services Act	\$0	(\$50,000)	GF

Language:

Page 323, line 16, strike "\$2,059,796" and insert "\$2,009,796".

Page 324, unstrike lines 1 through 2.

Page 324, line 2, after "effective" strike "on July 1, 2021" and insert "upon a date approved by the General Assembly in the 2022 Session".

Page 324, strike lines 3 through 6, and insert:

"C. Out of this appropriation, \$50,000 the second year is provided to the Office of Children's Services for training and technical assistance activities in overseeing the Children's Services Act. OCS is authorized to use a portion of these funds to hire a facilitator to assist with the workgroup created pursuant to Senate Bill 1313."

Explanation:

(This amendment reduces \$50,000 the second year from the general fund that was provided in the introduced budget for administration of a rate setting process for special education private day programs. Language modifies the effective date for rate setting to a future date to be determined by the 2022 General Assembly pending the results of the final rate setting study. Out of the \$100,000 general fund provided in the introduced budget, \$50,000 remains in the Office of Children's Services to be used for training and technical assistance related to the

Children's Services Act (CSA) program. The agency is allowed to use a portion of this funding for a facilitator for the stakeholder workgroup pursuant to Senate Bill 1313, that would determine how to move special education private day funding from CSA to the Department of Education.)

		Item 295 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$1,600,000	GF

Language:

Page 325, line 25, strike "\$985,000" and insert "\$2,585,000".

Page 325, strike lines 40 through 54.

Page 326, strike lines 1 through 7.

Page 326, unstrike lines 8 through 36.

Page 326, line 11, after "adolescent psychiatrists;" insert:

"psychiatric physician assistants; psychiatric pharmacists;".

Explanation:

(This amendment restores funding and language that was provided in Chapter 1289, 2020 Acts of Assembly, to establish the Behavioral Health Loan Repayment Program. Funding for this item was unallotted in April, 2020 and eliminated in Chapter 56, 2020 Special Session I Acts of Assembly. The program would increase the number of Virginia behavioral health practitioners through the establishment of an educational loan repayment incentive that complements and coordinates with existing efforts to recruit and retain Virginia behavioral health practitioners. The program would allow for a variety of behavioral health practitioners to receive a student loan repayment award from the Commonwealth in exchange for providing service to Virginia communities that are otherwise underserved. Practitioners would receive loan repayment for up to 25 percent of student loan debt for each year of health care service provided to the Commonwealth. Maximum loan repayment amounts per year are dependent upon the type of behavioral health professional applying and shall not exceed the total student loan debt. Participating practitioners will have an initial two-year minimum participation obligation and may renew for a third and fourth year. This provides the practitioner with the opportunity to fully pay off their student loan debt while providing four years of service to the Commonwealth. In addition, this amendment also strikes outdated language related to developing a plan for increasing the number of behavioral health practitioners.)

		Item 295 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$35,000	GF

Language:

Page 325, line 24, strike "\$985,000" and insert "\$1,020,000".

Page 326, after line 52, insert:

"E. Out of this appropriation, \$35,000 the second year from the general fund is provided for the Nurse Loan Repayment Program to provide loan repayments for certified nurse aides. The total loan repayment allowed per certified nurse aide is limited to no more than \$1,000."

Explanation:

(This amendment provides \$35,000 from the general fund the second year to assist with funding for the expansion of the nurse loan repayment program pursuant to Senate Bill 1147 that adds certified nurse aides as eligible for the program. The additional funding is dedicated to supporting certified nurse aides and is capped at \$1,000.)

		Item 295 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$500,000	GF

Language:

Page 325, line 25, strike "\$985,000" and insert "\$1,485,000".

Page 326, unstrike lines 37 through 52.

Page 326, line 37, after "D.", insert "1.".

Page 326, line 37, strike "\$500,000 the first year and".

Page 326, line 51, strike "2020" and insert "2021".

Page 326, after line 52, insert:

"2. The Virginia Health Workforce Development Authority shall develop the process for the consideration of requests for funding from the Nursing Preceptor Incentive Program."

Explanation:

(This amendment provides \$500,000 the second year from the general fund to the Virginia Department of Health to establish a Nursing Preceptor Incentive Program. The department would report to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021 on the progress of establishing the Nursing Preceptor Incentive Program. The Virginia Health Workforce Development Authority is directed to develop a process for the consideration of requests for funding from the Nursing Preceptor Incentive Program.)

Health and Human Resources FY20-21 FY21-22

Department of Health 0.00 30.00 FTE

Language:

Explanation:

(This amendment adds 30 full-time positions funded through nongeneral funds in the Virginia Department of Health's Office of Emergency Medical Services (OEMS). These positions would backfill the loss of positions based on implementation of the agency's shared business services. However, the positions are needed to enable OEMS to work directly with the Regional Emergency Medical Services (EMS) Councils. Each council maintains a Board of Directors but they are staffed by OEMS. In addition, the OEMS will have responsibility for directly managing two regional EMS councils (Shenandoah and Rappahannock). Funding will be shifted within OEMS to cover the cost of the added positions.)

		Item 297 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$51,146	GF

Language:

Page 328, line 2, strike "\$16,001,106" and insert "\$16,052,252".

Explanation:

(This amendment provides \$51,146 from the general fund the second year to support a work group to develop a plan for establishing a Fetal and Infant Mortality Review at the Virginia Department of Health. Such a team would function in a similar manner to the child fatality review team but focus on tracking and investigating selected fetal and infant deaths in the Commonwealth and make recommendations to reduce preventable deaths.)

		Item 299 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 0.00	\$395,818 3.00	GF FTE

Language:

Page 328, line 34, strike "\$213,178,894" and insert "\$213,574,712".

Explanation:

(This amendment provides \$395,818 the second year from the general fund in the Office of Epidemiology to support three additional positions to the \$3.1 million included in the

introduced budget provided to sustain and expand the Virginia Department of Health's communicable disease surveillance and investigation efforts across the Office of Epidemiology and the 35 local health districts in Virginia. The introduced budget funded 12 epidemiologists and 12 communicable disease nurses across the 35 health districts and funded two epidemiology program managers in the Office of Epidemiology in the Central Office. This amendment funds three additional epidemiology program managers in the Office of Epidemiology. A separate amendment funds a total of 35 epidemiologists and 35 communicable disease nurses, one of each position, in every local health district, to expand the Commonwealth's communicable disease surveillance and investigation capabilities.)

		Item 299 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	(\$30,184,899) \$18,002,665	(\$59,123,029) \$59,123,029	GF NGF

Language:

Page 328, line 35, strike "\$140,808,393" and insert "\$128,626,159".

Page 330, line 24, strike "\$30,184,899", and insert "\$18,002,665".

Page 330, line 25, strike "the general fund", and insert "federal funds".

Explanation:

(This amendment supplants \$30.2 million the first year and \$59.1 million the second year to support the Commonwealth's mass vaccination efforts in response to the COVID-19 pandemic. On December 27, 2020, the federal Consolidated Appropriations Act, 2021 was signed into law which provides substantial federal assistance to support states in vaccine administration efforts. Virginia's share of this funding is \$77.1 million. With this additional federal support, the general fund added in the introduced budget can be supplanted with federal funds. The new federal funding is not sufficient to fully supplant all \$89.3 million of general fund in the biennium, however, Coronavirus Relief Funds (CRF) allocated to the Department of Medical Assistance Services have not been fully utilized, so this amendment includes \$12.2 million in the first year in CRF funds. A separate amendment in Central Appropriations reflects the change in the allocation of CRF funds.)

		Item 299 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	(\$722,472)	GF

Language:

Page 328, line 35, strike "\$213,178,894" and insert "\$212,456,422".

Page 330, line 16, strike "\$1,444,944", and insert "\$722,472". Page 330, line 18, strike "\$1,008,000", and inset "\$504,000". Page 330, line 21, strike "\$436,944", and insert "\$218,472".

Explanation:

(This amendment reduces \$722,472 the second year from the general fund for COVID-19 data modeling, being provided by the University of Virginia's Biocomplexity Institute and the RAND corporation to provide epidemiologic analysis for the state's response to COVID-19. This amendment removes half the second year funding assuming that after calendar year 2021, the need for such modeling would likely be mitigated as the COVID-19 pandemic subsides as a result of widespread use of vaccines. In the case that COVID-19 continues as a public health emergency for longer and the need for such modeling continues, additional funding can be provided for this purpose in the 2022 General Assembly Session.)

Item 300 #1s

Health and Human Resources

Department of Health

Language

Language:

Page 332, after line 11, insert:

"H. The provisions of § 32.1-102.4 (B), Code of Virginia, shall not apply to nursing homes."

Explanation:

(This amendment clarifies that the charity care provisions of § 32.1-102.4 (B), Code of Virginia, do not apply to nursing facilities. This is a technical amendment to ensure the appropriate interpretation of the charity care requirements, based on changes made in the 2020 General Assembly Session, are applied consistent with the intent of that legislation.)

		Item 300 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$88,914	GF

Language:

Page 330, line 45, strike "\$21,336,679" and insert "\$21,425,593".

Explanation:

(This amendment provides \$88,914 the second year from the general fund to fund one position for administering the Behavioral Health Loan Repayment program and the Nursing Preceptor Incentive program funded in companion amendments in Item 295.)

Item 300 #3s

Health and Human Resources

Department of Health

Language

Language:

Page 332, strike lines 8 through 11.

Explanation:

(This amendment eliminates language included in the introduced budget that directs the Virginia Department of Health to provide administrative and technical support to the Virginia Partners in Prayer Program at a cost of up to \$20,000 in funding from the agency's existing budget. The department already provides support to this program and therefore this language is unnecessary.)

		Item 301 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$250,000	GF

Language:

Page 332, line 13, strike "\$163,353,397" and insert "\$163,603,397".

Page 332, line 40, after "C." insert "1."

Page 332, after line 44, insert:

"2. Out of this appropriation, \$250,000 the second year from the general fund is provided to establish a comprehensive adult program for sickle cell disease."

Explanation:

(This amendment adds \$250,000 the second year from the general fund for a comprehensive adult sickle cell disease (SCD) program. Historically, Virginia has funded SCD care for children but not for adults. This amendment would establish a comprehensive adult program for sickle cell disease within the Office of Family Health Services and would create a model of care based on the structure of existing pediatric care models.)

		Item 302 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	(\$7,364,304)	GF
Language:			

Page 333, line 20, strike "\$292,637,694" and insert "\$285,273,390".

Page 336, strike lines 9 through 16, and insert:

"H. Out of this appropriation, \$2,835,696 the second year from the general fund shall be provided to address revisions to the JLARC rate formula for the Cooperative Health Budget. These revisions and the changes in the local match rates shall be phased in over a three-year period beginning in the second year and shall be fully phased in by fiscal year 2024."

Explanation:

(This amendment reduces \$7.4 million the second year from the general fund by phasing in over three years the increase in local matching funds and the increase in state support for the updates to the local health department cooperative funding formula. The introduced budget provided \$10.2 million, which included funding for a hold harmless in fiscal year 2022, to implement the funding formula changes. This amendment reflects one-third of the state impact in the second year as the changes are phased in over three years and, as such, a hold harmless is not necessary.)

		Item 302 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 0.00	\$5,470,677 46.00	GF FTE

Language:

Page 333, line 20, strike "\$292,637,694" and insert "\$298,108,371".

Explanation:

(This amendment provides \$5.5 million the second year from the general fund in the Office of Epidemiology to support 46 additional positions in addition to the \$3.1 million included in the introduced budget provided to sustain and expand the Virginia Department of Health's communicable disease surveillance and investigation efforts across the Office of Epidemiology and the 35 local health districts in Virginia. The introduced budget funded 12 epidemiologists and 12 communicable disease nurses across the 35 health districts and funded two epidemiology program managers in the Office of Epidemiology in the Central Office. This amendment funds a total of 35 epidemiologists and 35 communicable disease nurses, one of each position, in every local health district, to expand the Commonwealth's communicable disease surveillance and investigation capabilities. A separate amendment funds three additional epidemiology program managers in the Office of Epidemiology)

Health and Human Resources FY20-21 FY21-22

Department of Health \$0 \$75,889 GF

Language:

Page 333, line 20, strike "\$292,637,694" and insert "\$292,713,583".

Explanation:

(This amendment restores \$75,889 the second year from the general fund that was originally appropriated in Chapter 1289 during the 2020 Regular Session, but then eliminated in Chapter 56 in the 2020 Special Session I. This funding supports local health districts that may experience cost increases due to moving to new facilities or rent increases in existing facilities. Health districts do not fully control the process for determining when and where their facilities will be located.)

		Item 302 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	0.00	-44.00	FTE

Language:

Explanation:

(This amendment eliminates 44 positions for the Loudoun County Health District, which pursuant to Senate Bill 1221 converts the health district from a state-administered health district to a locally-administered one. Therefore, existing employees would transition to become local employees.)

		Item 304 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 0.00	\$250,000 3.00	GF FTE

Language:

Page 340, line 34, strike "\$33,588,623" and insert "\$33,838,623".

Explanation:

(This amendment provides \$250,000 the second year from the general fund and three positions for the Virginia Department of Health to handle testing plans and test results of lead water testing by local schools and provides funding for the department to handle the testing plans and test results of lead water testing submitted to the agency from child care facilities.)

Item 305 #1s

Health and Human Resources	FY20-21	FY21-22	
Department of Health	0.00	-1.00	FTE

Language:

Explanation:

(This amendment eliminates one general fund position included in the introduced budget that was added with the funding for a wastewater infrastructure manager. This funding is a restoration of funding previously approved in Chapter 1289 from the 2020 Session. However, the funding for this position was eliminated in Chapter 56 in the Special Session I, but the position was not removed. Therefore, this amendment corrects the position level for the agency.)

		Item 307 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$60,000	GF

Language:

Page 341, line 27, strike "\$38,204,545" and insert "\$38,264,545".

Explanation:

(This amendment provides \$60,000 the second year from the general fund for the continued work of the Virginia Department of Health's Office of Drinking Water to continue its study of the occurrence of perfluorooctanoic acid (PFOA), perfluorooctane sulfonate (PFOS), and other perfluoroalkyl and polyfluoroalkyl substances (PFAS) in the Commonwealth's public drinking water and to develop recommendations for specific maximum contaminant levels for PFOA, PFOS, and other PFAS for inclusion in regulations of the Board of Health applicable to waterworks.)

		Item 307 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	(\$7,100,000)	(\$6,250,000)	GF

Language:

Page 341, line 27, strike "\$34,810,621" and insert "\$27,710,621".

Page 341, line 27, strike "\$38,204,545" and insert "\$31,954,545".

Page 343, line 1, after "appropriation,", strike "\$6,500,000 the first year and \$12,500,000", and

insert "\$6.250,000".

Page 343, line 4, after "pandemic.", insert:

"The Virginia Department of Health shall allocate no less than 20 percent of funding from state or federal sources dedicated for COVID-19 communications to use for outreach and communications to high-risk populations that have been adversely impacted by the COVID-19 pandemic more so than the general population and for which traditional communication mediums are not as effective. The department shall use such funding for alternative methods of communication, such as outreach coordinators going into communities, providing information pamphlets as part of meal pick-ups at schools, grants to community organizations, and other more effective ways at reaching high-risk populations. This funding shall also be used to translate communication materials into other languages; however the department shall not use Google Translate in any communications to non-English speakers."

Page 343, line 5, strike "Out of this appropriation,", and insert "No less than".

Page 343, line 5, strike "from the general fund" and insert "Coronavirus Relief Funds".

Explanation:

(This amendment supplants \$7.1 million the first year and \$6.3 million the second year from the general fund with available federal Coronavirus Relief Funds (CRF). The CRF funds allocated to the Department of Medical Assistance Services have not all been fully utilized, so this amendment supplants \$7.1 million in the first year and \$6.3 million the second year in CRF funds. A separate amendment in Central Appropriations reflects the change in the allocation of CRF funds. In addition, language directs a portion of the state and federal funding allocated for COVID-19 communications to be used for outreach and communications to high-risk populations that have been adversely impacted by the COVID-19 pandemic more so than the general population and for which traditional communication mediums are not as effective. It also directs the funding to be used for translation services.)

		Item 307 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 \$0	\$746,562 \$3,567,494	

Language:

Page 341, line 27, strike "\$38,204,545" and insert "\$42,518,601".

Page 341, line 52, strike "\$26,736", insert "\$773,298".

Page 341, line 52, strike "\$240,625", insert "\$3,808,119".

Page 342, line 30, "after annually", insert "by September 1 of each year".

Page 342, line 33, after "plans", insert "and their patient records viewed".

Explanation:

(This amendment adds \$746,562 from the general fund and \$3.6 million from enhanced federal

matching funds to continue funding for the Emergency Department Care Coordination (EDCC) Program and to complete several enhancements that should result in cost savings to the Commonwealth and higher-quality care for Medicaid and FAMIS beneficiaries. Language is also modified to clarify reporting requirements. The 2017 General Assembly established the EDCC program to provide a single, statewide technology solution that connects all hospital emergency departments in the Commonwealth to facilitate real-time communication and collaboration among physicians, other healthcare providers and other clinical and care management personnel for patients receiving services in hospital emergency departments for the purpose of improving the quality of patient care services. The program also integrates the state's Prescription Monitoring Program and the Advance Healthcare Directive Registry. To date, 100 percent of hospital emergency departments and health plans and 74 percent of physicians participate in the EDCC Program. The program has been successful in supporting a reduction in opioid overdoses since its implementation and has been critical in coordinating responses to the COVID-19 pandemic. The program has been funded through federal HITECH funding which provides a 90 percent federal match rate, but the federal funding will end on September 30, 2021. However, Medicaid Management Information Systems funding could be used to assist in leveraging additional federal dollars to complete the system enhancements at a 75 percent federal match rate.)

		Item 309 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health Professions	\$0	\$532,000	NGF

Language:

Page 343, line 34, strike "\$36,027,084" and insert "\$36,559,084".

Page 343, after line 51, insert:

"C. The Department of Health Professions shall have authority to increase fees from permitted pharmaceutical processors to generate up to \$532,000 the second year from nongeneral funds, as needed, for the implementation of botanical cannabis legislation pursuant to Senate Bill 1333 in the 2021 Session."

Explanation:

(This amendment provides up to \$532,000 the second year from nongeneral funds and language providing the Department of Health Professions with the authority to increase fees from permitted pharmaceutical processors pursuant to the implementation of botanical cannabis legislation pursuant to Senate Bill 1333.)

Item 309 #2s

Health and Human Resources

Department of Health Professions

Language

Language:

Page 343, after line 51, insert:

"C. The Department of Health Professions shall study and make recommendations regarding the oversight and regulation of advanced practice registered nurses (APRNs). The department shall review recommendations of the National Council of State Boards of Nursing, analyze the oversight and regulations governing the practice of APRNs in other states, and review research on the impact of statutes and regulations on practice and patient outcomes. The department shall report its findings to the Governor and General Assembly by November 1, 2021."

Explanation:

(This amendment directs the Department of Health Professions to study and make recommendations regarding the oversight and regulations of advanced practice registered nurses.)

		Item 312 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$11,136,631 \$20,682,315	

Language:

Page 344, line 38, strike "\$250,286,516" and insert "\$282,105,462". Page 345, after line 37, insert:

"H. The Department of Medical Assistance Services shall amend the Title XXI Children's Health Insurance Program (CHIP) State Plan for the Family Access to Medical Insurance Security (FAMIS) program to elect the "unborn child option" for purposes of prenatal coverage under the CHIP program, without regard to the pregnant woman's immigration or citizenship status if other applicable state eligibility requirements are met. The department shall have the authority to implement this change effective July 1, 2021, or consistent with the effective date in the State Plan Amendment approved by the Centers for Medicare and Medicaid Services (CMS), and prior to completion of any regulatory process necessary to implement this change."

Explanation:

(This amendment provides \$11.1 million from the general fund and \$20.7 million from federal funds to amend the Virginia Family Access to Medical Insurance Security (FAMIS) State Plan to allow the payment for prenatal care for all children regardless of the expectant mother's immigration status pursuant to provisions in Title XXI of the federal 2009 CHIP Reauthorization Act that includes care of all children who upon birth will be U.S. citizens, U.S. nationals, or qualified aliens. A companion amendment in the Medicaid program (Item 313) reduces expenditures by \$13.4 million general fund and \$13.4 million in matching federal

Medicaid funds based on the adoption of this change. Consequently, the provision of these services results in a savings of \$2,292,083 to the general fund and the receipt of \$7,253,601 more from matching federal funds for the CHIP program, which has a higher federal match rate than the Medicaid program.)

		Item 312 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	(\$1,834,183) \$1,834,183	(\$5,378,570) \$5,378,570	GF NGF

Language:

Explanation:

(This amendment captures \$1.8 million from the general fund the first year and \$5.4 million from the general fund the second year and a corresponding increase in federal Children's Health Insurance Program (CHIP) matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for the fourth quarter of fiscal year 2021 and the first and second quarters of fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The CHIP match rate factors in the Medicaid match rate and is therefore increased as a result. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022.)

		Item 313 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$6,304,339 \$6,304,339	\$60,829,273 \$60,829,273	

Language:

Page 345, line 39, strike "\$16,291,925,668" and insert "\$16,304,534,346".

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,616,581,861".

Page 375, strike line 57 and insert:

"2. Effective May 1, 2021, the Department of Medical Assistance Services shall increase the rates for agency- and consumer-directed personal care, respite and companion services in the home and community based services waivers and Early Periodic Screening, and Diagnosis and

Treatment (EPSDT) program by 6.4 percent. The department shall have the authority to implement these changes prior to completion of any regulatory process undertaken in order to effect such change.

- 3. Effective November 1, 2021, the Department of Medical Assistance Services shall increase the rates for agency- and consumer-directed personal care, respite and companion services in the home and community based services waivers and Early Periodic Screening, and Diagnosis and Treatment (EPSDT) program by 14.3 percent. The department shall have the authority to implement these changes prior to completion of any regulatory process undertaken in order to effect such change.
- 4. The Governor shall include in the introduced budget for the 2022 Session, submitted pursuant to § 2.2-1509, Code of Virginia, appropriations to support additional rate increases for agency-and consumer-directed personal care, respite and companion services that reflect additional increases in the state minimum wage such that the rates: (i) maintain the existing differential between the consumer-directed Rest-of-State rate above the state minimum wage; (ii) maintain the differential between the Northern Virginia and the Rest-of-State rate; and (iii) for agency-directed services are increased by the same percentage increase applied to consumer-directed services based on the prior provisions."

Page 376, strike lines 1 through 5.

Explanation:

(This amendment adds \$6.3 million from the general fund and \$6.3 million from federal Medicaid matching funds the first year and \$60.8 million the second year from the general fund and \$60.8 million in federal Medicaid matching funds the second year to increase provider rates for personal care, respite care, and companionship services provided in Medicaid waiver programs by 6.4 percent on May 1, 2021 and 14.3 percent effective November 1, 2021. These rate increases will supplement increases provided in Chapter 1289, 2020 Acts of Assembly, which provided a five percent increase beginning July 1, 2020 and a scheduled two percent increase on July 1, 2021. These rate increases enable providers to cover the direct costs for hourly care and ensure the safety of the patients and compliance with minimum wage increases scheduled to increase to \$9.50 per hour on May 1, 2021 and to \$11.00 per hour on January 1, 2022. In addition, language directs the Governor to include the appropriate rate increases as a result of changes in the state minimum wage for these services in his budget for the 2022-24 biennium for consideration in the 2022 General Assembly)

Item 313 #2s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services (DMAS) shall convene a

workgroup and make recommendations on a Medicaid home visiting benefit to support members' health, access to care and health equity. The workgroup shall include representatives from DMAS, Managed Care Organizations, the Virginia Department of Health, the Department of Health Professions, licensed and unlicensed providers of maternal and child health services, stakeholder groups, and community organizations. The workgroup shall: (i) analyze federal and state regulations and funding mechanisms impacting establishment of a Medicaid home visiting benefit; (ii) review home visiting strategies and benefits implemented in other state Medicaid programs; (iii) analyze and make recommendations on appropriate services and rates to be included in a Medicaid home visiting benefit; and (iv) project estimated costs over the next five years. The department shall report on the results and recommendations of the workgroup to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by December 1, 2021."

Explanation:

(This amendment directs the Department of Medical Assistance Services to convene a workgroup and make recommendations on a Medicaid home visiting benefit.)

Item 313 #3s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services shall defer the next scheduled nursing facility rate rebasing for one year in order to utilize the calendar year 2021 cost reports as the base year. The deferred year's rates would reflect the prior year rates inflated according to the existing reimbursement regulations. The department shall have the authority to implement these changes effective July 1, 2021, and prior to the completion of any regulatory process undertaken in order to effect such change."

Explanation:

(This amendment adds language directing the Department of Medical Assistance Services to defer nursing home rebasing until it is able to use 2021 cost reports in the rebasing process. Under the current rebasing schedule, cost reports ending in calendar year 2020 will be the basis of the next nursing facility rate rebasing. Due to the COVID-19 pandemic, the 2020 base year cost reports will contain significant variable data that reflects a mixture of increased costs (some permanent, others temporary) and reduced occupancy. Given the COVID-19 vaccination program, it is expected that the 2021 cost reports would not be as affected by the increased costs associated with the COVID-19 pandemic, and would better reflect ongoing Medicaid costs as a basis for the next rebasing.)

Item 313 #4s

Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance	\$0	\$10,747,100	GF
Services	\$0	\$10,747,100	NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,516,417,515". Page 349, line 46, strike "455" and insert "1,095".

Explanation:

(This amendment adds \$10.7 million from the general fund and a like amount of federal Medicaid matching funds in the second year to increase the number of Family and Individual Support (FIS) waiver slots by 650 in the second year bringing the total number of waiver slots funded in fiscal year 2022 to 1,200 in order to address the Priority One waiting list.)

Item 313 #5s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services shall modify its contracts with managed care organizations to require annual reporting, with regard to Medicaid Community Mental Health Rehabilitation Services, on: (i) the total number of provider terminations by year since FY 2018 and the number terminated with and without cause; (ii) the locality the terminated providers served; and (iii) the number of Medicaid members the providers were serving prior to termination of their provider contract. The department shall modify its contracts with the managed care organizations to require compliance with these provisions, effective July 1, 2021, such that the first reporting of this information by the managed care organizations shall be submitted by September 1, 2021."

Explanation:

(This amendment directs the Department of Medical Assistance Services to require its managed care organizations annually report, with regard to Medicaid Community Mental Health Rehabilitation Services, on information related to provider terminations.)

Item 313 #6s

Health and Human Resources

FY20-21

FY21-22

Department of Medical Assistance \$0 \$5,200,000 NGF Services

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,500,123,315". Page 364, after line 2, insert:

"9. The department shall amend the State plan for Medical Assistance to implement a supplemental inpatient payment for Lake Taylor Transitional Care Hospital based on the difference between Medicaid reimbursement and the inpatient Upper Payment Limit for non-state, government-owned hospitals. The department shall include in its contracts with managed care organizations a percentage increase for Lake Taylor Transitional Care Hospital consistent with the fee-for-service supplemental payment percentage increase. The department shall adjust capitation payments to Medicaid managed care organizations to fund this percentage increase. Both the contract changes and capitation rate adjustments shall be compliant with 42 C.F.R. 438.6(c)(1)(iii) and subject to Centers for Medicare and Medicaid Services approval. Prior to submitting the State Plan Amendment or making the managed care contract changes, Lake Taylor Transitional Care Hospital shall enter into an agreement with the department to transfer the non-federal share for these payments. The department shall have the authority to implement these reimbursement changes consistent with the effective date(s) approved by the Centers for Medicare and Medicaid Services (CMS). No payments shall be made without CMS approval."

Explanation:

(This amendment adds \$5.2 million in nongeneral fund appropriation the second year and language to allow Lake Taylor Transitional Care Hospital, a non-state, government-owned hospital located in Norfolk, Virginia, to provide the Commonwealth's share of funding through an intergovernmental transfer process to procure supplemental payments from the Medicaid program.)

		Item 313 #7s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$57,210 \$57,210	GF NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,495,037,735". Page 376, unstrike lines 29 through 34.

Explanation:

(This amendment provides \$57,210 from the general fund and a like amount of federal Medicaid matching funds the second year for the Department of Medical Assistance Services to amend the State Plan for Medical Assistance or any waiver under Title XIX of the Social

Security Act to increase the income eligibility for participation in the Medicaid Works Program to 138 percent of the Federal Poverty Level (FPL). Current eligibility for the Medicaid Works Program is 80 percent of the FPL and was not adjusted to 138 percent of FPL when Medicaid Expansion was implemented.)

Item 313 #8s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 365, line 10, strike "December 15, 2020", insert "November 1, 2021".

Explanation:

(This amendment extends the workgroup established to evaluate strategies to reduce unnecessary utilization by Medicaid members of emergency departments. The workgroup was delayed in beginning its deliberations and the additional time would result in better recommendations for consideration by the 2022 General Assembly Session.)

Item 313 #9s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after insert 17, insert:

"AAAAAA. The Department of Medicaid Assistance Services shall analyze utilization of Transportation Network Companies (TNC) Type II Non-Emergency Medical Transportation (NEMT) providers in the Medicaid fee-for-service program and the department's contracted managed care organizations (MCOs). The department shall identify any barriers to patient access to TNC Type II NEMT services. In its review of barriers to access TNC Type II NEMT benefits, the department shall identify any gaps in TNC Type II service contracting between the department's contracted MCOs, or their transportation brokers, and TNC Type II NEMT providers. Additionally, the department shall examine the eligible patient population for TNC Type II NEMT services to ensure all clinically indicated Medicaid beneficiaries are eligible for TNC Type II NEMT services. Further, the department shall examine the necessity of TNC Type II operating requirements and identify any extraneous service requirements limiting TNC Type II services. The department shall report its findings and recommendations to the Chairs of House Appropriations and Senate Finance and Appropriations Committees by October 1, 2021."

Explanation:

(This amendment directs the Department of Medical Assistance Services to analyze the usage of Transportation Network Companies (TNC) Type II Non-Emergency Medical Transportation (NEMT) providers in the Medicaid fee-for-service program and the department's contracted managed care organizations (MCOs) and to identify any barriers to accessing such providers.)

		Item 313 #10s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	\$0 \$0	\$2,682,089 GF \$4,186,201 NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,501,791,605".

Explanation:

(This amendment provides \$2.7 million from the general fund and \$4.2 million from nongeneral funds the second year for Medicaid-related costs of remote patient monitoring serves provided via telemedicine for Medicaid recipients with medically necessary conditions pursuant to Senate Bill 1338.)

		Item 313 #11s	5
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$354,766 \$354,766	GF NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,495,632,847".

Page 376, unstrike lines 20 through 28.

Page 376, line 27, strike "2020" and insert "2021".

Explanation:

(This amendment restores \$354,766 the second year from the general fund and a like amount of federal Medicaid matching funds to increase supplemental physician payments for physicians employed at Children's National Medical Center, a freestanding children's hospital serving the Northern Virginia region. Funding for this Item was provided in Chapter 1289, 2020 Acts of Assembly, unallotted in April, 2020 and eliminated in Chapter 56, 2020 Special Session I Acts of Assembly.)

Item 313 #12s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services is authorized to amend the State Plan under Title XIX of the Social Security Act to add coverage for the current procedural terminology (CPT) codes for Applied Behavioral Analysis that were added to the CPT list in January 2019, or any future updates to these CPT codes. The department shall have the authority to implement related programmatic changes to service definitions, prior authorization and utilization review criteria, provider qualifications, and reimbursement rates for the Behavioral Therapy Program. The department shall have the authority to implement these changes effective December 1, 2021, and prior to completion of any regulatory process to effect such changes."

Explanation:

(This amendment adds Medicaid coverage for Applied Behavioral Analysis services that were added to current procedural terminology costs in January 2019 and provides authority to the Department of Medical Assistance Services to make changes based on future updates, including necessary changes to rates, service definitions, and other programmatic requirements.)

		Item 313 #13s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	\$0 \$0	\$119,995 GF \$119,995 NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,495,163,305".

Page 374, unstrike lines 14 through 23.

Page 374, line 22, strike "2020", insert "2021".

Explanation:

(This amendment restores \$119,995 from the general fund and a like amount of matching federal Medicaid funds the second year and language to require the Department of Medical Assistance Services to modify nursing facility capital reimbursement for a nursing facility that became a free-standing facility because the hospital with which the facility was associated relocated. Funding for this Item was provided in Chapter 1289, 2020 Acts of Assembly, unallotted in April, 2020 and eliminated in Chapter 56, 2020 Special Session I Acts of

Assembly.)

Item 313 #14s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services, in coordination with the Department of Behavioral Health and Developmental Services, shall submit a request to the Centers for Medicare and Medicaid Services to amend its 1915(c) Home & Community-Based Services (HCBS) waivers to allow telehealth and virtual and/or distance learning as a permanent service option and accommodation for individuals on the Community Living, Family and Individual Services and Building Independence Waivers. The amendment, at a minimum, shall include all services currently authorized for telehealth and virtual options during the COVID-19 pandemic. The departments shall actively work with the established Developmental Disability Waiver Advisory Committee and other appropriate stakeholders in the development of the amendment including service elements and rate methodologies. The department shall have the authority to implement these changes prior to the completion of the regulatory process."

Explanation:

(This amendment adds language directing the Department of Medical Assistance Services to request amendments from the federal Centers for Medicare and Medicaid Services to the Home and Community Based Services Waivers to permanently continue telehealth and virtual and/or distance learning as service options for disabled individuals receiving these waiver services.)

Item 313 #15s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 380, line 38, after "participation." strike the remainder of the line. Page 380, strike line 39.

Explanation:

(This amendment directs the Department of Medical Assistance Services to continue to allow, beyond the COVID-19 emergency, Medicaid agency-directed personal care and respite services to conduct telephonic supervisory visits by a licensed nurse. The department's forms would be used to document the interaction during these phone calls and to meet the standards already

established by the department to include verbal consent, authorization, and confirmation of participation.)

Item 313 #16s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. Notwithstanding any other provision of law, any nursing facility that increases rates or adds special charges for residents, regardless of payer and including private pay, since the beginning of the Public Health Emergency (PHE) as declared by the U.S. Secretary of Health and Human Services and during such emergency, that are 10 percent or more higher than the monthly charge in the last full month prior to the month in which the PHE began is hereby ineligible for the additional per diem rates authorized in Item 380 Paragraph LLLLL in this Act and for financial assistance provided in Item 479.10 in this Act from federal Coronavirus Relief Funds. Upon a complaint by a resident or family member that a nursing facility has imposed higher rates or special charges that exceed the threshold, the Department of Medical Assistance Services shall verify such action with the nursing facility. If the department determines the nursing facility has increased rates or imposed new charges that exceed the threshold, the department shall no longer make additional payments pursuant to the previously mentioned provisions of this Act. Furthermore, the department shall adjust future payments to such nursing facility to recapture prior payments made under those same provisions."

Explanation:

(This amendment provides that any nursing facility in the Commonwealth that has increased their rates or added additional charges during the public health emergency for COVD-19, that are 10 percent or higher than charges prior to the emergency are not eligible for the \$20 per day per diem or any financial assistance from the Coronavirus Relief Fund as authorized in the current budget. These additional funding streams were provided by the Commonwealth to assist nursing facilities during the public health emergency to help offset losses of revenue or higher costs related to the pandemic. This amendment ensures that nursing facilities that dramatically increase their charges or add special fees to recover costs during the public health emergency are not eligible for the additional state assistance and would also be required to repay any prior payments from that assistance.)

Item 313 #17s

Health and Human Resources

FY20-21

FY21-22

Department of Medical Assistance	\$0	(\$13,428,714)	GF
Services	\$0	(\$13,428,714)	NGF

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,468,065,887".

Explanation:

(This amendment captures savings of \$13.4 million from the general fund and \$13.4 million from federal matching Medicaid funds from extending the provision for the payment of prenatal care for pregnant women through the Medicaid program regardless of the expectant mother's status, pursuant to provisions in Title XXI of the federal 2009 CHIP Reauthorization Act that includes care of all children without regard for an expectant mother's citizenship status who would otherwise be eligible under state requirements. A companion amendment in the FAMIS program (Item 312) adds language and funding for this initiative. The provision of these services results in a savings of \$2.3 million to the general fund and the receipt of \$7.3 million more from matching federal funds for the CHIP program, which has a higher federal match rate than the Medicaid program.)

		Item 313 #18s	5
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$34,718 \$34,718	GF NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,494,992,751". Pag 376, unstrike lines 35 through 39.

Explanation:

(This amendment provides \$34,718 from the general fund and a like amount of federal Medicaid matching funds the second year to add tobacco cessation services to the Medicaid program for adults not otherwise currently covered. The federal Patient Protection and Affordable Care Act (ACA) requires that Medicaid provide coverage for prevention services, including tobacco cessation, for individuals enrolled pursuant to the ACA. This amendment allows all adults in Medicaid to have access to tobacco cessation services.)

		Item 313 #19s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	(\$39,410,177) \$39,410,177	\$0 GF \$0 NGF

Page 346, line 54, strike "\$527,992,971" and insert "\$567,403,148". Page 347, after line 7, insert:

"4. Any repayment by managed care organizations resulting from exceeding their profit caps for not meeting the medical loss ratios pursuant to their contracts with the Department of Medical Assistance Services, shall be deposited to the Health Care Fund."

Explanation:

(This amendment reduces \$39.4 million from the general fund the first year and adds an equivalent amount of nongeneral fund appropriation to reflect a non-participating tobacco manufacture's payment to the Health Care Fund. Since this fund is used as state match for Medicaid, additional revenue to the fund offsets the general fund match for Medicaid. In addition, language also clarifies that repayments from managed care organizations are to be deposited to the Health Care Fund.)

		Item 313 #20s	8
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	(\$114,851,105) \$114,851,105	(+)	GF NGF

Language:

Explanation:

(This amendment captures \$114.9 million from the general fund the first year and \$191.6 million from the general fund the second year and a corresponding increase in federal Medicaid matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for the fourth quarter of fiscal year 2021 and the first and second quarters of fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of fiscal year 2022. The general fund savings for the state behavioral health facilities are also factored into these numbers and total \$808,764 the first year and \$1.8 million the second year.)

Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance	\$0	(\$104,168)	
Services	\$0	\$2,314,798	

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,497,133,945".

Page 346, line 55, strike "\$496,601,500", insert "\$496,705,668".

Page 382, after line 17, insert:

"AAAAAA. The Department shall amend the State Plan for Medical Assistance to allow payment of medical assistance services delivered to Medicaid-eligible students when such services qualify for reimbursement by the Virginia Medicaid program and may be provided by school divisions, regardless of whether the student receiving care has an individualized education program or whether the health care service is included in a student's individualized education program. Such services shall include those covered under the state plan for medical assistance services or by the Early and Periodic Screening, Diagnostic, and Treatment (EPSDT) benefit as specified in § 1905(r) of the federal Social Security Act, and shall include a provision for payment of medical assistance for health care services provided through telemedicine services, as defined in § 38.2-3418.16. No health care provider who provides health care services through telemedicine shall be required to use proprietary technology or applications in order to be reimbursed for providing telemedicine services."

Explanation:

(This amendment provides \$2.2 million from federal Medicaid funds the second year related to expanding Medicaid coverage of school-based services outside of a student's individualized education program. There is no state match required as the local schools certify local expenditures as the state match. The amendment also reflects a savings of \$104,168 from the general fund the second year since five percent of the additional federal funding for school-based services is retained by the state and deposited to the Health Care Fund, which offset general fund costs.)

		Item 313 #22s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	(\$800,000) (\$800,000)	\$0 GF \$0 NGF

Language:

Page 345, line 39, strike "\$16,291,925,668" and insert "\$16,290,325,668".

Page 346, line 7, after "A.", strike "1.".

Page 346, strike lines 11 through 16.

Explanation:

(This amendment reduces \$800,000 from the general fund the first year and a like amount of federal Medicaid matching funds to reflect the temporary pause in Medicaid billing by the Commonwealth Center for Children and Adolescents (CCCA) as a result of not meeting the accreditation standards necessary to bill for Medicaid. The facility is expected to begin billing again in the second year once accreditation is achieved, so this amendment adjusts the appropriation in the first year only. Language is also removed that would have allowed these funds to be transferred to CCCA; however such action is unnecessary as the facility is using special funds to cover the loss in Medicaid revenue.)

Item	313	#23s

Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance	\$0	(\$10,062,988)	
Services	\$0	(\$38,332)	

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,484,821,995".

Explanation:

(This amendment removes \$10.1 million from the general fund and \$38,332 in federal Medicaid matching funds to reflect the updated timeline for the phase-in of community behavioral health services in Medicaid. The redesign of these services was originally planned to be implemented beginning January 2021, however the budget impacts of COVID-19 resulted in delaying the beginning date to July 1, 2021. In the Special Session budget, Chapter 56, the full second year funding was restored, but the funding did not reflect the revised timeline. This amendment adjusts funding for these services to reflect the updated timeline including how the various services will be phased in over the next year.)

Item 313 #24s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 381, line 38, strike "\$16" and insert "\$13".

Page 381, line 41, after "fee.", insert:

"The department is authorized to set the administration fee at the same level as the Medicare reimbursement rates for COVID-19 vaccines if federal law or policy changes such that the Medicaid costs of the COVID-19 vaccine administration fee is 100 percent reimbursed from federal funds."

Explanation:

(This amendment corrects the fee amount included in the introduced budget for Medicaid reimbursement for vaccine administration in a pharmacy. Currently, Medicaid does not have a vaccine administration fee for pharmacists. The introduced budget directs the Department of Medical Assistance Services to establish such a fee and sets the fee at \$16. The department's budget assumptions for this fee was \$13. This amendment corrects the language to reflect the \$13 administration fee as reflected in the actual appropriation otherwise there would be a \$7.1 million general fund shortfall in Medicaid. Language is added that if the federal government decides to provide a 100 percent federal match for vaccine administration fees related to COVID-19, then the department is authorized to increase the fee to the Medicare reimbursement level.)

		Item 314 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$15,000,000	\$0	GF

Language:

Page 382, line 18, strike "\$821,702" and insert "\$15,821,702". Page 382, after line 34, insert:

"C. The Department of Medical Assistance Services, in collaboration with the Department of Behavioral Health and Developmental Services and the Department of Planning and Budget, shall develop criteria to provide support payments to Medicaid Developmental Disability Waiver providers that have experienced a significant disruption in operations and revenue during the COVID-19 public health emergency (PHE). The criteria shall include: (i) first priority to provide financial support for providers that have received no other state or federal assistance to date during the PHE, including supported employment providers; (ii) the second priority shall be other waiver providers that have received some limited assistance from state and federal sources, including day support providers, but may need additional assistance; (iii) the third priority to support other waiver providers that are still in operation and are at risk of closing due to the PHE disruption and for which the Commonwealth needs to maintain an adequate provider network such that when the PHE emergency ends there are sufficient providers to meet the service needs of Medicaid members; (iv) a support payment amount to be based on the provider's Medicaid monthly revenue and service authorizations prior to the PHE, however, for the period of assistance provided to the provider the payment may not exceed 65 percent of that prior monthly revenue; (v) the time period for assistance, which may not exceed three months; and (vi) other relevant criteria to meet the intent of this funding. The department shall issue guidelines within 20 days of enactment of this Act. After the guidelines are issued providers shall have 30 days to submit their applications consistent with the guidelines. The department shall determine the appropriate payments based on the guidelines, which shall be

prorated if funding is insufficient, and begin making the payments no later than 30 days after the provider submission deadline. Any unexpended balance as of June 30, 2021 shall not revert to the general fund, but shall be reappropriated for this purpose into FY 2022."

Explanation:

(This amendment adds \$15.0 million the first year from the general fund to provide support payments to Medicaid Developmental Disability Waiver providers. Payments would help support the stabilization of providers during COVID-19 and would retain community service capacity after the pandemic ends.)

		Item 315 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	(\$1,762,463) \$1,762,463	(\$5,250,333) \$5,250,333	GF NGF

Language:

Explanation:

(This amendment captures \$1.8 million from the general fund the first year and \$5.3 million from the general fund the second year and a corresponding increase in federal Children's Health Insurance Program (CHIP) matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for the fourth quarter of fiscal year 2021 and the first and second quarters of fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The CHIP match rate factors in the Medicaid match rate and is therefore increased as a result. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022.)

		Item 317 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0	(\$1,000,000)	GF

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$282,053,336".

Explanation:

(This amendment reduces by \$1.0 million general fund the second year funding provided in the introduced budget for managed care operational changes. This action leaves \$1.2 million from the general fund for this purpose. The agency's administrative budget can absorb any additional costs necessary to implement these changes.)

		Item 317 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$53,247 \$103,361	GF NGF

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$283,209,944".

Explanation:

(This amendment provides \$53,247 general fund and \$103,361 in federal Medicaid matching funds in the second year to fund the costs of an on-going orientation program for consumer-directed home care workers providing in-home care under the state's Medicaid program pursuant to Senate Bill 1102.)

Item 317 #3s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 355, line 10, after "EE.", insert "1.".

Page 355, line 12, after "Social Security Act to", delete the remainder of the line, and insert: "merge the Commonwealth Coordinated Care Plus and Medallion 4.0 managed care programs, effective July 1, 2022, into a single, streamlined managed care program that links seamlessly with the fee-for-service program, ensuring an efficient and well-coordinated Virginia Medicaid delivery system that provides high-quality care to its members and adds value for providers and the Commonwealth. The department shall have authority to implement necessary changes upon federal approval and prior to the completion of any regulatory process undertaken in order to effect such change.

2. The Department of Medical Assistance shall undertake a review of current contracts and staffing to determine the operational savings that would result from merging the Commonwealth Coordinated Care Plus and Medallion 4.0 managed care programs and may use such administrative savings that are available to fund other upfront costs associated with

merging the two managed care programs. The department shall report on its review of such administrative cost savings and merger-related costs by October 1, 2021 to the Department of Planning and Budget and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees."

Page 355 strike lines 13 through 54.

Page 356, strike lines 1 through 28.

Explanation:

(This amendment directs the Department of Medical Assistance Services to merge the Commonwealth Coordinated Care Plus and Medallion 4.0 managed care programs, effective July 1, 2022, into a single, streamlined managed care program that links seamlessly with the fee-for-service program, ensuring an efficient and well-coordinated Virginia Medicaid delivery system that provides high-quality care to its members and adds value for providers and the Commonwealth.)

		Item 317 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$50,000 \$50,000	GF NGF

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$283,153,336".

Page 385, after line 21, insert:

- "3. The Department of Medical Assistance Services shall post on its website the complete State Plan for Medical Assistance along with all amendments in an easily searchable format to be accessible to the public.
- 4. Within five days of any submission of a state plan amendment to the Centers for Medicare and Medicaid Services, the Department of Medical Assistance Services shall post such submission in its website. The department shall also post any federal approval documents once the state plan amendment is approved.
- 5. The department shall publish a document on its website, updated annually, that lists all policy changes, including their fiscal impact, for the Medicaid program for the preceding fiscal year."

Explanation:

(This amendment provides \$50,000 from the general fund and \$50,000 from federal Medicaid matching funds to support efforts to publish the complete State Plan for Medical Assistance and amendments and other supporting materials on the Department of Medical Assistances Services' website to be more publicly accessible.)

Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance	(\$742,622)	(\$427,900) GF
Services	\$742,622	\$427,900 NGF

Page 391, after line 24, insert:

"II. The Department of Planning and Budget shall, if the public health emergency is extended into the third or fourth quarters of the second year extending the enhanced federal match pursuant to the federal Families First Coronavirus Response Act, calculate the general fund savings in the Children's Health Insurance Program administrative appropriation and unallot such amount. These savings shall revert to the general fund at the end of the fiscal year."

Explanation:

(This amendment captures \$742,622 from the general fund the first year and \$427,900 from the general fund the second year and a corresponding increase in federal Children's Health Insurance Program (CHIP) matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for fiscal year 2021 and the first and second quarters of state fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022. The CHIP match rate factors in the Medicaid match rate and therefore results in a higher match. Unlike Medicaid the administrative expenditures of the CHIP program uses the same match rate as used for medical spending. Language is also included directing the Department of Planning and Budget to unallot any additional savings that results from an extension of the PHE.)

		Item 317 #6s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0 0.00	(\$1,166,180) (\$6,959,211) -4.00	GF NGF FTE

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$274,927,945". Page 391, strike lines 18 through 24.

Explanation:

(This amendment removes \$1.2 from the general fund and \$7.0 million from nongeneral funds and four positions to fund the fiscal impact of legislation to establish the Virginia Facilitated Enrollment Program. No bill was introduced in the Senate.)

Item 318 #1s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 392, line 36, after "department" strike the remainder of the line and insert: "for up to six months beyond the duration of the Governor's emergency declaration for COVID-19."

Explanation:

(This amendment modifies the extension allowed on conditional licenses for providers licensed by the Department of Behavioral Health and Developmental Services. A conditional license is granted to a provider who has successfully completed the application process but has not yet admitted individuals for service. It is time limited and can only be renewed one time for a total of twelve months. Due to the COVID-19 pandemic conditional licenses could be extended beyond 12 months, until December 31, 2020. This amendment allows an extension for up to six months past the duration of the Governor's emergency declaration.)

		Item 320 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$175,000	GF

Language:

Page 393, line 43, strike "\$110,804,911" and insert "\$110,979,911". Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services (DBHDS), in coordination with the Department of Medical Assistance Services (DMAS), shall contract with a vendor to review all current Medicaid Developmental Disability (DD) waiver provider rates and rate methodologies to ensure an adequate network of quality DD Waiver providers. DBHDS shall submit a rate rebase report with recommendations, including the projected fiscal impact on the Commonwealth, to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021. At a minimum, the report shall include a thorough review and alignment of each service rate methodology to reflect current service definitions and documentation requirements, the impact of minimum wage increases,

appropriate Bureau of Labor Statistics wage percentiles, and appropriate rate differentials for high cost areas of Virginia. All rate assumptions shall be based on the level of need. The vendor shall specifically evaluate the rates for the Supported Living Residential waiver service to ensure appropriate utilization of that service. The department shall actively work with the established DBHDS Provider Issue Resolution Workgroup in the development of its report."

Explanation:

(This amendment adds \$175,000 the second year from the general fund to require the Department of Behavioral Health and Developmental Services in coordination with the Department of Medical Assistance Services to review all current Medicaid Developmental Disability (DD) waiver provider rates and rate methodologies to ensure an adequate network of quality DD Waiver providers. The agency is required to report on its findings by September 1, 2021, to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees.)

Item 320 #2s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services shall continue the Temporary Detention Order Evaluator Workgroup established during the 2020 Session in Senate Bill 768. The workgroup shall report its implementation plan to the Governor, and Chairs of House Health, Welfare, and Institutions Committee, Senate Education and Health Committee, and Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by December 1, 2021."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to continue the Temporary Detention Order Evaluator Workgroup established during the 2020 Session in Senate Bill 768. The workgroup shall report its implementation plan to the Governor, and Chairs of House Health, Welfare, and Institutions Committee, Senate Education and Health Committee, and Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by December 1, 2021.)

Item 320 #3s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services shall establish a workgroup to review the current processes and barriers to sharing relevant patient information between community hospitals and Community Services Boards for shared patients subject to an Emergency Custody Order and under evaluation for a Temporary Detention Order. The department shall report its findings and recommendations to the Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by December 1, 2021.)

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to establish a workgroup to review the current processes and barriers to sharing relevant patient information between community hospitals and Community Services Boards for shared patients subject to an Emergency Custody Order and under evaluation for a Temporary Detention Order.)

Item 320 #4s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language

Page 398, line 17, strike "non-narcotic," and insert "appropriate".

Explanation:

(This amendment eliminates language requiring the use of non-narcotic, long-acting injectable prescription drug treatment regimens used for drug treatment court programs. Instead, it requires the use of appropriate long-acting injectable prescription drug treatment regimens.)

Item 320 #5s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services shall preserve historic microfiche records at Central State Hospital and work with interested partners to digitize such records to be added to the Central State Hospital Digital Library and Archives Project in order to make such information publicly available to researchers or other interested parties."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to preserve historic microfiche records at Central State Hospital and work with interested partners to digitize such records to be added to the Central State Hospital Digital Library and Archives Project.)

		Item 320 #6s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$3,750,000	GF

Language:

Page 393, line 43, strike "\$110,804,911" and insert "\$114,554,911".

Page 398, line 33, unstrike "\$7,500,000" and strike "\$3,750,000".

Page 398, line 35, after "hospital care", insert "or to increase capacity in the community for patients on the Extraordinary Barriers List".

Explanation:

(This amendment provides \$3.8 million the second year from the general fund to fully restore funding for alternative inpatient options to state behavioral health hospital care through the establishment of two-year pilot projects to reduce census pressures on state hospitals. A total of \$7.5 million each year was approved in the 2020 Session, but the funding was unalloted due to the revenue impact from the Coronavirus Pandemic. Half the funding was restored in the Special Session in Chapter 56 and this amendment restores the remaining amount. The amendment also clarifies that pilot projects could also include the option to build community capacity for patients on the Extraordinary Barriers List in order to relieve census pressure.)

		Item 320 #7s
Health and Human Resources	FY20-21	FY21-22
Department of Behavioral Health and Developmental Services	0.00	-5.00 FTE

Language:

Explanation:

(This amendment reduces the position level for the Department of Behavioral Health and Developmental Services for five positions that were added with the funding for administrative costs of STEP-VA. This funding is a restoration of funding previously approved in Chapter 1289 from the 2020 Session. However, the funding for these positions was eliminated in

Chapter 56 in the Special Session I, but the position level was not reduced. Therefore, this amendment corrects the position level for the agency.)

Item 320 #8s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services, in collaboration with the Virginia Treatment Center for Children (VTCC), shall examine and develop strategies to better utilize VTCC in assisting with relief for the census pressures on the Commonwealth Center for Children and Adolescents (CCCA). The strategies to be examined shall include, but are not limited to: (i) diversion strategies when CCCA is near capacity; (ii) increasing the number of Temporary Detention Order admissions; and (iii) operating as a step-down facility from CCCA. The department shall report its finding and recommendations to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees and the the Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by July 1, 2021."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to work with the Virginia Treatment Center for Children (VTCC) to examine and develop strategies that better utilize VTCC in assisting with relief on the census pressures on the Commonwealth Center for Children and Adolescents (CCCA).

		Item 321 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$150,000	GF

Language:

Page 399, line 42, strike "\$61,270,529" and insert "\$61,420,529".

Page 401, unstrike lines 5 through 8.

Page 401, line 5, strike "\$150,000 the first year and".

Explanation:

(This amendment provides \$150,000 the second year from the general fund to provide for the transportation costs of patients discharged from state hospitals that were admitted under a

Temporary Detention Order (TDO). Oftentimes individuals under a TDO are transported to a state facility that is hours away from the individual's home location and therefore upon discharge may have difficulty getting transportation back to their home location. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

Item 321 #2s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 401, line 24, after "services.", insert:

"The funding shall only be provided to members that are accredited by the Council on Accreditation of Peer Recovery Support Services (CAPRSS)."

Explanation:

(This amendment modifies the criteria to receive funds from a passthrough grant to the Virginia Association of Recovery Residencies such that the funds can only be provided to members that are accredited by the Council on Accreditation of Peer Recovery Support Services (CAPRSS).)

		Item 321 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$143,260	GF

Language:

Page 399, line 42, strike "\$61,270,529" and insert "\$61,413,789".

Explanation:

(This amendment restores \$143,260 from the general fund the second year to expand the Adverse Childhood Experiences (ACE) initiative. It supports a full-time Central Office position to: (i) provide oversight over 100 ACE Interface Master Trainers across the Commonwealth; (ii) plan and develop additional ACE Interface Master Trainer Cohorts in their region; and (iii) plan and facilitate monthly learning community meetings for each training cohort; etc. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

Item 321 #4s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 400, line 11, after "services.". insert:

"The Department of Behavioral Health and Developmental Services shall report quarterly on the the distributions of this funding to Community Services Boards (CSB). The report shall include the amounts distributed to each CSB, the total annual allocation of funding for each CSB, the current available funding balance for each CSB, and any reallocations of funding between CSBs. The report shall be submitted to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than 20 days after the end of each fiscal quarter."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to report quarterly on the distribution of funding provided to Community Services Boards (CSB) for the purchase of acute inpatient or community-based psychiatric services (known as LIPOS).)

Item 322 #1s

Health and Human Resources

Grants to Localities Language

Language:

Page 405, line 35, after "regimens.", insert:

"In expending any amount, the department shall prioritize allocation of the funding to any portion of medication assisted treatment that are not covered by insurance."

Page 405, line 36, strike "non-narcotic, non-addictive," and insert "appropriate".

Page 405, line 37, after "probation;" insert "or".

Page 405, line 38, after "jail", strike "; or (iii)", and insert:

". The department shall ensure that a portion of the funding is used for appropriate prescription drug treatment regimens for individuals who are".

Explanation:

(This amendment modifies language in the introduced budget related to medication assisted treatment for individuals who are addicted to opioids. Language changes require that a portion of the funding be used for appropriate prescription drug treatment regimens, instead of solely non-narcotic, long-acting injectable prescription drug treatment regimens. This change allows the service provider to prescribe the most effective treatment option for the individual obtaining

treatment services. The amendment also clarifies that funding should be prioritized to those who are not otherwise insured.)

		Item 322 #2s	
Health and Human Resources	FY20-21	FY21-22	
Grants to Localities	\$0	\$2,100,800	GF

Language:

Page 402, line 8, strike "\$554,715,057" and insert "\$556,815,857".

Page 405, line 53, after "first year and", unstrike "\$3,700,800".

Page 406, line 1, strike "\$1,600,000".

Explanation:

(This amendment restores \$2.1 million the second year from the general fund to expand forensic discharge planning services at three additional jails with a high percentage of inmates with serious mental illness. The General Assembly provided \$1.6 million from the general fund in fiscal year 2020 for discharge planning services at two local jails. Discharge planning includes linking inmates with serious mental illness to community providers for treatment and housing and other needed services as they transition from jails to the community. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

		Item 326 #1s	
Health and Human Resources	FY20-21	FY21-22	
Mental Health Treatment Centers	(\$2,142,601) \$2,142,601	(\$4,285,202) \$4,285,202	GF NGF

Language:

Page 408, line 27,, strike "the general fund" and insert "federal funds".

Page 408, line 28, after "facilities." insert:

"The department shall coordinate with the Virginia Department of Health (VDH) and local health districts as appropriate to coordinate its testing and surveillance activities in order to access federal ELC Enhancing Detection Expansion grant funding provided to VDH through the Centers for Disease Control."

Explanation:

(This amendment supplants \$2.1 million the first year and \$4.3 million the second year from the general fund with federal funds from the ELC Enhancing Detection Expansion grant funding

provided to the Virginia Department of Health (VDH) through the Centers for Disease Control. The federal Consolidated Appropriations Act, 2021 that was signed into law on December 27, 2020 and provides substantial federal funding to states for COVID-19 testing. The estimate for Virginia is about \$490 million. This amendment directs the Department of Behavioral Health and Developmental Services to coordinate its COVID-19 surveillance activities for state facilities with VDH in order to access those federal funds.)

		Item 326 #2s	
Health and Human Resources	FY20-21	FY21-22	
Mental Health Treatment Centers	\$0	\$765,428	GF

Language:

Page 408, line 2, strike "\$286,799,776" and insert "\$287,565,204".

Explanation:

(This amendment restores \$765,428 the second year from the general fund to provide critical clinical staffing at the Commonwealth Center for Children and Adolescents. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

		Item 339 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department for Aging and Rehabilitative Services	\$0	\$1,000,000	GF

Language:

Page 411, line 47, strike "\$100,388,245" and insert "\$101,388,245". Page 413, line 15, after "first year and", unstrike "\$6,976,719."

Page 413, line 16, strike "\$5,976,719".

Explanation:

(This amendment provides \$1.0 million from the general fund the second year to increase contracts with community based brain injury service providers to provide for cost increases to bring salaries in line with market demands in order to ensure staff retention, address existing case management waiting lists, enhance virtual programming, expand partnerships, and development responses to the high-intensity needs of Virginians with brain injury, especially those related to COVID-19 and the opioid epidemic.)

Item 339 #2s

Health and Human Resources	FY20-21	FY21-22
Department for Aging and Rehabilitative Services	\$0	\$425,000 GF

Language:

Page 411, line 47, strike "\$100,388,245" and insert "\$100,813,245".

Page 413, line 6, unstrike the second "\$5,521,858".

Page 413, line 6, strike the second "\$5,096,858".

Explanation:

(This amendment adds \$425,000 the second year from the general fund to support 17 Centers for Independent Living that provide independent living services including independent living skills training, advocacy, information and referral, peer mentoring, and transition services to people with significant disabilities. Transition services include youth transition services, services for individuals trying to transition from nursing facilities and other institutions, and services to prevent institutionalization.)

		Item 340 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department for Aging and Rehabilitative Services	\$0	\$150,000	GF

Language:

Page 414, line 9, strike "\$36,139,218" and insert "\$36,289,218".

Page 415, unstrike lines 28 through 35.

Page 415, line 28, strike "\$150,000 the first year and".

Page 415, line 35, after "year." insert:

"The services provided through this program shall be coordinated with the Department of Behavioral Health and Developmental Services to help address the inappropriate utilization of state psychiatric beds by patients with dementia."

Explanation:

(This amendment restores funding the second year from the general fund budget to provide an interdisciplinary plan of care and dementia care management for 50 Virginia residents diagnosed with dementia. The service would be provided through a partnership with the memory and aging care clinic at the University of Virginia and the Alzheimer's Association. Funding of \$150,000 each year of the 2020-22 biennium was provided in Chapter 1289, 2020 Acts of Assembly to serve 50 individuals diagnosed with dementia. That funding was unallotted

and subsequently eliminated in Chapter 56, 2020 Special Session I Acts of Assembly. This amendment would restore the funding in fiscal year 2022 enabling the program to serve 50 individuals with dementia.)

		Item 350 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0	\$1,400,000 \$15,928,859	GF NGF

Language:

Page 420, line 44, strike "\$152,429,363" and insert "\$169,758,222".

Page 423, line 4, after "O." insert "1."

Page 423, line 6, strike "2020." and insert:

"2021 and 18 percent effective July 1, 2022."

Page 423, after line 6, insert:

"2. The Department of Social Services shall develop a plan to increase the standards of assistance by 18 percent annually until they equal 50 percent of the federal poverty level."

Explanation:

(This amendment adds \$1.4 million from the general fund and \$15.9 million from the Temporary Assistance to Needy Families (TANF) block grant the second year to increase the standards of assistance by 18 percent beginning in fiscal year 2022. Language requires the Department of Social Services to develop a plan to increase the standards of assistance by 18 percent annually until the standards equal 50 percent of the federal poverty level, which is estimated to take four years. TANF eligibility is tied to the monthly cash assistance value. In 1985, a three person family in the City of Richmond could be eligible with net income below 48 percent of the federal poverty level. To qualify now, the family's net income must be less than 30 percent of the federal poverty level.)

Item 350 #2s

Health and Human Resources

Department of Social Services

Language

Language:

Page 424, after line 6, insert:

"W. The Department of Social Services shall develop demonstration projects with a goal of assisting families to earn a living wage and escape poverty. The projects shall include a benefit cliff pilot, a pilot to demonstrate the role of location as a key determinant for health and economic success, and a pilot program to determine the impact of cashing out key program

supports. The Department of Social Services shall determine the authority needed to implement the demonstration projects as outlined in the Social Security Act, the Food and Agriculture Act of 1977, and other federal legislation that provides for the development and testing of demonstration projects. The department shall identify federal and private grants that are available to help fund these demonstration projects. The department shall prepare a report on each proposed demonstration project that describes the project, identifies a funding amount needed, and indicates whether federal funds or other private funding is available to help implement such projects and shall submit the report to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021."

Explanation:

(This amendment adds language directing the Department of Social Services to develop several anti-poverty demonstration projects for consideration by the General Assembly.)

		Item 350 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$2,120,420	NGF

Language:

Page 420, line 44, strike "\$152,429,363" and insert "\$154,549,783".

Page 424, after line 6, insert:

"W. Out of this appropriation, \$2,120,420 the second year from the Temporary Assistance to Needy Families (TANF) block grant shall be provided for the Department of Social Services to implement a program so that TANF-eligible individuals may save funds in an individual development account established for the purposes of home purchase, education, starting a business, transportation, or self-sufficiency. The TANF funds shall be deposited to the individual development accounts at a match rate determined by the department."

Explanation:

(This amendment adds \$2.1 million the second year from the Temporary Assistance to Needy Families block grant for the implementation of individual development accounts for TANF recipients by the Department of Social Services.)

		Item 350 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$25,000	GF

Language:

Page 420, line 45, strike "\$152,429,363" and insert "\$152,454,363".

Page 424, after line 6, insert:

"W. Out of this appropriation, \$25,000 from the general fund the second year shall be deposited to the Virginia Digital Equity Pilot Program Fund."

Explanation:

(This amendment provides \$25,000 from the general fund the second year to fund Senate Bill 1462 which establishes a broadband pilot project in order to augment federal assistance funding (FCC Lifeline) for households currently participating in the Supplemental Nutrition Assistance Program. This funding would serve between 150 to 300 households with commensurate administrative funding to start the program. Funds of \$5.75 per household augments the \$9.25 federal assistance to equal the cost to afford low cost internet access of \$15 a month).

		Item 350 #5s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0	(\$1,000,000) (\$1,436,986)	GF NGF

Language:

Page 420, line 45, strike "\$152,429,363" and insert "\$149,992,377".

Explanation:

(This amendment reduces \$1.0 million from the general fund and \$1.4 million from federal funds the second year for the Supplemental Nutrition Employment and Training Program. This amendment reduces that additional funding by about half, thereby reducing the number of localities to which the program would expand.)

		Item 351 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0	\$2,150,048 \$2,175,528	GF NGF

Language:

Page 424, line 9, strike "\$525,386,732" and insert "\$529,712,308".

Page 425, unstrike lines 7 through 11.

Page 425, line 8, strike "each" and insert "the second".

Page 425, line 9, strike "2021" and insert "2022".

Explanation:

(This amendment provides \$2.2 million from the general fund and \$2.2 million from nongeneral

funds the second year to increase the minimum pay band for local departments of social services positions in fiscal year 2022. The pay band minimum would increase by 20 percent for family services positions and 15 percent for all other benefit program services positions, self sufficiency services positions, and administration positions that are currently below the new minimum threshold.)

		Item 351 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$84,004	GF
•	\$0	\$51,203	NGF

Language:

Page 424, line 9, strike "\$525,386,732" and insert "\$525,521,939".

Explanation:

(This amendment provides \$84,004 from the general fund and \$51,203 from nongeneral funds the second year to fund the fiscal impact on local departments of socials services' workload associated with the Senate Bill 1321, which expands the stepparent adoption provisions to allow a person who is not the child's stepparent but has a legitimate interest in the child to file a joint petition for adoption with the child's birth parent or parent by adoption.)

		Item 351 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0 0.00	(\$376,910) (\$2,709,766) -2.00	GF NGF FTE

Language:

Page 424, line 9, strike "\$525,386,732" and insert "\$522,300,056".

Explanation:

(This amendment removes \$376,910 from the general fund and \$2.5 million from nongeneral funds that would have funded the fiscal impact of legislation to establish the Virginia Facilitated Enrollment Program. No bill was introduced in the Senate.)

		Item 353 #1s	
Health and Human Resources	FY20-21	FY21-22	

Department of Social Services \$0 \$4,400,000 GF

Language:

Page 426, line 8, strike "\$41,177,762" and insert "\$45,577,762".

Page 426, line 23, after "\$1,420 per month", insert "and effective July 1, 2021, a rate of \$1,704 per month".

Explanation:

(This amendment provides \$4.4 million from the general fund the second year to increase the Auxiliary Grant (AG) rate, a state supplement that provides maintenance and care to aged, blind, and disabled adults residing in assisted living facilities (ALF), adult foster care homes, or supportive housing settings, by 20 percent on July 1, 2021.)

		Item 353 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$750,000	GF

Language:

Page 426, line 8, strike "\$41,177,762" and insert "\$41,927,762".

Page 427, after line 6, insert:

"G. Out of this appropriation, \$750,000 the second year from the general fund shall be transferred to the Virginia Sexual and Domestic Violence Prevention Fund."

Explanation:

(This amendment provides \$750,000 the second year from the general fund for the Virginia Sexual and Domestic Violence Prevention Fund that was created pursuant to Senate Bill 297 in the 2020 Session. The program would be administered by the Department of Social Services and the Department of Health. The fund will award grants on a competitive basis to local sexual and domestic violence agencies engaged in evidence-informed sexual and domestic violence prevention work.)

		Item 354 #1s
Health and Human Resources	FY20-21	FY21-22
Department of Social Services	\$0 \$0	(\$260,406) GF (\$99,594) NGF

Language:

Page 427, line 8, strike "\$261,995,376" and insert "\$261,635,376".

Explanation:

(This amendment reduces \$260,406 from the general fund and \$99,594 from nongeneral funds the second year for the fiscal impact of Senate Bill 1338, which establishes the State-Funded Kinship Guardianship Assistance Program (State KinGAP), which allows payments to be made to relatives, including fictive kin, who receive custody of a child. A corresponding amendment in the Children's Services Act reflects the increase in maintenance payments from the State KinGAP program while this amendment reflects the decrease in Title IV-E program costs as some children will transition to the State-Funded Kinship Guardship Assistance Program, which has lower maintenance payments than the Title IV-E program.)

Item 354 #2s

Health and Human Resources

Department of Social Services

Language

Language:

Page 431, after line 23, insert:

"X. The Department of Social Services shall create a diversion program supporting relative and fictive kin families who have received temporary physical and legal custody from the court that makes use of all federal and state monies available to provide a payment to relative and fictive kin families who have temporary custody through a court order. The department shall report on the program to the Chairs of the House Health, Welfare and Institutions, House Appropriations, Senate Rehabilitation and Social Services, and Senate Finance and Appropriations Committees by December 1, 2021, with the steps needed to implement the diversion program."

Explanation:

(This amendment adds language directing the Department of Social Services to create a diversion program supporting relative and fictive kin families who receive custody of a child from the court and report the steps to implement such program to the House Health, Welfare and Institutions, House Appropriations, Senate Rehabilitation and Social Services, and Senate Finance and Appropriations Committees by December 1, 2021.)

Item 354 #3s

Health and Human Resources FY20-21 FY21-22

Department of Social Services \$0 \$3,500,000 NGF

Language:

Page 427, line 8, strike "\$261,995,376" and insert "\$265,495,376". Page 431, after line 23, insert:

"X. The Department of Social Services shall extend payments to children aging out of the Fostering Futures program past age 21 through September 30, 2021."

Explanation:

(This amendment adds language to extend payments to children aging out of the Fostering Futures program through September 30, 2021. The recently passed federal Coronavirus Relief and Response Supplemental Appropriations Act has extended the John H. Chafee Foster Care Program for Successful Transition to Adulthood funding through September 2021. These funds can be used to support children aging out of services during the COVID-19 pandemic.)

		Item 354 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$2,923,178) \$2,923,178	(\$5,846,356) \$5,846,356	GF NGF

Language:

Explanation:

(This amendment captures \$2.9 million the first year and \$5.8 million the second year from the general fund and includes an offsetting amount of federal Title IV-E foster care funding to reflect a higher federal match rate. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared COVID-19 public health emergency (PHE) expires. The federal Title IV-E programs used the Medicaid match rate to determine the state and federal share of the program. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022.)

		Item 356 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$500,000	NGF

Language:

Page 431, line 35, strike "\$60,957,967" and insert "\$61,457,967".

Page 432, line 45, after "first year and", strike "\$1,500,000", and insert "\$2,000,000".

Explanation:

(This amendment provides \$500,000 from the Temporary Assistance to Needy Families block grant funds the second year to Northern Virginia Family Services. The additional funds are for the specific purpose of providing services and wrap-around supports to Virginians suffering impacts from COVID-19 in all northern Virginia counties, cities and towns. Services and supports include but are not limited to: the provision of food, financial assistance to prevent homelessness, access to health and mental health care, childcare and workforce development programs.)

		Item 356 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$500,000	\$0	GF

Language:

Page 431, line 34, strike "\$61,857,967" and insert "\$62,357,967".

Page 435, unstrike lines 40 through 42.

Page 435, line, 42, after "support.", insert:

Explanation:

(This amendment provides \$500,000 the first year from the general fund for the Laurel Center to support the rehabilitation of an existing freight station building located adjacent to the emergency shelter. This building will house the Empowerment Program and provide a safe and convenient location for job readiness and skills training for survivors of domestic violence.)

		Item 356 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$200,000	GF

Language:

Page 431, line 34, strike "\$60,957,967" and insert "\$61,157,967".

Page 431, after line 32, insert:

"Out of this appropriation, \$200,000 the second year from the general fund is provided for the Department of Social Services to increase interpretation and translation services to help immigrants in Virginia access local resources through 2-1-1, including healthcare, housing, and other social services."

Explanation:

[&]quot;Any unexpended balance as of June 30, 2021, shall not revert to the general fund but shall be reappropriated in FY 2022."

(This amendment provides \$200,000 the second year from the general fund for the Department of Social Services to increase interpretation and translation services to help immigrants in Virginia access local resources through 2-1-1, including healthcare, housing, and other social services.)

		Item 356 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$250,000)	\$0	GF

Language:

Page 431, line 35, strike "\$61,857,967" and insert "\$61,607,967". Page 436, strike lines 10 through 11.

Explanation:

(This amendment removes \$250,000 from the general fund the first year for a new passthrough grant provided in the introduced budget for Children's Harbor, a child care provider in Hampton Roads, to expand services on the Eastern Shore.)

		Item 357 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$45,653) (\$103,404)	(\$182,610) \$0	GF NGF

Language:

Page 436, line 13, strike "\$43,640,542" and insert "\$43,491,485". Page 436, line 13, strike "\$16,494,086" and insert "\$16,311,476".

Explanation:

(This amendment reduces \$45,656 from the general fund and \$103,404 from nongeneral funds the first year and \$182,610 from the general fund the second year for the purchase of personal protective equipment (PPE) for licensing inspectors related to COVD-19. The Virginia Department of Emergency Management should be able to secure PPE for the Department of Social Services' licensing inspectors based on the substantial funding that agency has received to purchase PPE.)

Item 359 #1s

Health and Human Resources

Department of Social Services

Language

Language:

Page 440, after line 43, insert:

"K. The Department of Social Services as administrator of the federal Community Services Block Grant shall establish an interagency working group to develop recommendations for implementation of local criminal justice diversion programs. Each diversion program should offer standards for providing persons charged with lower-level offenses alternatives to arrest, conviction or incarceration for lower-level offenses. The scope of these programs shall not include behavioral health issues as those priorities are being addressed elsewhere. The working group should include the appropriate offices and agencies of Health and Human Resources, Commerce and Trade, Public Safety and Homeland Security and the Governor's Chief Diversity, Equity and Inclusion Officer. The interagency working group shall work with community action agencies, local governments including local law enforcement, representatives of the judicial system, civil rights organizations as well as other stakeholders to develop locallybased solutions. The recommendations shall provide for two-generation whole family strategies that deal with meeting the needs of the potential offender and his or her entire family by addressing issues related to poverty, including homelessness. The Department of Social Services shall submit its recommendations to the Chairs of the House Appropriations Committee and the Senate Finance and Appropriations Committee no later than September 30, 2021."

Explanation:

(This amendment establishes an interagency working group to develop recommendations for local criminal justice diversion programs to provide alternatives to arrest, conviction or incarceration for lower-level offenses.)

		Item 359 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$923,804) (\$1,224,577)	(\$923,804) (\$1,224,577)	GF NGF

Language:

Page 438, line 39, strike "\$121,912,263" and insert "\$119,763,882". Page 438, line 39, strike "\$118,755,668" and insert "\$116,607,287".

Explanation:

(This amendment removes \$923,804 from the general fund and \$1.2 million nongeneral funds each year included in the introduced budget for Virginia Case Management System non-Medicaid modifications.)

Item 373 #1s

Natural Resources FY20-21 FY21-22

Department of Conservation and \$0 \$170,000 GF Recreation

Language:

Page 447, line 9, strike "\$84,750,087" and insert "\$84,920,087".

Explanation:

(This amendment restores \$170,000 GF the second year that was unalloted in Chapter 1289 for the environmental education experiences identified in Item 373, Paragraph K. This request recognizes the commitment Virginia made to its students and the importance environmental education plays in developing and nurturing watershed stewards. These funds provide for meaningful interactions and learning opportunities for students throughout the Chesapeake Bay watershed.)

Item 373 #2s

Natural Resources

Department of Conservation and Recreation

Language

Language:

Page 451, following line 36, insert:

"R. The Department of Conservation and Recreation and the Virginia Department of Agriculture and Consumer Services shall convene a workgroup including the Department of Forestry, the Virginia Department of Transportation, the Department of Wildlife Resources, the Virginia Native Plant Society, the Virginia Association of Counties, the Virginia Municipal League, Blue Ridge PRISM, the Audubon Society of Northern Virginia, the Virginia Nursery and Landscape Association, the Virginia Agribusiness Council, the Virginia Farm Bureau Federation, and a representative of the Virginia Cooperative Extension Program with expertise in crop and weed sciences to assess the sale and use of invasive plant species in the retail, landscape, greenhouse, and nursery industries and consider measures to reduce or eliminate the sale and use of invasive plant species in the Commonwealth and promote the sale and use of native plants. In conducting this assessment, the workgroup shall examine measures to reduce, mitigate, and eliminate the continued sale and use of invasive species as identified in the list of Virginia invasive plant species maintained by the Department of Conservation and Recreation. The workgroup shall evaluate measures including: (i) labeling plants as invasive plant species at the point of sale; (ii) taxing the sale of invasive plant species and applying revenues to the removal of invasive plant species or the restoration of sites for native habitat; (iii) adding

invasive plant species currently being offered for sale to the list of plants declared to be noxious weeds by the Board of Agriculture and Consumer Services through regulations adopted pursuant to Chapter 8 (§ 3.2-800 et seq.) of Title 3.2 of the Code of Virginia (the Noxious Weed List); (iv) supporting education and outreach, including state partnerships with nonprofit organizations dedicated to the preservation of Virginia's natural heritage, regarding the reduction of the use of invasive plant species and the promotion of the use of noninvasive or native plant species as substitutes; and (v) introducing measures to increase the use of native plants on properties and projects owned by localities or the Commonwealth. The workgroup shall submit its assessment, including any recommendations regarding statutory changes and changes to regulations or guidance documents adopted by relevant agencies, including changes related to the placement of plant species on the Noxious Weed Lists, to the Governor and the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations no later than December 1, 2021."

Explanation:

(This amendment directs the creation of a stakeholder workgroup to assess the sale and use of invasive plant species in the retail, landscape, greenhouse, and nursery industries and consider measures to reduce or eliminate the sale and use of invasive plant species in the Commonwealth and promote the sale and use of native plants.)

		Item 374 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$0	\$1,511,600	GF

Language:

Page 451, line 38, strike "\$81,225,147" and insert "\$82,736,747".

Page 454, following line 29, insert:

Explanation:

(This amendment provides \$1.5 million GF in the second year for the costs of connecting Mason Neck State Park to the local municipal drinking water system.)

		Item 374 #2s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$1,412,000	\$0	GF

[&]quot;S. Included in the amounts for this Item is \$1,511,600 the second year from the general fund to connect Mason Neck State Park to a public drinking water supply system."

Language:

Page 451, line 38, strike "\$72,102,316" and insert "\$73,514,316".

Page 452, following line 36, insert:

"4. In addition to the amounts provided in paragraph C.1., the Department is authorized to provide \$1,412,000 in the first year from the general fund for the modernization of the Rhododendron Restaurant and lodge unit repairs."

Explanation:

(This amendment provides \$1.4 million GF in the first year to modernize and repair lodge units and the Rhododendron Restaurant. This investment will support a key component of the Breaks Interstate Park's five-year strategic plan, make the park more sustainable from a maintenance perspective, and significantly boost overnight visitation.)

		Item 374 #3s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$0	(\$3,500,000)	GF

Language:

Page 451, line 38, strike "\$81,225,147" and insert "\$77,725,147".

Page 452, line 6, following "Virginia." insert:

"Out of the amounts in the second year, and pursuant to the provisions of § 4-13.00 of this Act, the Virginia Land Conservation Foundation shall reserve an amount up to \$5,000,000 for the identification and evaluation of a project focused on the preservation of tribal lands of the Chickahominy Tribe and for the conservation and preservation of River Farm in Alexandria, Virginia. Consideration of these projects shall be consistent with the policies and practices of the Foundation pursuant to § 10.1-1020, Code of Virginia."

Page 454, strike line 17 through line 18.

Explanation:

(This amendment provides for the consideration of one-time funding of up to \$5.0 million for the identification, acquisition of preservation of tribal lands for the Chickahominy Tribe and for the conservation and preservation of River Farm.)

		Item 374 #4s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$0	(\$4,000,000)	GF

Language:

Page 451, line 38, strike "\$81,225,147" and insert "\$77,225,147".

Page 454, line 19, strike "\$5,000,000" and insert "\$1,000,000",

Page 454, line 21, following "Cemetery" insert "."

Page 454 line 21, strike "and creation of".

Page 454, line 22, strike "the Harmony Living Shoreline Memorial".

Page 454, line 22, strike "all aspects of".

Page 454, line 23, strike "the project to include but not limited to".

Page 454, line 25, following "Farm", strike "," and insert "and".
Page 454, line 27, following "located", strike "," insert "." and strike the remainder of the line.

Page 454, strike line 28 through line 29.

Explanation:

(This amendment limits the scope of Project Harmony to the removal and relocation of the Columbian Harmony tombstones from Virginia to the New Harmony Cemetery in Landover, Maryland.)

		Item 374 #5s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$740,000	\$0	GF

Language:

Page 451, line 38, strike "\$72,102,316" and insert "\$72,842,316".

Page 451, line 49, strike "\$35,624,459" and insert "\$36,364,459".

Page 454, unstrike line 4 through line 6.

Explanation:

(This amendment restores \$740,000 GF in the first year as a source of matching funds for the construction of Riverfront Park in the City of Danville. This funding was approved in the 2020 Session and subsequently unallotted.)

		Item 377 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Environmental Quality	\$0 0.00	\$115,000 1.00	GF FTE

Language:

Page 455, line 31, strike "\$52,900,649" and insert "\$53,015,649".

Explanation:

(This amendment provides \$115,000 the second year from the general fund and one position for the Department of Environmental Quality to administer permit-by-rule applications for energy storage projects pursuant to the provisions of SB 1207 of the 2021 General Assembly.)

		Item 377 #2s	
Natural Resources	FY20-21	FY21-22	
Department of Environmental Quality	\$0 0.00	\$230,000 2.00	GF FTE

Language:

Page 455, line 31, strike "\$52,900,649" and insert "\$53,130,649".

Page 457, following line 41, insert:

"M. The Director of the Department of Environmental Quality shall convene a working group for the purpose of developing an annual or project-based fee schedule for the the review of erosion and sediment control plans related to solar energy project applications. The working group shall include representatives of (i) private sector companies that own or operate solar energy facilities, (ii) local governments that permit solar facilities, and (iii) other stakeholders determined by the Department to be necessary to the development of the fee schedule."

Explanation:

(This amendment provides the funding for staffing dedicated to the erosion and sediment control program to review for solar project permitting and provide this service to projects upon the request of a locality pursuant to the provisions of SB 1258 of the 2021 General Assembly.)

Item 377 #3s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 457, following line 41, insert:

"M. The Department of Environmental Quality, in consultation with the Department of Agriculture and Consumer Services and the Department of Forestry, shall establish a workgroup to review the practice of retiring agricultural land for the generation of nutrient credits and determine its impact on agricultural sustainability, farmland retention, farmland preservation, and functions of the nutrient credit exchange in the Virginia portion of the Chesapeake Bay watershed and its subwatersheds. If it is determined that there is impact on farmland retention/availability, the report should include recommendations regarding how the nutrient

credit trading regulations and/or underlying statutory authority should be changed to help reduce the loss of prime farmland. If the land for nutrient credits is converted to forestland, the workgroup should identify what protections are in the nutrient credit trading regulations to ensure the forestland is managed under a forestry management plan and/or noxious weed or invasive species are controlled. The review shall be completed and provided to the Chairs of the House Committee on Agriculture, Chesapeake and Natural Resources, the Senate Committee on Agriculture, Conservation, and Natural Resources and the Virginia delegation of the Chesapeake Bay Commission by December 1, 2021. The workgroup shall include representatives of the Virginia Agribusiness Council, Virginia Farm Bureau, the Chesapeake Bay Commission, Virginia Cooperative Extension, the Virginia Department of Transportation, Home Builders Association of Virginia, Virginia Association for Commercial Real Estate, representatives from local Soil and Water Conservation Districts, representatives of local governments, local economic development officials, and other stakeholders deemed appropriate by the Department."

Explanation:

(This amendment directs the creation of a multi-agency workgroup to review the practice of retiring agricultural land for the generation of nutrient credits and determine its impact on agricultural sustainability, farmland retention, farmland preservation, and functions of the nutrient credit exchange in the Virginia portion of the Chesapeake Bay watershed and its subwatersheds.)

Item 377 #4s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 457, following line 41, insert:

"M. The Department of Environmental Quality shall convene a workgroup, in conjunction with the Department of Health and the Department of Agriculture and Consumer Services, to conduct research and complete a single collaborative report that provides findings and recommendations related to: (i) the location, frequency, and severity of harmful algae blooms in Virginia waters; (ii) the factors that lead to the formation and occurrence of harmful algae blooms; and, (iii) plans and strategies for state agencies to lead or support appropriate mitigation efforts. The workgroup shall provide its findings to the Chairs of the House Agriculture, Chesapeake and Natural Resources Committee and Senate Agriculture, Conservation and Natural Resources Committee no later than September 1, 2021."

Explanation:

(This amendment directs DEQ, VDACS, and VDH to convene a joint workgroup to provide a report on the prevalence of harmful algae blooms in Virginia waters and strategies to address

their occurrence.)

Item 377 #5s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 457, unstrike line 20 through line 29.

Page 457, line 30, unstrike "Virginia." and strike the remainder of the line.

Page 457, strike line 31 through line 41.

Explanation:

(This amendment reverses the proposed water quality enhancement fee included in SB 1100, as introduced.)

Item 378 #1s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 458, line 31, following "Department." strike the remainder of the line.

Page 458, strike line 32 through line 37.

Explanation:

(This amendment removes a proposed modification to the language agreed to by the General Assembly regarding the regulation of hydrofluorocarbons in existing manufacturing processes in Virginia.)

Item 382 #1s

Natural Resources

Department of Game and Inland Fisheries

Language

Language:

Page 461, following line 25, insert:

"Notwithstanding § 29.1-113, Code of Virginia, access permit fees at boat ramps owned or managed by the Department of Wildlife Resources shall not be assessed prior to July 1, 2022, pending a study by the Department on the costs and benefits of such fees and the impact of said

fees on recreational users in Virginia. As part of this study, the Department shall convene a stakeholder group for the purpose of developing and providing recommendations on access permit fees, various alternatives, and other issues related to the use and maintenance of Department-owned boat ramp facilities. The stakeholder work group shall be composed of representatives of registered boat owners, paddlecraft liveries, outdoor outfitters, and other non-registered vessel recreational users of such boat ramps, or other affected parties the Department deems necessary. The Department shall report the study findings, conclusions and recommendations to the Governor and Chairs of the House Appropriations and Agriculture, Chesapeake and Natural Resources Committees and the Chairs of the Senate Finance and Appropriations and Agriculture, Conservation and Natural Resources Committees by October 1, 2021."

Explanation:

(This amendment delays by one fiscal year the authority of the Department of Wildlife Resources to assess fees for access to boat ramps it owns or manages.)

Item 383 #1s

Natural Resources

Department of Game and Inland Fisheries

Language

Language:

Page 461, strike line 35 through line 44, and renumber accordingly.

Explanation:

(This amendment removes an obsolete requirement that the Department of Wildlife Resources notify specific Standing Committee Chairs prior to consolidation of regional offices or reorganization of divisional responsibilities.)

Item 383 #2s

Natural Resources

Department of Game and Inland Fisheries

Language

Language:

Page 462, following line 22, insert:

"F. The Directors of the Departments of Wildlife Resources and Conservation and Recreation shall assess the feasibility of developing the Rapidan Wildlife Management Area into a State Park and provide a copy of its assessment to the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations no later than November 1, 2021. This assessment shall include, but not be limited to, the impact on wildlife

currently within the management area; any restrictions of deeds, easements, covenants or grant funding used in the initial acquisition of the wildlife management area; capital costs for developing recreational access and overnight accommodations; ongoing operational costs of the proposed facility; and an anticipated timeline for phased access to public recreational facilities within the existing master planning process."

Explanation:

(This amendment requires the DWR and DCR to assess the feasibility of converting the existing Rapidan Wildlife Management Area into a State Park.)

		Item 385 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Historic Resources	\$0 0.00	\$250,000 1.00	GF FTE

Language:

Page 462, line 43, strike "\$8,020,283" and insert "\$8,270,283".

Page 466, following line 12, insert:

"W. The Department of Historic Resources is authorized to enter into an agreement with one or more Virginia-based Historically Black Colleges and Universities to provide paid internships to enrolled students for data collection and outreach activities to expand Virginia's historical property catalogue to include underrepresented African American and indigenous communities. Included within the amounts in this Item, \$100,000 the second year from the general fund is provided for an initial cohort group in Fiscal Year 2022."

Explanation:

(This amendment provides \$250,000 to the Department of Historic Resources for the expansion of Virginia's historical property catalogue to include underrepresented African American and indigenous communities. Included in these amounts is \$110,000 to support one new FTE, \$100,000 for a grant program to provide paid internships in partnership with Virginia's HBCU's to conduct fieldwork, and \$40,000 for cultural data enrichments and database enhancements for the Virginia Cultural Resources Information System.)

		Item 386 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Historic Resources	\$139,328	\$139,328	GF

Language:

Page 466, line 14, strike "\$973,912" and insert "\$1,113,240".

Page 466, line 14, strike "\$973,912" and insert "\$1,113,240".

Page 466, strike line 22 through line 23, and insert:

"As a condition of receipt of funding in this Item, and consistent with the provisions of § 4-13.00 of this Act, it is the intent of the General Assembly that all funding designated in Item 385 shall be provided by the department as expressly stated therein."

Explanation:

(This amendment restores operational and administrative support at the Department of Historic Resources that was approved during the 2020 Session, but subsequently unallotted. In addition the language requires the Department to comply with the explicit intent of the General Assembly, and in a manner consistent with all current laws, in administering the funding designated in Item 385. Included within this amount is funding for in-band compensation adjustments for a variety of professional staff as recommended by the Department of Human Resources.)

Item 391 #1s

Public Safety and Homeland Security

Secretary of Public Safety and Homeland Security

Language

Language:

Page 471, after line 13, insert:

"G. The Secretary, in consultation with the Department of Planning and Budget, and the Secretary of Finance, as well as appropriate public safety or other agency staff, shall evaluate existing funding that has been previously authorized for the enforcement of laws related to controlled substance prohibition. The Secretary shall identify, for controlled substances which have recently been decriminalized or legalized, sources of funding that are authorized for enforcement activities, including funding dedicated to patrol, arrests, incarceration, training, or other activities, that may be saved and reallocated towards other programs. The Secretary shall report on the information required in this paragraph to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by December 1, 2021."

Explanation:

(This amendment directs the Secretary of Public Safety and Homeland Security to evaluate available base funding in the budget in order to identify savings pursuant to the reduction or elimination of penalties for the possession of controlled substances that may be reallocated to drug treatment and rehabilitation programs.)

Item 391 #2s

Public Safety and Homeland Security

Secretary of Public Safety and Homeland Security

Language

Language:

Page 471, after line 13, insert:

"G. The Secretary of Public Safety and Homeland Security shall, as part of the workgroup created pursuant to Senate Bill 1339, include a comprehensive review of all systems and processes necessary for the expungement or sealing of police or court records. The review shall examine improvements in technology, systems improvements, and automation of manual processes that would improve the cost-effectiveness of expungements and sealing of records, including consideration of the technology, systems and processes utilized by other states, in order to accommodate the requirements of the legislation passed by the General Assembly. The Secretary shall also report on the projected costs of changes to systems and processes for consideration in the 2022 General Assembly Session. The Secretary shall report on the workgroup's findings and recommendations consistent with the date in the legislation."

Explanation:

(This amendment directs the Secretary of Public Safety and Homeland Security as part of the workgroup created pursuant to Senate Bill 1339 to include a comprehensive review of all systems and processes necessary for the expungement or sealing of police or court records and to report on the costs of needed improvements to the General Assembly.)

		Item 394 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Virginia Alcoholic Beverage Control Authority	\$0 0.00	\$1,000,000 10.00	NGF FTE

Language:

Page 471, line 42, strike "\$24,692,092" and insert "\$25,692,092".

Explanation:

(This amendment provides \$1.0 million from nongeneral funds and ten positions the second year for the Virginia Alcoholic Beverage Control Authority's Bureau of Law Enforcement to increase enforcement related to direct shipment of alcohol.)

Item 394 #2s

Public Safety and Homeland Security

Virginia Alcoholic Beverage Control Authority

Language

Language:

Page 472, strike lines 18 through 23.

Explanation:

(This amendment eliminates authority provided in the introduced budget to authorize a Treasury Loan for the Virginia Alcoholic Beverage Control Authority to fund the start-up costs associated with legislation related to the legalization and governance of adult use marijuana products. The legislation passed by the Senate establishes a new authority making the language no longer necessary.)

Item 400 #1s

Public Safety and Homeland Security

Department of Corrections

Language

Language:

Page 478, line 25, after "\$500.", insert:

"The department shall amend its contract with the vendor that operates the Lawrenceville Correctional Center to require that this funding is provided as a bonus for correctional officers and require an accounting of the funding to the department. The department shall report on the use of this funding, including the number of correctional officers provided a bonus and, if applicable, any balances remaining to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by June 30, 2021."

Explanation:

(This amendment requires the Department of Corrections to ensure that the funding included in the introduced budget to provide a \$500 bonus for each correctional officer at Lawrenceville Correctional Center, employed by a private contractor that operates the facility, is expended only for such purpose.)

		Item 400 #2s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0	\$250,000	GF

Language:

Page 476, line 30, strike "\$821,331,843" and insert "\$821,581,843".

Page 478, after line 25 insert:

"N. Out of this appropriation, \$250,000 from the general fund in the second year is provided for the Department of Corrections to provide expanded video visitation services for inmates."

Explanation:

(This amendment provides \$250,000 from the general fund in the second year for expanded and subsidized prison family video visitation services. The purpose of this budget amendment is to promote public safety by allowing constructive reintegration for offenders with their families and children, to increase family and community ties prior to the release of the offender, and decrease the likelihood of recidivism.)

		Item 400 #3s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0 0.00	\$471,420 5.00	GF FTE

Language:

Page 476, line 30, strike "\$821,331,843" and insert "\$821,803,263".

Page 478, after line 25 insert:

"N. Out of this appropriation, \$471,420 the second year from the general fund is provided to fund five positions to implement the recommendations of the Secretary of Public Safety and Homeland Security's workgroup on Access to Sex Offender Treatment."

Explanation:

(This amendment adds \$471,420 from the general fund and five positions in the second year to partially fund the short-term recommendations that were included in the Secretary of Public Safety and Homeland Security's November 2020 Report "Access to Sex Offender Treatment in Virginia Prisons." Funding is included for two offender management positions to screen for sex offender risk at the beginning of an inmate's sentence and three positions in the sex offender services unit to increase the number of sex offender treatment providers dedicated to the provision of sex offender services.)

		Item 400 #4s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0 0.00	\$100,000 1.00	GF FTE

Language:

Page 476, line 30, strike "\$821,331,843" and insert "\$821,431,843".

Page 478, after line 25, insert:

"N. Out of this appropriation, \$100,000 the second year from the general fund is provided for one coordinator position to review and oversee the modifications to the operating procedures in the Department of Corrections related to restrictive housing pursuant to required changes included in legislation passed in the 2021 General Assembly. As part of the the review of the

use of restricted housing, the department shall report on current restrictive housing procedures related to medical and mental health evaluations, existing staff that support such evaluations, and report on how legislative changes would impact the current workload of such staff. The department, in collaboration with the Department of Juvenile Justice that shall also conduct a review of its procedures and staffing impacts, shall report on its findings and any potential cost impact to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021."

Explanation:

(This amendment provides \$100,000 the second year from the general fund for a coordinator position to review and oversee the modifications to the operating procedures in the Department of Corrections related to restrictive housing pursuant to required changes included in legislation passed in the 2021 General Assembly (Senate Bill 1301). The Department of Juvenile Justice would also need to review their procedures. A report on the findings and any potential cost impact would be made to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021.)

		Item 402 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0	\$350,000	GF

Language:

Page 479, line 35, strike "\$180,965,434" and insert "\$181,315,434".

Page 482, line 26, after "first year" insert "and \$350,000 the second year."

Page 483, after line 6, insert:

- "23. Senate Bill 1113, 2021 Session -- \$50,000
- 24. Senate Bill 1336, 2021 Session -- \$50,000
- 25. Senate Bill 1461, 2021 Session -- \$50,000
- 26. Senate Bill 1310, 2021 Session -- \$50,000
- 27. Senate Bill 1339, 2021 Session -- \$50,000
- 28. Senate Bill 1395, 2021 Session -- \$50,000
- 29. Senate Bill 1406, 2021 Session -- \$50,000."

Explanation:

(This amendment provides \$350,000 from the general fund in the second year to the Corrections Special Reserve Fund to reflect the estimated impact on utilization of beds in the Commonwealth's adult correctional centers resulting in changes in criminal sentencing pursuant to seven bills that passed the Senate.)

Public Safety and Homeland Security

Department of Corrections

Language

Language:

Page 481, line 54, after "Q." insert "1.".

Page 482, after line 3, insert:

"2. Prior to the award of any contract to a vendor for implementation of electronic health records, the Department of Corrections shall report on the total costs of implementing electronic health records at all of its facilities based on the selected vendor. The department shall certify that it has sufficient on-going funding for full implementation at all facilities prior to awarding a contract. If the department has insufficient base funding, then it shall not award any contract until sufficient funding has been appropriated by the General Assembly. The department shall report on all funding currently budgeted for the project, the timeline for implementation, and the interoperability of the system. The department shall utilize its nongeneral funds appropriated for this purpose prior to using the general fund appropriation. The department shall report on the project and provide its certification to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees prior to the award of any contract."

Explanation:

(This amendment provides that the Department of Corrections report on the costs of implementing electronic health records across its facilities and certify that it has sufficient funding for full implementation prior to awarding any contract to a vendor.)

		Item 402 #3s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0 0.00	\$160,000 2.00	GF FTE

Language:

Page 479, line 35, strike "\$180,965,434" and insert "\$181,125,434".

Page 483, after line 31, insert:

"Y. Out of this appropriation, \$160,000 the second year from the general fund is provided to the Department of Corrections for two full-time staff to support the Board of Local and Regional Jails."

Explanation:

(This amendment provides \$160,000 the second year from the general fund to support two positions dedicated to the Board of Local and Regional Jails pursuant to Senate Bill 1363, which expands the Board's oversight to include the state correctional system.)

Item 410 #1s

FY21-22	
\$2,500,000	ЗF
	\$2,500,000

Language:

Page 493, line 24, strike "\$28,699,285" and insert "\$31,199,285".
Page 494, line 24, after "first year" insert "and \$2,500,000 the second year."

Explanation:

(This amendment adds \$2.5 million the second year from the general fund to continue funding for the Emergency Shelter Upgrade Assistance Fund to aid local governments in proactively preparing for emergency sheltering situations on an on-going basis.)

		Item 411 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Emergency Management	(\$41,965,587)	\$0	GF

Language:

Page 494, line 37, strike "\$65,063,392" and insert "\$23,097,805".

Page 495, line 31, strike "\$41,965,587 the first year and".

Page 495, line 35, strike "\$31,144,081 the first year and".

Page 495, line 39, strike "\$569,833 the first year and".

Page 495, line 42, strike "\$8,050,173 the first year and".

Page 495, line 46, strike "\$2,201,500 the first year and".

Page 495, line 53, after "report", strike "the".

Page 496, line 1, strike "use of the funds in FY 2021 along with".

Explanation:

(This amendment reduces \$42.0 million the first year from the general fund that was provided in the introduced budget to the Virginia Department of Emergency Management for coordinating response and recovery efforts related to the COVID-19 pandemic. The Commonwealth recently received federal reimbursement for prior COVID-19 related expenses that were paid from the federal Coronavirus Relief Fund (CRF), authorized pursuant to the Coronavirus Aid, Relief, and Economic Security (CARES) Act of 2020. This federal funding allows the general fund provided in the introduced budget in the first year to be supplanted with newly available federal CRF funds. A separate amendment in Central Appropriations modifies the allocations from the CRF for this purpose.)

Item 423 #1s

Public Safety and Homeland Security

Department of Juvenile Justice

Language

Language:

Page 506, line 21, strike "6." and insert "C.1."

Page 506, after line 24, insert:

"2. In procuring any new security systems or components for the existing facility at Bon Air from such funds available in this Item, the Department shall consider ways to reuse the system procured in a future facility. To that end, the Department shall work with the Department of General Services to plan for reuse of a previously acquired security system in any future new facility constructed, to the extent feasible."

Explanation:

(This amendment directs the Department of Juvenile Justice to consider potential reuse of a security system in a new facility in the future, in the acquisition and procurement of such system for the existing facility at Bon Air.)

		Item 425 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0 0.00	\$501,990 5.00	GF FTE

Language:

Page 507, line 7, strike "\$74,070,135" and insert "\$74,572,125".

Page 509, after line 10, insert:

"Q. Out of this appropriation, \$501,990 the second year from the general fund is provided to the Department of State Police for five positions for cold case investigators to support efforts to resolve such cases."

Explanation:

(This amendment provides \$501,990 the second year from the general fund and five positions in the second year to support five cold case investigators. Legislation in the 2020 Session established a cold case file database and this funding will support ongoing work in the program.)

Item 425 #2s

Public Safety and Homeland Security

Department of State Police

Language

Language:

Page 509, after line 10, insert:

"Q. The Superintendent of State Police shall report on the feasibility of establishing a registry for determining eligibility to lawfully possess a firearm for on-site rental use at a sport shooting range, based on existing state and federal laws concerning possession of firearms by persons with a history of mental illness. The report shall consider, at a minimum: (i) the information technology changes needed to collect the necessary information to determine if the renter of a firearm for on-site use is prohibited from possessing a firearm under any applicable state or federal law; (ii) the appropriate form or mechanism for collection of information to determine the mental health and criminal history of customers of sport shooting ranges; (iii) the reasonable timeline by which the registry can be implemented; and (iv) any necessary costs for implementation of a mental health background check registry for on-site firearms rentals. The department shall report to the General Assembly on the information required in this paragraph by September 30, 2021."

Explanation:

(This amendment requires the Superintendent of State Police to report on the steps needed to establish a registry for determining eligibility to rent a firearm for on-site use at sport shooting ranges.)

		Item 426 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0	(\$2,310,700)	GF

Language:

Page 509, line 13, strike "\$300,057,282" and insert "\$297,746,582".

Page 511, line 16, after "R.", insert "1.".

Page 511, line 16, strike "\$9,488,184" and insert "\$7,177,484"

Page 511, after line 16, insert:

"The Department of Planning and Budget shall unallot the appropriation provided in Paragraph R.1. at the beginning of FY 2022. The Department of State Police shall provide additional information and justification on the increase in funding for vehicle replacements. This information shall include a detailed overview of specific types of vehicles by various automobile manufacturers that are available for the agency's use and the rationale for the agency's preferred manufacturer and type of vehicle over the others. As part of this information, the department shall provide detailed information on the specific equipment needed for the vehicles and whether certain vehicle types or vehicles from different manufacturers alter the

type or cost of equipment modifications needed for the vehicles. No sooner than 30 days after this information is provided to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees, the Department of Planning and Budget is authorized to allot the funding."

Explanation:

(This amendment reduces \$2.3 million the second year from the general fund for the replacement of State Police vehicles. The introduced budget proposed \$9.5 million in additional funding. This additional funding is on top of existing base funding of \$12.7 million spent on vehicle replacement and repair. The adjustment is based on maintaining the current turnover rate of 350 vehicles per year at approximately 130,000 miles. Funding is provided for the department to purchase SUVs and associated equipment. Language is included to unallot the appropriation until such time as the department provides additional information and justification on the increase in funding for vehicle replacements, which would include a detailed overview of specific types of vehicles by various automobile manufacturers that are available for the agency's use and the rationale for the agency's preferred manufacturer and type of vehicle over the others.)

		Item 426 #2s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0	\$28,547,132	NGF

Language:

Page 509, line 12, strike "\$300,057,282" and insert "\$328,604,414".

Page 511, after line 28 insert:

"S. Out of this appropriation, \$28,547,132 the second year shall be provided from the Public Safety Trust Fund to support the Department of State Police's Compensation, Compression, and Retention Plan for State Troopers. The department is authorized to implement the plan effective in the pay period beginning August 10, 2021."

Explanation:

(This amendment provides \$28.5 million from the Public Safety Trust Fund in the second year to fully fund the Department of State Police's Trooper Compensation Plan. The Public Safety Trust Fund is created pursuant to Senate Bill 1211 and is supported by an additional vehicle registration fee of \$4.00 per year beginning July 1, 2021. The Fund will be used to establish a sworn pay structure that promotes an inclusive, diverse, and well-qualified workforce by fully addressing compensation issues that negatively impact staffing, retention, and pay compression issues at the Department of State Police.)

Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0	\$1,145,246	GF

Language:

Page 511, line 30, strike "\$32,359,876" and insert "\$33,505,122".

Explanation:

(This amendment provides \$1.1 million from the general fund the second year for the Department of State Police to coordinate and provide training to state and local law enforcement agencies in Drug Recognition Expert techniques in order to provide effective detection and enforcement for driving under the influence of drugs, including marijuana. Approximately half of the funding is intended for course seats for political subdivisions, community colleges, and university law enforcement agencies. Funding is also provided for travel, meals, equipment, and related expenses.)

Item 427 #2s

Public Safety and Homeland Security

Department of State Police

Language

Language:

Page 512, after line 19 insert:

"D. Notwithstanding § 18.2-308.02, Code of Virginia, a person who has successfully completed online training for a Concealed Handgun Permit prior to January 1, 2021, is eligible to apply for such a permit if the person was not able to obtain an appointment with the county or city circuit court clerk due to COVID-19 restrictions. The online training course must have been successfully completed prior to January 1, 2021. The provisions of this paragraph shall expire on December 31, 2021."

Explanation:

(This language amendment authorizes a citizen to apply for a concealed handgun permit if they have demonstrated competence via online training, under certain circumstances, if their application has been impacted due to restrictions arising from the COVID-19 pandemic.)

		Item 429 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Virginia Parole Board	\$0	\$84,638	GF

Language:

Page 512, line 37, strike "\$2,648,168" and insert "\$2,732,806".

Explanation:

(This amendment provides \$84,638 the second year from the general fund for the costs associated with implementing Senate Bill 1103.)

Item 436 #1s

Transportation

Department of Motor Vehicles

Language

Language:

Page 523, following line 53, insert:

"U. Consistent with the provisions of § 4-13.00 of this Act, the definitions found in §46.2-1600, Code of Virginia, on June 30, 2021, shall remain in full force and effect until June 30, 2022."

Explanation:

(This amendment retains the current definitions found in the Code for "rebuilt vehicle" and "nonrepairable vehicle" and omits the definition of "cosmetic vehicle" allowing DMV customers to apply for salvage and nonrepairable certificates without submitting an estimated cost of repair.)

		Item 442 #1s	
Transportation	FY20-21	FY21-22	
Department of Rail and Public Transportation	\$22,397,000	\$0	NGF

Language:

Page 525, line 38, strike "\$699,845,958" and insert "\$722,242,958".

Page 527, following line 26, insert:

"J. Out of the amounts in this Item, and consistent with the provisions of § 4-13.00 of this Act, \$22,397,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board to the Washington Metropolitan Area Transit Authority (WMATA) Capital Fund established pursuant to § 33.2-3401, Code of Virginia to ensure Commonwealth meets its share of the dedicated regional funding for WMATA based on the 2018 needs assessment."

Explanation:

(This amendment allocates \$22.4 million NGF in the first year from public funds made

available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act to fulfill the Commonwealth's portion of the \$500 million annual commitment to the WMATA Capital program.)

Item 442 #2s

Transportation

Department of Rail and Public Transportation

Language

Language:

Page 527, following line 26, insert:

"J. Any entity established pursuant to Chapter 31 of Title 33.2, Code of Virginia, shall, as a condition of receipt of funding pursuant to § 33.2-1526.1(d)(3), Code of Virginia, and included in this Item, adopt the petition of the Fairfax County Board of Supervisors that was unanimously approved in Action Item 9 on September 15, 2020 and any private sector participating entity contributions described therein related to the naming of the "McLean - Capital One Hall" station."

Explanation:

(This amendment conditions the receipt of funding provided to the Northern Virginia Transportation Commission for distribution to the Washington Metropolitan Area Transit Authority for capital purposes and operating assistance, on adopting the petition of Fairfax County related to the naming of the McLean-Capital One Hall.)

		Item 442 #3s	
Transportation	FY20-21	FY21-22	
Department of Rail and Public Transportation	\$5,000,000	\$0	NGF

Language:

Page 525, line 38, strike "\$699,845,958" and insert "\$704,845,958".

Page 527, following line 26, insert:

"J. Out of the amounts in this Item, and consistent with the provisions of § 4-13.00 of this Act, \$5,000,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board to the Transit Ridership Incentive Program established pursuant to § 33.2-1526.3, Code of Virginia. Of these amounts the Board shall allocate \$2,500,000 to support the establishment of programs to reduce the impact of fares on low-income individuals, including reduced-fare programs and elimination of fares and \$2,500,000 for regional connectivity programs focused on congestion reduction and mitigation

through the provision of long-distance commuter routes."

Explanation:

(This amendment provides \$5.0 million in federal funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act to support transit incentives focused on fare and congestion reduction programs.)

		Item 443 #1s	
Transportation	FY20-21	FY21-22	
Department of Rail and Public Transportation	\$0 \$137,603,000	(\$50,000,000) \$0	GF NGF

Language:

Page 527, line 27, strike "\$137,107,434" and insert "\$274,710,434".

Page 527, line 28, strike "\$247,112,195" and insert "\$197,112,195".

Page 528, strike line 39 through line 43, and insert:

"H. Out of the amounts in this Item, and consistent with the provisions of § 4-13.00 of this Act, \$137,603,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board for the purpose of extending intercity passenger rail service from Roanoke, Virginia to the Blacksburg-Christiansburg, Virginia area and increasing the frequency of intercity passenger rail service along the I-81/Route 29 Corridor from Washington, DC. The Secretary shall provide an assessment of both total project costs and the incremental costs resulting from modelling conducted to assess any infrastructure or network costs needed to service a rail station in Bedford, Virginia to the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations no later than October 15, 2021."

Explanation:

(This amendment provides \$137.6 million from the Coronavirus Response and Relief Supplemental Appropriations Act to ensure long-term sustainable growth of intercity passenger and commuter rail services across the Commonwealth.)

Item 446 #1s

Transportation

Department of Transportation

Language

Language:

Page 529, line 46, strike "first year" and insert "second year".

Page 529, line 48, strike "fifth" and insert "eighth". Page 529, line 49, strike "2021" and insert "2022".

Explanation:

(This amendment moves funding that was previously authorized to support the participation in the Mobility Talks International (MTI) conference in the first year to the second year of the biennium. The 2020 General Assembly approved \$50,000 in funds to support participation in the 2021 annual MTI conference, which was not required due to COVID. The next conference is scheduled to be held in Washington D.C. in January 2022.)

		Item 447 #1s	
Transportation	FY20-21	FY21-22	
Department of Transportation	\$0 \$40,000,000	(\$5,000,000) \$0	GF NGF

Language:

Page 530, line 29, strike "\$3,650,896,554" and insert "\$3,690,896,554". Page 530, line 29, strike "\$3,783,863,742" and insert "\$3,778,863,742".

Page 532, strike line 5 through line 8 and insert:

"H. Out of the amounts in this Item, and consistent with the provisions of §4-13.00 of this Act, \$40,000,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be used by the Board to support the statewide planning, development and construction of multi-use trails. The Board shall give priority to ensuring the safety of existing trails and developing new trails with a total length in excess of 35 miles and shall seek to ensure geographic diversity and multimodal connectivity in the use of these funds. Of these amounts: (i) up to \$17,500,000 shall be allocated for the Eastern Shore-Bay Coast Railway Rails to Trails project; (ii) up to \$17,500,000 shall be allocated for safety enhancements to the Washington & Old Dominion Trail at Wiehle Avenue and improving the intermodal connectivity of the Cross County Trail at Difficult Run with Great Falls Park; (iii) up to \$2,500,000 for the Fall Line Trail; and (iv) up to \$2,500,000 for the Ivy Creek Trail and the completion of the Tobacco Heritage Trail from Skipwith Road to Rudds Creek. The Office of Intermodal Planning and Investment shall coordinate a policy working group comprised of representatives of the Departments of Transportation, Rail and Public Transportation, Conservation and Recreation, the Statewide Trails Advisory Committee and the staff of the House Appropriations and Senate Finance and Appropriations Committees to recommend a prioritization process for the identification of new multi-use trail opportunities, a master planning process, and funding needs assessment. A report of its recommendations shall be communicated to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than October 15, 2021."

Explanation:

(This amendment provides for a one-time capitalization of \$50,000,000 for a statewide multiuse trail initiative using federal funds available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act.)

		Item 447 #2s	
ansportation	FY20-21	FY21-22	
Department of Transportation	\$10,000,000	\$0	NGF

Language:

Transportation

Page 530, line 29, strike "\$3,650,896,554" and insert "\$3,660,896,554".

Page 531, line 23, following "Virginia." insert:

"In addition to these amounts, and consistent with the provisions of § 4-13.00 of this Act, \$10,000,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board for a connected infrastructure urban redevelopment demonstration project within and adjacent to the Virginia Tech campus in the City of Falls Church, Virginia."

Explanation:

(This amendment provides \$10.0 million in federal funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act to the Transportation Partnership Opportunity Fund for an urban smart infrastructure test bed in partnership with the City of Falls Church and Virginia Tech.)

Item 461 #1s

Veterans and Defense Affairs

FY20-21

FY21-22

Secretary of Veterans and Defense **Affairs**

Language:

Page 545, strike lines 8 through 10, and insert:

"Out of this appropriation, up to \$190,000 the second year from the general fund shall be used to support a Military Liaison position under the Secretariat."

Explanation:

(This amendment reallocates existing funding of \$190,000 the second year from the general fund that is currently provided for matching funds for the U.S. Department of Defense Office of Economic Adjustment grants, to support an ongoing Military Liaison position. The U.S. Department of Defense has reorganized the Office of Economic Adjustment and the grants are no longer expected. Therefore, the funding is not needed for the grant matching requirement.)

		Item 464 #1s	
Veterans and Defense Affairs	FY20-21	FY21-22	
Department of Veterans Services	\$0	\$147,500	GF
•	0.00	1.00	F

Language:

Page 547, line 24, strike "\$22,451,304" and insert "\$22,598,804".

Explanation:

(This amendment provides \$147,500 in the second year from the general fund to support a liaison to military spouses position in the Department of Veterans Services, pursuant to Senate Bill 1150. This funding includes \$22,300 in one-time equipment funding. The ongoing cost for the position is \$125,200.)

		Item 477 #2s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	\$0	\$20,311,179	GF

Language:

Page 554, line 9, strike "\$138,332,991" and insert "\$158,644,170".

Page 561, strike lines 12 through 23.

Page 565, line 1, strike "1".

Page 565, strike lines 5 through 51.

Page 565, line 1, following II. insert:

"The Governor is hereby authorized to allocate a sum of up to \$118,067,180 the second year from this appropriation, to the extent necessary to offset any downward revisions of the general fund revenue estimate prepared for fiscal years 2021 and 2022, after the enactment by the General Assembly of the 2021 Appropriation Act. If within five days of the preliminary close of the fiscal year ending on June 30, 2021, the Comptroller's analysis does not determine that a revenue re-forecast is required pursuant to § 2.2-1503.3, Code of Virginia, then such appropriation shall be used only for employee compensation purposes as stated in paragraphs JJ., KK., and LL. below.

2. Furthermore, \$139,781,368 provided in Item 145 of this act to support the state share of a three percent salary adjustment the second year for funded Standards of Quality instructional and support positions, Academic Year Governor's School Program instructional and support

positions, and Regional Alternative Education Program instructional and support positions shall be unallotted, if the Comptroller determines that general fund revenues did not meet or exceed the forecast for fiscal year 2021 and the actions authorized in paragraphs JJ., KK., and LL. are not effectuated.

- JJ.1. Contingent on the provisions of paragraph II.1. above, \$91,069,662 from the general fund the second year is provided to increase the base salary of the following employees by three percent on August 10, 2021:
- a. Full-time and other classified employees of the Executive Department subject to the Virginia Personnel Act;
- b. Full-time employees of the Executive Department not subject to the Virginia Personnel Act, except officials elected by popular vote;
- c. Any official whose salary is listed in § 4-6.01 of this act, subject to the ranges specified in the agency head salary levels in § 4-6.01 c;
- d. Full-time staff of the Governor's Office, the Lieutenant Governor's Office, the Attorney General's Office, Cabinet Secretaries' Offices, including the Deputy Secretaries, the Virginia Liaison Office, and the Secretary of the Commonwealth's Office;
- e. Heads of agencies in the Legislative Department;
- f. Full-time employees in the Legislative Department, other than officials elected by popular vote;
- g. Legislative Assistants as provided for in Item 1 of this act;
- h. Judges and Justices in the Judicial Department;
- i. Heads of agencies in the Judicial Department;
- j. Full-time employees in the Judicial Department;
- k. Commissioners of the State Corporation Commission and the Virginia Workers' Compensation Commission, the Chief Executive Officer of the Virginia College Savings Plan, and the Directors of the Virginia Lottery, and the Virginia Retirement System; and
- l. Full-time employees of the State Corporation Commission, the Virginia College Savings Plan, the Virginia Lottery, Virginia Workers' Compensation Commission, and the Virginia Retirement System.
- 2.a. Employees in the Executive Department subject to the Virginia Personnel Act shall receive the salary increases authorized in this paragraph only if they attained at least a rating of "Contributor" on their latest performance evaluation.
- b. Salary increases authorized in this paragraph for employees in the Judicial and Legislative Departments, employees of Independent agencies, and employees of the Executive Department not subject to the Virginia Personnel Act shall be consistent with the provisions of this

paragraph, as determined by the appointing or governing authority. However, notwithstanding anything herein to the contrary, the governing authorities of those state institutions of higher education with employees not subject to the Virginia Personnel Act may implement salary increases for such employees that may vary based on performance and other employment-related factors. The appointing or governing authority shall certify to the Department of Human Resource Management that employees receiving the awards are performing at levels at least comparable to the eligible employees as set out in subparagraph 2.a. of this paragraph.

- 3. The Department of Human Resource Management shall increase the minimum and maximum salary for each band within the Commonwealth's Classified Compensation Plan by three percent on August 10, 2021. No salary increase shall be granted to any employee as a result of this action. The department shall develop policies and procedures to be used in instances when employees fall below the entry level for a job classification due to poor performance. Movement through the revised pay band shall be based on employee performance.
- 4. The following agency heads, at their discretion, may utilize agency funds or the funds provided pursuant to this paragraph to implement the provisions of new or existing performance-based pay plans:
- a. The heads of agencies in the Legislative and Judicial Departments;
- b. The Commissioners of the State Corporation Commission and the Virginia Workers' Compensation Commission;
- c. The Attorney General;
- d. The Director of the Virginia Retirement System;
- e. The Director of the Virginia Lottery;
- f. The Director of the University of Virginia Medical Center;
- g. The Chief Executive Officer of the Virginia College Savings Plan; and
- h. The Executive Director of the Virginia Port Authority.
- 5. The base rates of pay, and related employee benefits, for wage employees may be increased by up to three percent no earlier than August 10, 2021. The cost of such increases for wage employees shall be borne by existing funds appropriated to each agency.
- 6. The governing authorities of those state institutions of higher education with employees may provide a salary adjustment based on performance and other employment-related factors, as long as the increases do not exceed the three percent increase, on average. No higher education administrators, executive officers, presidents, or faculty employees earning \$150,000 or more shall receive a pay adjustment provided for in this paragraph or otherwise provided for in this act.
- KK.1. Contingent on the provisions of paragraph II.1. above, the appropriations in this Item include funds to increase the base salary of the following employees by three percent on

September 1, 2021, provided that the governing authority of such employees use such funds to support salary increases for the following listed employees.

- a. Locally-elected constitutional officers;
- b. General Registrars and members of local electoral boards;
- c. Full-time employees of locally-elected constitutional officers and,
- d. Full-time employees of Community Services Boards, Centers for Independent Living, secure detention centers supported by Juvenile Block Grants, juvenile delinquency prevention and local court service units, local social services boards, local pretrial services act and Comprehensive Community Corrections Act employees, and local health departments where a memorandum of understanding exists with the Virginia Department of Health.
- 2. Out of the appropriation for Supplements to Employee Compensation is included \$24,125,830 the second year from the general fund to support the costs associated with the salary increase provided in this paragraph.
- LL. Contingent on the provisions of paragraph II.1. above, included in the appropriation for this item is \$2,871,689 the second year from the general fund to provide a three percent increase in base pay for adjunct faculty at Virginia two-year and four-year public colleges and higher education institutions, effective August 10, 2021."

Explanation:

(This amendment provides \$20.3 million GF the second year, above the \$97.8 million included in the Introduced Budget, to provide a three percent salary adjustment for state employees, including adjunct faculty at higher eduction institutions, effective August 10, 2021, for the September 1, 2021 paycheck, and state-supported local employees, effective September 1, 2021.)

		Item 477 #3s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	\$52,000	\$72,000	GF

Language:

Page 554, line 9, strike "(\$17,993,004)" and insert "(\$17,941,004)". Page 554, line 9, strike "\$138,332,991" and insert "\$138,404,991".

Explanation:

(This amendment provides \$52,000 GF the first year and \$72,000 GF the second year to cover estimated costs associated with SB 1275 of the 2021 General Assembly Session, adding EMS workers to § 65.1 402, related to the presumption as to death or disability from respiratory

disease, hypertension, or heart disease, cancer, contingent upon its final passage.)

		Item 477 #4s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	\$172,900	\$185,100	GF

Language:

Page 554, line 9, strike "(\$17,993,004)" and insert "(\$17,820,104)". Page 554, line 9, strike "\$138,332,991" and insert "\$138,518,091".

Explanation:

(This amendment provides \$172,900 GF the first year and \$185,100 GF the second year for the fiscal impact associated with SB 1375 of the 2021 General Assembly Session, which adds COVID-19 as a presumption under the Workers' Compensation Act for firefighters, emergency medical services personnel, law-enforcement, and correctional officers, contingent upon its final passage.)

		Item 479 #4s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	(\$8,000,000)	\$7,000,000	GF

Language:

Page 567, line 23, strike "\$14,869,500" and insert "\$6,869,500".

Page 567, line 23, strike "\$44,188,052" and insert "\$51,188,052".

Page 571, strike line 41 through line 44 and renumber remaining paragraphs.

Page 572, strike line 16 through line 18 and insert:

"U. Out of this appropriation is included up to \$35,000,000 the second year from the general fund for the purpose of funding marijuana legalization and criminal record expungement legislation adopted during the 2020 Special Session I and the 2021 Regular Session and not otherwise explicitly funded in this act. Of these amounts, up to \$3,500,000 in the second year is provided for the initial operating costs of the Virginia Cannabis Control Authority, upon the determination of the Secretary of Finance with the advice and consent of the Director of the Office of Diversity, Equity and Inclusion."

Explanation:

(This amendment consolidates and accounts for marijuana legalization and criminal records expungement legislation adopted during the 2020 Special Session I and the 2021 Regular Session of the General Assembly. SB 1100, as introduced, provided \$36.0 million for these initiatives.)

Item 479 #5s

Central Appropriations

Central Appropriations

Language

Language:

Page 569, after line 25, insert:

"5. The funding contained in paragraph H.1. above shall remain unallotted until such time as the City of Richmond provides to the Department of General Services detailed planning information related to the planning and development of the Slavery and Freedom Heritage Site in Richmond, including Lumpkin's Pavilion and Slave Trail improvements. Upon the receipt of such planning information by the Department of General Services and subsequent approval by the Secretaries of Finance and Administration, the Department of Planning and Budget shall reallot the funds to be made available to the City of Richmond for expenses incurred in such planning and development."

Explanation:

(This amendment unallots \$10 million GF over the biennium until the City of Richmond provides approved plans regarding the planning and development of the Slavery and Freedom Heritage Site, including Lumpkin's Pavilion and Slave Trail improvements.)

Item 479.10 #1s

Central Appropriations

Central Appropriations

Language

Language:

Page 576, line 2, after "account.", strike "Utility customers may only receive a direct payment" Page 576, strike line 3.

Explanation:

(This amendment removes the restriction that utility customers may only receive assistance from the COVID-19 Utility Assistance Fund once.)

Item 479.10 #2s

Central Appropriations

FY20-21

FY21-22

Central Appropriations

\$25,000,000

\$0 NGF

Language:

Page 572, line 19, strike "\$0" and insert "\$25,000,000".

Page 578, line 11, after "C.", insert "1.".

Page 578, after line 26, insert:

"2. Out of the revenues deposited to the COVID-19 Relief Fund created pursuant to § 2.2-115.1 of the Code of Virginia, \$25,000,000 the first year is hereby appropriated for the Department of Small Business and Supplier Diversity for the purpose of awarding grants to small businesses through the Rebuild Virginia program. Eligible grant recipients shall meet the small business definition of § 2.2-1604 of the Code of Virginia.

Explanation:

(This amendment appropriates \$25.0 million NGF from the COVID-19 Relief Fund for the purpose of awarding grants to small businesses through the Rebuild Virginia program.)

Item 479.10 #3s

Central Appropriations

Central Appropriations

Language

Language:

Page 579, after line 7, insert:

"I.1. There is hereby created a Joint Subcommittee for Federal Relief Funds Oversight of the House Appropriations and Senate Finance and Appropriations Committees to review, provide advice to the Governor and provide oversight of federal funds provided to the Commonwealth in response to the Coronavirus Pandemic whereby discretion is provided to the state on how such funds shall be allocated, hereby known as "federal relief funds". The joint subcommittee shall consist of eight members, including the Chair of the House Appropriations Committee and three members of the House Appropriations Committee selected by the Chair, the Chair of the Senate Finance and Appropriations Committee and three members of the Senate Finance and Appropriations Committee selected by the Chair. The Secretaries of Finance, Commerce and Trade, Health and Human Resources, and Public Safety and Homeland Security shall also be available to provide technical assistance to the joint subcommittee. The joint subcommittee shall be staffed by staff from the House Appropriations and Senate Finance and Appropriations Committees

2. The joint subcommittee shall work with the Governor to consider and examine all currently allocated and potential uses of federal relief funds in providing assistance to state agencies, institutions of higher education, local governments, private non-profit or for-profit businesses, and citizens of the Commonwealth that have been negatively impacted economically due to the effects of the Coronavirus Pandemic. The joint subcommittee shall make recommendations, consistent with the requirements of federal law and any federal regulations or guidance, to the Governor for his consideration. The Governor shall, within 10 days of the enactment of this Act, provide information to the joint subcommittee on previous allocations of federal relief funds,

expenditures to date, current unexpended allocations, and requests under consideration.

3. Prior to the allocation or appropriation of federal relief funds, the Governor shall submit his proposal or proposed allocation to the joint subcommittee, which shall review and make recommendations on such proposals to the Governor within 15 days. The Governor shall notify the Chairs of the Senate Finance and Appropriations Committee and the House Appropriations Committee in writing within five days concerning his final decision to appropriate and allocate federal relief funds having previously been considered by the joint subcommittee."

Explanation:

(This amendment establishes a Joint Subcommittee for Federal Relief Funds Oversight to allow the General Assembly to provide oversight of the substantial federal relief funds the federal government has been providing to the state in response to the Coronavirus Pandemic. The Joint Subcommittee would review proposed allocations of funding and make appropriate recommendations to the Governor, however the Governor would make the final decision on how the funds should be allocated. The Joint Subcommittee provides an opportunity for the General Assembly to provide input into the process for allocating federal relief funds and to provide oversight of how such funding is expended.)

Item 479.10 #4s

Central Appropriations

Central Appropriations

Language

Language:

Page 573, line 49, strike "\$55,640,872", insert "\$17,036,718".

Page 573, line 50, strike "\$9,256,178", insert "\$2,470,552".

Page 573, line 52, strike "\$73,056,734", insert "\$103,500,000".

Page 574, line 1, strike "\$25,000,000", insert "\$17,467,766".

Page 574, line 40, strike "\$22,052,445", insert "\$34,234,679"

Page 574, line 45, strike "\$3,450,000", insert "\$10,550,000".

Page 574, line 3, strike "\$116,261,410" and insert "\$133,511,410".

Page 574, line 50, strike "\$30,000,000" and insert "\$12,750,000".

Page 575, line 10, strike "\$2,127,357,769", insert "\$2,121,161,255".

Page 575, line 5, strike "\$15,000,000", insert "\$12,000,000".

Page 577, line 15, strike "\$73,056,734", insert "\$103,500,000".

Page 577, line 22, strike "\$25,000,000", insert "\$17,467,766".

Explanation:

(This amendment modifies Coronavirus Relief Fund allocations.)

Central Appropriations

Central Appropriations

Language

Language:

Page 579, after line 7, insert:

"1. If within five days of preliminary close of the fiscal year ending June 30, 2021, the Comptroller determines that general fund revenues for fiscal year 2021 exceeded the official forecast, then it is the intent of the General Assembly that the Governor shall allocate such additional revenue to fund the following items in priority order in fiscal year 2022:

Agency - Title	FY 2022
Direct Aid - Fund One Additional SOQ Support Position	\$47,236,642
Department of Medical Assistance Services - Increase DD Waiver Provider Rates to Reflect Minimum Wage Impact	\$20,000,000
Department of Accounts Transfer Payments - Virginia Retirement System Payment to Increase Actuarial Funding Status	\$200,000,000
Total	\$267.236.642

- 2. The Director, Department of Planning and Budget, shall have authority to adjust the general fund appropriations in the second year in the appropriate agencies to reflect items funded as a result of paragraph 1.
- 3. Should the additional revenue pursuant to paragraph 1 be insufficient to fund all the items listed, then the Governor shall fund those items listed in priority order for which the additional revenue is sufficient to fully fund."

Explanation:

(This amendment provides that any additional revenue for fiscal year 2021 as a result of the higher than forecast general fund revenues in the fiscal year will be used for the listed high priority funding items in fiscal year 2022.)

Item 481 #1s

Central Appropriations

Central Appropriations

Language

Language:

Page 580, line 1, after "Code of Virginia," insert "for the 2020-22 biennium".

Page 580, line 2, after "MOU." insert "The designated reviewers shall propose any needed

technical adjustments for consideration during the 2022 Session."

Explanation:

(This amendment clarifies proposed technical language regarding eligibility for "Tech Talent" grants pursuant to the approved Memorandums of Understanding.)

Item C-0 #1s

General Conditions

General Conditions Language

Language:

Page 609, after line 15, insert:

"9. Notwithstanding any other provision of law, a public institution of higher education may participate in the United States Department of Education Historically Black College and University Capital Financing Program (HBCU Program), and use federal grant and contract funds as permitted by the Program."

Explanation:

(This amendment is intended to ensure that Virginia's public historically black colleges and universities (HBCU) may access the United States Department of Education Historically Black College and University Capital Financing Program (HBCU Program). The HBCU Program requires that institutions be able to use revenue from federal grants and contracts via administrative offset if the pledged revenues are insufficient to pay debt service. The HBCU Program provides HBCUs with low-cost capital to finance infrastructure improvements, facilitating the repair, renovation, and construction of classrooms, libraries, laboratories, dormitories, instructional equipment, and research instrumentation.)

			Item C-17.10	#1s
Capital Projects		FY20-21	FY21-22	
James Madison V	University	\$0	\$6,579,237	NGF
Language:				
Page 618, after line 2	22, insert:			
"C-17.10 Improvements: Phase I	East Campus Steam Plant,	\$0	\$6,579,237	
Fund Sources:	Higher Education Operating	\$0	\$1,973,771	
	Bond Proceeds	\$0	\$4,605,466 "	

Explanation:

(This amendment provides \$4.6 million in bond proceeds for Phase 1 of improvements and upgrades to the East Campus Steam Plant project at James Madison University. Total project cost for this phase of work is \$6.6 million and includes almost \$2.0 million supported with auxiliary nongeneral funds.)

Item C-22.10 #1s

Education

Virginia Commonwealth University

Language

Language:

Page 619, strike lines 37 through 39 and insert:

"The provisions of Item C-13.10 A., B., C.2., C.3., and D., Chapter 854, 2019 Acts of Assembly, as it relates to the Virginia Commonwealth University acquisition of the Virginia Alcoholic Beverage Control Authority property are hereby extended for the 2020-22 Biennium. The provisions of Item C-13.10 C.1. shall be replaced as follows:

C. 1. The Authority shall offer for sale the Property to VCU at the fair market value determined by the lower of the two appraisals received pursuant to paragraph B. Any other conditions of the transfer, as negotiated between the Authority and VCU, shall be based on usual and customary terms for such intergovernmental transfers. If the Authority and VCU cannot agree on the terms, other than price as price shall be the fair market value described herein, of the sale of the Property, the Authority may sell the property pursuant to § 2.2-1156 and on such terms as it determines are in the best interest of the Authority, however any sale terms negotiated shall be presented to VCU, prior to any sale, to enable VCU to make a counter offer to the Authority. Should such counter offer be received by the Authority, the Authority shall consider the VCU counter offer and if determined favorable by the Authority, may sell the property to VCU.".

Explanation:

(This amendment specifies additional fair market value parameters regarding Virginia Commonwealth University acquisition of the Virginia Alcoholic Beverage Control Authority property on Hermitage Road in Richmond.)

Education FY20-21 FY21-22
Virginia Community College System \$500,000 \$0 GF

Language:

Page 620, after line 14, insert:

"C-23.5 Planning: Amherst and Campbell Halls, Central Virginia Community College
Fund Sources: General \$500,000 \$0"

Explanation:

(This amendment provides \$500,000 GF for planning for renovations of Amherst and Campbell Halls on Central Virginia Community College's campus. Amherst (54,042 Square Feet) and Campbell (19,550 square feet) Halls comprise the core classroom space for Central Virginia Community College. Amherst Hall was built in 1968 and Campbell Hall was completed in 1974 and neither have had a major renovation.)

	Item C-36.30		#1s
Education	FY20-21	FY21-22	
Frontier Culture Museum of Virginia	\$1,300,000	\$0	GF

Language:

Page 622, line 7, strike "\$0" and insert "\$1,300,000".

Page 622, line 12, strike the first instance of "\$0" and insert "\$1,300,000".

Explanation:

(This amendment provides an additional \$1.3 million GF the first year for the project named "Improvements: Construct Crossing Gallery (18316)". Detailed planning for the project was initially funded in the 2019 Session by Chapter 854, Item C-21-10. The additional funding is requested to support the additional services required of the project architect/engineer to complete the required revisions and complete the project planning through the preliminary design and cost estimate phase.)

	Item C-36		#1s
Education	FY20-21	FY21-22	
The Science Museum of Virginia	\$0	\$7,506,000	NGF
Language:			
Page 622, after line 16, insert:			
"C-36.45 Improvements: Community Green Space	\$0	\$7,506,000	
Fund Sources: Special	\$0	\$7,506,000 "	

Explanation:

(This amendment authorizes the Science Museum of Virginia to seek private funding of up to \$7.5 million for a community green space project. Situated on the Pulse corridor and a key element of the Richmond 300 Plan, this project will transform the area in front of the Museum into a vibrant and active community resource to serve the needs of a diverse, growing citizenry.)

Item C-40 #1s

Natural Resources

Department of Conservation and Recreation

Language

Language:

Page 623, line 23, after "Staunton River," insert "Kiptopeke,"

Explanation:

(This amendment updates the list for acquisitions of land for state parks for in-holdings and contiguous properties, consistent with funding included in the budget as introduced, to include Kiptopeke.)

		Item C-64 #1s	
Central Appropriations	FY20-21	FY21-22	
Central Capital Outlay	\$0 \$0	\$37,000,000 GF (\$37,000,000) NGF	

Language:

Page 628, line 26, strike "\$137,000,000" and insert "\$100,000,000".

Page 628, line 27, after "A." insert "1."

Page 628, line 27, strike the second instance of "\$137,000,000" and insert "\$100,000,000".

Page 628. after line 26, insert "Fund Sources: General \$0 \$37,000,000".

Page 628, after line 30, insert:

"2. Out of this appropriation, \$37,000,000 the second year from the general fund is designated for capital costs of maintenance reserve projects."

Explanation:

(This amendment changes the source of funding of a portion of the \$137.0 million provided in the second year of the biennium for maintenance reserve from tax-supported debt to general fund cash.)

Item C-66 #1s

Central Appropriations	FY20-21	FY21-22
Central Capital Outlay	\$8,450,000	\$0 GF

Language:

Page 632, line 48, strike "\$1,517,750" and insert "\$9,967,750".

Page 633, line 4, after "B." insert "1.".

Page 633, after line 9, insert "Replace Division 6 Headquarters".

Page 633, after line 36, insert:

"2. Out of this appropriation, \$4,500,000 the first year from the general fund is designated for planning the Fine and Performing Arts Center project at the University of Mary Washington, \$2,100,000 the first year from the general fund is designated for planning the Center for Leadership & Ethics Facility, Phase 2 project at Virginia Military Institute, and \$1,850,000 the first year from the general fund is designated for planning the Division 6 Headquarters project for the State Police."

Explanation:

(This amendment provides general funds for the planning the following projects that were included in the planning pool in the 2020 regular session: \$4.5 million for the University of Mary Washington for detailed planning for the construction of a Fine and Performing Arts Center; and \$2.1 million for Virginia Military Institute for planning for the Center for Leadership & Ethics Facility, Phase 2 project; and also \$1.85 million for planning the State Police Division 6 Headquarters replacement project.)

Item C-66.10 #1s

Central Appropriations	FY20-21	FY21-22	
Central Capital Outlay	\$0	(\$58,500,000)	NGF

Language:

Page 634, line 6, strike "\$58,500,000" and insert "\$0". Page 634, strike lines 5 through 21.

Explanation:

(This amendment maintains the scope for the replacement of Central State Hospital at 252 beds, reversing the proposed \$58.5 million increase in debt authorization for the expansion of an additional 48 beds.)

Item C-72 #1s

Central Appropriations

Central Capital Outlay

Language

Language:

Page 640, line 46, after "a capital project" insert "or projects".

Page 640, line 48, strike "and buildings".

Page 640, strike line 51 and insert "Authority each approve a public private partnership or other contractual agreement with respect to such a capital project or projects. The"

Explanation:

(This amendment would allow the bond proceeds authorized by this Item to be used for improvements that support multiple projects and projects performed as a P3 or a direct lease agreement with the Virginia Port Authority. Use of the funds will still require approval of both the Virginia Port Authority Board of Commissioners and the Major Employment & Investment Project Approval Commission, while providing the flexibility needed by the Virginia Port Authority and the Virginia Economic Development Partnership to attract multiple users and make the Portsmouth Marine Terminal an offshore wind industry hub.)

Item C-72.70 #1s

Capital Projects

Department of State Police

Language

Language:

Page 641, after line 14, insert:

"The title of the project for the Department of the State Police previously authorized in Item C-45, Chapter 2, 2018 Acts of Assembly, Special Session I, as "Construct Area 13 Barracks" is hereby changed to "Acquire, Renovate or Construct Area 13 Barracks".

Explanation:

(This amendment makes a title change to an existing project.)

Item C-76 #1s

Central Appropriations

Central Capital Outlay

Language

Language:

Page 644, after line 33, insert:

"James Madison University (216)

East Campus Steam Plant, Phase 1 XXXXX C-17.10 \$4,605,466". Page 645, line 18, strike "\$100,007,208" and insert "\$104,612,674".

Explanation:

(This amendment updates the Virginia College Building Authority bond table for the Committee's amendments.)

Item 3-1.01 #1s

Transfers

Interfund Transfers

Language

Language:

Page 650, line 6, strike "\$139,500,745" and insert "\$159,500,745". Page 650, line 7, strike "\$132,700,000" and insert "\$134,400,000".

Explanation:

(This amendment increases the estimated transfer to the general fund of net profits from ABC gross store sales by \$20.0 million the first year and \$1.7 million the second year, based upon higher assumed sales and profits in the current fiscal year.)

Item 3-2.03 #1s

Working Capital Funds and Lines of Credit

Lines of Credit Language

Language:

Page 657, strike lines 6 through 7.

Explanation:

(This amendment eliminates a \$1.0 million line of credit proposed in the introduced budget for Virginia Industries for the Blind.)

Item 3-5.15 #1s

Adjustments and Modifications to Tax Collections

Provider Coverage Assessment

Language

Language:

Page 661, line 29, strike "1.08" and insert "1.02".

Explanation:

(This amendment modifies the coverage assessment amount collected from hospitals to cover the state share of Medicaid expansion costs. Current law requires the collection of 108 percent of the state share of the full costs of expanded Medicaid coverage. The original intent was to ensure that collections were sufficient to cover the full cost of expansion and not rely on any general fund dollars to expand Medicaid. Rate assessment language has since been amended to allow the Department of Medical Assistance Services (DMAS) to adjust assessments with each quarter to account for differences between estimates and actual spending, however, the 108 percent collection requirement is applied to the estimate for the entire year and not just the remaining quarters. In addition, DMAS now has more experience with expansion and hospital overpayments in the first two years of the program have far exceeded the additional eight percent add-on. Reducing the add-on from eight to two percent would reduce the amount of overpayment made each year by the hospitals into the Health Care Coverage Assessment Fund.)

Item 3-5.16 #1s

Adjustments and Modifications to Tax Collections

Provider Payment Rate Assessment

Language

Language:

Page 662, line 51, after "equivalent to", strike the remainder of the line. Page 662, line 52, before "subject to CMS", insert: "the maximum managed care directed payment amount as allowed by CMS,".

Explanation:

(This amendment modifies language related to the hospital provider payment rate assessment to change the methodology to enable the hospitals to obtain additional federal dollars with no additional cost to the Commonwealth. Current budget language limits supplemental payments by the upper payment limit gap for fee-for-service claims and extrapolates that gap to managed care claims. Because over 90 percent of Medicaid enrollees are in managed care, this gap can be highly variable on an annual basis and can limit total rate enhancement. The federal Centers for Medicare and Medicaid Services (CMS) has provided other states the flexibility to structure managed care directed payments as a percentage of Medicare. However, the introduced budget language is highly prescriptive and limits the department's ability to explore and apply an alternate methodology which could potentially stabilize or increase total supplemental payments.)

Item 3-5.23 #1s

Finance

Language

Language:

Page 664, after line 37, insert:

"§ 3-5.23 CORPORATE INCOME TAX INFORMATIONAL REPORTING

- A.1. Corporations that are members of a unitary business must file a report, in a manner prescribed by the Tax Commissioner, for the unitary combined group containing the unitary combined net income of such group. The report shall be based on taxable year 2019 computations and include, at a minimum the difference in tax owed as a result of filing a unitary combined report, computed according to the method or methods specified by the Tax Commissioner, compared to the tax owed under the current filing requirements.
- 2. "Unitary business" means a single economic enterprise made up either of separate parts of a single business entity or of a commonly controlled group of business entities that are sufficiently interdependent, integrated, and interrelated through their activities so as to provide a synergy and mutual benefit that produces a sharing or exchange of value among them and a significant flow of value to the separate parts. A "unitary business" includes that part of the business that meets the definition in this section and is conducted by a taxpayer through the taxpayer's interest in a partnership, whether the interest in that partnership is held directly or indirectly through a series of partnerships or other pass-through entities. A "unitary business" shall not include persons subject to, or that would be subject to if doing business in the Commonwealth, the insurance premiums license tax under Chapter 25 (§ 58.1-2500 et seq.), Code of Virginia, or the bank franchise tax under Chapter 12 (§ 58.1-1200 et seq.).
- 3. The report must be submitted to the Department of Taxation on or before June 1, 2021, which date shall not be extended.
- 4. Members of a unitary combined group shall exclude as a member and disregard the income and apportionment factors of any corporation incorporated in a foreign jurisdiction (a "foreign corporation") if the average of its property, payroll and sales factors outside the United States is eighty percent (80%) or more. If a foreign corporation is includible as a member in the unitary combined group, to the extent that such foreign corporation's income is subject to the provisions of a federal income tax treaty, such income is not includible in the unitary combined group net income. Such member shall also not include in the unitary combined report any expenses or apportionment factors attributable to income that is subject to the provisions of a federal income tax treaty. For purposes of this paragraph, "federal income tax treaty" means a comprehensive income tax treaty between the United States and a foreign jurisdiction, other than a foreign

jurisdiction which the organization for economic co-operation and development has determined has not committed to the internationally agreed tax standard, or has committed to the international agreed tax standard but has not yet substantially implemented that standard, as identified in the then-current organization for economic co-operation and development progress report.

B. Any corporation required to submit such report to the Department of Taxation that fails to do so on or before June 1, 2021, or that makes a material omission or misstatement in connection with such report shall be subject to a penalty of \$10,000. The Tax Commissioner shall have the authority to waive such penalty upon a determination that the requirement would cause an undue hardship. All requests for waiver shall be transmitted to the Tax Commissioner in writing.

C. The Tax Commissioner shall on or before December 1, 2021, based on the information provided in income tax returns and the data submitted under this section, submit a report to the Chair of the Senate Finance and Appropriations Committee, the Chair of the House Appropriations Committee, and the Chair of the House Finance Committee."

Explanation:

(This amendment requires corporations that are members of a unitary business to file an informational report with the Department of Taxation for the unitary combined group containing the unitary combined net income of the group. The report must be based on taxable year 2019 computations and include, at a minimum the difference in tax owed as a result of filing a unitary combined report compared to the tax owed under the current filing requirements. Any corporation required to submit such report that fails to do so on or before June 1, 2021, or that makes a material omission or misstatement in connection with such report is subject to a penalty of \$10,000, unless waived by the Tax Commissioner upon a determination that the requirement would cause an undue hardship.)

Item 4-5.10 #1s

Special Conditions and Restrictions on Expenditures

Surplus Property Transfers for Economic Development

Language

Language:

Page 698, line 1, strike "2020" and insert "2021".

Explanation:

(This amendment extends by one year the deadline for a property sale between the Commonwealth of Virginia and the Eastern Virginia Regional Industrial Facility Authority.)

Item 4-14 #1s

Effective Date

Effective Date Language

Language:

Page 728, strike lines 56 through 60 and insert:

"3. If rent is unpaid when due, or if a payment under the terms of a payment plan is unpaid when due, the landlord shall, pursuant to § 55.1-1202, Code of Virginia, serve a written notice on the tenant that informs the tenant of the Virginia Rent and Mortgage Relief Program and provides the website address and statewide telephone number for that program. The written notice shall also provide information on how to reach 2-1-1 Virginia to determine whether there are any other available federal, state and local rent relief programs. The written notice shall also inform the tenant that the landlord shall apply for rental assistance on the tenant's behalf within 14 days of serving the notice on the tenant, unless the tenant pays in full, enters into a payment plan or informs the landlord that they have already applied for rental assistance. The landlord shall apply for rental assistance on behalf of the tenant no later than 14 days after serving the written notice on the tenant, unless they receive the full amount owed by the tenant or confirmation from the tenant that the tenant has applied for rental assistance before the 14th day, or they have entered into a payment plan with the tenant. If the tenant has applied for rental assistance, the landlord shall cooperate with the tenant's application, by providing all information and documentation required to complete the application, including but not limited to the W-9 IRS form and any supporting affidavits. If a tenant who has not paid in full or entered into a payment plan with the landlord within 14 days after the written notice is served refuses to apply for rental assistance and also refuses to cooperate with the landlord in providing information and documentation required to complete the application made by the landlord, or if such tenant is determined ineligible for rental assistance, or there are no longer funds available through any federal, state or local rental assistance program, the landlord may take action to obtain possession of the tenant's dwelling unit as provided in § 55.1-1251, Code of Virginia."

Page 729, strike lines 1 through 20.

Page 729, strike lines 28 through 30 and insert:

"f. Nothing in this section shall void any judgment for possession validly obtained by a landlord prior to November 18, 2020; however, a landlord shall not initiate, maintain, or advance any legal process to obtain possession of a dwelling unit for non-payment of the rent unless the landlord complies with the provisions of this Section 8."

Explanation:

(This amendment updates the language included in Chapter 56, 2020 Special Session I, Acts of Assembly to clarify the requirements for landlords and tenants to apply for rental assistance

prior to any evictions-related action occurring. It also removes the 45-day deadline for approval of a pending application for rental assistance.)

Item 4-14 #2s

Effective Date

Effective Date

Language

Language:

Page 727, after line 34, insert:

- "k. In addition to the relief provided pursuant to clause 7.j., within 60 days after the enactment of this act, a Phase II Utility shall forgive all such utility's jurisdictional customer balances more than 30 days in arrears as of December 31, 2020.
- 1. In the utility's 2021 triennial review, the provisions of clause 7.k. shall be excluded from the utility's cost of service for purposes of determining any test period earnings and determining any future rates of the utility. In determining any customer bill credits, in the utility's 2021 triennial review, the Commission shall first offset any amounts pursuant to clause 7.k. against the total earnings for the 2017 through 2020 test periods that are determined to be above the utility's authorized earnings band. Such offset shall be made prior to any offset to customer bill credits by customer credit reinvestment offsets.
- 2. Each Phase II Utility shall, no later than November 1, 2021, submit a report to the Governor, the Chairs of the House Committees on Labor and Commerce and Appropriations, and the Senate Committees on Commerce and Labor and Finance and Appropriations, and the Chair of the Commission on Electric Utility Regulation, detailing all actions by it pursuant to this act to forgive customer balances."

Explanation:

(This amendment requires that Dominion Energy write off debt owed as of December 31, 2020.)

Item 4-14 #3s

Effective Date

Language

Language:

Page 735, strike lines 50 through 51 and insert:

11. That §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402 of the Code of Virginia are amended and reenacted as follows:

§ 58.1-301. Conformity to Internal Revenue Code.

- A. Any term used in this chapter shall have the same meaning as when used in a comparable context in the laws of the United States relating to federal income taxes, unless a different meaning is clearly required.
- B. Any reference in this chapter to the laws of the United States relating to federal income taxes shall mean the provisions of the Internal Revenue Code of 1954, and amendments thereto, and other provisions of the laws of the United States relating to federal income taxes, as they existed on December 31, 2019 2020, except for:
- 1. The special depreciation allowance for certain property provided for under §§ 168(k), 168(l), 168(m), 1400L, and 1400N of the Internal Revenue Code;
- 2. The carry-back of certain net operating losses for five years under § 172(b)(1)(H) of the Internal Revenue Code;
- 3. The original issue discount on applicable high yield discount obligations under § 163(e)(5)(F) of the Internal Revenue Code:
- 4. The deferral of certain income under § 108(i) of the Internal Revenue Code. For Virginia income tax purposes, income from the discharge of indebtedness in connection with the reacquisition of an "applicable debt instrument" (as defined under § 108(i) of the Internal Revenue Code) reacquired in the taxable year shall be fully included in the taxpayer's Virginia taxable income for the taxable year, unless the taxpayer elects to include such income in the taxpayer's Virginia taxable income ratably over a three-taxable-year period beginning with taxable year 2009 for transactions completed in taxable year 2009, or over a three-taxable-year period beginning with taxable year 2010 for transactions completed in taxable year 2010 on or before April 21, 2010. For purposes of such election, all other provisions of § 108(i) of the Internal Revenue Code shall apply mutatis mutandis. No other deferral shall be allowed for income from the discharge of indebtedness in connection with the reacquisition of an "applicable debt instrument";
- 5. For taxable years beginning on and after January 1, 2019, the provisions of § 11046 of the federal Tax Cuts and Jobs Act, P.L. 115-97 (2017), related to the suspension of the overall limitation on itemized deductions under § 68(f) of the Internal Revenue Code; and
- 6. The provisions of § 103 of Division Q of the federal Further Consolidated Appropriations Act, 2020, P.L. 116-94 (2019), related to the reduction in the medical expense deduction floor For taxable years beginning on and after January 1, 2017, but before January 1, 2018, and for taxable years beginning on and after January 1, 2019, the 7.5 percent of federal adjusted gross income threshold set forth in § 213(a) of the Internal Revenue Code that is used for purposes of computing the deduction allowed for expenses for medical care pursuant to § 213 of the Internal Revenue Code. For such taxable years, the threshold utilized for Virginia income tax purposes to compute the deduction allowed for expenses for medical care pursuant to § 213 of the Internal Revenue Code shall be 10 percent of federal adjusted gross income;
- 7. The provisions of §§ 2303(a) and 2303(b) of the federal Coronavirus Aid, Relief, and

Economic Security Act, P.L. 116-136 (2020), related to the net operating loss limitation and carryback:

- 8. The provisions of § 2304(a) of the federal Coronavirus Aid, Relief, and Economic Security Act, P.L. 116-136 (2020), related to a loss limitation applicable to taxpayers other than corporations;
- 9. The provisions of § 2306 of the federal Coronavirus Aid, Relief, and Economic Security Act, P.L. 116-136 (2020), related to the limitation on business interest; and
- 10. The provisions of §§ 276(a), 276(b)(2), 276(b)(3), 278(a)(2), 278(a)(3), 278(b)(2), 278(b)(3), 278(c)(2), 278(c)(3), 278(d)(2), and 278(d)(3) of the federal Consolidated Appropriations Act, P.L. 116-260 (2020), related to deductions, tax attributes, and basis increases for certain loan forgiveness and other business financial assistance.

The Department of Taxation is hereby authorized to develop procedures or guidelines for implementation of the provisions of this section, which procedures or guidelines shall be exempt from the provisions of the Administrative Process Act (§ 2.2-4000 et seq.).

§ 58.1-322.02. Virginia taxable income; subtractions.

In computing Virginia taxable income pursuant to § 58.1-322, to the extent included in federal adjusted gross income, there shall be subtracted:

- 1. Income derived from obligations, or on the sale or exchange of obligations, of the United States and on obligations or securities of any authority, commission, or instrumentality of the United States to the extent exempt from state income taxes under the laws of the United States, including, but not limited to, stocks, bonds, treasury bills, and treasury notes but not including interest on refunds of federal taxes, interest on equipment purchase contracts, or interest on other normal business transactions.
- 2. Income derived from obligations, or on the sale or exchange of obligations, of the Commonwealth or of any political subdivision or instrumentality of the Commonwealth.
- 3. Benefits received under Title II of the Social Security Act and other benefits subject to federal income taxation solely pursuant to § 86 of the Internal Revenue Code.
- 4. Up to \$20,000 of disability income, as defined in § 22(c)(2)(B)(iii) of the Internal Revenue Code; however, any person who claims a deduction under subdivision 5 of § 58.1-322.03 may not also claim a subtraction under this subdivision.
- 5. The amount of any refund or credit for overpayment of income taxes imposed by the Commonwealth or any other taxing jurisdiction.
- 6. The amount of wages or salaries eligible for the federal Work Opportunity Credit which was not deducted for federal purposes on account of the provisions of § 280C(a) of the Internal Revenue Code.
- 7. Any amount included therein less than \$600 from a prize awarded by the Virginia Lottery.
- 8. The wages or salaries received by any person for active and inactive service in the National Guard of the Commonwealth of Virginia, not to exceed the amount of income derived from 39 calendar days of such service or \$3,000, whichever amount is less; however, only those persons in the ranks of O3 and below shall be entitled to the deductions specified in this subdivision.
- 9. Amounts received by an individual, not to exceed \$1,000 for taxable years beginning on or

before December 31, 2019, and \$5,000 for taxable years beginning on or after January 1, 2020, as a reward for information provided to a law-enforcement official or agency, or to a nonprofit corporation created exclusively to assist such law-enforcement official or agency, in the apprehension and conviction of perpetrators of crimes. This subdivision shall not apply to the following: an individual who is an employee of, or under contract with, a law-enforcement agency, a victim or the perpetrator of the crime for which the reward was paid, or any person who is compensated for the investigation of crimes or accidents.

- 10. The amount of "qualified research expenses" or "basic research expenses" eligible for deduction for federal purposes, but which were not deducted, on account of the provisions of § 280C(c) of the Internal Revenue Code and which shall be available to partners, shareholders of S corporations, and members of limited liability companies to the extent and in the same manner as other deductions may pass through to such partners, shareholders, and members. 11. Any income received during the taxable year derived from a qualified pension, profitsharing, or stock bonus plan as described by § 401 of the Internal Revenue Code, an individual retirement account or annuity established under § 408 of the Internal Revenue Code, a deferred compensation plan as defined by § 457 of the Internal Revenue Code, or any federal government retirement program, the contributions to which were deductible from the taxpayer's federal adjusted gross income, but only to the extent the contributions to such plan or program were subject to taxation under the income tax in another state.
- 12. Any income attributable to a distribution of benefits or a refund from a prepaid tuition contract or savings trust account with the Virginia College Savings Plan, created pursuant to Chapter 7 (§ 23.1-700 et seq.) of Title 23.1. The subtraction for any income attributable to a refund shall be limited to income attributable to a refund in the event of a beneficiary's death, disability, or receipt of a scholarship.
- 13. All military pay and allowances, to the extent included in federal adjusted gross income and not otherwise subtracted, deducted, or exempted under this section, earned by military personnel while serving by order of the President of the United States with the consent of Congress in a combat zone or qualified hazardous duty area that is treated as a combat zone for federal tax purposes pursuant to § 112 of the Internal Revenue Code.
- 14. For taxable years beginning before January 1, 2015, the gain derived from the sale or exchange of real property or the sale or exchange of an easement to real property which results in the real property or the easement thereto being devoted to open-space use, as that term is defined in § 58.1-3230, for a period of time not less than 30 years. To the extent that a subtraction is taken in accordance with this subdivision, no tax credit under this chapter for donating land for its preservation shall be allowed for three years following the year in which the subtraction is taken.
- 15. Fifteen thousand dollars of military basic pay for military service personnel on extended active duty for periods in excess of 90 days; however, the subtraction amount shall be reduced dollar-for-dollar by the amount by which the taxpayer's military basic pay exceeds \$15,000 and shall be reduced to zero if such military basic pay amount is equal to or exceeds \$30,000.
- 16. The first \$15,000 of salary for each federal and state employee whose total annual salary from all employment for the taxable year is \$15,000 or less.
- 17. Unemployment benefits taxable pursuant to § 85 of the Internal Revenue Code.

18. Any amount received as military retirement income by an individual awarded the Congressional Medal of Honor.

19. Items of income attributable to, derived from, or in any way related to (i) assets stolen from, hidden from, or otherwise lost by an individual who was a victim or target of Nazi persecution or (ii) damages, reparations, or other consideration received by a victim or target of Nazi persecution to compensate such individual for performing labor against his will under the threat of death, during World War II and its prelude and direct aftermath. This subtraction shall not apply to assets acquired with such items of income or with the proceeds from the sale of assets stolen from, hidden from, or otherwise lost to, during World War II and its prelude and direct aftermath, a victim or target of Nazi persecution. The provisions of this subdivision shall only apply to an individual who was the first recipient of such items of income and who was a victim or target of Nazi persecution, or a spouse, surviving spouse, or child or stepchild of such victim. As used in this subdivision:

"Nazi regime" means the country of Nazi Germany, areas occupied by Nazi Germany, those European countries allied with Nazi Germany, or any other neutral European country or area in Europe under the influence or threat of Nazi invasion.

"Victim or target of Nazi persecution" means any individual persecuted or targeted for persecution by the Nazi regime who had assets stolen from, hidden from, or otherwise lost as a result of any act or omission in any way relating to (i) the Holocaust, (ii) World War II and its prelude and direct aftermath, (iii) transactions with or actions of the Nazi regime, (iv) treatment of refugees fleeing Nazi persecution, or (v) the holding of such assets by entities or persons in the Swiss Confederation during World War II and its prelude and aftermath. A "victim or target of Nazi persecution" also includes any individual forced into labor against his will, under the threat of death, during World War II and its prelude and direct aftermath.

- 20. The military death gratuity payment made after September 11, 2001, to the survivor of deceased military personnel killed in the line of duty, pursuant to 10 U.S.C. Chapter 75; however, the subtraction amount shall be reduced dollar-for-dollar by the amount that the survivor may exclude from his federal gross income in accordance with § 134 of the Internal Revenue Code.
- 21. The death benefit payments from an annuity contract that are received by a beneficiary of such contract, provided that (i) the death benefit payment is made pursuant to an annuity contract with an insurance company and (ii) the death benefit payment is paid solely by lump sum. The subtraction under this subdivision shall be allowed only for that portion of the death benefit payment that is included in federal adjusted gross income.
- 22. Any gain recognized from the sale of launch services to space flight participants, as defined in 49 U.S.C. § 70102, or launch services intended to provide individuals with the training or experience of a launch, without performing an actual launch. To qualify for a deduction under this subdivision, launch services must be performed in Virginia or originate from an airport or spaceport in Virginia.
- 23. Any gain recognized as a result of resupply services contracts for delivering payload, as defined in 49 U.S.C. § 70102, entered into with the Commercial Orbital Transportation Services division of the National Aeronautics and Space Administration or other space flight entity, as defined in § 8.01-227.8, and launched from an airport or spaceport in Virginia.

24. Any income taxed as a long-term capital gain for federal income tax purposes, or any income taxed as investment services partnership interest income (otherwise known as investment partnership carried interest income) for federal income tax purposes. To qualify for a subtraction under this subdivision, such income shall be attributable to an investment in a "qualified business," as defined in § 58.1-339.4, or in any other technology business approved by the Secretary of Administration, provided that the business has its principal office or facility in the Commonwealth and less than \$3 million in annual revenues in the fiscal year prior to the investment. To qualify for a subtraction under this subdivision, the investment shall be made between the dates of April 1, 2010, and June 30, 2020. No taxpayer who has claimed a tax credit for an investment in a "qualified business" under § 58.1-339.4 shall be eligible for the subtraction under this subdivision for an investment in the same business.

25. For taxable years beginning on and after January 1, 2014, any income of an account holder for the taxable year taxed as (i) a capital gain for federal income tax purposes attributable to such person's first-time home buyer savings account established pursuant to Chapter 12 (§ 36-171 et seq.) of Title 36 and (ii) interest income or other income for federal income tax purposes attributable to such person's first-time home buyer savings account.

Notwithstanding the statute of limitations on assessments contained in § 58.1-312, any subtraction taken under this subdivision shall be subject to recapture in the taxable year or years in which moneys or funds withdrawn from the first-time home buyer savings account were used for any purpose other than the payment of eligible costs by or on behalf of a qualified beneficiary, as provided under § 36-174. The amount subject to recapture shall be a portion of the amount withdrawn in the taxable year that was used for other than the payment of eligible costs, computed by multiplying the amount withdrawn and used for other than the payment of eligible costs by the ratio of the aggregate earnings in the account at the time of the withdrawal to the total balance in the account at such time.

However, recapture shall not apply to the extent of moneys or funds withdrawn that were (i) withdrawn by reason of the qualified beneficiary's death or disability; (ii) a disbursement of assets of the account pursuant to a filing for protection under the United States Bankruptcy Code, 11 U.S.C. §§ 101 through 1330; or (iii) transferred from an account established pursuant to Chapter 12 (§ 36-171 et seq.) of Title 36 into another account established pursuant to such chapter for the benefit of another qualified beneficiary.

For purposes of this subdivision, "account holder," "eligible costs," "first-time home buyer savings account," and "qualified beneficiary" mean the same as those terms are defined in § 36-171.

26. For taxable years beginning on and after January 1, 2015, any income for the taxable year attributable to the discharge of a student loan solely by reason of the student's death. For purposes of this subdivision, "student loan" means the same as that term is defined under § 108(f) of the Internal Revenue Code.

27. a. Income, including investment services partnership interest income (otherwise known as investment partnership carried interest income), attributable to an investment in a Virginia venture capital account. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2018, but before December 31, 2023. No subtraction shall be allowed under this subdivision for an investment in a company that is owned or operated by a

family member or an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision 24 or a tax credit under § 58.1-339.4 for the same investment.

b. As used in this subdivision 27:

"Qualified portfolio company" means a company that (i) has its principal place of business in the Commonwealth; (ii) has a primary purpose of production, sale, research, or development of a product or service other than the management or investment of capital; and (iii) provides equity in the company to the Virginia venture capital account in exchange for a capital investment. "Qualified portfolio company" does not include a company that is an individual or sole proprietorship.

"Virginia venture capital account" means an investment fund that has been certified by the Department as a Virginia venture capital account. In order to be certified as a Virginia venture capital account, the operator of the investment fund shall register the investment fund with the Department prior to December 31, 2023, (i) indicating that it intends to invest at least 50 percent of the capital committed to its fund in qualified portfolio companies and (ii) providing documentation that it employs at least one investor who has at least four years of professional experience in venture capital investment or substantially equivalent experience. "Substantially equivalent experience" includes, but is not limited to, an undergraduate degree from an accredited college or university in economics, finance, or a similar field of study. The Department may require an investment fund to provide documentation of the investor's training, education, or experience as deemed necessary by the Department to determine substantial equivalency. If the Department determines that the investment fund employs at least one investor with the experience set forth herein, the Department shall certify the investment fund as a Virginia venture capital account at such time as the investment fund actually invests at least 50 percent of the capital committed to its fund in qualified portfolio companies.

28. a. Income attributable to an investment in a Virginia real estate investment trust. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2019, but before December 31, 2024. No subtraction shall be allowed for an investment in a trust that is managed by a family member or an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision 24 or 27 or a tax credit under § 58.1-339.4 for the same investment.

b. As used in this subdivision 28:

"Distressed" means satisfying the criteria applicable to a locality described in subdivision E 2 of $\S 2.2-115$.

"Double distressed" means satisfying the criteria applicable to a locality described in subdivision E 3 of § 2.2-115.

"Virginia real estate investment trust" means a real estate investment trust, as defined in 26 U.S.C. § 856, that has been certified by the Department as a Virginia real estate investment trust. In order to be certified as a Virginia real estate investment trust, the trustee shall register the trust with the Department prior to December 31, 2024, indicating that it intends to invest at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed. If the Department determines that the trust satisfies the preceding criteria, the Department shall certify the trust as a Virginia real estate

investment trust at such time as the trust actually invests at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed.

29. For taxable years beginning on and after January 1, 2019, any gain recognized from the taking of real property by condemnation proceedings.

30. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, up to \$100,000 of all grant funds received by the taxpayer under the Rebuild Virginia program established by the Governor and administered by the Department of Small Business and Supplier Diversity.

§ 58.1-322.03. Virginia taxable income; deductions.

In computing Virginia taxable income pursuant to § 58.1-322, there shall be deducted from Virginia adjusted gross income as defined in § 58.1-321:

- 1. a. The amount allowable for itemized deductions for federal income tax purposes where the taxpayer has elected for the taxable year to itemize deductions on his federal return, but reduced by the amount of income taxes imposed by the Commonwealth or any other taxing jurisdiction and deducted on such federal return and increased by an amount that, when added to the amount deducted under § 170 of the Internal Revenue Code for mileage, results in a mileage deduction at the state level for such purposes at a rate of 18 cents per mile; or
- b. Provided that the taxpayer has not itemized deductions for the taxable year on his federal income tax return: (i) for taxable years beginning before January 1, 2019, and on and after January 1, 2026, \$3,000 for single individuals and \$6,000 for married persons (one-half of such amounts in the case of a married individual filing a separate return) and (ii) for taxable years beginning on and after January 1, 2019, but before January 1, 2026, \$4,500 for single individuals and \$9,000 for married persons (one-half of such amounts in the case of a married individual filing a separate return). For purposes of this section, any person who may be claimed as a dependent on another taxpayer's return for the taxable year may compute the deduction only with respect to earned income.
- 2. a. A deduction in the amount of \$930 for each personal exemption allowable to the taxpayer for federal income tax purposes.
- b. Each blind or aged taxpayer as defined under § 63(f) of the Internal Revenue Code shall be entitled to an additional personal exemption in the amount of \$800.
- The additional deduction for blind or aged taxpayers allowed under this subdivision shall be allowable regardless of whether the taxpayer itemizes deductions for the taxable year for federal income tax purposes.
- 3. A deduction equal to the amount of employment-related expenses upon which the federal credit is based under § 21 of the Internal Revenue Code for expenses for household and dependent care services necessary for gainful employment.
- 4. An additional \$1,000 deduction for each child residing for the entire taxable year in a home under permanent foster care placement as defined in § 63.2-908, provided that the taxpayer can also claim the child as a personal exemption under § 151 of the Internal Revenue Code.
- 5. a. A deduction in the amount of \$12,000 for individuals born on or before January 1, 1939.

b. A deduction in the amount of \$12,000 for individuals born after January 1, 1939, who have attained the age of 65. This deduction shall be reduced by \$1 for every \$1 that the taxpayer's adjusted federal adjusted gross income exceeds \$50,000 for single taxpayers or \$75,000 for married taxpayers. For married taxpayers filing separately, the deduction shall be reduced by \$1 for every \$1 that the total combined adjusted federal adjusted gross income of both spouses exceeds \$75,000.

For the purposes of this subdivision, "adjusted federal adjusted gross income" means federal adjusted gross income minus any benefits received under Title II of the Social Security Act and other benefits subject to federal income taxation solely pursuant to § 86 of the Internal Revenue Code, as amended.

- 6. The amount an individual pays as a fee for an initial screening to become a possible bone marrow donor, if (i) the individual is not reimbursed for such fee or (ii) the individual has not claimed a deduction for the payment of such fee on his federal income tax return.
- 7. a. A deduction shall be allowed to the purchaser or contributor for the amount paid or contributed during the taxable year for a prepaid tuition contract or college savings trust account entered into with the Virginia College Savings Plan, pursuant to Chapter 7 (§ 23.1-700 et seg.) of Title 23.1. Except as provided in subdivision b, the amount deducted on any individual income tax return in any taxable year shall be limited to \$4,000 per prepaid tuition contract or college savings trust account. No deduction shall be allowed pursuant to this subdivision 7 if such payments or contributions are deducted on the purchaser's or contributor's federal income tax return. If the purchase price or annual contribution to a college savings trust account exceeds \$4,000, the remainder may be carried forward and subtracted in future taxable years until the purchase price or college savings trust contribution has been fully deducted; however, except as provided in subdivision b, in no event shall the amount deducted in any taxable year exceed \$4,000 per contract or college savings trust account. Notwithstanding the statute of limitations on assessments contained in § 58.1-312, any deduction taken hereunder shall be subject to recapture in the taxable year or years in which distributions or refunds are made for any reason other than (i) to pay qualified higher education expenses, as defined in § 529 of the Internal Revenue Code or (ii) the beneficiary's death, disability, or receipt of a scholarship. For the purposes of this subdivision, "purchaser" or "contributor" means the person shown as such on the records of the Virginia College Savings Plan as of December 31 of the taxable year. In the case of a transfer of ownership of a prepaid tuition contract or college savings trust account, the transferee shall succeed to the transferor's tax attributes associated with a prepaid tuition contract or college savings trust account, including, but not limited to, carryover and recapture of deductions.
- b. A purchaser of a prepaid tuition contract or contributor to a college savings trust account who has attained age 70 shall not be subject to the limitation that the amount of the deduction not exceed \$4,000 per prepaid tuition contract or college savings trust account in any taxable year. Such taxpayer shall be allowed a deduction for the full amount paid for the contract or contributed to a college savings trust account, less any amounts previously deducted.

 8. The total amount an individual actually contributed in funds to the Virginia Public School Construction Grants Program and Fund, established in Chapter 11.1 (§ 22.1-175.1 et seq.) of Title 22.1, provided that the individual has not claimed a deduction for such amount on his

federal income tax return.

- 9. An amount equal to 20 percent of the tuition costs incurred by an individual employed as a primary or secondary school teacher licensed pursuant to Chapter 15 (§ 22.1-289.1 et seq.) of Title 22.1 to attend continuing teacher education courses that are required as a condition of employment; however, the deduction provided by this subdivision shall be available only if (i) the individual is not reimbursed for such tuition costs and (ii) the individual has not claimed a deduction for the payment of such tuition costs on his federal income tax return.
- 10. The amount an individual pays annually in premiums for long-term health care insurance, provided that the individual has not claimed a deduction for federal income tax purposes, or, for taxable years beginning before January 1, 2014, a credit under § 58.1-339.11. For taxable years beginning on and after January 1, 2014, no such deduction for long-term health care insurance premiums paid by the individual during the taxable year shall be allowed if the individual has claimed a federal income tax deduction for such taxable year for long-term health care insurance premiums paid by him.
- 11. Contract payments to a producer of quota tobacco or a tobacco quota holder, or their spouses, as provided under the American Jobs Creation Act of 2004 (P.L. 108-357), but only to the extent that such payments have not been subtracted pursuant to subsection D of § 58.1-402, as follows:
- a. If the payment is received in installment payments, then the recognized gain may be subtracted in the taxable year immediately following the year in which the installment payment is received.
- b. If the payment is received in a single payment, then 10 percent of the recognized gain may be subtracted in the taxable year immediately following the year in which the single payment is received. The taxpayer may then deduct an equal amount in each of the nine succeeding taxable years.
- 12. An amount equal to 20 percent of the sum paid by an individual pursuant to Chapter 6 (§ 58.1-600 et seq.), not to exceed \$500 in each taxable year, in purchasing for his own use the following items of tangible personal property: (i) any clothes washers, room air conditioners, dishwashers, and standard size refrigerators that meet or exceed the applicable energy star efficiency requirements developed by the U.S. Environmental Protection Agency and the U.S. Department of Energy; (ii) any fuel cell that (a) generates electricity using an electrochemical process, (b) has an electricity-only generation efficiency greater than 35 percent, and (c) has a generating capacity of at least two kilowatts; (iii) any gas heat pump that has a coefficient of performance of at least 1.25 for heating and at least 0.70 for cooling; (iv) any electric heat pump hot water heater that yields an energy factor of at least 1.7; (v) any electric heat pump that has a heating system performance factor of at least 8.0 and a cooling seasonal energy efficiency ratio of at least 13.0; (vi) any central air conditioner that has a cooling seasonal energy efficiency ratio of at least 13.5; (vii) any advanced gas or oil water heater that has an energy factor of at least 0.65; (viii) any advanced oil-fired boiler with a minimum annual fuel-utilization rating of 85; (ix) any advanced oil-fired furnace with a minimum annual fuel-utilization rating of 85; and (x) programmable thermostats.
- 13. The lesser of \$5,000 or the amount actually paid by a living donor of an organ or other living tissue for unreimbursed out-of-pocket expenses directly related to the donation that arose

within 12 months of such donation, provided that the donor has not taken a medical deduction in accordance with the provisions of § 213 of the Internal Revenue Code for such expenses. The deduction may be taken in the taxable year in which the donation is made or the taxable year in which the 12-month period expires.

- 14. For taxable years beginning on and after January 1, 2013, the amount an individual age 66 or older with earned income of at least \$20,000 for the year and federal adjusted gross income not in excess of \$30,000 for the year pays annually in premiums for (i) a prepaid funeral insurance policy covering the individual or (ii) medical or dental insurance for any person for whom individual tax filers may claim a deduction for such premiums under federal income tax laws. As used in this subdivision, "earned income" means the same as that term is defined in § 32(c) of the Internal Revenue Code. The deduction shall not be allowed for any portion of such premiums paid for which the individual has (a) been reimbursed, (b) claimed a deduction for federal income tax purposes, (c) claimed a deduction or subtraction under another provision of this section, or (d) claimed a federal income tax credit or any income tax credit pursuant to this chapter.
- 15. For taxable years beginning on and after January 1, 2018, 20 percent of business interest disallowed as a deduction pursuant to § 163(j) of the Internal Revenue Code. For purposes of this subdivision, "business interest" means the same as that term is defined under § 163(j) of the Internal Revenue Code.
- 16. For taxable years beginning on and after January 1, 2019, the actual amount of real and personal property taxes imposed by the Commonwealth or any other taxing jurisdiction not otherwise deducted solely on account of the dollar limitation imposed on individual deductions by § 164(b)(6)(B) of the Internal Revenue Code.
- 17. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, up to \$100,000 of the amount that is not deductible when computing federal adjusted gross income solely on account of the portion of subdivision B 10 of \S 58.1-301 related to Paycheck Protection Program loans.

§ 58.1-402. Virginia taxable income.

A. For purposes of this article, Virginia taxable income for a taxable year means the federal taxable income and any other income taxable to the corporation under federal law for such year of a corporation adjusted as provided in subsections B, C, D, E, and G, and H.

For a regulated investment company and a real estate investment trust, such term means the "investment company taxable income" and "real estate investment trust taxable income," respectively, to which shall be added in each case any amount of capital gains and any other income taxable to the corporation under federal law which shall be further adjusted as provided in subsections B, C, D, E, and G, and H.

- B. There shall be added to the extent excluded from federal taxable income:
- 1. Interest, less related expenses to the extent not deducted in determining federal taxable income, on obligations of any state other than Virginia, or of a political subdivision of any such other state unless created by compact or agreement to which the Commonwealth is a party;
- 2. Interest or dividends, less related expenses to the extent not deducted in determining federal

taxable income, on obligations or securities of any authority, commission or instrumentality of the United States, which the laws of the United States exempt from federal income tax but not from state income taxes;

- 3. [Repealed.]
- 4. The amount of any net income taxes and other taxes, including franchise and excise taxes, which are based on, measured by, or computed with reference to net income, imposed by the Commonwealth or any other taxing jurisdiction, to the extent deducted in determining federal taxable income;
- 5. Unrelated business taxable income as defined by § 512 of the Internal Revenue Code;
- 6. [Repealed.]
- 7. The amount required to be included in income for the purpose of computing the partial tax on an accumulation distribution pursuant to § 667 of the Internal Revenue Code;
- 8. a. For taxable years beginning on and after January 1, 2004, the amount of any intangible expenses and costs directly or indirectly paid, accrued, or incurred to, or in connection directly or indirectly with one or more direct or indirect transactions with one or more related members to the extent such expenses and costs were deductible or deducted in computing federal taxable income for Virginia purposes. This addition shall not be required for any portion of the intangible expenses and costs if one of the following applies:
- (1) The corresponding item of income received by the related member is subject to a tax based on or measured by net income or capital imposed by Virginia, another state, or a foreign government that has entered into a comprehensive tax treaty with the United States government;
- (2) The related member derives at least one-third of its gross revenues from the licensing of intangible property to parties who are not related members, and the transaction giving rise to the expenses and costs between the corporation and the related member was made at rates and terms comparable to the rates and terms of agreements that the related member has entered into with parties who are not related members for the licensing of intangible property; or
- (3) The corporation can establish to the satisfaction of the Tax Commissioner that the intangible expenses and costs meet both of the following: (i) the related member during the same taxable year directly or indirectly paid, accrued or incurred such portion to a person who is not a related member, and (ii) the transaction giving rise to the intangible expenses and costs between the corporation and the related member did not have as a principal purpose the avoidance of any portion of the tax due under this chapter.
- b. A corporation required to add to its federal taxable income intangible expenses and costs pursuant to subdivision a may petition the Tax Commissioner, after filing the related income tax return for the taxable year and remitting to the Tax Commissioner all taxes, penalties, and interest due under this article for such taxable year including tax upon any amount of intangible expenses and costs required to be added to federal taxable income pursuant to subdivision a, to consider evidence relating to the transaction or transactions between the corporation and a related member or members that resulted in the corporation's taxable income being increased, as required under subdivision a, for such intangible expenses and costs.

If the corporation can demonstrate to the Tax Commissioner's sole satisfaction, by clear and convincing evidence, that the transaction or transactions between the corporation and a related member or members resulting in such increase in taxable income pursuant to subdivision a had

a valid business purpose other than the avoidance or reduction of the tax due under this chapter, the Tax Commissioner shall permit the corporation to file an amended return. For purposes of such amended return, the requirements of subdivision a shall not apply to any transaction for which the Tax Commissioner is satisfied (and has identified) that the transaction had a valid business purpose other than the avoidance or reduction of the tax due under this chapter. Such amended return shall be filed by the corporation within one year of the written permission granted by the Tax Commissioner and any refund of the tax imposed under this article shall include interest at a rate equal to the rate of interest established under § 58.1-15 and such interest shall accrue as provided under § 58.1-1833. However, upon the filing of such amended return, any related member of the corporation that subtracted from taxable income amounts received pursuant to subdivision C 21 shall be subject to the tax imposed under this article on that portion of such amounts for which the corporation has filed an amended return pursuant to this subdivision. In addition, for such transactions identified by the Tax Commissioner herein by which he has been satisfied by clear and convincing evidence, the Tax Commissioner may permit the corporation in filing income tax returns for subsequent taxable years to deduct the related intangible expenses and costs without making the adjustment under subdivision a.

The Tax Commissioner may charge a fee for all direct and indirect costs relating to the review of any petition pursuant to this subdivision, to include costs necessary to secure outside experts in evaluating the petition. The Tax Commissioner may condition the review of any petition pursuant to this subdivision upon payment of such fee.

No suit for the purpose of contesting any action of the Tax Commissioner under this subdivision shall be maintained in any court of this Commonwealth.

- c. Nothing in subdivision B 8 shall be construed to limit or negate the Department's authority under § 58.1-446;
- 9. a. For taxable years beginning on and after January 1, 2004, the amount of any interest expenses and costs directly or indirectly paid, accrued, or incurred to, or in connection directly or indirectly with one or more direct or indirect transactions with one or more related members to the extent such expenses and costs were deductible or deducted in computing federal taxable income for Virginia purposes. This addition shall not be required for any portion of the interest expenses and costs, if:
- (1) The related member has substantial business operations relating to interest-generating activities, in which the related member pays expenses for at least five full-time employees who maintain, manage, defend or are otherwise responsible for operations or administration relating to the interest-generating activities; and
- (2) The interest expenses and costs are not directly or indirectly for, related to or in connection with the direct or indirect acquisition, maintenance, management, sale, exchange, or disposition of intangible property; and
- (3) The transaction giving rise to the expenses and costs between the corporation and the related member has a valid business purpose other than the avoidance or reduction of taxation and payments between the parties are made at arm's length rates and terms; and
- (4) One of the following applies:
- (i) The corresponding item of income received by the related member is subject to a tax based on or measured by net income or capital imposed by Virginia, another state, or a foreign

government that has entered into a comprehensive tax treaty with the United States government; (ii) Payments arise pursuant to a pre-existing contract entered into when the parties were not related members provided the payments continue to be made at arm's length rates and terms; (iii) The related member engages in transactions with parties other than related members that generate revenue in excess of \$2 million annually; or

(iv) The transaction giving rise to the interest payments between the corporation and a related member was done at arm's length rates and terms and meets any of the following: (a) the related member uses funds that are borrowed from a party other than a related member or that are paid, incurred or passed-through to a person who is not a related member; (b) the debt is part of a regular and systematic funds management or portfolio investment activity conducted by the related member, whereby the funds of two or more related members are aggregated for the purpose of achieving economies of scale, the internal financing of the active business operations of members, or the benefit of centralized management of funds; (c) financing the expansion of the business operations; or (d) restructuring the debt of related members, or the pass-through of acquisition-related indebtedness to related members.

b. A corporation required to add to its federal taxable income interest expenses and costs pursuant to subdivision a may petition the Tax Commissioner, after filing the related income tax return for the taxable year and remitting to the Tax Commissioner all taxes, penalties, and interest due under this article for such taxable year including tax upon any amount of interest expenses and costs required to be added to federal taxable income pursuant to subdivision a, to consider evidence relating to the transaction or transactions between the corporation and a related member or members that resulted in the corporation's taxable income being increased, as required under subdivision a, for such interest expenses and costs.

If the corporation can demonstrate to the Tax Commissioner's sole satisfaction, by clear and convincing evidence, that the transaction or transactions between the corporation and a related member or members resulting in such increase in taxable income pursuant to subdivision a had a valid business purpose other than the avoidance or reduction of the tax due under this chapter and that the related payments between the parties were made at arm's length rates and terms, the Tax Commissioner shall permit the corporation to file an amended return. For purposes of such amended return, the requirements of subdivision a shall not apply to any transaction for which the Tax Commissioner is satisfied (and has identified) that the transaction had a valid business purpose other than the avoidance or reduction of the tax due under this chapter and that the related payments between the parties were made at arm's length rates and terms. Such amended return shall be filed by the corporation within one year of the written permission granted by the Tax Commissioner and any refund of the tax imposed under this article shall include interest at a rate equal to the rate of interest established under § 58.1-15 and such interest shall accrue as provided under § 58.1-1833. However, upon the filing of such amended return, any related member of the corporation that subtracted from taxable income amounts received pursuant to subdivision C 21 shall be subject to the tax imposed under this article on that portion of such amounts for which the corporation has filed an amended return pursuant to this subdivision. In addition, for such transactions identified by the Tax Commissioner herein by which he has been satisfied by clear and convincing evidence, the Tax Commissioner may permit the corporation in filing income tax returns for subsequent taxable years to deduct the related interest expenses and costs without making the adjustment under subdivision a.

The Tax Commissioner may charge a fee for all direct and indirect costs relating to the review of any petition pursuant to this subdivision, to include costs necessary to secure outside experts in evaluating the petition. The Tax Commissioner may condition the review of any petition pursuant to this subdivision upon payment of such fee.

No suit for the purpose of contesting any action of the Tax Commissioner under this subdivision shall be maintained in any court of this Commonwealth.

- c. Nothing in subdivision B 9 shall be construed to limit or negate the Department's authority under § 58.1-446.
- d. For purposes of subdivision B 9:

"Arm's-length rates and terms" means that (i) two or more related members enter into a written agreement for the transaction, (ii) such agreement is of a duration and contains payment terms substantially similar to those that the related member would be able to obtain from an unrelated entity, (iii) the interest is at or below the applicable federal rate compounded annually for debt instruments under § 1274(d) of the Internal Revenue Code that was in effect at the time of the agreement, and (iv) the borrower or payor adheres to the payment terms of the agreement governing the transaction or any amendments thereto.

"Valid business purpose" means one or more business purposes that alone or in combination constitute the motivation for some business activity or transaction, which activity or transaction improves, apart from tax effects, the economic position of the taxpayer, as further defined by regulation.

- 10. a. For taxable years beginning on and after January 1, 2009, the amount of dividends deductible under §§ 561 and 857 of the Internal Revenue Code by a Captive Real Estate Investment Trust (REIT). For purposes of this subdivision, a REIT is a Captive REIT if:
- (1) It is not regularly traded on an established securities market;
- (2) More than 50 percent of the voting power or value of beneficial interests or shares of which, at any time during the last half of the taxable year, is owned or controlled, directly or indirectly, by a single entity that is (i) a corporation or an association taxable as a corporation under the Internal Revenue Code; and (ii) not exempt from federal income tax pursuant to § 501(a) of the Internal Revenue Code; and
- (3) More than 25 percent of its income consists of rents from real property as defined in § 856(d) of the Internal Revenue Code.
- b. For purposes of applying the ownership test of subdivision 10 a (2), the following entities shall not be considered a corporation or an association taxable as a corporation:
- (1) Any REIT that is not treated as a Captive REIT;
- (2) Any REIT subsidiary under § 856 of the Internal Revenue Code other than a qualified REIT subsidiary of a Captive REIT:
- (3) Any Listed Australian Property Trust, or an entity organized as a trust, provided that a Listed Australian Property Trust owns or controls, directly or indirectly, 75 percent or more of the voting or value of the beneficial interests or shares of such trust; and
- (4) Any Qualified Foreign Entity.
- c. For purposes of subdivision B 10, the constructive ownership rules prescribed under § 318(a) of the Internal Revenue Code, as modified by § 856(d)(5) of the Internal Revenue Code, shall

apply in determining the ownership of stock, assets, or net profits of any person.

- d. For purposes of subdivision B 10:
- "Listed Australian Property Trust" means an Australian unit trust registered as a Management Investment Scheme, pursuant to the Australian Corporations Act, in which the principal class of units is listed on a recognized stock exchange in Australia and is regularly traded on an established securities market.
- "Qualified Foreign Entity" means a corporation, trust, association or partnership organized outside the laws of the United States and that satisfies all of the following criteria:
- (1) At least 75 percent of the entity's total asset value at the close of its taxable year is represented by real estate assets, as defined in § 856(c)(5)(B) of the Internal Revenue Code, thereby including shares or certificates of beneficial interest in any REIT, cash and cash equivalents, and U.S. Government securities;
- (2) The entity is not subject to a tax on amounts distributed to its beneficial owners, or is exempt from entity level tax;
- (3) The entity distributes, on an annual basis, at least 85 percent of its taxable income, as computed in the jurisdiction in which it is organized, to the holders of its shares or certificates of beneficial interest;
- (4) The shares or certificates of beneficial interest of such entity are regularly traded on an established securities market or, if not so traded, not more than 10 percent of the voting power or value in such entity is held directly, indirectly, or constructively by a single entity or individual; and
- (5) The entity is organized in a country that has a tax treaty with the United States.
- e. For taxable years beginning on or after January 1, 2016, for purposes of subdivision B 10, any voting power or value of the beneficial interests or shares in a REIT that is held in a segregated asset account of a life insurance corporation as described in § 817 of the Internal Revenue Code shall not be taken into consideration when determining if such REIT is a Captive REIT.
- 11. For taxable years beginning on or after January 1, 2016, to the extent that tax credit is allowed for the same donation pursuant to § 58.1-439.12:12, any amount claimed as a federal income tax deduction for such donation under § 170 of the Internal Revenue Code, as amended or renumbered.
- C. There shall be subtracted to the extent included in and not otherwise subtracted from federal taxable income:
- 1. Income derived from obligations, or on the sale or exchange of obligations, of the United States and on obligations or securities of any authority, commission or instrumentality of the United States to the extent exempt from state income taxes under the laws of the United States including, but not limited to, stocks, bonds, treasury bills, and treasury notes, but not including interest on refunds of federal taxes, interest on equipment purchase contracts, or interest on other normal business transactions.
- 2. Income derived from obligations, or on the sale or exchange of obligations of this Commonwealth or of any political subdivision or instrumentality of this Commonwealth.
- 3. Dividends upon stock in any domestic international sales corporation, as defined by § 992 of the Internal Revenue Code, 50 percent or more of the income of which was assessable for the

preceding year, or the last year in which such corporation has income, under the provisions of the income tax laws of the Commonwealth.

- 4. The amount of any refund or credit for overpayment of income taxes imposed by this Commonwealth or any other taxing jurisdiction.
- 5. Any amount included therein by the operation of the provisions of § 78 of the Internal Revenue Code (foreign dividend gross-up).
- 6. The amount of wages or salaries eligible for the federal Targeted Jobs Credit which was not deducted for federal purposes on account of the provisions of § 280C(a) of the Internal Revenue Code.
- 7. Any amount included therein by the operation of § 951 of the Internal Revenue Code (subpart F income) or, for taxable years beginning on and after January 1, 2018, § 951A of the Internal Revenue Code (Global Intangible Low-Taxed Income).
- 8. Any amount included therein which is foreign source income as defined in § 58.1-302.
- 9. [Repealed.]
- 10. The amount of any dividends received from corporations in which the taxpaying corporation owns 50 percent or more of the voting stock.
- 11. [Repealed.]
- 12, 13. [Expired.]
- 14. For taxable years beginning on or after January 1, 1995, the amount for "qualified research expenses" or "basic research expenses" eligible for deduction for federal purposes, but which were not deducted, on account of the provisions of § 280C(c) of the Internal Revenue Code.
- 15. For taxable years beginning on or after January 1, 2000, the total amount actually contributed in funds to the Virginia Public School Construction Grants Program and Fund established in Chapter 11.1 (§ 22.1-175.1 et seq.) of Title 22.1.
- 16. For taxable years beginning on or after January 1, 2000, but before January 1, 2015, the gain derived from the sale or exchange of real property or the sale or exchange of an easement to real property which results in the real property or the easement thereto being devoted to open-space use, as that term is defined in § 58.1-3230, for a period of time not less than 30 years. To the extent a subtraction is taken in accordance with this subdivision, no tax credit under this chapter for donating land for its preservation shall be allowed for three years following the year in which the subtraction is taken.
- 17. For taxable years beginning on and after January 1, 2001, any amount included therein with respect to § 58.1-440.1.
- 18. For taxable years beginning on and after January 1, 1999, income received as a result of (i) the "Master Settlement Agreement," as defined in § 3.2-3100; and (ii) the National Tobacco Grower Settlement Trust dated July 19, 1999, by (a) tobacco farming businesses; (b) any business holding a tobacco marketing quota, or tobacco farm acreage allotment, under the Agricultural Adjustment Act of 1938; or (c) any business having the right to grow tobacco pursuant to such a quota allotment.
- 19, 20. [Repealed.]
- 21. For taxable years beginning on and after January 1, 2004, any amount of intangible expenses and costs or interest expenses and costs added to the federal taxable income of a corporation pursuant to subdivision B 8 or B 9 shall be subtracted from the federal taxable

income of the related member that received such amount if such related member is subject to Virginia income tax on the same amount.

- 22. For taxable years beginning on and after January 1, 2009, any gain recognized from the sale of launch services to space flight participants, as defined in 49 U.S.C. § 70102, or launch services intended to provide individuals the training or experience of a launch, without performing an actual launch. To qualify for a deduction under this subdivision, launch services must be performed in Virginia or originate from an airport or spaceport in Virginia.
- 23. For taxable years beginning on and after January 1, 2009, any gain recognized as a result of resupply services contracts for delivering payload, as defined in 49 U.S.C. § 70102, entered into with the Commercial Orbital Transportation Services division of the National Aeronautics and Space Administration or other space flight entity, as defined in § 8.01-227.8, and launched from an airport or spaceport in Virginia.
- 24. For taxable years beginning on or after January 1, 2011, any income taxed as a long-term capital gain for federal income tax purposes, or any income taxed as investment services partnership interest income (otherwise known as investment partnership carried interest income) for federal income tax purposes. To qualify for a subtraction under this subdivision, such income must be attributable to an investment in a "qualified business," as defined in § 58.1-339.4, or in any other technology business approved by the Secretary of Administration, provided the business has its principal office or facility in the Commonwealth and less than \$3 million in annual revenues in the fiscal year prior to the investment. To qualify for a subtraction under this subdivision, the investment must be made between the dates of April 1, 2010, and June 30, 2020. No taxpayer who has claimed a tax credit for an investment in a "qualified business" under § 58.1-339.4 shall be eligible for the subtraction under this subdivision for an investment in the same business.
- 25. a. Income, including investment services partnership interest income (otherwise known as investment partnership carried interest income), attributable to an investment in a Virginia venture capital account. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2018, but before December 31, 2023. No subtraction shall be allowed under this subdivision for an investment in a company that is owned or operated by an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision C 24 for the same investment.

b. As used in this subdivision 25:

"Qualified portfolio company" means a company that (i) has its principal place of business in the Commonwealth; (ii) has a primary purpose of production, sale, research, or development of a product or service other than the management or investment of capital; and (iii) provides equity in the company to the Virginia venture capital account in exchange for a capital investment. "Qualified portfolio company" does not include a company that is an individual or sole proprietorship.

"Virginia venture capital account" means an investment fund that has been certified by the Department as a Virginia venture capital account. In order to be certified as a Virginia venture capital account, the operator of the investment fund shall register the investment fund with the Department prior to December 31, 2023, (i) indicating that it intends to invest at least 50 percent of the capital committed to its fund in qualified portfolio companies and (ii) providing

documentation that it employs at least one investor who has at least four years of professional experience in venture capital investment or substantially equivalent experience. "Substantially equivalent experience" includes, but is not limited to, an undergraduate degree from an accredited college or university in economics, finance, or a similar field of study. The Department may require an investment fund to provide documentation of the investor's training, education, or experience as deemed necessary by the Department to determine substantial equivalency. If the Department determines that the investment fund employs at least one investor with the experience set forth herein, the Department shall certify the investment fund as a Virginia venture capital account at such time as the investment fund actually invests at least 50 percent of the capital committed to its fund in qualified portfolio companies.

26. a. Income attributable to an investment in a Virginia real estate investment trust. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2019, but before December 31, 2024. No subtraction shall be allowed for an investment in a trust that is managed by an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision C 24 or 25 for the same investment.

b. As used in this subdivision 26:

"Distressed" means satisfying the criteria applicable to a locality described in subdivision E 2 of § 2.2-115.

"Double distressed" means satisfying the criteria applicable to a locality described in subdivision E 3 of § 2.2-115.

"Virginia real estate investment trust" means a real estate investment trust, as defined in 26 U.S.C. § 856, that has been certified by the Department as a Virginia real estate investment trust. In order to be certified as a Virginia real estate investment trust, the trustee shall register the trust with the Department prior to December 31, 2024, indicating that it intends to invest at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed. If the Department determines that the trust satisfies the preceding criteria, the Department shall certify the trust as a Virginia real estate investment trust at such time as the trust actually invests at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed.

- 27. For taxable years beginning on and after January 1, 2019, any gain recognized from the taking of real property by condemnation proceedings.
- 28. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, up to \$100,000 of all grant funds received by the taxpayer under the Rebuild Virginia program established by the Governor and administered by the Department of Small Business and Supplier Diversity.
- D. For taxable years beginning on and after January 1, 2006, there shall be subtracted from federal taxable income contract payments to a producer of quota tobacco or a tobacco quota holder as provided under the American Jobs Creation Act of 2004 (P.L. 108-357) as follows:

 1. If the payment is received in installment payments, then the recognized gain, including any gain recognized in taxable year 2005, may be subtracted in the taxable year immediately following the year in which the installment payment is received.

- 2. If the payment is received in a single payment, then 10 percent of the recognized gain may be subtracted in the taxable year immediately following the year in which the single payment is received. The taxpayer may then deduct an equal amount in each of the nine succeeding taxable years.
- E. Adjustments to federal taxable income shall be made to reflect the transitional modifications provided in § 58.1-315.
- F. Notwithstanding any other provision of law, the income from any disposition of real property which is held by the taxpayer for sale to customers in the ordinary course of the taxpayer's trade or business, as defined in § 453(1)(1)(B) of the Internal Revenue Code, of property made on or after January 1, 2009, may, at the election of the taxpayer, be recognized under the installment method described under § 453 of the Internal Revenue Code, provided that (i) the election relating to the dealer disposition of the property has been made on or before the due date prescribed by law (including extensions) for filing the taxpayer's return of the tax imposed under this chapter for the taxable year in which the disposition occurs, and (ii) the dealer disposition is in accordance with restrictions or conditions established by the Department, which shall be set forth in guidelines developed by the Department. Along with such restrictions or conditions, the guidelines shall also address the recapture of such income under certain circumstances. The development of the guidelines shall be exempt from the Administrative Process Act (§ 2.2-4000 et seq.).
- G. For taxable years beginning on and after January 1, 2018, there shall be deducted to the extent included in and not otherwise subtracted from federal taxable income 20 percent of business interest disallowed as a deduction pursuant to § 163(j) of the Internal Revenue Code. For purposes of this subsection, "business interest" means the same as that term is defined under § 163(j) of the Internal Revenue Code.
- H. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, there shall be deducted to the extent not otherwise subtracted from federal taxable income up to \$100,000 of the amount that is not deductible when computing federal taxable income solely on account of the portion of subdivision B 10 of § 58.1-301 related to Paycheck Protection Program loans."

Page 736, strike lines 1 through 32.

Explanation:

(This amendment advances Virginia's date of conformity with the Internal Revenue Code from December 31, 2019, to December 31, 2020. The amendment generally conforms Virginia's tax code to both the federal Coronavirus Aid, Relief, and Economic Security (CARES) Act and the Consolidated Appropriations Act (CAA), 2021, with certain exceptions consistent with SB 1146. The amendment provides a deduction of up to \$100,000 for business expenses funded by forgiven loans under the Paycheck Protection Program (PPP) and provides an income tax subtraction of up to \$100,000 for grant funds received under the Rebuild Virginia program.)

Item 4-14 #4s

Effective Date

Language

Language:

Page 732, after line 11, insert:

"I. Notwithstanding paragraph G. above or any other provision of law, the Chairs of the Virginia Redistricting Commission shall keep the Senate President Pro Tempore, the Senate Minority Leader, the Speaker of the House of Delegates, the House Minority Leader, and the Governor informed about the timing of availability of United States Bureau of the Census data as it relates to the tabulation of the population for reapportionment purposes pursuant to P.L. 94-171, and options for redistricting and its impact on elections for the House of Delegates."

Page 732, after line 38, insert:

"F. Notwithstanding paragraph C. above, or any other provision of law, the daily compensation and reimbursement for reasonable and necessary expenses for legislative and non-legislative members of the Virginia Redistricting Commission for attendance at an official meeting shall be set at the same amounts provided for legislative members in paragraphs B.4.d. and B.5. of Item 1 of the this act."

Explanation:

(This amendment provides for and directs the Chairs of the Virginia Redistricting Commission to communicate information related to the availability of U.S. Census data and its impact on elections for the House of Delegates, and includes language clarifying that members of the Virginia Redistricting Commission shall receive daily compensation and reimbursement for reasonable and necessary expenses for attendance at an official meeting equal to that which is generally provided to members for attendance at any official meeting.)

The reading of the amendments was waived.

On motion of Senator Howell, the uncontested committee amendments were agreed to.

Item 9 #1s was taken up and, on motion of Senator Petersen, was agreed to.

Item 40 #1s was taken up and, on motion of Senator Petersen, was agreed to.

Item 41 #1s was taken up and, on motion of Senator Petersen, was agreed to.

Item 42 #1s was taken up and, on motion of Senator Petersen, was agreed to.

Item 43 #1s was taken up and, on motion of Senator Petersen, was agreed to.

Item 48 #2s was taken up and, on motion of Senator Petersen, was agreed to.

Item 48 #3s was taken up and, on motion of Senator Petersen, was agreed to.

Item 57 #1s was taken up and, on motion of Senator Petersen, was agreed to.

Item 73 #1s was taken up and, on motion of Senator Petersen, was agreed to.

Item 73 #2s was taken up and, on motion of Senator Petersen, was agreed to.

Item 105 #1s was taken up and, on motion of Senator Marsden, was agreed to.

Item 112 #1s was taken up and, on motion of Senator Marsden, was agreed to.

Item 112 #2s was taken up and, on motion of Senator Marsden, was agreed to.

Item 130 #1s was taken up and, on motion of Senator Marsden, was agreed to.

Item 130 #2s was taken up and, on motion of Senator Marsden, was agreed to.

Item 135 #1s was taken up and, on motion of Senator Marsden, was agreed to.

Item 135 #2s was taken up and, on motion of Senator Marsden, was agreed to.

Item 196 #1s was taken up and, on motion of Senator Locke, was agreed to.

Senator Howell moved that the Rules be suspended and the third reading of the title of **S.B. 1100** as required by Article IV, Section 11, of the Constitution, be dispensed with.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1100, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--31.

NAYS--Chase, Cosgrove, Kiggans, McDougle, Obenshain, Peake, Stanley, Suetterlein--8. RULE 36--0.

STATEMENTS ON VOTE

Senator DeSteph stated that he was abstaining pursuant to Rule 36 on Item 105 #1s, but voting on **S.B. 1100** as a whole.

Senator Edwards stated that he was abstaining pursuant to Rule 36 on Item 266 #1s, but voting on **S.B. 1100** as a whole.

Senator Locke stated that she was abstaining pursuant to Rule 36 on Item 303, but voting on **S.B. 1100** as a whole.

Senator Lucas stated that she was abstaining pursuant to Rule 36 on Item 313, Item 313 #4s, Item 320 #1s, Item 479.20 #1s, and Item 482.20, but voting on **S.B. 1100** as a whole.

Senator McClellan stated that she was abstaining pursuant to Rule 36 on Item 114, Item 114 #5s, and Item 479.10, but voting on **S.B. 1100** as a whole.

Senator McDougle stated that he was abstaining pursuant to Rule 36 on Item 214 #1s, but voting on **S.B. 1100** as a whole.

Senator McPike stated that he was abstaining pursuant to Rule 36 on Item 385, Item 386 #1s, and Item C-70, but voting on **S.B. 1100** as a whole.

Senator Norment stated that he was abstaining pursuant to Rule 36 on Item 266 #1s and Item 313, but voting on **S.B. 1100** as a whole.

Senator Petersen stated that he was abstaining pursuant to Rule 36 on Item 105 #1s and Item 266 #1s, but voting on **S.B. 1100** as a whole.

Senator Stuart stated that he was abstaining pursuant to Rule 36 on Item 374 #4s, but voting on **S.B.** 1100 as a whole.

Senator Surovell stated that he was abstaining pursuant to Rule 36 on Item 135, Item 374, and Item 374 #3s, but voting on **S.B. 1100** as a whole.

Senator DeSteph stated that he wished to be recorded as voting nay on the question of agreeing to Item 36~#1s, Item 40~#1s, Item 57~#1s, Item 73~#1s, Item 73~#2s, Item 113~#1s, Item 114~#3s, Item 134~#2s, Item 138~#3s, Item 291~#1s, Item 312~#1s, Item 313~#17s, Item 350~#3s, Item 356~#1s, Item 356~#2s, Item 356~#3s, Item 356~#4s, Item 359~#1s, Item 385~#1s, Item 391~#1s, Item 391~#2s, Item 394~#2s, Item 400~#2s, Item 400~#4s, Item 402~#1s, Item 402~#3s, Item 425~#2s, Item 426~#2s, Item 479~#4s, Item 4-14~#2s to 8.B. 1100.

Senator Dunnavant stated that she wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 391 #2s, and Item 402 #1s to **S.B. 1100.**

Senator Hanger stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 391 #2s, Item 479 #4s, and Section 4.5.04 of part 4 to **S.B. 1100.**

Senator Newman stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 134 #2s, Item 138 #3s, Item 139 #1s, Item 150, Item 291 #1s, Item 313 #17s, Item 350 #3s, Item 356 #1s, Item 374 #2s, Item 391 #2s, Item 394 #2s, Item 402 #1s, Item 425 #2s, Item 479 #4s, Section 4.5.04 of part 4, and Item 302 F 1-4 to **S.B. 1100.**

Senator Norment stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, and Item C-22.10 #1s to **S.B. 1100.**

Senator Pillion stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 391 #2s, Item 402 #1s, Item 425 #2s, Item 479 #4s, and Section 4.5.04 of part 4 to **S.B. 1100.**

Senator Reeves stated that he wished to be recorded as voting nay on the question of agreeing to Item 36 #1s, Item 40 #1s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 113 #1s, Item 114 #3s, Item 134 #2s, Item 138 #3s, Item 291 #1s, Item 312 #1s, Item 313 #17s, Item 350 #3s, Item 356 #1s, Item 356 #2s, Item 356 #2s, Item 356 #4s, Item 359 #1s, Item 374 #2s, Item 385 #1s, Item 391 #1s, Item 391 #2s, Item 394 #2s, Item 400 #2s, Item 400 #4s, Item 402 #1s, Item 402 #3s, Item 425 #2s, Item 426 #2s, Item 479 #4s, Item 4-14 #1s, Item 4-14 #2s, Section 4.5.04 of part 4, and Item 302 F 1-4 to **S.B. 1100.**

Senator Ruff stated that he wished to be recorded as voting nay on the question of agreeing to Item $40 \, #1s$, Item $57 \, #1s$, Item $73 \, #1s$, Item $73 \, #2s$, Item $139 \, #1s$, Item $145 \, #9s$, Item $164 \, #1s$, Item $291 \, #1s$, Item $312 \, #1s$, Item $313 \, #17s$, Item $391 \, #1s$, Item $391 \, #2s$, Item $492 \, #1s$, Item 492

Senator Stuart stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 134 #2s, Item 138 #3s, Item 313 #17s, Item 374 #2s, Item 391 #2s, Item 402 #1s, Item 425 #2s, Item 479 #4s, Section 4.5.04 of part 4, and Item 302 F 1-4 to **S.B. 1100.**

Senator Vogel stated that she wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 402 #1s, and Item 425 #2s to **S.B. 1100.**

On motion of Senator Lucas, the Senate adjourned until Monday, February 15, 2021, at 12 m.

The Delma

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

MONDAY, FEBRUARY 15, 2021

The Senate met at 12 m. and was called to order by the President pro tempore, Senator L. Louise Lucas.

The Honorable Ghazala F. Hashmi, Tenth Senatorial District, offered the following prayer:

I offer today a prayer that was composed by Muslim leaders in Great Britain, and I give attribution to them here by acknowledging their names: Dr. Waqar Azmi, Imam Shahid Raza, Mustafa Field, and Maulana Qamaruzzaman Azmi.

A Muslim Prayer for Our Nation

Bismillah-ir-Rahman-ir-Rahim

In the name of God, the Most Gracious and the Most Merciful:

O Lord, unite our nation around the principles of justice, peace, love, and faith.

Put peace and love in our hearts for the diversity that makes our nation so beautiful.

Most Merciful, we pray for our nation to remain tolerant and loving, remove prejudice from our hearts, and allow us to love our brothers and sisters in humanity.

Allow our government to remain accountable to the people, give [our leaders] vision and wisdom, as they take decisions affecting peace in our world so that they may uphold peace in the world, advance the welfare of our nation, and deal kindly and justly with all our communities.

O Lord, Most Strong, give us the strength to protect and care for our neighbors.

Make our hearts and minds aware of our heritage, fulfilling duties and responsibilities as a citizen.

Lord, Most Merciful, allow us to show kindness to those most vulnerable in society.

Protect us from evil, inspire and guide us in defending those open to abuse.

Lord, Most Generous, allow us to give in charitable activity, and to help those most in need.

Make us more conscious of what is not good for us.

Lord, Most Merciful, Most Generous, please give us the patience to continue to learn from one another and work towards a more peaceful and kind world.

Make our hearts generous so that we may treat others as we wish to be treated ourselves. Help us to share that which we have with others, for Your sake. Strengthen us, love us, and be kind to us all. Ameen.

The Pledge of Allegiance to the Flag of the United States of America was led by Michael Jackson, Assistant Coordinator of Committee Operations, Senate Clerk's Office.

THE PRESIDENT PRESIDING

The President assumed the Chair.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

After the roll call, Senator Mason notified the Clerk of his presence.

On motion of Senator Bell, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--34. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, DeSteph, Petersen, Stanley--4. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 12, 2021

THE HOUSE OF DELEGATES HAS PASSED WITH A SUBSTITUTE THE FOLLOWING SENATE BILL:

S.B. 1155. A BILL to create a six-year capital outlay plan for projects to be funded entirely or partially from general fund–supported resources and to repeal Chapter 1134 of the Acts of Assembly of 2020.

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING HOUSE BILL:

H.B. 1800. A BILL to amend and reenact Chapter 1289 of the 2020 Acts of Assembly, as amended by Chapter 56 of the 2020 Acts of Assembly, Special Session I, which appropriated funds for the 2020-22 Biennium and provided a portion of revenues for the two years ending, respectively, on the thirtieth day of June, 2021, and the thirtieth day of June, 2022.

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING SENATE BILLS:

- **S.B. 1134.** A BILL to amend and reenact §§ 2 and 4 of the first enactment of Chapter 265 and §§ 2 and 4 of the first enactment of Chapter 408 of the Acts of Assembly of 1992, relating to the issuance of Commonwealth of Virginia Article X, Section 9 (c) Refunding Bonds, subject to the provisions of Article X, Sections 9 (a) and 9 (c) of the Constitution of Virginia; emergency. EMERGENCY
- **S.B. 1145.** A BILL to authorize the issuance of bonds, in an amount up to \$34,136,000 plus financing costs, pursuant to Article X, Section 9 (c) of the Constitution of Virginia, for paying costs of acquiring, constructing and equipping revenue-producing capital projects at institutions of higher learning of the Commonwealth; to authorize the Treasury Board, by and with the consent of the Governor, to fix the details of such bonds, to provide for the sale of such bonds, and to issue notes to borrow money in anticipation of the issuance of the bonds; to provide for the pledge of the net revenues of such capital projects and the full faith, credit and taxing power of the Commonwealth for the payment of such bonds; to provide that the interest income on such bonds and notes shall be exempt from all taxation by the Commonwealth and any political subdivision thereof; emergency. EMERGENCY
- **S.B. 1156.** A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 22.17, consisting of a section numbered 59.1-284.38, relating to creation of the Technology Development Grant Fund.

S.B. 1251. A BILL to amend and reenact §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308 of the Code of Virginia, relating to Virginia Retirement System; technical amendments.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows: YEAS--33. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McPike, Morrissey, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--33.

NAYS--Deeds, McDougle, Norment, Petersen, Stanley--5. RULE 36--0.

The House bill communicated as passed by the House of Delegates, the first reading of its title required by the Constitution having been dispensed with, was referred as follows:

H.B. 1800 was referred to the Committee on Finance and Appropriations.

COMMITTEE REPORTS

Senator Edwards, from the Committee on the Judiciary, presented the following reports:

SENATE OF VIRGINIA

February 15, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following persons are qualified to be elected to the respective circuit court judgeships as follows:

The Honorable Robert G. MacDonald, of Chesapeake, as a judge of the First Judicial Circuit for a term of eight years commencing March 16, 2021.

The Honorable Tyneka L. D. Flythe, of Newport News, as a judge of the Seventh Judicial Circuit for a term of eight years commencing April 1, 2021.

The Honorable Holly B. Smith, of Gloucester, as a judge of the Ninth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Jacqueline S. McClenney, of Richmond, as a judge of the Thirteenth Judicial Circuit for a term of eight years commencing April 1, 2021.

The Honorable Rondelle D. Herman, of Henrico, as a judge of the Fourteenth Judicial Circuit for a term of eight years commencing March 16, 2021.

Kathleen M. Uston, Esquire, of Alexandria, as a judge of the Eighteenth Judicial Circuit for a term of eight years commencing April 1, 2021.

Tania L. Saylor Peterson, Esquire, of Fairfax County, as a judge of the Nineteenth Judicial Circuit for a term of eight years commencing July 1, 2021.

The Honorable Petula C. A. Metzler, of Prince William, as a judge of the Thirty-first Judicial Circuit for a term of eight years commencing May 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

SENATE OF VIRGINIA

February 15, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following persons are qualified to be elected to the respective general district court judgeships as follows:

Linda L. Bryant, Esquire, of Chesapeake, as a judge of the First Judicial District for a term of six years commencing March 16, 2021.

Tanya L. Lomax, Esquire, of Chesapeake, as a judge of the First Judicial District for a term of six years commencing April 1, 2021.

Tameeka M. Williams, Esquire, of Norfolk, as a judge of the Fourth Judicial District for a term of six years commencing April 1, 2021.

Helivi L. Holland, Esquire, of Suffolk, as a judge of the Fifth Judicial District for a term of six years commencing April 1, 2021.

Charisse M. Mullen, Esquire, of Hampton, as a judge of the Seventh Judicial District for a term of six years commencing April 1, 2021.

Joshua P. DeFord, Esquire, of James City County, as a judge of the Ninth Judicial District for a term of six years commencing November 1, 2021.

Devika E. Davis, Esquire, of Henrico, as a judge of the Thirteenth Judicial District for a term of six years commencing June 1, 2021.

Jane M. Reynolds, Esquire, of Prince William, as a judge of the Fifteenth Judicial District for a term of six years commencing April 16, 2021.

Gary H. Moliken, Esquire, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing May 1, 2021.

Jessica H. Foster, Esquire, of Fauquier, as a judge of the Twentieth Judicial District for a term of six years commencing July 1, 2021.

Ché C. Rogers, Esquire, of Prince William, as a judge of the Thirty-first Judicial District for a term of six years commencing May 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

SENATE OF VIRGINIA

February 15, 2021

TO THE SENATE OF VIRGINIA:

The Committee on the Judiciary hereby certifies that the following persons are qualified to be elected to the respective juvenile and domestic relations district court judgeships as follows:

Shawn W. Overbey, Esquire, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing April 16, 2021.

Kimberly A. Kurkjian, Esquire, of Newport News, as a judge of the Seventh Judicial District for a term of six years commencing August 1, 2021.

Brian J. Smalls, Esquire, of Williamsburg, as a judge of the Ninth Judicial District for a term of six years commencing April 1, 2021.

Mara M. Matthews, Esquire, of James City County, as a judge of the Ninth Judicial District for a term of six years commencing July 1, 2021.

Stacy E. Lee, Esquire, of Henrico, as a judge of the Fourteenth Judicial District for a term of six years commencing May 1, 2021.

Melissa S. Cardoce, Esquire, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing April 1, 2021.

Melinda L. VanLowe, Esquire, of Fairfax County, as a judge of the Nineteenth Judicial District for a term of six years commencing April 1, 2021.

Heather P. Ferguson, Esquire, of Salem, as a judge of the Twenty-third Judicial District for a term of six years commencing April 1, 2021.

Robert C. Hagan, Jr., Esquire, of Botetourt, as a judge of the Twenty-fifth Judicial District for a term of six years commencing June 1, 2021.

Respectfully submitted,

/s/ John S. Edwards, Chairman Committee on the Judiciary

The following bills, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- H.B. 1806 (one thousand eight hundred six) with amendments.
- H.B. 1852 (one thousand eight hundred fifty-two).
- H.B. 1866 (one thousand eight hundred sixty-six).
- H.B. 1867 (one thousand eight hundred sixty-seven).
- H.B. 1911 (one thousand nine hundred eleven) with substitute.
- **H.B. 1912** (one thousand nine hundred twelve) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1936 (one thousand nine hundred thirty-six) with substitute.
- **H.B. 1992** (one thousand nine hundred ninety-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2064 (two thousand sixty-four).
 - H.B. 2081 (two thousand eighty-one).
- **H.B. 2133** (two thousand one hundred thirty-three) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **H.B.** 2150 (two thousand one hundred fifty).
 - H.B. 2236 (two thousand two hundred thirty-six).

The following bills, having been considered by the committee in session, were reported by Senator Lewis from the Committee on Local Government:

- H.B. 1749 (one thousand seven hundred forty-nine).
- H.B. 1764 (one thousand seven hundred sixty-four).
- H.B. 1778 (one thousand seven hundred seventy-eight) with amendments.
- H.B. 1783 (one thousand seven hundred eighty-three).
- **H.B.** 1858 (one thousand eight hundred fifty-eight).
- H.B. 1898 (one thousand eight hundred ninety-eight).
- H.B. 1919 (one thousand nine hundred nineteen) with substitute.
- H.B. 2042 (two thousand forty-two) with amendments.
- H.B. 2054 (two thousand fifty-four).
- H.B. 2091 (two thousand ninety-one).
- H.B. 2095 (two thousand ninety-five).
- **H.B. 2180** (two thousand one hundred eighty).
- H.B. 2186 (two thousand one hundred eighty-six).
- H.B. 2201 (two thousand two hundred one).
- H.B. 2217 (two thousand two hundred seventeen).
- H.B. 2257 (two thousand two hundred fifty-seven).
- H.B. 2287 (two thousand two hundred eighty-seven).
- **H.B. 2323** (two thousand three hundred twenty-three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- H.B. 1912, H.B. 1992, H.B. 2133, and H.B. 2323 were rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.R. 508. Celebrating the life of Edward Paul Crapol.

Patrons--Mason, Norment, Bell, Howell, Lewis and Spruill

S.R. 509. Commending Walt Whitman Middle School.

Patron--Surovell

- S.R. 510. Commending Montclair Elementary School.
 Patron--Surovell
- S.R. 511. Commending Swans Creek Elementary School.
 Patron--Surovell
- **S.R. 512.** Commending Featherstone Elementary School. Patron--Surovell
- **S.R. 513.** Celebrating the life of Charles R. Hooff III. Patron--Surovell
- **S.R. 514.** Commending the Science Museum of Virginia. Patrons--Hashmi, Bell, Howell, Lewis and Morrissey

CALENDAR

HOUSE BILLS ON THIRD READING

Senator Saslaw moved that the following House bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- H.B. 1790 (one thousand seven hundred ninety).
- H.B. 1798 (one thousand seven hundred ninety-eight).
- H.B. 1808 (one thousand eight hundred eight).
- H.B. 1812 (one thousand eight hundred twelve).
- H.B. 1814 (one thousand eight hundred fourteen).
- H.B. 1816 (one thousand eight hundred sixteen).
- H.B. 1824 (one thousand eight hundred twenty-four).
- H.B. 1827 (one thousand eight hundred twenty-seven).
- H.B. 1830 (one thousand eight hundred thirty).
- H.B. 1842 (one thousand eight hundred forty-two).
- H.B. 1848 (one thousand eight hundred forty-eight).
- H.B. 1873 (one thousand eight hundred seventy-three).
- H.B. 1885 (one thousand eight hundred eighty-five).
- H.B. 1891 (one thousand eight hundred ninety-one).
- H.B. 1905 (one thousand nine hundred five).
- **H.B. 1913** (one thousand nine hundred thirteen).
- **H.B. 1918** (one thousand nine hundred eighteen).
- **H.B. 1943** (one thousand nine hundred forty-three).
- H.B. 1944 (one thousand nine hundred forty-four).
- H.B. 1967 (one thousand nine hundred sixty-seven).
- H.B. 1971 (one thousand nine hundred seventy-one).
- H.B. 1981 (one thousand nine hundred eighty-one).
- H.B. 1998 (one thousand nine hundred ninety-eight).H.B. 2009 (two thousand nine).
- H.B. 2012 (two thousand twelve).
- H.B. 2031 (two thousand thirty-one).
- H.B. 2046 (two thousand forty-six).
- H.B. 2119 (two thousand one hundred nineteen).

- H.B. 2140 (two thousand one hundred forty).
- H.B. 2161 (two thousand one hundred sixty-one).
- H.B. 2169 (two thousand one hundred sixty-nine).
- H.B. 2170 (two thousand one hundred seventy).
- H.B. 2171 (two thousand one hundred seventy-one).
- H.B. 2182 (two thousand one hundred eighty-two).
- H.B. 2202 (two thousand two hundred two).
- H.B. 2222 (two thousand two hundred twenty-two).
- H.B. 2233 (two thousand two hundred thirty-three).
- H.B. 2308 (two thousand three hundred eight).
- H.B. 2310 (two thousand three hundred ten).
- H.B. 2314 (two thousand three hundred fourteen).
- H.B. 2320 (two thousand three hundred twenty).

The motion was agreed to.

H.B. 1808 (one thousand eight hundred eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 37.2-304 of the Code of Virginia, relating to Commissioner of Behavioral Health and Developmental Services; reports to designated protection and advocacy system.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1918 (one thousand nine hundred eighteen) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding a section numbered 22.1-205.1, relating to high school student parking passes; valid driver's license or driver privilege card required.

The reading of the substitute was waived.

Senator Norment moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

Senator Norment offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 22.1-205 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-205.1, relating to student driver safety.

On motion of Senator Norment, the reading of the substitute was waived.

On motion of Senator Norment, the substitute was agreed to.

The substitute was ordered to be engrossed.

1. Line 117, engrossed, after life

insert

H.B. 2012 (two thousand twelve) was taken up.

The following amendments proposed by the Committee on the Judiciary were offered:

```
strike

, or

insert

, [a comma]

2. Line 117, engrossed, after health
strike

, or normal development
```

, or normal development

On motion of Senator Edwards, the amendments were agreed to.

The amendments were ordered to be engrossed.

The reading of the amendments was waived.

H.B. 2320 (two thousand three hundred twenty) was taken up.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 55.1-703 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 55.1-708.2, relating to property; required disclosures for buyer to exercise due diligence; flood risk report.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

The substitute was ordered to be engrossed.

Senator Saslaw moved that the passage of the House bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following House bills were passed en bloc with their titles:

- H.B. 1790 (one thousand seven hundred ninety).
- H.B. 1798 (one thousand seven hundred ninety-eight).
- H.B. 1808 (one thousand eight hundred eight) with substitute.
- H.B. 1814 (one thousand eight hundred fourteen).
- H.B. 1816 (one thousand eight hundred sixteen).
- H.B. 1824 (one thousand eight hundred twenty-four).
- H.B. 1827 (one thousand eight hundred twenty-seven).
- H.B. 1830 (one thousand eight hundred thirty).
- H.B. 1848 (one thousand eight hundred forty-eight).
- H.B. 1873 (one thousand eight hundred seventy-three).
- H.B. 1885 (one thousand eight hundred eighty-five).
- H.B. 1891 (one thousand eight hundred ninety-one).
- H.B. 1905 (one thousand nine hundred five).
- H.B. 1913 (one thousand nine hundred thirteen).
- H.B. 1918 (one thousand nine hundred eighteen) with substitute.
- H.B. 1944 (one thousand nine hundred forty-four).
- H.B. 1967 (one thousand nine hundred sixty-seven).
- H.B. 1998 (one thousand nine hundred ninety-eight).
- H.B. 2009 (two thousand nine).
- H.B. 2012 (two thousand twelve) with amendments.
- H.B. 2119 (two thousand one hundred nineteen).
- H.B. 2140 (two thousand one hundred forty).
- **H.B. 2161** (two thousand one hundred sixty-one).
- H.B. 2169 (two thousand one hundred sixty-nine).
- H.B. 2170 (two thousand one hundred seventy).
- H.B. 2171 (two thousand one hundred seventy-one).
- **H.B. 2182** (two thousand one hundred eighty-two).
- H.B. 2202 (two thousand two hundred two).
- H.B. 2222 (two thousand two hundred twenty-two).
- H.B. 2233 (two thousand two hundred thirty-three).
- H.B. 2308 (two thousand three hundred eight).
- H.B. 2310 (two thousand three hundred ten).
- H.B. 2314 (two thousand three hundred fourteen).
- H.B. 2320 (two thousand three hundred twenty) with substitute.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1812 (one thousand eight hundred twelve), on motion of Senator Barker, was passed with its title.

The recorded vote is as follows: YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Ruff, Saslaw, Spruill, Surovell, Vogel--27.

NAYS--Chase, Cosgrove, DeSteph, Hanger, McDougle, Newman, Obenshain, Peake, Reeves, Stanley, Stuart, Suetterlein--12.

RULE 36--0.

H.B. 1842 (one thousand eight hundred forty-two) was taken up.

The following amendment proposed by the Committee on General Laws and Technology was offered:

1. Line 74, engrossed, after by strike regulation insert resolution

The reading of the amendment was waived.

On motion of Senator Barker, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 1842, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows: YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--27.

NAYS--Cosgrove, DeSteph, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Stuart, Suetterlein--12.

RULE 36--0.

H.B. 1943 (one thousand nine hundred forty-three), on motion of Senator McPike, was passed with its title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1971 (one thousand nine hundred seventy-one), on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--38.

NAYS--Suetterlein--1.

RULE 36--0.

H.B. 1981 (one thousand nine hundred eighty-one) was taken up.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 47, engrossed, after may

strike

the remainder of line 47 and through address on line 48

insert

consent in writing to the landlord addressing

2. Line 49, engrossed, after landlord.

insert

In the case of a tenant who has provided notice that he does not want nonemergency repairs made during the state of emergency due to a communicable disease of public health threat, the landlord may nonetheless enter the dwelling unit to do nonemergency repairs and maintenance with at least seven days' written notice to the tenant and at a time consented to by the tenant, no more than once every six months, provided that the employees and agents sent by the landlord are wearing all appropriate and reasonable personal protective equipment as required by state law. Furthermore, if the landlord is required to conduct maintenance or an inspection pursuant to the agreement for the loan or insurance policy that covers the dwelling unit, the tenant shall allow such maintenance or inspection, provided that the employees and agents sent by the landlord are wearing all appropriate personal protective equipment as required by state law.

The reading of the amendments was waived.

On motion of Senator Barker, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 1981, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Pillion, Ruff, Saslaw, Spruill, Surovell, Vogel--27.

NAYS--Chase, Cosgrove, DeSteph, Hanger, McDougle, Obenshain, Peake, Petersen, Reeves, Stanley, Stuart, Suetterlein--12.

RULE 36--0.

H.B. 2031 (two thousand thirty-one) was taken up.

The following amendment proposed by the Committee on General Laws and Technology was offered:

1. After line 37, engrossed

insert

C. Nothing in this section shall apply to commercial air service airports.

The reading of the amendment was waived.

On motion of Senator Barker, the amendment was agreed to.

H.B. 2031, on motion of Senator Surovell, was passed by for the day.

H.B. 2046 (two thousand forty-six) was taken up.

The following amendment proposed by the Committee on General Laws and Technology was offered:

1. At the beginning of line 78, engrossed

strike

status as a veteran

insert

military status

The reading of the amendment was waived.

On motion of Senator Barker, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 2046, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows: YEAS--27. NAYS--11. RULE 36--1.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Suetterlein, Surovell, Vogel--27.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley--11.

RULE 36--Stuart--1.

STATEMENT ON VOTE

Senator Suetterlein stated that he voted yea on the question of the passage of **H.B. 2046**, whereas he intended to vote nay.

- **H.B. 1737** (one thousand seven hundred thirty-seven), on motion of Senator Lucas, was passed by temporarily.
 - H.B. 1811 (one thousand eight hundred eleven) was read by title the third time.

The following amendments proposed by the Committee on General Laws and Technology were offered:

1. Line 14, engrossed, after a

strike

public body

insert

state agency

2. Line 16, engrossed, after such

strike

public body

insert

state agency

3. After line 17, engrossed

insert

C. When in the course of procuring goods, if a local public body receives two or more bids for products that are Energy Star certified, meet FEMP-designated efficiency requirements, appear on FEMP's Low Standby Power Product List, or are WaterSense certified, such local public body is encouraged to select among those bids.

The reading of the amendments was waived.

On motion of Senator Barker, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 1811, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Stuart, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--14.

RULE 36--0.

- **H.B. 1817** (one thousand eight hundred seventeen), on motion of Senator Ebbin, was passed by for the day.
- **H.B. 1821** (one thousand eight hundred twenty-one) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Petersen, Pillion, Reeves, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--31.

NAYS--Chase, Cosgrove, DeSteph, Newman, Obenshain, Peake, Ruff, Stanley--8. RULE 36--0.

H.B. 1847 (one thousand eight hundred forty-seven) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100 of the Code of Virginia, relating to sports betting; technical amendments.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1847, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Reeves, Saslaw, Spruill, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, Hanger, McDougle, Newman, Norment, Obenshain, Peake, Ruff, Stanley, Stuart, Suetterlein--13.

RULE 36--0.

H.B. 1849 (one thousand eight hundred forty-nine) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--21, NAYS--18, RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1878 (one thousand eight hundred seventy-eight) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1882 (one thousand eight hundred eighty-two) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--Locke, Lucas--2.

RULE 36--0.

H.B. 1904 (one thousand nine hundred four) was read by title the third time and, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1931 (one thousand nine hundred thirty-one) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows: YEAS--28. NAYS--11. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Saslaw, Spruill, Stuart, Surovell, Vogel--28.

NAYS--Chase, Hanger, Mason, McDougle, Newman, Obenshain, Peake, Reeves, Ruff, Stanley, Suetterlein--11.

RULE 36--0.

H.B. 1940 (one thousand nine hundred forty) was read by title the third time.

The following amendments proposed by the Committee on Education and Health were offered:

1. Line 7, engrossed, Title, after *civic* strike or political

2. Line 148, engrossed, after civic

strike

or political

3. Line 149, engrossed, after absent for such

strike

purposes

insert

purpose. Local school boards may require that the student provide advance notice of the intended absence and require that the student provide documentation of participation in a civic event

The reading of the amendments was waived.

On motion of Senator Lucas, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 1940, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Reeves, Saslaw, Spruill, Stuart, Surovell, Vogel--26.

NAYS--Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Stanley, Suetterlein--13.

RULE 36--0.

H.B. 2002 (two thousand two) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 16.1-260 and 63.2-1903 of the Code of Virginia, relating to child support; health care coverage.

The reading of the substitute was waived.

On motion of Senator Edwards, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2002, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.B. 2014 (two thousand fourteen) was read by title the third time.

Senator Barker moved that H.B. 2014 be passed with its title.

The question was put on passing H.B. 2014 with its title.

H.B. 2014 was defeated with its title.

The recorded vote is as follows:

YEAS--18. NAYS--21. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Spruill--18.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--21.

RULE 36--0.

H.B. 2085 (two thousand eighty-five) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--McDougle, Newman, Obenshain--3.

RULE 36--0.

H.B. 2128 (two thousand one hundred twenty-eight) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows: YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

H.B. 2168 (two thousand one hundred sixty-eight) was read by title the third time.

The following amendments proposed by the Committee on the Judiciary were offered:

1. Line 51, engrossed, after *Services* strike

the remainder of line 51 and through Board on line 52

2. Line 99, engrossed, after Services

strike

the remainder of line 99 and through Board on line 100

3. Line 105, engrossed, after of

insert

up to

4. Line 110, engrossed, after *penalty of* insert

up to

The reading of the amendments was waived.

On motion of Senator Edwards, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 2168, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--2.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Dunnavant, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--36.

NAYS--0.

RULE 36--DeSteph, Stuart--2.

RECONSIDERATION

Senator Surovell moved to reconsider the vote by which H.B. 2014 (two thousand fourteen) was defeated with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- H.B. 2014, on motion of Senator Surovell, was passed by for the day.
- **H.B. 2172** (two thousand one hundred seventy-two) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 2175 (two thousand one hundred seventy-five) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-2223.5, relating to housing protections; foreclosures; manufactured housing.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2175, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--35. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Chase, DeSteph, Obenshain, Ruff--4.

RULE 36--0.

H.B. 2229 (two thousand two hundred twenty-nine), on motion of Senator Newman, was passed by for the day.

H.B. 2249 (two thousand two hundred forty-nine) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 17.1-275, 55.1-1200, 55.1-1204, 55.1-1206, 55.1-1208, 55.1-1211, 55.1-1226, 64.2-2008, and 64.2-2012 of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; landlord charges for security deposits, insurance premiums for damage insurance, and insurance premiums for renter's insurance; filing of information regarding resident agent appointed by nonresident property owner.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

H.B. 2249, on motion of Senator Petersen, was passed by for the day.

H.B. 2258 (two thousand two hundred fifty-eight) was read by title the third time.

Senator Edwards moved that H.B. 2258 be passed with its title.

The question was put on passing H.B. 2258 with its title.

H.B. 2258 was defeated with its title.

The recorded vote is as follows:

YEAS--18. NAYS--21. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Mason, McClellan, McPike, Saslaw, Spruill, Surovell--18.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Marsden, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--21.

RULE 36--0.

H.B. 2290 (two thousand two hundred ninety) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows: YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18. RULE 36--0.

RECONSIDERATION

Senator Morrissey moved to reconsider the vote by which **H.B. 2258** (two thousand two hundred fifty-eight) was defeated with its title.

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- H.B. 2258, on motion of Senator Morrissey, was passed by for the day.
- **H.B. 1737** (one thousand seven hundred thirty-seven) was taken up, read by title the third time and, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

H.B. 1845 (one thousand eight hundred forty-five).

H.B. 1846 (one thousand eight hundred forty-six).

H.B. 1850 (one thousand eight hundred fifty).

```
H.B. 1851 (one thousand eight hundred fifty-one).
```

H.B. 1879 (one thousand eight hundred seventy-nine).

H.B. 1901 (one thousand nine hundred one).

H.B. 1960 (one thousand nine hundred sixty).

H.B. 1961 (one thousand nine hundred sixty-one).

H.B. 1962 (one thousand nine hundred sixty-two).

H.B. 1973 (one thousand nine hundred seventy-three).

H.B. 2010 (two thousand ten).

H.B. 2024 (two thousand twenty-four).

H.B. 2069 (two thousand sixty-nine).

H.B. 2071 (two thousand seventy-one).

H.B. 2092 (two thousand ninety-two).

H.B. 2131 (two thousand one hundred thirty-one).

H.B. 2197 (two thousand one hundred ninety-seven).

H.B. 2213 (two thousand two hundred thirteen).

H.B. 2216 (two thousand two hundred sixteen).

H.B. 2266 (two thousand two hundred sixty-six).

H.B. 2294 (two thousand two hundred ninety-four).

H.B. 1801 (one thousand eight hundred one).

H.B. 1805 (one thousand eight hundred five).

H.B. 1813 (one thousand eight hundred thirteen).

H.B. 1828 (one thousand eight hundred twenty-eight).

H.B. 1903 (one thousand nine hundred three).

H.B. 1932 (one thousand nine hundred thirty-two).

H.B. 2146 (two thousand one hundred forty-six).

H.B. 2262 (two thousand two hundred sixty-two).

H.B. 2312 (two thousand three hundred twelve). H.B. 2318 (two thousand three hundred eighteen).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

```
H.B. 1845 (one thousand eight hundred forty-five).
```

H.B. 1846 (one thousand eight hundred forty-six).

H.B. 1850 (one thousand eight hundred fifty).

H.B. 1851 (one thousand eight hundred fifty-one).

H.B. 1879 (one thousand eight hundred seventy-nine).

H.B. 1901 (one thousand nine hundred one).

H.B. 1960 (one thousand nine hundred sixty).

H.B. 1961 (one thousand nine hundred sixty-one).

H.B. 1962 (one thousand nine hundred sixty-two).

```
H.B. 1973 (one thousand nine hundred seventy-three).
```

H.B. 2010 (two thousand ten).

H.B. 2024 (two thousand twenty-four).

H.B. 2069 (two thousand sixty-nine).

H.B. 2071 (two thousand seventy-one).

H.B. 2092 (two thousand ninety-two).

H.B. 2131 (two thousand one hundred thirty-one).

H.B. 2197 (two thousand one hundred ninety-seven).

H.B. 2213 (two thousand two hundred thirteen).

H.B. 2216 (two thousand two hundred sixteen).

H.B. 2266 (two thousand two hundred sixty-six).

H.B. 2294 (two thousand two hundred ninety-four).

H.B. 1801 (one thousand eight hundred one).

H.B. 1805 (one thousand eight hundred five).

H.B. 1813 (one thousand eight hundred thirteen).

H.B. 1828 (one thousand eight hundred twenty-eight).

H.B. 1903 (one thousand nine hundred three).

H.B. 1932 (one thousand nine hundred thirty-two).

H.B. 2146 (two thousand one hundred forty-six).

H.B. 2262 (two thousand two hundred sixty-two).

H.B. 2312 (two thousand three hundred twelve).

H.B. 2318 (two thousand three hundred eighteen).

HOUSE JOINT RESOLUTIONS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House joint resolutions be waived:

```
H.J.R. 522 (five hundred twenty-two).
```

H.J.R. 567 (five hundred sixty-seven).

H.J.R. 526 (five hundred twenty-six).

H.J.R. 527 (five hundred twenty-seven).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House joint resolutions were passed by for the day:

```
H.J.R. 522 (five hundred twenty-two).
```

H.J.R. 567 (five hundred sixty-seven).

H.J.R. 526 (five hundred twenty-six).

H.J.R. 527 (five hundred twenty-seven).

On motion of Senator Lucas, the Senate adjourned until tomorrow at $12\ m.$

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

TUESDAY, FEBRUARY 16, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable William M. Stanley, Jr., Twentieth Senatorial District, offered the following prayer:

Mighty and loving Father, Creator of heaven and earth, we praise You and adore You. You are the King of Kings. We humbly ask Your forgiveness for our sins, to have mercy on us, O Lord. As we are gathered here today for our session, send us Your Holy Spirit to be our guide and give us the wisdom to understand every topic that we are going to discuss and every vote that we are going to take. Enlighten our minds and let Your love be upon us, and may today's session bring both success and growth to this body. Bless us as we pray and bless the entire General Assembly family. Give us inspiration to meet the needs of the future. Give us the wisdom to know Thy will and the courage to perform our duties. We thank You Father for this precious time that You have given us together. All of this we pray through our Lord Jesus Christ, Your Son who lives and reigns with You in unity of the Holy Spirit, one God forever. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Ophelia Anwah, Senate Committee Operations Intern, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

After the roll call, Senator McDougle notified the Clerk of his presence.

On motion of Senator Locke, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--34. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, McDougle, Petersen--3.

RULE 36--0.

HOUSE COMMUNICATIONS

The following communications were received:

In the House of Delegates February 15, 2021

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING SENATE BILL:

S.B. 1157. A BILL to amend and reenact § 15.2-1400 of the Code of Virginia, relating to time of certain local elections.

/s/ Suzette Denslow Clerk of the House of Delegates

> In the House of Delegates February 15, 2021

THE HOUSE OF DELEGATES HAS PASSED WITH AMENDMENTS THE FOLLOWING SENATE BILLS:

- **S.B. 1132.** A BILL to amend and reenact § 22.1-98 of the Code of Virginia, relating to public schools; severe weather conditions and other emergency situations; unscheduled remote learning days.
- **S.B. 1393.** A BILL to amend and reenact §§ 15.2-961 and 15.2-961.1 of the Code of Virginia, relating to replacement and conservation of trees during development.

THE HOUSE OF DELEGATES HAS PASSED WITH A SUBSTITUTE THE FOLLOWING SENATE BILL:

S.B. 1146. A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.

THE HOUSE OF DELEGATES HAS AGREED TO WITH A SUBSTITUTE THE FOLLOWING SENATE JOINT RESOLUTION:

S.J.R. 272. Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters and the right to vote; persons not entitled to vote.

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING SENATE BILLS:

- **S.B. 1111.** A BILL to amend and reenact § 24.2-606 of the Code of Virginia, relating to elections; preservation of order at the polls; powers of officers of election.
- **S.B. 1112.** A BILL to amend and reenact §§ 58.1-439.12:08 and 58.1-439.12:11 of the Code of Virginia, relating to research and development tax credits.
- **S.B. 1143.** A BILL to extend certain wetlands permits through 2021.
- **S.B. 1161.** A BILL to amend and reenact § 10.1-502 of the Code of Virginia, relating to Soil and Water Conservation Board; membership.
- **S.B. 1162.** A BILL to amend and reenact §§ 58.1-339.3 and 58.1-439.5 of the Code of Virginia, relating to tax credit; agricultural best management practices.
- **S.B. 1175.** A BILL to amend and reenact § 22.1-32 of the Code of Virginia, relating to Brunswick County school board; appointed school board salaries.
- **S.B. 1194.** A BILL to amend and reenact the second enactment of Chapter 574 of the Acts of Assembly of 2017, relating to produce safety; sunset.

- **S.B. 1196.** A BILL to amend and reenact §§ 22.1-253.13:5 and 22.1-298.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-298.7, relating to teachers and other licensed school board employees; cultural competency.
- **S.B. 1199.** A BILL to amend the Code of Virginia by adding in Chapter 10.1 of Title 10.1 a section numbered 10.1-1016.1 and by adding in Chapter 17 of Title 10.1 a section numbered 10.1-1705.1, relating to conservation easements; construction.
- **S.B. 1201.** A BILL to amend and reenact §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660, relating to tax exemptions for energy storage systems.
- **S.B. 1204.** A BILL providing a management agreement between the Commonwealth and George Mason University pursuant to the Restructured Higher Education Financial and Administrative Operations Act (§ 23.1-1000 et seq.).
- **S.B. 1210.** A BILL to direct the Department of Environmental Quality to convene working groups to revise permit fee schedules.
- S.B. 1252. A BILL to amend and reenact §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1 of the Code of Virginia, relating to sunset of coal tax credits.
- **S.B. 1280.** A BILL to amend the Code of Virginia by adding in Article 2 of Chapter 6 of Title 10.1 a section numbered 10.1-613.6, relating to dams; negotiated settlement agreements.
- **S.B. 1281.** A BILL to amend and reenact § 24.2-110 of the Code of Virginia, relating to elections; qualifications of the general registrar, residency.
- **S.B. 1387.** A BILL to amend and reenact § 23.1-506 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 23.1-505.1, relating to the State Council of Higher Education for Virginia; eligibility for in-state tuition.
- S.B. 1395. A BILL to amend and reenact §§ 24.2-105, as it shall become effective, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 24.2-104.1, by adding a section numbered 24.2-1005.2, and by adding in Title 24.2 a chapter numbered 1.1, consisting of sections numbered 24.2-125 through 24.2-131; and to repeal § 24.2-124, as it shall become effective, of the Code of Virginia, relating to elections; prohibited discrimination in voting and elections administration; required process for enacting certain covered practices; civil causes of action; penalties.
- **S.B. 1402.** A BILL to amend and reenact § 29.1-311 of the Code of Virginia, relating to trout fishing in stocked waters.
- S.B. 1403. A BILL to amend the Code of Virginia by adding a section numbered 58.1-609.14, relating to sales tax; exemption for personal protective equipment; emergency.
 EMERGENCY
- **S.B. 1404.** A BILL to amend and reenact § 62.1-44.15:29.1 of the Code of Virginia, relating to Stormwater Local Assistance Fund; grant requirements.
- **S.B. 1405.** A BILL to amend the Code of Virginia by adding a section numbered 23.1-2911.2, relating to the establishment of the Get Skilled, Get a Job, Give Back (G3) Fund and Program.

- **S.B. 1411.** A BILL to amend and reenact § 3.2-1905 of the Code of Virginia, relating to the excise tax on peanuts.
- S.B. 1453. A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 10.1 an article numbered 5, consisting of sections numbered 10.1-1332 and 10.1-1333, by adding in Chapter 1 of Title 33.2 a section numbered 33.2-120, by adding in Article 2 of Chapter 2 of Title 33.2 a section numbered 33.2-221.1, by adding a title numbered 45.2, containing a subtitle numbered I, consisting of chapters numbered 1 through 4, containing sections numbered 45.2-100 through 45.2-402, a subtitle numbered II, consisting of chapters numbered 5 through 10, containing sections numbered 45.2-500 through 45.2-1051, a subtitle numbered III, consisting of chapters numbered 11 through 15, containing sections numbered 45.2-1100 through 45.2-1505, a subtitle numbered IV, consisting of a chapter numbered 16, containing sections numbered 45.2-1600 through 45.2-1649, and a subtitle numbered V, consisting of chapters numbered 17 through 21, containing sections numbered 45.2-1700 through 45.2-2119, by adding sections numbered 55.1-1820.1, 55.1-1951.1, and 55.1-2133.1, and by adding in Title 56 a chapter numbered 29, consisting of sections numbered 56-614 through 56-624, and to repeal Chapter 6.1 (§§ 11-34.1 through 11-34.4) of Title 11, Title 45.1 (§§ 45.1-161.1 through 45.1-399), §§ 62.1-195.1 and 62.1-195.3, and Title 67 (§§ 67-100 through 67-1700) of the Code of Virginia, relating to administration of the Department of Mines, Minerals and Energy, coal mining, mineral mines, gas and oil, and other sources of energy and energy policy.

THE HOUSE OF DELEGATES HAS AGREED TO THE FOLLOWING SENATE JOINT RESOLUTIONS:

- **S.J.R. 270.** Proposing an amendment to Section 15-A of Article I of the Constitution of Virginia, relating to marriage; repeal of same-sex marriage prohibition; affirmative right to marry.
- S.J.R. 5001. Commending the East Coast Surfing Championship.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communications from the House of Delegates was waived.

The recorded vote is as follows: YEAS--34. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, DeSteph, McDougle, Petersen, Stanley--5. RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Saslaw from the Committee on Commerce and Labor:

- H.B. 1807 (one thousand eight hundred seven).
- **H.B. 1818** (one thousand eight hundred eighteen) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1829 (one thousand eight hundred twenty-nine).
 - H.B. 1834 (one thousand eight hundred thirty-four) with substitute.
 - H.B. 1892 (one thousand eight hundred ninety-two).
 - H.B. 1896 (one thousand eight hundred ninety-six).
 - H.B. 1907 (one thousand nine hundred seven).
 - H.B. 1923 (one thousand nine hundred twenty-three) with amendments.
- **H.B. 1925** (one thousand nine hundred twenty-five) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1942 (one thousand nine hundred forty-two).
 - H.B. 1965 (one thousand nine hundred sixty-five) with amendments.
- **H.B. 1985** (one thousand nine hundred eighty-five) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1994 (one thousand nine hundred ninety-four).
 - H.B. 2008 (two thousand eight) with substitute.
- **H.B. 2032** (two thousand thirty-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2034 (two thousand thirty-four).
 - H.B. 2036 (two thousand thirty-six).
- **H.B. 2040** (two thousand forty) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2207** (two thousand two hundred seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2219** (two thousand two hundred nineteen) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2282 (two thousand two hundred eighty-two).
 - H.B. 2304 (two thousand three hundred four) with substitute.
- **H.B. 2332** (two thousand three hundred thirty-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills, having been considered by the committee in session, were recommended for rereferral by the Committee on Commerce and Labor:

- **H.B. 1832** (one thousand eight hundred thirty-two) with the recommendation that it be rereferred to the Committee on Transportation.
- **H.B. 1884** (one thousand eight hundred eighty-four) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2250** (two thousand two hundred fifty) with the recommendation that it be rereferred to the Committee on Agriculture, Conservation and Natural Resources.
- **H.B. 2269** (two thousand two hundred sixty-nine) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bills, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- H.B. 1763 (one thousand seven hundred sixty-three) with amendments.
- H.B. 1774 (one thousand seven hundred seventy-four).
- H.B. 1776 (one thousand seven hundred seventy-six).
- **H.B.** 1796 (one thousand seven hundred ninety-six).
- H.B. 1800 (one thousand eight hundred) with amendments.

- H.B. 1895 (one thousand eight hundred ninety-five).
- H.B. 1899 (one thousand eight hundred ninety-nine).
- H.B. 1916 (one thousand nine hundred sixteen).
- H.B. 1935 (one thousand nine hundred thirty-five) with substitute.
- H.B. 1953 (one thousand nine hundred fifty-three).
- H.B. 1969 (one thousand nine hundred sixty-nine).
- H.B. 1986 (one thousand nine hundred eighty-six).
- H.B. 1999 (one thousand nine hundred ninety-nine).
- H.B. 2006 (two thousand six).
- H.B. 2059 (two thousand fifty-nine).
- H.B. 2060 (two thousand sixty).
- H.B. 2123 (two thousand one hundred twenty-three).
- H.B. 2138 (two thousand one hundred thirty-eight).
- H.B. 2174 (two thousand one hundred seventy-four) with substitute.
- H.B. 2177 (two thousand one hundred seventy-seven) with amendment.
- H.B. 2178 (two thousand one hundred seventy-eight) with amendment.
- H.B. 2179 (two thousand one hundred seventy-nine).
- H.B. 2181 (two thousand one hundred eighty-one).
- H.B. 2185 (two thousand one hundred eighty-five).
- H.B. 2194 (two thousand one hundred ninety-four).
- H.B. 2204 (two thousand two hundred four).
- H.B. 2206 (two thousand two hundred six) with substitute.
- H.B. 2261 (two thousand two hundred sixty-one).
- H.B. 2263 (two thousand two hundred sixty-three).
- H.B. 2273 (two thousand two hundred seventy-three) with substitute.

H.B. 1818, H.B. 1884, H.B. 1925, H.B. 1985, H.B. 2032, H.B. 2040, H.B. 2207, H.B. 2219, H.B. 2269, and H.B. 2332 were rereferred to the Committee on Finance and Appropriations.

- H.B. 1832 was rereferred to the Committee on Transportation.
- H.B. 2250 was rereferred to the Committee on Agriculture, Conservation and Natural Resources.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.R. 515. Commending Pocahontas County.

Patron--Deeds

S.R. 516. Commending Greek Orthodox churches in Virginia.

Patron--Dunnavant

S.R. 517. Commending the Virginia Women's Institute for Leadership at Mary Baldwin University. Patron--Hanger

CALENDAR

UNFINISHED BUSINESS—SENATE

S.B. 1155 (one thousand one hundred fifty-five) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to create a six-year capital outlay plan for projects to be funded entirely or partially from general fund-supported resources and to repeal Chapter 1134 of the Acts of Assembly of 2020.

On motion of Senator Howell, the substitute was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

HOUSE BILLS ON THIRD READING

Senator Saslaw moved that the following House bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- H.B. 1845 (one thousand eight hundred forty-five).
- H.B. 1846 (one thousand eight hundred forty-six).
- H.B. 1850 (one thousand eight hundred fifty).
- H.B. 1851 (one thousand eight hundred fifty-one).
- H.B. 1879 (one thousand eight hundred seventy-nine).
- H.B. 1901 (one thousand nine hundred one).
- H.B. 1960 (one thousand nine hundred sixty).
- H.B. 1961 (one thousand nine hundred sixty-one).
- H.B. 1962 (one thousand nine hundred sixty-two).
- H.B. 1973 (one thousand nine hundred seventy-three).
- H.B. 2010 (two thousand ten).
- H.B. 2024 (two thousand twenty-four).
- H.B. 2069 (two thousand sixty-nine).
- H.B. 2071 (two thousand seventy-one).
- H.B. 2092 (two thousand ninety-two).
- H.B. 2131 (two thousand one hundred thirty-one).
- H.B. 2197 (two thousand one hundred ninety-seven).
- H.B. 2213 (two thousand two hundred thirteen).
- H.B. 2216 (two thousand two hundred sixteen).
- H.B. 2266 (two thousand two hundred sixty-six).
- H.B. 2294 (two thousand two hundred ninety-four).

The motion was agreed to.

H.B. 2031 (two thousand thirty-one) was taken up, the committee amendment having been agreed to on February 15, 2021.

Senator McDougle offered the following amendments:

1. Line 14, engrossed, after technology;

strike

local

2. Line 23, engrossed

strike

all of line 23 and through agencies on line 24

insert

such purchase or deployment of facial recognition technology is expressly authorized by statute. For purposes of this section, a statute that does not refer to facial recognition technology shall not be construed to provide express authorization. Such statute shall require that any facial recognition technology purchased or deployed by the local law-enforcement agency be maintained under the exclusive control of such local law-enforcement agency and that any data contained by such facial recognition technology be kept confidential, not be disseminated or resold, and be accessible only by a search warrant issued pursuant to Chapter 5 (§ 19.2-52 et seq.) of Title 19.2 or an administrative or inspection warrant issued pursuant to law

3. Line 25, engrossed, after $\it approval$

strike

of governing body

4. Line 33, engrossed, after unless

strike

the remainder of line 33, all of line 34, and through technology on line 35

insert

such purchase or deployment of facial recognition technology is expressly authorized by statute. For purposes of this section, a statute that does not refer to facial recognition technology shall not be construed to provide express authorization. Such statute shall require that any facial recognition technology purchased or deployed by the campus police department be maintained under the exclusive control of such campus police department and that any data contained by such facial recognition technology be kept confidential, not be disseminated or resold, and be accessible only by a search warrant issued pursuant to Chapter 5 (§ 19.2-52 et seq.) of Title 19.2 or an administrative or inspection warrant issued pursuant to law

5. Line 40, engrossed, after by

strike

the remainder of line 40 and all of line 41

insert

an act of the General Assembly

On motion of Senator McDougle, the reading of the amendments was waived.

On motion of Senator McDougle, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 1846 (one thousand eight hundred forty-six) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Transportation was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 46.2-334.01, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to license restrictions for minors; use of handheld personal communications devices.

The reading of the substitute was waived.

On motion of Senator Marsden, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1850 (one thousand eight hundred fifty) was taken up.

The following amendments proposed by the Committee on Transportation were offered:

1. Line 21, engrossed, after limits strike

the remainder of line 21

2. Line 24, engrossed, after *pounds* insert

, provided that such weight is on the power unit

The reading of the amendments was waived.

On motion of Senator Marsden, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 2071 (two thousand seventy-one) was taken up.

The following amendment proposed by the Committee on Transportation was offered:

1. Line 20, engrossed, after *be* insert

or the project sponsor has committed that the design will be

The reading of the amendment was waived.

On motion of Senator Marsden, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 2197 (two thousand one hundred ninety-seven) was taken up.

The following amendments proposed by the Committee on Rehabilitation and Social Services were offered:

```
1. At the beginning of line 5, Title, engrossed
    strike
              options
    insert
              and develop recommendations
2. Line 12, engrossed, after § 1.
    strike
              The
    insert
              That the
3. Line 18, engrossed, after Governor
    strike
              and
    insert
              , [a comma]
4. Line 18, engrossed, after Assembly
    insert
              , and the Chairmen of the Senate Committee on Rehabilitation and Social
              Services and the House Committee on Health, Welfare and Institutions
```

The reading of the amendments was waived.

On motion of Senator Favola, the amendments were agreed to.

The amendments were ordered to be engrossed.

Senator Saslaw moved that the passage of the House bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following House bills were passed en bloc with their titles:

```
H.B. 2031 (two thousand thirty-one) with amendments.
H.B. 1845 (one thousand eight hundred forty-five).
H.B. 1846 (one thousand eight hundred forty-six) with substitute.
H.B. 1850 (one thousand eight hundred fifty) with amendments.
H.B. 1851 (one thousand eight hundred fifty-one).
H.B. 1879 (one thousand eight hundred seventy-nine).
H.B. 1901 (one thousand nine hundred one).
H.B. 1960 (one thousand nine hundred sixty).
H.B. 1961 (one thousand nine hundred sixty-one).
H.B. 1962 (one thousand nine hundred sixty-two).
H.B. 1973 (one thousand nine hundred seventy-three).
H.B. 2010 (two thousand ten).
H.B. 2024 (two thousand twenty-four).
H.B. 2071 (two thousand seventy-one) with amendment.
H.B. 2092 (two thousand ninety-two).
```

H.B. 2131 (two thousand one hundred thirty-one).

H.B. 2197 (two thousand one hundred ninety-seven) with amendments.

H.B. 2216 (two thousand two hundred sixteen).

H.B. 2294 (two thousand two hundred ninety-four).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 2069 (two thousand sixty-nine), on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 2213 (two thousand two hundred thirteen) was taken up.

The following amendments proposed by the Committee on Rules were offered:

1. Line 7, engrossed, Title, after *Commonwealth* strike

, and to prohibit the issuance of certain mining permits

2. Line 24, engrossed

strike

all of lines 24, 25, and 26

The reading of the amendments was waived.

On motion of Senator Locke, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 2213, on motion of Senator Locke, was passed with its title.

The recorded vote is as follows: YEAS--23. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--16.

RULE 36--0.

H.B. 2266 (two thousand two hundred sixty-six) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 4.1-206, 4.1-206.3, as it shall become effective, 4.1-231, 4.1-231.1, as it shall become effective, 4.1-233, and 4.1-233.1, as it shall become effective, of the Code of Virginia, relating to alcoholic beverage control; local special events license.

The reading of the substitute was waived.

On motion of Senator Favola, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2266, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1817 (one thousand eight hundred seventeen) was read by title the third time.

Senator Dunnavant offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 54.1-2957 and 54.1-2957.01 of the Code of Virginia, relating to practice of certified nurse midwives.

On motion of Senator Dunnavant, the reading of the substitute was waived.

On motion of Senator Dunnavant, the substitute was agreed to.

H.B. 1817, on motion of Senator Lucas, was passed by temporarily.

H.B. 2014 (two thousand fourteen) was taken up.

Senator Barker offered the following amendment:

1. Line 131, engrossed, after enforceable.

insert

Notwithstanding the requirements of this section, a landlord with four or fewer rental dwelling units, or up to a 10 percent interest in four or fewer rental dwelling units, may limit a tenant's use of the right of redemption to once per lease period, provided that the landlord provides written notice of such limitation to the tenant.

On motion of Senator Barker, the reading of the amendment was waived.

Senator Barker moved that the amendment be agreed to.

- H.B. 2014, on motion of Senator Barker, was passed by temporarily.
- **H.B. 2229** (two thousand two hundred twenty-nine) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--30. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Obenshain, Peake, Pillion, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--30.

NAYS--Chase, Hanger, Kiggans, Newman, Norment, Petersen, Reeves, Ruff, Suetterlein--9. RULE 36--0.

H.B. 2249 (two thousand two hundred forty-nine) was taken up, the committee substitute having been agreed to on February 15, 2021.

The substitute was ordered to be engrossed.

H.B. 2249, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Obenshain, Peake, Pillion, Ruff, Saslaw, Spruill, Surovell--26.

NAYS--Chase, Cosgrove, Deeds, DeSteph, Kiggans, Newman, Norment, Petersen, Reeves, Stanley, Stuart, Suetterlein, Vogel--13.

RULE 36--0.

H.B. 2258 (two thousand two hundred fifty-eight) was taken up.

Senator Surovell offered the following amendment:

1. Line 15, engrossed, after citizens

strike

, and to monitor and evaluate the administration and effectiveness of $\S\S~19.2\text{-}152.13$ and 19.2-152.14

Senator Surovell withdrew the amendment.

Senator Morrissey offered the following amendments:

1. Line 13, engrossed, after 19.2-152.14. strike

the remainder of line 13, all of lines 14 and 15, and through 19.2-152.14. on line 16

2. Line 18, engrossed, after *may* strike

the remainder of line 18, all of line 19, and through provisions, on line 20

On motion of Senator Morrissey, the reading of the amendments was waived.

On motion of Senator Morrissey, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 2258, on motion of Senator Morrissey, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--19. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--19.

RULE 36--0.

H.B. 1801 (one thousand eight hundred one) was read by title the third time.

Senator Marsden moved that H.B. 1801 be passed with its title.

The question was put on passing H.B. 1801 with its title.

H.B. 1801 was defeated with its title.

The recorded vote is as follows:

YEAS--18. NAYS--20. RULE 36--0.

YEAS--Barker, Bell, Boysko, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, Morrissey, Petersen, Ruff, Saslaw, Stuart--18.

NAYS--Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Hanger, Kiggans, McClellan, McDougle, McPike, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Spruill, Suetterlein, Surovell, Vogel--20. RULE 36--0.

H.B. 1805 (one thousand eight hundred five) was read by title the third time.

The following amendments proposed by the Committee on Rehabilitation and Social Services were offered:

1. Line 38, engrossed, after that

strike

caused by

insert

which is related to a history of discrimination for factors such as

2. Line 39, engrossed, after that

strike

restricts

insert

can affect

3. Line 43, engrossed, after for

insert

and meet the needs of

4. At the beginning of line 46, engrossed

strike

prioritize providing

insert

use available resources to provide

5. Line 46, engrossed, after economic

strike

or

insert

needs and those with the greatest

6. Line 46, engrossed, after social

strike

need

insert

needs

The reading of the amendments was waived.

On motion of Senator Favola, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 1805, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Peake, Petersen, Reeves, Saslaw, Spruill, Surovell--25.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Obenshain, Pillion, Ruff, Stanley, Stuart, Suetterlein, Vogel--14.

RULE 36--0.

H.B. 1813 (one thousand eight hundred thirteen) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--33. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Suetterlein, Surovell, Vogel--33.

NAYS--Chase, DeSteph, Newman, Obenshain, Stanley, Stuart--6.

RULE 36--0.

H.B. 1828 (one thousand eight hundred twenty-eight) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--29. NAYS--10. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--29.

NAYS--Chase, DeSteph, Dunnavant, McDougle, Obenshain, Reeves, Ruff, Stanley, Stuart, Suetterlein--10.

RULE 36--0.

- **H.B. 1903** (one thousand nine hundred three), on motion of Senator Suetterlein, was passed by for the day.
- **H.B. 1932** (one thousand nine hundred thirty-two), on motion of Senator Boysko, was passed by for the day.
 - H.B. 2146 (two thousand one hundred forty-six) was read by title the third time.

The following amendment proposed by the Committee on Transportation was offered:

1. Line 19, engrossed, after highway

insert

, but shall not include releasing or retrieving any dog in furtherance of hunting deer

The reading of the amendment was waived.

Senator Marsden moved that the amendment be agreed to.

The question was put on agreeing to the amendment.

The yeas and nays were called for by the Chair.

The recorded vote is as follows: YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Stuart, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--17.

RULE 36--0.

The amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 2146, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Stuart, Surovell, Vogel--24.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--15.

RULE 36--0.

H.B. 2262 (two thousand two hundred sixty-two), on motion of Senator Deeds, was passed by for the day.

H.B. 2312 (two thousand three hundred twelve) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-2216, as it is currently effective and as it shall become effective, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-210, as it is currently effective and as it shall become effective, 4.1-210, as it is currently effective and as it shall become effective, 4.1-210, as it is currently effective and as it shall become effective, 4.1-221, as it is currently effective and as it shall become effective, 4.1-221, as it is currently effective and as it shall become effective, 4.1-222, 4.1-2224, 4.1-225, 4.1-227, as it is currently effective and as it shall become

effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29, consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6, consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, containing chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Chapter 7 of Title 18.2 an article numbered 1.4, consisting of sections numbered 18.2-265.22 through 18.2-265.28, by adding a section numbered 19.2-392.2:1, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

The reading of the substitute was waived.

On motion of Senator Favola, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2312, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--23. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--16.

RULE 36--0.

H.B. 2318 (two thousand three hundred eighteen) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1817 (one thousand eight hundred seventeen) was taken up.

The substitute was ordered to be engrossed.

H.B. 1817, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--33. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--33.

NAYS--Chase, DeSteph, Kiggans, Locke, Peake, Reeves--6.

RULE 36--0.

RECONSIDERATION

Senator Ruff moved to reconsider the vote by which **H.B. 1801** (one thousand eight hundred one) was defeated with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.B. 1801, on motion of Senator Ruff, was passed by for the day.

H.B. 2014 (two thousand fourteen) was taken up and, on motion of Senator Barker, was passed by for the day.

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

```
H.B. 1749 (one thousand seven hundred forty-nine).
```

H.B. 1764 (one thousand seven hundred sixty-four).

H.B. 1783 (one thousand seven hundred eighty-three).

H.B. 1806 (one thousand eight hundred six).

H.B. 1858 (one thousand eight hundred fifty-eight).

H.B. 1866 (one thousand eight hundred sixty-six).

H.B. 1867 (one thousand eight hundred sixty-seven).

H.B. 1898 (one thousand eight hundred ninety-eight).

H.B. 2064 (two thousand sixty-four).

H.B. 2091 (two thousand ninety-one).

H.B. 2095 (two thousand ninety-five).

H.B. 2150 (two thousand one hundred fifty).

H.B. 2180 (two thousand one hundred eighty).

H.B. 2186 (two thousand one hundred eighty-six).

H.B. 2236 (two thousand two hundred thirty-six).

H.B. 2257 (two thousand two hundred fifty-seven).

H.B. 2287 (two thousand two hundred eighty-seven).

H.B. 1778 (one thousand seven hundred seventy-eight).

H.B. 1852 (one thousand eight hundred fifty-two).

H.B. 1911 (one thousand nine hundred eleven).

H.B. 1919 (one thousand nine hundred nineteen).

H.B. 1936 (one thousand nine hundred thirty-six).

H.B. 2042 (two thousand forty-two).

H.B. 2054 (two thousand fifty-four).

H.B. 2081 (two thousand eighty-one).

H.B. 2201 (two thousand two hundred one).

H.B. 2217 (two thousand two hundred seventeen).

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

```
H.B. 1749 (one thousand seven hundred forty-nine).
```

H.B. 1764 (one thousand seven hundred sixty-four).

H.B. 1783 (one thousand seven hundred eighty-three).

H.B. 1806 (one thousand eight hundred six).

H.B. 1858 (one thousand eight hundred fifty-eight).

H.B. 1866 (one thousand eight hundred sixty-six).

H.B. 1867 (one thousand eight hundred sixty-seven).

H.B. 1898 (one thousand eight hundred ninety-eight).

H.B. 2064 (two thousand sixty-four).

H.B. 2091 (two thousand ninety-one).

```
H.B. 2095 (two thousand ninety-five).
```

H.B. 2150 (two thousand one hundred fifty).

H.B. 2180 (two thousand one hundred eighty).

H.B. 2186 (two thousand one hundred eighty-six).

H.B. 2236 (two thousand two hundred thirty-six).

H.B. 2257 (two thousand two hundred fifty-seven).

H.B. 2287 (two thousand two hundred eighty-seven).

H.B. 1778 (one thousand seven hundred seventy-eight).

H.B. 1852 (one thousand eight hundred fifty-two).

H.B. 1911 (one thousand nine hundred eleven).

H.B. 1919 (one thousand nine hundred nineteen).

H.B. 1936 (one thousand nine hundred thirty-six).

H.B. 2042 (two thousand forty-two).

H.B. 2054 (two thousand fifty-four).

H.B. 2081 (two thousand eighty-one).

H.B. 2201 (two thousand two hundred one).

H.B. 2217 (two thousand two hundred seventeen).

HOUSE JOINT RESOLUTIONS ON THIRD READING

- **H.J.R. 522** (five hundred twenty-two) was read by title the third time and, on motion of Senator Locke, was agreed to.
- **H.J.R. 567** (five hundred sixty-seven) was read by title the third time and, on motion of Senator Locke, was agreed to.
- **H.J.R. 526** (five hundred twenty-six) was read by title the third time and, on motion of Senator Locke, was agreed to.
 - H.J.R. 527 (five hundred twenty-seven) was read by title the third time.

The following amendments proposed by the Committee on Rules were offered:

1. Line 26, engrossed, after Council, strike

and

2. Line 27, engrossed, after Federation

insert

, the Virginia Chapter of the American Society of Landscape Architects, an individual from the School of Plant and Environmental Sciences at the Virginia Polytechnic Institute and State University who has expertise in invasive species, local government associations, and such other stakeholders as the Department of Conservation and Recreation, jointly with the Virginia Department of Agriculture and Consumer Services, deem appropriate

The reading of the amendments was waived.

On motion of Senator Locke, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.J.R. 527, on motion of Senator Locke, was agreed to.

SUPPLEMENTAL CALENDAR NO. 1

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- H.B. 1763 (one thousand seven hundred sixty-three).
- H.B. 1774 (one thousand seven hundred seventy-four).
- H.B. 1776 (one thousand seven hundred seventy-six).
- H.B. 1800 (one thousand eight hundred).
- H.B. 1916 (one thousand nine hundred sixteen).
- H.B. 1935 (one thousand nine hundred thirty-five).
- H.B. 1953 (one thousand nine hundred fifty-three).
- H.B. 1986 (one thousand nine hundred eighty-six).
- H.B. 2006 (two thousand six).
- H.B. 2059 (two thousand fifty-nine).
- H.B. 2060 (two thousand sixty).
- H.B. 2177 (two thousand one hundred seventy-seven).
- H.B. 2178 (two thousand one hundred seventy-eight).
- H.B. 2179 (two thousand one hundred seventy-nine).
- H.B. 2181 (two thousand one hundred eighty-one).
- H.B. 2185 (two thousand one hundred eighty-five).
- H.B. 2204 (two thousand two hundred four).
- H.B. 2273 (two thousand two hundred seventy-three).
- **H.B.** 1796 (one thousand seven hundred ninety-six).
- H.B. 1895 (one thousand eight hundred ninety-five).
- H.B. 1899 (one thousand eight hundred ninety-nine).
- H.B. 1969 (one thousand nine hundred sixty-nine).
- H.B. 1999 (one thousand nine hundred ninety-nine).
- H.B. 2123 (two thousand one hundred twenty-three).
- H.B. 2138 (two thousand one hundred thirty-eight).
- **H.B. 2174** (two thousand one hundred seventy-four).
- H.B. 2194 (two thousand one hundred ninety-four).
- H.B. 2206 (two thousand two hundred six).
- H.B. 2261 (two thousand two hundred sixty-one).
- H.B. 2263 (two thousand two hundred sixty-three).

The motion was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

- H.B. 1763 (one thousand seven hundred sixty-three).
- **H.B.** 1774 (one thousand seven hundred seventy-four).
- **H.B.** 1776 (one thousand seven hundred seventy-six).
- H.B. 1800 (one thousand eight hundred).
- H.B. 1916 (one thousand nine hundred sixteen).
- H.B. 1935 (one thousand nine hundred thirty-five).
- H.B. 1953 (one thousand nine hundred fifty-three).
- H.B. 1986 (one thousand nine hundred eighty-six).
- H.B. 2006 (two thousand six).
- H.B. 2059 (two thousand fifty-nine).
- H.B. 2060 (two thousand sixty).
- H.B. 2177 (two thousand one hundred seventy-seven).
- H.B. 2178 (two thousand one hundred seventy-eight).
- H.B. 2179 (two thousand one hundred seventy-nine).
- H.B. 2181 (two thousand one hundred eighty-one).
- H.B. 2185 (two thousand one hundred eighty-five).
- H.B. 2204 (two thousand two hundred four).
- H.B. 2273 (two thousand two hundred seventy-three).
- H.B. 1796 (one thousand seven hundred ninety-six).
- H.B. 1895 (one thousand eight hundred ninety-five).
- H.B. 1899 (one thousand eight hundred ninety-nine).
- H.B. 1969 (one thousand nine hundred sixty-nine).
- H.B. 1999 (one thousand nine hundred ninety-nine).
- H.B. 2123 (two thousand one hundred twenty-three).
- H.B. 2138 (two thousand one hundred thirty-eight).
- H.B. 2174 (two thousand one hundred seventy-four).
- H.B. 2194 (two thousand one hundred ninety-four).
- H.B. 2206 (two thousand two hundred six).
- H.B. 2261 (two thousand two hundred sixty-one).
- H.B. 2263 (two thousand two hundred sixty-three).

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m. The Clerk was ordered to receive the committee reports.

COMMITTEE REPORTS

The following bill, having been considered by the committee in session, was reported by Senator Saslaw from the Committee on Commerce and Labor:

H.B. 2330 (two thousand three hundred thirty) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bill, having been considered by the committee in session, was reported by Senator Barker from the Committee on General Laws and Technology:

H.B. 2307 (two thousand three hundred seven) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.

The following bill, having been considered by the committee in session, was reported by Senator Favola from the Committee on Rehabilitation and Social Services:

- **H.B. 2191** (two thousand one hundred ninety-one) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- H.B. 2191, H.B. 2307, and H.B. 2330 were rereferred to the Committee on Finance and Appropriations.

Jacke Ochran

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

WEDNESDAY, FEBRUARY 17, 2021

The Senate met at 12 m. and was called to order by the President pro tempore, Senator L. Louise Lucas.

The Honorable Mamie E. Locke, Second Senatorial District, offered the following prayer:

Less than a year ago we had not heard of the Coronavirus, a soon-to-be pandemic that would soon plague our world. New words and phrases would enter our nomenclature: social distancing, testing, masking up. All this would become our new normal. Through it all God remained present in our lives. I offer this reading from Psalms 121, verses 1-3, "I will lift up mine eyes into the hills from whence cometh my help. My help cometh from the Lord, which made heaven and earth. He will not suffer thy foot to be moved; He that keepeth thee will not slumber."

Dear Lord, during this time of uncertainty in the pandemic, make sure that we all test positive for faith. Make sure that we keep distance from doubt, and make sure that we mask-up from fear. Lord, may we continue to trust in You and may each person who hears my voice be abundantly blessed, today and always. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Corinne Sloan, Assistant Journal Clerk, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

On motion of Senator Hashmi, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--34. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Deeds, DeSteph, Petersen, Stanley--4.

RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 16, 2021 THE HOUSE OF DELEGATES HAS PASSED WITH AMENDMENTS THE FOLLOWING SENATE BILLS:

- **S.B. 1212.** A BILL to amend the Code of Virginia by adding in Title 33.2 a chapter numbered 38, consisting of sections numbered 33.2-3800 through 33.2-3816, relating to creation of the New River Valley Passenger Rail Station Authority.
- **S.B. 1215.** A BILL to amend the Code of Virginia by adding a section numbered 55.1-1243.1 and to repeal § 55.1-1243 of the Code of Virginia, relating to Virginia Residential Landlord and Tenant Act; tenant remedies for exclusion from dwelling unit, interruption of services, or actions taken to make premises unsafe.
- S.B. 1254. A BILL to amend and reenact §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100 of the Code of Virginia, relating to sports betting; technical amendments.
- **S.B. 1350.** A BILL to amend and reenact §§ 33.2-214.2 and 33.2-353 of the Code of Virginia, relating to transportation projects; resiliency.
- S.B. 1374. A BILL to establish the Carbon Sequestration Task Force; report.
- **S.B. 1413.** A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to provision of broadband services by investor-owned electric utilities.

THE HOUSE OF DELEGATES HAS PASSED WITH SUBSTITUTES THE FOLLOWING SENATE BILLS:

- **S.B. 1327.** A BILL to amend and reenact §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-2223.5, relating to housing protections; foreclosures; manufactured housing.
- **S.B. 1366.** A BILL to amend and reenact §§ 51.5-134 and 51.5-135 of the Code of Virginia, relating to aging services; economic and social need.
- **S.B. 1375.** A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability from COVID-19.
- **S.B. 1406.** A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-122, 4.1-124, as it is currently effective and as it shall become effective, 4.1-128, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-202, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324,

4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, $18.2 - 265.1, \quad 18.2 - 265.2, \quad 18.2 - 265.3, \quad 18.2 - 287.2, \quad 18.2 - 308.03, \quad 18.2 - 308.09, \quad 18.2 - 308.012, \quad 18$ 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29, consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6, consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, containing chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Chapter 7 of Title 18.2 an article numbered 1.4, consisting of sections numbered 18.2-265.22 through 18.2-265.28, by adding a section numbered 19.2-392.2:1, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

- **S.B. 1410.** A BILL to amend and reenact §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1226, and 55.1-1310 of the Code of Virginia, relating to public accommodations, employment, and housing; prohibited discrimination on the basis of status as active military or a military spouse.
- **S.B. 1469.** A BILL to amend and reenact § 2.2-212 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-507.3, by adding in Chapter 22 of Title 2.2 an article numbered 12, consisting of sections numbered 2.2-2365 through 2.2-2376, and by adding in Article 3.1 of Chapter 1 of Title 51.1 a section numbered 51.1-124.40, relating to establishing an Opioid Abatement Authority.
- **S.B. 1471.** A BILL to amend and reenact §§ 4.1-206, 4.1-206.3, as it shall become effective, 4.1-231, 4.1-231.1, as it shall become effective, 4.1-233, and 4.1-233.1, as it shall become effective, of the Code of Virginia, relating to alcoholic beverage control; local special events license.
- **S.B. 1472.** A BILL to require the Department of Medical Assistance Services to establish a work group to study and develop recommendations for the permanent use of virtual supports and increasing access to virtual supports and services for individuals with intellectual and developmental disabilities.

THE HOUSE OF DELEGATES HAS PASSED WITH A SUBSTITUTE WITH AMENDMENT THE FOLLOWING SENATE BILL:

S.B. 1247. A BILL to amend and reenact § 56-599 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 26 of Title 45.1 a section numbered 45.1-394.1, relating to public disclosure of electric generating facility closures; integrated resource plans.

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING SENATE BILLS:

- **S.B. 1147.** A BILL to amend and reenact § 32.1-122.6:04 of the Code of Virginia, relating to Nurse Loan Repayment Program; certified nurse aide.
- **S.B. 1154.** A BILL to amend and reenact § 37.2-304 of the Code of Virginia, relating to Commissioner of Behavioral Health and Developmental Services; reports to designated protection and advocacy system.
- S.B. 1183. A BILL to amend and reenact §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953 of the Code of Virginia, relating to the Property Owners' Association Act; the Condominium Act; use of electronic means for meetings and voting.
- **S.B. 1187.** A BILL to amend and reenact § 54.1-3482 of the Code of Virginia, relating to the Department of Health Professions; practice of physical therapy.
- **S.B. 1205.** A BILL to amend and reenact §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909 of the Code of Virginia, relating to programs to address career fatigue and wellness in certain health care providers; civil immunity.

EMERGENCY

- S.B. 1221. A BILL to provide for the operation of the Loudoun County local health department.
- **S.B. 1259.** A BILL to amend and reenact §§ 56-539 and 56-542 of the Code of Virginia, relating to Virginia Highway Corporation Act; alteration of certificate of authority; powers and duties of the State Corporation Commission.
- **S.B. 1269.** A BILL to amend and reenact § 38.2-3407.15:2 of the Code of Virginia, relating to health insurance; authorization of drug prescribed for the treatment of a mental disorder.
- **S.B. 1276.** A BILL to amend and reenact § 38.2-3451 of the Code of Virginia, relating to health insurance; essential health benefits; abortion coverage.
- **S.B. 1287.** A BILL to amend and reenact §§ 18.2-340.19 and 18.2-340.28 of the Code of Virginia, relating to the Charitable Gaming Board; regulations; electronic pull tabs.
- **S.B. 1299.** A BILL to amend and reenact §§ 4.1-119, as it is currently effective and as it shall become effective, 4.1-204, as it is currently effective and as it shall become effective, 4.1-206.1, as it shall become effective, 4.1-207, 4.1-210, 4.1-212.1, as it is currently effective and as it shall become effective, and 4.1-221 of the Code of Virginia, relating to alcoholic beverage control; sale and delivery of mixed beverages and pre-mixed wine for off-premises consumption.

- **S.B. 1307.** A BILL to amend and reenact §§ 32.1-325 and 32.1-326.3 of the Code of Virginia, relating to Department of Medical Assistance Services; school-based health services; telemedicine.
- **S.B. 1334.** A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to pilot program for broadband capacity to unserved areas of the Commonwealth; municipal broadband authorities.
- **S.B. 1379.** A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-574, relating to Humane Cosmetics Act; civil penalties.
- **S.B. 1389.** A BILL to amend and reenact § 55.1-703 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 55.1-708.2, relating to property; required disclosures for buyer to exercise due diligence; flood risk report.
- **S.B. 1420.** A BILL to amend and reenact § 1 of the first enactment of Chapters 358 and 382 of the Acts of Assembly of 2013, as amended by Chapter 803 of the Acts of Assembly of 2017, and as amended by Chapters 1187, 1188, 1189, 1193, 1194, and 1239 of the Acts of Assembly of 2020, relating to electric utilities; nonjurisdictional customers; third party power purchase agreements.
- **S.B. 1421.** A BILL to amend and reenact § 37.2-403 of the Code of Virginia, relating to brain injury; definition.
- **S.B. 1429.** A BILL to authorize the Commonwealth to lease a portion of property previously used by the Department of Behavioral Health and Developmental Services as the Southwestern Virginia Mental Health Institute and to amend and reenact §§ 1 and 2 of Chapter 678 of the Acts of Assembly of 2019.
- **S.B. 1464.** A BILL to amend and reenact § 54.1-3446 of the Code of Virginia, relating to Drug Control Act; Schedule I.
- **S.B. 1470.** A BILL to amend and reenact § 46.2-600.1 of the Code of Virginia, relating to vehicle registration; special communication needs indicator.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--35. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--Deeds, McDougle, Petersen--3.

RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Petersen from the Committee on Agriculture, Conservation and Natural Resources:

- **H.B. 1750** (one thousand seven hundred fifty) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 1751** (one thousand seven hundred fifty-one) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1760 (one thousand seven hundred sixty).
 - H.B. 1804 (one thousand eight hundred four).
 - H.B. 1819 (one thousand eight hundred nineteen).
- **H.B. 1833** (one thousand eight hundred thirty-three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - **H.B. 1836** (one thousand eight hundred thirty-six) with substitute.
 - **H.B.** 1837 (one thousand eight hundred thirty-seven).
- **H.B. 1855** (one thousand eight hundred fifty-five) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1902 (one thousand nine hundred two).
 - H.B. 1928 (one thousand nine hundred twenty-eight).
 - H.B. 1958 (one thousand nine hundred fifty-eight).
 - H.B. 1982 (one thousand nine hundred eighty-two).
 - H.B. 1983 (one thousand nine hundred eighty-three).
 - H.B. 2030 (two thousand thirty) with substitute.
 - H.B. 2068 (two thousand sixty-eight) with substitute.
 - H.B. 2078 (two thousand seventy-eight).
- **H.B. 2129** (two thousand one hundred twenty-nine) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2148** (two thousand one hundred forty-eight) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2159 (two thousand one hundred fifty-nine).
 - H.B. 2187 (two thousand one hundred eighty-seven).
- **H.B. 2203** (two thousand two hundred three) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2250 (two thousand two hundred fifty).
 - H.B. 2275 (two thousand two hundred seventy-five).
 - H.B. 2302 (two thousand three hundred two) with substitute.
 - H.B. 2311 (two thousand three hundred eleven).
- H.B. 1750, H.B. 1751, H.B. 1833, H.B. 1855, H.B. 2129, H.B. 2148, and H.B. 2203 were rereferred to the Committee on Finance and Appropriations.

The following bills and joint resolution, having been considered by the committee in session, were reported by Senator Howell from the Committee on Finance and Appropriations:

- H.B. 1818 (one thousand eight hundred eighteen).
- H.B. 1820 (one thousand eight hundred twenty).
- H.B. 1833 (one thousand eight hundred thirty-three)
- H.B. 1884 (one thousand eight hundred eighty-four).
- H.B. 1893 (one thousand eight hundred ninety-three).

- H.B. 1912 (one thousand nine hundred twelve).
- H.B. 1925 (one thousand nine hundred twenty-five).
- H.B. 1979 (one thousand nine hundred seventy-nine) with substitute.
- H.B. 1985 (one thousand nine hundred eighty-five).
- H.B. 1989 (one thousand nine hundred eighty-nine) with substitute.
- H.B. 1992 (one thousand nine hundred ninety-two) with substitute.
- H.B. 1993 (one thousand nine hundred ninety-three).
- H.B. 1995 (one thousand nine hundred ninety-five).
- H.B. 2032 (two thousand thirty-two).
- H.B. 2040 (two thousand forty) with substitute.
- H.B. 2047 (two thousand forty-seven).
- H.B. 2101 (two thousand one hundred one) with substitute.
- H.B. 2110 (two thousand one hundred ten).
- H.B. 2111 (two thousand one hundred eleven).
- H.B. 2117 (two thousand one hundred seventeen).
- H.B. 2118 (two thousand one hundred eighteen) with substitute.
- H.B. 2133 (two thousand one hundred thirty-three).
- H.B. 2165 (two thousand one hundred sixty-five).
- H.B. 2166 (two thousand one hundred sixty-six) with amendments.
- H.B. 2167 (two thousand one hundred sixty-seven).
- H.B. 2207 (two thousand two hundred seven).
- H.B. 2212 (two thousand two hundred twelve).
- H.B. 2218 (two thousand two hundred eighteen).
- H.B. 2219 (two thousand two hundred nineteen).
- H.B. 2223 (two thousand two hundred twenty-three).
- H.B. 2230 (two thousand two hundred thirty).
- H.B. 2252 (two thousand two hundred fifty-two).
- H.B. 2269 (two thousand two hundred sixty-nine).
- H.B. 2293 (two thousand two hundred ninety-three).
- H.B. 2307 (two thousand three hundred seven).
- H.B. 2321 (two thousand three hundred twenty-one).
- H.B. 2323 (two thousand three hundred twenty-three).
- H.B. 2330 (two thousand three hundred thirty) with amendment.
- H.B. 2332 (two thousand three hundred thirty-two).
- H.J.R. 542 (five hundred forty-two) with substitute.

The following bill, having been considered by the committee in session, was reported by Senator Howell from the Committee on Finance and Appropriations:

H.B. 2058 (two thousand fifty-eight).

The following bills and joint resolutions, having been considered by the committee in session, were reported by Senator Deeds from the Committee on Privileges and Elections:

- H.B. 1810 (one thousand eight hundred ten).
- H.B. 1888 (one thousand eight hundred eighty-eight) with substitute.
- **H.B. 1890** (one thousand eight hundred ninety) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 1921 (one thousand nine hundred twenty-one).
 - H.B. 1952 (one thousand nine hundred fifty-two).

- H.B. 1968 (one thousand nine hundred sixty-eight).
- H.B. 2020 (two thousand twenty) with amendment.
- H.B. 2125 (two thousand one hundred twenty-five).
- H.B. 2198 (two thousand one hundred ninety-eight).
- H.B. 2324 (two thousand three hundred twenty-four).
- H.J.R. 555 (five hundred fifty-five) with substitute.
- H.J.R. 582 (five hundred eighty-two).
- H.B. 1890 was rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.R. 518. Commending Colonel Dean E. Gould.

Patrons--Reeves and Bell

S.R. 519. Commending the Virginia Wing of the Civil Air Patrol.

Patrons--Reeves and Bell

CALENDAR

UNFINISHED BUSINESS—SENATE

- **S.B. 1132** (one thousand one hundred thirty-two) was taken up with the amendment proposed by the House of Delegates as follows:
 - 1. Line 53, engrossed, after services.

strike

School divisions may not

insert

No school division shall

On motion of Senator Suetterlein, the amendment was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- **S.B.** 1146 (one thousand one hundred forty-six) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, and 58.1-322.03 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.

Senator Howell moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows: YEAS--2. NAYS--37. RULE 36--0.

YEAS--Bell, Spruill--2.

NAYS--Barker, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

RULE 36--0.

S.B. 1393 (one thousand three hundred ninety-three) was taken up with the amendments proposed by the House of Delegates as follows:

1. Line 65, engrossed, after locality's

strike

Municipal Separate Storm Sewer System

insert

municipal separate storm sewer system

2. Line 166, engrossed, after plans.

strike

Any

insert

If a locality is located within a nonattainment area, any

3. Line 297, engrossed, after 15.2-961.1 of the Code of Virginia

strike

the remainder of line 297 and through by on line 298

insert

and the amendments to such sections provided in the first enactment of

4. Line 298, engrossed, after recommend

strike

changes to those

insert

amendments to those statutes or the adoption of new Code

5. Line 300, engrossed, after planting,

strike

or

insert

and

6. Line 308, engrossed, after to the

strike

Chairman

insert

Chairmen

On motion of Senator Marsden, the amendments were agreed to.

The recorded vote is as follows: YEAS--31. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--31.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Obenshain, Peake, Reeves, Suetterlein--8. RULE 36--0.

S.J.R. 272 (two hundred seventy-two) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, as follows:

SENATE JOINT RESOLUTION NO. 272 AMENDMENT IN THE NATURE OF A SUBSTITUTE

Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters; felon disenfranchisement; automatic restoration of political rights.

RESOLVED by the Senate, the House of Delegates concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 1 of Article II of the Constitution of Virginia as follows:

ARTICLE II FRANCHISE AND OFFICERS

Section 1. Qualifications of voters.

In elections by the people, the qualifications of voters shall be as follows: Each voter shall be a citizen of the United States, shall be eighteen years of age, shall fulfill the residence requirements set forth in this section, and shall be registered to vote pursuant to this article. No person who has been convicted of a felony shall be qualified to vote unless his civil rights have been restored by the Governor or other appropriate authority or until completion of his sentence of imprisonment, at which time, without further action required of him, his political rights, including the right to vote, shall be restored. As prescribed by law, no person adjudicated to be mentally incompetent shall be qualified to vote until his competency has been reestablished.

The residence requirements shall be that each voter shall be a resident of the Commonwealth and of the precinct where he votes. Residence, for all purposes of qualification to vote, requires both domicile and a place of abode. The General Assembly may provide for persons who are employed overseas, and their spouses and dependents residing with them, and who are qualified to vote except for relinquishing their place of abode in the Commonwealth while overseas, to vote in the Commonwealth subject to conditions

and time limits defined by law. The General Assembly may provide for persons who are qualified to vote except for having moved their residence from one precinct to another within the Commonwealth to continue to vote in a former precinct subject to conditions and time limits defined by law. The General Assembly may also provide, in elections for President and Vice President of the United States, alternatives to registration for new residents of the Commonwealth.

Any person who will be qualified with respect to age to vote at the next general election shall be permitted to register in advance and also to vote in any intervening primary or special election.

Senator Locke moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows: YEAS--0. NAYS--38. RULE 36--0.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

RULE 36--0.

HOUSE BILLS ON THIRD READING

H.B. 1986 (one thousand nine hundred eighty-six), on motion of Senator Spruill, was passed by for the day.

Senator Saslaw moved that the following House bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- H.B. 1749 (one thousand seven hundred forty-nine).
- H.B. 1763 (one thousand seven hundred sixty-three).
- H.B. 1764 (one thousand seven hundred sixty-four).
- H.B. 1774 (one thousand seven hundred seventy-four).
- H.B. 1776 (one thousand seven hundred seventy-six).
- H.B. 1783 (one thousand seven hundred eighty-three).
- H.B. 1800 (one thousand eight hundred).
- H.B. 1806 (one thousand eight hundred six).
- H.B. 1858 (one thousand eight hundred fifty-eight).
- H.B. 1866 (one thousand eight hundred sixty-six).
- H.B. 1867 (one thousand eight hundred sixty-seven).
- H.B. 1898 (one thousand eight hundred ninety-eight).
- H.B. 1916 (one thousand nine hundred sixteen).
- H.B. 1935 (one thousand nine hundred thirty-five).
- H.B. 1953 (one thousand nine hundred fifty-three).
- H.B. 2006 (two thousand six).

```
H.B. 2059 (two thousand fifty-nine).
H.B. 2060 (two thousand sixty).
H.B. 2064 (two thousand sixty-four).
H.B. 2091 (two thousand ninety-one).
H.B. 2095 (two thousand ninety-five).
H.B. 2150 (two thousand one hundred fifty).
H.B. 2177 (two thousand one hundred seventy-seven).
H.B. 2178 (two thousand one hundred seventy-eight).
H.B. 2179 (two thousand one hundred seventy-nine).
H.B. 2180 (two thousand one hundred eighty).
H.B. 2181 (two thousand one hundred eighty-one).
H.B. 2185 (two thousand one hundred eighty-five).
H.B. 2186 (two thousand one hundred eighty-six).
H.B. 2204 (two thousand two hundred four).
H.B. 2236 (two thousand two hundred thirty-six).
H.B. 2257 (two thousand two hundred fifty-seven).
```

H.B. 2273 (two thousand two hundred seventy-three). H.B. 2287 (two thousand two hundred eighty-seven).

The motion was agreed to.

insert

H.B. 1763 (one thousand seven hundred sixty-three) was taken up.

The following amendments proposed by the Committee on Finance and Appropriations were offered:

```
    Line 19, engrossed, after 1998, insert
        but before January 1, 2025,
    Line 38, engrossed, after 2021, insert
        but before January 1, 2025,
    Line 103, engrossed, after 1998, insert
        but before January 1, 2025,
    Line 108, engrossed, after 2021, insert
        but before January 1, 2025,
    Line 146, engrossed, after annually
```

, and such aggregate cap shall be administered by the Department of Conservation and Recreation. Tax credits shall be issued to qualified taxpayers on a first-come, first-served basis in accordance with procedures established by the Department of Conservation and Recreation

The reading of the amendments was waived.

Senator Hanger requested that, pursuant to Senate Rule 31, there be a division of the amendments for consideration.

On motion of Senator Hanger, amendments Nos. 1, 2, 3, and 4 were agreed to.

Senator Hanger moved that amendment No. 5 be rejected.

The question was put on agreeing to amendment No. 5.

Amendment No. 5 was rejected.

The amendments were ordered to be engrossed.

H.B. 1806 (one thousand eight hundred six) was taken up.

The following amendments proposed by the Committee on the Judiciary were offered:

1. Line 69, engrossed, after Department,

insert

or within 60 days of such transfer,

2. Line 71, engrossed, after determine.

strike

the remainder of line 71 and all of lines 72 through 77

The reading of the amendments was waived.

On motion of Senator Edwards, the amendments were agreed to.

Senator Stanley offered the following amendment:

1. Line 66, engrossed, after *Department*)

strike

the remainder of line 66 and through Department on line 67

On motion of Senator Stanley, the reading of the amendment was waived.

On motion of Senator Stanley, the amendment was agreed to.

The amendments were ordered to be engrossed.

H.B. 1953 (one thousand nine hundred fifty-three) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 54.1-2900, 54.1-3005, 54.1-3303, and 54.1-3408 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 54.1-2957.04, relating to licensed certified midwives; licensure; practice.

The reading of the substitute was waived.

On motion of Senator Locke, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2177 (two thousand one hundred seventy-seven) was taken up.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. Line 85, engrossed strike
all of lines 85, 86, and 87

The reading of the amendment was waived.

On motion of Senator Howell, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 2178 (two thousand one hundred seventy-eight) was taken up.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. Line 46, engrossed, after *Academic* strike

18142

insert

18412

The reading of the amendment was waived.

On motion of Senator Howell, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 2273 (two thousand two hundred seventy-three) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

The reading of the substitute was waived.

On motion of Senator Howell, the substitute was agreed to.

The substitute was ordered to be engrossed.

Senator Saslaw moved that the passage of the House bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following House bills were passed en bloc with their titles:

```
H.B. 1749 (one thousand seven hundred forty-nine).
```

H.B. 1763 (one thousand seven hundred sixty-three) with amendments.

H.B. 1764 (one thousand seven hundred sixty-four).

H.B. 1774 (one thousand seven hundred seventy-four).

H.B. 1776 (one thousand seven hundred seventy-six).

H.B. 1783 (one thousand seven hundred eighty-three).

H.B. 1806 (one thousand eight hundred six) with amendments.

H.B. 1858 (one thousand eight hundred fifty-eight).

H.B. 1866 (one thousand eight hundred sixty-six).

H.B. 1953 (one thousand nine hundred fifty-three) with substitute.

H.B. 2059 (two thousand fifty-nine).

H.B. 2060 (two thousand sixty).

H.B. 2064 (two thousand sixty-four).

H.B. 2091 (two thousand ninety-one).

H.B. 2150 (two thousand one hundred fifty).

H.B. 2177 (two thousand one hundred seventy-seven) with amendment.

H.B. 2178 (two thousand one hundred seventy-eight) with amendment.

H.B. 2179 (two thousand one hundred seventy-nine).

H.B. 2180 (two thousand one hundred eighty).

H.B. 2181 (two thousand one hundred eighty-one).

H.B. 2185 (two thousand one hundred eighty-five).

H.B. 2186 (two thousand one hundred eighty-six).

H.B. 2236 (two thousand two hundred thirty-six). **H.B. 2257** (two thousand two hundred fifty-seven).

H.B. 2273 (two thousand two hundred seventy-three) with substitute.

H.B. 2287 (two thousand two hundred eighty-seven).

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1800 (one thousand eight hundred) was taken up.

The following amendments proposed by the Committee on Finance and Appropriations as substituted for House amendments were offered:

Item 0 #1s

Revenues

Revenues Language

Language:

Page 1, strike lines 34 through 38, and insert:

	First Year	Second Year	Total
Unreserved Beginning Balance	\$2,874,058,799	\$0	\$2,874,058,799
Additions to Balance	(\$1,284,491,604)	\$29,850,000	(\$1,254,641,604)
Official Revenue Estimates	\$21,941,832,509	\$22,577,524,514	\$44,519,357,023
Transfer	\$659,277,131	\$643,217,349	\$1,302,494,480
Total General Fund Resources Available for Appropriation	\$24,190,676,835	\$23,250,591,863	\$47,441,268,698

The appropriations made in this act from nongeneral fund revenues are based upon the following:

	First Year	Second Year	Total
Balance, June 30, 2020	\$6,915,611,972		\$6,915,611,972
Official Revenue Estimates	\$39,115,808,362	\$39,883,177,181	\$78,998,985,543
Lottery Proceeds Fund	\$685,031,123	\$690,903,334	\$1,375,934,457
Internal Service Fund	\$2,127,455,883	\$2,293,917,698	\$4,421,373,581
Bond Proceeds	\$2,479,633,162	\$294,775,137	\$2,774,408,299
Revenues Available for			
Total Nongeneral Fund Revenues Available for Appropriation	\$51,323,540,502	\$43,162,773,350	\$94,486,313,852
TOTAL PROJECTED REVENUES	\$75,514,217,337	\$66,413,365,213	\$141,927,582,550

Page 2, strike lines 1 through 15.

Explanation:

(This amendment reflects the net total of general and nongeneral fund revenue, transfer and balance adjustments encompassed in the Committee amendment package to SB 1100.)

Item 1 #1s

Legislative Department

General Assembly of Virginia

Language

Language:

Page 14, after line 14, insert:

- "Z.1. The Chair of the Senate Finance and Appropriations Committee shall appoint six members from the Senate Committee on Finance and Appropriations and the Chairman of the House Appropriations Committee shall appoint three members from the House Committee on Appropriations and three members of the House Committee on Finance to a Joint Subcommittee on Tax Policy. The Joint Subcommittee shall elect a chairman and vice-chairman from among its membership.
- 2. The goals and objectives of the Joint Subcommittee shall include (i) evaluating the fiscal impact of amendments to tax brackets, tax rates, credits, deductions, and exemptions, as well as any other factors it deems relevant to making Virginia's individual income tax system more fair and equitable; (ii) giving consideration to the fairness, certainty, convenience of payment, economy in collection, simplicity, neutrality, and economic efficiency of the Commonwealth's tax policies and any changes thereto; and (iii) recommending whether the General Assembly should amend the Code of Virginia.
- 3. To assist the Joint Subcommittee, the Chair of the Joint Subcommittee may appoint a workgroup which includes the staff of the House Committee on Finance, the House Committee on Appropriations, the Senate Committee on Finance and Appropriations, and any other stakeholders deemed appropriate. All agencies of the Commonwealth shall provide technical assistance to the Joint Subcommittee, upon request."

Explanation:

(This amendment establishes a Joint Subcommittee on Tax Policy to evaluate and make recommendations on potential changes to Virginia's tax policies, including changes to tax brackets, tax rates, credits, deductions, and exemptions, and any other changes it deems necessary. The Joint Subcommittee will consider factors such as equity, certainty, convenience of payment, economy in collection, simplicity, neutrality, economic efficiency and any other factors it deems relevant to the Commonwealth's tax policies.)

Item 1 #2s

Legislative Department

General Assembly of Virginia

Language

Language:

Page 14, after line 14, insert:

"Z. The staff of the Senate Fiscal Office, in collaboration with the staff of the Senate Committee on Finance and Appropriations, shall perform a comprehensive review of the compensation and benefits currently provided to legislative assistants, including a review of what might be considered best practies in other states regarding the compensation and benefits, professional growth, and training opportunities provided to legislative assistants, and make recommendations on any changes to the compensation and benefits, professional growth and training opportunities as a result of such review, and deliver a report on the staffs' findings and recommendations to the Clerk of the Senate and Chairman of the Senate Committee on Finance and Appropriations by November 1, 2021."

Explanation:

(This amendment directs the staff of the Senate Fiscal Office and the staff of the Senate Finance and Appropriations Committee to perform a review of the compensation and benefits currently provided to legislative assistants, and to make any recommendations regarding the structure of compensation and benefits to the Clerk of the Senate and the Chairman of the Senate Finance and Appropriations Committee by November 1, 2021.)

		Item 4 #1s	
Legislative Department	FY20-21	FY21-22	
Division of Capitol Police	\$0	\$846,907	GF

Language:

Page 15, line 44, strike "\$13,270,924" and insert "\$14,117,831".

Explanation:

(This amendment provides funding of \$846,907 GF the second year to increase the starting salaries of Division of Capitol Police police officers following graduation, manage salary compression, increase the starting salary of communications officers, and increase the salary of support and wage employees.)

		Item 9 #1s	
Legislative Department	FY20-21	FY21-22	
Dr. Martin Luther King, Jr. Memorial Commission	\$0	\$50,000	GF

Language:

Page 18, line 20, strike "\$50,643" and insert "\$100,643".

Page 18, after line 26, insert:

Included within the appropriation for this Item is \$50,000 the second year from the general fund for the Dr. Martin Luther King, Jr. Memorial Commission to complete a pre-planning study to locate a memorial tribute to the late Senator Yvonne Miller on Virginia's Capitol Square or another location. The Department of General Services shall consult with the Commission, if requested by the Commission, to provide its capital project pre-planning expertise and Capitol Square operation and maintenance knowledge to the Commission as it formulates its study findings. The Commission will complete its pre-planning study and report its findings to the Governor, and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than November 1, 2021. The Department of General Services shall be compensated for its services provided to the Commission from the funds authorized in this Item."

Explanation:

(This amendment provides \$50,000 GF the second year for the Dr. Martin Luther King, Jr. Memorial Commission, with the assistance of the Department of General Services, to begin the process of creating a memorial in tribute to the late Senator Yvonne Miller.)

			Item 29.1 #1s	
Legislative Depar	tment	FY20-21	FY21-22	
Behavioral Hea	lth Commission	\$0 0.00	\$348,744 4.00	GF FTE
Language:				
Page 26, after line 3	31, insert:			
	"LEGISL	ATIVE DEPARTMENT		
	§ 1-8.5. Beha	vioral Health Commission		
29.1 Behavioral Health	Commission	\$0	\$348,774	
Fund Sources:	General	\$0	\$348,774 "	

Explanation:

(This amendment provides \$348,774 GF the second year for the establishment of a legislative Behavioral Health Commission as provided for in Senate Bill 1273. The requested amount is for four positions and associated office-related costs that are phased-in over the year to reflect the timing necessary to startup the Commission.)

[&]quot;Authority: Title 30, Chapter 27, Code of Virginia.

Item 33 #1s

Legislative DepartmentFY20-21FY21-22Virginia Commission on
Intergovernmental Cooperation\$42,397\$66,377GF

Language:

Page 31, line 15, strike "\$780,935" and insert "\$823,332". Page 31, line 15, strike "\$780,935" and insert "\$847,312".

Explanation:

(This amendment provides \$42,397 GF the first year and \$66,377 GF the second year to fully fund the amount projected for dues specified in this Item.)

		Item 36 #1s	
Judicial Department	FY20-21	FY21-22	
Supreme Court	\$0	\$100,000	GF

Language:

Page 33, line 4, strike "\$14,594,927" and insert "\$14,694,927".

Explanation:

(This amendment provides \$100,000 GF the second year to fund the potential increase in the use of court-appointed experts caused by the passage of SB 1315 of the 2021 General Assembly, which broadens defendants' ability to introduce evidence regarding their mental state at the time of an alleged defense, dependent upon its final passage.)

Item 39 #1s

Judicial Department

Supreme Court Language

Language:

Page 36, after line 9, insert:

"P. The Office of the Executive Secretary of the Supreme Court shall prepare and distribute evaluation forms in all Circuit Court cases that are overseen by a retired judge for the purpose of collecting information on the number and types of cases referred to retired judges, and use such information to prepare and annually publish a report to be distributed to the members of the House Committee on Courts of Justice and the Senate Committee on the Judiciary, on or

about January 1, each year."

Explanation:

(This amendment requires the Supreme Court of Virginia (SCV) to distribute evaluation forms in all Circuit Court cases that are overseen by a retired judge, to collect the results of the evaluation forms and publish the findings to the members of the House Courts of Justice Committee and Senate Committee on the Judiciary. As a part of that report, the SCV will also report back on the number and types of cases referred to retired judges.)

		Item 40 #1s	
Judicial Department	FY20-21	FY21-22	
Court of Appeals of Virginia	\$1,064,609 27.00	\$2,736,885 25.00	GF FTE

Language:

Page 36, line 24, strike "\$10,183,547" and insert "\$11,248,156". Page 36, line 24, strike "\$15,460,379" and insert "\$18,197,264".

Explanation:

(This amendment provides \$1.1 million GF and 27 FTE positions in FY 2021 and \$2.7 million GF and 25 FTE positions in FY 2022 for fixed costs and the prorated portion for 27 judicial staff positions in the first year and funding for two additional judgeships and 23 support positions, including clerk, IT and support staff, additional transcripts, and office accommodations in the second year associated with SB 1261 of the 2020 General Assembly Session, which expands the jurisdiction of the Court of Appeals. This is in addition to the funding of \$235,419 GF the first year and \$4.9 million GF for four judgeships and 27 staff positions in the second year included in the Governor's Introduced Budget, SB 1100.)

		Item 41 #1s	
Judicial Department	FY20-21	FY21-22	
Circuit Courts	\$0	(\$421,117)	GF

Language:

Page 37, line 12, strike "\$112,595,520" and insert "\$112,174,403".

Explanation:

(This amendment reflects the estimated Criminal Fund savings (court-appointed attorney costs) from opening a public defender office in Chesterfield County. A companion amendment to Item 48 (Indigent Defense Commission), requests funding to establish a public defender office for

Chesterfield County. Savings reflect estimated first year cost savings prorated for six months.)

		Item 42 #1s	
Judicial Department	FY20-21	FY21-22	
General District Courts	\$0	(\$486,803)	GF

Language:

Page 39, line 18, strike "\$128,797,150" and insert "\$128,310,347".

Explanation:

(The amendment reflects the estimated Criminal Fund savings (court-appointed attorney costs) from opening a public defender office in Chesterfield County. A companion amendment to Item 48 (Indigent Defense Commission), requests funding to establish a public defender office for Chesterfield County. Savings reflect estimated first year cost savings prorated for six months.)

		Item 43 #1s	
Judicial Department	FY20-21	FY21-22	
Juvenile and Domestic Relations District Courts	\$0	(\$171,931)	GF

Language:

Page 40, line 20, strike "\$107,020,623" and insert "\$106,848,692".

Explanation:

(The amendment reflects the estimated Criminal Fund savings (court-appointed attorney costs) from opening a public defender office in Chesterfield County. A companion amendment to Item 48 (Indigent Defense Commission), requests funding to establish a public defender office for Chesterfield County. Savings reflect estimated first year cost savings prorated for six months.)

		Item 48 #1s	
Judicial Department	FY20-21	FY21-22	
Indigent Defense Commission	\$0	(\$2,909,010)	GF

Language:

Page 43, line 10, strike "\$63,148,850" and insert "\$60,239,840".

Page 43, after line 34, insert:

"E. Within the appropriation for this Item, \$1,019,506 in the second year from the general fund

for Capital Indigent Defense Services (32702), shall be reallocated to Criminal Indigent Defense Services (32701), and the remaining funding contained within (32702) shall revert to the general fund, pursuant to Senate Bill 1165, of the 2021 General Assembly Session, which repeals the death penalty."

Explanation:

(This amendment redirects \$1.0 million GF the second year for Capital Indigent Defense Services to Criminal Indigent Defense Services within the Indigent Defense Commission, and reverts remaining funding currently appropriated for Capital Indigent Defense Services, pursuant to SB 1165, of the 2021 General Assembly Session repealing the death penalty, contingent upon its final passage.)

		Item 48 #2s	
Judicial Department	FY20-21	FY21-22	
Indigent Defense Commission	\$0 0.00	\$3,164,584 33.00	GF FTE

Language:

Page 43, line 10, strike "\$63,148,850" and insert "\$66,313,434".

Explanation:

(The amendment provides \$3.2 million GF and 33 positions in FY 2022, which represents the cost and staffing necessary to establish a public defender office for Chesterfield County, pursuant to SB 1442, of the 2021 General Assembly Session, contingent upon its final passage. Companion amendments to Items 41, 42, and 43 reduce estimated Criminal Fund (GF) court-appointed attorney prorated costs based on projected first year savings from opening the office, and a companion amendment to Item 48 redirects a portion of funding currently appropriated for Capital Indigent Defense Services, due to the passage of SB 1165, of the 2021 General Assembly Session, which repealed the death penalty.)

	Item 48 #3s		
Judicial Department	FY20-21	FY21-22	
Indigent Defense Commission	\$0 0.00	\$824,277 8.00	GF FTE

Language:

Page 43, line 10, strike "\$63,148,850" and insert "\$63,973,127".

Explanation:

(This amendment provides \$824,277 GF and 8.0 FTE attorney positions in FY 2022 related to SB 1261 of the 2021 General Assembly Session, which expands the jurisdiction of the Court of Appeals, contingent on its final passage.)

	Item 49 #1s		
Judicial Department	FY20-21	FY21-22	
Virginia Criminal Sentencing	\$0	\$333,200	GF
Commission	0.00	2.00	FTE

Language:

Page 43, line 45, strike "\$1,240,651" and insert "\$1,573,851".

Explanation:

(This amendment provides \$333,200 GF and 2.0 FTE positions in FY 2022 for costs associated with SB 1391 of the 2021 General Assembly, which is a recommendation of the Virginia State Crime Commission, contingent upon its final passage.)

	Item 57 #1s		
Executive Offices	FY20-21	FY21-22	
Attorney General and Department of Law	\$0 0.00	\$7,547,029 61.00	GF FTE

Language:

Page 48, line 42, strike "\$38,488,923" and insert "\$46,035,952".

Explanation:

(This amendment provides \$7.5 million GF and 61.00 FTE positions, including 48 attorney positions and 13 administrative support staff positions, associated with SB 1261 of the 2021 General Assembly Session, which expands the jurisdiction of the Court of Appeals, contingent on its final passage.)

		Item 68 #1s	
Administration	FY20-21	FY21-22	
Compensation Board	\$0	\$600,000	GF

Language:

Page 55, line 45, strike "\$499,752,342" and insert "\$500,352,342".

Explanation:

(This amendment provides \$600,000 GF the second year for the Compensation Board to contract for services to be provided by the Virginia Center for Policing Innovation to provide automated protective order notification services as an enhancement to the Statewide Automated Victim Notification System (SAVIN).)

Item 73 #1s

Administration

Compensation Board

Language

Language:

Page 69, after line 6, insert:

"O. Notwithstanding § 17.1-275, Code of Virginia, and any other provision of law, the marriage license fee shall be increased from \$30 to \$50. All marriage license fee revenue collected by the circuit court clerks, less \$30 per marriage license, shall be used exclusively to support compensation increases for the clerk and deputy clerks of the circuit courts."

Explanation:

(This amendment increases the marriage license fee to \$50 and specifies that any additional revenues shall be used exclusively for compensation increases for the clerk and deputy clerks of the circuit courts.)

Item 73 #2s

Administration

Compensation Board

Language

Language:

Page 69, after line 6, insert:

"O. Notwithstanding § 17.1-275, Code of Virginia, and any other provision of law, the divorce filing fee shall be increased from \$86 to \$100. All divorce filing fee revenue collected by the circuit court clerks, less \$86 per divorce filing, and less any revenue directed to the Courts Technology Fund established under §17.1-132, Code of Virginia, shall be used exclusively to support compensation increases for the clerk and deputy clerks of the circuit courts."

Explanation:

(This amendment increases the divorce filing fee to \$100 and specifies that any additional revenues, less that already directed to the Courts Technology Fund, shall be used exclusively for compensation increases for the clerk and deputy clerks of the circuit courts.)

		Item 75 #2s	
Administration	FY20-21	FY21-22	
Compensation Board	\$0	\$250,000	GF

Language:

Page 70, line 16, strike "\$5,514,904" and insert "\$5,764,904". Page 73, after line 41, insert:

"V.1. Pursuant to SB 1226 of the 2021 General Assembly, the Compensation Board shall work with the Virginia Association of Commonwealth's Attorneys to examine the staffing standards used to determine the number of positions needed for allocation to Commonwealth's Attorneys' offices statewide and revise the duties and workload measures used as the basis for determining the allocation of new positions, such that the standards are not based solely on metrics related to felony charges and convictions. The examination shall identify funding needs to support staffing for statutorily prescribed duties while also identifying funding needs for participation in special programs, discretionary duties, and current local supplemental funds allocated. To assist in this goal, the Compensation Board shall contract with the National Center for State Courts to perform a time study as to the comprehensive duties and responsibilities of Commonwealth's Attorneys' offices including, but not limited to, "in-court" obligations, the use of diversion programs and specialty dockets, expungement/rights restoration volume as well as other obligations reflected in the *Code of Virginia* (e.g. duties prescribed under §15.2-1627, et seq). The study shall identify objective metrics for inclusion in the revised staffing standard. Included within this appropriation is \$250,000 in the second year from the general fund for the purpose of contracting with the Center to perform the study. All Commonwealth's Attorneys shall participate in the study as needed and identified by the Compensation Board and the National Center for State Courts.

2. The Compensation Board shall provide a status report on the progress of the study and participants to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021. The Compensation Board shall deliver a report containing the results of the study, anticipated costs, and staffing standards methodology revisions under review or approved by the Board to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2022."

Explanation:

(This amendment provides \$250,000 GF in FY 2022 for the Compensation Board to contract with the National Center for State Courts to assist the Compensation Board in a study, working in collaboration with the Virginia Association of Commonwealth's Attorneys, of the staffing standards for Commonwealth's Attorney's offices, to revise the workload measures used as the basis for the allocation of new positions so that they are not based solely on metrics related to felony charges and convictions. The budget language is supplemental to the provisions of SB 1226 of the 2021 General Assembly.)

Item 75 #3s

Administration

Compensation Board

Language

Language:

Page 73, after line 41, insert:

"V. The Compensation Board shall review the plan to be developed by the Department of Criminal Justice Services by July 1, 2021 outlining law enforcement agencies' roles and engagement with the development of the Mental Health Awareness Response and Community Understanding Services Alert System, established pursuant to House Bill 5043 and Senate Bill 5038 of the 2020 Special Session I of the General Assembly, and shall survey sheriffs' offices to determine anticipated costs to support staffing and training needs to meet the requirements established by the plan. The Compensation Board shall provide a report to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021 of the findings of the survey and estimated costs to meet the requirements established by the plan."

Explanation:

(This amendment adds language directing the Compensation Board to review the plan to be developed by the Department of Criminal Justice Services by July 1, 2021 outlining law enforcement agencies' roles and engagement with the development of the Mental Health Awareness Response and Community Understanding Services Alert System, established pursuant to House Bill 5043 and Senate Bill 5038 of the 2020 Special Session I of the General Assembly and to provide a report to the Chairs of the Money Committees by November 1, 2021.)

Item 83 #2s

Administration

Department of Human Resource Management

Language

Language:

Page 82, after line 28, insert:

"N.1. In order to ensure the continuity of state government operations during the COVID-19 pandemic, the Department of Human Resource Management (DHRM), and with the assistance of all other state agencies upon their request, shall establish the criteria and develop a return-towork policy for all state employees presently working remotely on either a partial or full-time basis, pursuant to applicable federal and state guidelines.

2. Furthermore, DHRM, in collaboration with the Department of General Services, the Virginia Information Technologies Agency, and any other state agency upon request, shall examine the Commonwealth's existing telework policies, and how agency program and service delivery tools and methodologies employed during the COVID-19 pandemic may inform future policy objectives regarding the use of telework and alternative work schedules as a means of achieving administrative efficiencies, and reducing cost, and sustaining the hiring and retention of a highly qualified workforce. DHRM shall report to the Governor and the Chairmen of the House Appropriations and Senate Finance and Appropriations Committees on its findings, and suggested policies, instructions and guidelines by September 1, 2021."

Explanation:

(This amendment directs the Department of Human Resource Management, in collaboration with other state agencies, to develop return-to-work policies for state government employees to ensure the continuity and performance of state government operations, to examine agency performance and sevice delivery, and to examine the Commonwealth's telework policies and how they may inform future policy objectives. A report on such findings and recommendations is due to the Governor and the Chairmen of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021.)

		Item 87 #1s	
Administration	FY20-21	FY21-22	
Department of Elections	\$0	\$2,534,575	GF

Language:

Page 85, line 51, strike "\$6,275,378" and insert "\$8,809,953".

Explanation:

(This amendment restores the full reimbursement for compensation for general registrars and electoral board members that was included in the budget as passed in March 2020, but unallotted in April and removed during the 2020 Special Session I.)

Item 92 #1s

Administration

Virginia Information Technologies Agency

Language

Language:

Page 90, after line 41, insert:

"F. The Virginia Information Technologies Agency shall provide a network infrastructure report to the House Appropriations Committee, Senate Finance and Appropriations Committee, and

Joint Legislative Audit and Review Commission by November 1 of each year. The report shall indicate whether the Commonwealth's network infrastructure is adequate to meet the needs of state agencies, and if not, identify any needed upgrades. For each network infrastructure upgrade identified, the report shall specify the estimated cost and whether the upgrade is to the portion of the network maintained by the Virginia Information Technologies Agency or another state agency."

Explanation:

(This amendment directs the Virginia Information Technologies Agency (VITA) to report, by November 1 of each year, on the adequacy of the Commonwealth's network infrastructure maintained by either VITA or another state agency, and any potential upgrades needed, and associated costs. This language is a recommendation of the Joint Legislative Audit and Review Commission.)

		Item 95 #1s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Agriculture and Consumer Services	\$0	(\$100,000)	GF

Language:

Page 94, line 13, strike "\$5,642,932" and insert "\$5,542,932".

Page 94, line 20, strike "\$600,000" and insert "\$500,000".

Page 94, line 23, strike "legislation to be", and insert "the provisions of Senate Bill 1188 of".

Page 94, line 24, strike "considered by".

Explanation:

(This amendment reduces by \$0.1 million GF the proposed deposit to the Virginia Agriculture Food Assistance Program established pursuant to SB 1188 of the 2020 General Assembly. The Committee amendments provide a total of \$500,000 for this new initiative beginning in FY 22.)

		Item 97 #1s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Agriculture and Consumer Services	\$250,000	\$0	GF

Language:

Page 94, line 40, strike "\$23,620,243" and insert "\$23,870,243". Page 96, unstrike line 15 through line 16.

Explanation:

(This amendment restores \$250,000 GF in the first year for the Holiday Lake 4-H Center that was approved in 2020 Regular Session and subsequently unallotted. Holiday Lake is an evacuation location for several public school systems in Central Virginia.)

		Item 97 #2s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Agriculture and Consumer Services	\$0	\$131,649	GF

Language:

Page 94, line 41, strike "\$21,892,069" and insert "\$22,023,718". Page 96, line 8, strike "\$1,120,226" and insert "\$1,620,226".

Explanation:

(This amendment increases funding for the International Marketing activities of the Virginia Department of Agriculture and Consumer Services by \$0.1 million GF in the second year to restore reductions included in Chapter 1289 of the 2020 General Assembly.)

Item 105 #1s

Agriculture and Forestry

Department of Agriculture and Consumer Services

Language

Language:

Page 99, following line 18, insert:

"D. The Office of the State Inspector General shall, with the assistance of the Office of Charitable and Regulatory Programs, review the regulatory structure of charitable gaming in Virginia, to include, at a minimum: (i) current permitting requirements and exemptions, (ii) net revenue dedicated to charitable activities and which types of gaming revenue is excluded from this calculation, (iii) charitable gaming occurring in remote locations not located in the same jurisdiction as the registered address of the charitable organization, (iv) enforcement of the "social quarters" and "members and guests" limitation, (v) the structure of the Charitable Gaming Board including any changes needed to prevent conflicts of interest, (vi) the adequacy of enforcement and resources dedicated to oversight activities of the Office of Charitable and Regulatory Programs, and (vii) whether regulation of charitable gaming would be more appropriately vested with the Virginia Lottery. The Office of the State Inspector General shall report on their findings to the General Assembly no later than December 1, 2021."

Explanation:

(This amendment directs the Office of the State Inspector General to investigate the operations

of the Charitable Gaming Board and recommend any changes needed to prevent conflicts of interest, including an assessment of whether responsibility for the regulation of charitable gaming should be placed under the Virginia Lottery.)

		Item 107 #1s	
Agriculture and Forestry	FY20-21	FY21-22	
Department of Forestry	\$0	(\$316,146)	GF

Language:

Page 99, line 47, strike "\$37,431,710" and insert "\$37,115,564".

Explanation:

(This amendment removes funding for a proposed new initiative between the Department of Forestry and the Virginia Department of Emergency Management. A companion amendment reverses the proposed new initiative at VDEM.)

		Item 112 #1s	
Commerce and Trade	FY20-21	FY21-22	
Economic Development Incentive Payments	\$0	\$6,330,000	GF

Language:

Page 105, line 19, strike "\$58,585,483" and insert "\$64,915,483".

Page 107, after line 47, insert:

"O. Out of the appropriation in this Item, \$6,330,000 the second year from the general fund shall be deposited to a special, nonreverting fund for the award of grants to a qualified shipping and logistics company in a qualified locality in accordance with legislation to be considered by the 2022 General Assembly and subject to performance metrics agreed to in a memorandum of understanding with the Commonwealth."

Explanation:

(This amendment provides \$6.3 million the second year from the general fund for the award of grants to a qualified shipping and logistics company. This is part of an incentives package for an economic development project endorsed by the Major Employment and Investment (MEI) Project Approval Commission.)

Commerce and Trade	FY20-21	FY21-22	
Economic Development Incentive Payments	\$0	\$1,000,000 G	F

Language:

Page 105, line 19, strike "\$58,585,483" and insert "\$59,585,483". Page 106, line 23, strike "\$3,000,000" and insert "\$4,000,000".

Explanation:

(This amendment provides \$1.0 million the second year from the general fund for the Governor's Motion Picture Opportunity Fund, restoring second year funding that was previously unallotted.)

		Item 113 #1s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	(\$15,700,000)	\$0	GF

Language:

Page 108, line 13, strike "\$168,760,089" and insert "\$153,060,089".

Page 109, line 5, strike "\$70,700,000" and insert "\$55,000,000".

Page 109, line 15, strike "\$28,200,000" and insert "\$12,500,000".

Page 109, after line 48, insert:

"5. It is the intent of the General Assembly that funds from Virginia's allocation of moneys distributed under the federal Consolidated Appropriations Act, P.L. 116-260 (2020) for rental assistance be designated to continue the Virginia Rent and Mortgage Relief Program, and that such funds be given priority over general fund appropriations provided in paragraph E.1. of this item in supporting the continuation of the Virginia Rent and Mortgage Relief Program."

Explanation:

(This amendment reduces the GF appropriation to the Virginia Housing Trust Fund intended to support the Virginia Rent and Mortgage Relief Program and directs that funding distributed under the federal Consolidated Appropriations Act, P.L. 116-260 (2020) for rental assistance be designated as the primary source of funding for the Commonwealth's rental assistance program.)

Item 114 #1s

Commerce and Trade FY20-21 FY21-22

Department of Housing and \$0 \$500,000 GF Community Development

Language:

Page 110, line 42, strike "\$128,538,362" and insert "\$129,038,362".

Page 112, line 32 strike "\$2,500,000" and insert "\$3,000,000".

Page 112, line 34, strike" each year" and insert "the first year and \$1,500,000 the second year.

Explanation:

(This amendment would provide an additional \$500,000 GF the second year for the Industrial Revitalization Fund and designate the increase for the removal, renovation or modernization of port-related buildings and facilities in the cities of Portsmouth, Norfolk, Newport News, Richmond, or Front Royal.)

		Item 114 #2s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0	\$500,000	GF

Language:

Page 110, line 42, strike "\$128,538,362" and insert "\$129,038,362".

Page 114, after line 46, insert:

"O. Out of the amounts in this Item, \$500,000 the second year from the general fund is provided for the Lenowisco Planning District Commission and Cumberland Plateau Planning District Commission designated for initiatives intended to expand education and telehealth access. Such funds for grants shall be managed by the Virginia Coalfield Economic Development Authority."

Explanation:

(This amendment would provide \$500,000 GF the second year for the Lenowisco Planning District Commission and Cumberland Plateau Planning District Commission designated for initiatives intended to expand education and telehealth access. Funding for grants would be managed by the Virginia Coalfield Economic Development Authority.)

		Item 114 #3s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0 0.00	\$10,000,000 3.00	GF FTE

Language:

Page 110, line 42, strike "\$128,538,362" and insert "\$138,538,362". Page 114, after line 46, insert:

- "O.1. Out of the amounts in this Item, \$10,000,000 the second year from the general fund is provided to establish a special, non-reverting Virginia Community Development Financial Institutions (CDFI) Fund to provide grants to community development financial institutions (CDFIs), community development enterprises (CDE), or other such similar entities as permitted by law, whose primary purpose is to provide financing in the form of loans, grants or forgivable loans to small businesses or community revitalization real estate projects in Virginia. The Fund shall consist of any funds appropriated to it by the general appropriation act and revenue from any other source, public or private. The Fund shall be established on the books of the Comptroller, and any funds remaining in the Fund at the end of a biennium shall not revert to the general fund but shall remain in the Fund. Interest earned on the Fund shall be credited to the Fund. Of the amounts included in this paragraph, up to \$300,000 the second year from the general fund is provided to the Department for administrative costs.
- 2. The Department is hereby authorized to develop appropriate criteria and guidelines for the use of funding provided to the Virginia Community Development Financial Institution Fund. The Department shall award grant funding based on these criteria and guidelines and may enter into a contractual agreement with eligible CDFIs or similar private entities to make grants and lending to small businesses adversely impacted by the COVID pandemic. An eligible qualifying CDFI shall be a community development bank, community development credit union, or other similar private entity that the Department finds is (i) established to conduct business legally within the Commonwealth; (ii) subject to oversight by federal or state financial institutions or insurance regulatory agencies, as appropriate; and (iii) eligible for certification by the U.S. Department of Treasury as a community development financial institution or other similar charter or principles which require support of small businesses.
- 3. The community development bank, community development credit union, or other similar organization is intended to be a source of targeted lending and investment with the capacity to provide a high degree of leveraging for economic development and business support activities within communities throughout the Commonwealth. These activities may include loans and investments to start or expand small businesses, operating and working capital, property renovation or development, and financial services with a focus on small businesses impacted by the COVID pandemic. The entity may also provide services that help ensure that credit is used effectively, such as technical assistance to small businesses and credit counseling to consumers.
- 4. On or before December 1 of each year, the Department shall report to the Secretary of Commerce and Trade, the Governor, and the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations on such other matters regarding the Fund as the Department may deem appropriate, including the amount of funding committed to projects from the Fund, or other items as may be requested by any of the foregoing persons to whom such report is to be submitted."

Explanation:

(This amendment establishes the Virginia CDFI Fund (the Fund) within the Department of Housing and Community Development which may be comprised of appropriations, grants and loans from federal, state and private sources, and other types of financial assistance, to provide capital through grants to community development financial institutions (CDFIs), community development enterprises (CDE), or other such similar entities as permitted by law, whose primary purpose is to provide financing in the form of loans, grants or forgivable loans to small business or community revitalization real estate projects in Virginia. The amendment provides \$10.0 million GF the second year to capitalize the Fund.)

Item 114 #4s

Commerce and Trade

Department of Housing and Community Development

Language

Language:

Page 114, line 46, after "basis.", insert:

"O. The Commission on Local Government shall review the fiscal effects of mandatory property tax exemptions on the capacity of local governments to deliver essential services to the public. As part of the review, the Commission shall ascertain the impact of the exemptions on property tax collections, the shift to and reliance on other local revenues to compensate for exempted properties, the additional fiscal stress placed on non-exempted properties and non-exempted local taxpayers, and the ability of local governments to meet spending needs. In addition, the Commission shall develop and include in its review potential recommendations to mitigate the fiscal impacts on local governments tied to these state tax exemption initiatives. The Commission shall report its findings to the Governor, the Joint Subcommittee on Local Government Fiscal Stress, and the Chairs of the Senate Committee on Finance and Appropriations, the House Committee on Finance, and the House Committee on Appropriations by November 1, 2021."

Explanation:

(This amendment directs the Commission on Local Government to undertake a review of the effects of mandatory property tax exemptions on local governments and recommend potential options for mitigating their fiscal impacts.)

Item 114 #5s

Commerce and Trade

Department of Housing and Community Development

Language

Language:

Page 113, line 8, after "private sector", insert ", except as provided for in paragraph L.6. of this

Page 113, after line 36, insert:

"6. The Department shall create a pilot program within VATI, with awards not to exceed 10 percent of total available VATI funds in fiscal year 2022, to which public broadband authorities may apply without investment from the private sector."

Explanation:

(This amendment directs the Department of Housing and Community Development to create a pilot program within VATI, with awards not to exceed 10 percent of total available VATI funds in FY22, to which public broadband authorities may apply without investment from the private sector. This is the recommendation of the Broadband Advisory Council.)

		Item 114 #6s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0	\$294,000	GF
_			

```
Language:
Page 110, line 42, strike "$128,538,362" and insert "$128,832,362".
Page 111, line 11, strike "$75,971" and insert "$89,971".
Page 111, line 15, strike "$75,971" and insert "$89,971".
Page 111, line 20, strike "$75,971" and insert "$89,971".
Page 111, line 22, strike "$75,971" and insert "$89,971".
Page 111, line 24, strike "$75,971" and insert "$89,971".
Page 111, line 26, strike "$75,971" and insert "$89,971".
Page 111, line 28, strike "$75,971" and insert "$89,971".
Page 111, line 30, strike "$151,943" and insert "$165,943".
Page 111, line 32, strike "$75,971" and insert "$89,971".
Page 111, line 34, strike "$75,971" and insert "$89,971".
Page 111, line 36, strike "$75,971" and insert "$89,971".
Page 111, line 38, strike "$75,971" and insert "$89,971".
Page 111, line 40, strike "$75,971" and insert "$89,971".
Page 111, line 42, strike "$75,971" and insert "$89,971".
Page 111, line 44, strike "$113,957" and insert "$127,957".
Page 111, line 46, strike "$75,971" and insert "$89,971".
Page 112, line 1, strike "$75,971" and insert "$89,971".
Page 112, line 3, strike "$75,971" and insert "$89,971".
Page 112, line 5, strike "$75,971" and insert "$89,971".
```

Page 112, line 7, strike "\$75,971" and insert "\$89,971". Page 112, line 9, strike "\$151,943" and insert "\$165,943".

Explanation:

(This amendment provides an additional \$294,000 the second year from the general fund to provide each of the 21 Planning District Commissions with an increase of \$14,000 per year. The 2020 Appropriation Act provided this amount in both years of the biennium; however, the increase was unallotted due to COVID-19 impacts to the state budget.)

		Item 115 #1s	
Commerce and Trade	FY20-21	FY21-22	
Department of Housing and Community Development	\$0	\$250,000	GF

Language:

Page 114, line 48, strike "\$14,789,114" and insert "\$15,039,114". Page 114, line 56, strike "\$14,500,000" and insert "\$14,750,000".

Explanation:

(This amendment restores \$250,000 GF the second year that was unallotted and subsequently reduced in the Enterprise Zone program.)

		Item 120 #1s	
Commerce and Trade	FY20-21	FY21-22	
Department of Labor and Industry	(\$175,073)	(\$500,290)	GF

Language:

Page 116, line 14, strike "\$1,773,255" and insert "\$1,598,182".

Page 116, line 14, strike "\$2,520,193" and insert "\$2,019,903".

Page 116, line 18, strike "\$596,794" and insert "\$421,721".

Page 116, line 18, strike "\$1,343,732" and insert "\$843,442".

Explanation:

(This amendment removes funding in the amounts of \$175,073 the first year and \$500,290 the second year from the general fund for the Labor and Employment Law Division. This is funding that was included in Chapter 1289, 2020 Acts of Assembly, to support implementation and enforcement of the provisions of SB 481 (2020), which ultimately did not pass.)

Commerce and Trade

Department of Small Business and Supplier Diversity

Language

Language:

Page 122, after line 37, insert:

"I.1.Notwithstanding § 2.2-1604, Code of Virginia, "Small business" shall be defined as a business that is at least 51 percent independently owned and controlled by one or more individuals, or in the case of a cooperative association organized pursuant to Chapter 3 (§ 13.1-301 et seq.) of Title 13.1 as a nonstock corporation, is at least 51 percent independently controlled by one or more members, who are U.S. citizens or legal resident aliens and, together with affiliates, has 250 or fewer employees or average annual gross receipts of \$10 million or less averaged over the previous three years. One or more of the individual owners or members shall control both the management and daily business operations of the small business.

2. Notwithstanding § 2.2-4310, Code of Virginia, and for purposes of the Commonwealth's SWaM program certification, "Small business" shall mean (i) a business, independently owned and controlled by one or more individuals, or (ii) in the case of a cooperative association organized pursuant to Chapter 3 (§ 13.1-301 et seq.) of Title 13.1 as a nonstock corporation, controlled by one or more members, who are U.S. citizens or legal resident aliens, and together with affiliates, has 250 or fewer employees, or annual gross receipts of \$10 million or less averaged over the previous three years. One or more of the individual owners or members shall control both the management and daily business operations of the small business."

Explanation:

(This amendment amends the definition of small business to include certain co-ops.)

		Item 130 #1s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Economic Development Partnership	\$0	\$2,000,000	GF

Language:

Page 123, line 44, strike "\$40,302,309" and insert "\$42,302,309".

Page 124, line 49, after "the first year and", strike "\$5,020,387" and insert "\$7,020,387".

Explanation:

(This amendment provides an additional \$2.0 million GF the second year for the Talent Accelerator Program to provide custom recruitment and training services for high-value projects creating new jobs in Virginia. Funding supports staff compensation and benefits, contract trainers, program support, client recruitment, travel and other expenses.)

Item 130 #2s

Commerce and Trade	FY20-21	FY21-22	
Virginia Economic Development Partnership	\$0	\$930,000 GF	

Language:

Page 123, line 44, strike "\$40,302,309" and insert "\$41,232,309".

Page 125, after line 4, insert:

"M. Out of the amounts in this item, \$930,000 the second year from the general fund is provided to support implementation of Virginia's International Trade Plan."

Explanation:

(This amendment provides \$930,000 GF the second year to support initiatives identified in the International Trade Plan.)

		Item 131 #1s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Employment Commission	\$0	(\$6,752,701)	GF

Language:

Page 125, line 13, strike "\$574,596,796" and insert "\$567,844,095".

Page 126, line 11, strike "\$7,502,701" and insert "\$750,000".

Explanation:

(This amendment updates the estimated interest on federal cash advances for unemployment insurance benefits.)

Item 134 #1s

Commerce and Trade

Virginia Tourism Authority

Language

Language:

Page 127, line 27, strike "Coalfield Regional Tourism Authority" and insert "Heart of Appalachia Tourism Authority".

Explanation:

(This amendment updates a reference to the Coalfield Regional Tourism Authority to the Heart

of Appalachia Tourism Authority in accordance with the name change effectuated in SB 1399. This is a technical amendment.)

		Item 134 #2s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Tourism Authority	\$0	\$2,140,000	GF

Language:

Page 126, line 39, strike "\$21,093,272" and insert "\$23,233,272".

Page 128, after line 20, insert:

"M. Out of the amounts in this Item, \$2,140,000 the second year from the general fund is provided for grants to promote tourism in accordance with the provisions of Senate Bill 1398, as enacted during the 2021 Regular Session of the General Assembly."

Explanation:

(This amendment provides \$2,140,000 the second year from the general fund for grants to promote tourism pursuant to SB 1398. This is the amount of estimated GF revenue raised by sales tax imposed on accommodations fees.)

		Item 135 #1s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Innovation Partnership Authority	\$0	\$1,400,000	GF

Language:

Page 128, line 28, strike "\$40,450,000" and insert "\$41,850,000".

Page 132, line 16, strike "\$925,000" and insert "\$1,925,000".

Page 132, line 29, after "and" strike the remainder of the line.

Page 132, strike lines 30 through 31, and insert:

"(iv) grants to CCAM for seedling research project costs that enable CCAM to market new research programs to prospective and existing industry members. These funds shall not revert back to the general fund at the end of the fiscal year."

Page 132, line 39, after "facility." insert: "These funds shall not revert back to the general fund at the end of the fiscal year."

Page 132, after line 39, insert:

"4. Out of the appropriation in this item, VIPA shall provide \$400,000 the second year from the general fund to CCAM for the purposes of: (i) attracting federal funds for research projects to be conducted at CCAM, including marketing, travel, grant proposal writing, and business development costs; (ii) matching funds for federal research programs; and (iii) federal research

program costs not reimbursable on federal research awards. These funds shall not revert back to the general fund at the end of the fiscal year."

Page 132, line 40, strike "4" and insert "5".

Page 132, line 42, strike "5" and insert "6".

Page 132, line 48, after "disbursements;" insert:

"(iv) all efforts and costs associated with obtaining federal research grants".

Page 132, line 48, strike "iv" and insert "v".

Explanation:

(This amendment provides an increase of \$1.0 million GF in rent, operating support and maintenance for the Center for Advanced Manufacturing (CCAM), bringing the total to \$1.9 million in both FY 2021 and FY 2022. It includes increased flexibility on currently appropriated matching funds and provides an additional \$400,000 in GF resources for CCAM to attract federal funds.)

		Item 135 #2s	
Commerce and Trade	FY20-21	FY21-22	
Virginia Innovation Partnership Authority	\$0	\$100,000	GF

Language:

Page 128, line 28, strike "\$40,450,000" and insert "\$40,550,000". Page 134, after line 16, insert:

"S.1. Out of the appropriation in this item, \$100,000 the second year from the general fund is provided for the Virginia Nuclear Energy Consortium Authority (VNECA) for the purpose of developing a proposal to create a nuclear research and innovation hub in Virginia. In creating this proposal, VNECA shall convene a workgroup that includes, but is not limited to, the Department of Minerals. Mines and Energy, the Virginia Economic Development Partnership.

Department of Minerals, Mines and Energy, the Virginia Economic Development Partnership, the Virginia Innovation Partnership Authority, Virginia public colleges and universities, and relevant industry representatives.

2. VNECA shall submit a report that includes planning activities and the final proposal to the Secretary of Commerce and Trade, Secretary of Education, Chairs of the House Appropriations Committee, the House Labor and Commerce Committee, the Senate Finance and Appropriations Committee, and the Senate Commerce and Labor Committee no later than November 1, 2021."

Explanation:

(This amendment provides \$100,000 GF the second year for the Virginia Nuclear Energy Consortium Authority as a one-time planning grant to develop a nuclear research and

innovation hub in Virginia.)

Item 137 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 139, after line 29, insert:

"Q. The Superintendent of Public Instruction shall convene a workgroup to make recommendations on the staffing standards for school nurses in local school divisions. The workgroup shall include recommendations on: (i) the appropriate staffing ratio of school nurses; and (ii) the desired qualifications and training for school nurses. The workgroup shall include at least (i) the Secretary of Education; (ii) two local school division representatives, including one superintendent; (iii) a member of a local school board; (iv) two school nurses; and (v) one member of the Board of Education. The recommendations shall be submitted to the Governor and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than October 1, 2021. Such recommendations shall detail any necessary legislative or budgetary changes to implement recommendations."

Explanation:

(This amendment directs the Superintendent of Public Instruction to convene a workgroup to make recommendations on the staffing standards for school nurses. The recommendations shall include the appropriate school nurse staffing ratio and the required qualifications and training for school nurses.)

Item 137 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 139, after line 29, insert:

"Q. To support the development and administration of a growth assessment system, \$10,000,000 in federal relief funds are provided from the federal funds reserved by the Department of Education from the Coronavirus Response and Relief Supplemental Appropriations Act, (P.L. 116-260) for such purpose. The system shall include diagnostic assessments in the fall and winter and a growth measurement assessment administered in the spring to measure student progress and competency in grades three through eight in English and mathematics for use in the 2021-2022 and 2022-2023 school year, pursuant to the provisions in Senate Bill 1357 of the 2021 General Assembly."

Explanation:

(This amendment provides \$10.0 million in federal relief funds for the Department of Education to provide and administer a diagnostic assessment tool for use in the 2021-22 and 2022-23 school year pursuant to the provisions of Senate Bill 1357.)

Item 138 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 11, insert:

"6. The Department of Education shall collect and publish data annually from each private special education day school on: (i) the number of teachers who are not fully endorsed in the content that they are teaching; (ii) the number of teachers who have less than one year of classroom experience; (iii) the number of teachers who are provisionally licensed; (iv) the type of academic credentials attained by each teacher and in what subjects; (v) the number of career and technical education credentials conferred by each school on its graduating students in each of the three prior academic years; (vi) each school's accreditation status, including the accrediting body; and (vii) the number of incidents of restraint and seclusion occurring in each of the previous three academic years."

Page 141, after line 14, insert:

"I. The Board of Education shall develop and promulgate regulations for private special education day schools on restraint and seclusion that establish the same requirements for restraint and seclusion as those for public schools."

Explanation:

(This amendment implements recommendation 8 and 9 from the 2020 JLARC report on the "Review of the Children's Services Act and Private Special Education Day School Costs".)

Item 138 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall develop and maintain a robust statewide plan for

improving (i) its ongoing oversight of local practices related to transition planning and services and (ii) technical assistance and guidance provided for post-secondary transition planning and services. The plan shall articulate how the Department will reliably and comprehensively assess the compliance and quality of transition plans for students with disabilities in Virginia on an ongoing basis and detail how the Department will communicate findings to local school division staff and local school boards. The plan shall be submitted to the Senate Education and Health and the House Education committees no later than December 1, 2022. The Department shall update those committees annually detailing its progress on implementing the plan."

Explanation:

(This amendment implements recommendation 9 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 138 #3s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall develop clear and simplified guidance, in multiple languages, for families of students with disabilities explaining (i) the limitations of the applied studies diploma; (ii) key curriculum and testing decisions that reduce the likelihood of a student being able to obtain a standard diploma; and (iii) that pursuit of an applied studies diploma may preclude a student's ability to pursue a standard diploma."

Explanation:

(This amendment implements recommendation 10 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 138 #4s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall revise the state's special education complaint procedures and practices to ensure the Department requires and enforces corrective actions that (i) achieve full and appropriate remedies for school divisions' non-compliance with special education laws and regulations, including, at a minimum, requiring school divisions to provide compensatory

services to students with disabilities when the Department determines divisions did not provide legally obligated services; and (ii) ensure that relevant personnel understand how to avoid similar non-compliance in the future."

Explanation:

(This amendment implements recommendation 21 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 138 #5s

Education

Department of Education, Central Office Operations

Language

Language:

Page 141, after line 14, insert:

"I. The Department of Education shall (i) conduct a one-time targeted review of the transition sections of a random sample of students' individualized education programs (IEPs) in each school division; (ii) communicate its findings to each local school division, school board, and local special education advisory committee; and (iii) ensure local school divisions correct any IEPs that are found out of compliance. The Department shall submit to the Senate Education and Health and the House Education Committees a letter certifying that school divisions have corrected all instances of non-compliance identified through these reviews, no later than May 1, 2022."

Explanation:

(This amendment implements recommendation 8 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

		Item 138 #6s	
Education	FY20-21	FY21-22	
Department of Education, Central Office Operations	\$0	\$300,000	GF

Language:

Page 139, line 30, strike "\$17,352,182" and insert "\$17,652,182".

Page 141. after line 14, insert:

"I. Out of this appropriation, \$300,000 the second year from the general fund is provided for the Department of Education to: (i) provide training and guidance documents to local school divisions on the development of individualized education programs (IEPs); (ii) develop a required training module for each individual who participates in an IEP meeting that

comprehensively addresses and explains in detail each IEP team member's respective role in the IEP meeting, the IEP development process, and components of effective IEPs; and (iii) conduct structured reviews of a sample of IEPs annually to verify that the IEPs are in compliance with state and federal laws and regulations."

Explanation:

(This amendment provides \$300,000 GF the second year for costs associated with implementation of Senate Bill 1288.)

		Item 139 #1s	
Education	FY20-21	FY21-22	
Department of Education, Central Office Operations	\$0	(\$1,812,000)	GF

Language:

Page 141, line 15, strike "\$39,750,487" and insert "\$37,938,487".

Explanation:

(This amendment captures savings in the amount of \$1.8 million GF from the reduction of Standards of Learning assessments to the minimum federal requirements, contingent on the passage of Senate Bill 1401.)

Item 141 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 143, after line 46, insert:

"F. To support a school division's needs for an analytics solution to evaluate student progress and determine instructional gaps, the Department of Education may provide funds out of this appropriation as one-time grants to divisions to support the costs of such analytics solution."

Explanation:

(This amendment allows the Department of Education to provide one-time grants to divisions to support the costs of an analytics solution to evaluate student progress and determine instructional gaps.)

Item 141 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 143, after line 46, insert:

"F. Virtual Education Supports

To continue the expansion of Virtual Virginia's Outreach Program initiated or expanded with federal funds from the Coronavirus Aid, Relief, and Economic Security (CARES) Act (P.L. 116-136), \$7,000,000 in Governor's Education Emergency Relief (GEER) funds from the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) are provided to support the continued expansion costs in the second year."

Explanation:

(This amendment provides \$7.0 million in Governor's Education Emergency Relief Funds from the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) to support the continued expansion of Virtual Virginia's learning management system and additional free course enrollment slots during the 2021-22 school year.)

Item 142 #1s

Education

Department of Education, Central Office Operations

Language

Language:

Page 144, after line 28, insert:

- "G. 1. The Board of Education shall review and update its regulations of general education K–12 teacher preparation programs to ensure graduates are required to demonstrate proficiency in: (i) differentiating instruction for students depending on their needs; (ii) understanding the role of general education teachers on the IEP team; (iii) implementing effective models of collaborative instruction, including co-teaching; and (iv) understanding the goals and benefits of inclusive education for all students.
- 2. The Board of Education shall review and update its regulations governing administrator preparation programs to ensure graduates are required to demonstrate comprehension of: (i) key special education laws and regulations; (ii) individualized education program (IEP) development; (iii) the roles and responsibilities of special education teachers; and (iv) appropriate behavior management practices."

Explanation:

(This amendment implements recommendation 15 and 17 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

Item 142 #2s

Education

Department of Education, Central Office Operations

Language

Language:

Page 144, after line 28, insert:

"G. The Department of Education shall develop and maintain a statewide strategic plan for recruiting and retaining special education teachers. At a minimum, the strategic plan shall (i) use data analyses to determine divisions' specific staffing needs on an ongoing basis; (ii) evaluate the potential effectiveness of strategies for addressing recruitment and retention challenges, including tuition assistance, differentiated pay for special education teachers, and the expansion of special education teacher mentorships; and (iii) estimate the costs of implementing each strategy, including the extent to which federal funds could be used to support implementation. The Department shall submit its plan to the Chairs of the House Education Committee and the Senate Education and Health Committee no later than November 1, 2021, and update those committees annually on its progress implementing the plan."

Explanation:

(This amendment implements recommendation 19 from the 2020 JLARC report on "K-12 Special Education in Virginia".)

		Item 144 #1s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$300,000	GF

Language:

Page 146, line 23, strike "\$43,069,426" and insert "\$43,369,426".

Explanation:

(This amendment provides \$300,000 GF the second year to restore the VPI Minority Fellowship. The introduced budget included the language, but the appropriated amount was not included in the Item.)

Item 144 #2s

Education	FY20-21	FY21-22
Direct Aid to Public Education	\$0	\$250,000 GF

Language:

Page 146, line 22, strike "\$43,069,426" and insert "\$43,319,426".

Page 147, after line 4, insert: "Dual Enrollment Passport Pilots \$0 \$250,000"

Page 157, after line 31, insert:

"UU. Out of this appropriation, \$250,000 the second year from the general fund is provided for grants to support one-time pilot programs to school divisions to redesign dual enrollment course offerings to align/link to the Passport and Uniform Certificate of General Studies offered by Virginia's community colleges. Divisions awarded such grants shall collaborate with the local community college to effectively redesign the local school division's dual enrollment course offerings and increase the number of qualified teachers to teach dual enrollment courses. Divisions applying shall include: (i) an explanation of why such dual enrollment pilot program is warranted; (ii) the dual enrollment courses currently offered by the division; (iii) the projected student enrollment in dual enrollment courses; and (iv) the number of the division's employed staff qualified to teach dual enrollment and the number currently teaching a dual enrollment course. The Department of Education may consider in the awarding of a grant: (i) the division's local composite index; (ii) the level of misalignment in the division's dual enrollment course offerings to the Passport and Uniform Certificate of General Studies; and (iii) the division's level of dual enrollment course availability and current student enrollment in those courses. The Department of Education shall report, along with the divisions and community colleges, the components of the redesign and efforts to increase availability and participation in dual enrollment courses to the General Assembly by November 1, 2022. The Department of Education and the Virginia Community College System shall use these pilot programs to provide a comprehensive guide to every school division and community college to assist with aligning high school dual enrollment course offerings to the Passport and Uniform Certificate of General Studies."

Explanation:

Language:

(This amendment provides \$250,000 GF the second year for grants to support one-time pilot programs to redesign a division's dual enrollment course offerings to align/link to the Passport and Uniform Certificate of General Studies offered by Virginia's community colleges.)

		Item 144 #3s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$250,000	GF

Page 146, line 23, strike "\$43,069,426" and insert "\$43,319,426".

Page 146, after line 33, insert: "Active Learning Grants \$0 \$250,000".

Page 157, after line 31, insert:

"UU. Out of this appropriation, \$250,000 the second year from the general fund shall be provided for grants to school divisions for encouraging active in-class, remote and hybrid learning for students in pre-kindergarten through the second grade. School divisions seeking to apply for this grant shall submit a proposal to the Department of Education outlining the intended use of funds and a projected number of students to be served. The Department shall establish criteria for awarding these funds. The funds may be used to purchase a platform featuring on-demand activities that integrate math and English Standards of learning content into movement-rich activities that can be used at school, home and on all devices (i.e., computers, tablets and phones)."

Explanation:

(This amendment provides \$250,000 GF the second year to encourage active learning for students in pre-K through 2nd grade. The Department of Education will establish criteria by which to award these funds to school divisions. Local school divisions will use the grant funds to provide active learning curriculum that relate math and language Standard of Learning content into movement rich activities for students in pre-K through 2nd grade.)

		Item 145 #1s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	(\$15,897,461)	\$0	GF

Language:

Page 157, line 33, strike "\$7,722,644,551" and insert "\$7,706,747,090".

Page 159, line 12 strike "\$88,303,142" and insert "\$72,405,681".

Page 181, line 27 strike "\$71,703,142" and insert "\$55,805,681".

Explanation:

(This amendment captures \$15.9 million GF the first year from the undistributed balance in the Virginia Preschool Initiative program after updating final participation. The introduced budget retained \$21.0 million GF the first year in the undistributed balance to ensure that there were sufficient funds available to school divisions that exercised the enrollment flexibility contained in *Superintendent's Memo #282-20*.)

 Item 145 #2s

 Education
 FY20-21
 FY21-22

Direct Aid to Public Education

\$145,636

\$0 GF

Language:

Page 157, line 33, strike "\$7,722,644,551" and insert "\$7,722,790,187". Page 158, line 45, strike "\$18,993,450" and insert "\$19,139,086".

Explanation:

(This amendment updates enrollment for Governor's School in FY 2021.)

Item	145	#39

Education FY20-21 FY21-22
Direct Aid to Public Education (\$2,500,000) \$0 GF

Language:

Page 157, line 33, strike "\$7,722,644,551" and insert "\$7,720,144,551".

Page 159, line 10, strike "\$2,500,000" and insert "\$0".

Page 196, line 28, strike "\$2,500,000 the first year and".

Explanation:

(This amendment captures savings of \$2.5 million GF the first year for the School Meals Expansion program. Under a federal waiver for the 2020-21 school year, all divisions are providing school meals at no cost to all students. These funds are no longer needed to eliminate the cost of reduced price school meals.)

Item	1 4 5	411-
пеш	143	#45

Education	FY20-21	FY21-22
Direct Aid to Public Education	\$0	(\$2,784,143) GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,855,855,978".

Page 170, line 9, after "Eight", strike "and the counties of Accomack and Northampton".

Page 170, line 17, after "Eight", strike "and the counties of Accomack and Northampton".

Page 170, line 21, after "Eight", strike "and the counties of Accomack and Northampton".

Explanation:

(This amendment removes Accomack and Northampton counties from the full cost of competing adjustment as included in the introduced budget.)

Item 145 #5s

Education	FY20-21	FY21-22
Direct Aid to Public Education	\$0	\$582,000 GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,859,222,121".

Page 159, after line 23, insert:

Out of this appropriation, \$582,000 the second year from the general fund is provided as an incentive for the consolidation of the Alleghany County and Covington City school divisions. Such payments shall be provided for no more than five fiscal years, beginning in fiscal year 2022."

Explanation:

(This amendment provides \$582,000 GF the second year as an incentive for the Alleghany County and Covington City school divisions to consolidate. Such payments are intended to be set at this amount for five years, with no adjustments. The Board of Education approved the consolidation of these divisions to be effective July 1, 2022.)

		Item 145 #6s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$59,303,635	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,917,943,756".

Page 158, strike line 43 and insert: "Compensation Supplement \$0 \$139,160,538".

Page 175, strike lines 8 through 12, and insert:

"6) Out of the appropriation included in paragraph C.44. of this Item, \$455,787 the second year from the Lottery Proceeds Fund are provided for a compensation supplement payment equal to 3.0 percent of base pay on August 1, 2021, for Regional Alternative Education Program instructional and support positions, as referenced in paragraph C.44. of this Item."

Page 191, strike lines 26 through 30, and insert:

"j. Out of the appropriation included in paragraph C.44. of this Item, \$620,830 the second year from the general fund is provided in the Academic Year Governor's School funding allocation

[&]quot;Alleghany County - Covington City School Division Consolidation Incentive \$0 \$582,000" Page 198, after line 24, insert:

[&]quot;45. Alleghany County - Covington City School Division Consolidation Incentive

to increase the per pupil amount the second year as an add-on for a compensation supplement payment equal to 3.0 percent of base pay on August 1, 2021, for Academic Year Governor's School instructional and support positions."

Page 197, strike lines 33 through 52, and insert: "44. Compensation Supplement

"a. Out of this appropriation, \$139,781,368 the second year from the general fund and \$455,787 the second year from the Lottery Proceeds Fund is provided for the state share of a payment of the following salary increases for funded SOQ instructional and support positions. Funded SOQ instructional positions shall include the teacher, school counselor, librarian, instructional aide, principal, and assistant principal positions funded through the SOQ staffing standards for each school division in the biennium. This amount includes \$620,830 the second year from the general fund referenced in paragraph C. 27. j. for the Academic Year Governor's Schools for the state share of a payment of the following salary increases for instructional and support positions, and this amount includes \$455,787 the second year from the Lottery Proceeds Fund referenced in paragraph C. 9. f. 6) for Regional Alternative Education Programs for the state share of a payment equivalent to a 3.0 percent salary increase effective August 1, 2021, for SOQ instructional and support positions.

b. It is the intent that the instructional and support position salaries are increased in school divisions throughout the state by at least an average of 3.0 percent during the second year. Sufficient funds are appropriated in this act to finance, on a statewide basis, the state share of a 3.0 percent salary increase the second year for funded SOQ instructional and support positions, effective August 1, 2021, to school divisions that certify to the Department of Education that salary increases of a minimum average of 3.0 percent have been or will have been provided during the 2020-2022 biennium, either in the first year or in the second year or through a combination of the two years, to instructional and support personnel. The state funds for which the division is eligible to receive shall be matched by the local government, based on the composite index of local ability-to-pay, which shall be calculated using an effective date of August 1, 2021, as the basis for the local match requirement for both funded SOQ instructional and support positions.

c. This funding is not intended as a mandate to increase salaries."

Page 198, strike line 1.

Explanation:

(This amendment provides \$139.8 million the second year from the general fund and \$455,787 the second year from the Lottery Proceeds fund to provide a 3.0 percent salary increase for SOQ-recognized instructional and support positions, to become effective August 1, 2021. This represents an addition of \$59.3 million the second year to provide a 3.0 percent salary increase in lieu of the 2.0 percent bonus that was proposed in the budget as introduced.)

		Item 145 #7s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$29,997,718	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,888,637,839".

Page 158, line 47, strike "\$133,057,800" and insert "\$163,055,518".

Page 159, line 41, strike "\$97,331,981" and insert "\$67,334,263".

Page 160, after line 16, insert: "Learning Loss Instructional Supports \$0 \$29,997,717".

Page 198, after line 4, insert:

"45. Learning Loss Instructional Supports

An additional state payment estimated at \$29,997,717 the second year from the Lottery Proceeds Fund shall be disbursed to support the state share of \$117.18 per pupil the second year based on the estimated number of federal Free Lunch participants, in support of one-time programs and initiatives to address learning loss resulting from the COVID-19 pandemic. These funds shall cover expenses for: extended school year programs; summer learning programs; supplemental afterschool programs; remedial instructional supports, including activities to differentiate instruction to meet the needs of students; targeted supports and services for students with disabilities, students in grades pre-kindergarten through third, at-risk students, and English language learners; and modifications to facilities to assist with COVID-19 mitigation strategies for in-person learning. No local match is required to receive these state funds."

Explanation:

(This amendment provides \$30.0 million from the Lottery Proceeds Fund the second year to support one-time programs and initiatives to address learning loss experienced by students due to the COVID-19 pandemic.)

		Item 145 #8s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$49,230,162	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,907,870,283".

Page 161, line 7, after "pupils)", strike "school nurses,".

Page 161, strike lines 11 through 15.

Explanation:

(This amendment funds the state's share of three specialized student support positions per 1,000 students. Specialized student support positions, consistent with Senate Bill 1257, includes school social workers, school psychologists, school nurses, licensed behavior analysts, licensed assistant behavior analysts, and other licensed health and behavioral positions.)

		Item 145 #9s	
Education	FY20-21	FY21-22	
Direct Aid to Public Education	\$0	\$2,699,993	GF

Language:

Page 157, line 33, strike "\$7,858,640,121" and insert "\$7,861,340,114".

Page 158, line 47, strike "\$133,057,800" and insert "\$135,757,793".

Page 159, line 41, strike "\$97,331,981" and insert "\$94,631,998".

Page 160, after line 16, insert: "Supplemental Support Accomack & Northampton \$0 \$2,700,000"

Page 198, after line 4, insert:

"45. Supplemental Support Accomack & Northampton

An additional state payment of \$2,700,000 the second year from the Lottery Proceeds Fund shall be disbursed to provide one-time support to Accomack and Northampton school divisions for teacher recruitment and retention efforts."

Explanation:

(This amendment provides \$2.7 million from the Lottery Proceeds Fund the second year to provide one-time support to Accomack and Northampton school divisions for recruitment and retention efforts.)

Item 145 #10s

Education

Direct Aid to Public Education

Language

Language:

Page 188, line 6, strike "c." and insert "c.1)". Page 188, after line 20, insert:

"2) The Board of Education shall make recommendations for: (i) appropriate staffing and funding levels necessary for State Operated Programs (SOP) in regional and local detention centers to provide a quality education program; (ii) implementation of appropriate efficiencies in staffing practices in such programs; (iii) statutory and regulatory changes needed to

implement the Board's findings; and (iv) appropriate programs to redirect any potential savings realized from implementation of the Board's findings.

In developing such recommendations, the Board shall consider: (i) the dramatic decrease in the Average Daily Population in detention centers over the course of two decades without a comparable decrease in state funding; (ii) establishing a system-wide staffing ratio that is comparable to those provided in Regional Alternative Education Programs and aligned with the staffing requirements provided in the federal Prison Rape Elimination Act; (iii) implementing efficiencies, such as sharing SOP instructional staff with participating school divisions, hiring part-time teachers and dually-certified teachers and principals, and utilizing lead teachers in lieu of full-time principals in programs with a low average daily population; (iv) changes to SOP operating agreements to facilitate more efficient staffing practices and to clarify the role of the state and school divisions in hiring and supervising SOP instructional staff; (v) increasing the use of enhanced distance learning; and (vi) the draft recommendations deliberated by the Commission on Youth from the 2020 study.

The Board shall convene a workgroup to assist in the development of findings and recommendations and shall include staff members from the Senate Finance and Appropriations Committee, House Appropriations Committee, Department of Planning and Budget, the Virginia Department of Education, the Department of Juvenile Justice, President of the Virginia Juvenile Detention Association or his/her designee, the chair of the Virginia Commission on Youth or his/her designee, and anyone else the Board deems as appropriate to serve on the workgroup. Findings and recommendations shall be reported to the House Appropriations Committee and the Senate Finance and Appropriations Committee by November 1, 2021."

Explanation:

(This amendment directs the Board of Education to develop recommendations for funding state operated programs in local and regional juvenile detention centers, which have experienced a significant decline in population in recent decades without an accompanying adjustment to state funding levels. This recommendation is from the Commission on Youth's 2020 study on "State Operated Programs Education in Juvenile Detention Centers".)

Item 145 #11s

Education

Direct Aid to Public Education

Language

Language:

Page 169, after line 27, insert:

"31. Each school division shall offer in-person instruction options to a student if requested by the parent or guardian in the 2021-2022 school year."

Explanation:

(This amendment requires all school divisions to offer in-person instruction options in the 2021-2022 school year to a student if the parent or guardian requests.)

		Item 150 #1s	
Education	FY20-21	FY21-22	
State Council of Higher Education for Virginia	\$150,000	\$300,000	GF

Language:

Page 200, line 25, strike "\$92,448,559" and insert "\$92,598,559".

Page 200, line 25, strike "\$105,568,934" and insert "\$105,868,934".

Page 202, line 26, strike the first instance of "\$1,980,000" and insert "\$2,130,000".

Page 202, line 26, strike the second instance of "\$1,980,000" and insert "\$2,280,000".

Explanation:

(This amendment provides additional funding based on estimated additional participation by qualifying veterans or their spouses or children for the Virginia Military Survivors and Dependent Education Program (VMSDEP), which provides assistance to Virginia military service members killed, missing in action, taken prisoner, or rated at least 90 percent permanently disabled as direct result of military service, by waiving tuition and required fees and providing a stipend to their dependents.)

		Item 152 #1s	
Education	FY20-21	FY21-22	
State Council of Higher Education for Virginia	\$0	\$250,000	GF

Language:

Page 205, line 6, strike "\$18,335,818" and insert "\$18,585,818".

Page 207, after line 55, insert:

"Q. Out of this appropriation, \$250,000 the second year from the general fund is designated for the Guidance to Postsecondary Success program. The program coordinates statewide efforts to increase college access and student success."

Explanation:

(This amendment restores \$250,000 GF the second year provided during the 2020 regular session for SCHEV to implement the Guidance to Postsecondary Success (GPS) program.

These funds were later unalloted. The purpose of this new initiative is to increase student transitions from high school to postsecondary education and also help students find the right fit for their future and in terms of costs. Each year, 25,000 high school graduates do not enroll in postsecondary education within 16 months, including significantly higher percentages for African-American, Hispanic and economically disadvantaged students. The pandemic and virtual learning environments in high schools are expected to exacerbate these inequitable outcomes in the coming year. Working closely with the Virginia Department of Education, SCHEV will seek to increase these enrollment rates through the development of materials and informational resources and the coordination of awareness events, professional development opportunities for secondary staff and partnerships with institutions, schools, school divisions and non-profit organizations. In addition, these funds will be used as a match for an upcoming SCHEV application for a federally-funded Gaining Early Awareness and Readiness for Undergraduate Programs (GEAR UP) grant that also will focus on school division-specific and statewide outreach programs, as well as, required scholarships for eligible students.)

		Item 152 #2s	
Education	FY20-21	FY21-22	
State Council of Higher Education for	\$0	\$300,000	GF
Virginia	0.00	1.00	FTE

Language:

Page 205, line 6, strike "\$18,335,818" and insert "\$18,635,818".

Page 208, after line 36, insert:

- "U. 1. Out of this appropriation, \$300,000 the second year from the general fund is designated to support related costs of undertaking a review of higher education costs, funding needs, appropriations and efficiencies.
- 2. The State Council of Higher Education, in consultation with representatives from House Appropriations Committee, Senate Finance and Appropriations Committee, Department of Planning and Budget, Secretary of Finance, and Secretary of Education, as well as representatives of public higher education institutions, shall review methodologies to determine higher education costs, funding needs, and appropriations in Virginia. The review shall identify and recommend: (1) methods to determine appropriate costs; (2) measures of efficiency and effectiveness; (3) provisions for any new reporting requirements; (4) strategies to allocate limited public resources based on outcomes that align with state needs related to affordability, access, completion, and workforce alignment, including with regard to nonresident pricing; (5) the impact of funding on underrepresented student populations; and (6) a timeline for implementation.

- 3. The review shall build on existing efforts including the assessment of base adequacy, recommendations provided through the Strategic Finance Plan, and peer institution comparisons to determine if existing funding models should be updated or replaced. It shall also build on promising practices and include input from Virginia's institutions, policy makers, and other education experts.
- 4. The Council shall present a proposed workplan to the Joint Subcommittee on the Future Competitiveness of Higher Education in Virginia by August 15, 2021. The Council shall submit a preliminary report and any related recommendations to the Governor and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by December 1, 2021 with a final report by July 1, 2022."

Explanation:

(This amendment restores funding for a review to benchmark costs for higher education using historic trends and comparisons to other state, institution, and national data and to identify methods to allocate additional funds to higher education when available.)

		Item 152 #3s	
Education	FY20-21	FY21-22	
State Council of Higher Education for Virginia	\$0	\$200,000	GF

Language:

Page 205, line 5, strike "\$18,335,818" and insert "\$18,535,818".

Explanation:

(This amendment provides \$200,000 GF for SCHEV to develop and maintain an application portal related to implementation of Senate Bill 1387, which provides that students who meet the criteria in the Code of Virginia that determine eligibility for in-state tuition regardless of their citizenship or immigration status shall be afforded the same educational benefits, including access to state financial assistance programs, as any other individual who is eligible for in-state tuition. The bill has a delayed effective date of August 1, 2022, but directs the State Council of Higher Education for Virginia, in coordination with institutions of higher education in the Commonwealth, to promulgate regulations to implement the provisions of the bill and develop the application portal in the interim. The portal will require additional privacy and security to collect financial aid data that SCHEV and institutions do not currently collect.)

Education

State Council of Higher Education for Virginia

Language

Language:

Page 207, after line 21, insert:

"In addition, the office of the workforce development advisor shall also have access to wage records collected by the Council."

Explanation:

(This amendment allows the office of the workforce development advisor access to wage records under SCHEV's existing authority, as a result of language adopted in 2019 specifying certain needed data exchange with the U.S. Census in order to get wage outcomes for graduates working outside the Commonwealth.)

		Item 156 #1s	
Education	FY20-21	FY21-22	
Christopher Newport University	\$0 \$0 0.00	\$1,019,000 \$651,500 19.00	GF NGF FTE

Language:

Page 209, line 30, strike "\$81,019,468" and insert "\$82,689,968".

Explanation:

(This amendment provides Christopher Newport University with on-going support for operations and maintenance (O&M) costs related to the opening of the new Fine Arts Center. This request is also a part of the university's six year plan. Construction will be completed in spring 2021, the building will open to the public in summer 2021 and classes will be held starting in the fall 2021 semester.)

		Item 164 #1s	
Education	FY20-21	FY21-22	
Richard Bland College	\$0	(\$299,000)	GF

Language:

Page 214, line 22, strike "\$15,180,047" and insert "\$14,881,047".

Explanation:

(This amendment removes proposed new general funds that had been requested to cover debt payments from a previous renovation of classroom space to dorms, in order that Richard Bland College could revert the space back to classroom use.)

		Item 168 #1s	
Education	FY20-21	FY21-22	
Virginia Institute of Marine Science	\$0	\$225,000	GF
	0.00	2.20	FTE

Language:

Page 216, line 38, strike "\$27,010,448" and insert "\$27,235,448".

Explanation:

(This amendment provides funding for science-based guidance on the management of existing and emerging disease threats to critical fishery and aquaculture resources in the Commonwealth and Chesapeake Bay region. This initiative was funded in the 2020 General Assembly Session but subsequently unallotted. Recent outbreaks of disease have damaged economically important and ecologically sensitive marine resources in the Commonwealth and nation. The pathogens responsible for these outbreaks are not well known, their risks to marine life and potential to spread remain poorly understood, and their ecological impacts have been difficult to assess with existing resources. To meet the challenge of diseases in marine systems, this initiative will provide science-based guidance on the management of existing and emerging disease threats to critical fishery and aquaculture resources in the Commonwealth and Chesapeake Bay region.)

		Item 168 #2s	
Education	FY20-21	FY21-22	
Virginia Institute of Marine Science	\$0 0.00	\$250,000 2.70	GF FTE

Language:

Page 216, line 38, strike "\$27,010,448" and insert "\$27,260,448".

Explanation:

(This amendment provides funding to provide scientific data for the management of saltwater fisheries in the Commonwealth. This initiative was funded in the 2020 General Assembly Session but subsequently unalloted. Long-standing VIMS fisheries surveys have provided scientific data for the management of saltwater fisheries in the Commonwealth of Virginia. Recent reductions in state and federal funds resulted in reductions in five of the most crucial surveys.)

		Item 183 #1s	
Education	FY20-21	FY21-22	
Norfolk State University	\$0	\$1,250,000	GF

Language:

Page 227, line 13, strike "\$100,293,110" and insert "\$101,543,110".

Page 228, after line 34, insert:

"H. Out of this appropriation, \$1,250,000 in the second year from the general fund is designated for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation."

Explanation:

(This amendment provides \$1.25 million GF beginning in the second year for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation.)

		Item 184 #1s	
Education	FY20-21	FY21-22	
Norfolk State University	(\$1,729,795)	\$0	GF

Language:

Page 228, line 37, strike "\$21,647,706" and insert "\$19,917,911".

Page 229, line 3, strike "\$3,459,590" and insert "\$1,729,795".

Page 229, line 23, after "program." strike the remainder of the line.

Page 229, strike lines 24 and 25.

Explanation:

(This amendment adjusts first year funding for the new VCAN affordability initiative approved during the 2020 Regular Session. The funding was unalloted at the Reconvened Session, then restored, for FY 2021, during the Special Session. The required detailed budget and implementation plan was submitted in December 2020 and to be impactful beginning spring semester 2021. Funding remains for the spring semester and for the 2021-22 academic year. The amendment provides for the release of first year funding by removing existing language that required certain approvals prior to release of the funding.)

Education	FY20-21	FY21-22	
Old Dominion University	\$0	\$1,250,000	GF

Language:

Page 230, line 6, strike "\$326,951,395" and insert "\$328,201,395".

Page 231, after line 47, insert:

"K. Out of this appropriation, \$1,250,000 in the second year from the general fund is designated for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation."

Explanation:

(This amendment provides \$1.25 million GF beginning in the second year from for the development and operation of a joint School of Public Health in partnership with Old Dominion University and Norfolk State University. This is to be a recurring appropriation.)

		Item 191 #1s	
Education	FY20-21	FY21-22	
Radford University	\$0	\$2,000,000	GF

Language:

Page 233, line 41, strike "\$135,081,721" and insert "\$137,081,721".

Explanation:

(This amendment provides \$2.0 million GF in additional funding for Radford University for operating support to further reduce tuition and fees for Virginians attending Radford University Carilion (RUC) in Roanoke to close the differential gap in tuition between the two campus sites, along with additional funding provided in FY 2021 and FY 2022 in another Item. The 2019 General Assembly authorized Radford University to establish RUC, a partnership with Carilion Clinic to strengthen the Roanoke region as a destination for health education and research, contributing to the overall growth and economic development of Southwest Virginia. Increased investment by the Commonwealth is intended to provide additional accessibility and affordability for Virginians enrolled in high-demand, high-paying health professions programs, and increase the number of health care professionals living and working in Southwest Virginia.)

		Item 196 #1s	
Education	FY20-21	FY21-22	
University of Mary Washington	\$0	\$568,000	GF

Language:

Page 236, line 10, strike "\$80,597,650" and insert "\$81,165,650".

Page 237, after line 15, insert:

"Out of this appropriation, \$568,000 the second year from the general fund is designated to support an educational partnership between regional K12; community college; University of Mary Washington and industry to develop a curriculum that accelerates time to degree; lowers cost; eliminates the skills gap and reduces reliance on student debt in the areas of Education, Healthcare, and Cybersecurity."

Explanation:

(This amendment provides \$568,000 GF for the University of Mary Washington's Workforce Development Initiative. This initiative partners three K-12 school districts, a community college, a public liberal arts university, and industry to develop a curriculum that accelerates time to degree, lowers cost of attendance, eliminates the skills gap that exists between educational experiences and works, reduces the reliance on student load debt to finance both degree competencies required to compete in the 21st century, and produces career-ready graduates whose experiences will empower them both socially and economically into their futures.)

		Item 205 #1s	
Education	FY20-21	FY21-22	
University of Virginia	\$0	(\$1,000,000)	GF

Language:

Page 241, line 48, strike "\$578,028,122" and insert "\$577,028,122". Page 242, line 14, strike "\$4,112,500" and insert "\$3,112,500".

Explanation:

(This amendment reallocates proposed additional funding for the Center for Focused Ultrasound to restorations of items from the 2020 Regular Session.)

		Item 214 #1s	
Education	FY20-21	FY21-22	
Virginia Commonwealth University	\$192,793	\$0	GF

Language:

Page 246, line 8, strike "\$661,732,918" and insert "\$661,925,711".

Page 248, after line 48, insert:

"Q. Out of this appropriation, \$192,753 the first year from the general fund shall be provided to

the L. Douglas Wilder School of Government and Public Affairs at Virginia Commonwealth University to support the Research Institute for Social Equity (RISE) addressing issues of racism and racial equity in public policy."

Explanation:

(This amendment provides \$192,753 GF the first year to the L. Douglas Wilder School of Government and Public Affairs at Virginia Commonwealth University to support the Research Institute for Social Equity (RISE) addressing issues of racism and racial equity in public policy. The L. Douglas Wilder School is a recognized national leader in advancing racial justice through multidisciplinary scholarship, research, training, and engagement. The creation of the Research Institute for Social Equity (RISE) within the Wilder School will serve as a vehicle to advance racial equity and social justice through multidisciplinary scholarship, research, training, and engagement. The goal of RISE is to be the national leader in advancing social justice to inform public policy, governance and practice to improve conditions for marginalized voices within our society, including communities of color, LGBTQIA individuals, people with disabilities, incarcerated populations, survivors of domestic violence, and women.)

		Item 214 #2s	
Education	FY20-21	FY21-22	
Virginia Commonwealth University	\$0	\$150,000	GF

Language:

Page 246, line 8, strike "\$661,732,918" and insert "\$661,882,918". Page 246, line 50, strike "\$386,685" and insert "\$536,685".

Page 247, after line 9, insert:

"All other funding support for the center shall be maintained by the university at least at the level provided in fiscal year 2019."

Explanation:

(This amendment restores funding in FY 2022 to the Virginia Center on Aging that was removed in the 2020 Special Session, and provides clarifying language.)

		Item 216 #1s	
Education	FY20-21	FY21-22	
Virginia Commonwealth University	(\$5,000,000)	(\$2,500,000)	GF

Language:

Page 249, line 26, strike "\$331,699,678" and insert "\$326,699,678".

Page 249, line 26, strike "\$331,699,678" and insert "\$329,199,678".

Page 249, line 39, strike "\$17,500,000" and insert "\$12,500,000."

Page 249, line 40, strike "\$17,500,000" and insert "\$15,000,000."

Page 249, line 41, after "support of cancer research" insert ", including pediatric cancer treatment".

Explanation:

(This amendment provides an additional \$2.5 million for Massey Cancer Center in FY 2022, consistent with the conference report during the 2020 Regular Session. The FY 2021 amount is level funded.)

		Item 220 #1s	
Education	FY20-21	FY21-22	
Virginia Community College System	\$0	\$385,177	GF

Language:

Page 251, line 14, strike "\$939,748,443" and insert "\$940,133,620".

Page 254, after line 1, insert:

"X. Out of this appropriation, \$385,177 the second year from the general fund is designated for costs of two associate degree programs in Physical Therapy Assistant and Surgical Technology that have transferred to Virginia Western Community College as a result of the merger of Radford University and the Jefferson College of Health Sciences authorized in Chapter 60 of the 2019 Acts of Assembly."

Explanation:

(This amendment provides funding for two programs that have transferred to Virginia Western Community College as a result of the merger of Radford University and the Jefferson College of Health Sciences authorized in Chapter 60 of the 2019 Acts of Assembly.)

		Item 220 #2s	
Education	FY20-21	FY21-22	
Virginia Community College System	\$0 0.00	\$5,000,000 60.00	GF FTE

Language:

Page 251, line 14, strike "\$939,748,443" and insert "\$944,748,443".

Page 254, line 9, strike "\$1,500,000" and insert "\$6,500,000".

Page 254, line 10, after "designated for" insert "advising,".

Explanation:

(This amendment provides \$5.0 million GF for Virginia community colleges to hire 60 additional advisors to assist students in identifying appropriate programs, resources and support, related to the G3 initiative.)

		Item 226 #1s	
Education	FY20-21	FY21-22	
Virginia Military Institute	\$0 0.00	\$103,048 1.00	GF FTE

Language:

Page 263, line 2, strike "\$44,354,698" and insert "\$44,457,746".

Explanation:

(This amendment restores funds for additional faculty positions to implement a redesigned required three-hour course in theory and practice of leadership, Leadership in Organizations. All cadets enroll in Leadership in Organizations, a required, 3-credit hour course in the theory and practice of leadership. This course will be redesigned to lessen the time commitment to leadership theory in order to enrich the application component of the course. In addition to curricular enhancements, this course will have a new course director who will oversee the development of: new training modules for current faculty and the training of Leadership Fellows who will teach the course.)

		Item 226 #2s	
Education	FY20-21	FY21-22	
Virginia Military Institute	\$0	\$126,000	GF

Language:

Page 263, line 2, strike "\$44,354,698" and insert "\$44,480,698".

Explanation:

(This amendment restores funds for the Math Education and Resource Center (MERC) and the Miller Academic Center (MAC) both of which were originally funded through private funds. The MERC was created to specifically address cadets' performance in mathematics with a particular emphasis on incoming freshmen and those in STEM majors. The MAC facilitates cadets' academic success and timely progress toward a degree by providing academic tutors and other academic support services.)

Item 236 #1s

Education	FY20-21	FY21-22	
Virginia Cooperative Extension and Agricultural Experiment Station	\$0	\$1,600,000	GF

Language:

Page 269, line 15, strike "\$93,864,832" and insert "\$95,464,832".

Explanation:

(This amendment supports the enhancement of the Virginia Cooperative Extension and Agricultural Experiment Station in order to support the resiliency and growth of the agricultural and forestry economy in the Commonwealth. Specific needs addressed in this multi-year request include improving internet connectivity, including one-time and on-going costs. Additional needs include modernizing research equipment, adding critical personnel, and supporting the market competitiveness of extension agent salaries.)

		Item 238 #1s	
Education	FY20-21	FY21-22	
Virginia State University	(\$1,886,745)	\$0	GF

Language:

Page 272, line 7, strike "\$19,677,956" and insert "\$17,791,211". Page 272, line 22, strike "\$3,773,490" and insert "\$1,886,745". Page 272, strike lines 42 and 43.

Explanation:

(This amendment adjusts first year funding for the new VCAN affordability initiative approved during the 2020 Regular Session. The funding was unalloted at the Reconvened Session, then restored, for FY 2021, during the Special Session. The required detailed budget and implementation plan included a timeline beginning with January 2021 for creating a pipeline for fall 2021. Funding remains for the spring semester and for the 2021-22 academic year. The amendment provides for the release of funding by removing existing language that required certain approvals prior to release of the funding.)

		Item 244 #1s	
Education	FY20-21	FY21-22	
Jamestown-Yorktown Foundation	\$0	\$412,484	GF

Language:

Page 275, line 8, strike "\$19,920,791" and insert "\$20,333,275".

Explanation:

(This amendment provides \$412,484 GF the second year to enhance digital marketing and social media capabilities to support key marketing strategies, and the development, promotion and maintenance of a new, re-focused website. Marketing funds support non-general fund revenue generation that has historically provided 50 percent of the agency's operating budget.)

		Item 244 #2s	
Education	FY20-21	FY21-22	
Jamestown-Yorktown Foundation	\$0	\$471,820	GF

Language:

Page 275, line 8, strike "\$19,920,791" and insert "\$20,392,611".

Explanation:

(This amendment provides \$471,820 GF the second year to support frontline personnel engaged in direct service delivery of museum and educational programming. These funds provide a competitive base salary and hourly rate for these employees relative to the local market and addresses retention challenges.)

		Item 247 #1s	
Education	FY20-21	FY21-22	
The Library Of Virginia	\$0	\$1,000,000	GF

Language:

Page 276, line 31, strike "\$17,233,584" and insert "\$18,233,584".

Explanation:

(This amendment restores \$1.0 million GF the second year to support additional statewide state aid to local public libraries that was approved at the 2020 Regular Session and then unallotted (Chapter 1289). State aid to local public libraries is currently funded in the second year at only 62 percent of the amount required. This amendment represents an additional step towards fully funding the state library aid formula over time.)

Education	FY20-21	FY21-22	
Virginia Museum of Fine Arts	\$0	(\$9,750,000)	GF

Language:

Page 279, line 23, strike "\$54,497,207" and insert "\$44,747,207".

Page 280, after line 4, insert:

"F. Out of this appropriation, \$1,000,000 in the second year from the general fund is provided to support the development of a plan for transforming Monument Avenue. The museum shall work with community stakeholders to develop the plan and utilize the recommendations from the *Report of the Monuments Work Group (2016)* on the best practices to foster constructive dialogues. The plan shall be reported to the Governor, Secretary of Education, and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2022."

Explanation:

(This amendment provides \$1.0 million GF the second year to support the museum's development of a plan for transforming Monument Avenue.)

		Item 256 #1s	
Education	FY20-21	FY21-22	
New College Institute	\$0	\$95,000	GF

Language:

Page 281, line 20, strike "\$4,318,700" and insert "\$4,413,700".

Explanation:

(This amendment restores funding in the second year for staffing at the New College Institute.)

		Item 257 #1s	
Education	FY20-21	FY21-22	
Institute for Advanced Learning and Research	\$0	\$95,000	GF

Language:

Page 282, line 14, strike "\$6,415,193" and insert "\$6,510,193".

Explanation:

(This amendment restores funding in the second year for staffing at the Institute for Advanced

Learning and Research.)

		Item 258 #1s	
Education	FY20-21	FY21-22	
Roanoke Higher Education Authority	\$0	\$98,817	GF

Language:

Page 282, line 43, strike "\$1,478,720" and insert "\$1,577,537".

Explanation:

(This amendment restores funding in the second year for the Roanoke Higher Education Center for one-time funding of \$50,873 for equipment and installation of blue light telephones, and a security camera system. Additionally, this budget amendment provides on-going funding of \$47,944 for additional hours of coverage from security officers.)

		Item 258 #2s	
Education	FY20-21	FY21-22	
Roanoke Higher Education Authority	\$0	\$213,254	GF

Language:

Page 282, line 43, strike "\$1,478,720" and insert "\$1,691,974".

Explanation:

(This amendment restores funding in the second year for the Roanoke Higher Education Center for one-time funding of \$66,898 and on-going funding of \$146,356 for the development and maintenance of a student success center.)

		Item 259 #1s	
Education	FY20-21	FY21-22	
Southern Virginia Higher Education Center	\$0	\$388,972	GF

Language:

Page 283, line 8, strike "\$7,949,697" and insert "\$8,338,669".

Explanation:

(This amendment restores funding in the second year for one-time funding of equipment in the

amount of \$293,972 and on-going staffing funding of \$95,000 for personnel and technical training equipment to support high-demand workforce training programs. Equipment includes Zoom classroom equipment and software to allow students access to quality distance and virtual training courses and upgrades to Amatrol training equipment, including new electrical wiring learning system.)

		Item 260 #1s	
Education	FY20-21	FY21-22	
Southwest Virginia Higher Education Center	\$0	\$95,000	GF

Language:

Page 284, line 16, strike "\$3,386,650" and insert "\$3,481,650".

Explanation:

(This amendment restores funding for staffing at the Southwest Virginia Higher Education Center.)

		Item 260 #2s	
Education	FY20-21	FY21-22	
Southwest Virginia Higher Education Center	\$0	\$1,000,000	GF

Language:

Page 284, line 16, strike "\$3,386,650" and insert "\$4,386,650".

Page 284, after line 32, insert:

"B. Out of the appropriation for this item, \$1,000,000 the second year from the general fund shall be deposited to the Virginia Rural Information Technology Apprenticeship Grant Fund, as established in § 23.1-3129.1 Code of Virginia, for the purpose of awarding grants on a competitive basis from the Fund to small, rural information technology businesses in qualifying localities to establish apprenticeship programs."

Explanation:

(This amendment restores funding for the Southwest Virginia Higher Education Center to develop and implement the Rural Information Technology Grant Apprenticeship Program. The Center will need to develop guidelines, criteria, an application process, and accountability reporting.)

Item 262.80 #1s

Education	FY20-21	FY21-22
Maintain Affordable Access	\$0	\$73,500,000 GF

Language:

Page 287, line 16, strike "\$0" and insert "\$73,500,000". Page 287, strike lines 16 through 43 and insert:

"Item 262.80

Educational and General Programs (10000)	\$60,000,000	\$73,500,000
Higher Education Instruction (10001)	\$60,000,000	\$73,500,000

Fund Sources: General

Authority: Discretionary Inclusion

A. Out of this appropriation, \$60,000,000 the first year and \$73,500,000 the second year from the general fund is designated to maintain affordable access to public colleges and universities. Institutions may use these funds for operational support, to enhance financial aid, or to address the impacts of the COVID-19 pandemic. For purposes of base budget development, these appropriated funds shall be transferred to the individual institution. Allocations from this item are as follows:

Institution	FY 2021 Allocation	FY 2022 Allocation
Christopher Newport University	\$2,400,000	\$2,400,000
College of William and Mary	3,500,000	3,500,000
George Mason University	0	9,000,000
James Madison University	5,700,000	5,700,000
Longwood University	1,500,000	1,500,000
University of Mary Washington	3,300,000	3,300,000
Norfolk State University	2,000,000	2,000,000
Old Dominion University	0	4,500,000
Radford University	4,900,000	4,900,000
University of Virginia	3,000,000	3,000,000
University of Virginia's College at Wise	1,000,000	1,000,000

Virginia Commonwealth University	10,000,000	10,000,000
Virginia Military Institute	1,000,000	1,000,000
Virginia Polytechnic Institute & State University	4,000,000	4,000,000
Virginia State University	1,700,000	1,700,000
Richard Bland College	1,000,000	1,000,000
Virginia Community College System	15,000,000	15,000,000
Total	\$60,000,000	\$73,500,000"

- B. Pursuant to Item 479.10 of this Act, \$17,250,000 from the Coronavirus Relief Fund is provided for the costs of conducting COVID-19 tests at Virginia's public colleges and universities.
- C. To address student affordability, \$22,000,000 in Governor's Education Emergency Relief funds from the Coronavirus Response and Relief Supplemental Appropriations Act, (P.L. 116-260) shall be allocated to public institutions of higher education for need-based undergraduate financial aid in the second year.
- D. To provide additional operational relief to institutions of higher education, the following reporting and procurement policies shall be modified accordingly:
- 1. Pursuant to § 4-2.01.b.11 of this act, for future reporting on fiscal year 2023 and beyond required reporting on intercollegiate athletic revenues and expenses, specifically related to the share of athletic revenues from school funds and student fees, as set out in § 23.1-1309, Code of Virginia, fiscal years 2020, 2021, and 2022 shall be excluded from the calculated five-year rolling average of the change in generated revenue and student fees also outlined in § 23.1-1309, Code of Virginia.
- 2. Consistent with the 2019 updates to the Virginia Public Procurement Act, institutions of higher education that have entered into memoranda of understanding or management agreements with the state are permitted to conform their Request for Proposal advertising rules to that of § 2.2-4302.2.A.2."

Explanation:

(This amendment provides funding to Virginia's public colleges and universities to meet the challenges of the COVID-19 pandemic, support operations, and address access and affordability. Specifically, the amendment provides the following: 1) Continues first-year one-time support of \$60.0 million GF provided in Chapter 56, 2020 Special Session I, into the second year of the biennium, and includes funding for George Mason University and Old Dominion University, for operational support, to enhance financial aid, and to address the impacts of the COVID-19 pandemic; 2) Allocates an additional \$17.3 million from the Coronavirus Relief Fund to address the costs of testing for COVID-19; 3) Allocates \$22.0 million from "GEER2" for additional financial aid, and 4) Provides relief from certain

procurement and other requirements to provide institutions with additional flexibility and to generate cost savings.)

Item 266 #1s

Finance

Department of Accounts

Language

Language:

Page 292, after line 39, insert:

"E.1. There is hereby created in the state treasury a special, nonreverting fund to be known as the Opioid Abatement Fund. All funds appropriated to the Fund, all funds designated by the Attorney General under § 2.2-507.3 from settlements, judgments, verdicts, and other court orders relating to claims regarding the manufacturing, marketing, distribution, or sale of opioids, and any gifts, donations, grants, bequests, and other funds received on the Fund's behalf shall be paid into the state treasury and credited to the Fund. Interest earned on moneys in the Fund shall remain in the Fund and be credited to it. Any moneys remaining in the Fund at the end of each fiscal year, including interest thereon, shall not revert to the general fund but shall remain in the Fund.

2. The provisions contained in this paragraph shall be in effect until July 1, 2021, at which time any balances remaining in this Fund shall transfer to the Opioid Abatement Fund created pursuant to SB 1469 of the 2021 General Assembly, and subject to the provisions thereof."

Explanation:

(This amendment establishes the Opioid Abatement Fund, consistent with SB 1469 of the 2021 General Assembly, which shall become effective July 1, 2021. Establishment of this Fund will allow for the deposit of any opioid related litigation settlement moneys received by the Commonwealth until such time as the Fund and associated Authority established under SB 1469 are effectuated.)

		Item 282 #1s	
Finance	FY20-21	FY21-22	
Department of Taxation	\$0	(\$69,816)	GF

Language:

Page 301, line 17, strike "\$61,659,588" and insert "\$61,589,772".

Page 301, line 16, unstrike the stricken language.

Page 301, line 17, strike "\$61,659,588".

Page 301, line 18, unstrike the stricken language.

Page 301, line 21, strike "\$12,371,531".

Page 301, line 25, unstrike the striken language.

Page 301, line 26, strike "\$50,819,573".

Explanation:

(This amendment removes funding included in the Introduced Budget for the Department of Taxation to make systems changes related to the Virginia Facilitated Enrollment Program.)

		Item 284 #1s	
Finance	FY20-21	FY21-22	
Department of Taxation	\$0	(\$164,819)	GF
Language:			
Page 306, line 51, strike "\$52,406,035" and insert Page 306, line 50, unstrike the striken language. Page 306, line 51, strike "\$52,406,035". Page 306, line 52, unstrike the striken language. Page 306, line 53, strike "\$31,261,776". Page 307, line 1, unstrike the stricken language. Page 307, line 2, strike "\$21,144,259". Page 307, line 3, unstrike the stricken language.	'\$52,241,216".		

Page 307, line 36, unstrike the stricken language.

Page 307, line 31, unstrike the stricken language. Page 307, line 32, strike "\$120,185,117".

Page 307, line 37, strike "\$107,702,426".

Page 307, line 4, strike "\$52,252,581".

Explanation:

(This amendment removes funding included in the Introduced Budget for the Department of Taxation to make systems changes related to the Virginia Facilitated Enrollment Program.)

		Item 288 #1s	
Finance	FY20-21	FY21-22	
Treasury Board	\$4,000,000	\$0	GF

Language:

Page 310, line 34, strike "\$850,158,182" and insert "\$854,158,182".

Page 315, after line 41, insert:

"I. Out of this appropriation, \$4,000,000 the first year from the general fund is provided for the defeasance of the outstanding bonds on the Central Virginia Training Center."

Explanation:

(This amendment provides \$4 million GF in FY 2021 for the defeasance of the bonds of outstanding bonds on the Central Virginia Training Center.)

Item 291 #1s

Health and Human Resources

Secretary of Health and Human Resources

Language

Language:

Page 318, after line 53, insert:

"F. The Secretary of Health and Human Resources, or his designee, shall convene a workgroup of appropriate agencies within the secretariat and other stakeholders, as necessary, to research and recommend strategies for the financing of health care services for undocumented immigrant children. The workgroup shall: (i) identify the number of children who would qualify and their geographic location; (ii) demonstrate the impact a lack of health care coverage has on these children; (iii) determine the financial burden carried by hospital systems and other healthcare facilities that currently provide care for these children; (iv) identify the existing barriers these children face when trying to access essential medical services in a timely manner; (v) identify the long-term health impacts to children who do not have health care coverage and the future cost the Commonwealth will incur as a result; and (vi) recommend options for providing health care coverage to these children and the approximate cost to the Commonwealth."

Explanation:

(This amendment directs the Secretary of Health and Human Resources to convene a workgroup to research and recommend strategies for the financing of health care services for undocumented immigrant children. Approximately 13,000 immigrant children in Virginia lack health insurance and approximately 9,000 of those children live under 200 percent of the federal poverty level. They do not qualify for CHIP-funded (Children's Health Insurance Program) health insurance due to their immigration status.)

Item 291 #2s

Health and Human Resources

Secretary of Health and Human Resources

Language

Language:

Page 318, after line 53, insert:

- "F.1. It is the intent of the General Assembly that the Virginia Department for the Aging (VDA) be reestablished as an agency of the Commonwealth under the Office of the Secretary of Health and Human Resources beginning July 1, 2022. This agency shall oversee policies and programs impacting older Virginians and provide a leadership role across state government in evaluating the impact the aging population has on state services.
- 2. The Secretary of Health and Human Resources, or his designee, shall convene a workgroup that includes representatives from the Department for Aging and Rehabilitative Services, Area Agencies on Aging, the Department of Planning and Budget, the Division of Legislative Services, appropriate staff from the House Appropriations and Senate Finance and Appropriations Committees, and other appropriate stakeholders. The workgroup shall: (i) review other state aging departments and best practices for establishing an agency fully capable of leading across state government with regard to impacts from an aging population; (ii) review and develop an optimal organizational structure for the new agency; (iii) develop a transition plan for transferring staff, funding and making other operational changes to establish the new agency; (iv) draft legislation for consideration by the 2022 General Assembly; (v) determine potential costs to create the new agency; and (vi) develop draft changes to the Appropriation Act. The workgroup shall at a minimum evaluate transitioning aging services, adult services, adult protective services and auxiliary grant programs to the new agency. In addition, the workgroup shall examine any other aging-related programs in the Health and Human Resources Secretariat and make recommendations for inclusion in the new agency.
- 3. The workgroup shall provide all deliverables and report on its findings by October 1, 2021, to the Governor, the Department of Planning and Budget, and the Chairs of House Appropriations and Senate Finance and Appropriations Committees. The Department for Aging and Rehabilitative Services shall provide this information to the Governor to be considered for inclusion in the 2022-2024 introduced budget."

Explanation:

(This amendment provides that it is the intent of the General Assembly that a new agency for aging services be established as an agency of the Commonwealth under the Office of the Secretary of Health and Human Resources, beginning July 1, 2022. The language creates a workgroup to consider the details of such a transition and to make the appropriate recommendations to the General Assembly.)

Item 291 #3s

Health and Human Resources

Secretary of Health and Human Resources

Language

Language:

Page 324, after line 6, insert:

"D. The Office of Children's Services shall develop a plan to modify its staffing and operations to ensure effective local implementation of the Children's Services Act. The plan shall include

any new or different staff positions required, how those positions will be used to monitor and improve effectiveness, and the estimated cost of implementing these changes. The plan shall be submitted to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021.

E. The Office of Children's Services shall collect annually from each local Children's Services Act program the number of program staff by full- and part-time status and the administrative budget broken out by state and local funding to understand local program resources and target technical assistance to the most under-sourced local programs."

Explanation:

(This amendment requires the Office of Children's Services (OCS) to develop a plan to modify its staffing and operations to ensure effective local implementation of the Children's Services Act (CSA). It also directs OCS to collect annually from each local CSA program the number of program staff by full- and part-time status and information on local administrative budgets to better understand local program sources. These actions are based on recommendations from the Joint Legislative Audit and Review Commission's November 2020 study on CSA.)

Item 292 #1s

Health and Human Resources

Children's Services Act

Language

Language:

Page 323, strike lines 1 through 5.

Explanation:

(This amendment eliminates the annual two percent rate cap on increases that localities may pay for private day special services under the Children's Services Act. This rate cap was temporary until the completion of the rate setting study, which will be complete in fiscal year 2021.)

Item 292 #2s

Health and Human Resources

Children's Services Act

Language

Language:

Page 323, strike lines 6 through 14.

Explanation:

(This amendment removes language that allows localities to adjust daily or monthly rates for the 2020-2021 school year for virtual or distance learning provided by a private school serving

students with disabilities under the Children's Services Act.)

		Item 292 #3s	
Health and Human Resources	FY20-21	FY21-22	
Children's Services Act	\$0	\$305,357	GF

Language:

Page 319, line 4, strike "\$384,786,416" and insert "\$385,091,773".

Explanation:

(This amendment provides \$305,357 the second year from the general fund for the fiscal impact of Senate Bill 1338, which establishes the State Kinship Guardianship Assistance program, which allows payments to be made to relatives, including fictive kin, who receive custody of a child. A corresponding amendment in the Department of Social Services reduces funding through the Title IV-E program as some children will transition to the State-Funded Kinship Guardship Assistance Program.)

		Item 293 #1s	
Health and Human Resources	FY20-21	FY21-22	
Children's Services Act	\$0	(\$50,000)	GF

Language:

Page 323, line 16, strike "\$2,059,796" and insert "\$2,009,796".

Page 324, unstrike lines 1 through 2.

Page 324, line 2, after "effective" strike "on July 1, 2021" and insert "upon a date approved by the General Assembly in the 2022 Session".

Page 324, strike lines 3 through 6, and insert:

"C. Out of this appropriation, \$50,000 the second year is provided to the Office of Children's Services for training and technical assistance activities in overseeing the Children's Services Act. OCS is authorized to use a portion of these funds to hire a facilitator to assist with the workgroup created pursuant to Senate Bill 1313."

Explanation:

(This amendment reduces \$50,000 the second year from the general fund that was provided in the introduced budget for administration of a rate setting process for special education private day programs. Language modifies the effective date for rate setting to a future date to be determined by the 2022 General Assembly pending the results of the final rate setting study. Out of the \$100,000 general fund provided in the introduced budget, \$50,000 remains in the Office of Children's Services to be used for training and technical assistance related to the

Children's Services Act (CSA) program. The agency is allowed to use a portion of this funding for a facilitator for the stakeholder workgroup pursuant to Senate Bill 1313, that would determine how to move special education private day funding from CSA to the Department of Education.)

		Item 295 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$1,600,000	GF

Language:

Page 325, line 25, strike "\$985,000" and insert "\$2,585,000".

Page 325, strike lines 40 through 54.

Page 326, strike lines 1 through 7.

Page 326, unstrike lines 8 through 36.

Page 326, line 11, after "adolescent psychiatrists;" insert:

"psychiatric physician assistants; psychiatric pharmacists;".

Explanation:

(This amendment restores funding and language that was provided in Chapter 1289, 2020 Acts of Assembly, to establish the Behavioral Health Loan Repayment Program. Funding for this item was unallotted in April, 2020 and eliminated in Chapter 56, 2020 Special Session I Acts of Assembly. The program would increase the number of Virginia behavioral health practitioners through the establishment of an educational loan repayment incentive that complements and coordinates with existing efforts to recruit and retain Virginia behavioral health practitioners. The program would allow for a variety of behavioral health practitioners to receive a student loan repayment award from the Commonwealth in exchange for providing service to Virginia communities that are otherwise underserved. Practitioners would receive loan repayment for up to 25 percent of student loan debt for each year of health care service provided to the Commonwealth. Maximum loan repayment amounts per year are dependent upon the type of behavioral health professional applying and shall not exceed the total student loan debt. Participating practitioners will have an initial two-year minimum participation obligation and may renew for a third and fourth year. This provides the practitioner with the opportunity to fully pay off their student loan debt while providing four years of service to the Commonwealth. In addition, this amendment also strikes outdated language related to developing a plan for increasing the number of behavioral health practitioners.)

		Item 295 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$35,000	GF

Language:

Page 325, line 24, strike "\$985,000" and insert "\$1,020,000".

Page 326, after line 52, insert:

"E. Out of this appropriation, \$35,000 the second year from the general fund is provided for the Nurse Loan Repayment Program to provide loan repayments for certified nurse aides. The total loan repayment allowed per certified nurse aide is limited to no more than \$1,000."

Explanation:

(This amendment provides \$35,000 from the general fund the second year to assist with funding for the expansion of the nurse loan repayment program pursuant to Senate Bill 1147 that adds certified nurse aides as eligible for the program. The additional funding is dedicated to supporting certified nurse aides and is capped at \$1,000.)

		Item 295 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$500,000	GF

Language:

Page 325, line 25, strike "\$985,000" and insert "\$1,485,000".

Page 326, unstrike lines 37 through 52.

Page 326, line 37, after "D.", insert "1.".

Page 326, line 37, strike "\$500,000 the first year and".

Page 326, line 51, strike "2020" and insert "2021".

Page 326, after line 52, insert:

Explanation:

(This amendment provides \$500,000 the second year from the general fund to the Virginia Department of Health to establish a Nursing Preceptor Incentive Program. The department would report to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by November 1, 2021 on the progress of establishing the Nursing Preceptor Incentive Program. The Virginia Health Workforce Development Authority is directed to develop a process for the consideration of requests for funding from the Nursing Preceptor Incentive Program.)

Health and Human Resources FY20-21 FY21-22

[&]quot;2. The Virginia Health Workforce Development Authority shall develop the process for the consideration of requests for funding from the Nursing Preceptor Incentive Program."

Department of Health 0.00 30.00 FTE

Language:

Explanation:

(This amendment adds 30 full-time positions funded through nongeneral funds in the Virginia Department of Health's Office of Emergency Medical Services (OEMS). These positions would backfill the loss of positions based on implementation of the agency's shared business services. However, the positions are needed to enable OEMS to work directly with the Regional Emergency Medical Services (EMS) Councils. Each council maintains a Board of Directors but they are staffed by OEMS. In addition, the OEMS will have responsibility for directly managing two regional EMS councils (Shenandoah and Rappahannock). Funding will be shifted within OEMS to cover the cost of the added positions.)

		Item 297 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$51,146	GF

Language:

Page 328, line 2, strike "\$16,001,106" and insert "\$16,052,252".

Explanation:

(This amendment provides \$51,146 from the general fund the second year to support a work group to develop a plan for establishing a Fetal and Infant Mortality Review at the Virginia Department of Health. Such a team would function in a similar manner to the child fatality review team but focus on tracking and investigating selected fetal and infant deaths in the Commonwealth and make recommendations to reduce preventable deaths.)

		Item 299 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 0.00	\$395,818 3.00	GF FTE

Language:

Page 328, line 34, strike "\$213,178,894" and insert "\$213,574,712".

Explanation:

(This amendment provides \$395,818 the second year from the general fund in the Office of Epidemiology to support three additional positions to the \$3.1 million included in the

introduced budget provided to sustain and expand the Virginia Department of Health's communicable disease surveillance and investigation efforts across the Office of Epidemiology and the 35 local health districts in Virginia. The introduced budget funded 12 epidemiologists and 12 communicable disease nurses across the 35 health districts and funded two epidemiology program managers in the Office of Epidemiology in the Central Office. This amendment funds three additional epidemiology program managers in the Office of Epidemiology. A separate amendment funds a total of 35 epidemiologists and 35 communicable disease nurses, one of each position, in every local health district, to expand the Commonwealth's communicable disease surveillance and investigation capabilities.)

		Item 299 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	(\$30,184,899) \$18,002,665	(\$59,123,029) \$59,123,029	GF NGF

Language:

Page 328, line 35, strike "\$140,808,393" and insert "\$128,626,159". Page 330, line 24, strike "\$30,184,899", and insert "\$18,002,665". Page 330, line 25, strike "the general fund", and insert "federal funds".

Explanation:

(This amendment supplants \$30.2 million the first year and \$59.1 million the second year to support the Commonwealth's mass vaccination efforts in response to the COVID-19 pandemic. On December 27, 2020, the federal Consolidated Appropriations Act, 2021 was signed into law which provides substantial federal assistance to support states in vaccine administration efforts. Virginia's share of this funding is \$77.1 million. With this additional federal support, the general fund added in the introduced budget can be supplanted with federal funds. The new federal funding is not sufficient to fully supplant all \$89.3 million of general fund in the biennium, however, Coronavirus Relief Funds (CRF) allocated to the Department of Medical Assistance Services have not been fully utilized, so this amendment includes \$12.2 million in the first year in CRF funds. A separate amendment in Central Appropriations reflects the change in the allocation of CRF funds.)

		Item 299 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	(\$722,472)	GF

Language:

Page 328, line 35, strike "\$213,178,894" and insert "\$212,456,422".

Page 330, line 16, strike "\$1,444,944", and insert "\$722,472". Page 330, line 18, strike "\$1,008,000", and inset "\$504,000". Page 330, line 21, strike "\$436,944", and insert "\$218,472".

Explanation:

(This amendment reduces \$722,472 the second year from the general fund for COVID-19 data modeling, being provided by the University of Virginia's Biocomplexity Institute and the RAND corporation to provide epidemiologic analysis for the state's response to COVID-19. This amendment removes half the second year funding assuming that after calendar year 2021, the need for such modeling would likely be mitigated as the COVID-19 pandemic subsides as a result of widespread use of vaccines. In the case that COVID-19 continues as a public health emergency for longer and the need for such modeling continues, additional funding can be provided for this purpose in the 2022 General Assembly Session.)

Item 300 #1s

Health and Human Resources

Department of Health

Language

Language:

Page 332, after line 11, insert:

"H. The provisions of § 32.1-102.4 (B), Code of Virginia, shall not apply to nursing homes."

Explanation:

(This amendment clarifies that the charity care provisions of § 32.1-102.4 (B), Code of Virginia, do not apply to nursing facilities. This is a technical amendment to ensure the appropriate interpretation of the charity care requirements, based on changes made in the 2020 General Assembly Session, are applied consistent with the intent of that legislation.)

		Item 300 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$88,914	GF

Language:

Page 330, line 45, strike "\$21,336,679" and insert "\$21,425,593".

Explanation:

(This amendment provides \$88,914 the second year from the general fund to fund one position for administering the Behavioral Health Loan Repayment program and the Nursing Preceptor Incentive program funded in companion amendments in Item 295.)

Item 300 #3s

Health and Human Resources

Department of Health

Language

Language:

Page 332, strike lines 8 through 11.

Explanation:

(This amendment eliminates language included in the introduced budget that directs the Virginia Department of Health to provide administrative and technical support to the Virginia Partners in Prayer Program at a cost of up to \$20,000 in funding from the agency's existing budget. The department already provides support to this program and therefore this language is unnecessary.)

		Item 301 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$250,000	GF

Language:

Page 332, line 13, strike "\$163,353,397" and insert "\$163,603,397".

Page 332, line 40, after "C." insert "1."

Page 332, after line 44, insert:

Explanation:

(This amendment adds \$250,000 the second year from the general fund for a comprehensive adult sickle cell disease (SCD) program. Historically, Virginia has funded SCD care for children but not for adults. This amendment would establish a comprehensive adult program for sickle cell disease within the Office of Family Health Services and would create a model of care based on the structure of existing pediatric care models.)

		Item 302 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	(\$7,364,304)	GF

Language:

[&]quot;2. Out of this appropriation, \$250,000 the second year from the general fund is provided to establish a comprehensive adult program for sickle cell disease."

Page 333, line 20, strike "\$292,637,694" and insert "\$285,273,390".

Page 336, strike lines 9 through 16, and insert:

"H. Out of this appropriation, \$2,835,696 the second year from the general fund shall be provided to address revisions to the JLARC rate formula for the Cooperative Health Budget. These revisions and the changes in the local match rates shall be phased in over a three-year period beginning in the second year and shall be fully phased in by fiscal year 2024."

Explanation:

(This amendment reduces \$7.4 million the second year from the general fund by phasing in over three years the increase in local matching funds and the increase in state support for the updates to the local health department cooperative funding formula. The introduced budget provided \$10.2 million, which included funding for a hold harmless in fiscal year 2022, to implement the funding formula changes. This amendment reflects one-third of the state impact in the second year as the changes are phased in over three years and, as such, a hold harmless is not necessary.)

		Item 302 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 0.00	\$5,470,677 46.00	GF FTE

Language:

Page 333, line 20, strike "\$292,637,694" and insert "\$298,108,371".

Explanation:

(This amendment provides \$5.5 million the second year from the general fund in the Office of Epidemiology to support 46 additional positions in addition to the \$3.1 million included in the introduced budget provided to sustain and expand the Virginia Department of Health's communicable disease surveillance and investigation efforts across the Office of Epidemiology and the 35 local health districts in Virginia. The introduced budget funded 12 epidemiologists and 12 communicable disease nurses across the 35 health districts and funded two epidemiology program managers in the Office of Epidemiology in the Central Office. This amendment funds a total of 35 epidemiologists and 35 communicable disease nurses, one of each position, in every local health district, to expand the Commonwealth's communicable disease surveillance and investigation capabilities. A separate amendment funds three additional epidemiology program managers in the Office of Epidemiology)

Item 302 #3s

Health and Human Resources FY20-21 FY21-22

Department of Health \$0 \$75,889 GF

Language:

Page 333, line 20, strike "\$292,637,694" and insert "\$292,713,583".

Explanation:

(This amendment restores \$75,889 the second year from the general fund that was originally appropriated in Chapter 1289 during the 2020 Regular Session, but then eliminated in Chapter 56 in the 2020 Special Session I. This funding supports local health districts that may experience cost increases due to moving to new facilities or rent increases in existing facilities. Health districts do not fully control the process for determining when and where their facilities will be located.)

		Item 302 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	0.00	-44.00	FTE

Language:

Explanation:

(This amendment eliminates 44 positions for the Loudoun County Health District, which pursuant to Senate Bill 1221 converts the health district from a state-administered health district to a locally-administered one. Therefore, existing employees would transition to become local employees.)

		Item 304 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 0.00	\$250,000 3.00	GF FTE

Language:

Page 340, line 34, strike "\$33,588,623" and insert "\$33,838,623".

Explanation:

(This amendment provides \$250,000 the second year from the general fund and three positions for the Virginia Department of Health to handle testing plans and test results of lead water testing by local schools and provides funding for the department to handle the testing plans and test results of lead water testing submitted to the agency from child care facilities.)

Item 305 #1s

Health and Human Resources	FY20-21	FY21-22	
Department of Health	0.00	-1.00	FTE

Language:

Explanation:

(This amendment eliminates one general fund position included in the introduced budget that was added with the funding for a wastewater infrastructure manager. This funding is a restoration of funding previously approved in Chapter 1289 from the 2020 Session. However, the funding for this position was eliminated in Chapter 56 in the Special Session I, but the position was not removed. Therefore, this amendment corrects the position level for the agency.)

		Item 307 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0	\$60,000	GF

Language:

Page 341, line 27, strike "\$38,204,545" and insert "\$38,264,545".

Explanation:

(This amendment provides \$60,000 the second year from the general fund for the continued work of the Virginia Department of Health's Office of Drinking Water to continue its study of the occurrence of perfluorooctanoic acid (PFOA), perfluorooctane sulfonate (PFOS), and other perfluoroalkyl and polyfluoroalkyl substances (PFAS) in the Commonwealth's public drinking water and to develop recommendations for specific maximum contaminant levels for PFOA, PFOS, and other PFAS for inclusion in regulations of the Board of Health applicable to waterworks.)

		Item 307 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	(\$7,100,000)	(\$6,250,000)	GF

Language:

Page 341, line 27, strike "\$34,810,621" and insert "\$27,710,621".

Page 341, line 27, strike "\$38,204,545" and insert "\$31,954,545".

Page 343, line 1, after "appropriation,", strike "\$6,500,000 the first year and \$12,500,000", and

insert "\$6,250,000".

Page 343, line 4, after "pandemic.", insert:

"The Virginia Department of Health shall allocate no less than 20 percent of funding from state or federal sources dedicated for COVID-19 communications to use for outreach and communications to high-risk populations that have been adversely impacted by the COVID-19 pandemic more so than the general population and for which traditional communication mediums are not as effective. The department shall use such funding for alternative methods of communication, such as outreach coordinators going into communities, providing information pamphlets as part of meal pick-ups at schools, grants to community organizations, and other more effective ways at reaching high-risk populations. This funding shall also be used to translate communication materials into other languages; however the department shall not use Google Translate in any communications to non-English speakers."

Page 343, line 5, strike "Out of this appropriation,", and insert "No less than".

Page 343, line 5, strike "from the general fund" and insert "Coronavirus Relief Funds".

Explanation:

(This amendment supplants \$7.1 million the first year and \$6.3 million the second year from the general fund with available federal Coronavirus Relief Funds (CRF). The CRF funds allocated to the Department of Medical Assistance Services have not all been fully utilized, so this amendment supplants \$7.1 million in the first year and \$6.3 million the second year in CRF funds. A separate amendment in Central Appropriations reflects the change in the allocation of CRF funds. In addition, language directs a portion of the state and federal funding allocated for COVID-19 communications to be used for outreach and communications to high-risk populations that have been adversely impacted by the COVID-19 pandemic more so than the general population and for which traditional communication mediums are not as effective. It also directs the funding to be used for translation services.)

		Item 307 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Health	\$0 \$0	\$746,562 \$3,567,494	GF NGF

Language:

Page 341, line 27, strike "\$38,204,545" and insert "\$42,518,601".

Page 341, line 52, strike "\$26,736", insert "\$773,298".

Page 341, line 52, strike "\$240,625", insert "\$3,808,119".

Page 342, line 30, "after annually", insert "by September 1 of each year".

Page 342, line 33, after "plans", insert "and their patient records viewed".

Explanation:

(This amendment adds \$746,562 from the general fund and \$3.6 million from enhanced federal

matching funds to continue funding for the Emergency Department Care Coordination (EDCC) Program and to complete several enhancements that should result in cost savings to the Commonwealth and higher-quality care for Medicaid and FAMIS beneficiaries. Language is also modified to clarify reporting requirements. The 2017 General Assembly established the EDCC program to provide a single, statewide technology solution that connects all hospital emergency departments in the Commonwealth to facilitate real-time communication and collaboration among physicians, other healthcare providers and other clinical and care management personnel for patients receiving services in hospital emergency departments for the purpose of improving the quality of patient care services. The program also integrates the state's Prescription Monitoring Program and the Advance Healthcare Directive Registry. To date, 100 percent of hospital emergency departments and health plans and 74 percent of physicians participate in the EDCC Program. The program has been successful in supporting a reduction in opioid overdoses since its implementation and has been critical in coordinating responses to the COVID-19 pandemic. The program has been funded through federal HITECH funding which provides a 90 percent federal match rate, but the federal funding will end on September 30, 2021. However, Medicaid Management Information Systems funding could be used to assist in leveraging additional federal dollars to complete the system enhancements at a 75 percent federal match rate.)

	Item 309 #1s	
FY20-21	FY21-22	

Department of Health Professions

Health and Human Resources

\$0 \$532,000 NGF

Language:

Page 343, line 34, strike "\$36,027,084" and insert "\$36,559,084".

Page 343, after line 51, insert:

"C. The Department of Health Professions shall have authority to increase fees from permitted pharmaceutical processors to generate up to \$532,000 the second year from nongeneral funds, as needed, for the implementation of botanical cannabis legislation pursuant to Senate Bill 1333 in the 2021 Session."

Explanation:

(This amendment provides up to \$532,000 the second year from nongeneral funds and language providing the Department of Health Professions with the authority to increase fees from permitted pharmaceutical processors pursuant to the implementation of botanical cannabis legislation pursuant to Senate Bill 1333.)

Item 309 #2s

Health and Human Resources

Department of Health Professions

Language

Language:

Page 343, after line 51, insert:

"C. The Department of Health Professions shall study and make recommendations regarding the oversight and regulation of advanced practice registered nurses (APRNs). The department shall review recommendations of the National Council of State Boards of Nursing, analyze the oversight and regulations governing the practice of APRNs in other states, and review research on the impact of statutes and regulations on practice and patient outcomes. The department shall report its findings to the Governor and General Assembly by November 1, 2021."

Explanation:

(This amendment directs the Department of Health Professions to study and make recommendations regarding the oversight and regulations of advanced practice registered nurses.)

		Item 312 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$11,136,631 \$20,682,315	

Language:

Page 344, line 38, strike "\$250,286,516" and insert "\$282,105,462". Page 345, after line 37, insert:

"H. The Department of Medical Assistance Services shall amend the Title XXI Children's Health Insurance Program (CHIP) State Plan for the Family Access to Medical Insurance Security (FAMIS) program to elect the "unborn child option" for purposes of prenatal coverage under the CHIP program, without regard to the pregnant woman's immigration or citizenship status if other applicable state eligibility requirements are met. The department shall have the authority to implement this change effective July 1, 2021, or consistent with the effective date in the State Plan Amendment approved by the Centers for Medicare and Medicaid Services (CMS), and prior to completion of any regulatory process necessary to implement this change."

Explanation:

(This amendment provides \$11.1 million from the general fund and \$20.7 million from federal funds to amend the Virginia Family Access to Medical Insurance Security (FAMIS) State Plan to allow the payment for prenatal care for all children regardless of the expectant mother's immigration status pursuant to provisions in Title XXI of the federal 2009 CHIP Reauthorization Act that includes care of all children who upon birth will be U.S. citizens, U.S. nationals, or qualified aliens. A companion amendment in the Medicaid program (Item 313) reduces expenditures by \$13.4 million general fund and \$13.4 million in matching federal

Medicaid funds based on the adoption of this change. Consequently, the provision of these services results in a savings of \$2,292,083 to the general fund and the receipt of \$7,253,601 more from matching federal funds for the CHIP program, which has a higher federal match rate than the Medicaid program.)

		Item 312 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	(\$1,834,183) \$1,834,183	(\$5,378,570) \$5,378,570	GF NGF

Language:

Explanation:

(This amendment captures \$1.8 million from the general fund the first year and \$5.4 million from the general fund the second year and a corresponding increase in federal Children's Health Insurance Program (CHIP) matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for the fourth quarter of fiscal year 2021 and the first and second quarters of fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The CHIP match rate factors in the Medicaid match rate and is therefore increased as a result. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022.)

		Item 313 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$6,304,339 \$6,304,339	\$60,829,273 \$60,829,273	

Language:

Page 345, line 39, strike "\$16,291,925,668" and insert "\$16,304,534,346".

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,616,581,861".

Page 375, strike line 57 and insert:

"2. Effective May 1, 2021, the Department of Medical Assistance Services shall increase the rates for agency- and consumer-directed personal care, respite and companion services in the home and community based services waivers and Early Periodic Screening, and Diagnosis and

Treatment (EPSDT) program by 6.4 percent. The department shall have the authority to implement these changes prior to completion of any regulatory process undertaken in order to effect such change.

- 3. Effective November 1, 2021, the Department of Medical Assistance Services shall increase the rates for agency- and consumer-directed personal care, respite and companion services in the home and community based services waivers and Early Periodic Screening, and Diagnosis and Treatment (EPSDT) program by 14.3 percent. The department shall have the authority to implement these changes prior to completion of any regulatory process undertaken in order to effect such change.
- 4. The Governor shall include in the introduced budget for the 2022 Session, submitted pursuant to § 2.2-1509, Code of Virginia, appropriations to support additional rate increases for agency-and consumer-directed personal care, respite and companion services that reflect additional increases in the state minimum wage such that the rates: (i) maintain the existing differential between the consumer-directed Rest-of-State rate above the state minimum wage; (ii) maintain the differential between the Northern Virginia and the Rest-of-State rate; and (iii) for agency-directed services are increased by the same percentage increase applied to consumer-directed services based on the prior provisions."

Page 376, strike lines 1 through 5.

Explanation:

(This amendment adds \$6.3 million from the general fund and \$6.3 million from federal Medicaid matching funds the first year and \$60.8 million the second year from the general fund and \$60.8 million in federal Medicaid matching funds the second year to increase provider rates for personal care, respite care, and companionship services provided in Medicaid waiver programs by 6.4 percent on May 1, 2021 and 14.3 percent effective November 1, 2021. These rate increases will supplement increases provided in Chapter 1289, 2020 Acts of Assembly, which provided a five percent increase beginning July 1, 2020 and a scheduled two percent increase on July 1, 2021. These rate increases enable providers to cover the direct costs for hourly care and ensure the safety of the patients and compliance with minimum wage increases scheduled to increase to \$9.50 per hour on May 1, 2021 and to \$11.00 per hour on January 1, 2022. In addition, language directs the Governor to include the appropriate rate increases as a result of changes in the state minimum wage for these services in his budget for the 2022-24 biennium for consideration in the 2022 General Assembly)

Item 313 #2s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services (DMAS) shall convene a

workgroup and make recommendations on a Medicaid home visiting benefit to support members' health, access to care and health equity. The workgroup shall include representatives from DMAS, Managed Care Organizations, the Virginia Department of Health, the Department of Health Professions, licensed and unlicensed providers of maternal and child health services, stakeholder groups, and community organizations. The workgroup shall: (i) analyze federal and state regulations and funding mechanisms impacting establishment of a Medicaid home visiting benefit; (ii) review home visiting strategies and benefits implemented in other state Medicaid programs; (iii) analyze and make recommendations on appropriate services and rates to be included in a Medicaid home visiting benefit; and (iv) project estimated costs over the next five years. The department shall report on the results and recommendations of the workgroup to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by December 1, 2021."

Explanation:

(This amendment directs the Department of Medical Assistance Services to convene a workgroup and make recommendations on a Medicaid home visiting benefit.)

Item 313 #3s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services shall defer the next scheduled nursing facility rate rebasing for one year in order to utilize the calendar year 2021 cost reports as the base year. The deferred year's rates would reflect the prior year rates inflated according to the existing reimbursement regulations. The department shall have the authority to implement these changes effective July 1, 2021, and prior to the completion of any regulatory process undertaken in order to effect such change."

Explanation:

(This amendment adds language directing the Department of Medical Assistance Services to defer nursing home rebasing until it is able to use 2021 cost reports in the rebasing process. Under the current rebasing schedule, cost reports ending in calendar year 2020 will be the basis of the next nursing facility rate rebasing. Due to the COVID-19 pandemic, the 2020 base year cost reports will contain significant variable data that reflects a mixture of increased costs (some permanent, others temporary) and reduced occupancy. Given the COVID-19 vaccination program, it is expected that the 2021 cost reports would not be as affected by the increased costs associated with the COVID-19 pandemic, and would better reflect ongoing Medicaid costs as a basis for the next rebasing.)

Item 313 #4s

Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance	\$0	\$10,747,100	GF
Services	\$0	\$10,747,100	NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,516,417,515". Page 349, line 46, strike "455" and insert "1,095".

Explanation:

(This amendment adds \$10.7 million from the general fund and a like amount of federal Medicaid matching funds in the second year to increase the number of Family and Individual Support (FIS) waiver slots by 650 in the second year bringing the total number of waiver slots funded in fiscal year 2022 to 1,200 in order to address the Priority One waiting list.)

Item 313 #5s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services shall modify its contracts with managed care organizations to require annual reporting, with regard to Medicaid Community Mental Health Rehabilitation Services, on: (i) the total number of provider terminations by year since FY 2018 and the number terminated with and without cause; (ii) the locality the terminated providers served; and (iii) the number of Medicaid members the providers were serving prior to termination of their provider contract. The department shall modify its contracts with the managed care organizations to require compliance with these provisions, effective July 1, 2021, such that the first reporting of this information by the managed care organizations shall be submitted by September 1, 2021."

Explanation:

(This amendment directs the Department of Medical Assistance Services to require its managed care organizations annually report, with regard to Medicaid Community Mental Health Rehabilitation Services, on information related to provider terminations.)

Item 313 #6s

Health and Human Resources

FY20-21

FY21-22

Department of Medical Assistance	\$0	\$5,200,000	NGF
Services			

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,500,123,315". Page 364, after line 2, insert:

"9. The department shall amend the State plan for Medical Assistance to implement a supplemental inpatient payment for Lake Taylor Transitional Care Hospital based on the difference between Medicaid reimbursement and the inpatient Upper Payment Limit for non-state, government-owned hospitals. The department shall include in its contracts with managed care organizations a percentage increase for Lake Taylor Transitional Care Hospital consistent with the fee-for-service supplemental payment percentage increase. The department shall adjust capitation payments to Medicaid managed care organizations to fund this percentage increase. Both the contract changes and capitation rate adjustments shall be compliant with 42 C.F.R. 438.6(c)(1)(iii) and subject to Centers for Medicare and Medicaid Services approval. Prior to submitting the State Plan Amendment or making the managed care contract changes, Lake Taylor Transitional Care Hospital shall enter into an agreement with the department to transfer the non-federal share for these payments. The department shall have the authority to implement these reimbursement changes consistent with the effective date(s) approved by the Centers for Medicare and Medicaid Services (CMS). No payments shall be made without CMS approval."

Explanation:

(This amendment adds \$5.2 million in nongeneral fund appropriation the second year and language to allow Lake Taylor Transitional Care Hospital, a non-state, government-owned hospital located in Norfolk, Virginia, to provide the Commonwealth's share of funding through an intergovernmental transfer process to procure supplemental payments from the Medicaid program.)

		Item 313 #7s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$57,210 \$57,210	GF NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,495,037,735". Page 376, unstrike lines 29 through 34.

Explanation:

(This amendment provides \$57,210 from the general fund and a like amount of federal Medicaid matching funds the second year for the Department of Medical Assistance Services to amend the State Plan for Medical Assistance or any waiver under Title XIX of the Social

Security Act to increase the income eligibility for participation in the Medicaid Works Program to 138 percent of the Federal Poverty Level (FPL). Current eligibility for the Medicaid Works Program is 80 percent of the FPL and was not adjusted to 138 percent of FPL when Medicaid Expansion was implemented.)

Item 313 #8s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 365, line 10, strike "December 15, 2020", insert "November 1, 2021".

Explanation:

(This amendment extends the workgroup established to evaluate strategies to reduce unnecessary utilization by Medicaid members of emergency departments. The workgroup was delayed in beginning its deliberations and the additional time would result in better recommendations for consideration by the 2022 General Assembly Session.)

Item 313 #9s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after insert 17, insert:

"AAAAAA. The Department of Medicaid Assistance Services shall analyze utilization of Transportation Network Companies (TNC) Type II Non-Emergency Medical Transportation (NEMT) providers in the Medicaid fee-for-service program and the department's contracted managed care organizations (MCOs). The department shall identify any barriers to patient access to TNC Type II NEMT services. In its review of barriers to access TNC Type II NEMT benefits, the department shall identify any gaps in TNC Type II service contracting between the department's contracted MCOs, or their transportation brokers, and TNC Type II NEMT providers. Additionally, the department shall examine the eligible patient population for TNC Type II NEMT services to ensure all clinically indicated Medicaid beneficiaries are eligible for TNC Type II NEMT services. Further, the department shall examine the necessity of TNC Type II operating requirements and identify any extraneous service requirements limiting TNC Type II services. The department shall report its findings and recommendations to the Chairs of House Appropriations and Senate Finance and Appropriations Committees by October 1, 2021."

Explanation:

(This amendment directs the Department of Medical Assistance Services to analyze the usage of Transportation Network Companies (TNC) Type II Non-Emergency Medical Transportation (NEMT) providers in the Medicaid fee-for-service program and the department's contracted managed care organizations (MCOs) and to identify any barriers to accessing such providers.)

		Item 313 #10s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$2,682,089 GF \$4,186,201 NGF	

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,501,791,605".

Explanation:

(This amendment provides \$2.7 million from the general fund and \$4.2 million from nongeneral funds the second year for Medicaid-related costs of remote patient monitoring serves provided via telemedicine for Medicaid recipients with medically necessary conditions pursuant to Senate Bill 1338.)

		Item 313 #11s	3
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$354,766 \$354,766	

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,495,632,847".

Page 376, unstrike lines 20 through 28.

Page 376, line 27, strike "2020" and insert "2021".

Explanation:

(This amendment restores \$354,766 the second year from the general fund and a like amount of federal Medicaid matching funds to increase supplemental physician payments for physicians employed at Children's National Medical Center, a freestanding children's hospital serving the Northern Virginia region. Funding for this Item was provided in Chapter 1289, 2020 Acts of Assembly, unallotted in April, 2020 and eliminated in Chapter 56, 2020 Special Session I Acts of Assembly.)

Item 313 #12s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services is authorized to amend the State Plan under Title XIX of the Social Security Act to add coverage for the current procedural terminology (CPT) codes for Applied Behavioral Analysis that were added to the CPT list in January 2019, or any future updates to these CPT codes. The department shall have the authority to implement related programmatic changes to service definitions, prior authorization and utilization review criteria, provider qualifications, and reimbursement rates for the Behavioral Therapy Program. The department shall have the authority to implement these changes effective December 1, 2021, and prior to completion of any regulatory process to effect such changes."

Explanation:

(This amendment adds Medicaid coverage for Applied Behavioral Analysis services that were added to current procedural terminology costs in January 2019 and provides authority to the Department of Medical Assistance Services to make changes based on future updates, including necessary changes to rates, service definitions, and other programmatic requirements.)

		Item 313 #13s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	\$0 \$0	\$119,995 GF \$119,995 NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,495,163,305".

Page 374, unstrike lines 14 through 23.

Page 374, line 22, strike "2020", insert "2021".

Explanation:

(This amendment restores \$119,995 from the general fund and a like amount of matching federal Medicaid funds the second year and language to require the Department of Medical Assistance Services to modify nursing facility capital reimbursement for a nursing facility that became a free-standing facility because the hospital with which the facility was associated relocated. Funding for this Item was provided in Chapter 1289, 2020 Acts of Assembly, unallotted in April, 2020 and eliminated in Chapter 56, 2020 Special Session I Acts of

Assembly.)

Item 313 #14s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. The Department of Medical Assistance Services, in coordination with the Department of Behavioral Health and Developmental Services, shall submit a request to the Centers for Medicare and Medicaid Services to amend its 1915(c) Home & Community-Based Services (HCBS) waivers to allow telehealth and virtual and/or distance learning as a permanent service option and accommodation for individuals on the Community Living, Family and Individual Services and Building Independence Waivers. The amendment, at a minimum, shall include all services currently authorized for telehealth and virtual options during the COVID-19 pandemic. The departments shall actively work with the established Developmental Disability Waiver Advisory Committee and other appropriate stakeholders in the development of the amendment including service elements and rate methodologies. The department shall have the authority to implement these changes prior to the completion of the regulatory process."

Explanation:

(This amendment adds language directing the Department of Medical Assistance Services to request amendments from the federal Centers for Medicare and Medicaid Services to the Home and Community Based Services Waivers to permanently continue telehealth and virtual and/or distance learning as service options for disabled individuals receiving these waiver services.)

Item 313 #15s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 380, line 38, after "participation." strike the remainder of the line. Page 380, strike line 39.

Explanation:

(This amendment directs the Department of Medical Assistance Services to continue to allow, beyond the COVID-19 emergency, Medicaid agency-directed personal care and respite services to conduct telephonic supervisory visits by a licensed nurse. The department's forms would be used to document the interaction during these phone calls and to meet the standards already

established by the department to include verbal consent, authorization, and confirmation of participation.)

Item 313 #16s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 382, after line 17, insert:

"AAAAAA. Notwithstanding any other provision of law, any nursing facility that increases rates or adds special charges for residents, regardless of payer and including private pay, since the beginning of the Public Health Emergency (PHE) as declared by the U.S. Secretary of Health and Human Services and during such emergency, that are 10 percent or more higher than the monthly charge in the last full month prior to the month in which the PHE began is hereby ineligible for the additional per diem rates authorized in Item 380 Paragraph LLLLL in this Act and for financial assistance provided in Item 479.10 in this Act from federal Coronavirus Relief Funds. Upon a complaint by a resident or family member that a nursing facility has imposed higher rates or special charges that exceed the threshold, the Department of Medical Assistance Services shall verify such action with the nursing facility. If the department determines the nursing facility has increased rates or imposed new charges that exceed the threshold, the department shall no longer make additional payments pursuant to the previously mentioned provisions of this Act. Furthermore, the department shall adjust future payments to such nursing facility to recapture prior payments made under those same provisions."

Explanation:

(This amendment provides that any nursing facility in the Commonwealth that has increased their rates or added additional charges during the public health emergency for COVD-19, that are 10 percent or higher than charges prior to the emergency are not eligible for the \$20 per day per diem or any financial assistance from the Coronavirus Relief Fund as authorized in the current budget. These additional funding streams were provided by the Commonwealth to assist nursing facilities during the public health emergency to help offset losses of revenue or higher costs related to the pandemic. This amendment ensures that nursing facilities that dramatically increase their charges or add special fees to recover costs during the public health emergency are not eligible for the additional state assistance and would also be required to repay any prior payments from that assistance.)

Item 313 #17s

Health and Human Resources

FY20-21

FY21-22

Department of Medical Assistance	\$0	(\$13,428,714)	GF
Services	\$0	(\$13,428,714)	NGF

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,468,065,887".

Explanation:

(This amendment captures savings of \$13.4 million from the general fund and \$13.4 million from federal matching Medicaid funds from extending the provision for the payment of prenatal care for pregnant women through the Medicaid program regardless of the expectant mother's status, pursuant to provisions in Title XXI of the federal 2009 CHIP Reauthorization Act that includes care of all children without regard for an expectant mother's citizenship status who would otherwise be eligible under state requirements. A companion amendment in the FAMIS program (Item 312) adds language and funding for this initiative. The provision of these services results in a savings of \$2.3 million to the general fund and the receipt of \$7.3 million more from matching federal funds for the CHIP program, which has a higher federal match rate than the Medicaid program.)

		Item 313 #18s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	\$0 \$0	\$34,718 GF \$34,718 NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,494,992,751". Pag 376, unstrike lines 35 through 39.

Explanation:

(This amendment provides \$34,718 from the general fund and a like amount of federal Medicaid matching funds the second year to add tobacco cessation services to the Medicaid program for adults not otherwise currently covered. The federal Patient Protection and Affordable Care Act (ACA) requires that Medicaid provide coverage for prevention services, including tobacco cessation, for individuals enrolled pursuant to the ACA. This amendment allows all adults in Medicaid to have access to tobacco cessation services.)

		Item 313 #19s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	(\$39,410,177) \$39,410,177	\$0 GF \$0 NGF

Page 346, line 54, strike "\$527,992,971" and insert "\$567,403,148". Page 347, after line 7, insert:

"4. Any repayment by managed care organizations resulting from exceeding their profit caps for not meeting the medical loss ratios pursuant to their contracts with the Department of Medical Assistance Services, shall be deposited to the Health Care Fund."

Explanation:

(This amendment reduces \$39.4 million from the general fund the first year and adds an equivalent amount of nongeneral fund appropriation to reflect a non-participating tobacco manufacture's payment to the Health Care Fund. Since this fund is used as state match for Medicaid, additional revenue to the fund offsets the general fund match for Medicaid. In addition, language also clarifies that repayments from managed care organizations are to be deposited to the Health Care Fund.)

		Item 313 #20s	S
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	(\$114,851,105) \$114,851,105	(\$191,551,022) \$191,551,022	GF NGF

Language:

Explanation:

(This amendment captures \$114.9 million from the general fund the first year and \$191.6 million from the general fund the second year and a corresponding increase in federal Medicaid matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for the fourth quarter of fiscal year 2021 and the first and second quarters of fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of fiscal year 2022. The general fund savings for the state behavioral health facilities are also factored into these numbers and total \$808,764 the first year and \$1.8 million the second year.)

Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance	\$0	(\$104,168)	
Services	\$0	\$2,314,798	

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,497,133,945".

Page 346, line 55, strike "\$496,601,500", insert "\$496,705,668".

Page 382, after line 17, insert:

"AAAAAA. The Department shall amend the State Plan for Medical Assistance to allow payment of medical assistance services delivered to Medicaid-eligible students when such services qualify for reimbursement by the Virginia Medicaid program and may be provided by school divisions, regardless of whether the student receiving care has an individualized education program or whether the health care service is included in a student's individualized education program. Such services shall include those covered under the state plan for medical assistance services or by the Early and Periodic Screening, Diagnostic, and Treatment (EPSDT) benefit as specified in § 1905(r) of the federal Social Security Act, and shall include a provision for payment of medical assistance for health care services provided through telemedicine services, as defined in § 38.2-3418.16. No health care provider who provides health care services through telemedicine shall be required to use proprietary technology or applications in order to be reimbursed for providing telemedicine services."

Explanation:

(This amendment provides \$2.2 million from federal Medicaid funds the second year related to expanding Medicaid coverage of school-based services outside of a student's individualized education program. There is no state match required as the local schools certify local expenditures as the state match. The amendment also reflects a savings of \$104,168 from the general fund the second year since five percent of the additional federal funding for school-based services is retained by the state and deposited to the Health Care Fund, which offset general fund costs.)

		Item 313 #22	S
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	(\$800,000) (\$800,000)	\$0 \$0	GF NGF

Language:

Page 345, line 39, strike "\$16,291,925,668" and insert "\$16,290,325,668".

Page 346, line 7, after "A.", strike "1.".

Page 346, strike lines 11 through 16.

Explanation:

(This amendment reduces \$800,000 from the general fund the first year and a like amount of federal Medicaid matching funds to reflect the temporary pause in Medicaid billing by the Commonwealth Center for Children and Adolescents (CCCA) as a result of not meeting the accreditation standards necessary to bill for Medicaid. The facility is expected to begin billing again in the second year once accreditation is achieved, so this amendment adjusts the appropriation in the first year only. Language is also removed that would have allowed these funds to be transferred to CCCA; however such action is unnecessary as the facility is using special funds to cover the loss in Medicaid revenue.)

		Item 313 #23s
Health and Human Resources	FY20-21	FY21-22
Department of Medical Assistance Services	\$0 \$0	(\$10,062,988) GF (\$38,332) NGF

Language:

Page 345, line 39, strike "\$17,494,923,315" and insert "\$17,484,821,995".

Explanation:

(This amendment removes \$10.1 million from the general fund and \$38,332 in federal Medicaid matching funds to reflect the updated timeline for the phase-in of community behavioral health services in Medicaid. The redesign of these services was originally planned to be implemented beginning January 2021, however the budget impacts of COVID-19 resulted in delaying the beginning date to July 1, 2021. In the Special Session budget, Chapter 56, the full second year funding was restored, but the funding did not reflect the revised timeline. This amendment adjusts funding for these services to reflect the updated timeline including how the various services will be phased in over the next year.)

Item 313 #24s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 381, line 38, strike "\$16" and insert "\$13".

Page 381, line 41, after "fee.", insert:

"The department is authorized to set the administration fee at the same level as the Medicare reimbursement rates for COVID-19 vaccines if federal law or policy changes such that the Medicaid costs of the COVID-19 vaccine administration fee is 100 percent reimbursed from federal funds."

Explanation:

(This amendment corrects the fee amount included in the introduced budget for Medicaid reimbursement for vaccine administration in a pharmacy. Currently, Medicaid does not have a vaccine administration fee for pharmacists. The introduced budget directs the Department of Medical Assistance Services to establish such a fee and sets the fee at \$16. The department's budget assumptions for this fee was \$13. This amendment corrects the language to reflect the \$13 administration fee as reflected in the actual appropriation otherwise there would be a \$7.1 million general fund shortfall in Medicaid. Language is added that if the federal government decides to provide a 100 percent federal match for vaccine administration fees related to COVID-19, then the department is authorized to increase the fee to the Medicare reimbursement level.)

		Item 314 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$15,000,000	\$0	GF

Language:

Page 382, line 18, strike "\$821,702" and insert "\$15,821,702". Page 382, after line 34, insert:

"C. The Department of Medical Assistance Services, in collaboration with the Department of Behavioral Health and Developmental Services and the Department of Planning and Budget, shall develop criteria to provide support payments to Medicaid Developmental Disability Waiver providers that have experienced a significant disruption in operations and revenue during the COVID-19 public health emergency (PHE). The criteria shall include: (i) first priority to provide financial support for providers that have received no other state or federal assistance to date during the PHE, including supported employment providers; (ii) the second priority shall be other waiver providers that have received some limited assistance from state and federal sources, including day support providers, but may need additional assistance; (iii) the third priority to support other waiver providers that are still in operation and are at risk of closing due to the PHE disruption and for which the Commonwealth needs to maintain an adequate provider network such that when the PHE emergency ends there are sufficient providers to meet the service needs of Medicaid members; (iv) a support payment amount to be based on the provider's Medicaid monthly revenue and service authorizations prior to the PHE, however, for the period of assistance provided to the provider the payment may not exceed 65 percent of that prior monthly revenue; (v) the time period for assistance, which may not exceed three months; and (vi) other relevant criteria to meet the intent of this funding. The department shall issue guidelines within 20 days of enactment of this Act. After the guidelines are issued providers shall have 30 days to submit their applications consistent with the guidelines. The department shall determine the appropriate payments based on the guidelines, which shall be

prorated if funding is insufficient, and begin making the payments no later than 30 days after the provider submission deadline. Any unexpended balance as of June 30, 2021 shall not revert to the general fund, but shall be reappropriated for this purpose into FY 2022."

Explanation:

(This amendment adds \$15.0 million the first year from the general fund to provide support payments to Medicaid Developmental Disability Waiver providers. Payments would help support the stabilization of providers during COVID-19 and would retain community service capacity after the pandemic ends.)

		Item 315 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	(\$1,762,463) \$1,762,463	(\$5,250,333) \$5,250,333	GF NGF

Language:

Explanation:

(This amendment captures \$1.8 million from the general fund the first year and \$5.3 million from the general fund the second year and a corresponding increase in federal Children's Health Insurance Program (CHIP) matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for the fourth quarter of fiscal year 2021 and the first and second quarters of fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The CHIP match rate factors in the Medicaid match rate and is therefore increased as a result. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022.)

		Item 317 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0	(\$1,000,000)	GF

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$282,053,336".

Explanation:

(This amendment reduces by \$1.0 million general fund the second year funding provided in the introduced budget for managed care operational changes. This action leaves \$1.2 million from the general fund for this purpose. The agency's administrative budget can absorb any additional costs necessary to implement these changes.)

		Item 317 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$53,247 \$103,361	GF NGF

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$283,209,944".

Explanation:

(This amendment provides \$53,247 general fund and \$103,361 in federal Medicaid matching funds in the second year to fund the costs of an on-going orientation program for consumer-directed home care workers providing in-home care under the state's Medicaid program pursuant to Senate Bill 1102.)

Item 317 #3s

Health and Human Resources

Department of Medical Assistance Services

Language

Language:

Page 355, line 10, after "EE.", insert "1.".

Page 355, line 12, after "Social Security Act to", delete the remainder of the line, and insert:

"merge the Commonwealth Coordinated Care Plus and Medallion 4.0 managed care programs, effective July 1, 2022, into a single, streamlined managed care program that links seamlessly with the fee-for-service program, ensuring an efficient and well-coordinated Virginia Medicaid delivery system that provides high-quality care to its members and adds value for providers and the Commonwealth. The department shall have authority to implement necessary changes upon federal approval and prior to the completion of any regulatory process undertaken in order to effect such change.

2. The Department of Medical Assistance shall undertake a review of current contracts and staffing to determine the operational savings that would result from merging the Commonwealth Coordinated Care Plus and Medallion 4.0 managed care programs and may use such administrative savings that are available to fund other upfront costs associated with

merging the two managed care programs. The department shall report on its review of such administrative cost savings and merger-related costs by October 1, 2021 to the Department of Planning and Budget and the Chairs of the House Appropriations and Senate Finance and Appropriations Committees."

Page 355 strike lines 13 through 54.

Page 356, strike lines 1 through 28.

Explanation:

(This amendment directs the Department of Medical Assistance Services to merge the Commonwealth Coordinated Care Plus and Medallion 4.0 managed care programs, effective July 1, 2022, into a single, streamlined managed care program that links seamlessly with the fee-for-service program, ensuring an efficient and well-coordinated Virginia Medicaid delivery system that provides high-quality care to its members and adds value for providers and the Commonwealth.)

		Item 317 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0	\$50,000 \$50,000	GF NGF

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$283,153,336". Page 385, after line 21, insert:

- "3. The Department of Medical Assistance Services shall post on its website the complete State Plan for Medical Assistance along with all amendments in an easily searchable format to be accessible to the public.
- 4. Within five days of any submission of a state plan amendment to the Centers for Medicare and Medicaid Services, the Department of Medical Assistance Services shall post such submission in its website. The department shall also post any federal approval documents once the state plan amendment is approved.
- 5. The department shall publish a document on its website, updated annually, that lists all policy changes, including their fiscal impact, for the Medicaid program for the preceding fiscal year."

Explanation:

(This amendment provides \$50,000 from the general fund and \$50,000 from federal Medicaid matching funds to support efforts to publish the complete State Plan for Medical Assistance and amendments and other supporting materials on the Department of Medical Assistances Services' website to be more publicly accessible.)

Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance	(\$742,622)	(\$427,900)	
Services	\$742,622	\$427,900	

Page 391, after line 24, insert:

"II. The Department of Planning and Budget shall, if the public health emergency is extended into the third or fourth quarters of the second year extending the enhanced federal match pursuant to the federal Families First Coronavirus Response Act, calculate the general fund savings in the Children's Health Insurance Program administrative appropriation and unallot such amount. These savings shall revert to the general fund at the end of the fiscal year."

Explanation:

(This amendment captures \$742,622 from the general fund the first year and \$427,900 from the general fund the second year and a corresponding increase in federal Children's Health Insurance Program (CHIP) matching funds to reflect the savings to the Commonwealth pursuant to enhanced federal matching funds for fiscal year 2021 and the first and second quarters of state fiscal year 2022. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared public health emergency (PHE) due to COVID-19 expires. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022. The CHIP match rate factors in the Medicaid match rate and therefore results in a higher match. Unlike Medicaid the administrative expenditures of the CHIP program uses the same match rate as used for medical spending. Language is also included directing the Department of Planning and Budget to unallot any additional savings that results from an extension of the PHE.)

		Item 317 #6s	
Health and Human Resources	FY20-21	FY21-22	
Department of Medical Assistance Services	\$0 \$0 0.00	(\$1,166,180) (\$6,959,211) -4.00	GF NGF FTE

Language:

Page 383, line 13, strike "\$283,053,336" and insert "\$274,927,945". Page 391, strike lines 18 through 24.

Explanation:

(This amendment removes \$1.2 from the general fund and \$7.0 million from nongeneral funds and four positions to fund the fiscal impact of legislation to establish the Virginia Facilitated Enrollment Program. No bill was introduced in the Senate.)

Item 318 #1s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 392, line 36, after "department" strike the remainder of the line and insert: "for up to six months beyond the duration of the Governor's emergency declaration for COVID-19."

Explanation:

(This amendment modifies the extension allowed on conditional licenses for providers licensed by the Department of Behavioral Health and Developmental Services. A conditional license is granted to a provider who has successfully completed the application process but has not yet admitted individuals for service. It is time limited and can only be renewed one time for a total of twelve months. Due to the COVID-19 pandemic conditional licenses could be extended beyond 12 months, until December 31, 2020. This amendment allows an extension for up to six months past the duration of the Governor's emergency declaration.)

		Item 320 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$175,000	GF

Language:

Page 393, line 43, strike "\$110,804,911" and insert "\$110,979,911". Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services (DBHDS), in coordination with the Department of Medical Assistance Services (DMAS), shall contract with a vendor to review all current Medicaid Developmental Disability (DD) waiver provider rates and rate methodologies to ensure an adequate network of quality DD Waiver providers. DBHDS shall submit a rate rebase report with recommendations, including the projected fiscal impact on the Commonwealth, to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021. At a minimum, the report shall include a thorough review and alignment of each service rate methodology to reflect current service definitions and documentation requirements, the impact of minimum wage increases,

appropriate Bureau of Labor Statistics wage percentiles, and appropriate rate differentials for high cost areas of Virginia. All rate assumptions shall be based on the level of need. The vendor shall specifically evaluate the rates for the Supported Living Residential waiver service to ensure appropriate utilization of that service. The department shall actively work with the established DBHDS Provider Issue Resolution Workgroup in the development of its report."

Explanation:

(This amendment adds \$175,000 the second year from the general fund to require the Department of Behavioral Health and Developmental Services in coordination with the Department of Medical Assistance Services to review all current Medicaid Developmental Disability (DD) waiver provider rates and rate methodologies to ensure an adequate network of quality DD Waiver providers. The agency is required to report on its findings by September 1, 2021, to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees.)

Item 320 #2s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services shall continue the Temporary Detention Order Evaluator Workgroup established during the 2020 Session in Senate Bill 768. The workgroup shall report its implementation plan to the Governor, and Chairs of House Health, Welfare, and Institutions Committee, Senate Education and Health Committee, and Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by December 1, 2021."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to continue the Temporary Detention Order Evaluator Workgroup established during the 2020 Session in Senate Bill 768. The workgroup shall report its implementation plan to the Governor, and Chairs of House Health, Welfare, and Institutions Committee, Senate Education and Health Committee, and Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by December 1, 2021.)

Item 320 #3s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services shall establish a workgroup to review the current processes and barriers to sharing relevant patient information between community hospitals and Community Services Boards for shared patients subject to an Emergency Custody Order and under evaluation for a Temporary Detention Order. The department shall report its findings and recommendations to the Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by December 1, 2021.)

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to establish a workgroup to review the current processes and barriers to sharing relevant patient information between community hospitals and Community Services Boards for shared patients subject to an Emergency Custody Order and under evaluation for a Temporary Detention Order.)

Item 320 #4s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language

Page 398, line 17, strike "non-narcotic," and insert "appropriate".

Explanation:

(This amendment eliminates language requiring the use of non-narcotic, long-acting injectable prescription drug treatment regimens used for drug treatment court programs. Instead, it requires the use of appropriate long-acting injectable prescription drug treatment regimens.)

Item 320 #5s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services shall preserve historic microfiche records at Central State Hospital and work with interested partners to digitize such records to be added to the Central State Hospital Digital Library and Archives Project in order to make such information publicly available to researchers or other interested parties."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to preserve historic microfiche records at Central State Hospital and work with interested partners to digitize such records to be added to the Central State Hospital Digital Library and Archives Project.)

		Item 320 #6s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$3,750,000	GF

Language:

Page 393, line 43, strike "\$110,804,911" and insert "\$114,554,911".

Page 398, line 33, unstrike "\$7,500,000" and strike "\$3,750,000".

Page 398, line 35, after "hospital care", insert "or to increase capacity in the community for patients on the Extraordinary Barriers List".

Explanation:

(This amendment provides \$3.8 million the second year from the general fund to fully restore funding for alternative inpatient options to state behavioral health hospital care through the establishment of two-year pilot projects to reduce census pressures on state hospitals. A total of \$7.5 million each year was approved in the 2020 Session, but the funding was unalloted due to the revenue impact from the Coronavirus Pandemic. Half the funding was restored in the Special Session in Chapter 56 and this amendment restores the remaining amount. The amendment also clarifies that pilot projects could also include the option to build community capacity for patients on the Extraordinary Barriers List in order to relieve census pressure.)

		Item 320 #7s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and	0.00	-5.00	FTE

Language:

Explanation:

(This amendment reduces the position level for the Department of Behavioral Health and Developmental Services for five positions that were added with the funding for administrative costs of STEP-VA. This funding is a restoration of funding previously approved in Chapter 1289 from the 2020 Session. However, the funding for these positions was eliminated in

Chapter 56 in the Special Session I, but the position level was not reduced. Therefore, this amendment corrects the position level for the agency.)

Item 320 #8s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 399, after line 39, insert:

"II. The Department of Behavioral Health and Developmental Services, in collaboration with the Virginia Treatment Center for Children (VTCC), shall examine and develop strategies to better utilize VTCC in assisting with relief for the census pressures on the Commonwealth Center for Children and Adolescents (CCCA). The strategies to be examined shall include, but are not limited to: (i) diversion strategies when CCCA is near capacity; (ii) increasing the number of Temporary Detention Order admissions; and (iii) operating as a step-down facility from CCCA. The department shall report its finding and recommendations to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees and the the Joint Subcommittee to Study Mental Health Services in the Commonwealth in the 21st Century by July 1, 2021."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to work with the Virginia Treatment Center for Children (VTCC) to examine and develop strategies that better utilize VTCC in assisting with relief on the census pressures on the Commonwealth Center for Children and Adolescents (CCCA).

		Item 321 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$150,000	GF

Language:

Page 399, line 42, strike "\$61,270,529" and insert "\$61,420,529".

Page 401, unstrike lines 5 through 8.

Page 401, line 5, strike "\$150,000 the first year and".

Explanation:

(This amendment provides \$150,000 the second year from the general fund to provide for the transportation costs of patients discharged from state hospitals that were admitted under a

Temporary Detention Order (TDO). Oftentimes individuals under a TDO are transported to a state facility that is hours away from the individual's home location and therefore upon discharge may have difficulty getting transportation back to their home location. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

Item 321 #2s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 401, line 24, after "services.", insert:

"The funding shall only be provided to members that are accredited by the Council on Accreditation of Peer Recovery Support Services (CAPRSS)."

Explanation:

(This amendment modifies the criteria to receive funds from a passthrough grant to the Virginia Association of Recovery Residencies such that the funds can only be provided to members that are accredited by the Council on Accreditation of Peer Recovery Support Services (CAPRSS).)

		Item 321 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Behavioral Health and Developmental Services	\$0	\$143,260	GF

Language:

Page 399, line 42, strike "\$61,270,529" and insert "\$61,413,789".

Explanation:

(This amendment restores \$143,260 from the general fund the second year to expand the Adverse Childhood Experiences (ACE) initiative. It supports a full-time Central Office position to: (i) provide oversight over 100 ACE Interface Master Trainers across the Commonwealth; (ii) plan and develop additional ACE Interface Master Trainer Cohorts in their region; and (iii) plan and facilitate monthly learning community meetings for each training cohort; etc. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

Item 321 #4s

Health and Human Resources

Department of Behavioral Health and Developmental Services

Language

Language:

Page 400, line 11, after "services.". insert:

"The Department of Behavioral Health and Developmental Services shall report quarterly on the the distributions of this funding to Community Services Boards (CSB). The report shall include the amounts distributed to each CSB, the total annual allocation of funding for each CSB, the current available funding balance for each CSB, and any reallocations of funding between CSBs. The report shall be submitted to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than 20 days after the end of each fiscal quarter."

Explanation:

(This amendment directs the Department of Behavioral Health and Developmental Services to report quarterly on the distribution of funding provided to Community Services Boards (CSB) for the purchase of acute inpatient or community-based psychiatric services (known as LIPOS).)

Item 322 #1s

Health and Human Resources

Grants to Localities Language

Language:

Page 405, line 35, after "regimens.", insert:

"In expending any amount, the department shall prioritize allocation of the funding to any portion of medication assisted treatment that are not covered by insurance."

Page 405, line 36, strike "non-narcotic, non-addictive," and insert "appropriate".

Page 405, line 37, after "probation;" insert "or".

Page 405, line 38, after "jail", strike "; or (iii)", and insert:

". The department shall ensure that a portion of the funding is used for appropriate prescription drug treatment regimens for individuals who are".

Explanation:

(This amendment modifies language in the introduced budget related to medication assisted treatment for individuals who are addicted to opioids. Language changes require that a portion of the funding be used for appropriate prescription drug treatment regimens, instead of solely non-narcotic, long-acting injectable prescription drug treatment regimens. This change allows the service provider to prescribe the most effective treatment option for the individual obtaining

treatment services. The amendment also clarifies that funding should be prioritized to those who are not otherwise insured.)

		Item 322 #2s	
Health and Human Resources	FY20-21	FY21-22	
Grants to Localities	\$0	\$2,100,800	GF

Language:

Page 402, line 8, strike "\$554,715,057" and insert "\$556,815,857". Page 405, line 53, after "first year and", unstrike "\$3,700,800". Page 406, line 1, strike "\$1,600,000".

Explanation:

(This amendment restores \$2.1 million the second year from the general fund to expand forensic discharge planning services at three additional jails with a high percentage of inmates with serious mental illness. The General Assembly provided \$1.6 million from the general fund in fiscal year 2020 for discharge planning services at two local jails. Discharge planning includes linking inmates with serious mental illness to community providers for treatment and housing and other needed services as they transition from jails to the community. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

		Item 326 #1s	
Health and Human Resources	FY20-21	FY21-22	
Mental Health Treatment Centers	(\$2,142,601) \$2,142,601	(\$4,285,202) \$4,285,202	GF NGF

Language:

Page 408, line 27,, strike "the general fund" and insert "federal funds".

Page 408, line 28, after "facilities." insert:

"The department shall coordinate with the Virginia Department of Health (VDH) and local health districts as appropriate to coordinate its testing and surveillance activities in order to access federal ELC Enhancing Detection Expansion grant funding provided to VDH through the Centers for Disease Control."

Explanation:

(This amendment supplants \$2.1 million the first year and \$4.3 million the second year from the general fund with federal funds from the ELC Enhancing Detection Expansion grant funding

provided to the Virginia Department of Health (VDH) through the Centers for Disease Control. The federal Consolidated Appropriations Act, 2021 that was signed into law on December 27, 2020 and provides substantial federal funding to states for COVID-19 testing. The estimate for Virginia is about \$490 million. This amendment directs the Department of Behavioral Health and Developmental Services to coordinate its COVID-19 surveillance activities for state facilities with VDH in order to access those federal funds.)

		Item 326 #2s	
Health and Human Resources	FY20-21	FY21-22	
Mental Health Treatment Centers	\$0	\$765,428	GF

Language:

Page 408, line 2, strike "\$286,799,776" and insert "\$287,565,204".

Explanation:

(This amendment restores \$765,428 the second year from the general fund to provide critical clinical staffing at the Commonwealth Center for Children and Adolescents. These funds were included in the budget passed in March during the 2020 Session, but subsequently unallotted in April due to the COVID-19 pandemic and eliminated in the Chapter 56 budget adopted in the 2020 Special Session.)

		Item 339 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department for Aging and Rehabilitative Services	\$0	\$1,000,000	GF

Language:

Page 411, line 47, strike "\$100,388,245" and insert "\$101,388,245". Page 413, line 15, after "first year and", unstrike "\$6,976,719." Page 413, line 16, strike "\$5,976,719".

Explanation:

(This amendment provides \$1.0 million from the general fund the second year to increase contracts with community based brain injury service providers to provide for cost increases to bring salaries in line with market demands in order to ensure staff retention, address existing case management waiting lists, enhance virtual programming, expand partnerships, and development responses to the high-intensity needs of Virginians with brain injury, especially those related to COVID-19 and the opioid epidemic.)

1	tem	330	#28

Health and Human Resources	FY20-21	FY21-22
Department for Aging and Rehabilitative Services	\$0	\$425,000 GF

Page 411, line 47, strike "\$100,388,245" and insert "\$100,813,245".

Page 413, line 6, unstrike the second "\$5,521,858".

Page 413, line 6, strike the second "\$5,096,858".

Explanation:

(This amendment adds \$425,000 the second year from the general fund to support 17 Centers for Independent Living that provide independent living services including independent living skills training, advocacy, information and referral, peer mentoring, and transition services to people with significant disabilities. Transition services include youth transition services, services for individuals trying to transition from nursing facilities and other institutions, and services to prevent institutionalization.)

		Item 340 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department for Aging and Rehabilitative Services	\$0	\$150,000	GF

Language:

Page 414, line 9, strike "\$36,139,218" and insert "\$36,289,218".

Page 415, unstrike lines 28 through 35.

Page 415, line 28, strike "\$150,000 the first year and".

Page 415, line 35, after "year." insert:

"The services provided through this program shall be coordinated with the Department of Behavioral Health and Developmental Services to help address the inappropriate utilization of state psychiatric beds by patients with dementia."

Explanation:

(This amendment restores funding the second year from the general fund budget to provide an interdisciplinary plan of care and dementia care management for 50 Virginia residents diagnosed with dementia. The service would be provided through a partnership with the memory and aging care clinic at the University of Virginia and the Alzheimer's Association. Funding of \$150,000 each year of the 2020-22 biennium was provided in Chapter 1289, 2020 Acts of Assembly to serve 50 individuals diagnosed with dementia. That funding was unallotted

and subsequently eliminated in Chapter 56, 2020 Special Session I Acts of Assembly. This amendment would restore the funding in fiscal year 2022 enabling the program to serve 50 individuals with dementia.)

		Item 350 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0	\$1,400,000 \$15,928,859	GF NGF

Language:

Page 420, line 44, strike "\$152,429,363" and insert "\$169,758,222".

Page 423, line 4, after "O." insert "1."

Page 423, line 6, strike "2020." and insert:

"2021 and 18 percent effective July 1, 2022."

Page 423, after line 6, insert:

"2. The Department of Social Services shall develop a plan to increase the standards of assistance by 18 percent annually until they equal 50 percent of the federal poverty level."

Explanation:

(This amendment adds \$1.4 million from the general fund and \$15.9 million from the Temporary Assistance to Needy Families (TANF) block grant the second year to increase the standards of assistance by 18 percent beginning in fiscal year 2022. Language requires the Department of Social Services to develop a plan to increase the standards of assistance by 18 percent annually until the standards equal 50 percent of the federal poverty level, which is estimated to take four years. TANF eligibility is tied to the monthly cash assistance value. In 1985, a three person family in the City of Richmond could be eligible with net income below 48 percent of the federal poverty level. To qualify now, the family's net income must be less than 30 percent of the federal poverty level.)

Item 350 #2s

Health and Human Resources

Department of Social Services

Language

Language:

Page 424, after line 6, insert:

"W. The Department of Social Services shall develop demonstration projects with a goal of assisting families to earn a living wage and escape poverty. The projects shall include a benefit cliff pilot, a pilot to demonstrate the role of location as a key determinant for health and economic success, and a pilot program to determine the impact of cashing out key program

supports. The Department of Social Services shall determine the authority needed to implement the demonstration projects as outlined in the Social Security Act, the Food and Agriculture Act of 1977, and other federal legislation that provides for the development and testing of demonstration projects. The department shall identify federal and private grants that are available to help fund these demonstration projects. The department shall prepare a report on each proposed demonstration project that describes the project, identifies a funding amount needed, and indicates whether federal funds or other private funding is available to help implement such projects and shall submit the report to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021."

Explanation:

(This amendment adds language directing the Department of Social Services to develop several anti-poverty demonstration projects for consideration by the General Assembly.)

		Item 350 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$2,120,420	NGF

Language:

Page 420, line 44, strike "\$152,429,363" and insert "\$154,549,783".

Page 424, after line 6, insert:

"W. Out of this appropriation, \$2,120,420 the second year from the Temporary Assistance to Needy Families (TANF) block grant shall be provided for the Department of Social Services to implement a program so that TANF-eligible individuals may save funds in an individual development account established for the purposes of home purchase, education, starting a business, transportation, or self-sufficiency. The TANF funds shall be deposited to the individual development accounts at a match rate determined by the department."

Explanation:

(This amendment adds \$2.1 million the second year from the Temporary Assistance to Needy Families block grant for the implementation of individual development accounts for TANF recipients by the Department of Social Services.)

		Item 350 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$25,000	GF

Language:

Page 420, line 45, strike "\$152,429,363" and insert "\$152,454,363".

Page 424, after line 6, insert:

"W. Out of this appropriation, \$25,000 from the general fund the second year shall be deposited to the Virginia Digital Equity Pilot Program Fund."

Explanation:

(This amendment provides \$25,000 from the general fund the second year to fund Senate Bill 1462 which establishes a broadband pilot project in order to augment federal assistance funding (FCC Lifeline) for households currently participating in the Supplemental Nutrition Assistance Program. This funding would serve between 150 to 300 households with commensurate administrative funding to start the program. Funds of \$5.75 per household augments the \$9.25 federal assistance to equal the cost to afford low cost internet access of \$15 a month).

		Item 350 #5s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0	(\$1,000,000) (\$1,436,986)	GF NGF

Language:

Page 420, line 45, strike "\$152,429,363" and insert "\$149,992,377".

Explanation:

(This amendment reduces \$1.0 million from the general fund and \$1.4 million from federal funds the second year for the Supplemental Nutrition Employment and Training Program. This amendment reduces that additional funding by about half, thereby reducing the number of localities to which the program would expand.)

		Item 351 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0	\$2,150,048 \$2,175,528	GF NGF

Language:

Page 424, line 9, strike "\$525,386,732" and insert "\$529,712,308".

Page 425, unstrike lines 7 through 11.

Page 425, line 8, strike "each" and insert "the second".

Page 425, line 9, strike "2021" and insert "2022".

Explanation:

(This amendment provides \$2.2 million from the general fund and \$2.2 million from nongeneral

funds the second year to increase the minimum pay band for local departments of social services positions in fiscal year 2022. The pay band minimum would increase by 20 percent for family services positions and 15 percent for all other benefit program services positions, self sufficiency services positions, and administration positions that are currently below the new minimum threshold.)

		Item 351 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$84,004	GF
•	\$0	\$51,203	NGF

Language:

Page 424, line 9, strike "\$525,386,732" and insert "\$525,521,939".

Explanation:

(This amendment provides \$84,004 from the general fund and \$51,203 from nongeneral funds the second year to fund the fiscal impact on local departments of socials services' workload associated with the Senate Bill 1321, which expands the stepparent adoption provisions to allow a person who is not the child's stepparent but has a legitimate interest in the child to file a joint petition for adoption with the child's birth parent or parent by adoption.)

		Item 351 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0 \$0 0.00	(\$376,910) (\$2,709,766) -2.00	GF NGF FTE

Language:

Page 424, line 9, strike "\$525,386,732" and insert "\$522,300,056".

Explanation:

(This amendment removes \$376,910 from the general fund and \$2.5 million from nongeneral funds that would have funded the fiscal impact of legislation to establish the Virginia Facilitated Enrollment Program. No bill was introduced in the Senate.)

Health and Human Resources	FY20-21	FY21-22	
		Item 353 #1s	

Department of Social Services

\$0 \$4,400,000 GF

Language:

Page 426, line 8, strike "\$41,177,762" and insert "\$45,577,762".

Page 426, line 23, after "\$1,420 per month", insert "and effective July 1, 2021, a rate of \$1,704 per month".

Explanation:

(This amendment provides \$4.4 million from the general fund the second year to increase the Auxiliary Grant (AG) rate, a state supplement that provides maintenance and care to aged, blind, and disabled adults residing in assisted living facilities (ALF), adult foster care homes, or supportive housing settings, by 20 percent on July 1, 2021.)

		Item 353 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$750,000	GF

Language:

Page 426, line 8, strike "\$41,177,762" and insert "\$41,927,762".

Page 427, after line 6, insert:

"G. Out of this appropriation, \$750,000 the second year from the general fund shall be transferred to the Virginia Sexual and Domestic Violence Prevention Fund."

Explanation:

(This amendment provides \$750,000 the second year from the general fund for the Virginia Sexual and Domestic Violence Prevention Fund that was created pursuant to Senate Bill 297 in the 2020 Session. The program would be administered by the Department of Social Services and the Department of Health. The fund will award grants on a competitive basis to local sexual and domestic violence agencies engaged in evidence-informed sexual and domestic violence prevention work.)

		Item 354 #1s
Health and Human Resources	FY20-21	FY21-22
Department of Social Services	\$0 \$0	(\$260,406) GF (\$99,594) NGF

Language:

Page 427, line 8, strike "\$261,995,376" and insert "\$261,635,376".

Explanation:

(This amendment reduces \$260,406 from the general fund and \$99,594 from nongeneral funds the second year for the fiscal impact of Senate Bill 1338, which establishes the State-Funded Kinship Guardianship Assistance Program (State KinGAP), which allows payments to be made to relatives, including fictive kin, who receive custody of a child. A corresponding amendment in the Children's Services Act reflects the increase in maintenance payments from the State KinGAP program while this amendment reflects the decrease in Title IV-E program costs as some children will transition to the State-Funded Kinship Guardship Assistance Program, which has lower maintenance payments than the Title IV-E program.)

Item 354 #2s

Health and Human Resources

Department of Social Services

Language

Language:

Page 431, after line 23, insert:

"X. The Department of Social Services shall create a diversion program supporting relative and fictive kin families who have received temporary physical and legal custody from the court that makes use of all federal and state monies available to provide a payment to relative and fictive kin families who have temporary custody through a court order. The department shall report on the program to the Chairs of the House Health, Welfare and Institutions, House Appropriations, Senate Rehabilitation and Social Services, and Senate Finance and Appropriations Committees by December 1, 2021, with the steps needed to implement the diversion program."

Explanation:

(This amendment adds language directing the Department of Social Services to create a diversion program supporting relative and fictive kin families who receive custody of a child from the court and report the steps to implement such program to the House Health, Welfare and Institutions, House Appropriations, Senate Rehabilitation and Social Services, and Senate Finance and Appropriations Committees by December 1, 2021.)

Item 354 #3s FY21-22

Health and Human Resources

\$0 \$3,500,000 NGF

FY20-21

Department of Social Services

Language:

Page 427, line 8, strike "\$261,995,376" and insert "\$265,495,376". Page 431, after line 23, insert:

"X. The Department of Social Services shall extend payments to children aging out of the Fostering Futures program past age 21 through September 30, 2021."

Explanation:

(This amendment adds language to extend payments to children aging out of the Fostering Futures program through September 30, 2021. The recently passed federal Coronavirus Relief and Response Supplemental Appropriations Act has extended the John H. Chafee Foster Care Program for Successful Transition to Adulthood funding through September 2021. These funds can be used to support children aging out of services during the COVID-19 pandemic.)

		Item 354 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$2,923,178) \$2,923,178	(\$5,846,356) \$5,846,356	GF NGF

Language:

Explanation:

(This amendment captures \$2.9 million the first year and \$5.8 million the second year from the general fund and includes an offsetting amount of federal Title IV-E foster care funding to reflect a higher federal match rate. The federal Families First Coronavirus and Response Act, passed in March 2020, increased the federal match rate for Medicaid by 6.2 percentage points until the end of the quarter in which the declared COVID-19 public health emergency (PHE) expires. The federal Title IV-E programs used the Medicaid match rate to determine the state and federal share of the program. The PHE is extended in 90-day increments and was recently extended into the fourth quarter of state fiscal year 2021. In addition, the Acting U.S. Secretary of Health and Human Services has informed all state Governors of their intention to extend the PHE through the end of calendar year 2021 to provide greater budget certainty to states. Therefore, this amendment also captures the savings from the first and second quarters of state fiscal year 2022.)

		Item 356 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$500,000	NGF

Language:

Page 431, line 35, strike "\$60,957,967" and insert "\$61,457,967".

Page 432, line 45, after "first year and", strike "\$1,500,000", and insert "\$2,000,000".

Explanation:

(This amendment provides \$500,000 from the Temporary Assistance to Needy Families block grant funds the second year to Northern Virginia Family Services. The additional funds are for the specific purpose of providing services and wrap-around supports to Virginians suffering impacts from COVID-19 in all northern Virginia counties, cities and towns. Services and supports include but are not limited to: the provision of food, financial assistance to prevent homelessness, access to health and mental health care, childcare and workforce development programs.)

		Item 356 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$500,000	\$0	GF

Language:

Page 431, line 34, strike "\$61,857,967" and insert "\$62,357,967".

Page 435, unstrike lines 40 through 42.

Page 435, line, 42, after "support.", insert:

Explanation:

(This amendment provides \$500,000 the first year from the general fund for the Laurel Center to support the rehabilitation of an existing freight station building located adjacent to the emergency shelter. This building will house the Empowerment Program and provide a safe and convenient location for job readiness and skills training for survivors of domestic violence.)

		Item 356 #3s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	\$0	\$200,000	GF

Language:

Page 431, line 34, strike "\$60,957,967" and insert "\$61,157,967".

Page 431, after line 32, insert:

"Out of this appropriation, \$200,000 the second year from the general fund is provided for the Department of Social Services to increase interpretation and translation services to help immigrants in Virginia access local resources through 2-1-1, including healthcare, housing, and other social services."

Explanation:

[&]quot;Any unexpended balance as of June 30, 2021, shall not revert to the general fund but shall be reappropriated in FY 2022."

(This amendment provides \$200,000 the second year from the general fund for the Department of Social Services to increase interpretation and translation services to help immigrants in Virginia access local resources through 2-1-1, including healthcare, housing, and other social services.)

		Item 356 #4s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$250,000)	\$0	GF

Language:

Page 431, line 35, strike "\$61,857,967" and insert "\$61,607,967". Page 436, strike lines 10 through 11.

Explanation:

(This amendment removes \$250,000 from the general fund the first year for a new passthrough grant provided in the introduced budget for Children's Harbor, a child care provider in Hampton Roads, to expand services on the Eastern Shore.)

		Item 357 #1s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$45,653) (\$103,404)	(\$182,610) \$0	GF NGF

Language:

Page 436, line 13, strike "\$43,640,542" and insert "\$43,491,485". Page 436, line 13, strike "\$16,494,086" and insert "\$16,311,476".

Explanation:

(This amendment reduces \$45,656 from the general fund and \$103,404 from nongeneral funds the first year and \$182,610 from the general fund the second year for the purchase of personal protective equipment (PPE) for licensing inspectors related to COVD-19. The Virginia Department of Emergency Management should be able to secure PPE for the Department of Social Services' licensing inspectors based on the substantial funding that agency has received to purchase PPE.)

Item 359 #1s

Health and Human Resources

Department of Social Services

Language

Language:

Page 440, after line 43, insert:

"K. The Department of Social Services as administrator of the federal Community Services Block Grant shall establish an interagency working group to develop recommendations for implementation of local criminal justice diversion programs. Each diversion program should offer standards for providing persons charged with lower-level offenses alternatives to arrest, conviction or incarceration for lower-level offenses. The scope of these programs shall not include behavioral health issues as those priorities are being addressed elsewhere. The working group should include the appropriate offices and agencies of Health and Human Resources, Commerce and Trade, Public Safety and Homeland Security and the Governor's Chief Diversity, Equity and Inclusion Officer. The interagency working group shall work with community action agencies, local governments including local law enforcement, representatives of the judicial system, civil rights organizations as well as other stakeholders to develop locallybased solutions. The recommendations shall provide for two-generation whole family strategies that deal with meeting the needs of the potential offender and his or her entire family by addressing issues related to poverty, including homelessness. The Department of Social Services shall submit its recommendations to the Chairs of the House Appropriations Committee and the Senate Finance and Appropriations Committee no later than September 30, 2021."

Explanation:

(This amendment establishes an interagency working group to develop recommendations for local criminal justice diversion programs to provide alternatives to arrest, conviction or incarceration for lower-level offenses.)

		Item 359 #2s	
Health and Human Resources	FY20-21	FY21-22	
Department of Social Services	(\$923,804) (\$1,224,577)	(\$923,804) (\$1,224,577)	GF NGF

Language:

Page 438, line 39, strike "\$121,912,263" and insert "\$119,763,882". Page 438, line 39, strike "\$118,755,668" and insert "\$116,607,287".

Explanation:

(This amendment removes \$923,804 from the general fund and \$1.2 million nongeneral funds each year included in the introduced budget for Virginia Case Management System non-Medicaid modifications.)

Item 373 #1s

Natural Resources FY20-21 FY21-22 Department of Conservation and \$0 \$170,000 GF Recreation

Language:

Page 447, line 9, strike "\$84,750,087" and insert "\$84,920,087".

Explanation:

(This amendment restores \$170,000 GF the second year that was unalloted in Chapter 1289 for the environmental education experiences identified in Item 373, Paragraph K. This request recognizes the commitment Virginia made to its students and the importance environmental education plays in developing and nurturing watershed stewards. These funds provide for meaningful interactions and learning opportunities for students throughout the Chesapeake Bay watershed.)

Item 373 #2s

Natural Resources

Department of Conservation and Recreation

Language

Language:

Page 451, following line 36, insert:

"R. The Department of Conservation and Recreation and the Virginia Department of Agriculture and Consumer Services shall convene a workgroup including the Department of Forestry, the Virginia Department of Transportation, the Department of Wildlife Resources, the Virginia Native Plant Society, the Virginia Association of Counties, the Virginia Municipal League, Blue Ridge PRISM, the Audubon Society of Northern Virginia, the Virginia Nursery and Landscape Association, the Virginia Agribusiness Council, the Virginia Farm Bureau Federation, and a representative of the Virginia Cooperative Extension Program with expertise in crop and weed sciences to assess the sale and use of invasive plant species in the retail, landscape, greenhouse, and nursery industries and consider measures to reduce or eliminate the sale and use of invasive plant species in the Commonwealth and promote the sale and use of native plants. In conducting this assessment, the workgroup shall examine measures to reduce, mitigate, and eliminate the continued sale and use of invasive species as identified in the list of Virginia invasive plant species maintained by the Department of Conservation and Recreation. The workgroup shall evaluate measures including: (i) labeling plants as invasive plant species at the point of sale; (ii) taxing the sale of invasive plant species and applying revenues to the removal of invasive plant species or the restoration of sites for native habitat; (iii) adding

invasive plant species currently being offered for sale to the list of plants declared to be noxious weeds by the Board of Agriculture and Consumer Services through regulations adopted pursuant to Chapter 8 (§ 3.2-800 et seq.) of Title 3.2 of the Code of Virginia (the Noxious Weed List); (iv) supporting education and outreach, including state partnerships with nonprofit organizations dedicated to the preservation of Virginia's natural heritage, regarding the reduction of the use of invasive plant species and the promotion of the use of noninvasive or native plant species as substitutes; and (v) introducing measures to increase the use of native plants on properties and projects owned by localities or the Commonwealth. The workgroup shall submit its assessment, including any recommendations regarding statutory changes and changes to regulations or guidance documents adopted by relevant agencies, including changes related to the placement of plant species on the Noxious Weed Lists, to the Governor and the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations no later than December 1, 2021."

Explanation:

(This amendment directs the creation of a stakeholder workgroup to assess the sale and use of invasive plant species in the retail, landscape, greenhouse, and nursery industries and consider measures to reduce or eliminate the sale and use of invasive plant species in the Commonwealth and promote the sale and use of native plants.)

		Item 374 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$0	\$1,511,600	GF

Language:

Page 451, line 38, strike "\$81,225,147" and insert "\$82,736,747".

Page 454, following line 29, insert:

Explanation:

(This amendment provides \$1.5 million GF in the second year for the costs of connecting Mason Neck State Park to the local municipal drinking water system.)

		Item 374 #2s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$1,412,000	\$0	GF

[&]quot;S. Included in the amounts for this Item is \$1,511,600 the second year from the general fund to connect Mason Neck State Park to a public drinking water supply system."

Language:

Page 451, line 38, strike "\$72,102,316" and insert "\$73,514,316".

Page 452, following line 36, insert:

"4. In addition to the amounts provided in paragraph C.1., the Department is authorized to provide \$1,412,000 in the first year from the general fund for the modernization of the Rhododendron Restaurant and lodge unit repairs."

Explanation:

(This amendment provides \$1.4 million GF in the first year to modernize and repair lodge units and the Rhododendron Restaurant. This investment will support a key component of the Breaks Interstate Park's five-year strategic plan, make the park more sustainable from a maintenance perspective, and significantly boost overnight visitation.)

		Item 374 #3s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$0	(\$3,500,000)	GF

Language:

Page 451, line 38, strike "\$81,225,147" and insert "\$77,725,147".

Page 452, line 6, following "Virginia." insert:

"Out of the amounts in the second year, and pursuant to the provisions of § 4-13.00 of this Act, the Virginia Land Conservation Foundation shall reserve an amount up to \$5,000,000 for the identification and evaluation of a project focused on the preservation of tribal lands of the Chickahominy Tribe and for the conservation and preservation of River Farm in Alexandria, Virginia. Consideration of these projects shall be consistent with the policies and practices of the Foundation pursuant to § 10.1-1020, Code of Virginia."

Page 454, strike line 17 through line 18.

Explanation:

(This amendment provides for the consideration of one-time funding of up to \$5.0 million for the identification, acquisition of preservation of tribal lands for the Chickahominy Tribe and for the conservation and preservation of River Farm.)

		Item 374 #4s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$0	(\$4,000,000)	GF

Language:

Page 451, line 38, strike "\$81,225,147" and insert "\$77,225,147".

Page 454, line 19, strike "\$5,000,000" and insert "\$1,000,000",

Page 454, line 21, following "Cemetery" insert "."

Page 454 line 21, strike "and creation of".

Page 454, line 22, strike "the Harmony Living Shoreline Memorial".

Page 454, line 22, strike "all aspects of".

Page 454, line 23, strike "the project to include but not limited to".

Page 454, line 25, following "Farm", strike "," and insert "and".

Page 454, line 27, following "located", strike "," insert "." and strike the remainder of the line.

Page 454, strike line 28 through line 29.

Explanation:

(This amendment limits the scope of Project Harmony to the removal and relocation of the Columbian Harmony tombstones from Virginia to the New Harmony Cemetery in Landover, Maryland.)

		Item 374 #5s	
Natural Resources	FY20-21	FY21-22	
Department of Conservation and Recreation	\$740,000	\$0	GF

Language:

Page 451, line 38, strike "\$72,102,316" and insert "\$72,842,316".

Page 451, line 49, strike "\$35,624,459" and insert "\$36,364,459".

Page 454, unstrike line 4 through line 6.

Explanation:

(This amendment restores \$740,000 GF in the first year as a source of matching funds for the construction of Riverfront Park in the City of Danville. This funding was approved in the 2020 Session and subsequently unallotted.)

		Item 377 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Environmental Quality	\$0 0.00	\$115,000 1.00	GF FTE

Language:

Page 455, line 31, strike "\$52,900,649" and insert "\$53,015,649".

Explanation:

(This amendment provides \$115,000 the second year from the general fund and one position for the Department of Environmental Quality to administer permit-by-rule applications for energy storage projects pursuant to the provisions of SB 1207 of the 2021 General Assembly.)

		Item 377 #2s	
Natural Resources	FY20-21	FY21-22	
Department of Environmental Quality	\$0 0.00	\$230,000 2.00	GF FTE

Language:

Page 455, line 31, strike "\$52,900,649" and insert "\$53,130,649".

Page 457, following line 41, insert:

"M. The Director of the Department of Environmental Quality shall convene a working group for the purpose of developing an annual or project-based fee schedule for the the review of erosion and sediment control plans related to solar energy project applications. The working group shall include representatives of (i) private sector companies that own or operate solar energy facilities, (ii) local governments that permit solar facilities, and (iii) other stakeholders determined by the Department to be necessary to the development of the fee schedule."

Explanation:

(This amendment provides the funding for staffing dedicated to the erosion and sediment control program to review for solar project permitting and provide this service to projects upon the request of a locality pursuant to the provisions of SB 1258 of the 2021 General Assembly.)

Item 377 #3s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 457, following line 41, insert:

"M. The Department of Environmental Quality, in consultation with the Department of Agriculture and Consumer Services and the Department of Forestry, shall establish a workgroup to review the practice of retiring agricultural land for the generation of nutrient credits and determine its impact on agricultural sustainability, farmland retention, farmland preservation, and functions of the nutrient credit exchange in the Virginia portion of the Chesapeake Bay watershed and its subwatersheds. If it is determined that there is impact on farmland retention/availability, the report should include recommendations regarding how the nutrient

credit trading regulations and/or underlying statutory authority should be changed to help reduce the loss of prime farmland. If the land for nutrient credits is converted to forestland, the workgroup should identify what protections are in the nutrient credit trading regulations to ensure the forestland is managed under a forestry management plan and/or noxious weed or invasive species are controlled. The review shall be completed and provided to the Chairs of the House Committee on Agriculture, Chesapeake and Natural Resources, the Senate Committee on Agriculture, Conservation, and Natural Resources and the Virginia delegation of the Chesapeake Bay Commission by December 1, 2021. The workgroup shall include representatives of the Virginia Agribusiness Council, Virginia Farm Bureau, the Chesapeake Bay Commission, Virginia Cooperative Extension, the Virginia Department of Transportation, Home Builders Association of Virginia, Virginia Association for Commercial Real Estate, representatives from local Soil and Water Conservation Districts, representatives of local governments, local economic development officials, and other stakeholders deemed appropriate by the Department."

Explanation:

(This amendment directs the creation of a multi-agency workgroup to review the practice of retiring agricultural land for the generation of nutrient credits and determine its impact on agricultural sustainability, farmland retention, farmland preservation, and functions of the nutrient credit exchange in the Virginia portion of the Chesapeake Bay watershed and its subwatersheds.)

Item 377 #4s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 457, following line 41, insert:

"M. The Department of Environmental Quality shall convene a workgroup, in conjunction with the Department of Health and the Department of Agriculture and Consumer Services, to conduct research and complete a single collaborative report that provides findings and recommendations related to: (i) the location, frequency, and severity of harmful algae blooms in Virginia waters; (ii) the factors that lead to the formation and occurrence of harmful algae blooms; and, (iii) plans and strategies for state agencies to lead or support appropriate mitigation efforts. The workgroup shall provide its findings to the Chairs of the House Agriculture, Chesapeake and Natural Resources Committee and Senate Agriculture, Conservation and Natural Resources Committee no later than September 1, 2021."

Explanation:

(This amendment directs DEQ, VDACS, and VDH to convene a joint workgroup to provide a report on the prevalence of harmful algae blooms in Virginia waters and strategies to address

their occurrence.)

Item 377 #5s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 457, unstrike line 20 through line 29.

Page 457, line 30, unstrike "Virginia." and strike the remainder of the line.

Page 457, strike line 31 through line 41.

Explanation:

(This amendment reverses the proposed water quality enhancement fee included in SB 1100, as introduced.)

Item 378 #1s

Natural Resources

Department of Environmental Quality

Language

Language:

Page 458, line 31, following "Department." strike the remainder of the line. Page 458, strike line 32 through line 37.

Explanation:

(This amendment removes a proposed modification to the language agreed to by the General Assembly regarding the regulation of hydrofluorocarbons in existing manufacturing processes in Virginia.)

Item 382 #1s

Natural Resources

Department of Game and Inland Fisheries

Language

Language:

Page 461, following line 25, insert:

"Notwithstanding § 29.1-113, Code of Virginia, access permit fees at boat ramps owned or managed by the Department of Wildlife Resources shall not be assessed prior to July 1, 2022, pending a study by the Department on the costs and benefits of such fees and the impact of said

fees on recreational users in Virginia. As part of this study, the Department shall convene a stakeholder group for the purpose of developing and providing recommendations on access permit fees, various alternatives, and other issues related to the use and maintenance of Department-owned boat ramp facilities. The stakeholder work group shall be composed of representatives of registered boat owners, paddlecraft liveries, outdoor outfitters, and other non-registered vessel recreational users of such boat ramps, or other affected parties the Department deems necessary. The Department shall report the study findings, conclusions and recommendations to the Governor and Chairs of the House Appropriations and Agriculture, Chesapeake and Natural Resources Committees and the Chairs of the Senate Finance and Appropriations and Agriculture, Conservation and Natural Resources Committees by October 1, 2021."

Explanation:

(This amendment delays by one fiscal year the authority of the Department of Wildlife Resources to assess fees for access to boat ramps it owns or manages.)

Item 383 #1s

Natural Resources

Department of Game and Inland Fisheries

Language

Language:

Page 461, strike line 35 through line 44, and renumber accordingly.

Explanation:

(This amendment removes an obsolete requirement that the Department of Wildlife Resources notify specific Standing Committee Chairs prior to consolidation of regional offices or reorganization of divisional responsibilities.)

Item 383 #2s

Natural Resources

Department of Game and Inland Fisheries

Language

Language:

Page 462, following line 22, insert:

"F. The Directors of the Departments of Wildlife Resources and Conservation and Recreation shall assess the feasibility of developing the Rapidan Wildlife Management Area into a State Park and provide a copy of its assessment to the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations no later than November 1, 2021. This assessment shall include, but not be limited to, the impact on wildlife

currently within the management area; any restrictions of deeds, easements, covenants or grant funding used in the initial acquisition of the wildlife management area; capital costs for developing recreational access and overnight accommodations; ongoing operational costs of the proposed facility; and an anticipated timeline for phased access to public recreational facilities within the existing master planning process."

Explanation:

(This amendment requires the DWR and DCR to assess the feasibility of converting the existing Rapidan Wildlife Management Area into a State Park.)

		Item 385 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Historic Resources	\$0 0.00	\$250,000 1.00	GF FTE

Language:

Page 462, line 43, strike "\$8,020,283" and insert "\$8,270,283".

Page 466, following line 12, insert:

"W. The Department of Historic Resources is authorized to enter into an agreement with one or more Virginia-based Historically Black Colleges and Universities to provide paid internships to enrolled students for data collection and outreach activities to expand Virginia's historical property catalogue to include underrepresented African American and indigenous communities. Included within the amounts in this Item, \$100,000 the second year from the general fund is provided for an initial cohort group in Fiscal Year 2022."

Explanation:

(This amendment provides \$250,000 to the Department of Historic Resources for the expansion of Virginia's historical property catalogue to include underrepresented African American and indigenous communities. Included in these amounts is \$110,000 to support one new FTE, \$100,000 for a grant program to provide paid internships in partnership with Virginia's HBCU's to conduct fieldwork, and \$40,000 for cultural data enrichments and database enhancements for the Virginia Cultural Resources Information System.)

		Item 386 #1s	
Natural Resources	FY20-21	FY21-22	
Department of Historic Resources	\$139,328	\$139,328	GF

Language:

Page 466, line 14, strike "\$973,912" and insert "\$1,113,240".

Page 466, line 14, strike "\$973,912" and insert "\$1,113,240".

Page 466, strike line 22 through line 23, and insert:

"As a condition of receipt of funding in this Item, and consistent with the provisions of § 4-13.00 of this Act, it is the intent of the General Assembly that all funding designated in Item 385 shall be provided by the department as expressly stated therein."

Explanation:

(This amendment restores operational and administrative support at the Department of Historic Resources that was approved during the 2020 Session, but subsequently unallotted. In addition the language requires the Department to comply with the explicit intent of the General Assembly, and in a manner consistent with all current laws, in administering the funding designated in Item 385. Included within this amount is funding for in-band compensation adjustments for a variety of professional staff as recommended by the Department of Human Resources.)

Item 391 #1s

Public Safety and Homeland Security

Secretary of Public Safety and Homeland Security

Language

Language:

Page 471, after line 13, insert:

"G. The Secretary, in consultation with the Department of Planning and Budget, and the Secretary of Finance, as well as appropriate public safety or other agency staff, shall evaluate existing funding that has been previously authorized for the enforcement of laws related to controlled substance prohibition. The Secretary shall identify, for controlled substances which have recently been decriminalized or legalized, sources of funding that are authorized for enforcement activities, including funding dedicated to patrol, arrests, incarceration, training, or other activities, that may be saved and reallocated towards other programs. The Secretary shall report on the information required in this paragraph to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by December 1, 2021."

Explanation:

(This amendment directs the Secretary of Public Safety and Homeland Security to evaluate available base funding in the budget in order to identify savings pursuant to the reduction or elimination of penalties for the possession of controlled substances that may be reallocated to drug treatment and rehabilitation programs.)

Item 391 #2s

Public Safety and Homeland Security

Secretary of Public Safety and Homeland Security

Language

Language:

Page 471, after line 13, insert:

"G. The Secretary of Public Safety and Homeland Security shall, as part of the workgroup created pursuant to Senate Bill 1339, include a comprehensive review of all systems and processes necessary for the expungement or sealing of police or court records. The review shall examine improvements in technology, systems improvements, and automation of manual processes that would improve the cost-effectiveness of expungements and sealing of records, including consideration of the technology, systems and processes utilized by other states, in order to accommodate the requirements of the legislation passed by the General Assembly. The Secretary shall also report on the projected costs of changes to systems and processes for consideration in the 2022 General Assembly Session. The Secretary shall report on the workgroup's findings and recommendations consistent with the date in the legislation."

Explanation:

(This amendment directs the Secretary of Public Safety and Homeland Security as part of the workgroup created pursuant to Senate Bill 1339 to include a comprehensive review of all systems and processes necessary for the expungement or sealing of police or court records and to report on the costs of needed improvements to the General Assembly.)

		Item 394 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Virginia Alcoholic Beverage Control Authority	\$0 0.00	\$1,000,000 10.00	NGF FTE

Language:

Page 471, line 42, strike "\$24,692,092" and insert "\$25,692,092".

Explanation:

(This amendment provides \$1.0 million from nongeneral funds and ten positions the second year for the Virginia Alcoholic Beverage Control Authority's Bureau of Law Enforcement to increase enforcement related to direct shipment of alcohol.)

Item 394 #2s

Public Safety and Homeland Security

Virginia Alcoholic Beverage Control Authority

Language

Language:

Page 472, strike lines 18 through 23.

Explanation:

(This amendment eliminates authority provided in the introduced budget to authorize a Treasury Loan for the Virginia Alcoholic Beverage Control Authority to fund the start-up costs associated with legislation related to the legalization and governance of adult use marijuana products. The legislation passed by the Senate establishes a new authority making the language no longer necessary.)

Item 400 #1s

Public Safety and Homeland Security

Department of Corrections

Language

Language:

Page 478, line 25, after "\$500.", insert:

"The department shall amend its contract with the vendor that operates the Lawrenceville Correctional Center to require that this funding is provided as a bonus for correctional officers and require an accounting of the funding to the department. The department shall report on the use of this funding, including the number of correctional officers provided a bonus and, if applicable, any balances remaining to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by June 30, 2021."

Explanation:

(This amendment requires the Department of Corrections to ensure that the funding included in the introduced budget to provide a \$500 bonus for each correctional officer at Lawrenceville Correctional Center, employed by a private contractor that operates the facility, is expended only for such purpose.)

		Item 400 #2s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0	\$250,000	GF

Language:

Page 476, line 30, strike "\$821,331,843" and insert "\$821,581,843".

Page 478, after line 25 insert:

"N. Out of this appropriation, \$250,000 from the general fund in the second year is provided for the Department of Corrections to provide expanded video visitation services for inmates."

Explanation:

(This amendment provides \$250,000 from the general fund in the second year for expanded and subsidized prison family video visitation services. The purpose of this budget amendment is to promote public safety by allowing constructive reintegration for offenders with their families and children, to increase family and community ties prior to the release of the offender, and decrease the likelihood of recidivism.)

		Item 400 #3s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0 0.00	\$471,420 5.00	GF FTE

Language:

Page 476, line 30, strike "\$821,331,843" and insert "\$821,803,263".

Page 478, after line 25 insert:

"N. Out of this appropriation, \$471,420 the second year from the general fund is provided to fund five positions to implement the recommendations of the Secretary of Public Safety and Homeland Security's workgroup on Access to Sex Offender Treatment."

Explanation:

(This amendment adds \$471,420 from the general fund and five positions in the second year to partially fund the short-term recommendations that were included in the Secretary of Public Safety and Homeland Security's November 2020 Report "Access to Sex Offender Treatment in Virginia Prisons." Funding is included for two offender management positions to screen for sex offender risk at the beginning of an inmate's sentence and three positions in the sex offender services unit to increase the number of sex offender treatment providers dedicated to the provision of sex offender services.)

		Item 400 #4s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0	\$100,000	GF
•	0.00	1.00	FTE

Language:

Page 476, line 30, strike "\$821,331,843" and insert "\$821,431,843".

Page 478, after line 25, insert:

"N. Out of this appropriation, \$100,000 the second year from the general fund is provided for one coordinator position to review and oversee the modifications to the operating procedures in the Department of Corrections related to restrictive housing pursuant to required changes included in legislation passed in the 2021 General Assembly. As part of the the review of the

use of restricted housing, the department shall report on current restrictive housing procedures related to medical and mental health evaluations, existing staff that support such evaluations, and report on how legislative changes would impact the current workload of such staff. The department, in collaboration with the Department of Juvenile Justice that shall also conduct a review of its procedures and staffing impacts, shall report on its findings and any potential cost impact to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021."

Explanation:

(This amendment provides \$100,000 the second year from the general fund for a coordinator position to review and oversee the modifications to the operating procedures in the Department of Corrections related to restrictive housing pursuant to required changes included in legislation passed in the 2021 General Assembly (Senate Bill 1301). The Department of Juvenile Justice would also need to review their procedures. A report on the findings and any potential cost impact would be made to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees by September 1, 2021.)

		Item 402 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0	\$350,000	GF

Language:

Page 479, line 35, strike "\$180,965,434" and insert "\$181,315,434".

Page 482, line 26, after "first year" insert "and \$350,000 the second year."

Page 483, after line 6, insert:

- "23. Senate Bill 1113, 2021 Session -- \$50,000
- 24. Senate Bill 1336, 2021 Session -- \$50,000
- 25. Senate Bill 1461, 2021 Session -- \$50,000
- 26. Senate Bill 1310, 2021 Session -- \$50,000
- 27. Senate Bill 1339, 2021 Session -- \$50,000
- 28. Senate Bill 1395, 2021 Session -- \$50,000
- 29. Senate Bill 1406, 2021 Session -- \$50,000."

Explanation:

(This amendment provides \$350,000 from the general fund in the second year to the Corrections Special Reserve Fund to reflect the estimated impact on utilization of beds in the Commonwealth's adult correctional centers resulting in changes in criminal sentencing pursuant to seven bills that passed the Senate.)

Public Safety and Homeland Security

Department of Corrections

Language

Language:

Page 481, line 54, after "Q." insert "1.".

Page 482, after line 3, insert:

"2. Prior to the award of any contract to a vendor for implementation of electronic health records, the Department of Corrections shall report on the total costs of implementing electronic health records at all of its facilities based on the selected vendor. The department shall certify that it has sufficient on-going funding for full implementation at all facilities prior to awarding a contract. If the department has insufficient base funding, then it shall not award any contract until sufficient funding has been appropriated by the General Assembly. The department shall report on all funding currently budgeted for the project, the timeline for implementation, and the interoperability of the system. The department shall utilize its nongeneral funds appropriated for this purpose prior to using the general fund appropriation. The department shall report on the project and provide its certification to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees prior to the award of any contract."

Explanation:

(This amendment provides that the Department of Corrections report on the costs of implementing electronic health records across its facilities and certify that it has sufficient funding for full implementation prior to awarding any contract to a vendor.)

		Item 402 #3s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Corrections	\$0 0.00	\$160,000 2.00	GF FTE

Language:

Page 479, line 35, strike "\$180,965,434" and insert "\$181,125,434".

Page 483, after line 31, insert:

"Y. Out of this appropriation, \$160,000 the second year from the general fund is provided to the Department of Corrections for two full-time staff to support the Board of Local and Regional Jails."

Explanation:

(This amendment provides \$160,000 the second year from the general fund to support two positions dedicated to the Board of Local and Regional Jails pursuant to Senate Bill 1363, which expands the Board's oversight to include the state correctional system.)

Item 410 #1s

Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Emergency	\$0	\$2,500,000	GF
Management			

Language:

Page 493, line 24, strike "\$28,699,285" and insert "\$31,199,285".
Page 494, line 24, after "first year" insert "and \$2,500,000 the second year."

Explanation:

(This amendment adds \$2.5 million the second year from the general fund to continue funding for the Emergency Shelter Upgrade Assistance Fund to aid local governments in proactively preparing for emergency sheltering situations on an on-going basis.)

		Item 411 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of Emergency Management	(\$41,965,587)	\$0	GF

Language:

Page 494, line 37, strike "\$65,063,392" and insert "\$23,097,805".

Page 495, line 31, strike "\$41,965,587 the first year and".

Page 495, line 35, strike "\$31,144,081 the first year and".

Page 495, line 39, strike "\$569,833 the first year and".

Page 495, line 42, strike "\$8,050,173 the first year and".

Page 495, line 46, strike "\$2,201,500 the first year and".

Page 495, line 53, after "report", strike "the".

Page 496, line 1, strike "use of the funds in FY 2021 along with".

Explanation:

(This amendment reduces \$42.0 million the first year from the general fund that was provided in the introduced budget to the Virginia Department of Emergency Management for coordinating response and recovery efforts related to the COVID-19 pandemic. The Commonwealth recently received federal reimbursement for prior COVID-19 related expenses that were paid from the federal Coronavirus Relief Fund (CRF), authorized pursuant to the Coronavirus Aid, Relief, and Economic Security (CARES) Act of 2020. This federal funding allows the general fund provided in the introduced budget in the first year to be supplanted with newly available federal CRF funds. A separate amendment in Central Appropriations modifies the allocations from the CRF for this purpose.)

Item 423 #1s

Public Safety and Homeland Security

Department of Juvenile Justice

Language

Language:

Page 506, line 21, strike "6." and insert "C.1."

Page 506, after line 24, insert:

"2. In procuring any new security systems or components for the existing facility at Bon Air from such funds available in this Item, the Department shall consider ways to reuse the system procured in a future facility. To that end, the Department shall work with the Department of General Services to plan for reuse of a previously acquired security system in any future new facility constructed, to the extent feasible."

Explanation:

(This amendment directs the Department of Juvenile Justice to consider potential reuse of a security system in a new facility in the future, in the acquisition and procurement of such system for the existing facility at Bon Air.)

		Item 425 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0	\$501,990	GF
	0.00	5.00	FTE

Language:

Page 507, line 7, strike "\$74,070,135" and insert "\$74,572,125".

Page 509, after line 10, insert:

"Q. Out of this appropriation, \$501,990 the second year from the general fund is provided to the Department of State Police for five positions for cold case investigators to support efforts to resolve such cases."

Explanation:

(This amendment provides \$501,990 the second year from the general fund and five positions in the second year to support five cold case investigators. Legislation in the 2020 Session established a cold case file database and this funding will support ongoing work in the program.)

Public Safety and Homeland Security

Department of State Police

Language

Language:

Page 509, after line 10, insert:

"Q. The Superintendent of State Police shall report on the feasibility of establishing a registry for determining eligibility to lawfully possess a firearm for on-site rental use at a sport shooting range, based on existing state and federal laws concerning possession of firearms by persons with a history of mental illness. The report shall consider, at a minimum: (i) the information technology changes needed to collect the necessary information to determine if the renter of a firearm for on-site use is prohibited from possessing a firearm under any applicable state or federal law; (ii) the appropriate form or mechanism for collection of information to determine the mental health and criminal history of customers of sport shooting ranges; (iii) the reasonable timeline by which the registry can be implemented; and (iv) any necessary costs for implementation of a mental health background check registry for on-site firearms rentals. The department shall report to the General Assembly on the information required in this paragraph by September 30, 2021."

Explanation:

(This amendment requires the Superintendent of State Police to report on the steps needed to establish a registry for determining eligibility to rent a firearm for on-site use at sport shooting ranges.)

		Item 426 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0	(\$2,310,700)	GF

Language:

Page 509, line 13, strike "\$300,057,282" and insert "\$297,746,582".

Page 511, line 16, after "R.", insert "1.".

Page 511, line 16, strike "\$9,488,184" and insert "\$7,177,484"

Page 511, after line 16, insert:

"The Department of Planning and Budget shall unallot the appropriation provided in Paragraph R.1. at the beginning of FY 2022. The Department of State Police shall provide additional information and justification on the increase in funding for vehicle replacements. This information shall include a detailed overview of specific types of vehicles by various automobile manufacturers that are available for the agency's use and the rationale for the agency's preferred manufacturer and type of vehicle over the others. As part of this information, the department shall provide detailed information on the specific equipment needed for the vehicles and whether certain vehicle types or vehicles from different manufacturers alter the

type or cost of equipment modifications needed for the vehicles. No sooner than 30 days after this information is provided to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees, the Department of Planning and Budget is authorized to allot the funding."

Explanation:

(This amendment reduces \$2.3 million the second year from the general fund for the replacement of State Police vehicles. The introduced budget proposed \$9.5 million in additional funding. This additional funding is on top of existing base funding of \$12.7 million spent on vehicle replacement and repair. The adjustment is based on maintaining the current turnover rate of 350 vehicles per year at approximately 130,000 miles. Funding is provided for the department to purchase SUVs and associated equipment. Language is included to unallot the appropriation until such time as the department provides additional information and justification on the increase in funding for vehicle replacements, which would include a detailed overview of specific types of vehicles by various automobile manufacturers that are available for the agency's use and the rationale for the agency's preferred manufacturer and type of vehicle over the others.)

		Item 426 #2s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0	\$28,547,132	NGF

Language:

Page 509, line 12, strike "\$300,057,282" and insert "\$328,604,414".

Page 511, after line 28 insert:

"S. Out of this appropriation, \$28,547,132 the second year shall be provided from the Public Safety Trust Fund to support the Department of State Police's Compensation, Compression, and Retention Plan for State Troopers. The department is authorized to implement the plan effective in the pay period beginning August 10, 2021."

Explanation:

(This amendment provides \$28.5 million from the Public Safety Trust Fund in the second year to fully fund the Department of State Police's Trooper Compensation Plan. The Public Safety Trust Fund is created pursuant to Senate Bill 1211 and is supported by an additional vehicle registration fee of \$4.00 per year beginning July 1, 2021. The Fund will be used to establish a sworn pay structure that promotes an inclusive, diverse, and well-qualified workforce by fully addressing compensation issues that negatively impact staffing, retention, and pay compression issues at the Department of State Police.)

Public Safety and Homeland Security	FY20-21	FY21-22	
Department of State Police	\$0	\$1,145,246	GF

Language:

Page 511, line 30, strike "\$32,359,876" and insert "\$33,505,122".

Explanation:

(This amendment provides \$1.1 million from the general fund the second year for the Department of State Police to coordinate and provide training to state and local law enforcement agencies in Drug Recognition Expert techniques in order to provide effective detection and enforcement for driving under the influence of drugs, including marijuana. Approximately half of the funding is intended for course seats for political subdivisions, community colleges, and university law enforcement agencies. Funding is also provided for travel, meals, equipment, and related expenses.)

Item 427 #2s

Public Safety and Homeland Security

Department of State Police

Language

Language:

Page 512, after line 19 insert:

"D. Notwithstanding § 18.2-308.02, Code of Virginia, a person who has successfully completed online training for a Concealed Handgun Permit prior to January 1, 2021, is eligible to apply for such a permit if the person was not able to obtain an appointment with the county or city circuit court clerk due to COVID-19 restrictions. The online training course must have been successfully completed prior to January 1, 2021. The provisions of this paragraph shall expire on December 31, 2021."

Explanation:

(This language amendment authorizes a citizen to apply for a concealed handgun permit if they have demonstrated competence via online training, under certain circumstances, if their application has been impacted due to restrictions arising from the COVID-19 pandemic.)

		Item 429 #1s	
Public Safety and Homeland Security	FY20-21	FY21-22	
Virginia Parole Board	\$0	\$84,638	GF

Language:

Page 512, line 37, strike "\$2,648,168" and insert "\$2,732,806".

Explanation:

(This amendment provides \$84,638 the second year from the general fund for the costs associated with implementing Senate Bill 1103.)

Item 436 #1s

Transportation

Department of Motor Vehicles

Language

Language:

Page 523, following line 53, insert:

"U. Consistent with the provisions of § 4-13.00 of this Act, the definitions found in §46.2-1600, Code of Virginia, on June 30, 2021, shall remain in full force and effect until June 30, 2022."

Explanation:

(This amendment retains the current definitions found in the Code for "rebuilt vehicle" and "nonrepairable vehicle" and omits the definition of "cosmetic vehicle" allowing DMV customers to apply for salvage and nonrepairable certificates without submitting an estimated cost of repair.)

		Item 442 #1s	
Transportation	FY20-21	FY21-22	
Department of Rail and Public Transportation	\$22,397,000	\$0	NGF

Language:

Page 525, line 38, strike "\$699,845,958" and insert "\$722,242,958".

Page 527, following line 26, insert:

"J. Out of the amounts in this Item, and consistent with the provisions of § 4-13.00 of this Act, \$22,397,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board to the Washington Metropolitan Area Transit Authority (WMATA) Capital Fund established pursuant to § 33.2-3401, Code of Virginia to ensure Commonwealth meets its share of the dedicated regional funding for WMATA based on the 2018 needs assessment."

Explanation:

(This amendment allocates \$22.4 million NGF in the first year from public funds made

available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act to fulfill the Commonwealth's portion of the \$500 million annual commitment to the WMATA Capital program.)

Item 442 #2s

Transportation

Department of Rail and Public Transportation

Language

Language:

Page 527, following line 26, insert:

"J. Any entity established pursuant to Chapter 31 of Title 33.2, Code of Virginia, shall, as a condition of receipt of funding pursuant to § 33.2-1526.1(d)(3), Code of Virginia, and included in this Item, adopt the petition of the Fairfax County Board of Supervisors that was unanimously approved in Action Item 9 on September 15, 2020 and any private sector participating entity contributions described therein related to the naming of the "McLean - Capital One Hall" station."

Explanation:

(This amendment conditions the receipt of funding provided to the Northern Virginia Transportation Commission for distribution to the Washington Metropolitan Area Transit Authority for capital purposes and operating assistance, on adopting the petition of Fairfax County related to the naming of the McLean-Capital One Hall.)

		Item 442 #3s	
Transportation	FY20-21	FY21-22	
Department of Rail and Public Transportation	\$5,000,000	\$0	NGF

Language:

Page 525, line 38, strike "\$699,845,958" and insert "\$704,845,958".

Page 527, following line 26, insert:

"J. Out of the amounts in this Item, and consistent with the provisions of § 4-13.00 of this Act, \$5,000,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board to the Transit Ridership Incentive Program established pursuant to § 33.2-1526.3, Code of Virginia. Of these amounts the Board shall allocate \$2,500,000 to support the establishment of programs to reduce the impact of fares on low-income individuals, including reduced-fare programs and elimination of fares and \$2,500,000 for regional connectivity programs focused on congestion reduction and mitigation

through the provision of long-distance commuter routes."

Explanation:

(This amendment provides \$5.0 million in federal funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act to support transit incentives focused on fare and congestion reduction programs.)

		Item 443 #1s	
Transportation	FY20-21	FY21-22	
Department of Rail and Public Transportation	\$0 \$137,603,000	(\$50,000,000) \$0	GF NGF

Language:

Page 527, line 27, strike "\$137,107,434" and insert "\$274,710,434". Page 527, line 28, strike "\$247,112,195" and insert "\$197,112,195".

Page 528, strike line 39 through line 43, and insert:

"H. Out of the amounts in this Item, and consistent with the provisions of § 4-13.00 of this Act, \$137,603,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board for the purpose of extending intercity passenger rail service from Roanoke, Virginia to the Blacksburg-Christiansburg, Virginia area and increasing the frequency of intercity passenger rail service along the I-81/Route 29 Corridor from Washington, DC. The Secretary shall provide an assessment of both total project costs and the incremental costs resulting from modelling conducted to assess any infrastructure or network costs needed to service a rail station in Bedford, Virginia to the Chairs of the House Committee on Appropriations and the Senate Committee on Finance and Appropriations no later than October 15, 2021."

Explanation:

(This amendment provides \$137.6 million from the Coronavirus Response and Relief Supplemental Appropriations Act to ensure long-term sustainable growth of intercity passenger and commuter rail services across the Commonwealth.)

Item 446 #1s

Transportation

Department of Transportation

Language

Language:

Page 529, line 46, strike "first year" and insert "second year".

Page 529, line 48, strike "fifth" and insert "eighth". Page 529, line 49, strike "2021" and insert "2022".

Explanation:

(This amendment moves funding that was previously authorized to support the participation in the Mobility Talks International (MTI) conference in the first year to the second year of the biennium. The 2020 General Assembly approved \$50,000 in funds to support participation in the 2021 annual MTI conference, which was not required due to COVID. The next conference is scheduled to be held in Washington D.C. in January 2022.)

		Item 447 #1s	
Transportation	FY20-21	FY21-22	
Department of Transportation	\$0 \$40,000,000	(\$5,000,000) \$0	GF NGF

Language:

Page 530, line 29, strike "\$3,650,896,554" and insert "\$3,690,896,554". Page 530, line 29, strike "\$3,783,863,742" and insert "\$3,778,863,742".

Page 532, strike line 5 through line 8 and insert:

"H. Out of the amounts in this Item, and consistent with the provisions of §4-13.00 of this Act, \$40,000,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be used by the Board to support the statewide planning, development and construction of multi-use trails. The Board shall give priority to ensuring the safety of existing trails and developing new trails with a total length in excess of 35 miles and shall seek to ensure geographic diversity and multimodal connectivity in the use of these funds. Of these amounts: (i) up to \$17,500,000 shall be allocated for the Eastern Shore-Bay Coast Railway Rails to Trails project; (ii) up to \$17,500,000 shall be allocated for safety enhancements to the Washington & Old Dominion Trail at Wiehle Avenue and improving the intermodal connectivity of the Cross County Trail at Difficult Run with Great Falls Park; (iii) up to \$2,500,000 for the Fall Line Trail; and (iv) up to \$2,500,000 for the Ivy Creek Trail and the completion of the Tobacco Heritage Trail from Skipwith Road to Rudds Creek. The Office of Intermodal Planning and Investment shall coordinate a policy working group comprised of representatives of the Departments of Transportation, Rail and Public Transportation, Conservation and Recreation, the Statewide Trails Advisory Committee and the staff of the House Appropriations and Senate Finance and Appropriations Committees to recommend a prioritization process for the identification of new multi-use trail opportunities, a master planning process, and funding needs assessment. A report of its recommendations shall be communicated to the Chairs of the House Appropriations and Senate Finance and Appropriations Committees no later than October 15, 2021."

Explanation:

(This amendment provides for a one-time capitalization of \$50,000,000 for a statewide multiuse trail initiative using federal funds available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act.)

		Item 447 #2s	
Transportation	FY20-21	FY21-22	
Department of Transportation	\$10,000,000	\$0	NGF

Language:

Page 530, line 29, strike "\$3,650,896,554" and insert "\$3,660,896,554".

Page 531, line 23, following "Virginia." insert:

"In addition to these amounts, and consistent with the provisions of § 4-13.00 of this Act, \$10,000,000 the first year from public funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act (P.L. 116-260) shall be allocated by the Commonwealth Transportation Board for a connected infrastructure urban redevelopment demonstration project within and adjacent to the Virginia Tech campus in the City of Falls Church, Virginia."

Explanation:

(This amendment provides \$10.0 million in federal funds made available for Highway Infrastructure Programs by the Coronavirus Response and Relief Supplemental Appropriations Act to the Transportation Partnership Opportunity Fund for an urban smart infrastructure test bed in partnership with the City of Falls Church and Virginia Tech.)

		Item 461 #1s	
Veterans and Defense Affairs	FY20-21	FY21-22	

Secretary of Veterans and Defense Affairs

Language:

Page 545, strike lines 8 through 10, and insert:

"Out of this appropriation, up to \$190,000 the second year from the general fund shall be used to support a Military Liaison position under the Secretariat."

Explanation:

(This amendment reallocates existing funding of \$190,000 the second year from the general fund that is currently provided for matching funds for the U.S. Department of Defense Office of

Economic Adjustment grants, to support an ongoing Military Liaison position. The U.S. Department of Defense has reorganized the Office of Economic Adjustment and the grants are no longer expected. Therefore, the funding is not needed for the grant matching requirement.)

		Item 464 #1s	
Veterans and Defense Affairs	FY20-21	FY21-22	
Department of Veterans Services	\$0 0.00	\$147,500 1.00	GF FTE

Language:

Page 547, line 24, strike "\$22,451,304" and insert "\$22,598,804".

Explanation:

(This amendment provides \$147,500 in the second year from the general fund to support a liaison to military spouses position in the Department of Veterans Services, pursuant to Senate Bill 1150. This funding includes \$22,300 in one-time equipment funding. The ongoing cost for the position is \$125,200.)

		Item 477 #2s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	\$0	\$20,311,179	GF

Language:

Page 554, line 9, strike "\$138,332,991" and insert "\$158,644,170".

Page 561, strike lines 12 through 23.

Page 565, line 1, strike "1".

Page 565, strike lines 5 through 51.

Page 565, line 1, following II. insert:

"The Governor is hereby authorized to allocate a sum of up to \$118,067,180 the second year from this appropriation, to the extent necessary to offset any downward revisions of the general fund revenue estimate prepared for fiscal years 2021 and 2022, after the enactment by the General Assembly of the 2021 Appropriation Act. If within five days of the preliminary close of the fiscal year ending on June 30, 2021, the Comptroller's analysis does not determine that a revenue re-forecast is required pursuant to § 2.2-1503.3, Code of Virginia, then such appropriation shall be used only for employee compensation purposes as stated in paragraphs JJ., KK., and LL. below.

2. Furthermore, \$139,781,368 provided in Item 145 of this act to support the state share of a three percent salary adjustment the second year for funded Standards of Quality instructional and support positions, Academic Year Governor's School Program instructional and support

positions, and Regional Alternative Education Program instructional and support positions shall be unallotted, if the Comptroller determines that general fund revenues did not meet or exceed the forecast for fiscal year 2021 and the actions authorized in paragraphs JJ., KK., and LL. are not effectuated.

- JJ.1. Contingent on the provisions of paragraph II.1. above, \$91,069,662 from the general fund the second year is provided to increase the base salary of the following employees by three percent on August 10, 2021:
- a. Full-time and other classified employees of the Executive Department subject to the Virginia Personnel Act;
- b. Full-time employees of the Executive Department not subject to the Virginia Personnel Act, except officials elected by popular vote;
- c. Any official whose salary is listed in § 4-6.01 of this act, subject to the ranges specified in the agency head salary levels in § 4-6.01 c;
- d. Full-time staff of the Governor's Office, the Lieutenant Governor's Office, the Attorney General's Office, Cabinet Secretaries' Offices, including the Deputy Secretaries, the Virginia Liaison Office, and the Secretary of the Commonwealth's Office;
- e. Heads of agencies in the Legislative Department;
- f. Full-time employees in the Legislative Department, other than officials elected by popular vote:
- g. Legislative Assistants as provided for in Item 1 of this act;
- h. Judges and Justices in the Judicial Department;
- i. Heads of agencies in the Judicial Department;
- j. Full-time employees in the Judicial Department;
- k. Commissioners of the State Corporation Commission and the Virginia Workers' Compensation Commission, the Chief Executive Officer of the Virginia College Savings Plan, and the Directors of the Virginia Lottery, and the Virginia Retirement System; and
- l. Full-time employees of the State Corporation Commission, the Virginia College Savings Plan, the Virginia Lottery, Virginia Workers' Compensation Commission, and the Virginia Retirement System.
- 2.a. Employees in the Executive Department subject to the Virginia Personnel Act shall receive the salary increases authorized in this paragraph only if they attained at least a rating of "Contributor" on their latest performance evaluation.
- b. Salary increases authorized in this paragraph for employees in the Judicial and Legislative Departments, employees of Independent agencies, and employees of the Executive Department not subject to the Virginia Personnel Act shall be consistent with the provisions of this

paragraph, as determined by the appointing or governing authority. However, notwithstanding anything herein to the contrary, the governing authorities of those state institutions of higher education with employees not subject to the Virginia Personnel Act may implement salary increases for such employees that may vary based on performance and other employment-related factors. The appointing or governing authority shall certify to the Department of Human Resource Management that employees receiving the awards are performing at levels at least comparable to the eligible employees as set out in subparagraph 2.a. of this paragraph.

- 3. The Department of Human Resource Management shall increase the minimum and maximum salary for each band within the Commonwealth's Classified Compensation Plan by three percent on August 10, 2021. No salary increase shall be granted to any employee as a result of this action. The department shall develop policies and procedures to be used in instances when employees fall below the entry level for a job classification due to poor performance. Movement through the revised pay band shall be based on employee performance.
- 4. The following agency heads, at their discretion, may utilize agency funds or the funds provided pursuant to this paragraph to implement the provisions of new or existing performance-based pay plans:
- a. The heads of agencies in the Legislative and Judicial Departments;
- b. The Commissioners of the State Corporation Commission and the Virginia Workers' Compensation Commission;
- c. The Attorney General;
- d. The Director of the Virginia Retirement System;
- e. The Director of the Virginia Lottery;
- f. The Director of the University of Virginia Medical Center;
- g. The Chief Executive Officer of the Virginia College Savings Plan; and
- h. The Executive Director of the Virginia Port Authority.
- 5. The base rates of pay, and related employee benefits, for wage employees may be increased by up to three percent no earlier than August 10, 2021. The cost of such increases for wage employees shall be borne by existing funds appropriated to each agency.
- 6. The governing authorities of those state institutions of higher education with employees may provide a salary adjustment based on performance and other employment-related factors, as long as the increases do not exceed the three percent increase, on average. No higher education administrators, executive officers, presidents, or faculty employees earning \$150,000 or more shall receive a pay adjustment provided for in this paragraph or otherwise provided for in this act.
- KK.1. Contingent on the provisions of paragraph II.1. above, the appropriations in this Item include funds to increase the base salary of the following employees by three percent on

September 1, 2021, provided that the governing authority of such employees use such funds to support salary increases for the following listed employees.

- a. Locally-elected constitutional officers;
- b. General Registrars and members of local electoral boards;
- c. Full-time employees of locally-elected constitutional officers and,
- d. Full-time employees of Community Services Boards, Centers for Independent Living, secure detention centers supported by Juvenile Block Grants, juvenile delinquency prevention and local court service units, local social services boards, local pretrial services act and Comprehensive Community Corrections Act employees, and local health departments where a memorandum of understanding exists with the Virginia Department of Health.
- 2. Out of the appropriation for Supplements to Employee Compensation is included \$24,125,830 the second year from the general fund to support the costs associated with the salary increase provided in this paragraph.
- LL. Contingent on the provisions of paragraph II.1. above, included in the appropriation for this item is \$2,871,689 the second year from the general fund to provide a three percent increase in base pay for adjunct faculty at Virginia two-year and four-year public colleges and higher education institutions, effective August 10, 2021."

Explanation:

(This amendment provides \$20.3 million GF the second year, above the \$97.8 million included in the Introduced Budget, to provide a three percent salary adjustment for state employees, including adjunct faculty at higher eduction institutions, effective August 10, 2021, for the September 1, 2021 paycheck, and state-supported local employees, effective September 1, 2021.)

		Item 477 #3s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	\$52,000	\$72,000	GF

Language:

Page 554, line 9, strike "(\$17,993,004)" and insert "(\$17,941,004)". Page 554, line 9, strike "\$138,332,991" and insert "\$138,404,991".

Explanation:

(This amendment provides \$52,000 GF the first year and \$72,000 GF the second year to cover estimated costs associated with SB 1275 of the 2021 General Assembly Session, adding EMS workers to § 65.1 402, related to the presumption as to death or disability from respiratory

disease, hypertension, or heart disease, cancer, contingent upon its final passage.)

		Item 477 #4s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	\$172,900	\$185,100	GF

Language:

Page 554, line 9, strike "(\$17,993,004)" and insert "(\$17,820,104)". Page 554, line 9, strike "\$138,332,991" and insert "\$138,518,091".

Explanation:

(This amendment provides \$172,900 GF the first year and \$185,100 GF the second year for the fiscal impact associated with SB 1375 of the 2021 General Assembly Session, which adds COVID-19 as a presumption under the Workers' Compensation Act for firefighters, emergency medical services personnel, law-enforcement, and correctional officers, contingent upon its final passage.)

		Item 479 #4s	
Central Appropriations	FY20-21	FY21-22	
Central Appropriations	(\$8,000,000)	\$7,000,000	GF

Language:

Page 567, line 23, strike "\$14,869,500" and insert "\$6,869,500".

Page 567, line 23, strike "\$44,188,052" and insert "\$51,188,052".

Page 571, strike line 41 through line 44 and renumber remaining paragraphs.

Page 572, strike line 16 through line 18 and insert:

"U. Out of this appropriation is included up to \$35,000,000 the second year from the general fund for the purpose of funding marijuana legalization and criminal record expungement legislation adopted during the 2020 Special Session I and the 2021 Regular Session and not otherwise explicitly funded in this act. Of these amounts, up to \$3,500,000 in the second year is provided for the initial operating costs of the Virginia Cannabis Control Authority, upon the determination of the Secretary of Finance with the advice and consent of the Director of the Office of Diversity, Equity and Inclusion."

Explanation:

(This amendment consolidates and accounts for marijuana legalization and criminal records expungement legislation adopted during the 2020 Special Session I and the 2021 Regular Session of the General Assembly. SB 1100, as introduced, provided \$36.0 million for these initiatives.)

Item 479 #5s

Central Appropriations

Central Appropriations

Language

Language:

Page 569, after line 25, insert:

"5. The funding contained in paragraph H.1. above shall remain unallotted until such time as the City of Richmond provides to the Department of General Services detailed planning information related to the planning and development of the Slavery and Freedom Heritage Site in Richmond, including Lumpkin's Pavilion and Slave Trail improvements. Upon the receipt of such planning information by the Department of General Services and subsequent approval by the Secretaries of Finance and Administration, the Department of Planning and Budget shall reallot the funds to be made available to the City of Richmond for expenses incurred in such planning and development."

Explanation:

(This amendment unallots \$10 million GF over the biennium until the City of Richmond provides approved plans regarding the planning and development of the Slavery and Freedom Heritage Site, including Lumpkin's Pavilion and Slave Trail improvements.)

Item 479.10 #1s

Central Appropriations

Central Appropriations

Language

Language:

Page 576, line 2, after "account.", strike "Utility customers may only receive a direct payment" Page 576, strike line 3.

Explanation:

(This amendment removes the restriction that utility customers may only receive assistance from the COVID-19 Utility Assistance Fund once.)

Item 479.10 #2s

Central Appropriations

FY20-21

FY21-22

Central Appropriations

\$25,000,000

\$0 NGF

Language:

Page 572, line 19, strike "\$0" and insert "\$25,000,000".

Page 578, line 11, after "C.", insert "1.".

Page 578, after line 26, insert:

"2. Out of the revenues deposited to the COVID-19 Relief Fund created pursuant to § 2.2-115.1 of the Code of Virginia, \$25,000,000 the first year is hereby appropriated for the Department of Small Business and Supplier Diversity for the purpose of awarding grants to small businesses through the Rebuild Virginia program. Eligible grant recipients shall meet the small business definition of § 2.2-1604 of the Code of Virginia.

Explanation:

(This amendment appropriates \$25.0 million NGF from the COVID-19 Relief Fund for the purpose of awarding grants to small businesses through the Rebuild Virginia program.)

Item 479.10 #3s

Central Appropriations

Central Appropriations

Language

Language:

Page 579, after line 7, insert:

"I.1. There is hereby created a Joint Subcommittee for Federal Relief Funds Oversight of the House Appropriations and Senate Finance and Appropriations Committees to review, provide advice to the Governor and provide oversight of federal funds provided to the Commonwealth in response to the Coronavirus Pandemic whereby discretion is provided to the state on how such funds shall be allocated, hereby known as "federal relief funds". The joint subcommittee shall consist of eight members, including the Chair of the House Appropriations Committee and three members of the House Appropriations Committee selected by the Chair, the Chair of the Senate Finance and Appropriations Committee and three members of the Senate Finance and Appropriations Committee selected by the Chair. The Secretaries of Finance, Commerce and Trade, Health and Human Resources, and Public Safety and Homeland Security shall also be available to provide technical assistance to the joint subcommittee. The joint subcommittee shall be staffed by staff from the House Appropriations and Senate Finance and Appropriations Committees.

2. The joint subcommittee shall work with the Governor to consider and examine all currently allocated and potential uses of federal relief funds in providing assistance to state agencies, institutions of higher education, local governments, private non-profit or for-profit businesses, and citizens of the Commonwealth that have been negatively impacted economically due to the effects of the Coronavirus Pandemic. The joint subcommittee shall make recommendations, consistent with the requirements of federal law and any federal regulations or guidance, to the Governor for his consideration. The Governor shall, within 10 days of the enactment of this Act, provide information to the joint subcommittee on previous allocations of federal relief funds,

expenditures to date, current unexpended allocations, and requests under consideration.

3. Prior to the allocation or appropriation of federal relief funds, the Governor shall submit his proposal or proposed allocation to the joint subcommittee, which shall review and make recommendations on such proposals to the Governor within 15 days. The Governor shall notify the Chairs of the Senate Finance and Appropriations Committee and the House Appropriations Committee in writing within five days concerning his final decision to appropriate and allocate federal relief funds having previously been considered by the joint subcommittee."

Explanation:

(This amendment establishes a Joint Subcommittee for Federal Relief Funds Oversight to allow the General Assembly to provide oversight of the substantial federal relief funds the federal government has been providing to the state in response to the Coronavirus Pandemic. The Joint Subcommittee would review proposed allocations of funding and make appropriate recommendations to the Governor, however the Governor would make the final decision on how the funds should be allocated. The Joint Subcommittee provides an opportunity for the General Assembly to provide input into the process for allocating federal relief funds and to provide oversight of how such funding is expended.)

Item 479.10 #4s

Central Appropriations

Central Appropriations

Language

Language:

Page 573, line 49, strike "\$55,640,872", insert "\$17,036,718".

Page 573, line 50, strike "\$9,256,178", insert "\$2,470,552".

Page 573, line 52, strike "\$73,056,734", insert "\$103,500,000".

Page 574, line 1, strike "\$25,000,000", insert "\$17,467,766".

Page 574, line 40, strike "\$22,052,445", insert "\$34,234,679"

Page 574, line 45, strike "\$3,450,000", insert "\$10,550,000".

Page 574, line 3, strike "\$116,261,410" and insert "\$133,511,410".

Page 574, line 50, strike "\$30,000,000" and insert "\$12,750,000".

Page 575, line 10, strike "\$2,127,357,769", insert "\$2,121,161,255".

Page 575, line 5, strike "\$15,000,000", insert "\$12,000,000".

Page 577, line 15, strike "\$73,056,734", insert "\$103,500,000".

Page 577, line 22, strike "\$25,000,000", insert "\$17,467,766".

Explanation:

(This amendment modifies Coronavirus Relief Fund allocations.)

Central Appropriations

Central Appropriations

Language

Language:

Page 579, after line 7, insert:

"1. If within five days of preliminary close of the fiscal year ending June 30, 2021, the Comptroller determines that general fund revenues for fiscal year 2021 exceeded the official forecast, then it is the intent of the General Assembly that the Governor shall allocate such additional revenue to fund the following items in priority order in fiscal year 2022:

Agency - Title	FY 2022
Direct Aid - Fund One Additional SOQ Support Position	\$47,236,642
Department of Medical Assistance Services - Increase DD Waiver Provider Rates to Reflect Minimum Wage Impact	\$20,000,000
Department of Accounts Transfer Payments - Virginia Retirement System Payment to Increase Actuarial Funding Status	\$200,000,000
Total	\$267,236,642

- 2. The Director, Department of Planning and Budget, shall have authority to adjust the general fund appropriations in the second year in the appropriate agencies to reflect items funded as a result of paragraph 1.
- 3. Should the additional revenue pursuant to paragraph 1 be insufficient to fund all the items listed, then the Governor shall fund those items listed in priority order for which the additional revenue is sufficient to fully fund."

Explanation:

(This amendment provides that any additional revenue for fiscal year 2021 as a result of the higher than forecast general fund revenues in the fiscal year will be used for the listed high priority funding items in fiscal year 2022.)

Item 481 #1s

Central Appropriations

Central Appropriations

Language

Language:

Page 580, line 1, after "Code of Virginia," insert "for the 2020-22 biennium".

Page 580, line 2, after "MOU." insert "The designated reviewers shall propose any needed

technical adjustments for consideration during the 2022 Session."

Explanation:

(This amendment clarifies proposed technical language regarding eligibility for "Tech Talent" grants pursuant to the approved Memorandums of Understanding.)

Item C-0 #1s

General Conditions

General Conditions Language

Language:

Page 609, after line 15, insert:

"9. Notwithstanding any other provision of law, a public institution of higher education may participate in the United States Department of Education Historically Black College and University Capital Financing Program (HBCU Program), and use federal grant and contract funds as permitted by the Program."

Explanation:

(This amendment is intended to ensure that Virginia's public historically black colleges and universities (HBCU) may access the United States Department of Education Historically Black College and University Capital Financing Program (HBCU Program). The HBCU Program requires that institutions be able to use revenue from federal grants and contracts via administrative offset if the pledged revenues are insufficient to pay debt service. The HBCU Program provides HBCUs with low-cost capital to finance infrastructure improvements, facilitating the repair, renovation, and construction of classrooms, libraries, laboratories, dormitories, instructional equipment, and research instrumentation.)

			Item C-17.10 #1s	
Capital Projects		FY20-21	FY21-22	
James Madison	University	\$0	\$6,579,237	NGF
Language:				
Page 618, after line	22, insert:			
"C-17.10 Improvements: Phase I	East Campus Steam Plant,	\$0	\$6,579,237	
Fund Sources:	Higher Education Operating	\$0	\$1,973,771	
	Bond Proceeds	\$0	\$4,605,466 "	

Explanation:

(This amendment provides \$4.6 million in bond proceeds for Phase 1 of improvements and upgrades to the East Campus Steam Plant project at James Madison University. Total project cost for this phase of work is \$6.6 million and includes almost \$2.0 million supported with auxiliary nongeneral funds.)

Item C-22.10 #1s

Education

Virginia Commonwealth University

Language

Language:

Page 619, strike lines 37 through 39 and insert:

"The provisions of Item C-13.10 A., B., C.2., C.3., and D., Chapter 854, 2019 Acts of Assembly, as it relates to the Virginia Commonwealth University acquisition of the Virginia Alcoholic Beverage Control Authority property are hereby extended for the 2020-22 Biennium. The provisions of Item C-13.10 C.1. shall be replaced as follows:

C. 1. The Authority shall offer for sale the Property to VCU at the fair market value determined by the lower of the two appraisals received pursuant to paragraph B. Any other conditions of the transfer, as negotiated between the Authority and VCU, shall be based on usual and customary terms for such intergovernmental transfers. If the Authority and VCU cannot agree on the terms, other than price as price shall be the fair market value described herein, of the sale of the Property, the Authority may sell the property pursuant to § 2.2-1156 and on such terms as it determines are in the best interest of the Authority, however any sale terms negotiated shall be presented to VCU, prior to any sale, to enable VCU to make a counter offer to the Authority. Should such counter offer be received by the Authority, the Authority shall consider the VCU counter offer and if determined favorable by the Authority, may sell the property to VCU.".

Explanation:

(This amendment specifies additional fair market value parameters regarding Virginia Commonwealth University acquisition of the Virginia Alcoholic Beverage Control Authority property on Hermitage Road in Richmond.)

		Item C-23.5 #	1s
Education	FY20-21	FY21-22	
Virginia Community College System	\$500,000	\$0	GF

Language:

Page 620, after line 14, insert:

"C-23.5 Planning: Amho Central Virginia Commo		\$500,000	\$0
Fund Sources:	General	\$500,000	\$0 "

Explanation:

(This amendment provides \$500,000 GF for planning for renovations of Amherst and Campbell Halls on Central Virginia Community College's campus. Amherst (54,042 Square Feet) and Campbell (19,550 square feet) Halls comprise the core classroom space for Central Virginia Community College. Amherst Hall was built in 1968 and Campbell Hall was completed in 1974 and neither have had a major renovation.)

		Item C-36.30 #	[‡] 1s
Education	FY20-21	FY21-22	
Frontier Culture Museum of Virginia	\$1,300,000	\$0	GF

Language:

Page 622, line 7, strike "\$0" and insert "\$1,300,000".
Page 622, line 12, strike the first instance of "\$0" and insert "\$1,300,000".

Explanation:

(This amendment provides an additional \$1.3 million GF the first year for the project named "Improvements: Construct Crossing Gallery (18316)". Detailed planning for the project was initially funded in the 2019 Session by Chapter 854, Item C-21-10. The additional funding is requested to support the additional services required of the project architect/engineer to complete the required revisions and complete the project planning through the preliminary design and cost estimate phase.)

			Item C-36.45 #1s	
Education		FY20-21	FY21-22	
The Science	Museum of Virginia	\$0	\$7,506,000	NGF
Language:				
Page 622, after li	ne 16, insert:			
"C-36.45 Improvements: Community Green Space		\$0	\$7,506,000	
Fund Sources:	Special	\$0	\$7,506,000 "	

Explanation:

(This amendment authorizes the Science Museum of Virginia to seek private funding of up to \$7.5 million for a community green space project. Situated on the Pulse corridor and a key element of the Richmond 300 Plan, this project will transform the area in front of the Museum into a vibrant and active community resource to serve the needs of a diverse, growing citizenry.)

Item C-40 #1s

Natural Resources

Department of Conservation and Recreation

Language

Language:

Page 623, line 23, after "Staunton River," insert "Kiptopeke,"

Explanation:

(This amendment updates the list for acquisitions of land for state parks for in-holdings and contiguous properties, consistent with funding included in the budget as introduced, to include Kiptopeke.)

	Item C-64		
Central Appropriations	FY20-21	FY21-22	
Central Capital Outlay	\$0 \$0	\$37,000,000 (\$37,000,000)	GF NGF

Language:

Page 628, line 26, strike "\$137,000,000" and insert "\$100,000,000".

Page 628, line 27, after "A." insert "1."

Page 628, line 27, strike the second instance of "\$137,000,000" and insert "\$100,000,000".

Page 628. after line 26, insert "Fund Sources: General \$0 \$37,000,000".

Page 628, after line 30, insert:

"2. Out of this appropriation, \$37,000,000 the second year from the general fund is designated for capital costs of maintenance reserve projects.".

Explanation:

(This amendment changes the source of funding of a portion of the \$137.0 million provided in the second year of the biennium for maintenance reserve from tax-supported debt to general fund cash.)

Item C-66 #1s

Central Appropriations	FY20-21	FY21-22
Central Capital Outlay	\$8,450,000	\$0 GF

Language:

Page 632, line 48, strike "\$1,517,750" and insert "\$9,967,750".

Page 633, line 4, after "B." insert "1.".

Page 633, after line 9, insert "Replace Division 6 Headquarters".

Page 633, after line 36, insert:

"2. Out of this appropriation, \$4,500,000 the first year from the general fund is designated for planning the Fine and Performing Arts Center project at the University of Mary Washington, \$2,100,000 the first year from the general fund is designated for planning the Center for Leadership & Ethics Facility, Phase 2 project at Virginia Military Institute, and \$1,850,000 the first year from the general fund is designated for planning the Division 6 Headquarters project for the State Police."

Explanation:

(This amendment provides general funds for the planning the following projects that were included in the planning pool in the 2020 regular session: \$4.5 million for the University of Mary Washington for detailed planning for the construction of a Fine and Performing Arts Center; and \$2.1 million for Virginia Military Institute for planning for the Center for Leadership & Ethics Facility, Phase 2 project; and also \$1.85 million for planning the State Police Division 6 Headquarters replacement project.)

Item C-66.10 #1s

Central Appropriations	FY20-21	FY21-22	
Central Capital Outlay	\$0	(\$58,500,000)	NGF

Language:

Page 634, line 6, strike "\$58,500,000" and insert "\$0". Page 634, strike lines 5 through 21.

Explanation:

(This amendment maintains the scope for the replacement of Central State Hospital at 252 beds, reversing the proposed \$58.5 million increase in debt authorization for the expansion of an additional 48 beds.)

Item C-72 #1s

Central Appropriations

Central Capital Outlay

Language

Language:

Page 640, line 46, after "a capital project" insert "or projects".

Page 640, line 48, strike "and buildings".

Page 640, strike line 51 and insert "Authority each approve a public private partnership or other contractual agreement with respect to such a capital project or projects. The"

Explanation:

(This amendment would allow the bond proceeds authorized by this Item to be used for improvements that support multiple projects and projects performed as a P3 or a direct lease agreement with the Virginia Port Authority. Use of the funds will still require approval of both the Virginia Port Authority Board of Commissioners and the Major Employment & Investment Project Approval Commission, while providing the flexibility needed by the Virginia Port Authority and the Virginia Economic Development Partnership to attract multiple users and make the Portsmouth Marine Terminal an offshore wind industry hub.)

Item C-72.70 #1s

Capital Projects

Department of State Police

Language

Language:

Page 641, after line 14, insert:

"The title of the project for the Department of the State Police previously authorized in Item C-45, Chapter 2, 2018 Acts of Assembly, Special Session I, as "Construct Area 13 Barracks" is hereby changed to "Acquire, Renovate or Construct Area 13 Barracks".

Explanation:

(This amendment makes a title change to an existing project.)

Item C-76 #1s

Central Appropriations

Central Capital Outlay

Language

Language:

Page 644, after line 33, insert:

"James Madison University (216)

East Campus Steam Plant, Phase 1 XXXXX C-17.10 \$4,605,466".

Page 645, line 18, strike "\$100,007,208" and insert "\$104,612,674".

Explanation:

(This amendment updates the Virginia College Building Authority bond table for the Committee's amendments.)

Item 3-1.01 #1s

Transfers

Interfund Transfers

Language

Language:

Page 650, line 6, strike "\$139,500,745" and insert "\$159,500,745". Page 650, line 7, strike "\$132,700,000" and insert "\$134,400,000".

Explanation:

(This amendment increases the estimated transfer to the general fund of net profits from ABC gross store sales by \$20.0 million the first year and \$1.7 million the second year, based upon higher assumed sales and profits in the current fiscal year.)

Item 3-2.03 #1s

Working Capital Funds and Lines of Credit

Lines of Credit

Language

Language:

Page 657, strike lines 6 through 7.

Explanation:

(This amendment eliminates a \$1.0 million line of credit proposed in the introduced budget for Virginia Industries for the Blind.)

Item 3-5.15 #1s

Adjustments and Modifications to Tax Collections

Provider Coverage Assessment

Language

Language:

Page 661, line 29, strike "1.08" and insert "1.02".

Explanation:

(This amendment modifies the coverage assessment amount collected from hospitals to cover the state share of Medicaid expansion costs. Current law requires the collection of 108 percent of the state share of the full costs of expanded Medicaid coverage. The original intent was to ensure that collections were sufficient to cover the full cost of expansion and not rely on any general fund dollars to expand Medicaid. Rate assessment language has since been amended to allow the Department of Medical Assistance Services (DMAS) to adjust assessments with each quarter to account for differences between estimates and actual spending, however, the 108 percent collection requirement is applied to the estimate for the entire year and not just the remaining quarters. In addition, DMAS now has more experience with expansion and hospital overpayments in the first two years of the program have far exceeded the additional eight percent add-on. Reducing the add-on from eight to two percent would reduce the amount of overpayment made each year by the hospitals into the Health Care Coverage Assessment Fund.)

Item 3-5.16 #1s

Adjustments and Modifications to Tax Collections

Provider Payment Rate Assessment

Language

Language:

Page 662, line 51, after "equivalent to", strike the remainder of the line.

Page 662, line 52, before "subject to CMS", insert:

"the maximum managed care directed payment amount as allowed by CMS,".

Explanation:

(This amendment modifies language related to the hospital provider payment rate assessment to change the methodology to enable the hospitals to obtain additional federal dollars with no additional cost to the Commonwealth. Current budget language limits supplemental payments by the upper payment limit gap for fee-for-service claims and extrapolates that gap to managed care claims. Because over 90 percent of Medicaid enrollees are in managed care, this gap can be highly variable on an annual basis and can limit total rate enhancement. The federal Centers for Medicare and Medicaid Services (CMS) has provided other states the flexibility to structure managed care directed payments as a percentage of Medicare. However, the introduced budget language is highly prescriptive and limits the department's ability to explore and apply an alternate methodology which could potentially stabilize or increase total supplemental payments.)

Item 3-5.23 #1s

Finance

Language

Language:

Page 664, after line 37, insert:
"§ 3-5.23 CORPORATE INCOME TAX INFORMATIONAL REPORTING

- A.1. Corporations that are members of a unitary business must file a report, in a manner prescribed by the Tax Commissioner, for the unitary combined group containing the unitary combined net income of such group. The report shall be based on taxable year 2019 computations and include, at a minimum the difference in tax owed as a result of filing a unitary combined report, computed according to the method or methods specified by the Tax Commissioner, compared to the tax owed under the current filing requirements.
- 2. "Unitary business" means a single economic enterprise made up either of separate parts of a single business entity or of a commonly controlled group of business entities that are sufficiently interdependent, integrated, and interrelated through their activities so as to provide a synergy and mutual benefit that produces a sharing or exchange of value among them and a significant flow of value to the separate parts. A "unitary business" includes that part of the business that meets the definition in this section and is conducted by a taxpayer through the taxpayer's interest in a partnership, whether the interest in that partnership is held directly or indirectly through a series of partnerships or other pass-through entities. A "unitary business" shall not include persons subject to, or that would be subject to if doing business in the Commonwealth, the insurance premiums license tax under Chapter 25 (§ 58.1-2500 et seq.), Code of Virginia, or the bank franchise tax under Chapter 12 (§ 58.1-1200 et seq.).
- 3. The report must be submitted to the Department of Taxation on or before June 1, 2021, which date shall not be extended.
- 4. Members of a unitary combined group shall exclude as a member and disregard the income and apportionment factors of any corporation incorporated in a foreign jurisdiction (a "foreign corporation") if the average of its property, payroll and sales factors outside the United States is eighty percent (80%) or more. If a foreign corporation is includible as a member in the unitary combined group, to the extent that such foreign corporation's income is subject to the provisions of a federal income tax treaty, such income is not includible in the unitary combined group net income. Such member shall also not include in the unitary combined report any expenses or apportionment factors attributable to income that is subject to the provisions of a federal income tax treaty. For purposes of this paragraph, "federal income tax treaty" means a comprehensive income tax treaty between the United States and a foreign jurisdiction, other than a foreign

jurisdiction which the organization for economic co-operation and development has determined has not committed to the internationally agreed tax standard, or has committed to the international agreed tax standard but has not yet substantially implemented that standard, as identified in the then-current organization for economic co-operation and development progress report.

B. Any corporation required to submit such report to the Department of Taxation that fails to do so on or before June 1, 2021, or that makes a material omission or misstatement in connection with such report shall be subject to a penalty of \$10,000. The Tax Commissioner shall have the authority to waive such penalty upon a determination that the requirement would cause an undue hardship. All requests for waiver shall be transmitted to the Tax Commissioner in writing.

C. The Tax Commissioner shall on or before December 1, 2021, based on the information provided in income tax returns and the data submitted under this section, submit a report to the Chair of the Senate Finance and Appropriations Committee, the Chair of the House Appropriations Committee, and the Chair of the House Finance Committee."

Explanation:

(This amendment requires corporations that are members of a unitary business to file an informational report with the Department of Taxation for the unitary combined group containing the unitary combined net income of the group. The report must be based on taxable year 2019 computations and include, at a minimum the difference in tax owed as a result of filing a unitary combined report compared to the tax owed under the current filing requirements. Any corporation required to submit such report that fails to do so on or before June 1, 2021, or that makes a material omission or misstatement in connection with such report is subject to a penalty of \$10,000, unless waived by the Tax Commissioner upon a determination that the requirement would cause an undue hardship.)

Item 4-5.10 #1s

Special Conditions and Restrictions on Expenditures

Surplus Property Transfers for Economic Development

Language

Language:

Page 698, line 1, strike "2020" and insert "2021".

Explanation:

(This amendment extends by one year the deadline for a property sale between the Commonwealth of Virginia and the Eastern Virginia Regional Industrial Facility Authority.)

Item 4-14 #1s

Effective Date

Effective Date Language

Language:

Page 728, strike lines 56 through 60 and insert:

"3. If rent is unpaid when due, or if a payment under the terms of a payment plan is unpaid when due, the landlord shall, pursuant to § 55.1-1202, Code of Virginia, serve a written notice on the tenant that informs the tenant of the Virginia Rent and Mortgage Relief Program and provides the website address and statewide telephone number for that program. The written notice shall also provide information on how to reach 2-1-1 Virginia to determine whether there are any other available federal, state and local rent relief programs. The written notice shall also inform the tenant that the landlord shall apply for rental assistance on the tenant's behalf within 14 days of serving the notice on the tenant, unless the tenant pays in full, enters into a payment plan or informs the landlord that they have already applied for rental assistance. The landlord shall apply for rental assistance on behalf of the tenant no later than 14 days after serving the written notice on the tenant, unless they receive the full amount owed by the tenant or confirmation from the tenant that the tenant has applied for rental assistance before the 14th day, or they have entered into a payment plan with the tenant. If the tenant has applied for rental assistance, the landlord shall cooperate with the tenant's application, by providing all information and documentation required to complete the application, including but not limited to the W-9 IRS form and any supporting affidavits. If a tenant who has not paid in full or entered into a payment plan with the landlord within 14 days after the written notice is served refuses to apply for rental assistance and also refuses to cooperate with the landlord in providing information and documentation required to complete the application made by the landlord, or if such tenant is determined ineligible for rental assistance, or there are no longer funds available through any federal, state or local rental assistance program, the landlord may take action to obtain possession of the tenant's dwelling unit as provided in § 55.1-1251, Code of Virginia."

Page 729, strike lines 1 through 20.

Page 729, strike lines 28 through 30 and insert:

"f. Nothing in this section shall void any judgment for possession validly obtained by a landlord prior to November 18, 2020; however, a landlord shall not initiate, maintain, or advance any legal process to obtain possession of a dwelling unit for non-payment of the rent unless the landlord complies with the provisions of this Section 8."

Explanation:

(This amendment updates the language included in Chapter 56, 2020 Special Session I, Acts of Assembly to clarify the requirements for landlords and tenants to apply for rental assistance

prior to any evictions-related action occurring. It also removes the 45-day deadline for approval of a pending application for rental assistance.)

Item 4-14 #2s

Effective Date

Effective Date Language

Language:

Page 727, after line 34, insert:

- "k. In addition to the relief provided pursuant to clause 7.j., within 60 days after the enactment of this act, a Phase II Utility shall forgive all such utility's jurisdictional customer balances more than 30 days in arrears as of December 31, 2020.
- 1. In the utility's 2021 triennial review, the provisions of clause 7.k. shall be excluded from the utility's cost of service for purposes of determining any test period earnings and determining any future rates of the utility. In determining any customer bill credits, in the utility's 2021 triennial review, the Commission shall first offset any amounts pursuant to clause 7.k. against the total earnings for the 2017 through 2020 test periods that are determined to be above the utility's authorized earnings band. Such offset shall be made prior to any offset to customer bill credits by customer credit reinvestment offsets.
- 2. Each Phase II Utility shall, no later than November 1, 2021, submit a report to the Governor, the Chairs of the House Committees on Labor and Commerce and Appropriations, and the Senate Committees on Commerce and Labor and Finance and Appropriations, and the Chair of the Commission on Electric Utility Regulation, detailing all actions by it pursuant to this act to forgive customer balances."

Explanation:

(This amendment requires that Dominion Energy write off debt owed as of December 31, 2020.)

Item 4-14 #3s

Effective Date

Language

Language:

Page 735, strike lines 50 through 51 and insert:

11. That §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402 of the Code of Virginia are amended and reenacted as follows:

§ 58.1-301. Conformity to Internal Revenue Code.

- A. Any term used in this chapter shall have the same meaning as when used in a comparable context in the laws of the United States relating to federal income taxes, unless a different meaning is clearly required.
- B. Any reference in this chapter to the laws of the United States relating to federal income taxes shall mean the provisions of the Internal Revenue Code of 1954, and amendments thereto, and other provisions of the laws of the United States relating to federal income taxes, as they existed on December 31, 2019 2020, except for:
- 1. The special depreciation allowance for certain property provided for under §§ 168(k), 168(l), 168(m), 1400L, and 1400N of the Internal Revenue Code;
- 2. The carry-back of certain net operating losses for five years under § 172(b)(1)(H) of the Internal Revenue Code;
- 3. The original issue discount on applicable high yield discount obligations under § 163(e)(5)(F) of the Internal Revenue Code:
- 4. The deferral of certain income under § 108(i) of the Internal Revenue Code. For Virginia income tax purposes, income from the discharge of indebtedness in connection with the reacquisition of an "applicable debt instrument" (as defined under § 108(i) of the Internal Revenue Code) reacquired in the taxable year shall be fully included in the taxpayer's Virginia taxable income for the taxable year, unless the taxpayer elects to include such income in the taxpayer's Virginia taxable income ratably over a three-taxable-year period beginning with taxable year 2009 for transactions completed in taxable year 2009, or over a three-taxable-year period beginning with taxable year 2010 for transactions completed in taxable year 2010 on or before April 21, 2010. For purposes of such election, all other provisions of § 108(i) of the Internal Revenue Code shall apply mutatis mutandis. No other deferral shall be allowed for income from the discharge of indebtedness in connection with the reacquisition of an "applicable debt instrument";
- 5. For taxable years beginning on and after January 1, 2019, the provisions of § 11046 of the federal Tax Cuts and Jobs Act, P.L. 115-97 (2017), related to the suspension of the overall limitation on itemized deductions under § 68(f) of the Internal Revenue Code; and
- 6. The provisions of § 103 of Division Q of the federal Further Consolidated Appropriations Act, 2020, P.L. 116-94 (2019), related to the reduction in the medical expense deduction floor For taxable years beginning on and after January 1, 2017, but before January 1, 2018, and for taxable years beginning on and after January 1, 2019, the 7.5 percent of federal adjusted gross income threshold set forth in § 213(a) of the Internal Revenue Code that is used for purposes of computing the deduction allowed for expenses for medical care pursuant to § 213 of the Internal Revenue Code. For such taxable years, the threshold utilized for Virginia income tax purposes to compute the deduction allowed for expenses for medical care pursuant to § 213 of the Internal Revenue Code shall be 10 percent of federal adjusted gross income;
- 7. The provisions of §§ 2303(a) and 2303(b) of the federal Coronavirus Aid, Relief, and

Economic Security Act, P.L. 116-136 (2020), related to the net operating loss limitation and carryback;

- 8. The provisions of § 2304(a) of the federal Coronavirus Aid, Relief, and Economic Security Act, P.L. 116-136 (2020), related to a loss limitation applicable to taxpayers other than corporations;
- 9. The provisions of § 2306 of the federal Coronavirus Aid, Relief, and Economic Security Act, P.L. 116-136 (2020), related to the limitation on business interest; and
- 10. The provisions of §§ 276(a), 276(b)(2), 276(b)(3), 278(a)(2), 278(a)(3), 278(b)(2), 278(b)(3), 278(c)(2), 278(c)(3), 278(d)(2), and 278(d)(3) of the federal Consolidated Appropriations Act, P.L. 116-260 (2020), related to deductions, tax attributes, and basis increases for certain loan forgiveness and other business financial assistance.

The Department of Taxation is hereby authorized to develop procedures or guidelines for implementation of the provisions of this section, which procedures or guidelines shall be exempt from the provisions of the Administrative Process Act (§ 2.2-4000 et seq.).

§ 58.1-322.02. Virginia taxable income; subtractions.

In computing Virginia taxable income pursuant to § 58.1-322, to the extent included in federal adjusted gross income, there shall be subtracted:

- 1. Income derived from obligations, or on the sale or exchange of obligations, of the United States and on obligations or securities of any authority, commission, or instrumentality of the United States to the extent exempt from state income taxes under the laws of the United States, including, but not limited to, stocks, bonds, treasury bills, and treasury notes but not including interest on refunds of federal taxes, interest on equipment purchase contracts, or interest on other normal business transactions.
- 2. Income derived from obligations, or on the sale or exchange of obligations, of the Commonwealth or of any political subdivision or instrumentality of the Commonwealth.
- 3. Benefits received under Title II of the Social Security Act and other benefits subject to federal income taxation solely pursuant to § 86 of the Internal Revenue Code.
- 4. Up to \$20,000 of disability income, as defined in § 22(c)(2)(B)(iii) of the Internal Revenue Code; however, any person who claims a deduction under subdivision 5 of § 58.1-322.03 may not also claim a subtraction under this subdivision.
- 5. The amount of any refund or credit for overpayment of income taxes imposed by the Commonwealth or any other taxing jurisdiction.
- 6. The amount of wages or salaries eligible for the federal Work Opportunity Credit which was not deducted for federal purposes on account of the provisions of § 280C(a) of the Internal Revenue Code.
- 7. Any amount included therein less than \$600 from a prize awarded by the Virginia Lottery.
- 8. The wages or salaries received by any person for active and inactive service in the National Guard of the Commonwealth of Virginia, not to exceed the amount of income derived from 39 calendar days of such service or \$3,000, whichever amount is less; however, only those persons in the ranks of O3 and below shall be entitled to the deductions specified in this subdivision.
- 9. Amounts received by an individual, not to exceed \$1,000 for taxable years beginning on or

before December 31, 2019, and \$5,000 for taxable years beginning on or after January 1, 2020, as a reward for information provided to a law-enforcement official or agency, or to a nonprofit corporation created exclusively to assist such law-enforcement official or agency, in the apprehension and conviction of perpetrators of crimes. This subdivision shall not apply to the following: an individual who is an employee of, or under contract with, a law-enforcement agency, a victim or the perpetrator of the crime for which the reward was paid, or any person who is compensated for the investigation of crimes or accidents.

- 10. The amount of "qualified research expenses" or "basic research expenses" eligible for deduction for federal purposes, but which were not deducted, on account of the provisions of § 280C(c) of the Internal Revenue Code and which shall be available to partners, shareholders of S corporations, and members of limited liability companies to the extent and in the same manner as other deductions may pass through to such partners, shareholders, and members. 11. Any income received during the taxable year derived from a qualified pension, profit-sharing, or stock bonus plan as described by § 401 of the Internal Revenue Code, an individual retirement account or annuity established under § 408 of the Internal Revenue Code, a deferred compensation plan as defined by § 457 of the Internal Revenue Code, or any federal government retirement program, the contributions to which were deductible from the taxpayer's federal adjusted gross income, but only to the extent the contributions to such plan or program were subject to taxation under the income tax in another state.
- 12. Any income attributable to a distribution of benefits or a refund from a prepaid tuition contract or savings trust account with the Virginia College Savings Plan, created pursuant to Chapter 7 (§ 23.1-700 et seq.) of Title 23.1. The subtraction for any income attributable to a refund shall be limited to income attributable to a refund in the event of a beneficiary's death, disability, or receipt of a scholarship.
- 13. All military pay and allowances, to the extent included in federal adjusted gross income and not otherwise subtracted, deducted, or exempted under this section, earned by military personnel while serving by order of the President of the United States with the consent of Congress in a combat zone or qualified hazardous duty area that is treated as a combat zone for federal tax purposes pursuant to § 112 of the Internal Revenue Code.
- 14. For taxable years beginning before January 1, 2015, the gain derived from the sale or exchange of real property or the sale or exchange of an easement to real property which results in the real property or the easement thereto being devoted to open-space use, as that term is defined in § 58.1-3230, for a period of time not less than 30 years. To the extent that a subtraction is taken in accordance with this subdivision, no tax credit under this chapter for donating land for its preservation shall be allowed for three years following the year in which the subtraction is taken.
- 15. Fifteen thousand dollars of military basic pay for military service personnel on extended active duty for periods in excess of 90 days; however, the subtraction amount shall be reduced dollar-for-dollar by the amount by which the taxpayer's military basic pay exceeds \$15,000 and shall be reduced to zero if such military basic pay amount is equal to or exceeds \$30,000.
- 16. The first \$15,000 of salary for each federal and state employee whose total annual salary from all employment for the taxable year is \$15,000 or less.
- 17. Unemployment benefits taxable pursuant to § 85 of the Internal Revenue Code.

18. Any amount received as military retirement income by an individual awarded the Congressional Medal of Honor.

19. Items of income attributable to, derived from, or in any way related to (i) assets stolen from, hidden from, or otherwise lost by an individual who was a victim or target of Nazi persecution or (ii) damages, reparations, or other consideration received by a victim or target of Nazi persecution to compensate such individual for performing labor against his will under the threat of death, during World War II and its prelude and direct aftermath. This subtraction shall not apply to assets acquired with such items of income or with the proceeds from the sale of assets stolen from, hidden from, or otherwise lost to, during World War II and its prelude and direct aftermath, a victim or target of Nazi persecution. The provisions of this subdivision shall only apply to an individual who was the first recipient of such items of income and who was a victim or target of Nazi persecution, or a spouse, surviving spouse, or child or stepchild of such victim. As used in this subdivision:

"Nazi regime" means the country of Nazi Germany, areas occupied by Nazi Germany, those European countries allied with Nazi Germany, or any other neutral European country or area in Europe under the influence or threat of Nazi invasion.

"Victim or target of Nazi persecution" means any individual persecuted or targeted for persecution by the Nazi regime who had assets stolen from, hidden from, or otherwise lost as a result of any act or omission in any way relating to (i) the Holocaust, (ii) World War II and its prelude and direct aftermath, (iii) transactions with or actions of the Nazi regime, (iv) treatment of refugees fleeing Nazi persecution, or (v) the holding of such assets by entities or persons in the Swiss Confederation during World War II and its prelude and aftermath. A "victim or target of Nazi persecution" also includes any individual forced into labor against his will, under the threat of death, during World War II and its prelude and direct aftermath.

- 20. The military death gratuity payment made after September 11, 2001, to the survivor of deceased military personnel killed in the line of duty, pursuant to 10 U.S.C. Chapter 75; however, the subtraction amount shall be reduced dollar-for-dollar by the amount that the survivor may exclude from his federal gross income in accordance with § 134 of the Internal Revenue Code.
- 21. The death benefit payments from an annuity contract that are received by a beneficiary of such contract, provided that (i) the death benefit payment is made pursuant to an annuity contract with an insurance company and (ii) the death benefit payment is paid solely by lump sum. The subtraction under this subdivision shall be allowed only for that portion of the death benefit payment that is included in federal adjusted gross income.
- 22. Any gain recognized from the sale of launch services to space flight participants, as defined in 49 U.S.C. § 70102, or launch services intended to provide individuals with the training or experience of a launch, without performing an actual launch. To qualify for a deduction under this subdivision, launch services must be performed in Virginia or originate from an airport or spaceport in Virginia.
- 23. Any gain recognized as a result of resupply services contracts for delivering payload, as defined in 49 U.S.C. § 70102, entered into with the Commercial Orbital Transportation Services division of the National Aeronautics and Space Administration or other space flight entity, as defined in § 8.01-227.8, and launched from an airport or spaceport in Virginia.

24. Any income taxed as a long-term capital gain for federal income tax purposes, or any income taxed as investment services partnership interest income (otherwise known as investment partnership carried interest income) for federal income tax purposes. To qualify for a subtraction under this subdivision, such income shall be attributable to an investment in a "qualified business," as defined in § 58.1-339.4, or in any other technology business approved by the Secretary of Administration, provided that the business has its principal office or facility in the Commonwealth and less than \$3 million in annual revenues in the fiscal year prior to the investment. To qualify for a subtraction under this subdivision, the investment shall be made between the dates of April 1, 2010, and June 30, 2020. No taxpayer who has claimed a tax credit for an investment in a "qualified business" under § 58.1-339.4 shall be eligible for the subtraction under this subdivision for an investment in the same business.

25. For taxable years beginning on and after January 1, 2014, any income of an account holder for the taxable year taxed as (i) a capital gain for federal income tax purposes attributable to such person's first-time home buyer savings account established pursuant to Chapter 12 (§ 36-171 et seq.) of Title 36 and (ii) interest income or other income for federal income tax purposes attributable to such person's first-time home buyer savings account.

Notwithstanding the statute of limitations on assessments contained in § 58.1-312, any subtraction taken under this subdivision shall be subject to recapture in the taxable year or years in which moneys or funds withdrawn from the first-time home buyer savings account were used for any purpose other than the payment of eligible costs by or on behalf of a qualified beneficiary, as provided under § 36-174. The amount subject to recapture shall be a portion of the amount withdrawn in the taxable year that was used for other than the payment of eligible costs, computed by multiplying the amount withdrawn and used for other than the payment of eligible costs by the ratio of the aggregate earnings in the account at the time of the withdrawal to the total balance in the account at such time.

However, recapture shall not apply to the extent of moneys or funds withdrawn that were (i) withdrawn by reason of the qualified beneficiary's death or disability; (ii) a disbursement of assets of the account pursuant to a filing for protection under the United States Bankruptcy Code, 11 U.S.C. §§ 101 through 1330; or (iii) transferred from an account established pursuant to Chapter 12 (§ 36-171 et seq.) of Title 36 into another account established pursuant to such chapter for the benefit of another qualified beneficiary.

For purposes of this subdivision, "account holder," "eligible costs," "first-time home buyer savings account," and "qualified beneficiary" mean the same as those terms are defined in § 36-171.

26. For taxable years beginning on and after January 1, 2015, any income for the taxable year attributable to the discharge of a student loan solely by reason of the student's death. For purposes of this subdivision, "student loan" means the same as that term is defined under § 108(f) of the Internal Revenue Code.

27. a. Income, including investment services partnership interest income (otherwise known as investment partnership carried interest income), attributable to an investment in a Virginia venture capital account. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2018, but before December 31, 2023. No subtraction shall be allowed under this subdivision for an investment in a company that is owned or operated by a

family member or an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision 24 or a tax credit under § 58.1-339.4 for the same investment.

b. As used in this subdivision 27:

"Qualified portfolio company" means a company that (i) has its principal place of business in the Commonwealth; (ii) has a primary purpose of production, sale, research, or development of a product or service other than the management or investment of capital; and (iii) provides equity in the company to the Virginia venture capital account in exchange for a capital investment. "Qualified portfolio company" does not include a company that is an individual or sole proprietorship.

"Virginia venture capital account" means an investment fund that has been certified by the Department as a Virginia venture capital account. In order to be certified as a Virginia venture capital account, the operator of the investment fund shall register the investment fund with the Department prior to December 31, 2023, (i) indicating that it intends to invest at least 50 percent of the capital committed to its fund in qualified portfolio companies and (ii) providing documentation that it employs at least one investor who has at least four years of professional experience in venture capital investment or substantially equivalent experience. "Substantially equivalent experience" includes, but is not limited to, an undergraduate degree from an accredited college or university in economics, finance, or a similar field of study. The Department may require an investment fund to provide documentation of the investor's training, education, or experience as deemed necessary by the Department to determine substantial equivalency. If the Department determines that the investment fund employs at least one investor with the experience set forth herein, the Department shall certify the investment fund as a Virginia venture capital account at such time as the investment fund actually invests at least 50 percent of the capital committed to its fund in qualified portfolio companies.

28. a. Income attributable to an investment in a Virginia real estate investment trust. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2019, but before December 31, 2024. No subtraction shall be allowed for an investment in a trust that is managed by a family member or an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision 24 or 27 or a tax credit under § 58.1-339.4 for the same investment.

b. As used in this subdivision 28:

"Distressed" means satisfying the criteria applicable to a locality described in subdivision E 2 of $\S 2.2-115$.

"Double distressed" means satisfying the criteria applicable to a locality described in subdivision E 3 of § 2.2-115.

"Virginia real estate investment trust" means a real estate investment trust, as defined in 26 U.S.C. § 856, that has been certified by the Department as a Virginia real estate investment trust. In order to be certified as a Virginia real estate investment trust, the trustee shall register the trust with the Department prior to December 31, 2024, indicating that it intends to invest at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed. If the Department determines that the trust satisfies the preceding criteria, the Department shall certify the trust as a Virginia real estate

investment trust at such time as the trust actually invests at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed.

29. For taxable years beginning on and after January 1, 2019, any gain recognized from the taking of real property by condemnation proceedings.

30. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, up to \$100,000 of all grant funds received by the taxpayer under the Rebuild Virginia program established by the Governor and administered by the Department of Small Business and Supplier Diversity.

§ 58.1-322.03. Virginia taxable income; deductions.

In computing Virginia taxable income pursuant to § 58.1-322, there shall be deducted from Virginia adjusted gross income as defined in § 58.1-321:

- 1. a. The amount allowable for itemized deductions for federal income tax purposes where the taxpayer has elected for the taxable year to itemize deductions on his federal return, but reduced by the amount of income taxes imposed by the Commonwealth or any other taxing jurisdiction and deducted on such federal return and increased by an amount that, when added to the amount deducted under § 170 of the Internal Revenue Code for mileage, results in a mileage deduction at the state level for such purposes at a rate of 18 cents per mile; or
- b. Provided that the taxpayer has not itemized deductions for the taxable year on his federal income tax return: (i) for taxable years beginning before January 1, 2019, and on and after January 1, 2026, \$3,000 for single individuals and \$6,000 for married persons (one-half of such amounts in the case of a married individual filing a separate return) and (ii) for taxable years beginning on and after January 1, 2019, but before January 1, 2026, \$4,500 for single individuals and \$9,000 for married persons (one-half of such amounts in the case of a married individual filing a separate return). For purposes of this section, any person who may be claimed as a dependent on another taxpayer's return for the taxable year may compute the deduction only with respect to earned income.
- 2. a. A deduction in the amount of \$930 for each personal exemption allowable to the taxpayer for federal income tax purposes.
- b. Each blind or aged taxpayer as defined under § 63(f) of the Internal Revenue Code shall be entitled to an additional personal exemption in the amount of \$800.
- The additional deduction for blind or aged taxpayers allowed under this subdivision shall be allowable regardless of whether the taxpayer itemizes deductions for the taxable year for federal income tax purposes.
- 3. A deduction equal to the amount of employment-related expenses upon which the federal credit is based under § 21 of the Internal Revenue Code for expenses for household and dependent care services necessary for gainful employment.
- 4. An additional \$1,000 deduction for each child residing for the entire taxable year in a home under permanent foster care placement as defined in § 63.2-908, provided that the taxpayer can also claim the child as a personal exemption under § 151 of the Internal Revenue Code.
- 5. a. A deduction in the amount of \$12,000 for individuals born on or before January 1, 1939.

b. A deduction in the amount of \$12,000 for individuals born after January 1, 1939, who have attained the age of 65. This deduction shall be reduced by \$1 for every \$1 that the taxpayer's adjusted federal adjusted gross income exceeds \$50,000 for single taxpayers or \$75,000 for married taxpayers. For married taxpayers filing separately, the deduction shall be reduced by \$1 for every \$1 that the total combined adjusted federal adjusted gross income of both spouses exceeds \$75,000.

For the purposes of this subdivision, "adjusted federal adjusted gross income" means federal adjusted gross income minus any benefits received under Title II of the Social Security Act and other benefits subject to federal income taxation solely pursuant to § 86 of the Internal Revenue Code, as amended.

- 6. The amount an individual pays as a fee for an initial screening to become a possible bone marrow donor, if (i) the individual is not reimbursed for such fee or (ii) the individual has not claimed a deduction for the payment of such fee on his federal income tax return.
- 7. a. A deduction shall be allowed to the purchaser or contributor for the amount paid or contributed during the taxable year for a prepaid tuition contract or college savings trust account entered into with the Virginia College Savings Plan, pursuant to Chapter 7 (§ 23.1-700 et seq.) of Title 23.1. Except as provided in subdivision b, the amount deducted on any individual income tax return in any taxable year shall be limited to \$4,000 per prepaid tuition contract or college savings trust account. No deduction shall be allowed pursuant to this subdivision 7 if such payments or contributions are deducted on the purchaser's or contributor's federal income tax return. If the purchase price or annual contribution to a college savings trust account exceeds \$4,000, the remainder may be carried forward and subtracted in future taxable years until the purchase price or college savings trust contribution has been fully deducted; however, except as provided in subdivision b, in no event shall the amount deducted in any taxable year exceed \$4,000 per contract or college savings trust account. Notwithstanding the statute of limitations on assessments contained in § 58.1-312, any deduction taken hereunder shall be subject to recapture in the taxable year or years in which distributions or refunds are made for any reason other than (i) to pay qualified higher education expenses, as defined in § 529 of the Internal Revenue Code or (ii) the beneficiary's death, disability, or receipt of a scholarship. For the purposes of this subdivision, "purchaser" or "contributor" means the person shown as such on the records of the Virginia College Savings Plan as of December 31 of the taxable year. In the case of a transfer of ownership of a prepaid tuition contract or college savings trust account, the transferee shall succeed to the transferor's tax attributes associated with a prepaid tuition contract or college savings trust account, including, but not limited to, carryover and recapture of deductions.

b. A purchaser of a prepaid tuition contract or contributor to a college savings trust account who has attained age 70 shall not be subject to the limitation that the amount of the deduction not exceed \$4,000 per prepaid tuition contract or college savings trust account in any taxable year. Such taxpayer shall be allowed a deduction for the full amount paid for the contract or contributed to a college savings trust account, less any amounts previously deducted. 8. The total amount an individual actually contributed in funds to the Virginia Public School Construction Grants Program and Fund, established in Chapter 11.1 (§ 22.1-175.1 et seq.) of Title 22.1, provided that the individual has not claimed a deduction for such amount on his

federal income tax return.

- 9. An amount equal to 20 percent of the tuition costs incurred by an individual employed as a primary or secondary school teacher licensed pursuant to Chapter 15 (§ 22.1-289.1 et seq.) of Title 22.1 to attend continuing teacher education courses that are required as a condition of employment; however, the deduction provided by this subdivision shall be available only if (i) the individual is not reimbursed for such tuition costs and (ii) the individual has not claimed a deduction for the payment of such tuition costs on his federal income tax return.
- 10. The amount an individual pays annually in premiums for long-term health care insurance, provided that the individual has not claimed a deduction for federal income tax purposes, or, for taxable years beginning before January 1, 2014, a credit under § 58.1-339.11. For taxable years beginning on and after January 1, 2014, no such deduction for long-term health care insurance premiums paid by the individual during the taxable year shall be allowed if the individual has claimed a federal income tax deduction for such taxable year for long-term health care insurance premiums paid by him.
- 11. Contract payments to a producer of quota tobacco or a tobacco quota holder, or their spouses, as provided under the American Jobs Creation Act of 2004 (P.L. 108-357), but only to the extent that such payments have not been subtracted pursuant to subsection D of § 58.1-402, as follows:
- a. If the payment is received in installment payments, then the recognized gain may be subtracted in the taxable year immediately following the year in which the installment payment is received.
- b. If the payment is received in a single payment, then 10 percent of the recognized gain may be subtracted in the taxable year immediately following the year in which the single payment is received. The taxpayer may then deduct an equal amount in each of the nine succeeding taxable years.
- 12. An amount equal to 20 percent of the sum paid by an individual pursuant to Chapter 6 (§ 58.1-600 et seq.), not to exceed \$500 in each taxable year, in purchasing for his own use the following items of tangible personal property: (i) any clothes washers, room air conditioners, dishwashers, and standard size refrigerators that meet or exceed the applicable energy star efficiency requirements developed by the U.S. Environmental Protection Agency and the U.S. Department of Energy; (ii) any fuel cell that (a) generates electricity using an electrochemical process, (b) has an electricity-only generation efficiency greater than 35 percent, and (c) has a generating capacity of at least two kilowatts; (iii) any gas heat pump that has a coefficient of performance of at least 1.25 for heating and at least 0.70 for cooling; (iv) any electric heat pump hot water heater that yields an energy factor of at least 1.7; (v) any electric heat pump that has a heating system performance factor of at least 8.0 and a cooling seasonal energy efficiency ratio of at least 13.0; (vi) any central air conditioner that has a cooling seasonal energy efficiency ratio of at least 13.5; (vii) any advanced gas or oil water heater that has an energy factor of at least 0.65; (viii) any advanced oil-fired boiler with a minimum annual fuel-utilization rating of 85; (ix) any advanced oil-fired furnace with a minimum annual fuel-utilization rating of 85; and (x) programmable thermostats.
- 13. The lesser of \$5,000 or the amount actually paid by a living donor of an organ or other living tissue for unreimbursed out-of-pocket expenses directly related to the donation that arose

within 12 months of such donation, provided that the donor has not taken a medical deduction in accordance with the provisions of § 213 of the Internal Revenue Code for such expenses. The deduction may be taken in the taxable year in which the donation is made or the taxable year in which the 12-month period expires.

- 14. For taxable years beginning on and after January 1, 2013, the amount an individual age 66 or older with earned income of at least \$20,000 for the year and federal adjusted gross income not in excess of \$30,000 for the year pays annually in premiums for (i) a prepaid funeral insurance policy covering the individual or (ii) medical or dental insurance for any person for whom individual tax filers may claim a deduction for such premiums under federal income tax laws. As used in this subdivision, "earned income" means the same as that term is defined in § 32(c) of the Internal Revenue Code. The deduction shall not be allowed for any portion of such premiums paid for which the individual has (a) been reimbursed, (b) claimed a deduction for federal income tax purposes, (c) claimed a deduction or subtraction under another provision of this section, or (d) claimed a federal income tax credit or any income tax credit pursuant to this chapter.
- 15. For taxable years beginning on and after January 1, 2018, 20 percent of business interest disallowed as a deduction pursuant to § 163(j) of the Internal Revenue Code. For purposes of this subdivision, "business interest" means the same as that term is defined under § 163(j) of the Internal Revenue Code.
- 16. For taxable years beginning on and after January 1, 2019, the actual amount of real and personal property taxes imposed by the Commonwealth or any other taxing jurisdiction not otherwise deducted solely on account of the dollar limitation imposed on individual deductions by § 164(b)(6)(B) of the Internal Revenue Code.
- 17. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, up to \$100,000 of the amount that is not deductible when computing federal adjusted gross income solely on account of the portion of subdivision B 10 of § 58.1-301 related to Paycheck Protection Program loans.

§ 58.1-402. Virginia taxable income.

A. For purposes of this article, Virginia taxable income for a taxable year means the federal taxable income and any other income taxable to the corporation under federal law for such year of a corporation adjusted as provided in subsections B, C, D, E, and G, and H.

For a regulated investment company and a real estate investment trust, such term means the "investment company taxable income" and "real estate investment trust taxable income," respectively, to which shall be added in each case any amount of capital gains and any other income taxable to the corporation under federal law which shall be further adjusted as provided in subsections B, C, D, E, and G, and H.

- B. There shall be added to the extent excluded from federal taxable income:
- 1. Interest, less related expenses to the extent not deducted in determining federal taxable income, on obligations of any state other than Virginia, or of a political subdivision of any such other state unless created by compact or agreement to which the Commonwealth is a party;
- 2. Interest or dividends, less related expenses to the extent not deducted in determining federal

taxable income, on obligations or securities of any authority, commission or instrumentality of the United States, which the laws of the United States exempt from federal income tax but not from state income taxes;

- 3. [Repealed.]
- 4. The amount of any net income taxes and other taxes, including franchise and excise taxes, which are based on, measured by, or computed with reference to net income, imposed by the Commonwealth or any other taxing jurisdiction, to the extent deducted in determining federal taxable income;
- 5. Unrelated business taxable income as defined by § 512 of the Internal Revenue Code;
- 6. [Repealed.]
- 7. The amount required to be included in income for the purpose of computing the partial tax on an accumulation distribution pursuant to § 667 of the Internal Revenue Code;
- 8. a. For taxable years beginning on and after January 1, 2004, the amount of any intangible expenses and costs directly or indirectly paid, accrued, or incurred to, or in connection directly or indirectly with one or more direct or indirect transactions with one or more related members to the extent such expenses and costs were deductible or deducted in computing federal taxable income for Virginia purposes. This addition shall not be required for any portion of the intangible expenses and costs if one of the following applies:
- (1) The corresponding item of income received by the related member is subject to a tax based on or measured by net income or capital imposed by Virginia, another state, or a foreign government that has entered into a comprehensive tax treaty with the United States government;
- (2) The related member derives at least one-third of its gross revenues from the licensing of intangible property to parties who are not related members, and the transaction giving rise to the expenses and costs between the corporation and the related member was made at rates and terms comparable to the rates and terms of agreements that the related member has entered into with parties who are not related members for the licensing of intangible property; or
- (3) The corporation can establish to the satisfaction of the Tax Commissioner that the intangible expenses and costs meet both of the following: (i) the related member during the same taxable year directly or indirectly paid, accrued or incurred such portion to a person who is not a related member, and (ii) the transaction giving rise to the intangible expenses and costs between the corporation and the related member did not have as a principal purpose the avoidance of any portion of the tax due under this chapter.
- b. A corporation required to add to its federal taxable income intangible expenses and costs pursuant to subdivision a may petition the Tax Commissioner, after filing the related income tax return for the taxable year and remitting to the Tax Commissioner all taxes, penalties, and interest due under this article for such taxable year including tax upon any amount of intangible expenses and costs required to be added to federal taxable income pursuant to subdivision a, to consider evidence relating to the transaction or transactions between the corporation and a related member or members that resulted in the corporation's taxable income being increased, as required under subdivision a, for such intangible expenses and costs.

If the corporation can demonstrate to the Tax Commissioner's sole satisfaction, by clear and convincing evidence, that the transaction or transactions between the corporation and a related member or members resulting in such increase in taxable income pursuant to subdivision a had

a valid business purpose other than the avoidance or reduction of the tax due under this chapter, the Tax Commissioner shall permit the corporation to file an amended return. For purposes of such amended return, the requirements of subdivision a shall not apply to any transaction for which the Tax Commissioner is satisfied (and has identified) that the transaction had a valid business purpose other than the avoidance or reduction of the tax due under this chapter. Such amended return shall be filed by the corporation within one year of the written permission granted by the Tax Commissioner and any refund of the tax imposed under this article shall include interest at a rate equal to the rate of interest established under § 58.1-15 and such interest shall accrue as provided under § 58.1-1833. However, upon the filing of such amended return, any related member of the corporation that subtracted from taxable income amounts received pursuant to subdivision C 21 shall be subject to the tax imposed under this article on that portion of such amounts for which the corporation has filed an amended return pursuant to this subdivision. In addition, for such transactions identified by the Tax Commissioner herein by which he has been satisfied by clear and convincing evidence, the Tax Commissioner may permit the corporation in filing income tax returns for subsequent taxable years to deduct the related intangible expenses and costs without making the adjustment under subdivision a.

The Tax Commissioner may charge a fee for all direct and indirect costs relating to the review of any petition pursuant to this subdivision, to include costs necessary to secure outside experts in evaluating the petition. The Tax Commissioner may condition the review of any petition pursuant to this subdivision upon payment of such fee.

No suit for the purpose of contesting any action of the Tax Commissioner under this subdivision shall be maintained in any court of this Commonwealth.

- c. Nothing in subdivision B 8 shall be construed to limit or negate the Department's authority under § 58.1-446;
- 9. a. For taxable years beginning on and after January 1, 2004, the amount of any interest expenses and costs directly or indirectly paid, accrued, or incurred to, or in connection directly or indirectly with one or more direct or indirect transactions with one or more related members to the extent such expenses and costs were deductible or deducted in computing federal taxable income for Virginia purposes. This addition shall not be required for any portion of the interest expenses and costs, if:
- (1) The related member has substantial business operations relating to interest-generating activities, in which the related member pays expenses for at least five full-time employees who maintain, manage, defend or are otherwise responsible for operations or administration relating to the interest-generating activities; and
- (2) The interest expenses and costs are not directly or indirectly for, related to or in connection with the direct or indirect acquisition, maintenance, management, sale, exchange, or disposition of intangible property; and
- (3) The transaction giving rise to the expenses and costs between the corporation and the related member has a valid business purpose other than the avoidance or reduction of taxation and payments between the parties are made at arm's length rates and terms; and
- (4) One of the following applies:
- (i) The corresponding item of income received by the related member is subject to a tax based on or measured by net income or capital imposed by Virginia, another state, or a foreign

government that has entered into a comprehensive tax treaty with the United States government; (ii) Payments arise pursuant to a pre-existing contract entered into when the parties were not related members provided the payments continue to be made at arm's length rates and terms; (iii) The related member engages in transactions with parties other than related members that generate revenue in excess of \$2 million annually; or

- (iv) The transaction giving rise to the interest payments between the corporation and a related member was done at arm's length rates and terms and meets any of the following: (a) the related member uses funds that are borrowed from a party other than a related member or that are paid, incurred or passed-through to a person who is not a related member; (b) the debt is part of a regular and systematic funds management or portfolio investment activity conducted by the related member, whereby the funds of two or more related members are aggregated for the purpose of achieving economies of scale, the internal financing of the active business operations of members, or the benefit of centralized management of funds; (c) financing the expansion of the business operations; or (d) restructuring the debt of related members, or the pass-through of acquisition-related indebtedness to related members.
- b. A corporation required to add to its federal taxable income interest expenses and costs pursuant to subdivision a may petition the Tax Commissioner, after filing the related income tax return for the taxable year and remitting to the Tax Commissioner all taxes, penalties, and interest due under this article for such taxable year including tax upon any amount of interest expenses and costs required to be added to federal taxable income pursuant to subdivision a, to consider evidence relating to the transaction or transactions between the corporation and a related member or members that resulted in the corporation's taxable income being increased, as required under subdivision a, for such interest expenses and costs.

If the corporation can demonstrate to the Tax Commissioner's sole satisfaction, by clear and convincing evidence, that the transaction or transactions between the corporation and a related member or members resulting in such increase in taxable income pursuant to subdivision a had a valid business purpose other than the avoidance or reduction of the tax due under this chapter and that the related payments between the parties were made at arm's length rates and terms, the Tax Commissioner shall permit the corporation to file an amended return. For purposes of such amended return, the requirements of subdivision a shall not apply to any transaction for which the Tax Commissioner is satisfied (and has identified) that the transaction had a valid business purpose other than the avoidance or reduction of the tax due under this chapter and that the related payments between the parties were made at arm's length rates and terms. Such amended return shall be filed by the corporation within one year of the written permission granted by the Tax Commissioner and any refund of the tax imposed under this article shall include interest at a rate equal to the rate of interest established under § 58.1-15 and such interest shall accrue as provided under § 58.1-1833. However, upon the filing of such amended return, any related member of the corporation that subtracted from taxable income amounts received pursuant to subdivision C 21 shall be subject to the tax imposed under this article on that portion of such amounts for which the corporation has filed an amended return pursuant to this subdivision. In addition, for such transactions identified by the Tax Commissioner herein by which he has been satisfied by clear and convincing evidence, the Tax Commissioner may permit the corporation in filing income tax returns for subsequent taxable years to deduct the related interest expenses and costs without making the adjustment under subdivision a.

The Tax Commissioner may charge a fee for all direct and indirect costs relating to the review of any petition pursuant to this subdivision, to include costs necessary to secure outside experts in evaluating the petition. The Tax Commissioner may condition the review of any petition pursuant to this subdivision upon payment of such fee.

No suit for the purpose of contesting any action of the Tax Commissioner under this subdivision shall be maintained in any court of this Commonwealth.

- c. Nothing in subdivision B 9 shall be construed to limit or negate the Department's authority under § 58.1-446.
- d. For purposes of subdivision B 9:

"Arm's-length rates and terms" means that (i) two or more related members enter into a written agreement for the transaction, (ii) such agreement is of a duration and contains payment terms substantially similar to those that the related member would be able to obtain from an unrelated entity, (iii) the interest is at or below the applicable federal rate compounded annually for debt instruments under § 1274(d) of the Internal Revenue Code that was in effect at the time of the agreement, and (iv) the borrower or payor adheres to the payment terms of the agreement governing the transaction or any amendments thereto.

"Valid business purpose" means one or more business purposes that alone or in combination constitute the motivation for some business activity or transaction, which activity or transaction improves, apart from tax effects, the economic position of the taxpayer, as further defined by regulation.

- 10. a. For taxable years beginning on and after January 1, 2009, the amount of dividends deductible under §§ 561 and 857 of the Internal Revenue Code by a Captive Real Estate Investment Trust (REIT). For purposes of this subdivision, a REIT is a Captive REIT if:
- (1) It is not regularly traded on an established securities market;
- (2) More than 50 percent of the voting power or value of beneficial interests or shares of which, at any time during the last half of the taxable year, is owned or controlled, directly or indirectly, by a single entity that is (i) a corporation or an association taxable as a corporation under the Internal Revenue Code; and (ii) not exempt from federal income tax pursuant to § 501(a) of the Internal Revenue Code; and
- (3) More than 25 percent of its income consists of rents from real property as defined in § 856(d) of the Internal Revenue Code.
- b. For purposes of applying the ownership test of subdivision 10 a (2), the following entities shall not be considered a corporation or an association taxable as a corporation:
- (1) Any REIT that is not treated as a Captive REIT;
- (2) Any REIT subsidiary under § 856 of the Internal Revenue Code other than a qualified REIT subsidiary of a Captive REIT;
- (3) Any Listed Australian Property Trust, or an entity organized as a trust, provided that a Listed Australian Property Trust owns or controls, directly or indirectly, 75 percent or more of the voting or value of the beneficial interests or shares of such trust; and
- (4) Any Qualified Foreign Entity.
- c. For purposes of subdivision B 10, the constructive ownership rules prescribed under § 318(a) of the Internal Revenue Code, as modified by § 856(d)(5) of the Internal Revenue Code, shall

apply in determining the ownership of stock, assets, or net profits of any person.

d. For purposes of subdivision B 10:

"Listed Australian Property Trust" means an Australian unit trust registered as a Management Investment Scheme, pursuant to the Australian Corporations Act, in which the principal class of units is listed on a recognized stock exchange in Australia and is regularly traded on an established securities market.

"Qualified Foreign Entity" means a corporation, trust, association or partnership organized outside the laws of the United States and that satisfies all of the following criteria:

- (1) At least 75 percent of the entity's total asset value at the close of its taxable year is represented by real estate assets, as defined in § 856(c)(5)(B) of the Internal Revenue Code, thereby including shares or certificates of beneficial interest in any REIT, cash and cash equivalents, and U.S. Government securities;
- (2) The entity is not subject to a tax on amounts distributed to its beneficial owners, or is exempt from entity level tax;
- (3) The entity distributes, on an annual basis, at least 85 percent of its taxable income, as computed in the jurisdiction in which it is organized, to the holders of its shares or certificates of beneficial interest;
- (4) The shares or certificates of beneficial interest of such entity are regularly traded on an established securities market or, if not so traded, not more than 10 percent of the voting power or value in such entity is held directly, indirectly, or constructively by a single entity or individual; and
- (5) The entity is organized in a country that has a tax treaty with the United States.
- e. For taxable years beginning on or after January 1, 2016, for purposes of subdivision B 10, any voting power or value of the beneficial interests or shares in a REIT that is held in a segregated asset account of a life insurance corporation as described in § 817 of the Internal Revenue Code shall not be taken into consideration when determining if such REIT is a Captive REIT.
- 11. For taxable years beginning on or after January 1, 2016, to the extent that tax credit is allowed for the same donation pursuant to § 58.1-439.12:12, any amount claimed as a federal income tax deduction for such donation under § 170 of the Internal Revenue Code, as amended or renumbered.
- C. There shall be subtracted to the extent included in and not otherwise subtracted from federal taxable income:
- 1. Income derived from obligations, or on the sale or exchange of obligations, of the United States and on obligations or securities of any authority, commission or instrumentality of the United States to the extent exempt from state income taxes under the laws of the United States including, but not limited to, stocks, bonds, treasury bills, and treasury notes, but not including interest on refunds of federal taxes, interest on equipment purchase contracts, or interest on other normal business transactions.
- 2. Income derived from obligations, or on the sale or exchange of obligations of this Commonwealth or of any political subdivision or instrumentality of this Commonwealth.
- 3. Dividends upon stock in any domestic international sales corporation, as defined by § 992 of the Internal Revenue Code, 50 percent or more of the income of which was assessable for the

preceding year, or the last year in which such corporation has income, under the provisions of the income tax laws of the Commonwealth.

- 4. The amount of any refund or credit for overpayment of income taxes imposed by this Commonwealth or any other taxing jurisdiction.
- 5. Any amount included therein by the operation of the provisions of § 78 of the Internal Revenue Code (foreign dividend gross-up).
- 6. The amount of wages or salaries eligible for the federal Targeted Jobs Credit which was not deducted for federal purposes on account of the provisions of § 280C(a) of the Internal Revenue Code.
- 7. Any amount included therein by the operation of § 951 of the Internal Revenue Code (subpart F income) or, for taxable years beginning on and after January 1, 2018, § 951A of the Internal Revenue Code (Global Intangible Low-Taxed Income).
- 8. Any amount included therein which is foreign source income as defined in § 58.1-302.
- 9. [Repealed.]
- 10. The amount of any dividends received from corporations in which the taxpaying corporation owns 50 percent or more of the voting stock.
- 11. [Repealed.]
- 12, 13. [Expired.]
- 14. For taxable years beginning on or after January 1, 1995, the amount for "qualified research expenses" or "basic research expenses" eligible for deduction for federal purposes, but which were not deducted, on account of the provisions of § 280C(c) of the Internal Revenue Code.
- 15. For taxable years beginning on or after January 1, 2000, the total amount actually contributed in funds to the Virginia Public School Construction Grants Program and Fund established in Chapter 11.1 (§ 22.1-175.1 et seq.) of Title 22.1.
- 16. For taxable years beginning on or after January 1, 2000, but before January 1, 2015, the gain derived from the sale or exchange of real property or the sale or exchange of an easement to real property which results in the real property or the easement thereto being devoted to open-space use, as that term is defined in § 58.1-3230, for a period of time not less than 30 years. To the extent a subtraction is taken in accordance with this subdivision, no tax credit under this chapter for donating land for its preservation shall be allowed for three years following the year in which the subtraction is taken.
- 17. For taxable years beginning on and after January 1, 2001, any amount included therein with respect to § 58.1-440.1.
- 18. For taxable years beginning on and after January 1, 1999, income received as a result of (i) the "Master Settlement Agreement," as defined in § 3.2-3100; and (ii) the National Tobacco Grower Settlement Trust dated July 19, 1999, by (a) tobacco farming businesses; (b) any business holding a tobacco marketing quota, or tobacco farm acreage allotment, under the Agricultural Adjustment Act of 1938; or (c) any business having the right to grow tobacco pursuant to such a quota allotment.
- 19, 20. [Repealed.]
- 21. For taxable years beginning on and after January 1, 2004, any amount of intangible expenses and costs or interest expenses and costs added to the federal taxable income of a corporation pursuant to subdivision B 8 or B 9 shall be subtracted from the federal taxable

income of the related member that received such amount if such related member is subject to Virginia income tax on the same amount.

- 22. For taxable years beginning on and after January 1, 2009, any gain recognized from the sale of launch services to space flight participants, as defined in 49 U.S.C. § 70102, or launch services intended to provide individuals the training or experience of a launch, without performing an actual launch. To qualify for a deduction under this subdivision, launch services must be performed in Virginia or originate from an airport or spaceport in Virginia.
- 23. For taxable years beginning on and after January 1, 2009, any gain recognized as a result of resupply services contracts for delivering payload, as defined in 49 U.S.C. § 70102, entered into with the Commercial Orbital Transportation Services division of the National Aeronautics and Space Administration or other space flight entity, as defined in § 8.01-227.8, and launched from an airport or spaceport in Virginia.
- 24. For taxable years beginning on or after January 1, 2011, any income taxed as a long-term capital gain for federal income tax purposes, or any income taxed as investment services partnership interest income (otherwise known as investment partnership carried interest income) for federal income tax purposes. To qualify for a subtraction under this subdivision, such income must be attributable to an investment in a "qualified business," as defined in § 58.1-339.4, or in any other technology business approved by the Secretary of Administration, provided the business has its principal office or facility in the Commonwealth and less than \$3 million in annual revenues in the fiscal year prior to the investment. To qualify for a subtraction under this subdivision, the investment must be made between the dates of April 1, 2010, and June 30, 2020. No taxpayer who has claimed a tax credit for an investment in a "qualified business" under § 58.1-339.4 shall be eligible for the subtraction under this subdivision for an investment in the same business.
- 25. a. Income, including investment services partnership interest income (otherwise known as investment partnership carried interest income), attributable to an investment in a Virginia venture capital account. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2018, but before December 31, 2023. No subtraction shall be allowed under this subdivision for an investment in a company that is owned or operated by an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision C 24 for the same investment.

b. As used in this subdivision 25:

"Qualified portfolio company" means a company that (i) has its principal place of business in the Commonwealth; (ii) has a primary purpose of production, sale, research, or development of a product or service other than the management or investment of capital; and (iii) provides equity in the company to the Virginia venture capital account in exchange for a capital investment. "Qualified portfolio company" does not include a company that is an individual or sole proprietorship.

"Virginia venture capital account" means an investment fund that has been certified by the Department as a Virginia venture capital account. In order to be certified as a Virginia venture capital account, the operator of the investment fund shall register the investment fund with the Department prior to December 31, 2023, (i) indicating that it intends to invest at least 50 percent of the capital committed to its fund in qualified portfolio companies and (ii) providing

documentation that it employs at least one investor who has at least four years of professional experience in venture capital investment or substantially equivalent experience. "Substantially equivalent experience" includes, but is not limited to, an undergraduate degree from an accredited college or university in economics, finance, or a similar field of study. The Department may require an investment fund to provide documentation of the investor's training, education, or experience as deemed necessary by the Department to determine substantial equivalency. If the Department determines that the investment fund employs at least one investor with the experience set forth herein, the Department shall certify the investment fund as a Virginia venture capital account at such time as the investment fund actually invests at least 50 percent of the capital committed to its fund in qualified portfolio companies.

26. a. Income attributable to an investment in a Virginia real estate investment trust. To qualify for a subtraction under this subdivision, the investment shall be made on or after January 1, 2019, but before December 31, 2024. No subtraction shall be allowed for an investment in a trust that is managed by an affiliate of the taxpayer. No subtraction shall be allowed under this subdivision for a taxpayer who has claimed a subtraction under subdivision C 24 or 25 for the same investment.

b. As used in this subdivision 26:

"Distressed" means satisfying the criteria applicable to a locality described in subdivision E 2 of § 2.2-115.

"Double distressed" means satisfying the criteria applicable to a locality described in subdivision E 3 of \S 2.2-115.

"Virginia real estate investment trust" means a real estate investment trust, as defined in 26 U.S.C. § 856, that has been certified by the Department as a Virginia real estate investment trust. In order to be certified as a Virginia real estate investment trust, the trustee shall register the trust with the Department prior to December 31, 2024, indicating that it intends to invest at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed. If the Department determines that the trust satisfies the preceding criteria, the Department shall certify the trust as a Virginia real estate investment trust at such time as the trust actually invests at least 90 percent of trust funds in Virginia and at least 40 percent of trust funds in real estate in localities that are distressed or double distressed.

- 27. For taxable years beginning on and after January 1, 2019, any gain recognized from the taking of real property by condemnation proceedings.
- 28. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, up to \$100,000 of all grant funds received by the taxpayer under the Rebuild Virginia program established by the Governor and administered by the Department of Small Business and Supplier Diversity.
- D. For taxable years beginning on and after January 1, 2006, there shall be subtracted from federal taxable income contract payments to a producer of quota tobacco or a tobacco quota holder as provided under the American Jobs Creation Act of 2004 (P.L. 108-357) as follows:

 1. If the payment is received in installment payments, then the recognized gain, including any gain recognized in taxable year 2005, may be subtracted in the taxable year immediately following the year in which the installment payment is received.

- 2. If the payment is received in a single payment, then 10 percent of the recognized gain may be subtracted in the taxable year immediately following the year in which the single payment is received. The taxpayer may then deduct an equal amount in each of the nine succeeding taxable years.
- E. Adjustments to federal taxable income shall be made to reflect the transitional modifications provided in § 58.1-315.
- F. Notwithstanding any other provision of law, the income from any disposition of real property which is held by the taxpayer for sale to customers in the ordinary course of the taxpayer's trade or business, as defined in § 453(l)(1)(B) of the Internal Revenue Code, of property made on or after January 1, 2009, may, at the election of the taxpayer, be recognized under the installment method described under § 453 of the Internal Revenue Code, provided that (i) the election relating to the dealer disposition of the property has been made on or before the due date prescribed by law (including extensions) for filing the taxpayer's return of the tax imposed under this chapter for the taxable year in which the disposition occurs, and (ii) the dealer disposition is in accordance with restrictions or conditions established by the Department, which shall be set forth in guidelines developed by the Department. Along with such restrictions or conditions, the guidelines shall also address the recapture of such income under certain circumstances. The development of the guidelines shall be exempt from the Administrative Process Act (§ 2.2-4000 et seq.).
- G. For taxable years beginning on and after January 1, 2018, there shall be deducted to the extent included in and not otherwise subtracted from federal taxable income 20 percent of business interest disallowed as a deduction pursuant to § 163(j) of the Internal Revenue Code. For purposes of this subsection, "business interest" means the same as that term is defined under § 163(j) of the Internal Revenue Code.
- H. For taxable years beginning on and after January 1, 2020, but before January 1, 2021, there shall be deducted to the extent not otherwise subtracted from federal taxable income up to \$100,000 of the amount that is not deductible when computing federal taxable income solely on account of the portion of subdivision B 10 of § 58.1-301 related to Paycheck Protection Program loans."

Page 736, strike lines 1 through 32.

Explanation:

(This amendment advances Virginia's date of conformity with the Internal Revenue Code from December 31, 2019, to December 31, 2020. The amendment generally conforms Virginia's tax code to both the federal Coronavirus Aid, Relief, and Economic Security (CARES) Act and the Consolidated Appropriations Act (CAA), 2021, with certain exceptions consistent with SB 1146. The amendment provides a deduction of up to \$100,000 for business expenses funded by forgiven loans under the Paycheck Protection Program (PPP) and provides an income tax subtraction of up to \$100,000 for grant funds received under the Rebuild Virginia program.)

Item 4-14 #4s

Effective Date

Language

Language:

Page 732, after line 11, insert:

"I. Notwithstanding paragraph G. above or any other provision of law, the Chairs of the Virginia Redistricting Commission shall keep the Senate President Pro Tempore, the Senate Minority Leader, the Speaker of the House of Delegates, the House Minority Leader, and the Governor informed about the timing of availability of United States Bureau of the Census data as it relates to the tabulation of the population for reapportionment purposes pursuant to P.L. 94-171, and options for redistricting and its impact on elections for the House of Delegates."

Page 732, after line 38, insert:

"F. Notwithstanding paragraph C. above, or any other provision of law, the daily compensation and reimbursement for reasonable and necessary expenses for legislative and non-legislative members of the Virginia Redistricting Commission for attendance at an official meeting shall be set at the same amounts provided for legislative members in paragraphs B.4.d. and B.5. of Item 1 of the this act."

Explanation:

(This amendment provides for and directs the Chairs of the Virginia Redistricting Commission to communicate information related to the availability of U.S. Census data and its impact on elections for the House of Delegates, and includes language clarifying that members of the Virginia Redistricting Commission shall receive daily compensation and reimbursement for reasonable and necessary expenses for attendance at an official meeting equal to that which is generally provided to members for attendance at any official meeting.)

The reading of the amendments was waived.

On motion of Senator Howell, the amendments were agreed to.

H.B. 1800, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows: YEAS--30. NAYS--9. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--30.

NAYS--Chase, Cosgrove, Kiggans, McDougle, Norment, Obenshain, Peake, Stanley, Suetterlein--9. RULE 36--0.

STATEMENTS ON VOTE

Senator DeSteph stated that he was abstaining pursuant to Rule 36 on Item 105 #1s, but voting on **H.B. 1800** as a whole.

Senator Edwards stated that he was abstaining pursuant to Rule 36 on Item 266 #1s, but voting on **H.B. 1800** as a whole.

Senator Locke stated that she was abstaining pursuant to Rule 36 on Item 303, but voting on **H.B.** 1800 as a whole.

Senator Lucas stated that she was abstaining pursuant to Rule 36 on Item 313, Item 313 #4s, Item 320 #1s, Item 479.20 #1s, and Item 482.20, but voting on **H.B. 1800** as a whole.

Senator McClellan stated that she was abstaining pursuant to Rule 36 on Item 114, Item 114 #5s, and Item 479.10, but voting on **H.B. 1800** as a whole.

Senator McDougle stated that he was abstaining pursuant to Rule 36 on Item 214 #1s, but voting on **H.B. 1800** as a whole.

Senator McPike stated that he was abstaining pursuant to Rule 36 on Item 385, Item 386 #1s, and Item C-70, but voting on **H.B. 1800** as a whole.

Senator Norment stated that he was abstaining pursuant to Rule 36 on Item 266 #1s and Item 313, but voting on **H.B. 1800** as a whole.

Senator Petersen stated that he was abstaining pursuant to Rule 36 on Item 105 #1s and Item 266 #1s, but voting on **H.B. 1800** as a whole.

Senator Stuart stated that he was abstaining pursuant to Rule 36 on Item 374 #4s, but voting on **H.B.** 1800 as a whole.

Senator Surovell stated that he was abstaining pursuant to Rule 36 on Item 135, Item 374, and Item 374 #3s, but voting on **H.B. 1800** as a whole.

Senator DeSteph stated that he wished to be recorded as voting nay on the question of agreeing to Item 36 #1s, Item 40 #1s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 113 #1s, Item 114 #3s, Item 134 #2s, Item 138 #3s, Item 291 #1s, Item 312 #1s, Item 313 #17s, Item 350 #3s, Item 356 #1s, Item 356 #2s, Item 356 #3s, Item 356 #4s, Item 359 #1s, Item 385 #1s, Item 391 #1s, Item 391 #2s, Item 394 #2s, Item 400 #2s, Item 400 #4s, Item 402 #1s, Item 402 #3s, Item 425 #2s, Item 426 #2s, Item 479 #4s, Item 4-14 #1s, and Item 4-14 #2s to **H.B. 1800.**

Senator Dunnavant stated that she wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 391 #2s, and Item 402 #1s to **H.B. 1800.**

Senator Hanger stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 391 #2s, Item 479 #4s, and Section 4.5.04 of part 4 to **H.B. 1800.**

Senator Newman stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 134 #2s, Item 138 #3s, Item 139 #1s, Item 150, Item 291 #1s, Item 313 #17s, Item 350 #3s, Item 356 #1s, Item 374 #2s, Item 391 #2s, Item 394 #2s, Item 402 #1s, Item 425 #2s, Item 479 #4s, Section 4.5.04 of part 4, and Item 302 F 1-4 to **H.B. 1800.**

Senator Norment stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, and Item C-22.10 #1s to **H.B. 1800**.

Senator Pillion stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 391 #2s, Item 402 #1s, Item 425 #2s, Item 479 #4s, and Section 4.5.04 of part 4 to **H.B. 1800**.

Senator Reeves stated that he wished to be recorded as voting nay on the question of agreeing to Item 36 #1s, Item 40 #1s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 113 #1s, Item 114 #3s, Item 134 #2s, Item 138 #3s, Item 291 #1s, Item 312 #1s, Item 313 #17s, Item 350 #3s, Item 356 #1s, Item 356 #2s, Item 356 #3s, Item 356 #4s, Item 359 #1s, Item 374 #2s, Item 385 #1s, Item 391 #1s, Item 391 #2s, Item 394 #2s, Item 400 #2s, Item 400 #4s, Item 402 #1s, Item 402 #3s, Item 425 #2s, Item 426 #2s, Item 479 #4s, Item 4-14 #1s, Item 4-14 #2s, Section 4.5.04 of part 4, and Item 302 F 1-4 to **H.B. 1800.**

Senator Ruff stated that he wished to be recorded as voting nay on the question of agreeing to Item $40 \, \#1s$, Item $57 \, \#1s$, Item $73 \, \#1s$, Item $73 \, \#2s$, Item $139 \, \#1s$, Item $145 \, \#9s$, Item $164 \, \#1s$, Item $291 \, \#1s$, Item $312 \, \#1s$, Item $313 \, \#17s$, Item $391 \, \#1s$, Item $391 \, \#2s$, Item $402 \, \#1s$, Item $45 \, \#2s$, Item $479 \, \#4s$, Item $414 \, \#1s$, Item $414 \, \#2s$, Section $45.04 \, of$ part $411 \, \#1s$, Item $411 \, \#1s$, Item 4

Senator Stuart stated that he wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 73 #1s, Item 73 #2s, Item 134 #2s, Item 138 #3s, Item 313 #17s, Item 374 #2s, Item 391 #2s, Item 402 #1s, Item 425 #2s, Item 479 #4s, Section 4.5.04 of part 4, and Item 302 F 1-4 to **H.B. 1800.**

Senator Vogel stated that she wished to be recorded as voting nay on the question of agreeing to Item 40 #1s, Item 48 #3s, Item 57 #1s, Item 402 #1s, and Item 425 #2s to **H.B. 1800.**

H.B. 1867 (one thousand eight hundred sixty-seven), on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.B. 1898 (one thousand eight hundred ninety-eight), on motion of Senator McPike, was passed with its title.

The recorded vote is as follows: YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

NAYS--Chase, Suetterlein--2. RULE 36--0.

THE PRESIDENT PRESIDING

The President assumed the Chair.

H.B. 1916 (one thousand nine hundred sixteen), on motion of Senator Howell, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1. RULE 36--0.

H.B. 1935 (one thousand nine hundred thirty-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.

The reading of the substitute was waived.

On motion of Senator Howell, the substitute was agreed to.

Senator McDougle offered the following amendments to the substitute:

1. Line 6, substitute, Title, after *reenact* strike §§

insert

ş

2. Line 6, substitute, Title, after 58.1-301 strike

, 58.1-322.02, 58.1-322.03, and 58.1-402

3. Line 10, substitute, after That

strike

§§

insert

§

4. Line 10, substitute, after 58.1-301

strike

, 58.1-322.02, 58.1-322.03, and 58.1-402

5. Line 10, substitute, after Virginia

strike

are

insert

is

6. Line 52, substitute, after corporations;

insert

and

7. Line 54, substitute, after interest

strike

the remainder of line 54, all of lines 55, 56, and 57, and through assistance on line 58

8. Line 62, substitute

strike

all of lines 62 through 754

On motion of Senator McDougle, the reading of the amendments was waived.

Senator McDougle moved that the amendments be agreed to.

The question was put on agreeing to the amendments.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--17. NAYS--22. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22. RULE 36--0.

The amendments were rejected.

The substitute was ordered to be engrossed.

H.B. 1935, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows: YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1. RULE 36--0.

STATEMENT ON VOTE

Senator Chase stated that she voted nay on the question of the passage of **H.B. 1935**, whereas she intended to vote yea.

H.B. 2006 (two thousand six), on motion of Senator Howell, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--1. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

NAYS--Chase--1.

RULE 36--Suetterlein--1.

H.B. 2095 (two thousand ninety-five), on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--38.

NAYS--Suetterlein--1.

RULE 36--0.

H.B. 2204 (two thousand two hundred four), on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

 $NAYS\text{--}Chase, \, De Steph\text{--}2.$

RULE 36--0.

RECESS

At 1:05 p.m., Senator Saslaw moved that the Senate recess until 2:00 p.m.

The motion was agreed to.

The hour of 2:00 p.m. having arrived, the Chair was resumed.

HOUSE BILLS ON THIRD READING

H.B. 2014 (two thousand fourteen) was taken up, amendment No. 1 offered by Senator Barker having been offered on February 16, 2021.

Senator Barker withdrew amendment No. 1.

Senator Barker offered the following amendment No. 2:

2. Line 131, engrossed, after enforceable. insert

A landlord owning four or fewer rental dwelling units is exempt from the requirements of this section.

On motion of Senator Barker, the reading of amendment No. 2 was waived.

Senator Barker moved that amendment No. 2 be agreed to.

- H.B. 2014, on motion of Senator Petersen, was passed by temporarily.
- **H.B. 1801** (one thousand eight hundred one) was taken up and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--23. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, Deeds, DeSteph, Hanger, Kiggans, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Reeves, Suetterlein--14.

RULE 36--0.

H.B. 1903 (one thousand nine hundred three) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Vogel--17.

RULE 36--0.

H.B. 1932 (one thousand nine hundred thirty-two), on motion of Senator Favola, was recommitted to the Committee on Rehabilitation and Social Services.

H.B. 2262 (two thousand two hundred sixty-two) was read by title the third time.

Senator Deeds offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 46.2-839 and 46.2-905 of the Code of Virginia, relating to traffic regulation; bicycles.

On motion of Senator Deeds, the reading of the substitute was waived.

Senator Deeds moved that the substitute be agreed to.

The question was put on agreeing to the substitute.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22. NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment,

RULE 36--0.

The substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2262, on motion of Senator Deeds, was passed with its title.

Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1778 (one thousand seven hundred seventy-eight) was read by title the third time.

The following amendments proposed by the Committee on Local Government were offered:

1. Line 16, engrossed, after clutter,

insert

except on land zoned for or in active farming operation,

2. Line 18, engrossed, after *clutter*, insert

except on land zoned for or in active farming operation,

3. Line 25, engrossed, after clutter,

insert

except on land zoned for or in active farming operation,

The reading of the amendments was waived.

On motion of Senator Lewis, the amendments were agreed to.

The amendments were ordered to be engrossed.

Senator Lewis moved that **H.B. 1778** be passed with its title.

The question was put on passing **H.B.** 1778 with its title.

H.B. 1778 was defeated with its title.

The recorded vote is as follows:

YEAS--19. NAYS--20. RULE 36--0.

YEAS--Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Saslaw, Spruill--19.

NAYS--Barker, Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Surovell, Vogel--20. RULE 36--0.

- **H.B. 1796** (one thousand seven hundred ninety-six), on motion of Senator Surovell, was passed by temporarily.
- **H.B. 1852** (one thousand eight hundred fifty-two) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Saslaw, Spruill, Suetterlein, Surovell, Vogel--34.

NAYS--Ruff, Stuart--2.

RULE 36--0.

H.B. 1895 (one thousand eight hundred ninety-five) was read by title the third time.

The following amendment proposed by the Committee on the Judiciary was offered:

1. Line 43, engrossed, after agreements.

strike

the remainder of line 43 and through agreement. on line 44

The reading of the amendment was waived.

On motion of Senator Edwards, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 1895, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--19. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Spruill, Surovell--19.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1899 (one thousand eight hundred ninety-nine) was read by title the third time and, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

- **H.B. 1911** (one thousand nine hundred eleven), on motion of Senator Surovell, was passed by for the day.
 - H.B. 1919 (one thousand nine hundred nineteen) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Local Government was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding a section numbered 15.2-958.3:1, relating to local green banks.

The reading of the substitute was waived.

On motion of Senator Lewis, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1919, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Ruff, Saslaw, Spruill, Surovell, Vogel--25.

NAYS--Chase, DeSteph, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Stanley, Stuart, Suetterlein--13.

RULE 36--0.

H.B. 1936 (one thousand nine hundred thirty-six) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 18.2-58 of the Code of Virginia, relating to robbery; penalties.

The reading of the substitute was waived.

On motion of Senator Edwards, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1936, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1969 (one thousand nine hundred sixty-nine) was read by title the third time and, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

H.B. 1999 (one thousand nine hundred ninety-nine) was read by title the third time and, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

- H.B. 2042 (two thousand forty-two), on motion of Senator Surovell, was passed by temporarily.
- H.B. 2054 (two thousand fifty-four), on motion of Senator Surovell, was passed by temporarily.
- H.B. 2081 (two thousand eighty-one), on motion of Senator Surovell, was passed by for the day.
- **H.B. 2123** (two thousand one hundred twenty-three) was read by title the third time and, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 2138 (two thousand one hundred thirty-eight) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--19. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--19.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

- **H.B. 2174** (two thousand one hundred seventy-four), on motion of Senator Stanley, was passed by for the day.
- **H.B. 2194** (two thousand one hundred ninety-four) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--34.

NAYS--Obenshain, Peake, Suetterlein--3.

RULE 36--0.

- H.B. 2201 (two thousand two hundred one), on motion of Senator Favola, was passed by temporarily.
- H.B. 2206 (two thousand two hundred six) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 63.2-1911, as it is currently effective, of the Code of Virginia and to temporarily expand the Child Care Subsidy Program to provide financial assistance for child care to families in need during the public health emergency; emergency.

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2206, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

- **H.B. 2217** (two thousand two hundred seventeen), on motion of Senator Favola, was passed by temporarily.
- **H.B. 2261** (two thousand two hundred sixty-one) was read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--6. RULE 36--1.

YEAS--Bell, Boysko, Chase, Deeds, DeSteph, Dunnavant, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Vogel--31.

NAYS--Barker, Ebbin, McPike, Petersen, Suetterlein, Surovell--6.

RULE 36--Cosgrove--1.

- **H.B. 2263** (two thousand two hundred sixty-three), on motion of Senator Stanley, was passed by for the day.
 - H.B. 2014 (two thousand fourteen) was taken up.

Senator Barker withdrew amendment No. 2.

H.B. 2014, on motion of Senator Barker, was passed by for the day.

RECONSIDERATION

Senator Barker moved to reconsider the vote by which **H.B.** 1778 (one thousand seven hundred seventy-eight) was defeated with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

- H.B. 1778, on motion of Senator Barker, was passed by for the day.
- **H.B.** 1796 (one thousand seven hundred ninety-six) was taken up, read by title the third time and, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--28. NAYS--9. RULE 36--1.

YEAS--Bell, Boysko, Chase, DeSteph, Dunnavant, Edwards, Favola, Hanger, Hashmi, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Vogel--28.

NAYS--Barker, Deeds, Ebbin, Howell, McPike, Morrissey, Petersen, Suetterlein, Surovell--9. RULE 36--Cosgrove--1.

H.B. 2042 (two thousand forty-two) was taken up and read by title the third time.

The following amendments proposed by the Committee on Local Government were offered:

1. Line 66, engrossed, after locality's

strike

Municipal Separate Storm Sewer System

insert

municipal separate storm sewer system

2. Line 67, engrossed, after Area

insert

to address recurrent flooding

3. Line 167, engrossed, after plans.

strike

Any

insert

If a locality is located within a nonattainment area, any

4. Line 283, engrossed, after Area

insert

to address recurrent flooding

5. Line 290, engrossed

strike

all of line 290

insert

2. That the Secretary of Natural Resources and Secretary of Agriculture and Forestry (the Secretaries) shall convene a stakeholder work group (the Work Group) for the purpose of developing and providing recommendations to state and local governments related to policies that encourage the conservation of mature trees and tree cover on sites being developed, increase tree canopy cover in communities, and encourage the planting of trees. The Work Group shall also examine the Commonwealth's existing enabling statutes and their use related to the preservation, planting, and replacement of trees during the land development process, including §§ 15.2-961 and 15.2-961.1 of the Code of Virginia and the amendments to such sections provided in the first enactment of this act, and recommend amendments to those statutes or the adoption of new Code sections that would enhance the preservation, planting, and replacement of trees during the land development process and increase incentives for the preservation, planting, and replacement of trees during the land development process. The Work Group shall be composed of representatives of the residential and commercial development industries, representatives of agricultural and forestry industries, professional environmental technical experts, representatives of environmental and conservation organizations, representatives of local governments, solar developers, and other affected parties so that the various stakeholders are represented in the Work Group. No later than October 1, 2021, the Secretaries shall provide a report containing the Work Group's detailed findings, recommendations, and draft legislation to encourage the conservation of tree cover and mature trees, and the planting of trees, to the Chairmen of the House Committee on Agriculture, Chesapeake and Natural Resources, the House Committee on Counties, Cities and Towns,

the Senate Committee on Agriculture, Conservation and Natural Resources, and the Senate Committee on Local Government.

3. That the provisions of the first enactment of this act shall not become effective unless reenacted by the 2022 Session of the General Assembly.

The reading of the amendments was waived.

On motion of Senator Lewis, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 2042, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--11. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--26.

NAYS--Chase, DeSteph, McDougle, Newman, Obenshain, Peake, Reeves, Ruff, Stanley, Stuart, Suetterlein--11.

RULE 36--0.

H.B. 2054 (two thousand fifty-four) was taken up, read by title the third time and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--16.

RULE 36--0.

H.B. 2201 (two thousand two hundred one) was taken up, read by title the third time and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--3. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--34.

NAYS--Chase, DeSteph, Reeves--3.

RULE 36--Suetterlein--1.

H.B. 2217 (two thousand two hundred seventeen) was taken up, read by title the third time and, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows: YEAS--33. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, Morrissey, Newman, Norment, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Vogel--33.

NAYS--McDougle, McPike, Obenshain, Petersen, Suetterlein, Surovell--6. RULE 36--0.

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- H.B. 1807 (one thousand eight hundred seven).
- H.B. 1829 (one thousand eight hundred twenty-nine).
- H.B. 1834 (one thousand eight hundred thirty-four).
- H.B. 1892 (one thousand eight hundred ninety-two).
- H.B. 1907 (one thousand nine hundred seven).
- H.B. 1923 (one thousand nine hundred twenty-three).
- H.B. 1942 (one thousand nine hundred forty-two).
- H.B. 1994 (one thousand nine hundred ninety-four).
- H.B. 2008 (two thousand eight).
- H.B. 2034 (two thousand thirty-four).
- H.B. 2036 (two thousand thirty-six).
- H.B. 2282 (two thousand two hundred eighty-two).
- H.B. 2304 (two thousand three hundred four).
- H.B. 1896 (one thousand eight hundred ninety-six).
- H.B. 1965 (one thousand nine hundred sixty-five).

The motion was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

- H.B. 1807 (one thousand eight hundred seven).
- H.B. 1829 (one thousand eight hundred twenty-nine).
- H.B. 1834 (one thousand eight hundred thirty-four).
- H.B. 1892 (one thousand eight hundred ninety-two).
- H.B. 1907 (one thousand nine hundred seven).
- H.B. 1923 (one thousand nine hundred twenty-three).
- H.B. 1942 (one thousand nine hundred forty-two).
- H.B. 1994 (one thousand nine hundred ninety-four).

- H.B. 2008 (two thousand eight).
- H.B. 2034 (two thousand thirty-four).
- H.B. 2036 (two thousand thirty-six).
- H.B. 2282 (two thousand two hundred eighty-two).
- H.B. 2304 (two thousand three hundred four).
- H.B. 1896 (one thousand eight hundred ninety-six).
- H.B. 1965 (one thousand nine hundred sixty-five).

SUPPLEMENTAL CALENDAR NO. 1

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- H.B. 1818 (one thousand eight hundred eighteen).
- H.B. 1820 (one thousand eight hundred twenty).
- H.B. 1893 (one thousand eight hundred ninety-three).
- H.B. 1925 (one thousand nine hundred twenty-five).
- H.B. 1985 (one thousand nine hundred eighty-five).
- H.B. 1989 (one thousand nine hundred eighty-nine).
- H.B. 1995 (one thousand nine hundred ninety-five).
- H.B. 2078 (two thousand seventy-eight).
- H.B. 2101 (two thousand one hundred one).
- H.B. 2110 (two thousand one hundred ten).
- H.B. 2111 (two thousand one hundred eleven).
- H.B. 2117 (two thousand one hundred seventeen).
- H.B. 2133 (two thousand one hundred thirty-three).
- H.B. 2165 (two thousand one hundred sixty-five).
- H.B. 2166 (two thousand one hundred sixty-six).
- H.B. 2167 (two thousand one hundred sixty-seven).
- H.B. 2207 (two thousand two hundred seven).
- H.B. 2212 (two thousand two hundred twelve).
- H.B. 2219 (two thousand two hundred nineteen).
- H.B. 2223 (two thousand two hundred twenty-three).
- H.B. 2230 (two thousand two hundred thirty).
- H.B. 2252 (two thousand two hundred fifty-two).
- H.B. 2269 (two thousand two hundred sixty-nine).H.B. 2293 (two thousand two hundred ninety-three).
- H.B. 2323 (two thousand three hundred twenty-three).
- H.B. 1833 (one thousand eight hundred thirty-three).
- H.B. 1884 (one thousand eight hundred eighty-four).
- H.B. 1902 (one thousand nine hundred two).
- H.B. 1912 (one thousand nine hundred twelve).
- H.B. 1979 (one thousand nine hundred seventy-nine).
- H.B. 1992 (one thousand nine hundred ninety-two).
- H.B. 1993 (one thousand nine hundred ninety-three).
- H.B. 2032 (two thousand thirty-two).
- H.B. 2040 (two thousand forty).
- H.B. 2047 (two thousand forty-seven).
- H.B. 2118 (two thousand one hundred eighteen).
- H.B. 2159 (two thousand one hundred fifty-nine).

```
H.B. 2218 (two thousand two hundred eighteen).
```

H.B. 2307 (two thousand three hundred seven).

H.B. 2321 (two thousand three hundred twenty-one).

H.B. 2330 (two thousand three hundred thirty).

H.B. 2332 (two thousand three hundred thirty-two).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

```
H.B. 1818 (one thousand eight hundred eighteen).
```

H.B. 1820 (one thousand eight hundred twenty).

H.B. 1893 (one thousand eight hundred ninety-three).

H.B. 1925 (one thousand nine hundred twenty-five).

H.B. 1985 (one thousand nine hundred eighty-five).

H.B. 1989 (one thousand nine hundred eighty-nine).

H.B. 1995 (one thousand nine hundred ninety-five).

H.B. 2078 (two thousand seventy-eight).

H.B. 2101 (two thousand one hundred one).

H.B. 2110 (two thousand one hundred ten).

H.B. 2111 (two thousand one hundred eleven).

H.B. 2117 (two thousand one hundred seventeen).

H.B. 2133 (two thousand one hundred thirty-three).

H.B. 2165 (two thousand one hundred sixty-five). **H.B. 2166** (two thousand one hundred sixty-six).

H.B. 2167 (two thousand one hundred sixty-seven).

H.B. 2207 (two thousand two hundred seven).

H.B. 2212 (two thousand two hundred twelve).

H.B. 2219 (two thousand two hundred nineteen).

H.B. 2223 (two thousand two hundred twenty-three).

H.B. 2230 (two thousand two hundred thirty).

H.B. 2252 (two thousand two hundred fifty-two).

H.B. 2269 (two thousand two hundred sixty-nine).

H.B. 2293 (two thousand two hundred ninety-three).

H.B. 2323 (two thousand three hundred twenty-three).

H.B. 1833 (one thousand eight hundred thirty-three).

H.B. 1884 (one thousand eight hundred eighty-four).

H.B. 1902 (one thousand nine hundred two).

H.B. 1912 (one thousand nine hundred twelve).

H.B. 1979 (one thousand nine hundred seventy-nine).

H.B. 1992 (one thousand nine hundred ninety-two).

H.B. 1993 (one thousand nine hundred ninety-three).

- H.B. 2032 (two thousand thirty-two).
- H.B. 2040 (two thousand forty).
- H.B. 2047 (two thousand forty-seven).
- H.B. 2118 (two thousand one hundred eighteen).
- H.B. 2159 (two thousand one hundred fifty-nine).
- H.B. 2218 (two thousand two hundred eighteen).
- H.B. 2307 (two thousand three hundred seven).
- H.B. 2321 (two thousand three hundred twenty-one).
- H.B. 2330 (two thousand three hundred thirty).
- H.B. 2332 (two thousand three hundred thirty-two).

HOUSE JOINT RESOLUTION ON SECOND READING

H.J.R. 542 (five hundred forty-two) was read by title the second time.

LEGISLATION SIGNED BY PRESIDING OFFICER

The President of the Senate as required by Article IV, Section 11, of the Constitution, on the date recorded below, signed the following bills that had been passed by both houses and duly enrolled:

February 17, 2021

- **H.B. 1737.** An Act to amend and reenact § 54.1-2957 of the Code of Virginia, relating to nurse practitioners; practice without a practice agreement.
- **H.B. 1790.** An Act to amend and reenact § 22.1-98 of the Code of Virginia, relating to public schools; severe weather conditions and other emergency situations; unscheduled remote learning days.
- **H.B. 1798.** An Act to amend and reenact § 22.1-32 of the Code of Virginia, relating to Brunswick County school board; appointed school board salaries.
- **H.B. 1812.** An Act to amend and reenact §§ 2.2-3711, 58.1-4100, 58.1-4109, 58.1-4110, 58.1-4114, 58.1-4122, 58.1-4124, and 58.1-4125 of the Code of Virginia, relating to casino gaming; technical amendments.
- **H.B. 1814.** An Act to amend and reenact § 34-29 of the Code of Virginia, relating to garnishment of wages; protected portion of disposable earnings.
- **H.B. 1816.** An Act to amend and reenact §§ 55.1-1800, 55.1-1815, 55.1-1816, 55.1-1832, 55.1-1900, 55.1-1935, 55.1-1949, 55.1-1952, and 55.1-1953 of the Code of Virginia, relating to the Property Owners' Association Act; the Condominium Act; use of electronic means for meetings and voting.
- **H.B. 1821.** An Act to amend and reenact § 18.2-251.03 of the Code of Virginia, relating to arrest and prosecution when experiencing or reporting overdoses.
- **H.B. 1824.** An Act to amend and reenact § 55.1-703 of the Code of Virginia, relating to the Virginia Residential Property Disclosure Act; required disclosures for buyer to beware; mold.
- **H.B. 1827.** An Act to amend and reenact § 22.1-9 of the Code of Virginia, relating to the Board of Education; membership; geographic representation.

- **H.B. 1830.** An Act to amend and reenact § 2.2-2282 of the Code of Virginia, relating to the Virginia Small Business Financing Authority; Board of Directors; membership; small business lending experience.
- **H.B. 1848.** An Act to amend and reenact §§ 2.2-3902, 2.2-3905, and 51.5-41 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-3905.1, relating to the Virginia Human Rights Act; discrimination on the basis of disability.
- H.B. 1849. An Act related to apprenticeship training programs; report.
- **H.B. 1873.** An Act to amend and reenact § 37.2-403 of the Code of Virginia, relating to brain injury; definition.
- **H.B. 1878.** An Act to amend and reenact §§ 16.1-256 and 16.1-260 of the Code of Virginia, relating to juvenile intake and petition; appeal to a magistrate on a finding of no probable cause.
- H.B. 1882. An Act to amend and reenact § 55.1-319 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 55.1-318.1, relating to deeds of trust; amendment to loan document; statement of interest rate of a refinanced mortgage.
- H.B. 1885. An Act to require the Department of Education to perform a comprehensive review of computer science standards, courses, and course pathways in public schools; report.
- H.B. 1891. An Act to amend and reenact § 2.2-1201 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 12 of Title 2.2 a section numbered 2.2-1212, relating to the Department of Human Resource Management; duties of the Department; annual safety and disaster awareness training.
- **H.B. 1904.** An Act to amend and reenact §§ 22.1-253.13:5 and 22.1-298.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-298.7, relating to teachers and other licensed school board employees; cultural competency.
- **H.B. 1905.** An Act to amend and reenact § 22.1-200.03 of the Code of Virginia, relating to economic education and financial literacy required in middle and high school grades; employment arrangements.
- **H.B. 1913.** An Act to amend and reenact §§ 8.01-581.16, 54.1-2400.6, and 54.1-2909 of the Code of Virginia, relating to programs to address career fatigue and wellness in certain health care providers; civil immunity.

EMERGENCY

- **H.B. 1931.** An Act to amend and reenact § 2.2-3708.2 of the Code of Virginia, relating to the Virginia Freedom of Information Act; electronic meetings.
- **H.B. 1943.** An Act to amend and reenact §§ 18.2-340.19 and 18.2-340.28 of the Code of Virginia, relating to the Charitable Gaming Board; regulations; electronic pull tabs.
- **H.B. 1944.** An Act to amend and reenact § 58.1-4110 of the Code of Virginia, relating to casino gaming; requirements for issuance of operator's license; human trafficking training.
- **H.B. 1967.** An Act to amend and reenact § 2.2-2240.3 of the Code of Virginia, relating to the Virginia Jobs Investment Program and Fund; minimum wage requirements.

- **H.B. 1971.** An Act to amend and reenact § 36-96.3:2 of the Code of Virginia, relating to the Virginia Fair Housing Law; reasonable accommodations; disability-related requests for parking.
- **H.B. 1998.** An Act to amend and reenact § 22.1-137.2 of the Code of Virginia, relating to public schools; lock-down drills; annual requirement.
- **H.B. 2009.** An Act to revert certain property upon which the Chamberlin Hotel at Fort Monroe is located to the Commonwealth and to repeal § 1 of Chapter 809 of the Acts of Assembly of 1998, as amended by Chapter 713 of the Acts of Assembly of 2004.
- **H.B. 2085.** An Act to amend and reenact § 44-146.19 of the Code of Virginia, relating to Emergency Services and Disaster Law; local and interjurisdictional emergency operations plans.
- **H.B. 2119.** An Act to amend and reenact § 22.1-205 of the Code of Virginia, relating to student driver education program; parent/student component exemption.
- **H.B. 2128.** An Act to amend and reenact § 18.2-308.2:2, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to sale and transfer of firearms; criminal history record information checks.
- **S.B. 1111.** An Act to amend and reenact § 24.2-606 of the Code of Virginia, relating to elections; preservation of order at the polls; powers of officers of election.
- **S.B. 1112.** An Act to amend and reenact §§ 58.1-439.12:08 and 58.1-439.12:11 of the Code of Virginia, relating to research and development tax credits.
- **S.B. 1134.** An Act to amend and reenact §§ 2 and 4 of the first enactment of Chapter 265 and §§ 2 and 4 of the first enactment of Chapter 408 of the Acts of Assembly of 1992, relating to the issuance of Commonwealth of Virginia Article X, Section 9 (c) Refunding Bonds, subject to the provisions of Article X, Sections 9 (a) and 9 (c) of the Constitution of Virginia; emergency.

EMERGENCY

- **S.B. 1143.** An Act to extend certain wetlands permits through 2021.
- **S.B. 1145.** An Act to authorize the issuance of bonds, in an amount up to \$34,136,000 plus financing costs, pursuant to Article X, Section 9 (c) of the Constitution of Virginia, for paying costs of acquiring, constructing and equipping revenue-producing capital projects at institutions of higher learning of the Commonwealth; to authorize the Treasury Board, by and with the consent of the Governor, to fix the details of such bonds, to provide for the sale of such bonds, and to issue notes to borrow money in anticipation of the issuance of the bonds; to provide for the pledge of the net revenues of such capital projects and the full faith, credit and taxing power of the Commonwealth for the payment of such bonds; to provide that the interest income on such bonds and notes shall be exempt from all taxation by the Commonwealth and any political subdivision thereof; emergency.

EMERGENCY

- **S.B. 1156.** An Act to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 22.17, consisting of a section numbered 59.1-284.38, relating to creation of the Technology Development Grant Fund
- **S.B. 1157.** An Act to amend and reenact § 15.2-1400 of the Code of Virginia, relating to time of certain local elections.

- **S.B. 1161.** An Act to amend and reenact § 10.1-502 of the Code of Virginia, relating to Soil and Water Conservation Board; membership.
- **S.B. 1162.** An Act to amend and reenact §§ 58.1-339.3 and 58.1-439.5 of the Code of Virginia, relating to tax credit; agricultural best management practices.
- **S.B. 1175.** An Act to amend and reenact § 22.1-32 of the Code of Virginia, relating to Brunswick County school board; appointed school board salaries.
- **S.B. 1194.** An Act to amend and reenact the second enactment of Chapter 574 of the Acts of Assembly of 2017, relating to produce safety; sunset.
- **S.B. 1196.** An Act to amend and reenact §§ 22.1-253.13:5 and 22.1-298.1 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-298.7, relating to teachers and other licensed school board employees; cultural competency.
- **S.B. 1199.** An Act to amend the Code of Virginia by adding in Chapter 10.1 of Title 10.1 a section numbered 10.1-1016.1 and by adding in Chapter 17 of Title 10.1 a section numbered 10.1-1705.1, relating to conservation easements; construction.
- **S.B. 1201.** An Act to amend and reenact §§ 58.1-2600, 58.1-2628, 58.1-2636, and 58.1-3660, relating to tax exemptions for energy storage systems.
- **S.B. 1210.** An Act to direct the Department of Environmental Quality to convene working groups to revise permit fee schedules.
- S.B. 1251. An Act to amend and reenact §§ 51.1-124.3, 51.1-157, 51.1-168, 51.1-301, and 51.1-308 of the Code of Virginia, relating to Virginia Retirement System; technical amendments.
- **S.B. 1252.** An Act to amend and reenact §§ 58.1-433.1, 58.1-439.2, and 58.1-2626.1 of the Code of Virginia, relating to sunset of coal tax credits.
- **S.B. 1280.** An Act to amend the Code of Virginia by adding in Article 2 of Chapter 6 of Title 10.1 a section numbered 10.1-613.6, relating to dams; negotiated settlement agreements.
- **S.B. 1281.** An Act to amend and reenact § 24.2-110 of the Code of Virginia, relating to elections; qualifications of the general registrar, residency.
- **S.B. 1387.** An Act to amend and reenact § 23.1-506 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 23.1-505.1, relating to the State Council of Higher Education for Virginia; eligibility for in-state tuition.
- S.B. 1395. An Act to amend and reenact §§ 24.2-105, as it shall become effective, 24.2-306, 24.2-649, 24.2-1000, 24.2-1005, and 24.2-1005.1 of the Code of Virginia; to amend the Code of Virginia by adding a section numbered 24.2-104.1, by adding a section numbered 24.2-1005.2, and by adding in Title 24.2 a chapter numbered 1.1, consisting of sections numbered 24.2-125 through 24.2-131; and to repeal § 24.2-124, as it shall become effective, of the Code of Virginia, relating to elections; prohibited discrimination in voting and elections administration; required process for enacting certain covered practices; civil causes of action; penalties.
- **S.B. 1402.** An Act to amend and reenact § 29.1-311 of the Code of Virginia, relating to trout fishing in stocked waters.

- **S.B. 1403.** An Act to amend the Code of Virginia by adding a section numbered 58.1-609.14, relating to sales tax; exemption for personal protective equipment; emergency. EMERGENCY
- **S.B. 1404.** An Act to amend and reenact § 62.1-44.15:29.1 of the Code of Virginia, relating to Stormwater Local Assistance Fund; grant requirements.
- **S.B. 1405.** An Act to amend the Code of Virginia by adding a section numbered 23.1-2911.2, relating to the establishment of the Get Skilled, Get a Job, Give Back (G3) Fund and Program.
- **S.B. 1411.** An Act to amend and reenact § 3.2-1905 of the Code of Virginia, relating to the excise tax on peanuts.

On motion of Senator Lucas, the Senate adjourned until tomorrow at 12 m.

ssan Carle Delsan

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

THURSDAY, FEBRUARY 18, 2021

The Senate met at 12 m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Stephen D. Newman, Twenty-third Senatorial District, offered the following prayer:

In Matthew 6, Christ told us how to pray. Some of you may join me.

Our Father, which art in heaven, hallowed be Thy name. Thy kingdom come, Thy will be done, on Earth as it is in heaven. Give us this day our daily bread and forgive us our debts as we forgive our debtors. And lead us not into temptation, but deliver us from evil. For thine is the kingdom, the power, and the glory forever. Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Corinne Sloan, Assistant Journal Clerk, Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell.

A quorum was present.

After the roll call, Senator Vogel notified the Clerk of her presence.

On motion of Senator Surovell, the reading of the Journal was waived.

The recorded vote is as follows:

YEAS--32. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stuart, Suetterlein, Surovell--32.

NAYS--Deeds, DeSteph, McDougle, Petersen, Stanley--5. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 17, 2021

THE HOUSE OF DELEGATES HAS PASSED WITH AMENDMENTS THE FOLLOWING SENATE BILLS:

- **S.B. 1242.** A BILL to amend and reenact § 19.2-3.1 of the Code of Virginia, relating to personal appearance by two-way electronic video and audio communication; entry of plea or nolle prosequi; adjudication of probation violations.
- **S.B. 1399.** A BILL to amend and reenact §§ 15.2-5500, 15.2-5505, 15.2-5506, and 45.1-246 of the Code of Virginia, relating to the Tourism Development Authority; name change.

- **S.B. 1415.** A BILL to amend and reenact §§ 16.1-253 and 16.1-253.2 of the Code of Virginia, relating to violations of protective orders; preliminary child protective order.
- **S.B. 1456.** A BILL to amend and reenact §§ 16.1-248.1, 16.1-249, 16.1-278.7, and 16.1-278.8 of the Code of Virginia, relating to juveniles; eligibility for commitment to the Department of Juvenile Justice; eligibility for predispositional confinement in a secure facility.
- **S.B. 1465.** A BILL to amend and reenact § 18.2-325, as it is currently effective and as it shall become effective, of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-331.1, relating to illegal gambling; skills games; civil penalty; enforcement by localities and Attorney General.
- **S.B. 1475.** A BILL to amend and reenact § 19.2-56 of the Code of Virginia, relating to execution of search warrants; emergency. EMERGENCY

THE HOUSE OF DELEGATES HAS PASSED WITH SUBSTITUTES THE FOLLOWING SENATE BILLS:

- **S.B. 1104.** A BILL to amend and reenact §§ 53.1-136 and 53.1-155 of the Code of Virginia, relating to parole; notice and certification; monthly reports.
- **S.B. 1113.** A BILL to amend and reenact § 18.2-60 of the Code of Virginia, relating to communicating threats of death or bodily injury to a person with intent to intimidate; penalty.
- **S.B. 1297.** A BILL to amend and reenact §§ 63.2-1603 and 63.2-1609 of the Code of Virginia, relating to emergency order for adult protective services; acts of violence, force, or threat or financial exploitation; penalty.
- **S.B. 1336.** A BILL to amend the Code of Virginia by adding a section numbered 18.2-271.5, relating to restricted permits to operate a motor vehicle; ignition interlock systems.
- **S.B. 1397.** A BILL to amend and reenact § 53.1-136 of the Code of Virginia, relating to conditional release of geriatric or terminally ill inmates; notice to attorney for the Commonwealth.

THE HOUSE OF DELEGATES HAS REJECTED THE AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1811.** A BILL to amend the Code of Virginia by adding a section numbered 2.2-4328.1, relating to the Virginia Public Procurement Act; preference for energy-efficient and water-efficient goods.
- **H.B. 2168.** A BILL to amend and reenact § 18.2-325, as it is currently effective and as it shall become effective, of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-331.1, relating to illegal gambling; skills games; civil penalty; enforcement by localities and Attorney General.

THE HOUSE OF DELEGATES HAS REJECTED THE SUBSTITUTE PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILL:

H.B. 1847. A BILL to amend and reenact §§ 58.1-4030, 58.1-4031, 58.1-4032, 58.1-4039, and 58.1-4100 of the Code of Virginia, relating to sports betting; technical amendments.

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING SENATE BILLS:

- **S.B. 1122.** A BILL to amend and reenact §§ 8.01-9, 8.01-407, 16.1-77, 16.1-305, 17.1-213, 19.2-389, as it is currently effective and as it shall become effective, 46.2-301, 46.2-301.1, 46.2-411, and 53.1-21 of the Code of Virginia and to repeal Article 9 (§§ 46.2-355.1 through 46.2-363) of Chapter 3 of Title 46.2 of the Code of Virginia, relating to habitual offenders; repeal.
- **S.B. 1128.** A BILL to amend and reenact § 44 and § 133, as amended, of Chapter 34 of the Acts of Assembly of 1918 and to repeal § 61 of Chapter 34 of the Acts of Assembly of 1918, which provided a charter for the City of Norfolk, relating to general updates.
- **S.B. 1152.** A BILL to amend and reenact §§ 1 and 4 of the charter of the Town of Appomattox, which was granted by order of the Circuit Court of the County of Appomattox on June 2, 1925, and as amended by Chapter 43 of the Acts of Assembly of 1980, relating to election and appointment of officers; time of election.
- **S.B. 1206.** A BILL to amend and reenact § 16.1-300 of the Code of Virginia, relating to confidentiality of juvenile records; exceptions.
- **S.B. 1213.** A BILL to amend and reenact § 18.2-271.1 of the Code of Virginia, relating to driver's license suspensions; restricted licenses; drug offenses.
- **S.B. 1216.** A BILL to amend and reenact §§ 3.1, as amended, 3.2, 3.7, as amended, and 5.1 of Chapter 669 of the Acts of Assembly of 1972, which provided a charter for the Town of Crewe in Nottoway County, and to repeal Chapter 6 (§§ 6.1 through 6.4) of Chapter 669 of the Acts of Assembly of 1972, relating to town council; elections and powers.
- **S.B. 1248.** A BILL to amend and reenact § 16.1-356 of the Code of Virginia, relating to juveniles; competency evaluation; receipt of court order.
- **S.B. 1256.** A BILL to amend and reenact §§ 9.1-102, 9.1-108, and 9.1-112, as they shall become effective, of the Code of Virginia, relating to membership on Criminal Justice Services Board and Committee on Training; law-enforcement training.
- **S.B. 1267.** A BILL to amend and reenact §§ 2.02, 5.021, and 6.02, § 12.01, as amended, and § 15.03 of Chapter 227, as amended, of the Acts of Assembly of 1954, which provided a charter for the City of Covington, and to amend and reenact § 22.1-32 of the Code of Virginia, relating to consolidated school board of Alleghany County and the City of Covington; school board salaries.
- **S.B. 1272.** A BILL to amend the Code of Virginia by adding a section numbered 19.2-169.3:1, relating to disposition of the unrestorably incompetent defendant; capital murder charge; inpatient custody of the Commissioner of the Department of Behavioral Health and Developmental Services.
- **S.B. 1296.** A BILL to amend and reenact § 44-146.18 of the Code of Virginia, relating to the State Coordinator of Emergency Management; establishment of Emergency Management Equity Working Group.
- **S.B. 1298.** A BILL to amend the Code of Virginia by adding in Chapter 24 of Title 15.2 an article numbered 3, consisting of sections numbered 15.2-2413.1 through 15.2-2413.11, relating to tourism improvement districts.

- **S.B. 1300.** A BILL to direct the Board of Local and Regional Jails to review services provided to immates during pregnancy, pregnancy termination, labor and delivery, and postpartum recovery; report.
- **S.B. 1309.** A BILL to amend and reenact § 15.2-2114.01 of the Code of Virginia, relating to local stormwater assistance; flood mitigation and protection.
- **S.B. 1400.** A BILL to authorize a quitclaim and release of interest and the conveyance of an easement by the Board of Wildlife Resources in Tazewell County.
- **S.B. 1426.** A BILL to amend and reenact §§ 19.2-305.1, 19.2-305.2, 19.2-349, and 19.2-354 of the Code of Virginia, relating to orders of restitution; enforcement.
- **S.B. 1431.** A BILL to amend and reenact § 19.2-169.1 of the Code of Virginia, relating to unrestorably incompetent defendant; competency report.
- **S.B. 1457.** A BILL to amend and reenact § 15.2-2306 of the Code of Virginia, relating to preservation of historic sites.

THE HOUSE OF DELEGATES HAS AGREED TO THE AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1842.** A BILL to amend and reenact §§ 55.1-1819 and 55.1-1959 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 55.1-1819.1 and 55.1-1960.1, relating to the Property Owners' Association Act; the Condominium Act; rulemaking authority of property owners' associations and unit owners' associations; smoking.
- **H.B. 1940.** A BILL to amend and reenact § 22.1-254 of the Code of Virginia, relating to the Department of Education; guidelines on excused student absences; civic engagement.
- H.B. 1981. A BILL to amend and reenact § 55.1-1229 of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; access to dwelling unit during certain declared states of emergency.
- **H.B. 2012.** A BILL to amend and reenact §§ 16.1-253 and 16.1-253.2 of the Code of Virginia, relating to violations of protective orders; preliminary child protective order.
- **H.B. 2046.** A BILL to amend and reenact §§ 36-96.3 and 36-96.17 of the Code of Virginia, relating to the Virginia Fair Housing Law; unlawful discriminatory housing practices.

THE HOUSE OF DELEGATES HAS AGREED TO THE SUBSTITUTES PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1808.** A BILL to amend and reenact § 37.2-304 of the Code of Virginia, relating to Commissioner of Behavioral Health and Developmental Services; reports to designated protection and advocacy system.
- **H.B. 1918.** A BILL to amend and reenact § 22.1-205 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-205.1, relating to student driver safety.
- **H.B. 2002.** A BILL to amend and reenact §§ 16.1-260 and 63.2-1903 of the Code of Virginia, relating to child support; health care coverage.

- **H.B. 2175.** A BILL to amend and reenact §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-2223.5, relating to housing protections; foreclosures; manufactured housing.
- **H.B. 2320.** A BILL to amend and reenact § 55.1-703 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 55.1-708.2, relating to property; required disclosures for buyer to exercise due diligence; flood risk report.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows: YEAS--33. NAYS--4. RULE 36--0.

YEAS--Barker, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--33.

NAYS--Deeds, DeSteph, McDougle, Petersen--4. RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Lucas from the Committee on Education and Health:

- H.B. 1747 (one thousand seven hundred forty-seven).
- H.B. 1823 (one thousand eight hundred twenty-three).
- H.B. 1831 (one thousand eight hundred thirty-one).
- H.B. 1838 (one thousand eight hundred thirty-eight).
- H.B. 1865 (one thousand eight hundred sixty-five).
- H.B. 1894 (one thousand eight hundred ninety-four).
- H.B. 1909 (one thousand nine hundred nine) with amendments.
- H.B. 1930 (one thousand nine hundred thirty) with amendment.
- H.B. 1950 (one thousand nine hundred fifty).
- H.B. 1963 (one thousand nine hundred sixty-three).
- H.B. 1976 (one thousand nine hundred seventy-six).
- H.B. 1980 (one thousand nine hundred eighty).
- H.B. 1987 (one thousand nine hundred eighty-seven) with substitute.
- H.B. 1988 (one thousand nine hundred eighty-eight) with substitute.
- **H.B. 2007** (two thousand seven) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2013 (two thousand thirteen).
 - H.B. 2019 (two thousand nineteen) with substitute.
- **H.B. 2027** (two thousand twenty-seven) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2039 (two thousand thirty-nine).

- H.B. 2061 (two thousand sixty-one).
- H.B. 2079 (two thousand seventy-nine).
- H.B. 2086 (two thousand eighty-six).
- H.B. 2105 (two thousand one hundred five).
- H.B. 2116 (two thousand one hundred sixteen) with substitute.
- H.B. 2120 (two thousand one hundred twenty).
- **H.B. 2124** (two thousand one hundred twenty-four) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2135 (two thousand one hundred thirty-five).
 - H.B. 2154 (two thousand one hundred fifty-four).
 - H.B. 2162 (two thousand one hundred sixty-two).
 - H.B. 2176 (two thousand one hundred seventy-six).
 - H.B. 2220 (two thousand two hundred twenty).
 - H.B. 2238 (two thousand two hundred thirty-eight).
 - H.B. 2299 (two thousand two hundred ninety-nine) with substitute.
 - H.B. 2300 (two thousand three hundred) with amendments.
 - H.B. 2316 (two thousand three hundred sixteen).

The following bills, having been considered by the committee in session, were reported by Senator Barker from the Committee on General Laws and Technology:

- H.B. 1843 (one thousand eight hundred forty-three).
- H.B. 1876 (one thousand eight hundred seventy-six) with amendment.
- H.B. 1889 (one thousand eight hundred eighty-nine) with substitute.
- H.B. 1900 (one thousand nine hundred).
- H.B. 1908 (one thousand nine hundred eight).
- **H.B. 2001** (two thousand one) with amendment with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2004** (two thousand four) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2025 (two thousand twenty-five).
 - H.B. 2029 (two thousand twenty-nine) with substitute.
- **H.B. 2053** (two thousand fifty-three) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2072 (two thousand seventy-two).
- **H.B. 2074** (two thousand seventy-four) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2098 (two thousand ninety-eight).
 - H.B. 2130 (two thousand one hundred thirty).
 - H.B. 2147 (two thousand one hundred forty-seven).
- **H.B. 2227** (two thousand two hundred twenty-seven) with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2288 (two thousand two hundred eighty-eight) with amendment.
- **H.B. 2322** (two thousand three hundred twenty-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2327 (two thousand three hundred twenty-seven).

The following bills, having been considered by the committee in session, were recommended for rereferral by the Committee on General Laws and Technology:

- **H.B. 1864** (one thousand eight hundred sixty-four) with the recommendation that it be rereferred to the Committee on the Judiciary.
- **H.B. 2155** (two thousand one hundred fifty-five) with the recommendation that it be rereferred to the Committee on the Judiciary.

The following bills and joint resolution, having been considered by the committee in session, were reported by Senator Edwards from the Committee on the Judiciary:

- H.B. 1853 (one thousand eight hundred fifty-three).
- H.B. 1991 (one thousand nine hundred ninety-one) with amendments.
- **H.B. 2017** (two thousand seventeen) with amendments with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2018 (two thousand eighteen).
 - H.B. 2038 (two thousand thirty-eight) with substitute.
 - H.B. 2055 (two thousand fifty-five) with amendments.
- **H.B. 2099** (two thousand ninety-nine) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2113** (two thousand one hundred thirteen) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
- **H.B. 2132** (two thousand one hundred thirty-two) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2139 (two thousand one hundred thirty-nine) with substitute.
 - H.B. 2190 (two thousand one hundred ninety) with substitute.
 - H.B. 2192 (two thousand one hundred ninety-two).
 - H.B. 2193 (two thousand one hundred ninety-three) with amendment.
 - H.B. 2234 (two thousand two hundred thirty-four).
- **H.B. 2276** (two thousand two hundred seventy-six) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2295 (two thousand two hundred ninety-five) with substitute.
 - H.B. 2298 (two thousand two hundred ninety-eight).
 - **H.B. 2317** (two thousand three hundred seventeen).
- **H.B. 2331** (two thousand three hundred thirty-one) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.J.R. 629 (six hundred twenty-nine).
 - H.B. 1864 and H.B. 2155 were rereferred to the Committee on the Judiciary.
- H.B. 2001, H.B. 2004, H.B. 2007, H.B. 2017, H.B. 2027, H.B. 2053, H.B. 2074, H.B. 2099, H.B. 2113, H.B. 2124, H.B. 2132, H.B. 2227, H.B. 2276, H.B. 2322, and H.B. 2331 were rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, were presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.R. 520. Celebrating the life of Richard Blane Byrd.

Patron--Deeds

S.R. 521. Celebrating the life of Jamie Beletz.

Patron--McPike

S.R. 522. Celebrating the life of Thomas F. Cleary, M.D.

Patron--Surovell

S.R. 523. Celebrating the life of Carlton Farquhar Andrus.

Patron--Surovell

JOINT ORDER FOR ELECTIONS

Senator Edwards moved that, pursuant to **H.J.R. 5001** (five thousand one), the special and continuing joint order relating to judicial elections be suspended until Tuesday, February 23, 2021, at the conclusion of the morning hour.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

Senator Edwards was ordered to inform the House of Delegates thereof.

HOUSE COMMUNICATION

The following communication was received and read:

In the House of Delegates February 18, 2021

THE HOUSE OF DELEGATES HAS POSTPONED THE SPECIAL AND CONTINUING JOINT ORDER UNTIL TUESDAY, FEBRUARY 23, 2021, AT THE CONCLUSION OF THE MORNING HOUR, PURSUANT TO RULE 6 OF HJR 5001.

/s/ Suzette Denslow Clerk of the House of Delegates

CALENDAR

UNFINISHED BUSINESS—SENATE

S.B. 1212 (one thousand two hundred twelve) was taken up with the amendment proposed by the House of Delegates as follows:

1. Line 121, engrossed, after *therein* insert

. However, nothing in this subdivision shall be construed to provide the authority with the power of condemnation

On motion of Senator Edwards, the amendment was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

S.B. 1215 (one thousand two hundred fifteen) was taken up with the amendment proposed by the House of Delegates as follows:

1. Line 25, engrossed, after tenant's

strike

position

insert

petition

On motion of Senator Ebbin, the amendment was agreed to.

The recorded vote is as follows:

YEAS--24. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Peake, Petersen, Reeves, Saslaw, Spruill, Surovell--24.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Obenshain, Pillion, Ruff, Stanley, Stuart, Suetterlein--13.

RULE 36--0.

STATEMENT ON VOTE

Senator Kiggans stated that she voted nay on the question of agreeing to the amendment proposed by the House of Delegates to S.B. 1215, whereas she intended to vote yea.

- **S.B. 1247** (one thousand two hundred forty-seven) was taken up with the amendment in the nature of a substitute with amendment proposed by the House of Delegates, the title of the substitute, printed separately, and amendment being as follows:
- A BILL to amend and reenact § 56-599 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 26 of Title 45.1 a section numbered 45.1-394.1, relating to public disclosure of electric generating facility closures; integrated resource plans.
 - 1. Line 18, substitute, after pursuant

insert

to

On motion of Senator Deeds, the substitute with amendment was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

S.B. 1254 (one thousand two hundred fifty-four) was taken up with the amendments proposed by the House of Delegates as follows:

1. Line 217, engrossed, after 7.

insert

In issuing permits to operate sports betting platforms prior to July 1, 2025, the Director shall give substantial and preferred consideration to any applicant that demonstrates in its application (i) a description of any equity interest owned by minority individuals or minority-owned businesses, (ii) a detailed plan to achieve increased minority equity investment, (iii) a description of all efforts made to seek equity investment from minority individuals or minority-owned businesses, or (iv) a plan detailing efforts made to solicit participation of minority individuals or minority-owned businesses in the applicant's purchase of goods and services related to the sports betting platform or to provide assistance to a historically disadvantaged community or historically black colleges and universities located within the Commonwealth. As used in this subdivision, "historically black colleges and universities," "minority individual," and "minority-owned business" mean the same as those terms are defined in § 2.2-1604.

8.

2. Line 218, engrossed, after 5,

strike

or

3. Line 218, engrossed, after 6

insert

, or D 7

Senator McPike moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

The recorded vote is as follows: YEAS--0. NAYS--38. RULE 36--0.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

RULE 36--0.

- **S.B. 1327** (one thousand three hundred twenty-seven) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact §§ 8.01-463, 36-139, 55.1-320, 55.1-321, and 55.1-1303 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 15.2-2223.5, relating to housing protections; foreclosures; manufactured housing.

On motion of Senator McClellan, the substitute was agreed to.

The recorded vote is as follows:

YEAS--37. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--37.

NAYS--DeSteph--1.

RULE 36--0.

- **S.B. 1350** (one thousand three hundred fifty) was taken up with the amendment proposed by the House of Delegates as follows:
 - 1. Line 20, engrossed, after *be* insert

or the project sponsor has committed that the design will be

On motion of Senator Lewis, the amendment was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

S.B. 1366 (one thousand three hundred sixty-six) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend and reenact §§ 51.5-134 and 51.5-135 of the Code of Virginia, relating to aging services; economic and social need.

Senator Barker moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows:

YEAS--3. NAYS--36. RULE 36--0.

YEAS--McClellan, Petersen, Saslaw--3.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

RULE 36--0.

STATEMENT ON VOTE

Senator Saslaw stated that he voted yea on the question of agreeing to the substitute proposed by the House of Delegates to **S.B. 1366**, whereas he intended to vote nay.

- **S.B. 1374** (one thousand three hundred seventy-four) was taken up with the amendments proposed by the House of Delegates as follows:
 - 1. Line 17, engrossed, after from

insert

the University of Virginia,

2. Line 21, engrossed, after *Foundation*, insert

Shellfish Growers of Virginia,

On motion of Senator Lewis, the amendments were agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

- **S.B. 1375** (one thousand three hundred seventy-five) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability from COVID-19.

Senator Saslaw moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows: YEAS--7. NAYS--32. RULE 36--0.

YEAS--Bell, Cosgrove, Dunnavant, McClellan, McDougle, Obenshain, Vogel--7.

NAYS--Barker, Boysko, Chase, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McPike, Morrissey, Newman, Norment, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--32. RULE 36--0.

S.B. 1406 (one thousand four hundred six) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 4.1-101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-122, 4.1-124, as it is currently effective and as it shall become effective, 4.1-128, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-202, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-390, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29 consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6 consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, consisting of chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Title 19.2 a chapter numbered 23.2, consisting of sections numbered 19.2-392.5 through 19.2-392.13, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

Senator Ebbin moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows:

YEAS--0. NAYS--39. RULE 36--0.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

RULE 36--0.

RECONSIDERATION

Senator Ebbin moved to reconsider the vote by which the amendment proposed by the House of Delegates to **S.B. 1215** (one thousand two hundred fifteen) was agreed to.

The motion was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

On motion of Senator Ebbin, the amendment was agreed to.

The recorded vote is as follows:

YEAS--24. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Suetterlein, Surovell--24.

NAYS--Chase, Cosgrove, DeSteph, Hanger, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Vogel--15.

RULE 36--0.

- **S.B. 1410** (one thousand four hundred ten) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact §§ 2.2-2901.1, 2.2-3004, 2.2-3900, 2.2-3901, 2.2-3902, 2.2-3904, 2.2-3905, 15.2-853, 15.2-854, 15.2-965, 15.2-1500.1, 15.2-1507, 15.2-1604, 22.1-295.2, 22.1-306, 36-96.1 through 36-96.3, 36-96.4, 36-96.6, 55.1-1208, and 55.1-1310 of the Code of Virginia, relating to public accommodations, employment, and housing; prohibited discrimination on the basis of status as active military or a military spouse.

On motion of Senator Bell, the substitute was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- **S.B. 1413** (one thousand four hundred thirteen) was taken up with the amendments proposed by the House of Delegates as follows:
 - 1. Line 68, engrossed, after responsible

strike

to obtain

insert

for obtaining

2. Line 70, engrossed, after providers

strike

A

3. Line 70, engrossed, after nongovernmental

insert

An

4. Line 70, engrossed, after responsible

strike

to obtain

insert

for obtaining

On motion of Senator Boysko, the amendments were agreed to.

The recorded vote is as follows: YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

NAYS--DeSteph, Suetterlein--2.

RULE 36--0.

- **S.B. 1469** (one thousand four hundred sixty-nine) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact § 2.2-212 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 2.2-507.3, by adding in Chapter 22 of Title 2.2 an article numbered 12, consisting of sections numbered 2.2-2365 through 2.2-2376, and by adding in Article 3.1 of Chapter 1 of Title 51.1 a section numbered 51.1-124.40, relating to establishing an Opioid Abatement Authority.

Senator Barker moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows: YEAS--0. NAYS--38. RULE 36--1.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

RULE 36--Petersen--1.

- **S.B. 1471** (one thousand four hundred seventy-one) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact §§ 4.1-206, 4.1-206.3, as it shall become effective, 4.1-231, 4.1-231.1, as it shall become effective, 4.1-233, 4.1-233.1, as it shall become effective, and 4.1-308 of the Code of Virginia, relating to alcoholic beverage control; outdoor refreshment areas.

Senator Dunnavant moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows: YEAS--0. NAYS--39. RULE 36--0.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

RULE 36--0.

- **S.B. 1472** (one thousand four hundred seventy-two) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to require the Department of Medical Assistance Services to establish a work group to study options for the permanent use of virtual supports and increasing access to virtual supports and services for individuals with intellectual and developmental disabilities.

On motion of Senator Suetterlein, the substitute was agreed to.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

HOUSE BILLS ON THIRD READING

Senator Saslaw moved that the following House bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- H.B. 1986 (one thousand nine hundred eighty-six).
- H.B. 1807 (one thousand eight hundred seven).
- H.B. 1818 (one thousand eight hundred eighteen).
- H.B. 1820 (one thousand eight hundred twenty).
- H.B. 1829 (one thousand eight hundred twenty-nine).
- H.B. 1834 (one thousand eight hundred thirty-four).
- H.B. 1892 (one thousand eight hundred ninety-two).
- H.B. 1893 (one thousand eight hundred ninety-three).
- H.B. 1907 (one thousand nine hundred seven).
- H.B. 1923 (one thousand nine hundred twenty-three).
- H.B. 1925 (one thousand nine hundred twenty-five).
- H.B. 1942 (one thousand nine hundred forty-two).
- H.B. 1985 (one thousand nine hundred eighty-five).
- H.B. 1989 (one thousand nine hundred eighty-nine).
- H.B. 1994 (one thousand nine hundred ninety-four).
- H.B. 1995 (one thousand nine hundred ninety-five).
- H.B. 2008 (two thousand eight).

- H.B. 2034 (two thousand thirty-four).
- H.B. 2036 (two thousand thirty-six).
- H.B. 2078 (two thousand seventy-eight).
- H.B. 2101 (two thousand one hundred one).
- H.B. 2110 (two thousand one hundred ten).
- H.B. 2111 (two thousand one hundred eleven).
- H.B. 2117 (two thousand one hundred seventeen).
- H.B. 2133 (two thousand one hundred thirty-three).
- H.B. 2165 (two thousand one hundred sixty-five).
- H.B. 2166 (two thousand one hundred sixty-six).
- H.B. 2167 (two thousand one hundred sixty-seven).
- H.B. 2207 (two thousand two hundred seven).
- H.B. 2212 (two thousand two hundred twelve).
- H.B. 2219 (two thousand two hundred nineteen).
- H.B. 2223 (two thousand two hundred twenty-three).
- H.B. 2230 (two thousand two hundred thirty).
- H.B. 2252 (two thousand two hundred fifty-two).
- H.B. 2269 (two thousand two hundred sixty-nine).
- H.B. 2282 (two thousand two hundred eighty-two).
- H.B. 2293 (two thousand two hundred ninety-three).
- H.B. 2304 (two thousand three hundred four).
- H.B. 2323 (two thousand three hundred twenty-three).

The motion was agreed to.

H.B. 1818 (one thousand eight hundred eighteen) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 65.2-402 of the Code of Virginia, relating to workers' compensation; presumption of compensation for certain diseases; applicable to salaried and volunteer emergency medical services personnel.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1834 (one thousand eight hundred thirty-four) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 56-599 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 26 of Title 45.1 a section numbered 45.1-394.1, relating to public disclosure of electric generating facility closures; integrated resource plans.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1985 (one thousand nine hundred eighty-five) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability of health care providers from COVID-19.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1989 (one thousand nine hundred eighty-nine) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Article 3 of Chapter 2 of Title 32.1 a section numbered 32.1-48.001, relating to public health emergency; emergency medical services agencies; real-time access to information.

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2008 (two thousand eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 38.2-3407.15:2 of the Code of Virginia, relating to health insurance; authorization of drug prescribed for the treatment of a mental disorder.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2111 (two thousand one hundred eleven) was taken up.

Senator Lucas offered the following amendment:

1. Line 29, engrossed, after midwives and

strike

two individuals who are certified midwives

insert

one certified midwife and one certified professional midwife

On motion of Senator Lucas, the reading of the amendment was waived.

On motion of Senator Lucas, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 2117 (two thousand one hundred seventeen) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-5211 and 2.2-5212 of the Code of Virginia, relating to Children's Services Act; special education programs.

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2133 (two thousand one hundred thirty-three) was taken up.

The following amendments proposed by the Committee on the Judiciary were offered:

1. Line 21, engrossed, after encouraged,

strike

forced, intimidated, or deceived

insert

or induced

2. Line 22, engrossed, after 18.2-346,

insert

by means of force, fraud, deceit, intimidation, or coercion, or in which the person induced to perform such an act is younger than 18 years of age, for money or its equivalent

The reading of the amendments was waived.

On motion of Senator Edwards, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 2166 (two thousand one hundred sixty-six) was taken up.

The following amendments proposed by the Committee on Finance and Appropriations were offered:

1. Line 121, engrossed, after days

insert

; in prescribing the terms of the order, including its length, the judge or special justice shall consider the impact on the person's opportunities and obligations, including education and employment

2. Line 175, engrossed, after days

insert

; in prescribing the terms of the order, including its length, the judge or special justice shall consider the impact on the person's opportunities and obligations, including education and employment

3. Line 215, engrossed, after to the plan

strike

and

insert

and

4. Line 220, engrossed, after person

strike

the remainder of line 220, all of line 221, and through met on line 222

5. Line 388, engrossed, after either

strike

(i)

insert

(i)

6. Line 476, engrossed, after days

insert

; in prescribing the terms of the order, including its length, the judge or special justice shall consider the impact on the person's opportunities and obligations, including education and employment

The reading of the amendments was waived.

On motion of Senator Howell, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 2167 (two thousand one hundred sixty-seven) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Rehabilitation and Social Services was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 53.1-136 and 53.1-155 of the Code of Virginia, relating to discretionary parole and conditional release; notice and certification; monthly reports.

The reading of the substitute was waived.

On motion of Senator Favola, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2207 (two thousand two hundred seven) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 65.2-402.1 of the Code of Virginia, relating to workers' compensation; presumption as to death or disability from COVID-19.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

The substitute was ordered to be engrossed.

Senator Saslaw moved that the passage of the House bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following House bills were passed en bloc with their titles:

- H.B. 1986 (one thousand nine hundred eighty-six).
- H.B. 1807 (one thousand eight hundred seven).
- H.B. 1818 (one thousand eight hundred eighteen) with substitute.
- H.B. 1820 (one thousand eight hundred twenty).
- H.B. 1829 (one thousand eight hundred twenty-nine).
- H.B. 1834 (one thousand eight hundred thirty-four) with substitute.
- H.B. 1892 (one thousand eight hundred ninety-two).
- H.B. 1925 (one thousand nine hundred twenty-five).
- H.B. 1942 (one thousand nine hundred forty-two).
- **H.B.** 1985 (one thousand nine hundred eighty-five) with substitute.
- H.B. 1989 (one thousand nine hundred eighty-nine) with substitute.
- H.B. 1994 (one thousand nine hundred ninety-four).
- **H.B. 1995** (one thousand nine hundred ninety-five).
- H.B. 2008 (two thousand eight) with substitute.
- H.B. 2034 (two thousand thirty-four).
- **H.B. 2036** (two thousand thirty-six).
- H.B. 2078 (two thousand seventy-eight).
- H.B. 2110 (two thousand one hundred ten).
- **H.B. 2111** (two thousand one hundred eleven) with amendment.
- H.B. 2117 (two thousand one hundred seventeen) with substitute.
- H.B. 2133 (two thousand one hundred thirty-three) with amendments.
- **H.B. 2165** (two thousand one hundred sixty-five).
- H.B. 2166 (two thousand one hundred sixty-six) with amendments.
- H.B. 2167 (two thousand one hundred sixty-seven) with substitute.
- H.B. 2207 (two thousand two hundred seven) with substitute.

```
H.B. 2212 (two thousand two hundred twelve).
```

H.B. 2219 (two thousand two hundred nineteen).

H.B. 2223 (two thousand two hundred twenty-three).

H.B. 2230 (two thousand two hundred thirty).

H.B. 2252 (two thousand two hundred fifty-two).

H.B. 2293 (two thousand two hundred ninety-three).

H.B. 2323 (two thousand three hundred twenty-three).

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell--38.

NAYS--0.

RULE 36--0.

RECONSIDERATION

Senator Cosgrove moved to reconsider the vote by which the following House bills were passed en bloc with their titles:

- H.B. 1986 (one thousand nine hundred eighty-six).
- H.B. 1807 (one thousand eight hundred seven).
- H.B. 1818 (one thousand eight hundred eighteen) with substitute.
- H.B. 1820 (one thousand eight hundred twenty).
- H.B. 1829 (one thousand eight hundred twenty-nine).
- H.B. 1834 (one thousand eight hundred thirty-four) with substitute.
- H.B. 1892 (one thousand eight hundred ninety-two).
- H.B. 1925 (one thousand nine hundred twenty-five).
- H.B. 1942 (one thousand nine hundred forty-two).
- H.B. 1985 (one thousand nine hundred eighty-five) with substitute.
- H.B. 1989 (one thousand nine hundred eighty-nine) with substitute.
- H.B. 1994 (one thousand nine hundred ninety-four).
- H.B. 1995 (one thousand nine hundred ninety-five).
- H.B. 2008 (two thousand eight) with substitute.
- H.B. 2034 (two thousand thirty-four).
- H.B. 2036 (two thousand thirty-six).
- H.B. 2078 (two thousand seventy-eight).
- H.B. 2110 (two thousand one hundred ten).
- H.B. 2111 (two thousand one hundred eleven) with amendment.
- H.B. 2117 (two thousand one hundred seventeen) with substitute.
- H.B. 2133 (two thousand one hundred thirty-three) with amendments.
- H.B. 2165 (two thousand one hundred sixty-five).
- H.B. 2166 (two thousand one hundred sixty-six) with amendments.
- H.B. 2167 (two thousand one hundred sixty-seven) with substitute.
- H.B. 2207 (two thousand two hundred seven) with substitute.
- H.B. 2212 (two thousand two hundred twelve).
- H.B. 2219 (two thousand two hundred nineteen).
- H.B. 2223 (two thousand two hundred twenty-three).

```
H.B. 2230 (two thousand two hundred thirty).
```

H.B. 2252 (two thousand two hundred fifty-two).

H.B. 2293 (two thousand two hundred ninety-three).

H.B. 2323 (two thousand three hundred twenty-three).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed en bloc with their titles:

```
H.B. 1986 (one thousand nine hundred eighty-six).
```

H.B. 1807 (one thousand eight hundred seven).

H.B. 1818 (one thousand eight hundred eighteen) with substitute.

H.B. 1820 (one thousand eight hundred twenty).

H.B. 1829 (one thousand eight hundred twenty-nine).

H.B. 1834 (one thousand eight hundred thirty-four) with substitute.

H.B. 1892 (one thousand eight hundred ninety-two).

H.B. 1925 (one thousand nine hundred twenty-five).

H.B. 1942 (one thousand nine hundred forty-two).

H.B. 1985 (one thousand nine hundred eighty-five) with substitute.

H.B. 1989 (one thousand nine hundred eighty-nine) with substitute.

H.B. 1994 (one thousand nine hundred ninety-four).

H.B. 1995 (one thousand nine hundred ninety-five).

H.B. 2008 (two thousand eight) with substitute.

H.B. 2034 (two thousand thirty-four).

H.B. 2036 (two thousand thirty-six).

H.B. 2078 (two thousand seventy-eight).

H.B. 2110 (two thousand one hundred ten).

H.B. 2111 (two thousand one hundred eleven) with amendment.

H.B. 2117 (two thousand one hundred seventeen) with substitute.

H.B. 2133 (two thousand one hundred thirty-three) with amendments.

H.B. 2165 (two thousand one hundred sixty-five).

H.B. 2166 (two thousand one hundred sixty-six) with amendments.

H.B. 2167 (two thousand one hundred sixty-seven) with substitute.

H.B. 2207 (two thousand two hundred seven) with substitute.

H.B. 2212 (two thousand two hundred twelve).

H.B. 2219 (two thousand two hundred nineteen).

H.B. 2223 (two thousand two hundred twenty-three).

H.B. 2230 (two thousand two hundred thirty).

H.B. 2252 (two thousand two hundred fifty-two).

H.B. 2293 (two thousand two hundred ninety-three).

H.B. 2323 (two thousand three hundred twenty-three).

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1893 (one thousand eight hundred ninety-three) was taken up.

The following amendment proposed by the Committee on Transportation was offered:

 Line 140, engrossed, after higher strike

educaton

insert

education

The reading of the amendment was waived.

On motion of Senator Marsden, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 1893, on motion of Senator Marsden, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1.

RULE 36--0.

H.B. 1907 (one thousand nine hundred seven), on motion of Senator Saslaw, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1.

RULE 36--0.

H.B. 1923 (one thousand nine hundred twenty-three) was taken up.

The following amendments proposed by the Committee on Commerce and Labor were offered:

1. Line 46, engrossed, after responsible

strike

to obtain

insert

for obtaining

2. Line 48, engrossed, after responsible

strike

to obtain

insert

for obtaining

The reading of the amendments was waived.

On motion of Senator Saslaw, the amendments were agreed to.

H.B. 1923, on motion of Senator Suetterlein, was passed by temporarily.

H.B. 2101 (two thousand one hundred one) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend the second enactment of Chapter 525 of the Acts of Assembly of 2020, relating to GO Virginia Grants; matching funds; sunset.

The reading of the substitute was waived.

On motion of Senator Howell, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2101, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

NAYS--Chase, Suetterlein--2.

RULE 36--0.

H.B. 2269 (two thousand two hundred sixty-nine), on motion of Senator Saslaw, was passed with its title.

The recorded vote is as follows:

YEAS--36. NAYS--1. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--36.

NAYS--Chase--1.

RULE 36--Suetterlein--1.

H.B. 2282 (two thousand two hundred eighty-two), on motion of Senator Saslaw, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--Chase--1.

RULE 36--0.

RECONSIDERATION

Senator Dunnavant moved to reconsider the vote by which **H.B. 2269** (two thousand two hundred sixty-nine) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 2269, on motion of Senator Saslaw, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--1. RULE 36--1.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

NAYS--Chase--1.

RULE 36--Suetterlein--1.

H.B. 2304 (two thousand three hundred four) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 56-585.1:9 of the Code of Virginia, relating to provision of broadband services by investor-owned electric utilities.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2304, on motion of Senator Saslaw, was passed with its title.

The recorded vote is as follows:

YEAS--38. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--38.

NAYS--Suetterlein--1.

RULE 36--0.

STATEMENT ON VOTE

Senator DeSteph stated that he voted yea on the question of the passage of **H.B. 2304**, whereas he intended to vote nay.

H.B. 2014 (two thousand fourteen) was taken up.

Senator Barker offered the following amendment No. 3:

3. Line 131, engrossed, after enforceable.

insert

Notwithstanding the requirements of this section, a landlord with four or fewer rental dwelling units, or up to a 10 percent interest in four or fewer rental dwelling units, may limit a tenant's use of the right of redemption to once per lease period, provided that the landlord provides written notice of such limitation to the tenant.

On motion of Senator Barker, the reading of the amendment was waived.

On motion of Senator Barker, the amendment was agreed to.

The amendment was ordered to be engrossed.

H.B. 2014, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Stuart, Suetterlein--14.

RULE 36--0.

H.B. 1778 (one thousand seven hundred seventy-eight) was taken up, the committee amendments having been agreed to and ordered to be engrossed on February 17, 2021.

H.B. 1778, on motion of Senator Lewis, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

H.B. 1911 (one thousand nine hundred eleven) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 20-106 of the Code of Virginia, relating to no-fault divorces; corroboration requirement.

The reading of the substitute was waived.

Senator Edwards moved that the substitute be agreed to.

The question was put on agreeing to the substitute.

The yeas and nays were called for by the Chair.

The recorded vote is as follows:

YEAS--18. NAYS--21. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Edwards, Hanger, Kiggans, Lucas, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Suetterlein, Vogel--18.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Favola, Hashmi, Howell, Lewis, Locke, Marsden, Mason, McClellan, McDougle, McPike, Reeves, Saslaw, Spruill, Stanley, Stuart, Surovell--21. RULE 36--0.

The substitute was rejected.

H.B. 1911, on motion of Senator Surovell, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Peake, Petersen, Reeves, Ruff, Saslaw, Spruill, Stuart, Surovell, Vogel--27.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Newman, Norment, Obenshain, Pillion, Stanley, Suetterlein--12.

RULE 36--0.

H.B. 2081 (two thousand eighty-one) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

- H.B. 2174 (two thousand one hundred seventy-four) was passed by temporarily.
- **H.B. 2263** (two thousand two hundred sixty-three), on motion of Senator Surovell, was passed by for the day.
- **H.B. 1833** (one thousand eight hundred thirty-three) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Ruff, Saslaw, Spruill, Surovell--26.

NAYS--DeSteph, Dunnavant, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Stanley, Stuart, Suetterlein, Vogel--13.

RULE 36--0.

STATEMENT ON VOTE

Senator Chase stated that she voted yea on the question of the passage of H.B. 1833, whereas she intended to vote nay.

H.B. 1884 (one thousand eight hundred eighty-four) was read by title the third time and, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--30. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Petersen, Pillion, Reeves, Saslaw, Spruill, Stanley, Surovell, Vogel--30.

NAYS--Chase, DeSteph, McDougle, Norment, Obenshain, Peake, Ruff, Stuart--8.

RULE 36--0.

H.B. 1896 (one thousand eight hundred ninety-six) was read by title the third time and, on motion of Senator Saslaw, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

- H.B. 1902 (one thousand nine hundred two), on motion of Senator Petersen, was passed by for the day.
- **H.B. 1912** (one thousand nine hundred twelve) was read by title the third time and, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Obenshain, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Barker, Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

RECONSIDERATION

Senator Obenshain moved to reconsider the vote by which **H.B. 1912** (one thousand nine hundred twelve) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1912, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--19. RULE 36--0.

YEAS--Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill--20.

NAYS--Barker, Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Surovell, Vogel--19.

RULE 36--0.

RECONSIDERATION

Senator Kiggans moved to reconsider the vote by which **H.B. 1884** (one thousand eight hundred eighty-four) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1884, on motion of Senator Favola, was passed with its title.

The recorded vote is as follows:

YEAS--27. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Reeves, Saslaw, Spruill, Suetterlein, Surovell--27.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Obenshain, Peake, Pillion, Ruff, Stanley, Stuart, Vogel--12.

RULE 36--0.

H.B. 1965 (one thousand nine hundred sixty-five), on motion of Senator Newman, was passed by for the day.

- **H.B. 1979** (one thousand nine hundred seventy-nine), on motion of Senator Howell, was passed by for the day.
- **H.B. 1992** (one thousand nine hundred ninety-two), on motion of Senator Edwards, was passed by temporarily.
- **H.B. 1993** (one thousand nine hundred ninety-three) was read by title the third time and, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--24. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--24.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--15.

RULE 36--0.

H.B. 2032 (two thousand thirty-two) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-3905, 40.1-2, 40.1-29, 40.1-49.3, and 40.1-49.8 of the Code of Virginia, relating to the employees providing domestic service; the Virginia Human Rights Act; application of laws applicable to employee safety and payment of wages.

The reading of the substitute was waived.

On motion of Senator Saslaw, the substitute was agreed to.

The substitute was ordered to be engrossed.

Senator Saslaw moved that **H.B. 2032** be passed with its title.

- H.B. 2032, on motion of Senator McClellan, was passed by temporarily.
- H.B. 2040 (two thousand forty), on motion of Senator Obenshain, was passed by for the day.
- H.B. 2047 (two thousand forty-seven) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 19.2-120, 19.2-163.03, and 19.2-299 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 16 of Title 19.2 a section numbered 19.2-271.6, relating to criminal proceedings; consideration of mental condition and intellectual and developmental disabilities.

The reading of the substitute was waived.

On motion of Senator McClellan, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2047, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows:

YEAS--25. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Stuart, Surovell, Vogel--25.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Mason, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Suetterlein--14.

RULE 36--0.

RECONSIDERATION

Senator Lewis moved to reconsider the vote by which **H.B. 1912** (one thousand nine hundred twelve) was passed with its title.

The motion was agreed to by unanimous consent.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1912, on motion of Senator Edwards, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

RECONSIDERATION

Senator Kiggans moved to reconsider the vote by which H.B. 1993 (one thousand nine hundred ninety-three) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1993, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Surovell, Vogel--26.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Stuart, Suetterlein--13.

RULE 36--0.

- **H.B. 2118** (two thousand one hundred eighteen), on motion of Senator Barker, was passed by for the day.
- **H.B. 2159** (two thousand one hundred fifty-nine) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

H.B. 2218 (two thousand two hundred eighteen) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Education and Health was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-250.1, 54.1-2519, 54.1-2521, 54.1-2903, 54.1-3408.3, and 54.1-3442.5 through 54.1-3442.8 of the Code of Virginia, relating to pharmaceutical processors; cannabis products.

The reading of the substitute was waived.

On motion of Senator Lucas, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2218, on motion of Senator Lucas, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Suetterlein, Surovell, Vogel--31.

NAYS--Chase, DeSteph, McDougle, Newman, Obenshain, Peake, Reeves, Stuart--8. RULE 36--0.

RECONSIDERATION

Senator Boysko moved to reconsider the vote by which H.B. 2159 (two thousand one hundred fifty-nine) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 2159, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--17. RULE 36--0.

YEAS--Barker, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell, Vogel--22.

NAYS--Bell, Chase, Cosgrove, DeSteph, Dunnavant, Hanger, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--17.

RULE 36--0.

- **H.B. 2307** (two thousand three hundred seven), on motion of Senator Surovell, was passed by for the day.
 - H.B. 2321 (two thousand three hundred twenty-one) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 2.2-200, 2.2-204, 2.2-205, 2.2-205.1, 2.2-435.6, 2.2-435.8, 2.2-435.9, 2.2-435.10, 2.2-2471, 2.2-2471.1, 2.2-2472, 2.2-2472.2, 2.2-2472.3, and 30-377 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 2 of Title 2.2 an article numbered 6.1, consisting of sections numbered 2.2-214.2 and 2.2-214.3; and to repeal § 2.2-435.7 of the Code of Virginia, relating to Governor's Secretaries; Secretary of Workforce created.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2321, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--18. RULE 36--1.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--Morrissey--1.

- **H.B. 2330** (two thousand three hundred thirty), on motion of Senator Saslaw, was passed by for the day.
- **H.B. 2332** (two thousand three hundred thirty-two), on motion of Senator Saslaw, was passed by for the day.
 - H.B. 1923 (one thousand nine hundred twenty-three) was taken up.

The amendments were ordered to be engrossed.

H.B. 1923, on motion of Senator Boysko, was passed with its title.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

- **H.B. 2174** (two thousand one hundred seventy-four) was taken up and, on motion of Senator Howell, was passed by for the day.
 - H.B. 1992 (one thousand nine hundred ninety-two) was taken up and read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on the Judiciary was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-308.09, 18.2-308.2:1, as it is currently effective and as it shall become effective, 18.2-308.2:2, as it is currently effective and as it shall become effective, 18.2-308.2:3, as it is currently effective and as it shall become effective, and 19.2-386.28 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-308.1:8, relating to purchase, possession, or transportation of firearms following conviction for assault and battery of a family or household member; penalties.

The reading of the substitute was waived.

Senator Edwards moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 18.2-308.09, 18.2-308.2:1, as it is currently effective and as it shall become effective, 18.2-308.2:2, as it is currently effective and as it shall become effective, 18.2-308.2:3, as it is currently effective and as it shall become effective, and 19.2-386.28 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-308.1:8, relating to purchase, possession, or transportation of firearms following conviction for assault and battery of a family or household member; penalties.

The reading of the substitute was waived.

On motion of Senator Edwards, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1992, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--19. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, Lewis, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--19.

RULE 36--0.

H.B. 2032 (two thousand thirty-two) was taken up and, on motion of Senator McClellan, was passed with its title.

The recorded vote is as follows: YEAS--21. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissev, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- H.B. 1804 (one thousand eight hundred four).
- H.B. 1837 (one thousand eight hundred thirty-seven).
- H.B. 1921 (one thousand nine hundred twenty-one).
- H.B. 1982 (one thousand nine hundred eighty-two).
- H.B. 1983 (one thousand nine hundred eighty-three).
- H.B. 2068 (two thousand sixty-eight).
- H.B. 2187 (two thousand one hundred eighty-seven).
- H.B. 2302 (two thousand three hundred two).
- H.B. 1760 (one thousand seven hundred sixty).
- H.B. 1810 (one thousand eight hundred ten).
- H.B. 1819 (one thousand eight hundred nineteen).
- **H.B. 1836** (one thousand eight hundred thirty-six).
- H.B. 1888 (one thousand eight hundred eighty-eight).
- H.B. 1928 (one thousand nine hundred twenty-eight).
- H.B. 1952 (one thousand nine hundred fifty-two).
- H.B. 1958 (one thousand nine hundred fifty-eight).
- H.B. 1968 (one thousand nine hundred sixty-eight).
- H.B. 2020 (two thousand twenty).
- H.B. 2030 (two thousand thirty).
- H.B. 2125 (two thousand one hundred twenty-five).
- H.B. 2198 (two thousand one hundred ninety-eight).
- H.B. 2250 (two thousand two hundred fifty).
- H.B. 2275 (two thousand two hundred seventy-five).
- H.B. 2311 (two thousand three hundred eleven).
- H.B. 2324 (two thousand three hundred twenty-four).

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

- H.B. 1804 (one thousand eight hundred four).
- H.B. 1837 (one thousand eight hundred thirty-seven).
- H.B. 1921 (one thousand nine hundred twenty-one).
- H.B. 1982 (one thousand nine hundred eighty-two).
- H.B. 1983 (one thousand nine hundred eighty-three).
- H.B. 2068 (two thousand sixty-eight).
- H.B. 2187 (two thousand one hundred eighty-seven).
- H.B. 2302 (two thousand three hundred two).
- H.B. 1760 (one thousand seven hundred sixty).
- H.B. 1810 (one thousand eight hundred ten).
- H.B. 1819 (one thousand eight hundred nineteen).
- H.B. 1836 (one thousand eight hundred thirty-six).
- $\textbf{H.B. 1888} \ (one \ thousand \ eight \ hundred \ eighty-eight).$
- H.B. 1928 (one thousand nine hundred twenty-eight).
- H.B. 1952 (one thousand nine hundred fifty-two). H.B. 1958 (one thousand nine hundred fifty-eight).
- H.B. 1968 (one thousand nine hundred sixty-eight).
- H.B. 2020 (two thousand twenty).
- H.B. 2030 (two thousand thirty).
- H.B. 2125 (two thousand one hundred twenty-five).
- H.B. 2198 (two thousand one hundred ninety-eight).
- H.B. 2250 (two thousand two hundred fifty).
- H.B. 2275 (two thousand two hundred seventy-five).
- H.B. 2311 (two thousand three hundred eleven).
- H.B. 2324 (two thousand three hundred twenty-four).

HOUSE JOINT RESOLUTION ON THIRD READING

H.J.R. 542 (five hundred forty-two) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

Directing the Joint Legislative Audit and Review Commission to study transit equity and modernization in the Commonwealth. Report.

The reading of the substitute was waived.

On motion of Senator Locke, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.J.R. 542, on motion of Senator Locke, was agreed to.

HOUSE JOINT RESOLUTIONS ON SECOND READING

- H.J.R. 555 (five hundred fifty-five) was read by title the second time.
- H.J.R. 582 (five hundred eighty-two) was read by title the second time.

MEMORIAL RESOLUTIONS

H.J.R. 706 (seven hundred six), on motion of Senator Saslaw, was agreed to by a unanimous standing vote.

On motion of Senator Saslaw, the questions on agreeing to the Senate resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate resolutions were taken up, ordered to be engrossed, and agreed to en bloc by a unanimous standing vote:

S.R. 508 (five hundred eight).

S.R. 513 (five hundred thirteen).

COMMENDING RESOLUTIONS

On motion of Senator Saslaw, the questions on agreeing to the House joint resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following House joint resolutions were taken up and agreed to en bloc:

H.J.R. 718 (seven hundred eighteen).

H.J.R. 763 (seven hundred sixty-three).

On motion of Senator Saslaw, the questions on agreeing to the Senate resolutions that follow were considered en bloc.

On motion of Senator Saslaw, the following Senate resolutions were taken up, ordered to be engrossed, and agreed to en bloc:

S.R. 506 (five hundred six).

S.R. 507 (five hundred seven).

S.R. 509 (five hundred nine).

S.R. 510 (five hundred ten).

S.R. 511 (five hundred eleven).

S.R. 512 (five hundred twelve).

S.R. 514 (five hundred fourteen).

S.R. 515 (five hundred fifteen).

S.R. 516 (five hundred sixteen).

S.R. 517 (five hundred seventeen).

S.R. 518 (five hundred eighteen).

S.R. 519 (five hundred nineteen).

SUPPLEMENTAL CALENDAR NO. 1

HOUSE BILL ON SECOND READING

H.B. 2058 (two thousand fifty-eight) was read by title the second time.

HOUSE COMMUNICATION

The following communication was received and read:

In the House of Delegates February 18, 2021

THE HOUSE OF DELEGATES HAS PASSED WITH AMENDMENTS THE FOLLOWING SENATE BILL:

S.B. 1163. A BILL to amend and reenact §§ 58.1-334, 58.1-337, 58.1-432, and 58.1-436 of the Code of Virginia, relating to tax credits of agricultural equipment.

THE HOUSE OF DELEGATES HAS PASSED WITH SUBSTITUTES THE FOLLOWING SENATE BILLS:

- **S.B. 1320.** A BILL to amend and reenact § 54.1-2900 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 54.1-2957.04, relating to licensed certified midwives; licensure; practice.
- **S.B. 1392.** A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-581, relating to Consumer Data Protection Act.
- **S.B. 1423.** A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

THE HOUSE OF DELEGATES HAS REJECTED THE SENATE AMENDMENTS AS SUBSTITUTED FOR HOUSE AMENDMENTS TO THE FOLLOWING HOUSE BILL:

H.B. 1800. A BILL to amend and reenact Chapter 1289 of the 2020 Acts of Assembly, as amended by Chapter 56 of the 2020 Acts of Assembly, Special Session I, which appropriated funds for the 2020-22 Biennium and provided a portion of revenues for the two years ending, respectively, on the thirtieth day of June, 2021, and the thirtieth day of June, 2022.

THE HOUSE OF DELEGATES HAS INSISTED ON ITS SUBSTITUTE AND HAS REQUESTED A COMMITTEE OF CONFERENCE ON THE FOLLOWING SENATE BILL:

S.B. 1146. A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, 58.1-322.03, and 58.1-402 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.

THE HOUSE OF DELEGATES HAS REJECTED THE SUBSTITUTES PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1935.** A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, and 58.1-322.03 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.
- **H.B. 2273.** A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

H.B. 2312. A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-122, 4.1-124, as it is currently effective and as it shall become effective, 4.1-128, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-202, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-390, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29 consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6 consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, consisting of chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Title 19.2 a chapter numbered 23.2, consisting of sections numbered 19.2-392.5 through 19.2-392.13, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

THE HOUSE OF DELEGATES HAS AGREED TO THE AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1763.** A BILL to amend and reenact §§ 58.1-339.3 and 58.1-439.5 of the Code of Virginia, relating to tax credit; agricultural best management practices.
- **H.B. 1895.** A BILL to amend and reenact §§ 19.2-353.5 through 19.2-355 of the Code of Virginia, relating to fines and costs; accrual of interest; deferral or installment payment agreements.

THE HOUSE OF DELEGATES HAS AGREED TO THE SUBSTITUTE PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILL:

H.B. 1953. A BILL to amend and reenact § 54.1-2900 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 54.1-2957.04, relating to licensed certified midwives; licensure; practice.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

Jesan Carle Gelson

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Lucas, the Senate adjourned until tomorrow at 11:00 a.m.

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate

FRIDAY, FEBRUARY 19, 2021

The Senate met at 11:00 a.m. and was called to order by Lieutenant Governor Justin E. Fairfax.

The Honorable Jeremy S. McPike, Twenty-ninth Senatorial District, offered the following prayer:

These words this morning are all offered and written by my father-in-law, Emmel Myskowski. God of love, You created us and You sustain us in the human struggle for goodness. You call us to live as brothers and sisters. Give us the strength daily to be instruments of peace and prosperity. Enable us to see everyone in our constituencies as brother or sister. Make us sensitive to the plea of our Virginia citizens who entreat us to daily find and implement ways to assist their legitimate needs. Help us to be strong enough to diffuse the violence of our tongues and our hands. Renew our hearts and minds, so that our way of life will always be that of shalom, peace, salaam, and we say, Amen.

The Pledge of Allegiance to the Flag of the United States of America was led by Corinne Sloan, Assistant Journal Clerk. Senate Clerk's Office.

The roll was called and the following Senators answered to their names:

Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel.

A quorum was present.

After the roll call, Senator McClellan notified the Clerk of her presence.

On motion of Senator Stuart, the reading of the Journal was waived.

The recorded vote is as follows: YEAS--31. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Ruff, Saslaw, Stuart, Suetterlein, Surovell, Vogel--31.

NAYS--Deeds, DeSteph, McDougle, Petersen, Reeves, Stanley--6. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 18, 2021

THE HOUSE OF DELEGATES HAS PASSED WITH AMENDMENTS THE FOLLOWING SENATE BILLS:

S.B. 1322. A BILL to amend and reenact § 8.01-225 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 22.1-274.6, relating to public schools; seizure management and action plan; biennial training.

S.B. 1439. A BILL to amend and reenact § 22.1-254 of the Code of Virginia, relating to the Department of Education; guidelines on excused student absences; civic engagement.

THE HOUSE OF DELEGATES HAS PASSED WITH SUBSTITUTES THE FOLLOWING SENATE BILLS:

- **S.B. 1169.** A BILL to amend the Code of Virginia by adding a section numbered 22.1-205.1, relating to high school student parking passes; valid driver's license or driver privilege card required.
- **S.B. 1193.** A BILL to amend the Code of Virginia by adding in Title 3.2 a chapter numbered 33.1, consisting of sections numbered 3.2-3304 through 3.2-3307, relating to Dairy Producer Margin Coverage Premium Assistance Program.
- **S.B. 1338.** A BILL to amend and reenact §§ 32.1-325 and 38.2-3418.16 of the Code of Virginia, relating to telemedicine services; remote patient monitoring services.
- **S.B. 1349.** A BILL to amend and reenact § 2.2-3802 of the Code of Virginia, relating to the Government Data Collection and Dissemination Practices Act; exemptions; email addresses of licensed professionals.
- **S.B. 1354.** A BILL to amend and reenact §§ 10.1-1186.01, 62.1-44.19:13, and 62.1-44.19:14 of the Code of Virginia, relating to Chesapeake Bay Phase III Watershed Improvement Plan; nutrient removal; regulations.
- **S.B. 1461.** A BILL to amend the Code of Virginia by adding a section numbered 18.2-474.2, relating to bribery in correctional facilities; penalty.

THE HOUSE OF DELEGATES HAS PASSED WITH A SUBSTITUTE WITH AMENDMENT THE FOLLOWING SENATE BILL:

S.B. 1130. A BILL to amend the Code of Virginia by adding in Article 5 of Chapter 36 of Title 58.1 a section numbered 58.1-3668, relating to personal property tax exemption; motor vehicle of a disabled veteran.

THE HOUSE OF DELEGATES HAS REJECTED THE AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1805.** A BILL to amend and reenact §§ 51.5-134 and 51.5-135 of the Code of Virginia, relating to aging services; economic and social need.
- **H.B. 2197.** A BILL to require the Department of Medical Assistance Services to establish a work group to study options for the permanent use of virtual supports and increasing access to virtual supports and services for individuals with intellectual and developmental disabilities.

THE HOUSE OF DELEGATES HAS REJECTED THE SUBSTITUTES PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1817.** A BILL to amend and reenact §§ 54.1-2957 and 54.1-2957.01 of the Code of Virginia, relating to practice of certified nurse midwives.
- **H.B. 2266.** A BILL to amend and reenact §§ 4.1-206, 4.1-206.3, as it shall become effective, 4.1-231, 4.1-231.1, as it shall become effective, 4.1-233, 4.1-233.1, as it shall become effective, and 4.1-308 of the Code of Virginia, relating to alcoholic beverage control; outdoor refreshment areas.

THE HOUSE OF DELEGATES HAS INSISTED ON ITS SUBSTITUTE AND HAS REQUESTED A COMMITTEE OF CONFERENCE ON THE FOLLOWING SENATE JOINT RESOLUTION:

S.J.R. 272. Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters and the right to vote; persons not entitled to vote.

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING SENATE BILLS:

- **S.B. 1098.** A BILL to amend and reenact § 5.1-5 of the Code of Virginia, relating to aircraft registration; unmanned aircraft.
- **S.B. 1102.** A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 10 of Title 32.1 a section numbered 32.1-331.04, relating to Department of Medical Assistance Services; personal care aides; orientation program.
- S.B. 1158. A BILL to amend and reenact §§ 58.1-439.12:06, 58.1-439.12:09, and 58.1-439.12:10 of the Code of Virginia, relating to Port of Virginia tax credits; sunset.
- **S.B. 1225.** A BILL to amend the Code of Virginia by adding in Article 5 of Chapter 9 of Title 15.2 a section numbered 15.2-986 and by adding a section numbered 22.1-79.9, relating to promotion of broadband service for educational purposes.
- **S.B. 1391.** A BILL to amend and reenact § 2.2-3802 of the Code of Virginia and to amend the Code of Virginia by adding in Article 1 of Chapter 9 of Title 19.2 a section numbered 19.2-134.1, relating to pretrial data collection.
- **S.B. 1436.** A BILL to amend the Code of Virginia by adding in Article 2 of Chapter 1 of Title 23.1 a section numbered 23.1-110, by adding in Article 3 of Chapter 1 of Title 32.1 a section numbered 32.1-23.3, and by adding a section numbered 54.1-2400.03, relating to Department of Health; Eligible Health Care Provider Reserve Directory.
- **S.B. 1458.** A BILL to amend and reenact §§ 2.2-436 and 2.2-437 of the Code of Virginia, relating to Secretary of Commerce and Trade; Identity Management Standards Advisory Council.

THE HOUSE OF DELEGATES HAS AGREED TO THE AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1850.** A BILL to amend and reenact § 46.2-1129.2 of the Code of Virginia, relating to motor vehicle weight limits; vehicles powered primarily by electric battery power or fueled primarily by natural gas.
- **H.B. 2031.** A BILL to amend the Code of Virginia by adding in Article 1 of Chapter 17 of Title 15.2 a section numbered 15.2-1723.2 and by adding a section numbered 23.1-815.1, relating to facial recognition technology; authorization of use by local law-enforcement agencies and public institutions of higher education.

- **H.B. 2071.** A BILL to amend and reenact §§ 33.2-214.2 and 33.2-353 of the Code of Virginia, relating to transportation projects; resiliency.
- **H.B. 2213.** A BILL to require the establishment of a work group to study the mining and processing of gold in the Commonwealth; report.
- **H.B. 2258.** A BILL to amend and reenact § 19.2-387.3 of the Code of Virginia, relating to Substantial Risk Order Registry; maintenance and access.

THE HOUSE OF DELEGATES HAS AGREED TO THE SUBSTITUTES PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 1846.** A BILL to amend and reenact § 46.2-334.01, as it is currently effective and as it shall become effective, of the Code of Virginia, relating to license restrictions for minors; use of handheld personal communications devices.
- H.B. 2249. A BILL to amend and reenact §§ 17.1-275, 55.1-1200, 55.1-1204, 55.1-1206, 55.1-1208, 55.1-1211, 55.1-1226, 64.2-2008, and 64.2-2012 of the Code of Virginia, relating to the Virginia Residential Landlord and Tenant Act; landlord charges for security deposits, insurance premiums for damage insurance, and insurance premiums for renter's insurance; filing of information regarding resident agent appointed by nonresident property owner.

THE HOUSE OF DELEGATES HAS AGREED TO THE AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE JOINT RESOLUTION:

H.J.R. 527. Requesting the Department of Conservation and Recreation, jointly with the Virginia Department of Agriculture and Consumer Services, to study the sale and use of invasive plant species. Report.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--33. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--33.

NAYS--Deeds, DeSteph, McDougle--3.

RULE 36--0.

COMMITTEE REPORTS

The following bills, having been considered by the committee in session, were reported by Senator Favola from the Committee on Rehabilitation and Social Services:

- H.B. 1874 (one thousand eight hundred seventy-four).
- H.B. 1957 (one thousand nine hundred fifty-seven).
- H.B. 2035 (two thousand thirty-five).
- H.B. 2065 (two thousand sixty-five).
- H.B. 2070 (two thousand seventy).

The following bills, having been considered by the committee in session, were reported by Senator Marsden from the Committee on Transportation:

- H.B. 1832 (one thousand eight hundred thirty-two).
- H.B. 1841 (one thousand eight hundred forty-one).
- H.B. 1854 (one thousand eight hundred fifty-four).
- H.B. 1868 (one thousand eight hundred sixty-eight).
- H.B. 1887 (one thousand eight hundred eighty-seven).
- H.B. 1926 (one thousand nine hundred twenty-six).
- H.B. 2075 (two thousand seventy-five).
- **H.B. 2163** (two thousand one hundred sixty-three) with substitute with the recommendation that it be rereferred to the Committee on Finance and Appropriations.
 - H.B. 2284 (two thousand two hundred eighty-four).

The following bills and joint resolutions, having been considered by the committee in session, were reported by Senator Locke from the Committee on Rules:

- H.B. 1789 (one thousand seven hundred eighty-nine).
- H.B. 1990 (one thousand nine hundred ninety) with amendment.
- H.B. 2208 (two thousand two hundred eight).
- H.J.R. 525 (five hundred twenty-five).
- H.J.R. 537 (five hundred thirty-seven).
- H.J.R. 538 (five hundred thirty-eight) with amendments.
- H.J.R. 562 (five hundred sixty-two).
- H.J.R. 563 (five hundred sixty-three).
- H.J.R. 578 (five hundred seventy-eight).
- H.J.R. 579 (five hundred seventy-nine).
- H.J.R. 583 (five hundred eighty-three).
- H.J.R. 596 (five hundred ninety-six).
- H.J.R. 604 (six hundred four).
- H.J.R. 605 (six hundred five).
- H.J.R. 606 (six hundred six).
- **H.B. 2163** was rereferred to the Committee on Finance and Appropriations.

INTRODUCTION OF LEGISLATION

The following, by leave, was presented and laid on the Clerk's Desk under Senate Rule 26 (g):

S.R. 524. Celebrating the life of the Honorable Clyde H. Perdue, Jr. Patron--Stanley

CALENDAR

UNFINISHED BUSINESS—HOUSE

H.B. 1800 (one thousand eight hundred) was taken up.

On motion of Senator Howell, the Senate insisted on its amendments as substituted for House amendments and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

H.B. 1811 (one thousand eight hundred eleven) was taken up.

On motion of Senator Barker, the Senate insisted on its amendments and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.B. 1847 (one thousand eight hundred forty-seven) was taken up.

On motion of Senator Barker, the Senate insisted on its substitute and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.B. 1935 (one thousand nine hundred thirty-five) was taken up.

On motion of Senator Howell, the Senate insisted on its substitute and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.B. 2168 (two thousand one hundred sixty-eight) was taken up.

On motion of Senator Edwards, the Senate insisted on its amendments and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

H.B. 2273 (two thousand two hundred seventy-three) was taken up.

On motion of Senator Howell, the Senate insisted on its substitute and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

H.B. 2312 (two thousand three hundred twelve) was taken up.

On motion of Senator Favola, the Senate insisted on its substitute and respectfully requested a committee of conference.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

UNFINISHED BUSINESS—SENATE

S.B. 1104 (one thousand one hundred four) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend and reenact §§ 53.1-136 and 53.1-155 of the Code of Virginia, relating to parole; notice and certification; monthly reports; discretionary early consideration.

Senator Obenshain moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows:

YEAS--0. NAYS--38. RULE 36--0.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

RULE 36--0.

S.B. 1113 (one thousand one hundred thirteen) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend and reenact § 18.2-60 of the Code of Virginia, relating to communicating threats of death or bodily injury to a person with intent to intimidate; penalty.

On motion of Senator Spruill, the substitute was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

STATEMENT ON VOTE

Senator Suetterlein stated that he voted yea on the question of agreeing to the substitute proposed by the House of Delegates to **S.B. 1113**, whereas he intended to vote nay.

S.B. 1163 (one thousand one hundred sixty-three) was taken up with the amendments proposed by the House of Delegates as follows:

Line 32, engrossed, after before January 1, strike
 2025
 insert
 2026
 Line 115, engrossed, after before January 1, strike
 2025

On motion of Senator Hanger, the amendments were agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

2026

insert

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0. RULE 36--0.

 $S.B.\ 1242$ (one thousand two hundred forty-two) was taken up with the amendments proposed by the House of Delegates as follows:

1. Line 5, engrossed, Title, after *prosequi* insert or dismissal

2. Line 5, engrossed, Title, after prosequi;

strike

the remainder of line 5 and through violations on line 6

insert

revocation proceedings

3. Line 20, engrossed, after prosequi

insert

or dismissal

4. At the beginning of line 21, engrossed

strike

adjudication of an alleged violation of probation

insert

a revocation proceeding pursuant to § 19.2-306

On motion of Senator Edwards, the amendments were agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1297 (one thousand two hundred ninety-seven) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend and reenact §§ 63.2-1603, 63.2-1606, and 63.2-1609 of the Code of Virginia, relating to emergency order for adult protective services; acts of violence, force, or threat or financial exploitation; penalty.

On motion of Senator Obenshain, the substitute was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1320 (one thousand three hundred twenty) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend and reenact §§ 54.1-2900, 54.1-3005, 54.1-3303, and 54.1-3408 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 54.1-2957.04, relating to licensed certified midwives; licensure; practice.

On motion of Senator Lucas, the substitute was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

- **S.B. 1336** (one thousand three hundred thirty-six) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend the Code of Virginia by adding a section numbered 18.2-271.5, relating to restricted permits to operate a motor vehicle; ignition interlock systems.

On motion of Senator Stuart, the substitute was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

- **S.B. 1392** (one thousand three hundred ninety-two) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-581, relating to Consumer Data Protection Act.

On motion of Senator Marsden, the substitute was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Vogel--36.

NAYS--Petersen, Surovell--2.

RULE 36--0.

- **S.B. 1397** (one thousand three hundred ninety-seven) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact § 53.1-136 of the Code of Virginia, relating to parole and conditional release; notice and certification.

On motion of Senator Norment, the substitute was agreed to.

The recorded vote is as follows: YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0. RULE 36--0.

- S.B. 1399 (one thousand three hundred ninety-nine) was taken up with the amendments proposed by the House of Delegates as follows:
 - 1. Line 4, introduced, Title, after 15.2-5500, insert

15.2-5501,

2. Line 11, introduced, after 15.2-5500, insert

15.2-5501,

On motion of Senator Pillion, the amendments were agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

- S.B. 1415 (one thousand four hundred fifteen) was taken up with the amendments proposed by the House of Delegates as follows:
 - 1. Line 117, engrossed, after life strike

, [the comma]

insert

2. Line 117, engrossed, after health strike

, or normal development

Senator Stanley moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

The recorded vote is as follows:

YEAS--3. NAYS--35. RULE 36--0.

YEAS--Hashmi, Locke, Lucas--3.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Howell, Kiggans, Lewis, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

RULE 36--0.

- **S.B. 1423** (one thousand four hundred twenty-three) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

Senator McPike moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows:

YEAS--0. NAYS--37. RULE 36--0.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--37.

RULE 36--0.

- **S.B. 1456** (one thousand four hundred fifty-six) was taken up with the amendment proposed by the House of Delegates as follows:
 - 1. Line 114, engrossed, after 4 insert

, but under no circumstances shall such juvenile be detained pursuant to this section in a secure detention facility

On motion of Senator Marsden, the amendment was agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1465 (one thousand four hundred sixty-five) was taken up with the amendments proposed by the House of Delegates as follows:

1. Line 51, engrossed, after *Services* insert

or the Charitable Gaming Board

2. Line 98, engrossed, after Services

insert

or the Charitable Gaming Board

3. Line 103, engrossed, after *of* [strike

up to

4. Line 108, engrossed, after *penalty of* [strike

up to

Senator Reeves moved that the amendments be rejected.

The question was put on agreeing to the amendments.

The amendments were rejected.

The recorded vote is as follows:

YEAS--0. NAYS--37. RULE 36--0.

YEAS--0.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

RULE 36--0.

S.B. 1475 (one thousand four hundred seventy-five) was taken up with the amendments proposed by the House of Delegates as follows:

1. Line 14, engrossed, after shall

strike

only

2. Line 15, engrossed, after warrant

insert

only

3. Line 107, engrossed, after unless

insert

circumstances require the issuance of

4. Line 107, engrossed, after *unless the warrant* strike

was issued

5. Line 108, engrossed, after *subsection*, strike

when circumstances necessitate,

On motion of Senator Stuart, the amendments were agreed to.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

S.B. 1146 (one thousand one hundred forty-six) was taken up.

On motion of Senator Howell, the Senate acceded to the request of the House of Delegates for a committee of conference on the bill.

The recorded vote is as follows:

YEAS--38. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--38.

NAYS--0.

RULE 36--0.

CONFERENCE PROCEDURES

Senator Howell, Chair of the Committee on Finance and Appropriations, appointed Senators Howell, Barker, and Newman, the conferees on the part of the Senate for **S.B. 1146** (one thousand one hundred forty-six).

HOUSE BILLS ON THIRD READING

On motion of Senator Saslaw, the following House bills were passed by temporarily:

H.B. 1804 (one thousand eight hundred four).

H.B. 1837 (one thousand eight hundred thirty-seven).

H.B. 1921 (one thousand nine hundred twenty-one).

H.B. 1982 (one thousand nine hundred eighty-two).

H.B. 1983 (one thousand nine hundred eighty-three).

H.B. 2058 (two thousand fifty-eight).

- H.B. 2068 (two thousand sixty-eight).
- H.B. 2187 (two thousand one hundred eighty-seven).
- H.B. 2302 (two thousand three hundred two).
- **H.B. 2174** (two thousand one hundred seventy-four), on motion of Senator Saslaw, was passed by temporarily.
- **H.B. 2263** (two thousand two hundred sixty-three), on motion of Senator Saslaw, was passed by temporarily.
- **H.B. 1902** (one thousand nine hundred two), on motion of Senator Petersen, was passed by temporarily.
- **H.B. 1965** (one thousand nine hundred sixty-five), on motion of Senator Saslaw, was passed by temporarily.
- **H.B. 1979** (one thousand nine hundred seventy-nine), on motion of Senator Saslaw, was passed by temporarily.
 - H.B. 2040 (two thousand forty) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 60.2-619 and 60.2-633 of the Code of Virginia, relating to unemployment compensation; continuation of benefits; repayment of overpayments.

The reading of the substitute was waived.

Senator Saslaw moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 60.2-619 and 60.2-633 of the Code of Virginia, relating to unemployment compensation; continuation of benefits; repayment of overpayments.

The reading of the substitute was waived.

On motion of Senator Howell, the substitute was agreed to.

The substitute was ordered to be engrossed.

Senator Saslaw moved that H.B. 2040 be passed with its title.

H.B. 2040, on motion of Senator Barker, was passed by temporarily.

H.B. 2118 (two thousand one hundred eighteen), on motion of Senator Saslaw, was passed by temporarily.

H.B. 2307 (two thousand three hundred seven) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on General Laws and Technology was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-581, relating to Consumer Data Protection Act.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2307, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--32. NAYS--7. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Vogel--32.

NAYS--Deeds, McPike, Morrissey, Petersen, Pillion, Suetterlein, Surovell--7.

RULE 36--0.

UNFINISHED BUSINESS—SENATE RECONSIDERATION

Senator Suetterlein moved to reconsider the vote by which the substitute proposed by the House of Delegates to **S.B. 1392** (one thousand three hundred ninety-two) was agreed to.

The motion was agreed to.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

On motion of Senator Marsden, the substitute was agreed to.

The recorded vote is as follows: YEAS--32. NAYS--7. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Vogel--32.

NAYS--Deeds, McPike, Morrissey, Petersen, Pillion, Suetterlein, Surovell--7. RULE 36--0.

HOUSE BILLS ON THIRD READING

H.B. 2330 (two thousand three hundred thirty) was read by title the third time.

The following amendment proposed by the Committee on Finance and Appropriations was offered:

1. After line 307, engrossed insert

4. That the provisions of this act shall not become effective unless reenacted by the 2022 Session of the General Assembly.

The reading of the amendment was waived.

Senator Mason moved that the amendment be rejected.

The question was put on agreeing to the amendment.

The amendment was rejected.

Senator Mason offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 56-576 and 56-585.6 of the Code of Virginia, relating to electric utilities; Percentage of Income Payment Program.

On motion of Senator Mason, the reading of the substitute was waived.

On motion of Senator Mason, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2330, on motion of Senator Mason, was passed with its title.

The recorded vote is as follows: YEAS--20. NAYS--19. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Kiggans, McDougle, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--19. RULE 36--0.

H.B. 2332 (two thousand three hundred thirty-two) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Commerce and Labor was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 38.2-4214 and 38.2-4319 of the Code of Virginia and to amend the Code of Virginia by adding in Title 38.2 a chapter numbered 66, consisting of sections numbered 38.2-6600 through 38.2-6606, relating to the Commonwealth Health Reinsurance Program; established; special fund established; federal waiver application.

The reading of the substitute was waived.

On motion of Senator Barker, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 2332, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows: YEAS--26. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Spruill, Surovell, Vogel--26.

NAYS--Chase, Cosgrove, DeSteph, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein--13.

RULE 36--0.

Senator Saslaw moved that the following House bills, the titles of the bills having been printed in the Calendar for their third reading as required by Article IV, Section 11, of the Constitution, be placed before the Senate by number only:

- H.B. 1804 (one thousand eight hundred four).
- H.B. 1837 (one thousand eight hundred thirty-seven).
- H.B. 1921 (one thousand nine hundred twenty-one).
- H.B. 1982 (one thousand nine hundred eighty-two).
- H.B. 1983 (one thousand nine hundred eighty-three).
- H.B. 2058 (two thousand fifty-eight).
- H.B. 2068 (two thousand sixty-eight).
- H.B. 2187 (two thousand one hundred eighty-seven).
- H.B. 2302 (two thousand three hundred two).

The motion was agreed to.

H.B. 2068 (two thousand sixty-eight) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 3.2-303, 3.2-304, and 3.2-310 of the Code of Virginia and to amend the Code of Virginia by adding in Chapter 3.1 of Title 3.2 a section numbered 3.2-311, relating to Local Food and Community Infrastructure Grant Program.

The reading of the substitute was waived.

Senator Petersen moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

H.B. 2302 (two thousand three hundred two) was taken up.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to require that farmers market food and beverage sales be considered essential during a declared state of emergency.

The reading of the substitute was waived.

On motion of Senator Petersen, the substitute was agreed to.

The substitute was ordered to be engrossed.

Senator Saslaw moved that the passage of the House bills that follow be considered en bloc.

The motion was agreed to.

On motion of Senator Saslaw, the following House bills were passed en bloc with their titles:

- H.B. 1804 (one thousand eight hundred four).
- H.B. 1837 (one thousand eight hundred thirty-seven).
- H.B. 1921 (one thousand nine hundred twenty-one).
- H.B. 1982 (one thousand nine hundred eighty-two).
- H.B. 2058 (two thousand fifty-eight).
- H.B. 2068 (two thousand sixty-eight).
- H.B. 2187 (two thousand one hundred eighty-seven).
- H.B. 2302 (two thousand three hundred two) with substitute.

The recorded vote is as follows:

YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 1983 (one thousand nine hundred eighty-three), on motion of Senator Stuart, was passed with its title.

The recorded vote is as follows: YEAS--39. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Dunnavant, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--39.

NAYS--0.

RULE 36--0.

H.B. 2174 (two thousand one hundred seventy-four) was taken up and read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 23.1-701 of the Code of Virginia and to amend the Code of Virginia by adding in Title 2.2 a chapter numbered 27.1, consisting of sections numbered 2.2-2744 through 2.2-2757, relating to state-facilitated IRA savings program; establishment.

The reading of the substitute was waived.

On motion of Senator Howell, the substitute was agreed to.

Senator Suetterlein offered an amendment in the nature of a substitute, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 23.1-701 of the Code of Virginia and to amend the Code of Virginia by adding in Title 2.2 a chapter numbered 27.1, consisting of sections numbered 2.2-2744 through 2.2-2757, relating to state-facilitated IRA savings program; establishment.

Senator Suetterlein withdrew the substitute.

Senator Suetterlein offered the following amendments to the committee substitute:

1. Line 22, committee substitute, after basis

strike

the remainder of line 22, all of lines 23, 24, and 25, and through implementation on line 26

2. Line 29, committee substitute, after Code.

strike

the remainder of line 29 and all of line 30

3. Line 96, committee substitute, after chapter

strike

the remainder of line 96 and through annually on line 97

4. Line 135, committee substitute, after Any

strike

employer that is not an

5. Line 136, committee substitute, after However,

strike

such

insert

the eligible

6. Line 147, committee substitute, after employers

strike

shall enroll

insert

may facilitate the participation of

7. Line 151, committee substitute, after *D*.

strike

Participation in the Program shall be mandatory for eligible employers. Eligible employers

insert

An eligible employer who facilitates the participation of its eligible employees in the Program pursuant to subsection A

8. Line 152, committee substitute, after Plan.

strike

Eligible

insert

Such

9. Line 155, committee substitute, after of

strike

an eligible

insert

a participating

10. Line 291, committee substitute, after legislation,

insert

and

11. Line 292, committee substitute, after businesses

strike

the remainder of line 292, all of line 293, and through act on line 294

On motion of Senator Suetterlein, the reading of the amendments was waived.

Senator Suetterlein moved that the amendments be agreed to.

The question was put on agreeing to the amendments.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows: YEAS--19. NAYS--20. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Dunnavant, Hanger, Hashmi, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--19.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20. RULE 36--0.

The amendments were rejected.

STATEMENT ON VOTE

Senator Hashmi stated that she voted yea on the question of agreeing to the amendments offered by Senator Suetterlein to the substitute to **H.B. 2174**, whereas she intended to vote nay.

The substitute was ordered to be engrossed.

Senator Howell moved that H.B. 2174 be passed with its title.

- H.B. 2174, on motion of Senator Petersen, was passed by temporarily.
- **H.B. 2263** (two thousand two hundred sixty-three) was taken up and, on motion of Senator Surovell, was passed by for the day.
- **H.B. 1902** (one thousand nine hundred two), on motion of Senator Petersen, was passed by temporarily.
 - H.B. 1965 (one thousand nine hundred sixty-five) was taken up and read by title the third time.

The following amendments proposed by the Committee on Commerce and Labor were offered:

1. Line 66, engrossed, after regulations for

insert

an Advanced Clean Cars Program that includes

2. Line 67, engrossed, after program

insert

only for motor vehicles with a gross vehicle weight of 14,000 pounds or less

3. Line 71, engrossed

strike

all of lines 71 and 72

4. After line 82, engrossed

insert

5. That the regulations required to be adopted by the State Air Pollution Control Board (Board) pursuant to § 10.1-1307.04 of the Code of Virginia, as created by this act, shall allow any motor vehicle manufacturer to establish a Virginia-specific zero-emission vehicle (ZEV) credit account in the ZEV Credit System and to make a one-time deposit into its account a number of proportional credits equal to its 2025 model year starting

California credit balance multiplied by the ratio of the average number of passenger cars and light-duty trucks that a manufacturer produced and delivered for sale in Virginia to the average number of passenger cars and light-duty trucks the manufacturer produced and delivered for sale in California during the time period selected by the manufacturer for calculation of its ZEV requirement for the first effective model year. The deposit shall be made only after all credit obligations for model year 2024 and earlier have been satisfied. While manufacturers may trade or sell these proportional credits to any other manufacturer, these credits may be used to meet up to 18 percent of its ZEV program credit requirements in any model year, unless and until the required regulations are superseded by regulations updating the Advanced Clean Car Program. The Board shall not award or provide any vehicle manufacturer with any other form of ZEV program credits or credit balance prior to the effective date or at the beginning of the compliance period of the regulations required by § 10.1-1307.04 of the Code of Virginia, as created by this act. As part of any update to the required regulations to ensure compliance of the ZEV program with the federal Clean Air Act (42 U.S.C. § 7401 et. seq.), the Board shall adjust, if necessary, restrictions on the use of the proportional credits remaining in manufacturers' Virginia accounts in order to ensure that the percentage of ZEVs required to be delivered for sale under Virginia's ZEV program is approximately equivalent to, but does not exceed, the percentage required under California's ZEV program, taking into account only existing ZEV credit banks, any changes in restrictions on their use, and the effects of new regulatory requirements on the amount and timing of ZEVs required to be delivered for sale.

The reading of the amendments was waived.

On motion of Senator Spruill, the amendments were agreed to.

Senator Newman offered the following amendments:

```
1. Line 73, engrossed, after 2. That strike
```

the remainder of line 73, all of lines 74 and 75, and through **Regulations** on line 76

insert

, except for the third enactment of this act, the provisions of this act shall not become effective unless reenacted by the 2022 Session of the General Assembly

2. Line 77, engrossed, after 3. That

strike

the remainder of line 77, all of line 78, and through 2024 on line 79

insert

the State Air Pollution Control Board shall promulgate regulations to implement the provisions of this act; however, such regulations shall not become effective unless the provisions of this act are reenacted by the 2022 Session of the General Assembly

3. Line 80, engrossed, after 4.

insert

That, notwithstanding the provisions of § 10.1-1307.04 of the Code of Virginia, as created by this act, beginning July 1, 2022, the State Air Pollution Control Board shall not have the authority to adopt by regulation and enforce any new model year standards relating to the control of emissions from new motor vehicles or new motor vehicle engines or amend any existing standards unless such authority is expressly granted by the General Assembly in an act of assembly.

5.

On motion of Senator Newman, the reading of the amendments was waived.

Senator Newman moved that the amendments be agreed to.

The question was put on agreeing to the amendments.

The yeas and nays were called for; and, being desired by one-fifth of the Senators present, the yeas and nays were ordered.

The recorded vote is as follows:

YEAS--18. NAYS--20. RULE 36--0.

YEAS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Vogel--18.

NAYS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Spruill, Surovell--20.

RULE 36--0.

The amendments were rejected.

The committee amendments were ordered to be engrossed.

H.B. 1965, on motion of Senator Spruill, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stanley, Stuart, Suetterlein--15.

RULE 36--0.

H.B. 1979 (one thousand nine hundred seventy-nine) was taken up and read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Finance and Appropriations was offered, having been printed separately, with its title reading as follows:

A BILL to amend the Code of Virginia by adding in Title 67 a chapter numbered 18, consisting of sections numbered 67-1800 through 67-1806, relating to electric vehicle rebate program; creation and funding; report.

The reading of the substitute was waived.

On motion of Senator Howell, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1979, on motion of Senator Howell, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

H.B. 2040 (two thousand forty) was taken up.

Senator Barker offered the following amendment to the substitute:

1. Line 143, substitute, after act.

insert

No employer shall be responsible for benefits charges related to an overpayment that is waived pursuant to this act.

On motion of Senator Barker, the reading of the amendment was waived.

On motion of Senator Barker, the amendment was agreed to.

The substitute with amendment was ordered to be engrossed.

H.B. 2040, on motion of Senator Barker, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--18. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--18.

RULE 36--0.

- **H.B. 2118** (two thousand one hundred eighteen) was taken up and, on motion of Senator Barker, was passed by for the day.
- **H.B. 1760** (one thousand seven hundred sixty) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--30. NAYS--8. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Surovell, Vogel--30.

NAYS--Chase, Cosgrove, McDougle, Norment, Obenshain, Reeves, Stuart, Suetterlein--8. RULE 36--0.

H.B. 1810 (one thousand eight hundred ten) was read by title the third time and, on motion of Senator Deeds, was passed with its title.

The recorded vote is as follows:

YEAS--34. NAYS--4. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--34.

NAYS--Chase, Cosgrove, DeSteph, Reeves--4.

RULE 36--0.

H.B. 1819 (one thousand eight hundred nineteen) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--7. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Locke, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell--31.

NAYS--Chase, Lucas, McDougle, Obenshain, Peake, Suetterlein, Vogel--7. RULE 36--0.

H.B. 1836 (one thousand eight hundred thirty-six) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Agriculture, Conservation and Natural Resources was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact § 2.2-215 of the Code of Virginia, relating to the Secretary of Natural Resources.

The reading of the substitute was waived.

On motion of Senator Petersen, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1836, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows: YEAS--25. NAYS--13. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Stuart, Surovell, Vogel--25.

NAYS--Chase, DeSteph, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein--13.

RULE 36--0.

H.B. 1888 (one thousand eight hundred eighty-eight) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Privileges and Elections was offered, having been printed separately, with its title reading as follows:

A BILL to amend and reenact §§ 24.2-101, 24.2-416.1, 24.2-603, 24.2-653.1, 24.2-704, 24.2-705, 24.2-706 through 24.2-711, and 24.2-712 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 24.2-103.2, 24.2-667.1, and 24.2-707.1, relating to absentee voting; procedural and process reforms; availability and accessibility reforms; penalty.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.B. 1888, on motion of Senator Deeds, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Locke, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

H.B. 1928 (one thousand nine hundred twenty-eight) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

H.B. 1952 (one thousand nine hundred fifty-two), on motion of Senator Surovell, was passed by for the day.

H.B. 1958 (one thousand nine hundred fifty-eight) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows: YEAS--32. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--32.

NAYS--Chase, McDougle, Obenshain, Peake, Suetterlein--5. RULE 36--0.

RECONSIDERATION

Senator Stanley moved to reconsider the vote by which **H.B. 1819** (one thousand eight hundred nineteen) was passed with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Edwards--1.

RULE 36--0.

H.B. 1819, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--31. NAYS--5. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--31.

NAYS--Chase, McDougle, Obenshain, Peake, Suetterlein--5. RULE 36--0.

H.B. 1968 (one thousand nine hundred sixty-eight) was read by title the third time and, on motion of Senator Deeds, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

- H.B. 2020 (two thousand twenty), on motion of Senator Suetterlein, was passed by for the day.
- **H.B. 2030** (two thousand thirty), on motion of Senator Petersen, was passed by for the day.
- **H.B. 2125** (two thousand one hundred twenty-five) was read by title the third time and, on motion of Senator Deeds, was passed with its title.

The recorded vote is as follows:

YEAS--20. NAYS--17. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--17.

RULE 36--0.

- **H.B. 2198** (two thousand one hundred ninety-eight), on motion of Senator Vogel, was passed by for the day.
- **H.B. 2250** (two thousand two hundred fifty) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--22. NAYS--15. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Saslaw, Spruill, Stanley, Surovell--22.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Lucas, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Stuart, Suetterlein, Vogel--15.

RULE 36--0.

H.B. 2275 (two thousand two hundred seventy-five) was read by title the third time.

Senator Petersen moved that H.B. 2275 be passed with its title.

The question was put on passing H.B. 2275 with its title.

H.B. 2275 was defeated with its title.

The recorded vote is as follows:

YEAS--18. NAYS--19. RULE 36--0.

YEAS--Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--18.

NAYS--Barker, Chase, Cosgrove, DeSteph, Hanger, Kiggans, Lucas, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--19.

RULE 36--0.

H.B. 2311 (two thousand three hundred eleven) was read by title the third time and, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

H.B. 2324 (two thousand three hundred twenty-four) was read by title the third time.

Senator Deeds moved that H.B. 2324 be passed with its title.

The question was put on passing H.B. 2324 with its title.

H.B. 2324 was defeated with its title, having failed to receive the necessary affirmative votes required by Article IV, Section 13, of the Constitution.

The recorded vote is as follows:

YEAS--19. NAYS--14. RULE 36--4.

YEAS--Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--19.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, Norment, Obenshain, Peake, Pillion, Ruff, Stanley, Stuart, Suetterlein, Vogel--14.

RULE 36--Barker, McDougle, Newman, Reeves--4.

H.B. 2174 (two thousand one hundred seventy-four) was taken up.

Senator Petersen offered the following amendments to the substitute:

1. Line 20, committee substitute, after *employed* insert

at least thirty hours a week

2. Line 23, committee substitute, after *or more* insert

eligible

On motion of Senator Petersen, the reading of the amendments was waived.

Senator Petersen moved that the amendments be agreed to.

H.B. 2174, on motion of Senator McClellan, was passed by temporarily.

H.B. 1902 (one thousand nine hundred two) was taken up and read by title the third time.

Senator Petersen moved that H.B. 1902 be passed with its title.

H.B. 1902, on motion of Senator Petersen, was passed by temporarily.

HOUSE BILLS ON SECOND READING

Senator Saslaw moved that the Rules be suspended and the second reading of the titles of the following House bills as required by Article IV, Section 11, of the Constitution, be dispensed with:

- H.B. 1747 (one thousand seven hundred forty-seven).
- **H.B. 1838** (one thousand eight hundred thirty-eight).
- H.B. 1843 (one thousand eight hundred forty-three).
- H.B. 1865 (one thousand eight hundred sixty-five).
- H.B. 1876 (one thousand eight hundred seventy-six).
- H.B. 1894 (one thousand eight hundred ninety-four).
- H.B. 1950 (one thousand nine hundred fifty).
- H.B. 1976 (one thousand nine hundred seventy-six).
- H.B. 1987 (one thousand nine hundred eighty-seven).
- H.B. 1988 (one thousand nine hundred eighty-eight).
- H.B. 2018 (two thousand eighteen).
- H.B. 2039 (two thousand thirty-nine).
- H.B. 2079 (two thousand seventy-nine).
- H.B. 2086 (two thousand eighty-six).
- H.B. 2098 (two thousand ninety-eight).
- H.B. 2105 (two thousand one hundred five).
- H.B. 2116 (two thousand one hundred sixteen).
- H.B. 2120 (two thousand one hundred twenty).
- H.B. 2135 (two thousand one hundred thirty-five).
- H.B. 2154 (two thousand one hundred fifty-four).
- H.B. 2162 (two thousand one hundred sixty-two).
- H.B. 2190 (two thousand one hundred ninety).
- H.B. 2192 (two thousand one hundred ninety-two).
- H.B. 2193 (two thousand one hundred ninety-three).
- H.B. 2220 (two thousand two hundred twenty).
- H.B. 2234 (two thousand two hundred thirty-four).
- H.B. 2238 (two thousand two hundred thirty-eight). H.B. 2298 (two thousand two hundred ninety-eight).
- H.B. 2299 (two thousand two hundred ninety-nine).
- **11.B.** 2299 (two thousand two hundred limiting-limiting)
- H.B. 2300 (two thousand three hundred).
- H.B. 2316 (two thousand three hundred sixteen).
- H.B. 2317 (two thousand three hundred seventeen).
- H.B. 1823 (one thousand eight hundred twenty-three).
- H.B. 1831 (one thousand eight hundred thirty-one).
- H.B. 1853 (one thousand eight hundred fifty-three).
- H.B. 1889 (one thousand eight hundred eighty-nine).
- H.B. 1900 (one thousand nine hundred).
- H.B. 1908 (one thousand nine hundred eight).
- H.B. 1930 (one thousand nine hundred thirty).
- H.B. 1963 (one thousand nine hundred sixty-three).
- H.B. 1980 (one thousand nine hundred eighty).
- H.B. 1991 (one thousand nine hundred ninety-one).
- H.B. 2013 (two thousand thirteen).
- H.B. 2019 (two thousand nineteen).
- H.B. 2025 (two thousand twenty-five).
- H.B. 2029 (two thousand twenty-nine).
- H.B. 2038 (two thousand thirty-eight).

```
H.B. 2055 (two thousand fifty-five).
```

H.B. 2061 (two thousand sixty-one).

H.B. 2072 (two thousand seventy-two).

H.B. 2130 (two thousand one hundred thirty).

H.B. 2139 (two thousand one hundred thirty-nine).

H.B. 2147 (two thousand one hundred forty-seven).

 $\textbf{H.B. 2176} \ (two \ thousand \ one \ hundred \ seventy-six).$

H.B. 2288 (two thousand two hundred eighty-eight).

H.B. 2295 (two thousand two hundred ninety-five).

H.B. 2327 (two thousand three hundred twenty-seven).

The motion was agreed to.

The recorded vote is as follows:

YEAS--32. NAYS--0. RULE 36--0.

YEAS--Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--32.

NAYS--0.

RULE 36--0.

On motion of Senator Saslaw, the following House bills were passed by for the day:

```
H.B. 1747 (one thousand seven hundred forty-seven).
```

H.B. 1838 (one thousand eight hundred thirty-eight).

H.B. 1843 (one thousand eight hundred forty-three).

H.B. 1865 (one thousand eight hundred sixty-five).

H.B. 1876 (one thousand eight hundred seventy-six).

H.B. 1894 (one thousand eight hundred ninety-four). **H.B. 1950** (one thousand nine hundred fifty).

H.B. 1976 (one thousand nine hundred seventy-six).

H.B. 1987 (one thousand nine hundred eighty-seven).

H.B. 1988 (one thousand nine hundred eighty-eight).

H.B. 2018 (two thousand eighteen).

H.B. 2039 (two thousand thirty-nine).

H.B. 2079 (two thousand seventy-nine).

H.B. 2086 (two thousand eighty-six).

H.B. 2098 (two thousand ninety-eight).

H.B. 2105 (two thousand one hundred five).

H.B. 2116 (two thousand one hundred sixteen).

H.B. 2120 (two thousand one hundred twenty).

H.B. 2135 (two thousand one hundred thirty-five).

 $\textbf{H.B. 2154} \ (two \ thousand \ one \ hundred \ fifty-four).$

H.B. 2162 (two thousand one hundred sixty-two).

H.B. 2190 (two thousand one hundred ninety).

H.B. 2192 (two thousand one hundred ninety-two).

H.B. 2193 (two thousand one hundred ninety-three).

H.B. 2220 (two thousand two hundred twenty).

H.B. 2234 (two thousand two hundred thirty-four).

H.B. 2238 (two thousand two hundred thirty-eight).

H.B. 2298 (two thousand two hundred ninety-eight).

- H.B. 2299 (two thousand two hundred ninety-nine).
- H.B. 2300 (two thousand three hundred).
- H.B. 2316 (two thousand three hundred sixteen).
- H.B. 2317 (two thousand three hundred seventeen).
- H.B. 1823 (one thousand eight hundred twenty-three).
- H.B. 1831 (one thousand eight hundred thirty-one).
- H.B. 1853 (one thousand eight hundred fifty-three).
- H.B. 1889 (one thousand eight hundred eighty-nine).
- H.B. 1900 (one thousand nine hundred).
- H.B. 1908 (one thousand nine hundred eight).
- H.B. 1930 (one thousand nine hundred thirty).
- H.B. 1963 (one thousand nine hundred sixty-three).
- H.B. 1980 (one thousand nine hundred eighty).
- H.B. 1991 (one thousand nine hundred ninety-one).
- H.B. 2013 (two thousand thirteen).
- H.B. 2019 (two thousand nineteen).
- H.B. 2025 (two thousand twenty-five).
- H.B. 2029 (two thousand twenty-nine).
- H.B. 2038 (two thousand thirty-eight).
- H.B. 2055 (two thousand fifty-five).
- H.B. 2061 (two thousand sixty-one).
- H.B. 2072 (two thousand seventy-two).
- H.B. 2130 (two thousand one hundred thirty).
- H.B. 2139 (two thousand one hundred thirty-nine).
- H.B. 2147 (two thousand one hundred forty-seven).
- H.B. 2176 (two thousand one hundred seventy-six).
- H.B. 2288 (two thousand two hundred eighty-eight).
- H.B. 2295 (two thousand two hundred ninety-five).
- H.B. 2327 (two thousand three hundred twenty-seven).

HOUSE JOINT RESOLUTIONS ON THIRD READING

H.J.R. 555 (five hundred fifty-five) was read by title the third time.

The amendment in the nature of a substitute proposed by the Committee on Privileges and Elections was offered, having been printed separately, as follows:

HOUSE JOINT RESOLUTION NO. 555 AMENDMENT IN THE NATURE OF A SUBSTITUTE

Proposing an amendment to Section 1 of Article II of the Constitution of Virginia, relating to qualifications of voters and the right to vote; persons not entitled to vote.

RESOLVED by the House of Delegates, the Senate concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 1 of Article II of the Constitution of Virginia as follows:

ARTICLE II FRANCHISE AND OFFICERS

Section 1. Qualifications of voters.

In elections by the people, the qualifications of voters shall be as follows: Each voter shall be a citizen of the United States, shall be eighteen years of age, shall fulfill the residence requirements set forth in this section, and shall be registered to vote pursuant to this article. No Every person who meets these qualifications shall have the fundamental right to vote in the Commonwealth, and such right shall not be abridged by law, except that no person who has been convicted of a felony shall be qualified entitled to vote unless his civil rights have been restored by the Governor or other appropriate authority. As prescribed by law during any period of imprisonment, no person adjudicated to be mentally incompetent shall be qualified to vote until his competency has been reestablished and no person who has been adjudicated by a court of competent jurisdiction to lack the capacity to understand the act of voting shall be entitled to vote until his capacity has been reestablished as prescribed by law.

The residence requirements shall be that each voter shall be a resident of the Commonwealth and of the precinct where he votes. Residence, for all purposes of qualification to vote, requires both domicile and a place of abode. The General Assembly may provide for persons who are employed overseas, and their spouses and dependents residing with them, and who are qualified to vote except for relinquishing their place of abode in the Commonwealth while overseas, to vote in the Commonwealth subject to conditions and time limits defined by law. The General Assembly may provide for persons who are qualified to vote except for having moved their residence from one precinct to another within the Commonwealth to continue to vote in a former precinct subject to conditions and time limits defined by law. The General Assembly may also provide, in elections for President and Vice President of the United States, alternatives to registration for new residents of the Commonwealth.

Any person who will be qualified with respect to age to vote at the next general election shall be permitted to register in advance and also to vote in any intervening primary or special election.

The reading of the substitute was waived.

On motion of Senator Deeds, the substitute was agreed to.

The substitute was ordered to be engrossed.

H.J.R. 555, on motion of Senator Deeds, was agreed to.

The recorded vote is as follows:

YEAS--20. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Surovell--20.

NAYS--Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Stuart, Suetterlein, Vogel--16.

RULE 36--0.

STATEMENT ON VOTE

Senator Chase stated that she was recorded as not voting on the question of agreeing to **H.J.R. 555**, whereas she intended to vote nay.

H.J.R. 582 (five hundred eighty-two) was read by title the third time.

HOUSE JOINT RESOLUTION NO. 582

Proposing an amendment to Section 15-A of Article I of the Constitution of Virginia, relating to marriage; repeal of same-sex marriage prohibition; affirmative right to marry.

RESOLVED by the House of Delegates, the Senate concurring, a majority of the members elected to each house agreeing, That the following amendment to the Constitution of Virginia be, and the same hereby is, proposed and referred to the General Assembly at its first regular session held after the next general election of members of the House of Delegates for its concurrence in conformity with the provisions of Section 1 of Article XII of the Constitution of Virginia, namely:

Amend Section 15-A of Article I of the Constitution of Virginia as follows:

ARTICLE I BILL OF RIGHTS

Section 15-A. Marriage Fundamental right to marry.

That only a union between one man and one woman may be a marriage valid in or recognized by this Commonwealth and its political subdivisions the right to marry is a fundamental right, inherent in the liberty of persons, and marriage is one of the vital personal rights essential to the orderly pursuit of happiness.

This Commonwealth and its political subdivisions shall not create or recognize a legal status for relationships of unmarried individuals that intends to approximate the design, qualities, significance, or effects of marriage. Nor shall this Commonwealth or its political subdivisions create or recognize another union, partnership, or other legal status to which is assigned the rights, benefits, obligations, qualities, or effects of marriage and agents shall issue marriage licenses, recognize marriages, and treat all marriages equally under the law regardless of the sex or gender of the parties to the marriage.

Religious organizations and clergy acting in their religious capacity shall have the right to refuse to perform any marriage.

H.J.R. 582, on motion of Senator Ebbin, was agreed to.

The recorded vote is as follows:

YEAS--22. NAYS--12. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Petersen, Saslaw, Surovell, Vogel--22. NAYS--Chase, DeSteph, Hanger, McDougle, Newman, Obenshain, Peake, Pillion, Reeves, Ruff,

RULE 36--0.

Stanley, Stuart--12.

HOUSE JOINT RESOLUTION ON SECOND READING

H.J.R. 629 (six hundred twenty-nine) was read by title the second time.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 19, 2021

THE HOUSE OF DELEGATES HAS PASSED WITH AMENDMENTS THE FOLLOWING SENATE BILL:

S.B. 1398. A BILL to amend and reenact §§ 58.1-602, 58.1-603, as it is currently effective and as it may become effective, 58.1-3819, as it shall become effective, 58.1-3829, as it shall become effective, 58.1-3824, 58.1-3825, 58.1-3825.2, 58.1-3825.3, as it shall become effective, 58.1-3842, and 58.1-3843 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 2.2-2320.2 and 58.1-612.2 and by adding in Article 6 of Chapter 38 of Title 58.1 a section numbered 58.1-3818.8, relating to retail sales and transient occupancy taxes on room rentals.

THE HOUSE OF DELEGATES HAS PASSED WITH SUBSTITUTES THE FOLLOWING SENATE BILLS:

- **S.B. 1127.** A BILL to amend and reenact §§ 18.2-334.2, 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.25, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.34 of the Code of Virginia, relating to charitable gaming; conduct of instant bingo, network bingo, pull tabs, and seal cards.
- **S.B. 1197.** A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 58.1 an article numbered 13.4, consisting of sections numbered 58.1-439.29 and 58.1-439.30, relating to Virginia housing opportunity tax credit.
- **S.B. 1302.** A BILL to amend and reenact §§ 37.2-311.1, as it shall become effective, 56-484.12, 56-484.17, and 56-484.17:1 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 37.2-311.2, 37.2-311.3, and 37.2-311.4 and by adding in Article 7 of Chapter 15 of Title 56 a section numbered 56-484.18:1, relating to crisis call centers; Crisis Call Center Fund established.

THE HOUSE OF DELEGATES HAS INSISTED ON ITS SUBSTITUTE AND HAS REQUESTED A COMMITTEE OF CONFERENCE ON THE FOLLOWING SENATE BILL:

S.B. 1406. A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-205, as it is currently effective and as it shall become effective, 4.1-205, as it is currently effective and as it shall

become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29, consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6, consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, containing chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Chapter 7 of Title 18.2 an article numbered 1.4, consisting of sections numbered 18.2-265.22 through 18.2-265.28, by adding a section numbered 19.2-392.2:1, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

THE HOUSE OF DELEGATES HAS PASSED THE FOLLOWING SENATE BILL:

S.B. 1417. A BILL to amend and reenact § 3.2-6591 of the Code of Virginia and to amend the Code of Virginia by adding in Article 13 of Chapter 65 of Title 3.2 a section numbered 3.2-6593.1, relating to animal testing facilities; adoption of dogs and cats; civil penalty.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow

Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

RECESS

At 2:40 p.m., Senator Saslaw moved that the Senate recess until 3:50 p.m.

The motion was agreed to.

The hour of 3:50 p.m. having arrived, the Chair was resumed.

COMMITTEE REPORT

The following bill, having been considered by the committee in session, was reported by Senator Edwards from the Committee on the Judiciary:

H.B. 1864 (one thousand eight hundred sixty-four).

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 19, 2021

THE HOUSE OF DELEGATES HAS INSISTED ON ITS SUBSTITUTE AND HAS REQUESTED A COMMITTEE OF CONFERENCE ON THE FOLLOWING SENATE BILL:

S.B. 1423. A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.

THE HOUSE OF DELEGATES HAS ACCEDED TO THE REQUEST OF THE SENATE FOR COMMITTEES OF CONFERENCE ON THE FOLLOWING HOUSE BILLS:

- **H.B. 1800.** A BILL to amend and reenact Chapter 1289 of the 2020 Acts of Assembly, as amended by Chapter 56 of the 2020 Acts of Assembly, Special Session I, which appropriated funds for the 2020-22 Biennium and provided a portion of revenues for the two years ending, respectively, on the thirtieth day of June, 2021, and the thirtieth day of June, 2022.
- **H.B. 1935.** A BILL to amend and reenact §§ 58.1-301, 58.1-322.02, and 58.1-322.03 of the Code of Virginia, relating to conformity of the Commonwealth's taxation system with the Internal Revenue Code.

- **H.B. 2168.** A BILL to amend and reenact § 18.2-325, as it is currently effective and as it shall become effective, of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 18.2-331.1, relating to illegal gambling; skills games; civil penalty; enforcement by localities and Attorney General.
- **H.B. 2273.** A BILL to amend and reenact § 58.1-609.3 of the Code of Virginia, relating to sales and use tax exemption for data centers.
- H.B. 2312. A BILL to amend and reenact §§ 2.2-221, 2.2-507, 2.2-511, 2.2-1119, 2.2-2818, 2.2-2905, 2.2-3114, 2.2-3705.3, 2.2-3711, 2.2-3802, 2.2-4024, 3.2-1010, 3.2-3906, 3.2-4112, 3.2-4113, 3.2-4114, 3.2-4114.2, 3.2-4116, 4.1-100, as it is currently effective and as it shall become effective, 4.1-101.01, 4.1-101.02, 101.07, 4.1-101.09, 4.1-101.010, 4.1-101.1, 4.1-103, as it is currently effective and as it shall become effective, 4.1-104, 4.1-105, 4.1-106, 4.1-107, 4.1-111, as it is currently effective and as it shall become effective, 4.1-112.2, 4.1-113.1, 4.1-115, 4.1-116, 4.1-118, 4.1-119, as it is currently effective and as it shall become effective, 4.1-122, 4.1-124, as it is currently effective and as it shall become effective, 4.1-128, 4.1-200, 4.1-201, as it is currently effective and as it shall become effective, 4.1-202, 4.1-205, as it is currently effective and as it shall become effective, 4.1-206, 4.1-206.1, 4.1-206.2, 4.1-206.3, 4.1-207, 4.1-207.1, 4.1-208, 4.1-212, as it is currently effective and as it shall become effective, 4.1-213, 4.1-215, as it is currently effective and as it shall become effective, 4.1-216, as it is currently effective and as it shall become effective, 4.1-216.1, 4.1-222, 4.1-224, 4.1-225, 4.1-227, as it is currently effective and as it shall become effective, 4.1-230, as it is currently effective and as it shall become effective, 4.1-231, 4.1-240, 4.1-300, 4.1-302, 4.1-303, 4.1-310, as it is currently effective and as it shall become effective, 4.1-310.1, as it is currently effective and as it shall become effective, 4.1-320, 4.1-323, 4.1-324, 4.1-325, as it is currently effective and as it shall become effective, 4.1-325.2, as it is currently effective and as it shall become effective, 4.1-329, 4.1-336, 4.1-337, 4.1-338, 4.1-348, 4.1-349, 4.1-350, 4.1-351, 4.1-352, 4.1-353, 4.1-354, 5.1-13, 9.1-101, as it is currently effective and as it shall become effective, 9.1-400, 9.1-500, 9.1-801, 9.1-1101, 15.2-1627, 15.2-2820, 16.1-69.40:1, 16.1-69.48:1, as it is currently effective and as it shall become effective, 16.1-228, 16.1-260, 16.1-273, 16.1-278.8:01, 16.1-278.9, 17.1-276, 17.1-293.1, 17.1-323, 17.1-413, 17.1-502, 18.2-46.1, 18.2-57, 18.2-247, 18.2-248, 18.2-248.01, 18.2-251, 18.2-251.02, 18.2-251.03, 18.2-251.1:1, 18.2-251.1:2, 18.2-251.1:3, 18.2-252, 18.2-254, 18.2-255, 18.2-255.1, 18.2-255.2, 18.2-258, 18.2-258.02, 18.2-258.1, 18.2-265.1, 18.2-265.2, 18.2-265.3, 18.2-287.2, 18.2-308.03, 18.2-308.09, 18.2-308.012, 18.2-308.016, 18.2-308.1:5, 18.2-308.4, 18.2-371.2, 18.2-460, 18.2-474.1, 19.2-66, 19.2-72, 19.2-74, 19.2-81, 19.2-81.1, 19.2-83.1, 19.2-188.1, 19.2-303, 19.2-303.01, 19.2-310.7, 19.2-340, 19.2-386.22 through 19.2-386.25, 19.2-389, as it is currently effective and as it shall become effective, 19.2-390, 19.2-392.02, as it is currently effective and as it shall become effective, 19.2-392.1, 19.2-392.2, 19.2-392.4, 22.1-206, 22.1-277.08, 23.1-609, 23.1-1301, 24.2-233, 33.2-613, 46.2-105.2, 46.2-347, 48-17.1, 51.1-212, 53.1-231.2, 54.1-2903, 54.1-3408.3, 54.1-3442.6, 54.1-3442.8, 58.1-3, 59.1-148.3, 65.2-107, 65.2-402, and 65.2-402.1 of the Code of Virginia; to amend the Code of Virginia by adding in Chapter 24 of Title 2.2 an article numbered 29 consisting of sections numbered 2.2-2499.1 through 2.2-2499.4, by adding sections numbered 3.2-4117.1 and 3.2-4117.2, by adding in Chapter 41.1 of Title 3.2 a section numbered 3.2-4122, by adding in Chapter 51 of Title 3.2 an article numbered 6 consisting of sections numbered 3.2-5145.6 through 3.2-5145.9, by adding in Title 4.1 a subtitle numbered II, consisting of chapters numbered 6 through 15, consisting of sections numbered 4.1-600 through 4.1-1503, by adding in Article 2 of Chapter 1 of Title 6.2 a section numbered 6.2-107.1, by adding in Title 19.2 a chapter numbered 23.2, consisting of sections numbered 19.2-392.5 through 19.2-392.13, and by adding a section numbered 46.2-341.20:7; and to repeal §§ 18.2-248.1, 18.2-250.1, 18.2-251.1, and 19.2-389.3 of the Code of Virginia, relating to marijuana; legalization of simple possession; penalties.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows: YEAS--33. NAYS--2. RULE 36--0.

YEAS--Bell, Boysko, Chase, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--33.

NAYS--McDougle, Petersen--2.

RULE 36--0.

SUPPLEMENTAL CALENDAR NO. 1

CONFERENCE PROCEDURES

Senator Howell, Chair of the Committee on Finance and Appropriations, appointed Senators Howell, Barker, Lucas, Locke, Deeds, Norment, Hanger, Saslaw (Advisor), and Newman (Advisor), the conferees on the part of the Senate for **H.B. 1800** (one thousand eight hundred).

Senator Howell, Chair of the Committee on Finance and Appropriations, appointed Senators Howell, Barker, and Newman, the conferees on the part of the Senate for **H.B. 1935** (one thousand nine hundred thirty-five).

Senator Edwards, Chair of the Committee on the Judiciary, appointed Senators Reeves, Morrissey, and Stanley, the conferees on the part of the Senate for **H.B. 2168** (two thousand one hundred sixty-eight).

Senator Howell, Chair of the Committee on Finance and Appropriations, appointed Senators McPike, Ebbin, and Ruff, the conferees on the part of the Senate for **H.B. 2273** (two thousand two hundred seventy-three).

Senator Favola, Chair of the Committee on Rehabilitation and Social Services, appointed Senators Ebbin, Lucas, McPike, Surovell, and Dunnavant, the conferees on the part of the Senate for **H.B. 2312** (two thousand three hundred twelve).

UNFINISHED BUSINESS—SENATE

S.B. 1127 (one thousand one hundred twenty-seven) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend and reenact §§ 18.2-340.16, 18.2-340.19, 18.2-340.23, 18.2-340.26:1, 18.2-340.27, 18.2-340.28, 18.2-340.28:1, and 18.2-340.34 of the Code of Virginia, relating to charitable gaming; conduct of instant bingo, network bingo, pull tabs, and seal cards.

On motion of Senator Reeves, the substitute was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--0. RULE 36--1.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--0.

RULE 36--DeSteph--1.

- **S.B. 1197** (one thousand one hundred ninety-seven), on motion of Senator Surovell, was passed by temporarily.
- **S.B. 1302** (one thousand three hundred two) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:
- A BILL to amend and reenact §§ 37.2-311.1, as it shall become effective, 56-484.12, 56-484.17, and 56-484.17:1 of the Code of Virginia and to amend the Code of Virginia by adding sections numbered 37.2-311.2 through 37.2-311.6, relating to crisis call centers; Crisis Call Center Fund established.

On motion of Senator McPike, the substitute was agreed to.

The recorded vote is as follows:

YEAS--36. NAYS--1. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

NAYS--Cosgrove--1.

RULE 36--0.

- **S.B. 1398** (one thousand three hundred ninety-eight) was taken up with the amendments proposed by the House of Delegates as follows:
 - 1. Line 28, engrossed, after revenue

strike

raised by

insert

collected from

2. Line 29, engrossed, after 58.1

insert

, after accounting for all designations and distributions of such revenue under § 58.1-638,

On motion of Senator Norment, the amendments were agreed to.

The recorded vote is as follows: YEAS--31. NAYS--6. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Peake, Petersen, Pillion, Ruff, Saslaw, Spruill, Stanley, Stuart, Surovell, Vogel--31.

NAYS--Chase, McDougle, Newman, Obenshain, Reeves, Suetterlein--6. RULE 36--0.

S.B. 1406 (one thousand four hundred six) was taken up.

On motion of Senator Ebbin, the Senate acceded to the request of the House of Delegates for a committee of conference on the bill.

The recorded vote is as follows: YEAS--27. NAYS--10. RULE 36--0.

YEAS--Barker, Bell, Boysko, Cosgrove, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Norment, Peake, Petersen, Ruff, Saslaw, Spruill, Stuart, Vogel--27.

NAYS--Chase, DeSteph, McDougle, Newman, Obenshain, Pillion, Reeves, Stanley, Suetterlein, Surovell--10.

RULE 36--0.

S.B. 1423 (one thousand four hundred twenty-three) was taken up.

On motion of Senator McPike, the Senate acceded to the request of the House of Delegates for a committee of conference on the bill.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0. RULE 36--0.

CONFERENCE PROCEDURES

Senator Favola, Chair of the Committee on Rehabilitation and Social Services, appointed Senators Ebbin, Lucas, McPike, Surovell, and Dunnavant, the conferees on the part of the Senate for **S.B. 1406** (one thousand four hundred six).

Senator Howell, Chair of the Committee on Finance and Appropriations, appointed Senators McPike, Ebbin, and Ruff, the conferees on the part of the Senate for **S.B. 1423** (one thousand four hundred twenty-three).

HOUSE BILLS ON THIRD READING

H.B. 2174 (two thousand one hundred seventy-four) was taken up.

On motion of Senator Petersen, the amendments were agreed to.

The substitute with amendments was ordered to be engrossed.

H.B. 2174, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--23. NAYS--14. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hashmi, Howell, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Pillion, Ruff, Saslaw, Spruill, Surovell, Vogel--23.

NAYS--Chase, Cosgrove, DeSteph, Hanger, Kiggans, McDougle, Newman, Norment, Obenshain, Peake, Reeves, Stanley, Stuart, Suetterlein--14.

RULE 36--0.

STATEMENT ON VOTE

Senator Ruff stated that he voted yea on the question of the passage of H.B. 2174, whereas he intended to vote nay.

H.B. 1902 (one thousand nine hundred two) was taken up.

Senator Petersen offered the following amendments:

1. Line 33, engrossed, after organization,

strike

or

2. Line 33, engrossed, after group

insert

, or state or local government entity

3. Line 33, engrossed, after services.

strike

the remainder of line 33 and all of line 34

On motion of Senator Petersen, the reading of the amendments was waived.

On motion of Senator Petersen, the amendments were agreed to.

The amendments were ordered to be engrossed.

H.B. 1902, on motion of Senator Petersen, was passed with its title.

The recorded vote is as follows:

YEAS--21. NAYS--16. RULE 36--0.

YEAS--Barker, Bell, Boysko, Deeds, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Lewis, Marsden, Mason, McClellan, McPike, Morrissey, Petersen, Saslaw, Spruill, Stuart, Surovell--21.

NAYS--Chase, Cosgrove, DeSteph, Kiggans, Lucas, McDougle, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Stanley, Suetterlein, Vogel--16.

RULE 36--0.

HOUSE BILL ON SECOND READING

H.B. 1909 (one thousand nine hundred nine) was read by title the second time.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 19, 2021

THE HOUSE OF DELEGATES HAS REJECTED THE SUBSTITUTES PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILLS:

- **H.B. 2330.** A BILL to amend and reenact §§ 56-576 and 56-585.6 of the Code of Virginia, relating to electric utilities; Percentage of Income Payment Program.
- **H.B. 2332.** A BILL to amend and reenact §§ 38.2-4214 and 38.2-4319 of the Code of Virginia and to amend the Code of Virginia by adding in Title 38.2 a chapter numbered 66, consisting of sections numbered 38.2-6600 through 38.2-6607, relating to the Commonwealth Health Reinsurance Program; established; special fund established; assessment; federal waiver application.

THE HOUSE OF DELEGATES HAS AGREED TO THE SUBSTITUTE PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILL:

H.B. 2307. A BILL to amend the Code of Virginia by adding in Title 59.1 a chapter numbered 52, consisting of sections numbered 59.1-571 through 59.1-581, relating to Consumer Data Protection Act.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows: YEAS--35. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--McDougle, Petersen--2. RULE 36--0.

UNFINISHED BUSINESS—SENATE

S.B. 1197 (one thousand one hundred ninety-seven) was taken up with the amendment in the nature of a substitute proposed by the House of Delegates, and printed separately, the title reading as follows:

A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 58.1 an article numbered 13.4, consisting of sections numbered 58.1-439.29 and 58.1-439.30, relating to Virginia housing opportunity tax credit.

Senator McClellan moved that the substitute be rejected.

The question was put on agreeing to the substitute.

The substitute was rejected.

The recorded vote is as follows: YEAS--1. NAYS--36. RULE 36--0.

YEAS--Hashmi--1.

NAYS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--36.

RULE 36--0.

UNFINISHED BUSINESS—HOUSE

H.B. 2330 (two thousand three hundred thirty) was taken up.

On motion of Senator Saslaw, the Senate insisted on its substitute and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

H.B. 2332 (two thousand three hundred thirty-two) was taken up.

On motion of Senator Saslaw, the Senate insisted on its substitute and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

HOUSE BILL ON THIRD READING RECONSIDERATION

Senator Stuart moved to reconsider the vote by which H.B. 2275 (two thousand two hundred seventy-five) was defeated with its title.

The motion was agreed to.

The recorded vote is as follows:

YEAS--35. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--0.

RULE 36--0.

H.B. 2275, on motion of Senator Stuart, was passed by for the day.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 19, 2021

THE HOUSE OF DELEGATES HAS INSISTED ON ITS SUBSTITUTE AND HAS REQUESTED A COMMITTEE OF CONFERENCE ON THE FOLLOWING SENATE BILL:

S.B. 1197. A BILL to amend the Code of Virginia by adding in Chapter 3 of Title 58.1 an article numbered 13.4, consisting of sections numbered 58.1-439.29 and 58.1-439.30, relating to Virginia housing opportunity tax credit.

THE HOUSE OF DELEGATES HAS REJECTED THE AMENDMENTS PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILL:

H.B. 1902. A BILL to amend and reenact §§ 10.1-1414 and 10.1-1422.01 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 10.1-1424.3, relating to expanded polystyrene food service containers; prohibition; civil penalty.

THE HOUSE OF DELEGATES HAS REJECTED THE SUBSTITUTE WITH AMENDMENT PROPOSED BY THE SENATE TO THE FOLLOWING HOUSE BILL:

H.B. 2040. A BILL to amend and reenact §§ 60.2-528.1, 60.2-619, and 60.2-633 of the Code of Virginia, relating to unemployment compensation; failure to respond; continuation of benefits; repayment of overpayments.

THE HOUSE OF DELEGATES HAS ACCEDED TO THE REQUEST OF THE SENATE FOR COMMITTEES OF CONFERENCE ON THE FOLLOWING HOUSE BILLS:

- **H.B. 2330.** A BILL to amend and reenact §§ 56-576 and 56-585.6 of the Code of Virginia, relating to electric utilities; Percentage of Income Payment Program.
- **H.B. 2332.** A BILL to amend and reenact §§ 38.2-4214 and 38.2-4319 of the Code of Virginia and to amend the Code of Virginia by adding in Title 38.2 a chapter numbered 66, consisting of sections numbered 38.2-6600 through 38.2-6607, relating to the Commonwealth Health Reinsurance Program; established; special fund established; assessment; federal waiver application.

IN WHICH ACTION IT REQUESTS THE CONCURRENCE OF THE SENATE.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows:

YEAS--35. NAYS--2. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--35.

NAYS--McDougle, Petersen--2.

RULE 36--0.

UNFINISHED BUSINESS—HOUSE

H.B. 2040 (two thousand forty) was taken up.

On motion of Senator Saslaw, the Senate insisted on its substitute with amendment and respectfully requested a committee of conference.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

CONFERENCE PROCEDURES

Senator Saslaw, Chair of the Committee on Commerce and Labor, appointed Senators Mason, Lewis, and Norment, the conferees on the part of the Senate for **H.B. 2330** (two thousand three hundred thirty).

Senator Saslaw, Chair of the Committee on Commerce and Labor, appointed Senators Barker, Deeds, and Newman, the conferees on the part of the Senate for **H.B. 2332** (two thousand three hundred thirty-two).

UNFINISHED BUSINESS—SENATE

S.B. 1197 (one thousand one hundred ninety-seven) was taken up.

On motion of Senator McPike, the Senate acceded to the request of the House of Delegates for a committee of conference on the bill.

The recorded vote is as follows:

YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0.

RULE 36--0.

CONFERENCE PROCEDURES

Senator Howell, Chair of the Committee on Finance and Appropriations, appointed Senators Locke, Lucas, and Vogel, the conferees on the part of the Senate for **S.B. 1197** (one thousand one hundred ninety-seven).

UNFINISHED BUSINESS—HOUSE

H.B. 1902 (one thousand nine hundred two) was taken up.

On motion of Senator Petersen, the Senate insisted on its amendments and respectfully requested a committee of conference.

The recorded vote is as follows: YEAS--37. NAYS--0. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, Deeds, DeSteph, Ebbin, Edwards, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McDougle, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Petersen, Pillion, Reeves, Ruff, Saslaw, Spruill, Stanley, Stuart, Suetterlein, Surovell, Vogel--37.

NAYS--0. RULE 36--0.

HOUSE COMMUNICATION

The following communication was received:

In the House of Delegates February 19, 2021

THE HOUSE OF DELEGATES HAS ACCEDED TO THE REQUEST OF THE SENATE FOR COMMITTEES OF CONFERENCE ON THE FOLLOWING HOUSE BILLS:

- **H.B. 1902.** A BILL to amend and reenact §§ 10.1-1414 and 10.1-1422.01 of the Code of Virginia and to amend the Code of Virginia by adding a section numbered 10.1-1424.3, relating to expanded polystyrene food service containers; prohibition; civil penalty.
- **H.B. 2040.** A BILL to amend and reenact §§ 60.2-528.1, 60.2-619, and 60.2-633 of the Code of Virginia, relating to unemployment compensation; failure to respond; continuation of benefits; repayment of overpayments.

/s/ Suzette Denslow Clerk of the House of Delegates

On motion of Senator Saslaw, the Rules were suspended and the reading of the communication from the House of Delegates was waived.

The recorded vote is as follows: YEAS--31. NAYS--3. RULE 36--0.

YEAS--Barker, Bell, Boysko, Chase, Cosgrove, DeSteph, Ebbin, Favola, Hanger, Hashmi, Howell, Kiggans, Lewis, Lucas, Marsden, Mason, McClellan, McPike, Morrissey, Newman, Norment, Obenshain, Peake, Reeves, Ruff, Saslaw, Stanley, Stuart, Suetterlein, Surovell, Vogel--31.

NAYS--Deeds, McDougle, Petersen--3. RULE 36--0.

CONFERENCE PROCEDURES

Senator Petersen, Chair of the Committee on Agriculture, Conservation and Natural Resources, appointed Senators Marsden, Petersen, and Stuart, the conferees on the part of the Senate for **H.B. 1902** (one thousand nine hundred two).

Senator Saslaw, Chair of the Committee on Commerce and Labor, appointed Senators Spruill, Lewis, and Obenshain, the conferees on the part of the Senate for **H.B. 2040** (two thousand forty).

On motion of Senator Lucas, the Senate adjourned until Monday, February 22, 2021, at 10:00 a.m.

Susan Chile Gelson

Justin E. Fairfax President of the Senate

Susan Clarke Schaar Clerk of the Senate